AUDIT & ACCOUNTING

# GUIDE

**JULY 1, 2016** 

Depository and Lending
Institutions: Banks and
Savings Institutions, Credit
Unions, Finance Companies,
and Mortgage Companies



AUDIT & ACCOUNTING

# GUIDE

July 1, 2016

Depository and Lending
Institutions: Banks and
Savings Institutions, Credit
Unions, Finance Companies,
and Mortgage Companies



Copyright © 2016 by American Institute of Certified Public Accountants, Inc. New York, NY 10036-8775

All rights reserved. For information about the procedure for requesting permission to make copies of any part of this work, please e-mail copyright@aicpa.org with your request. Otherwise, requests should be written and mailed to the Permissions Department, AICPA, 220 Leigh Farm Road, Durham, NC 27707-8110.

1234567890 AAP 19876

ISBN 978-1-94354-674-9

# **Preface**

(Updated as of July 1, 2016)

# **About AICPA Audit and Accounting Guides**

This AICPA Audit and Accounting Guide has been developed by the AICPA Guides Combination Task Force to assist management in the preparation of their financial statements in conformity with U.S. generally accepted accounting principles (GAAP) and to assist practitioners in performing and reporting on their audit or their attestation engagements.

AICPA Guides may include sections at the end of individual chapters or following the last chapter. These sections will be entitled either "Supplement" or "Appendix." A supplement is a reproduction, in whole or in part, of authoritative guidance originally issued by a standard setting body (including regulatory bodies) and applicable to entities or engagements within the purview of that standard setter, independent of the authoritative status of the applicable AICPA Guide. An appendix is included for informational purposes and has no authoritative status.

The Financial Reporting Executive Committee (FinREC) is the designated senior committee of the AICPA authorized to speak for the AICPA in the areas of financial accounting and reporting. Conforming changes made to the financial accounting and reporting guidance contained in this guide are approved by the FinREC Chair (or his or her designee). Updates made to the financial accounting and reporting guidance in this guide exceeding that of conforming changes are approved by the affirmative vote of at least two-thirds of the members of FinREC.

This guide does the following:

- Identifies certain requirements set forth in the FASB *Accounting Standards Codification*® (ASC).
- Describes FinREC's understanding of prevalent or sole industry practice concerning certain issues. In addition, this guide may indicate that FinREC expresses a preference for the prevalent or sole industry practice, or it may indicate that FinREC expresses a preference for another practice that is not the prevalent or sole industry practice; alternatively, FinREC may express no view on the matter.
- Identifies certain other, but not necessarily all, industry practices concerning certain accounting issues without expressing FinREC's views on them.
- Provides guidance that has been supported by FinREC on the accounting, reporting, or disclosure treatment of transactions or events that are not set forth in FASB ASC.

Accounting guidance for nongovernmental entities included in an AICPA Audit and Accounting Guide is a source of nonauthoritative accounting guidance. As discussed later in this preface, FASB ASC is the authoritative source of U.S. accounting and reporting standards for nongovernmental entities, in addition to guidance issued by the SEC.

Auditing guidance related to generally accepted auditing standards (GAAS) included in an AICPA Audit and Accounting Guide is recognized as an

©2016, AICPA AAG-DEP

interpretive publication as defined in AU-C section 200, Overall Objectives of the Independent Auditor and the Conduct of an Audit in Accordance With Generally Accepted Auditing Standards (AICPA, Professional Standards). Interpretive publications are recommendations on the application in specific circumstances, including engagements for entities in specialized industries.

An interpretive publication is issued under the authority of the AICPA Auditing Standards Board (ASB) after all ASB members have been provided an opportunity to consider and comment on whether the proposed interpretive publication is consistent with GAAS. The members of the ASB have found the auditing guidance in this guide to be consistent with existing GAAS.

Although interpretive publications are not auditing standards, AU-C section 200 requires the auditor to consider applicable interpretive publications in planning and performing the audit because interpretive publications are relevant to the proper application of GAAS in specific circumstances. If the auditor does not apply the auditing guidance in an applicable interpretive publication, the auditor should document how the requirements of GAAS were complied with in the circumstances addressed by such auditing guidance.

The ASB is the designated senior committee of the AICPA authorized to speak for the AICPA on all matters related to auditing. Conforming changes made to the auditing guidance contained in this guide are approved by the ASB Chair (or his or her designee) and the Director of the AICPA Audit and Attest Standards Staff. Updates made to the auditing guidance in this guide exceeding that of conforming changes are issued after all ASB members have been provided an opportunity to consider and comment on whether the guide is consistent with the Statements on Auditing Standards (SASs).

Any auditing guidance in a guide or chapter appendix, while not authoritative, is considered an "other auditing publication." In applying such auditing guidance, the auditor should, exercising professional judgment, assess the relevance and appropriateness of such guidance to the circumstances of the audit. Although the auditor determines the relevance of other auditing guidance, auditing guidance in an appendix to a guide or an appendix to a guide chapter has been reviewed by the AICPA Audit and Attest Standards staff and the auditor may presume that it is appropriate.

Attestation guidance included in an AICPA Audit and Accounting Guide is recognized as an attestation interpretation as defined in AT-C section 105, Concepts Common to All Attestation Engagements (AICPA, Professional Standards). Attestation interpretations are recommendations on the application of Statements on Standards for Attestation Engagements (SSAEs) in specific circumstances, including engagements for entities in specialized industries. Attestation interpretations are issued under the authority of the ASB. The members of the ASB have found the attestation guidance in this guide to be consistent with existing SSAEs.

A practitioner should be aware of and consider attestation interpretations applicable to his or her attestation engagement. If the practitioner does not apply the guidance included in an applicable AICPA Audit and Accounting Guide, the practitioner should be prepared to explain how he or she complied with the SSAE provisions addressed by such attestation guidance.

Any attestation guidance in a guide or chapter appendix, while not authoritative, is considered an "other attestation publication". In applying such guidance, the practitioner should, exercising professional judgment, assess the relevance and appropriateness of such guidance to the circumstances of the

engagement. Although the practitioner determines the relevance of other attestation guidance, such guidance in an appendix to a guide or an appendix to a guide chapter has been reviewed by the AICPA Audit and Attest Standards staff and the practitioner may presume that it is appropriate.

The ASB is the designated senior committee of the AICPA authorized to speak for the AICPA on all matters related to attestation. Conforming changes made to the attestation guidance contained in this guide are approved by the ASB Chair (or his or her designee) and the Director of the AICPA Audit and Attest Standards Staff. Updates made to the attestation guidance in this guide exceeding that of conforming changes are issued after all ASB members have been provided an opportunity to consider and comment on whether the guide is consistent with the SSAEs.

# **Purpose and Applicability**

This AICPA Audit and Accounting Guide has been prepared to assist financial institutions in preparing financial statements in conformity with GAAP and to assist independent accountants in reporting on financial statements (and, as discussed in appendix A, "FDI Act Reporting Requirements," other written management assertions) of those entities.

Chapters of the guide are generally organized by financial statement line item into four sections:

- a. An *introduction* that describes the general transactions and risks associated with the area. (The introduction does not address all possible transactions in each area.)
- b. Regulatory matters that may be of relevance in the preparation and audit of financial statements. Other regulatory matters may exist that require attention in the preparation and audit of financial statements following the general guidance on regulatory matters. Further, the guide does not address regulations that are not relevant to the preparation and audit of financial statements and certain of the regulatory requirements discussed may not be applicable to uninsured institutions.
- c. Accounting and financial reporting guidance that addresses accounting and financial reporting issues. FASB ASC 105, Generally Accepted Accounting Principles, establishes FASB ASC as the source of authoritative GAAP recognized by FASB to be applied by nongovernmental entities.
- d. Auditing guidance that includes objectives, planning, internal control over financial reporting and possible tests of controls, and substantive tests.

# Scope

This guide applies to all banks, savings institutions, credit unions, finance companies, and other entities (including entities with trade receivables). That population includes the following:

- a. Finance companies, including finance company subsidiaries
- b. Entities that do not consider themselves to be finance companies that engage in transactions that involve lending to or financing the

©2016. AICPA AAG-DEP

- activities of others (including trade receivables and independent and captive financing activities of all kinds of entities)
- c. Depository institutions insured by the FDIC's Deposit Insurance Fund or the National Credit Union Administration's National Credit Union Share Insurance Fund
- d. Bank holding companies
- e. Savings and loan association holding companies
- f. Branches and agencies of foreign banks regulated by U.S. federal banking regulatory agencies
- g. State chartered banks, credit unions, and savings institutions that are not federally insured
- h. Foreign financial institutions whose financial statements are purported to be prepared in conformity with GAAP
- i. Mortgage companies
- j. Entities that do not consider themselves to be mortgage companies that engage in transactions that involve mortgage activities or transactions
- k. Corporate credit unions
- l. Financing and lending activities of insurance companies

This guide does not apply to the following:

- a. Investment companies, broker dealers in securities, employee benefit plans and similar entities that carry loans and trade receivables at fair value with the unrealized gains and losses included in earnings
- b. Governmental or federal entities that follow the principles of GASB or the Federal Accounting Standards Advisory Board

As used in this guide, the term *depository institution* means a bank, credit union, and savings institution. The terms *financial institutions* or *institutions* refer to all entities covered by this guide.

As stated in the previous list, this guide applies to the financing activities of all kinds of enterprises. Certain entities may have financing activities but are not otherwise covered by this guide—for example, the financing subsidiary, unit, or division of a manufacturing company or retailer. Only those sections and chapters of this guide related to financing activities are intended to apply to such entities. The remaining portions are not intended to apply to such entities, but may otherwise be useful to financial statement preparers and auditors.

Certain terms are used interchangeably throughout the guide as follows:

- Credit unions often refer to shares, dividends on shares, and members, which are equivalent to deposits, interest on deposits, and customers for banks and savings institutions.
- Finance companies often refer to *finance receivables*, which are equivalent to *loans* or *loans receivable* for other entities. A *credit officer* of a finance company is the same as a *loan officer*.
- A *supervisory committee* of a credit union is the functional equivalent of an *audit committee* of other entities.

#### Limitations

In July 1990, the AICPA's Board of Directors authorized the AICPA staff to make conforming changes to the Audit and Accounting Guides with the approval of the chairman of the ASB or the chairman of FinREC, as appropriate. The board resolution defines *conforming changes* as "revisions intended to effect changes necessitated by the issuance of authoritative pronouncements." Conforming changes are carefully and judiciously made and normally limited to items that result from the issuance of new authoritative literature. Conforming changes also include nonaccounting and nonauditing revisions that modify, add, or delete regulatory guidance and industry background information in response to changes in the regulatory and industry environment. Conforming changes do not include recent legislative programs or other governmental measures or industry actions that may have been taken as a result of the current economic environment.

# Recognition

# AICPA Senior Committees Auditing Standards Board

Sandra Johnigan, Member Michael J. Santay, *Chair* 

#### **Financial Reporting Executive Committee**

Muneera Carr, Member Richard C. Paul, *Chair* 

The AICPA gratefully acknowledges those members of the AICPA Depository and Lending Institutions Expert Panel (2015–2016) who reviewed or otherwise contributed to the development of this edition of the guide:

Sydney Garmong, Chair
David C. Britt
Bret Dooley
Larry Gee
Dom Giuffrida
Ryan Richards
Jeffrey C. Skumin
Gregory C. Talbott

In addition to the 2015–2016 expert panel members listed, the AICPA gratefully acknowledges those who reviewed and otherwise contributed to the development of this guide: Danielle Austin, Kevin Chiu, Michael Gonzales, Ken Harris, Jeff Honeycutt, Matthew Kinkaid, John G. Klinge, Mike Lundberg, Bryan W. Mogensen, Sarah V. Nawrocki, Randy Oberdiek, Douglas J. Orth, Amber Sarb, Shuchi Satwah, Chris Vallez, and Joanne Wakim.

#### **AICPA Staff**

Teresa Brenan

Technical Manager

Accounting and Auditing—Member Learning and Competency

Kim Kushmerick Senior Technical Manager Accounting Standards

©2016, AICPA AAG-DEP

#### Salome J. Tinker Senior Technical Manager Accounting Standards and

Staff Liaison to the AICPA Depository and Lending Institutions Expert Panel

### Guidance Considered in This Edition

This edition of the guide has been modified by the AICPA staff to include certain changes necessary due to the issuance of authoritative guidance since the guide was originally issued, and other revisions as deemed appropriate. Authoritative guidance issued through July 1, 2016, has been considered in the development of this edition of the guide. However, this guide does not include all audit, accounting, reporting, and other requirements applicable to an entity or a particular engagement. This guide is intended to be used in conjunction with all applicable sources of authoritative guidance.

Authoritative guidance that is issued and effective for entities with fiscal years ending on or before July 1, 2015, is incorporated directly in the text of this guide. Authoritative guidance issued but not yet effective for fiscal years ending on or before July 1, 2015 but becoming effective on or before December 31, 2016, is also presented directly in the text of the guide, but shaded gray and accompanied by a footnote indicating the effective date of the new guidance. The distinct presentation of this content is intended to aid the reader in differentiating content that may not be effective for the reader's purposes (as part of the guide's "dual guidance" treatment of applicable new guidance).

Authoritative guidance issued but not yet effective as of the date of the guide and not becoming effective until after December 31, 2016, is referenced in a "guidance update" box; that is, boxed text that contains summary information on the guidance issued but not yet effective.

In updating this guide, all guidance issued up to and including the following was considered, but not necessarily incorporated, as determined based on applicability:

- FASB Accounting Standards Update (ASU) No. 2016-13, Financial Instruments Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments
- SAS No. 131, Amendment to Statement on Auditing Standards No. 122 Section 700, Forming an Opinion and Reporting on Financial Statements (AICPA, Professional Standards, AU-C sec. 700)
- Interpretation No. 3, "Reporting on Audits Conducted in Accordance With Auditing Standards Generally Accepted in the United States of America and International Standards on Auditing," of AU-C section 700, Forming an Opinion and Reporting on Financial Statements (AICPA, Professional Standards, AU-C sec. 9700 par. .08–.13)
- Revised interpretations issued through July 1, 2015, including Interpretation No. 3, "Appropriateness of Identifying No Significant Deficiencies or No Material Weaknesses in an Interim Communication," of AU-C section 265, Communicating Internal Control Related Matters Identified in an Audit (AICPA, Professional Standards, AU-C sec. 9265 par. .08-.10)

- Statement of Position 13-2, Performing Agreed-Upon Procedures Engagements That Address the Completeness, Mapping, Consistency, or Structure of XBRL-Formatted Information (AICPA, Professional Standards, AUD sec. 55)
- SSAE No. 18, Attestation Standards: Clarification and Recodification (AICPA, Professional Standards)
- Interpretation No. 1, "Third-Party Due Diligence Services Related to Asset-Backed Securitizations: SEC Release No. 34-72936," of AT section 201, Agreed-Upon Procedures Engagements (AICPA, Professional Standards, AT sec. 9201 par. .01-.19)
- PCAOB Auditing Standard No. 18, Related Parties (AICPA, PCAOB Standards and Related Rules)

Users of this guide should consider guidance issued subsequent to those items listed previously to determine their effect, if any, on entities and engagements covered by this guide. In determining the applicability of recently issued guidance, its effective date should also be considered.

The changes made to this edition of the guide are identified in appendix H, "Schedule of Changes Made to the Text From the Previous Edition." The changes do not include all those that might be considered necessary if the guide were subjected to a comprehensive review and revision.

PCAOB quoted content is from PCAOB Auditing Standards and PCAOB Staff Audit Practice Alerts, ©2015, Public Company Accounting Oversight Board. All rights reserved. Used by permission.

FASB standards quoted are from the FASB  $Accounting\ Standards\ Codification\$ ©2015, Financial  $Accounting\ Foundation$ . All rights reserved. Used by permission.

# **FASB ASC Pending Content**

# Presentation of Pending Content in FASB ASC

Amendments to FASB ASC (issued in the form of ASUs) are initially incorporated into FASB ASC in "pending content" boxes below the paragraphs being amended with links to the transition information. The pending content boxes are meant to provide users with information about how the guidance in a paragraph will change as a result of the new guidance.

Pending content applies to different entities at different times due to varying fiscal year-ends, and because certain guidance may be effective on different dates for public and nonpublic entities. As such, FASB maintains amended guidance in pending content boxes within FASB ASC until the "roll-off" date. Generally, the "roll-off" date is six months following the latest fiscal year end for which the original guidance being amended could still be applied.

# Presentation of FASB ASC Pending Content in AICPA Audit and Accounting Guides

Amended FASB ASC guidance that is included in pending content boxes in FASB ASC on July 1, 2015, is referenced as "Pending Content" in this guide. Readers should be aware that "Pending Content" referenced in this guide will eventually be subjected to FASB's roll-off process and no longer be labeled as "Pending Content" in FASB ASC (as discussed in the previous paragraph).

©2016, AICPA AAG-DEP

# Terms Used to Define Professional Requirements in This AICPA Audit and Accounting Guide

Any requirements described in this guide are normally referenced to the applicable standards or regulations from which they are derived. Generally, the terms used in this guide describing the professional requirements of the referenced standard setter (for example, the ASB) are the same as those used in the applicable standards or regulations (for example, *must* or *should*). However, where the accounting requirements are derived from FASB ASC, this guide uses *should*, whereas FASB uses *shall*. In its resource document "About the Codification" that accompanies FASB ASC, FASB states that it considers the terms *should* and *shall* to be comparable terms and to represent the same concept—the requirement to apply a standard.

Readers should refer to the applicable standards and regulations for more information on the requirements imposed by the use of the various terms used to define professional requirements in the context of the standards and regulations in which they appear.

Certain exceptions apply to these general rules, particularly in those circumstances where the guide describes prevailing or preferred industry practices, or both, for the application of a standard or regulation. In these circumstances, the applicable senior committee responsible for reviewing the guide's content believes the guidance contained herein is appropriate for the circumstances.

# Applicability of GAAS and PCAOB Standards

Appendix A, "Council Resolution Designating Bodies to Promulgate Technical Standards," of the AICPA Code of Professional Conduct recognizes both the ASB and the PCAOB as standard setting bodies designated to promulgate auditing, attestation, and quality control standards. Paragraph .01 of the "Compliance With Standards Rule" (AICPA, *Professional Standards*, ET sec. 1.310.001 and 2.310.001) requires an AICPA member who performs an audit to comply with the applicable standards.

Audits of the financial statements of those entities not subject to the oversight authority of the PCAOB (that is, those entities whose audits are not within the PCAOB's jurisdiction—hereinafter referred to as *nonissuers*) are to be conducted in accordance with GAAS as issued by the ASB. The ASB develops and issues standards in the form of SASs through a due process that includes deliberation in meetings open to the public, public exposure of proposed SASs, and a formal vote. The SASs and their related interpretations are codified in the AICPA's *Professional Standards*. In citing GAAS and their related interpretations, references generally use section numbers within the codification of currently effective SASs and not the original statement number, as appropriate.

Audits of the financial statements of those entities subject to the oversight authority of the PCAOB (that is, those entities whose audits are within the PCAOB's jurisdiction—hereinafter referred to as *issuers*) are to be conducted in accordance with standards established by the PCAOB, a private sector, non-profit corporation created by the Sarbanes-Oxley Act of 2002. The SEC has oversight authority over the PCAOB, including the approval of its rules, standards, and budget. In citing the auditing standards of the PCAOB, references generally use section numbers within the reorganized PCAOB auditing standards and not the original standard number, as appropriate.

The auditing content in this guide primarily discusses GAAS issued by the ASB and is applicable to audits of nonissuers.

Users of this guide may find the tool developed by the PCAOB's Office of the Chief Auditor helpful in identifying comparable PCAOB Standards. The tool is available at pcaobus.org/Standards/Auditing/Pages/FindAnalogousStandards.aspx.

Considerations for audits of issuers in accordance with PCAOB standards may also be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

# **Applicability of Quality Control Standards**

QC section 10, A Firm's System of Quality Control (AICPA, Professional Standards), addresses a CPA firm's responsibilities for its system of quality control for its accounting and auditing practice. A system of quality control consists of policies that a firm establishes and maintains to provide it with reasonable assurance that the firm and its personnel comply with professional standards, as well as applicable legal and regulatory requirements. The policies also provide the firm with reasonable assurance that reports issued by the firm are appropriate in the circumstances. This section applies to all CPA firms with respect to engagements in their accounting and auditing practice.

AU-C section 220, Quality Control for an Engagement Conducted in Accordance With Generally Accepted Auditing Standards (AICPA, Professional Standards), addresses the auditor's specific responsibilities regarding quality control procedures for an audit of financial statements. When applicable, it also addresses the responsibilities of the engagement quality control reviewer.

Because of the importance of audit quality, we have added a new appendix, appendix D, "Overview of Statements on Quality Control Standards," to this guide. Appendix D summarizes key aspects of the quality control standard. This summarization should be read in conjunction with QC section 10 and AU-C section 220, and the quality control standards issued by the PCAOB, as applicable.

# Alternatives Within U.S. GAAP

The Private Company Council (PCC), established by the Financial Accounting Foundation's Board of Trustees in 2012, and FASB, working jointly, will mutually agree on a set of criteria to decide whether and when alternatives within U.S. GAAP are warranted for private companies. Based on those criteria, the PCC reviews and proposes alternatives within U.S. GAAP to address the needs of users of private company financial statements. These U.S. GAAP alternatives may be applied to those entities that are not public business entities, not-for-profits, or employee benefit plans.

The FASB ASC Master Glossary defines a *public business entity* as follows:

A public business entity is a business entity meeting any one of the criteria below. Neither a not-for-profit entity nor an employee benefit plan is a business entity.

a. It is required by the SEC to file or furnish financial statements, or does file or furnish financial statements (including voluntary filers), with the SEC (including other

©2016, AICPA AAG-DEP

- entities whose financial statements or financial information are required to be or are included in a filing).
- b. It is required by the Securities Exchange Act of 1934 (the Act), as amended, or rules or regulations promulgated under the Act, to file or furnish financial statements with a regulatory agency other than the SEC.
- c. It is required to file or furnish financial statements with a foreign or domestic regulatory agency in preparation for the sale of or for purposes of issuing securities that are not subject to contractual restrictions on transfer.
- d. It has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market.
- e. It has one or more securities that are not subject to contractual restrictions on transfer, and it is required by law, contract, or regulation to prepare U.S. GAAP financial statements (including footnotes) and make them publicly available on a periodic basis (for example, interim or annual periods). An entity must meet both of these conditions to meet this criterion.

An entity may meet the definition of a public business entity solely because its financial statements or financial information is included in another entity's filing with the SEC. In that case, the entity is only a public business entity for purposes of financial statements that are filed or furnished with the SEC.

Considerations related to alternatives for private companies may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Private Companies That Elect to Use Standards as Issued by the Private Company Council.

# **AICPA.org Website**

The AICPA encourages you to visit the website at www.aicpa.org and the Financial Reporting Center at www.aicpa.org/FRC. The Financial Reporting Center supports members in the execution of high-quality financial reporting. Whether you are a financial statement preparer or a member in public practice, this center provides exclusive member-only resources for the entire financial reporting process, and provides timely and relevant news, guidance and examples supporting the financial reporting process, including accounting, preparing financial statements and performing compilation, review, audit, attest or assurance and advisory engagements. Certain content on the AICPA's websites referenced in this guide may be restricted to AICPA members only.

# Select Recent Developments Significant to This Guide

# **AICPA's Ethics Codification Project**

The AICPA's Professional Ethics Executive Committee (PEEC) restructured and codified the AICPA Code of Professional Conduct (code) so that members

and other users of the code can apply the rules and reach appropriate conclusions more easily and intuitively. This is referred to as the AICPA Ethics Codification Project.

Although PEEC believes it was able to maintain the substance of the existing AICPA ethics standards through this process and limited substantive changes to certain specific areas that were in need of revision, the numeric citations and titles of interpretations have all changed. In addition, the ethics rulings are no longer in a question and answer format but rather, have been drafted as interpretations, incorporated into interpretations as examples, or deleted where deemed appropriate. For example,

- Rule 101, *Independence* [ET sec. 101 par. .01], is referred to as the "Independence Rule" [ET sec. 1.200.001] in the revised code.
- The content from the ethics ruling entitled "Financial Services Company Client has Custody of a Member's Assets" [ET sec. 191 par. .081–.082] is incorporated into the "Brokerage and Other Accounts" interpretation [ET sec. 1.255.020] found under the subtopic "Depository, Brokerage, and Other Accounts" [ET sec. 1.255] of the "Independence" topic [ET sec. 1.200].

The revised code was effective December 15, 2014, and is available at http://pub.aicpa.org/codeofconduct/Ethics.aspx. References to the code have been updated in this guide.

To assist users in locating in the revised code content from the prior code, PEEC created a mapping document. The mapping document is available in Excel format in appendix D in the revised code.

# **Attestation Clarity Project**

To address concerns over the clarity, length, and complexity of its standards, the ASB established clarity drafting conventions and undertook a project to redraft all the standards it issues in clarity format. The redrafting of Statements on Standards for Attestation Engagements (SSAEs or attestation standards) in SSAE No. 18, Attestation Standards: Clarification and Recodification, represents the culmination of that process.

The attestation standards are developed and issued in the form of SSAEs and are codified into sections. SSAE No. 18 recodifies the "AT" section numbers designated by SSAE Nos. 10–17 using the identifier "AT-C" to differentiate the sections of the clarified attestation standards (AT-C sections) from the attestation standards that are superseded by SSAE No. 18 (AT sections).

The AT sections in AICPA Professional Standards remain effective through April 2017, by which time substantially all engagements for which the AT sections were still effective are expected to be completed.

©2016. AICPA AAG-DEP

# **TABLE OF CONTENTS**

Chapter		Paragraph
1	Industry Overview—Banks and Savings Institutions	.01118
	Description of Business	.0107
	Regulation and Oversight	.0869
	Regulatory Background	.1523
	Deposit Insurance Fund	.2430
	The Dodd-Frank Act	.3169
	Regulatory Capital Matters	.7091
	Capital Adequacy	.7280
	Prompt Corrective Action	.8191
	Annual Independent Audits and Reporting Requirements	.9211 <i>7</i>
	Additional Regulatory Requirements Concerning the Sarbanes-Oxley Act, Corporate Governance, and Services Outsourced to External Auditors	
	Other Reporting Considerations	.118
2	Industry Overview—Credit Unions	.0162
	Description of Business	.0110
	The Board of Directors	.07
	The Supervisory Committee	.0809
	The Credit Committee	.10
	Charters, Bylaws, and Minutes	.1117
	Financial Structure	.1213
	Credit Union System	.1416
	Corporate Credit Unions	.17
	Regulation and Oversight	.1823
	Government Supervision	.1820
	NCUA	.2123
	Regulatory Capital Matters	.2454
	Natural Person Credit Unions	.2444
	Corporate Credit Unions	.4554
	Annual Audits	.5562
	Other Reporting Considerations	.62
3	Industry Overview—Finance Companies	.0114
-	Description of Business	.0111
	Debt Financing	.0811
	Regulation and Oversight	.1214

Chapter		Paragraph
4	Industry Overview—Mortgage Companies	.0137
	Description of Business	.0120
	Regulation and Oversight	.2132
	Reporting Considerations	
	HUD Programs	
	Asset Servicing for Investors	
5	Audit Considerations and Certain Financial Reporting Matters	.01257
	Overview	.0109
	An Audit of Financial Statements	.0405
	Audit Risk	.0609
	Terms of Engagement	.1013
	Audit Planning	.1415
	Materiality	.1620
	Performance Materiality	.1920
	Use of Assertions in Assessment of Risks of Material	
	Misstatement	
	Risk Assessment Procedures	.2375
	Risk Assessment Procedures and Related Activities	.2532
	Understanding the Entity and Its Environment, Including	00.75
	the Entity's Internal Control	
	Risk Assessment and the Design of Further Audit Procedures	.7698
	Identifying and Assessing the Risks of Material Misstatement	. <i>77</i> 81
	Designing and Performing Further Audit Procedures	.8298
	Evaluation of Misstatements Identified During the Audit	.99101
	Audit Documentation	.102109
	Timely Preparation of Audit Documentation	.105
	Documentation of the Audit Procedures Performed and Audit Evidence Obtained	.106108
	Assembly and Retention of the Final Audit File	
	Using the Work of an Auditor's Specialist	
	Using the Work of a Management's Specialist	
	Processing of Transactions by Service Organizations	
	Consideration of Fraud in a Financial Statement Audit	
	Professional Skepticism	
	Discussion Among the Engagement Team	
	Risk Assessment Procedures and Related Activities	
	Identification and Assessment of the Risks of Material	
	Misstatement Due to Fraud	.137139
	Responses to the Assessed Risks of Material Misstatement	
	Due to Fraud	
	Evaluation of Audit Evidence	
	Auditor Unable to Continue the Engagement	148

of Contents	xvii

	<b>Table of Contents</b>	xvii
Chapter		Paragraph
5	Audit Considerations and Certain Financial Reporting Matters—continued	
	Communications to Management and With Those Charged With Governance	.149
	Communications to Regulatory and Enforcement Authorities	.150
	Documentation	.151
	Compliance With Laws and Regulations	
	Responsibility for Compliance With Laws and Regulations	
	The Auditor's Consideration of Compliance With Laws and Regulations	
	Going-Concern Considerations	
	Evaluating Whether Substantial Doubt Exists	
	Written Representations	
	Consideration of the Effects on the Auditor's Report	
	Documentation	
	Written Representations	
	Written Representations as Audit Evidence	.101
	Management From Whom Written Representations Are Requested	.182183
	Written Representations About Management's Responsibilities and Other Written Representations	.184186
	Date of, and Period(s) Covered by, Written Representations	.187
	Form of Written Representations	.188
	Doubt About the Reliability of Written Representations and Requested Written Representations Not Provided	.189
	Information Other Than Financial Statements	.190192
	Certain Financial Reporting Matters	.193211
	Disclosures of Certain Significant Risks and Uncertainties	
	Segment Reporting	
	Regulation and Supervision of Depository Institutions	
	Introduction	
	Rule Making	
	Examinations	
	Enforcement	
	Planning	
	Detection of Errors and Fraud	.238239
	Evaluation of Contingent Liabilities and Related Disclosures	.240
	Going-Concern Considerations	.241242
	Regulatory Reporting Matters—Interpretation and Reporting Related to GAAP	
	Auditor and Examiner Relationship	
	and the second s	

©2016, AICPA **Contents** 

### xviii

Chapter		Paragraph
6	Cash and Cash Equivalents	.0142
	Introduction	.0108
	CIPC and Cash Equivalents	.02
	Deposits With Other Depository Institutions	.0305
	Balances With Federal Reserve Banks and FHLBs	.06
	Federal Funds Sold	.07
	Cash on Hand	.08
	Accounting and Financial Reporting	
	Definition of Cash and Cash Equivalents	.0911
	Classification of Cash Flows	.1222
	Acquisition and Sales of Certain Securities and Loans	.2324
	Gross and Net Cash Flows	.2528
	Cash Receipts and Payments Related to Hedging Activities	.29
	Financial Statement Presentation and Disclosure	.3033
	Auditing	.3442
	Objectives	.34
	Planning	.35
	Internal Control Over Financial Reporting and Possible	.55
	Tests of Controls	.3640
	Substantive Tests	.4142
7	Investments in Debt and Equity Securities	.01163
	Introduction	.0165
	U.S. Government and Agency Obligations	.0507
	Municipal Obligations	.0811
	Asset-Backed Securities	.1249
	Other Structured Credit Products	.5059
	Issues of International Organizations and Foreign	
	Governments	.60
	Other Securities	.61
	Transfers of Securities	
	Regulatory Matters	
	Bank Accounting Advisory Series	.82
	Accounting and Financial Reporting	
	Introduction	.8391
	OTTI	.92103
	Unrealized Gains and Losses	
	Premiums and Discounts	.107113
	Interest Income	.114
	Consolidation	.115
	Special Areas	
	Transfers and Servicing of Securities	.119121
	Troubled Debt Restructurings	.122124

xix

Chapter		Paragraph
7	Investments in Debt and Equity Securities—continued	
	Loans and Debt Securities Acquired With Deteriorated Credit Quality	125
	Financial Statement Presentation and Disclosure	
	Auditing	138163
	Planning	
	Internal Control Over Financial Reporting and Possible Tests of Control	144-154
	Substantive Tests	
8	Loans	.01230
Ü	Introduction	
	The Lending Process	
	Credit Strategy	
	Credit Risk	
	Lending Policies and Procedures	
	Types of Lending	
	Commercial, Industrial, and Agricultural Loans	
	Consumer Loans	
	Residential Real Estate Loans	
	Lease Financing	
	Trade Financing	
	CRE and Construction Loans	
	Foreign Loans	
	Loans Involving More Than One Lender	
	Regulatory Matters	
	Real Estate Lending Standards	
	Retail Credit Loans and Residential Mortgage Loans	
	Troubled Debt Restructurings	
	Credit Card Lending	
	Nontraditional Mortgage Products	
	Correspondent Concentration Risks	
	Leveraged Lending	
	Income Recognition on Problem Loans	
	Credit Union Lending Restrictions	
	Lending Statutes	
	Uniform Commercial Code	
	Bank Accounting Advisory Series	
	Accounting and Financial Reporting	
	Interest Income, Delinquency Fees, Prepayment Fees,	07101
	and Rebates	95100
	Loan Fees, Costs, Discounts, and Premiums	
	TDRs	
	.=	5 50

Chapter		Paragraph
8	Loans—continued	
	Real Estate Investments	.139
	Lease Financing	.140
	Foreign Loans	
	Commitments	
	Financial Statement Presentation and Disclosure	
	Auditing	.182230
	Objectives	
	Planning	
	Internal Control Over Financial Reporting and Possible	
	Tests of Controls	
	Substantive Tests	.199230
9	Credit Losses	.01112
	Introduction	.0122
	Management's Methodology	.0412
	Loan Reviews	.1315
	Loans Individually Evaluated for Impairment	.16
	Loans Collectively Evaluated for Impairment	.1 <i>7</i> 18
	Estimating Overall Credit Losses	
	Regulatory Matters	.2338
	Credit Unions	.3438
	Accounting and Financial Reporting	.3962
	Sources of Applicable Guidance	.3941
	Allowance for Loan Losses	.4248
	Loss Contingencies	.4952
	Financial Statement Presentation and Disclosure of Loan	
	Impairment and Allowance for Credit Losses	.5362
	Auditing	
	Objectives	
	Planning and Risk Assessment	.6468
	Internal Control Over Financial Reporting and Possible	/O 75
	Tests of Controls	
	Substantive Tests	.76112
10	Transfers and Servicing and Variable Interest Entities	.01127
	Introduction	
	Asset-Backed Securitizations	
	Loan Participations and Loan Syndications	
	Loan Servicing	.08
	Regulatory Matters	
	VA No-Bids and Private Mortgage Agencies	.1619

Chapter		Paragraph
10	Transfers and Servicing and Variable Interest Entities—continued	
	Accounting and Financial Reporting	20113
	Mortgage Loans and MBSs Held for Sale	
	Transfers and Servicing of Financial Assets	
	Sale of Financial Assets	
	Transfers of Loans With Recourse	.4950
	Servicing Assets and Liabilities	.5169
	Secured Borrowings and Collateral	.7073
	Loans Not Previously HFS	.7478
	Financial Statement Presentation	
	Financial Statement Disclosure	.8696
	VIEs	.97113
	Auditing	114127
	Objectives	114
	Planning	
	Internal Control Over Financial Reporting and Possible	
	Tests of Controls	.119125
	Substantive Tests	.126127
11	Real Estate Investments, Real Estate Owned, and Other	
11	Foreclosed Assets	.0173
	Introduction	
	Foreclosed Assets	
	Real Estate Investments	
	Regulatory Matters	
	Accounting and Financial Reporting	
	Foreclosed Assets	. 1723
	Accounting and Reporting for Long-Lived Assets to Be Disposed of by Sale	.2433
	Accounting and Reporting for Long-Lived Assets	.2400
	to Be Held and Used	.3440
	Real Estate Investments	.4143
	Sale of Real Estate Assets	
	Development Costs	
	Allocation of Income and Equity Among Parties	
	to a Joint Venture	.5055
	Auditing	5673
	Objectives	56
	Planning	
	Internal Control Over Financial Reporting and Possible	
	Tests of Controls	.5861
	Substantive Tests	.6273

Chapter		Paragraph
12	Other Assets, Other Liabilities, and Other Investments	.0191
	Introduction	.0110
	Premises and Equipment, Net	.03
	FHLB or FRB Stock	.0405
	Identifiable Intangibles	.0607
	Goodwill	.08
	Customers' Liabilities on Acceptances	.09
	Other Miscellaneous Items	.10
	Regulatory Matters	
	Accounting and Financial Reporting	
	Premises and Equipment, Net	.1626
	FHLB or FRB Stock	.2734
	Goodwill and Other Intangible Assets	.3550
		.5152
	Exit or Disposal Activities	
	Asset Retirement Obligations	.5356
	Customers' Liabilities on Acceptances	.57 .58
	Mortgage Servicing Advances	.50 .59
	Impairment of Long-Lived Assets	
	NCUSIF Deposit	.6066
	Other Investments	.6772
	Contributed Assets	.7376
	Auditing	. <i>77</i> 91
	Objectives	.7778
	Planning	.79
	Internal Control Over Financial Reporting and Possible	00.00
	Tests of Controls	.8082
	Substantive Tests	.8391
13	Deposits	.0170
	Introduction	.0132
	Demand Deposits	.04
	Savings Deposits	.05
	Time Deposits	.0610
	Brokered Deposits	.1113
	Dormant Accounts	.14
	Closed Accounts	
	Other Deposit Services	.16
	The Payments Function and Services	.1 <i>7</i> 32
	Regulatory Matters	
	- •	.3338
	Limitations on Brokered Deposits	
	Classification of Deposits of Credit Unions	
	Accounting and Financial Reporting	
	Auditing	.4770
	Objectives	.47

	<b>Table of Contents</b>	xxiii
Chapter		Paragraph
13	Deposits—continued	
	Planning Internal Control Over Financial Reporting and Possible	.4849
	Tests of Controls	.5055
	Substantive Tests	
14	Federal Funds and Repurchase Agreements	.0169
	Introduction	0124
	Federal Funds Purchased	.0203
	Repos	.0425
	Regulatory Matters	26
	Accounting and Financial Reporting	2747
	Auditing	4869
	Objectives	48
	Planning	
	Internal Control Over Financial Reporting and Possible	
	Tests of Controls	
	Substantive Tests	.5969
15	Debt	.0177
	Introduction	0131
	Long Term Debt	.0609
	Short Term Debt	.1031
	Regulatory Matters	3233
	Accounting and Financial Reporting	3457
	Auditing	5877
	Objectives	58
	Planning	.59
	Internal Control Over Financial Reporting and Possible	
	Tests of Controls	
	Substantive Tests	.6477
16	Income Taxes	.0136
	Introduction	
	Banks and Savings Institutions	
	Other	
	Credit Union	
	Regulatory Matters	
	Accounting and Financial Reporting	1830
	Deferred Tax Assets and Liabilities	
	Temporary Differences	
	Financial Statement Presentation and Disclosure	

xxiii

©2016, AICPA Contents

Chapter		Paragraph
16	Income Taxes—continued	
	Auditing	.3136
	Objectives	.31
	Planning	.32
	Internal Control Over Financial Reporting and Possible	
	Tests of Controls	.3334
	Substantive Tests	.3536
1 <i>7</i>	Equity and Disclosures Regarding Capital Matters	.01115
	Introduction	.01
	Banks and Savings Institutions	.0246
	Introduction	.02
	Equity	.0310
	Holding Company Equity and Regulatory Capital	.1127
	Disclosures for Banks and Savings Institutions	.2837
	Disclosure for Holding Companies	.3840
	Illustrative Disclosures for Banks and Savings Institutions (The example disclosures that follow are for	
	illustrative purposes only)	.4146
	Credit Unions	.4764
	Introduction	.47
	Members' Equity	.4854
	New Credit Unions and Low Income Designated Credit Unions	.55
	Disclosures for Natural Person Credit Unions	.5660
	Illustrative Disclosures for Natural Person Credit Unions	.6164
	Corporate Credit Unions	.6579
	Introduction	.65
	Equity	.6671
	Disclosures for Corporate Credit Unions	.7276
	Illustrative Disclosures for Corporate Credit Unions	.7779
	Mortgage Companies and Mortgage Banking Activities	
	Introduction	.80
	Disclosure for Mortgage Companies and Mortgage Banking Activities	.8183
	Illustrative Disclosures for Mortgage Companies and	
	Mortgage Banking Activities	.84
	Regulatory Capital Matters for All Entities	.8588
	Regulatory Capital Disclosures for Branches of Foreign Institutions	.8586
	Regulatory Capital Disclosures for Trust Operations	.8788
	Auditing	.89115
	Banks, Savings Institutions, and Credit Unions	.89105
	Mortgage Companies and Activities	.106115

XXV

Chapter		Paragraph
18	Derivative Instruments: Futures, Forwards, Options, Swaps, and Other Derivative Instruments	.01101
	Introduction	.0169
	Risks Inherent in Derivatives	.0513
	Types of Derivatives	.1452
	Uses of Derivatives to Alter Risk	.5364
	Variations on Basic Derivatives	.6569
	Regulatory Matters	.7078
	Accounting and Financial Reporting	.79100
	Financial Statement Disclosures	.95100
	Auditing Considerations	.101
19	Business Combinations	.0141
	Introduction	.01
	Regulatory Matters	.0207
	Accounting and Financial Reporting	.0830
	Purchase of a Loan or Group of Loans	.1820
	Loans and Debt Securities Acquired With Deteriorated	
	Credit Quality	.21
	Special Considerations in Applying the Acquisition Method to Combinations of Mutual Entities	.2225
	Impairment and Disposal Accounting for Certain Acquired Long Term Customer Relationship	
	Intangible Assets	.2627
	Branch Acquisitions	.2830
	Auditing Considerations	.3141
20	Fair Value	.0168
	Introduction	.01
	Accounting and Financial Reporting	.0263
	Definition of Fair Value	.0227
	Valuation Techniques	.2830
	Present Value Techniques	.3133
	The Fair Value Hierarchy	.3437
	Measuring the Fair Value of Investments in Certain Entities That Calculate Net Asset Value per Share	
	(or Its Equivalent)	.3841
	Measuring Fair Value When the Volume or Level of Activity for an Asset or a Liability Has Significantly	
	Decreased	.4247
	Using Quoted Prices Provided by Third Parties	.48
	Disclosures	.4960
	Fair Value Option	.6163
	Auditing Considerations	.6468

#### xxvi

Chapter		Paragraph
21	Trust and Asset Management Activities	.0130
	Introduction	0112
	Personal Trusts	
	Corporate Trusts	
	Employee Benefit Trusts	
	Common or Collective Trust Funds	
	Regulatory Matters	
	Accounting and Financial Reporting	
	Auditing	1730
	Objectives	17
	Planning	.18
	Internal Control Over Financial Reporting and Possible Tests of Controls	1921
	Financial Reporting Controls of the Trust	
	Substantive Tests Related to Financial Statement Audits	2324
	Substantive Procedures Related to the Trust	2529
	Audits of Unit Investment Trusts	30
22	Insurance Activities	.0170
	Introduction	
	Types of Insurance Coverage	
	Credit Life	
	Credit Accident and Health	
	Property and Liability	
	Writing Policies	
	Commissions	
	Regulatory Matters	
	Accounting	
	Premium Income	
	Acquisition Costs	
	Investment Portfolios	
	State Laws	
	Commissions	
	Consolidation Policy	
	Financial Statement Presentation	
	Auditing	/0.70
23	Reporting Considerations	.0138
	Introduction	
	Forming an Opinion on the Financial Statements	
	Reports	0537
	Unmodified Opinion	0506

	<b>Table of Contents</b>	xxvii
Chapter		Paragraph
23	Reporting Considerations—continued	
	EOM and Other-Matter Paragraphs Added to the Independent Auditor's Report	
	Modified Opinions  Financial Statements Prepared in Accordance With a	.1321
	Special Purpose Framework	.2223
	Members' Shares Reported as Equity	.24
	Communication of Internal Control Related Matters	
	Reports on Supervisory Committee Audits	.2833
	Example Report on the Application of Agreed-Upon Procedures Performed in Connection With a	
	Supervisory Committee Audit	.34
	Example Reports on the FDIC Loss Sharing Purchase	05.04
	and Assumption Transactions	.3536
	Small Business Lending Fund Auditor Certification Guidance	.37
	Appendix A—Illustrative Unqualified PCAOB Reports	.38
Appendix		
Α	FDI Act Reporting Requirements	
В	Regulatory Reporting Matters—Interpretation and Reporting Related to U.S. GAAP	
С	Information Sources	
D	Overview of Statements on Quality Control Standards	
Е	The New Revenue Recognition Standard: FASB ASU No. 2014-09	
F	The New Leases Standard: FASB ASU No. 2016-02	
G	Accounting for Financial Instruments	
Н	Schedule of Changes Made to the Text From the Previous Edition	
Index of Pro	onouncements and Other Technical Guidance	
Subject Inde	ex	

©2016, AICPA **Contents** 

# Chapter 1

# Industry Overview—Banks and Savings Institutions

Gray shaded text in this chapter reflects guidance issued but not yet effective as of the date of this guide, July 1, 2016, but becoming effective on or prior to December 31, 2016, exclusive of any option to early adopt ahead of the mandatory effective date. Unless otherwise indicated, all unshaded text reflects guidance that was already effective as of the date of this guide.

# **Description of Business**

1.01 Banks and savings institutions provide a link between entities that have capital and entities that need capital. They accept deposits from entities with idle funds and lend to entities with investment or spending needs. This process of financial intermediation benefits the economy by increasing the supply of money available for investment and spending. It also provides an efficient means for the payment and transfer of funds between entities.

1.02 Government, at both the federal and state levels, has long recognized the importance of financial intermediation by offering banks and savings institutions special privileges and protections. These incentives—such as access to credit through the Board of Governors of the Federal Reserve System (Federal Reserve) and federal insurance of deposits—have not been similarly extended to commercial enterprises. Accordingly, the benefits and responsibilities associated with their public role as financial intermediaries have brought banks and savings institutions under significant governmental oversight. Federal and state regulations affect every aspect of banks and savings institutions' operations. Similarly, legislative and regulatory developments in the last decade have radically changed the business environment for banks and savings institutions.

1.03 Although banks and savings institutions continue in their traditional role as financial intermediaries, the ways in which they carry out that role became increasingly complex in the most recent decade. Under continuing pressure to operate profitably, the industry adopted innovative approaches to carrying out the basic process of gathering and lending funds. The management of complex assets and liabilities, development of additional sources of income, reactions to technological advances, responses to changes in regulatory policy, and competition for deposits all added to the risks and complexities of the business of banking. These include the following:

- Techniques for managing assets and liabilities that allow institutions to manage financial risks and maximize income have evolved.
- Income, traditionally derived from the excess of interest collected over interest paid, became dependent on fees and other income streams from specialized transactions and services.

©2016. AICPA AAG-DEP 1.03

- Technological advances accommodated complex transactions, such as the sale of securities backed by cash flows from other financial assets.
- Regulatory policy alternately fostered or restricted innovation. Institutions have looked for new transactions to accommodate changes in the amount of funds they generally must keep in reserve or to achieve the desired levels of capital in relation to their assets.
- Regulatory policy has expanded and become increasingly complex in response to increasing complexities in the industry and recent economic recessions.
- **1.04** In addition, competition arose from within the industry, and also from other competitors such as investment companies, brokers and dealers in securities, insurers, and financial subsidiaries of commercial enterprises. These entities increased business directly with potential depositors and borrowers in transactions traditionally executed through banks and savings institutions. This disintermediation increased the need for innovative approaches to attracting depositors and borrowers.
- 1.05 This disintermediation also led to a sharp increase in consolidation within the financial institution industry, which created several large and highly complex financial holding companies. With the changes previously mentioned and the increased size of many financial institutions, a dramatic shift in lending, capital market activities, and sources of funding occurred. During this transformation of the industry, the regulatory system issued additional guidance in an effort to keep pace with the changes in the industry.
- 1.06 The economic recession, which officially began in 2007, revealed vulnerabilities in financial institutions and the regulatory system that contributed to unprecedented strain and stress on financial institutions and in financial markets. As a result, certain financial institutions either failed or came close to failure and many additional widespread repercussions affected or continue to affect this industry. Total assets of "problem" institutions reached their highest levels since 1993 during the first quarter of 2010, per the FDIC's Quarterly Banking Profile. In addition, the number of bank failures reached the highest level since 1992. The economic crisis fueled the demand for financial reform. As a result, on July 21, 2010, the president signed the Dodd-Frank Wall Street Reform and Consumer Protection Act (the Dodd-Frank Act) into law in response to weaknesses in the financial services industry that were believed to have contributed to the economic recession. See further discussion of the Dodd-Frank Act beginning at paragraph 1.31.
- **1.07** The innovation and complexity related to this industry creates a constantly changing body of business and economic risks. These risk factors, and related considerations for auditors, are identified and discussed throughout this guide.

# Regulation and Oversight

1.08 As previously discussed, the importance of financial intermediation has driven governments to play a role in the banking and savings institutions industry. Banks and savings institutions have been given unique privileges and protections, including the insurance of their deposits by the federal government

through the FDIC and access to the Federal Reserve's discount window and payments system. (See chapter 2, "Industry Overview—Credit Unions," of this guide for the roles and responsibilities of the National Credit Union Administration [NCUA]). Currently, the federal oversight of institutions receiving these privileges falls to the following three agencies:

- a. The Federal Reserve, established in 1913 as the central bank of the United States, which has supervisory responsibilities for bank and saving and loan holding companies, state chartered banks that are members of the Federal Reserve, and foreign banking organizations operating in the United States
- b. The FDIC, established in 1934 to restore confidence in the banking system through the federal insurance of deposits, which has supervisory responsibilities for state chartered banks and savings institutions that are not members of the Federal Reserve
- c. The Office of the Comptroller of the Currency (OCC), created in 1863, which regulates and provides federal charters for national banks and federal savings associations
- **1.09** The Federal Reserve and the FDIC are independent agencies of the federal government. The OCC is a bureau of the U.S. Department of Treasury (Treasury). Each state has a banking department and are members of an organization called the Conference of State Bank Supervisors.
- **1.10** Although each agency has its own jurisdiction and authority, the collective regulatory and supervisory responsibilities of federal and state banking agencies include the following:
  - Establishing (either directly or as a result of legislative mandate) the rules and regulations that govern institutions' operations
  - Supervising institutions' operations and activities
  - Reviewing and approving organization, conversion, consolidation, merger, or other changes in control of the institutions and their branches
  - Appraising (in part through on-site examinations) institutions' financial condition, the safety and soundness of operations, the quality of management, the adequacy and quality of capital, asset quality, liquidity needs, and compliance with laws and regulations
- 1.11 Given the nature of their duties to consider a bank's risk characteristics and loss behavior, the banking agencies also have significant influence in aiding banks and savings institutions with technical details on the application of U.S. generally accepted accounting principles (GAAP) in regulatory reporting. For example, the agencies also have certain authority over the activities of auditors serving the industry. Further, the Federal Reserve, the FDIC, the OCC, and the NCUA constitute the Federal Financial Institutions Examination Council (FFIEC). The FFIEC sets forth uniform examination and supervisory guidelines in certain areas related to banks' and savings institutions' and credit unions' activities, including those involving regulatory reporting matters.
- 1.12 This chapter discusses the current regulatory approach to the supervision of banks and savings institutions and provides an overview of major areas of regulation and related regulatory reporting. Legislative efforts over time to regulate, deregulate, and reregulate banks and savings institutions

are also addressed in this chapter. Other specific regulatory considerations are identified throughout this guide in the relevant chapters.

- 1.13 In addition to supervision and regulation by the federal and state banking agencies, publicly held holding companies are generally subject to the requirements of federal securities laws, including the Securities Act of 1933 and the Securities Exchange Act of 1934 (the 1934 Act). Holding companies whose securities are registered under the 1934 Act must comply with its reporting requirements through periodic filings with the SEC. Publicly held institutions that are not part of a holding company are required under Section 12(i) of the 1934 Act to make equivalent filings directly with their primary federal regulators. Each of the agencies has regulations that provide for the adoption of forms, disclosure rules, and other registration requirements equivalent to those of the SEC as mandated by the 1934 Act.
- 1.14 Both the Financial Institutions Reform, Recovery, and Enforcement Act of 1989 (FIRREA) and the Federal Deposit Insurance Corporation Improvement Act of 1991 (FDICIA) were adopted to protect the federal deposit insurance funds through the early detection and intervention in problem institutions, with an emphasis on capital adequacy.

# Regulatory Background

- 1.15 Declining real estate markets in the mid-1980s contributed heavily to widespread losses in the savings institutions industry, evidenced by the insolvency of the savings industry's federal deposit insurance fund. The FIRREA provided funds for the resolution of thrift institutions, replaced the existing regulatory structure, introduced increased regulatory capital requirements, established limitations on certain investments and activities, and enhanced regulators' enforcement authority. The FIRREA redefined responsibilities for federal deposit insurance by designating separate insurance funds, the Bank Insurance Fund (BIF), and the Savings Associations Insurance Fund (SAIF). The FIRREA also established the Resolution Trust Corporation (RTC) to dispose of the assets of failed thrifts. The RTC is no longer in existence and its work is now being done by the FDIC.
- 1.16 As the 1980s came to a close, record numbers of bank failures began to drain the BIF. The FDICIA provided additional funding for the BIF but also focused the least-cost resolution of and prompt corrective action (PCA) for troubled institutions and improved supervision and examinations. The FDICIA also focused the regulatory enforcement mechanism on capital adequacy. Many of the FDICIA's provisions were amendments or additions to the existing Federal Deposit Insurance Act (FDI Act).
- **1.17** In April 2006, the FDIC merged the BIF and the SAIF to form the Deposit Insurance Fund (DIF). This action was pursuant to the provisions in the Federal Deposit Insurance Reform Act of 2005 (Reform Act). Under the Reform Act, the FDIC may set the designated reserve ratio, calculated as the target insurance fund size as a percentage of estimated insured deposits, within a range of 1.15 percent to 1.50 percent of estimated insured deposits.
- 1.18 A desire to allow banks to serve a broad spectrum of customer financial needs caused Congress to pass legislation in 1999. The Gramm-Leach-Bliley Act (also known as the Financial Services Modernization Act) changed the types of activities that are permissible for bank holding company affiliates and for subsidiaries of banks. The bill created so-called financial holding

companies that may engage in a broad array of activities. Financial holding company affiliates could provide insurance as principal, agent, or broker and may issue annuities. These affiliates may engage in expanded underwriting, dealing in, or making a market in securities, as well as engage in expanded merchant banking activities. The legislation affirmed the concept of functional regulation.

- 1.19 Federal banking regulators continue to be the primary supervisors of the banking affiliates of financial holding companies and state insurance authorities supervise the insurance companies, and the SEC and securities self-regulatory organizations supervise the securities business. Each functional regulator determines appropriate capital standards for the companies it supervises. The Treasury and the Federal Reserve have the authority to approve additional activities to be permissible for financial holding companies. To maintain financial holding company status, all of a bank holding company's insured deposit taking subsidiaries must be "well capitalized," "well managed," and have at least a satisfactory Community Reinvestment Act rating.
- **1.20** In 1970, the Bank Secrecy Act (BSA) was enacted to address the problem of money laundering. The BSA authorized the Treasury to issue regulations requiring financial institutions to file reports, keep certain records, implement anti-money-laundering programs and compliance procedures, and report suspicious transactions to the government. (See Title 31 U.S. *Code of Federal Regulations* [CFR] Chapter X). These regulations, promulgated under the authority of the BSA, and subsequently the USA-Patriot Act of 2001, are intended to help federal authorities detect, deter, and prevent criminal activity. The Financial Crimes Enforcement Network (FinCEN), an arm of the Treasury, administers these regulations.
- 1.21 On December 2, 2014, the FFIEC released the revised Bank Secrecy Act/Anti-Money Laundering (BSA/AML) Examination Manual (manual). The revised manual provides current guidance on risk-based policies, procedures, and processes for banking organizations to comply with the BSA and safeguard operations from money laundering and terrorist financing. The manual has been updated to further clarify supervisory expectations and incorporate regulatory changes since the manual's 2010 update.
- 1.22 In 2002, the Sarbanes-Oxley Act was enacted in response to high-profile business failures which called into question the effectiveness of the CPA profession's self-regulatory process as well as the effectiveness of the audit to uphold the public trust in the capital markets. The requirements of the Sarbanes-Oxley Act and the SEC regulations implementing the Act are wide-ranging. The banking regulatory agencies also passed regulations implementing certain provisions of the Sarbanes-Oxley Act. Paragraphs 1.105–.117 provide additional information regarding regulatory issuances related to the Sarbanes-Oxley Act. In addition, the Sarbanes-Oxley Act created the PCAOB, which has the authority to set and enforce auditing, attestation, quality control, and ethics (including independence) standards for auditors of issuers. It also is empowered to inspect the auditing operations of public accounting firms that audit issuers as well as impose disciplinary and remedial sanctions for violations of the board's rules, securities laws, and professional auditing and accounting standards.
- **1.23** Key economic issues affecting the regulations are centered on the ability of financial institutions to operate profitably—for example, the costs

and benefits of regulations, the effects of unemployment and future corporate layoff plans, levels of interest rates, and the availability of credit.

# **Deposit Insurance Fund**

- 1.24 On October 7, 2008, the FDIC established a Restoration Plan for the DIF to return the DIF to its statutorily mandated minimum reserve ratio of 1.15 percent within 5 years. In February 2009, the FDIC amended its Restoration Plan to extend the restoration period from 5 to 7 years. Congress then amended the statute governing the Restoration Plan, in May 2009, to allow the FDIC up to 8 years to return the DIF reserve ratio to 1.15 percent. In September 2009, the FDIC amended the Restoration Plan consistent with the statutory change and, pursuant to the amended Restoration Plan, adopted a uniform 3 basis point increase in initial assessment rates effective January 1, 2011.
- 1.25 The Dodd-Frank Act requires the FDIC to set a designated reserve ratio of not less than 1.15 percent for any year and to increase the level of the DIF to 1.35 percent of estimated insured deposits by September 30, 2020.¹ In March 2016, the FDIC approved a final rule, effective July 1, to increase the DIF to the statutorily required minimum level of 1.35 on institutions with total consolidated assets of \$10 billion or more while providing credits to institutions that have assets or less than \$10 billion. Readers are encouraged to consult the full text of this final rule on FDIC's website at www.fdic.org. The Dodd-Frank Act also called for a revision to the definition of the deposit insurance assessment base. The intent of changing the assessment base was to shift a greater percentage of overall total assessments away from community institutions and toward the largest institutions.
- 1.26 In response to the provisions of the Dodd-Frank Act, in February 2011, the FDIC's board of directors, through the issuance of Financial Institution Letter (FIL)-8-2011, adopted the final rule *Deposit Insurance Assessment Base, Assessment Rate Adjustments, Dividends, Assessment Rates, and Large Bank Pricing Methodology* to redefine the deposit insurance assessment base, as required by the Dodd-Frank Act; alter the assessment rates; implement the Dodd-Frank Act's DIF dividend provisions; and revise the risk-based assessment system for all large insured depository institutions (IDIs).<sup>2</sup> The final rule
  - redefines the *deposit insurance assessment base* as average consolidated total assets minus average tangible equity (the assessment base had previously been defined as total domestic deposits).
  - makes generally conforming changes to the unsecured debt and brokered deposit adjustments to assessment rates.
  - creates a depository institution debt adjustment.
  - eliminates the secured liability adjustment.

 $<sup>^{1}</sup>$  The Deposit Insurance Fund (DIF) is used to (a) insure the deposits of, and protect the depositors of, failed FDIC-insured institutions and (b) resolve failed FDIC-insured institutions upon appointment of the FDIC as receiver. The reserve ratio represents the ratio of the net worth of the DIF to aggregate estimated insured deposits of FDIC-insured institutions. The DIF is funded primarily through deposit insurance assessments.

 $<sup>^2</sup>$  A large insured depository institution (IDI) has at least \$10 billion in total assets. In general, a highly complex IDI will be (a) an IDI (other than a credit card bank) with more than \$50 billion in total assets that is controlled by a parent or an intermediate parent company with more than \$500 billion in total assets or (b) a processing bank or trust company with at least \$10 billion in total assets.

- adopts a new assessment rate schedule which became effective April 1, 2011, and, in lieu of dividends, other rate schedules when the reserve ratio reaches certain levels.
- 1.27 In addition, the final rule establishes a new methodology for calculating deposit insurance assessment rates for highly complex and other large IDIs (commonly referred to as the Large Bank Pricing Rule). The new methodology combines capital adequacy, asset quality, management, earnings, liquidity, and sensitivity to market risk (CAMELS) ratings and financial measures to produce a score that is converted into an institution's assessment rate. The Large Bank Pricing Rule authorizes the FDIC to adjust, up or down, an institution's total score by 15 basis points. The final rule became effective on April 1, 2011. For further information, readers can access the final rule on the FDIC website at www.fdic.gov.
- 1.28 In September 2011, the FDIC adopted guidelines describing the process that the FDIC will follow to determine whether to make an adjustment, to determine the size of any adjustment, and to notify an institution of an adjustment made to its assessment rate score, as allowed under the Large Bank Pricing Rule. The guidelines also provide examples of circumstances that might give rise to an adjustment. Further information on the guidelines can be found in FIL-64-2011, Assessments: Assessment Rate Adjustment Guidelines, on the FDIC website at www.fdic.gov.
- 1.29 In October 2012, the FDIC's board of directors, through the issuance of FIL-44-2012, Assessments: Final Rule on Assessments, Large Bank Pricing, adopted a final rule to amend and clarify definitions related to higher risk assets as used by the deposit insurance pricing scorecards for large and highly complex IDIs. The rule applies only to institutions with \$10 billion or more in assets. Specifically, the rule revises the definition of certain higher risk assets, such as leveraged loans and subprime consumer loans; clarifies the timing of identifying an asset as higher risk; clarifies the way securitizations (including those that meet the definition of nontraditional mortgage loans) are identified as higher risk; and further defines terms that are used in the large bank pricing rule adopted in February 2011. The final rule became effective on April 1, 2013. For further information, readers are encouraged to access the final rule in FIL-44-2012 on the FDIC website at www.fdic.gov.
- **1.30** In November 2014, the FDIC issued the *Assessments* final rule to revise the FDIC's risk-based deposit insurance assessment system to reflect changes in the regulatory capital rules. The final rule
  - conforms the capital ratios and ratio thresholds in the small institution assessment system to the new PCA capital ratios and ratio thresholds.
  - conforms the assessment base calculation for custodial banks to the new asset risk weights using the standardized approach in the regulatory capital rules.
  - requires that all highly complex institutions measure counterparty exposure for the assessment purposes using the Basel III standardized approach credit equivalent amount for derivatives and the Basel III standardized approach exposure amount for securities financing transactions in the regulatory capital rules.

For further information, readers can access the final rule in FIL-57-2014, Assessments: Final Rule, on the FDIC website at www.fdic.gov.

### The Dodd-Frank Act

- 1.31 The Dodd-Frank Act was signed into law by President Obama on July 21, 2010. It aims to promote U.S. financial stability by improving accountability and transparency in the financial system, putting an end to the belief that certain financial institutions were too big to fail, protecting American taxpayers by ending bailouts, and protecting consumers from abusive financial services practices. The Dodd-Frank Act contains many provisions; some highlights that may be of particular interest to readers are summarized in the following sections.
- **1.32** A copy of the full Dodd-Frank Act, as signed by the president, can be found at www.gpo.gov. The AICPA is also following any developments related to the Dodd-Frank Act on its website at www.aicpa.org on the "Federal Issues" page under "Advocacy."

### Financial Stability Oversight Council

1.33 The Dodd-Frank Act created a new systemic risk regulator called the Financial Stability Oversight Council (FSOC). The two main goals of the FSOC are to identify risks to the financial stability of the United States banking system and to promote market discipline by eliminating the moral hazard of "too big to fail." To meet these goals, the FSOC has many powers to identify any company, product, or activity that could threaten U.S. financial stability. The FSOC is chaired by the Secretary of the Treasury, and voting members are heads of nine federal financial regulatory agencies, including chairmen of the Federal Reserve, the FDIC, and the SEC, among others. The FSOC is authorized to facilitate regulatory coordination, facilitate information sharing and collection, designate nonbank financial companies for consolidated supervision, designate systemic financial market utilities and systemic payment, clearing or settlement activities, and recommend stricter standards for the largest, most interconnected firms, break up firms that pose a "grave threat" to financial stability, and recommend Congress close specific gaps in regulation. Further information on the FSOC and proposed rulings can be found at www.treasury.gov/initiatives/Pages/FSOC-index.aspx.

# Leverage and Risk-Based Capital Requirements

1.34 Title 1, "Financial Stability," of the Dodd-Frank Act requires the appropriate federal banking agencies to establish minimum leverage and risk-based capital requirements, on a consolidated basis, for IDIs, depository institution holding companies, and nonbank financial companies supervised by the Federal Reserve. The minimum leverage and risk-based capital requirements for IDIs established by the agencies under this section of the Dodd-Frank Act should not be less than the generally applicable requirements, which should serve as a floor for any capital requirements that the agencies may require, nor be quantitatively lower than the generally applicable requirements that were in effect for IDIs as of the date of enactment. The provisions of Section 171 of the Dodd-Frank Act regarding trust preferred securities can be found in paragraph 17.20 of this guide.

**1.35** Title VI, "Improvements to Regulation," of the Dodd-Frank Act mandates stronger capital requirements for all IDIs, depository institution holding companies, and any company that controls an IDI and provides that any company in control be accountable for the financial strength of that entity.

### Consumer Financial Protection Bureau

1.36 The Consumer Financial Protection Bureau (CFPB) is an independent agency that consolidates much of the federal regulation of financial services offered to consumers. The CFPB is expected to ensure that consumers receive clear, accurate information to shop for mortgages, credit cards, and other financial products (but not products subject to securities or insurance regulations); to provide consumers with one dedicated advocate; and to protect them from hidden fees and deceptive practices. The CFPB also oversees the enforcement of federal laws intended to ensure the fair, equitable, and nondiscriminatory access to credit for individuals. The director of the CFPB replaces the director of the Office of Thrift Supervision (OTS) on the FDIC board. The CFPB is led by an independent director appointed by the president and confirmed by the Senate and has a dedicated budget in the Federal Reserve.

1.37 The CFPB has the authority to examine and enforce regulations for banks and credit unions with assets of over \$10 billion; all mortgage-related businesses (nondepository institution lenders, servicers, mortgage brokers, and foreclosure operators); providers of payday loans; student lenders; and other nonbank financial entities, such as debt collectors and consumer reporting agencies. Banks and credit unions with assets of \$10 billion or less will be examined for consumer compliance by the appropriate regulator. The CFPB also is able to autonomously write rules for consumer protections governing all financial institutions (banks and nonbanks) offering consumer financial services or products.

**1.38** For further information on the CFPB and the progress the agency has made since its inception, readers can access the CFPB website at www .consumerfinance.gov.

# **Derivatives Trading**

1.39 The Dodd-Frank Act provided the SEC and the Commodity Futures Trading Commission (CFTC) with the authority to regulate over-the-counter derivatives and required central clearing and exchange trading for derivatives. The SEC has regulatory authority over specific security-based swaps (including credit default swaps), and the CFTC has primary regulatory authority over all other swaps, including energy-rate swaps, interest-rate swaps, and broad-based security group or index swaps. Standardized swaps will be traded on an exchange or in other centralized trading facilities, which will promote transparency; standardized derivatives will also have to be handled by central clearinghouses. The Dodd-Frank Act requires all cleared swaps to be traded on a registered exchange or board of trade.<sup>3</sup>

©2016, AICPA AAG-DEP 1.39

<sup>&</sup>lt;sup>3</sup> The SEC has proposed numerous rulings related to the provisions on derivative trading included in the Dodd-Frank Wall Street Reform and Consumer Protection Act (Dodd-Frank Act). Readers are encouraged to visit the "Dodd-Frank Act Rulemaking: Derivatives" page on the SEC website to access further information.

- **1.40** The Dodd-Frank Act also provided regulators the authority to impose capital and margin requirements on swap dealers and major swap participants. The credit exposure from derivative transactions will be considered in banks' lending limits.
- **1.41** Banks are allowed to continue engaging in principal transactions involving interest-rate, foreign-exchange, gold, silver, and investment-grade credit default swaps, subject to Section 619 of the Dodd-Frank Act (commonly referred to as the Volcker Rule) limitations on proprietary trading. See discussion of the Volcker Rule in paragraph 1.54. For commodities, most other metals, energy, and equities, banks must shift their swap operations to a separately capitalized affiliate within the holding entity.

## **Lending Limits**

- 1.42 Section 610 of the Dodd-Frank Act revises the statutory definition of loans and extensions of credit to include credit exposures arising from derivative transactions, repurchase agreements, reverse repurchase agreements, securities lending transactions, and securities borrowing transactions (collectively, securities financing transactions). This revised definition also is applicable to all savings associations.
- 1.43 In June 2013, the OCC finalized its lending limits interim rule, which consolidated the lending limits rules applicable to national banks and savings associations, removed the separate OCC regulation governing lending limits for savings associations, and implemented Section 610 of the Dodd-Frank Act. The final rule outlines the methods that banks can choose from to measure credit exposures of derivative transactions and securities financing transactions. A bank may choose which method it will use; however, the OCC may specify that a bank use a particular method for safety and soundness reasons. Banks may request OCC approval to use a different method to calculate credit exposure for certain transactions. If the Model Method<sup>5</sup> is used, the OCC must approve the use of the model and any subsequent changes to an approved model. The final rule continues to provide that loans and extensions of credit, including those that arise from derivative transactions and securities financing transactions, must be consistent with safe and sound banking practices.
- **1.44** *Derivative transactions.* Banks can generally choose to measure the credit exposure of derivatives transactions through
  - the Conversion Factor Matrix Method.<sup>6</sup>

<sup>&</sup>lt;sup>4</sup> In November 2015, the Office of the Comptroller of the Currency (OCC), the Board of Governors of the Federal Reserve System (Federal Reserve), the FDIC, the Farm Credit Administration, and the Federal Housing Finance Agency (collectively, the agencies) issued the final rule *Margin and Capital Requirements for Covered Swap Entities* to implement Sections 731 and 764 of the Dodd-Frank Act. The final regulations establish minimum margin and capital requirements for registered swap dealers, major swap participants, security-based swap dealers, and major security-based swap participants for which one of the agencies is the prudential regulator. The final rule was effective April 1, 2016. Readers may access the full text of the regulation from any of the agencies websites.

 $<sup>^5</sup>$  Under the Model Method, the credit exposure of a derivative transaction should equal the sum of the current credit exposure of the derivative transaction and the potential future credit exposure of the derivative transaction. See Title 12 U.S. *Code of Federal Regulations* (CFR) Part 32.9 for further discussion on the calculation of current credit exposure and the potential future credit exposure.

<sup>&</sup>lt;sup>6</sup> Under the Conversion Factor Matrix Method, credit exposure arising from a derivative transaction should equal and remain fixed at the potential future credit exposure of the derivative transaction, which should equal the product of the notional principal amount of the derivative transaction and a fixed multiplicative factor utilizing the conversion factor matrix found in Table 1 to 12 CFR 32.9.

- the Current Exposure Method.<sup>7</sup>
- an OCC-approved internal model.
- **1.45** For credit derivatives (transactions in which banks buy or sell credit protection against loss on a third-party reference entity), the final rule provides a special rule for calculating credit exposure based on exposure to the counterparty and reference entity.
- **1.46** Securities financing transactions. The final rule specifically exempts securities financing transactions relating to Type I securities (such as U.S. or state government obligations) from the lending limits calculations. For other securities financing transactions, banks can choose to measure credit exposure by the following methods:
  - Locking in the attributable exposure based on the type of transaction
  - Using an OCC-approved internal model
  - Using the Basel Collateral Haircut Method<sup>8</sup>

1.47 Information for community banks. The final rule minimizes the compliance burden on small and midsize banks of measuring the credit exposure of derivative transactions and securities financing transactions by providing different options for measuring the exposures for each transaction type. The options permit banks to adopt compliance alternatives that fit their size and risk management requirements, consistent with safety and soundness and the goals of the statute. Community banks should note that derivative transactions include interest rate swaps; however, community banks may use the Conversion Factor Matrix Method, which is an easy-to-use lookup table that locks in the attributable exposure at the execution of the transaction. The simplest calculation of securities financing transactions, excluding those related to Type 1 securities, is the Basic Method, which locks in the attributable exposure based on the type of transaction.

### Securitization

**1.48** The Dodd-Frank Act requires changes to rules and regulations for securitization transactions. The Dodd-Frank Act also requires entities that sponsor products such as mortgage-backed securities to retain at least 5 percent of the credit risk, unless the underlying loans meet standards that reduce the risk. It also requires these sponsors to disclose more information about the underlying assets, including analysis of the quality of the underlying assets.

©2016. AICPA AAG-DEP 1.48

<sup>&</sup>lt;sup>7</sup> Under the Current Exposure Method, credit exposure for derivative transactions is calculated by adding the current exposure (the greater of zero or the mark-to-market value) and the potential future credit exposure (calculated by multiplying the notional amount by a specified conversion factor taken from Table 4 of the Advanced Approaches Appendix of the capital rules, which varies based on the type and remaining maturing of the contract) of the derivative transactions. The current exposure method incorporates additional calculations for netting arrangements and collateral and utilizes multipliers that are more tailored to compute the potential future credit exposure of derivative transactions.

<sup>&</sup>lt;sup>8</sup> The Basel collateral haircut method applies standard supervisory haircuts (the percentage reduction of the amount that will be repaid to creditors) for measuring counterparty credit risk for such transactions under the capital rules' Basel II Advanced Internal Ratings-Based Approach or the Basel III Advanced Approaches.

- 1.49 In January 2011, the SEC adopted new rules related to representations and warranties in asset-backed securities offerings, as outlined in Release No. 33-9175, Disclosure for Asset-Backed Securities Required by Section 943 of the Dodd-Frank Wall Street Reform and Consumer Protection Act. These rules require securitizers of asset-backed securities to disclose fulfilled and unfulfilled repurchase requests. The rules also require nationally recognized statistical rating organizations to include information regarding the representations, warranties and enforcement mechanisms available to investors in an asset-backed securities offering in any report accompanying a credit rating issued in connection with such offering, including a preliminary credit rating.
- **1.50** Pursuant to Section 945 of the Dodd-Frank Act, the SEC issued Release No. 33-9176, *Issuer Review of Assets in Offerings of Asset-Backed Securities*, which requires any issuer registering the offer and sale of an asset-backed security to perform a review of the assets underlying the asset-backed security. In addition, the rule amended Regulation AB by requiring an asset-backed security issuer to disclose the nature, findings, and conclusion of its review of the assets.
- **1.51** Section 942(a) of the Dodd-Frank Act eliminated the automatic suspension of the duty to file under Section 15(d) of the 1934 Act for asset-backed securities issuers and granted the SEC the authority to issue rules providing for the suspension or termination of such duty. To implement Section 942(a), the SEC issued Release No. 34-65148, Suspension of the Duty to File Reports for Classes of Asset-Backed Securities Under Section 15(d) of the Securities Exchange Act of 1934, which establishes rules to provide certain thresholds for suspension of the reporting obligations for asset-backed securities issuers.
- 1.52 In October 2014, the OCC, the Federal Reserve, the FDIC, the SEC, the Federal Housing Finance Agency, and the U.S. Department of Housing and Urban Development (HUD) issued the joint final rule *Credit Risk Retention* to implement the credit risk retention requirements of Section 15G of the 1934 Act, as added by Section 941 of the Dodd-Frank Act. Section 15G generally requires the securitizer of asset-backed securities to retain not less than 5 percent of the credit risk of the assets collateralizing the asset-backed securities. Section 15G includes a variety of exemptions for these requirements, including an exemption for asset-backed securities that are collateralized exclusively by residential mortgages that qualify as qualified residential mortgages. The final rule became effective February 23, 2015.
- 1.53 In connection with making amendments to its "safe harbor" rule that were necessary due to the implementation of FASB Statement No. 166, Accounting for Transfers of Financial Assets, an amendment of FASB Statement No. 140 (codified in FASB Accounting Standards Codification [ASC] 860, Transfers and Servicing), and FASB Statement No. 167, Amendments to FASB Interpretation No. 46(R) (codified in FASB ASC 810, Consolidation), the FDIC included a condition to qualify for the safe harbor that, among other conditions, sponsors must retain an economic interest of no less than 5 percent of the credit risk of the financial assets underlying a securitization until the joint interagency regulations that are required to be adopted under the Dodd-Frank Act

<sup>&</sup>lt;sup>9</sup> In August 2011, the SEC made a technical correction to the final ruling due to an incorrect paragraph reference in an instruction to Rule 15Ga-1. See Release No. 33-9175A, Disclosure for Asset-Backed Securities Required by Section 943 of the Dodd-Frank Wall Street Reform and Consumer Protection Act.

become effective. The sponsor is not permitted to hedge the credit risk of the retained interest but may hedge certain other risks (such as interest rate and currency). Other conditions are also necessary to qualify for the safe harbor. The rule grandfathers the previous safe harbor rule for transfers of financial assets on or prior to December 31, 2010. For further information on the FDIC's safe harbor rule, readers are encouraged to access the September 27, 2010, board minutes from the "FDIC Board Meetings" page on the FDIC website.

### Volcker Rule

**1.54** The Volcker Rule prohibits banking entities and affiliated companies from proprietary trading; acquiring or retaining any equity, partnership, or other ownership interest in a hedge fund or private equity fund; and sponsoring a hedge fund or private equity fund. Proprietary trading consists of transactions made by an entity that affect the entity's own account but not the accounts of its clients. Banks are allowed to make de minimis investments in hedge funds and private equity funds using no more than 3 percent of their tier 1 capital in all such funds combined. Also, a bank's investment in a private fund may not exceed 3 percent of the fund's total ownership interest. Nonbank financial institutions supervised by the Federal Reserve also have restrictions on proprietary trading, hedge fund investments, and private equity investments. See discussion on final rulings enacted as a result of the Volcker Rule in paragraphs 18.77–.78 of this guide.

## **Thrift Regulations**

- **1.55** The Dodd-Frank Act abolished the OTS, which had been the federal supervisor for federal savings associations and thrift holding companies. Its authority for federal savings associations and rulemaking for all savings associations was transferred to the OCC, its authority for state savings associations was transferred to the FDIC, and its authority for thrift holding companies (also known as savings and loan holding companies or SLHCs) was transferred to the Federal Reserve. However, the thrift charter has been preserved. In January 2011, the Federal Reserve, the FDIC, the OCC, and the OTS issued a *Joint Implementation Plan* to provide an overview of actions taken by the agencies to efficiently and effectively implement Sections 301–326 of the Dodd-Frank Act. The transfer of authority took place on July 21, 2011, and certain regulations have been enacted in response, as subsequently discussed.
- **1.56** In July 2011, the OCC issued an interim final rule that republishes regulations issued by the OTS, prior to its transfer of powers, that the OCC has authority to promulgate and enforce. This rule, which was effective immediately, renumbers and issues these former OTS regulations as new OCC regulations (recodified in Chapter I at Parts 100–197), with nomenclature and other technical amendments to reflect the OCC supervision of federal savings associations. These newly issued OCC regulations supersede the OTS regulations for purposes of the OCC supervision of federal savings associations.
- 1.57 In August 2011, the Federal Reserve issued an interim final rule establishing regulations for SLHCs. This rule provides for the corresponding transfer from the OTS to the Federal Reserve of the regulations necessary for the Federal Reserve to administer the statutes governing SLHCs. The three components to the rule include new Regulation LL (Part 238), which sets forth regulations generally governing SLHCs; new Regulation MM (Part 239), which sets forth regulations governing SLHCs in mutual form; and technical

amendments to current Federal Reserve regulations necessary to accommodate the transfer of supervisory authority for SLHCs from the OTS to the Federal Reserve.

- **1.58** In August 2011, the FDIC published an interim final rule reissuing and redesigning certain transferring OTS regulations. In republishing these rules, the FDIC only made technical changes to existing OTS regulations. The OTS regulations were recodified in Chapter III at Parts 390–391.
- 1.59 In December 2011, the OCC issued Bulletin OCC 2011-47, OTS Integration: Supervisory Policy Integration Process, to outline the process that the OCC intends to follow to fully integrate the OTS policy guidance documents into a common set of supervisory policies that applies to both national banks and federal savings associations. Phase 1 involves rescinding a significant number of documents including OTS documents that transmitted or summarized rules, interagency guidance, or Examination Handbook sections that are no longer useful because of the elimination of the OTS, the passage of time, or duplicate existing OCC guidance. The OCC has announced the rescission through numerous bulletins. Phase II focuses on guidance that requires further review, substantive revision, or combination or guidance that is considered unique to federal savings associations. Readers are encouraged to access the "OTS Integration" page on the OCC website for further developments on the integration of the two agencies. 10

### Resolution Plans 11

- **1.60** The FDIC and the Federal Reserves issued a joint rule to implement Section 165(d) of the Dodd-Frank Act. This rule requires bank holding companies with assets of \$50 billion or more and companies designated as systemically important by the FSOC to report periodically to the FDIC and the Federal Reserve the company's plan for its rapid and orderly resolution in the event of material financial distress or failure.
- **1.61** The goal of this rule is to achieve a rapid and orderly resolution of an organization that would not cause a systemic risk to the financial system. The final rule also establishes specific standards for the resolution plans (commonly referred to as *living wills*), including requiring a strategic analysis of the plan's components; a description of the range of specific actions to be taken in the resolution; and analyses of the company's organization, material entities, interconnections and interdependencies, and management information systems, among other elements.
- **1.62** The rule requires companies to update their plans annually. A company that experiences a material event after a plan is submitted has 45 days to notify regulators of the event.

Office of Thrift Supervision (OTS) policies and guidance remain applicable to federal savings associations until rescinded, superseded, or revised. In some cases, the OCC may amend an OTS rule, policy, or practice that is cross-referenced in more than one document or affects only a portion of a document. If overlapping guidance exists, any guidance or regulation issued by the OCC after July 21, 2011, that specifically includes federal savings associations in its scope will prevail. If a document has not been rescinded, but a portion of the content no longer applies, the superseded portion will be grayed out electronically.

<sup>&</sup>lt;sup>11</sup> In December 2014, the FDIC issued guidance for resolutions plans that IDIs with assets greater than \$50 billion must submit periodically to the FDIC. The guidance includes direction regarding the elements that should be discussed in a fully developed resolution strategy and the cost analysis, clarification regarding assumptions made in the plan, and a list of significant obstacles to an orderly and least costly resolution that institutions should address.

- **1.63** Separately, the FDIC's board of directors approved a complementary final rule under the FDI Act to require IDIs with \$50 billion or more in total assets to submit periodic contingency plans to the FDIC for resolution in the event of the depository institution failure. The final rule became effective on April 1, 2012.
- **1.64** The final rule requires these IDIs to submit a resolution plan that will enable the FDIC, as receiver, to resolve the bank to ensure that depositors receive access to their insured deposits within one business day of the institution's failure, maximize the net present value return from the sale or disposition of its assets, and minimize the amount of any loss to be realized by the institution's creditors.
- **1.65** Both the final rule related to certain bank holding companies and systemically important companies and the final rule related to certain IDIs can be found on the FDIC website at www.fdic.gov.

### Stress Testing

- 1.66 Section 165(i) of the Dodd-Frank Act requires certain companies to conduct annual stress tests (commonly referred to as Dodd-Frank Act Stress Testing) in accordance with the regulations proposed by their respective primary financial regulatory agencies, as well as semiannual company-run stress tests. Specifically, it requires the primary financial regulatory agency to define the stress tests; establish methodologies for the conduct of the stress tests, which must include at least three different sets of conditions (baseline, adverse, and severely adverse); establish the form and content of the report that institutions are required to submit; and instruct the institution to publish a summary of the results of the Dodd-Frank Act institutional stress test.
- **1.67** In May 2012, the Federal Reserve, the OCC, and the FDIC jointly issued final supervisory guidance on stress testing for banking organizations with more than \$10 billion in total consolidated assets that became effective on July 23, 2012. The guidance highlights the importance of stress testing as an ongoing risk management practice that supports a banking organization's forward-looking assessment of its risks. In addition, the guidance highlights five principles that should be part of a banking organization's stress testing framework. The framework should (a) include activities and exercises that are tailored to the exposures, activities, and risks of the organization; (b) employ multiple conceptually sound activities and approaches; (c) be forward looking and flexible; (d) be clear, actionable, well supported, and used in the decision making process, and (e) include strong governance and effective internal control. Furthermore, the guidance discusses four types of stress testing approaches and applications, which include scenario analysis, sensitivity analvsis, enterprise-wide stress testing, and reverse stress testing. Readers can access the supervisory guidance from any of the agencies' websites.
- **1.68** In conjunction with the release of stress testing guidance, the Federal Reserve, the FDIC, and the OCC also released a statement to clarify that community banks are not required or expected to conduct the type of stress testing required of larger organizations. However, the statement also noted that all banking organizations, regardless of size, should have the capacity to analyze the potential impact of adverse outcomes on their financial condition. Examples of such interagency guidance that addresses potential adverse outcomes as a part of sound risk management practices include, but are not limited to,

interest rate risk (IRR) management, commercial real estate concentrations, and funding and liquidity management.

**1.69** On October 9, 2012, the Federal Reserve, the FDIC, and the OCC issued final rules on company-run stress testing for companies with more than \$10 billion in total assets as required by the Dodd-Frank Act. Readers can access the stress test requirements of each agency from the respective agencies' websites. <sup>12</sup>

# **Regulatory Capital Matters**

- 1.70 Capital is the primary tool used by regulators to monitor the financial health of insured financial institutions. Regulatory intervention is focused primarily on an institution's capital levels relative to regulatory standards. The agencies have a uniform framework for PCA, as well as specific capital adequacy guidelines set forth by each agency. <sup>13</sup>
- 1.71 In addition to assessing financial statement disclosures, which are discussed in chapter 17, "Equity and Disclosures Regarding Capital Matters," of this guide, the auditor considers regulatory capital from the perspective that noncompliance or expected noncompliance with regulatory capital requirements may be a condition, when considered with other factors, that could indicate substantial doubt about an entity's ability to continue as a going concern. This discussion provides an overview to help auditors understand regulatory capital requirements. Capital regulations are complex, and their application by management requires a thorough understanding of specific requirements and the potential impact of noncompliance. Accordingly, the auditor should consult the relevant regulations and regulatory guidance, as necessary, when considering regulatory capital matters.

# Capital Adequacy

1.72 The FDIC, the OCC, and the Federal Reserve historically had common capital adequacy guidelines which differed in some respects from those of the OTS, prior to its transfer of powers, involving minimum (a) leverage capital and (b) risk-based capital requirements. <sup>14</sup> Capital adequacy guidelines are now substantially the same for banks and savings associations. A summary of the general requirements follows. Specific requirements are set forth in Title 12, Banks and Banking, of U.S. CFR and in the instructions for the FFIEC's Consolidated Reports of Condition and Income (Call Report) and the Federal Reserve's Consolidated Financial Statements for Holding Companies—FR Y-9C. The reports are required to be filed quarterly and contain certain financial

<sup>&</sup>lt;sup>12</sup> In March 2014, the OCC, the Federal Reserve, and the FDIC issued Supervisory Guidance for Banking Organizations With Total Consolidated Assets of More Than \$10 Billion but Less Than \$50 Billion.

 $<sup>^{13}</sup>$  This chapter discusses federal capital requirements. Separate state requirements may exist that also should be considered for purposes of assessing the entity's ability to continue as a going concern.

<sup>&</sup>lt;sup>14</sup> In accordance with FASB Accounting Standards Codification 942-505-50-1G, savings institution holding companies are not subject to regulatory capital requirements separate from those of their subsidiaries. Bank holding companies do have capital requirements separate from those of their subsidiaries. Chapter 17, "Equity and Disclosures Regarding Capital Matters," of this guide provides additional guidance.

information, including information used in calculating regulatory capital ratios and amounts.  $^{15}$ 

1.73 The OCC, the Federal Reserve, and the FDIC established a minimum common equity tier 1 capital ratio of 4.5 percent, tier 1 capital ratio of 6 percent, total capital ratio of 8 percent, and leverage ratio of 4 percent. The capital rules limit capital distributions and certain discretionary bonus payments if banks do not maintain a capital conservation buffer of common equity tier 1 capital above minimum capital requirements. Advanced approaches organizations (defined as banking organizations with \$250 billion or more in total consolidated assets or total consolidated on-balance sheet foreign exposure of \$10 billion or more) must also maintain a minimum supplementary leverage ratio of 3 percent. Although advanced approaches banking organizations are not required to comply with the minimum supplementary leverage ratio until January 1, 2018, they were required to begin reporting the ratio as of January 1, 2015. By statute, the FDIC and the OCC also require all federal and state savings associations to maintain a tangible capital requirement of 1.5 percent of assets. The advanced approaches and standardized capital ratio calculations can be found at 12 CFR 3.10 (OCC), 12 CFR 217.10 (Federal Reserve), and 12 CFR 324.10 (FDIC).

### © Update 1-1 Regulatory: Regulatory Capital Rules

The regulatory final rule Regulatory Capital Rules: Regulatory Capital, Enhanced Supplementary Leverage Ratio Standards for Certain Bank Holding Companies and Their Subsidiary Insured Depository Institutions, issued by the Federal Reserve, the OCC, and the FDIC in April 2014, will become effective on January 1, 2018. Banking organizations subject to the requirements were required to calculate and publicly disclose the ratio beginning January 1, 2015.

The final regulatory rule strengthens the agencies' supplementary leverage ratio standards for large, interconnected U.S. banking organizations and is applicable to any U.S. top-tier bank holding company with more than \$700 billion in total consolidated assets or more than \$10 trillion in assets under custody and any IDI subsidiary of these bank holding companies. The final rule establishes enhanced supplementary leverage ratio standards for covered bank holding companies and their subsidiary IDIs. Among other provisions of the final rule, an IDI that is a subsidiary of a covered bank holding company must maintain a supplementary leverage ratio of at least 6 percent to be well capitalized under the agencies' PCA framework.

Readers are encouraged to consult the full text of this final rule at any of the respective agencies' websites.

**1.74** Risk-based capital standards of the FDIC, the OCC, and the Federal Reserve explicitly identify concentrations of credit risk, risks of nontraditional

©2016, AICPA AAG-DEP 1.74

<sup>&</sup>lt;sup>15</sup> Banking agencies provide additional regulatory capital guidance through examination manuals and other communications, such as Supervision and Regulation (SR) letters issued by the Federal Reserve, Financial Institution Letters issued by the FDIC, and Bulletins issued by the OCC. The OTS provided additional regulatory capital guidance through examination manuals and other communications such as CEO memos, thrift bulletins, and regulatory bulletins. Readers are encouraged to visit the "OTS Integration" page of the OCC website for further information regarding OTS documents, which have either been rescinded or maintained.

activities, and IRR as qualitative factors to be considered in examiner assessments of an institution's overall capital adequacy; however, the standards require no specific quantitative measure of such risks.

- 1.75 The FDIC, the OCC, and the Federal Reserve have augmented their IRR requirements through a joint policy statement, *Joint Agency Policy Statement on Interest Rate Risk*, that explains how examiners will assess institutions' IRR exposure. <sup>16,17</sup> The policy statement also suggests that institutions with complex systems for measuring IRR may seek assurance about the institution's risk management process from internal and external auditors.
- 1.76 The Market Risk Rule (MRR) establishes risk-based regulatory capital requirements for bank holding companies, state member banks, SLHCs, national banks, federal savings associations, and state savings associations (collectively, banking organizations) with significant exposure to certain market risks. The MRR implements the Amendment to the Capital Accord (Market Risk Amendment or MRA) to incorporate market risks issued by the Basel Committee on Banking Supervision (Basel Committee) in 1996 and modified in 1997, 2005, 2009, and 2010. The MRR is set forth at 12 CFR 217, subpart F (Federal Reserve), 12 CFR 3, subpart F (OCC), and 12 CFR 324, subpart F (FDIC).
- 1.77 The effect of the market risk capital rules is that any banking organization regulated by the federal banking agencies, with significant exposure to market risk, generally must measure that risk using its own internal value at risk model, and hold a commensurate amount of capital. The amount of capital required to be held includes tier 1 and tier 2 capital. The regulatory capital requirements only apply to banking organizations whose trading activity on a worldwide consolidated basis equals 10 percent or more of the total assets or totals \$1 billion or more.
- 1.78 In June 2012, the OCC, the Federal Reserve, and the FDIC amended the market risk capital rule. The amendment revises the calculation of market risk to better characterize the risks facing a particular institution and to help ensure the adequacy of capital related to the institution's market risk-related positions. Under the amendment, additional charges were implemented for stressed VaR, credit risk, correlation trading, and other securitizations. The amendment became effective on January 1, 2013, and can be accessed from any of the agencies' websites.
- **1.79** Institutions are required to report certain financial information to regulators in quarterly Call Reports, which include amounts used in calculations of the institution's various regulatory capital ratios and amounts.
- **1.80** Under the capital adequacy standards of the OCC, the Federal Reserve, and the FDIC, a banking organization must deduct certain assets from common equity tier 1 capital. A banking organization is permitted to net associated deferred tax liability against some of those assets prior to making the deduction from tier 1 capital, if the deferred tax liability is associated with the assets and the deferred tax liability would be extinguished if the associated asset becomes impaired or is derecognized under GAAP. Deductions from common equity tier 1 capital include goodwill and other intangible, deferred tax

<sup>&</sup>lt;sup>16</sup> Federal Register Vol. 61, No. 124 [26 June 1996], pp. 33166–33172.

 $<sup>^{17}</sup>$  The OCC incorporated the joint policy statement into its "Interest Rate Risk" booklet of the  ${\it Comptroller's\,Handbook}.$ 

assets that arise from net operating loss and tax credit carryforwards, gains on sale in connection with a securitization, any defined benefit pension fund net asset held by entities that are not depository institutions (unless the banking organizations has unrestricted and unfettered access to the assets in that fund), investments in a banking organization's own capital instruments, mortgage servicing rights (above certain levels) and investments in the capital of unconsolidated financial institutions (above certain levels).

# **Prompt Corrective Action**

- 1.81 The FDICIA made capital an essential tool for regulators to monitor the financial health of insured banks and savings institutions. Regulatory intervention is now focused primarily on an institution's capital levels relative to regulatory standards. In Section 38, "Rules, Regulations, and Orders," of the FDI Act, the FDICIA added (to the existing capital adequacy guidelines set forth by each agency) a uniform framework for prompt corrective regulatory action. Holding companies are not subject to the PCA provisions.
- **1.82** Section 38 provides for supervisory action at certain institutions based on their capital levels. Each institution falls into one of five regulatory capital categories (see paragraph 1.85) based primarily on four capital measures, total risk-based capital; tier 1 based capital; common equity tier 1 capital; and leverage ratios. <sup>18</sup> These capital ratios are defined in the same manner for Section 38 purposes as under the respective agencies' capital adequacy guidelines and regulations. For savings associations, tier 1 leverage capital is comparable to core capital.
- 1.83 Regulations also specify a minimum requirement for tangible equity, which is defined as tier 1 capital plus outstanding perpetual preferred stock not included in tier 1 capital. In calculating the tangible capital ratio, the regulations specify specific deductions that should be applied to total assets included in the ratio denominator.
- **1.84** An institution may be reclassified between certain capital categories if its condition or an activity is deemed by regulators to be *unsafe or unsound*. A change in an institution's capital category initiates certain mandatory—and possibly additional discretionary—action by regulators.
  - 1.85 Under Section 38 of the FDI Act, an institution is considered
    - a. well capitalized if its capital level significantly exceeds the required minimum level for each relevant capital measure;
    - adequately capitalized if its capital levels meets the required minimum level for each relevant capital measure;
    - undercapitalized if its capital level fails to meet the required minimum level for each relevant capital measure;
    - d. significantly undercapitalized if its capital level is significantly below the required minimum level for each relevant capital measure; and
    - e. critically undercapitalized if its capital level fails to meet any level specified under subsection (c)(3)(A) of Section 38 of the FDI Act.

©2016, AICPA AAG-DEP 1.85

 $<sup>^{18}\,</sup>$  With respect to an advanced approaches national bank or advanced approaches federal savings association, on January 1, 2018, and thereafter, the leverage measure also includes the supplementary leverage ratio.

1.86 The PCA levels are defined as follows:

Category	Total Risk-Based Capital Ratio (Percent)	Tier 1 Risk-Based Capital Ratio (Percent)	Common Equity Tier 1 Risk-Based Capital Ratio (Percent)	Leverage Ratio* (Percent)
Well capitalized	≥10 and	≥8 and	>6.5 and	$\geq 5$
Adequately capitalized	≥8 and	≥6 and	>4.5 and	≥4
Undercapitalized	<8 or	<6 or	<4.5 or	<4
Significantly undercapitalized	<6 or	<4 or	<3 or	<3

- \* With respect to an advanced approaches national bank or advanced approaches federal savings association, on January 1, 2018, and thereafter, the leverage measure also includes capital adequacy guidelines for the supplementary leverage ratio in determination of both adequate capitalization and undercapitalization.
- **1.87** Critically undercapitalized institutions are those having a ratio of tangible equity to total assets of 2 percent or less.
- **1.88** An institution will not be considered well capitalized if it is under a capital-related cease-and-desist order, formal agreement, capital directive, or PCA capital directive.
- **1.89** Actions that may be taken under the PCA provisions range from the restriction or prohibition of certain activities to the appointment of a receiver or conservator of the institution's net assets.
- **1.90** Regulators will also require undercapitalized institutions to submit a plan for restoring the institution to an acceptable capital category. For example, each undercapitalized institution is generally required to submit a plan that specifies the following:
  - Steps the institution will take to become adequately capitalized
  - Targeted capital levels for each year of the plan
  - How the institution will comply with other restrictions or requirements put into effect
  - Types and levels of activities in which the institution will engage
- **1.91** Noncompliance or expected noncompliance with regulatory capital requirements may be a condition that, when considered with other factors, could indicate substantial doubt about an entity's ability to continue as a going concern. The implementation of the PCA provisions warrants similar attention by independent accountants when considering an institution's ability to remain a going concern.

# **Annual Independent Audits and Reporting Requirements**

- 1.92 The primary source of annual independent audits and reporting requirements is Section 36, Early Identification of Needed Improvements in Financial Management, of the FDI Act. In 1991, Section 112 of the FDICIA added Section 36 of the FDI Act. 12 CFR 363 (Part 363) of the FDIC's regulations implements Section 36 of the FDI Act. Part 363 was initially adopted by the FDIC's Board of Directors in 1993 and was most recently amended in 2013. Section 36 and Part 363 also establish minimum qualifications for auditors that provide audit and attest services to IDIs. Section 36 and Part 363 apply to each FDIC IDI having total assets of \$500 million or more at the beginning of its fiscal year. The requirements specified in Section 36 and Part 363 are in addition to any other statutory and regulatory requirements otherwise applicable to an IDI.
- **1.93** Notwithstanding the requirements of Section 36 of the FDI Act and Part 363, the Federal Reserve requires certain bank holding companies to submit audited financial statements (under authority of 12 CFR 225.5 [Regulation Y]).
- 1.94 Also, audit requirements for savings associations, state savings associations, and SLHCs are set forth in 12 CFR 162.4 (OCC), 12 CFR 238.5 (Federal Reserve), and 12 CFR 390.322 (FDIC). In general, the OCC, the Federal Reserve, and the FDIC may require an independent audit of any such entity that they supervise when needed for any identified safety and soundness reason. However, audits for safety and soundness are required as follows:
  - Savings associations supervised by the OCC, regardless of size, with a composite safety and soundness CAMELS rating of 3, 4, or 5
  - SLHCs supervised by the Federal Reserve, which control savings association subsidiary(ies) with aggregate consolidated assets of \$500 million or more
  - State savings associations supervised by the FDIC, regardless of size, with a composite safety and soundness CAMELS rating of 3, 4, or 5

12 CFR 162.4 (OCC), 12 CFR 238.5 (Federal Reserve), and 12 CFR 390.322 (FDIC) provide that these audits should be conducted by an independent public accountant who is in compliance with the AICPA Code of Professional Conduct and meets the independence requirements and interpretations of the SEC. 19

**1.95** Part 363, "Annual Independent Audits and Reporting Requirements," of the FDIC's rules and regulations, which implements Section 36 of the FDI Act, also includes guidelines and interpretations (guidelines) to facilitate a better understanding of, and full compliance with, the provisions of the Section 36. On July 20, 2009, a final rule which amended the regulation and guidelines in Part 363 was published in the *Federal Register* (Vol. 74, No. 137 [20 July 2009], pp. 35726–35761). The final rule applies to Part 363 Annual Reports with filing deadlines on or after the effective date of the amendments, which was August 6, 2009. The compliance date for the provision of the final

©2016, AICPA AAG-DEP 1.95

 $<sup>^{19}</sup>$  12 CFR 162.4 (OCC), 12 CFR 238.5 (Federal Reserve), and 12 CFR 390.322 (FDIC) have not been updated to include a reference to the independence requirements of the PCAOB or independent public accountants registered with the PCAOB.

rule that requires institutions' boards of directors to develop and adopt written criteria pertaining to audit committee member independence was delayed until December 31, 2009. The provision of the final rule that requires the consolidated total assets of a holding company's IDI subsidiaries to comprise 75 percent or more of the holding company's consolidated total assets in order for an institution to be eligible to comply with Part 363 at the holding company level became effective for fiscal years ending on or after June 15, 2010.

- 1.96 Part 363 applies to any IDI with total assets above certain thresholds and requires annual independent audits, assessments of the effectiveness of internal control over financial reporting, and compliance with laws and regulations pertaining to insider loans and dividend restrictions, the establishment of independent audit committees, and related reporting requirements. The asset size threshold for reporting on an institution's internal control is \$1 billion and the threshold for the other requirements generally is \$500 million. The FDIC's FIL-33-2009, Annual Audit and Reporting Requirements: Final Amendments to Part 363, issued on June 23, 2009, provides a summary of the final rule and highlights certain amended annual and other reporting requirements. The general requirements, as amended, are summarized in the following text.
- **1.97** Annual reporting requirements. According to Sections 363.2 and 363.4, management is required to prepare and file a Part 363 Annual Report that includes the following:<sup>20</sup>
  - a. Comparative financial statements in accordance with GAAP, which should be audited by an independent public accountant.
  - b. A management report that must contain the following:
    - i. A statement of management's responsibilities for preparing the institution's annual financial statements, for establishing and maintaining an adequate internal control structure and procedures for financial reporting, and for complying with laws and regulations relating to safety and soundness pertaining to insider loans and dividend restrictions, which are designated by the FDIC and the appropriate federal banking agency.
    - ii. An assessment by management of the institution's compliance with the designated laws and regulations pertaining to insider loans and dividend restrictions during such fiscal year. The assessment must state management's conclusion regarding compliance and disclose any noncompliance with these laws and regulations. The assessment must clearly state whether the institution has or has not complied with these regulations. Disclosure is not dependent on the degree or materiality of any noncompliance. Statements such as "management believes that the institution complied, in all material respects with the designated safety and soundness laws and regulations" do not present a definitive and unconditional conclusion regarding compliance as envisioned under Part 363.

<sup>&</sup>lt;sup>20</sup> The reporting requirements may be satisfied for certain subsidiaries through reporting by their holding companies. These exemptions are discussed in Section 363.1(b) of the rule.

- iii. For an institution with consolidated total assets of \$1 billion or more at the beginning of its fiscal year, an assessment by management of the effectiveness of such internal control structure and procedures as of the end of such fiscal year. (See paragraphs 1.110–.111 for additional information regarding the internal control reporting requirements.)
- c. The management report must be signed by the CEO and the chief accounting officer or the CFO at the insured depository level or the holding company level as specified in Section 363.2(c).
- **1.98** *Independent public accountant.* As amended, Section 363.3 clarifies the independence standards applicable to accountants and requires the following:
  - a. Each IDI should engage an independent public accountant to audit and report on its annual financial statements in accordance with generally accepted auditing standards or the PCAOB's auditing standards, if applicable, and Section 37 of the FDI Act.
  - b. For each IDI with total assets of \$1 billion or more at the beginning of the institution's fiscal year, the independent public accountant who audits the institution's financial statements should examine, attest to, and report separately on the assertion of management concerning the effectiveness of the institution's internal control structure and procedures for financial reporting. The attestation and report should be made in accordance with attestation standards established by the AICPA or the PCAOB's auditing standards, if applicable. The accountant's report must not be dated prior to the date of the management report and management's assessment of the effectiveness of internal control over financial reporting.
  - c. When the independent public accountant performing services under Part 363 ceases to be the institution's accountant, the accountant must provide the FDIC, the appropriate Federal banking agency, and any appropriate State bank supervisor with written notification of such termination within 15 days after the occurrence of such an event. Guideline 20 to Part 363 provides additional guidance regarding an independent public accountant's notice of termination.
  - d. The auditors must report certain communications on a timely basis to the audit committee. The requirements for communications with audit committees, consistent with the requirements under Section 363.3(d), are set forth in the applicable professional standards. The applicable AICPA professional standards, which include AU-C section 260, The Auditor's Communication With Those Charged With Governance; AU-C section 240, Consideration of Fraud in a Financial Statement Audit; AU-C section 265, Communicating Internal Control Related Matters Identified in an Audit; and AT section 501, An Examination of an Entity's Internal Control Over Financial Reporting That Is Integrated With an Audit of Its Financial Statements (AICPA, Professional Standards), provide guidance regarding certain matters required to be communicated to those charged with governance, such as audit committees. PCAOB AS 1301,

Communications with Audit Committees, and AS 2201, An Audit of Internal Control Over Financial Reporting That Is Integrated with An Audit of Financial Statements (AICPA, PCAOB Standards and Related Rules), address the requirements for communication of certain matters to audit committees for audits of public companies.

The auditors must report certain communications on a timely basis to the audit committee. The requirements for communications with audit committees, consistent with the requirements under Section 363.3(d), are set forth in the applicable professional standards. The applicable AICPA professional standards, which include AU-C section 260, The Auditor's Communication With Those Charged With Governance; AU-C section 240, Consideration of Fraud in a Financial Statement Audit; AU-C section 265, Communicating Internal Control Related Matters Identified in an Audit; and AU-C section 940, An Audit of Internal Control Over Financial Reporting That Is Integrated With an Audit of Its Financial Statements (AICPA, Professional Standards), provide guidance regarding certain matters required to be communicated to those charged with governance, such as audit committees. PCAOB AS 1301, Communications with Audit Committees, and AS 2201, An Audit of Internal Control Over Financial Reporting That Is Integrated with An Audit of Financial Statements (AICPA, PCAOB Standards and Related Rules), address the requirements for communication of certain matters to audit committees for audits of public companies.<sup>21</sup>

- e. The auditors must retain the working papers related to the audit of the IDI's financial statements and, if applicable, the evaluation of the institution's internal control over financial reporting for seven years from the report release date, unless a longer period of time is required by law.
- f. The auditors must comply with the independence standards and interpretations of the AICPA, the SEC, and the PCAOB. To the extent that any of the rules within any one of these independence standards (AICPA, SEC, and PCAOB) is more or less restrictive than the corresponding rule in the other independence standards, auditors must comply with the more restrictive rule.
- g. Prior to commencing any services for an IDI under Part 363, the independent public accountant must have received a peer review, or be enrolled in a peer review program, that meets acceptable guidelines. Acceptable peer reviews include peer reviews

<sup>&</sup>lt;sup>21</sup> Statement on Auditing Standards (SAS) No. 130, An Audit of Internal Control Over Financial Reporting That Is Integrated With an Audit of Financial Statements (AICPA, Professional Standards, AU-C sec. 940), was issued in November 2015. The SAS is effective for integrated audits for periods ending on or after December 15, 2016.

The Auditing Standards Board concluded that, because engagements performed under AT section 501, An Examination of an Entity's Internal Control Over Financial Reporting That Is Integrated With an Audit of Its Financial Statements (AICPA, Professional Standards), as well as related attestation Interpretation No. 1, "Reporting Under Section 112 of the Federal Deposit Insurance Corporation Improvement Act" (AICPA, Professional Standards, AT sec. 9501 par. .01–.07), are required to be integrated with an audit of financial statements, it is appropriate to move the content of AT section 501 from the attestation standards into generally accepted auditing standards.

Readers are encouraged to consult the full text of this SAS on the AICPA's website at www.aicpa.org.

in accordance with the AICPA's Peer Review standards and inspections conducted by the PCAOB. For auditors required to conduct their audits in accordance with PCAOB standards, registration with the PCAOB is mandatory. Within 15 days of receiving notification that a peer review has been accepted or a PCAOB inspection report has been issued, or before commencing any audit under this part, whichever is earlier, the independent public accountant must file two copies of the most recent peer review report and the public portion of the most recent PCAOB inspection report, if any, accompanied by any letters of comments, response, and acceptance, with the FDIC. Also, within 15 days of the PCAOB making public a previously nonpublic portion of an inspection report, the independent public accountant must file two copies of the previously nonpublic portion of the inspection report with the FDIC.

- **1.99** Filing and notice requirements. As amended, Section 363.4 extends the annual report filing deadline for nonpublic institutions and includes the following requirements:
  - a. A Part 363 Annual Report must contain the following:
    - i. Audited comparative annual financial statements
    - ii. The independent public accountant's report thereon
    - iii. A management report (see appendix B to Part 363 for illustrative management reports)
    - iv. For an institution with consolidated total assets of \$1 billion or more at the beginning of its fiscal year, an assessment by management of the effectiveness of such internal control structure and procedures as of the end of such fiscal year
    - v. If applicable, the independent public accountant's attestation report on management's assessment concerning the institution's internal control structure and procedures for financial reporting

Generally, the filing deadline for a Part 363 Annual Report is 120 days after the end of the fiscal year for an institution that is neither a public company nor a subsidiary of a public company, and 90 days after the end of the fiscal year for an institution that is a public company or a subsidiary of public company.

- b. Except for the Part 363 Annual Report and the peer reviews and inspection reports, as previously described, which should be available for public inspection, all other reports and notifications required under Part 363 are exempt from public disclosure by the FDIC.
- c. Institutions must file with the FDIC a copy of any management letter or other report issued by its independent public accountant with respect to such institution and the services provided by such accountant pursuant to Part 363 within 15 days after receipt. (See Section 363.4(c) for examples of such reports.)
- **1.100** Audit committees. Section 363.5 and Guidelines 27 to 35 to Part 363 provide guidance, address the composition requirements for audit committees, specify the audit committee's duties regarding the independent public accountant, require audit committees to ensure that audit engagement letters

do not contain unsafe and unsound limitation of liability provisions, and require boards of directors to develop and apply written criteria for evaluating audit committee members' independence.

- 1.101 General qualifications. Section 36(g)(3)(A) of the FDI Act provides that all audit services required by Section 36 should be performed by an independent public accountant who has agreed to provide regulators with access to audit documentation related to such services, if requested; and has received a peer review that meets guidelines acceptable to the FDIC. Guideline 13 to Part 363 also requires accountants to agree to provide copies of audit documentation to regulators. Interpretation No. 1, "Providing Access to or Copies of Audit Documentation (AICPA, Professional Standards, AU-C section 230, Audit Documentation (AICPA, Professional Standards, AU-C sec. 9230 par. .01–.15), and AU-C section 230 provide additional information to auditors.
- **1.102** Enforcement actions against auditors. In August 2003, the FDIC, the OCC, the Federal Reserve, and the OTS jointly issued final rules that establish procedures under which the agencies can remove, suspend, or bar an accountant or firm from performing audit and attestation services for IDIs subject to the annual audit and reporting requirements of Section 36 of the FDI Act. The final rule can be accessed at www.fdic.gov/news/news/financial/2003/fil0366.html.
- 1.103 Under the final rules, certain violations of law, negligent conduct, reckless violations of professional standards, or lack of qualifications to perform auditing services may be considered good cause to remove, suspend, or bar an accountant or firm from providing audit and attestation services for institutions subject to Section 36 of the FDI Act and Part 363. In addition, the rules prohibit an accountant or accounting firm from performing these services if the accountant or firm has been removed, suspended, or debarred by one of the agencies, or if the SEC or the PCAOB takes certain disciplinary actions against the accountant or firm. The rules also permit immediate suspensions of accountants and firms in limited circumstances.
- **1.104** Communication with independent auditors. Section 36(h) of the FDI Act and Guideline 17 to Part 363 require an institution to provide its auditor with certain information including copies of the institution's most recent reports of condition and examination; any supervisory memorandum of understanding or written agreement with any federal or state regulatory agency; and a report of any action initiated or taken by Federal or State banking regulators.

# Additional Regulatory Requirements Concerning the Sarbanes-Oxley Act, Corporate Governance, and Services Outsourced to External Auditors

- 1.105 In connection with the Sarbanes-Oxley Act of 2002, the SEC issued regulations implementing sections of the act, addressing various areas such as certification of financial statements, auditor independence, non-U.S. GAAP financial measures, accounting firms' record retention, audit committees, influencing auditors, and other matters. These regulations are not unique to financial institutions. Management, the board of directors, the audit committee, and auditors generally should be aware of the requirements of the Sarbanes-Oxley Act and the implementing SEC regulations.
- 1.106 In addition to the previously mentioned regulations, in June 2003, the SEC adopted rules requiring companies subject to the reporting requirements of the 1934 Act, other than registered investment companies, to assess

the effectiveness of their internal control and include in their annual reports a report of management on the company's internal control over financial reporting. The rule also mandates quarterly reports on changes in internal control. See paragraphs 1.108–.111 and 1.117 for additional information regarding these rules.

**1.107** The banking regulatory agencies also implemented regulations in connection with the Sarbanes-Oxley Act. These regulations can affect nonpublic as well as public entities. These regulations include the following:

- On March 17, 2003, the FDIC, the OTS, the OCC, and the Federal Reserve issued *Interagency Policy Statement on the Internal Audit Function and Its Outsourcing*. <sup>22</sup> This policy statement reflects the passage of the Sarbanes-Oxley Act and prohibits an external auditor from providing internal audit services during the same period for which the external auditor expresses an opinion on the financial statements. This prohibition applies to banks, savings associations, and their holding companies that
  - have a class of securities registered with either the SEC or the respective savings association agency under Section 12 of the 1934 Act or are required to file reports with the SEC under Section 15(d) of that act (commonly referred to as *public companies*) and, therefore, required to have an external audit.
  - are savings associations and banks with assets of \$500 million or more that are subject to the FDIC's external audit and reporting requirements under Part 363.
  - are savings associations and savings association holding companies that are required to have an external audit by their respective primary federal regulator pursuant to 12 CFR 162 (OCC), 12 CFR 238.5 (Federal Reserve), or Subpart R to 12 CFR 390 (FDIC).

For all other banks, savings associations, and their holding companies that have external audits of their financial statements but are not mandated to do so, the policy encourages such organizations to follow the internal audit outsourcing prohibition in Section 201 of the Sarbanes-Oxley Act when the SEC's regulations implementing this prohibition take effect.

On March 5, 2003, the FDIC issued FIL-17-2003, Corporate Governance, Audits, and Reporting Requirements, and the Federal Reserve, the OCC, and the OTS, in May 2003, issued Statement on Application of Recent Corporate Governance Initiatives to Non-Public Banking Organizations. This letter and statement require or recommend that certain nonpublic financial institutions comply with certain sections of the Sarbanes-Oxley Act. Familiarity with this guidance is recommended for external auditors.

©2016, AICPA AAG-DEP 1.107

<sup>&</sup>lt;sup>22</sup> In January 2013, the Federal Reserve issued SR letter 13-1, Supplemental Policy Statement on the Internal Audit Function and Its Outsourcing, to provide institutions with additional guidance related to interagency guidance that was issued in 2003. Building upon the 2003 interagency guidance, the supplemental guidance addresses characteristics, governance, and operational effectiveness of an institution's internal audit function. Further, this supplemental guidance explains changes over the past several years in banking regulations related to auditor independence and limitations placed on the external auditor.

- On August 12, 2003, the FDIC, the OCC, the Federal Reserve, and the OTS jointly issued final rules that establish procedures under which the agencies could remove, suspend, or bar an accountant or firm from performing audit and attestation services for IDIs subject to the annual audit and reporting requirements of Section 36. Section 36 applies to institutions with \$500 million or more in total assets.
- Effective April 1, 2003, the Federal Reserve adopted a final rule to reflect the amendments made to Section 12(i) of the 1934 Act. These amendments vest the Federal Reserve with the authority to administer and enforce several of the enhanced reporting, disclosure, and corporate governance obligations imposed by the Sarbanes-Oxley Act in respect to state member banks that have a class of securities registered under the 1934 Act.
- On June 30, 2005, the FFIEC issued the BSA/AML manual. The
  manual was the result of a collaborative effort of the federal banking agencies and the Treasury's FinCEN. The manual does not set
  new standards; instead, it is a compilation of existing regulatory
  requirements, supervisory expectations, and sound practices in
  the BSA/AML area.
- On November 28, 2005, the FDIC amended Part 363 of its regulations by raising the asset-size threshold from \$500 million to \$1 billion for internal control assessments by management and external auditors. For institutions between \$500 million and \$1 billion in assets, the audit committee of its board of directors should be outside directors, the majority of whom should be independent of management of the institution.
- In June 2009, as previously noted, the FDIC's Board of Directors approved amendments to Part 363 of its regulations. Among other requirements, the amendments require both management's assessment and the auditor's report on internal control over financial reporting to disclose the internal control framework used by management and the auditor and to identify all material weaknesses that have been identified that have not been remediated as of the end of the institution's fiscal year. See the following for additional information.

1.108 Sarbanes-Oxley Act Section 404 and Part 363. Public companies that are subject to Section 36 of the FDI Act and Part 363 (more than \$500 million in assets) must prepare reports for the SEC, the FDIC, and other regulators that are similar in nature. Section 404(a) of the Sarbanes-Oxley Act mandates that registrants (a) take responsibility for establishing and maintaining adequate internal control structure and procedures and (b) assess their effectiveness at the end of each fiscal year. According to the SEC's final rule Management's Report on Internal Control Over Financial Reporting and Certification of Disclosure in Exchange Act Periodic Reports, management generally must create a Management's Annual Internal Control Report as part of the Annual Report. (Quarterly updating is necessary only if the internal control environment has changed or is likely to change materially.) The report must contain the following:

- A statement of management's responsibility for establishing and maintaining adequate internal control over financial reporting for the company.
- A statement identifying the framework used by management to evaluate the effectiveness of this internal control.
- Management's assessment of the effectiveness of internal control as of the end of the company's most recent fiscal year, including a statement about whether internal control over financial reporting is effective.
- Disclosure of any material weaknesses. Management is not permitted to conclude that the registrant's internal control over financial reporting is effective if there are one or more material weaknesses in the issuer's internal control over financial reporting.
- A statement that its auditor has issued an attestation report on management's assessment, which is normally included in the company's annual report.

1.109 The SEC coordinated with the FDIC to eliminate any unnecessary duplication between the aforementioned requirements and Section 36 of the FDI Act and Part 363. Many internal control requirements of the Sarbanes-Oxley Act were structured after Section 36 of the FDI Act and Part 363. A comparison of Sarbanes-Oxley and the Part 363 management requirements are indicated in the following table for clarity.

Sarbanes-Oxley	Federal Deposit Insurance Corporation Improvement Act of 1991	
A statement of management's responsibility for establishing and maintaining adequate internal control over financial reporting for the company	Insured depository institutions (IDIs) with at least \$500 million in total assets, a statement of management's responsibility for establishing and maintaining an adequate internal control structure and procedures for financial reporting (Financial reporting generally must encompass both financial statements prepared in accordance with U.S. generally accepted accounting principles and those prepared for regulatory purposes.)	
Not required by Sarbanes-Oxley	IDIs with at least \$500 million in total assets, a statement of management's responsibility for preparing the institution's financial statements	
Not required by Sarbanes-Oxley	IDIs with at least \$500 million in total assets, a statement of management's responsibility for complying with designated laws and regulations relating to safety and soundness pertaining to insider loans and dividend restrictions	

(continued)

Sarbanes-Oxley	Federal Deposit Insurance Corporation Improvement Act of 1991
A statement identifying the framework used by management to evaluate the effectiveness of internal control over financial reporting	IDIs with \$1 billion or more in total assets, a statement identifying the internal control framework used by management to evaluate the effectiveness of internal control over financial reporting
Management's assessment of the effectiveness of internal control over financial reporting as of the end of the company's most recent fiscal year	IDIs with \$1 billion or more in total assets, a statement expressing management's conclusion concerning whether internal control over financial reporting is effective as of the end of its fiscal year
Disclosure of any material weakness (and the related stipulation that management is not permitted to conclude that the company's internal control over financial reporting is effective if there are one or more material weaknesses)	For IDIs with \$1 billion or more in total assets, management must disclose all material weaknesses in internal control over financial reporting, if any, that it has identified that have not been remediated prior to the IDI's fiscal year-end.  Management is precluded from concluding that the institution's internal control over financial reporting is effective if there are one or more material weaknesses
A statement that a registered public accounting firm has issued an attestation report on the effectiveness of internal control over financial reporting	Not required by Part 363
Inclusion of the registered public accounting firm's attestation report on the effectiveness of internal control over financial reporting in the annual report	For IDIs with \$1 billion or more in total assets, the management report component of the annual report must include the independent public accountant's attestation report concerning the effectiveness of the institution's internal control structure over financial reporting

1.110 IDIs with \$1 billion or more in total assets as of the beginning of its fiscal year that are subject to both Part 363 and the SEC's rules implementing Section 404 of Sarbanes-Oxley Act (as well as holding companies permitted to file an internal control report on behalf of their IDI subsidiaries in satisfaction of the FDIC and SEC regulations) can choose to either prepare two separate management reports to satisfy the FDIC's and Sarbanes-Oxley Act Section 404 requirements or prepare a single management report that satisfies both the FDIC and Sarbanes-Oxley Act Section 404 requirements.

- 1.111 If a single report is prepared it must contain the following combined requirements of the preceding chart:
  - A statement of management's responsibility for preparing the registrant's annual financial statements, for establishing and maintaining adequate internal control over financial reporting for the registrant, and for the institution's compliance with laws and regulations relating to safety and soundness designated by the FDIC and the appropriate federal banking agencies.
  - A statement identifying the framework used by management to evaluate the effectiveness of the registrant's internal control over financial reporting as required by the 1934 Act Rule 13a-15 or 15d-15.
  - Management's assessment of the effectiveness of the registrant's internal control over financial reporting as of the end of the registrant's most recent fiscal year, including a statement regarding whether or not management has concluded that the registrant's internal control over financial reporting is effective, and of the institution's compliance with the designated safety and soundness laws and regulations pertaining to insider loans and dividend restrictions during the fiscal year. This discussion must include disclosure of any material weakness in the registrant's internal control over financial reporting identified by management and disclosure of any instances of noncompliance with the designated safety and soundness laws and regulations pertaining to insider loans and dividend restrictions.
  - A statement that the registered public accounting firm that audited the financial statements included in the registrant's annual report, has issued an attestation report on the effectiveness of the registrant's internal control over financial reporting.

Finally, it is important to note that the institution or holding company will have to provide the registered public accounting firm's attestation report on management's assessment in its annual report filed under the 1934 Act. For purposes of the report of management and the attestation report, financial reporting generally must encompass both financial statements prepared in accordance with GAAP and those prepared for regulatory reporting purposes.

- **1.112** Section 404(b) of the Sarbanes-Oxley Act and Part 363 require the external auditor to attest to, and publicly report on the effectiveness of the company's internal control and procedures for financial reporting. Section 404(b) states, that any such attestation should not be the subject of a separate engagement. Auditors are expected to expand their scope in relation to internal control.
- 1.113 In September 2010, the SEC issued Final Rule Release No. 33-9142, Internal Control Over Financial Reporting in Exchange Act Periodic Reports of Non-Accelerated Filers, to conform its rules to Section 404(c) of the Sarbanes-Oxley Act, as added by Section 989G of the Dodd-Frank Act. Section 404(c) provides that Section 404(b) of the Sarbanes-Oxley Act should not apply with respect to any audit report prepared for an issuer that is neither an accelerated filer nor a large accelerated filer as defined in Rule 12b-2 under the 1934 Act. Prior to enactment of the Dodd-Frank Act, a nonaccelerated filer would have

©2016, AICPA AAG-DEP 1.113

been required, under existing SEC rules, to include an attestation report of its registered public accounting firm on internal control over financial reporting in the filer's annual report filed with the SEC for fiscal years ending on or after June 15, 2010. For further information on conforming changes adopted as a result of this ruling, Final Rule Release No. 33-9142 can be accessed on the SEC website at www.sec.gov. Notwithstanding the SEC's final rule, IDIs subject to Part 363 of the FDIC's rules and regulations must continue to comply with the requirements of Section 363.3(b) regarding the independent public accountant's attestation report on management's assessment of the effectiveness of internal control over financial reporting.

- **1.114** For an institution that is a public company or a subsidiary of a public company that is required to comply with the auditor attestation requirement of Section 404 of the Sarbanes-Oxley Act, the auditor's report would be prepared in accordance with AS 2201.
- **1.115** Generally, for an institution that is not a public company or a subsidiary of a public company, the auditor's report would be prepared in accordance with AT section 501.

Generally, for an institution that is not a public company or a subsidiary of a public company, the auditor's report would be prepared in accordance with AU-C section  $940.^{23}$ 

- **1.116** Guideline 18A of Part 363 of the FDIC's regulations provides additional guidance regarding the standards that auditors should follow when reporting on internal control.
- 1.117 Section 404 of the Sarbanes-Oxley Act does not specify where the management report might appear. However, SEC Final Rule Release No. 33-8238, Management's Reports on Internal Control Over Financial Reporting and Certification of Disclosure in Exchange Act Periodic Reports, explains that it is important for management's report to be in close proximity to the corresponding attestation report issued by the company's registered public accounting firm. Positioning the report near the company's Management's Discussion and Analysis disclosure or immediately preceding the company's financial statements would be two appropriate locations.

# **Other Reporting Considerations**

- **1.118** Banks and savings institutions often engage auditors to perform assurance services other than those required by Section 36 of the FDI Act. Such engagements may relate to the following:
  - a. Student loans. Lenders participating in the Federal Family Education Loan Program may be required to engage an auditor to examine and report on management's assertions regarding compliance with certain U.S. Department of Education requirements. This examination is performed in accordance with (i) Government Auditing Standards (also known as the Yellow Book) issued by the Comptroller General of the United States, (ii) AT section 601, Compliance Attestation (AICPA, Professional Standards), and (iii) the Audit Guide Compliance Audits (Attestation Engagements) for

<sup>&</sup>lt;sup>23</sup> See footnote 21.

- Lenders and Lender Servicers Participating in the Federal Family Education Loan Program issued by the U.S. Department of Education. This examination requirement applies to lenders with origination levels exceeding a specified dollar amount.<sup>24</sup>
- b. Federal Home Loan Mortgage Corporation (Freddie Mac) borrowings. Banks or savings institutions that are members of the Freddie Mac system may borrow from their respective district Federal Home Loan Bank. Borrowings are generally secured by the pledging of assets, often in the form of a blanket lien. The district banks maintain separate and distinct credit policies that have varying requirements concerning a member bank's engagement of auditors to render assurance services relating to the adequacy of collateral maintenance levels. It is incumbent on the auditor to ascertain the professional standards that may be applicable to the requested services. The engagement generally takes the form of (i) an agreedupon procedures engagement performed in accordance with AT section 201, Agreed-Upon Procedures Engagements (AICPA, Professional Standards), or (ii) an audit engagement performed in accordance with AU-C section 806, Reporting on Compliance With Aspects of Contractual Agreements or Regulatory Requirements in Connection With Audited Financial Statements (AICPA, Professional Standards).
- c. Loan servicing. Lenders who service mortgage loans for others may be required to engage an auditor to examine management's assertions about compliance with minimum servicing standards set forth in the Uniform Single Attestation Program for Mortgage Bankers (USAP). Companies that are issuers or servicers, or both, of publicly registered commercial-mortgage backed securities and private label residential-mortgage backed securities must also submit reports prepared in accordance with Item 1122; and compliance with applicable servicing criteria, of Regulation AB, Asset-Backed Securities, published by the SEC in 2004.<sup>25</sup> The Item 1122 engagement largely encompasses and expands upon the USAP engagement. Both the USAP and Regulation AB are attestation engagements performed in accordance with AT section 601 as further described in paragraphs 4.36–.37 of this guide.
- d. HUD programs. To the extent that a bank or savings institution originates or services HUD loans through a subsidiary that is designated a nonsupervised mortgagee, or a supervised mortgagee, compliance with the Consolidated Audit Guide for Audits of HUD Programs is required, as further described in paragraphs 4.33–.35 of this guide.
- e. FDIC Loss Sharing Purchase and Assumption Transactions. The FDIC's Resolutions Handbook states that a loss sharing transaction is a purchase and assumption (P&A) transaction that the FDIC

<sup>&</sup>lt;sup>24</sup> Readers are encouraged to visit the National Council of Higher Education Loan Program's website for the most recent audit guide and related amendments.

<sup>&</sup>lt;sup>25</sup> In September 2014, the SEC published Regulation AB II, which incorporates significant revisions to Regulation AB and other rules governing the offering process, disclosure, and reporting for asset-backed securities. Regulation AB II became effective November 24, 2014. A one-year transition period was adopted by the SEC for all the new rules under Regulation AB II with the exception of those rules related to asset-level disclosure that fall under a two-year transition period.

commonly uses as a resolution tool for handling failed institutions with more than \$500 million in assets. A P&A is a resolution transaction in which a healthy institution purchases some or all of the assets of a failed bank or thrift and assumes some or all of the liabilities, including all insured deposits. The Resolutions Handbook also states that a loss sharing P&A uses the basic P&A structure, except for the provision regarding transferred assets. Instead of selling some or all of the assets to the acquirer at a discounted price, the FDIC agrees to share in future loss experienced by the acquirer on a fixed pool of assets (covered assets). The *Resolutions* Handbook for P&A agreements requires that "[w]ithin 90 days after each calendar year end, the acquiring bank must furnish the FDIC a report signed by its independent public accountant containing specified statements<sup>26</sup> relative to the accuracy of any computations made regarding shared loss assets. AICPA Technical Questions and Answers (Q&A) section 9110.16, "Example Reports on Federal Deposit Insurance Corporation Loss Sharing Purchase and Assumption Transactions" (AICPA, Technical Questions and Answers), provides examples of how the auditor might respond.

### © Update 1-2 Attestation: Clarification and Recodification

Statement on Standards for Attestation Engagements (SSAE) No. 18, Attestation Standards: Clarification and Recodification (AICPA, Professional Standards), issued in April 2016, is effective for practitioners' reports dated on or after May 1, 2017.

To address concerns over the clarity, length, and complexity of its standards, the Auditing Standards Board established clarity drafting conventions and undertook a project to redraft all the standards it issues in clarity format. The redrafting of SSAEs (or attestation standards) in SSAE No. 18 represents the culmination of that process. The attestation standards are developed and issued in the form of SSAEs and are codified into sections. SSAE No. 18 recodifies the "AT" section numbers designated by SSAE Nos. 10–17 using the identifier "AT-C" to differentiate the sections of the clarified attestation standards (AT-C sections) from the attestation standards that are superseded by SSAE No. 18 (AT sections). The AT sections in AICPA *Professional Standards* remain effective through April 2017, by which time substantially all engagements for which the AT sections were still effective are expected to be completed.

This edition of the guide has not been updated to reflect changes as a result of this SSAE, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this SSAE on the AICPA's website at www.aicpa.org.

<sup>&</sup>lt;sup>26</sup> The term *specified statements* is not defined in the FDIC's *Resolutions Handbook*. The practitioner is advised to read the terms of the loss share agreement and confirm that the audit requirement in that agreement provides for the receipt of a report expressing negative assurance.

# Chapter 2

# Industry Overview—Credit Unions

# **Description of Business**

**2.01** The first credit union in the United States was organized in 1908. Although credit unions originally arose within communities, greater success was achieved by organizing credit unions to serve employee groups—particularly government employees, teachers, railway workers, and telephone company employees. A credit union is a member-owned financial cooperative, democratically controlled by its members and operating for the purpose of providing financial services to its members. Credit unions are organized as not-for-profit entities and are exempt from certain taxes. A credit union is subject to rules that restrict membership to the entity.

**2.02** In 1934, Congress passed the Federal Credit Union Act (FCUA), establishing a federal regulatory system, which authorized the formation of federally chartered credit unions in all states. In 1970, the National Credit Union Administration (NCUA), an independent governmental agency, was created by Congress to charter, supervise, and regulate federal credit unions. It also extended supervision to state chartered federally insured credit unions. Other legislative initiatives that have affected credit unions include the following:

- In 1970, the National Credit Union Share Insurance Fund (NCUSIF) was formed to insure credit union share (deposit) accounts up to applicable limits in all federal credit unions and federally insured state-chartered credit unions.
- In 1977, legislation was passed expanding services available to credit union members, including share certificates and mortgage lending.
- The Depository Institution Deregulation and Monetary Control Act of 1980.
- The Financial Institutions Reform, Recovery, and Enforcement Act of 1989.
- The Credit Union Membership Access Act of 1998 (CUMAA).
- The Gramm-Leach-Bliley Act of 1999 (also known as the Financial Services Modernization Act).
- The Financial Services Regulatory Relief Act of 2006.
- The Emergency Economic Stabilization Act of 2008.
- The Helping Families Save Their Homes Act of 2009.
- The Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010.
- The Consumer Financial Protection Bureau (CFPB) regulations.

**2.03** Each credit union is organized around a defined *field of membership*, and each member shares a *common bond* of affiliation with other members. The field of membership is a key characteristic of a credit union and is defined, in its charter or bylaws, as those who may belong to it and use its services. All credit unions, except corporate credit unions, are referred to as *natural person credit unions*. The common bond is a characteristic of the members themselves.

©2016, AICPA AAG-DEP 2.03

Congress, in the FCUA, has recognized three types of membership fields: single common bond, multiple common bond, and community. Single common bond credit unions consist of one group that has a common bond of occupation or association. Multiple common bond credit unions include multiple groups, each of which has a common bond of occupation or association. Community membership fields are defined by the FCUA as persons or organizations within a well-defined local community, neighborhood, or rural district.

- **2.04** In early 1998, the United States Supreme Court ruled that NCUA had strayed from the original intent of Congress as reflected in the FCUA passed in 1934 relating to common bond affiliation for credit union membership. This ruling had the effect of restricting future membership in federal credit unions. On August 7, 1998, legislation was signed into law that eased membership restrictions on credit unions and allowed them to expand. The legislation, known as the CUMAA, permits occupation-based credit unions to take in groups of members from unrelated companies under certain circumstances.
- **2.05** The CUMAA also establishes three important new requirements with respect to financial statements and audits. First, all federally insured credit unions with assets of \$500 million or more must obtain an annual independent audit of their financial statements by a CPA or public accountant licensed by the appropriate state or jurisdiction. Second, all federally insured credit unions with assets of \$10 million or more must follow U.S. generally accepted accounting principles (GAAP) for all reports or statements required to be filed with the NCUA board. Third, for any federal credit union with assets of more than \$10 million that uses an independent auditor who is compensated for his or her services to perform a financial statement audit, the audit is subject to state accounting laws, including licensing requirements.
- **2.06** The CUMAA further addressed minimum capital (net worth) requirements and prompt corrective action (PCA) to restore capital. Net worth standards based on a percentage of assets were established for federally insured credit unions, as well as risk-based net worth standards for complex credit unions as defined by the NCUA. The NCUA also developed PCA regulations, as well as regulations concerning other areas such as new field of membership rules, and supervisory committee audit rules as required by this legislation. In addition, the CUMAA placed restrictions on member business lending and restricted conversions to mutual savings banks.

### The Board of Directors

**2.07** The board of directors establishes the general operation of a credit union and ensures that it follows applicable laws and regulations and adheres to its bylaws. In addition, the board is responsible for ensuring that a credit union maintains its financial stability, follows good business practices, and is properly insured and bonded. As membership organizations, credit unions are democratically controlled. Federal and state laws require that a board of directors be elected by the membership on the basis of one member, one vote. The board of directors, in turn, appoints the supervisory committee; however, based on state law, not all state chartered credit unions require a supervisory committee. The supervisory committee, which is similar to an audit committee, plays a major role in monitoring a credit union's financial affairs. A credit committee may be appointed or elected to oversee the lending transactions. Other committees may include an asset-liability committee, a marketing or member-relations committee, an educational committee, and various ad hoc

committees. Credit unions depend heavily on member volunteers to set policy, make decisions, and sometimes even to operate them. Some officials (board members and board appointed persons) of state chartered credit unions may receive compensation for services, as allowed by law. However, federally chartered credit unions, except as expressly stated in Part 701.33(b) of the NCUA regulations, are generally prohibited from compensating officials.

# The Supervisory Committee

**2.08** The supervisory or audit committee is responsible for ensuring that member funds are protected, financial records and operations are in order, and elected officials carry out their duties properly. Supervisory committee responsibilities are prescribed in Part 715 of the NCUA regulations. In addition, the supervisory committee is generally responsible for overseeing the financial reporting process and ensuring that management has established effective internal controls. Section 115 of the FCUA (Banks and Banking, U.S. Code 12, Section 1761d) states

The supervisory committee shall make or cause to be made an annual audit and shall submit a report of that audit to the board of directors and a summary of the report to the members at the next annual meeting of the credit union; shall make or cause to be made such supplemental audits as it deems necessary or as may be ordered by the board, and submit reports of the supplementary audits to the board of directors.

Similar requirements exist for most state-chartered credit unions, depending on state laws. The supervisory committee may engage an independent auditor to audit and report on the credit union's financial statements.

**2.09** Supervisory committees play an important role in developing and maintaining strong operational and financial management at credit unions. As credit unions continue to broaden the nature and scope of the activities in which they are involved, it is important that supervisory committees meet regularly to carefully review operational and financial goals, internal control, financial statements, and examiners' and auditors' reports. Lack of supervisory committee involvement in credit union operations may be an early indicator of potential problems for a credit union.

### The Credit Committee

**2.10** The credit committee (composed of volunteers either appointed or elected by the board of directors) establishes and monitors a credit union's lending policies, approves loan applications, and provides credit-counseling services to members. This committee may delegate some of its loan-granting authority to one or more loan officers employed by the credit union in accordance with the bylaws. Many credit unions have amended their bylaws to eliminate the elected credit committee. In these instances, the board of directors assumes credit committee responsibilities and generally delegates its responsibility to loan officers employed by the credit union.

# Charters, Bylaws, and Minutes

**2.11** The NCUA issues charters for federally chartered credit unions and prescribes the form of bylaws of such credit unions. State regulatory authorities establish the form of the charter and bylaws for state-chartered credit unions.

©2016, AICPA AAG-DEP 2.11

The regulatory authorities generally require monthly meetings of the board of directors and other volunteer committees.

#### Financial Structure

- **2.12** Because they are nonstock cooperatives, credit unions' primary source of funds is members' share and savings account deposits. To be entitled to membership, each member must generally own at least one share in the credit union. Members' shares or share accounts are savings accounts that represent the members' ownership in the credit union. Credit unions pay interest (commonly referred to as *dividends*) on shares. This interest cannot be guaranteed (as interest on deposits can), but ordinarily must be declared by the board and may be paid from current earnings or undivided earnings.
- **2.13** Credit unions use the funds from these shares and other members' savings accounts to make loans to members and to make investments. In general, loans to members make up the bulk of credit union assets. Funds not needed to meet member loan demand and operating expenses are invested.

# **Credit Union System**

- **2.14** Credit unions—through their state and national trade associations, service organizations, and corporate credit unions—make up the credit union system. Most credit unions are affiliated with the system through membership in their state or combined state credit union leagues. In turn, credit union leagues or associations belong to the Credit Union National Association, Inc. (CUNA), the principal trade association for credit unions in the United States, and CUNA belongs to the World Council of Credit Unions, an international credit union organization. On the national level, for-profit affiliates of CUNA (including the CUNA Strategic Services, Inc., and the CUNA Mutual Group) provide a wide variety of products and services to credit unions on a fee basis.
- **2.15** The National Association of State Credit Union Supervisors (NASCUS) was founded in 1965 and serves both state-chartered credit unions and the state credit union regulators who supervise them. NASCUS is the primary resource and voice of the 47 state governmental agencies that charter, regulate, and examine the nation's state-chartered credit unions. (Delaware, South Dakota, and Wyoming have no laws permitting state-chartered credit unions.) NASCUS is the only organization dedicated to the defense and promotion of the state credit union charter and the autonomy of state credit union regulatory agencies. NASCUS promotes a dual-chartering system and the advancement of the autonomy and expertise of state credit union regulatory agencies.
- **2.16** Other national credit union associations include the National Association of Federal Credit Unions, the Credit Union Executives Society, and other associations serving similar credit unions such as educational, defense-related, or aerospace credit unions. These groups may also provide such services as supplies, marketing, insurance, fund transfers, and investment instruments through their affiliates.

# **Corporate Credit Unions**

**2.17** A *corporate credit union* is defined as a credit union organized by natural person credit unions to offer central liquidity, investment, back office processing, deposit and lending facilities for natural person credit unions similar to those provided by other correspondent banking service providers.

# Regulation and Oversight

# **Government Supervision**

- **2.18** Credit unions operate under either a federal or state charter and, therefore, are subject to government supervision and regulation, including periodic examinations by supervisory agency examiners. Federally chartered credit unions are supervised by the NCUA, which is also responsible for administering the NCUSIF. The NCUSIF provides share insurance to all federal credit unions and federally insured, state-chartered credit unions, and insures each deposit up to a specified amount. Each federally insured credit union is required to maintain a deposit with the NCUSIF in an amount equal to 1 percent of its total insured shares.
- **2.19** State-chartered credit unions are supervised by the regulatory agency of the chartering state. Most state-chartered credit unions are ordinarily required to obtain NCUSIF share insurance coverage. Such credit unions are subject to a periodic insurance examination by the NCUSIF generally performed jointly with their state supervisory authority. Credit unions are allowed to obtain insurance from other sources that are sponsored by a private insurer, depending on state laws. Participation in an insurance program is mandatory for all credit unions.
- 2.20 Credit unions are subject to the federal, state, and local laws applicable to financial institutions in general. Most state laws allow and follow federal parity whereby they will basically follow NCUA rules and regulations. Such laws include the Uniform Commercial Code, the Truth-in-Lending Laws. the Uniform Consumer Code, Truth-in-Savings regulations, CFPB regulations, and various federal and state tax codes. As financial institutions, they are also subject to a wide variety of federal regulations issued by such agencies as the Treasury Department, the Board of Governors of the Federal Reserve System (Federal Reserve), and the IRS. Rules and regulations issued by the federal and state regulatory agencies address such issues as accounting practices, qualifications for membership, interest rate controls, permissible investments, consumer-protection issues, liquidity reserves, and other operational aspects. The Sarbanes-Oxley Act of 2002 and the SEC implementing regulations do not specifically apply to federal credit unions. However, the NCUA issued Letter to Federal Credit Unions No. 03-FCU-07, Guidance on Selected Provisions of the Sarbanes-Oxley Act of 2002 for Federal Credit Unions (FCUs), in October 2003 to provide a summary of certain provisions within the Sarbanes-Oxley Act of 2002 that NCUA believes are relevant to federal credit unions.

### **NCUA**

- **2.21** The NCUA issues regulations for both federal credit unions and federally insured, state-chartered credit unions. Federally insured, state-chartered credit unions sign an insurance agreement with the NCUA when they secure federal insurance that stipulates the regulations by which they agree to be bound. NCUA publications that provide useful background information to credit union auditors include the following:
  - The FCUA
  - NCUA letters to credit unions, legal opinions, and regulatory updates

©2016, AICPA AAG-DEP 2.21

- Federal Credit Union Bylaws
- NCUA Chartering and Field of Membership Manual
- NCUA Examiner's Guide
- Accounting Manual for Federal Credit Unions (and interim Accounting Bulletins)
- Supervisory Committee Guide for Federal Credit Unions
- The Federal Credit Union Handbook

Many of the previously mentioned documents can be found at the NCUA's website at www.ncua.gov.

**2.22** Credit unions with under \$10 million in assets are provided the *Accounting Manual for Federal Credit Unions*, which includes some regulatory accounting practices as a guide in accounting for financial transactions and reporting. In accordance with the CUMAA, credit unions with \$10 million or more in assets must follow GAAP in the Call Reports they file with the NCUA. These credit unions should not look to this manual, but should seek the advice of an independent accountant to gain a full understanding of GAAP and its implementation. The manual may be adopted by federally insured, state-chartered credit unions under \$10 million in assets at the option of the credit unions and their state supervisor.

**2.23** The NCUA administers a Central Liquidity Facility (CLF). The CLF is a mixed ownership government corporation created to improve the general financial stability of credit unions by serving as a liquidity lender to credit unions experiencing unusual or unexpected liquidity shortfalls. Membership is voluntary and is open to all natural person and corporate credit unions that purchase a prescribed amount of CLF stock. The NCUA encourages all credit unions to participate in either the CLF or have direct access to Federal Reserve borrowings.

# **Regulatory Capital Matters**

### Natural Person Credit Unions

# Capital Adequacy

### © Update 2-1 Regulatory: Risk-Based Capital

The NCUA regulatory final rule *Risk-Based Capital*, issued in October 2015, is effective on January 1, 2019, for federally insured, natural person credit unions with assets over \$100 million.

The final regulation restructures the NCUA's PCA regulations and makes various revisions, including amending the agency's current risk-based net worth requirement by replacing it with a new risk-based capital ratio for federally insured natural person credit unions. The risk-based capital requirement set forth in the final rule is more consistent with the NCUA's risk-based capital measure for corporate credit unions and, as the law requires, more comparable to the regulatory risk-based capital measures used by the FDIC, the Federal Reserve, and the Office of the Comptroller of the Currency.

The final rule also eliminates several provisions in the NCUA's current PCA regulations, including provisions relating to regular reserve accounts, risk-mitigation credits, and alternative risk-weights.

This edition of the guide has not been updated to reflect changes as a result of this final rule, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this regulation on NCUA's website at www.ncua.gov.

- **2.24** Title III of the CUMAA established a system of tiered net worth requirements for all insured natural person credit unions. These requirements did not take effect until August 2000. The act required that the NCUA establish a net worth standard for insured credit unions as well as risk-based capital standards for complex credit unions as defined by the NCUA. A separate system of PCA was mandated for new credit unions. A *new credit union* is defined as a federally insured credit union that both has been in operation for less than ten years and has \$10 million or less in total assets. A summary of general requirements follows. In 2000, the NCUA published PCA guidelines in the *Federal Register* with respect to the risk-based net worth requirement (RBNWR). Specific requirements are set forth in Title 12 U.S. *Code of Federal Regulations* (CFR) Parts 700, 702, 741, and 747.
- **2.25** Under the existing net worth standard, a credit union's *net worth*, the numerator of the net worth ratio, is defined in 12 CFR 702.2 as the retained earnings balance of the credit union at quarter-end as determined under GAAP together with any amounts that were previously retained earnings of any other credit union with which the credit union has combined.<sup>1</sup>
- **2.26** A credit union's total assets, the denominator of the net worth ratio, is calculated in any one of four methods. It may be (a) the average of the quarter-end balances of the four most recent quarters, (b) the monthly average over the quarter, (c) the daily average over the quarter, or (d) the quarter-end balance. A credit union may elect a method from the four options to apply for each quarter. Whatever method is chosen for a quarter generally must be used consistently for all PCA measures other than the RBNWR.
- **2.27** Credit unions with a ratio of less than 7 percent net worth to total assets and any complex credit union, as defined in the following, not meeting its RBNWR will be required to increase net worth quarterly by an amount of earnings equivalent to at least 1/10 percent (0.1 percent) of total assets for the current quarter. Earnings are required to be transferred quarterly from current earnings to the statutory (regular) reserve. Not all states that have state-chartered credit unions permit this transfer. As in some states, legislation may be necessary to enact change. Separate calculations may also be required for state-chartered credit unions subject to state-imposed capital requirements

©2016, AICPA AAG-DEP 2.27

¹ In response to the FASB projects on business combinations, Congress passed the Regulatory Relief Act of 2006, which changed the definition of net worth to include premerger retained earnings. The change was necessary to correct the unintended regulatory capital consequences in mutual to mutual combinations of the recent accounting guidance. The recent accounting guidance working in conjunction with the existing prompt correction action rule resulted in the dilution of the retained earnings of the continuing credit union (its net worth) by the total assets of both the continuing and target credit unions without benefit of the fair value of the acquiree's equity or member interests. The preferred remedy of redefining net worth as retained earnings plus equity acquired in combination was not advanced by Congress; thus, the premerger retained earnings outcome binding on current combinations was a compromise remedy.

and may be significantly different from the federal requirements. However, a credit union's net worth category may be downgraded if any supervisory or safety and soundness issues are identified at the credit union by the NCUA or any applicable state regulatory authority.

- **2.28** In 1998, Congress amended the FCUA to require the NCUA board to adopt a system of PCA to be applied to federally insured credit unions that become undercapitalized. The new FCUA provision imposes a series of progressively more stringent restrictions and requirements indexed to five net worth categories. The provision also mandates a separate system for new credit unions and additional RBNWRs for complex credit unions. See 12 CFR 702 for details of the system of PCA.
- **2.29** A credit union is defined as *complex*, and a RBNWR is applicable only if the credit union meets both of the following criteria as reflected in its most recent Call Report:
  - The credit union's quarter-end total assets exceed \$50 million.
  - The credit union's RBNWR exceeds 6 percent.
- **2.30** Under the PCA regulations of the NCUA, a credit union is classified in a net worth category as presented in exhibit 2-1.

### Exhibit 2-1

### **Net Worth Classifications**

Classification	Net Worth Ratio	Prompt Corrective Action
Well capitalized	7% or greater	None
Adequately capitalized	>6% but <7%	Earnings transfer
Undercapitalized— first tier	>5% but <6%	Mandatory for level
Undercapitalized— second tier	>4% but <5%	Mandatory and discretionary for level
Significantly undercapitalized	Either  a. >2% but <4% b. or >4% but <5% and either  i. fails to submit an acceptable net worth restoration plan; or  ii. materially fails to implement a restoration plan approved by the NCUA board.	Mandatory and discretionary for level
Critically undercapitalized	<2%	Mandatory and discretionary for level

- **2.31** Under PCA regulations, the net worth category of a credit union that is not considered *complex* is determined by calculating the ratio of the credit union's net worth (equal to retained earnings as defined under GAAP) to total assets (computed under any of the four methods described previously) based on the net worth categories in exhibit 2-1.
- **2.32** A credit union that is considered *complex* must compare its net worth ratio to its RBNWR to determine its net worth category. If a complex credit union's net worth ratio exceeds its RBNWR, the net worth category under the PCA regulations is determined by calculating its net worth ratio and based on the net worth categories in exhibit 2-1.
- **2.33** A complex credit union whose RBNWR is greater than its net worth ratio cannot be classified as either "well capitalized" or "adequately capitalized" under PCA regulations. If its net worth ratio is 6 percent or more, the actual net worth classification under PCA regulations would be the first tier of "undercapitalized." For a complex credit union with a net worth ratio less than 6 percent, any net worth restoration plan submitted by the credit union would have to consider the RBNWR if that requirement were greater than 6 percent.
- **2.34** The RBNWR is computed by multiplying the end-of-quarter balances of the credit union's risk-portfolio components (as defined in the regulations) by prescribed percentages (the standard calculation). If the standard calculation produces a RBNWR that is larger than the credit union's net worth ratio, the credit union can recalculate its RBNWR using some or all of the alternative components approach. In the alternative components approach, the maturities of several of the risk-portfolio components are used to produce a more detailed set of calculations, again each with a prescribed risk percentage. If the alternative components approach produces a RBNWR that is less than the credit union's net worth ratio, the credit union would have met its RBNWR. If the alternative components approach produces a RBNWR that is larger than the credit union's net worth ratio, the credit union may apply to the NCUA for a risk mitigation credit (explained in the following) to reduce its calculated RB-NWR. If the credit union fails to obtain an adequate amount of risk mitigation credit to reduce its RBNWR below its net worth ratio, it would have failed its RBNWR. The RBNWR ratio is the sum of all components for each category at the calculation date.
- **2.35** As noted previously, a credit union that fails to meet its applicable RBNWR using either the standard or alternative calculations may apply to the NCUA board for a risk mitigation credit against the credit union's RBNWR. A risk mitigation credit may be granted by the NCUA board based upon proof from the credit union of mitigation of credit risk or interest rate risk. The amount of the credit and the period that the credit can be used by the credit union is up to the discretion of the NCUA board. A risk mitigation credit may be denied based on the information presented by the credit union or based on other subjective factors considered by the board of the NCUA. A risk mitigation credit may be withdrawn by the NCUA board at any time.
- **2.36** Beyond the net worth and RBNWR related actions noted previously, the NCUA board may reclassify a well capitalized credit union as adequately capitalized and may require an adequately capitalized or undercapitalized credit union to comply with certain mandatory or discretionary supervisory actions as if it were in the next lower net worth category in the following circumstances:

©2016. AICPA AAG-DEP 2.36

- The NCUA board determines that the credit union is in an unsafe or unsound condition.
- The NCUA board determines that the credit union has not corrected a material unsafe and unsound practice of which it was, or should have been, aware.
- **2.37** Actions that may be taken under the PCA provisions can include both mandatory and discretionary actions for each level of capitalization below well capitalized. Actions can range from setting earnings aside to build net worth to restricting or prohibiting certain activities.
- **2.38** According to regulations, credit unions classified as undercapitalized, significantly undercapitalized, or critically undercapitalized must submit a net worth restoration plan for restoring the credit union to adequate capitalization. Among other things, the plan could
  - specify a quarterly timetable of steps the credit union will take to become adequately capitalized,
  - contain a specific timetable for increasing net worth for each quarter of the plan,
  - specify the amount of earnings equivalent the credit union will transfer to its reserve account on a quarterly basis,
  - detail how the credit union will comply with other restrictions or requirements put into effect,
  - set forth the types and levels of activities that the union will engage, and
  - include pro forma statements covering the next two years at a minimum.

#### New Credit Unions

- **2.39** Under the FCUA, a new credit union is classified as
  - a. well capitalized if it has a net worth ratio of 7 percent or greater,
  - b. adequately capitalized if it has a net worth ratio of 6 percent or more but less than 7 percent,
  - c. moderately capitalized if it has a net worth ratio of 3.5 percent or more but less than 6 percent,
  - d. marginally capitalized if it has a net worth ratio of 2 percent or more but less than 3.5 percent,
  - e. minimally capitalized if it has a net worth ratio of 0 percent or greater but less than 2 percent, and
  - f. uncapitalized if it has a net worth of less than 0 percent.
- **2.40** The NCUA board may reclassify a well capitalized, adequately capitalized, or moderately capitalized new credit union to the next lower net worth category if they determine the credit union is in an unsafe or unsound condition, or the credit union has not corrected a material unsafe or unsound condition.
- **2.41** New credit unions classified as moderately, marginally, minimally, or undercapitalized generally must file a revised business plan. At a minimum, the following items should be included in the business plan:
  - Outline steps the credit union will take to become adequately capitalized.

- Set specific quarterly targets for increasing net worth for each year of the plan.
- Set forth the amount of earnings equivalent the credit union will transfer to its reserve account.
- Detail how the credit union will comply with other restrictions or requirements put into effect.
- **2.42** Actions that may be taken under the PCA provisions for new credit unions include both mandatory and discretionary actions ranging from the restriction or prohibition of certain activities to the appointment of a receiver or conservator.

### Notice and Effective Date of Net Worth Classification

- **2.43** A federally insured credit union should have notice of its net worth ratio (including any applicable RBNWR) and should be classified within the corresponding net worth category as of the earliest to occur of the following:<sup>2</sup>
  - The last day of the calendar month following the end of the calendar quarter
  - The date the credit union received subsequent written notice from NCUA or, if state-chartered from the appropriate state official, of a decline in net worth category due to correction of an error or misstatement in the credit union's most recent Call Report
  - The date the credit union received written notice from the NCUA board or, if state-chartered, the appropriate state official, of reclassification based on safety and soundness grounds
- **2.44** Noncompliance or expected noncompliance with regulatory net worth requirements may be a condition that, when considered with other factors, could indicate substantial doubt about an entity's ability to continue as a going concern. The implementation of the PCA provisions warrants similar attention by independent accountants when considering a credit union's ability to remain a going concern. In addition, when a credit union has met its RBNWR through the use of a risk mitigation credit, the subjectivity involved in granting and maintaining the credit may also warrant attention by independent accountants.

### **Corporate Credit Unions**

**2.45** Corporate credit unions have regulatory capital requirements that are different from those of other credit unions. Corporate credit unions are not covered by the net worth requirements applicable to other credit unions by virtue of the CUMAA. By statute, the equity or capital of a corporate credit union consists of reserves, undivided earnings, and contributed capital. The NCUA has established a regulatory capital requirement applicable to all corporate credit unions. A state-chartered corporate credit union is also subject to its applicable state law capital requirement, if any. For NCUA regulatory purposes, corporate credit union total capital, as defined in 12 CFR 704.2,

©2016, AICPA AAG-DEP 2.45

 $<sup>^2</sup>$  Readers may refer to Title 12 U.S. Code of Federal Regulations (CFR) Part 702.101 for additional information.

consists of the sum of tier 1 and tier 2 capital,<sup>3</sup> less the corporate credit union's equity investments not otherwise deducted when calculating tier 1 capital. *Tier 1 capital*, as defined in 12 CFR 704.2, represents the sum of (1) retained earnings; (2) perpetual contributed capital (PCC); (3) the retained earnings of any acquired credit union, or of an integrated set of activities and assets, calculated at the point of acquisition, if the acquisition was a mutual combination; and (4) minority interests in the equity accounts of credit union service organization (CUSOs) that are fully consolidated less all of the following:

- a. The amount of the corporate credit union's intangible assets that exceed one half percent of its moving daily average net assets. However, the NCUA (on its own initiative, upon petition by the applicable state regulator, or upon application from a corporate credit union) may direct the corporate credit union to add back some of these assets.
- b. Investments, both equity and debt, in unconsolidated CUSOs.
- c. An amount equal to any PCC or nonperpetual capital account (NCA) that the corporate credit union maintains at another corporate credit union.
- d. Beginning on October 20, 2016, and ending on October 20, 2020, any amount of PCC that causes PCC minus retained earnings, all divided by moving daily net average assets, to exceed two percent.
- e. Beginning after October 20, 2020, any amount of PCC that causes PCC to exceed retained earnings.
- **2.46** The essential function of PCC and nonperpetual capital is to serve as an additional reserve of capital to absorb losses in excess of retained earnings. Therefore, when there is a retained earnings deficit in a corporate credit union, the PCC and nonperpetual capital are depleted to the extent necessary to resolve the deficit. PCC and nonperpetual capital are at-risk capital reserves, and, because they are so designated, a corporate credit union has no legal obligation or authorization as a going concern to restore, replenish, or recoup depleted paid-in capital and membership capital out of future retained earnings, even if retained earnings substantially improve.
- **2.47** For NCUA purposes, a corporate credit union must maintain at all times (a) a leverage ratio<sup>4</sup> of 4.0 percent or greater, (b) a tier 1 risk-based capital ratio<sup>5</sup> of 4.0 percent or greater, and (c) a total risk-based capital ratio<sup>6</sup> of 8.0 percent or greater.

AAG-DEP 2.46

<sup>&</sup>lt;sup>3</sup> Tier 2 capital, as defined in 12 CFR 704.2, is the sum of nonperpetual capital accounts, as amortized under 12 CFR 704.3(b)(3); allowance for loan and lease losses calculated under U.S. generally accepted accounting principles to a maximum of 1.25 percent of risk-weighted assets; any perpetual contributed capital from tier 1 capital; and 45 percent of unrealized gains on available-for-sale equity securities with readily determinable fair values.

 $<sup>^4</sup>$  Leverage ratio, as defined in 12 CFR 704.2, means the ratio of tier 1 capital to moving daily average net assets. The moving daily average net assets, as defined in 12 CFR 704.2, means the average of daily average net assets for the month being measured and the previous 11 months.

 $<sup>^5</sup>$  Tier 1 risk-based capital ratio, as defined in 12 CFR 704.2, means the ratio of tier 1 capital to the moving monthly average net risk-weighted assets. The moving monthly average net risk-weighted assets, as defined in 12 CFR 704.2, means the average of the net risk-weighted assets for the month being measured and the previous 11 months. Measurements must be taken on the last day of each month.

 $<sup>^6</sup>$  Total risk-based capital ratio, as defined in 12 CFR 704.2, means the ratio of total capital to moving monthly average net risk-weighted assets.

- **2.48** NCUA may also establish different minimum capital requirements for an individual corporate credit union based on its circumstances.
- **2.49** In 2010, the NCUA amended the corporate credit union rules to strengthen the corporate credit union system regulation. As part of these amendments, the NCUA adopted new PCA provisions similar to those currently applicable to banks. Each corporate credit union will be assigned to one of five capital categories. See 12 CFR 704.4 for details of the system of PCA.
- **2.50** Under the PCA regulations of the NCUA, a corporate credit union is classified in a capital category as follows:
  - a. Well capitalized if it has a total risk-based capital ratio of 10 percent or greater; has a tier 1 risk-based capital ratio of 6 percent or greater; has a leverage ratio of 5 percent or greater; and is not subject to any written agreement, order, capital directive, or PCA directive issued by the NCUA to meet and maintain a specific capital level for any capital measure.
  - b. Adequately capitalized if it has a total risk-based capital ratio of 8 percent or greater; has a tier 1 risk-based capital ratio of 4 percent or greater; has a leverage ratio of 4 percent or greater; and does not meet the definition of a well capitalized corporate credit union.
  - c. Undercapitalized if it has a total risk-based capital ratio that is less than 8 percent; has a tier 1 risk-based capital ratio that is less than 4 percent; or has a leverage ratio that is less than 4 percent.
  - d. Significantly undercapitalized if it has a total risk-based capital ratio that is less than 6 percent; has a tier 1 risk-based capital ratio that is less than 3 percent; or has a leverage ratio that is less than 3 percent.
  - e. Critically undercapitalized if it has a total risk-based capital ratio that is less than 4 percent; has a tier 1 risk-based capital ratio that is less than 2 percent; or has a leverage ratio that is less than 2 percent.
- **2.51** The NCUA may reclassify a well capitalized corporate credit union as adequately capitalized and may require an adequately capitalized or undercapitalized corporate credit union to comply with certain mandatory or discretionary supervisory actions as if the corporate credit union were in the next lower capital category in the following circumstances:
  - a. Unsafe or unsound condition. The NCUA has determined, after notice and opportunity for hearing pursuant to 12 CFR 704.4(h)(1), that the corporate credit union is in an unsafe or unsound condition.
  - b. Unsafe or unsound practice. The NCUA has determined, after notice and an opportunity for hearing pursuant to 12 CFR 704.4(h)(1), that the corporate credit union received a less-than-satisfactory CAMEL rating (three or lower, in other words) for any rating category (other than in a rating category specifically addressing capital adequacy) and has not corrected the conditions that served as the basis for the less than satisfactory rating. Ratings under this paragraph refer to the most recent ratings (as determined either on-site or off-site by the most recent examination) of which the corporate credit union has been notified in writing.

- **2.52** The potential consequences of failing to meet capital standards include restrictions on activities, restrictions on investments and asset growth, restrictions on the payment of dividends, restrictions on executive compensation, requirements to elect new directors or dismiss management, and possible conservatorship.
- **2.53** According to PCA regulations, corporate credit unions classified as undercapitalized, significantly undercapitalized, or critically undercapitalized must submit a written capital restoration plan for restoring the credit union to adequate capitalization. Among other things, the plan must include
  - the steps the corporate credit union will take to become adequately capitalized;
  - the levels of capital to be attained during each year in which the plan will be in effect;
  - how the corporate credit union will comply with the restrictions or requirements that are put into effect;
  - the types and levels of activities in which the corporate credit union will engage; and
  - a description of the steps the corporate credit union will take to correct the unsafe or unsound condition or practice if required to submit a capital restoration plan as the result of a reclassification of the corporate credit union pursuant to paragraph 2.51.
- **2.54** Noncompliance or expected noncompliance with regulatory capital requirements may be a condition that, when considered with other factors, could indicate substantial doubt about an entity's ability to continue as a going concern. The implementation of the PCA provisions warrants similar attention by independent accountants when considering a corporate credit union's ability to remain a going concern.

### **Annual Audits**

- **2.55** As discussed in paragraph 2.05, the CUMAA requires that all federally insured credit unions with assets of \$10 million or more must follow GAAP for all reports or statements required to be filed with the NCUA board and obtain one of the following four services:
  - a. If the credit union is federally insured with assets of \$500 million or more, a financial statement audit performed in accordance with generally accepted auditing standards by a CPA or public accountant licensed by the appropriate state or jurisdiction in which the audit is conducted.
  - b. If the credit union is federally chartered with assets of more than \$10 million but less than \$500 million, the credit union has four options:
    - i. Follow the requirement in item a.
    - ii. Obtain an opinion audit on the credit union's balance sheet performed by an independent accountant licensed by the state or jurisdiction in which the audit is conducted.
    - iii. Obtain an examination of management's assertions regarding controls over Call Reporting conducted by an independent accountant licensed by the state or jurisdiction in which the audit is conducted.

- iv. Obtain a supervisory committee audit that meets the minimum requirements of the Supervisory Committee Guide.
- **2.56** For any federal credit union with assets of more than \$10 million that uses an independent accountant who is compensated for his or her services to perform a financial statement audit, the audit is subject to state accounting laws, including licensing requirements.
- **2.57** Although GAAP basis accounting is not mandated for internal reporting, GAAP is required for Call Reports filed with the NCUA board for credit unions with assets of \$10 million or more.<sup>7</sup>
- **2.58** A federally-chartered credit union with \$10 million or less in total assets must obtain an annual Supervisory Committee audit.
- 2.59 The minimum requirements for a supervisory committee audit of federally chartered credit unions are prescribed by 12 CFR 715.8 State-chartered credit unions are subject to the audit requirements established by state regulatory agencies if they are more stringent than 12 CFR 715 requirements. To satisfy regulatory requirements for a supervisory committee audit, the supervisory committee may perform the necessary procedures itself or it may engage an independent accountant to perform procedures that are necessary to fulfill the federal or state requirements. Because the types of engagement can differ so significantly, it is important for the independent accountant to establish a clear understanding of the nature of an engagement to perform a supervisory committee audit.
- **2.60** The NCUA requires CUSOs to obtain a separate financial statement audit from a CPA in accordance with generally accepted auditing standards before a federally insured credit union can invest or lend to that CUSO. A wholly owned CUSO is not required to obtain a separate annual financial statement audit if it is included in the annual consolidated financial statement audit of the investing federally insured credit union. See 12 CFR 712.3 for additional information.
- **2.61** The NCUA requires internal control and reporting requirements for corporate credit unions similar to those required for banks under the FDIC Improvement Act of 1991 and the Sarbanes-Oxley Act of 2002. The most significant revisions, which became effective January 1, 2012, require a corporate credit union to
  - ensure that its annual financial statements and regulatory reports reflect all material correcting adjustments necessary to conform with GAAP as identified by the independent public accountant.
  - prepare an annual management report, signed by the CEO and the chief accounting officer or CFO, that contains
    - a statement of management's responsibility for preparing the financial statements, for establishing and maintaining an adequate internal control structure, and for

©2016, AICPA AAG-DEP 2.61

Readers may refer to 12 CFR 741.6(b) for additional information.

<sup>&</sup>lt;sup>8</sup> Under the National Credit Union Administration's (NCUA's) supervisory committee audits regulations, all federal credit unions engaging in derivatives activities are required to have a financial statement audit, regardless of asset size (see 12 CFR 715.5). NCUA derivatives regulations are set forth in 12 CFR 703 Subpart B.

- complying with safety and soundness laws and regulations;
- an assessment of compliance with such laws and regulations; and
- an assessment of the effectiveness of the internal control structure. The assessment requirement is effective January 1, 2013, and therefore would be applicable to management reports for the calendar year 2012 and thereafter.
- ensure that its independent public accountant
  - reports to the supervisory committee all critical accounting policies;
  - retains for seven years the working papers related to an audit;
  - complies with the independence standards and interpretations of the AICPA;
  - has an acceptable peer review;
  - notifies the NCUA if the independent public accountant ceases being a corporate credit union's independent accountant; and
  - reports separately to the supervisory committee on management's assertions concerning the effectiveness of internal control structure. The requirement is effective January 1, 2014, and therefore would be applicable to management reports prepared for the calendar year 2013 and thereafter.
- ensure that it files a copy of its annual report to the NCUA within 180 days after the end of the calendar year, which the NCUA will make available for public inspection.
- provide the NCUA with a copy of any letter or report issued by its independent public accountant.
- inform the NCUA when it engages an independent public accountant or loses an independent public accountant through dismissal or resignation.
- provide a notice to NCUA of late filing of the annual report.
- submit a summary of its annual report to the membership.
- ensure that its supervisory committee consists of members who are independent of the corporate credit union.
- supervise the independent public accountant.
- ensure that audit engagement letters do not contain unsafe and unsound limitation of liability provisions.

See 12 CFR 704.15 for further information.

### **Other Reporting Considerations**

**2.62** The independent accountant may be requested to perform assurance services other than those required by the CUMAA to the extent that a credit union may be

- a. originating or holding student loans,
- b. servicing residential mortgage loans for others,
- c. originating or servicing Federal Housing Administration or Government National Mortgage Association loans subject to generally accepted government auditing standards (the U.S. Department of Housing and Urban Development),
- d. borrowing from a district Federal Home Loan Bank,
- e. participating in an automated-teller-machine network,
- f. originating or receiving automated-clearinghouse transactions,
- g. using outside technology partners, and
- h. subject to the provisions of the Bank Secrecy Act.

©2016, AICPA AAG-DEP 2.62

# Chapter 3

# Industry Overview — Finance Companies 1

# **Description of Business**

- **3.01** Finance companies provide lending and financing services to consumers (consumer financing) and to business entities (commercial financing). Many finance companies engage solely in consumer or commercial financing activities; others provide both types.
- **3.02** Manufacturers, retailers, wholesalers, and various other business entities may provide financing to encourage customers to buy their products and services. Such financing, generally known as *captive finance activity*, may be provided directly by those companies or through affiliated companies. Although most such companies originally financed only their own products and services, many have expanded their financing activities to include a wide variety of products and services sold by unaffiliated businesses.
- **3.03** Consumer finance activities comprise direct and indirect consumer loans, including auto, credit card, and mortgage loans and retail sales financing. Direct loan companies provide loans directly to the customers. Indirect lenders typically are purchasing retail installment contracts from retail establishments or are directly purchasing loans from other companies. Many companies that provide consumer financing also offer a variety of insurance services or other ancillary products to their borrowers.
- **3.04** Insurance and other ancillary products and services. Many companies engaged in consumer finance activities also offer insurance coverage to their customers. Such coverage may include life insurance to repay remaining loan balances if borrowers die: accident and health insurance to continue loan payments if borrowers become sick or disabled for an extended period of time: debt cancellation products to cover the remaining obligation on certain events covered in the contract; and property insurance to protect the values of loan collateral against damage, theft, or destruction. Some lenders may also sell extended service contracts, auto clubs, or other ancillary products or receive commissions for those products. Some lenders may provide insurance through subsidiaries. Others act as brokers and, if licensed, often receive commissions from independent insurers or third-party service providers. Lenders also may receive retrospective rate credits on group policies issued by independent insurers. In still other instances, policies may be written by independent insurance companies and then reinsured by the insurance subsidiaries of finance companies.
- **3.05** Commercial finance companies often provide a wide range of services, including factoring arrangements, revolving loans, installment and term loans, floor plan loans, portfolio purchase agreements, and lease financing to a variety of clients, including manufacturers, wholesalers, retailers, and service organizations. Many commercial finance activities are called *asset based financial services* because of the lenders' reliance on collateral.

©2016, AICPA AAG-DEP 3.05

<sup>&</sup>lt;sup>1</sup> This guide covers entities under FASB *Accounting Standards Codification* 942-10-15-2, which includes finance companies and finance company subsidiaries.

- **3.06** Commercial loans are either secured by various types of assets, including notes and accounts receivable; inventories; and property, plant, and equipment; or they are unsecured.
- **3.07** Increased competition has come from both within the industry and from nontraditional players such as investment companies, brokers and dealers in securities, insurers, and financial subsidiaries of commercial entities. These entities now do business directly with potential customers in transactions traditionally executed through finance companies. This disintermediation has increased the need for innovative approaches to attracting customers. It has also led to an increased need for more complex financing structures such as use of tax oriented vehicles, the ability to offer longer term financing than traditional banks, and a higher level of asset knowledge to take more aggressive residual positions and collateral risk.

### **Debt Financing**

- **3.08** The basic activity of finance companies is borrowing money at wholesale interest rates and lending at a markup. Strong credit ratings foster the ability to attract wholesale funds at a competitive cost. Accordingly, in order to qualify for high credit ratings, it is common for finance companies to structure financing transactions according to predetermined rating agency credit criteria. A credit rating represents a measure of the general creditworthiness of an obligor with respect to a particular debt security or financial obligation, based on relevant risk factors. Historically, the credit ratings from rating agencies such as Standard & Poor's Rating Services, Moody's Investor Service, and Fitch Ratings have been used to differentiate an obligor's credit quality.
- **3.09** Unlike most depository institutions, finance companies typically do not utilize low-cost customer deposits as a significant source of funding. Accordingly, access to a variety of funding sources is vital to market access, liquidity, and funding cost effectiveness. Typical short-term funding sources include commercial paper and bank credit facilities. Senior debt, senior subordinated debt, and junior subordinated debt are typical medium term to long term funding sources. It is common for these types of funding sources to contain restrictive covenants.
- **3.10** Securitization is often utilized by finance companies to expand and diversify their funding sources. In some markets, securitization has reduced entry barriers and increased competition. Securitization involves the sale, generally to a trust, of a portfolio of loan receivables. Asset-backed certificates are then sold by the trust to investors through a private placement or public offering. Typically, the finance company will retain the servicing rights for the loans sold to the trust. A subordinated interest in the trust is also typically retained by the finance company, serving as a credit enhancement to the asset-backed certificates. Such structures provide the opportunity for less credit worthy companies to obtain funding at competitive levels through the asset-backed and other structural characteristics of securitization vehicles.
- **3.11** The Risk Management Association, an organization of bank lending officers, has developed financial information questionnaires for lenders engaged in retail sales financing, direct cash lending, commercial financing, captive financing activities, and mortgage banking. Finance companies generally complete and submit the questionnaires to credit grantors as an integral part of the process of obtaining credit lines with commercial banks and other lenders. The information is used to analyze the quality of the operations

and creditworthiness of finance companies. More information can be found at www.rmahq.org.

# Regulation and Oversight

- **3.12** Publicly held finance companies are generally subject to requirements of federal securities laws, including the Securities Act of 1933, the Securities Exchange Act of 1934 (the 1934 Act), and the Sarbanes-Oxley Act of 2002. Companies whose securities are registered under the 1934 Act must comply with its reporting requirements through periodic filings with the SEC.
- **3.13** Numerous state and federal statutes affect finance companies' operations. Some statutes apply only to specific types of activities. Regulations affecting finance companies generally are limited to matters such as loan amounts, repayment terms, interest rates, collateral, compliance with state laws, and consumer loan product disclosure requirements; they generally do not address financial accounting and reporting. Certain of the more significant state and federal laws related to consumer lending are discussed in chapter 8, "Loans," of this guide.
- **3.14** Lending and other activities of finance companies are also generally subject to oversight by the Consumer Financial Protection Bureau (CFPB). The CFPB was created by the Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010.

©2016. AICPA AAG-DEP 3.14

# Chapter 4

# Industry Overview—Mortgage Companies

# **Description of Business**

- **4.01** As a result of the relative imbalance between the supply and demand for residential mortgage funds, mortgage banking entities play an integral role in providing mortgage capital based on housing finance demands of the general public in various geographic locations. The market where mortgage banking entities originate loans to borrowers is referred to as the *primary market*. The market where originated loans and mortgage-backed securities (MBS) trade is referred to as the *secondary market*.
- 4.02 The principal participants in the secondary market for residential financing are government sponsored entities (GSEs), such as the Federal Home Loan Mortgage Corporation (Freddie Mac) and the Federal National Mortgage Association (Fannie Mae). Also active in the secondary market are federal agencies such as the Government National Mortgage Association (Ginnie Mae), the U.S. Department of Veterans' Affairs (VA), and the Federal Housing Administration (FHA). These entities participate in the secondary market as issuers, investors, or guarantors of asset-backed securities (ABSs) such as MBSs, real estate mortgage investment conduits, and collateralized mortgage obligations. Private entities have also been active in the secondary market as issuers, investors, and guarantors, though the extent of their participation has varied significantly in response to market conditions. (Chapter 7, "Investments in Debt and Equity Securities," of this guide describes ABS transactions and considerations for investors in ABSs.)
- **4.03** Freddie Mac and Fannie Mae primarily purchase conventional fixed and variable rate residential mortgage loans, originated by private entities, and Ginnie Mae generally purchases pools of government insured residential mortgage loans. These secondary market participants typically buy and sell originated loans, securitize them into MBSs, and sell the securities to investors.
- **4.04** MBSs became more prominent with the creation of Ginnie Mae in 1968 and the subsequent issuance of the first Ginnie Mae pass-through securities. Nontraditional mortgage investors were more inclined to invest in Ginnie Mae pass-through securities as a result of government guarantees on both the underlying mortgage collateral and on the securities themselves. During this same time period, Freddie Mac began selling pass-through securities backed by conventional residential mortgages. By the mid-1970s, the investment community accepted MBSs as viable securities collateralized by residential mortgages.
- **4.05** Beginning in the early 1970s, secondary market activities for all mortgage lenders increased substantially as a result of the establishment of Freddie Mac and the new involvement of Fannie Mae and Freddie Mac with the conventional secondary market. Prior to that time, Fannie Mae was one of the few national secondary mortgage market participants through its whole loan purchase and sale programs related to government loans.
- **4.06** In 2008, Fannie Mae and Freddie Mac experienced dramatic repercussions as a result of the financial crisis and were placed into conservatorship of the Federal Housing Finance Agency (FHFA). In September 2008, the U.S.

Treasury Department acquired \$1 billion of preferred shares in each GSE and has since provided additional capital as necessary. The future state of the GSEs is uncertain and has been the subject of legislative focus since they were placed into conservatorship.

- **4.07** The securities markets play a significant role in the execution and pricing of residential MBSs. In addition, the markets handle a significant volume of residential mortgage backed transactions. As a result, securities markets influence mortgage pricing on a national scale and also influence the design of various mortgage products.
- **4.08** With the dominant role of the mortgage securities markets and economic changes throughout the mortgage lending industry, nontraditional participants in the secondary market (as opposed to the traditional bank and thrift portfolio lenders) continue to evolve. Securities underwriters, commercial banks, financial guaranty companies, insurance companies, and real estate investment trusts play various roles in the mortgage banking industry. In addition, mortgage lending entities have securitized other types of loan products, such as commercial mortgage loans, subprime residential mortgage loans, and home equity loans (junior lien mortgages).
- **4.09** The Mortgage Partnership Finance (MPF) Program, which is available through most Federal Home Loan Banks (FHLBs), provides FHLB member institutions an alternative method for funding home mortgages for their customers. Under the MPF Program, the lender originates loans for, or sells loans to, the respective FHLB. The lender retains some or all of the credit risk and customer relationship (through servicing) inherent in the loan, and shifts the interest rate risk and prepayment risk to the FHLB. The lender receives a credit enhancement fee from the FHLB in exchange for managing the credit risk of the loan. Effectively, the FHLBs offer an alternative funding strategy to the traditional secondary mortgage market, particularly for smaller entities that do not have the desire or ability to hedge the associated interest rate risk and prepayment risk. The Mortgage Purchase Program (MPP) was introduced in 2000 to further support the FHLBs' mission of expanding housing finance opportunities in the several districts for members that originate and hold mortgages on their books. The MPP, similar to the MPF program shifts the interest rate risk and prepayment risk to the FHLB while the member retains the customer relationship and credit risk of the loan.
- **4.10** Many mortgage banking entities are subsidiaries of banks or bank holding companies. Mortgage banking is generally compatible with a bank's financing operations, and the bank is an obvious resource for the mortgage banking entity's financing requirements. A mortgage banker typically draws upon its warehouse line of credit, whereby mortgage loans are funded by advances from the credit line, and are "warehoused" in the portfolio as security for the credit line until the credit line is paid down through the subsequent sale of the mortgage loans into the secondary market. The interest margin between the rate a mortgage bank can fund its operations and the rate at which it can extend mortgage financing is critical to the financial success of the entity.
- **4.11** In turn, access to the secondary mortgage market is an important source of liquidity for banks and savings institutions. Many institutions have deposit bases that are keyed to variable rates and, therefore, are particularly sensitive to interest rate risk. A variable rate deposit base cannot fund long term, fixed rate assets without creating significant loss exposure in rising

interest rate environments. Therefore, sales of mortgage loans in the secondary market are an important source of liquidity and income. In addition, income streams created from servicing and other ancillary fees are an important source of funds to many institutions. Access to the secondary market also provides opportunities to restructure existing long-term loan portfolios, in order to meet asset/liability objectives or capital requirements.

- **4.12** Mortgage banking activities primarily consist of two separate but interrelated activities, namely, (a) the origination or acquisition of mortgage loans for the purpose of selling those loans to permanent investors in the secondary market, and (b) the subsequent servicing of those loans. Mortgage loans are acquired for sale to permanent investors from a variety of sources, including in-house origination and purchases from third-party correspondents.
- **4.13** Residential mortgage loans may be sold to investors with or without the right to service such loans (that is, sold *servicing released* or *servicing retained*, respectively). Fannie Mae, Freddie Mac, Ginnie Mae, and securitization vehicles do not have their own servicing functions, so mortgage companies that sell loans to those entities in the secondary market typically sell the loans servicing retained or sell the servicing to another party. Servicing rights offer the servicer additional and potentially significant sources of income in the form of servicing fees, late fees, float earnings, and other ancillary fees. Servicing fees are typically expressed as a percentage of a loan's unpaid principal balance, such as 0.25 percent or 25 basis points per year, which is deducted from the amounts due to the investor (that is, the servicing fee is a "strip" of the loan's stated coupon interest). Servicing also provides intangible benefits such as customer relationships, which may yield future sources of revenue such as refinancing fees.
- **4.14** The servicing function includes collecting payments from borrowers, transmitting insurance and tax payments to the related recipients, remitting payments to investors, performing the collection and loss mitigation functions for delinquent loans, and handling all phases of foreclosure proceedings. The precise nature of the servicing function is dependent on the specific requirements of the investor and the pooling and servicing agreement or similar agreement.
- **4.15** The servicing rights attributable to a mortgage portfolio are generally viewed as a primary asset of a mortgage banking entity. The value of the servicing asset is a function of the anticipated life of the servicing right (how long the loan is expected to be outstanding and serviced) and the estimated net servicing revenues attributable to the servicing function as compared to the adequate servicing compensation that another substitute servicer would demand in the market. The market for purchases and sales of loan servicing rights is limited; thus, there is a limited degree of liquidity and observable market prices for servicing rights. Even when there is a price discovery for servicing rights, an entity needs to consider the relevance of observable prices because the transactions may involve unique terms or may reflect features that are not an attribute of the unit of account being measured (for example, intangible assets). Furthermore, servicing portfolios are subject to significant volatility in valuation as unanticipated periods of rapid prepayments, increases in loan losses, increases in market servicing costs, and changes in the discount rates can cause substantial declines in the value of the servicing asset. Accordingly, the assumptions upon which the value of servicing transactions is based are critical to the reported financial results of the servicing entity. Refer to FASB

Accounting Standards Codification (ASC) 860, Transfers and Servicing, for accounting requirements relating to servicing rights. See paragraphs 4.21–.32 for regulatory guidance about servicing assets.

- **4.16** FASB ASC 948, Financial Services—Mortgage Banking, establishes accounting and reporting standards for mortgage banking entities and entities that engage in certain mortgage banking activities. Some of the items subject to the guidance in FASB ASC 948 are financial instruments. In addition, FASB ASC 820, Fair Value Measurement, provides guidance that may be applied to certain mortgage related items such as derivative loan commitments, derivative sales contracts, loans held for sale, and servicing rights. FASB ASC 825, Financial Instruments, allows entities to choose, at specified election dates, to measure eligible items at fair value (the fair value option) with gains and losses recorded in earnings at each subsequent reporting date. Additionally, FASB ASC 860-50-30-1 requires an entity at initial recognition to measure a servicing asset or servicing liability that qualifies for separate recognition regardless of whether explicit consideration was exchanged at fair value. In accordance with FASB ASC 860-50-35-1, an entity should subsequently measure each class of servicing asset and servicing liability using either the amortization method or the fair value measurement method. Paragraphs 2 and 4 of FASB ASC 860-50-35 state that the election should be made separately for each class of servicing assets and servicing liabilities and the entity should apply the same subsequent measurement method to each servicing asset and servicing liability in a class. A detailed discussion on the accounting for servicing assets and servicing liabilities can be found in chapter 10, "Transfers and Servicing and Variable Interest Entities," of this guide.
- **4.17** Some mortgage banking entities elect to account for loans held for sale at fair value under the fair value option in accordance with FASB ASC 825. Because derivative instruments are often used to mitigate the risks of subsequent changes in fair value of the loans between the commitment date and the sale, applying the fair value election eliminates the operational burden of achieving fair value hedge accounting in accordance with the requirements of FASB ASC 815, *Derivatives and Hedging*, because both the loans and the derivative instruments used to hedge the risks of changes in fair value are recorded at fair value through earnings.
- **4.18** The magnitude of interest rate movements and the speed with which they can occur make risk management in a mortgage banking entity complex and difficult because they influence the demand for loans, the qualification of borrowers for loans, prices of loans, and how long the loans will be outstanding before they are repaid (for example, prepayment risk). Risk mitigation strategies and operating plans, as well as sophisticated reporting systems that provide the information needed to carry out the plans and strategies, are used to monitor and control the interest risk exposure of mortgage banking operations.
- **4.19** In addition to the interest rate risk inherent in the mortgage loan pipeline (the inventory of loan commitments in various stages of process) and prepayment risk inherent in the servicing asset, an entity that sells loans it originated or purchased also is subject to recourse or repurchase risk under

<sup>&</sup>lt;sup>1</sup> FASB Accounting Standards Codification (ASC) 820, Fair Value Measurement, defines fair value, establishes a framework for measuring fair value, and requires certain disclosures about fair value measurements. A summary of the guidance in FASB ASC 820 may be found in chapter 20, "Fair Value," of this guide.

specific provisions or representations and warranties in the sale agreement. Recourse risk is the risk that an investor may either reject a loan or mandate the mortgage lender to repurchase the loan or reimburse the investor for credit-related losses if there is a defect related to the underwriting or documentation of the loan that contributes to a subsequent loss or if the loan becomes delinquent within a specified amount of time after purchase. This risk varies based on the source and underwriting procedures of the loan, terms of the sale, and servicing agreement with each investor. An entity will establish a liability measured at fair value at the date of transfer related to these various representations and warranties that reflect management's estimate of losses for loans for which it has repurchase or make-whole obligations and will adjust that contingent liability subsequently based on actual claims activity and changes in expectations. Many investors have filed lawsuits against private issuers of MBSs over defects in the underwriting process and inadequate disclosures over credit risks inherent in the underlying loans.

**4.20** Mortgage banking is a complex financial services business requiring advanced analytical skills, financial modeling, and forecasting abilities. The necessary level of computer systems support for mortgage banking operations is significant. Access to large volumes of accurate data that is instantly available is paramount in managing the risks of mortgage banking. The resources necessary to compete effectively have made it difficult for the small, independent firm to survive, and the medium to large size mortgage banking operations are often subsidiaries of larger institutions, both financial and non-financial.

# Regulation and Oversight

- **4.21** Publicly held mortgage companies are generally subject to requirements of federal securities laws, including the Securities Act of 1933, the Securities Exchange Act of 1934 (the 1934 Act), and the Sarbanes-Oxley Act of 2002. Companies whose securities are registered under the 1934 Act must comply with its reporting requirements through periodic filings with the SEC.
- **4.22** Virtually all states have enacted laws governing the conduct of mortgage lenders and mortgage servicers, and have created regulatory bodies to oversee the industry. The majority of all jurisdictions have licensing requirements for mortgage brokering, lending, and servicing. The scope of these requirements can vary significantly. Certain states simply require that an entity register with a state before participating in a certain mortgage related activity. Other regulations require compliance with strict regulations concerning recordkeeping, office location, accounting, and origination and servicing procedures.
- **4.23** The mortgage lending process is regulated by both state and federal law. Regulations are generally designed to protect the consumer from unfair lending practices, and noncompliance with the regulations may result in financial liability, including the imposition of civil money penalties and reimbursements to borrowers, where applicable. Certain of the more significant regulations are discussed in chapter 8, "Loans," and chapter 9, "Credit Losses," of this guide. On February 25, 2003, the Office of the Comptroller of the Currency (OCC); the FDIC; the Board of Governors of the Federal Reserve System (Federal Reserve) (collectively, the federal banking agencies); and the Office of Thrift Supervision (OTS), prior to its transfer of powers to the OCC, the FDIC,

and the Federal Reserve,<sup>2</sup> issued *Interagency Advisory on Mortgage Banking*. This important document discusses examination concerns about the valuation and modeling of servicing assets and discusses the need to determine if an impaired servicing asset should be written off. In May 2005, the federal banking agencies, along with the OTS, and National Credit Union Administration (NCUA) issued *Interagency Advisory on Accounting and Reporting for Commitments to Originate and Sell Mortgage Loans*. This advisory provides guidance related to the origination of mortgage loans that will be held for resale, and the sale of mortgage loans under mandatory delivery and best efforts contracts.

- **4.24** On October 4, 2006, the federal banking agencies, along with the OTS, and the NCUA jointly issued *Interagency Guidance on Nontraditional Mortgage Product Risks*. The guidance discusses how institutions can offer nontraditional mortgage products in a safe and sound manner and in a way that clearly discloses the benefits and risks to borrowers. On June 8, 2007, the federal banking agencies, along with the OTS, and the NCUA jointly published guidance entitled *Illustrations of Consumer Information for Nontraditional Mortgage Products*. The illustrations are intended to assist institutions in implementing the consumer protection portion of the *Interagency Guidance on Nontraditional Mortgage Product Risks* (interagency guidance).
- **4.25** On May 29, 2008, the federal banking agencies, along with the OTS, and the NCUA published *Illustrations of Consumer Information for Hybrid Adjustable Rate Mortgage Products*. The illustrations are intended to assist institutions in implementing the consumer protection portion of the *Interagency Statement on Subprime Mortgage Lending* adopted on July 10, 2007, and to provide information to consumers on hybrid adjustable rate mortgage products as recommended by that interagency statement. The illustrations are not model forms and institutions may choose not to use them.
- **4.26** In March 2009, the Treasury Department announced guidelines under the Home Affordable Mortgage Program to promote sustainable loan modifications for homeowners at risk of losing their homes due to foreclosure. Modifications come in the form of lower monthly payments and principal reductions. Also in March 2009, the FHFA, which regulates Fannie Mae and Freddie Mac, created the Home Affordable Refinance Program (HARP). HARP incentivizes mortgage servicers to modify loans to give healthy borrowers access to market interest rates in instances in which the loan collateral is worth less than the balance on the mortgage. The FHA and VA also have comparable loan modification programs. These modification programs were introduced as temporary responses to the financial crisis and, as such, had expiration dates in the short term; however, the programs have been extended past those dates and continue to be available to borrowers at risk of foreclosure.
- **4.27** In February 2014, the OCC issued an updated "Mortgage Banking" booklet of the *Comptroller's Handbook*. The booklet provides updated guidance to examiners and bankers on assessing the quantity of risk associated with mortgage banking and the quality of mortgage banking risk management. It also addresses changes to the functional area of production, secondary marketing, servicing, and mortgage servicing rights to incorporate regulatory changes. Furthermore, it addresses, among other statutory and regulatory changes, amendments to Regulation X and Regulation Z (such as, new

**AAG-DEP 4.24** 

<sup>&</sup>lt;sup>2</sup> See chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide for further discussion on the Office of Thrift Supervision transfer of powers.

servicing-related standards and requirements and ability-to-repay requirements) issued by the Consumer Financial Protection Bureau. The booklet applies to all banks engaged in mortgage banking activities. Readers are encouraged to view this publication under the "Publications—Comptroller Handbook" page at www.occ.gov.

- **4.28** In addition, in connection with various lending programs that a mortgage lender may be involved in, specific program requirements may be applicable. Certain common requirements are discussed in the following paragraphs.
- **4.29** The U.S. Department of Housing and Urban Development (HUD) sponsors a broad range of programs designed to revitalize urban neighborhoods, stimulate housing construction, encourage home ownership opportunities, and provide safe and affordable housing. The programs are carried out through various forms of federal financial assistance, including direct loans and mortgage insurance. The FHA was established by Congress in 1934 and is part of HUD. The FHA was created to encourage lenders to make residential mortgage loans by providing mortgage insurance. To participate in the FHA mortgage insurance program, a mortgage lender must obtain HUD approval by meeting various requirements prescribed by HUD, including maintaining minimum net worth requirements. Net worth requirements vary depending on the program.
- **4.30** To obtain approval to sell and service mortgage loans for Fannie Mae or Freddie Mac, or both, a mortgage lender must meet various requirements including maintaining an acceptable net worth. Upon approval, a mortgage lender enters into a selling and servicing contract and must comply with the terms of the respective selling and servicing guides, which set forth detailed requirements regarding underwriting, mortgage delivery, and servicing.
- **4.31** Ginnie Mae was created by Congress as part of HUD. Ginnie Mae's primary role is to guarantee MBSs issued by Ginnie Mae approved lenders and backed principally by FHA insured and VA-guaranteed loans. To obtain Ginnie Mae approval, a mortgage lender must meet various eligibility requirements as prescribed in chapter 2, "Eligibility Requirements—Approval as a Ginnie Mae Issuer," of the *Ginnie Mae MBS Guide*. Among the requirements, an issuer must be an approved FHA mortgagee in good standing and maintain specific net worth, liquidity, and institution-wide capital requirements.
- **4.32** Mortgage lenders may also enter into agreements with private investors to sell and service mortgage loans. Such agreements set forth various standards applicable to the transaction and may include minimum financial or net worth requirements.

# **Reporting Considerations**

# **HUD Programs**

**4.33** To participate in HUD programs, a nonsupervised mortgagee (a lender other than a financial institution that is a member of the Federal Reserve or whose accounts are insured by the FDIC or the NCUA) and a supervised mortgagee must comply with the requirements of the Consolidated Audit Guide for Audits of HUD Programs, issued by the HUD Office of the Inspector General. The guide requires that the engagement be performed in accordance

with Government Auditing Standards and contains (a) suggested procedures for testing an entity's compliance with laws and regulations affecting HUD-assisted programs, (b) a requirement to test internal control over compliance in all HUD-related audits, (c) the basic financial statements and types of supplementary information presented with an entity's basic financial statements, and (d) an auditor's reporting responsibilities and illustrative reports on the basic financial statements and supplementary information, internal control, and compliance with laws and regulations.<sup>3</sup>

**4.34** In August 2002, HUD released the Final Uniform Financial Reporting Standards Rule (Title 24 U.S. Code of Federal Regulations Part 5) requiring electronic submission of the financial statement package required for annual mortgagee recertification. In order to ensure the integrity of this audited financial information, mortgagees' auditors are required to attest to the data electronically. Refer to HUD Mortgagee Letter No. 2003-03, Mandatory Electronic Submission of Financial Statement Package for Annual Mortgagee Recertification. The required information and associated attestation were historically performed on the Lender Assessment Subsystem (LASS). In 2014, LASS was replaced by the Lender Electronic Assessment Portal.

**4.35** HUD releases Mortgagee Letters to notify lenders about amendments to FHA operations, policies, and procedures. Readers are encouraged to visit the "Mortgagee Letters" page at www.hud.gov to assess the applicability of the releases and how the releases might impact accounting and financial reporting matters, as well as audit requirements.

### **Asset Servicing for Investors**

**4.36** Lenders that service residential mortgage loans for investors may be required to engage an auditor to provide assurance relating to management's written assertions about compliance with the SEC Regulation AB or the minimum servicing standards set forth in the *Uniform Single Attestation Program for Mortgage Bankers* (USAP), or both. The USAP examination engagement is performed in accordance with AT section 601, *Compliance Attestation* (AICPA, *Professional Standards*). USAP was developed by the Mortgage Bankers Association and is intended to provide the minimum servicing standards with which an investor should expect a servicing entity to comply.

### © Update 4-1 Attestation: Clarification and Recodification

Statement on Standards for Attestation Engagements (SSAE) No. 18, Attestation Standards: Clarification and Recodification (AICPA, Professional Standards), issued in April 2016, is effective for practitioners' reports dated on or after May 1, 2017.

<sup>&</sup>lt;sup>3</sup> With the publication of the U.S. Department of Housing and Urban Development (HUD) Handbook 4000.1, effective September 2015, the Federal Housing Administration (FHA) implemented minor changes to the Quality Control Plan requirements. For fiscal year 2015, timeliness in implementing the 60 day requirements (previously, a 90 day requirement) during the last quarter of 2015 will not constitute material noncompliance. However, if the lender did not adhere to the post-closing requirements in place prior to the effective date of the handbook, the auditor should report this as a material noncompliance. Going forward into fiscal year 2016, lenders are expected to comply with all quality control requirements, and failure to do so should be reported as material noncompliance. Readers are reminded that the FHA Resource Center is available to answer handbook related questions. Additionally, the Lender Approval and Recertification division of HUD is available to address specific questions and concerns.

To address concerns over the clarity, length, and complexity of its standards, the Auditing Standards Board established clarity drafting conventions and undertook a project to redraft all the standards it issues in clarity format. The redrafting of SSAEs (or attestation standards) in SSAE No. 18 represents the culmination of that process. The attestation standards are developed and issued in the form of SSAEs and are codified into sections. SSAE No. 18 recodifies the "AT" section numbers designated by SSAE Nos. 10–17 using the identifier "AT-C" to differentiate the sections of the clarified attestation standards (AT-C sections) from the attestation standards that are superseded by SSAE No. 18 (AT sections). The AT sections in AICPA *Professional Standards* remain effective through April 2017, by which time substantially all engagements for which the AT sections were still effective are expected to be completed.

This edition of the guide has not been updated to reflect changes as a result of this SSAE, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this SSAE on the AICPA's website at www.aicpa.org.

**4.37** Regulation AB<sup>4</sup> codifies requirements for registration, disclosure, and reporting for all publicly registered ABS, including MBS. Regulation AB requires the issuance of an attestation report on assessment of compliance with servicing criteria for ABSs, establishes the required disclosures associated with the securities registration process, establishes the reporting requirements for ABSs, and necessitates an annual servicing assertion.

©2016, AICPA AAG-DEP 4.37

<sup>&</sup>lt;sup>4</sup> Readers are encouraged to view the full text of SEC Regulation AB, issued December 22, 2004, at www.sec.gov/rules/final/33-8518.htm. In addition, readers may refer to footnote 25 in paragraph 1.118 of this guide for further discussion on subsequent revisions to Regulation AB. Readers may also access the latest Regulation AB interpretations of the staff of the Division of Corporation Finance under the "Division of Corporation Finance—Compliance and Disclosures Interpretations" page at www.sec.gov.

### Chapter 5

# Audit Considerations and Certain Financial Reporting Matters<sup>1</sup>

Gray shaded text in this chapter reflects guidance issued but not yet effective as of the date of this guide, July 1, 2016, but becoming effective on or prior to December 31, 2016, exclusive of any option to early adopt ahead of the mandatory effective date. Unless otherwise indicated, all unshaded text reflects guidance that was already effective as of the date of this guide.

### Overview

**5.01** AU-C section 200, Overall Objectives of the Independent Auditor and the Conduct of an Audit in Accordance With Generally Accepted Auditing Standards (AICPA, Professional Standards), addresses the independent auditor's overall responsibilities when conducting an audit of financial statements in accordance with generally accepted auditing standards (GAAS). Specifically, it sets out the overall objectives of the independent auditor (the auditor) and explains the nature and scope of an audit designed to enable the auditor to meet those objectives. It also explains the scope, authority, and structure of GAAS and includes requirements establishing the general responsibilities of the auditor applicable in all audits, including the obligation to comply with GAAS.

**5.02** Paragraph .12 of AU-C section 200 states that the overall objectives of the auditor, in conducting an audit of financial statements, are to

- a. obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error, thereby enabling the auditor to express an opinion on whether the financial statements are presented fairly, in all material respects, in accordance with an applicable financial reporting framework; and
- b. report on the financial statements, and communicate as required by GAAS, in accordance with the auditor's findings.

**5.03** Depository and lending institutions are subject to certain risks as a result of the regulatory environment and the current economic climate in which these entities operate as well as the complex nature of these entities and the transactions in which these entities are engaged. This chapter provides guidance on the application of the auditor's overall objectives, including the risk assessment process and general auditing considerations for depository and lending institutions.

©2016, AICPA AAG-DEP 5.03

<sup>&</sup>lt;sup>1</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards (GAAS) issued by the Auditing Standards Board (ASB) and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of *issuers* and *nonissuers* as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

### An Audit of Financial Statements

**5.04** Consistent with the guidance presented in paragraph .04 of AU-C section 200, the purpose of an audit of a deposit and lending institution's financial statements is to provide financial statement users with an opinion by the auditor on whether the financial statements are presented fairly, in all material respects, in accordance with an applicable financial reporting framework, which enhances the degree of confidence that intended users can place in the financial statements. An audit conducted in accordance with GAAS and relevant ethical requirements enables the auditor to form that opinion. As the basis for the auditor's opinion, paragraph .06 of AU-C section 200 states that GAAS require the auditor to obtain reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error. Reasonable assurance is a high, but not absolute, level of assurance. It is obtained when the auditor has obtained sufficient appropriate audit evidence to reduce audit risk (for purposes of GAAS, that is, the risk that the auditor expresses an inappropriate opinion when the financial statements are materially misstated) to an acceptably low level.

**5.05** Paragraphs .08 and .10 of AU-C section 200 state that GAAS contain objectives, requirements, and application and other explanatory material that are designed to support the auditor in obtaining reasonable assurance. GAAS require that the auditor exercise professional judgment and maintain professional skepticism throughout the planning and performance of the audit and, among other things,

- identify and assess risks of material misstatement, whether due to fraud or error, based on an understanding of the entity and its environment, including the entity's internal control.
- obtain sufficient appropriate audit evidence about whether material misstatements exist, through designing and implementing appropriate responses to the assessed risks.
- form an opinion on the financial statements, or determine that an opinion cannot be formed, based on an evaluation of the audit evidence obtained.

The auditor also may have certain other communication and reporting responsibilities to users, management, those charged with governance, or parties outside the entity, regarding matters arising from the audit. These responsibilities may be established by GAAS or by applicable law or regulation.

Considerations for Audits Performed in Accordance With PCAOB  $Standards^2$ 

PCAOB Staff Audit Practice Alert No. 10, Maintaining and Applying Professional Skepticism in Audits (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance, sec. 400.10), reminds auditors of the requirement to appropriately apply professional skepticism throughout their audits, which includes an attitude of a questioning mind and a critical assessment of audit evidence. This practice alert highlights: (1) professional skepticism and due professional care; (2) impediments to the application of professional skepticism; (3) promoting professional skepticism via an appropriate system of quality

<sup>&</sup>lt;sup>2</sup> PCAOB Staff Audit Practice Alerts are not rules of the board and do not reflect any board determination or judgment about the conduct of any particular firm, auditor, or any other person.

control; (4) the importance of supervision to the application of professional skepticism; and (5) the appropriate application of professional skepticism.

### **Audit Risk**

- **5.06** Paragraph .A36 of AU-C section 200 explains that audit risk is a function of the risks of material misstatement and detection risk. The assessment of risks is based on audit procedures to obtain information necessary for that purpose and evidence obtained throughout the audit. The assessment of risks is a matter of professional judgment, rather than a matter capable of precise measurement.
- **5.07** Paragraphs .A38—.A40 of AU-C section 200 provide further explanation on the two levels of the risks of material misstatement. The risks of material misstatement exist at the overall financial statement level and the assertion level for classes of transactions, account balances, and disclosures. Risks of material misstatement at the overall financial statement level refer to risks of material misstatement that relate pervasively to the financial statements as a whole and potentially affect many assertions. Risks of material misstatement at the assertion level are assessed in order to determine the nature, timing, and extent of further audit procedures necessary to obtain sufficient appropriate audit evidence. This evidence enables the auditor to express an opinion on the financial statements at an acceptably low level of audit risk.
- **5.08** Paragraph .A44 of AU-C section 200 states that GAAS do not ordinarily refer to inherent risk and control risk separately but rather to a combined assessment of the risks of material misstatement. However, the auditor may make separate or combined assessments of inherent and control risk depending on preferred audit techniques or methodologies and practical considerations. The assessment of the risks of material misstatement may be expressed in quantitative terms, such as in percentages or in nonquantitative terms. In any case, the need for the auditor to make appropriate risk assessments is more important than the different approaches by which they may be made.
- **5.09** Paragraphs .A41–.A44 and .A46–.A47 of AU-C section 200 provide further guidance on the two components of the risk of material misstatement (inherent risk and control risk) and characteristics of detection risk.

### **Terms of Engagement**

- **5.10** The scope of services rendered by auditors generally depends on the types of reports to be issued as a result of the engagement. Paragraphs .09–.10 of AU-C section 210, *Terms of Engagement* (AICPA, *Professional Standards*), states that the auditor should agree upon the terms of the audit engagement with management or those charged with governance, as appropriate. The agreed-upon terms of the audit engagement should be documented in an audit engagement letter or other suitable form of written agreement (see paragraph .10 of AU-C section 210 for a listing of agreed-upon terms that should be included). Both management and the auditor have an interest in documenting the agreed-upon terms of the audit engagement before the commencement of the audit to help avoid misunderstandings with respect to the audit as stated in paragraph .A22 of AU-C section 210.
- **5.11** In accordance with paragraphs .A23-.A24 of AU-C section 210, the form and content of the audit engagement letter may vary for each entity.

©2016, AICPA AAG-DEP 5.11

When relevant, additional services to be provided, such as those relating to regulatory requirements (see further discussion on these engagements in the section "Annual Independent Audits and Reporting Requirements" beginning in paragraph 1.92 of this guide), could be included in the audit engagement letter. In addition, the engagement letter may also include any additional legal or contractual requirements, such as the following:

- Auditing the financial statements of common trust funds and applying agreed-upon procedures related to trust activities. (Chapter 21, "Trust and Asset Management Activities," of this guide includes a description of trust services and activities.)
- Reporting on management's assertions about compliance with the requirements of the Consolidated Audit Guide for Audits of HUD Programs, compliance with the minimum servicing standards set forth in the Uniform Single Attestation Program for Mortgage Bankers, and compliance with servicing criteria for asset-backed securities as required by Regulation AB. (See chapter 4, "Industry Overview—Mortgage Companies," of this guide)
- Applying minimum agreed-upon procedures to assist the supervisory committee in fulfilling its responsibilities. (The scope of services is expanded beyond the minimum procedures. See chapter 2, "Industry Overview—Credit Unions," and chapter 23, "Reporting Considerations," of this guide)
- Reporting on management's assertions about compliance with certain Department of Education requirements relative to student loan activities.<sup>3</sup> (See chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide.)
- Reporting on the controls at banks and savings institutions or credit unions functioning as service organizations in accordance with Statement on Standards for Attestation Engagements No. 16, Reporting on Controls at a Service Organization (AICPA, Professional Standards, AT sec. 801).<sup>4</sup> (See chapter 10, "Transfers and Servicing and Variable Interest Entities," chapter 13, "Deposits," and chapter 20, "Fair Value," of this guide as they relate to loan servicing, deposits, and trust activities, respectively.)

**5.12** In February 2006, the Office of the Comptroller of the Currency (OCC), the Board of Governors of the Federal Reserve System (Federal Reserve), the FDIC, the Office of Thrift Supervision (prior to its transfer of powers to the OCC, the Federal Reserve, and the FDIC),<sup>5</sup> and the National Credit Union Administration (NCUA) published the *Interagency Advisory on the Unsafe and Unsound Use of Limitation of Liability Provisions in External Audit* 

<sup>&</sup>lt;sup>3</sup> Readers are encouraged to visit the National Council of Higher Education Loan Program's website (www.nchelp.org/) for the most recent audit guide and related amendments, if applicable.

<sup>&</sup>lt;sup>4</sup> The AICPA Guide Service Organizations: Reporting on Controls at a Service Organization Relevant to User Entities' Internal Control Over Financial Reporting contains information for practitioners reporting on controls at a service organization that affect user entities' internal control over financial reporting. Also, the AICPA Guide Reporting on Controls at a Service Organization Relevant to Security, Availability, Processing Integrity, Confidentiality, or Privacy (SOC 2®) summarizes the three SOC engagements and provides detailed guidance on planning, performing, and reporting on SOC 2® engagements.

<sup>&</sup>lt;sup>5</sup> See chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide for further discussion on the Office of Thrift Supervision (OTS) transfer of powers.

Engagement Letters. The advisory was issued because the federal agencies had observed an increase in the types and frequency of provisions in financial institutions' external audit engagement letters limiting the auditor's liability. Examples of these provisions included, but were not limited to, indemnifying the external auditor against claims made by third parties, releasing the external auditor from liability for claims or potential claims that might be asserted by the client financial institution, or limiting the remedies available to the client financial institution. The federal agencies believe that when financial institutions agree to limit their external auditors' liability, either in provisions in engagement letters or in provisions that accompany alternative dispute resolution agreements, such provisions may weaken the external auditor's objectivity, impartiality, and performance. In this regard, the Professional Ethics Executive Committee issued Interpretation No. 501-8, "Failure to Follow Requirements of Governmental Bodies, Commissions, or Other Regulatory Agencies on Indemnification of Liability Provisions in Connection With Audit and Other Attest Services" (AICPA, Professional Standards, ET sec. 501 par. .09). This interpretation provides that including prohibited limitation of liability provisions in engagement letters is an act discreditable to the profession.

5.13 The advisory informs financial institutions' boards of directors, audit committees, and management that they should not enter into agreements that incorporate unsafe and unsound external auditor limitation of liability provisions with respect to engagements for financial statement audits, audits of internal control over financial reporting, and attestations on management's assessment of internal control over financial reporting. It applies to all audits of financial institutions, regardless of whether an institution is public or a non-public company. However, the advisory does not apply to non-audit services; audits of financial institutions' 401K plans, pension plans, and other similar audits; services performed by accountants who are not engaged to perform financial institutions' audits; and other service providers. Readers may access the full text of this advisory from any of the federal agencies' websites.

### **Audit Planning**

**5.14** AU-C section 300, *Planning an Audit* (AICPA, *Professional Standards*), addresses the auditor's responsibilities to plan an audit of financial statements. AU-C section 300 is written in the context of recurring audits. Matters related to planning audits of group financial statements are addressed in AU-C section 600, *Special Considerations—Audits of Group Financial Statements (Including the Work of Component Auditors)* (AICPA, *Professional Standards*). Planning activities involve performing preliminary engagement activities; establishing an overall audit strategy and communicating with those charged with governance an overview of the planned scope and timing of the audit; developing a detailed, written audit plan; determining direction and supervision of engagement team members and review of their work; and determining the extent of involvement of professionals with specialized skills. Adequate planning benefits the audit of financial statements in several ways, including the following:

- Helping the auditor identify and devote appropriate attention to important areas of the audit
- Helping the auditor identify and resolve potential problems on a timely basis

©2016. AICPA AAG-DEP 5.14

- Helping the auditor properly organize and manage the audit engagement so that it is performed in an effective and efficient manner
- Assisting in the selection of engagement team members with appropriate levels of capabilities and competence to respond to anticipated risks and allocating team member responsibilities
- Facilitating the direction and supervision of engagement team members and the review of their work
- Assisting, when applicable, in coordination of work done by auditors of components and specialists

Paragraph .A1 of AU-C section 300 further explains that the nature, timing, and extent of planning activities will vary according to the size and complexity of the entity, the key engagement team members' previous experience with the entity, and changes in circumstances that occur during the audit.

**5.15** In accordance with paragraph .09 of AU-C section 300, the auditor should develop an audit plan that includes a description of the nature and extent of planned risk assessment procedures, as determined under AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards) (see discussion of risk assessment procedures in paragraphs 5.23-.75); the nature, timing, and extent of planned further audit procedures at the relevant assertion level, as determined under AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards) (see discussion of planned further audit procedures in paragraphs 5.82-.98); and, other planned audit procedures that are required to be carried out so that the engagement complies with GAAS. Paragraph .A2 of AU-C section 300 explains that planning is not a discrete phase of an audit, but rather a continual and iterative process that often begins shortly after (or in connection with) the completion of the previous audit and continues until the completion of the current audit engagement.

### **Materiality**

**5.16** AU-C section 320, *Materiality in Planning and Performing an Audit* (AICPA, *Professional Standards*), addresses the auditor's responsibility to apply the concept of materiality in planning and performing an audit of financial statements. AU-C section 450, *Evaluation of Misstatements Identified During the Audit* (AICPA, *Professional Standards*), explains how materiality is applied in evaluating the effect of identified misstatements on the audit and the effect of uncorrected misstatements, if any, on the financial statements (see paragraphs 5.99–.101 for a discussion of evaluation of misstatements).

**5.17** Paragraphs .04 and .06 of AU-C section 320 state that the auditor's determination of materiality is a matter of professional judgment and is influenced by the auditor's perception of the financial information needs of users of financial statements. In planning the audit, the auditor makes judgments about the size of misstatements that will be considered material. Although it is not practicable to design audit procedures to detect misstatements that could be material solely because of their nature (that is, qualitative considerations), the auditor considers not only the size but also the nature of uncorrected misstatements, and the particular circumstances of their occurrence, when evaluating their effect on the financial statements.

**5.18** In accordance with paragraphs .10 and .A5 of AU-C section 320, the auditor should determine materiality for the financial statements as a whole when establishing the overall audit strategy. Determining materiality involves the exercise of professional judgment. A percentage is often applied to a chosen benchmark as a starting point in determining materiality for the financial statements as a whole. If, in the specific circumstances of the entity, one or more particular classes of transactions, account balance, or disclosures exist for which misstatements of lesser amounts than materiality for the financial statements as a whole could reasonably be expected to influence the economic decisions of users, then, taken on the basis of the financial statements, the auditor also should determine the materiality level or levels to be applied to those particular classes of transactions, account balances, or disclosures. See paragraphs .A12–.A13 of AU-C section 320 for further application guidance on materiality level or levels for particular classes of transactions, account balances, or disclosures.

### **Performance Materiality**

**5.19** Paragraph .A14 of AU-C section 320 explains that planning the audit solely to detect individual material misstatements overlooks the fact that the aggregate of individually immaterial misstatements may cause the financial statements to be materially misstated and leaves no margin for possible undetected misstatements. Therefore, in accordance with paragraph .11 of AU-C section 320, the auditor should determine performance materiality for purposes of assessing the risks of material misstatement and determining the nature, timing, and extent of further audit procedures. Performance materiality, for purposes of GAAS, is defined in AU-C section 320 as the amount or amounts set by the auditor at less than materiality for the financial statements as a whole to reduce to an appropriately low level the probability that the aggregate of uncorrected and undetected misstatements exceeds materiality for the financial statements as a whole. If applicable, performance materiality also refers to the amount or amounts set by the auditor at less than the materiality level or levels for particular classes of transactions, account balances, or disclosures. Performance materiality is to be distinguished from tolerable misstatement, which is the application of performance materiality to a particular sampling procedure.<sup>6</sup>

**5.20** Paragraph .A14 of AU-C section 320 goes on to explain that the determination of performance materiality is not a simple mechanical calculation and involves the exercise of professional judgment. It is affected by the auditor's understanding of the entity, updated during the performance of the risk assessment procedures, and the nature and extent of misstatements identified in previous audits and, thereby, the auditor's expectations regarding misstatements in the current period.

# Use of Assertions in Assessment of Risks of Material Misstatement

**5.21** Paragraphs .A113-.A118 of AU-C section 315 discuss the use of assertions in assessment of risks of material misstatement. In representing

©2016, AICPA AAG-DEP 5.21

<sup>&</sup>lt;sup>6</sup> AU-C section 530, *Audit Sampling* (AICPA, *Professional Standards*), defines tolerable misstatement and provides further application guidance about the concept.

that the financial statements are in accordance with the applicable financial reporting framework, management implicitly or explicitly makes assertions regarding the recognition, measurement, presentation, and disclosure of the various elements of financial statements and related disclosures. Assertions used by the auditor to consider the different types of potential misstatements that may occur fall into the following categories and may take the following forms.

### **Categories of Assertions**

	Description of Assertions			
	Classes of Transactions and Events During the Period	Account Balances at the End of the Period	Presentation and Disclosure	
Occurrence/ Existence	Transactions and events that have been recorded have occurred and pertain to the entity.	Assets, liabilities, and equity interests exist.	Disclosed events and transactions have occurred.	
Rights and Obligations	_	The entity holds or controls the rights to assets, and liabilities are the obligations of the entity.	Disclosed events and transactions pertain to the entity.	
Completeness	All transactions and events that should have been recorded have been recorded.	All assets, liabilities, and equity interests that should have been recorded have been recorded.	All disclosures that should have been included in the financial statements have been included.	
Accuracy/ Valuation and Allocation	Amounts and other data relating to recorded transactions and events have been recorded appropriately.	Assets, liabilities, and equity interests are included in the financial statements at appropriate amounts and any resulting valuation or allocation adjustments are recorded appropriately.	Financial and other information is disclosed fairly and at appropriate amounts.	
Cut-off	Transactions and events have been recorded in the correct accounting period.	_	_	
Classification and Under- standability	Transactions and events have been recorded in the proper accounts.	_	Financial information is appropriately presented and described and information in disclosures is expressed clearly.	

**5.22** According to paragraph .A116 of AU-C section 315, the auditor should use relevant assertions for classes of transactions, account balances, and disclosures in sufficient detail to form a basis for the assessment of risks of material misstatement and the design and performance of further audit procedures. The auditor should use relevant assertions in assessing risks by relating the identified risks to what can go wrong at the relevant assertion, taking account of relevant controls that the auditor intends to test, and designing further audit procedures that are responsive to the assessed risks.

### **Risk Assessment Procedures**

- **5.23** AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control.
- **5.24** Obtaining an understanding of the entity and its environment, including the entity's internal control (referred to hereafter as an *understanding of the entity*), is a continuous, dynamic process of gathering, updating, and analyzing information throughout the audit. As stated in paragraph .A1 of AU-C section 315, the understanding of the entity establishes a frame of reference within which the auditor plans the audit and exercises professional judgment throughout the audit when, for example
  - assessing risks of material misstatement of the financial statements:
  - determining materiality in accordance with AU-C section 320;
  - considering the appropriateness of the selection and application of accounting policies and the adequacy of financial statement disclosures;
  - identifying areas for which special audit consideration may be necessary (for example, related party transactions, the appropriateness of management's use of the going concern assumption, considering the business purpose of transactions, or the existence of complex and unusual transactions);
  - developing expectations for use when performing analytical procedures;
  - responding to the assessed risks of material misstatement, including designing and performing further audit procedures to obtain sufficient appropriate audit evidence; and
  - evaluating the sufficiency and appropriateness of audit evidence obtained, such as the appropriateness of assumptions and management's oral and written representations.

### Risk Assessment Procedures and Related Activities

**5.25** In accordance with paragraph .05 of AU-C section 315, the auditor should perform risk assessment procedures to provide a basis for the identification and assessment of risks of material misstatement at the financial statement and relevant assertion levels. Risk assessment procedures by themselves, however, do not provide sufficient appropriate audit evidence on which to base the audit opinion. For purposes of GAAS, *risk assessment procedures* are defined in AU-C section 315 as audit procedures performed to obtain an

©2016, AICPA AAG-DEP 5.25

understanding of the entity and its environment, including the entity's internal control, to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels.

- **5.26** The auditor is required to exercise professional judgment<sup>7</sup> to determine the extent of the required understanding of the entity. Paragraph .A3 of AU-C section 315 states that the auditor's primary consideration is whether the understanding of the entity that has been obtained is sufficient to meet the objectives of AU-C section 315. The depth of the overall understanding that is required by the auditor is less than that possessed by management in managing the entity.
- **5.27** Paragraph .06 of AU-C section 315 states that the risk assessment procedures should include the following:
  - Inquiries of management, appropriate individuals within the internal audit function (if such function exists), and others within the entity who, in the auditor's professional judgment, may have information that is likely to assist in identifying risks of material misstatement due to fraud or error
  - Analytical procedures
  - Observation and inspection

### **Analytical Procedures**

- **5.28** Paragraphs .A7-.A10 of AU-C section 315 provide additional explanation for analytical procedures performed during the risk assessment process. Analytical procedures performed as risk assessment procedures may identify aspects of the entity of which the auditor was unaware and may assist in assessing the risks of material misstatement in order to provide a basis for designing and implementing responses to the assessed risks. Analytical procedures may enhance the auditor's understanding of the institution's business and the significant transactions and events that have occurred since the prior audit and help to identify the existence of unusual transactions or events and amounts, ratios, and trends that might indicate matters that have audit implications.
- **5.29** Ratios, operating statistics, and other analytical information that may be useful in assessing an institution's position relative to other similar institutions and to industry norms, as well as in identifying unusual relationships between data about the institution itself, are generally readily available. Ratios and statistics developed for use by management or regulators often can be effectively used by the auditor in performing analytical procedures for risk assessment purposes. Many institutions disclose analytical information in their annual and quarterly reports. Other sources of information that may be useful for risk assessment purposes are the institution's Call Reports and the disclosures made by publicly held institutions in accordance with the SEC's Industry Guide No. 3, Statistical Disclosures by Bank Holding Companies. The Uniform Bank Performance Reports, published by the Federal Financial Institutions Examination Council (FFIEC), and various reports published by the

<sup>&</sup>lt;sup>7</sup> Paragraph .18 of AU-C section 200, Overall Objectives of the Independent Auditor and the Conduct of an Audit in Accordance With Generally Accepted Auditing Standards (AICPA, Professional Standards), requires the auditor to exercise professional judgment in planning and performing an audit.

FDIC contain industry data and statistics. There are also several sources of industry data published by private companies. Many of these reports use a peer group format. It is important to understand the relevance of any peer group data to the client institution before making any judgments.

- **5.30** A number of the ratios that may be useful to the auditor in an audit of the financial statements of an institution are listed here with a brief description of the information they provide:
  - Investments to total assets. Measures the mix of earning assets
  - *Loans to total assets*. Measures the mix of earning assets
  - Investments by type divided by total investments. Measures the composition of investment portfolio
  - Loans to deposits. Indicates the funding sources for the loan base
  - Loans by type to total loans. Measures the composition of loan portfolio and of lending strategy and risk
  - Allowance for loan losses to total loans. Measures loan portfolio credit risk coverage
  - Loan loss recoveries to prior-year write-offs. Indicates write-off policy and measure recovery experience
  - Classified loans to total loans. Indicates asset quality
  - Investment income to average total securities. Measures investment portfolio yield
  - Allowance for loan losses to classified loans. Measures management's estimate of losses
  - Loan income to average net loans. Measures loan portfolio yield
  - Total deposit interest expense to average total deposits. Measures costs of deposit funds
  - Overhead to total revenue (net interest income plus noninterest income). Measures operating efficiency
  - *Net income to average total assets.* Measures return on assets
  - *Net income to average capital.* Measures return on equity
  - Capital ratios. Measures financial strength and regulatory compliance
  - Noninterest income to total revenue (net interest income plus noninterest income). Measures the extent of noninterest income
  - Liabilities to shareholders' equity. Measures the extent equity can cover creditors' claims in the event of liquidation

### Discussion Among the Engagement Team

**5.31** In accordance with paragraph .11 of AU-C section 315, the engagement partner and other key engagement team members should discuss the susceptibility of the entity's financial statements to material misstatement and the application of the applicable financial reporting framework to the entity's

©2016, AICPA AAG-DEP 5.31

facts and circumstances. The engagement partner should determine which matters are to be communicated to engagement team members not involved in the discussion. Paragraph .A14 of AU-C section 315 states this discussion may be held concurrently with the discussion among the engagement team that is required by AU-C section 240, Consideration of Fraud in a Financial Statement Audit (AICPA, Professional Standards), to discuss the susceptibility of the entity's financial statements to fraud. Paragraphs 5.129–.132 further address the discussion among the engagement team about the risks of fraud.

### Additional Guidance

**5.32** In addition to the requirements discussed previously, paragraphs .07–.10 of AU-C section 315 address additional requirements on risk assessment procedures and related activities. Additional application and explanatory material regarding risk assessment requirements can be found in paragraphs .A1–.A16 of AU-C section 315.

# Understanding the Entity and Its Environment, Including the Entity's Internal Control

- **5.33** Paragraph .12 of AU-C section 315 states that the auditor should obtain an understanding of the following:
  - a. Relevant industry, regulatory, and other external factors, including the applicable financial reporting framework.
  - b. The nature of the entity, including
    - i. its operations;
    - ii. its ownership and governance structures;
    - iii. the types of investments that the entity is making and plans to make, including investments in entities formed to accomplish specific objectives; and
    - iv. the way that the entity is structured and how it is financed, to enable the auditor to understand the classes of transactions, account balances, and disclosures to be expected in the financial statements.
  - c. The entity's selection and application of accounting policies, including the reasons for changes thereto. The auditor should evaluate whether the entity's accounting policies are appropriate for its business and consistent with the applicable financial reporting framework and accounting policies used in the relevant industry.
  - d. The entity's objectives and strategies and those related business risks that may result in risks of material misstatement.
  - e. The measurement and review of the entity's financial performance.

Appendix A, "Understanding the Entity and Its Environment," of AU-C section 315 contains examples of matters that the auditor may consider in obtaining an understanding of the entity and its environment. Appendix B, "Internal Control Components," of AU-C section 315 contains a detailed explanation of the internal control components.

### Understanding of the Client's Business

**5.34** As previously discussed in paragraph 5.33, in addition to an understanding of the industry, including matters such as those described in chapter 1,

chapter 2, chapter 3, "Industry Overview—Finance Companies," and chapter 4 of this guide, the auditor should obtain an understanding of the nature of an entity and the entity's objectives and strategies and those related business risks that may result in risks of material misstatement. With regard to financial institutions, such matters include risk management strategies, organizational structure, product lines and services, capital structure, locations, and other operating characteristics. Paragraph .A32 of AU-C section 315 identifies examples of matters that the auditor may consider when obtaining an understanding of the entity's objectives, strategies, and related business risks that may result in a risk of material misstatement of the financial statements. For issuers, the auditor should also obtain an understanding of the operating segments of the business, as defined by FASB Accounting Standards Codification (ASC) 280-10-50.

**5.35** An understanding of the entity may also be obtained or supplemented by reading documents such as the following:

- The charter and bylaws of the institution
- Minutes of meetings of the board of directors, audit committee, credit committee or loan officers, or both, and other appropriate committees
- Prior-year and interim financial statements and other relevant reports, such as recently issued registration statements
- Risk management strategies and reports, such as interest rate, asset quality, and liquidity reports
- Organizational charts
- Operating policies, including strategies for lending and investing
- Regulatory examination reports
- Correspondence with regulators
- Periodic regulatory financial reports: FFIEC Consolidated Reports of Condition and Income or NCUA Call Reports (collectively, Call Reports)
- Sales brochures and other marketing materials
- Capital or business plans
- Internal reports and financial information utilized by management to make segment-related decisions
- Significant or unusual contracts entered into by the entity
- **5.36** Related parties. Obtaining an understanding of a client's business should also include performing the procedures set forth in AU-C section 550, Related Parties (AICPA, Professional Standards), to determine the existence of related-party relationships and transactions with such parties. The FASB ASC glossary defines related parties as
  - a. affiliates of the institution (according to the FASB ASC glossary, an affiliated entity is an entity that directly or indirectly controls, is controlled by, or is under common control with another entity; also, a party with which the entity may deal if one party has the ability to exercise significant influence over the other's operating and financial policies as discussed in FASB ASC 323-10-15);

©2016. AICPA AAG-DEP 5.36

- b. entities for which investments would be required, absent the election of the fair value option under the "Fair Value Option" subsections of FASB ASC 825-10-15, to be accounted for by the equity method by the institution;
- c. trusts for the benefit of employees, such as pension and profitsharing trusts, that are managed by or are under the trusteeship of management of the institution;
- d. principal owners of the institution and members of their immediate families;
- e. management of the institution and members of their immediate families:
- f. other parties with which the institution may deal if one party controls or can significantly influence the management or operating policies of the other to an extent that one of the transacting parties might be prevented from fully pursuing its own separate interests; and
- g. other parties that can significantly influence the management or operating policies of the transacting parties or that have an ownership interest in one of the transacting parties and can significantly influence the other to an extent that one or more of the transacting parties might be prevented from fully pursuing its own separate interests

**5.37** Paragraph .A2 of AU-C section 550 states that the substance of a particular transaction may be significantly different from its form. Accordingly, financial statements prepared in accordance with U.S. generally accepted accounting principles (GAAP) generally recognize the substance of particular transactions rather than merely their legal form. Paragraph .A45 of AU-C section 550 explains that it will generally not be possible to determine whether a particular transaction would have taken place if the parties had not been related, or assuming it would have taken place, what the terms and manner of settlement would have been. Accordingly, it is difficult to substantiate representations that a transaction was consummated on terms equivalent to those that prevail in arm's length transactions. Paragraphs .A47 and .A49 of AU-C section 550 further state that the preparation and fair presentation of the financial statements requires management to substantiate an assertion included in financial statements that a related party transaction was conducted on terms equivalent to those prevailing in an arm's length transaction. If the auditor believes that management's assertions are unsubstantiated or the auditor cannot obtain sufficient appropriate audit evidence to support the assertions, the auditor, in accordance with AU-C section 705, Modifications to the Opinion in the Independent Auditor's Report (AICPA, Professional Standards), considers the implications for the audit, including the opinion in the auditor's report. AU-C section 705 addresses the auditor's responsibility to issue an appropriate report in circumstances when, in forming an opinion in accordance with

<sup>&</sup>lt;sup>8</sup> FASB Accounting Standards Codification (ASC) 850-10-50-5 states that if representations are made about transactions with related parties, the representations should not imply that the related party transactions were consummated on terms equivalent to those that prevail in arm's length transactions unless such representations can be substantiated.

AU-C section 700, Forming an Opinion and Reporting on Financial Statements (AICPA, Professional Standards), the auditor concludes that a modification to the auditor's opinion on the financial statements is necessary. Chapter 23 of this guide provides additional discussion on auditor reports.

- **5.38** Regulation O loans. Part 215, "Loans to Executive Officers, Directors, and Principal Shareholders of Member Banks," of the U.S. Code of Federal Regulations (CFR), commonly referred to as Regulation O, governs any extension of credit made by a member bank to an executive officer, director, or principal shareholder of the member bank, of any company of which the member bank is a subsidiary, and of any other subsidiary of that company. It also applies to any extension of credit made by a member bank to a company controlled by such a person, or to a political or campaign committee that benefits or is controlled by such a person. In general, Part 215.4 states that no member bank may extend credit to any insider of the bank or insider of its affiliates unless the extension of credit
  - is made on substantially the same terms (including interest rates and collateral) as, and following credit underwriting procedures that are not less stringent than, those prevailing at the time for comparable transactions by the bank with other persons that are not covered by this part and who are not employed by the bank and
  - does not involve more than the normal risk of repayment or present other unfavorable features.
- **5.39** Management of a financial institution would generally be expected to be able to support that their related party loans were conducted on terms equivalent to those prevailing in an arm's length transaction. In instances where a bank has made such a related party loan, the auditor should perform procedures to verify this assertion, including reviewing management's documentation as well as the regulatory examination report, which would identify instances where there are possible Regulation O violations.

## **Industry Risk Factors**

**5.40** As previously discussed in paragraph 5.33a, auditors should obtain an understanding of the relevant industry risk factors as a part of the evaluation of the entity and its environment. No list of risk factors covers all of the complex characteristics that affect transactions in the industry. However, some of those risk factors are competition for business, innovations in financial instruments, and the role of regulatory policy. Emerging regulatory and accounting guidance is discussed throughout this guide. Other primary risk factors (discussion to follow) involve the sensitivity of an institution's earnings to changes in interest rates, liquidity, asset quality, fiduciary, and processing risk. Auditors should obtain an understanding of such risk factors when planning the audit of an institution's financial statements. Practical considerations of these risk factors for certain transactions are provided in each chapter where appropriate.

©2016, AICPA AAG-DEP 5.40

<sup>&</sup>lt;sup>9</sup> An important source of such information is the AICPA's Audit Risk Alert series.

- **5.41** *Interest rate risk (IRR)*. <sup>10,11</sup> In general, financial institutions derive their income primarily from the excess of interest collected over interest paid. The rates of interest an institution earns on its assets and owes on its liabilities generally are established contractually for a period of time. Market interest rates change over time. Accordingly, an institution is exposed to lower profit margins (or losses) if it cannot adapt to interest rate changes.
- **5.42** For example, assume an institution's assets carry intermediate or long term fixed rates. Assume those assets were funded with short term liabilities. Also assume that interest rates rise by the time the short term liabilities are refinanced. The increase in the institution's interest expense on the new liabilities—which carry new, higher rates—will not be offset if assets continue to earn at the long term fixed rates. Accordingly, the institution's profits would decrease on the transaction because the institution will either have lower net interest income or, possibly, net interest expense. Similar risks exist if assets are subject to contractual interest rate ceilings, or rate sensitive assets are funded by longer term, fixed rate liabilities in a decreasing rate environment.

<sup>10</sup> The Federal Financial Institutions Examination Council's (FFIEC's) Advisory on Interest Rate Risk Management, issued in January 2010, reminds institutions of supervisory expectations for sound practices to manage interest rate risk (IRR). This advisory reiterates the importance of effective corporate governance, policies and procedures, risk measuring and monitoring systems, stress testing, and internal control related to the IRR exposures of depository institutions. It also clarifies elements of existing guidance and describes some IRR management techniques used by effective risk managers. For the complete text of the advisory see the FFIEC website at www.ffiec.gov.

In January 2012, the Office of the Comptroller of the Currency (OCČ), the Board of Governors of the Federal Reserve System (Federal Reserve), the FDIC, the National Credit Union Administration (NCUA), and State Liaison Committee issued the Interagency Advisory on Interest Rate Risk Management Frequently Asked Questions. This document was created to clarify points in the 2010 interagency Advisory on Interest Rate Risk Management by providing responses to the most common questions. The responses address IRR exposure measurement and reporting, model risk management, stress testing, assumption development, and model and systems validation. Financial institution management should consider the responses in the context of their institution's complexity, risk profile, business model, and scope of operations. Readers can access the guidance from any of the respective agencies' websites.

- <sup>11</sup> Due to changes in balance sheet composition for federally insured credit unions (FICUs) and increased uncertainty in the financial market, there is a heightened importance for FICUs to incorporate strong policies and programs addressing the credit union's management of controls for IRR. In January 2012, the NCUA issued a final rule, which became effective September 30, 2012, requiring FICUs to develop and adopt a written policy on IRR management and a program to effectively implement that policy as part of credit unions' asset liability management responsibilities. There are asset size and activity triggers for how the written IRR policy requirements would apply based on the following guidelines:
  - A FICU with assets of more than \$50 million must adopt a written IRR policy and implement an effective IRR program.
  - A FICU with assets of \$10 million or more but not greater than \$50 million must adopt a
    written IRR policy and implement an effective IRR program if the total of first mortgage
    loans it holds combined with total investments with maturities greater than 5 years, as
    reported by the FICU on its most recent call report, is equal to or greater than 100 percent
    of its net worth.
  - A FICU with assets less than \$10 million is not required to comply regardless of the amount
    of first mortgage loans and total investments with maturities greater than 5 years it holds.

The final rule provides discussion on the roles and responsibilities of the FICU's board of directors and management in establishing and implementing the IRR policy and program; risk management systems, methods, and valuation measures; internal control; decision making informed by IRR measurement systems; and guidelines addressing the adequacy and effectiveness of the policy and program.

The rule also provides additional guidance for large credit unions with complex or high risk balance sheets. *Large credit unions* are defined as institutions with assets of at least \$500 million. Readers can access this final rule from the NCUA website at www.ncua.gov. Readers are also encouraged to access NCUA Letters to Credit Unions No. 12-CU-11, *Interest Rate Risk Policy and Program Frequently Asked Questions*.

- **5.43** Several techniques might be used by an institution to minimize interest-rate risk. One approach is for the institution to continually analyze and manage assets and liabilities based on their payment streams and interest rates, the timing of their maturities, and their sensitivity to actual or potential changes in market interest rates. Such activities fall under the broad definition of asset/liability management.
- **5.44** One technique used in asset/liability management is measurement of an institution's asset/liability gap—that is, the difference between the cash flow amounts of interest-sensitive assets and liabilities that will be refinanced (or repriced) during a given period. For example, if the asset amount to be repriced exceeds the corresponding liability amount for a certain day, month, year, or longer period, the institution is in an asset-sensitive gap position. In this situation, net interest income would increase if market interest rates rose and decrease if market interest rates fell. If, alternatively, more liabilities than assets will reprice, the institution is in a liability-sensitive position. Accordingly, net interest income would decline when rates rose and increase when rates fell. Such gap analysis assumes that assets and liabilities will be repriced only when they mature—it does not consider opportunities to reprice principal or interest cash flows before maturity. Also, these examples assume that interest rate changes for assets and liabilities are of the same magnitude, whereas actual interest rate changes generally differ in magnitude for assets and liabilities.
- **5.45** Duration analysis is a technique that builds on gap analysis by adding consideration of the average life of a stream of cash flows. The duration of an asset or liability is measured by weighting cash flow amounts based on their timing. Accordingly, duration analysis adds a measure of the effect of the timing of interest rate changes on earnings.
- **5.46** Another technique used to analyze IRR involves simulation models. These models measure the effect of changes in interest rates on either net interest income or on the economic value of equity. Net interest income models measure the sensitivity of changes in net interest income as a result of different interest rate scenarios. The economic value of equity measures the difference in the market value of an institution's financial assets, liabilities, and off-balance-sheet instruments as a result of change in the interest rate environment. Simulation analysis involves the projection of various interest rate scenarios over future periods. To determine market value, the estimated cash flows for each rate scenario are discounted to arrive at a present value calculation for each rate scenario. The resulting range of probable risk exposures reflects both current and expected IRR. The rate scenarios often reflect variations of factors such as the mix of assets and liabilities and related pricing strategies. As with gap and duration analyses, if the assumptions are not valid, the results may not provide an accurate reflection of the institution's IRR.
  - **5.47** Several ways an institution can affect IRR includes the following:
    - Selling existing assets or repaying certain liabilities
    - Matching repricing periods for new assets and liabilities—for example, by shortening terms of new loans or investments
    - Hedging existing assets, liabilities, firm commitments, or forecasted transactions
- **5.48** An institution might also invest in more complex financial instruments intended to hedge or otherwise change IRR. Interest rate swaps, futures

©2016, AICPA AAG-DEP 5.48

contracts, options on futures, and other such derivative instruments often are used for this purpose. Because these instruments are sensitive to interest rate changes, they generally require management expertise to be effective. Accounting and regulatory guidance for these instruments continue to evolve. Chapter 18, "Derivative Instruments: Futures, Forwards, Options, Swaps, and Other Derivative Instruments," of this guide discusses specific accounting and regulatory guidance in this area, as well as related audit considerations.

- **5.49** Financial institutions are subject to a related risk—prepayment risk—in falling rate environments. For example, mortgage loans and other receivables may be prepaid by a debtor so that the debtor may refund its obligations at new, lower rates. Prepayments of assets carrying the old, higher rates reduce the institution's interest income and overall asset yields. Prepayment risk is discussed further in chapter 7, "Investments in Debt and Equity Securities," of this guide.
- **5.50** Liquidity risk. <sup>12,13,14</sup> A large portion of an institution's liabilities may be short term or due on demand, although most of its assets may be invested in long term loans or investments. Accordingly, the institution needs to have in place sources of cash to meet short term demands. These funds can be obtained in cash markets, by borrowing, or by selling assets. Also, the secondary mortgage, repurchase agreement, and Euro-markets have become increasingly important sources of liquidity for banks and savings institutions. However, if an institution resorts to sales of assets or loans to obtain liquidity, immediate losses will be incurred when the effective rates those assets carry are below market rates at the time of sale. Related audit considerations are addressed in chapter 7 of this guide.

<sup>&</sup>lt;sup>12</sup> In March 2010, the OCC, the OTS, the FDIC, the Federal Reserve, and the NCUA issued Interagency Policy Statement on Funding and Liquidity Risk Management to provide sound practices for managing funding and liquidity risk and strengthening liquidity risk management practices. The policy statement emphasizes the importance of cash flow projections, diversified funding sources, stress testing, a cushion of liquid assets and a formal, well-developed contingency funding plan as primary tools for measuring and managing liquidity risk. The agencies expect each financial institution to manage funding and liquidity risk using processes and systems that are commensurate with the institution's complexity, risk profile and scope of operations. This guidance can be found in Federal Register Vol. 75, No. 54 [22 March 2010], pp. 13656–13666.

<sup>&</sup>lt;sup>13</sup> In October 2013, the NCUA issued a final rule, Liquidity and Contingency Funding Plans, requiring FICUs with less than \$50 million in assets to maintain a basic written policy that provides a credit union board-approved framework for managing liquidity and a list of contingent liquidity sources that can be employed under adverse circumstances. The rule requires FICUs with assets of \$50 million or more to have a contingency funding plan that clearly sets out strategies for addressing liquidity shortfalls in emergency situations. Finally, the rule requires FICUS with assets of \$250 million or more to have access to a backup federal liquidity source for emergency situations. The guidance can be found in Federal Register Vol. 78, No. 210 [30 October 2013], pp. 64879–64883.

<sup>&</sup>lt;sup>14</sup> In September 2014, the OCC, the Federal Reserve, and the FDIC issued a final rule implementing a quantitative liquidity requirement consistent with the liquidity coverage ratio standard established by the Basel Committee on Banking Supervision. The final rule requires a company subject to the rule to maintain an amount of high-quality liquid assets (the numerator of the ratio) that is no less than 100 percent of its total net cash outflows over a prospective 30 calendar day period (the denominator of the ratio). The final rule applies to large and internationally active banking organizations, generally, bank holding companies, certain savings and loan holding companies, and depository institutions with \$250 billion or more in total assets or \$10 billion or more in on-balance sheet foreign exposure and to their consolidated subsidiaries that are depository institutions with \$10 billion or more in total consolidated assets. The final rule focuses on these financial institutions because of their complexity, funding profiles, and potential risk to the financial system. Therefore, the agencies do not intend to apply the final rule to community banks. The final rule became effective January 1, 2015, with transition periods for compliance with the requirements of the rule. The final rule can be accessed from any of the respective agencies' websites.

- **5.51** The composition of an institution's deposits also affects liquidity and IRR because large volumes of deposits can be withdrawn over a short period of time. For example, institutions are also subject to reputation risk. If an institution receives adverse publicity, it may have difficulty retaining deposits and, therefore, become dependent on other forms of borrowing at a higher cost of funds. (Chapter 13 of this guide addresses audit considerations for deposits.)
- **5.52** Asset-quality risk. Financial institutions have generally suffered their most severe losses as a result of the loss of expected cash flows due to loan defaults and inadequate collateral. For example, significant credit losses on real estate loans have occurred, due largely to downturns in regional and national real estate markets, but also because of other general economic conditions and higher-risk lending activities. Chapter 9, "Credit Losses," of this guide addresses credit losses.
- **5.53** Other financial assets are subject to other impairment issues—similar to credit quality—that involve subjective determinations. For example, increased prepayments of principal during periods of falling interest rates have a significant impact on the economic value of assets such as mortgage servicing rights.
- **5.54** Auditors who audit financial statements of financial institutions should give particular attention to the assessment of impairment of financial assets. The auditor should focus on the methods used, assumptions made, and conclusions reached by management (and outside specialists relied on by management, such as appraisers) in assessing impairment of financial assets. Practical guidance is provided in subsequent chapters.
- **5.55** Fiduciary risk. Many financial institutions activities involve custody of financial assets, management of such assets, or both. Fiduciary responsibilities are the focus of activities such as servicing the collateral behind assetbacked securities, managing mutual funds, and administering trusts. These activities expose the institution to the risk of loss arising from failure to properly process transactions or handle the related assets on behalf of third parties. Related audit considerations are addressed in subsequent chapters.
- **5.56** Processing risk. Large volumes of transactions must be processed by most financial institutions, generally over short periods of time. Demands placed on both computerized and manual systems can be great. These demands increase the risk that the accuracy and timeliness of related information could be impaired.
- **5.57** Financial institutions utilize information systems to process large volumes of transactions (for example, arising from banks' electronic funds transfer and check processing operations) on an accurate and timely basis. Related considerations are discussed in subsequent chapters.

## The Entity's Internal Control<sup>15,16</sup>

**5.58** As explained in paragraph .A44 of AU-C section 315, the way in which internal control is designed, implemented, and maintained varies with

©2016, AICPA AAG-DEP 5.58

<sup>&</sup>lt;sup>15</sup> The AICPA's Technical Questions and Answers (Q&A) section 8200, Internal Control (AICPA, Technical Questions and Answers), provides nonauthoritative guidance to auditors. For more information, visit the AICPA website at www.aicpa.org.

<sup>&</sup>lt;sup>16</sup> This section discusses the consideration of internal control in a financial statement audit; it does not address reporting on a written management assertion about internal control over financial reporting.

an entity's size and complexity. The assets of financial institutions generally are more negotiable and more liquid than those of other entities. As a result, they may be subject to greater risk of loss. In addition, the operations of financial institutions are characterized by a high volume of transactions; as a result, the effectiveness of internal control is a significant audit consideration.

- **5.59** Paragraphs .13—.14 of AU-C section 315 states that the auditor should obtain an understanding of internal control relevant to the audit. Although most controls relevant to the audit are likely to relate to financial reporting, not all controls that relate to financial reporting are relevant to the audit. It is a matter of the auditor's professional judgment whether a control, individually or in combination with others, is relevant to the audit. When obtaining an understanding of controls that are relevant to the audit, the auditor should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. Paragraph .A42 of AU-C section 315 further explains that an understanding of internal control assists the auditor in identifying types of potential misstatements and factors that affect the risks of material misstatement and in designing the nature, timing, and extent of further audit procedures.
- **5.60** Purpose of internal control. Paragraph .A44 of AU-C section 315 explains that internal control is designed, implemented, and maintained to address identified business risks that threaten the achievement of any of the entity's objectives that concern (a) the reliability of the entity's financial reporting, (b) the effectiveness and efficiency of its operations, and (c) its compliance with applicable laws and regulations.
- **5.61** *Division of internal control.* For purposes of GAAS, internal control is divided into the following five components:
  - a. Control environment sets the tone of an institution, influencing the control consciousness of its people. It is the foundation for all other components of internal control, providing discipline and structure.
  - b. Risk assessment is the institution's identification, analysis, and management of risks relevant to the preparation and fair presentation of financial statements.
  - c. Information system, including the related business processes relevant to financial reporting and communication consists of the procedures and records designed and established to
    - i. initiate, authorize, record, process, and report entity transactions (as well as events and conditions) and maintain accountability for the related assets, liabilities, and equity;
    - resolve incorrect processing of transactions (for example, automated suspense files and procedures followed to clear suspense items out on a timely basis);
    - iii. process and account for system overrides or bypasses to controls;
    - iv. transfer information from transaction processing systems to the general ledger;
    - v. capture information relevant to financial reporting for events and conditions other than transactions, such as

- the depreciation and amortization of assets and changes in the recoverability of accounts receivables; and
- vi. ensure information required to be disclosed by the applicable financial reporting framework is accumulated, recorded, processed, summarized, and appropriately reported in the financial statements.
- d. Control activities are the policies and procedures that help ensure management directives are carried out.
- e. Monitoring is a process that assesses the quality of internal control performance over time.

Audit requirements and application guidance related to the preceding components can be found in paragraphs .15–.25 and .A71–.A107, respectively, of AU-C section 315.

- **5.62** Controls relevant to the audit. Paragraphs .A61–.A62 of AU-C section 315 state a direct relationship exists between an entity's objectives and the controls it implements to provide reasonable assurance about their achievement. The entity's objectives and, therefore, controls relate to financial reporting, operations, and compliance; however, not all of these objectives and controls are relevant to the auditor's risk assessment. Factors relevant to the auditor's professional judgment about whether a control, individually or in combination with others, is relevant to the audit may include such matters as the following:
  - Materiality
  - The significance of the related risk
  - The institution's size
  - The nature of the institution's business, including its organization and ownership characteristics
  - The diversity and complexity of the institution's operations
  - Applicable legal and regulatory requirements
  - The circumstances and the applicable component of internal control
  - The nature and complexity of the systems that are part of the institution's internal control, including the use of service organizations
  - Whether and how a specific control, individually or in combination with other controls, prevents, or detects and corrects, material misstatements
- **5.63** Paragraph .A64 of AU-C section 315 states that the controls relating to operations and compliance objectives also may be relevant to an audit if they relate to data the auditor evaluates or uses in applying audit procedures. For example, controls pertaining to nonfinancial data that the auditor may use in analytical procedures, such as production statistics, or controls pertaining to detecting noncompliance with laws and regulations that may have a direct effect on the determination of material amounts and disclosures in the financial statements, such as compliance with income tax laws and regulations used to determine the income tax provision, may be relevant to an audit.
- **5.64** *IT considerations*. Financial institutions' operations are characterized by large volumes of transactions and, therefore, generally rely heavily on computers. AU-C section 315 establish standards and provide guidance for auditors who have been engaged to audit an entity's financial statements when

©2016, AICPA AAG-DEP 5.64

significant information is transmitted, processed, maintained, or accessed electronically.

Considerations for Audits Performed in Accordance With PCAOB Standards<sup>17</sup>

PCAOB Staff Audit Practice Alert No. 11, Considerations for Audits of Internal Control Over Financial Reporting (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance, sec. 400.11), highlights certain requirements of the auditing standards of the PCAOB in aspects of audits of internal control over financial reporting in which significant auditing deficiencies have been cited frequently in PCAOB inspection reports. Among other topics, the alert specifically addresses PCAOB standards regarding the consideration of IT in audits of internal control, including when testing controls that use system-generated data and reports and evaluating deficiencies in IT general controls.

**5.65** Paragraph .A54 of AU-C section 315 states that an entity's use of IT may affect any of the five components of internal control relevant to the achievement of the entity's financial reporting, operations, or compliance objectives, and its operating units or business functions. The auditor might consider matters such as

- the extent that information technology is used for significant accounting applications;
- the complexity of the institution's information technology, including whether outside service organizations are used;
- the organizational structure for information technology, including the extent that online terminals and networks are used;
- the physical security controls over computer equipment:
- controls over information technology (for example, program changes and access to data files), operations, and systems;
- the availability of data; and
- the use of information technology assisted audit techniques to increase the efficiency and effectiveness of performing procedures. (Using information technology assisted audit techniques may also provide the auditor with an opportunity to apply certain procedures to an entire population of accounts or transactions. In addition, in some accounting systems, it may be difficult or impossible for the auditor to analyze certain data or test specific control procedures without information technology assistance.)

**5.66** Some of the accounting data and corroborating audit evidence may be available only in electronic form. For example, entities may use electronic data interchange or image processing systems. In image processing systems, documents are scanned and converted into electronic images to facilitate storage and reference, and the source documents may not be retained after conversion. Certain electronic evidence may exist at a certain point in time. However, such evidence may not be retrievable after a specified period of time if files are changed and if backup files do not exist. Therefore, the auditor might consider the time during which information exists or is available in determining the

<sup>&</sup>lt;sup>17</sup> See footnote 2.

nature, timing, and extent of his or her substantive tests and, if applicable, tests of controls.

- **5.67** Information technology may be performed solely by the institution, shared with others, or provided by an independent organization supplying specific data processing services for a fee. AU-C section 402, *Audit Considerations Relating to an Entity Using a Service Organization* (AICPA, *Professional Standards*), addresses the user auditor's responsibility when auditing the financial statements of entities that obtain services that are part of its information system from another organization (see further discussion in paragraphs 5.120–122).
- **5.68** The auditor should consider whether specialized skills are needed to consider the effect of information technology on the audit, to understand the internal control, or to design and perform audit procedures. If specialized skills are needed, the auditor should seek the assistance of someone possessing such skills who may be either on the audit staff or an outside professional. If the use of such a professional is planned, the auditor should have sufficient information technology related knowledge to communicate the desired objectives to the information technology professional, to evaluate whether the specific procedures will meet the auditor's objectives, and to evaluate the results of the procedures applied as they relate to the nature, timing, and extent of other planned audit procedures.<sup>18</sup>
- **5.69** System upgrades, conversions, and changes in technology have occurred with increasing frequency in the industry to accommodate the many changes in the nature and complexity of products and services offered, ongoing changes in accounting rules, continually evolving regulations, and mergers and acquisitions. A number of system changes may affect internal control. For example, merging institutions with incompatible computer systems can have a significant negative impact on the surviving institution's internal control. In addition to obtaining the understanding of ongoing or planned changes in processing controls that is necessary to plan the audit, the auditor may find it necessary to consider the effect of system changes on
  - a. controls over the accurate conversion of data to new or upgraded systems;
  - b. the effectiveness of data provided to perform analyses, such as those of the institution's performance versus its plan for assetliability management; and
  - c. the adequacy of the institution's disaster recovery plan and system.
- **5.70** Communication with those charged with governance. AU-C section 260, The Auditor's Communication With Those Charged With Governance (AICPA, Professional Standards), addresses the auditor's responsibility to communicate with those charged with governance in an audit of financial statements. Although this section applies regardless of an entity's governance structure or size, particular considerations apply when all of those charged with governance are involved in managing an entity. This section does not establish requirements regarding the auditor's communication with an entity's management or owners unless they are also charged with a governance role.

©2016, AICPA AAG-DEP 5.70

<sup>&</sup>lt;sup>18</sup> The requirements for the auditor's use of a specialist are described in AU-C section 620, *Using the Work of an Auditor's Specialist* (AICPA, *Professional Standards*).

- **5.71** AU-C section 265, Communicating Internal Control Related Matters Identified in an Audit (AICPA, Professional Standards), addresses the auditor's responsibility to appropriately communicate to those charged with governance and management deficiencies in internal control that the auditor has identified in an audit of financial statements. In particular, AU-C section 265
  - defines the terms deficiency in internal control, significant deficiency, and material weakness.
  - provides guidance on evaluating the severity of deficiencies in internal control identified in an audit of financial statements.
  - requires the auditor to communicate, in writing, to management and those charged with governance significant deficiencies and material weaknesses identified in an audit.
- **5.72** Paragraphs .11–.13 of AU-C section 265 state that the auditor should communicate in writing to those charged with governance on a timely basis significant deficiencies and material weaknesses identified during the audit, including those that were remediated during the audit. The auditor also should communicate to management at an appropriate level of responsibility, on a timely basis
  - a. in writing, significant deficiencies and material weaknesses that the auditor has communicated or intends to communicate to those charged with governance, unless it would be inappropriate to communicate directly to management in the circumstances.
  - b. in writing or orally, other deficiencies in internal control identified during the audit that have not been communicated to management by other parties and that, in the auditor's professional judgment, are of sufficient importance to merit management's attention. If other deficiencies in internal control are communicated orally, the auditor should document the communication.

The communication referred to should be made no later than 60 days following the report release date. However, paragraph .A15 of AU-C section 265 further explains that the communication is best made by the report release date because receipt of such communication may be an important factor in enabling those charged with governance to discharge their oversight responsibilities.

- **5.73** In accordance with paragraph .03 of AU-C section 265, nothing in AU-C section 265 precludes the auditor from communicating to those charged with governance or management other internal control matters that auditor has identified during the audit.
- **5.74** The appendix, "Examples of Circumstances That May Be Deficiencies, Significant Deficiencies, or Material Weaknesses," of AU-C section 265 includes examples of circumstances that may be deficiencies, significant deficiencies, or material weaknesses.
- **5.75** AU-C section 265 is not applicable if the auditor is engaged to report on the effectiveness of an entity's internal control over financial reporting under AT section 501, An Examination of an Entity's Internal Control Over Financial Reporting That Is Integrated With an Audit of Its Financial Statements (AICPA, Professional Standards).

AU-C section 265 is not applicable if the auditor is engaged to perform an audit of internal control over financial reporting that is integrated with an audit of financial statements. In such circumstances, AU-C section 940, An

Audit of Internal control Over Financial Reporting That Is Integrated With an Audit of Financial Statements (AICPA, Professional Standards), applies.<sup>19</sup>

# Risk Assessment and the Design of Further Audit Procedures

**5.76** As discussed in paragraph 5.25, risk assessment procedures allow the auditor to gather the information necessary to obtain an understanding of the entity and its environment including its internal control. This knowledge provides a basis for assessing the risks of material misstatement of the financial statements. These risk assessments are then used to design further audit procedures, such as tests of controls and substantive tests. This section provides guidance on assessing the risks of material misstatement and how to design further audit procedures that effectively respond to those risks.

## Identifying and Assessing the Risks of Material Misstatement

- **5.77** To provide a basis for designing and performing further audit procedures, paragraphs .26—.27 of AU-C section 315 state that the auditor should identify and assess the risks of material misstatement at the financial statement level and at the relevant assertion level for classes of transactions, account balances, and disclosures. For this purpose, the auditor should
  - a. identify risks throughout the process of obtaining an understanding of the entity and its environment, including relevant controls that relate to the risks, by considering the classes of transactions, account balances, and disclosures in the financial statements (see further discussion in paragraph 5.79);
  - assess the identified risks and evaluate whether they relate more pervasively to the financial statements as a whole and potentially affect many assertions;
  - relate the identified risks to what can go wrong at the relevant assertion level, taking account of relevant controls that the auditor intends to test; and
  - d. consider the likelihood of misstatement, including the possibility of multiple misstatements, and whether the potential misstatement is of a magnitude that could result in a material misstatement.

**5.78** Paragraph .A108 of AU-C section 315 explains that the risks of material misstatement at the financial statement level refer to risks that relate pervasively to the financial statements as a whole and potentially affect many assertions. Risks of this nature are not necessarily risks identifiable with

©2016, AICPA AAG-DEP 5.78

<sup>&</sup>lt;sup>19</sup> Statement on Auditing Standards (SAS) No. 130, An Audit of Internal Control Over Financial Reporting That Is Integrated With an Audit of Financial Statements, was issued in November 2015. The SAS is effective for integrated audits for periods ending on or after December 15, 2016.

The ASB concluded that, because engagements performed under AT section 501, An Examination of an Entity's Internal Control Over Financial Reporting That Is Integrated With an Audit of Its Financial Statements (AICPA, Professional Standards), as well as related attestation Interpretation No. 1, "Reporting Under Section 112 of the Federal Deposit Insurance Corporation Improvement Act" (AICPA, Professional Standards, AT sec. 9501 par. .01–.07), are required to be integrated with an audit of financial statements, it is appropriate to move the content of AT section 501 from the attestation standards into GAAS.

Readers are encouraged to consult the full text of this SAS on the AICPA's website at www.aicpa.org.

specific assertions at the class of transactions, account balance, or disclosure level. Rather, they represent circumstances that may increase the risks of material misstatement at the assertion level (for example, through management override of internal control). Financial statement level risks may be especially relevant to the auditor's consideration of the risks of material misstatement arising from fraud.

**5.79** Process of identifying risks of material misstatement. Paragraph .A120 of AU-C section 315 explains that information gathered by performing risk assessment procedures, including the audit evidence obtained in evaluating the design of controls and determining whether they have been implemented, is used as audit evidence to support the risk assessment. The risk assessment determines the nature, timing, and extent of further audit procedures to be performed.

#### Risks that Require Special Audit Consideration

- **5.80** Paragraphs .28—.29 of AU-C section 315 state that as part of the risk assessment described in paragraph .26 of AU-C section 315 (see paragraph 5.77), the auditor should determine whether any of the risks identified are, in the auditor's professional judgment, a significant risk. In exercising this judgment, the auditor should exclude the effects of identified controls related to the risk. In addition, the auditor should consider at least
  - a. whether the risk is a risk of fraud;
  - b. whether the risk is related to recent significant economic, accounting, or other developments and, therefore, requires specific attention:
  - c. the complexity of transactions;
  - d. whether the risk involves significant transactions with related parties;
  - the degree of subjectivity in the measurement of financial information related to the risk, especially those measurements involving a wide range of measurement uncertainty; and
  - f. whether the risk involves significant transactions that are outside the normal course of business for the entity or that otherwise appear to be unusual.
- **5.81** If the auditor has determined that a significant risk exists, paragraph .30 of AU-C section 315 states that the auditor should obtain an understanding of the entity's controls, including control activities, relevant to that risk and, based on that understanding, evaluate whether such controls have been suitably designed and implemented to mitigate such risks. See paragraphs 5.90 and 5.93 for discussion over further audit procedures pertaining to significant risks.

## **Designing and Performing Further Audit Procedures**

**5.82** AU-C section 330 addresses the auditor's responsibility to design and implement responses to the risks of material misstatement identified and assessed by the auditor in accordance with AU-C section 315 and to evaluate the audit evidence obtained in an audit of financial statements.

## **Overall Responses**

**5.83** Paragraph .05 of AU-C section 330 states that the auditor should design and implement overall responses to address the assessed risks of

material misstatement at the financial statement level. Paragraph .A1 of AU-C section 330 states that overall responses to address the assessed risks of material misstatement at the financial statement level may include emphasizing to the audit team the need to maintain professional skepticism, assigning more experienced staff or those with specialized skills or using specialists, providing more supervision, incorporating additional elements of unpredictability in the selection of further audit procedures to be performed, or making general changes to the nature, timing, or extent of further audit procedures (for example, performing substantive procedures at period end instead of at an interim date or modifying the nature of audit procedures to obtain more persuasive audit evidence). Financial institutions are subject to certain risks that are less prevalent in commercial, industrial, and other nonfinancial businesses. and they operate in a particularly volatile and highly regulated environment. Accordingly, the auditor might design appropriate overall responses to that higher risk with personnel who have appropriate relevant experience and provide more extensive supervision. See paragraphs 5.06-.09 for more guidance regarding the auditor's overall responses to audit risk.

**5.84** Paragraphs .A2–.A3 of AU-C section 330 go on to explain that the assessment of the risks of material misstatement at the financial statement level and, thereby, the auditor's overall responses are affected by the auditor's understanding of the control environment. An effective control environment may allow the auditor to have more confidence in internal control and the reliability of audit evidence generated internally within the entity and, thus, for example, allow the auditor to conduct some audit procedures at an interim date rather than at the period-end. Deficiencies in the control environment, however, have the opposite effect (for example, the auditor may respond to an ineffective control environment by

- conducting more audit procedures as of the period-end rather than at an interim date,
- obtaining more extensive audit evidence from substantive procedures, and
- increasing the number of locations to be included in the audit scope).

Such considerations, therefore, have a significant bearing on the auditor's general approach (for example, an emphasis on substantive procedures [substantive approach] or an approach that uses tests of controls as well as substantive procedures [combined approach]).

#### Further Audit Procedures

**5.85** Further audit procedures provide important audit evidence to support an audit opinion. These procedures consist of tests of controls and substantive tests. Paragraph .06 of AU-C section 330 states that the auditor should design and perform further audit procedures whose nature, timing, and extent are based on, and are responsive to, the assessed risks of material misstatement at the relevant assertion level.

 ${f 5.86}$  In designing the further audit procedures to be performed, paragraph .07 of AU-C section 330 states that the auditor should

 consider the reasons for the assessed risk of material misstatement at the relevant assertion level for each class of transactions, account balance, and disclosure, including

©2016. AICPA AAG-DEP 5.86

- the likelihood of material misstatement due to the particular characteristics of the relevant class of transactions, account balance, or disclosure (the inherent risk) and
- ii. whether the risk assessment takes account of relevant controls (the control risk), thereby requiring the auditor to obtain audit evidence to determine whether the controls are operating effectively (that is, the auditor intends to rely on the operating effectiveness of controls in determining the nature, timing, and extent of substantive procedures), and
- b. obtain more persuasive audit evidence the higher the auditor's assessment of risk.
- **5.87** Tests of controls. In accordance with paragraph .08 of AU-C section 330, the auditor should design and perform tests of controls to obtain sufficient appropriate audit evidence about the operating effectiveness of relevant controls if (a) the auditor's assessment of risks of material misstatement at the relevant assertion level includes an expectation that the controls are operating effectively (that is, the auditor intends to rely on the operating effectiveness of controls in determining the nature, timing, and extent of substantive procedures)<sup>20</sup> or (b) when substantive procedures alone cannot provide sufficient appropriate audit evidence at the relevant assertion level. In accordance with paragraph .A21 of AU-C section 330, tests of controls are performed only on those controls that the auditor has determined are suitably designed to prevent, or detect and correct, a material misstatement in a relevant assertion. If substantially different controls were used at different times during the period under audit, each is considered separately.
- **5.88** Paragraph .A22 of AU-C section 330 states that the testing the operating effectiveness of controls is different from obtaining an understanding of and evaluating the design and implementation of controls. However, the same types of audit procedures are used. The auditor may, therefore, decide it is efficient to test the operating effectiveness of controls at the same time the auditor is evaluating their design and determining that they have been implemented.
- **5.89** Paragraph .A23 of AU-C section 330 states that although some risk assessment procedures may not have been specifically designed as tests of controls, they may nevertheless provide audit evidence about the operating effectiveness of the controls and, consequently, serve as tests of controls.
- **5.90** Timing of tests of controls over significant risks. One or more significant risks normally arise on most audits. <sup>21</sup> Paragraph .15 of AU-C section 330 states that if the auditor plans to rely on controls over a risk the auditor

<sup>&</sup>lt;sup>20</sup> Q&A section 8200.06, "The Meaning of Expectation of the Operating Effectiveness of Controls" (AICPA, Technical Questions and Answers), states that the phrase expectation of the operating effectiveness of controls means that the auditor's understanding of the five components of internal control has enabled him or her to initially assess control risk at less than maximum; and the auditor's strategy contemplates a combined approach of designing and performing tests of controls and substantive procedures.

<sup>&</sup>lt;sup>21</sup> According to paragraph .27 of AU-C section 240, Consideration of Fraud in a Financial Statement Audit (AICPA, Professional Standards), the auditor should treat those assessed risks of material misstatement due to fraud as significant risks. Paragraph .26 of AU-C section 240 states that there is a presumption that risks of fraud exist in revenue recognition. Paragraph .31 of AU-C section 240 states that the risk of management override of controls is present in all entities and is a risk of material misstatement due to fraud and thus, a significant risk. Thus, there are generally at least two significant risks in any financial statement audit.

has determined to be a significant risk, the auditor should test the operating effectiveness of those controls in the current period.

- **5.91** Substantive procedures. Irrespective of the assessed risks of material misstatement, the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure, in accordance with paragraph .18 of AU-C section 330.
- **5.92** Paragraph .21 of AU-C section 330 states that the auditor's substantive procedures should include audit procedures related to the financial statement closing process, such as
  - agreeing or reconciling the financial statements with the underlying accounting records and
  - examining material journal entries and other adjustments made during the course of preparing the financial statements.

Paragraph .A57 of AU-C section 330 states that the nature and extent of the auditor's examination of journal entries and other adjustments depends on the nature and complexity of the entity's financial reporting process and the related risks of material misstatement.

- **5.93** Substantive procedures responsive to significant risks. If the auditor has determined that an assessed risk of material misstatement at the relevant assertion level is a significant risk, paragraph .22 of AU-C section 330 states that the auditor should perform substantive procedures that are specifically responsive to that risk. When the approach to a significant risk consists only of substantive procedures, those procedures should include tests of details.
- **5.94** Substantive analytical procedures. AU-C section 520, Analytical Procedures (AICPA, Professional Standards), addresses the auditor's use of analytical procedures as substantive procedures (substantive analytical procedures). It also addresses the auditor's responsibility to perform analytical procedures near the end of the audit that assist the auditor when forming an overall conclusion on the financial statements.
- **5.95** As explained in paragraphs .A2—.A3 of AU-C section 520, analytical procedures include the consideration of comparisons of the entity's financial information with, for example, comparable information for prior periods, anticipated results of the entity (such as, budgets or forecasts) or expectations of the auditor, or similar industry information. Analytical procedures also include consideration of relationships, like elements of financial information that would be expected to conform to a predictable pattern based on recent history of the entity and industry or between financial information and relevant nonfinancial information (such as, payroll costs to number of employees). When designing and performing analytical procedures, either alone or in combination with tests of details, as substantive procedures, paragraph .05 of AU-C section 520 states that the auditor should
  - a. determine the suitability of particular substantive analytical procedures for given assertions, taking into account the assessed risks of material misstatement and tests of details, if any, for these assertions;
  - evaluate the reliability of data from which the auditor's expectation
    of recorded amounts or ratios is developed, taking into account
    the source, comparability, and nature and relevance of information
    available and controls over preparation;

©2016. AICPA AAG-DEP 5.95

- c. develop an expectation of recorded amounts or ratios and evaluate whether the expectation is sufficiently precise (taking into account whether substantive analytical procedures are to be performed alone or in combination with tests of details) to identify a misstatement that, individually or when aggregated with other misstatements, may cause the financial statements to be materially misstated; and
- d. determine the amount of any difference of recorded amounts from expected values that is acceptable without further investigation and compare the recorded amounts, or ratios developed from recorded amounts, with the expectations.
- **5.96** Paragraphs .A13-.A14 of AU-C section 520 explain that different types of analytical procedures provide different levels of assurance. The determination of the suitability of particular substantive analytical procedures is influenced by the nature of the assertion and the auditor's assessment of the risk of material misstatement. Paragraph .A8 of AU-C section 520 states that the effectiveness and efficiency of a substantive analytical procedure in addressing risks of material misstatement depends on, among other things, (a) the nature of the assertion, (b) the plausibility and predictability of the relationship, (c) the availability and reliability of the data used to develop the expectation, and (d) the precision of the expectation. For this reason, substantive analytical procedures alone are not well suited to detecting fraud. In addition, paragraph .A19 of AU-C section 520 notes that the auditor may consider testing the operating effectiveness of controls, if any, over the entity's preparation of information used by the auditor in performing the substantive analytical procedures in response to assessed risks. When such controls are effective, the auditor may have greater confidence in the reliability of the information and, therefore, in the results of analytical procedures. The operating effectiveness of controls over nonfinancial information may often be tested in conjunction with other tests of controls.
- 5.97 Paragraph .08 of AU-C section 520 states that when substantive analytical procedures have been performed, the auditor should include in the audit document the following:
  - a. The expectation referred to in paragraph .05c of AU-C section 520 (see paragraph 5.95c) and the factors considered in its development when that expectation or those factors are not otherwise readily determinable from the audit documentation
  - b. Results of the comparison referred to in paragraph .05d of AU-C section 520 (see paragraph 5.95d) of the recorded amounts, or ratios developed from recorded amounts, with the expectations
  - c. Any additional auditing procedures performed in accordance with paragraph .07 of AU-C section 520 relating to the investigation of fluctuations or relationships that are inconsistent with other relevant information or that differ from expected values by a significant amount and the results of such additional procedures

## **Evaluating the Sufficiency and Appropriateness of Audit Evidence**

**5.98** Paragraph .28 of AU-C section 330 states the auditor should conclude whether sufficient appropriate audit evidence has been obtained. In forming a conclusion, the auditor should consider all relevant audit evidence, regardless

of whether it appears to corroborate or to contradict the relevant assertions in the financial statements.

## **Evaluation of Misstatements Identified During the Audit**

- **5.99** Based on the results of substantive procedures, the auditor may identify misstatements in accounts or notes to the financial statements. AUC section 450 addresses the auditor's responsibility to evaluate the effect of identified misstatements on the audit and the effect of uncorrected misstatements, if any, on the financial statements. Paragraphs .05–.12 of AU-C section 450 address specific requirements the auditor should perform in relation to accumulation of identified misstatements, consideration of identified misstatements as the audit progresses, communication and correction of misstatements, evaluating the effect of uncorrected misstatements,  $^{22}$  and documentation.
- **5.100** The circumstances related to some misstatements may cause the auditor to evaluate them as material, individually or when considered together with other misstatements accumulated during the audit, even if they are below the materiality threshold for the financial statements as a whole. For example, a loan made to a related party of an otherwise immaterial amount could be material if there is a reasonable possibility that it could lead to a material contingent liability or a material loss of revenue. Paragraph .A23 of AU-C section 450 provides circumstances that the auditor may consider relevant in determining whether misstatements are material.
- **5.101** AU-C section 700 addresses the auditor's responsibility in forming an opinion on the financial statements based on the evaluation of the audit evidence obtained. The auditor's conclusion, required by AU-C section 700, takes into account the auditor's evaluation of uncorrected misstatements, if any, on the financial statements, in accordance with AU-C section 450.

## **Audit Documentation**

- **5.102** AU-C section 230, *Audit Documentation* (AICPA, *Professional Standards*), addresses the auditor's responsibility to prepare audit documentation for an audit of financial statements. The exhibit, "Audit Documentation Requirements in Other AU-C Sections," (see paragraph .A30 of AU-C section 230) lists other AU-C sections that contain specific documentation requirements and guidance. The specific documentation requirements of other AU-C sections do not limit the application of AU-C section 230. Law, regulation, or other standards may establish additional documentation requirements.
- $\bf 5.103~$  Paragraph .02 of AU-C section 230 states that audit documentation that meets the requirements of AU-C section 230 and the specific documentation requirements of other relevant AU-C sections provides

<sup>&</sup>lt;sup>22</sup> The SEC's Codification of Staff Accounting Bulletins topic 1N, "Considering the Effects of Prior Year Misstatements when Quantifying Misstatements in Current Year Financial Statements," provides guidance on the consideration of the effects of prior year misstatements in quantifying current year misstatements for the purpose of a materiality assessment. The bulletin points out that some registrants do not consider the effects of prior year errors on current year financial statements that allow the entity to report unadjusted (and improper) assets and liabilities. The topic also notes that an immaterial error on the balance sheet could be material on the income statement.

- a. evidence of the auditor's basis for a conclusion about the achievement of the overall objectives of the auditor;<sup>23</sup> and
- b. evidence that the audit was planned and performed in accordance with GAAS and applicable legal and regulatory requirements.

**5.104** For purposes of GAAS, *audit documentation*, as defined in paragraph .06 of AU-C section 230, is the record of audit procedures performed, relevant audit evidence obtained, and conclusions the auditor reached (terms such as *working papers* or *workpapers* are also sometimes used).

### **Timely Preparation of Audit Documentation**

**5.105** Paragraph .07 of AU-C section 230 states that the auditor should prepare audit documentation on a timely basis. Paragraph .A3 of AU-C section 230 further explains that preparing sufficient and appropriate audit documentation on a timely basis throughout the audit helps to enhance the quality of the audit and facilitates the effective review and evaluation of the audit evidence obtained and conclusions reached before the auditor's report is finalized. Documentation prepared at the time such work is performed or shortly thereafter is likely to be more accurate than documentation prepared at a much later time.<sup>24</sup>

## Documentation of the Audit Procedures Performed and Audit Evidence Obtained

**5.106** Paragraphs .08–.12 of AU-C section 230 address the auditor's responsibilities regarding documentation of the audit procedures performed and audit evidence obtained including form, content, and extent of audit documentation. In accordance with paragraph .08 of AU-C section 230, the auditor should prepare audit documentation that is sufficient to enable an experienced auditor, having no previous connection with the audit, to understand

- a. the nature, timing, and extent of the audit procedures performed to comply with GAAS and applicable legal and regulatory requirements; (Readers can find additional application and explanatory material in paragraphs .A8—.A9 of AU-C section 230)
- b. the results of the audit procedures performed, and the audit evidence obtained; and
- c. significant findings or issues arising during the audit, the conclusions reached thereon, and significant professional judgments made in reaching those conclusions. (Readers can find additional application and explanatory material in paragraphs .A10-.A13 of AU-C section 230.)

<sup>&</sup>lt;sup>23</sup> Paragraph .12 of AU-C section 200.

<sup>&</sup>lt;sup>24</sup> A firm of independent auditors has a responsibility to adopt a system of quality control policies and procedures to provide the firm with reasonable assurance that its personnel comply with applicable professional standards, including GAAS, and the firm's standards of quality in conducting individual audit engagements. Review of audit documentation and discussions with engagement team members are among the procedures a firm performs when monitoring compliance with the quality control policies and procedures that it has established. The elements of quality control are identified in QC section 10, A Firm's System of Quality Control (AICPA, Professional Standards). See also AU-C section 220, Quality Control for an Engagement Conducted in Accordance With Generally Accepted Auditing Standards (AICPA, Professional Standards).

As stated in paragraph .A5 of AU-C section 230, examples of audit documentation include audit plans, analyses, issues memorandums, summaries of significant findings or issues, letters of confirmation and representation, checklists, and correspondence (including e-mail) concerning significant findings or issues.

**5.107** For audit procedures related to the inspection of significant contracts or agreements, paragraph .10 of AU-C section 230 states that the auditor should include abstracts or copies of those contracts or agreements in the audit documentation.

5.108 In addition to the requirements discussed previously, paragraphs .13–.14 of AU-C section 230 address further documentation requirements about departures from relevant requirements and matters arising after the date of the auditor's report.

## Assembly and Retention of the Final Audit File

**5.109** Paragraphs .15–.19 of AU-C section 230 address an auditor's responsibilities regarding assembly and retention of the final audit file. Paragraph .16 of AU-C section 230 states that the auditor should assemble the audit documentation in an audit file and complete the administrative process of assembling the final audit file on a timely basis, no later than 60 days following the report release date. After the documentation completion date, paragraph .17 of AU-C section 230 prohibits the auditor from deleting or discarding audit documentation of any nature before the end of the specified retention period. If it is necessary to modify existing audit documentation or add new audit documentation after the documentation date, paragraph .18 of AU-C section 230 requires the auditor to document the specific reasons for making the changes and when and by whom the changes were made and reviewed.

## Using the Work of an Auditor's Specialist

**5.110** AU-C section 620, *Using the Work of an Auditor's Specialist* (AICPA, *Professional Standards*), addresses the auditor's responsibilities relating to the work of an individual or organization possessing expertise in a field other than accounting or auditing when that work is used to assist the auditor in obtaining sufficient appropriate audit evidence (defined as an *auditor's specialist* for purposes of GAAS). An auditor's specialist may be either an internal specialist (who is a partner or staff, including temporary staff, of the auditor's firm or a network firm) or an external specialist.

#### 5.111 AU-C section 620 does not address

- situations in which the engagement team includes a member or consults an individual or organization with expertise in a specialized area of accounting or auditing, which are addressed in AU-C section 220, Quality Control for an Engagement Conducted in Accordance With Generally Accepted Auditing Standards (AICPA, Professional Standards), and AU-C section 300,<sup>25,26</sup> or
- auditor's use of the work of an individual or organization possessing expertise in a field other than accounting or auditing, whose work in that field is used by the entity to assist the entity in

<sup>&</sup>lt;sup>25</sup> Paragraphs .A10 and .A20-.A22 of AU-C section 220.

<sup>&</sup>lt;sup>26</sup> Paragraph .12 of AU-C section 300, Planning an Audit (AICPA, Professional Standards).

preparing the financial statements (a management's specialist), which is addressed in AU-C section 500, *Audit Evidence* (AICPA, *Professional Standards*).<sup>27</sup>

- **5.112** In accordance with AU-C section 620, the objectives of the auditor are (a) to determine whether to use the work of an auditor's specialist and (b) if using the work of an auditor's specialist, to determine whether that work is adequate for the auditor's purposes. In reaching these objectives, the auditor should
  - determine the need for an auditor's specialist if expertise in a field other than accounting or auditing is necessary to obtain sufficient appropriate audit evidence.
  - evaluate the competence, capabilities, and objectivity of the auditor's specialist.
  - obtain a sufficient understanding of the field of expertise of the auditor's specialist to enable the auditor to (a) determine the nature, scope, and objectives of the work of the auditor's specialist for the auditor's purposes and (b) evaluate the adequacy of that work for the auditor's purposes.
- **5.113** Paragraph .09 of AU-C section 620 states that the auditor should evaluate whether the auditor's specialist has the necessary competence, capabilities, and objectivity for the auditor's purposes.
- **5.114** AU-C section 620 does not preclude the auditor from using a specialist who has a relationship with the client, including situations where the client has the ability to directly or indirectly control or significantly influence the specialist. However, paragraph .09 of AU-C section 620 states that, in the case of an auditor's external specialist, the evaluation of objectivity should include inquiry regarding interests and relationships that may create a threat to the objectivity of the auditor's specialist. If the auditor believes that a relationship between the entity and the auditor's specialist might impair the objectivity of the auditor way perform additional procedures with respect to some or all of the assumptions, methods, or findings of the auditor's specialist to determine that the findings are reasonable or may engage another specialist for that purpose.
- **5.115** Paragraph .10 of AU-C section 620 states that the auditor should obtain a sufficient understanding of the field of expertise of the auditor's specialist to enable the auditor to
  - determine the nature, scope, and objectives of the work of the auditor's specialist for the auditor's purposes and
  - evaluate the adequacy of that work for the auditor's purposes.

## Using the Work of a Management's Specialist

**5.116** AU-C section 500 addresses the auditor's use of the work of an individual or organization possessing expertise in a field other than accounting or auditing, whose work in that field is used by the entity to assist the entity in preparing the financial statements (defined as a *management's specialist*).

<sup>&</sup>lt;sup>27</sup> Paragraphs A35-A49 of AU-C section 500, Audit Evidence (AICPA, Professional Standards).

- **5.117** Information regarding the competence, capabilities, and objectivity of a management's specialist may come from a variety of sources, such as knowledge of that specialist's qualifications, membership in a professional body or industry association, license to practice, or other forms of external recognition (a listing of additional sources is addressed in paragraph .A39 of AU-C section 500). For example, if the auditor is using an appraisal of commercial real estate values in connection with the audit of financial statements, he or she should evaluate the appraiser's professional qualifications and his or her experience with commercial real estate. Further application and explanatory material regarding the reliability of information produced by a management's specialist is addressed in paragraphs .A35—.A49 of AU-C section 500.
- **5.118** In a number of cases, the specialist's work may have been prepared for another purpose (such as, an appraiser's report prepared for a loan origination). If information to be used as audit evidence has been prepared using the work of a management's specialist, paragraph .08 of AU-C section 500 states that the auditor should, to the extent necessary, taking into account the significance of that specialist's work for the auditor's purposes,
  - a. evaluate the competence, capabilities, and objectivity of that specialist;
  - b. obtain an understanding of the work of that specialist; and
  - evaluate the appropriateness of that specialist's work as audit evidence for the relevant assertion.

Furthermore, paragraph .17 of Interpretation No. 1, "The Use of Legal Interpretations As Audit Evidence to Support Management's Assertion That a Transfer of Financial Assets Has Met the Isolation Criterion in Paragraphs 7–14 of Financial Accounting Standards Board Accounting Standards Codification 860-10-40," of AU-C section 620 (AICPA, Professional Standards, AU-C sec. 9620 par. .01–.21), states that, in some cases, the auditor may decide it necessary to contact the specialist to determine that the specialist is aware that his or her work will be used for evaluating the assertions in the financial statements.

**5.119** The Audit Issues Task Force of the Auditing Standards Board issued Interpretation No. 1 of AU-C section 620.<sup>28</sup> The guidance relates to examples of legal opinions that auditors will need to obtain and review with regard to transfers of financial assets by banks subject to receivership or conservatorship

FDIC would not try to reclaim loans transferred into such transactions.

In light of the issuance of FASB Statement No. 166 and the FDIC's changes to the safe harbor, the ASB is currently in the process of revising the interpretation. Auditors should be alert for such revisions; however, the guidance in this interpretation continues to be relevant.

<sup>&</sup>lt;sup>28</sup> Interpretation No. 1, "The Use of Legal Interpretations As Audit Evidence to Support Management's Assertion That a Transfer of Financial Assets Has Met the Isolation Criterion in Paragraphs 7–14 of FASB Accounting Standards Codification 860-10-40," of AU-C section 620 (AICPA, Professional Standards, AU-C sec. 9620 par. .01–.21) has not been updated to reflect the issuance of FASB Statement No. 166, Accounting for Transfers of Financial Assets. FASB Statement No. 166 was incorporated into FASB ASC in FASB Accounting Standards Update (ASU) No. 2009-16, Transfers and Servicing (Topic 860): Accounting for Transfers of Financial Assets, and is discussed in FASB ASC 860, Transfers and Servicing.

In addition, this interpretation has not been updated for changes to the FDIC's safe harbor for financial assets transferred in connection with securitizations and participations. The FDIC's final amendments to the safe harbor, Treatment by the Federal Deposit Insurance Corporation as Conservator or Receiver of Financial Assets Transferred by an Insured Depository Institution in Connection With a Securitization or Participation After September 30, 2010 (www.fdic.gov/news/news/press/2010/pr10216.html), were issued in September 2010. The safe harbor provides important protections for securitizations and participations by confirming that in the event of a bank failure, the FDIC would not try to reclaim loans transferred into such transactions.

under provisions of the Federal Deposit Insurance Act (FDI Act). This interpretation is for auditing procedures related to transfers of financial assets that are accounted for under FASB ASC 860, *Transfers and Servicing*.

## **Processing of Transactions by Service Organizations**

**5.120** AU-C section 402 addresses the user auditor's responsibility for obtaining sufficient appropriate audit evidence in an audit of the financial statements of a user entity that uses one or more service organizations (for example, using a mortgage banker to service mortgages). Specifically, it expands on how the user auditor applies AU-C sections 315 and 330 in obtaining an understanding of the user entity, including internal control relevant to the audit, sufficient to identify and assess the risks of material misstatement and in designing and performing further audit procedures responsive to those risks.

**5.121** Paragraphs .03–.05 of AU-C section 402 state that services provided by a service organization are relevant to the audit of a user entity's financial statements when those services and the controls over them affect the user entity's information system, including related business processes, relevant to financial reporting. Although most controls at the service organization are likely to relate to financial reporting, other controls also may be relevant to the audit, such as controls over the safeguarding of assets. A service organization's services are part of a user entity's information system, including related business processes, relevant to financial reporting if these services affect any of the following:

- a. The classes of transactions in the user entity's operations that are significant to the user entity's financial statements;
- b. The procedures within both IT and manual systems by which the user entity's transactions are initiated, authorized, recorded, processed, corrected as necessary, transferred to the general ledger, and reported in the financial statements;
- c. The related accounting records, supporting information, and specific accounts in the user entity's financial statements that are used to initiate, authorize, record, process, and report the user entity's transactions. This includes the correction of incorrect information and how information is transferred to the general ledger; the records may be in either manual or electronic form;
- d. How the user entity's information system captures events and conditions, other than transactions, that are significant to the financial statements;
- The financial reporting process used to prepare the user entity's financial statements, including significant accounting estimates and disclosures; and
- f. Controls surrounding journal entries, including nonstandard journal entries used to record nonrecurring, unusual transactions, or adjustments.

The nature and extent of work to be performed by the user auditor regarding the services provided by a service organization depend on the nature and significance of those services to the user entity and the relevance of those services to the audit. **5.122** AU-C section 402 does not apply to services that are limited to processing an entity's transactions that are specifically authorized by the entity, such as the processing of checking account transactions by a bank or the processing of securities transactions by a broker (that is, when the user entity retains responsibility for authorizing the transactions and maintaining the related accountability). In addition, AU-C section 402 does not apply to the audit of transactions arising from an entity that holds a proprietary financial interest in another entity, such as a partnership, corporation, or joint venture, when the partnership, corporation, or joint venture performs no processing on behalf of the entity.

## Consideration of Fraud in a Financial Statement Audit 29

 ${f 5.123}$  AU-C section 240 addresses the auditor's responsibilities relating to fraud in an audit of financial statements. Specifically, it expands on how AU-C sections 315 and 330 are to be applied regarding risks of material misstatement due to fraud.

**5.124** Although fraud is a broad legal concept, for the purposes of GAAS, the auditor is primarily concerned with fraud that causes a material misstatement in the financial statements. In accordance with paragraph .03 of AU-C section 240, two types of intentional misstatements are relevant to the auditor:

- Misstatements resulting from fraudulent financial reporting
- Misstatements resulting from misappropriation of assets

Although the auditor may suspect or, in rare cases, identify the occurrence of fraud, the auditor does not make legal determinations of whether fraud has actually occurred.

**5.125** Paragraph .A1 of AU-C section 240 states that fraud, whether fraudulent financial reporting or misappropriation of assets, involves incentive or pressure to commit fraud, a perceived opportunity to do so, and some rationalization of the act.

## **Professional Skepticism**

**5.126** Consistent with paragraph .15 of AU-C section 200, paragraph .12 of AU-C section 240 states that the auditor should maintain professional skepticism throughout the audit, recognizing the possibility that a material misstatement due to fraud could exist, notwithstanding the auditor's past experience of the honesty and integrity of the entity's management and those charged with governance.

**5.127** Paragraphs .A9-.A10 of AU-C section 240 states that maintaining professional skepticism requires an ongoing questioning of whether the information and evidence obtained suggests that a material misstatement due to fraud may exist. It includes considering the reliability of the information to be used as audit evidence and the controls over its preparation and maintenance when relevant. Although the auditor cannot be expected to disregard past experience of the honesty and integrity of the entity's management and those charged with governance, the auditor's professional skepticism is particularly important in considering the risk of material misstatement due to fraud because there may have been changes in circumstances.

<sup>&</sup>lt;sup>29</sup> See footnote 22.

**5.128** When responses to inquiries of management, those charged with governance, or others are inconsistent or otherwise unsatisfactory (for example, vague or implausible), paragraph .14 of AU-C section 240 states that the auditor should further investigate the inconsistencies or unsatisfactory responses.

## **Discussion Among the Engagement Team**

- **5.129** AU-C section 315 requires a discussion among the key engagement team members (see detailed discussion at paragraph 5.31). Paragraph .15 of AU-C section 240 states this discussion should include an exchange of ideas or brainstorming among the engagement team members about how and where the entity's financial statements might be susceptible to material misstatement due to fraud, how management could perpetrate and conceal fraudulent financial reporting, and how assets of the entity could be misappropriated. The discussion should occur setting aside beliefs that the engagement team members may have that management and those charged with governance are honest and have integrity, and should, in particular, also address
  - a. known external and internal factors affecting the entity that may create an incentive or pressure for management or others to commit fraud, provide the opportunity for fraud to be perpetrated, and indicate a culture or environment that enables management or others to rationalize committing fraud;
  - b. the risk of management override of controls;
  - c. consideration of circumstances that might be indicative of earnings management or manipulation of other financial measures and the practices that might be followed by management to manage earnings or other financial measures that could lead to fraudulent financial reporting;
  - the importance of maintaining professional skepticism throughout the audit regarding the potential for material misstatement due to fraud; and
  - e. how the auditor might respond to the susceptibility of the entity's financial statements to material misstatement due to fraud.

Communication among the engagement team members about the risks of material misstatement due to fraud should continue throughout the audit, particularly upon discovery of new facts during the audit.

- **5.130** Paragraph .A12 of AU-C section 240 states that discussing the susceptibility of the entity's financial statements to material misstatement due to fraud with the engagement team
  - provides an opportunity for more experienced engagement team members to share their insights about how and where the financial statements may be susceptible to material misstatement due to fraud.
  - enables the auditor to consider an appropriate response to such susceptibility and to determine which members of the engagement team will conduct certain audit procedures.
  - permits the auditor to determine how the results of audit procedures will be shared among the engagement team and how to deal with any allegations of fraud that may come to the auditor's attention during the audit.

- **5.131** In addition, paragraph .A13 of AU-C section 240 states the discussion may include the following matters:
  - A consideration of management's involvement in overseeing employees with access to cash or other assets susceptible to misappropriation
  - A consideration of any unusual or unexplained changes in behavior or lifestyle of management or employees that have come to the attention of the engagement team
  - A consideration of the types of circumstances that, if encountered, might indicate the possibility of fraud
  - A consideration of how an element of unpredictability will be incorporated into the nature, timing, and extent of the audit procedures to be performed
  - A consideration of the audit procedures that might be selected to respond to the susceptibility of the entity's financial statements to material misstatement due to fraud and whether certain types of audit procedures are more effective than others
  - A consideration of any allegations of fraud that have come to the auditor's attention

A number of factors may influence the extent of the discussion and how it may occur. For example, if the audit involves more than one location, there could be multiple discussions with team members in differing locations. Another factor in planning the discussions is whether to include specialists assigned to the audit team.

**5.132** Exhibit 5-1, "Fraud Risk Factors," which appears at the end of this chapter, contains a list of fraud risk factors that auditors may consider as part of their planning and audit procedures. The purpose is for audit team members to communicate and share information obtained throughout the audit that may affect the assessment of the risks of material misstatement due to fraud or error or the audit procedures performed to address the risks.

#### Risk Assessment Procedures and Related Activities

**5.133** When performing risk assessment procedures and related activities to obtain an understanding of the entity and its environment, including the entity's internal control, required by AU-C section 315, paragraph .16 of AU-C section 240 states that the auditor should perform the procedures in paragraphs .17–.24 of AU-C section 240 to obtain information for use in identifying the risk of material misstatement due to fraud. As part of this work, the auditor should perform the following procedures:

- a. Hold fraud discussions with management, others within the entity, and those charged with governance (unless all those charged with governance are involved in managing the entity). See specific inquiries the auditor should make in paragraphs .17–.19 and .21 of AU-C section 240.
- b. Obtain an understanding of how those charged with governance exercise oversight of management's process for identifying and responding to the risks of fraud in the entity and the internal control that management has established to mitigate these risks, unless all those charged with governance are involved in managing the entity. (See paragraphs .20 and .A21-.A23 of AU-C section 240.)

- c. Evaluate whether unusual or unexpected relationships that have been identified (based on analytical procedures performed as part of risk assessment procedures) indicate risks of material misstatement due to fraud. (See paragraphs .22, .A24-.A26, and .A46 of AU-C section 240.)
- d. Consider whether other information obtained by the auditor indicates risks of material misstatement due to fraud. (See further application guidance in paragraph .A27 of AU-C section 240.)
- e. Evaluate whether the information obtained from the risk assessment procedures and related activities performed indicates that one or more fraud risk factors are present. (See paragraphs .24 and .A28–.A32 of AU-C section 240.)

#### **Evaluation of Fraud Risk Factors**

- **5.134** As indicated in paragraph 5.133e, the auditor may identify events or conditions that indicate incentives and pressures to perpetrate fraud, opportunities to carry out the fraud, or attitudes and rationalizations to justify a fraudulent action. Such events or conditions are referred to as *fraud risk factors*. Although fraud risk factors may not necessarily indicate the existence of fraud, paragraph .24 of AU-C section 240 states that they have often been present in circumstances in which frauds have occurred and, therefore, may indicate risks of material misstatement due to fraud.
- **5.135** Paragraph .A31 of AU-C section 240 states that the size, complexity, and ownership characteristics of the entity have a significant influence on the consideration of relevant fraud risk factors. Additional fraud risk factor considerations on large and smaller, less complex entities can be found in paragraphs .A31–.A32 of AU-C section 240.
- **5.136** Appendix A, "Examples of Fraud Risk Factors," of AU-C section 240 identifies examples of fraud risk factors that may be faced by auditors in a broad range of situations. Exhibit 5-1 at the end of this chapter contains a list of fraud risk factors specific to financial institutions. Remember that fraud risk factors are only one of several sources of information an auditor considers when identifying and assessing risks of material misstatement due to fraud.

#### Identification and Assessment of the Risks of Material Misstatement Due to Fraud

- **5.137** In accordance with AU-C section 315, paragraph .25 of AU-C section 240 states that the auditor should identify and assess the risks of material misstatement due to fraud at the financial statement level, and at the assertion level for classes of transactions, account balances, and disclosures.<sup>30</sup> The auditor's risk assessment should be ongoing throughout the audit, following the initial assessment.
- **5.138** Paragraph .26 of AU-C section 240 states that when identifying and assessing the risks of material misstatement due to fraud, the auditor should, based on a presumption that risks of fraud exist in revenue recognition, evaluate which types of revenue, revenue transactions, or assertions give rise to such risks. Paragraph .46 of AU-C section 240 specifies the documentation

<sup>&</sup>lt;sup>30</sup> This requirement provides a link between the auditor's consideration of fraud and assessment of risk and the auditor's procedures in response to those assessed risks.

required when the auditor concludes that the presumption is not applicable in the circumstances of the engagement and, accordingly, has not identified revenue recognition as a risk of material misstatement due to fraud. (See paragraphs .A33–.A35 of AU-C section 240 for application guidance of fraud risks in revenue recognition.<sup>31</sup>)

Considerations for Audits Performed in Accordance With PCAOB  $Standards^{32}$ 

PCAOB Staff Audit Practice Alert No. 12, Matters Related to Auditing Revenue in an Audit of Financial Statements (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance, sec. 400.12), highlights certain requirements of PCAOB standards relating to aspects of auditing revenue in which significant auditing deficiencies have been frequently observed by PCAOB Inspections staff. More specifically, the alert addresses, among other topics, responding to the risks of material misstatement due to fraud associated with revenue.

**5.139** Paragraph .27 of AU-C section 240 states that the auditor should treat those assessed risks of material misstatement due to fraud as significant risks and, accordingly, to the extent not already done so, the auditor should obtain an understanding of the entity's related controls, including control activities, relevant to such risks, including the evaluation of whether such controls have been suitably designed and implemented to mitigate such fraud risks. (See paragraphs .A36—.A37 of AU-C section 240 for application guidance on identifying and assessing the risks of material misstatement due to fraud and understanding the entity's related controls.)

#### Responses to the Assessed Risks of Material Misstatement Due to Fraud

### **Overall Responses**

**5.140** In accordance with AU-C section 330, paragraphs .28–.29 of AU-C section 240 state that the auditor should determine overall responses to address the assessed risks of material misstatement due to fraud at the financial statement level. Accordingly, the auditor should

- assign and supervise personnel, taking into account the knowledge, skill and ability of the individuals to be given significant engagement responsibilities and the auditor's assessment of the risks of material misstatement due to fraud for the engagement;
- b. evaluate whether the selection and application of accounting policies by the entity, particularly those related to subjective measurements and complex transactions, may be indicative of fraudulent financial reporting resulting from management's effort to manage earnings, or a bias that may create a material misstatement; and
- c. incorporate an element of unpredictability in the selection of the nature, timing, and extent of audit procedures.

 $<sup>^{31}</sup>$  For a discussion of indicators of improper revenue recognition and common techniques for overstating revenue and illustrative audit procedures, see the AICPA Audit Guide  $Auditing\ Revenue$  in  $Certain\ Industries$ .

<sup>32</sup> See footnote 2.

See paragraphs .A38—.A42 of AU-C section 240 for additional application guidance on overall responses to the assessed risks of material misstatement due to fraud.

#### Audit Procedures Responsive to Assessed Risks of Material Misstatement Due to Fraud at the Assertion Level

**5.141** In accordance with AU-C section 300, paragraph .30 of AU-C section 240 states that the auditor should design and perform further audit procedures whose nature, timing, and extent are responsive to the assessed risks of material misstatement due to fraud at the assertion level (See paragraphs .A43–.A46 for further application guidance.).

#### Audit Procedures Responsive to Risks Related to Management Override of Controls

**5.142** Even if specific risks of material misstatement due to fraud are not identified by the auditor, paragraph .32 of AU-C section 240 states that a possibility exists that management override of controls could occur. Accordingly, the auditor should address the risk of management override of controls apart from any conclusions regarding the existence of more specifically identifiable risks by designing and performing audit procedures to

- a. test the appropriateness of journal entries recorded in the general ledger and other adjustments made in preparation of the financial statements, including entries posted directly to financial statement drafts,
- b. review accounting estimates for biases and evaluate whether the circumstances producing the bias, if any, represent a risk of material misstatement due to fraud, and
- c. evaluate, for significant transactions that are outside the normal course of business for the entity or that otherwise appear to be unusual given the auditor's understanding of the entity and its environment and other information obtained during the audit, whether the business rationale (or lack thereof) of the transactions suggests that they may have been entered into to engage in fraudulent financial reporting or to conceal misappropriate of assets.
- **5.143** Other audit procedures. Paragraph .33 of AU-C section 240 states that the auditor should determine whether, in order to respond to the identified risks of management override of controls, the auditor needs to perform other audit procedures in addition to those specifically referred to previously (that is, when specific additional risks of management override exist that are not covered as part of the procedures performed to address the requirements in paragraph .32 of AU-C section 240.

#### **Evaluation of Audit Evidence**

**5.144** Paragraphs .34—.37 and .A56—.A62 of AU-C section 240 provide requirements and application guidance for evaluating audit evidence. As stated in paragraph .34 of AU-C section 240, the auditor should evaluate, at or near the end of the audit, whether the accumulated results of auditing procedures, including analytical procedures, that were performed as substantive tests or when forming an overall conclusion, affect the assessment of the risks of

material misstatement due to fraud made earlier in the audit or indicate a previously unrecognized risk of material misstatement due to fraud.

- **5.145** Paragraph .35 of AU-C section 240 states that, if the auditor identifies a misstatement, the auditor should evaluate whether such a misstatement is indicative of fraud. If such an indication exists, the auditor should evaluate the implications of the misstatement with regard to other aspects of the audit, particularly the auditor's evaluation of materiality, management and employee integrity, and the reliability of management representations, recognizing that an instance of fraud is unlikely to be an isolated occurrence. Furthermore, paragraph .36 of AU-C section 240 states that, if the auditor identifies a misstatement, whether material or not, and the auditor has reason to believe that it is, or may be, the result of fraud and that management (in particular, senior management) is involved, the auditor should reevaluate the assessment of the risks of material misstatement due to fraud and its resulting effect on the nature, timing, and extent of audit procedures to respond to the assessed risks. The auditor should also consider whether circumstances or conditions indicate possible collusion involving employees, management, or third parties when reconsidering the reliability of evidence previously obtained.
- **5.146** Paragraph .A60 of AU-C section 240 states that the implications of identified fraud depend on the circumstances. For example, an otherwise insignificant fraud may be significant if it involves senior management. In such circumstances, the reliability of evidence previously obtained may be called into question because there may be doubts about the completeness and truthfulness of representations made and genuineness of accounting records and documentation. There may also be a possibility of collusion involving employees, management, or third parties.
- **5.147** Paragraph .37 of AU-C section 240 states that if the auditor concludes that, or is unable to conclude whether, the financial statements are materially misstated as a result of fraud, the auditor should evaluate the implications for the audit. AU-C sections 450 and 700 address the evaluation and disposition of misstatements and the effect on the auditor's opinion in the auditor's report.

## **Auditor Unable to Continue the Engagement**

- **5.148** Paragraph .38 of AU-C section 240 states that, if, as a result of identified fraud or suspected fraud, the auditor encounters circumstances that bring into question the auditor's ability to continue performing the audit, the auditor should
  - a. determine the professional and legal responsibilities applicable in the circumstances, including whether a requirement exists for the auditor to report to the person or persons who engaged the auditor or, in some cases, to regulatory authorities;
  - consider whether it is appropriate to withdraw from the engagement, when withdrawal is possible under applicable law or regulation; and
  - c. if the auditor withdraws
    - discuss with the appropriate level of management and those charged with governance the auditor's withdrawal

©2016. AICPA AAG-DEP 5.148

from the engagement and the reasons for the withdrawal, and

ii. determine whether a professional or legal requirement exists to report to the person or persons who engaged the auditor or, in some cases, to regulatory authorities, the auditor's withdrawal from the engagement and the reasons for the withdrawal.

Given the nature of the circumstances and the need to consider the legal requirements, paragraph .A65 of AU-C section 240 states that the auditor may consider it appropriate to seek legal advice when deciding whether to withdraw from an engagement and in determining an appropriate course of action, including the possibility of reporting to regulators or others.<sup>33</sup> For additional application guidance, including examples of circumstances that may arise and bring into question the auditor's ability to continue performing the audit, see paragraphs .A63—.A65 of AU-C section 240.

## Communications to Management and With Those Charged With Governance

**5.149** Paragraph .39 of AU-C section 240 states that, if the auditor has identified a fraud or has obtained information that indicates that a fraud may exist, the auditor should communicate these matters on a timely basis to the appropriate level of management in order to inform those with primary responsibility for the prevention and detection of fraud of matters relevant to their responsibilities. As stated in paragraph .A67 of AU-C section 240, this is true even if the matter might be considered inconsequential (for example, a minor defalcation by an employee at a low level in the entity's organization). Unless all of those charged with governance are involved in managing the entity, paragraphs .40-.41 of AU-C section 240 state that, if the auditor has identified or suspects fraud involving (a) management, (b) employees who have significant roles in internal control, or (c) others, when the fraud results in a material misstatement in the financial statements, the auditor should communicate these matters to those charged with governance on a timely basis. If the auditor suspects fraud involving management, the auditor should communicate these suspicions to those charged with governance and discuss with them the nature, timing, and extent of audit procedures necessary to complete the audit. In addition, the auditor should communicate with those charged with governance any other matters related to fraud that are, in the auditor's professional judgment. relevant to their responsibilities. See paragraphs .A68-.A71 of AU-C section 240 for further application guidance concerning communications with those charged with governance.

## Communications to Regulatory and Enforcement Authorities

**5.150** If the auditor has identified or suspects a fraud, paragraph .42 of AU-C section 240 states that the auditor should determine whether the auditor has a responsibility to report the occurrence or suspicion to a party outside the entity. Although the auditor's professional duty to maintain the confidentiality

<sup>&</sup>lt;sup>33</sup> AU-C section 510, Opening Balances—Initial Audit Engagements, Including Reaudit Engagements (AICPA, Professional Standards), provides guidance on communications with an auditor replacing the existing auditor.

of client information may preclude such reporting, the auditor's legal responsibilities may override the duty of confidentiality in some circumstances.

#### **Documentation**

**5.151** Paragraphs .43–.46 of AU-C section 240 address requirements on certain items and events to be documented by the auditor in relation to assessed risks of material misstatement due to fraud.

## **Compliance With Laws and Regulations**

**5.152** AU-C section 250, Consideration of Laws and Regulations in an Audit of Financial Statements (AICPA, Professional Standards), addresses the auditor's responsibility to consider laws and regulations in an audit of financial statements. However, it does not apply to other assurance engagements in which the auditor is specifically engaged to test and report separately on compliance with specific laws and regulations.<sup>34</sup>

## Responsibility for Compliance With Laws and Regulations

#### Responsibility of Management

**5.153** In accordance with paragraph .03 of AU-C section 250, it is the responsibility of management, with the oversight of those charged with governance, to ensure that the entity's operations are conducted in accordance with the provisions of laws and regulations, including compliance with the provisions of laws and regulations that determine the reported amounts and disclosures in an entity's financial statements.

### Responsibility of the Auditor

**5.154** The requirements in AU-C section 250 are designed to assist the auditor in identifying material misstatement of the financial statements due to noncompliance with laws and regulations. However, paragraph .04 of AU-C section 250 recognizes that the auditor is not responsible for preventing noncompliance and cannot be expected to detect noncompliance with all laws and regulations. For purposes of discussion in AU-C section 250, the term *noncompliance* is defined as acts of omission or commission by the entity, either intentional or unintentional, which are contrary to the prevailing laws or regulations.

**5.155** The auditor is responsible for obtaining reasonable assurance that the financial statements as a whole are free from material misstatement, whether caused by fraud or error.<sup>35</sup> In conducting an audit of financial statements, the auditor takes into account the applicable legal and regulatory framework. Because of the inherent limitations of an audit, an unavoidable risk exists that some material misstatements in the financial statements may not be detected, even though the audit is properly planned and performed in accordance

<sup>&</sup>lt;sup>34</sup> AU-C section 935, *Compliance Audits* (AICPA, *Professional Standards*), is applicable when an auditor is engaged, or required by law or regulation, to perform a compliance audit in accordance with GAAS, the standards for financial audits under *Government Auditing Standards*, and a governmental audit requirement that requires an auditor to express an opinion on compliance.

<sup>35</sup> Paragraph .12 of AU-C section 200.

with GAAS.<sup>36</sup> In the context of laws and regulations, the potential effects of inherent limitations on the auditor's ability to detect material misstatements are greater for the reasons set forth in paragraph .05 of AU-C section 250. Paragraph .05 of AU-C section 250 further states that the further removed noncompliance is from the events and transactions reflected in the financial statements, the less likely the auditor is to become aware of, or recognize, the noncompliance.

- **5.156** Paragraph .06 of AU-C section 250 distinguishes the auditor's responsibilities regarding compliance with the following two categories of laws and regulations:
  - a. The provisions of those laws and regulations generally recognized to have a direct effect on the determination of material amounts and disclosures in the financial statements, such as tax and pension laws and regulations (see paragraph 5.157)
  - b. The provisions of other laws and regulations that do not have a direct effect on the determination of the amounts and disclosures in the financial statements but compliance with which may be
    - i. fundamental to the operating aspects of the business,
    - ii. fundamental to an entity's ability to continue its business, or
    - iii. necessary for the entity to avoid material penalties

(for example, compliance with the terms of an operating license, regulatory solvency requirements, or environmental regulations); therefore, noncompliance with such laws and regulations may have a material effect on the financial statements (see paragraphs 5.158–.160).

## The Auditor's Consideration of Compliance With Laws and Regulations

**5.157** Paragraph .A9 of AU-C section 250 states that certain laws and regulations are well established, known to the entity and within the entity's industry or sector, and relevant to the entity's financial statements. These laws and regulations generally are directly relevant to the determination of material amounts and disclosures in the financial statements and readily evident to the auditor. They could include those that relate to, for example

- tax laws affecting accruals and the amount recognized as expense in the accounting period.
- certain laws and regulations placing limits on the nature or amount of investments that institutions are permitted to hold. Such laws and regulations may affect the classification and valuation of assets.

For such laws and regulations, paragraph .13 of AU-C section 250 states that the auditor should obtain sufficient appropriate audit evidence regarding material amounts and disclosures in the financial statements that are determined by the provisions of those laws and regulations (see paragraph 5.156a).

<sup>&</sup>lt;sup>36</sup> Paragraph .A49 of AU-C section 200.

## Procedures to Identify Instances of Noncompliance — Other Laws and Regulations

- **5.158** As discussed in paragraphs .A12–.A14 of AU-C section 250, certain other laws and regulations may need particular attention by the auditor because they have a fundamental effect on the operations of the entity. Noncompliance with laws and regulations that have a fundamental effect on the operations of the entity may cause the entity to cease operations or call into question the entity's continuance as a going concern (for example, noncompliance with capital or investment requirements).
- **5.159** In addition, many laws and regulations relating principally to an institution's operating aspects do not directly affect the financial statements (their financial statement effect is indirect) and are not captured by the entity's information systems relevant to financial reporting. Their indirect effect may result from the need to disclose a contingent liability because of the allegation or determination of identified or suspected noncompliance. Those other laws or regulations may include those related to securities trading, occupational safety and health, food and drug administration, environmental protection, equal employment opportunities, and price-fixing or other antitrust violations.
- **5.160** For these other such laws and regulations, paragraph .14 of AU-C section 250 states that the auditor should perform the following audit procedures that may identify instances of noncompliance with other laws and regulations that may have a material effect on the financial statements (see paragraph 5.156b):
  - a. Inquiring of management and, when appropriate, those charged with governance about whether the entity is in compliance with such laws and regulations
  - b. Inspecting correspondence, if any, with the relevant licensing or regulatory authorities (additional application and explanatory material can be found at paragraph .A16 of AU-C section 250)

However, even when those procedures are performed, the auditor may not become aware of the existence of noncompliance unless there is evidence of noncompliance in the records, documents, or other information normally inspected in an audit of financial statements.

## Noncompliance Brought to the Auditor's Attention By Other Audit Procedures

- **5.161** During the audit, paragraph .15 of AU-C section 250 states that the auditor should remain alert to the possibility that other audit procedures applied may bring instances of noncompliance or suspected noncompliance with laws and regulations to the auditor's attention. For example, paragraph .A17 of AU-C section 250 states that such audit procedures may include reading minutes; inquiring of the institution's management and in-house or external legal counsel concerning litigation, claims, and assessments; performing substantive tests of details of classes of transactions, account balances, or disclosures.
- **5.162** Further discussion regarding audit procedures when noncompliance is identified or suspected, reporting of identified or suspected noncompliance, and documentation requirements can be found in paragraphs .17–.28 of AU-C section 250.

## Going-Concern Considerations <sup>37</sup>

**5.163** AU-C section 570, The Auditor's Consideration of an Entity's Ability to Continue as a Going Concern (AICPA, Professional Standards), addresses the auditor's responsibilities in an audit of financial statements with respect to evaluating whether there is substantial doubt about the entity's ability to continue as a going concern. This section applies to all audits of financial statements, regardless of whether the financial statements are prepared in accordance with a general purpose or a special purpose framework. This section does not apply to an audit of financial statements based on the assumption of liquidation (for example, when [a] an entity is in the process of liquidation, [b] the owners have decided to commence dissolution or liquidation, or [c] legal proceedings, including bankruptcy, have reached a point at which dissolution or liquidation is probable). The auditor's evaluation of an institution's ability to continue as a going concern may be one of the most complex and important portions of the audit. This section describes the unique issues that an auditor may encounter in evaluating an institution's ability to continue as a going concern.

Considerations for Audits Performed in Accordance With PCAOB Standards<sup>38</sup>

PCAOB Staff Audit Practice Alert No. 13, Matters Related to the Auditor's Consideration of a Company's Ability to Continue as a Going Concern (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance, sec. 400.13), addresses the professional standards applicable to the auditor's evaluation of a company's ability to continue as a going concern in light of recent changes to GAAP. The alert specifically highlights that in addition to adhering to the existing requirements in the PCAOB's interim auditing standard AS 2415, Consideration of an Entity's Ability to Continue as a Going Concern (AICPA, PCAOB Standards and Related Rules), auditors should assess management's going concern evaluation in accordance with the requirements of the applicable financial reporting framework.

**5.164** Financial institutions operate in a highly regulated environment. As a result, laws and regulations can have a significant effect on their operations. The enactment of the Financial Institutions Reform, Recovery, and Enforcement Act of 1989 and the FDIC Improvement Act of 1991 dramatically changed the regulatory environment in the banking and thrift industries and imposed new regulatory capital requirements that are far more stringent than previous requirements. Chapter 1 of this guide includes a discussion of regulatory capital requirements for banks and savings institutions and such requirements for credit unions are discussed in chapter 2 of this guide.

<sup>&</sup>lt;sup>37</sup> FASB ASU No. 2014-15, Presentation of Financial Statements—Going Concern (Subtopic 205-40) (Topic 205): Disclosure of Uncertainties about an Entity's Ability to Continue as a Going Concern, issued in August 2014, requires management to assess an entity's ability to continue as a going concern by incorporating and expanding upon certain principles that are currently in U.S. auditing standards. Specifically, the amendments (1) provide a definition of the term substantial doubt, (2) require an evaluation every reporting period including interim periods, (3) provide principles for considering the mitigating effect of management's plans, (4) require certain disclosures when substantial doubt is alleviated as a result of consideration of management's plans, (5) require an express statement and other disclosures when substantial doubt is not alleviated, and (6) require an assessment for a period of one year after the date that the financial statements are issued (or available to be issued).

FASB ASU No. 2014-15 is effective for the annual period ending after December 15, 2016, and for annual periods and interim periods thereafter. Early application is permitted.

<sup>38</sup> See footnote 2.

## **Evaluating Whether Substantial Doubt Exists**

**5.165** In accordance with paragraph .08 of AU-C section 570, the auditor should evaluate whether there is substantial doubt about an entity's ability to continue as a going concern for a *reasonable period of time* (defined in AU-C section 570 as a period of time not to exceed one year beyond the date of the financial statements being audited) based on the results of the audit procedures.

**5.166** When the applicable financial reporting framework includes a definition of substantial doubt about an entity's ability to continue as a going concern, Interpretation No. 1, "Definition of Substantial Doubt About an Entity's Ability to Continue as a Going Concern," of AU-C section 570 (AICPA, Professional Standards, AU-C sec. 9570, par. 01–.02), states that definition would be used by the auditor when applying the requirements of AU-C section 570. Interpretation No. 2, "Definition of Reasonable Period of Time," of AU-C section 570 (AICPA, Professional Standards, AU-C sec. 9570, par. 03–.05), provides guidance on how an auditor should apply the term reasonable period of time when the applicable financial reporting framework requires management to evaluate whether there are conditions and events that raise substantial doubt for a period of time greater than one year from the date of the financial statements. Specifically, Interpretation No. 2 states that the auditor's assessment of management's going concern evaluation would be for the same period of time as required by the applicable financial reporting framework.

### Identifying Conditions or Events That Indicate Substantial Doubt Could Exist

**5.167** As stated in paragraph .09 of AU-C section 570, the auditor should consider whether the results of procedures performed during the course of the audit identify conditions and events that, when considered in the aggregate, indicate there could be substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time. The auditor should consider the need to obtain additional information about such conditions and events, as well as the appropriate audit evidence to support information that mitigates the auditor's doubt.

**5.168** Paragraph .A1 of AU-C section 570 states that it is not necessary to design audit procedures solely to identify conditions or events that, when considered in the aggregate, indicate there could be substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time. The results of audit procedures designed and performed to identify and assess risk in accordance with AU-C section 315, gather audit evidence in response to assessed risks in accordance with AU-C section 330, and complete the audit are expected to be sufficient for that purpose. The following are examples of procedures normally performed in audits of the financial statements of financial institutions that may identify such conditions and events:

- Analytical procedures
- Review of subsequent events
- Review of compliance with the terms of debt and loan agreements
- Reading of minutes of meetings of stockholders, board of directors, and important committees of the board
- Inquiry of an entity's legal counsel about litigation, claims, and assessments

©2016, AICPA AAG-DEP 5.168

- Confirmation with related and third parties of the details of arrangements to provide or maintain financial support
- Review of the financial strength and liquidity of the parent company, if applicable
- Review of loans maturing in less than one year and entity's ability to refinance or pay off the loan
- Review of reports of significant examinations and related communications between examiners and the institution
- Review of compliance with regulatory capital requirements

**5.169** In performing such audit procedures as noted previously, paragraph .A2 of AU-C section 570 states that the auditor may identify information about certain conditions or events that, when considered in the aggregate, indicate there could be substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time. The significance of such conditions or events will depend on the circumstances, and some conditions or events may have significance only when viewed in conjunction with others. The following are examples of such conditions and events that may be encountered in audits of financial institutions:

- Recurring operating losses
- Indications of strained liquidity
- Failure to meet minimum regulatory capital requirements or to adhere to the terms of an approved capital plan
- Concerns expressed or actions taken by regulatory authorities regarding alleged unsafe or unsound practices
- Indications of strained relationships between management and regulatory authorities

### Considerations of Management's Plans When the Auditor Believes There Is Substantial Doubt

**5.170** If, after considering the identified conditions or events in the aggregate, the auditor believes that there is substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time, paragraph .10 of AU-C section 570 states that the auditor should obtain information about management's plans that are intended to mitigate the adverse effects of such conditions or events. The auditor should

- a. assess whether it is likely that the adverse effects would be mitigated by management's plans for a reasonable period of time;
- b. identify those elements of management's plans that are particularly significant to overcoming the adverse effects of the conditions or events and plan and perform procedures to obtain audit evidence about them, including, when applicable, considering the adequacy of support regarding the ability to obtain additional financing or the planned disposal of assets; and
- c. assess whether it is likely that such plans can be effectively implemented.
- **5.171** When prospective financial information is particularly significant to management's plans, paragraph .11 of AU-C section 570 states that the auditor should request management to provide that information and should

consider the adequacy of support for significant assumptions underlying that information. The auditor should give particular attention to assumptions that are

- material to the prospective financial information.
- especially sensitive or susceptible to change.
- inconsistent with historical trends.

The auditor's consideration should be based on knowledge of the entity, its business, and its management and should include (a) reading the prospective financial information and the underlying assumptions and (b) comparing prospective financial information from prior periods with actual results and comparing prospective information for the current period with results achieved to date. If the auditor becomes aware of factors, the effects of which are not reflected in such prospective financial information, the auditor should discuss those factors with management and, if necessary, request revisions of the prospective financial information.

### Consideration of Financial Statement Effects

**5.172** Paragraph .12 of AU-C section 570 states that when, after considering management's plans, the auditor concludes that there is substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time, the auditor should consider the possible effects on the financial statements and the adequacy of the related disclosures. In considering the adequacy of disclosure, paragraph .A4 of AU-C section 570 states that some of the information that might be disclosed includes the following:

- Principal conditions or events giving rise to the assessment of substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time
- The possible effects of such conditions or events
- Management's evaluation of the significance of those conditions or events and any mitigating factors
- Possible discontinuance of operations
- Management's plans (including relevant prospective financial information)
- Information about the recoverability or classification of recorded asset amounts or the amounts or classification of liabilities

**5.173** When the auditor concludes, primarily because of the auditor's consideration of management's plans, that substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time has been alleviated, paragraph .13 of AU-C section 570 states that the auditor should consider the need for, and evaluate the adequacy of, disclosure of the principal conditions or events that initially caused the auditor to believe there was substantial doubt. The auditor's consideration of disclosure should include the possible effects of such conditions and events, and any mitigating factors, including management's plans. The auditor may have to communicate with the regulator to assist with the auditor's assessment. (Refer to chapter 1 of this guide for a discussion of necessary communications with regulators.) Chapter 23 of this guide includes an illustration of a report that includes such an emphasis-of-matter paragraph.

©2016, AICPA AAG-DEP 5.173

**5.174** When the applicable financial reporting framework provides disclosure requirements related to management's evaluation of substantial doubt, Interpretation No. 4, "Consideration of Financial Statement Effects," of AUC section 570 (AICPA, *Professional Standards*, AU-C sec. 9570, par. 09–.10), states that the auditor's assessment of the financial statement effects under AU-C section 570 would be based on the disclosure requirements of the applicable financial reporting framework.

## Written Representations

- **5.175** If the auditor believes, before consideration of management's plans pursuant to paragraph .10 of AU-C section 570 (see paragraph 5.170), there is substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time, paragraph .14 of AU-C section 570 states that the auditor should obtain written representations from management
  - a. regarding its plans that are intended to mitigate the adverse effects of conditions or events that indicate there is substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time and the likelihood that those plans can be effectively implemented, and
  - b. that the financial statements disclose all the matters of which management is aware that are relevant to the entity's ability to continue as a going concern, including principal conditions or events and management's plans.

## Consideration of the Effects on the Auditor's Report

**5.176** Paragraphs .15—.16 of AU-C section 570 state that, if, after considering identified conditions and events and management's plans, the auditor concludes that substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time remains, the auditor should include an emphasis of matter paragraph<sup>39</sup> in the auditor's report to reflect that conclusion. The auditor's conclusion about the entity's ability to continue as a going concern should be expressed through the use of the phrase "substantial doubt about its (the entity's) ability to continue as a going concern or similar wording that includes the terms substantial doubt and going concern. In a going concern emphasis-of-matter paragraphs, the auditor should not use conditional language in expressing a conclusion concerning the existence of substantial doubt about the entity's ability to continue as going concern. Paragraph .A6 of AU-C section 570 provides an illustration of a going-concern emphasis-of-matter paragraph.

**5.177** The auditor's decision about whether modification of the standard report is appropriate may depend also on

- the institution's existing regulatory-capital position;
- the likelihood that the institution's regulatory-capital position will improve or deteriorate within the next 12 months;
- whether the plan has been accepted by regulatory authorities; and
- the auditor's assessment of the institution's ability to achieve its capital plan, if any.

 $<sup>^{39}</sup>$  Paragraphs 23.07–.13 of this guide address requirements concerning emphasis-of-matter paragraphs.

**5.178** Chapter 23 of this guide discusses circumstances that the auditor might disclaim an opinion on.

### **Documentation**

- **5.179** If the auditor believes, before consideration of management's plans pursuant to paragraph .10 of AU-C section 570 (see paragraph 5.170), there is substantial doubt about the ability of the entity to continue as a going concern for a reasonable period of time, paragraph .22 of AU-C section 570 states that the auditor should document the following:
  - The conditions or events that led the auditor to believe that there
    is substantial doubt about the entity's ability to continue as a
    going concern for a reasonable period of time.
  - The elements of management's plans that the auditor considered to be particularly significant to overcoming the adverse effects of the conditions or events.
  - The auditing procedures performed to evaluate the significant elements of management's plans and evidence obtained.
  - The auditor's conclusion as to whether substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time remains or is alleviated. If substantial doubt remains, the auditor also should document the possible effects of the conditions or events on the financial statements and the adequacy of the related disclosures. If substantial doubt is alleviated, the auditor also should document the conclusion as to the need for and, if applicable, the adequacy of disclosure of the principal conditions and events that initially caused the auditor to believe there was substantial doubt.
  - The auditor's conclusion with respect to the effects on the auditor's report.

## Written Representations

**5.180** AU-C section 580, *Written Representations* (AICPA, *Professional Standards*), addresses the auditor's responsibility to obtain written representations from management and, when appropriate, those charged with governance in an audit of financial statements.

## Written Representations as Audit Evidence

**5.181** According to paragraphs .03–.04 of AU-C section 580, written representations are necessary information that the auditor requires in connection with the audit of the entity's financial statements. Accordingly, similar to responses to inquiries, written representations are audit evidence. Although written representations provide necessary audit evidence, they complement other auditing procedures and do not provide sufficient appropriate audit evidence on their own about any of the matters with which they deal. Furthermore, obtaining reliable written representations does not affect the nature or extent of other audit procedures that the auditor applies to obtain audit evidence about the fulfillment of management's responsibilities or about specific assertions.

©2016, AICPA AAG-DEP 5.181

# Management From Whom Written Representations Are Requested

**5.182** As explained in paragraph .A2 of AU-C section 580, written representations are requested from those with overall responsibility for financial and operating matters whom the auditor believes are responsible for, and knowledgeable about, directly or through others in the organization, the matters covered by the representations, including the preparation and fair presentation of the financial statements. As such, in accordance with paragraph .09 of AU-C section 580, the auditor should request written representations from management with appropriate responsibilities for the financial statements and knowledge of the matters concerned.

**5.183** Paragraph .A2 of AU-C section 580 further states that those individuals with overall responsibility may vary depending on the governance structure of the entity; however, management (rather than those charged with governance) is often the responsible party. Written representations may therefore be requested from the entity's chief executive officer and chief financial officer or other equivalent persons in entities that do not use such titles. In some circumstances, however, other parties, such as those charged with governance, also are responsible for the preparation and fair presentation of the financial statements.

# Written Representations About Management's Responsibilities and Other Written Representations

**5.184** Paragraphs .10–.18 of AU-C section 580 discuss matters the auditor should request management to provide written representation about such as preparation and fair presentation of the financial statements, information provided and completeness of transactions, fraud, laws and regulations, uncorrected misstatements, litigation an claims, estimates, related party transactions, and subsequent events. If, in addition to such required representations and those addressed in other AU-C sections, <sup>40</sup> the auditor determines that it is necessary to obtain one or more written representations to support other audit evidence relevant to the financial statements or one or more specific assertions in the financial statements, paragraph .19 of AU-C section 580 states that the auditor should request such other written representations.

**5.185** Additional representations specific to banks and savings institutions, credit unions, or both that may be obtained include the following:

- All regulatory examination reports, supervisory correspondence, and similar materials from applicable regulatory agencies (particularly communications concerning supervisory actions or noncompliance with or deficiencies in the rules and regulations or supervisory actions) have been provided to the auditor.
- The classification of securities between held-to-maturity, available-for-sale, or trading categories accurately reflects management's ability and intent.

<sup>&</sup>lt;sup>40</sup> As stated in paragraph .02 of AU-C section 580, Written Representations (AICPA, Professional Standards), exhibit D, "List of AU-C Sections Containing Requirements for Written Representations," lists other AU-C sections containing subject matter-specific requirements for written representations. The specific requirements for written representations of other AU-C sections do not limit the application of this section.

- The methodology for determining fair value disclosures is based on reasonable assumptions.
- Adequate disclosure has been made of the status of the institution's capital plan filed with regulators, if applicable, and management believes it is in compliance with any formal agreements or orders in any memorandum of understanding or cease-and-desist order.
- Contingent assets and liabilities have been adequately disclosed in the financial statements.
- Related-party transactions have been entered into in compliance with existing regulations.
- Adequate provision has been made for any losses, costs, or expenses that may be incurred on securities, loans, or leases and real estate as of the balance sheet date.
- Other than temporary declines in the value of investment securities have been properly recognized in the financial statements.
- Commitments to purchase or sell securities under forwardplacement, financial-futures contracts, and standby commitments have been adequately disclosed in the financial statements.
- Sales with recourse have been adequately disclosed in the financial statements.
- Proper disclosure has been made regarding the nature, terms, and credit risk of financial instruments with off-balance-sheet risk.
- No transactions or activities are planned that would result in any recapture of the base-year, tax-basis bad debt reserves.
- Proper disclosure has been made regarding financial instruments with significant
  - off-balance-sheet risk and
  - individual or group concentrations of credit risk.

**5.186** Paragraph .A22 of AU-C section 580 states that management's representations may be limited to matters that are considered either individually or collectively material to the financial statements, provided management and the auditor have reached an understanding on materiality for this purpose. Materiality may be different for different representations. A discussion of materiality may be included explicitly in the representation letter in either qualitative or quantitative terms. Materiality considerations do not apply to those representations that are not directly related to amounts included in the financial statements (for example, management's representations about the premise underlying the audit). In addition, because of the possible effects of fraud on other aspects of the audit, materiality would not apply to management's acknowledgment regarding its responsibility for the design, implementation, and maintenance of internal control to prevent and detect fraud.

## Date of, and Period(s) Covered by, Written Representations

 $\bf 5.187$  Paragraph .20 of AU-C section 580 states that the date of the written representations should be as of the date of the auditor's report on the financial statements. The written representations should be for all financial statements and period(s) referred to in the auditor's report.

©2016. AICPA AAG-DEP 5.187

## Form of Written Representations

**5.188** In accordance with paragraph .21 of AU-C section 580, the written representations should be in the form of a representation letter addressed to the auditor.

# Doubt About the Reliability of Written Representations and Requested Written Representations Not Provided

- **5.189** Paragraph .25 of AU-C section 580 states that the auditor should disclaim an opinion on the financial statements in accordance with AU-C section 705 or withdraw from the engagement if
  - a. the auditor concludes that sufficient doubt exists about the integrity of management such that the written representations required by paragraphs .10-.11 of AU-C section 580 are not reliable or
  - b. management does not provide the written representations required by paragraphs .10-.11 of AU-C section 580.

### Information Other Than Financial Statements

- **5.190** An institution may publish various documents that contain information in addition to audited financial statements and the auditor's report thereon. AU-C section 720, Other Information in Documents Containing Audited Financial Statements (AICPA, Professional Standards), addresses the auditor's responsibility with respect to other information in documents containing audited financial statements and the auditor's report thereon. In the absence of any separate requirement in the particular circumstances of the engagement, the auditor's opinion on the financial statements does not cover other information, and the auditor has no responsibility for determining whether such information is properly stated. This section establishes the requirement for the auditor to read the other information of which the auditor is aware because the credibility of the audited financial statements may be undermined by material inconsistencies between the audited financial statements and other information.
- **5.191** In some circumstances, an auditor submits to the client or others a document that contains information in addition to the client's basic financial statements and the auditor's report thereon. AU-C section 725, Supplementary Information in Relation to the Financial Statements as a Whole (AICPA, Professional Standards), addresses the auditor's responsibility when engaged to report on whether supplementary information is fairly stated, in all material respects, in relation to the financial statements as a whole. The information covered by this section is presented outside the basic financial statements and is not considered necessary for the financial statements to be fairly presented in accordance with the applicable financial reporting framework. This section also may be applied, with the report wording adapted as necessary, when an auditor has been engaged to report on whether required supplementary information is fairly stated, in all material respects, in relation to the financial statements as a whole.
- **5.192** AU-C section 730, Required Supplementary Information (AICPA, Professional Standards), addresses the auditor's responsibility with respect to

information that a designated accounting standards setter requires to accompany an entity's basic financial statements (hereinafter referred to as *required supplementary information*). In the absence of any separate requirement in the particular circumstances of the engagement, the auditor's opinion on the basic financial statements does not cover required supplementary information.

## **Certain Financial Reporting Matters**

## **Disclosures of Certain Significant Risks and Uncertainties**

**5.193** FASB ASC 275-10-50-1<sup>41</sup> requires institutions to make disclosures in their financial statements about the risks and uncertainties existing as of the date of those statements in the following areas:

- a. The nature of their operations, including the activities in which the entity is currently engaged if principal operations have not commenced
- b. The use of estimates in the preparation of their financial statements
- c. Certain significant estimates
- d. Current vulnerability due to certain concentrations
- **5.194** An illustration of the application of these disclosure requirements by a bank or savings institution follows:

*Nature of operations.* ABC Institution operates seven branches in rural and suburban communities in the United States Midwest. The Institution's primary source of revenue is providing loans to customers that are predominantly small and middle-market businesses and middle-income individuals.

Use of estimates in the preparation of financial statements. The preparation of financial statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the financial statements and that affect the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

**5.195** The application of these disclosure requirements by a bank or savings institution is discussed and illustrated in the following paragraphs.

## **Certain Significant Estimates**

**5.196** As explained in FASB ASC 275-10-50-7, disclosures are required regarding estimates used in the determination of the carrying amounts of assets or liabilities or in disclosure of gain or loss contingencies, as described herein. FASB ASC 275-10-50-8 goes on to state that disclosure regarding an

©2016, AICPA AAG-DEP 5.196

<sup>&</sup>lt;sup>41</sup> See also paragraphs 1–2 of FASB ASC 825-10-55 for a discussion on concentrations involving loan product terms. The terms of certain loan products may increase a reporting entity's exposure to credit risk and thereby may result in a concentration of credit risk as that term is used in FASB ASC 825–10, either as an individual product type or as a group of products with similar features. Except as indicated in FASB ASC 825–10–50–22, FASB ASC 825–10–50–20 states that an entity should disclose all significant concentrations of credit risk arising from financial instruments, whether from an individual counterparty or groups of counterparties.

estimate should be made when known information available before the financial statements are issued or are available to be issued (as discussed in FASB ASC 855-10-25) indicates that both of the following criteria are met:

- a. It is at least reasonably possible that the estimate of the effect on the financial statements of a condition, situation, or set of circumstances that existed at the date of the financial statements will change in the near term due to one or more future confirming events.
- b. The effect of the change would be material to the financial statements
- **5.197** In accordance with FASB ASC 275-10-50-9, the disclosure should indicate the nature of the uncertainty and include an indication that it is at least reasonably possible that a change in the estimate will occur in the near term. If the estimate involves a loss contingency covered by FASB ASC 450-20, the disclosure also should include an estimate of the possible loss or range of loss, or state that such an estimate cannot be made. <sup>42</sup>
- **5.198** Following is an illustrative disclosure about the allowance for loan losses when no uncertainties meet the disclosure criteria established in FASB ASC 275-10-50-8 and FASB ASC 450-20-50-3.

Allowance for loan losses. The allowance for loan losses is established as losses are estimated to have occurred through a provision for loan losses charged to earnings. Loan losses are charged against the allowance when management believes the uncollectibility of a loan balance is confirmed. Subsequent recoveries, if any, are credited to the allowance.

The allowance for loan losses is evaluated on a regular basis by management and is based upon management's periodic review of the collectibility of the loans in light of historical experience, the nature and volume of the loan portfolio, adverse situations that may affect the borrower's ability to repay, estimated value of any underlying collateral, and prevailing economic conditions. This evaluation is inherently subjective as it relies on estimates that are susceptible to significant revision as more information becomes available.

**5.199** The following illustrates a paragraph that might be added to the illustrative disclosure in paragraph 5.198 to disclose an uncertainty that meets the disclosure criteria of FASB ASC 275-10-50-8, is a loss contingency covered by FASB ASC 450-20, and affects the estimate of loan losses for only some portion of the institution's loan portfolio:

<sup>&</sup>lt;sup>42</sup> FASB ASC 450-20-50-3 requires reporting entities to disclose certain contingencies, if there is at least a reasonable possibility that a loss or an additional loss may have been incurred and either of the following conditions exist:

a. An accrual is not made for a loss contingency because any of the conditions in FASB ASC 450-20-25-2 are not met.

 $b. \;\;$  An exposure to loss exists in excess of the amount accrued pursuant to the provisions of FASB ASC 450-20-30-1.

As stated in FASB ASC 450-20-50-4, the disclosure in FASB ASC 450-20–50–3 should include both of the following:

a. The nature of the contingency

b. An estimate of the possible loss or range of loss or a statement that such an estimate cannot be made

Three of the Institution's seven branches are in communities that were flooded in late 200X. These branches made loans to individuals and businesses affected by the flooding and the Institution considered the flood's effect in determining the adequacy of the allowance for loan losses. No estimate can be made of a range of amounts of loss that are reasonably possible with respect to that event. <sup>43</sup>

**5.200** The following illustrates a paragraph that might be added to the illustration in paragraph 5.198 to disclose an uncertainty that meets the disclosure criteria of FASB ASC 275-10-50-8 and is a loss contingency covered by FASB ASC 450-20:

The Institution lends primarily to individuals employed at ABC Air Force Base and businesses local to the base. On December 19, 20X3, the President of the United States ratified a plan that includes the closing of the base effective November 20X4. It is reasonably possible that a change in estimated loan losses will occur in the near term. No estimate can be made of a range of amounts of loss that are reasonably possible with respect to the base closing.

- **5.201** FASB ASC 275-10-50-15 gives examples of assets and liabilities and related revenues and expenses, and of disclosure of gain or loss contingencies included in financial statements that, based on facts and circumstances existing at the date of the financial statements, may be based on estimates that are particularly sensitive to change in the near term.
- **5.202** Besides valuation allowances for loans, examples of similar estimates often included in banks', savings institutions', and credit unions' financial statements include the following:
  - Impairment of long-lived assets, for example, assets related to marginal branches
  - Estimates involving assumed prepayments, for example, discounts or premiums on certain financial assets (such as securities or loans), mortgage servicing rights and excess servicing receivables, and mortgage related securities
  - Lives of identifiable intangible assets (for example, depositor or borrower relationships)

**5.203** For example, during 20X5, DEF Bank evaluated the profitability of its branch operations. DEF Bank determined that it will significantly change the extent or manner in which it uses a group of long-lived assets related to six of its branches. In applying FASB ASC 360, *Property, Plant, and Equipment*, DEF Bank determined that the sum of the estimated future cash flows (cash inflows less associated cash outflows) that are directly associated with and that are expected to arise as a direct result of the use and eventual disposition of the asset group, excluding interest charges, exceeds the carrying amount of the long-lived asset group. In addition, the carrying amount of the asset group does not exceed its fair value. Thus, an impairment loss has not been recognized

©2016, AICPA AAG-DEP 5.203

<sup>&</sup>lt;sup>43</sup> If a range of possible loss can be estimated, the last sentence might say: It is reasonably possible that in the near term loan losses with respect to that event could be \$5 million to \$7 million more than estimated in the allowance for loan losses. If the possible loss can be estimated, the last sentence might say: It is reasonably possible that in the near term loan losses with respect to that event could be \$6 million more than estimated in the allowance for loan losses.

under FASB ASC 360. The significant change in the extent or manner in which the assets are used, however, indicates that the estimate associated with the carrying amounts of those assets may be particularly sensitive in the near term. <sup>44</sup> Following is an illustrative disclosure:

Management of DEF Bank has reevaluated and will significantly change its use of a group of long-lived assets associated with six of its branches. It is reasonably possible that the Bank's estimate of the carrying amounts of these assets will change in the near term. No estimate can be made of a range of amounts of loss that are reasonably possible.

## **Current Vulnerability Due to Certain Concentrations**

**5.204** FASB ASC 275-10-50-16 requires institutions to disclose the concentrations described in FASB ASC 275-10-50-18 if, based on information known to management before the financial statements are issued or are available to be issued (as discussed in FASB ASC 855-10-25), all of the following criteria are met:

- a. The concentration exists at the date of the financial statements.
- b. The concentration makes the institution vulnerable to the risk of a near-term severe impact.
- c. It is at least reasonably possible that the events that could cause the severe impact will occur in the near term.

**5.205** FASB ASC 275, *Risks and Uncertainties*, does not address concentrations of financial instruments. However, as discussed in chapter 7, chapter 8, "Loans," and chapter 18 of this guide, and elsewhere in this guide, FASB ASC 825, *Financial Instruments*, includes the disclosure provisions about concentrations of credit risk.<sup>45</sup>

**5.206** The following concentrations described in FASB ASC 275-10-50-18 require disclosure if they meet the criteria of FASB ASC 275-10-50-16:

- a. Concentrations in the volume of business transacted with a particular customer, supplier, lender, grantor, or contributor
- Concentrations in revenue from particular products, services, or fund-raising events
- c. Concentrations in the available sources of supply of materials, labor, or services, or of licenses or other rights used in the entity's operations
- d. Concentrations in the market or geographic area in which an entity conducts its operations

**5.207** Examples of concentrations that may fall in one or more of these categories and that may exist at certain financial institutions include

 sale of a substantial portion of or all receivables or loan products to a single customer;

<sup>&</sup>lt;sup>44</sup> FASB ASC 360-10-35-21 requires that a long-lived asset (asset group) be tested for recoverability whenever events or changes in circumstances indicate that its carrying amount may not be recoverable. A significant adverse change in the extent or manner in which a long-lived asset (asset group) is being used or in its physical condition is an example of such an event or change in circumstances.

<sup>&</sup>lt;sup>45</sup> See footnote 41.

- loss of approved status as a seller to or servicer for a third party;
- concentration of revenue from issuances involving a third-party guarantee program;
- concentration of revenue from mortgage banking activities; and
- in the case of a credit union, membership in the institution is concentrated with employees of a specific industry or in a region.

**5.208** For example, assume a significant portion of GHI Institution's net income is from sales of originated loans. In 20X5, GHI Institution originated \$800 million of loans. GHI Institution sold the loans and servicing rights to a substantial portion of these loans to a single servicer, TCB. TCB has historically purchased a substantial portion of the loans and servicing originated by GHI Institution. Following is an illustrative disclosure:

A substantial portion of GHI Institution's loan and loan-servicingright originations is sold to a single servicer.

**5.209** Assume a significant portion of JKL Bank's revenues is from the origination of loans guaranteed by the Small Business Administration under its Section 7 program and sale of the guaranteed portions of those loans. Funding for the Section 7 program depends on annual appropriations by the U.S. Congress. The customer base for this lending specialization and the resulting profits depend on the continuation of the program. Following is an illustrative disclosure:

A substantial portion of JKL Bank's revenues is from origination of loans guaranteed by the Small Business Administration under its Section 7 program and sale of the guaranteed portions of those loans. Funding for the Section 7 program depends on annual appropriations by the U.S. Congress.

## **Segment Reporting**

**5.210** FASB ASC 280-10 provides guidance to public entities on how to report certain information about operating segments in complete sets of financial statements of the public entity and in condensed financial statements of interim periods issued to shareholders. Refer to FASB ASC 280, *Segment Reporting*, for further discussion and detail regarding segment reporting requirements.

**5.211** Per the FASB ASC glossary, a *public entity* is defined as a business entity or a not-for-profit entity that meets any of the following conditions:

- It has issued debt or equity securities or is a conduit bond obligor for conduit debt securities that are traded in a public market (a domestic or foreign stock exchange or an over-the-counter market, including local or regional markets).
- It is required to file financial statements with the SEC.
- It provides financial statements for the purpose of issuing any class of securities in a public market.

## Regulation and Supervision of Depository Institutions

### Introduction

**5.212** Laws and their implementing regulations affect the areas and ways in which certain financial institutions operate while creating standards with

which those institutions must comply. Some laws and regulations directly address the responsibilities of auditors.  $^{46}$ 

- **5.213** The primary objective of this section is to explain why and how auditors might consider regulatory matters in the audits of certain financial institutions. This chapter also addresses the overall regulatory approach and environment, and the relative responsibilities of those institutions, examiners, and auditors. Considerations auditors might give to specific areas of regulation are highlighted in subsequent chapters.
- ${f 5.214}$  Auditors might consider the effect regulations have on various engagements:
  - a. Acceptance of engagements in the affected industry
  - b. Planning activities (that is, development of the expected conduct and scope of an engagement)
  - c. Responsibility for detection of errors and irregularities
  - d. Evaluation of contingent liabilities and related disclosures
  - Consideration of an institution's ability to continue as a going concern
- **5.215** Paragraph .12 of AU-C section 315 indicates that auditors should obtain an understanding of relevant regulatory factors, including the applicable financial reporting framework. In that regard, it is helpful for auditors to be familiar with the nature and purpose of regulatory examinations—including the differences and relationship between examinations and financial statement audits.
- **5.216** Finally, an understanding of the regulatory environment in which these institutions operate is necessary to complement the auditor's knowledge of existing regulatory requirements. Because the regulatory environment is continually changing, the auditor might consider monitoring relevant regulatory changes and consider their implications in the audit process.
- **5.217** One primary objective of regulation is to maintain the strength of the financial system, in turn, promoting and enforcing the public role of certain financial institutions as financial intermediaries, protecting depositors, and preserving funds for federal deposit insurance. Regulations are generally associated with one or more of the following objectives: capital adequacy, asset quality, management competence, earnings, liquidity, and sensitivity to market risk.
- **5.218** Many laws and areas of regulation address the public role of certain financial institutions. For example, laws and regulations exist to ensure the availability of credit to all creditworthy applicants without discrimination and to satisfy the credit needs of low- and moderate-income neighborhoods in institutions' local communities.
- **5.219** Other regulations address directly these institution's operations and, therefore, have broader financial implications. For example, rules exist

<sup>&</sup>lt;sup>46</sup> Although the discussion in this chapter is focused on federal regulation, it also may be useful in considering state regulatory matters, especially the impact of regulatory matters on the auditor. Further, the guide does not address specific state regulations that may be relevant in the audit of financial statements.

that restrict the acceptance and renewal of brokered deposits based on a bank or savings institution's level of capitalization.

**5.220** In addition to the specific regulatory matters outlined in subsequent chapters, the three aspects of the regulatory process that are particularly important to auditors are rule making, examinations, and enforcement.

## **Rule Making**

**5.221** Regulations are created by the agencies based on their ongoing authority or as specifically mandated by legislation. Proposed rules and regulations are generally published for comment in the *Federal Register*, a daily publication of the federal government. Final rules also appear in the *Federal Register* and are codified in Title 12, *Banks and Banking*, of U.S. CFR. The *Federal Register* may be accessed at the Government Printing Office website. The rules applicable to a given institution depend on the institution's charter and other factors, such as whether it is federally insured and whether it is a member of the Federal Reserve System. Institutions are informed of new rules, policies, and guidance through publications of the agencies.

**5.222** Discussions of specific regulatory matters found throughout this guide should not be substituted for a complete reading of related regulations, rulings, or other documents where appropriate. It is important for auditors to keep apprised of recent changes in regulations, as the regulatory environment is constantly changing.

### **Examinations**

**5.223** As used in this guide, the term *audit* refers to an audit performed by an auditor for the purpose of expressing an opinion on an institution's financial statements, unless the context in which the term is used clearly indicates that the reference is to an internal audit. The term *examination* generally refers to an examination made by a regulatory authority. There are several types of regulatory examinations, including a Safety and Soundness Examination, an Information Systems Examination, a Trust Examination and a Compliance Examination. These examinations may be combined or performed separately. The purpose of the regulatory examination is to determine the safety and soundness of an institution. The term *examiner* as used in this guide means those individuals—acting on behalf of a regulatory agency—responsible for supervising the performance or preparation of reports of examination and, when appropriate, supervisory personnel at the district and national level.

**5.224** Federally insured financial institutions are required to have periodic full-scope, on-site examinations by the appropriate agency. In some cases the OCC and the Federal Reserve will perform off site examinations. In certain cases, an examination by a state regulatory agency is accepted. Full-scope and other examinations are intended primarily to provide early identification of problems at insured institutions rather than as a basis for expressing an opinion on fair presentation of an institution's financial statements.

**5.225** The scope of an examination is generally unique to each institution based on risk factors assessed by the examiner; however, general areas that might be covered include the following:

- Capital adequacy
- Asset quality

- Management
- Earnings
- Liquidity
- Sensitivity to market risk
- Funds management
- Internal systems and controls
- Consumer affairs
- Electronic data processing
- Fiduciary activities

**5.226** Examinations are sometimes targeted to a specific area of operations. Separate compliance examination programs also exist to address institutions' compliance with laws and regulations in areas such as consumer protection, insider transactions, and reporting under the Bank Secrecy and USA Patriot Acts.

**5.227** An examination generally begins with a review of various background material and information, including practices, policies or procedures established by an institution. The examiner compares these practices, policies, or procedures to regulatory and supervisory requirements and assesses the institution's adherence to sound fundamental principles in its day-to-day operations. Any additional detailed procedures considered necessary are then applied. A written report of procedures and findings is then prepared by the examiner. The relationship between the work of the examiner and that of the auditor is further discussed in the following paragraph.

**5.228** Results of examinations are also used in assigning the institution a rating under regulatory rating systems. The FFIEC has adopted the Uniform Financial Institutions Rating System, which bases an institution's composite CAMELS (the rating on component factors addressing capital adequacy, asset quality, management, earnings, liquidity, and sensitivity to market risk). Further, the Federal Reserve assigns BOPEC (the rating stands for the five key areas of supervisory concern: the condition of the BHC's bank subsidiaries, other nonbank subsidiaries, parent company, earnings, and capital adequacy) ratings to bank holding companies based on consideration of the bank's CAMELS rating, operation of significant nonbanking subsidiaries, the parent's strength and operations, earnings of the banking organization, and capital of the banking organization. Both systems involve a five-point rating scale, with one being the highest possible rating.

### **Enforcement**

**5.229** Regulatory enforcement is sometimes carried out through a written agreement between the regulator and the institution—ranging from the least severe commitment letter to a cease-and-desist order. Among other actions that can be taken, the agencies may enforce regulations by

- ordering an institution to cease and desist from certain practices or violations:
- removing an officer or prohibiting an officer from participating in the affairs of the institution or the industry;
- assessing civil money penalties; and
- terminating insurance of an institution's deposits.

- **5.230** The examination focus has shifted from complete reliance on transaction testing to an assessment of risks and each of the agencies has issued guidance on "supervision by risk," under which examiners identify the risks a bank faces and evaluate how the institution manages those risks. Derivative activities (including the use of credit derivatives), as well as the trading activities of banks, have also received increased scrutiny. In addition, recent losses involving fraud have led to a reemphasis on the identification of significant internal control weaknesses and other potential indicators of fraud.
- **5.231** Further, insured financial institutions may be subject to other mandatory and discretionary actions taken by regulators under prompt corrective action provisions of the FDI Act and the Federal Credit Union Act (FCUA). As described in chapters 1 and 2 of this guide, possible actions range from the restriction or prohibition of certain activities to appointment of a receiver or conservator of the institution's net assets.
- **5.232** Many enforcement actions—such as civil money penalties—apply not only to an insured financial institution but also to a broader class of institution-affiliated parties, which could include auditors. For example, regulatory agencies may assess civil money penalties of up to \$1 million<sup>47</sup> per day against an institution or institution-affiliated party that violates a written agreement or any condition imposed in writing by the agency, breaches a fiduciary duty, or engages in *unsafe* or *unsound* practices. Because the term *unsafe* or *unsound* is not defined in any law or regulation, the potential liability of institution-affiliated parties is great.
- **5.233** The FDI Act also authorizes the agencies that regulate banks and savings institutions—on a showing of good cause—to remove, suspend, or bar an auditor from performing engagements required under the FDI Act.
- **5.234** Due to the passage of Credit Union Membership Access Act of 1998 in 1998, the NCUA adopted stiffer net worth requirements and prompt corrective action regulations. Practitioners should understand these regulations and their effect on the credit union.
- **5.235** The NCUA is required to publicly disclose formal and informal enforcement orders and any modifications to or terminations of such orders. Publication may be delayed for a reasonable time if disclosure would seriously threaten the safety or soundness of the credit union.
- **5.236** Currently, federal and most state credit union regulators use a letter of understanding and agreement or similar contractual arrangement to formalize the negotiated agreement between the regulatory agency or agencies (the regional director represents the NCUA) and the credit union's board of directors concerning problems, the actions to be taken, and the timetable for completing each action. In dealing with a state-chartered, non-National Credit Union Share Insurance Fund–insured credit union, the state regulator will usually involve the appropriate state or private insurer.

### **Planning**

**5.237** AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's

<sup>&</sup>lt;sup>47</sup> Title 12 U.S. Code Section 1818(i)(2)(D).

internal control. The auditor should obtain knowledge about regulatory matters and developments as part of the understanding of an institution's business. The auditor might also consider the results of regulatory examinations, as discussed previously.

### **Detection of Errors and Fraud**

**5.238** AU-C section 240 addresses the auditor's responsibilities relating to fraud in an audit of financial statements. Specifically, it expands on how AU-C sections 315 and 330 are to be applied regarding risks of material misstatement due to fraud. Noncompliance with laws and regulations (for example, noncompliance with regulatory capital requirements) is one indicator of higher risk that is especially relevant in the industry. Events of noncompliance are often described in

- regulatory reports and
- cease-and-desist orders or other regulatory actions, whether formal or informal.

**5.239** In accordance with paragraph .A10 of AU-C section 250, the auditor's responsibility regarding misstatements resulting from noncompliance with laws and regulations having a direct effect on the determination of material amounts and disclosures in the financial statements is the same as that for misstatements caused by fraud or error. For purposes of AU-C section 250, noncompliance is defined as acts of omission or commission by the entity either intentional or unintentional, which are contrary to the prevailing laws or regulations. Such acts include transactions entered into by, or in the name of, the entity or on its behalf by those charged with governance, management, or employees. Noncompliance does not include personal misconduct (unrelated to the business activities of the entity) by those charged with governance, management, or employees of the entity.

## **Evaluation of Contingent Liabilities and Related Disclosures**

**5.240** Management's financial statement assertions include those about the completeness, presentation, and disclosure of liabilities. Because some areas of regulation relate more to operations than to financial reporting or accounting, consideration of compliance in those areas would normally be limited to the evaluation of disclosures of any contingent liability based on alleged or actual violation of the law.

## **Going-Concern Considerations**

**5.241** Paragraphs 5.163–.179 address going-concern considerations. In addition to the matters discussed in those paragraphs, the auditor's consideration might include regulatory matters such as the following:

- Noncompliance with laws and regulations
- Supervisory actions or regulatory changes that place limitations or restrictions on operating activities
- Classification of the institution under prompt corrective action provisions of the FDI Act and the FCUA (see chapters 1 and 2 of this guide)

**5.242** For example, regulatory changes in 1992 placed new restrictions on the acceptance of brokered deposits by certain banks and savings institutions.

This change had two implications. First, it potentially limited sources of liquidity and created a compliance requirement. An auditor auditing the financial statements of an institution subject to those restrictions would have needed to evaluate whether the effect on the institution's liquidity, when considered with other factors, raised substantial doubt about the institution's ability to remain a going concern for a reasonable period of time. The auditor would also have needed to consider the financial statement effects of any known event of noncompliance with the requirement itself. Examples of other events or conditions that would warrant the auditor's consideration include

- the continued existence of conditions that brought about previous regulatory actions or restrictions;
- effects of scheduled increases in deposit insurance premiums;
- failure to meet minimum regulatory capital requirements;
- limitations on the availability of borrowings through the Federal Reserve System discount window; and
- exposure to the institution posed by transactions with correspondent banks and related limitations on interbank liabilities.

## Regulatory Reporting Matters—Interpretation and Reporting Related to GAAP

**5.243** General purpose financial statements are prepared in accordance with GAAP. Every national bank and savings and loan association, state member bank and state chartered savings and loan association, and insured state nonmember bank is required to file FFIEC Call Reports. Every federally insured credit union is required to file the NCUA 5300 Call Report. Call Reports (for example, FFIEC and NCUA) present an institution's financial condition and results of operations on a consolidated basis in accordance with GAAP. These reports are used by regulators as a basis for supervisory action, a source of statistical information, and other such purposes. In 1997, the banking regulators adopted instructions for these reports that follow GAAP.

**5.244** FDI Act Section 37(a)(2) requires that reports and other regulatory filings for banks and savings institutions follow accounting principles that are uniform and consistent with GAAP. Regulatory reporting topics noted herein are consistent with acceptable practices under GAAP. The Call Report instructions explain certain specific reporting guidance in greater detail. Information may often be found in the appropriate entries in the "Glossary" section of the Call Report or, in more detail, in the GAAP standards. Financial institutions are encouraged to discuss specific events and transactions not covered by GAAP or the guidance in the regulatory report instructions with their primary supervisory agency for more technical detail on the application of the GAAP accounting standards.

**5.245** Appendix B, "Regulatory Reporting Matters—Interpretation and Reporting Related to U.S. GAAP," of this guide serves as an aid in specific selected areas and is not intended to be a comprehensive discussion of the principles of bank accounting or reporting.

**5.246** For financial institutions, the allowance for loan and lease losses (ALLL) is an area that requires judgment and is a focus of auditors and examiners. At the same time, the *Interagency Policy Statement on the Allowance for Loan and Lease Losses*, dated December 13, 2006, emphasizes that the ALLL should be consistent with GAAP. This policy statement reminds institutions

that the ALLL generally should not be based solely on a "standard percentage" of loans. To that end, the policy statement no longer references standardized loss estimates for classified loans. Banks should review the entry *allowance for loan and lease losses* in the "Glossary" section of the FFIEC's *Instructions for Preparation of Consolidated Reports of Condition and Income*, and the interagency policy statement on the ALLL.

5.247 Bank examiners will review the reasonableness of the range and management's best estimate within the range. The agencies find that an ALLL established in accordance with the December 13, 2006, Interagency Policy Statement on the Allowance for Loan and Lease Losses and the Interagency Policy Statement on Allowance for Loan and Lease Losses Methodologies and Documentation for Banks and Savings Institutions, issued July 2001 (2001 Policy Statement) as applicable, falls within the range of acceptable estimates determined in accordance with GAAP. The guidance in the 2001 Policy Statement was substantially adopted by the NCUA through its Interpretive Ruling and Policy Statement 02-3, Allowance for Loan and Lease Losses Methodologies and Documentation for Federally-Insured Credit Unions, in May 2002.

## **Auditor and Examiner Relationship**

- **5.248** Banking regulators conduct periodic on-site examinations to address broader regulatory and supervisory issues. There are some objectives shared by examiners and auditors, and coordination in consultation with the institution may be beneficial.
- **5.249** The primary objective of communicating with examiners is to ensure that auditors consider competent audit evidence produced by examiners before expressing an opinion on audited financial statements. In areas such as the adequacy of credit loss allowances and violations of laws or regulations, for example, information known to or judgments made by examiners generally should be made known to management and the auditor before financial statements are issued or an audit opinion is rendered. Such communication will minimize the possibility that a regulatory agency will subsequently require restatement—based on the examiner's additional knowledge or different judgment—of Call Reports and affect the general purpose financial statements, on which the auditor has already expressed an opinion, dated during or subsequent to the period in which a regulatory examination was being conducted.
- **5.250** FDI Act Section 36(h) requires that each bank and savings institution provide its auditor with copies of the institution's most recent Call Report and examination report (see 12 CFR 363). According to regulations, the institution must also provide the auditor with any of the following documents related to the period covered by the engagement:
  - a. Any memorandum of understanding or other written agreement between the institution and any federal or state banking agency
  - b. The report of any action initiated or taken by any federal or state banking agency, including any assessment of civil money penalties
- **5.251** The auditor might consider reviewing communications from examiners and, when appropriate, make inquiries of examiners. Specifically, the auditor could
  - a. request that management provide access to all reports of examination and related correspondence;

- b. review the reports of examination and related correspondence between examiners and the institution during the period under audit and through the date of the auditor's opinion;
- c. with prior approval of the institution, communicate with the examiners if their examination is still in process, the institution's appeal of an examination finding is outstanding, or their examination report is still pending; and
- d. with prior approval of the institution, consider attending, as an observer, the exit conference between the examiner and the institution's board of directors, its executive officers, or both.
- **5.252** The auditor's attendance at other meetings between examiners and representatives of the institution is based on prior approval by the regulatory agency.
- **5.253** Auditors may request a meeting with the appropriate regulatory representatives to inquire about supervisory matters relevant to the client institution. The management of the institution would generally be present at such a meeting, and matters discussed would generally be limited to findings already presented to management. Federal regulatory policy also permits meetings between examiners and auditors in the absence of the institution's management.<sup>48</sup>
- **5.254** Management refusal to furnish access to reports or correspondence, or to permit the auditor to communicate with the examiner, would ordinarily be a limitation on the scope of a financial statement audit sufficient to preclude an opinion. Refusal by an examiner to communicate with the auditor may create the same scope limitation, depending on the auditor's assessment of the circumstances. AU-C section 705 addresses how the form and content of the auditor's report is affected when the auditor expresses a modified opinion in the auditor's report. (For a detailed discussion on reports issued under the guidance of AU-C section 705, along with AU-C sections 700 and 706, *Emphasis-of-Matter Paragraphs and Other-Matter Paragraphs in the Independent Auditor's Report* [AICPA, *Professional Standards*], and related PCAOB requirements when performing integrated audits see chapter 23 of this guide.)
- **5.255** Examiners might request permission to attend the meeting between the auditor and representatives of the institution (for example, the audit committee of the board of directors) to review the auditor's report on the institution's financial statements. If such a request is made and management concurs, the auditor should be responsive to the request.
- **5.256** Examiners and others may, from time to time, request auditors of financial statements of banks and savings institutions to provide access to working papers and audit documentation. The FFIEC's *Interagency Policy Statement on External Auditing Programs for Banks and Savings Associations* states that the independent public auditor or other auditor of an institution should agree in the engagement letter to grant examiners access to all the auditor's working papers and other material pertaining to the institution prepared in the course of performing the completed external auditing program.

©2016, AICPA AAG-DEP 5.256

<sup>&</sup>lt;sup>48</sup> Related instructions to examiners were published in a July 23, 1992, *Interagency Policy Statement on Coordination and Communication Between External Auditors and Examiners*. On January 27, 1997, the division of Supervision of the FDIC issued a supervisory memo to encourage each region to improve communications, coordination, and working relationships with independent public accountants of FDIC-supervised institutions.

The FDIC issued guidance concerning the review of external auditor's working papers (Regional Director Memorandum No. 2000-019, *Reviews of External Auditors' Workpapers*, dated March 21, 2000.) Auditors who have been requested to provide such access should consider Interpretation No. 1, "Providing Access to or Copies of Audit Documentation to a Regulator," of AU-C section 230 (AICPA, *Professional Standards*, AU-C sec. 9230 par. .01–.15). The interpretation states when a regulator requests access to audit documentation pursuant to law, regulation, or audit contract, the auditor may take the following steps:

- Consider advising the client that the regulator has requested access to (and possibly copies of) the audit documentation and that the auditor intends to comply with such request.
- Make appropriate arrangements with the regulator for the review.
- Maintain control over the audit documentation.
- Consider submitting to the regulator a letter clarifying that an audit performed in accordance with GAAS is not intended to, and does not, satisfy a regulator's oversight responsibilities. An example of such a letter is illustrated in paragraph .06 of Interpretation No. 1 of AU-C section 230.

In addition, the interpretation addresses situations in which an auditor has been requested by a regulator to provide access to the audit documentation before the audit has been completed and the report released. Also, the interpretation notes that if a regulator engages an independent party, such as another independent public auditor, to perform the audit documentation review on behalf of the regulatory agency, there are some precautions auditors might consider observing.

**5.257** Information in examination reports, inspection reports, supervisory discussions—including summaries or quotations—is considered confidential. Such information may not be disclosed to any party without the written permission of the appropriate agency, and unauthorized disclosure of such information could subject the auditor to civil and criminal enforcement actions.

## Exhibit 5-1 Fraud Risk Factors

Two types of fraud are relevant to the auditor's consideration, namely, fraudulent financial reporting and the misappropriation of assets. For each of these types of fraud, the risk factors are further classified based on the three conditions generally present when material misstatements due to fraud occur, which are incentives/pressures, opportunities, and attitudes/rationalizations. Although the risk factors cover a broad range of situations, they are only examples and, accordingly, the auditor may identify additional or different risk factors. Also, the order of the examples of risk factors provided is not intended to reflect their relative importance or frequency of occurrence.

Although fraud is a broad legal concept, for the purposes of GAAS, paragraph .03 of AU-C section 240, Considerations of Fraud in a Financial Statement Audit (AICPA, Professional Standards), states that the auditor is primarily concerned with fraud that causes a material misstatement in the financial statements. Some of the following factors and conditions are present in entities in which specific circumstances do not present a risk of material misstatement. Also, specific controls may exist that mitigate the risk of material misstatement

due to fraud, even though risk factors or conditions are present. When identifying risk factors and other conditions, the auditors might assess whether those risk factors and conditions, individually and in combination, present a risk of material misstatement of the financial statements.

### Fraudulent Financial Reporting

The following are examples of risk factors that might result in misstatements arising from fraudulent financial reporting.

### **Incentives/Pressures**

- Financial stability or profitability is threatened by economic, industry, or entity operating conditions, such as (or as indicated by) the following:
  - a. High degree of competition or market saturation, accompanied by declining margins shown by the following:
    - An increase of competitor investment products that are close alternatives for the institution's deposit products (for example, mutual funds, insurance annuities, and mortgage loans), placing pressure on the institution's deposit rates
    - ii. Competitor product pricing that results in loss of customers or market share for such products as loan, deposit, trust, asset management, and brokerage offerings
  - b. High vulnerability to rapid changes, such as changes in technology, product obsolescence, or interest rates, exemplified by the following:
    - A failure or inability to keep pace with or to afford rapid changes in technology, if the financial stability or profitability of the particular institution is placed at risk due to that failure or inability
    - ii. Significant unexpected volatility (for example, in interest rates, foreign exchange rates, and commodity prices) in financial markets where the institution has a significant capital market presence and is exposed to loss of revenue or has not appropriately hedged its risk to price changes that effect proprietary positions
    - iii. Flattening yield curves or extremely high or low market interest rate environments
  - c. Significant declines in customer demand and increasing business failures in either the industry or overall economy, such as the following:
    - Deteriorating economic conditions (for example, declining corporate earnings, adverse exchange movements, and real estate prices) within industries or geographic regions in which the institution has significant credit concentrations
    - For credit unions, losing a very substantial portion of the membership base, which places considerable pressure on management insofar as

financial projections are often based on gaining new members and offering commercial loans

- d. Rapid growth or unusual profitability, especially compared to that of other peer financial institutions; for example, unusually large growth in the loan portfolio without a commensurate increase in the size of the allowance for loan and lease losses (ALLL)
- e. New and existing accounting, statutory, or regulatory requirements, such as the following:
  - i. Substantially weak CAMELS (capital adequacy, asset quality, management, earnings, liquidity, and sensitivity to market risk) or, for bankholding companies, BOPEC (bank's CAMELS rating, operation of significant nonbanking subsidiaries, parent's strength and operations, earnings of the banking organization, and capital of the banking organization) ratings.
  - ii. Regulatory capital requirements
- f. Decline in asset quality due to the following:
  - Borrowers affected by recessionary declines and layoffs
  - ii. Issuers affected by recessionary declines and industry factors
- 2. Excessive pressure exists for management or operating personnel to meet financial targets established by those charged with governance, including incentive goals:
  - a. Unrealistically aggressive loan goals and lucrative incentive programs for loan originations, shown by the following, for example:
    - i. Relaxation of credit standards
    - Excessive extension of credit standards with approved deviation from policy
    - iii. Excessive concentration of lending (particularly new lending)
    - iv. Excessive lending in new products
    - Excessive pricing concessions not linked to enhanced collateral positions or other business rational (for example, sales of other products or services)
    - vi. Excessive refinancing at lower rates that may delay the recognition of problem loans
  - b. Perceived or real adverse effects of reporting poor financial results on significant pending transactions, such as business combinations (For example, the acquisition of another institution has been announced in the press with the terms dependent on the future financial results of the acquiring institution.)

- c. Willingness by management to respond to these pressures by pursuing business opportunities for which the institution does not possess the needed expertise
- d. Excessive reliance on wholesale funding (brokered deposits)
- e. Speculative use of derivatives
- f. Failure to establish economic hedges against key risks (for example, interest rate) through effective asset liability committee processes
- g. Changes in a bank's loan loss accounting methodology that are not accompanied by observed changes in credit administration practices or credit conditions
- h. Frequent or unusual exceptions to credit policy
- i. Threat of a downgrade in the institution's overall regulatory rating (for example, CAMEL, MACRO [rating stands for management, asset quality, capital adequacy, risk management and operating results], or BOPEC) that could preclude expansion or growth plans
- j. Threat of failing to meet minimum capital adequacy requirements that could cause adverse regulatory actions
- 3. Management's or those charged with governance's personal net worth is threatened by the entity's financial performance arising from the following:
  - a. Heavy concentrations of their personal net worth in the entity
  - b. Bank is privately owned by one person or family whose net worth or income (from dividends) is dependent on the bank

### **Opportunities**

- 1. The nature of the industry or the entity's operations provides opportunities to engage in fraudulent financial reporting that can arise from the following:
  - a. Significant related party transactions not in the ordinary course of business or with related entities not audited or audited by another firm, such as the following:
    - Loans and other transactions with directors, officers, significant shareholders, affiliates, and other related parties, particularly those involving favorable terms
    - ii. Variable interest entities (VIEs)
    - iii. Certain types of lending practices such as, subprime and predatory lending by banks in an effort to obtain better yields
    - iv. Transfers of impaired assets
  - b. Assets, liabilities, revenues, or expenses based on significant estimates that involve subjective judgments or uncertainties that are difficult to corroborate (Significant estimates generally include the allowance for loan losses, and the valuation of servicing rights, residual interests,

- and deferred tax assets, fair value determinations, and the recognition of other impairment losses; for example, goodwill and investments)
- c. Significant, unusual, or highly complex transactions, especially those close to year end that pose difficult "substance over form" questions, such as the following:
  - i. Consolidation questions with VIEs
  - ii. Material amounts of complex financial instruments and derivatives held by the institution that are difficult to value, or the institution's use of complex collateral disposition schemes
- d. Frequent or unusual adjustments to the ALLL
- e. Loan sales that result in retained beneficial interests (Valuation of retained beneficial interests is based on estimates and assumptions and are susceptible to manipulation if not properly controlled.)
- f. Complex transactions that result in income or gains, such as sale and leasebacks, with arbitrarily short leaseback terms
- g. Deferred tax assets, arising from net operating loss carryforwards, without valuation allowances
- h. Deferral of loan origination costs that exceed the appropriate costs that may be deferred under FASB ASC 310-20
- Internal control components are deficient as a result of the following:
  - a. Inadequate monitoring of controls, including automated controls and controls over financial reporting, such as lack of oversight of critical processes in the following areas:
    - i. Cash and correspondent banks—Reconciliation and review
    - ii. Intercompany or interbranch cash or suspense accounts and "internal" demand deposit accounts (DDAs)—Monitoring of activity and resolution of aged items
    - iii. Lending—Lack of credit committee and lack of stringent underwriting procedures
    - iv. Treasury—Securities/derivatives valuation (selection of models, methodologies, and assumptions)
    - v. Regulatory compliance—Lack of knowledge of pertinent regulation
    - vi. Deposits—Lack of monitoring unusual and significant activity
  - b. Ineffective internal audit function
  - c. Lack of board-approved credit (underwriting and administration) or investment policies
  - d. Vacant staff positions remain unfilled for extended periods, thereby preventing the proper segregation of duties

- e. Lack of an appropriate system of authorization and approval of transactions in areas such as lending and investment, in which the policies and procedures for the authorization of transactions are not established at the appropriate level
- f. Lack of independent processes for the establishment and review of allowance for loan losses
- g. Lack of independent processes for the evaluation of other than temporary impairments
- h. Inadequate controls over transaction recording, including the setup of loans on systems
- Lack of controls over the perfection of interests in lending collateral
- j. Inadequate methods of identifying and communicating exceptions and variances from planned performance
- k. Inadequate accounting reconciliation policies and practices, including appropriate supervisory review, the monitoring of stale items and out of balance conditions, and the timeliness of write-offs
- Failure to establish adequate segregation of duties between approval transactions and the disbursement of funds
- m. Lack of control over the regulatory reporting process, in which key decision makers also have control over the process
- Lack of adequate reporting to the board of directors and executive management regarding credit, interest-rate, liquidity, and market risks
- o. Change from an internal audit function that has been outsourced to the external auditor or other provider to a new in-house internal audit department or another outsourcing provider

#### **Attitudes and Rationalizations**

- Known history of violations of securities laws or other laws and regulations, or claims against the entity, its senior management, or those charged with governance alleging fraud or violations of laws and regulations, such as the following:
  - a. The existence of a regulatory cease and desist order, memorandum of understanding, or other regulatory agreements (whether formal or informal) that concern management competence or internal control
  - b. Repeated criticisms or apparent violations cited in regulatory examination reports that management has ignored
- 2. Nonfinancial management's excessive participation in or preoccupation with the selection of accounting principles or the determination of significant estimates, such as the following:
  - a. Consideration of "business issues" (for example, share-holder expectations) in determining significant estimates

- b. Adjustments to the allowance for loan losses by senior management or the board for which there is no written documentation
- c. An unusual propensity to enter into complex asset disposition agreements
- 3. The disregard of control-related recommendations from internal or external auditors
- 4. A high level of customer complaints (especially when management does not fix the cause of them promptly)
- Indications that internal audit is not adequately staffed or trained, and does not have appropriate specialized skills given the environment
- 6. Indications that internal audit is not independent (authority and reporting relationships) and does not have adequate access to the audit committee (or equivalent)
- 7. Inappropriate scope of internal audit's activities (for example, the balance between financial and operational audits, coverage, and rotation of decentralized operations)
- 8. Limited authority of internal audit to examine all aspects of the client's operations or failure to exercise its authority
- 9. Failure by internal audit to adequately plan, perform risk assessments, or document the work performed or conclusions reached
- 10. Failure of internal audit to adhere to professional standards
- 11. Operating responsibilities assigned to internal audit
- 12. Inability to prepare accurate and timely financial reports, including interim reports
- 13. Failure of planning and reporting systems (such as business planning; budgeting, forecasting, and profit planning; and responsibility accounting) to adequately set forth management's plans and the results of actual performance
- 14. A low level of user satisfaction with information systems processing, including reliability and timeliness of reports
- 15. Understaffed accounting or information technology department, inexperienced or ineffective accounting or information technology personnel, or high turnover
- 16. Lack of timely and appropriate documentation for transactions
- 17. Dividend requirements by management or ownership frequently at or near the maximum allowable by law (In closely held companies, executive management/ownership combines high dividends with frequently substantial increases in cash salary or bonus compensation. The bank has been cited for dividend violations by regulatory authorities.)

### Misappropriation of Assets

Risk factors that relate to misstatements arising from the misappropriation of assets are also classified according to the three conditions generally present when fraud exists, namely, incentives/pressures, opportunity, and attitudes/rationalizations. Some of the risk factors related to misstatements arising from fraudulent financial reporting also may be present when misstatements arising from misappropriation of assets occur. For example, ineffective

monitoring of management and other deficiencies in internal control that are not effective may be present when misstatements due to either fraudulent financial reporting or the misappropriation of assets exist. The following sections show examples of risk factors related to misstatements arising from misappropriation of assets.

#### Incentives and Pressures

- 1. Adverse relationships between the institution and employees with access to cash or other assets susceptible to theft may motivate those employees to misappropriate those assets. For example, the following may create adverse relationships:
  - a. It is likely that the institution will be merged into or acquired by another institution and there is uncertainty regarding the employees' future employment opportunities.
  - b. The institution has recently completed a merger or acquisition, employees are working long hours on integration projects, and morale is low.
  - c. The institution is under regulatory scrutiny, and there is uncertainty surrounding the future of the institution.
- Members of executive management evidence personal financial distress through indications such as frequent informal "loans" or "salary advances" to key executive officers or their family members.

### **Opportunities**

- 1. Certain characteristics or circumstances may increase the susceptibility of assets to misappropriation. For example, opportunities to misappropriate assets increase when the following exist:
  - Large amounts of cash on hand and wire transfer capabilities
  - Easily convertible assets, such as bearer bonds or diamonds, that may be in safekeeping
  - c. Inadequate or ineffective physical security controls, for example, overliquid assets or information systems
  - d. Access to customer accounts
- 2. Inadequate internal control over assets may increase the susceptibility of misappropriation of those assets. For example, misappropriation of assets may occur because the following exist:
  - a. Inadequate management oversight of employees responsible for assets, such as the following:
    - Vacant branch manager positions or managers are away on leave without replacements for an inordinate amount of time, causing a considerable lack of management oversight.
    - ii. The independent risk management function does not have the appropriate level of sophistication or the capability to effectively monitor and measure the risks, such as capital markets trading activities.
    - Lack of adherence or enforcement of vacation policy.

- b. Inadequate job applicant screening and monitoring of employees, such as the following:
  - Federal Bureau of Investigation background checks, credit reports, and bonding eligibility screening are not incorporated into the hiring process for employees with access to significant assets susceptible to misappropriation.
  - A monitoring process does not identify employees who have access to assets susceptible to misappropriation and who are known to have financial difficulties.
- c. Inadequate segregation of duties or independent checks, such as the following:
  - Lack of independent monitoring of activity in internal DDAs and correspondent bank accounts
  - ii. No independent monitoring and resolution of customer exceptions/inquiries related to electronic funds transfer (EFT) transactions, loan disbursements/payments, customer deposit accounts, securities and derivatives transactions, and trust/fiduciary accounts
  - iii. Lack of key periodic independent reconciliations (in addition to reconciliations of subledgers to the general ledger) for wire transfer, treasury, trust, suspense accounts, automated teller machines, and cash
  - iv. Lack of segregation of duties in the following areas:
    - (1) EFT—Origination, processing, confirmation, and recordkeeping
    - (2) Lending—Relationship management, underwriting (including approval), processing, cash collection/disbursement, and recordkeeping; no periodic confirmation of customer loan information or indebtedness by personnel independent of the relationship officer.
    - (3) Treasury—Trading, processing, settlement, and recordkeeping. (The derivatives positions on the Treasury system are not priced by an independent operations area. The capital markets risk management process is not independent from the trading function. There is no independent confirmation of individual trades.)
    - (4) Trust—Relationship management, transaction authorization, transaction execution, settlement, custody, and account recordkeeping. (There is no

- annual review of the activity in trust accounts by an investment committee to ensure compliance with the terms of the trust agreement and bank investment guidelines.)
- (5) Fiduciary—Issuance, registration, transfer, cancellation, and recordkeeping
- (6) Charged-off loan accounts and recoveries
- (7) Dormant and inactive DDAs and the escheatment process.
- d. No independent mailing of customer statements or monitoring of "Do not Mail/Hold" statements
- e. Lack of control over new accounts
- f. Failure to reconcile "due from" bank accounts on a regular basis, and review open items
- g. Loans are purchased from loan brokers, but the loans are not reunderwritten before purchase
- h. Inadequate segregation of duties because the institution is small and has limited staff
- *i.* Lack of appropriate system of authorization and approval of transactions, such as the following:
  - No verification of EFT initiation and authorization, including those instances in which bank employees initiate a transaction on a customer's behalf
  - ii. Frequent underwriting exceptions to boardestablished credit authorization limits
  - Frequent instances of cash disbursements on loans that have not yet received all approvals or met all preconditions for funding
  - iv. Lack of board approval for significant loans or unusually high loan-officer approval limits (Be alert to the existence of multiple loans being funded just below a loan officer's limit.)
- j. Poor physical safeguards over cash, investments, customer information, or fixed assets, such as the following:
  - Lack of adequate physical security over the EFT operations area and customer records
  - Failure to appropriately limit access to the vault to authorized employees acting within the scope of their job
  - Lack of dual control over the vault, negotiable instruments (including travelers' checks and money orders), and blank-check stock
  - iv. Lack of accountability over negotiable instruments

### **Depository and Lending Institutions**

- k. Inadequate training of tellers and operations personnel regarding the following:
  - i. "Knowing your customer"
  - ii. Recognizing check fraud and kiting activities
  - iii. Controls over cash, negotiable instruments, and EFT

#### Attitudes and Rationalization

- 1. Disregard for the need for monitoring or reducing risks related to misappropriations of assets
- 2. Disregard for internal control over misappropriation of assets by overriding existing controls or by failing to take appropriate remedial action on known deficiencies in internal control
- 3. Behavior indicating displeasure or dissatisfaction with the entity or its treatment of the employee
- 4. Changes in behavior or lifestyle that may indicate assets have been misappropriated
- 5. The belief by some executives that their level of authority justifies a certain level of compensation and personal privileges
- 6. Tolerance of petty theft

## **Chapter 6**

## Cash and Cash Equivalents

### Introduction

**6.01** Cash and cash equivalents include cash items in the process of collection (CIPC), deposits with other financial institutions, including corporate credit unions, balances with the Federal Reserve Banks and the Federal Home Loan Banks (FHLBs), federal funds sold, and cash and cash equivalents on hand. Instruments that meet the definition of cash and cash equivalents, as defined in the FASB Accounting Standards Codification (ASC) glossary, are discussed in this chapter. Investments in debt and equity securities that are accounted for under FASB ASC 320, Investments—Debt and Equity Securities, are discussed in chapter 7, "Investments in Debt and Equity Securities," of this guide. Other investments are discussed in chapter 12, "Other Assets, Other Liabilities, and Other Investments," of this guide. The fair value option for financial assets and liabilities under FASB ASC 825, Financial Instruments, is addressed in chapter 20, "Fair Value," of this guide.

## **CIPC and Cash Equivalents**

**6.02** CIPC includes customer deposits drawn on other depository institutions that have not yet cleared, matured instruments (such as coupons and bonds), and other matured items temporarily held pending their liquidation. Such assets are received with deposits and other customer transactions. CIPC are eventually cleared through local clearinghouses, correspondent institutions (correspondents), or a Federal Reserve Bank. Collection of these items generally takes between one to five business days. A discussion of cash equivalents can be found in paragraph 6.09.

## **Deposits With Other Financial Institutions**

- **6.03** Correspondents are depository institutions that hold the account balances of other financial institutions and provide services to those institutions, such as check collection and item processing. Such accounts with balances due from other institutions are generally called "due from banks" and are maintained by depository institutions as a means of more efficient check clearing or to compensate the correspondent for other services provided to the depositor. Institutions that engage in international banking may maintain deposits with foreign depository institutions for the same reasons.
- **6.04** Many institutions also invest in nonnegotiable or negotiable certificates of deposit (CDs) of other depository institutions. These balances are generally interest bearing and insured up to \$250,000¹ and have a range of maturity options.
- **6.05** Many credit unions hold funds in corporate credit unions. Corporate credit unions, which are regulated by the National Credit Union Administration, provide investment, liquidity, and payment services to natural-person

©2016, AICPA AAG-DEP 6.05

<sup>&</sup>lt;sup>1</sup> On July 21, 2010, the Dodd-Frank Wall Street Reform and Consumer Protection Act made permanent the current maximum deposit insurance amount of \$250,000, which was originally scheduled to return to \$100,000 on January 1, 2014. The FDIC coverage limit applies per depositor, per insured depository institution, for each account ownership category.

credit unions. These corporate credit unions aggregate funds acquired from natural-person credit unions in order to facilitate the purchase of overnight and term investments. Overnight investments include cash management accounts and overnight certificates. Term investments, with maturities from two days to five years or longer, include fixed-rate and variable-rate shares and certificates.

### Balances With Federal Reserve Banks and FHLBs

**6.06** Federal regulations require depository institutions to set aside specified amounts of cash as reserves against transaction accounts and time deposits. These reserves may be held as vault cash, in a noninterest-bearing account with a district Federal Reserve Bank or FHLB, or as deposits with correspondents. Though one objective of reserve requirements is to safeguard liquidity in the banking system, institutions do not look to their reserves as a primary source of liquidity because regulations permit their depletion for only short periods and in limited circumstances. Rather, reserves are a primary tool of the Board of Governors of the Federal Reserve System (Federal Reserve) to effect monetary policy; by increasing or decreasing reserve requirements, the Federal Reserve can expand or contract the money supply. Depository institutions also may lend excess balances overnight and for short periods in the federal funds market.

### Federal Funds Sold

**6.07** Chapter 14, "Federal Funds and Repurchase Agreements," of this guide discusses federal funds and repurchase agreements, which can be either assets or liabilities, depending on which side of the transaction the institution participates.

### Cash on Hand

**6.08** Cash on hand consists primarily of coin and currency in vaults, in the institution's automated teller machines (ATMs), and maintained by tellers to meet customers' requests. Cash on hand generally represents a small percentage of a depository institution's total of cash and cash equivalent items.

## **Accounting and Financial Reporting**

## Definition of Cash and Cash Equivalents

**6.09** The FASB ASC glossary defines *cash* as currency on hand, demand deposits with banks or other financial institutions, and other kinds of accounts that have the general characteristics of demand deposits in that the customer may deposit additional funds at any time and also effectively may withdraw funds at any time without prior notice or penalty. *Cash equivalents* are defined in the FASB ASC glossary as short term, highly liquid investments that have both of the following characteristics:

- a. Readily convertible to known amounts of cash
- b. So near their maturity that they present insignificant risk of changes in value because of changes in interest rates

Generally, only investments with original maturities of three months or fewer qualify under that definition. Original maturity means original maturity to

the entity holding the investment. For example, both a three-month U.S. Treasury bill (T-bill) and a three-year U.S. Treasury note (T-note) purchased three months from maturity qualify as cash equivalents. However, a T-note purchased three years ago does not become a cash equivalent when its remaining maturity is three months. Examples of items commonly considered to be cash equivalents are T-bills, commercial paper, money market funds, and federal funds sold (for an entity with banking operations). Other examples may include CDs.

- **6.10** Only instruments that meet both of the previously mentioned criteria and are used as part of an institution's cash-management activities ordinarily would be included in cash equivalents. For example, T-bills purchased for an investment account would be part of the institution's investing activities (not cash-management activities) and would therefore be excluded from cash equivalents. The carrying amount of items classified as cash and cash equivalents generally approximates fair value because of the relatively short period of time between the origination of the instruments and their expected realization.
- **6.11** Investments, such as negotiable CDs and mutual funds that meet the definition of a *security* in the FASB ASC glossary are subject to the reporting, classification, and other provisions of FASB ASC 320. Investments subject to FASB ASC 320 are discussed in chapter 7 of this guide.

### Classification of Cash Flows

- **6.12** FASB ASC 230, *Statement of Cash Flows*, presents standards for reporting cash flows in general-purpose financial statements. FASB ASC 230-10-45-4 states the total amounts of cash and cash equivalents at the beginning and end of the period shown in the statement of cash flows should be the same amounts as similarly titled line items or subtotals shown in the statements of financial position as of those dates.<sup>2</sup>
- **6.13** FASB ASC 230-10-50-1 requires an entity to disclose its policy for determining which items are treated as cash equivalents. Any change to that policy is a change in accounting principle that should be effected by restating financial statements for earlier years presented for comparative purposes.
- ${f 6.14}$  The cash equivalents policy is generally disclosed in the accounting policy footnote.
- **6.15** Specific guidance for applying the direct method and the indirect method of reporting cash flows is provided in FASB ASC 230-10. In reporting cash flows from operating activities, FASB ASC 230-10-45-25 states that entities are encouraged to report major classes of gross cash receipts and gross cash payments and their arithmetic sum—the net cash flow from operating activities (the direct method). If the direct method of reporting net cash flow from operating activities is used, the reconciliation of net income of a business entity to net cash flow from operating activities should be provided in a separate schedule, as stated in FASB ASC 230-10-45-30. Paragraphs 1–5 of FASB

<sup>&</sup>lt;sup>2</sup> In April 2016, FASB issued proposed Accounting Standards Update (ASU) Statement of Cash Flows (Topic 230): Restricted Cash (a consensus of the FASB Emerging Issues Task Force) to provide guidance on the presentation of restricted cash or restricted cash equivalents in the statement of cash flows. Specifically, the amendments in this proposed ASU would require that a statement of cash flows explain the change during the period in the total of cash, cash equivalents, and amounts generally described as restricted cash or restricted cash equivalents. Readers are encouraged to visit the "Technical Agenda" page at www.fasb.org for the latest developments regarding FASB's project titled Restricted Cash and how it may impact the guidance in this chapter.

ASC 942-230-55 provide implementation guidance and illustrations regarding the statement of cash flows under the direct method for financial institutions.

### © Update 6-1 Accounting and Reporting: Credit Losses

FASB Accounting Standards Update (ASU) No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, issued in June 2016, is effective for fiscal years of public business entities that are SEC filers beginning after December 15, 2019, including interim periods within those fiscal years.

For all other public business entities, the amendments in FASB ASU No. 2016-13 are effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years.

For all other entities, including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960, *Plan Accounting—Defined Benefit Pension Plans*, through FASB ASC 965, *Plan Accounting—Health and Welfare Benefit Plans*, on plan accounting, FASB ASU No. 2016-13 is effective for fiscal years beginning after December 15, 2020, and interim periods within fiscal years beginning after December 15, 2021.

Early application is permitted for all entities as of the fiscal years beginning after December 15, 2018, including interim periods within those fiscal years.

FASB ASU No. 2016-13 creates FASB ASC 326, *Financial Instruments—Credit Losses*, to amend guidance on reporting credit losses for assets held at amortized cost basis and available-for-sale debt securities.

For assets held at amortized cost basis, FASB ASC 326 eliminates the probable initial recognition threshold in current U.S. generally accepted accounting principles (GAAP) and, instead, requires an entity to reflect its current estimate of all expected credit losses. The allowance for credit losses is a valuation account that is deducted from the amortized cost basis of the financial assets to present the net amount expected to be collected.

For available-for-sale debt securities, credit losses should be measured in a manner similar to current GAAP. However, FASB ASC 326 will require that credit losses be presented as an allowance rather than as a write-down.

FASB ASU No. 2016-13 affects entities holding financial assets and net investment in leases that are not accounted for at fair value through net income. The amendments affect loans, debt securities, trade receivables, net investments in leases, off-balance-sheet credit exposures, reinsurance receivables, and any other financial assets not excluded from the scope that have the contractual right to receive cash.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this paragraph will be updated, as needed, in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-13, see appendix G, "Accounting for Financial Instruments," of this guide.

**6.16** Examples of major classes of gross cash receipts reported in operating activities may include interest received and service charges collected. Examples

of gross cash disbursements may include interest paid and operating expenses paid.

- **6.17** Entities that choose not to provide information about major classes of operating cash receipts and payments by the direct method should determine and report the same amount for net cash flow from operating activities indirectly by adjusting net income of a business entity to reconcile it to net cash flow from operating activities (the indirect or reconciliation method), as stated in FASB ASC 230-10-45-28. If the indirect method is used, amounts of interest paid (net of amounts capitalized) and income taxes paid during the period should be disclosed, according to FASB ASC 230-10-50-2.
- **6.18** According to FASB ASC 230-10-45-10, a statement of cash flows should classify cash receipts and cash payments as resulting from investing, financing, or operating activities. The FASB ASC glossary defines *operating activities* as all transactions and other events that are not defined as investing or financing activities (see paragraphs 12–15 of FASB ASC 230-10-45). Operating activities generally involve producing and delivering goods and providing services. Cash flows from operating activities are generally the cash effects of transactions and other events that enter into the determination of net income.
- **6.19** Paragraphs 11–12 of FASB ASC 230-10-45 state that cash flows from the purchases, sales, and maturities of available-for-sale securities should be classified as cash flows from investing activities and reported gross in the statement of cash flows. Receipts from sales of loans that were not specifically acquired for resale are cash inflows from investing activities. That is, if loans were acquired as investments, cash receipts from sales of those loans should be classified as investing cash inflows regardless of a change in the purpose for holding those loans.

# © Update 6-2 Accounting and Reporting: Recognition and Measurement of Financial Assets and Financial Liabilities

FASB ASU No. 2016-01, Financial Instruments—Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities, issued in January 2016, is effective for fiscal years, and interim periods within those fiscal years, of a public business entity beginning after December 15, 2017.

For all other entities (including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting), FASB ASU No. 2016-01 is effective for fiscal years beginning after December 15, 2018, and interim periods within fiscal years beginning after December 15, 2019. Early application is permitted for all other entities as of the fiscal years beginning after December 15, 2017, including interim periods within those fiscal years.

©2016. AICPA AAG-DEP 6.19

<sup>&</sup>lt;sup>3</sup> FASB has an open project titled Statement of Cash Flows: Classification of Certain Cash Receipts and Cash Payments. The proposal addresses eight specific cash flow issues with the goal of reducing the existing diversity in practice. Those cash flow issues include debt prepayment or debt extinguishment costs, settlement of zero-coupon bonds, contingent consideration payments made after a business combination, proceeds from the settlement of insurance claims, proceeds from the settlement of corporate-owned life insurance policies, including bank-owned life insurance policies, distributions received from equity method investees, beneficial interests in securitization transactions, and predominant cash receipts and cash payments. Readers are encouraged to visit the "Technical Agenda" page at www.fasb.org for the latest developments regarding the project and how it may impact the guidance in this chapter.

All entities may adopt the presentation guidance in paragraphs 5–7 of FASB ASC 825-10-45 for financial statements of fiscal years or interim periods that have not yet been issued or that have not yet been made available for issuance. Furthermore, entities that are not public business entities may elect not to disclose the information about fair value of financial instruments required by the "General" subsection of FASB ASC 825-10-50 in financial statements of fiscal years or interim periods that have not yet been made available for issuance. Except as indicated previously in this paragraph, early application is not permitted.

FASB ASU No. 2016-01 addresses certain aspects of recognition, measurement, presentation, and disclosure of financial instruments and affects all entities that hold financial assets or owe financial liabilities. Among other provisions of the guidance, the amendments in this ASU supersede the guidance to classify equity securities with readily determinable fair values into different categories (that is, trading or available-for-sale) and require equity securities (including other ownership interests, such as partnerships, unincorporated joint ventures, and limited liability companies) to be measured at fair value with changes in the fair value recognized through net income. As such, upon the effective date of the amended guidance, an entity should classify cash flows from purchases and sales of equity securities on the basis of the nature and purpose for which it acquired the securities.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-01, see appendix G of this guide.

**6.20** Summarized in the following are some typical investing and financing cash flows that may be reported for a financial institution.

#### **Investing Activities**

Cash Inflows	Cash Outflows		
Loan principal payments	Loan originations		
Portfolio loan sale proceeds	Loan purchases		
Security sale and maturity proceeds (disclose separately for held-to-maturity securities and available-for-sale securities)*	Security purchases (disclose separately for held-to-maturity securities and available-for-sale securities)*		
Real estate sale proceeds	Investment in real estate held for development		
Net deposits withdrawn from other financial institutions	Net deposits placed with other financial institutions		
Sales of loan servicing rights	Purchases of loan servicing rights		
Net decrease in reverse repurchase agreements** (repos)	Net increase in repos*		

#### Investing Activities—continued

Decrease in National Credit Union Share Insurance Fund (NCUSIF) deposit	crease in NCUSIF deposit

<sup>\*</sup> See footnote 1.

#### **Financing Activities**

Cash Inflows	Cash Outflows	
Net increase in mortgage escrow deposits	Net decrease in mortgage escrow deposits	
Net certificates of deposit (CDs) issued	Net CDs matured	
Net increase in other deposit accounts	Net decrease in other deposit accounts	
Proceeds from Federal Home Loan Banks (FHLB) advances and other borrowings	Repayment of FHLB advances and other borrowings	
Net increase in short term borrowings (original maturity of three months or fewer)	Net decrease in short term borrowings (original maturity of three months or fewer)	
Proceeds from the issuance of common stock or other equity instruments	Reacquisition of equity instruments (for example, purchase of treasury stock)	
Net increase in repurchase agreements (repos) and dollar-roll repos	Net decrease in repos and dollar-roll repos	
	Dividends and other cash distributions to stockholders	

**6.21** Noncash investing and financing activities. According to paragraphs 3–4 of FASB ASC 230-10-50, information about all investing and financing activities of an entity during a period that affect recognized assets or liabilities but that do not result in cash receipts or cash payments in the period should be disclosed. Those disclosures may be either narrative or summarized in a schedule, and they should clearly relate the cash and noncash aspects of transactions involving similar items. Examples of noncash investing and financing activities are converting debt to equity; acquiring assets by assuming directly related liabilities, such as purchasing a building by incurring a mortgage to the seller; obtaining an asset by entering into a capital lease; obtaining a building or investment asset by receiving a gift; and exchanging noncash assets or liabilities for other noncash assets or liabilities.<sup>4</sup>

<sup>\*\*</sup> Repos are addressed in chapter 14, "Federal Funds and Repurchase Agreements," of this guide.

<sup>&</sup>lt;sup>4</sup> See footnote 3.

#### © Update 6-3 Accounting and Reporting: Leases

FASB ASU No. 2016-02, *Leases (Topic 842)*, issued in February 2016, is effective for fiscal years of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files financial statements with the SEC beginning after December 15, 2018, including interim periods within those fiscal years.

For all other entities, FASB ASU No. 2016-02 is effective for fiscal years beginning after December 15, 2019, and interim periods within fiscal years beginning after December 15, 2019.

Early application is permitted for all entities.

FASB ASU No. 2016-02 supersedes the lease requirements in FASB ASC 840, *Leases*, and creates FASB ASC 842, *Leases*, to establish the principles that lessees and lessors should apply to report useful information to users of financial statements about the amount, timing, and uncertainty of cash flows arising from a lease. FASB ASC 842 affects any entity that enters into a lease (as that term is defined in FASB ASU No. 2016-02), with some specified scope exceptions.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-02, see appendix F, "The New Leases Standard: FASB ASU No. 2016-02," of this guide.

**6.22** Other examples of noncash investing and financing activities for financial institutions may include

- originating a mortgage loan to finance the sale of foreclosed real estate or real estate held for development;
- acquiring a real estate property through, or in lieu of, foreclosure of the related loan:
- converting mortgage or other loans into mortgage-backed or other asset-backed securities (commonly referred to as securitizing loans); and
- selling or purchasing branch offices when the buyer assumes deposit liabilities in exchange for loans and other assets received from the seller, in which case only the cash paid, net of cash acquired or received, ordinarily should be reported as a cash outflow or inflow.

### Acquisition and Sales of Certain Securities and Loans

**6.23** Banks, brokers and dealers in securities, and other entities may carry securities and other assets in a trading account, as stated in paragraphs 18–21 of FASB ASC 230-10-45. Cash receipts and cash payments resulting from purchases and sales of securities classified as trading securities as discussed in FASB ASC 320 should be classified pursuant to FASB ASC 230 based on the nature and purpose for which the securities were acquired. Cash receipts and cash payments resulting from purchases and sales of other securities and

other assets should be classified as operating cash flows if those assets are acquired specifically for resale and are carried at fair value in a trading account. Cash receipts and cash payments resulting from acquisitions and sales of loans also should be classified as operating cash flows if those loans are acquired specifically for resale and are carried at fair value or at the lower of cost or fair value in accordance with FASB ASC 948, Financial Services—Mortgage Banking.

# © Update 6-4 Accounting and Reporting: Recognition and Measurement of Financial Assets and Financial Liabilities

FASB ASU No. 2016-01, issued in January 2016, is effective for fiscal years, and interim periods within those fiscal years, of a public business entity beginning after December 15, 2017.

For all other entities (including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting), FASB ASU No. 2016-01 is effective for fiscal years beginning after December 15, 2018, and interim period within fiscal years beginning after December 15, 2019. Early application is permitted for all other entities as of the fiscal years beginning after December 15, 2017, including interim periods within those fiscal years.

All entities may adopt the presentation guidance in paragraphs 5–7 of FASB ASC 825-10-45 for financial statements of fiscal years or interim periods that have not yet been issued or that have not yet been made available for issuance. Furthermore, entities that are not public business entities may elect not to disclose the information about fair value of financial instruments required by the "General" subsection of FASB ASC 825-10-50 in financial statements of fiscal years or interim periods that have not yet been made available for issuance. Except as indicated previously in this paragraph, early application is not permitted.

FASB ASU No. 2016-01 addresses certain aspects of recognition, measurement, presentation, and disclosure of financial instruments. Among other provisions of the guidance, the amendments in this ASU supersede the guidance to classify equity securities with readily determinable fair values into different categories (that is, trading or available-for-sale) and require equity securities (including other ownership interests, such as partnerships, unincorporated joint ventures, and limited liability companies) to be measured at fair value with changes in the fair value recognized through net income. Accordingly, upon the effective date of the amended guidance, cash receipts and cash payments resulting from purchases and sales of equity securities accounted for in accordance with FASB ASC 321, *Investments—Equity Securities*, should be classified pursuant to FASB ASC 230 based on the nature and purpose for which the securities were acquired.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-01, see appendix G of this guide.

©2016, AICPA AAG-DEP 6.23

**6.24** In applying the guidance stated in the previous paragraph, for the direct method, gross cash receipts and cash payments from these sources should be reported separately as operating cash flows consistent with FASB ASC 230-10-45-25. If the indirect method is used, only the net increases or decreases in loans and securities may be reported in reconciling net income to the net cash flow from operating activities consistent with FASB ASC 230-10-45-28.

### Gross and Net Cash Flows

- **6.25** Paragraphs 7–9 of FASB ASC 230-10-45 state that, generally, information about the gross amounts of cash receipts and cash payments during a period is more relevant than information about the net amounts of cash receipts and payments. However, the net amount of related receipts and payments provides sufficient information for the following:
  - Cash equivalents
  - Certain items for which turnover is quick, amounts are large, and maturity is short
  - Cash receipts and payments pertaining to any of the following providing that the original maturity of the asset or liability is three months or less:
    - Investments (other than cash equivalents)
    - Loans receivable
    - Debt
- **6.26** According to FASB ASC 230-10-45-8, for certain other items, such as demand deposits of a bank and customer accounts payable of a broker-dealer, the entity is substantively holding or disbursing cash on behalf of its customers. Only the net changes during the period in assets and liabilities with those characteristics need be reported because knowledge of the gross cash receipts and payments related to them may not be necessary to understand the entity's operating, investing, and financing activities.
- **6.27** Other items for which the institution is substantively holding, receiving, or disbursing cash on behalf of its customers, may include
  - negotiable order of withdrawal (NOW) and super NOW accounts,
  - savings deposits,
  - money market deposit accounts,
  - mortgage escrow funds, and
  - collections and remittances on loans serviced for others.
- **6.28** FASB ASC 942, Financial Services—Depository and Lending, provides industry-specific accounting and reporting guidance for depository and lending financial institutions. Paragraphs 1–2 of FASB ASC 942-230-45 state that banks, savings institutions, and credit unions are not required to report gross amounts of cash receipts and cash payments for (a) deposits placed with other financial institutions and withdrawals of those deposits, (b) time deposits accepted and the repayments of deposits, and (c) loans made to customers and principal collections of loans. When those entities constitute part of a consolidated entity, net amounts of cash receipts and cash payments for deposit or lending activities of those entities should be reported separate from gross amounts of cash receipts and cash payments for other investing and financing activities of the consolidated entity, including those of a subsidiary of a bank,

savings institution, or credit union that is not itself a bank, savings institution, or credit union.

### Cash Receipts and Payments Related to Hedging Activities

**6.29** FASB ASC 230-10-45-27 explains that cash flows from derivative instruments that are accounted for as fair value hedges or cash flow hedges may be classified in the same category as the cash flows from the items being hedged provided that the derivative instrument does not include an other-than-insignificant financing element at inception, other than a financing element inherently included in an at-the-market derivative instrument with no prepayments (that is, the forward points in an at-the-money forward contract) and that the accounting policy is disclosed. If the derivative instrument includes an other-than-insignificant financing element at inception, all cash inflows and outflows of the derivative instrument should be considered cash flows from financing activities by the borrower. If for any reason hedge accounting for an instrument that hedges an identifiable transaction or event is discontinued, then any cash flows after the date of discontinuance should be classified consistent with the nature of the instrument.

### Financial Statement Presentation and Disclosure

- **6.30** FASB ASC 942-305-50-1 states that restrictions on the use or availability of certain cash balances, such as deposits with a Federal Reserve Bank, FHLB, or correspondent financial institutions to meet reserve requirements or deposits under formal compensating balance agreements, should be disclosed in the notes to the financial statements.
- **6.31** FASB ASC 942-305-05-2 states that a financial institution that accepts deposits may have balances due from the same financial institution from which it has accepted a deposit, also called reciprocal balances. Reciprocal account balances, as explained in FASB ASC 942-305-45-1, should be offset if they will be offset in the process of collection or payment. Overdrafts of such accounts should be reclassified as liabilities, unless the financial institution has other accounts at the same financial institution against which such overdrafts can be offset.
- **6.32** The presentation of deposits in other depository institutions in the balance sheet varies among financial institutions. For example, if all or some portion of such deposits meet the definition of cash equivalent, as defined in the FASB ASC glossary, some institutions may combine all or the applicable portion of deposits in other institutions with cash and cash equivalents as the first line item in the balance sheet. Any portion of deposits not meeting the definition of cash equivalent may then be shown separately in the balance sheet, or it may be combined with other short term investments or other investments (if interest bearing); in either case, presented after cash and cash equivalents in the statement of financial condition. Alternatively, some institutions may segregate interest-bearing and noninterest-bearing deposits. Noninterest-bearing deposits that meet the definition of cash equivalent are typically combined with cash equivalents. Interest-bearing deposits in other institutions are presented separately in the balance sheet after cash and cash equivalents, or combined with other short term investments or other investments, regardless of whether all or some portion of such deposits meet the definition of cash equivalent. These practices are considered acceptable, provided that cash and cash equivalents in the balance sheet include only those instruments meeting the definition of cash

©2016, AICPA AAG-DEP 6.32

equivalents, and as discussed further in paragraph 6.09, herein, the institution discloses its policy used to classify items as cash equivalents.

**6.33** If deposits in other institutions are material, then deposits should be presented as a separate amount in the balance sheet, as stated in FASB ASC 942-210-45-4.

### Auditing<sup>5</sup>

### **Objectives**

 ${f 6.34}$  The primary audit objectives for cash are to obtain sufficient appropriate evidence that

- a. recorded balances exist and are owned by the institution;
- b. recorded balances are complete and stated at realizable amounts;
- c. balances are properly presented in the financial statements;
- d. restrictions on the availability or use of cash are appropriately identified and disclosed; and
- e. cash receipts, disbursements, and transfers between accounts are recorded in the proper period.

### **Planning**

**6.35** In accordance with AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (as described in chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide). Factors related to cash and cash equivalents that could influence the risks of material misstatement may include (a) cash and cash equivalents are generally negotiable, (b) involve large volumes of transactions, and (c) affect a large number of financial statement accounts.

# Internal Control Over Financial Reporting and Possible Tests of Controls

**6.36** AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13–.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit and, in doing so, should evaluate the design of those controls and determine whether

<sup>&</sup>lt;sup>5</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of issuers and nonissuers as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

**6.37** Because of the negotiability of the items included in cash, the large volume of activity in cash accounts, and the large number of accounts affected by cash transactions, the effectiveness of internal control in this area is an important factor in audit planning. Internal control over financial reporting and possible tests of controls related to the payments function, including wire transfers, are discussed in chapter 13, "Deposits," of this guide. Examples of control activities for cash balances include the following:

- Currency and coins are periodically counted and are reconciled to recorded amounts on a timely basis.
- Surprise counts of teller cash funds, vault cash, and cash items are performed periodically by persons other than those with related day-to-day responsibility.
- Tellers have exclusive access to and custody of their respective cash on hand.
- Access to night depositories (including ATM depositories) is under dual control (the control of more than one person), and at least two persons are present when the contents of depositories are removed, counted, listed, or otherwise processed.
- Cash transaction items are reviewed daily for propriety by an officer or a supervisory employee other than the custodian of the items.
- Each of the functions of draft issuance, register maintenance, and reconciliation is performed by a different employee.
- Confirmation requests received from depository institutions, supervisory examiners, and other parties are processed by an employee who does not also reconcile the subject account.
- Controls exist over access to and execution of official and certified checks.
- Controls exist over consignment items, such as traveler's checks or money orders that could easily be converted into cash.
- Cash and coin-counting equipment are periodically tested for accuracy.
- Currency that is mutilated or identified as counterfeit is segregated and reported.
- The replenishment of teller's cash is documented and reviewed by another employee.
- Vault cash is under the control of more than one person.
- Procedures exist for the credit evaluation of correspondent banking relationships.
- Records of ATM transactions are reconciled to their recording in books of entry on a daily basis.

- **6.38** AU-C section 330, *Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained* (AICPA, *Professional Standards*), addresses the auditor's responsibility to design and implement responses to the risks of material misstatement identified and assessed by the auditor in accordance with AU-C section 315 and to evaluate the audit evidence obtained in an audit of financial statements.
- **6.39** In accordance with paragraph .08 of AU-C section 330, the auditor should design and perform tests of controls to obtain sufficient appropriate audit evidence about the operating effectiveness of relevant controls if (a) the auditor's assessment of risks of material misstatement at the relevant assertion level includes an expectation that the controls are operating effectively or (b) substantive procedures alone do not provide sufficient appropriate audit evidence at the relevant assertion level. The following are examples of tests of controls to be considered:
  - Observing that the existing segregation of duties is adequate with respect to the handling and reconciliation of cash
  - Reading the documentation of surprise cash counts of teller, vault, ATM, and other cash on hand to determine whether documentation supports management's assertion that the surprise cash counts are performed periodically and in accordance with the institution's policies
  - Observing maintenance of control over mail receipts and supplies of consigned items
  - Inspecting and testing reconciliations to determine that they are performed and reviewed in a timely manner
- **6.40** Possible tests of controls related to electronic funds transfers are discussed in chapter 13 of this guide.

#### Substantive Tests

- **6.41** Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure, which for a financial institution would include cash and cash equivalents. In accordance with paragraph .445 of AU-C section 330, this requirement reflects the facts that (a) the auditor's assessment of risk is judgmental and may not identify all risks of material misstatement and (b) inherent limitations to internal control exist, including management override.
- **6.42** Substantive procedures that the auditor may consider include, but are not limited to, the following:
  - Counting cash and comparing the balances with tellers' records
  - Testing tellers' records for mathematical accuracy
  - Testing the reconciliations between recorded balances of cash due from correspondents and statements received from correspondents
  - Reviewing the cutoff of interbank transfers
  - Testing the reconciliations of subsidiary ledgers to the general ledger

- Testing the propriety of authorized accounts and signatures
- Reviewing the composition of suspense accounts, especially noting the recurring use and aging of reconciling items and any failure or inability to reconcile the cash account
- Confirming account balances with and reviewing the creditworthiness of correspondents
- Confirming consigned items with consignors
- Reviewing cash records for unusual transactions or adjustments
- Testing the propriety of due to and due from accounts set off in the balance sheet
- Testing fair value disclosures

©2016, AICPA AAG-DEP 6.42

### Chapter 7

## Investments in Debt and Equity Securities

Gray shaded text in this chapter reflects guidance issued but not yet effective as of the date of this guide, July 1, 2016, but becoming effective on or prior to December 31, 2016, exclusive of any option to early adopt ahead of the mandatory effective date. Unless otherwise indicated, all unshaded text reflects guidance that was already effective as of the date of this guide.

### Introduction

**7.01** Financial institutions acquire securities for various purposes. In addition to providing a source of income through investment or resale, securities are used to manage interest-rate and liquidity risk as part of an institution's overall asset/liability management strategies. They are also used in certain collateralized transactions. The most common securities acquired by institutions are described in the subsequent paragraphs. Investments that meet the definition of a *security* as defined in the FASB *Accounting Standards Codification* (ASC) glossary are discussed in this chapter. Other investments, including nonmarketable equity securities such as investments in Federal Home Loan Bank (FHLB) stock and Federal Reserve Bank stock, are discussed in chapter 12, "Other Assets, Other Liabilities, and Other Investments," of this guide.

**7.02** A *direct* relationship generally exists between risk and return (the higher the security's risk, the higher its expected yield). An *inverse* relationship generally exists between the security's liquidity and its yield; less liquid and longer-term securities generally have higher yields. Achieving the proper mix of safety, liquidity, and yield in an investment portfolio is one of the primary tasks of management. In managing their investment portfolios, financial institutions seek to maximize their returns without jeopardizing the liquidity the portfolios provide. Asset/liability management is discussed further in chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide.

**7.03** Management policies, adopted by the board of directors or its investment committee, establish authority and responsibility for investments in securities. Such policies may address investment objectives and guidelines, including specific position limits for each major type of investment, provisions for assessing risks of alternative investments, and policies on evaluating and selecting securities dealers and safekeeping agents. They also may set forth procedures for ensuring that management's investment directives are carried out and for gathering, analyzing, and communicating timely information about investment transactions.

**7.04** The institution generally should have procedures to analyze alternative securities (including complex derivative securities which are defined as securities whose value is derived from that of some underlying asset(s)) according to the institution's intent, with consideration of the level of management expertise, the sophistication of the institution's control procedures and monitoring systems, its asset/liability structure, and its capacity to maintain liquidity and absorb losses out of capital. For example, analyses prepared for derivative securities prior to purchase would generally include sensitivity analyses that show the effect on the carrying amount and net interest income

©2016, AICPA AAG-DEP 7.04

of various interest-rate, credit loss, and prepayment scenarios. Such analyses may also evaluate the effect of investment securities on the institution's overall exposure to interest-rate risk. An analysis might also be performed to evaluate the reasonableness of interest-rate, credit loss, and prepayment assumptions provided by the selling broker, and management may obtain price quotes from more than one broker prior to executing a trade. Management may also review contractual documents to ascertain the rights and obligations of all parties to the transaction, as well as the recourse available to each party.

### **U.S. Government and Agency Obligations**

**7.05** The Department of the Treasury (U.S. Treasury), as fiscal agent for the United States, routinely sells federal government debt securities called *treasuries or T-bills*. Backed by the full faith and credit of the United States, treasuries are generally considered the reference for the risk-free rate and are highly liquid. The income they provide is generally exempt from state and local taxes. Accordingly, treasuries are used by institutions as a primary source of liquidity.

**7.06** T-bills are the shortest term obligations, having original maturities of one year or less. T-bills are sold at a discount from their face value; income to T-bill investors is the difference between the purchase price and the face value. U.S. Treasury notes and bonds (T-notes and T-bonds, respectively) are longer-term obligations that pay interest in semiannual coupon payments. T-notes have original maturities between 1 and 10 years; T-bonds have maturities of 10 years or longer.

7.07 The debt of U.S. government agencies, such as the Government National Mortgage Association (Ginnie Mae), and government-sponsored enterprises (GSEs), such as the Federal Home Loan Mortgage Corporation (Freddie Mac) and Federal National Mortgage Association (Fannie Mae), trades at yields above treasury yields but historically below that of high credit quality corporate debt. The agencies and GSEs issue debentures, notes, and other debt securities having a wide variety of maturities and other features. The GSEs, as public shareholder owned companies, were placed into conservatorship on September 6, 2008, at the direction of the Secretary of the Treasury, the Chairman of the Board of Governors of the Federal Reserve System (Federal Reserve) and the Director of Federal Housing Finance Agency. This changed the perception of both GSEs from being implicitly government backed to explicitly government backed along with Ginnie Mae.

### **Municipal Obligations**

**7.08** State and local governments and their agencies (such as housing, school, or sewer authorities) issue notes and bonds of various maturities. Many municipal bonds are callable: they may be redeemed by the municipality before the scheduled maturity date. Tax anticipation notes, so named under the expectation that they will shortly be repaid with tax receipts, generally mature within one year and are usually purchased directly from the state or local government at a negotiated price. Revenue and bond anticipation notes are similarly issued and retired with certain expected revenues or proceeds from the expected sale of bonds. Municipal bonds may be either general obligation (that is, backed by the full taxing authority of the issuer) or limited obligation (that is, used to finance specific long term public projects, such as building a

school). Municipal bonds are purchased through a competitive bidding process or in the secondary market.

- **7.09** Municipal obligations vary significantly in risk. Credit quality depends heavily on the ability and willingness of the municipality to service its debt or the profitability of the particular project being financed. Liquidity also varies. A number of municipal obligations are traded actively on over-the-counter markets; others are thinly traded. Interest on most municipal obligations is exempt from taxes in the state of the municipality; exemption from federal income taxes depends on the extent to which the obligations benefit private parties rather than the public. (See chapter 16, "Income Taxes," of this guide for additional discussion of tax-exempt income.)
- **7.10** According to the FASB ASC glossary, a *conduit debt security* is defined as certain limited-obligation revenue bonds, certificates of participation, or similar debt instruments issued by a state or local governmental entity for the express purpose of providing financing for a specific third party (the conduit bond obligor) that is not part of the state or local government's financial reporting entity. Although conduit debt securities bear the name of the governmental entity that issues them, the governmental entity often has no obligation for such debt beyond the resources provided by a lease or loan agreement with the third party on whose behalf the securities are issued. Further, the conduit bond obligor is responsible for any future financial reporting requirements.
- **7.11** The definition of *public entity*, as stated in the FASB ASC glossary, includes a conduit bond obligor for conduit debt securities that are traded in a public market (a domestic or foreign stock exchange or an over-the-counter market, including local or regional markets).

#### **Asset-Backed Securities**

- **7.12** Asset-backed securities (ABSs) are financial instruments that derive their value and receive cash flows from other financial assets (such as mortgage loans or credit card receivables). ABSs historically provided a great level of liquidity to financial markets, allowed for a wide variety of innovative products, and, because they often involve incrementally more risk, offered better yields than other investments.
- **7.13** ABSs are highly versatile because cash flows from the underlying assets can be reconfigured through any number of structures for repayment to ABS investors. ABSs allow the issuer to enhance the marketability of the underlying assets, for example, by spreading liquidity and credit risk across broad pools, or by providing a higher yield to those investors willing to accept a higher concentration of the risks associated with specific cash flows from the collateral.
- **7.14** This chapter focuses on ABSs from the perspective of the security holder. Chapter 10, "Transfers and Servicing and Variable Interest Entities," and chapter 15, "Debt," of this guide discuss matters unique to depository institutions that issue ABSs.
- **7.15** A given ABS structure generally involves any number of investment classes (or tranches) with various degrees of risk and reward. Among other common characteristics, ABSs
  - are issued by both governmental and private issuers, including banks and savings institutions;

©2016, AICPA AAG-DEP 7.15

- generally include a credit allocation structure (for example, payout order of priority) or some form of credit enhancement to limit
  the credit risk of the underlying assets. For example, an issuer
  or third party may guarantee that the ABS principal and interest
  will be repaid as scheduled regardless of whether cash is received
  from payments on the underlying collateral; and
- are often issued in book-entry form. That is, no physical certificates change hands; rather, ownership is recorded on the investor's account.
- **7.16** The largest volumes of ABSs issued are backed by real estate mortgage loans (mortgages) and are called mortgage-backed securities (MBSs). Other types of collateral that have been used in ABS issuances include credit card receivables, treasuries (in synthetic structures), car loans, recreational vehicle loans, mobile home loans and trust preferred securities (TPSs) issued by financial institutions and insurance companies. MBSs and other mortgage securities are discussed in the following paragraphs to provide examples of risk characteristics and other matters that may be encountered with various forms of ABSs and their collateral.
- **7.17** *MBSs*. The simplest form of the ABS is the basic (or *plain vanilla*) MBS, created by pooling a group of similar mortgages. The collateral of a MBS may be residential real estate or commercial real estate. Most MBSs are issued with a stated minimum principal amount and interest rate and represent a pro rata share in the principal and interest cash flows to be received as the underlying mortgages are repaid by the mortgagors. The mortgages underlying the issuance typically have
  - a. the same type of collateral, such as single-family residential real estate:
  - b. fixed or adjustable interest rates within a specified range; and
  - c. maturities within a specified range.
- **7.18** MBSs are securities issued by a GSE or government agency (for example, Ginnie Mae, Freddie Mac, or Fannie Mae) or by private issuers (for example, banks, and mortgage banking entities). MBSs may be mortgage participation certificates or pass-through certificates, which represent an undivided interest in a pool of specific mortgage loans. Periodic payments on Ginnie Mae, Freddie Mac, and Fannie Mae participation certificates are backed by those agencies.
- **7.19** Risk characteristics of MBSs. Because the repayment of MBSs is contingent on repayment of the underlying loans, the risk characteristics of specific MBS issuances are driven by the risk characteristics of the underlying loans. For example, underlying mortgages insured by the Federal Housing Administration (FHA) would typically involve less credit risk than unguaranteed conventional mortgages.
- **7.20** More complex MBS structures. More complex MBS structures concentrate or dilute risk to create a range of possible investments with unique risks and rewards. As described in the following paragraphs, an understanding of the structure and nature of a specific MBS is necessary to understand the related risks. This understanding is important, for example, when evaluating whether an investment in a particular tranche of an MBS is other-than-temporarily impaired under FASB ASC 320, *Investments—Debt and Equity Securities*. Such

an understanding is also important in making an assessment of whether an embedded derivative exists that would require bifurcation under FASB ASC 815, *Derivatives and Hedging*.

- **7.21** *Credit risk.* To make a particular issuance of MBS more attractive to potential investors, the credit risk associated with mortgages underlying MBSs is generally reduced through some form of *credit enhancement*, such as
  - a. a letter of credit;
  - guarantee of scheduled principal or interest payments, often achieved through a transaction with a federal agency such as Ginnie Mae, or a GSE such as Freddie Mac or Fannie Mae;
  - guarantee of all or a portion of scheduled principal and interest payments through insurance of the pool by a private mortgage insurer;
  - d. overcollateralization of the issuance, where cash flows from the excess collateral are used to make up for delinquent collateral payments; and
  - e. a senior/subordinated (senior/sub) structure, in which one group of investors holds a subordinated interest in the pool by accepting all or a large portion of the related credit risk in return for a greater yield.
- **7.22** The degree of protection from credit risk offered by the various types of credit enhancement generally needs to be considered in relation to the characteristics of the collateral and, therefore, is unique to each security. Further, when credit risk is addressed through a credit enhancement, the security holder is still at risk that the third-party guarantor or private insurer could default on its responsibility. (The risk that another party to a transaction will default on its obligations under the transaction is referred to as *counterparty risk*.) Many MBS issuances carry credit ratings assigned by an independent rating agency.
- 7.23 Interest rate risk and prepayment risk. The overall return—or yield—earned on a mortgage depends on the amount of interest earned over the life of the loan and any discount received or premium paid at acquisition. Mortgage yields, therefore, are highly sensitive to the fact that most mortgages can be repaid before their scheduled maturity date without penalty. Although the owner of a mortgage receives the full amount of principal when prepaid, the interest income that would have been earned during the remaining period to maturity—net of any discount or premium amortization—is lost.
- **7.24** As with individual mortgages, the actual maturities and yields of MBSs depend on when the underlying mortgage principal and interest are repaid. If market interest rates fall below a mortgage's contractual interest rate, it is generally to the borrower's advantage to prepay the existing loan and obtain new financing at a new, lower rate. Accordingly, expected prepayments must be estimated to predict and account for the yield on MBSs.
- **7.25** In addition to changes in interest rates, actual mortgage prepayments depend on other factors such as loan types and maturities, the geographical location of the related properties (and associated regional economies), seasonality, age and mobility of borrowers, and whether the loans are assumable, as are certain loans insured by the FHA or guaranteed by the U.S. Department of Veterans' Affairs. Prepayments are also dependent on availability of liquidity in the market; during the recession that begin in 2007, interest rates

decreased as a result of market factors and the Federal Reserve's monetary policy, but prepayments did not increase as expected because financial institutions were approving less refinances than they would in a normalized market environment.

- **7.26** Some MBSs are backed by adjustable-rate mortgages (ARMs). Interest rates on ARMs change periodically based on an independent factor plus an interest-rate spread, which is expressed as a specified percentage (1 percent, also referred to as one *point*) or one one-hundredth of a percentage (.01 percent, also referred to as one *basis point*). For example, an ARM might carry a rate that changes every 6 months based on the average rate on 1 year treasuries plus 2 points. Annual increases in an ARM's interest rate are generally capped, as are total interest-rate increases over the life of the loan.
- **7.27** Yields on ARMs follow changes in stipulated interest rate indexes. This, and the fact that some ARMs may been issued with *teaser* rates that are significantly below market rates as a way to attract borrowers, make it more difficult to predict the overall risk of investments in ARM MBSs. The frequency of interest-rate adjustments, the index, the initial interest rate, and the annual and lifetime caps all generally should be considered. For example, credit risk may be higher for ARM MBSs because when interest rates rise borrowers' ability to pay is diminished as their monthly payments increase.
- **7.28** Changes in the indexed rates of certain ARMs lag behind changes in prevailing rates. When interest rates are falling, adjustable-rate MBSs generally trade at a premium, although frequently they are prepaid as borrowers seek to lock in lower fixed rates. Conversely, when interest rates are rising, adjustable-rate MBSs generally trade at a discount.
- **7.29** Other MBSs. Other MBSs add layers of complexity to the security structure to create investment classes that meet the needs of and are attractive to more potential investors. Security holders find certain investment classes attractive because they can purchase the cash flows they desire most, or can synthetically create a security with the desired interest rate and prepayment characteristics. As discussed previously, MBSs offer pro rata shares in principal and interest cash flows with stated principal amounts and interest rates, and are subject to credit, prepayment, and other risks. More complex MBSs are used to further restructure the cash flows and risks so that investment classes may offer features which include the following:
  - Different anticipated maturities
  - A single final payment (called a zero-coupon class) rather than monthly, quarterly, or semiannual installments
  - Floating interest rates, even though the underlying assets have fixed rates
  - Repayment on a specified schedule, unless mortgage prepayments go outside a prescribed range (called a *planned amortization class*)
  - Protection against faster but not slower prepayments (called a *targeted amortization class*)
  - Rights to interest cash flows only, called *interest-only securities* (IOs), or to principal cash flows only, called *principal-only securities* (POs)
  - Rights only to those cash flows remaining after all other classes have been repaid (a *residual interest* or *residual*)

- Lower risk of default and resulting higher investment ratings
- Higher yield (generally achieved through subordination)
- **7.30** These and other specialized classes—and the fact that some MBSs use pools of MBSs rather than pools of mortgages as collateral—make analysis of investments in MBSs complex. Accordingly, such instruments could expose an institution to substantial risk if not understood or effectively managed by the institution.
- 7.31 Two common forms of multiclass MBSs are collateralized mortgage obligations (CMOs) and real estate mortgage investment conduits (REMICs). CMOs are bonds secured by (and repaid with) the cash flows from collateral MBSs or mortgages and generally involve some form of credit enhancement. The collateral is generally transferred to a special-purpose entity (SPE) which may be organized as a trust, a corporation, or a partnership. The SPE is an entity created by an asset transferor or sponsor to carry out a specific purpose, activity or series of transactions directly related to its specific purpose. The SPE then becomes the issuer of the CMO (see consolidation considerations for SPEs and former qualifying special purpose entities in paragraph 7.115 and beginning in paragraph 10.97 of this guide. Accordingly, a security holder may invest in a CMO in equity form (for example, trust interests, stock, and partnership interests) or nonequity form (for example, participating debt securities). REMICs are a form of CMO specially designated for federal income tax purposes so that the related income is taxed only once (to the security holder). See chapter 16 of this guide.
- **7.32** Understanding the risks associated with a particular tranche of a MBS or other ABS often requires an understanding of the security structure, as documented in the offering document and related literature. The term *tranche* refers to one of several related securities offered that are collateralized by the same group of assets. Tranches from the same offering usually have different risks, rewards, or maturity characteristics.
- **7.33** *Risk analysis*. A discussion of the risks associated with every possible form of MBS or other ABS is beyond the scope of this Audit and Accounting Guide. A basic understanding of the relationship between interest and principal cash flows, in addition to an understanding of related risks (such as credit risk), is needed to analyze investments in MBSs. The following discussion uses IOs, POs, and residuals as examples of ABS classes for this purpose. The discussion of the senior/sub structure used in some issuances also highlights the importance of understanding the structure and the form of credit enhancement when evaluating an investment in mortgage products or other ABSs.
- **7.34** Investment classes that have a contractual right only to certain interest cash flows (interest classes), such as MBS IOs, are extremely interest-rate and prepayment sensitive and, therefore, carry the risk that the security holder's entire recorded investment could be lost. Investment classes weighted toward solely principal cash flows (principal classes), such as MBS POs, also carry special risks. The following discussion of related risk concepts can be applied to various other investments in MBSs or other ABSs.
- **7.35** *Interest classes and IOs.* Interest classes receive all, or substantially all, of the interest cash flows from the underlying collateral mortgages. Accordingly, they have been found to be useful vehicles for managing the interest-rate risk inherent in mortgage portfolios, because prepayments cause the value of IOs to move in the opposite direction from that of mortgages and traditional

©2016, AICPA AAG-DEP 7.35

fixed-income securities. However, because of the sensitivity of IOs to interest rates, the recorded investment in an IO may be lost if actual prepayments are higher than anticipated.

- **7.36** Changes in the prices (and, therefore, the values) of MBSs are heavily dependent on whether the collateral's interest rates are above or below prevailing interest rates. A mortgage will trade at a discount (a discount mortgage) when it carries an interest rate lower than prevailing interest rates. A mortgage that carries an interest rate above prevailing rates will trade at a premium (a premium mortgage).
- 7.37 An IO backed by a pool of premium mortgages may be a more useful tool for controlling interest-rate risk than one backed by a pool of discount mortgages because it often shows greater appreciation in value when interest rates increase and often does not suffer as significant a decrease in value when interest rates fall. Falling interest rates generally result in greater prepayments. Accordingly, the cash generated from an IO over its life usually decreases because interest is earned on a smaller remaining principal balance. Although the discounting of the stream of interest receipts at a lower interest rate increases the present value of each future dollar of interest, the negative effect of increased prepayments generally outweighs the positive discounting effect, and, therefore, the fair value of the IO generally declines. IOs generally increase in value in a rising rate environment because as prepayments slow, the related mortgage principal balance remains outstanding for a longer period, and, therefore, interest is earned for a longer period (although the present value of each of those future dollars is reduced by the higher discount rate).
- **7.38** Principal classes and POs. Principal classes are often issued at deep discounts from the contractual principal amount because the security holder receives no interest. In contrast to zero-coupon bonds, whose entire principal amount is paid at maturity, the principal amount of POs is paid periodically according to repayment of the underlying mortgage principal. If the security holder has the ability to hold the PO to maturity, only credit risk or counterparty default would prevent ultimate recovery of the recorded investment. The fair value of a PO is also dependent on the effects of prepayments and discounting, both of which are dependent on interest rates.
- **7.39** The fair value of a PO often tends to increase as prepayments accelerate, because the security holder receives the return of principal more quickly. Conversely, as prepayments slow, the value of the PO often tends to decline. A PO backed by discount mortgages tends to appreciate more as interest rates fall than would a PO backed by premium mortgages. However, when interest rates rise, a PO backed by discount mortgages would not decline in value as much as a PO backed by premium mortgages. The difference in fair values reflects the relationship between prepayment rates and the stated interest rates on the collateral backing the POs. Prepayments on discount POs are generally significantly lower than prepayments on premium POs. As interest rates decline, prepayments on both types of POs will accelerate. However, prepayments on premium POs do not increase as much, because prepayments on these instruments are usually already at a high level. Conversely, when interest rates rise, prepayments on underlying discount mortgages do not slow significantly, because they are usually already at a relatively low level, but prepayments on underlying premium mortgages decline sharply.
- **7.40** A decline or increase in interest rates similarly causes the fair value of expected cash flows from POs to increase or decrease, respectively, because

of related changes in the discount rate used to determine the present value of any future cash flows.

- **7.41** Because POs generally increase in value in response to declining interest rates, they are sometimes used to manage the interest-rate risk associated with investments in mortgage servicing rights, CMO or REMIC residuals, and IOs. However, institutions in liability-sensitive positions (that is, institutions whose liabilities will reprice more quickly than their assets) would be negatively affected by an increase in interest rates, and, therefore, the use of POs to manage the interest-rate risk of such assets may be counterproductive because such a strategy may increase the institution's overall exposure to interest-rate risk.
- **7.42** Residual classes. From a legal perspective, residuals represent an ownership interest in the underlying collateral, subject to the first lien and indenture of the other security holders. Residuals entitle the holder to the excess, if any, of the issuer's cash inflows (including reinvestment earnings) over cash outflows (which often include any debt service and administrative expenses). Three sources of residual cash flows typically exist:
  - a. The differential between interest cash flows on the collateral and interest payments on other investment classes
  - b. Any overcollateralization provided as a credit enhancement
  - c. Any income earned on reinvestment of other cash flows before they are distributed to other security holders (because payments on collateral mortgages are received monthly but some investment classes are repaid quarterly or semiannually, these receipts are reinvested in the interim)
- **7.43** Residuals are often designed to reduce the prepayment and credit risk of other classes and to provide security holders with the potential for high yields. Residuals may earn high yields if prepayments and credit losses of the underlying collateral are not greater than the rate assumed at the time the issuance was structured and sold. Residuals are particularly sensitive to prepayments, and the residual holder's recorded investment may be lost entirely if actual prepayments are higher than anticipated. They also represent the riskiest tranche of securities, as they are first in line to absorb credit losses, which is why they have the potential for high yields. As with POs and IOs, their fair values are dependent on the effect of discounting.
- **7.44** Typically residuals absorb credit losses. However, they could also absorb other risks such as interest rate risk and related prepayment risk. Although other investment classes may receive triple-A credit ratings, residuals are usually not rated, because they are so susceptible to interest-rate risk. Even if a residual is rated triple-A, such a rating often indicates only that the rating agency expects that the minimum required payments of principal, interest, or both will be received (that is, that credit risk is perceived to be low), not that a security holder will realize the anticipated yield.
- **7.45** As with other investment classes, the return on and fair value of a residual is dependent on the underlying collateral, the security structure, and its performance under varying interest-rate and prepayment scenarios. Residuals may carry fixed- or floating-interest rates.
- **7.46** The fair value of fixed-rate residuals often increases as interest rates increase and decreases as interest rates decline. The main source of cash flow on

a fixed-rate residual comes from the interest differential between the interest payments received on the underlying collateral mortgages and the interest payments made on other investment classes. Because short term classes usually carry lower interest rates than longer-term classes, residual cash flows from the interest differential tend to be greatest in earlier years after issuance, before the short term classes have been repaid. Accordingly, the longer the lower-rate classes remain outstanding, the greater the cash flow accruing to the residual class. As interest rates decline, prepayments accelerate, the interest differential narrows, and overall cash flows decline. Conversely, as interest rates climb, prepayments slow, generating a larger cash flow to residual holders.

**7.47** As with fixed-rate residuals, the main source of cash flow to floatingrate residuals is the interest differential between interest earned on the collateral and interest paid on other investment classes. However, because one or more of the classes is tied to a floating rate, the interest differential can change when the rates on floating-rate classes are reset. For example, when interest rates rise, the rate on the floating-rate class may be reset at a higher rate. More of the cash flows from the underlying collateral may then be paid to the floating-rate classes, leaving less cash flow for the residual. Higher interest rates also tend to cause prepayments to slow, and thereby increase the period over which the interest-differential income is earned by the residual holders. Conversely, when interest rates decline, rates on floating-rate classes decrease, but prepayments of premium mortgages would tend to accelerate. The loss of interest income as a result of prepayments would typically offset a widening of the interest differential stemming from the lower rate on the floating-rate class, thus reducing the cash flow to the residual. Thus, changes in interest rates produce two opposing effects on the fair value of floating-rate residuals. Whether the value of the residual actually declines or rises when interest rates change depends on the interrelationship between factors such as the interest on the floating-rate class and mortgage prepayment speeds.

**7.48** Senior/sub securities. The senior/sub form of credit enhancement is often used for conventional mortgages. A senior/sub issuance generally divides the offered securities into two risk classes, namely, a senior class and one or more subordinated classes. The subordinated classes, often retained by the sponsor of the ABS, provide credit protection to the senior class. When cash flows on the underlying mortgages are impaired, the cash is first directed to make principal and interest payments on the senior-class securities. Furthermore, some cash receipts may be held in a reserve fund to meet any future shortfalls of principal and interest to the senior class. The subordinated classes may not receive debt-service payments until all of the principal and interest payments have been made on the senior class and, where applicable, until a specified level of funds has been contributed to the reserve fund.

**7.49** Subordinated classes generally carry higher interest rates and are often unrated because of the higher credit risk. Accordingly, subordinated classes are not usually purchased to be held to maturity. The fair value of subordinated securities, like the fair value of other MBSs, depends on the nature of the underlying collateral and how changes in interest rates affect cash flows on the collateral. The fair value would also reflect any reserve fund priorities and the increased credit risk associated with the securities.

#### Other Structured Credit Products

**7.50** The following discussion addresses certain common characteristics of other structured credit products such as structured notes, TPSs, and pooled

TPSs. These general characteristics along with those unique to the various securities are important considerations when evaluating the related risks.

- **7.51** On April 30, 2009, the FDIC issued Financial Institution Letter (FIL)-20-2009, "Risk Management of Investments in Structured Credit Products." According to the FDIC FIL, the term *structured credit products* may be broadly defined to refer to all structured investment products where repayment is derived from the performance of the underlying assets or other reference assets, or by third parties that serve to enhance or support the structure. Such products include MBSs and CMOs, as previously discussed. These products also include, but are not limited to, structured investment vehicles, collateralized debt obligations (CDOs), including securities backed by TPSs, and other ABSs. As previously noted, a discussion of the risks associated with all forms of structured credit product is beyond the scope of this guide.
- **7.52** Structured notes. Structured notes are hybrid securities that combine fixed term, fixed or variable rate instruments, and derivative products. Structured notes are debt securities issued by corporations or GSEs, including FHLB, Fannie Mae, and Freddie Mac. Structured notes generally contain embedded options and have cash flows that are linked to the indexes of various financial variables, such as interest rates, foreign exchange rates, commodity prices, prepayment rates, and other financial variables. Structured notes can be linked to different market sectors or interest rate scenarios, such as the shape of the yield curve, the relationship between two different yield curves, or foreign exchange rates.
- 7.53 TPSs. Investments in TPSs are also hybrid instruments possessing characteristics typically associated with debt obligations. Although each issue of these securities may involve minor differences in terms, under the basic structure of TPSs a corporate issuer, such as a bank holding company, first organizes a business trust or other special purpose entity. This trust issues two classes of securities: common securities, all of which are purchased and held by the corporate issuer, and TPSs, which are sold to investors. The corporate issuer makes periodic interest payments on the subordinated debentures to the business trust, which uses these payments to pay periodic dividends on the TPSs to the investors. The subordinated debentures have a stated maturity and may also be redeemed under other circumstances. Most TPSs are subject to a mandatory redemption upon the repayment of the debentures. Because of the mandatory redemption provision in the typical TPS, investments in TPSs would normally be considered debt securities under FASB ASC 320.
- **7.54** TPSs have a payment deferral feature of up to 5 years and, in general, have a 30 year maturity. In the event of bankruptcy, TPSs are below all senior and subordinated debt, but above equity securities in priority. In addition, TPSs generally are rated by nationally recognized rating firms.
- **7.55** Pooled TPSs. In a pooled TPSs offering, an additional trust is added to the structure and is referred to as a business trust. A pooled TPS is a form of CDO backed by various TPSs. The business trust issues securities to investors and uses the proceeds to purchase all of the TPSs from the grantor trust. The TPSs are then securitized, as the business trust is the sole investor of the securities.
- **7.56** Trust preferred pools issue several classes, or tranches, of securities, which include senior, mezzanine, and residual (also referred to as *income*) tranches. The senior tranches have priority of payment, with the subordinated

residual interest and mezzanine positions absorbing losses earlier and serving as credit protection for the senior tranches. The TPSs include credit support provisions such that if performance in the underlying collateral deteriorates below certain levels, cash flows are diverted from the residual tranches to pay the senior and mezzanine tranches. Further deterioration may result in diversion of cash flows from the holders of the mezzanine tranches to pay the holders of the senior tranches. Securitization documents explain the priority of payments and the credit support levels of the various tranches.

- **7.57** Understanding the list of issuers that compose the portfolio, the current investment ratings, the status of the collateral (for example, the extent of any deferrals or defaults), and the credit quality of the institutions remaining in the pool are important aspects of these investments when considering the related risks.
- **7.58** The risks associated with pooled TPSs are heavily dependent upon the position in the securitization structure. It is important to understand the structure of the security, the priority of payments, and current credit support levels (for example, whether income is being diverted from the subordinate tranches to support the credit protection level of the senior tranches).
- **7.59** The underlying collateral for a significant number of structured credit products, including those previously addressed, has performed poorly during difficult credit environments. Due to the complex nature of these investments, the structural characteristics and the underlying collateral at the pool and individual investment level are important factors to be considered when evaluating the related risks and determining an appropriate valuation.

### Issues of International Organizations and Foreign Governments

**7.60** International financial institutions and foreign governments and their political subdivisions increasingly rely on international capital markets for funds. A significant portion of international debt securities is denominated in U.S. dollars. The credit risk and liquidity risk vary for different issues, though many are high quality and widely traded. Institutions have also obtained foreign debt securities of financially troubled countries in troubled debt restructurings; such securities are generally lower quality and not widely traded.

### **Other Securities**

**7.61** Other securities held by depository institutions, where permitted by applicable laws and regulations, include the following:

- Common-trust or mutual-investment funds
- Investments in negotiable certificates of deposit<sup>1</sup>
- Equity securities, including venture capital investments
- Corporate bonds and commercial paper

The credit quality and risk of these instruments are unique to the particular issuance. The financial strength of the issuer and other counterparties is a major determinant and may be evidenced by an investment rating.

<sup>&</sup>lt;sup>1</sup> Technical Questions & Answers section 2130.40, "Certificates of Deposit and FASB ASC 320" (AICPA, Technical Questions and Answers), addresses whether certificates of deposit fall within the scope of FASB Accounting Standards Codification (ASC) 320, Investments—Debt and Equity Securities.

#### Transfers of Securities

- **7.62** Short sales. Short sales are trading activities in which an institution transfers securities it does not own, with the intention of buying or borrowing securities at an agreed-upon future date to cover the transfer. Securities are "sold short" for protection against losses, for short term borrowing of funds, for arbitrage, or in anticipation of a decline in market prices.
- **7.63** Borrowing and lending securities. Sometimes an institution will borrow securities from a counterparty or from its trust customers' assets when the institution is obligated to deliver securities it does not own. Examples are in a short sale, to settle a repurchase agreement, or because a counterparty may have failed to deliver securities the institution needed for delivery to another counterparty. The institution, therefore, uses borrowed securities to fulfill its obligation until it actually receives the securities it has purchased. Institutions also may loan securities to a counterparty.
- **7.64** An institution may advance cash, pledge other securities, or issue letters of credit as collateral for borrowed securities. The amount of cash or other collateral deemed necessary may increase or decrease depending on changes in the value of the securities.
- **7.65** Repurchase and reverse repurchase agreements. Sometimes an institution will enter into an agreement with a counterparty to sell (or purchase) securities and then buy (or sell) them back at a specified date. If the repurchase agreement is for the same or substantially similar securities, the transaction is not recorded as a sale (or purchase). Instead, the securities continue to be recorded as assets of the transferor and the transaction is recorded as a secured borrowing. Repurchase and reverse repurchase agreements are discussed further in chapter 14, "Federal Funds and Repurchase Agreements," of this guide.

### **Regulatory Matters**

- **7.66** Federal laws and regulations place certain restrictions on the types of financial instruments that an institution may deal in, underwrite, purchase, and sell. Transactions in certain securities, such as those backed by the full faith and credit of the United States, are generally unrestricted. Holdings of other securities—of any one obligor—are generally limited based on capitalization. Restrictions on dealing in or underwriting in the security may also apply. Additional restrictions may apply to state-chartered institutions.
- **7.67** Banks and savings institutions. The Federal Financial Institutions Examination Council (FFIEC) issued the Supervisory Policy Statement on Investment Securities and End-User Derivatives Activities dated April 23, 1998, that was adopted by all of the federal banking agencies. The policy statement provides guidance to financial institutions on sound practices for managing the risks of investment securities and end-user derivatives activities. The guidance describes the practices that a prudent manager normally would follow, but it emphasizes that it is not intended to be a checklist and management should establish practices and maintain documentation appropriate to the institution's individual circumstances.
- **7.68** The FFIEC supervisory policy statement applies to all securities in held-to-maturity and available-for-sale accounts as described in FASB ASC 320, certificates of deposit held for investment purposes, and end-user derivative contracts not held in trading accounts. This guidance covers all securities

©2016, AICPA AAG-DEP 7.68

used for investment purposes, including money market instruments, fixed-rate and floating-rate notes and bonds, structured notes, mortgage pass-through and other ABSs and mortgage-derivative products. Similarly, the guidance covers all end-user derivative instruments used for nontrading purposes, such as swaps, futures, and options.

- **7.69** The FFIEC supervisory policy statement also describes sound principles and practices for managing and controlling the risks associated with investment activities. Institutions should fully understand and effectively manage the risks inherent in their investment activities. The policy statement emphasizes that failure to understand and adequately manage the risks in these areas constitutes an unsafe and unsound practice.
- **7.70** Board of director and senior management oversight is an integral part of an effective risk management program. The board of directors is responsible for approving major policies for conducting investment activities, including the establishment of risk limits. Senior management is responsible for the daily management of an institution's investments. Institutions with significant investment activities ordinarily should ensure that back-office, settlement, and transaction reconciliation responsibilities are conducted or managed by qualified personnel who are independent of those initiating risk taking positions.
- **7.71** An effective risk management process for investment activities includes (a) policies, procedures, and limits; (b) the identification, measurement, and reporting of risk exposures; and (c) a system of internal control. The policy statement identifies sound practices for managing specific risks involved in investment activities. These risks include
  - market risk,
  - credit risk,
  - liquidity risk,
  - operational (transaction) risk, and
  - legal risk.

In addition, institutions are reminded to follow any specific guidance or requirements from their primary supervisor related to these activities.

- **7.72** As previously noted, the FDIC issued FIL-20-2009, which clarifies the application of existing supervisory guidance to structured credit products, including the 1998 supervisory guidance discussed in paragraphs 7.67–.71. Topics addressed in FIL-20-2009 include investment suitability and due diligence, the use of external credit ratings, pricing and liquidity, and adverse classification of investment securities.
- **7.73** FIL-20-2009 addresses concerns about the nonagency structured credit market. However, some of the clarifications in this guidance also are relevant for agency securities. See paragraphs 7.51–.60 regarding further discussion of other structure credit products.
- **7.74** The FDIC's *Risk Management Manual of Examination Policies* includes a subchapter titled "Securities and Derivatives" which references these interagency guidance documents.
- **7.75** On December 13, 1999, the federal banking agencies, including the Office of Thrift Supervision (OTS) (prior to its transfer of powers to the Office of

the Comptroller of the Currency [OCC], the FDIC, and the Federal Reserve),<sup>2</sup> jointly released the Interagency Guidelines on Asset Securitization that highlight the risks associated with asset securitization and emphasize the agencies' concerns with certain retained interests generated from the securitization and sale of assets. The guidelines set forth the supervisory expectation that the value of retained interests in securitizations must be supported by objectively verifiable documentation of the assets' fair market value, utilizing reasonable, conservative valuation assumptions. Retained interests that do not meet such standards or that fail to meet the supervisory standards outlined in the guidance will be disallowed as assets of the bank for regulatory capital purposes. The guidance stresses the need for bank management to implement policies and procedures that include limits on the amount of retained interests that may be carried as a percentage of capital. Institutions that lack effective risk management programs or engage in practices deemed to present other safety and soundness concerns may be subject to more frequent supervisory review, limitations on retained interest holdings, more stringent capital requirements, or other supervisory response.

7.76 Federal and state savings associations. Thrift Bulletin (TB) 73a, Investing in Complex Securities, states that TPSs that otherwise meet the requirements of corporate debt securities³ are permissible investments for federal savings associations. Savings associations are, however, prohibited from purchasing TPSs or any other type of security from the parent holding company or any other affiliate. TB 73a can be found on the "Office of Thrift Supervision Thrift Bulletin" page at www.occ.gov and should be referred to for additional limitations and requirements for holding these securities. National banks and state nonmember banks are permitted to invest in trust preferred stock within certain limitations. (See OCC Interpretive Letter No. 777, April 8, 1997; FDIC FIL-16-99, February 16, 1999.) The Dodd-Frank Wall Street Reform and Consumer Protection Act (Dodd-Frank Act) did not change the prohibitions.

7.77 The Dodd-Frank Act authorized the FDIC to issue a final rule to establish standards of creditworthiness to replace previously used investment grade standards for corporate debt security investments held or acquired by savings associations. In July 2012, the FDIC's board of directors, through the issuance of FIL-34-2012, Investments in Corporate Debt Securities by Savings Associations, adopted the final rule Permissible Investments for Federal and State Savings Associations: Corporate Debt Securities. The final rule prohibits state and federal savings associations from acquiring or holding a corporate debt security when the security's issuer does not have an adequate capacity to meet all financial commitments under the security for the projected life of the security. In addition, the FDIC also issued final guidance that sets forth due diligence standards for determining the credit quality of a corporate debt security. Readers are encouraged to review the full text of the final rule and guidance in FIL-34-2012 from www.fdic.gov.

**7.78** The Volcker rule of the Dodd-Frank Act, among other things, prohibits banks and other financial institutions from acquiring or retaining any ownership interests in or sponsorship of *covered funds* as defined in the rule. A

 $<sup>^2</sup>$  See chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide for further discussion on the Office of Thrift Supervision transfer of powers.

<sup>&</sup>lt;sup>3</sup> The federal savings association requirements of corporate debt securities are set forth in Title 12 U.S. *Code of Federal Regulations* (CFR) Part 160.40 (Office of the Comptroller of Currency).

recent focus has been on whether collateralized loan obligations (CLOs) qualify as covered funds. Although not exempt from the requirements, the Federal Reserve has extended the compliance deadline for CLOs by two years. See paragraph 7.86 for accounting considerations resulting from the Volcker rule.

**7.79** Credit unions. Federal regulations describe investments allowed for federal credit unions.<sup>4,5</sup> These regulations explicitly prohibit federal credit unions from (a) purchasing mortgage servicing rights as an investment, (b) investing in stripped MBSs or certain securities that represent interests in stripped MBSs, (c) purchasing residual interests in CMOs, REMICs, or small-business-related securities, and (d) engaging in adjusted trading or short sales  $^6$ 

**7.80** Federally insured state-chartered credit unions are required under terms of the insurance agreement to establish an investment valuation reserve (displayed as an appropriation of retained earnings) for nonconforming investments. Nonconforming investments are those investments permissible under state law for a state-chartered credit union, but which are impermissible for federally chartered credit unions.

**7.81** In October 2010, the National Credit Union Administration (NCUA) issued amendments to its rule governing corporate credit unions. The major revisions involved corporate credit union capital, investments, asset-liability management, governance, and credit union service organization activities. In regards to investments, the final amendments now involve a rigorous investment screening process prior to purchase. Some of the significant changes within the process include (a) Nationally Recognized Statistical Rating Organization ratings screen, (b) additional prohibition of certain highly complex and leveraged securities (specifically, a CDO, net interest margin security, private label residential MBS, or security subordinated to any other securities in the issuance), (c) single obligor limits tightened from 50 percent of capital to 25 percent of capital, (d) portfolio weighted average life (WAL) not to exceed two years, and (e) portfolio WAL (assuming prepayment slowdown of 50 percent) not to exceed 2.5 years. In addition, some corporations may hold investments that are in violation of one or more of these new prohibitions, and these investments will be subject to the investment action plan provisions.

### **Bank Accounting Advisory Series**

**7.82** The OCC's *Bank Accounting Advisory Series* (BAAS) is updated periodically to express the Office of the Chief Accountant's current views on accounting topics of interest to national banks and federal savings associations. Banks prepare their Consolidated Reports of Condition and Income using U.S. generally accepted accounting principles (GAAP) and regulatory requirements. Accordingly, responses contained in the series are based on GAAP and

 $<sup>^4\,</sup>$  For further discussion of regulations related to permissible investments, see 12 CFR 703.14.

<sup>&</sup>lt;sup>5</sup> In January 2014, the National Credit Union Administration (NCUA) amended its investment regulations to permit federal credit unions to engage in limited derivatives activities for the purpose of mitigating interest rate risk (see 12 CFR 703.14[k]). NCUA derivatives regulations are set forth in 12 CFR 703 Subpart B.

 $<sup>^6</sup>$  For further discussion of regulations related to prohibited investment activities and prohibited investments, see 12 CFR 703.15-.16.

 $<sup>^7</sup>$  The requirements for corporate credit union investments and the investment action plan are set forth in 12 CFR 704.5 and 704.10.

regulatory requirements. These advisories are not official rules or regulations of the OCC; but rather, represent either interpretation by the OCC's Office of the Chief Accountant of GAAP, or OCC interpretations of regulatory capital requirements. Topic 1, "Investment Securities" of the BAAS includes interpretations on (a) investment in debt and equity securities and (b) other-than-temporary-impairment (OTTI). Readers are encouraged to view this publication under the "Publications—Bank Management" page at www.occ.gov.

### **Accounting and Financial Reporting**

# © Update 7-1 Accounting and Reporting: Recognition and Measurement of Financial Assets and Financial Liabilities

FASB Accounting Standards Update (ASU) No. 2016-01, Financial Instruments—Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities, issued in January 2016, is effective for fiscal years, and interim periods within those fiscal years, of a public business entity beginning after December 15, 2017.

For all other entities (including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting), FASB ASU No. 2016-01 is effective for fiscal years beginning after December 15, 2018, and interim periods within fiscal years beginning after December 15, 2019. Early application is permitted for all other entities as of the fiscal years beginning after December 15, 2017, including interim periods within those fiscal years.

All entities may adopt the presentation guidance in paragraphs 5–7 of FASB ASC 825-10-45 for financial statements of fiscal years or interim periods that have not yet been issued or that have not yet been made available for issuance. Furthermore, entities that are not public business entities may elect not to disclose the information about fair value of financial instruments required by the "General" subsection of FASB ASC 825-10-50 in financial statements of fiscal years or interim periods that have not yet been made available for issuance. Except as indicated previously in this paragraph, early application is not permitted.

FASB ASU No. 2016-01 addresses certain aspects of recognition, measurement, presentation, and disclosure of financial instruments and affects all entities that hold financial assets or owe financial liabilities. Among other provisions of the guidance, the amendments in this ASU supersede the guidance to classify equity securities with readily determinable fair values into different categories (that is, trading or available-for-sale) and require equity securities (including other ownership interests, such as partnerships, unincorporated joint ventures, and limited liability companies) to be measured at fair value with changes in the fair value recognized through net income (other than equity securities accounted for under the equity method of accounting or those that result in consolidation of an investee).

The amendments in this ASU also require an entity to present separately in other comprehensive income the portion of the total change in fair value of a liability resulting from a change in the instrument-specific credit risk when the entity has elected to measure the liability at fair value in accordance with the fair value option for financial instruments.

©2016, AICPA AAG-DEP 7.82

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-01, see appendix G, "Accounting for Financial Instruments," of this guide.

#### © Update 7-2 Accounting and Reporting: Credit Losses

FASB ASU No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, issued in June 2016, is effective for fiscal years of public business entities that are SEC filers beginning after December 15, 2019, including interim periods within those fiscal years.

For all other public business entities, the amendments in FASB ASU No. 2016-13 are effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years. For all other entities, including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting, FASB ASU No. 2016-13 is effective for fiscal years beginning after December 15, 2020, and interim periods within fiscal years beginning after December 15, 2021.

Early application is permitted for all entities as of the fiscal years beginning after December 15, 2018, *including* interim periods within those fiscal years.

FASB ASU No. 2016-13 creates FASB ASC 326, Financial Instruments—Credit Losses, to amend guidance on reporting credit losses for assets held at amortized cost basis and available-for-sale debt securities.

For assets held at amortized cost basis, FASB ASC 326 eliminates the probable initial recognition threshold in current GAAP and, instead, requires an entity to reflect its current estimate of all expected credit losses. The allowance for credit losses is a valuation account that is deducted from the amortized cost basis of the financial assets to present the net amount expected to be collected.

For available-for-sale debt securities, credit losses should be measured in a manner similar to current GAAP. However, FASB ASC 326 will require that credit losses be presented as an allowance rather than as a write-down.

FASB ASU No. 2016-13 affects entities holding financial assets and net investment in leases that are not accounted for at fair value through net income. The amendments affect loans, debt securities, trade receivables, net investments in leases, off-balance-sheet credit exposures, reinsurance receivables, and any other financial assets not excluded from the scope that have the contractual right to receive cash.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-13, see appendix G of this guide.

### Introduction

**7.83** FASB ASC 320 addresses accounting and reporting for investments in equity securities that have readily determinable fair values and for all investments in debt securities, according to FASB ASC 320-10-05-2.8 FASB ASC 320-10-25-1 requires that at acquisition, an entity should classify debt securities and equity securities into one of these three categories:

- a. Held-to-maturity securities. Investments in debt securities should be classified as held-to-maturity only if the reporting entity has the positive intent and ability to hold those securities to maturity. According to FASB ASC 320-10-35-1, investments in debt securities classified as held-to-maturity should be measured subsequently at amortized cost in the statement of financial position.
- b. Trading securities. If a security is acquired with the intent of selling it within hours or days, the security should be classified as trading. However, at acquisition an entity is not precluded from classifying as trading a security it plans to hold for a longer period. Classification of a security as trading should not be precluded simply because the entity does not intend to sell it in the near term. According to FASB ASC 320-10-35-1, trading securities should be measured subsequently at fair value in the statement of financial position and unrealized holding gains and losses should be included in earnings.
- c. Available-for-sale securities. Available-for-sale securities are investments in debt and equity securities that have readily determinable fair values not classified as held-to-maturity securities or trading securities. According to FASB 320-10-35-1, available-for-sale securities are measured subsequently at fair value in the statement of financial position, with unrealized holding gains and losses excluded from earnings and reported as other comprehensive income until realized except as indicated in the following sentence. All or a portion of the unrealized holding gain and loss of an available-for-sale security that is designated as being hedged in a fair value hedge should be recognized in earnings during the period of the hedge, pursuant to paragraphs 1–4 of FASB ASC 815-25-35.

**7.84** FASB ASC 825, *Financial Instruments*, allows entities to choose, at specified election dates, to measure eligible items at fair value (the fair value option) with gains and losses recorded in earnings at each subsequent reporting date. Chapter 20, "Fair Value," of this guide provides a summary of FASB ASC 825 but is not intended as a substitute for reading the guidance in FASB ASC 825. Investments in securities may contain embedded derivatives that, absent a fair value option election, could require bifurcation and separate accounting under FASB ASC 815.

**7.85** Paragraphs 4–18 of FASB ASC 320-10-25 describe certain circumstances not consistent with held-to-maturity classification and circumstances

©2016, AICPA AAG-DEP 7.85

<sup>&</sup>lt;sup>8</sup> In accordance with FASB ASC 320-10-20, equity securities are defined as any security representing an ownership interest in an entity (for example, common, preferred, or other capital stock) or the right to acquire (for example, warrants, rights, and call options) or dispose of (for example, put options) an ownership interest in an entity at fixed or determinable prices. Debt securities are defined as securities that represent a creditor relationship with an entity, such as treasury securities or corporate bonds.

that are consistent with held-to-maturity classification. FASB ASC 320-10-25-6 addresses changes in circumstances that may cause the entity to change its intent to hold a certain security to maturity without calling into question its intent to hold other debt securities to maturity in the future. Entities rarely transfer securities out of the held-to-maturity category because a single transfer may call into question management's intent and ability to hold the entire portfolio to maturity. FASB ASC 320-10-25-18 provides specific scenarios where sale or transfer of a held-to-maturity security will not call into question an investor's stated intent to hold other debt securities to maturity in the future.

- **7.86** Compliance with the provisions of the Volcker rule (see paragraph 7.78) may have current accounting consequences. For example, entities should evaluate whether a positive intent and ability to hold held-to-maturity debt securities to maturity still exists. In addition, if it is more likely than not that an entity will be required to sell a debt security with a fair value that is less than its amortized cost, the manner in which impairment is recognized could be impacted.
- **7.87** Paragraphs 10–16 of FASB ASC 320-10-35 address transfers of securities between categories. The transfer of a security between categories of investments should be accounted for at fair value. In addition, given the nature of a trading security, transfers into or from the trading category also should be rare. Paragraphs 75–77 of FASB ASC 860-10-55 give an example addressing whether a transferor has the option to classify debt securities as trading at the time of a transfer.
- **7.88** FASB ASC 948-310-40-1 states that, after the securitization of a mortgage loan held for sale that meets the conditions for a sale addressed in FASB ASC 860-10-40-5, any MBSs received by a transferor as proceeds should be classified in accordance with the provisions of FASB ASC 320. However, FASB ASC 948-310-35-3A states that a mortgage banking entity should classify as trading any retained MBSs that it commits to sell before or during the securitization process. An entity is prohibited from reclassifying loans as investment securities unless the transfer of those loans meets the conditions for sale accounting addressed in FASB ASC 860-10-40-5.
- **7.89** FASB ASC 740-20-45-11b states that the tax effects of gains and losses included in comprehensive income but excluded from net income (for example, changes in the unrealized holding gains and losses of securities classified as available-for-sale under FASB ASC 320), as defined in the FASB ASC glossary, occurring during the year should be charged or credited directly to other comprehensive income or to related components of shareholders' equity.
- **7.90** FASB ASC 320-10-25-5a states that a security should not be classified as held-to-maturity if that security can contractually be prepaid or otherwise settled in such a way that the holder of the security would not recover substantially all of its recorded investment. The justification for using historical-cost-based measurement for debt securities classified as held-to-maturity is that no matter how market interest rates fluctuate, the holder will recover its recorded investment and thus realize no gains or losses when the issuer pays

<sup>&</sup>lt;sup>9</sup> If the remaining held-to-maturity portfolio is tainted and reclassified to the available-for-sale classification, the length of the taint period under U.S. generally accepted accounting principles is not defined. The SEC recommended a two year time frame minimum for the tainting period. Refer to the guidance in the *Bank Accounting Advisory Series* Question 9 (December 2001) Topic 1A, "Investments in Debt and Equity Securities."

the amount promised at maturity. However, that justification does not extend to receivables purchased at a substantial premium over the amount at which they can be prepaid, and it does not apply to instruments whose payments derive from prepayable receivables but have no principal balance. Therefore, a callable debt security purchased at a significant premium might be precluded from held-to-maturity classification under FASB ASC 860-20-35-2 if it can be prepaid or otherwise settled in such a way that the holder of the security would not recover substantially all of its recorded investment. In addition, a mortgage-backed interest-only certificate should not be classified as held-tomaturity. Paragraphs 3-6 of FASB ASC 860-20-35 provide further guidance. Note that a debt security that is purchased late enough in its life such that. even if it was prepaid, the holder would recover substantially all of its recorded investment, could be initially classified as held-to-maturity if the conditions of FASB ASC 320-10-25-5 and FASB ASC 320-10-25-1 are met. (A debt security that can contractually be prepaid or otherwise settled in such a way that the holder of the security would not recover substantially all of its recorded investment may contain an embedded derivative. Therefore, such a security should be evaluated in accordance with FASB ASC 815-15 to determine whether it contains an embedded derivative that needs to be accounted for separately).

**7.91** See chapter 20 of this guide for a summary of FASB ASC 820, Fair Value Measurement.

#### OTTI

- **7.92** Paragraphs 17–35A of FASB ASC 320-10-35 provide guidance regarding the impairment of individual available-for-sale and held-to-maturity securities, including the scope of impairment guidance, the steps for identifying and accounting for impairment, and the recognition of an OTTI. See paragraphs 7.126–.133 and FASB ASC 320-10-50 for disclosure requirements related to OTTI.
- **7.93** For individual securities classified as either available-for-sale or held-to-maturity, an entity should determine whether a decline in fair value below the amortized cost basis is other than temporary as stated in FASB ASC 320-10-35-18. Providing a general allowance for unidentified impairment in a portfolio of securities is not appropriate. Paragraphs 20–29 of FASB ASC 320-10-35 provide guidance an entity should follow in assessing impairment of individual securities classified as either available-for-sale or held-to-maturity. If the fair value of an investment is less than its amortized cost basis at the balance sheet date of the reporting period for which impairment is assessed, the impairment is either temporary or other than temporary, as stated in FASB ASC 320-10-35-30.
- **7.94** Equity securities. FASB ASC 320-10-35-32A states that for equity securities, an entity should apply the guidance that is pertinent to the determination of whether an impairment is other than temporary, such as FASB ASC 323-10-35 and FASB ASC 325-40-35. If it is determined that the impairment is other than temporary, then an impairment loss should be recognized in earnings equal to the entire difference between the investment's cost and its fair value at the balance sheet date of the reporting period for which the assessment is made as stated in FASB ASC 320-10-35-34. The measurement of the impairment should not include partial recoveries after the balance sheet date. The fair value of the investment would then become the new amortized

cost basis and that cost basis should not be adjusted for subsequent recoveries in fair value.

- **7.95** *Debt securities.* For debt securities, readers may refer to paragraphs 33A–33I of FASB ASC 320-10-35 for guidance on evaluating whether an impairment is other than temporary. Paragraphs 34A–34E of FASB ASC 320-10-35 provide guidance on determining the amount of an OTTI recognized in earnings and other comprehensive income if the impairment is other than temporary. The following paragraphs highlight certain sections of this guidance.
- **7.96** If an entity intends to sell the impaired debt security (that is, it has decided to sell the security), an OTTI should be considered to have occurred. If an entity does not intend to sell the impaired debt security, the entity should consider available evidence to assess whether it more likely than not will be required to sell the security before the recovery of its amortized cost basis (for example, whether its regulatory obligations indicate that the security will be required to be sold before a forecasted recovery occurs). If the entity more likely than not will be required to sell the security before recovery of its amortized cost basis, an OTTI should be considered to have occurred.
- **7.97** In assessing whether the entire amortized cost basis of the security will be recovered, an entity should compare the present value of cash flows expected to be collected from the security with the amortized cost basis of the security. If the present value of cash flows expected to be collected is less than the amortized cost basis of the security, the entire amortized cost basis of the security will not be recovered (that is, a credit loss exists), and an OTTI should be considered to have occurred.
- **7.98** In determining whether a credit loss exists, an entity should use its best estimate of the present value of cash flows expected to be collected from the debt security. One way of estimating that amount would be to consider the methodology described in FASB ASC 310-10-35 for measuring an impairment on the basis of the present value of expected future cash flows. Briefly, the entity would discount the expected cash flows at the effective interest rate implicit in the security at the date of acquisition. All of the factors, listed in FASB ASC 320-10-35-33F should be considered when estimating whether a credit loss exists and the period over which the debt security is expected to recover. This list is not meant to be all inclusive.
- **7.99** In making its OTTI assessment, an entity should consider the factors in paragraphs 33G-33H in FASB ASC 320-10-35. An entity also should consider how other credit enhancements affect the expected performance of the security, including consideration of the current financial condition of the guarantor of a security (if the guarantee is not a separate contract as discussed in FASB ASC 320-10-35-23) or whether any subordinated interests are capable of absorbing estimated losses on the loans underlying the security. The remaining payment terms of the security could be significantly different from the payment terms in prior periods (such as some securities backed by nontraditional loans; see FASB ASC 825-10-55-1). Thus, an entity should consider whether a security backed by currently performing loans will continue to perform when required payments increase in the future (including balloon payments). An entity also should consider how the value of any collateral would affect the expected performance of the security. If the fair value of the collateral has declined, an entity should assess the effect of that decline on the ability of the entity to collect the balloon payment.

- **7.100** If an OTTI has occurred, the amount of the OTTI recognized in earnings depends on whether an entity intends to sell the security or more likely than not will be required to sell the security before recovery of its amortized cost basis less any current-period credit loss. If an entity does not intend to sell the debt security and it is not more likely than not that the entity will be required to sell the security before recovery of its amortized cost basis less any current-period credit loss, the OTTI should be separated into both of the following:
  - a. The amount representing the credit loss
  - b. The amount related to all other factors

The amount of the total OTTI related to the credit loss should be recognized in earnings. The amount of the total OTTI related to other factors should be recognized in other comprehensive income, net of applicable taxes.

- **7.101** Subsequent increases and decreases (if not an OTTI) in the fair value of available-for-sale securities should be included in other comprehensive income, according to FASB ASC 320-10-35-35.
- **7.102** The OTTI recognized in other comprehensive income for debt securities classified as held-to-maturity should be accreted over the remaining life of the debt security in a prospective manner on the basis of the amount and timing of future estimated cash flows, as stated in FASB ASC 320-10-35-35A. That accretion should increase the carrying value of the security and should continue until the security is sold, the security matures, or there is an additional OTTI that is recognized in earnings.
- **7.103** A decline in the value of a security that is other than temporary is also discussed in AU-C section 501, *Audit Evidence—Specific Consideration for Selected Items* (AICPA, *Professional Standards*), and the SEC *Codification of Staff Accounting Bulletins* topic 5(M), "Other Than Temporary Impairment of Certain Investments in Equity Securities."

#### **Unrealized Gains and Losses**

7.104 FASB ASC 220-10-45-1 requires an entity to report comprehensive income either in a single continuous financial statement or in two separate but consecutive financial statements. According to paragraphs 1A-1C of FASB ASC 220-10-45, an entity reporting comprehensive income in a single continuous financial statement should present its components in two sections: net income and other comprehensive income. If applicable, an entity should present in that financial statement a total amount for net income together with the components that make up net income, a total amount for other comprehensive income together with the components that make up other comprehensive income, and total comprehensive income. An entity reporting comprehensive income in two separate but consecutive statements should present components of and the total for net income in the statement of net income and components of and the total for other comprehensive income as well as total for comprehensive income in the statement of other comprehensive income, which should be presented immediately after the statement of net income. A reporting entity should begin the second statement with net income. An entity should present, either in a single continuous statement of comprehensive income or in a statement of net income and statement of other comprehensive income, all items that meet the definition of comprehensive income for the period in which those items are recognized. Components included in other comprehensive income should be

classified based on their nature. For related guidance, see paragraphs 10A–10B of FASB ASC 220-10-45.

**7.105** FASB ASC 220-10-45-15 requires that reclassification adjustments be made to avoid double counting of items in comprehensive income that are presented as part of net income for a period that also had been presented as part of other comprehensive income in that period or earlier periods. For example, gains on investment securities that were realized and included in net income of the current period that also had been included in other comprehensive income as unrealized holding gains in the period in which they arose must be deducted through other comprehensive income of the period in which they are included in net income to avoid including them in comprehensive income twice (see FASB ASC 320-10-40-2). Example 3 (see paragraphs 18–27 of FASB ASC 220-10-55) illustrates the presentation of reclassification adjustments in accordance with this paragraph.

**7.106** FASB ASC 220-10-45-17 requires an entity to provide information about the effects on net income of significant amounts reclassified out of each component of accumulated other comprehensive income if those amounts all are required under other FASB ASC topics to be reclassified to net income in their entirety in the same reporting period. An entity should provide this information together, in one location, either (a) on the face of the financial statement where net income is presented or (b) as a separate disclosure in the notes to the financial statements. FASB ASC 220-10-45-17A describes the information requirements for presentation on the face of the statement where net income is presented, and FASB ASC 220-10-45-17B describes the information requirements for disclosure in the notes to the financial statements. FASB ASC 220-10-45-18B states that nonpublic entities are not required to meet the requirements in paragraphs 17–17B of FASB ASC 220-10-45 for interim reporting periods but are required to meet them for annual reporting periods.

#### **Premiums and Discounts**

**7.107** An institution will often pay less (or more) for a security than the security's face value. Accretion of the resulting discount (or amortization of the premium), together with the stated coupon on the security, represents the effective rate of interest on the security, thereby reflecting the security's market yield.

**7.108** Dividend and interest income, <sup>10</sup> including amortization of the premium and discount arising at acquisition, for all three categories of investments in securities should be included in earnings, according to FASB ASC 320-10-35-4.

**7.109** For a debt security transferred into the held-to-maturity category from the available-for-sale category, the unrealized holding gain or loss at the date of the transfer should continue to be reported in a separate component

<sup>&</sup>lt;sup>10</sup> For debt securities for which other-than-temporary impairments were recognized in earnings, the difference between the new amortized cost basis and the cash flows expected to be collected should be accreted as interest income in accordance with applicable guidance as stated in FASB ASC 320-10-35-35.

FASB Accounting Standards Update (ASU) No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, issued in June 2016, creates FASB ASC 326, Financial Instruments—Credit Losses, to amend guidance on reporting credit losses for assets held at amortized cost basis and available for sale debt securities. Update 7-2 provides further details on the amendments and effective date of ASU No. 2016-13.

of shareholders' equity, such as accumulated other comprehensive income, but should be amortized over the remaining life of the security as an adjustment of yield in a manner consistent with the amortization of any premium or discount, as stated in item (d) in FASB ASC 320-10-35-10. The amortization of an unrealized holding gain or loss reported in equity will offset or mitigate the effect on interest income of the amortization of the premium or discount) for that held-to-maturity security. For a debt security transferred into the held-to-maturity category, the use of fair value may create a premium or discount that, under amortized cost accounting, should be amortized thereafter as an adjustment of yield pursuant to FASB ASC 310-20.

- **7.110** The guidance in FASB ASC 310-20 explicitly includes the accounting for discounts, premiums, and commitments fees associated with the purchase of loans and other debt securities such as corporate bonds, treasury notes and bonds, groups of loans, and loan-backed securities, as stated in item (b) in FASB ASC 310-20-15-2. FASB ASC 310-20-35-18 specifies that net fees or costs that are required to be recognized as yield adjustments over the life of the related loan(s) should be recognized by the interest method except as set forth in paragraphs 21–24 of FASB ASC 310-20-35. The objective of the interest method is to arrive at periodic interest income (including recognition of fees and costs) at a constant effective yield on the net investment in the receivable (that is, the principal amount of the receivable adjusted by unamortized fees or costs and purchase discount or premium).
- **7.111** FASB ASC 942-320-35-1 states that the period of amortization or accretion for debt securities should generally extend from the purchase date to the maturity date, not an earlier call date. FASB ASC 310-20-35-26 explains that if the entity holds a large number of similar loans for which prepayments are probable and the timing and the amount of the prepayments can be reasonably estimated, the entity may consider estimates of future principal prepayments in the calculation of the constant effective yield necessary to apply the interest method.
- **7.112** Certain ABSs may meet those conditions, and institutions may therefore consider estimates of prepayments in determining the amortization period for calculation of the constant effective yield.
- **7.113** In accordance with FASB ASC 310-20-35-26, if the institution anticipates prepayments in applying the interest method and a difference arises between the anticipated prepayments and the actual prepayments received, the effective yield should be recalculated to reflect actual payments to date and anticipated future payments. The net investment in the loans should be adjusted to the amount that would have existed had the new effective yield been applied since the acquisition of the loans. The investment in the loans should be adjusted to the new balance with a corresponding charge or credit to interest income.

#### Interest Income

**7.114** Guidance for recognizing interest income on debt securities is included within FASB ASC 310-20, 320-10, 325-40, and 310-30. The appropriate model to apply depends on a number of factors including, but not limited to, the nature of the investment, its creditworthiness, and whether the asset was originated or purchased in the secondary market.

#### Consolidation

**7.115** A reporting entity that holds a direct or indirect (explicit or implicit) variable interest in a legal entity must determine whether the guidance in the "Variable Interest Entities" subsections of FASB ASC 810-10 applies to that legal entity before considering other consolidation guidance. However, if a reporting entity does not have a direct or indirect (explicit or implicit) variable interest in a legal entity, then the reporting entity is not the primary beneficiary of that legal entity and is not required to provide disclosures for that legal entity under FASB ASC 810-10 "Variable Interest Entities" subsections. Variable interest entities (VIEs) may appear in various forms, such as TPSs, synthetic leases, asset-backed commercial paper conduits, and CDOs. See detailed discussion of consolidation considerations related to VIEs beginning in paragraph 10.97 of this guide.

## **Special Areas**

- **7.116** The following paragraphs list several specialized accounting issues involving investments in securities.
- **7.117** Cost basis of debt security received in restructuring. FASB ASC 310-40-40-8A states that the initial cost basis of a debt security of the original debtor received as part of a debt restructuring should be the security's fair value at the date of the restructuring. Any excess of the fair value of the security received over the net carrying amount of the loan should be recorded as a recovery on the loan. Any excess of the net carrying amount of the loan over the fair value of the security received should be recorded as a charge-off to the allowance for credit losses. Subsequent to the restructuring, the security received should be accounted for according to the provisions of FASB ASC 320.
- **7.118** Sales of marketable securities with put arrangements. Paragraphs 20–23 of FASB ASC 860-20-55 address transactions that involve the sale of a marketable security to a third-party buyer, with the buyer having an option to put the security back to the seller at a specified future date or dates for a fixed price. If the transfer is accounted for as a sale, a put option that enables the holder to require the writer of the option to reacquire for cash or other assets a marketable security or an equity instrument issued by a third party should be accounted for as a derivative by both the holder and the writer, provided the put option meets the definition of a derivative in FASB ASC 815-10-15-83 (including meeting the net settlement requirement, which may be met if the option can be net settled in cash or other assets or if the asset required to be delivered is readily convertible to cash).

#### Transfers and Servicing of Securities

- **7.119** FASB ASC 860-10 establishes accounting and reporting standards for transfers and servicing of financial assets, according to FASB ASC 860-10-05-1. FASB ASC 860-10-05-6 also provides an overview of the types of transfers addressed, including securitization, factoring, transfers of receivables with recourse, securities lending transactions, repurchase agreements, loan participation, and banker's acceptances. Refer to chapter 10 of this guide for additional guidance regarding transfers and servicing of securities.
- **7.120** Trade date accounting. FASB ASC 942-325-25-2 states that regular-way purchases and sales of securities should be recorded on the trade date. Gains and losses from regular-way security sales or disposals should be

recognized as of the trade date in the statement of operations for the period in which securities are sold or otherwise disposed of. $^{11}$ 

**7.121** Short sales. As stated in FASB ASC 942-405-25-1 and FASB ASC 942-405-35-1, the obligations incurred in short sales should be reported as liabilities. Such liabilities are generally called securities sold, not yet purchased. The obligations should be subsequently measured at fair value through the income statement at each reporting date. Interest on the short positions should be accrued periodically and reported as interest expense. The fair value adjustment should be classified in the income statement with gain and losses on securities, according to FASB ASC 942-405-45-1.

## **Troubled Debt Restructurings**

**7.122** In accordance with item (a) in FASB ASC 320-10-55-2, any loan that was restructured as a security in a troubled debt restructuring involving a modification of terms would be subject to the provisions of FASB ASC 320 if the debt instrument meets the definition of a security. See FASB ASC 310-40-40-8A for additional information.

7.123 FASB ASC 310-40 addresses measurement, derecognition, disclosure and implementation guidance issues concerning creditor's treatment of troubled debt restructurings. According to FASB ASC 310-40-15-4, receivables that may be involved in troubled debt restructurings commonly result from lending cash, or selling goods or services on credit. Examples are accounts receivable, notes, debentures and bonds (whether those receivables are secured or unsecured and whether they are convertible or nonconvertible), and related accrued interest, if any. FASB ASC 310-40-35-5 states that a creditor in a troubled debt restructuring involving only a modification of terms of a receivable—that is, not involving receipt of assets (including an equity interest in the debtor)—should account for the troubled debt restructuring in accordance with the provisions of FASB ASC 310-40. This topic is discussed in more detail in chapter 8, "Loans," of this guide.

**7.124** Troubled debt restructuring may involve debt securities, including instances in which there is a substitution of debtors, which is addressed in FASB ASC 310-40-25-2.

# Loans and Debt Securities Acquired With Deteriorated Credit Quality

**7.125** FASB ASC 310-30 provides recognition, measurement, and disclosure guidance regarding loans acquired with evidence of deterioration of credit quality since origination acquired by completion of a transfer for which it is probable, at acquisition, that the investor will be unable to collect all contractually required payments receivable, according to FASB ASC 310-30-05-1. FASB ASC 310-30 is discussed in more detail in chapter 8 of this guide.

#### Financial Statement Presentation and Disclosure

**7.126** *OTTIs.* Paragraphs 8A and 9A of FASB ASC 320-10-45 state that in periods in which an entity determines that a security's decline in fair value below its amortized cost basis is other than temporary, the entity should present

©2016, AICPA

<sup>&</sup>lt;sup>11</sup> Chapter 10, "Transfers and Servicing and Variable Interest Entities," of this guide discusses accounting for transfers of loans that have not been previously securitized.

the total OTTI in the statement of earnings with an offset for the amount of the total OTTI that is recognized in other comprehensive income, in accordance with FASB ASC 320-10-35-34D, if any. FASB ASC 320-10-55-21A illustrates the application of this guidance. An entity should separately present, in the financial statement in which the components of accumulated other comprehensive income are reported, amounts recognized therein related to held-tomaturity and available-for-sale debt securities for which a portion of an OTTI has been recognized in earnings.

- **7.127** FASB ASC 320-10-50 addresses disclosures about OTTIs for debt and equity securities and requires a detailed, risk-oriented breakdown of major security types and related information. Disclosures related to this guidance are required for all interim and annual periods.
- **7.128** Paragraphs 2 and 5 of FASB ASC 320-10-50 require that the following information about available-for-sale and held-to-maturity securities be disclosed separately for each of those categories:
  - a. For securities classified as available-for-sale, all reporting entities should disclose all of the following by major security type as of each date for which a statement of financial position is presented:
    - i. Amortized cost basis
    - ii. Aggregate fair value
    - iii. Total OTTI recognized in accumulated other comprehensive income
    - iv. Total gains for securities with net gains in accumulated other comprehensive income
    - v. Total losses for securities with net losses in accumulated other comprehensive income
    - vi. Information about the contractual maturities of those securities as of the date of the most recent statement of financial position presented.
  - b. For securities classified as held-to-maturity, all reporting entities should disclose all of the following by major security type as of each date for which a statement of financial position is presented:
    - i. Amortized cost basis
    - ii. Aggregate fair value
    - iii. Gross unrecognized holding gains
    - iv. Gross unrecognized holding losses
    - v. Net carrying amount
    - vi. Total OTTI recognized in accumulated other comprehensive income
    - vii. Gross gains and losses in accumulated other comprehensive income for any derivatives that hedged the forecasted acquisition of the held-to-maturity securities
    - viii. Information about the contractual maturities of those securities as of the date of the most recent statement of financial position presented.
- **7.129** Maturity information may be combined in appropriate groupings, as stated in paragraphs 3 and 5 of FASB ASC 320-10-50. In complying with

this requirement, financial institutions (see FASB ASC 942-320-50-1) should disclose the fair value and the net carrying amount (if different from fair value) of debt securities on the basis of at least the following four maturity groupings:

- a. Within one year
- b. After one year through five years
- c. After five years through ten years
- d. After ten years

Securities not due at a single maturity date, such as MBSs, may be disclosed separately rather than allocated over several maturity groupings; if allocated, the basis for allocation should be disclosed.

- **7.130** In accordance with FASB ASC 942-320-50-2, in complying with the disclosure requirements mentioned in paragraph 7.128, financial institutions should include all of the following major security types, though additional types also may be included as appropriate:
  - a. Equity securities, segregated by any one of the following:
    - i. Industry type
    - ii. Entity size
    - iii. Investment objective
  - b. Debt securities issued by the U.S. Treasury and other U.S. government corporations and agencies
  - c. Debt securities issued by states within the United States and political subdivisions of the states
  - d. Debt securities issued by foreign governments
  - e. Corporate debt securities
  - f. Residential MBSs
  - g. Commercial MBSs
  - h. CDOs
  - i. Other debt obligations
- **7.131** Paragraphs 4–5 of FASB ASC 942-320-50 explain that the carrying amount of investment assets that serve as collateral to secure public funds, securities sold under repos, and other borrowings, that are not otherwise disclosed under FASB ASC 860, *Transfers and Servicing*, should be disclosed in the notes to the financial statements. The notes to the financial statements should include an explanation of the institution's accounting policy for securities, including the basis for classification. An entity should disclose the carrying amount of securities deposited by insurance subsidiaries with state regulatory authorities, as stated in FASB ASC 944-320-50-1.
- **7.132** For all investments in an unrealized loss position, including those within the scope of FASB ASC 325-40, for which OTTI has not been recognized in earnings (including investments for which a portion of an OTTI has been recognized in other comprehensive income), FASB ASC 320-10-50-6 states that an entity should disclose all of the following items in its interim and annual financial statements:
  - a. As of each date for which a statement of financial position is presented, quantitative information aggregated by category of investment—each major security type that the entity discloses in

accordance with FASB ASC 320-10 and cost-method investments—in tabular form:

- The aggregate related fair value of investments with unrealized losses
- ii. The aggregate amount of unrealized losses (that is, the amount by which amortized cost basis exceeds fair value)
- b. As of the date of the most recent statement of financial position, additional information (in narrative form) that provides sufficient information to allow financial statements users to understand the quantitative disclosures and the information that the entity considered (both positive and negative) in reaching the conclusion that the impairment or impairments are not other than temporary. (The application of step 2 in FASB ASC 320-10-35-30 should provide insight into the entity's rationale for concluding that unrealized losses are not OTTIs. The disclosures required may be aggregated by investment categories, but individually significant unrealized losses generally should not be aggregated.) This disclosure could include all the following:
  - i. The nature of the investment(s)
  - ii. The cause(s) of the impairment(s)
  - iii. The number of investment positions that are in an unrealized loss position
  - iv. The severity and duration of the impairment(s)
  - v. Other evidence considered by the investor in reaching its conclusion that the investment is not other-than-temporarily impaired, including, for example, any of the following:
    - Performance indicators of the underlying assets in the security, including default rates, delinquency rates, or percentage of nonperforming assets
    - 2. Loan-to-collateral-value ratios
    - 3. Third-party guarantees
    - 4. Current levels of subordination
    - 5. Vintage
    - 6. Geographic concentration
    - 7. Industry analyst reports
    - 8. Sector credit ratings
    - 9. Volatility of the security's fair value
    - 10. Any other information that the investor considers relevant

**7.133** This guidance also requires disclosures regarding the significant inputs used in determining a credit loss, as well as rollforward of that amount each period. For interim and annual periods in which an OTTI of a debt security is recognized and only the amount related to a credit loss was recognized in earnings, an entity should disclose by major security type, the methodology and significant inputs used to measure the amount related to credit loss. FASB ASC

320-10-50-8A provides examples of significant inputs and FASB ASC 320-10-50-8B addresses the tabular rollforward of the amount related to credit losses recognized in earnings and provides certain items which should be disclosed within the rollforward.

**7.134** Sales or transfers. For each period for which the results of operations are presented FASB ASC 320-10-50-9 requires that the institution disclose all of the following:

- a. The proceeds from sales of available-for-sale securities and the gross realized gains and gross realized losses that have been included in earnings as a result of those sales
- b. The basis on which cost of a security sold or the amount reclassified out of accumulated other comprehensive income into earnings was determined (that is, specific identification, average cost, or other method used)
- c. The gross gains and gross losses included in earnings from transfers of securities from the avail-able-for-sale category into the trading category
- d. The amount of the net unrealized holding gain or loss on available-for-sale securities for the period that has been included in accumulated other comprehensive income and the amount of gains and losses reclassified out of accumulated other comprehensive income into earnings for the period<sup>12</sup>
- e. The portion of trading gains and losses for the period that relates to trading securities still held at reporting date

**7.135** In accordance with FASB ASC 320-10-50-10, for any sales of or transfers from securities classified as held-to-maturity, an entity should disclose all of the following in the notes to the financial statements for each period for which the results of operations are presented: (a) the net carrying amount of the sold or transferred security, (b) the net gain or loss in accumulated other

For an entity that prospectively applies the guidance, the only required transition disclosure will be to disclose, if applicable, both the nature and the amount of an item included in income from continuing operations after adoption that adjusts an extraordinary item previously classified and presented before the date of adoption. An entity retrospectively applying the guidance should provide the disclosures in paragraphs 1-2 of FASB ASC 250-10-50.

Readers are encouraged to read the full text of the ASU, available at www.fasb.org. Readers should apply the appropriate guidance based on their facts and circumstances.

<sup>&</sup>lt;sup>12</sup> FASB ASC 740-20-45-2 requires that income tax expense or benefit for the year be allocated among continuing operations, discontinued operations, extraordinary items, other comprehensive income, and items charged or credited directly to shareholders' equity (such as changes in the unrealized holding gains and losses of securities classified as available for sale, as stated in FASB ASC 740-20-45-11 [see chapter 16, "Income Taxes," of this guide]).

<sup>&</sup>quot;Pending Content" in FASB ASC 740-20-45-2 requires that income tax expense or benefit for the year be allocated among continuing operations, discontinued operations, other comprehensive income, and items charged or credited directly to shareholders' equity (such as changes in the unrealized holding gains and losses of securities classified as available for sale [see chapter 16, "Income Taxes," of this guide]).

FASB ASU No. 2015-01, Income Statement—Extraordinary and Unusual Items (Subtopic 225-20): Simplifying Income Statement Presentation by Eliminating the Concept of Extraordinary Items, was issued in January 2015. The amendments are effective for fiscal years, and interim periods within those fiscal years, beginning after December 15, 2015. A reporting entity may apply the amendments in this ASU prospectively. A reporting entity also may apply the amendments in this ASU retrospectively to all prior periods presented in the financial statements. Early adoption is permitted provided that the guidance is applied from the beginning of the fiscal year of adoption. The effective date is the same for both public business entities and all other entities.

comprehensive income for any derivative that hedged the forecasted acquisition of the held-to-maturity security, (c) the related realized or unrealized gain or loss, and (d) the circumstances leading to the decision to sell or transfer the security. Such sales or transfers should be rare, except for sales and transfers due to the changes in circumstances identified in items (a)–(f) in FASB ASC 320-10-25-6. FASB ASC 320-10-25-14 sets forth the conditions under which sales of debt securities may be considered as maturities for purposes of the disclosure requirements under FASB ASC 320-10-50-10.

**7.136** Concentration-of-credit-risk. The concentrations-of-credit-risk disclosures apply to debt securities as well as loans. FASB ASC 825-10-50-20 states that, except as indicated in FASB ASC 825-10-50-22, an entity should disclose all significant concentrations of credit risk arising from all financial instruments, whether from an individual counterparty or groups of counterparties.

**7.137** *Mutual funds*. Investments in mutual funds that invest only in U.S. government debt securities may be shown separately rather than grouped with other equity securities in the disclosures by major security type required by FASB ASC 942-320-50-2, according to FASB ASC 320-10-50-4.

# Auditing 13

#### **Objectives**

**7.138** The primary objectives of audit procedures in this area are to obtain reasonable assurance that

- a. securities, accrued interest, and discounts and premiums of the institution
  - exist at the balance sheet date (definitive securities are on hand or held by others in custody or safekeeping for the account of the institution) and are owned by the institution.
  - have been properly classified, described, and disclosed in the financial statements at appropriate amounts (including consideration of any other-than-temporary declines in value and disclosure of any securities pledged as collateral for other transactions).
- b. sales of securities and other transactions that occurred
  - i. have been recorded during the appropriate period.
  - ii. are properly classified, described, and disclosed.
- realized and unrealized gains and losses, and interest (including premium amortization and discount accretion), dividend, and other revenue components

<sup>&</sup>lt;sup>13</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of issuers and nonissuers as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

- have been included in the financial statements at appropriate amounts.
- ii. are properly classified, described, and disclosed.

#### Planning

**7.139** In accordance with AU-C section 315, *Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement* (AICPA, *Professional Standards*), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (see chapter 5 of this guide for further information).

7.140 The primary inherent risks related to investments—interest-rate risk, credit risk, and liquidity risk—are interrelated. For example, increases in market interest rates may affect other risk factors by decreasing marketability (that is, liquidity) or by increasing the credit risk of the issuer's obligations. The auditor's understanding of the relationship between the interest-rate environment and the market values of securities might include review of the institution's asset/liability and other risk management policies, which may provide useful information about the possible effects of interest rate and liquidity risks on the institution's securities.

**7.141** Another risk inherent to complex investments is the business risk that the institution does not properly understand the terms and economic substance of a significant complex investment. Such misunderstandings could result in the incorrect pricing of a transaction and improper accounting for the investment or related income. (See chapter 18, "Derivative Instruments: Futures, Forwards, Options, Swaps, and Other Derivative Instruments," of this guide.) Inquiry of a specialist could be considered by the auditor if a financial institution engaged in holding or trading such complex securities.

**7.142** If the preparation of the financial statements involves the use of expertise in a field other than accounting, paragraph A7 of AU-C section 620, Using the Work of an Auditor's Specialist (AICPA, Professional Standards), states that the auditor, who is skilled in accounting and auditing, may not possess the necessary expertise to audit those financial statements. The engagement partner is required by AU-C section 220, Quality Control for an Engagement Conducted in Accordance With Generally Accepted Auditing Standards (AICPA, Professional Standards), to be satisfied that the engagement team and any external auditor's specialists (defined in the auditing standards as an individual or organization possessing expertise in a field other than accounting or auditing, whose work in that field is used by the auditor to assist the auditor in obtaining sufficient appropriate audit evidence) who are not part of the engagement team, collectively, have the appropriate competence and capabilities to perform the audit engagement. 14 Further, the auditor is required by AU-C section 300, Planning an Audit (AICPA, Professional Standards), to ascertain the nature, timing, and extent of resources necessary to perform the

©2016, AICPA

<sup>&</sup>lt;sup>14</sup> Paragraph .16 of AU-C section 220, Quality Control for an Engagement Conducted in Accordance With Generally Accepted Auditing Standards (AICPA, Professional Standards).

engagement.<sup>15</sup> The auditor's determination of whether to use the work of an auditor's specialist, and, if so, when and to what extent, assists the auditor in meeting these requirements. As the audit progresses or as circumstances change, the auditor may need to revise earlier decisions about using the work of an auditor's specialist.

Considerations for Audits Performed in Accordance With PCAOB Standards<sup>16</sup>

PCAOB Staff Audit Practice Alert No. 11, Considerations for Audits of Internal Control Over Financial Reporting (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance, sec. 400.11), highlights certain requirements of the auditing standards of the PCAOB in aspects of audits of internal control in which significant auditing deficiencies have been cited frequently in PCAOB inspection reports. Specifically, the alert discusses, among other topics, using the work of others.

**7.143** Classification of investments in securities among the held-to-maturity, available-for-sale, and trading categories is important because it directly affects the accounting treatment. The classification of securities, which must occur at acquisition, ordinarily should be consistent with the institution's investment, asset/liability, and other risk management policies. The independent auditor should ascertain whether the accounting policies adopted by the entity for investments are in conformity with GAAP. In planning the audit, the independent auditor may consider reading the current year's interim financial statements, investment policy, and other financial information related to securities. The level of inherent risk for securities varies widely from institution to institution depending on, among other things, the nature and complexity of the securities and the extent and effectiveness of the institution's accounting and operational policies and procedures, as well as management's understanding and awareness of the risks. The following factors related to securities may, considered in the aggregate, indicate higher inherent risk:

- a. Significant concentrations of credit risk with one counterparty or within one geographic area
- b. Significant use of complex securities, particularly without relevant in-house expertise
- c. Excessively high volumes of borrowing or lending of securities
- d. Relatively high volatility in interest rates
- e. Changes in the terms of government guarantees
- f. Actual prepayment experience that differs significantly from that anticipated
- g. Declines in the values of collateral underlying securities
- h. Changes in guarantors' claims processing
- $i. \hspace{0.2cm} \textbf{Significant conversion options related to the collateral (for example, variable to fixed rates)} \\$
- j. Sales and transfers from the held-to-maturity securities portfolio

<sup>&</sup>lt;sup>15</sup> Item e in paragraph .08 of AU-C section 300, Planning an Audit (AICPA, Professional Standards).

<sup>&</sup>lt;sup>16</sup> PCAOB Staff Audit Practice Alerts are not rules of the board, and do not reflect any PCAOB determination or judgment about the conduct of any particular firm, auditor, or any other person.

- k. Uncertainty regarding the financial stability of an ABS servicer or of guarantors
- Uncertainty regarding the financial stability of a safekeeping agent or other third party holding the institution's securities
- m. Changes in accounting systems, including software and manual processes
- n. Differing assumptions used in determining fair values will result in different conclusions
- o. Significant reliance on outside parties.

# Internal Control Over Financial Reporting and Possible Tests of Control

- 7.144 AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13–.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.
- **7.145** Effective controls, as they relate to financial reporting of investments in securities, should provide assurance that
  - a. management's policies are adequate to provide for financial reporting in accordance with GAAP;
  - b. physical securities are on hand or held in custody or safekeeping by others in accordance with management's authorization;
  - misstatements caused by error or fraud in the processing of accounting information for investments in securities are prevented or detected, and corrected in a timely manner;
  - d. securities are monitored on an ongoing basis to determine whether recorded financial statement amounts necessitate adjustment, including other than temporary impairment; and
  - e. the presentation and disclosure of the fair value measurements of investment securities are in accordance with GAAP.
- **7.146** Control activities that would contribute to internal control over financial reporting in this area include the maintenance of management policies, adopted by the those charged with governance or its investment committee, that establish authority and responsibility for investments in securities.
- **7.147** Other control activities that contribute to strong internal control over financial reporting of securities include the following:
  - Procedures exist to identify and monitor credit risk, prepayment risk, and impairment.

©2016. AICPA AAG-DEP 7.147

- Those charged with governance—generally through an investment committee—oversee management's securities activities.
- Accounting entries supporting securities transactions are periodically reviewed by supervisory personnel to ensure that classification of securities was made and documented at acquisition (and date of transfer, if applicable) and is in accordance with the institution's investment policy and management's intent.
- Recorded securities are periodically reviewed and compared to safekeeping ledgers and custodial confirmations, on a timely basis, including immediate and thorough investigation and resolution of differences and appropriate supervisory review and approval of completed reconciliations.
- Current fair values of securities are determined in accordance with GAAP and reviewed on a timely basis.
- Securities loaned to other entities or pledged as collateral are designated as such in the accounting records.
- Lists of authorized signers are reviewed and updated periodically, and transaction documentation is compared to the authorized lists.
- There is appropriate segregation of duties among those who (a) execute securities transactions, (b) approve securities transactions, (c) have access to securities, and (d) post or reconcile related accounting records.
- Buy and sell orders are routinely compared to brokers' advices.
- Adjustments to securities accounts (for example, to recognize impairments) are reviewed and approved by the officials designated in management's policy.
- Periodic tests of interest and dividend income are performed by reference to supporting documentation, which may include using analytical procedures commonly referred to as *yield analysis*. (With this approach, actual yields during the period are compared to expected yields based on previous results and current market trends. Any significant differences should be investigated and explained.)
- Management maintains frequent, open dialogue with any thirdparty investment advisors as to the company's strategy and how that impacts investment decisions.
- Securities are monitored on an ongoing basis and factors affecting income recognition and the carrying amount of the securities are analyzed periodically to determine whether adjustments are necessary.

**7.148** AU-C section 540, Auditing Accounting Estimates, Including Fair Value Accounting Estimates, and Related Disclosures (AICPA, Professional Standards), addresses the auditor's responsibilities relating to accounting estimates, including fair value accounting estimates and related disclosures, in an audit of financial statements. Specifically, it expands on how AU-C section 315; AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards); and other relevant AU-C sections are to be applied with regard to accounting

estimates. It also includes requirements and guidance related to misstatements of individual accounting estimates and indicators of possible management bias.

- **7.149** When performing risk assessment procedures and related activities to obtain an understanding of the entity and its environment, including the entity's internal control, as required by AU-C section 315, paragraph .08 of AU-C section 540 states that the auditor should obtain an understanding of the following in order to provide a basis for the identification and assessment of the risks of material misstatement for accounting estimates:<sup>17</sup>
  - a. The requirements of the applicable financial reporting framework relevant to accounting estimates, including related disclosures.
  - b. How management identifies those transactions, events, and conditions that may give rise to the need for accounting estimates to be recognized or disclosed in the financial statements. In obtaining this understanding, the auditor should make inquiries of management about changes in circumstances that may give rise to new, or the need to revise existing, accounting estimates.
  - c. How management makes the accounting estimates and the inputs on which they are based, including
    - i. the method(s), including, when applicable, the model, used in making the accounting estimate;
    - ii. relevant controls:
    - iii. whether management has used a specialist;
    - iv. the assumptions underlying the accounting estimates;
    - whether there has been or ought to have been a change from the prior period in the method(s) or assumption(s) for making the accounting estimates and, if so, why; and
    - vi. whether and, if so, how management has assessed the effect of estimation uncertainty.

See paragraphs .A11–.A37 of AU-C section 540 for additional application guidance and other explanatory material as it relates to accounting estimate risk assessment procedures and related activities.

- **7.150** Many of the control activities for securities are often performed directly by senior management. Although management's close attention to securities transactions can be an effective factor in internal control, in accordance with paragraph .15 of AU-C section 240, Consideration of Fraud in a Financial Statement Audit (AICPA, Professional Standards), the engagement team should address the risk of management override of controls during the discussion among the key engagement team members.
- **7.151** AU-C section 330 addresses the auditor's responsibility to design and implement responses to the risks of material misstatement identified and assessed by the auditor in accordance with AU-C section 315 and to evaluate the audit evidence obtained in an audit of financial statements.
- **7.152** In accordance with paragraph .08 of AU-C section 330, the auditor should design and perform tests of controls to obtain sufficient appropriate audit evidence about the operating effectiveness of relevant controls if (a) the

<sup>&</sup>lt;sup>17</sup> Paragraphs .05-.06 and .12-.13 of AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards).

auditor's assessment of risks of material misstatement at the relevant assertion level includes an expectation that the controls are operating effectively or (b) substantive procedures alone do not provide sufficient appropriate audit evidence at the relevant assertion level. Examples of tests of controls that might be considered include

- reading minutes of meetings of the board of directors (and any investment committee), or attending a meeting of the board of directors or investment committee, for evidence of the board's periodic review of securities activities made so that the board may determine adherence to the institution's policy;
- comparing securities transactions, including transfers, to the institution's accounting policy to determine whether the institution is following its policy. For example, the independent accountant may include
  - testing that transactions have been executed in accordance with authorizations specified in the investment policy;
  - evaluating evidence that securities portfolios and related transactions (including impairments) are being monitored on a timely basis and reviewing supporting documentation; and
  - testing recorded purchases of securities, including classification of the securities, prices, and entries used to record related amounts (for example, use of trade versus settlement date, treatment of commissions, and premiums and discounts).
- recalculating a sample of premium and discount amortization amounts and gains and losses on sales;
- reviewing controls over accumulating information necessary for financial statement disclosures;
- testing the reconciliation process. The independent accountant might test whether reconciling differences are investigated and resolved and whether the reconciliations are reviewed and approved by supervisory personnel; and
- examine evidence that the company takes physical inventory and confirms safekeeping on a periodic basis, including reconciliation of differences.

**7.153** Many financial institutions outsource the determination of fair value measurements of investment securities to third party service organizations, such as a pricing service. AU-C section 402, *Audit Considerations Relating to an Entity Using a Service Organization* (AICPA, *Professional Standards*), addresses the user auditor's responsibility for obtaining sufficient appropriate audit evidence in an audit of the financial statements of a user entity that uses one or more service organizations. Specifically, it expands on how the user auditor applies AU-C sections 315 and 330 in obtaining an understanding of the user entity, including internal control relevant to the audit, sufficient to identify and assess the risks of material misstatement and in designing and performing further audit procedures responsive to those risks. If the user

auditor plans to use a type 2 report as audit evidence that controls at the service organization are operating effectively, paragraph .17 of AU-C section 402 states that the user auditor should determine whether the service auditor's report provides sufficient appropriate audit evidence about the effectiveness of the controls to support the user auditor's risk assessment by

- evaluating whether the type 2 report is for a period that is appropriate for the user auditor's purposes;
- b. determining whether complementary user entity controls identified by the service organization are relevant in addressing the risks of material misstatement relating to the relevant assertions in the user entity's financial statements and, if so, obtaining an understanding of whether the user entity has designed and implemented such controls and, if so, testing their operating effectiveness;
- c. evaluating the adequacy of the time period covered by the tests of controls and the time elapsed since the performance of the tests of controls; and
- d. evaluating whether the tests of controls performed by the service auditor and the results thereof, as described in the service auditor's report, are relevant to the assertions in the user entity's financial statements and provide sufficient appropriate audit evidence to support the user auditor's risk assessment.
- **7.154** The auditor may also consider the following, which is not intended to be an all-inclusive list of considerations:
  - Whether the pricing service determines fair value measurements in accordance with the requirements of FASB ASC 820?
  - If trades of identical securities in an active market are available, are the pricing service's fair value estimates equal to quoted market prices?
  - For trades of identical securities, does the third party pricing service evaluate whether or not the market is active?
  - What is the criteria used to evaluate whether the market is active?
  - When a model is used to determine fair value, does the pricing service maximize the use of observable inputs and minimize the use of unobservable inputs?

#### Substantive Tests

**7.155** Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure, which for a financial institution would include investments in debt and equity securities. In accordance with paragraph .A45 of AU-C section 330, this requirement reflects the facts that (a) the auditor's assessment of risk is judgmental and may not identify all risks of material misstatement and (b) inherent limitations to internal control exist, including management override.

**7.156** AU-C section 501 addresses specific considerations by the auditor in obtaining sufficient appropriate audit evidence, in accordance with AU-C section 330; AU-C section 500, *Audit Evidence* (AICPA, *Professional Standards*);

©2016, AICPA AAG-DEP 7.156

and other relevant AU-C sections, regarding, among other considerations, certain aspects of investments in securities and derivative instruments in an audit of financial statements. In addition, the companion AICPA Audit Guide *Special Considerations in Auditing Financial Instruments* specifically addresses derivatives and securities measured or disclosed at fair value, various methods for determining fair value as specified by GAAP, and evaluating audit evidence for the valuation assertion of derivatives and securities. Readers may consider this guidance when designing and performing substantive tests.

- **7.157** Paragraph .06 of AU-C section 540 states that the objective of the auditor is to obtain sufficient appropriate audit evidence about whether, in the context of the applicable financial reporting framework,
  - accounting estimates, including fair value accounting estimates, in the financial statements, whether recognized or disclosed, are reasonable; and
  - b. related disclosures in the financial statements are adequate.
- **7.158** Based on the assessed risks of material misstatement, paragraph .12 of AU-C section 540 states that the auditor should determine
  - a. whether management has appropriately applied the requirements of the applicable financial reporting framework relevant to the accounting estimate;
  - b. whether the methods for making the accounting estimates are appropriate and have been applied consistently; and
  - c. whether changes from the prior period, if any, in accounting estimates or the method for making them are appropriate in the circumstances.
- **7.159** In responding to the assessed risks of material misstatement, as required by AU-C section 330, paragraph .13 of AU-C section 540 states that the auditor should undertake one or more of the following, taking into account the nature of the accounting estimate: 18
  - a. Determine whether events occurring up to the date of the auditor's report provide audit evidence regarding the accounting estimate.
  - b. Test how management made the accounting estimate and the data on which it is based. In doing so, the auditor should evaluate whether
    - i. the method of measurement used is appropriate in the circumstances.
    - ii. the assumptions used by management are reasonable in light of the measurement objectives of the applicable financial reporting framework, and
    - iii. the data on which the estimate is based is sufficiently reliable for the auditor's purposes.
  - c. Test the operating effectiveness of the controls over how management made the accounting estimate, together with appropriate substantive procedures.

<sup>&</sup>lt;sup>18</sup> Paragraphs .05–.06 of AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards).

- d. Develop a point estimate or range to evaluate management's point estimate. For this purpose
  - i. if the auditor uses assumptions or methods that differ from management's, the auditor should obtain an understanding of management's assumptions or methods sufficient to establish that the auditor's point estimate or range takes into account relevant variables and to evaluate any significant differences from management's point estimate.
  - ii. if the auditor concludes that it is appropriate to use a range, the auditor should narrow the range, based on audit evidence available, until all outcomes within the range are considered reasonable.
- **7.160** In determining the matters identified in paragraph .12 of AU-C section 540 (see paragraph 7.158) or in responding to the assessed risks of material misstatement in accordance with paragraph .13 of AU-C section 540 (see paragraph 7.159), paragraph .14 of AU-C section 540 states that the auditor should consider whether specialized skills or knowledge with regard to one or more aspects of the accounting estimates are required in order to obtain sufficient appropriate audit evidence. Depending on the auditor's understanding of, and experience working with, the auditor's specialist or those other individuals with specialized skills or knowledge, paragraph .A107 of AU-C section 540 states that the auditor may consider it appropriate to discuss matters such as the requirements of the applicable financial reporting framework with the individuals involved to establish that their work is relevant for audit purposes.
- **7.161** Auditing interests in trusts held by a third-party trustee and reported at fair value. In circumstances in which the auditor determines that the nature and extent of auditing procedures should include verifying the existence and testing the measurement of investments held by a trust, simply receiving a confirmation from the trustee, either in aggregate or on an investment-byinvestment basis, does not in and of itself constitute adequate audit evidence with respect to the requirements for auditing the fair value of interests in trusts under AU-C section 540. In addition, receiving confirmation from the trustee for investments in aggregate does not constitute adequate audit evidence with respect to the existence assertion. Receiving confirmation from the trustee on an investment-by-investment basis, however, typically would constitute adequate audit evidence with respect to the existence assertion. Also, in discussing obtaining an understanding of how management identifies the need for accounting estimates, paragraph .A15 of AU-C section 540 states that the preparation and fair presentation of the financial statements requires management to determine whether a transaction, an event, or a condition gives rise to the need to make an accounting estimate and that all necessary accounting estimates have been recognized, measured, and disclosed in the financial statements in accordance with the applicable financial reporting framework.
- **7.162** In circumstances in which the auditor is unable to audit the existence or measurement of interests in trusts at the financial statement date, the auditor should consider whether that scope limitation requires the auditor to either qualify his or her opinion or to disclaim an opinion, as discussed in AU-C section 705, *Modifications to the Opinion in the Independent Auditor's Report* (AICPA, *Professional Standards*).

©2016. AICPA AAG-DEP 7.162

#### 7.163

Considerations for Audits Performed in Accordance With PCAOB  $Standards^{19}$ 

PCAOB Staff Audit Practice Alert No. 2, Matters Related to Auditing Fair Value Measurements of Financial Instruments and the Use of Specialists (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance, sec. 400.02), provides guidance on auditors' responsibilities for auditing fair value measurements of financial instruments and when using the work of specialists under the existing standards of the PCAOB. This alert is focused on specific matters that are likely to increase audit risk related to the fair value of financial instruments in a rapidly changing economic environment. This practice alert highlights certain requirements in the auditing standards related to fair value measurements and disclosures in the financial statements and certain aspects of GAAP that are particularly relevant to the economic environment.

PCAOB Staff Audit Practice Alert No. 4, Auditor Considerations Regarding Fair Value Measurements, Disclosures, and Other-Than-Temporary Impairments (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance, sec. 400.04), informs auditors about potential implications of FASB guidance on reviews of interim financial information and annual audits. This alert addresses the following topics: (a) reviews of interim financial information; (b) audits of financial statements, including integrated audits; (c) disclosures; and (d) auditor reporting considerations.

<sup>&</sup>lt;sup>19</sup> See footnote 16.

Loans 205

## **Chapter 8**

## Loans

# FASB Accounting Standards Update (ASU) No. 2014-09, Revenue from Contracts with Customers (Topic 606)

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14, Revenue from Contracts with Customers [Topic 606]—Deferral of the Effective Date) for annual reporting periods of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period. Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as of either:

- An annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- An annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB Accounting Standards Codification (ASC) 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605, *Revenue Recognition*, as well as guidance within the 900 series of industry-specific topics, including FASB ASC 942, *Financial Services—Depository and Lending*. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts).

Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

The AICPA has formed 16 industry task forces to assist in developing a new Accounting Guide on revenue recognition that will provide helpful hints and illustrative examples for how to apply the new standard. Revenue recognition implementation issues identified by the Depository and Lending Institutions Revenue Recognition Task Force will be available for informal comment, after review by the AICPA Financial Reporting Executive Committee, at the "Depository and Lending Institutions Revenue Recognition Task Force" page at www.aicpa.org.

Readers are encouraged to submit comments to revreccomments@aicpa.org. For more information on FASB ASU No. 2014-09, see appendix E, "The New Revenue Recognition Standard: FASB ASU No. 2014-09," of this guide.

©2016, AICPA AAG-DEP 8

#### Introduction

**8.01** Loans usually are the most significant assets of financial institutions and generate the largest portion of revenues. Like investments, an institution's management of its loans is an integral part of its asset/liability management strategy (discussed in chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide). Institutions originate loans, purchase loans or participating interests in loans, sell loans or portions of loans, and securitize loans (the latter two activities are discussed in chapter 10, "Transfers and Servicing and Variable Interest Entities," of this guide). The composition of loan portfolios differs considerably among institutions because lending activities are influenced by many factors, including the type of institution, management's objectives and philosophies regarding diversification and risk (credit strategy), the availability of funds, credit demand, interest-rate margins, and regulations. Further, the composition of a particular institution's loan portfolio may vary substantially over time.

## The Lending Process

**8.02** This section discusses certain characteristics of and considerations involved in the lending process. The specific features will vary from institution to institution. To plan and design audit procedures properly, the auditor needs to understand the institution's loan portfolio, lending processes, loan accounting policies, market specialty, and trade area, as well as other factors such as economic conditions.

#### **Credit Strategy**

**8.03** The institution's credit strategy includes its defined goals and objectives for loans, as well as the loan policies written to help achieve those goals and objectives. A guiding principle in credit strategy is to achieve profitable returns while managing risk within the loan portfolio. Credit strategy and policy are usually determined by senior management and approved by the board of directors.

**8.04** The objectives of a sound credit plan are to identify profitable markets, determine appropriate risk tolerance levels for each type of loan, set goals for portfolio growth or contraction, and establish limits on industry and geographic concentrations. The plan establishes the institution's credit underwriting standards. These underwriting standards should consider recently issued regulatory requirements. Management's procedures and controls should enable the monitoring of loan performance through periodic reporting and review in order to identify and monitor problem loan situations.

#### **Credit Risk**

**8.05** The overriding factor in making a loan is the amount of credit risk associated with the loan in relation to the potential reward. For individual loans, credit risk pertains to the borrower's ability and willingness to make contractual loan payments and the fair value of any collateral pledged to secure the loan; it is assessed before credit is granted or renewed and periodically throughout the loan term.

**8.06** An institution's credit exposure may be affected by external factors, such as the level of interest rates, unemployment, general economic conditions, real estate values, and trends in particular industries and markets. Internal

factors—such as an institution's underwriting practices, credit practices, training, risk management techniques, familiarity and experience with its loan products and customers, the relative mix and geographic concentration of its loan portfolio and the strength of its internal control—also have a significant effect on an institution's ability to control and monitor its credit exposure.

**8.07** Additional risks, however, are involved in the overall credit process, and the institution generally should assess them when developing a credit strategy, defining target markets, and designing proper controls over credit initiation and supervision. Those additional risks include the following:

- Collateral risk. The institution may be exposed to loss on collateralized loans if its security interest is not perfected or the collateral is not otherwise under the institution's control, and also, in the case of fraud, the collateral does not exist. Further risks are if the value of the collateral declines, or if environmental contingencies impair the value of the collateral or otherwise create liability for the institution.
- Concentration risk. Inadequate diversification of the loan portfolio in terms of different industries, geographic regions, loan products, terms of loan products, or the number of borrowers may result in significant losses. A high concentration of loans to companies in a single industry would constitute a concentration risk. For example, membership of credit unions may be limited to employees of one organization or to individuals of a geographic region. If the credit union's sponsoring organization is experiencing financial problems or is anticipating layoffs of employees, the credit union could be exposed to significant losses. A high concentration of loans whose contractual features may increase the exposure of the originator to risk of nonpayment or realization would also constitute a concentration risk. For example, interest-only loans are designed to allow the borrower to only pay interest in the early part of the loan's term, which may delay defaults. For these loans, evidence of risk to loss may not become apparent until the contractual provisions of the loans cause a change in required payments.
- Sovereign country risk. The economic, social, legal, and political conditions of a foreign country may unfavorably affect a borrower's ability to repay in the currency of the loan. Cross-border loans are those that borrowers must repay in a currency other than their local currency or to a lender in a different country. Losses may result if a country's foreign exchange reserves are insufficient to permit the timely repayment of cross-border loans by borrowers domiciled in that country, even if the borrowers possess sufficient local currency. In addition, foreign government decisions and associated events can affect business activities in a country as well as a borrower's ability to repay its loans.
- Foreign exchange risk. Changes in foreign exchange rates may
  affect lenders unfavorably. Fluctuations in foreign exchange rates
  could reduce the translated value of the cash flows, earnings, and
  equity investments in foreign currency denominated subsidiaries.
  Foreign exchange rate movements, if not effectively hedged, could
  also increase the funding costs of foreign operations as it is not

- uncommon for foreign operations to be funded by borrowings in currencies different than their functional currency.
- *Fraud risk*. Loans may expose the institution to loss by not being bona fide transactions.
- Insider risk. Loans to executive officers, directors, and principal shareholders of the institution and related interests of such insiders may expose the institution to loss if these loans are made to related individuals or companies, or both, with little credit history; if they lack an identified source of funds for repayment; or if they are made to newly organized or highly leveraged enterprises with insufficient collateral and inadequate financial information.
- Interest rate risk. The maturity and repricing characteristics of loans can have a significant impact on the interest-rate risk profile (and, therefore, interest income) of an institution. For example, an institution that holds primarily fixed-rate loans and finances itself with floating rate obligations could be adversely affected by a significant increase in interest rates.
- Legal and regulatory risk. Illegally granted loans, loans with usurious interest rates, and loans with terms that are not adequately disclosed to the borrower may expose the institution to loss.
- *Management risk*. Management's competence, judgment, and integrity in originating, disbursing, supervising, collecting, and reviewing loans could substantially affect the collectability of loans.
- Operational risk. Funds might be disbursed without proper loan authorization, collateral documentation, or loan documentation.
   Failure of the institution to evaluate and monitor potentially uncollectible loans also constitutes an operations risk.

## **Lending Policies and Procedures**

- **8.08** Well-defined lending policies and comprehensive procedures for implementing such policies can contribute significantly to the institution's internal controls over financial reporting as they relate to the lending process.
- **8.09** The lending function can be broadly divided into the categories of (a) credit origination and disbursement, (b) credit supervision, (c) collection, and (d) loan review.
- **8.10** *Credit origination and disbursement.* Credit origination involves all the processes from the original request for credit to the disbursement of funds to the customer. Specific control features to meet operational—rather than financial reporting—objectives for credit origination usually include the following:
  - Credit initiation, that is, obtaining complete and informative loan applications, including financial statements and the intended use of proceeds
  - Credit investigation, including the following:
    - Credit reports or other independent investigations
    - Proper analysis of customer credit information, including the determination of projected sources of loan servicing and repayment

209

- Loan approval (new and renewed loans):
  - Loan approval limits according to officer expertise, administrative authority, or both
  - Committee approval or board of director approval, or both, for loans exceeding prescribed limits
  - The segregation of duties between the loan approval function and the disbursement and collection functions
  - Collateral ownership and control verified, including lien searches and documentation of the priority of security interest
  - Collateral margin determined
- Documentation of credit, or the inspection of supporting documents for proper form, completeness, and accuracy by someone other than the lending officer
- Perfection of collateral interest or proper security filings and recording of liens as well as possession of tangible collateral such as jewelry or bearer bonds
- The disbursement of loan proceeds or, to the extent possible, control of the disbursement to ensure that proceeds are used for the borrower's stated loan purpose
- **8.11** *Credit supervision.* Loan officers are responsible for closely monitoring the loans in their portfolios and bringing problem loans to the attention of management. Their duties normally include obtaining and analyzing the borrower's periodic financial statements and credit histories, reassessing collateral values, making periodic visits to the customer's place of operation, or in the case of collateral-based lending the place of the collateral, and generally keeping abreast of industry trends and developments and of the customer's financial requirements and ability to perform. Management reports concerning loan activity, renewals, and delinquencies are vital to the timely identification of problem loans. Input from loan officers is also important for identifying when loans should be restructured, reserved for, or charged off.
- **8.12** *Collection.* Loans identified as problem loans under the institution's established criteria should be monitored, restructured, or liquidated, as appropriate. The institution normally attempts to work with the customer to remedy a delinquency. Traditional mortgage collection procedures are not as effective in high loan-to-value (LTV) products. Delinquent borrowers who have little or no equity in the property may not have the incentive to work with the lender; therefore, high LTV lenders must intervene early to reduce the risk of default and loss. Sometimes the debt is restructured to include terms the customer can satisfy; at other times, the institution obtains additional collateral to support the loan. However, when the loan is delinquent for a specified period of time, as normally defined in the institution's lending policy, the institution may begin legal proceedings such as foreclosure or repossession to recover any outstanding interest and principal.
- **8.13** Loan review. Periodic review by institution personnel of the credit process and of individual loans is essential in assessing the quality of the loan portfolio and the lending process. Loan review should be conducted by personnel who are independent of the credit origination, disbursement, supervision,

and collection functions. Depending on the complexity of the organizational structure, these personnel report directly to the board of directors, a designated committee of the board of directors, or to senior management. Loan review may be performed by specifically assigned staff or may be incorporated within an internal audit function, including the use of third party providers.

- **8.14** A loan review includes several distinct activities. The principal emphasis is on determining whether the loans adhere to the institution's written lending policies and is likely to perform in accordance with the agreed-on terms and conditions, including compliance with any restrictive covenants in a loan agreement. The review normally includes analyzing the borrower's financial statements, reviewing performance since origination or last renewal, and determining if sufficient credit information is available to assess the borrower's current financial condition and determine the debt service coverage ratio.
- **8.15** Loan file contents should be reviewed as part of the institution's internal loan review process to determine if credit reports, appraisals, and other third-party information existed before the credit or renewal was granted and if the quality of such information supported, and continues to support, the credit decision. If the loan is secured or guaranteed, the review should also determine that collateral is under control, security interest is perfected, guarantees have been executed properly, and the guarantor's credit worthiness should be evaluated. Also, the value of collateral should be estimated at the review date to assess the loan to value ratio and to identify deficiencies in collateral margins.
- **8.16** Loan reviews may identify weaknesses in the underwriting process or in the lending officers' skill in originating, supervising, and collecting loans. Loan review results should be documented and may be summarized in the form of subjective ratings of individual loans that are similar to regulatory examination classifications. In addition, loan review may reveal that individual loans are impaired and need a loss accrual or may identify other factors relevant in assessing the allowance for loan and lease losses (ALLL), as discussed in chapter 9, "Credit Losses," of this guide.

## **Types of Lending**

- **8.17** Lending institutions offer a variety of loan products to meet borrowers' needs and as part of their overall credit strategy and asset/liability management strategy. Loans may be made on a line-of-credit, installment, demand, time, or term basis. A brief description of each of those kinds of arrangements follows:
  - a. Line-of-credit arrangements. The institution provides the borrower with a maximum borrowing limit for a specified period. Lines-of-credit may be structured in a variety of ways. Letters of credit (discussed in paragraph 8.50), which are commonly used as credit enhancements for other forms of borrowing (such as commercial paper, performance guarantees, or trade financing), are agreements to lend a specified amount for a specified period (usually less than one year). Revolving credit agreements, which are also used in credit card lending, are agreements to lend up to a specified maximum amount for a specified period, usually more than one year, and provide that repayment of amounts previously borrowed under the agreement are available to the borrower for subsequent borrowing. Repayment schedules may be on an installment, demand,

Loans 211

time, or term basis, as discussed subsequently. Other line-of-credit arrangements are applied to

- construction, whereby the borrower may draw on the line as necessary to finance building costs to supplement (or pending the securing of) a construction loan;
- ii. liquidity, used by the borrower in overall management of its liquidity needs; and
- iii. warehousing, used by borrowers engaged in mortgage banking activities to fund origination of mortgage loans, generally pending sale of the loans to a secondary market investor.
- b. Installment loans. These loan contracts require periodic principal and interest payments. Installment loans may be made on either a simple interest or a discounted basis. The discounted basis means that interest (discount), credit-life insurance premiums, and other charges are generally added to the amount advanced to arrive at the face amount of the note. The discount, called unearned interest, is netted against the face amount of the note on the balance sheet and accreted into income over time to achieve a level yield.
- c. Demand loans. These have no fixed maturity date, are payable on demand of the lender, and generally have interest rates that change periodically. Demand loans generally require periodic interest payments.
- d. Time loans. These are made for a specific period of time. Interest is payable periodically, and principal is due at maturity. Such loans are often renewed at maturity in what is known as a "rollover." Interest rates, if fixed during the loan period, reprice when the loan is rolled over.
- e. Term loans. These are made for a specified term, generally in excess of one year, at a rate of interest that either is fixed or floats based on an independent index, such as the London Interbank Offered Rate, or prime or treasury rates. Repayment schedules are structured in a variety of ways. Some term loans are amortized on a regular installment schedule; others contain provisions for a large portion of the loan to be paid at maturity (a balloon payment); and still others may call for installments of irregular size and timing based on cash-flow projections.
- **8.18** Loans may be categorized in a variety of ways, depending on the institution. Institutions group loans in ways that are meaningful in their particular circumstances; for most, the groupings are based on the kind of borrower, the purpose of the loan, or other common risk characteristics. Some common categories of loans include (a) commercial, industrial, and agricultural; (b) consumer; (c) residential real estate; (d) lease financing; (e) trade financing; (f) commercial real estate (CRE) and construction; and (g) foreign.

## Commercial, Industrial, and Agricultural Loans

**8.19** Despite changes in corporate borrowing practices (and increased competition from other kinds of financial institutions), commercial, industrial, and agricultural loans (sometimes called C and I or business loans) are an

important part of many institutions' business. There are a wide variety of commercial, industrial, and agricultural loans. They include

- factoring the purchase, usually without recourse, of trade accounts receivable;
- revolving loans or short term working capital loans, which are generally used by manufacturing companies to finance the purchase of raw materials and other production needs until the finished goods are sold;
- asset-based financing, usually secured by current assets such as accounts receivable or inventories, including receivable portfolio purchases;
- seasonal loans, which are used to provide cash to businesses (such as farms and retailers) during low-revenue periods of the year;
- floor-plan financing, which is used by automobile and durable goods dealers to finance inventories;
- long term working capital loans;
- loans and leases to finance the purchase of equipment; and
- loans to finance major projects, such as the construction of refineries, pipelines, and mining facilities.
- **8.20** Large commercial loans may involve more than one lender (see the discussion of loan participations that follows). Commercial loans may be secured (that is, the institution holds a lien against pledged assets, such as securities, inventories, property and equipment, an interest in the business, or accounts receivable) or unsecured. Also, such loans may be guaranteed or endorsed by third parties, including agencies of the U.S. government such as the Small Business Administration or the Export-Import Bank. Compensating-balance arrangements and commitment fees are often associated with commercial, industrial, and agricultural lending and are important factors in determining the interest rates on such loans. Commercial loans include demand loans, term loans, and line-of-credit arrangements.
- **8.21** Factoring. Factoring is the purchase, usually without recourse, of trade accounts receivable. A company that purchases trade accounts receivable is commonly called a factor. Factors buy trade accounts receivable from clients. Clients' customers send their payments directly to factors, often by means of a lockbox arrangement. Factored accounts receivable are not collateral for loans to clients; rather, the receivables are purchased outright. Except in certain instances involving advance factoring, as described in the following paragraphs, no loan is made. However, clients continue to remain contractually responsible for customer claims related to defective merchandise.
- **8.22** Factors buy clients' invoices, net of trade and cash discounts granted to customers, and provide clients with services that include assuming the clients' responsibilities of credit review, bookkeeping, and collection. Factors also assume risks of credit losses when customer credit is approved before a client ships the goods. Usually, if factors do not approve customers' credit, shipments are made at clients' risk. Factors buying accounts with recourse, however, provide bookkeeping and collection services and assume no credit risk, unless both the client and its customers become insolvent. Factors receive fees for services rendered to the client, usually computed as a percentage of net receivables bought.

- **8.23** Factoring usually mandates that customer notification be placed on the face of invoices, indicating that accounts have been sold and that factors are to be paid directly. Under nonnotification contracts, customers continue to pay clients and normally are unaware of factor ownership of the related accounts.
- **8.24** Two types of factoring arrangements are maturity and advance. Maturity factoring requires factors to pay clients only when related accounts are due (generally based on average due dates) or collected. In contrast, advance factoring allows clients to draw cash advances against the balance of the receivables before they are due or collected. Factors charge interest from the date on which advances are drawn to the date on which receivables are due or collected, at rates usually based on a stipulated percentage over commercial banks' prime rates.
- **8.25** In calculating limits for payments under advance factoring arrangements, factors generally retain a reserve against unpaid receivables to cover claims, returns, allowances, and other adjustments. Reserves ordinarily are a percentage of outstanding receivables based on factors' experience and judgment. Overadvances occur when clients draw cash advances that exceed uncollected receivable balances. Factors may permit overadvances to finance clients' seasonal business requirements. Such overadvances often can be anticipated. Overadvances also may result from unanticipated chargebacks, such as those resulting from defective merchandise and price disputes, because clients continue to remain contractually responsible for such problems. Overadvances may be collateralized by other assets, such as inventory or fixed assets, or may be secured by personal guarantees. In certain circumstances, overadvances also may be unsecured. Overadvances generally are reduced when receivables from additional sales are factored.
- **8.26** Revolving loans. Revolving loans, sometimes called working capital loans, generally provide borrowers with the cash needed for business operations. The loans typically are collateralized by accounts receivable and generally cannot exceed agreed percentages of the face values of those receivables. Such loans may also be referred to as accounts receivable loans. Collections against such receivables usually are remitted daily by borrowers to the lenders. Depending on the terms of the agreements, new accounts receivable acquired by borrowers and pledged to lenders may immediately qualify as collateral.
- **8.27** Lenders' policies may permit eligible collateral for revolving loans to be expanded to include inventories if borrowers require additional cash. In such cases, additional advances may be referred to as *inventory loans*. Inventory loans supplementing accounts receivable loans are common when seasonal businesses generate relatively low amounts of accounts receivable but demand large inventories in anticipation of the selling season. When the inventories are sold, the loans are paid off or accounts receivable generated by the sales replace inventories as collateral for such loans.
- **8.28** Receivables portfolio purchase agreements. Unlike factoring arrangements, receivables portfolio purchases are bulk purchases of trade accounts or finance receivables, often intended to provide sellers with cash for operations or improved financial ratios. Because the buyers usually assume all credit risks, a stipulated percentage of the purchase price is often retained to absorb credit losses. Credit losses in excess of that amount are borne by the buyer.
- **8.29** Terms of portfolio purchase agreements vary. Some provide for single purchases; others provide for continuing purchases on a revolving basis.

In addition, customers may not be notified of purchases or may be notified and required to pay the buyer directly. Receivables acquired under this type of agreement generally are accounted for as assets owned by the buyer and are not considered to represent collateral for loans made to sellers. Instances where participating interests (such as senior interests) are purchased by buyers should be carefully evaluated to determine whether the buyer should recognize the underlying assets for accounting purposes.

**8.30** Floor plan loans. Floor plan loans, commonly called wholesale loans, are made to businesses to finance inventory purchases. Some lenders make floor plan loans primarily to induce dealers to allow the lenders to buy the retail contracts generated from sales of inventories. Inventories serve as collateral for floor plan loans, the amounts of which usually are limited to the wholesale values of the inventories. Unlike revolving loans collateralized by inventory, floor plan loans generally are collateralized by specific inventory items. They also require minimum payments known as *curtailments*, with balances becoming due when collateral is sold or at the end of stipulated periods.

#### **Consumer Loans**

- **8.31** Consumer loans are loans to individuals for household, family, and other personal expenditures. Commonly, such loans are made to finance purchases of consumer goods, such as automobiles, boats, household goods, vacations, and education. Interest rates and terms vary considerably depending on many factors, including whether the loan is secured or unsecured. The two most significant kinds of consumer lending are installment loans and revolving credit arrangements (credit card lending).
- **8.32** *Installment loans.* Consumer installment loans, which are generally secured by the item purchased, may be originated directly with an institution's customers (direct paper) or acquired indirectly from a dealer's customers (indirect paper or retail sales contracts).
- **8.33** Retail sales contracts. Many sales of consumer goods and services are financed through retail sales contracts. Those contracts are made, directly or through retailers and dealers, with individual consumers. The contracts often are sold to a lender. Retail sales contracts commonly are called three-party paper because they involve three parties, namely, an individual borrower, a dealer or distributor, and a lender.
- **8.34** Retail sales contracts usually are sold at a discount to a lender under terms that permit dealers or distributors to share a portion of the finance charges paid by borrowers. Provisions for dealers' shares of finance charges vary among lenders and dealers. Dealers' shares of finance charges may be based on stipulated percentages of the finance charges or the principal amounts of the retail contracts, on a fixed amount for each contract, or on other negotiated terms. The Office of the Comptroller of the Currency (OCC) frequently issues guidance on retail sales contracts.
- **8.35** Some agreements provide for a portion of the amounts due to dealers to be withheld to cover certain contingencies. Other agreements provide no such conditions. Amounts withheld from dealers may either be limited to or greater than the dealers' shares of finance charges. Dealer reserves represent liabilities for unpaid portions of dealers' shares of finance charges on retail contracts bought from dealers. Dealer holdbacks, which are not limited to dealers' shares of finance charges, also represent liabilities, but usually are for

amounts withheld from dealers on retail contracts with greater-than-normal credit risk. Such risks may relate to factors such as the types of collateral, excessive loan periods, or the credit ratings of the borrowers involved. Dealer reserves and holdbacks may be required even if applicable contracts are bought with recourse.

- **8.36** *Credit cards.* Credit card lending is a major business for many institutions. Institutions may participate in the credit card market in various ways. Some institutions may issue or make credit cards available directly to customers. Institutions may also sponsor cards that are issued by another institution. The sponsoring institution may take credit applications, perform credit checks, and have its name printed on the cards, but the issuing institution records the consumer loans and assumes the credit risk. Most credit card lending is on an unsecured basis, although some secured programs exist. Within geographic areas, there are service companies that centralize card issuance, process transactions, and maintain customer accounts.
- **8.37** Credit card holders receive prenumbered cards under a prearranged line of credit with the institution issuing the card. Though the terms of credit cards vary, an annual fee is often charged for the use of the card and interest is charged on outstanding balances. Cards typically carry a grace period during which no interest is charged if outstanding balances are paid in full. Furthermore, merchants are generally charged a transaction fee.
- **8.38** Many institutions that issue credit cards have agreements with one of the two major international bank card systems, Visa and Interbank (MasterCard). However, a number of financial institutions have independent plans. The main functions that the bank card systems perform are enrolling merchant members and providing authorization and clearing systems. The main functions that the issuing institutions perform are issuing cards, setting credit limits, billing, collections, and customer service.
- **8.39** Overdraft protection. Another type of revolving credit is overdraft protection on checking accounts. Overdraft protection is an agreement between an institution and its customer to provide a prearranged line of credit that is automatically drawn if the customer writes checks greater than the amount in his or her deposit account. Interest is charged on amounts outstanding, and fees for usage may also be charged.

#### **Residential Real Estate Loans**

- **8.40** Loans secured by one-to-four-family residential property of the borrower are generally referred to as *residential mortgage loans*. Repayment terms for residential mortgage loans may vary considerably. Such loans may be structured to provide for the full amortization of principal, partial amortization with a balloon payment at a specified date, or negative amortization. Interest rates may be fixed, variable, or a combination of both. Variable-rate loans generally are referred to as adjustable-rate mortgages (ARMs). In addition, institutions may require borrowers in certain circumstances to purchase private mortgage insurance to reduce the institution's credit risk.
- **8.41** Many different types of nonpurchase related first and second lien residential mortgage loans have become popular, including
  - reverse mortgages, which provide homeowners with monthly payments in return for increasing the principal amount of a loan

- (decreasing the equity the homeowner has), wherein the institution may eventually gain ownership of real estate;
- second-lien fixed-term (or closed end) loans through which homeowners borrow a portion of their equity (property value in excess of the first-lien balance) and repay such over a fixed period of time with fixed or variable interest rates;
- home equity lines of credit allow the homeowner to borrow on demand, a portion of their equity, repay such and reborrow, if desired;
- **8.42** The Federal Housing Administration (FHA) insures and the U.S. Department of Veterans' Affairs (VA) partially guarantees many residential real estate mortgages. The FHA sets minimum down payments and interest rates for FHA loans. FHA-insured borrowers pay an annual insurance premium computed each year on the loan balance at the beginning of the year. The VA guarantee program, which was initiated to enable veterans to obtain homes when they return from military service, provides certain features, including an interest-rate ceiling that is generally lower than prevailing market rates, a partial guarantee to the lender, a low (or no) down payment, and a prohibition against mortgage brokers' commissions. Residential mortgage loans that are not FHA-insured or VA-guaranteed are called *conventional loans*.
- **8.43** Chapters 4, "Industry Overview—Mortgage Companies," and 10 of this guide include further discussion on mortgage banking activities.

#### **Lease Financing**

#### © Update 8-1 Accounting and Reporting: Leases

FASB ASU No. 2016-02, *Leases (Topic 842)*, issued in February 2016, is effective for fiscal years of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files financial statements with the SEC beginning after December 15, 2018, including interim periods within those fiscal years.

For all other entities, FASB ASU No. 2016-02 is effective for fiscal years beginning after December 15, 2019, and interim periods within fiscal years beginning after December 15, 2019.

Early application is permitted for all entities.

<sup>&</sup>lt;sup>1</sup> The respective instructions for Schedule HC-R of the Board of Governors of the Federal Reserve System's (Federal Reserve's) Consolidated Financial Statements for Holding Companies—FR Y-9C or for Schedule RC-R of the Federal Financial Institutions Examination Council's Consolidated Reports of Condition and Income should be read in conjunction with the regulatory capital rules (for example, Title 12 U.S. Code of Federal Regulations [CFR] Part 217) issued by the reporting bank holding company and insured depository institution or primary federal supervisory authority for determination of the applicable risk weight. U.S. banking organizations are subject to the standardized or general capital rules and certain of these institutions also measure regulatory capital using qualified advanced approaches. For example, Item 4.a, Residential mortgage exposures, of Schedules HC-R Part II and RC-R Part II indicate that the 20 percent risk weight applies to residential loans secured by real estate including the carrying value of the guaranteed portion of held for sale (HFS) Federal Housing Administration and the U.S. Department of Veterans' Affairs mortgage loans included in closed end loans secured by first liens on 1-4 family residential properties. The 50 percent risk weight would apply for the unguaranteed portion of 1-4 family residential properties and multi-family (5 or more) residential properties if prudently underwritten. The 100 percent risk weight applies to the remaining loans.

FASB ASU No. 2016-02 supersedes the lease requirements in FASB ASC 840, *Leases*, and creates FASB ASC 842, *Leases*, to establish the principles that lessees and lessors should apply to report useful information to users of financial statements about the amount, timing, and uncertainty of cash flows arising from a lease. FASB ASC 842 affects any entity that enters into a lease (as that term is defined in FASB ASU No. 2016-02), with some specified scope exceptions.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-02, see appendix F, "The New Leases Standard: FASB ASU No. 2016-02," of this guide.

- **8.44** Institutions also may be involved in direct lease financing, in which an institution owns and leases personal property for the use of its customers at the customers' specific request. A typical lease agreement contains an option providing for the purchase of the leased property, at its fair value or at a specified price, by the lessee at the expiration of the lease. Such leases may be financing transactions (discussed in paragraph 8.140). Despite similarities between leases and other forms of installment loans, continuing legal and tax changes have resulted in language and procedures unique to leasing activities.
- **8.45** *Operating leases.* An operating lease is a rental agreement in which asset ownership resides with the lessor. At the end of the lease term, the lessee may renew the lease, purchase the equipment, or return it to the lessor. During the course of the lease, the lessee expenses the rental payment made. As these types of agreements are in substance usage agreements, the debt is allowed to remain off the lessee's balance sheet. The lessor records the equipment as an asset and is required to depreciate it. Operating leases generally run for periods considerably shorter than the useful lives of related assets. At the expiration of such leases, the assets generally are sold or leased again.
- **8.46** *Direct financing leasing.* Direct financing leases are similar to other forms of installment lending in that lessors generally do not retain benefits and risks incidental to ownership of the property subject to leases. Such arrangements are essentially financing transactions that permit lessees to acquire and use property.
- **8.47** Leveraged leasing. Leveraged leasing involves at least three parties, namely, a lessee, a long term creditor, and a lessor (commonly called the equity participant). The lessor may, however, be represented by an owner trustee. Finance companies and other lenders frequently enter into leveraged lease transactions as lessors or equity participants. A substantial portion of the purchase price of assets is supplied nonrecourse by unaffiliated long term lenders. If a lessee defaults on lease payments, the long term lender has no recourse to the lessor, but usually has recourse to the specific property being leased. The gross return to a finance company or lender is measured using the discounted net cash receipts generated from investment tax credits and the tax effects of timing differences resulting principally from the use of accelerated depreciation in tax returns, rental payments minus debt service costs, and the estimated residual values of equipment leased.

**8.48** Transactions and vendor leasing. Leasing arrangements also may be categorized as transactional, involving direct negotiations between a lessor and lessee, and as vendor leasing. Transactional lease financing tends to be a time-consuming and expensive process that is economically feasible only for transactions sufficiently large to generate profits in excess of the costs of preparing custom-made leases. Vendor leasing has developed to finance asset acquisitions that would not be profitable to finance with transactional leasing arrangements. Vendor leasing involves a third-party lessor that offers a vendor's or manufacturer's customers a basic finance package. The lessor usually establishes interest rates within given dollar ranges and uses a standardized credit scoring process to approve credit and keep documentation simple. As a result, vendors are promptly paid for sales and avoid the need to perform inhouse financing operations. Some lenders also may serve as *lease brokers*—that is, as intermediaries between lessors and lessees for a fee.

#### **Trade Financing**

**8.49** Trade financing is a specialized area of commercial lending frequently used by businesses that engage in international activities. Such financing includes open account financing, sales on consignment, documentary collections, advances against collections, letters of credit, bankers' acceptances, factoring, and forfeiting. Lending institutions charge fees for such arrangements. The most commonly used of these arrangements is the letter of credit.

**8.50** The two primary types of letters of credit are the commercial letter of credit and the standby letter of credit. A *commercial letter of credit* represents a commitment by the issuing institution to make payment for a specified buyer to a specified seller in accordance with terms stated in the letter of credit. Under a *standby letter of credit*, the issuing institution guarantees that the buyer will make payment. The issuing institution is not ordinarily expected to make payment; however, if it does make payment, the buyer is obligated under the agreement to repay the institution. Standby letters of credit are also used to guarantee the performance of U.S. companies under contracts with foreign corporations and foreign or domestic governments. Depending on the nature of the agreement, these transactions may involve a high degree of credit risk. Portions of loans for cross-border transactions are often guaranteed by the Export-Import Bank.

#### **CRE and Construction Loans**

**8.51** Loans made on real property such as office buildings, apartment buildings, shopping centers, industrial property, and hotels are generally referred to as *CRE loans*. Such loans are usually secured by mortgages or other liens on the related real property such as an assignment of rents or a lien on the property, plant, and equipment within the real property. Repayment terms on CRE loans vary considerably. Interest rates may be fixed or variable, and the loans may be structured for full, partial, or no amortization of principal (that is, periodic interest payments are required and the principal is to be paid in full at the loan maturity date). Some give the institution recourse to third parties, who guarantee repayment of all or a portion of the loans. Others are nonrecourse, that is, if the borrower cannot repay the loan, the lender has only the collateral as a source of repayment—the lender does not have recourse to any other source of repayment.

Loans 219

**8.52** Construction lending involves advances of money from a bank or savings institution to finance the construction of buildings or the development of raw land. The institution generally agrees to a specified loan amount, part of which will be disbursed to the borrower at the inception of the project and part of which will be disbursed as construction progresses, based on specified milestones that were agreed to by the institution and the borrower. Construction loans are usually made for the construction period only, which generally runs from one to seven years. Often, both interest and principal are payable at maturity. After construction is completed, the borrower commonly obtains long-term mortgage financing from the same or another financial institution. Large CRE and construction loans may involve more than one lender (discussed in paragraphs 8.55–.56).

**8.53** Certain real estate loan arrangements, in which the lender has virtually the same risks and potential rewards as those of the owners of the property, should be considered and accounted for as investments in real estate. Certain real estate acquisition, development, and construction (ADC) arrangements that should be accounted for as investments in real estate are discussed in chapter 11, "Real Estate Investments, Real Estate Owned, and Other Foreclosed Assets," of this guide.

#### **Foreign Loans**

**8.54** Foreign (or cross-border) loans are made primarily by larger institutions and consist of loans to foreign governments, loans to foreign banks and other financial institutions, and commercial and industrial loans. Foreign loans also include consumer and commercial lending, including real estate loans, made by foreign branches. Such loans may contain certain risks, not associated with domestic lending, such as foreign exchange and country or transfer risks, as described previously in paragraph 8.07. This type of lending exposes the institution to cross-border risk, which is the possibility that the borrowing country's exchange reserves are insufficient to support its repayment obligations.

## **Loans Involving More Than One Lender**

**8.55** Institutions sometimes receive requests for loans that exceed the institution's capacity or willingness to lend. In response, shared lending arrangements have been created. In a syndication lending arrangement, groups of institutions agree to provide a portion of a particular loan, with each institution being a direct creditor of the borrower but with uniform lending terms applied by all the institutions. One institution is typically appointed as the agent, or lead institution, having primary responsibility for communication and negotiation with the borrower. The lead institution may also service all loans in the group. In some cases, an institution that is part of syndication may subsequently transfer via an assignment all or a portion of their share of the loan to another institution. If the borrower approves the assignment, the new institution becomes an official party to the lending agreement (for example, direct creditor to the borrower). In a participation lending arrangement, a lead institution originates a loan for the entire amount and sells to other lenders (participating institutions) portions of the loan it originated. The lead institution disburses all funds, supervises the perfection of legal interests in the underlying collateral, and usually services the loan. Loan participations may be negotiated on either a recourse or nonrecourse basis. Also, a participation may be sold on terms that differ from the original loan terms. The participating institutions have rights to their respective share of the cash flows from the original loan but do not become an official party to the loan agreement (for example, direct creditor). The borrower may not even be aware of the participation agreement.

**8.56** In a loan syndication, the participating institutions arrange a lending syndicate in which the lead syndicator and participants in the syndication fund their respective portions of the loan. A syndication typically involves less risk to a lead institution than a participation because the lead institution funds only its portion—rather than the entire amount—of the loan at origination. A major difference between syndications and participations relates to the accounting by the agent or lead institution. Refer to paragraphs 19–20 of FASB ASC 310-20-25 for additional guidance. Depending on the nature of the loan participation agreement (particularly those that do not involve pari passu loan participations), the lead institution may be unable to derecognize the participations transferred to other banks under FASB ASC 860, *Transfers and Servicing*. See further discussion on transfers of financial assets within chapter 10 of this guide.

## **Regulatory Matters**

## Real Estate Lending Standards

**8.57** The Board of Governors of the Federal Reserve System (Federal Reserve), the OCC, and the FDIC (collectively, the federal banking agencies) have established real estate lending standards and related guidelines that describe the factors management should address in its real estate lending policies.<sup>2</sup> According to regulations, each institution is required to adopt and maintain written policies that establish limits and standards for extensions of credit related to real estate. The lending policies must establish

- a. portfolio diversification standards;
- b. underwriting standards, including LTV ratio limitations;
- c. loan administration policies; and
- d. documentation, approval, and reporting requirements to monitor compliance and appropriateness.
- **8.58** Management's policies are to be consistent with safe and sound banking practices, appropriate to the size of the institution and the nature and scope of its operation, and reviewed and approved by the institution's board of directors at least annually.
- **8.59** The policies also outline considerations for loan portfolio management, underwriting standards, loan administration, supervisory LTV limits and excluded transactions, policy exceptions, and supervisory review of real estate lending policies and practices.
- **8.60** On October 8, 1999, the federal banking agencies, including the Office of Thrift Supervision (OTS), prior to its transfer of powers to the federal

 $<sup>^2</sup>$  See 12 CFR Part 34 (Office of the Comptroller of the Currency [OCC]); Part 208 (Federal Reserve); and Part 365 (FDIC).

Loans 221

banking agencies,<sup>3</sup> jointly issued the *Interagency Guidance on High Loan-to-Value Residential Real Estate Lending*, which highlight the risks inherent in this activity and provides for supervisory limits and capital considerations. The guidelines set forth the supervisory expectation that high LTV portfolios, as defined, will not exceed 100 percent of total capital. Institutions that approach this limit will be subject to increased supervisory scrutiny. If the limit is exceeded, then its regulatory agency will determine if the activity represents a supervisory concern and take action accordingly. Policy and procedure guidelines provided in the 1992 *Interagency Guidelines for Real Estate Lending Policies* apply to these transactions.

**8.61** The federal banking agencies issued interagency guidance *Concentrations in Commercial Real Estate Lending, Sound Risk Management Practices*, which was effective on December 6, 2006. The guidance was intended to help ensure that institutions pursuing a significant CRE lending strategy remained healthy and profitable while continuing to serve the credit needs of their communities. The guidance encourages ongoing risk assessments and analysis of CRE lending policies. This includes evaluating the appropriateness of an institution's risk practices, as well as capital levels in relation to the size and complexity of its CRE portfolio. The guidance is applicable for state member banks and bank holding companies, as well as their nonbank subsidiaries.

**8.62** The federal banking agencies, along with the National Credit Union Administration (NCUA), adopted a Policy Statement on Prudent Commercial Real Estate Loan Workouts on October 30, 2009. This policy statement provides guidance for examiners, and for financial institutions working with CRE borrowers who are experiencing diminished operating cash flows, depreciated collateral values, or prolonged delays in selling or renting commercial properties. This guidance addresses supervisory expectations for an institution's risk management elements for loan workout programs, loan workout arrangements, classification of loans, 4 and regulatory reporting and accounting considerations. The statement also includes references and materials related to regulatory reporting, but it does not change existing regulatory reporting guidance provided in relevant interagency statements issued by the banking regulators or accounting requirements under U.S. generally accepted accounting principles (GAAP). The guidance includes a series of examples of CRE loan workouts, which are provided for illustrative purposes. This guidance replaces the *Inter*agency Policy Statement on the Review and Classification of Commercial Real Estate Loans (November 1991 and June 1993). Readers are encouraged to view the press release under the "Press Releases" page at www.ffiec.gov for additional information.

**8.63** Appraisals. On December 10, 2010, the federal banking agencies, along with the NCUA, issued *Interagency Appraisal and Evaluation Guidelines*, which replaced the guidelines issued in 1994. These guidelines describe the elements of a sound program for conducting appraisals and evaluations in compliance with the agencies' appraisal regulations. The guidelines provide additional clarification for when a real estate appraisal and evaluation is required to support a real estate-related financial transaction. Further, they

©2016, AICPA AAG-DEP 8.63

<sup>&</sup>lt;sup>3</sup> See chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide for further discussion on the Office of Thrift Supervision transfer of powers.

 $<sup>^4</sup>$  See paragraphs 9.08–.09 of this guide for additional information regarding the classification of loans.

explain the minimum regulatory appraisal standards and the supervisory expectations for the development and content of an evaluation, which is permitted in certain situations in lieu of an appraisal. The guidelines build on the existing federal regulatory framework and reaffirm long-standing supervisory expectations. They also incorporate the agencies' recent supervisory issuances and, in response to advances in information technology, clarify standards for the industry's appropriate use of analytical methods and technological tools in developing evaluations. The Dodd–Frank Wall Street Financial Reform and Consumer Protection Act of 2010 underscores the importance of sound real estate lending decisions; revisions to the guidelines may be necessary after regulations are adopted to implement the act. Financial institutions should review their appraisal and evaluation programs to ensure that the programs are consistent with the guidelines. Readers are encouraged to access the guidance from any of the agencies' websites.

- **8.64** In January 2013, the federal banking agencies, along with the Consumer Financial Protection Bureau (CFPB), the Federal Housing Finance Agency, and the NCUA, issued a final rule *Appraisals for Higher-Priced Mortgage Loans* to establish new appraisal requirements for higher-priced mortgage loans. Higher-priced mortgages are those mortgages with an annual percentage rate that exceeds the average prime offer rate by a specified percentage.<sup>5</sup>
- **8.65** For higher-priced mortgage loans, the rule requires creditors to use a licensed or certified appraiser who prepares a written appraisal report based on a physical inspection of the interior of the property. The rule also requires creditors to disclose to applicants information about the purpose of the appraisal and provide consumers with a free copy of any appraisal report. If the seller acquired the property for a lower price during the prior six months and the price difference exceeds certain thresholds, creditors will have to obtain a second appraisal at no cost to the consumer. This requirement for higher-priced home-purchase mortgage loans is intended to address fraudulent property flipping by seeking to ensure that the value of the property legitimately increased.
- **8.66** The rule exempts several types of loans, such as qualified mortgages; temporary bridge loans and construction loans; loans for new manufactured homes; and loans for mobile homes, trailers, and boats that are dwellings. The rule also has exemptions from the second appraisal requirement to facilitate loans in rural areas and other transactions. Readers can access the full text of this final rule from any of the respective agencies' websites.
- **8.67** The federal banking agencies and the NCUA require an appraisal by a state certified or licensed appraiser for all real estate-related financial transactions (as defined in the U.S. *Code of Federal Regulations*) having a value greater than \$250,000. The appraisal regulations do exempt certain real estate-related financial transactions from the appraisal requirement. The federal banking agencies and the NCUA also reserve the right to require an

<sup>&</sup>lt;sup>5</sup> In December 2013, the Federal Reserve, the Consumer Financial Protection Bureau (CFPB), the FDIC, the Federal Housing Finance Agency, the National Credit Union Administration (NCUA), and the OCC issued a supplemental rule to the appraisals for higher-priced mortgage loans. The supplemental rule provides an exemption to loans of \$25,000 or less and certain streamlined refinancings from the Dodd-Frank Wall Street Reform and Consumer Protection Act appraisal requirements. On January 1, 2015, the exemption threshold was adjusted to \$25,500 through a final ruling issued by the OCC, the Federal Reserve, and the CFPB.

 $<sup>^6</sup>$  See 12 CFR Part 34.43 (OCC); Part 225.63 (Federal Reserve); Parts 323.3 and 390.442 (FDIC); and Part 722.3 (NCUA).

appraisal under the appraisal regulations to address safety and soundness concerns in a transaction. As a matter of policy, the OCC and the FDIC use their supervisory authority to require problem federal and state savings associations and federal and state savings associations in troubled condition to obtain appraisals for all real estate-related transactions over \$100,000 (unless the transaction is otherwise exempt). The NCUA requires a written estimate of market value for all real estate-related transactions valued at the appraisal threshold or less, or that involve an existing extension of credit where there is either an advancement of new monies or a material change in the condition of the property.

## Retail Credit Loans and Residential Mortgage Loans

**8.68** On June 12, 2000, the federal banking agencies issued the *Revised Uniform Retail Credit Classification and Account Management Policy* (initially issued in 1999), which instructs institutions on the review and classification of retail credit loans and residential mortgage loans. Institutions should adopt the standards contained in the policy as part of their loan review program. The guidelines include requirements for the classification and charge-off of retail credit and mortgage loans, as well as fraudulent loans and bankruptcy cases. On April 30, 2009, the FDIC issued Financial Institution Letter-19-2009, *Classification Treatment for High Loan-to-Value (LTV) Residential Refinance Loans*, in which the FDIC affirmed that the standards in the *Uniform Retail Credit Classification and Account Management Policy* should be followed relative to the classification treatment for high LTV residential refinance loans. The guidance establishes that retail loan classifications should be based on the borrower's payment performance, not the value of the collateral, which can rise and fall as market conditions change.

**8.69** On March 26, 2001, the federal banking agencies, along with the NCUA, issued *Interagency Guidance on Certain Loans Held for Sale*, to provide instruction to institutions and examiners about the appropriate accounting and reporting treatment for certain loans that are sold directly from the loan portfolio or transferred to a held for sale (HFS) account. That guidance also addresses subsequent declines in value for loans within its scope and states the following:

After a loan or group of loans is transferred to the HFS account, those assets must be revalued at each subsequent reporting date until sold and reported at the lower of cost or fair value. Any declines in value (including those attributable to changes in credit quality) and recoveries of such declines in value occurring after the transfer to the HFS account should be accounted for as increases and decreases in a valuation allowance for HFS loans, not as adjustments to the allowance for loan and lease losses (ALLL). Changes in this valuation allowance should be reported in current earnings. The valuation allowance for HFS loans cannot be reduced below zero (that is cannot have a debit balance).

**8.70** In September 2009, the NCUA issued Letter to Credit Union 09-CU-19, *Evaluating Residential Real Estate Mortgage Loan Modification Program*, which provides certain financial reporting considerations for credit unions, as

©2016, AICPA AAG-DEP 8.70

<sup>&</sup>lt;sup>7</sup> See Federal Reserve Supervision and Regulation letter 00-8, Revised Uniform Retail Credit Classification and Account Management Policy, at www.federalreserve.gov.

well as several other considerations related to loan modifications. Readers are encouraged to view the full text of this letter under the "Letters to Credit Unions—2009" page at www.ncua.gov.

#### **Troubled Debt Restructurings**

- **8.71** In April 2012, the OCC issued Bulletin OCC 2012-10, Troubled Debt Restructurings: Supervisory Guidance on Accounting and Reporting Requirements, to national banks and federal savings associations to address many inquiries received from bankers and examiners on accounting and reporting requirements for troubled debt restructurings (TDRs), especially related to loan renewals and extensions of substandard commercial loans. The bulletin focuses on factors and key concepts to consider when evaluating loans for TDR designation and considerations for the appropriateness of accrual status and impairment analyses.
- **8.72** The bulletin specifically highlights that all substandard loans on accrual status that are renewed, extended, or otherwise modified should not automatically be considered TDRs. Rather, the institution needs to consider the totality of the transaction given the borrower's financial condition. As such, it is important for banks to establish an appropriate process for identification and analysis of TDRs and to document such an analysis. For example, the procedures should address the process for flagging a modified or renewed loan for review, considering factors to assess TDR status, designating responsibility for the TDR decision, and clearly documenting the facts and circumstances analyzed for each modification or renewal and the conclusion reached. Further guidance regarding TDRs that may assist when considering whether a loan modification or renewal is a TDR can be found in the interagency's *Policy Statement on Prudent Commercial Real Estate Loan Workouts* and the *Bank Accounting Advisory Series*.
- **8.73** The bulletin further reminds banks that renewals, extensions, or modifications deemed to be TDRs must be evaluated for the appropriate impairment measurement under FASB ASC 310-10 to ensure that the ALLL and accrual status are appropriate and consistent with the Federal Financial Institutions Examination Council's *Instructions for Preparation of Consolidated Reports of Condition and Income*. The bulletin also addresses separate considerations that should be given when disclosing of a loan as a TDR and its evaluation under FASB ASC 310-10. Readers can access Bulletin OCC 2012-10 from the OCC website at www.occ.gov.
- 8.74 In October 2013, the federal banking agencies and the NCUA issued Interagency Supervisory Guidance Addressing Certain Issues Related to Troubled Debt Restructurings. The supervisory guidance for financial institutions<sup>8</sup> addresses certain issues related to the accounting treatment and regulatory credit risk grade or classification of commercial and residential real estate loans that have undergone TDRs. The document reiterates key aspects of previously issued regulatory guidance (such as the interagency Policy Statement on Prudent Commercial Real Estate Loan Workouts) and discusses the definition

<sup>&</sup>lt;sup>8</sup> For purposes of this guidance, the term financial institution includes national banks, federal savings associations, and federal branches and agencies supervised by the OCC; state member banks, bank holding companies, savings and loan holding companies, and all other institutions for which the Federal Reserve is the primary federal supervisor; state nonmember banks, state savings associations, and insured state branches of foreign banks for which the FDIC is the primary federal supervisor; and federal credit unions and all other institutions for which the NCUA is the federal insurer.

of *collateral-dependent loans* and the circumstances under which a charge-off is required for TDRs. Readers can access the full text of this guidance from any of the respective agencies' websites.

**8.75** The Federal Financial Institutions Examination Council issued *Supplemental Instructions for September 2014 Call Reports* that address the accounting for subsequent restructurings of TDRs. Pursuant to this guidance, certain TDRs that are subsequently modified may no longer be considered to be TDRs if on the subsequent modification the borrower is no longer experiencing financial difficulties and no concession is granted to the borrower as a part of this modification. The subsequent modification must carry a market rate of interest for similar debt. Loans with principal forgiveness are considered to carry a continuing concession and therefore are not eligible for reconsideration of TDR status upon a subsequent modification. Loans that are no longer considered TDRs would be measured for impairment under FASB ASC 450-20.

#### **Credit Card Lending**

8.76 On January 8, 2003, the federal banking agencies issued Account Management and Loss Allowance Guidance for Credit Card Lending. The issuance communicated the expectations for prudent practices in a variety of account management, risk management, and loss allowance practices of institutions engaged in credit card lending. The account management portion of the guidance covers credit lines, overlimit practices, negative amortization, workout programs, and settlements. The loss allowance portion of the guidance covers a number of factors that should be considered by institutions when they estimate and account for their allowance for loan losses. On September 24, 2009, the OTS issued CEO Memo 321, No Interest, No Payment Credit Card Programs, which reminds savings associations of some of the specific requirements of the January 8, 2003, guidance, such as requiring a minimum payment from the borrower each month for all credit card programs, including private label arrangements with retailers. Readers are encouraged to visit the "OTS CEO Memos" page at www.occ.gov to review the full text of this memo.

8.77 For further information, on credit losses, see chapter 9 of this guide.

## **Nontraditional Mortgage Products**

8.78 Because of increased consumer demand for closed-end residential mortgage loan products that allow borrowers to defer repayments of principal and sometimes interest, mortgage institutions are offering nontraditional mortgage loans such as "interest only" mortgages, or mortgages with subprime interest rates. On October 4, 2006, the federal banking agencies and the NCUA adopted Interagency Guidance on Nontraditional Mortgage Product Risks including the use of subprime loans. The guidelines remind banks of the risks inherent in nontraditional mortgage lending and outline the types of risks and controls that are expected for an institution that enters this field of lending. Institutions should establish an appropriate ALLL for estimated credit losses inherent in their nontraditional mortgage loan portfolios. Capital levels should be commensurate with the risk characteristics of the nontraditional mortgage loan portfolios. Institutions should also use "stress tests" to analyze the performance of their nontraditional mortgage portfolios. On June 8, 2007, the federal banking agencies and the NCUA adopted Illustrations of Consumer Information for Nontraditional Mortgage Products to assist institutions in implementing the consumer protection portion of the *Interagency Guidance* on *Nontraditional Mortgage Product Risks*.

#### **Correspondent Concentration Risks**

**8.79** On May 4, 2010, the federal banking agencies issued *Correspondent Concentration Risks Interagency Guidance*. The interagency guidance outlines the agencies' expectations for identifying, monitoring, and managing correspondent concentration risks between financial institutions. The guidance also addresses the agencies' expectations relative to performing appropriate due diligence on all credit exposures to and funding transactions with other financial institutions.

## **Leveraged Lending**

**8.80** In March 2013, the federal banking agencies issued *Interagency* Guidance on Leveraged Lending. This guidance, which replaces the 2001 leveraged lending guidance, outlines high-level principles related to safe-and-sound leveraged lending activities, including underwriting considerations, assessing and documenting enterprise value, risk management expectations for credits awaiting distribution, stress-testing expectations, pipeline portfolio management, and risk management expectations for exposures held by the institution. This guidance applies to all financial institutions supervised by the federal banking agencies that engage in leveraged lending activities. The number of community banks with substantial involvement in leveraged lending is small: therefore, the agencies generally expect community banks to be largely unaffected by this guidance. In November 2014, the federal banking agencies subsequently released a frequently asked questions document to foster industry and examiner understanding of the 2013 leveraged lending guidance and to promote consistent application of the guidance in policy formulation, implementation, and regulatory supervisory assessments.

## **Income Recognition on Problem Loans**

**8.81** The federal banking regulators have issued guidance specifically for nonaccrual policies. Following issuance of FASB Statement No. 118, Accounting by Creditors for Impairment of a Loan-Income Recognition and Disclosures—an amendment of FASB Statement No. 114, which is codified in FASB ASC 310-10 and 310-40, the federal banking agencies announced they would retain their existing nonaccrual policies governing the recognition of interest income. This guidance was published in the Federal Register on February 10, 1995.

**8.82** NCUA guidelines state that loans delinquent for 3 months or more should be placed on nonaccrual status and that accrual of interest on loans should be reversed when the loan is determined to be a loss or when it becomes 12 months delinquent, whichever occurs first. State credit union regulators may also have specific requirements for the discontinuance and reversal of accrued income.

## **Credit Union Lending Restrictions**

**8.83** Credit unions can generally only make loans to members. Further restrictions include, but are not limited to, LTV limits, limits on loans to one borrower, limits on member business loans, and limits on loans to officers, directors, and employees.

#### **Lending Statutes**

**8.84** Certain of the more significant federal and state statutes related to consumer and mortgage lending activities follow:

- Home Mortgage Disclosure Act (HMDA). The HMDA requires that mortgage lenders compile and report to the institution's regulatory agency, certain information applicable to applications for home acquisition and improvement loans. The objectives of the regulation are to provide information to the public regarding whether the institution is serving the credit needs of the neighborhoods it serves, and to assist public officials in targeting private-sector investments to the areas in which they are most needed.
- Fair lending statutes. These statutes include the Equal Credit Opportunity Act and the Fair Housing Act, which prohibit discrimination in lending and housing-related activities, and the Fair Credit Reporting Act, which regulates consumer credit reporting activities.
- Real Estate Settlement Procedures Act (RESPA). The RESPA is administered by the CFPB and requires the disclosure of information to mortgage loan applicants about the costs and procedures involved in loan settlement.
- Direct consumer lending. State laws regulating consumer finance operations are designated as licensed-lending, small-loan, or consumer-financing statutes. Diverse state statutes usually regulate mortgage loans and other direct consumer loans. Each branch office of a company that makes direct consumer loans must be licensed by the state in which the office is located. State licensing authorities, many of which are divisions of state banking departments, examine loans to ascertain that they comply with statutory provisions and to determine whether rebates and refunds are properly computed.
- Retail sales financing. Laws governing retail sales financing may require offices to be licensed or registered. The laws vary widely among states. For example, all goods statutes may govern consumer goods loans; other goods laws may govern loans for consumer goods excluding automobiles. Additional statutes may affect revolving credit arrangements.
- Federal Consumer Credit Protection Act (Truth in Lending Act). The act, through Federal Reserve Regulation Z, requires disclosure of finance charges and annual percentage rates so that consumers can more readily compare various credit terms. It does not set maximum or minimum rates of charges.

#### **Uniform Commercial Code**

**8.85** The Uniform Commercial Code (UCC), fully adopted by all states, is a set of statutes designed to provide consistency among state laws concerning various commercial transactions. Article 9 of the UCC, which addresses secured transactions, contains especially significant laws that affect financing activities. It applies to two-party collateralized loan transactions as well as to sales of accounts receivable and retail sales contracts, which are essentially three-party transactions. Article 9 generally provides certain rights to the

secured parties and the debtors involved in secured transactions. The definition of a secured party includes a lender who obtains a security interest as well as a buyer of trade accounts receivable or retail sales contracts. Similarly, the definition of a debtor includes both the individual obligor and the seller of trade accounts receivable or retail sales contracts.

**8.86** Under Article 9, all transactions creating a security interest are treated alike. The article sets forth various procedures necessary to safeguard, or *perfect*, the potential creditor's interest in collateral against the interests of other creditors. According to Article 9, those procedures generally require that the creditor file a *financing statement* at a specified public office. The statement, available for public inspection, provides legal notice of a perfected security interest. Consequently, before making collateralized loans, prospective lenders generally search the public files to determine if other lenders have already filed financing statements against the collateral.

**8.87** For certain commercial financing activities, Article 9 permits *continuing general lien arrangements*, in which a security interest applies continuously to all present and future collateral of the type described in the financing statement for as long as the financing statement is effective. That provision simplifies, for example, maintaining security interests in purchased receivables and in collateral securing revolving loans. The underlying collateral becomes subject to the security interest as soon as it comes into existence or into the debtor's possession. The financing statement is generally effective for five years from the date of filing and then lapses, unless a *continuation statement* is filed within the six-month period before the expiration date. The continuation statement extends the security interest for another five years.

## **Bank Accounting Advisory Series**

**8.88** The OCCs *Bank Accounting Advisory Series* (BAAS) is updated periodically to express the Office of the Chief Accountant's current views on accounting topics of interest to national banks and federal savings associations. See further discussion of the BAAS in paragraph 7.82 of this guide. Topic 2, "Loans," of the BAAS includes interpretations and responses on (a) TDRs, (b) nonaccruals, (c) commitments, (d) origination fees and costs, (e) loans HFS, and (f) loan recoveries. Readers are encouraged to view this publication under the "Publications—Bank Management" page at www.occ.gov.

## **Accounting and Financial Reporting**

#### © Update 8-2 Accounting and Reporting: Credit Losses

FASB ASU No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, issued in June 2016, is effective for fiscal years of public business entities that are SEC filers beginning after December 15, 2019, including interim periods within those fiscal years.

For all other public business entities, the amendments in FASB ASU No. 2016-13 are effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years.

For all other entities, including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan

accounting, FASB ASU No. 2016-13 is effective for fiscal years beginning after December 15, 2020, and interim periods within fiscal years beginning after December 15, 2021.

Early application is permitted for all entities as of the fiscal years beginning after December 15, 2018, including interim periods within those fiscal years.

FASB ASU No. 2016-13 creates FASB ASC 326, Financial Instruments—Credit Losses, to amend guidance on reporting credit losses for assets held at amortized cost basis and available-for-sale debt securities.

For assets held at amortized cost basis, FASB ASC 326 eliminates the probable initial recognition threshold in current GAAP and, instead, requires an entity to reflect its current estimate of all expected credit losses. The allowance for credit losses is a valuation account that is deducted from the amortized cost basis of the financial assets to present the net amount expected to be collected.

For available-for-sale debt securities, credit losses should be measured in a manner similar to current GAAP. However, FASB ASC 326 will require that credit losses be presented as an allowance rather than as a write-down.

FASB ASU No. 2016-13 affects entities holding financial assets and net investment in leases that are not accounted for at fair value through net income. The amendments affect loans, debt securities, trade receivables, net investments in leases, off-balance-sheet credit exposures, reinsurance receivables, and any other financial assets not excluded from the scope that have the contractual right to receive cash.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-13, see appendix G, "Accounting for Financial Instruments," of this guide.

8.89 FASB ASC 310-10-35-47 states that loans and trade receivables that management has the intent and ability to hold for the foreseeable future or until maturity or payoff should be reported in the balance sheet at outstanding principal adjusted for any charge-offs, the allowance for loan losses (or the allowance for doubtful accounts), any deferred fees or costs on originated loans, and any unamortized premiums or discounts on purchased loans. (Chapter 9 of this guide addresses the allowance for loan losses.) FASB ASC 860-20-35-2 requires financial assets, except for instruments that are within the scope of FASB ASC 815-10, that can contractually be prepaid or otherwise settled in such a way that the holder would not recover substantially all of its recorded investment, should be subsequently measured like investments in debt securities classified as available for sale or trading under FASB ASC 320, Investments— Debt and Equity Securities. Examples of such financial assets include, but are not limited to, interest-only strips, other beneficial interests, loans, or other receivables. Readers may refer to FASB ASC 860-20 as well as chapter 10 of this guide for further guidance.

**8.90** Mortgage loans HFS should be reported at the lower of cost or fair value determined as of the balance sheet date, according to FASB ASC 948-310-35-1. If a mortgage loan has been the hedged item in a fair value hedge

(as addressed in FASB ASC 815, *Derivatives and Hedging*<sup>9</sup>), the loan's cost basis used in the lower-of-cost-or-fair-value accounting should reflect the adjustments of its carrying amount made pursuant to FASB ASC 815-25-35-1. In accordance with FASB ASC 948-310-40-1, after the securitization of a mortgage loan HFS that meets the conditions for a sale addressed in FASB ASC 860-10-40-5, any mortgage-backed securities received by the transferor as proceeds should be classified in accordance with the provisions of FASB ASC 320. However, FASB ASC 948-310-35-3A states that a mortgage banking entity should classify as trading any retained mortgage-backed securities that it commits to sell before or during the securitization process. An entity is prohibited from reclassifying loans as investment securities unless the transfer of those loans meets the conditions for sale accounting addressed in FASB ASC 860-10-40-5.

- **8.91** FASB ASC 310-10-35-48 states that nonmortgage loans HFS should be reported at the lower of cost or fair value. Chapter 10 of this guide addresses accounting and reporting at the time the decision is made to sell loans as well as treatment of loans held for investment. This chapter addresses accounting and reporting subsequent to a transfer into a HFS classification.
- **8.92** Mortgage and nonmortgage loans may qualify for application of the "Fair Value Option" subsections of FASB ASC 825-10 upon origination or purchase. The reason for such an election often is that the loan is being economically hedged with a derivative instrument that is required to be recorded at fair value. Application of the fair value option eliminates the need to qualify for hedge accounting. Those subsections, as stated in FASB ASC 825-10-05-5, address circumstances in which entities may choose, at specified election dates, to measure eligible items at fair value (the fair value option). See FASB ASC 825-10-15 for guidance on the scope of the "Fair Value Option" subsections of FASB ASC 825, *Financial Instruments*. See chapter 20, "Fair Value," of this guide for a summary of FASB ASC 825.
- **8.93** FASB ASC 310-10-25-3 states that transfers of receivables under factoring arrangements meeting the sale criteria of FASB ASC 860-10-40-5 should be accounted for by the factor as purchases of receivables. The acquisition of receivables and accounting for purchase discounts such as factoring commissions should be recognized in accordance with FASB ASC 310-20. Factoring commissions under these arrangements should be recognized over the

**AAG-DEP 8.91** 

<sup>&</sup>lt;sup>9</sup> In May 2010, FASB issued proposed Accounting Standard Update (ASU) Accounting for Financial Instruments and Revisions to the Accounting for Derivative Instruments and Hedging Activities—Financial Instruments (Topic 825) and Derivatives and Hedging (Topic 815). The main objective of this proposal is to provide financial statement users with a more timely and representative depiction of an entity's involvement in financial instruments while reducing the complexity in accounting for those instruments. It develops a consistent framework for classifying financial instruments, removes the threshold for recognizing credit impairments creating a single credit impairment model for both loans and debt securities, and makes changes to the requirements to qualify for hedge accounting.

On February 9, 2011, FASB issued a discussion paper, Invitation to Comment—Selected Issues about Hedge Accounting, to solicit input on the International Accounting Standards Board (IASB)'s exposure draft Hedge Accounting in order to improve, simplify, and converge the financial reporting requirements for hedging activities. Specifically, it requests stakeholders to comment on whether the IASB's exposure draft is a better starting point for any changes to U.S. generally accepted accounting principles (GAAP) as it relates to derivatives and hedging activities.

With the other major financial instruments related projects completed, FASB has re-engaged in the hedging project. As of the release of this guide, FASB staff are drafting an exposure draft for public comment.

Readers are encouraged to visit the "Technical Agenda" page at www.fasb.org for the latest developments regarding FASB's research project titled Accounting for Financial Instruments—Hedging and how it may impact the guidance in this chapter

period of the loan contract in accordance with FASB ASC 310-20. That period begins when the finance company (or an entity with financing activities, including trade receivables) funds a customer's credit and ends when the customer's account is settled.

**8.94** FASB ASC 310-10-25-8 requires that transfers not meeting the sale criteria in FASB ASC 860-10-40-5 should be accounted for as secured loans (that is, loans collateralized by customer accounts or receivables). FASB ASC 860-30-25-5 provides additional guidance in those situations.

#### Interest Income, Delinquency Fees, Prepayment Fees, and Rebates

- **8.95** Interest income on performing loans should be accrued and credited to interest income as it is earned, using the interest method.
- **8.96** Entities involved in transactions in which captive finance companies offer favorable financing to increase sales of related companies should account for such transactions under the guidance in FASB ASC 835-30. FASB ASC 835-30 provides guidance for the appropriate accounting when the face amount of a note does not reasonably represent the present value of the consideration given or received in an exchange.
- **8.97** Delinquency fees are amounts debtors pay because of late payment on loans. Such fees are generally small and are intended to represent additional interest to compensate the lender for the time value and additional collection costs associated with delinquencies.
- **8.98** Finance companies may charge various types of fees to customers in connection with lending transactions, including prepayment penalties—amounts borrowers pay to lenders, in addition to remaining outstanding principal, if borrowers pay off loans prior to contractual maturity.
- **8.99** Paragraphs 12–13 of FASB ASC 310-10-25 address recognition guidance related to prepayment and delinquency fees. Prepayment penalties should not be recognized in income until loans (or trade receivables, if applicable) are prepaid, except that the existence of prepayment penalties may affect the accounting resulting from the application of item (a) in FASB ASC 310-20-35-18. Delinquency fees should be recognized in income when chargeable, assuming collectibility is reasonably assured.
- **8.100** FASB 310-10-05-7 states that rebates represent refunds of portions of the precomputed finance charges on installment loans (or trade receivables, if applicable) that occur when payments are made ahead of schedule. Rebate calculations generally are governed by state laws and may differ from unamortized finance charges on installment loans or trade receivables because many states require rebate calculations to be based on the Rule of 78s or other methods instead of the interest method. FASB ASC 310-10-25-11 states that the accrual of interest income on installment loans or trade receivables should not be affected by the possibility that rebates may be calculated on a method different from the interest method, except that the possibility of rebates affects the accounting resulting from the application of item (a) in FASB ASC 310-20-35-18. Differences between rebate calculations and accrual of interest income merely adjust original estimates of interest income and should be recognized in income when loans or trade receivables are prepaid or renewed.

#### Loan Fees, Costs, Discounts, and Premiums

8.101 FASB ASC 310-20 provides guidance on the recognition, measurement, derecognition, and disclosure of nonrefundable fees, origination costs, and acquisition costs associated with lending activities and loan purchases. FASB ASC 310-20-35-2 states that loan origination fees deferred in accordance with FASB ASC 310-20-25-2 should be recognized over the life of the loan as an adjustment of yield (interest income). Likewise, direct loan origination costs deferred in accordance with FASB ASC 310-20-25-2 should be recognized as a reduction in the yield of the loan except as set forth in FASB ASC 310-20-35-12 (for a TDR)<sup>10</sup> FASB ASC 310-20-30-2 explains that loan origination fees and related direct loan origination costs for a given loan should be offset and only the net amount should be deferred. If the entity holds a large number of similar loans for which prepayments are probable and timing and amount of prepayments can be reasonably estimated, the entity may consider estimates of future principal prepayments in the calculation of the constant effective yield necessary to apply the interest method, in accordance with FASB ASC 310-20-35-26. Otherwise, the payment terms required by the loan contract should be used.

**8.102** Direct loan origination costs must be deferred irrespective of the existence of related loan fees. As defined in the FASB ASC glossary, *direct loan origination costs* include only incremental direct costs of loan origination incurred in transactions with independent third parties and certain costs directly related to specified activities performed by the lender for that loan. Unsuccessful loan origination efforts and other indirect costs, which include administrative costs, rent, depreciation, and all other occupancy and equipment costs, should be charged to expense as incurred, according to FASB ASC 310-20-25-3.

**8.103** FASB ASC 310-20-35-3 explains that, except as set forth in this paragraph, fees received for a commitment to originate or purchase a loan or group of loans should be, if the commitment is exercised, recognized over the life of the loan as an adjustment of yield or, if the commitment expires unexercised, recognized in income upon expiration of the commitment:

- a. If the entity's experience with similar arrangements indicates that the likelihood that the commitment will be exercised is remote, the commitment fee should be recognized over the commitment period on a straight-line basis as service fee income. If the commitment is subsequently exercised during the commitment period, the remaining unamortized commitment fee at the time of exercise should be recognized over the life of the loan as an adjustment of yield. The term remote is used here, consistent with its use in FASB ASC 450, Contingencies, to mean that the likelihood is slight that a loan commitment will be exercised before its expiration.
- b. If the amount of the commitment fee is determined retrospectively as a percentage of the line of credit available but unused in a previous period, if that percentage is nominal in relation to the stated interest rate on any related borrowing, and if that borrowing will bear a market interest rate at the date the loan is made, the commitment fee should be recognized as service fee income as of the determination date.

 $<sup>^{10}</sup>$  Paragraph 10.26 addresses the accounting guidance for loan origination fees and direct loan origination costs when a loan is held for resale.

- **8.104** FASB ASC 310-20 does not apply to fees and costs related to a commitment to originate, sell, or purchase loans that is accounted for as a derivative instrument under FASB ASC 815-10, as stated in item (d) in FASB ASC 310-20-15-3.
- **8.105** FASB ASC 310-20-25-22 considers that for a purchased loan or a group of loans, the initial investment should include the amount paid to the seller, net of fees paid or received. All other costs related to acquiring purchased loans or committing to purchase loans should be charged to expense as incurred. FASB ASC 310-20-35-15 explains that the difference between the initial investment and the related loan's principal amount at the date of purchase should be recognized as an adjustment of yield over the contractual life of the loan.
- **8.106** FASB ASC 310-20-35-21 explains that certain loan agreements provide no scheduled payment terms (demand loans). Other loan agreements provide the borrower with the option to make multiple borrowings up to a specified maximum amount, to repay portions of previous borrowings, and then reborrow under the same contract (revolving lines of credit). Paragraphs 22–23 of FASB ASC 310-20-35 stipulate that any net fees or costs for a loan that is payable at the lender's demand may be recognized as an adjustment of yield on a straight-line basis over a period that is consistent with (a) the understanding between the borrower and the lender or (b) if no understanding exists, the lender's estimate of the period over which the loan will remain outstanding. The net fees or costs on revolving lines of credit (or similar loan arrangements) should be recognized in income on a straight-line basis over the period the revolving line of credit is active, assuming that borrowings are outstanding for the maximum term provided in the loan contract.
- **8.107** Paragraphs 9–10 of FASB ASC 310-20-35 state that if the terms of the new loan resulting from a loan refinancing or restructuring other than a TDR are at least as favorable to the lender as the terms for comparable loans to other customers with similar collection risks who are not refinancing or restructuring a loan with the lender, the refinanced loan should be accounted for as a new loan. This condition would be met if the new loan's effective yield is at least equal to the effective yield for such loans and modifications of the original debt instrument are more than minor. Any unamortized net fees or costs and any prepayment penalties from the original loan should be recognized in interest income when the new loan is granted. If the refinancing or restructuring does not meet the condition set forth in FASB ASC 310-20-35-9 or if only minor modifications are made to the original loan contract, the unamortized net fees or costs from the original loan and any prepayment penalties should be carried forward as a part of the net investment in the new loan.
- **8.108** FASB ASC 310-20-35-11 states that a modification of a debt instrument should be considered more than minor under FASB ASC 310-20-35-10 if the present value of the cash flows under the terms of the new debt instrument is at least 10 percent different from the present value of the cash flows under the terms of the original instrument. If the difference between the present value of the cash flows under the terms of the new debt instrument and the present value of the remaining cash flows under the terms of the original debt instrument is less than 10 percent, a creditor should evaluate whether the modification is more than minor based on the specific facts and circumstances (and other relevant considerations) surrounding the modification. The guidance in

FASB ASC 470, *Debt*, should be used to calculate the present value of the cash flows for purposes of applying the 10 percent test.

**8.109** Paragraphs 17–25 of FASB ASC 310-20-35 also discuss a variety of other amortization matters, including the treatment of increasing, decreasing, and variable-rate loans.

#### Loans and Debt Securities Acquired With Deteriorated Credit Quality<sup>11</sup>

**8.110** FASB ASC 310-30 provides recognition, measurement, and disclosure guidance regarding loans acquired with evidence of deterioration of credit quality since origination acquired by completion of a transfer for which it is probable, at acquisition, that the investor will be unable to collect all contractually required payments receivable. <sup>12</sup>

**8.111** FASB ASC 310-30 permits an entity the option to aggregate and pool loans possessing common risk characteristics that are acquired together or during the same fiscal quarter. The term *common risk characteristics* is defined in FASB ASC glossary as loans with similar credit risk (for example, evidenced by similar Fair Isaac Company scores, an automated rating process for credit reports) or risk ratings that share one or more predominant risk characteristics, such as financial asset type, collateral type, size, interest rate, date of origination, term, and geographic location. In other words, the pooling of loans is permitted to be done on the basis of as few as, but no less than, two common attributes with similar credit risk or risk ratings as one required element and predominant risk characteristics as to the other required element. For example, it would not be appropriate to aggregate loans based solely on the collateral type of the loans without regard to their credit risk profile or risk rating.

**8.112** The guidance in FASB ASC 310-30 applies to all loans that are acquired by completion of a transfer and includes an individual loan, a pool of loans, a group of loans, and loans acquired in a purchase business combination, as stated in FASB ASC 310-30-15-3. See FASB ASC 310-30-15-2 for a list of scope exceptions. See also paragraph 19.20 of this guide for further discussion of the AICPA letter to the SEC addressing the scope of FASB ASC 310-30. For purposes of applying the recognition, measurement, and disclosure provisions of FASB ASC 310-30, for loans that are not accounted for as debt securities, FASB ASC 310-30-15-6 states that investors may aggregate loans acquired in the same fiscal quarter that have common risk characteristics and thereby use a composite interest rate and expectation of cash flows expected to be collected for the pool. To be eligible for aggregation, each loan first should be determined individually to meet the scope criteria of FASB ASC 310-30-15-2. After determining that certain acquired loans are within the scope, as defined in FASB ASC 310-30-15-2, the investor may evaluate whether such loans have common risk characteristics, thus permitting the aggregation of such loans into one or more pools.

<sup>&</sup>lt;sup>11</sup> The AICPA issued Technical Questions and Answers sections 2130.09–.37 (AICPA, Technical Questions and Answers), which address accounting for certain loans or debt securities acquired in a transfer. For additional information visit the AICPA website at www.aicpa.org.

<sup>&</sup>lt;sup>12</sup> Related financial reporting by liquidating banks is beyond the scope of this guidance. See FASB Accounting Standards Codification (ASC) 205-30 for guidance on when and how an entity should prepare its financial statements using the liquidation basis of accounting and describe the related disclosures that should be made.

- **8.113** Upon completion of a transfer of a credit impaired loan, FASB ASC 310-30 requires that the investor (transferee) should recognize the acquired loans initially at fair value. Subsequently, the excess of all cash flows expected at acquisition over the investor's initial investment in the loan should be accreted as interest income on a level-yield basis over the life of the loan (defined as the *accretable yield* in the FASB ASC glossary), according to FASB ASC 310-30-35-2.
- **8.114** FASB ASC 310-30-45-1 requires that the amount of accretable yield not be displayed in the balance sheet. In addition, the loan's contractually required payments receivable in excess of the amount of its cash flows expected at acquisition (defined as the *nonaccretable difference* in the FASB ASC glossary) should not be displayed in the balance sheet or recognized as an adjustment of yield, a loss accrual, or a valuation allowance for credit risk.
- **8.115** According to paragraphs 6–7 of FASB ASC 310-30-35, an increase in accretable yield establishes a higher effective interest rate and a different threshold for any subsequent impairment determination. The subsequent measurement and accounting depends on whether the loans are accounted for as a debt security or are not accounted for as a debt security.
- **8.116** For both loans accounted for and not accounted for as a debt security, an investor should continue to estimate cash flows expected to be collected over the life of the loan, in accordance with paragraphs 8 and 10 of FASB ASC 310-30-35, respectively.
- **8.117** For loans accounted for as a debt security, in accordance with FASB ASC 310-30-35-8, if upon subsequent evaluation:
  - a. The fair value of the debt security has declined below its amortized cost basis, an entity should determine whether the decline is other than temporary. An entity should apply the impairment of securities guidance in FASB ASC 320-10-35. The investor should consider both the timing and amount of cash flows expected to be collected in making a determination about whether there has been a decrease in cash flows expected to be collected.
  - b. Based on current information and events, there is a significant increase in cash flows previously expected to be collected or if actual cash flows are significantly greater than cash flows previously expected, the investor should recalculate the amount of accretable yield for the loan as (1) the excess of the revised cash flows expected to be collected, over (2) the sum of the initial investment, less (3) cash collected, less (4) other-than-temporary impairments, plus (5) the amount of yield accreted to date.
- **8.118** For loans not accounted for as a debt security, in accordance with FASB ASC 310-30-35-10, if, upon subsequent evaluation that is based on current information and events, it is probable that
  - a. the investor is unable to collect all cash flows expected at acquisition plus additional cash flows expected to be collected arising from changes in estimate after acquisition (in accordance with item (b)(ii), then the loan should be considered impaired for purposes of applying FASB ASC 450 or, if applicable, FASB ASC 310, Receivables.

- b. there is a significant increase in cash flows previously expected to be collected, or if actual cash flows are significantly greater than cash flows previously expected, the investor should
  - reduce any remaining valuation allowance (or allowance for loan losses) for the loan established after its acquisition for the increase in the present value of cash flows expected to be collected.
  - ii. recalculate the amount of accretable yield for the loan as (1) the excess of the revised cash flows expected to be collected, over (2) the sum of the initial investment, less (3) cash collected, less (4) write-downs, plus (5) the amount of yield accreted to date.
- **8.119** For both loans accounted for and not accounted for as a debt security, the investor should adjust the amount of accretable yield by reclassification from nonaccretable difference. The adjustment should be accounted for as a change in estimate in conformity with FASB ASC 250, *Accounting Changes and Error Corrections*, with the amount of periodic accretion adjusted over the remaining life of the loan, in accordance with paragraphs 9 and 11 of FASB ASC 310-30-35, respectively. In addition, for loans not accounted for as a debt security, the resulting yield adjustment should be used as the effective interest rate in any subsequent application of item (a) in FASB ASC 310-30-35-10 (see paragraph 8.118).
- 8.120 FASB ASC 310-30-35-14 states that if a loan's contractual interest rate varies based on subsequent changes in an independent factor, such as an index or rate, for example, the prime rate, the London Interbank Offered Rate, or the U.S. Treasury bill weekly average, that loan's contractually required payments receivable should be calculated based on the factor as it changes over the life of the loan. Projections of future changes in the factor should not be made for purposes of determining the effective interest rate or estimating cash flows expected to be collected. Increases in cash flows expected to be collected should be accounted for according to item (b) in FASB ASC 310-30-35-8 or item (b) in 310-30-35-10 (see paragraphs 8.117-.118). Decreases in cash flows expected to be collected resulting directly from a change in the contractual interest rate should be recognized prospectively as a change in estimate in conformity with FASB ASC 250 by reducing, for purposes of applying item (a) in FASB ASC 310-30-35-8 and item (a) in 310-30-35-10 (see paragraphs 8.117-.118), all cash flows expected to be collected at acquisition and the accretable yield. The investor should decrease the amount of accretable yield and the cash flows expected to be collected. Thus, for decreases in cash flows expected to be collected resulting directly from a change in the contractual interest rate, the effect will be to reduce prospectively the yield recognized rather than recognize a loss.
- **8.121** FASB ASC 310-30-30-1 explains that valuation allowances should reflect only those losses incurred by the investor after acquisition—that is, the present value of all cash flows expected at acquisition that ultimately are not to be received. For loans that are acquired by completion of a transfer, it is not appropriate, at acquisition, to establish a loss allowance. For loans acquired in a business combination, the initial recognition of those loans should be the present value of amounts to be received. The loss accrual or valuation allowance recorded by the investor should reflect only losses incurred by the

investor, rather than losses incurred by the transferor or the investor's estimate at acquisition of credit losses over the life of the loan.

**8.122** The prohibition of carrying over any valuation allowance previously recorded by the seller applies to the initial accounting of all loans acquired in a transfer that are within the scope of FASB ASC 310-30.

#### TDRs<sup>13</sup>

- **8.123** FASB ASC 310-40 addresses measurement, derecognition, disclosure, and implementation guidance issues concerning TDRs focused on the creditor's records. See paragraph 8.75 for a discussion on the FFIEC's Supplemental Instructions for September 2014 Call Reports that address the accounting for subsequent restructurings of TDRs.
- **8.124** For creditors, TDRs include certain modifications of terms of loans and receipt of assets from debtors in partial or full satisfaction of loans.
- **8.125** According to FASB ASC 310-40-15-5, a restructuring of a debt constitutes a TDR for purposes of FASB ASC 310-40 if the creditor for economic or legal reasons related to the debtor's financial difficulties grants a concession to the debtor that it would not otherwise consider.
- **8.126** Creditor's concession determination. According to paragraphs 13–14 of FASB ASC 310-40-15, a creditor has granted a concession when, as a result of the restructuring, it does not expect to collect all amounts due, including interest accrued at the original contract rate. In that situation, and if the payment of principal at original maturity is primarily dependent on the value of collateral, an entity should consider the current value of that collateral in determining whether the principal will be paid. A creditor may restructure a debt in exchange for additional collateral or guarantees from the debtor. In that situation, a creditor has granted a concession when the nature and amount of that additional collateral or guarantees received as part of a restructuring do not serve as adequate compensation for other terms of the restructuring. When additional guarantees are received in a restructuring, an entity should evaluate both a guarantor's ability and its willingness to pay the balance owed.
- **8.127** Paragraphs 15–16 of FASB ASC 310-40-15 state if a debtor does not otherwise have access to funds at a market rate for debt with similar risk characteristics as the restructured debt, the restructuring would be considered to be at a below-market rate, which may indicate that the creditor has granted a concession. In that situation, a creditor should consider all aspects of the restructuring in determining whether it has granted a concession. A temporary or permanent increase in the contractual interest rate as a result of a restructuring does not preclude the restructuring from being considered a concession because the new contractual interest rate on the restructured debt could still be below market interest rates for new debt with similar risk characteristics. In that situation, a creditor should consider all aspects of the restructuring in determining whether it has granted a concession.

<sup>&</sup>lt;sup>13</sup> The Center for Audit Quality issued the white paper Application of FASB Statement No. 114 to Modifications of Residential Mortgage Loans that Qualify as Troubled Debt Restructuring on December 23, 2008. The purpose of this nonauthoritative paper is to assist preparers and auditors by discussing questions related to the application of existing GAAP associated with the application of FASB Statement No. 114, Accounting by Creditors for Impairment of a Loan—an amendment of FASB Statements No. 5 and 15, which is codified in FASB ASC 310-10-35 and 310-40.

- **8.128** Paragraphs 17–18 of FASB ASC 310-40-15 provides guidance in evaluating whether a restructuring resulting in a delay in payment is insignificant; and therefore, would not be considered a concession.
- **8.129** *Debtor's financial difficulties determination.* In evaluating whether a receivable is a TDR, a creditor must determine whether the debtor is experiencing financial difficulties, according to FASB ASC 310-40-15-20. In making this determination, a creditor should consider the following indicators:
  - a. The debtor is currently in payment default on any of its debt. In addition, a creditor should evaluate whether it is probable that the debtor would be in payment default on any of its debt in the foreseeable future without the modification. That is, a creditor may conclude that a debtor is experiencing financial difficulties, even though the debtor is not currently in payment default.
  - b. The debtor has declared or is in the process of declaring bankruptcy.
  - c. There is substantial doubt concerning whether the debtor will continue to be a going concern.
  - d. The debtor has securities that have been delisted, are in the process of being delisted, or are under threat of being delisted from an exchange.
  - e. On the basis of estimates and projections that only encompass the debtor's current capabilities, the creditor forecasts that the debtor's entity-specific cash flows will be insufficient to service any of its debt (both interest and principal) in accordance with the contractual terms of the existing agreement for the foreseeable future.
  - f. Without the current modification, the debtor cannot obtain funds from sources other than the existing creditors at an effective interest rate equal to the current market interest rate for similar debt for a nontroubled debtor.

The preceding list of indicators is not intended to include all indicators of a debtor's financial difficulties.

- **8.130** Restructuring Characteristics. As stated in FASB ASC 310-40-15-9, a TDR may include, but is not necessarily limited to, one or a combination of the following (a) transfer from the debtor to the creditor of receivables from third parties, real estate, or other assets to satisfy fully or partially a debt (including a transfer resulting from foreclosure or repossession), (b) issuance or other granting of an equity interest to the creditor by the debtor to satisfy fully or partially a debt unless the equity interest is granted pursuant to existing terms for converting the debt into an equity interest, and (c) modification of terms of a debt.
- **8.131** FASB ASC 310-40-15-9 further states that modification of terms of debt may include one or a combination of any of the following:
  - a. Reduction (absolute or contingent) of the stated interest rate for the remaining original life of the debt
  - b. Extension of the maturity date or dates at a stated interest rate lower than the current market rate for new debt with similar risk
  - c. Reduction (absolute or contingent) of the face amount or maturity amount of the debt as stated in the instrument or other agreement
  - d. Reduction (absolute or contingent) of accrued interest

**8.132** *Modification of terms*. A creditor in a TDR involving only a modification of terms of a receivable—that is, not involving receipt of assets (including an equity interest in the debtor)—should account for the TDR in accordance with the provisions of FASB ASC 310, as explained in FASB ASC 310-40-35-5.

Loans

- **8.133** Item (a) in FASB ASC 320-10-55-2 clarifies that any loan that was restructured as a security in a TDR involving a modification of terms would be subject to the provisions of FASB ASC 320 if the debt instrument meets the definition of a *security* (as provided in the FASB ASC glossary). FASB 310-40-40-8A provides guidance on how to account for any excess of the fair value of the debt security received in restructuring over the net carrying amount of the loan at the date of the restructuring.
- **8.134** Partial Satisfaction of a Receivable. In accordance with FASB ASC 310-40-35-7, TDRs involving receipt of assets (including an equity interest in the debtor) in partial satisfaction of a receivable and a modification of terms of the remaining receivable should be accounted for as prescribed in FASB ASC 310 except that, first, the assets received should be accounted for as prescribed in paragraphs 2–4 of FASB ASC 310-40-40 (see paragraph 8.135) and the recorded investment in the receivable should be reduced by the fair value less cost to sell of the assets received. If cash is received in a partial satisfaction of a receivable, the recorded investment in the receivable should be reduced by the amount of cash received.
- **8.135** Receipts of assets. According to paragraphs 2–3 of FASB ASC 310-40-40, a creditor that receives from a debtor in full satisfaction of a receivable either (a) receivables from third parties, real estate, or other as-sets or (b) shares of stock or other evidence of an equity interest in the debtor, or both, should account for those assets (including an equity interest) at their fair value at the time of the restructuring (see FASB ASC 820, Fair Value Measurement, for guidance regarding fair value measurements and chapter 20 of this guide for a summary of FASB ASC 820). A creditor that receives long-lived assets that will be sold from a debtor in full satisfaction of a receivable should account for those assets at their fair value less cost to sell as that term is used in FASB ASC 360-10-35-43. The excess of the recorded investment in the receivable satisfied over the fair value of assets received (less cost to sell, if required) is a loss to be recognized. For purposes of this paragraph, losses, to the extent they are not offset against allowances for uncollectible amounts or other valuation accounts, should be included in measuring net income for the period.
- **8.136** *Impairment*. A loan restructured in a TDR is an impaired loan, as stated in FASB ASC 310-40-35-10. It should not be accounted for as a new loan because a TDR is part of a creditor's ongoing effort to recover its investment in the original loan. A loan usually will have been identified as impaired because the conditions specified in paragraphs 16–17 of FASB ASC 310-10-35 will have existed before a formal restructuring. FASB ASC 310-10-35-21 explains that some impaired loans have risk characteristics that are unique to an individual borrower and the creditor should apply the measurement methods described in FASB ASC 310-30-30-2; 310-10-35-22 through 310-10-35-28; and 310-10-35-37 on a loan-by-loan basis. However, some impaired loans may have risk characteristics in common with other impaired loans. A creditor may aggregate those loans and may use historical statistics, such as average recovery period and average amount recovered, along with a composite effective interest rate as a means of measuring impairment of those loans.

**8.137** Foreclosed Assets in TDRs. "Pending Content" in FASB ASC 310-40-40-6 explains that except in the circumstances described in FASB ASC 310-40-40-6A, a TDR that is in substance a repossession or foreclosure by the creditor (that is, the creditor receives physical possession of the debtor's assets regardless of whether formal foreclosure proceedings take place, or in which the creditor otherwise obtains one or more of the debtor's assets in place of all or part of the receivable) should be accounted for according to the provisions of FASB ASC 310-40-35-7 (see paragraph 8.134), paragraphs 2–4 of FASB ASC 310-40-40 and, if appropriate, FASB ASC 310-40-40-8 (see paragraph 8.138). See further discussion on the classification and measurement of certain government-guaranteed mortgage loans upon foreclosure in the "Foreclosed Assets" section of chapter 11 of this guide. For guidance on when a creditor should be considered to have received physical possession (resulting from an in substance repossession or foreclosure) of residential real estate properly collateralizing a consumer mortgage loan, see FASB ASC 310-40-55-10A.

# © Update 8-3 Accounting and Reporting: Revenue From Contracts With Customers

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14) for annual reporting periods of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period. Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as of either:

- An annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- An annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB ASC 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605 as well as guidance within the 900 series of industry-specific topics. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts).

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2014-09, see appendix E of this guide.

**8.138** Sale of Assets from a TDR. FASB ASC 310-40-40-8 states that a receivable from the sale of assets previously obtained in a TDR should be accounted for according to FASB ASC 835-30 regardless of whether the assets were obtained in satisfaction (full or partial) of a receivable to which FASB ASC 835, *Interest*, was not intended to apply. A difference, if any, between the amount of the new receivable and the carrying amount of the assets sold is a gain or loss on sale of assets.

#### Real Estate Investments

8.139 The "Acquisition, Development, and Construction Arrangements" subsections in FASB ASC 310-10 provide guidance for determining whether a lender should account for an ADC arrangement as a loan or as an investment in real estate or a joint venture. As explained in FASB ASC 310-10-05-9, lenders may enter into ADC arrangements in which they have virtually the same risks and potential rewards as those of owners or joint venturers. Loans granted to acquire operating properties sometimes grant the lender a right to participate in expected residual profit from the sale or refinancing of the property. The expected residual profit may take the form of a percentage of the appreciation of the property determined at the maturity of the loan. If the lender is expected to receive over 50 percent of the expected residual profit from the project, item (a) in FASB ASC 310-10-25-27 states that the lender should account for income or loss from the arrangement as a real estate investment as specified in FASB ASC 970, Real Estate—General. See chapter 11 of this guide. Effective January 1, 2015, high volatility commercial real estate loans carry a capital charge that is 50 percent higher than the capital charge for other commercial real estate loans.

## Lease Financina 14

**8.140** Accounting for leases by lessees and lessors is established in FASB ASC 840. FASB ASC 840-40-55-38 states that a transaction should be considered a sale-leaseback transaction subject to FASB ASC 840-40 if the preexisting lease is modified in connection with the sale, except for insignificant changes. Accordingly, transactions with modifications to the preexisting lease involving real estate should be accounted for in accordance with the guidance in FASB ASC 840-40 that addresses sale-leaseback transactions involving real estate. If the preexisting lease is not modified in conjunction with the sale, except for insignificant changes, profit should be deferred and recognized in accordance with FASB ASC 840-40-25-3.

#### **Foreign Loans**

- **8.141** Accounting for foreign loans is generally the same as for single-jurisdiction, domestic loans. However, unique issues arise regarding the accounting for restructured debt of developing countries and the recognition of interest income on such loans.
- **8.142** FASB ASC 942-310 addresses situations where a financially troubled country may suspend the payment of interest on its loans. Debt-equity swap programs are in place in several financially troubled countries, as stated

<sup>&</sup>lt;sup>14</sup> This edition of the guide has not been updated to reflect the changes as a result of FASB ASU No. 2016-02, *Leases (Topic 842)*, how-ever, this discussion will be updated in a future edition. See Update 8-1 for further details on the amendments and effective date of FASB ASU No. 2016-02.

in FASB ASC 942-310-05-3. Although the programs differ somewhat among the countries, the principal elements of each program generally are as follows:

- a. Holders of U.S. dollars-denominated debt of these countries can choose to convert that debt into approved local equity investments.
- b. The holders are credited with local currency, at the official exchange rate, approximately equal to the U.S. dollar debt.
- c. A discount from the official exchange rate is usually imposed as a transaction fee.
- d. The local currency credited to the holder must be used for an approved equity investment.
- e. The local currency is not available to the holders for any other purpose.
- f. Dividends on the equity investment can generally be paid annually, although there may be restrictions on the amounts of the dividends or on payment of dividends in the early years of the investment.
- g. Capital usually cannot be repatriated for several years, and although some countries permit the investment to be sold, the proceeds from any such sale are generally subject to similar repatriation restrictions.
- **8.143** In accordance with paragraph 1-2 of FASB ASC 942-310-30, debt/equity swaps (defined in the FASB ASC glossary as an exchange transaction of a monetary asset for a nonmonetary asset) should be measured at fair value at the date the transaction is agreed to by both parties. Because the secondary market for debt of financially troubled countries may be considered to be thin, it may not be the best indicator of the fair value of the equity investment or of net assets received. FASB ASC 942-310-30-3 provides factors to consider in measuring fair values of debt/equity swap transactions.
- **8.144** If amounts are received on a financially troubled country's loan on which the accrual of interest has been suspended, FASB ASC 942-310-35-2 notes that a determination should be made about whether the payment should be recorded as a reduction of the principal balance of the loan or as interest income.

#### Commitments

**8.145** Paragraphs 69–71 of FASB ASC 815-10-15 provide guidance on the types of loan commitments that are derivatives under FASB ASC 815-10 (and therefore required to be accounted for as derivatives) and those that are excluded from the scope. Notwithstanding the characteristics discussed in FASB ASC 815-10-15-83, loan commitments that relate to the origination of mortgage loans that will be HFS (as discussed in FASB ASC 948-310-25-3) should be accounted for as derivative instruments by the issuer of the loan commitment (that is, the potential lender). For the holder of a commitment to originate a loan (that is, the potential borrower), that loan commitment is not subject to the requirements of FASB ASC 815-10. For issuers of loan commitments to originate mortgage loans that will be held for investment purposes, as discussed in paragraphs 3–4 of FASB ASC 948-310-25, those loan commitments are not subject to FASB ASC 815-10.

- **8.146** Loan commitments to originate loans, excluded from the scope of FASB ASC 815-10, are accounted for under FASB ASC 948, *Financial Services—Mortgage Banking*, or FASB ASC 310-20, as appropriate.
- **8.147** However, commitments to purchase or sell mortgage loans or other types of loans at a future date must be evaluated under the definition of a derivative instrument to determine whether FASB ASC 815-10 applies, according to FASB ASC 815-10-15-70.
- **8.148** Commitments to originate mortgage loans that will be HFS generally qualify as derivative instruments and are recorded at fair value at inception and changes in fair value are recorded in current earnings. Chapters 10 and 18, "Derivative Instruments: Futures, Forwards, Options, Swaps, and Other Derivative Instruments," of this guide address commitments to sell loans. Chapter 9 and subsequent paragraphs of this guide address accounting for loss contingencies in conformity with FASB ASC 450.
- **8.149** FASB ASC 310-10-05-5 states that entities sometimes enter into forward standby commitments to purchase loans at a stated price in return for a standby commitment fee. In such an arrangement, settlement of the standby commitment is at the option of the seller of the loans and would result in delivery to the entity only if the contract price equals or exceeds the market price of the underlying loan or security on the settlement date. A standby commitment differs from a mandatory commitment in that the entity assumes all the market risks of ownership but shares in none of the rewards. A standby commitment is, in substance, a written put option that will be exercised only if the value of the loans is less than or equal to the strike price.
- **8.150** Many entities use standby commitments to supplement their normal loan origination volume. Such standby commitments may be subject to the scope of FASB ASC 815 if they satisfy the definition of a derivative in paragraphs 83–139 of FASB ASC 815-10-15. Standby commitments discussed in the previous paragraph that satisfy the definition of a derivative are recognized in the statement of financial position at inception and measured at the fair value of the commitment. In accordance with FASB ASC 815-10-35-2, changes in fair value of derivative instruments not designated in hedging relationships are recognized currently in earnings.
- **8.151** FASB ASC 310-10-30-7 states that if a standby commitment is viewed under FASB ASC 310-10-25-6 as part of the normal production of loans, an entity should record loans purchased under the standby commitment at cost on the settlement date, net of the standby commitment fee received, in conformity with FASB ASC 310-20. If a standby commitment is accounted for as a written option as discussed in FASB ASC 310-10-25-6, the option premium received (standby commitment fee) should be recorded as a liability representing the fair value of the standby commitment on the trade date.
- **8.152** As stated in FASB ASC 310-10-35-46, this guidance applies only to standby commitments to purchase loans. It does not apply to other customary kinds of commitments to purchase loans, nor does it apply to commitments to originate loans. See FASB ASC 310-10-35-46 for subsequent measurement guidance related to standby commitments to purchase loans.

#### Financial Statement Presentation and Disclosure

**8.153** *Loans and trade receivables.* FASB ASC 310-10-50-2 states that the summary of significant accounting policies should include the following:

- The basis for accounting for loans and trade receivables.
- The classification and method of accounting for interest-only strips, loans, other receivables, or retained interests in securitizations that can be contractually prepaid or otherwise settled in a way that the holder would not recover substantially all of its recorded investment.<sup>15</sup>
- The method used in determining the lower of cost or fair value of nonmortgage loans HFS (that is, aggregate or individual asset basis).<sup>16</sup>
- The method for recognizing interest income on loans and trade receivables, including a statement about the entity's policy for the treatment of related fees and costs, including the method of amortizing net deferred fees or costs.
- **8.154** FASB ASC 310-10-45-2 states that loans or trade receivables may be presented on the balance sheet as aggregate amounts. However, such receivables HFS should be a separate balance sheet category. Major categories of loans or trade receivables should be presented separately either in the balance sheet or in the notes to the financial statements.
- **8.155** FASB ASC 310-10-50-4 states that the allowance for credit losses (also referred to as the allowance for doubtful accounts) and, as applicable, any unearned income, any unamortized premiums and discounts, and any net unamortized deferred fees and costs should be disclosed in the financial statements.
- $8.156\,$  Except for credit card receivables, FASB ASC 310-10-50-4A requires that an entity disclose its policy for charging off uncollectible trade accounts receivable that have both a contractual maturity of one year or less and arose from the sale of goods or services.
- **8.157** Nonaccrual and past due financing receivables. FASB ASC 310-10-50-6 requires that a entity's summary of significant accounting policies for financing receivables should include
  - the policy for placing financing receivables, if applicable, on nonaccrual status (or discontinuing accrual of interest)
  - the policy for recording payments received on nonaccrual financing receivables (if applicable),
  - the policy for resuming accrual of interest, and
  - the policy for determining past due or delinquency status.

<sup>&</sup>lt;sup>15</sup> Interest-only strips, other interests that continue to be held by a transferor in securitizations, loans, other receivables, or other financial assets that can contractually be prepaid or otherwise settled in such a way that the holder would not recover substantially all of its recorded investment, except for instruments that are within the scope of FASB ASC 815-10, should be subsequently measured like investments in debt securities classified as available for sale or trading under FASB ASC 320, Investments—Debt and Equity Securities. Interest-only strips and similar interests that continue to be held by a transferor that meets the definition of securities are included in the scope of FASB ASC 320. Therefore, all relevant provisions of FASB ASC 320 (including the disclosure requirements) should be applied. As defined in the FASB ASC glossary, the recorded investment in the receivable is the face amount increased or decreased by applicable accrued interest and unamortized premium, discount, finance charges, or acquisition costs and may also reflect a previous write-down of the investment. See FASB ASC 860-20-35-2 for additional information.

<sup>&</sup>lt;sup>16</sup> A similar requirement exists for mortgage loans HFS.

- **8.158** According to paragraphs 7–7A of FASB ASC 310-10-50, an entity should provide both the following disclosures related to nonaccrual and past due financing receivables as of each balance sheet date: (a) the recorded investment in financing receivables on nonaccrual status and (b) the recorded investment in financing receivables past due 90 days or more and still accruing. An entity should also provide an analysis of the age of the recorded investment in financing receivables at the end of the reporting period that are past due, as determined by the entity's policy. The guidance in FASB ASC 310-10-50-7A does not apply to the financing receivables listed in FASB ASC 310-10-50-7B.
- **8.159** For trade receivables that do not accrue interest until a specified period has elapsed, nonaccrual status would be the point when accrual is suspended after the receivable becomes past due, as stated in FASB ASC 310-10-50-8.
- **8.160** The guidance in paragraphs 6–8 of FASB ASC 310-10-50 does not apply to loans acquired with deteriorated credit quality. Instead, disclosures of such loans should be in accordance with FASB ASC 310-30. In addition, the guidance in paragraphs 6–7A of FASB ASC 310-10-50 should be provided by class of financing receivable except for the financing receivables listed in FASB ASC 310-10-50-5B.
- **8.161** *Impaired loans*. See discussion on disclosure requirements of impaired loans in paragraph 9.55 of this guide.
- **8.162** *Credit quality*. According to paragraphs 28–30 of FASB ASC 310-10-50, an entity should provide information that enables financial statement users to (a) understand how and to what extent management monitors the credit quality of its financing receivables in an ongoing manner and (b) assess the quantitative and qualitative risks arising from the credit quality of its financing receivables. To meet this objective, an entity should provide quantitative and qualitative information by class about the credit quality of financing receivables, including all of the following:
  - A description of the credit quality indicator
  - The recorded investment in financing receivables by credit quality indicator
  - For each credit quality indicator, the date or range of dates in which the information was updated for that credit quality indicator

If an entity discloses internal risk ratings, then the entity should provide qualitative information on how those internal risk ratings relate to the likelihood of loss. This guidance does not apply to the financing receivables listed in FASB ASC 310-10-50-7B, as stated in FASB ASC 310-10-50-27.

- **8.163** *Modifications*. As required by FASB ASC 310-40-50-1, as of the date of each balance sheet presented, a creditor should disclose, either in the body of the financial statements or in the accompanying notes, the amount of commitments, if any, to lend additional funds to debtors owing receivables whose terms have been modified in TDRs.
- **8.164** Paragraphs 2–3 of FASB ASC 310-40-50 explains that information about an impaired loan that has been restructured in a TDR involving a modification of terms need not be included in the disclosures required by items (a)

and (c) in FASB ASC 310-10-50-15 in years after the restructuring if (a) the restructuring agreement specifies an interest rate equal to or greater than the rate that the creditor was willing to accept at the time of the restructuring for a new loan with comparable risk and (b) the loan is not impaired based on the terms specified by the restructuring agreement. That exception should be applied consistently for items (a) and (c) in FASB ASC 310-10-50-15 to all loans restructured in a TDR that meet the criteria in items (a) and (b) in FASB ASC 310-40-50-2.

- **8.165** For each period for which a statement of income is presented, FASB ASC 310-10-50-33 requires an entity to disclose the following about TDRs of financing receivables that occurred during the period:
  - a. By class of financing receivable, qualitative and quantitative information, including how the financing receivables were modified and the financial effects of the modifications
  - By portfolio segment, qualitative information about how such modifications are factored into the determination of the allowance for credit losses
- **8.166** For each period for which a statement of income is presented, FASB ASC 310-10-50-34 requires an entity to disclose the following for financing receivables modified as TDRs within the previous 12 months and for which there was a payment default during the period:
  - a. By class of financing receivable, qualitative and quantitative information about those defaulted financing receivables, including the types of financing receivables that defaulted and the amount of financing receivables that defaulted
  - By portfolio segment, qualitative information about how such defaults are factored into the determination of the allowance for credit losses
- **8.167** Assets serving as collateral. For required disclosures of the carrying amount of loans, trade receivables, securities, and financial instruments that serve as collateral for borrowings see FASB ASC 860-30-50-1A.
- **8.168** Transfers of financial assets. Accounting and financial reporting matters related to the sales or other dispositions of loans are addressed in chapter 10 of this guide.
- **8.169** *Lease financing*. <sup>17</sup> Paragraphs 4–5 of FASB ASC 840-10-50 require certain disclosures by lessors when leasing is a significant part of a lessor's business activities in terms of revenue, net income, or assets.
- **8.170** Fair value option. As of each date for which a statement of financial position is presented, item (e) in FASB ASC 825-10-50-28 states that entities should disclose for loans held as assets, for which the fair value option has been elected, all of the following:
  - a. The aggregate fair value of loans that are 90 days or more past due.

 $<sup>^{17}</sup>$  This edition of the guide has not been updated to reflect the changes as a result of FASB ASU No. 2016-02, however, this discussion will be updated in a future edition. See Update 8-1 for further details on the amendments and effective date of ASU No. 2016-02

247

- b. If the entity's policy is to recognize interest income separately from other changes in fair value, the aggregate fair value of loans in nonaccrual status.
- c. The difference between the aggregate fair value and the aggregate unpaid principal balance for loans that are 90 days or more past due, in nonaccrual status, or both.
- **8.171** For each period for which an income statement is presented, item (c) in FASB ASC 825-10-50-30 requires that entities disclose for loans and other receivables held as assets, for which the fair value option has been elected, both of the following:
  - The estimated amount of gains or losses included in earnings during the period attributable to changes in instrument-specific credit risk
  - How the gains or losses attributable to changes in instrumentspecific credit risk were determined
- **8.172** Concentrations of credit risk of all financial instruments. Paragraphs 20–21 of FASB ASC 825-10-50 require disclosures about all significant concentrations of credit risk arising from all financial instruments except for the instruments described in FASB ASC 825-10-50-22. The following should be disclosed for each significant concentration:
  - a. Information about the (shared) activity, region, or economic characteristic that identifies the concentration
  - b. The maximum amount of loss due to credit risk that, based on the gross fair value of the financial instrument, the entity would incur if parties to the financial instruments that make up the concentration failed completely to perform according to the terms of the contracts and the collateral or other security, if any, for the amount due proved to be of no value to the entity
  - c. With respect to collateral, the entity's policy of requiring collateral or other security to support financial instruments subject to credit risk, information about the entity's access to that collateral or other security, and the nature and a brief description of the collateral or other security supporting those financial instruments
  - d. With respect to master netting arrangements, the entity's policy of entering into master netting arrangements to mitigate the credit risk of financial instruments, information about the arrangements for which the entity is a party, and a brief description of the terms of those arrangements, including the extent to which they would reduce the entity's maximum amount of loss due to credit risk

An entity is encouraged, but not required, to disclose quantitative information about the market risks of financial instruments that is consistent with the way it manages or adjusts those risks. See FASB ASC 825-10-50-23 for possible disclosures.

- **8.173** Information about loan products and terms of loan products that identifies the concentration may also be disclosed.
- **8.174** Disclosure requirements under FASB ASC 825 which might be challenging for financial institutions, may include requirements related to loan fair values based on pricing models, the reconciliation of the beginning and ending balances for fair value measurements using significant unobservable inputs

(level 3), and nonrecurring measurements such as real estate owned. See FASB ASC 820-10-50 for the required disclosures for recurring and nonrecurring fair value measurements.

- **8.175** Off-balance-sheet credit risk. FASB ASC 942-825-50-1 states that off-balance-sheet credit risk refers to credit risk on off-balance-sheet loan commitments, standby letters of credit, financial guarantees, and other similar instruments, except those instruments within the scope of FASB ASC 815. For financial instruments with off-balance-sheet credit risk, except for those instruments within the scope of FASB ASC 815, an entity should disclose the following information:
  - a. The face or contract amount
  - b. The nature and terms, including, at a minimum, a discussion of the
    - i. credit and market risk of those instruments
    - ii. cash requirements of those instruments
    - iii. related accounting policy pursuant to FASB ASC 235-10
  - c. The entity's policy for requiring collateral or other security to support financial instruments subject to credit risk, information about the entity's access to that collateral or other security, and the nature and a brief description of the collateral or other security supporting those financial instruments
- **8.176** Examples of activities and financial instruments with off-balance-sheet credit risk include obligations for loans sold with recourse (with or without a floating-interest-rate provision), fixed-rate and variable-rate loan commitments, financial guarantees, note issuance facilities at floating rates, and letters of credit, as stated in FASB ASC 942-825-50-2.
- **8.177** Related party disclosures. FASB ASC 850, Related Party Disclosures, contains guidance on disclosures about transactions with various related parties. Institutions frequently make loans to parent and affiliated companies, directors, officers, and stockholders, as well as to entities with which directors, officers, and stockholders are affiliated. The aggregate amount of such loans should be disclosed. See paragraphs 5.38–.39 of this guide for Regulation O requirements regarding extension of credit to certain related parties.
- **8.178** Guarantees. FASB ASC 460, Guarantees, establishes the accounting and disclosure requirements to be met by a guarantor for certain guarantees issued and outstanding. Commercial letters of credit and other loan commitments, which are commonly thought of as guarantees of funding, are not included in the scope of FASB ASC 460, as stated in item (a) in FASB ASC 460-10-55-16. However, FASB ASC 460-10-55-2 provides that a financial standby letter of credit is an example of a guarantee contract under the scope of FASB ASC 460. A financial standby letter of credit is defined in the FASB ASC glossary as an irrevocable undertaking (typically by a financial institution) to guarantee payment of a specified financial obligation. See paragraphs 4–7 of FASB ASB 460-10-15 for types of guaranteed contracts included and excluded from the scope of FASB ASC 460.
- **8.179** In accordance with FASB ASC 460-10-25-4, at the inception of a guarantee, the guarantor should recognize in its statement of financial position a liability for that guarantee. FASB ASC 460-10 does not prescribe a

specific account for the guarantor's offsetting entry when it recognizes the liability at the inception of a guarantee. That offsetting entry depends on the circumstances in which the guarantee was issued. The liability that the guarantor initially recognized under FASB ASC 460-10-25-4 would typically be reduced (by a credit to earnings) as the guarantor is released from risk under the guarantee, as stated in FASB ASC 460-10-35-1.

Loans

- **8.180** In accordance with paragraphs 2–3 of FASB ASC 460-10-30, the objective of the initial measurement of a guarantee liability is the fair value of the guarantee at its inception. In the event that, at the inception of the guarantee, the guarantor is required to recognize a liability under FASB ASC 450-20-25 for the related contingent loss, the liability to be initially recognized for that guarantee should be the greater of (a) the amount that satisfies the fair value objective or (b) the contingent liability amount required to be recognized at inception of the guarantee in FASB ASC 450-20-30.
- **8.181** FASB ASC 460-10-50-4 requires a number of disclosures about a guarantor's obligations under guarantees. A guarantor should disclose all of the following information about each guarantee, or each group of similar guarantees, even if the likelihood of the guarantor's having to make any payments under the guarantee is remote:
  - *a*. The nature of the guarantee, including all of the following:
    - i. The approximate term of the guarantee
    - ii. How the guarantee arose
    - iii. The events or circumstances that would require the guarantor to perform under the guarantee
    - iv. The current status (that is, as of the date of the statement of financial position) of the payment/performance risk of the guarantee (for example, the current status of the payment/performance risk of a credit-risk-related guarantee could be based on either recently issued external credit ratings or current internal groupings used by the guarantor to manage its risk)
    - v. If the entity uses internal groupings for purposes of item (a)(iv), and how those groupings are determined and used for managing risk
  - b. The following information about the maximum potential amount of future payments under the guarantee, as appropriate:
    - i. The maximum potential amount of future payments (undiscounted) the guarantor could be required to make under the guarantee, which should not be reduced by the effect of any amounts that may possibly be recovered under recourse or collateralization provisions in the guarantee
    - ii. If the terms of the guarantee provide for no limitation to the maximum potential future payments under the guarantee
    - iii. If the guarantor is unable to develop an estimate of the maximum potential amount of future payments under its guarantee, the reasons why it cannot estimate the maximum potential amount

- c. The current carrying amount of the liability, if any, for the guarantor's obligations under the guarantee (including the amount if any, recognized under FASB ASC 450-20-30) regardless of whether the guarantee is freestanding or embedded in another contract
- d. The nature of any recourse provisions that would enable the guarantor to recover from third parties any of the amounts paid under the guarantee
- e. The nature of any assets held either as collateral or by third parties that, upon the occurrence of any triggering event or condition under the guarantee, the guaranter can obtain and liquidate to recover all or a portion of the amounts paid under the guarantee
- f. If estimable, the approximate extent to which the proceeds from liquidation of assets held either as collateral or by third parties would be expected to cover the maximum potential amount of future payments under the guarantee

## Auditing 18

#### **Objectives**

**8.182** The primary objectives of audit procedures in the loan area are to obtain sufficient appropriate evidence that

- a. loans exist and are owned by the entity as of the balance sheet date;
- b. the allowance for credit losses is adequate for estimated losses that have been incurred in the loan portfolio (audit procedures to satisfy this objective are discussed in chapter 9 of this guide);
- c. loans are properly classified, described, and disclosed in the financial statements, including fair values of loans and concentrations of credit risk:
- d. recorded loans include all such assets of the institution and the financial statements include all related transactions during the period;
- e. loan transactions are recorded in the proper period;
- f. loans HFS are properly classified and are stated at the lower of cost or fair value;
- g. interest income, fees, and costs and the related balance sheet accounts (accrued interest receivable, unearned discount, unamortized purchase premiums and discounts, and unamortized net deferred loan fees or costs) have been properly measured and recorded;
- credit commitments, letters of credit, guarantees, recourse provisions, and loans that collateralize borrowings are properly disclosed in the financial statements; and

<sup>&</sup>lt;sup>18</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board (ASB) and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of *issuers* and *nonissuers* as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

 transfers of loans have been properly accounted for as sales or secured borrowings under FASB ASC 860.

#### **Planning**

8.183 In accordance with AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (see chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide for additional information). As described earlier in this chapter, credit risk is normally the principal risk inherent in lending. The composition of an institution's loan portfolio, which can vary widely from institution to institution, is one of the most important factors in assessing the risks of material misstatement related to loans. For example, the risks associated with construction lending are very different from the risks associated with credit card lending. The current year's interim financial statements and other financial information (for example, board of directors' minutes, asset-classification reports, credit management reports, and reports of the institution's regulators) should be helpful in understanding an institution's credit strategy and loan portfolio characteristics and, thereby, in assessing the related risks of material misstatement. Those reports generally include information about such items as dollar amounts and types of loans; the volume of current originations by type and related net deferred loan fees or costs; identification of TDRs; ADC arrangements; purchases and sales of loans, including gains and losses; and wash sales, among others. Controls over loans should also include controls over allowances and write-offs. Readers may refer to chapter 9 of this guide for guidance.

**8.184** The following factors related to loans may be indicative of risks of material misstatement (and, often, higher control risk) for loans and related amounts:

- Lack of a formal written lending policy
- High rate of growth in the loan portfolio
- Concentration of lending authority in one individual
- Lack of personnel with skills and knowledge of a particular kind of loan, such as credit card or construction
- Significant changes in the composition of an institution's portfolio
- Poor underwriting standards and procedures
- Poor recordkeeping and monitoring of principal and interest receipts
- Significant nontraditional lending activities that involve a higher degree of risk, such as highly leveraged lending transactions
- Significant originations or purchases of loans outside the institution's normal activities or market area
- Sales of loans with significant recourse provisions
- Ambiguous transactions involving the sale or transfer of loans, especially when there is a lack of analysis prior to the transactions

- Failure of personnel to follow management's written lending policies for underwriting and documentation
- Loans that are continuously extended, restructured, or modified
- Loans that are of a type, customer, collateral, industry, or geographical location not authorized by management's written lending policies
- Loans of unusual size or with unusual interest rates or terms
- Significant concentrations of loans in a particular industry or geographic area, or with a particular borrower or relationship of borrowers
- The potential for insider abuse because of significant loans to the institution's officers, directors, shareholders, or other related parties that do not meet normal underwriting standards, such as nominee loans, loans with questionable collateral, and multiple transactions with a single related party or group of affiliated parties
- Significant concentrations of loan products with terms that give rise to a credit risk; such as, negative amortization loans, loans with high LTV ratios, multiple loans on the same collateral that when combined result in a high LTV ratio, and interest-only loans
- Additional charge-off of loans upon resolution exceeding the specific reserve established for impaired loans

# Internal Control Over Financial Reporting and Possible Tests of Controls <sup>19</sup>

**8.185** AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13–.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement

<sup>&</sup>lt;sup>19</sup> The FDIC's Financial Institution Letter-23-2009, *Annual Audit and Reporting Requirements: Final Amendments to Part 363*, provides guidance on the internal control attestation standards that auditors of insured institutions with \$1 billion or more in total assets should follow to comply with FDIC audit and reporting requirements in Part 363 of the FDIC regulations. For more information please refer to chapter 1 and appendix A, "FDI Act Reporting Requirements," of this guide and the FDIC website at www.fdic.gov.

As a result of the ASB's Attestation Clarity Project, the ASB has issued Statement on Auditing Standards (SAS) No. 130, An Audit of Internal Control Over Financial Reporting That Is Integrated With an Audit of Financial Statements (AICPA, Professional Standards, AU-C sec. 940). The ASB concluded that, because engagements performed under AT section 501, An Examination of an Entity's Internal Control Over Financial Reporting That Is Integrated With an Audit of Its Financial Statements (AICPA, Professional Standards), as well as related attestation interpretation No. 1, "Reporting Under Section 112 of the Federal Deposit Insurance Corporation Improvement Act" (AICPA, Professional Standards, AT sec. 9501), are required to be integrated with an audit of financial statements, it is appropriate to move the content of AT section 501 from the attestation standards into generally accepted auditing standards. SAS No. 130 is effective for integrated audits for periods ending on or after December 15, 2016.

253

at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

Loans

- **8.186** An understanding of the internal control over the financial reporting of loans may include controls over transactions such as granting credit, disbursing loan funds, applying loan payments, amortizing discounts, accruing interest income, purchased loans, participations, and syndications as those transactions relate to each significant type of lending activity. Also, procedures are needed to ensure that all appropriate liens have been filed.
- **8.187** Effective controls in this area should provide assurance that errors or fraud in management's financial statement assertions about the loan portfolio—including those due to the failure to execute lending transactions in accordance with management's written lending policies—are prevented or detected. For example, failure to document a second lien as mandated by management's written loan documentation policy could affect financial statement assertions about ownership and valuation.
- 8.188 Factors that contribute to an effective control environment may include
  - those charged with governance take an active role in monitoring lending policies and practices;
  - information systems which enforce the segregation of duties and the monitoring of activities, and maintain the integrity of information on which management relies upon to identify problem loans;
  - a well-defined lending approval and review system that includes established credit limits, limits and controls over the types of loans made, and limits on maturities of loans; and
  - a reporting system that provides the institution with the information needed to manage the loan portfolio including monitoring of payment and past due status and credit quality.
- **8.189** In obtaining an understanding of the entity and its environment, including its internal control, the auditor should obtain an understanding about the institution's accounting system as it relates to loans receivable, including the methods used by the institution when processing and recording new loans, applying loan payments, accruing interest, and amortizing discounts.
- **8.190** With respect to some risks, paragraph .31 of AU-C section 315 states that the auditor may judge that it is not possible or practicable to obtain sufficient appropriate audit evidence only from substantive procedures. Such risks may relate to the inaccurate or incomplete recording of routine and significant classes of transactions or account balances, the characteristics of which often permit highly automated processing with little or no manual intervention. In such cases, the entity's controls over such risks are relevant to the audit, and the auditor should obtain an understanding of them. Typical controls relating to loans include the following:
  - All loans and credit lines (including all new loans, renewals, extensions, and commitments) are approved by officers or committees in conformity with management's written lending policies and authority limits.

- An inventory of loan documents, including evidence of collateral and of the recording of liens, is monitored to ensure the timely receipt of necessary documents.
- Pertinent loan information is entered into the data-processing system on a timely basis and is independently verified to ensure accuracy.
- Subsidiary ledgers and trial balances are maintained and reconciled with the general ledger on a timely basis, differences found are investigated and resolved, and appropriate supervisory personnel review and approve completed reconciliations on a timely basis.
- Loans HFS are properly identified in the accounting records.
- Loans that were sold and are removed from the lender's balance sheet meet the requirements for removal under FASB ASC 860.
- Payments due for principal or interest are monitored for their eventual receipt, aging of delinquencies, and follow-up with late payers.
- There is segregation of duties among those who (a) approve loans, (b) control notes and collateral, (c) receive payments, (d) post subsidiary ledgers, and (e) reconcile subsidiary and general ledgers.
- Procedures are periodically performed to ensure that interest income is properly accrued and recorded.
- Notes and collateral on hand are kept in secure, locked, fireproof compartments. Negotiable collateral is kept under dual access control. Physical inventory and other processes are in place to identify losses or impairment of collateral.
- Construction loan advances are adequately documented, and periodic on-site inspections of properties are made to ensure construction progress is consistent with amounts advanced.
- **8.191** Loan files. Complete and accurate loan files are an element of internal control over financial reporting. Paragraph 8.192 details information that may be found in a loan file. The contents of the files vary, depending on the type of loan, the requirements of local law, and whether the institution intends to hold the loan or not. However, all loan files should contain a signed note. An inspection of the files supporting loans originated in prior audit periods, as well as new loans (including some of the loans still in the process of disbursement), generally permits the auditor to understand the institution's internal control in this area as a basis for planning substantive tests. It may also be useful to design dual-purpose tests in this area.
- **8.192** Following are items a loan file may contain. The specific items contained in a loan file will vary based on the size, complexity, and type of loan. The location of the contents listed will vary from one institution to another depending on the type of loan and a particular institution's policies and procedures:
  - a. Credit investigation/application/supervision section
    - i. Loan application
    - ii. Credit approval document that summarizes the following:
      - (1) Borrower

- (2) Amount of request, rate, payment terms, and fees
- (3) Purpose
- (4) Repayment sources (primary and secondary)
- (5) Collateral description and valuation
- (6) Guarantors
- (7) Other conditions and requirements of approval
- iii. Evidence of loan committee or other required approval (for example, borrower's board resolutions concerning loan approval) and date approval was granted
- iv. Recent financial statements of borrower, guarantor, or both
- v. Spreadsheets and other analyses of the financial situation of the borrower including analysis of debt service capability
- vi. Credit agency reports and other account information reports, as well as direct trade creditor references
- vii. Newspaper clippings about borrower
- viii. Various other pertinent data, including the borrower's history and forecasts
  - ix. Correspondence
  - x. Loan summary sheet, containing information such as the following:
    - (1) Lending committee approval date
    - (2) Drawdown amounts and dates
    - (3) Interest rates and adjustment dates
    - (4) Amount of undrawn commitment
    - (5) Rate of commitment fee and due dates
    - (6) Date commitment fee received
    - (7) Repayment terms
    - (8) Name of country risk
    - (9) Name and country of any guarantor
    - (10) Amount of participation fee (if applicable)
    - (11) Indication of overdue payments of interest, fees, or installments
  - xi. Memorandum to the file, by the lending officer, with description of the credit and commentary on its quality and potential future developments
- b. Loan documents section, including the following:
  - i. Signed loan agreement
  - ii. Legal opinion
  - iii. Signed note
  - iv. Signed mortgage or deed of trust, with evidence of recordation
  - v. Signed guarantee

- vi. Periodic report of collateral, including its location and value and any related environmental studies
- vii. Participation certificates and participation agreements (if applicable)
- viii. Evidence of insurance, including loss payable clauses that protect the bank's interest
  - ix. Approvals
  - x. Security agreements or other collateral pledge agreements, titles, or financing statements recorded in the proper jurisdictions to perfect lien position (nonpossessory collateral); negotiable collateral (such as stocks and bonds) with proper endorsements/assignments; hypothecation agreement for third-party pledge of collateral
  - xi. Collateral ledger used to record the instruments (including stocks and bonds, which are typically held in a vault separate from loan files or with an independent custodian) that secure a borrower's indebtedness
- **8.193** For commercial loans, the credit file usually contains the borrower's financial statements, memoranda about the borrower's financial or personal status, financial statements of guarantors (individual or corporate), internally prepared analyses of the credit, copies of supplemental agreements between the institution and the borrower, and other loan-related correspondence.
- **8.194** Files supporting either direct or indirect installment loans should include the borrower's application, discount sheet (loan computations), credit information, title or financing statement, evidence of the existence of an in-force insurance policy payable to the institution, and the note. Credit files are also maintained on dealers from whom the institution has purchased loan paper.
- **8.195** Mortgage loan files generally include the note, loan application, appraisal report, verifications of employment and assets, deed of trust, mortgage, title insurance or opinion, insurance policy, settlement statement, and VA guarantee or FHA insurance, if applicable.
- **8.196** Specific procedures the auditor may consider performing to test the operating effectiveness of controls for loans include
  - inspecting loan documents, including sales and participation agreements, to determine whether the institution's lending policies and procedures are being followed, for example, to test whether
    - loans are being approved by authorized officers or committees in accordance with the institution's lending policies:
    - credit investigations are performed;
    - credit limits are adhered to:
    - the institution's procedure to capture all required loan documents is functioning
    - the information recorded in the institution's dataprocessing system and used for management reporting

- is being tested by personnel independent of the preparer and is accurate; and
- loans sold meet the conditions in FASB ASC 860 for removal from the lender's balance sheet.
- testing the institution's reconciliation process. This testing might include the daily activity balancing process as well as the reconciliation of subsidiary ledgers with the general ledger. The auditor should test whether reconciling differences are appropriately investigated and resolved in a timely manner and whether the reconciliations are reviewed and approved by appropriate supervisory personnel.
- testing the accuracy and performing a review of delinquency reports to determine whether the institution initiates follow-up procedures on delinquent loans in accordance with its policies and whether the system identifies potentially troubled loans for purposes of assessing impairment.
- checking the accuracy and performing a review of concentration reports (such as loans to one borrower, in a particular region, or in a specific industry) and related-party loan reports.
- reviewing internal audit, loan review, and examination reports to identify control weaknesses and exceptions.
- observing or otherwise obtaining evidence that proper segregation of duties exists among those who approve, disburse, record, and reconcile loans.
- performing detailed tests of initial recording of loans, application of cash receipts, and changes in loan details (such as adjustment of rates for ARMs and maturity dates).

See paragraph 8.220 for procedures around controls identified at a service organization.

**8.197** Credit card activities. If the institution is involved in credit card operations, including credit card issuance and the processing of transactions, the auditor should consider internal control over financial reporting of credit card activities to the extent the auditor considers such internal control relevant to the audit. Audit procedures for testing financial statement assertions related to credit card activities depend on the degree of the institution's involvement in such activities. If the institution owns the customer receivables, the following may be appropriate:

- Review lending policies
- Confirm customer balances
- Test interest and service charges, collections, delinquencies, and charge-offs may be appropriate
- Test charge-off history and compare to reserve levels

If the institution only processes merchants' deposits and the resulting receivables are owned by other institutions, a review of the arrangements and a test of service fee income is generally performed.

**8.198** To the extent the institution relies on other entities for some processing activities, the auditor should refer to the guidance in AU-C section

402, Audit Considerations Relating to an Entity Using a Service Organization (AICPA, Professional Standards).

#### **Substantive Tests**

**8.199** Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards), states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure, which for a financial institution would include loans. In accordance with paragraph .445 of AU-C section 330, this requirement reflects the facts that (a) the auditor's assessment of risk is judgmental and may not identify all risks of material misstatement and (b) inherent limitations to internal control exist, including management override.

**8.200** Paragraph .06 of AU-C section 330 states that the auditor should design and perform further audit procedures whose nature, timing, and extent are based on, and are responsive to, the assessed risks of material misstatement at the relevant assertion level.

**8.201** Analytical procedures. AU-C section 520, Analytical Procedures (AICPA, Professional Standards), addresses the auditor's use of analytical procedures as substantive procedures (substantive analytical procedures). See chapter 5 of this guide for additional guidance regarding analytical procedures.

**8.202** Analytical procedures that the auditor may apply in the loan area include the analysis and evaluation of the following:

- Changes in the mix between different types of loans in the portfolio
- Comparison of the aging of past-due loans with similar aging of prior year
- Comparison of loan origination volume by month with that of prior periods
- Current-year income compared with expectations and prior-year income
- Average loan balances by type in the current year compared with those of the prior year
- Comparison of yields on loans to the institution's established lending rates or pricing policies
- Reasonableness of balance sheet accruals based upon underlying terms and amounts of corresponding loans
- Average yield throughout the period computed for each loan category on a monthly or quarterly basis

**8.203** In accordance with item a in paragraph .05 of AU-C section 520, when designing and performing analytical procedures, either alone or in combination with tests of details, as substantive procedures in accordance with AU-C section 330,  $^{20}$  the auditor should determine the suitability of particular substantive analytical procedures for given assertions, taking into account the assessed risks of material misstatement and tests of details, if any, for these

<sup>&</sup>lt;sup>20</sup> Paragraph .18 of AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards).

Loans 259

assertions. Analytical procedures may be used to help assess risk and identify areas that may require additional audit procedures. In using analytical procedures as a substantive test, the auditor might consider the implications of changes in important relationships and the extent of the difference between actual and expected results that can be accepted without further investigation. It is normally difficult to develop expectations to be used in with a sufficient level of precision analyzing yields on aggregated loans as a substantive test of related income amounts. Accordingly, analytical procedures in this area might be considered only as a supplement to other substantive procedures, except where an expected yield can be known with some precision (using computer-assisted audit techniques).

In accordance with item b in paragraph .05 of AU-C section 520, when designing and performing analytical procedures, either alone or in combination with tests of details, as substantive procedures in accordance with AU-C section 330,21 the auditor should evaluate the reliability and completeness of data from which the auditor's expectation of recorded amounts or ratios is developed, taking into account the source, comparability, and nature and relevance of information available and controls over preparation. System generated information should be tested for completeness and accuracy. Paragraph .A19 of AU-C section 520 further explains that the auditor may consider testing the operating effectiveness of controls, if any, over the entity's preparation of information used by the auditor in performing substantive analytical procedures in response to assessed risks. When such controls are effective, the auditor may have greater confidence in the reliability of the information and, therefore, in the results of analytical procedures. The operating effectiveness of controls over nonfinancial information may often be tested in conjunction with other tests of controls. Further requirements on the auditor's evaluation of data reliability can be found in paragraph .09 of AU-C section 500, Audit Evidence (AICPA, Professional Standards).

**8.205** When designing substantive analytical procedures, the auditor also might evaluate the risk of management override of controls. As part of this process, the auditor might consider evaluating whether such an override allowed adjustments outside of the normal period-end financial reporting process to have been made to the financial statements. Such adjustments might have resulted in artificial changes to the financial statement relationships being analyzed, causing the auditor to draw erroneous conclusions. For this reason, substantive analytical procedures alone are not well suited to detecting fraud. In addition, before using results obtained from substantive analytical procedures, the auditor could either test the design and operating effectiveness of controls over financial information used in the substantive analytical procedures or perform other procedures to support the completeness and accuracy of the underlying information.

**8.206** For significant risks of material misstatement, it is unlikely that audit evidence obtained from substantive analytical procedures alone will be sufficient.

**8.207** Subsidiary records. In accordance with paragraph .21 of AU-C section 330, the auditor's substantive procedures should include audit procedures relating to the financial statement closing process, such as agreeing or reconciling financial statement balances, including loan principal balances and related

<sup>&</sup>lt;sup>21</sup> See footnote 20.

accounts (accrued interest receivable, unearned discount, and net deferred loan fees and costs), with the underlying accounting records (trial balance, general ledger, and other subsidiary records). The auditor should test significant reconciling items.

- **8.208** Confirmation. Guidance on the extent and timing of audit procedures (that is, considerations involved in determining the number of items to confirm) is found in AU-C section 530, Audit Sampling (AICPA, Professional Standards), <sup>22</sup> and AU-C section 320, Materiality in Planning and Performing an Audit (AICPA, Professional Standards). Readers may also refer to the AICPA Audit Guide Audit Sampling that provides guidance to help auditors apply audit sampling in accordance with AU-C section 530. Guidance on the timing of audit procedures is included in AU-C section 330.<sup>23</sup>
- **8.209** AU-C section 505, *External Confirmations* (AICPA, *Professional Standards*), addresses the auditor's use of external confirmation procedures to obtain audit evidence, in accordance with the requirements of AU-C sections 320 and 500.
- **8.210** When using external confirmation procedures, item (c) in paragraph .07 of AU-C section 505 states that the auditor should maintain control over external confirmation requests, including designing the confirmation requests, determining that requests are properly directed to the appropriate confirming party and providing for responses being directly to the auditor. Paragraphs .A4-.A5 of AU-C section 505 further explain that the design of a confirmation request may directly affect the confirmation response rate and the reliability and nature of the audit evidence obtained from responses. In addition, a factor to consider when designing confirmation requests is the ability of the intended confirming party to confirm or provide the requested information. For example, respondents may not be able to confirm the balances of installment loans, but they may be able to confirm whether their payments are up to date, the amounts of the payments, the interest rate, and the term of their loans.
- **8.211** Auditors may use either positive or negative requests to confirm loans. However, paragraph .15 of AU-C section 505 states that negative confirmations provide less persuasive audit evidence than positive confirmations. Accordingly, the auditor should not use negative confirmation requests as the sole substantive audit procedure to address an assessed risk of material misstatement at the assertion level, unless all of the following are present:
  - a. The auditor has assessed the risk of material misstatement as low and has obtained sufficient appropriate audit evidence regarding the operating effectiveness of controls relevant to the assertion.
  - b. The population of items subject to negative confirmation procedures comprises a large number of small, homogeneous account balances, transactions, or conditions.
  - c. A very low exception rate is expected.
  - d. The auditor is not aware of circumstances or conditions that would cause recipients of negative confirmation requests to disregard such requests.

 $<sup>^{22}</sup>$  See footnote 11.

<sup>&</sup>lt;sup>23</sup> See footnote 11.

Loans 261

Positive confirmation procedures might be appropriate for larger loans and for loans that necessitate additional assurance or other related information in addition to the loan balance, such as amount and type of collateral.

- **8.212** Inspecting loan documents. Loan files vary considerably in content depending on the type of loan. Inspection of loan documents may provide evidence about the existence and ownership of the loan. It is important for the auditor to be alert when inspecting loan documents. Indicators such as notations could imply problems that merit further investigation or follow up. When loan documents are in the possession of an attorney or other outside parties, the auditor should consider confirming the existence and ownership of such documents.
- **8.213** When inspecting loan documents, the auditor may test the physical existence and read any evidence of assignment to the institution of the collateral that supports collateralized loans. For certain loans, the auditor might inspect collateral in the custody of the borrower, such as floor-plan merchandise. However, the auditor may conclude that a review of the reports of institution personnel who inspect collateral is sufficient audit evidence. The auditor may also consider examining or requesting confirmation of collateral not on hand. An inspection of loan documentation could include tests of the adequacy of both the current value of collateral in relation to the outstanding loan balance and, if needed, insurance coverage on the loan collateral.
- **8.214** While inspecting loan documents, the auditor should keep in mind the audit objectives discussed in chapter 9 of this guide. For example, reading the financial statements and other evidence of the financial condition of cosignatories and guarantors could be employed when the auditor tests guaranteed loans. Consideration might also be given to the institution's historical experience with enforcing guarantees or confirmation of terms with guarantor.
- **8.215** While inspecting loan documents, the auditor might look for evidence of approvals by the board of directors or loan committee as required by management's written lending policies, a comparison of loan amounts with appraisals, and an inspection of whether hazard and title coverage meets coverage requirements set in management's written policy. For loans generated under certain governmental programs and other special arrangements, the auditor may be engaged to perform the additional procedures required under the specific trust or servicing agreement.
- **8.216** Construction loans. Audit procedures should be responsive to the assessed risk related to the institution's construction lending practices. For example, the auditor might perform tests to determine whether construction loans are properly classified as loans rather than real estate investments. The auditor might test origination, approval, inspection, and disbursements made based on progress on the particular construction project. The auditor might perform on-site inspections of significant construction projects to review the collateral and to determine whether construction has progressed in accordance with the loan terms.
- **8.217** Lease financing. When confirming basic lease terms, the confirmation requests typically would include cancellation provisions, if any. Confirmation should ordinarily be requested from the lessee. For leveraged leases, the material aspects of the lease agreement, including information necessary for income tax purposes, may be requested from the lease trustee. Although alternative methods may be used for reporting income for tax purposes, the

auditor should determine that income for book purposes is being recorded in accordance with FASB ASC  $840.^{24}$ 

**8.218** Whole loans or participations purchased. Audit procedures for purchased loans should be similar to those for direct loans, except that requests for the confirmation of balances, collateral, and recourse provisions, if any, are usually sent to the originating or servicing institution. Loan files for purchased participations should be available at the institution and contain pertinent documents, or copies of them, including credit files supporting loans in which the institution has purchased participations from other banks or savings institutions. The auditor should consider confirming the actual status of borrower payments with the servicer. Although it is usually not practicable to confirm balances of serviced loans with the individual borrowers, the servicer's auditors often perform audit procedures on individual loans, such as confirmation with borrowers and examination of loan documents. AU-C section 402 addresses the user auditor's responsibility for obtaining sufficient appropriate audit evidence in an audit of the financial statements of a user entity that uses one or more service organizations. Statement on Standards for Attestation Engagements No. 16, Reporting on Controls at a Service Organization (AICPA, Professional Standards, AT sec. 801), 25 addresses examination engagements undertaken by a service auditor to report on controls at organizations that provide services to user entities when those controls are likely to be relevant to user entities' internal control over financial reporting. It complements AU-C section 402 in that reports prepared in accordance with this section may provide appropriate evidence under AU-C section 402.

**8.219** Paragraph .A24 of AU-C section 402 states that a type 1 or type 2 report (as defined in AU-C section 402) may assist the user auditor in obtaining a sufficient understanding to identify and assess the risks of material misstatement of the user entity's financial statements. A type 1 report, however, does not provide any evidence of the operating effectiveness of the relevant controls.

**8.220** When the user auditor's risk assessment includes an expectation that controls at the service organization are operating effectively, paragraph .16 of AU-C section 402 states that the user auditor should obtain audit evidence about the operating effectiveness of those controls from one or more of the following procedures:

- a. Obtaining and reading a type 2 report, if available
- b. Performing appropriate tests of controls at the service organization
- c. Using another auditor to perform tests of controls at the service organization on behalf of the user auditor

If the user auditor believes that the service auditor's report may not provide sufficient appropriate audit evidence (for example, if a service auditor's report does not contain a description of the service auditor's tests of controls and results thereof), paragraph .A38 of AU-C section 402 further explains that the

<sup>&</sup>lt;sup>24</sup> See paragraph 8.140.

<sup>&</sup>lt;sup>25</sup> The AICPA Guide Service Organizations: Reporting on Controls at a Service Organization Relevant to User Entities' Internal Control Over Financial Reporting contains information for practitioners reporting on controls at a service organization that affect user entities' internal control over financial reporting. Also, the AICPA Guide Reporting on Controls at a Service Organization Relevant to Security, Availability, Processing Integrity, Confidentiality, or Privacy (SOC 2®) summarizes the three SOC engagements and provides detailed guidance on planning, performing, and reporting on SOC 2 engagements.

user auditor may supplement his or her understanding of the service auditor's procedures and conclusions by contacting the service organization through the user entity to request a discussion with the service auditor about the scope and results of the service auditor's work. Also, if the user auditor believes it is necessary, the user auditor may contact the service organization through the user entity to request that the service auditor perform procedures at the service organization, or the user auditor may perform such procedures. Further guidance for a user auditor can be found in AU-C section 402.

- **8.221** Accrued interest receivable and interest income. Provided that a basis exists to rely on loan data recorded in the loan accounting system, interest income may be tested using computer-assisted audit techniques, the recomputation of accrued amounts for individual accounts, analytical procedures, or some combination thereof. If interest rates were relatively stable during a period and the amounts can be predicted with a sufficient level of precision, interest income can often be tested effectively by using analytical tests by type of loan. The auditor should consider average balances in principal accounts, related yields as compared to averages of rates offered and of rates on existing loans, and other factors and relationships. As discussed in paragraphs 8.201–.206, the effectiveness of such analytical procedures may vary.
- **8.222** Computer assisted audit techniques. Computer-assisted audit techniques may also be used to perform "exception/limit" checks of individual files for unusual or questionable items meriting further investigation. Examples include identifying unusual interest rates, balances, and payments, or testing the accuracy of delinquency reports.
- **8.223** Balance sheet classification of loans. Depending on the auditor's assessment of the risks of material misstatement, the auditor should consider whether any portion of loans is being HFS and, therefore, whether a corresponding valuation allowance or write-down to lower of cost or market value is necessary. Previous loan sale activity, types of loans sold, transactions subsequent to year-end, pending contracts, and management's intentions are factors that should be considered in identifying loans HFS.
- **8.224** Loan fees and costs. Depending on the auditor's assessment of risks of material misstatement, the auditor should review and test the propriety of the institution's deferral of loan origination fees and costs in accordance with FASB ASC 310-20, as well as evaluating the impact of not deferring loan costs and fees. The auditor should also consider performing a test of the amortization of net deferred loan fees or costs.
- **8.225** Undisbursed portion of mortgage loans. Financial institutions sometimes record loans at the gross amount with an offsetting account entitled loans in process (LIP). As funds are disbursed, the LIP account is reduced. Interest or fees on construction loans also may be debited to this account. The LIP account should be cleared when the loan is fully disbursed. LIP detailed ledgers should be reviewed to determine the propriety of accounting, including that for complex interest calculations. Unusual LIP balances, such as debit balances or balances outstanding for an excessive period of time (for example, over a year), may be indicative of problem loans.
- **8.226** A review of the LIP detailed activity may be performed in connection with the examination of the current-year loan files. Loans selected for testing may be traced to the LIP account. Construction loans selected for testing may be traced to the LIP ledger, and disbursements may be reviewed in connection

with the percentage of completion noted on inspection reports. In addition, if loan fees or interest are being capitalized (added to the loan balance) during construction, a review of the LIP ledgers may point out areas of concern. Based on the auditor's assessment of the risks of material misstatement, the auditor should consider whether to send confirmations to the borrower on any undisbursed loan balances.

**8.227** *TDRs*. The auditor should consider performing procedures to determine whether management has appropriately identified all TDRs, whether the accrual status is appropriate, whether they have been accounted for in conformity with FASB ASC 310-40, and whether management has appropriately measured impairment for TDRs under FASB ASC 310-10. Such tests may include procedures to determine whether possession of collateral has been taken as part of a TDR that is in substance a repossession or foreclosure by the creditor, that is, the creditor receives physical possession of the debtor's assets regardless of whether formal foreclosure proceedings take place (as discussed in FASB ASC 310-40-40-6 [see paragraph 8.137]). Auditors should also consider whether the entities have appropriate tracking and reporting processes in place to address disclosure requirements applicable to TDRs.

**8.228** In addition, auditors should consider reviewing substandard or watch-listed loans that have been renewed at terms similar to the original loan because these loans may involve borrowers that are experiencing some level of financial difficulty and, because of the deterioration in the loan's credit quality, may not otherwise qualify for the terms as offered in the renewal agreement. In these instances, the institution may have granted a concession because the interest rate for such a renewal is not indicative of a market rate, and, therefore, the renewal under such terms is a strong indicator that the loan should be accounted for as a TDR. In such cases, auditors should consider whether the institutions have appropriately documented their conclusions regarding TDR status and appropriately accounted for renewals of this nature. When the practical expedient for collateral dependent loans is not elected, the auditor may also want to review the assumptions of projected cash flows utilized in impairment measurements to determine the reasonableness of the estimates because this will drive the allocated allowance for such loans.

**8.229** Valuation. For loans for which there is a market price, the auditor may test fair-value disclosures by reference to third-party market quotations, including information received from brokers or dealers in loans. Fair-value estimates of loans for which there is no market price are highly subjective. There are a variety of methodologies that may be used by institutions to estimate fair values of loans. Most derive a fair value by discounting expected cash flows using appropriate interest rates. Some methodologies are relatively simple, such as methods that derive much of their data from the information used in estimating the allowance for credit losses, and some are relatively complex, such as option pricing models. In accordance with paragraph .06 of AU-C section 540, Auditing Accounting Estimates, Including Fair Value Accounting Estimates, and Related Disclosures (AICPA, Professional Standards), the objective of the auditor is to obtain sufficient appropriate audit evidence about whether, in the context of the applicable financial reporting framework, accounting estimates including fair value accounting estimates in the financial statements, whether recognized or disclosed, are reasonable and related disclosures in the financial statements are adequate. AU-C section 540 provides relevant guidance. AU-C section 620, Using the Work of an Auditor's Specialist (AICPA, Professional

Loans 265

Standards), addresses the auditor's responsibilities relating to the work of an individual or organization possessing expertise in a field other than accounting or auditing when that work is used to assist the auditor in obtaining sufficient appropriate audit evidence. AU-C section 620 does not address the auditor's use of the work of an individual or organization possessing expertise in a field other than accounting or auditing, whose work in that field is used by the entity to assist the entity in preparing the financial statements (a management's specialist), which is addressed in AU-C section 500.

#### 8.230

Considerations for Audits Performed in Accordance With PCAOB Standards<sup>26</sup>

PCAOB Staff Audit Practice Alert No. 2, Matters Related to Auditing Fair Value Measurements of Financial Instruments and the Use of Specialists (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance, sec. 400.02), provides guidance on auditors' responsibilities for auditing fair value measurements of financial instruments and when using the work of specialists under the existing standards of the PCAOB. This alert is focused on specific matters that are likely to increase audit risk related to the fair value of financial instruments in a rapidly changing economic environment. This practice alert highlights certain requirements in the auditing standards related to fair value measurements and disclosures in the financial statements and certain aspects of GAAP that are particularly relevant to the current economic environment.

PCAOB Staff Audit Practice Alert No. 4, Auditor Considerations Regarding Fair Value Measurements, Disclosures, and Other-Than-Temporary Impairments (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance, sec. 400.04), informs auditors about potential implications of recently issued FASB guidance on reviews of interim financial information and annual audits. This alert addresses the following topics: (a) reviews of interim financial information; (b) audits of financial statements, including integrated audits; (c) disclosures; and (d) auditor reporting considerations.

PCAOB Staff Audit Practice Alert No. 7, Auditor Considerations of Litigation and Other Contingencies Arising From Mortgage and Other Loan Activities (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance, sec. 400.07), advises auditors that the potential risks and costs associated with mortgage and foreclosure-related activities or exposures, such as those discussed in the SEC staff letters, could have implications for audits of financial statements or of internal control over financial reporting. These implications might include accounting for litigation or other loss contingencies and the related disclosures.

<sup>&</sup>lt;sup>26</sup> PCAOB Staff Audit Practice Alerts are not rules of the board and do not reflect any board determination or judgment about the conduct of any particular firm, auditor, or any other person.

# Chapter 9

# **Credit Losses**

#### © Update 9-1 Accounting and Reporting: Credit Losses

FASB Accounting Standards Update (ASU) No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, issued in June 2016, is effective for fiscal years of public business entities that are SEC filers beginning after December 15, 2019, including interim periods within those fiscal years.

For all other public business entities, the amendments in FASB ASU No. 2016-13 are effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years.

For all other entities, including not-for-profit entities and employee benefit plans within the scope of FASB Accounting Standards Codification (ASC) 960, Plan Accounting—Defined Benefit Pension Plans through FASB ASC 965, Plan Accounting—Health and Welfare Benefit Plans on plan accounting, FASB ASU No. 2016-13 is effective for fiscal years beginning after December 15, 2020, and interim periods within fiscal years beginning after December 15, 2021.

Early application is permitted for all entities as of the fiscal years beginning after December 15, 2018, *including* interim periods within those fiscal years.

FASB ASU No. 2016-13 creates FASB ASC 326, Financial Instruments—Credit Losses, to amend guidance on reporting credit losses for assets held at amortized cost basis and available-for-sale debt securities.

For assets held at amortized cost basis, FASB ASC 326 eliminates the probable initial recognition threshold in current U.S. generally accepted accounting principles (GAAP) and, instead, requires an entity to reflect its current estimate of all expected credit losses. The allowance for credit losses is a valuation account that is deducted from the amortized cost basis of the financial assets to present the net amount expected to be collected.

For available-for-sale debt securities, credit losses should be measured in a manner similar to current GAAP. However, FASB ASC 326 will require that credit losses be presented as an allowance rather than as a write-down.

FASB ASU No. 2016-13 affects entities holding financial assets and net investment in leases that are not accounted for at fair value through net income. The amendments affect loans, debt securities, trade receivables, net investments in leases, off-balance-sheet credit exposures, reinsurance receivables, and any other financial assets not excluded from the scope that have the contractual right to receive cash.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this chapter will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-13, see appendix G, "Accounting for Financial Instruments," of this guide.

©2016, AICPA AAG-DEP 9

#### Introduction

- **9.01** Financial institutions accept and manage significant amounts of credit risk. Declines in the value of underlying collateral have traditionally been the source of most credit losses incurred by financial institutions. The allowance for loan losses is an accounting estimate of credit losses inherent in an institution's loan portfolio that are incurred as of the balance sheet date. Chapter 8, "Loans," of this guide discusses the various types of loans institutions make or purchase, the lending process and related controls, financial reporting for loans, and the audit procedures for loans.
- **9.02** Institutions may also have off-balance-sheet financial instruments, such as commitments to extend credit, guarantees, and standby letters of credit that are subject to credit risk. Although liabilities related to credit losses associated with such off-balance-sheet financial instruments are not part of the allowance for loan losses, institutions' processes for evaluation and estimation of credit losses may include consideration of credit risk associated with those off-balance-sheet financial instruments, especially when the counterparty to an off-balance-sheet financial instrument is also a borrower. The information and guidance in this chapter, although generally referring to *loan* losses, may equally be useful in evaluating and estimating credit losses for off-balance-sheet financial instruments.
- **9.03** Careful planning, risk assessment and execution of audit procedures related to credit losses is essential due to the significance of the allowance for loan losses and related provision to institutions' financial statements and to the high degree of subjectivity involved in estimating these amounts, the high degree of regulatory guidance and oversight directed toward institutions' estimates of credit losses, and, consequently, the relatively high inherent audit risk associated with auditing such estimates.

# Management's Methodology

- **9.04** Management is responsible for estimating credit losses using a methodology that results in an estimate that is in accordance with GAAP. Management has a responsibility for developing, maintaining, and documenting a comprehensive, systematic, and consistently applied process for determining the allowance for loan losses. To fulfill that responsibility, management should ensure controls are in place to consistently estimate the allowance for loan losses using a methodology that results in an estimate that is in accordance with GAAP, and that complies with the institution's stated policies and procedures. Estimating credit losses is unavoidably subjective and involves management making careful judgments about collectibility and estimates of losses. Management's judgments often depend on micro- and macro-economic factors; past, current, and anticipated events based on conditions that existed at the balance sheet date; and realistic courses of action that management expects to take.
- **9.05** An institution's methodology for estimating credit losses should be well documented, with clear explanations of the supporting analyses and rationale. An institution's methodology for estimating credit losses is influenced by many factors, including the institution's size, organizational structure, business environment and strategy, management style, loan portfolio characteristics, loan administration procedures, and management information systems. Although different institutions may use different methodologies, there are

certain common elements included in any effective methodology. They include, subject to materiality considerations,

- a detailed and regular analysis of the loan portfolio and off-balancesheet financial instruments with credit risk:
- b. procedures for timely identification of problem credits;
- c. consistent use and application;
- d. consideration of all known relevant internal and external factors that may affect collectibility;
- e. consideration of all loans (whether on an individual or pool-of-loans basis) and other relevant credit exposures;
- f. consideration of the particular risks inherent in the different kinds of lending;
- g. consideration of the current collateral fair values less cost to sell, where applicable;
- performance by competent and well-trained personnel with oversight by appropriate levels of management, and, when appropriate, board committees;
- i. current, relevant, and reliable data;
- j. good documentation with clear explanations of the supporting analyses and rationale;
- k. identification of all subsequent events that provide additional evidence about conditions that existed at the balance sheet date;
- segregation of loans purchased or acquired such that an appropriate allowance for loan losses is recorded for deterioration in cash flows on loans scoped into FASB ASC 310-30 (purchased credit impaired);
- m. segregation of troubled loans that have been modified such that an appropriate allowance for loan losses is recorded for loans scoped into FASB ASC 310-40 (troubled debt restructurings by creditors); and
- consideration of the particular risks in the organization structure, underwriting or loan administration practices, where applicable.
- **9.06** Allowance methodologies that rely solely on mathematical calculations, such as a percentage of total loans based on historical experience or the similar allowance percentages of peer institutions, generally fail to contain all of the essential elements of an effective methodology because they do not involve a detailed analysis of an institution's particular transactions or consider the current economic environment. The allowance documentation should include a conclusion as well as the basis for conclusion on the appropriateness and reasonableness of the resulting point estimate.
- **9.07** The components of the loan portfolio and their related allowance for loan losses might have different risks, for instance if those components relate to different credit exposures, and the components of the allowance for loan losses are determined using different methodologies, or are subject to different accounting requirements. As discussed in the following paragraph, financial institutions have traditionally identified loans that are to be evaluated for collectibility by dividing the loan portfolio into different segments. Loans with similar risk characteristics should be grouped and evaluated together. The

defining risk characteristics of each segment may include different combinations of such factors as risk classification or rating, past-due status, type of loan, collateral, vintage, geographic regions, FICO scores, loan-to-value ratios, and lines of business. Appropriate segmentation provides for a more accurate assessment of the estimated loss in the portfolio by differentiating loss rates based on common risk factors.

**9.08** A key element of most methodologies for estimating credit losses related to commercial loans is a risk classification process that involves categorizing loans into risk categories or ratings. The categorization should be based on relevant information about the ability of borrowers to service the debt, such as current financial information, historical payment experience, credit documentation, public information, current trends, and the value of collateral, including its ability to generate cash flows. Many institutions classify loans using a risk rating system that incorporates as its most serious categories the supervisory classification system as follows:<sup>1</sup>

**substandard.** Assets classified as substandard are inadequately protected by the current sound worth and paying capacity of the obligor or of the collateral pledged, if any. Assets so classified must have a well-defined weakness or weaknesses that jeopardize the liquidation of the debt. They are characterized by the distinct possibility that the institution will sustain some loss if the deficiencies are not corrected.

**doubtful.** Assets classified as doubtful have all the weaknesses inherent in those classified as substandard, with the added characteristic that the weaknesses make collection or liquidation in full, on the basis of currently existing facts, conditions, and values, highly questionable and improbable. Doubtful is a transitional asset classification category; loans classified doubtful typically have a pending short term event that will provide evidence of the confirmed loss amount.

**loss.** Assets classified as loss are considered uncollectible and of such little value that their continuance as bankable assets is not warranted. This classification does not mean that the asset has absolutely no recovery or salvage value but, rather, that it is not practical or desirable to defer writing off this basically worthless asset even though partial recovery may be affected in the future.

**9.09** Some loans are listed as special mention.<sup>2</sup> Such loans have potential weaknesses that deserve management's close attention. If left uncorrected, these potential weaknesses may result in deterioration of the repayment prospects for the asset or of the institution's credit position at some future date. Special-mention loans are not adversely classified and do not expose an institution to sufficient risk to warrant adverse classification. However, these loans do generally represent an elevated collection risk for the institution and the method of estimating credit losses should consider the elevated collection risk accordingly.

<sup>&</sup>lt;sup>1</sup> Uniform Agreement on the Classification and Appraisal of Securities Held by Depository Institutions, October 29, 2013.

<sup>&</sup>lt;sup>2</sup> Interagency Statement on the Supervisory Definition of Special Mention Assets, June 10, 1993.

#### 9.10 Examples of potential weaknesses are

- poor underwriting or loan administration practices that result in significant defects in the loan agreement, security agreement, guarantee agreement, or other documentation and the deteriorating condition of or lack of control over collateral. In other words, these are conditions that may jeopardize the institution's ability to enforce loan terms or that reduce the protection afforded by secondary repayment sources.
- lack of information (if relevant) about the borrower or guarantors, including stale financial information or lack of current collateral valuations.
- economic or market conditions that in the future may affect the borrower's ability to meet scheduled repayments. These may be evidenced by adverse profitability, liquidity, or leverage trends in the borrower's financial statements.
- **9.11** The supervisory ratings special mention, substandard, doubtful, and loss identify different degrees of credit weakness. Credits that are not covered by these definitions are "pass" credits, for which no formal supervisory definition exists (for example, supervisory ratings do not distinguish among loans within the pass category). However, more precise monitoring of credit risk allows for better allowance estimates and enhances early warning and portfolio management. It is difficult to manage risk prospectively without some stratification of the pass ratings. The number of pass ratings a bank will find useful depends on the complexity of the portfolio and the objectives of the risk rating system. Less complex, community banks may find that a few pass ratings are sufficient to differentiate the risk among their pass-rated credits. Larger, more complex institutions will generally require the use of a larger number of pass ratings to achieve their risk identification and portfolio management objectives.
- **9.12** Although the Board of Governors of the Federal Reserve System (Federal Reserve), the FDIC, and the Office of the Comptroller of the Currency (OCC) do not require institutions to adopt identical classification definitions, institutions should classify their assets using a system that can be easily reconciled with the supervisory classification system.

#### Loan Reviews

**9.13** Loan review function. An effective loan review function should include internal controls to promptly identify loans with potential credit weaknesses and trends that affect the collectibility of the portfolio. Loan credit analyses focus on determining whether the loans are properly classified or rated according to their relative credit risk and were made in accordance with the institution's written lending policies and whether the borrower is likely to perform in accordance with the contractual terms and conditions of the loan. The review typically includes analysis of (a) loan performance since origination or the last renewal, (b) the current economic situation of a borrower or guarantor, and (c) estimates of current fair values of collateral. Borrower and guarantor financial statements are generally reviewed for weaknesses concerning financial resources, liquidity, future cash flows, and other financial information pertinent to the ability to repay the debt. Collateral is reviewed to determine whether it is under the institution's control, whether security interests have been perfected (which is a legal determination), and whether the

fair value less costs to sell is greater than the amount owed. Loan file contents are generally reviewed for completeness and conformity with the institution's written policies for loan documentation. The nature of the loan review system may vary based on an institution's size, complexity, loan types and management practices. Regardless of the structure, the lack of an internal loan review function and classification system may be considered to be an unsafe and unsound practice by regulators. The absence of an internal loan review function may also be an indicator of a significant deficiency or a material weakness, as defined in paragraph .07 of AU-C section 265, Communicating Internal Control Related Matters Identified in an Audit (AICPA, Professional Standards).

- **9.14** Foreign loans should be reviewed and require special consideration because of the transfer risk associated with cross-border lending. Transfer risk is the possibility that an asset cannot be serviced in the currency of payment because of a lack of, or restraint on the availability of, needed foreign exchange in the country of the obligor. Certain foreign loans are required by the Interagency Country Exposure Risk Committee (ICERC) pursuant to the International Supervision Act of 1983 to have allocated transfer risk reserves (ATRRs). ATRRs are minimum specific reserves related to loans in particular countries. Such reserves are minimums, and institutions may determine that a higher allowance for loan losses is necessary based on the assessment of probable losses in accordance with GAAP.
- **9.15** Loan evaluations performed by management for the purpose of risk rating and measurement of impairment (and tests of such by auditors to the extent they are performed as part of the audit engagement) should avoid the following:
  - Inadequate appraisals of collateral. This is the failure to critically review appraisals to understand the methods employed, assumptions made, and limitations inherent in the appraisal process, including undue reliance on management appraisals. Appraisal methods and assumptions may be inappropriate in the current circumstances. Going concern values generally are dramatically different from liquidation values. For example, real estate appraisals made on the income approach are not usually appropriate for incomplete projects or in circumstances in which operating conditions have changed.
  - Outdated or unreliable financial information. This is the reliance on old, incomplete, or inconsistent data to assess operating performance or financial capacity. Financial information should be current and complete, particularly for borrowers or collateral sensitive to cyclical fluctuations or who demonstrate significant growth or changes in operating philosophy and markets. Collateral values and liquidity often tend to decline in periods during which they are most needed to protect against loan losses. For example, if an oversupply in the real estate market causes lower-thanprojected occupancy rates (creating cash flow problems for the borrower), the value of the property will likely decline, diminishing the protection afforded by the collateral. Similar scenarios can be drawn for oil and gas reserves when energy prices decline, for specialized equipment (for example, drilling rigs, mining equipment, farm equipment, steel mills, and construction equipment) during specific industry slowdowns, for farmland during periods

- of depressed agricultural commodity and livestock prices, and for accounts receivable of a failing company.
- Excessive renewals or unrealistic terms. This is the reliance on current or performing-as-agreed status if the transaction has been structured to obscure weaknesses. Excessive renewals, unrealistic terms, and interest capitalization may be indications of such a structure. The purpose of a loan and performance against the original agreement should be critically reviewed.
- Personal bias. This is the bias of a reviewer for or against industries, companies, individuals, and products. For example, the involvement of a public personality in a venture could influence a reviewer to place more credibility than appropriate on the success of the venture.
- Overlooking self-dealing. This concerns directors or large share-holders of the institution who improperly use their position to obtain excessive extensions of credit on an unsound basis. In this situation, management is often unduly influenced by persons in these positions because management serves at the pleasure of the board and shareholders.
- Dependence on management representations. This is undue reliance on management and loan officer representations even though there is no supporting evidence. For example, such representations as "the guarantee is not signed but it is still good" or "the future prospects for this troubled borrower are promising" necessitate a critical review.

# Loans Individually Evaluated for Impairment

9.16 After segmenting the loan portfolio into groups of loans with similar risk characteristics, loans that are not recognized as being impaired in the individual loan impairment analyses, based on the definition, normally would be evaluated and measured for impairment on a collective basis. Institutions will further identify certain loans to be assessed for individual loan impairment. Loans that are generally assessed for individual impairment include large loans, and loans with especially heightened risk profiles, for example, certain risk categories. The individual loan impairment analyses may be performed by loan officers subject to review by an internal loan review function or may be performed by a credit risk management function. Some institutions outsource the loan review function to an independent third party, although the loan review function always remains a responsibility of management. An individual loan is deemed to be impaired when it is probable that the creditor will be unable to collect all the contractual interest and principal payments as scheduled in the loan agreement. Loans that are recognized as being impaired cannot be further evaluated and measured for impairment on a collective basis, regardless of whether such loans result in an impairment loss or not. For this reason it is important to distinguish between an assessment that concludes a loan is not impaired, and a loan that is impaired but that does not need any allowance or write down.

# **Loans Collectively Evaluated for Impairment**

**9.17** Loans not evaluated for impairment individually are included in groups (or pools) of homogeneous loans and evaluated for impairment on a

collective basis. The focus of the pool approach is generally on the historical loss experience for the pool. Loss experience, which is usually determined by reviewing the historical loss (charge-off) rate for each pool over a designated time period, is adjusted for changes in current trends and conditions. The time period used for determining the historical loss rate is often referred to as the look-back period. The look-back period should be reasonable, supported, and consistently applied. However, the look-back period should not be static; rather, it should be revised (either shortened or lengthened) as changes occur in the portfolio and its credit quality. Generally, management should determine the historical loss rate for each group of loans with similar risk characteristics based on the institution's own loss experience for loans in that group. Methods for calculating loss rates include average historical net charge-off rates, migration analysis (including roll rate analysis), and loss estimation models. The method used by an institution will depend on considerations such as the size of the institution and the nature, scope, and risk of its lending activities. Although historical loss experience provides a reasonable starting point for the analysis of loss rates, historical losses (or even recent trends in losses) do not by themselves form a sufficient basis to estimate the appropriate level of the allowance for loan losses. Management should also consider those qualitative or environmental factors that are likely to cause estimated credit losses associated with the institution's existing portfolio to differ from historical loss experience. Qualitative or environmental factors can include, but are not limited to,

- changes in the volume and severity of past due loans, the volume of nonaccrual loans, and the volume and severity of adversely classified or graded loans;
- changes in the nature and volume of the portfolio and in the terms of loans:
- changes in lending policies and procedures, including changes in underwriting standards and collection, charge-off and recovery practices not considered elsewhere in estimating credit losses;
- changes in the experience, ability, and depth of lending management and other relevant staff;
- changes in international, national, regional, and local economic and business conditions and developments that affect the collectibility of the portfolio, including the condition of various market segments;
- the existence and effect of any concentrations of credit risk and changes in the level of such concentrations;
- changes in the quality of the institution's loan review system;
- changes in the value of underlying collateral for collateraldependent loans; and
- the effect of other external factors, such as competition and legal and regulatory requirements on the level of estimated credit losses in the institution's existing portfolio.

**9.18** Qualitative adjustments may address limitations of the quantitative analysis of the allowance for loan losses based on historical loss experience. When qualitative adjustments are used, they should be supported by reliable evidence. The quantitative and qualitative analyses should interact with each other to ensure that the allowance for loan losses is appropriate and reflects credit losses incurred as of the balance sheet date.

# **Estimating Overall Credit Losses**

- **9.19** Careful judgment must be applied in assessing the risks as well as other relevant factors for each segment of loans to estimate the allowance amount to be recorded. FASB ASC 310, *Receivables*, contains different guidance for addressing groups of loans or individual loans. In addition, paragraphs 8–9 of FASB ASC 310-10-35 point out that FASB ASC 450-20 guidance on when a loss is recognized applies to both individual receivables and groups of receivables. Further, FASB ASC 310-10-35 addresses interaction with FASB ASC 450-20. The approach for determination of the allowance should be well documented and applied consistently from period to period, as stated in item (c) in FASB ASC 310-10-35-4.
- **9.20** In addition to the standards outlined above, SEC Staff Accounting Bulletin (SAB) No. 102, Selected Loan Loss Allowance Methodology and Documentation Issues, and the 2006 Federal Financial Institutions Examination Council (FFIEC) Interagency Policy Statement on the Allowance for Loan and Lease Losses provides further guidance regarding documentation requirements related to the allowance for loan losses.
- **9.21** Management often considers credit losses associated with certain off-balance-sheet financial instruments (such as commitments to extend credit, guarantees, and letters of credit) at the same time it considers credit losses associated with the loan portfolio. Although it is generally practical to consider credit losses on loans and other financial instruments at the same time, allowances necessary for off-balance-sheet financial instruments should be reported separately as liabilities and not as part of the allowance for loan losses (see paragraph 9.57).
- **9.22** Management should consider its overall allowance for loan losses and the liability for other credit exposures to be appropriate in accordance with GAAP only if such amounts are considered appropriate to cover estimated losses inherent in the loan portfolio and the portfolio of other financial instruments, respectively. An illustration of a worksheet for an allowance and liability calculation is shown in exhibit 9-1, "Worksheet for Estimating Credit Losses."

# Exhibit 9-1 Worksheet for Estimating Credit Losses

•	$Recorded \ Investment^{\dagger}$	$Estimated\ Credit\ Loss\ Amount^*$	
Category		High	Low
	\$	\$	\$
Allowance for Loan Losses			
I. Individually evaluated for impairment: $^{\ddagger}$			
Impairment identified $^{\parallel}$			
No impairment identified		N/A	N/A
II. Large groups of smaller-balance homogeneous loans collectively evaluated for impairment: #			
Credit card			
			(continued)

	Recorded.	Estimated Credit Loss Amount*	
Category	$Investment^{\dagger}$	High	Low
Residential mortgage			
Consumer			
Other			
III. Other large groups of loans containing unidentified, impaired loans **			
IV. Loans measured at fair value or at the lower of cost or fair value <sup>(2)</sup>		<u>N/A</u>	<u>N/A</u>
Total allowance for loan losses		\$	\$
Liability for Losses on Credit Instruments and Other Credit Exposures			
Standby letters of credit (1)			
Commitments (1)			
Loans sold with recourse			
Losses on guarantees (FASB Accounting Standards Codification [ASC] 460, Guarantees)			
Other			
Total liability for credit instruments and other credit exposures		\$	\$

<sup>\*</sup> For purposes of this worksheet, the estimated credit loss amount may be a specific amount or a range of estimated amounts. The measure of impairment under FASB Accounting Standards Codification (ASC) 310-10-35-26, is the creditor's best estimate based on reasonable and supportable assumptions and projections.

# Regulatory Matters

**9.23** The Federal Reserve, the FDIC, the OCC (collectively, the federal banking agencies), and the Office of Thrift Supervision's (prior to its transfer

 $<sup>^\</sup>dagger$  The total of amounts in this column generally should correspond to the institution's total loan portfolio.

<sup>&</sup>lt;sup>‡</sup> This category includes loans evaluated for impairment in conformity with FASB ASC 310-10-35.

This subcategory includes loans for which it is probable that the creditor will be unable to collect all amounts due according to the contractual terms of the loan agreement and, accordingly, for which impairment is measured in conformity with FASB ASC 310-10-35.

<sup>#</sup> This category comprises large groups of smaller-balance homogeneous loans that are collectively evaluated for impairment.

<sup>\*\*</sup> This category comprises large groups of all other loans not addressed in categories I or II and not individually considered impaired but that, on a portfolio basis, are believed to have some inherent but unidentified impairment.

<sup>(1)</sup> If subject to the scope of FASB ASC 815, *Derivatives and Hedging*, standby letters of credit and commitments should be excluded from the analysis. Credit exposure for instruments within the scope of FASB ASC 815 is captured by the fair value measurement of the instrument.

<sup>(2)</sup> Refer to FASB ASC 820, Fair Value Measurement, and FASB ASC 825, Financial Instruments, in chapter 20, "Fair Value," of this guide.

of powers to the federal banking agencies)<sup>3</sup> Policy Statement on Allowance for Loan and Lease Losses Methodologies and Documentation for Banks and Savings Institutions, published in the Federal Register on July 6, 2001, addresses (a) responsibilities of the board of directors and management, (b) importance of maintaining, and nature of, appropriate documentation supporting the reported amount of the allowance for loan and lease losses (ALLL), (c) written policies and procedures regarding the ALLL, and (d) appropriate ALLL methodologies, including validation of methodologies. The statement provides various examples of the areas addressed and includes six frequently asked questions along with agencies' interpretive responses. On July 6, 2001, the SEC issued parallel guidance in SAB No. 102, which expresses certain staff views on the development, documentation, and application of a systematic methodology as required by Financial Reporting Release No. 28 for determining ALLL in accordance with GAAP. In particular, the guidance focuses on the documentation the staff normally would expect registrants to prepare and maintain in support of the ALLL.

- **9.24** Guidance was provided to examiners in the federal banking agencies' December 12, 2006, joint issuance *Concentrations in Commercial Real Estate Lending, Sound Risk Management Practices*, which clarified final guidance on concentrations in commercial real estate lending. The guidance is intended to help ensure that institutions pursuing a significant commercial real estate lending strategy remain profitable while continuing to serve the credit needs of their communities. Other matters addressed include the following:
  - Small- to medium-sized banks facing strong competition should be aware of the risk of unanticipated earnings and capital volatility due to increased real estate loan concentrations. The agencies provide supervisory criteria including the use of numerical indicators in identifying institutions with potentially significant commercial real estate loan concentrations that may warrant greater supervisory scrutiny.
  - The guidance also serves to remind institutions that strong risk management practices and appropriate levels of capital are important elements of a sound lending program, particularly when an institution has a concentration in commercial real estate loans.
- 9.25 The interagency statements include the *Interagency Policy Statement* on the Allowance for Loan and Lease Losses, which was issued on December 13, 2006, by the federal banking agencies, along with the National Credit Union Administration (NCUA). This statement revised the 1993 policy statement on the ALLL to ensure consistency with GAAP. The revisions make the policy statement applicable to credit unions. The statement addresses (a) the nature and purpose of the allowance, (b) the related responsibilities of the board of directors and management and of the examiners, (c) loan review systems, and (d) international transfer risk matters. The federal banking agencies, along with the NCUA, also issued 16 frequently asked questions to assist institutions in complying with GAAP and ALLL supervisory guidance. (See the entry allowance for loan and lease losses, in the "Glossary" section of the FFIEC's

©2016, AICPA AAG-DEP 9.25

<sup>&</sup>lt;sup>3</sup> See chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide for further discussion on the Office of Thrift Supervision transfer of powers.

<sup>&</sup>lt;sup>4</sup> Federal Register Vol. 71, No. 238 [12 December 2006], pp. 74585-74588.

Instructions for Preparation of Consolidated Reports of Condition and Income, for additional information.)

- **9.26** On March 17, 2008, the FDIC issued Financial Institution Letter (FIL)-22-2008, *Managing Commercial Real Estate Concentrations in a Challenging Environment*, to re-emphasize the importance of strong capital and loan loss allowance levels, and robust credit risk management practices for institutions with concentrated commercial real estate exposures, consistent with the commercial real estate lending interagency guidance published on December 12, 2006, and the interagency policy statement on the ALLL issued on December 13, 2006.
- 9.27 On August 3, 2009, the FDIC issued FIL-43-2009, Allowance for Loan and Lease Losses: Residential Mortgages Secured by Junior Liens, which reminded financial institutions of several key points in the December 13, 2006, interagency guidance (see paragraph 9.25) and provided specific guidance for residential mortgages secured by junior liens. Institutions were reminded that, when estimating credit losses on each group of loans with similar risk characteristics under FASB ASC 450-20, they should consider their historical loss experience on the group, adjusted for changes in trends, conditions, and other relevant factors that affect repayment of the loans in the group as of the ALLL evaluation date. FDIC FIL-43-2009 stated that the need to consider all significant factors that affect the collectibility of loans is especially important for loans secured by junior liens on one-to-four family residential properties, both closed-end and open-end, in areas where there have been declines in the value of such properties. The letter notes that delaying the recognition of estimated credit losses is not appropriate and could delay appropriate loss mitigation activity, such as restructuring junior lien loans to more affordable payments or reducing principal on such loans to facilitate refinancing. The letter concluded by stating that examiners are evaluating the effectiveness of an institution's loss mitigation strategies for loans as part of their assessment of the institution's overall financial condition.
- **9.28** In January 2012, the federal banking agencies and the NCUA issued Interagency Supervisory Guidance on Allowance for Loan and Lease Losses Estimation Practices for Loans and Lines of Credit Secured by Junior Liens on 1-4 Family Residential Properties. This guidance reiterates key GAAP concepts and supervisory guidance related to the ALLL and loss estimation practices. For institutions that hold a significant junior lien portfolio, the guidance addresses the responsibilities of management in estimating the allowance and examiners' review of management's assessment. Although the discussion is specifically tailored in evaluating junior liens, the concepts and principles contained apply to estimating the ALLL for all types of loans. Readers can access the guidance from any of the federal banking agencies' websites.
- **9.29** The FDIC released FIL-49-2015, Advisory on Effective Risk Management Practices for Purchased Loans and Purchased Loan Participations, in November 2015, to update information contained in the FDIC Advisory on Effective Credit Risk Management Practices for Purchased Loan Participations (initially issued September 2012 through FIL-38-2012). The update Advisory addresses purchased loans and loan participations and reminds FDIC-supervised institutions of the importance of underwriting and administering loan purchases and loan participations in the same diligent manner as if the loans were directly originated by the purchasing institution. The advisory also reminds FDIC-supervised institutions that third party arrangements to

facilitate the purchase of loans and participations should be managed by an effective third-party risk management process. The advisory goes on to outline recommended practices relating to (a) policy guidelines for purchased loans and participations, (b) independent credit and collateral analysis, (c) profit analysis, (d) loan purchase and participation agreements, (e) ability to transfer sell, or assign interest, (f) due diligence and monitoring of purchased loans and participations in out-of-territory or unfamiliar markets, (g) due diligence of third parties, (h) financial reporting, (i) audit, (j) board approval and reporting, (k) and Bank Secrecy Act/Anti-Money Laundering. Readers can access FIL-49-2015 from the FDIC website at www.fdic.gov.

- **9.30** The OCC's *Bank Accounting Advisory Series* (BAAS) is updated periodically to express the Office of the Chief Accountant's current views on accounting topics of interest to national banks and federal savings associations. See further discussion of the BAAS in paragraph 7.82 of this guide. Topic 4, "Allowance for Loan and Lease Losses," includes interpretations and responses related to the ALLL. Readers are encouraged to view this publication under the "Publications—Bank Management" page at www.occ.gov.
- **9.31** Regulatory guidance provides that loans (and other assets) should be placed on nonaccrual (a) when the loan (or asset) is maintained on a cash basis because of deterioration in the financial condition of the borrower, (b) when payment in full of principal or interest is not expected, or (c) when principal or interest has been in default for a period of 90 days or more unless the asset is both well secured and in the process of collection. (See the entry nonaccrual status in the "Glossary" section of the FFIEC's Instructions for Preparation of Consolidated Reports of Condition and Income. See also appendix B, "Regulatory Reporting Matters—Interpretation and Reporting Related to U.S. GAAP," of this guide for additional information. Readers can also find questions and answers related to nonaccrual loans in the OCC's BAAS. See further discussion and a link to this publication in paragraph 9.30.)
- **9.32** Management should provide auditors with regulatory examination reports, which generally disclose classified loans and certain statistics regarding those classifications. If a regulatory examination is in process, the auditor should discuss the status and preliminary findings of the examination with institution management and the examiners. Communications with regulators are discussed further in chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide.
- **9.33** Effective March 30, 1993, the federal banking agencies established the *Interagency Policy Statement on Documentation for Loans to Small-and Medium-sized Businesses and Farms* to encourage lending to small and medium-sized businesses. The policy allows certain banks and savings institutions to establish a portfolio of loans exempt from certain documentation requirements. To qualify for the exemption, each loan may not exceed the lesser of \$900,000 or 3 percent of the institution's total capital, and the aggregate value of the loans may not exceed 20 percent of an institution's total capital. Examiners may not criticize the credit quality of an exempt loan on the basis of documentation and may not classify the loan unless it is delinquent by more than 60 days. The institution's management, however, is still required to fully evaluate the collectibility of exempt loans in determining the appropriateness in accordance with GAAP of loan loss allowances. (See paragraph 9.98.)

#### **Credit Unions**

- **9.34** Federal credit unions are required by Part 702 of the NCUA regulations to establish and maintain an allowance for loan losses. Federally insured state-chartered credit unions are usually required by their insurance agreement with the National Credit Union Share Insurance Fund (NCUSIF) to establish and maintain an allowance. The requirements for state-chartered credit unions that are not federally insured vary by state and insurer.
- **9.35** Credit unions should not base the justification of a lower allowance for loan losses on the maintenance of the regular reserve. Regulators have historically stated that the regular reserve has been established to cover loan losses. Although this may be true in a regulatory sense, the regular reserve constitutes an appropriation of undivided earnings and should not be considered in determining the amount of the allowance for loan losses under GAAP.
- **9.36** Some credit unions record a provision for loan losses equal to what would normally be transferred to the regular reserve from undivided earnings. Because the regular reserve may be reduced by the amount equal to the provision made, no regular reserve transfer is effectively made. The consequences of these actions may result in an overstatement of the allowance account. A credit union's allowance may be materially overstated due to strict adherence to this process.
- **9.37** For regulatory purposes, credit unions have historically used either the experience method or the adjustment method to calculate their allowance for loan losses. The NCUA issued Interpretive Ruling and Policy Statement (IRPS) 02-3, Allowance for Loan and Lease Losses Methodologies and Documentation for Federally-Insured Credit Unions, through Letter to Credit Unions No. 02-CU-09, Allowance for Loan and Lease Losses. The IRPS provides guidance on the design and implementation of ALLL methodologies and supporting documentation practices. Furthermore, it recognizes that credit unions should adopt methodologies and documentation practices that are appropriate for their size and complexity. As a follow up to IRPS 02-3, the NCUA issued Letter to Credit Unions No. 03-CU-01, Loan Charge-Off Guidance, to provide guidance on the systematic charge off on uncollectible loans. Although the application of the NCUA's methods may or may not result in substantially the same allowance as management's estimate for the allowance, management should report an allowance in the financial statements prepared under GAAP that is appropriate in accordance with GAAP to cover all estimated losses incurred at the statement-of-financial-condition date in the loan portfolio.
- **9.38** The NCUA also issued Accounting Bulletin No. 06-01 to distribute an interagency advisory addressing the ALLL (see paragraph 9.25 for further discussion on the advisory). The bulletin states that the policy statement applies to all credit unions supervised by the NCUA and insured by the NCUSIF. The policy statement is designed to supplement IRPS 02-03 and the *Update on Accounting for Loan Lease Losses*.

# **Accounting and Financial Reporting**

# Sources of Applicable Guidance

**9.39** FASB ASC 310-10 and FASB ASC 450-20 are the primary sources of guidance on accounting for the allowance for loan losses. FASB ASC 450-20 provides the basic guidance for recognition of impairment losses for all

receivables (except those receivables specifically addressed in other guidance, such as debt securities). In addition, see FASB ASC 310-40 which addresses loans subject to troubled debt restructurings, and which can interact with the guidance on impaired loans.

- **9.40** Paragraphs 12–40 of FASB ASC 310-10-35 provides specific guidance on measurement and disclosure for loans that are identified for evaluation and that are individually deemed to be impaired (because it is probable that the creditor will be unable to collect all the contractual interest and principal payments as scheduled in the loan agreement).
- **9.41** The "General" subsection of FASB ASC 310-10-35 addresses both the impairment concepts applicable to all receivables, with references to the guidance in FASB ASC 450-20 where appropriate, and the impairment concepts related to loans that are identified for evaluation and that are individually deemed to be impaired, as discussed in FASB ASC 310-10-35-2 (see paragraph 9.40). Paragraphs 33–36 of FASB ASC 310-10-35 also address the interaction between FASB ASC 310-10-35 and FASB ASC 450-20.

#### Allowance for Loan Losses

- **9.42** The following provides an overview of GAAP for loan impairments, as stated in FASB ASC 310-10-35-4:
  - a. It is usually difficult, even with hindsight, to identify any single event that made a particular loan uncollectible. However, the concept in GAAP is that impairment of receivables should be recognized when, based on all available information, it is probable that a loss has been incurred based on past events and conditions existing at the date of the financial statements.
  - b. Losses should not be recognized before it is probable that they have been incurred, even though it may be probable based on past experience that losses will be incurred in the future. It is inappropriate to consider possible or expected future trends that may lead to additional losses. Recognition of losses should not be deferred to periods after the period in which the losses have been incurred.
  - c. GAAP does not permit the establishment of allowances that are not supported by appropriate analyses. The approach for determination of the allowance should be well documented and applied consistently from period to period.
  - d. Under FASB ASC 450-20, the threshold for recognition of impairment should be the same whether the creditor has many loans or has only one loan. FASB ASC 310-10-35-9 requires that if the conditions of FASB ASC 450-20-25-2 are met, accrual should be made even though the particular receivables that are uncollectible may not be identifiable.
  - e. The guidance in FASB ASC 310-10-35 is more specific than FASB ASC 450-20 in that it requires certain methods of measurement for loans that are individually considered impaired, but it does not fundamentally change the recognition criteria for loan losses.
- **9.43** The allowance for loan losses should be appropriate in accordance with GAAP to cover probable credit losses related to specifically identified loans as well as probable credit losses inherent in the remainder of the loan portfolio.

- **9.44** The act of lending money generally is not the event that causes asset impairment, and future losses should not be provided for at the time loans are made. Credit losses are provided for when the events have occurred that cause the losses or loan impairment (for example, loss of employment, disability, or bankruptcy). As stated in FASB ASC 942-310-25-1, generally, a loan would be impaired at origination only if a faulty credit granting decision has been made or loan credit review procedures are inadequate or overly aggressive, in which case, the loss generally should be recognized at the date of loan origination.
- **9.45** Loans that are identified for evaluation or that are individually considered impaired. FASB ASC 310-10-35 provides guidance on measurement and disclosure of loans that are identified for evaluation and that are individually deemed to be impaired through a three-step process of identifying loans for evaluation (paragraphs 14–15 of FASB ASC 310-10-35), assessing whether a loan is impaired (paragraphs 16-19 of FASB ASC 310-10-35), and measurement of impairment if a loan is deemed impaired (paragraphs 20-36 of FASB ASC 310-10-35). In accordance with FASB ASC 310-10-35-13, this guidance applies to all loans that are identified for evaluation, uncollateralized as well as collateralized, except for (a) large groups of smaller-balance homogeneous loans that are collectively evaluated for impairment. (b) loans that are measured at fair value or at the lower of cost or fair value, (c) leases as defined in FASB ASC 840. Leases, and (d) debt securities as defined in FASB ASC 320. Investments—Debt and Equity Securities. This guidance does not address when a creditor should record a direct write-down of an impaired loan, nor does it address how a creditor should assess the overall adequacy of the allowance for credit losses.

#### © Update 9-2 Accounting and Reporting: Leases

FASB ASU No. 2016-02, *Leases (Topic 842)*, issued in February 2016, is effective for fiscal years of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files financial statements with the SEC beginning after December 15, 2018, including interim periods within those fiscal years.

For all other entities, FASB ASU No. 2016-02 is effective for fiscal years beginning after December 15, 2019, and interim periods within fiscal years beginning after December 15, 2019.

Early application is permitted for all entities.

FASB ASU No. 2016-02 supersedes the lease requirements in FASB ASC 840 and creates FASB ASC 842, *Leases*, to establish the principles that lessees and lessors should apply to report useful information to users of financial statements about the amount, timing, and uncertainty of cash flows arising from a lease. FASB ASC 842 affects any entity that enters into a lease (as that term is defined in FASB ASU No. 2016-02), with some specified scope exceptions.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-02, see appendix F, "The New Leases Standard: FASB ASU No. 2016-02," of this guide.

- **9.46** Paragraphs 20–36 of FASB ASC 310-10-35 address the measurement of impairment. FASB ASC 310-10-35-21 permits a creditor to aggregate impaired loans that have risk characteristics in common with other impaired loans and use historical statistics, such as average recovery period and average amount recovered, along with a composite effective interest rate as a means of measuring those impaired loans.
- **9.47** FASB ASC 310-10-35-22 states that when a loan is impaired (see paragraphs 16–17 of FASB ASC 310-10-35), a creditor should measure impairment based on the present value of expected future cash flows discounted at the loan's effective interest rate, except that as a practical expedient, a creditor may measure impairment based on a loan's observable market price or the fair value of the collateral if the loan is collateral dependent. As discussed in the FASB ASC glossary, a *collateral-dependent loan* is a loan for which the repayment is expected to be provided solely by the underlying collateral. Generally, repayment of an impaired loan would be expected to be provided solely by the sale or continued operation of the underlying collateral if cash flows from all other available sources (including guarantors) are expected to be no more than nominal.
- **9.48** FASB ASC 310-10-35-23 states that if a creditor uses the fair value of the collateral to measure impairment of a collateral-dependent loan and repayment or satisfaction of a loan is dependent on the sale of the collateral, the fair value of the collateral should be adjusted to consider estimated costs to sell. However, if repayment or satisfaction of the loan is dependent only on the operation, rather than the sale, of the collateral, the measure of impairment should not incorporate estimated costs to sell the collateral.

# **Loss Contingencies**

- **9.49** Paragraphs 1 and 3 of FASB ASC 450-20-25 state that when a loss contingency exists, the likelihood that the future event or events will confirm the loss or impairment of an asset can range from remote to probable. (*Probable*, according to the FASB ASC glossary, means the future event or events are likely to occur.) However, the conditions for accrual are not intended to be so rigid that they require virtual certainty before a loss is accrued.
- **9.50** FASB ASC 450-20-25-2 requires that an estimated loss from a loss contingency should be accrued by a charge to income if both of the following conditions are met:
  - a. Information available prior to issuance of the financial statements indicates that it is probable that an asset had been impaired or a liability had been incurred at the date of the financial statements. It is implicit in this condition that it must be probable that one or more future events will occur confirming the fact of the loss.
  - b. The *amount* of loss can be reasonably estimated.
- **9.51** FASB ASC 450-20 deals with uncertainty by requiring a probability threshold for recognition of a loss contingency and that the amount of the loss be reasonably estimable. As noted in FASB ASC 450-20-30-1, when both of those recognition criteria are met, and the reasonably estimable loss is a range, the accrual should be the amount that appears to be a better estimate than any other estimate within the range or the minimum amount in the range if no amount within the range is a better estimate than any other amount.

- 9.52 The following may be examples of loss contingencies:
  - Collectibility of receivables other than loans
  - Guarantees of indebtedness of others
  - Obligations of commercial banks under standby letters of credit
  - Agreements to repurchase receivables (or to repurchase the related property) that have been sold

# Financial Statement Presentation and Disclosure of Loan Impairment and Allowance for Credit Losses

- **9.53** Paragraphs 41–42 of FASB ASC 310-10-35 state that credit losses for loans and trade receivables, which may be for all or part of a particular loan or trade receivable, should be deducted from the allowance. The related loan or trade receivable balance should be charged off in the period in which the loans or trade receivables are deemed uncollectible (for example, the loss is confirmed). Recoveries of loans and trade receivables previously charged off should be recorded when received. Practices differ between entities as some industries typically credit recoveries directly to earnings. Financial institutions typically credit the allowance for loan losses for recoveries, which, in combination with the practice of frequently reviewing the adequacy of the allowance for loan losses, results in the same credit to earnings, albeit in an indirect manner.
- **9.54** Provisions for loan and other credit losses should be charged to operating income sufficient to maintain the allowance for loan losses or liabilities related to off-balance-sheet credit exposures at a level appropriate in accordance with GAAP—that is, management should verify that the allowance is appropriate to cover incurred losses in accordance with GAAP as of the balance sheet date. Thus, the primary measurement focus is on the appropriateness of the balance of the allowance and the liabilities, rather than on the resulting provision charged to income.
- **9.55** *Impaired loans.* Paragraphs 14A–20 of FASB ASC 310-10-50 require disclosure of information about loans that meet the definition of an impaired loan in paragraphs 16–17 of FASB ASC 310-10-35. Included are various disclosures about the recorded investment in the impaired loans, the total unpaid principal balance of the impaired loans, the entity's interest income recognition policy, the activity in the allowance for loan losses, the entity's loan impairment assessment policy, and factors considered in determining that the loan is impaired. (See FASB ASC 310-40-50 for disclosure information about an impaired loan that has been restructured in a troubled debt restructuring involving a modification of terms.)
- **9.56** Loans acquired with deteriorated credit quality. FASB ASC 310-30-50 provides disclosure guidance regarding loans with evidence of deterioration of credit quality since origination that are acquired by completion of a transfer for which it is probable, at acquisition, that the investor will be unable to collect all contractually required payments receivable, as explained in FASB ASC 310-30-05-1. Paragraphs 8.110–.122 of this guide provide additional information regarding loans and debt securities acquired with deteriorated credit quality.
- **9.57** Off-balance-sheet credit exposures. As stated in paragraphs 1–3 of FASB ASC 825-10-35, an accrual for credit loss on a financial instrument with off-balance-sheet risk should be recorded separately from a valuation account related to a recognized financial instrument. Credit losses for off-balance-sheet financial instruments should be deducted from the liability for credit

losses in the period in which the liability is settled. Off-balance-sheet financial instruments refer to off-balance-sheet loan commitments, standby letters of credit, financial guarantees, and other similar instruments with off-balance-sheet credit risk, except for instruments within the scope of FASB ASC 815-10. See chapter 20, "Fair Value," of this guide for a summary of FASB ASC 825, Financial Instruments.

- **9.58** FASB ASC 310-10-50-9 requires that in addition to disclosures required by FASB ASC 450-20, an entity should disclose a description of the accounting policies and methodology the entity used to estimate its liability for off-balance-sheet credit exposures and related charges for those credit exposures. Such a description should identify the factors that influenced management's judgment (for example, historical losses and existing economic conditions) and a discussion of risk elements relevant to particular categories of financial instruments.
- **9.59** Allowance for credit losses related to financing receivables. In accordance with FASB ASC 310-10-50-11B, an entity should disclose all of the following by portfolio segment: $^5$ 
  - a. A description of the entity's accounting policies and methodology used to estimate the allowance for credit losses, including all of the following:
    - A description of the factors that influenced management's judgment, including both historical losses and existing economic conditions
    - ii. A discussion of risk characteristics relevant to each portfolio segment
    - iii. Identification of any changes to the entity's accounting policies or methodology from the prior period and the entity's rationale for the change
  - b. A description of the policy for charging off uncollectible financing receivables
  - c. The activity in the allowance for credit losses for each period, including all of the following:
    - The balance in the allowance at the beginning and end of each period
    - ii. Current period provision
    - iii. Direct write-downs charged against the allowance
    - iv. Recoveries of amounts previously charged off
  - d. The quantitative effect of changes to allowance accounting policies or methodology on the current period loss provision
  - e. The amount of any significant purchases of financing receivables during each reporting period
  - f. The amount of any significant sales of financing receivables or reclassifications of financing receivables to held for sale during each reporting period

 $<sup>^5</sup>$  According to the FASB  $Accounting\ Standards\ Codification\ (ASC)\ glossary, a portfolio\ segment$  is defined as the level at which an entity develops and documents a systematic methodology to determine its allowance for credit losses. See paragraphs 21–22 of FASB ASC 310-10-55.

- g. The balance in the allowance for credit losses at the end of each period disaggregated on the basis of the entity's impairment method
- h. The recorded investment in financing receivables at the end of each period related to each balance in the allowance for credit losses, disaggregated on the basis of the entity's impairment methodology in the same manner as the disclosure in item g

See paragraph 8.163 for additional disclosures related to credit quality that are required pursuant to FASB ASC 310-10-50.

According to FASB ASC 310-10-50-11A, the preceding guidance does not apply to financing receivables listed in FASB ASC 310-10-50-7B or lessors' net investments in leveraged leases.

- **9.60** FASB ASC 310-10-50-11C states to disaggregate the information required by items g and h in the preceding paragraph on the basis of the impairment methodology, an entity should separately disclose the following amounts:
  - a. Amounts collectively evaluated for impairment (determined under FASB ASC 450-20)
  - b. Amounts individually evaluated for impairment (determined under FASB ASC 310-10-35)
  - c. Amounts related to loans acquired with deteriorated credit quality (determined under FASB ASC 310-30)
- **9.61** Risks and uncertainties. In accordance with FASB ASC 310-10-50-25, certain loan products have contractual terms that expose entities to risks and uncertainties that fall into one or more categories, as discussed in FASB ASC 275-10-50-1. See FASB ASC 275-10-50 for disclosure guidance related to those loan products.
- **9.62** *Credit unions*. A change from a method of calculating the allowance for loan losses that is not generally accepted (for example, a calculation used for regulatory purposes) to a method that is generally accepted and that results in an adjustment to the amount previously reported is considered a correction of an error and is reported as a prior-period adjustment, requiring restatement of prior-period financial statements, in accordance with FASB ASC 250-10-45-23. Such a change frequently arises when a credit union that has in the past undergone only supervisory committee audits initially undergoes an audit in accordance with generally accepted auditing standards.

# Auditing<sup>6</sup>

## **Objectives**

**9.63** The primary objectives of audit procedures for credit losses are to obtain sufficient appropriate evidence that

<sup>&</sup>lt;sup>6</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards (GAAS) issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of *issuers* and *nonissuers* as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

- a. the allowance for loan losses and liability for other credit exposures are appropriate in accordance with GAAP to cover the amount of probable credit losses inherent in the loan portfolio and off-balancesheet financial instruments, respectively, at the balance sheet date;
- b. credit losses and other items, such as loan charge-offs and recoveries, have been included in the financial statements at appropriate amounts and in the appropriate reporting periods; and
- c. disclosures are adequate.

The auditor generally achieves those objectives by testing management's estimates of the allowance based on available and relevant information regarding loan collectibility. The auditor could also develop an independent estimate as more fully discussed in paragraph 9.80. The auditor is not responsible for estimating the amount of the allowance or ascertaining the collectibility of each, or any, specific loan included in an institution's loan portfolio.

## **Planning and Risk Assessment**

**9.64** In accordance with AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (as described in chapter 5 of this guide). Because of the significance of loans to institutions' balance sheets, and because the estimation of loan losses is based on subjective judgments, auditors are likely to assess inherent risk related to the allowance for loan losses as high. Such assessment will influence engagement staffing, extent of supervision, overall scope and strategy, and degree of professional skepticism applied. Further, it is necessary for the auditor to gain familiarity with the applicable regulatory guidance, including guidance on the classification of credits, concentration of credits, foreign loans, and significant related parties.

**9.65** AU-C section 540, Auditing Accounting Estimates, Including Fair Value Accounting Estimates, and Related Disclosures (AICPA, Professional Standards), addresses the auditor's responsibilities relating to accounting estimates, including fair value accounting estimates and related disclosures, in an audit of financial statements. Specifically, it expands on how AU-C sections 315; 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards); and other relevant AU-C sections are to be applied with regard to accounting estimates. It also includes requirements and guidance related to misstatements of individual accounting estimates and indicators of possible management bias.

Considerations for Audits Performed in Accordance With PCAOB Standards

When performing an integrated audit of financial statements and internal control over financial reporting in accordance with PCAOB standards, the work that the auditor performs as part of the audit of internal control over financial reporting should necessarily inform the auditor's decisions about the approach he or she takes to auditing an estimate because, as part of the audit of internal control over financial reporting, the auditor would be required to obtain an understanding of the process management used to develop the estimate and to test controls over all relevant assertions related to the estimate.

PCAOB Staff Audit Practice Alert No. 11, Considerations for Audits of Internal Control Over Financial Reporting (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance, sec. 400.11), discusses the application of certain requirements of AS 2201, An Audit of Internal Control Over Financial Reporting That Is Integrated with An Audit of Financial Statements (AICPA, PCAOB Standards and Related Rules), and other PCAOB standards to specific aspects of the audit of internal control in light of the PCAOB's recent observations of auditing deficiencies. Specifically, this alert discusses the following topics:

- Risk assessment and the audit of internal control
- Selecting controls to test
- Testing management review controls
- IT considerations, including system-generated data and reports
- Roll-forward of controls tested at an interim date
- Using the work of others
- Evaluating identified control deficiencies
- **9.66** The audit procedures performed in connection with the allowance for loan losses typically are time-consuming and are most efficient if initiated early in the audit. Because of the subjective nature of the allowance for loan losses, experienced audit personnel, preferably with prior depository institution engagement experience and, if necessary, with knowledge of industries in which the institution's loans are concentrated, should closely supervise or perform the audit procedures. One example would be audit procedures over loan reviews. The assigned audit staff should also understand the lending environment, including credit strategy; credit risk; and the lending policies, procedures, and control environment of the institution, and should be familiar with known related parties and related-party transactions.
- **9.67** Another important consideration in obtaining an understanding of the entity and its environment is whether an institution's internal loan review and internal audit functions can be considered in the audit plan by the auditor, and permit the auditor to modify the nature, timing, and extent of procedures to be performed. Discussions with internal loan review and internal audit staff can provide the auditor with information concerning loan customers, related-party transactions, and account histories that may not be readily available elsewhere. Also, because the internal audit function is involved in evaluating accounting systems and control activities (as discussed in chapter 8 of this guide), it can provide the auditor with important control process descriptions and results of testing that are helpful in understanding internal control. Chapter 5 of this guide discusses consideration of the internal audit function.
- **9.68** Paragraph .A120 of AU-C section 315 states that information gathered by performing risk assessment procedures, including the audit evidence obtained in evaluating the design of controls and determining whether they

have been implemented, is used as audit evidence to support the risk assessment. The risk assessment determines the nature, timing, and extent of further audit procedures to be performed. As a part of the risk assessment with respect to the allowance for loan losses, the auditor should consider factors such as

- composition of the loan portfolio;
- identified potential problem loans, including loans classified by regulatory agencies;
- trends in loan volume by major categories, especially categories experiencing rapid growth, and in delinquencies and restructured loans;
- previous loss and recovery experience, including timeliness of charge-offs;
- concentrations of loans to individuals and their related interests, industries, and geographic regions;
- size of individual credit exposures (few, large loans versus numerous, small loans);
- quality of internal loan review and internal audit functions, and results of their work;
- total amount of loans and problem loans, including delinquent loans, by officer;
- lending, charge-off, collection, and recovery policies and procedures;
- loan documentation and compliance exceptions reports;
- loans with repayment terms structured (or restructured) such that collectability problems and concerns may not be evident until payments come due, such as construction loans with interest reserves included in the loan commitment amount:
- local, national, and international economic and environmental conditions:
- experience, competence, and depth of lending management and staff;
- results of regulatory examinations;
- discussions of significant lending activities included in board and loan committee minutes;
- related-party lending;
- organizational structure; and
- management information systems.

Individual components of the allowance for loan losses might have different risks (for example, components that reflect different credit exposures, are determined using different methodologies, or are subject to different accounting requirements). As a result, the risk assessment process should involve challenging management's methodology and process, including factors such as portfolio segmentation, look-back periods, loss emergence periods, qualitative factors, and other criteria and metrics used to estimate the allowance for loan losses.

# Internal Control Over Financial Reporting and Possible Tests of Controls

**9.69** AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements

through understanding the entity and its environment, including the entity's internal control. Paragraphs .13—.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

- **9.70** Effective controls related to estimating the allowance for loan losses may reduce the likelihood of material misstatement of the allowance for loan losses. The auditor should obtain an understanding of (a) how management developed the allowance for loan losses, (b) how the process has changed from prior periods, and (c) the institution's loan portfolio, lending process, loan accounting policies, market focus, trade area, and other relevant factors. Specific aspects of effective controls related to the allowance for loan losses may include the following:
  - Control environment, including management communication of the need for proper reporting of the allowance. The control environment strongly influences the effectiveness of the system of controls and affects the auditor's assessment of control risk. The control environment reflects the overall attitude, awareness, and action of the board of directors and management concerning the importance of control. The auditor might consider
    - the level of the board of directors' and management's integrity and ethical values;
    - the board and management's commitment to competence;
    - the level of involvement and quality of leadership provided by the board of directors, audit committee, and senior management in evaluating the allowance;
    - management's philosophy and operating style;
    - the organizational structure;
    - the assignment of authority and responsibility; and
    - human resource policies and practices.
  - Controls over the completeness and accuracy of data used, including the accumulation of relevant, sufficient, and reliable data on which to base management's estimate of the allowance. Management reports summarizing loan activity, renewals, and delinquencies are vital to the timely identification of problem loans. The institution's procedures and controls are important for identifying when loans should be placed on nonaccrual status, provided for, or charged off. Most institutions have written policies covering nonaccrual status, the timing of charge-offs, and transfers of loans to the special asset or workout department. Most institutions have policies and procedures for the gathering and analysis of information from and about debtors. Management should have controls over the completeness and accuracy of underlying data

- used in formulating the allowance estimate, such as segmented portfolio balances, loan risk classifications or ratings, charge-offs, delinquencies, and exceptions to underwriting standards.
- Controls over loan risk ratings. There may be multiple controls or elements of controls that operate over the appropriateness of loan risk ratings to mitigate the risk of using inaccurate ratings in formulating the allowance estimate. The controls should be designed such that the risk rating used for purposes of the allowance estimate is subject to review by a party other than the individual responsible for assigning the rating. The following are examples of controls that address the appropriateness of loan risk ratings; this is not intended to be an all-inclusive list:
  - Independent loan reviews. Loan reviews should be conducted by competent personnel who are independent of the underwriting, supervision, and collections functions. Alternatively, an independent loan review function may be outsourced to a competent third party. Although the specific lines of reporting for the loan review function often will depend on the complexity of the institution's organizational structure, the loan reviewers should report to a high level of management that is independent from the lending process in the institution. The loan review function is designed to test line management's identification and evaluation of existing and potential problem loans in a timely manner, including the appropriateness of individual loan risk ratings used in the allowance estimate. The loan review function should be designed to address considerations such as (a) risk assessment, including the determination of which groups of loans will be covered through credit file reviews by independent loan review in the current year; (b) selection of individual loans within those loan groups for review in the current year; (c) overall credit file review plan for the current year; (d) frequency of credit file reviews; (e) definition of *loan risk rating exceptions*; and (*f*) use of credit file review results. The sample of loans selected for review should be representative and unbiased.
  - The loan review function ordinarily conducts a series of activities, including but not limited to
    - conducting a loan grading risk assessment of the segmented risk rated loan portfolios;
    - preparing a loan file review plan;
    - determining the frequency of loan reviews;
    - monitoring loan grading risk assessment considerations to determine whether the review plan should be updated;
    - defining loan risk rating exceptions;
    - selecting a risk-based sample of loans for review and evaluating whether the assigned loan risk rating is appropriate;

- evaluating the appropriateness of the assigned loan risk rating for loans for which a loan review was performed;
- concluding whether the loan review results are appropriate and representative of the entire risk rated loan population, thereby mitigating the risk of inappropriate loan risk ratings in the population of graded loans in the current year; and
- communicating loan review results to management for consideration in the allowance estimate for the current year.
- Other reviews of loan risk ratings. In addition to independent loan reviews, institutions may have one or more controls that are relevant to the accuracy of loan risk ratings. Such controls may take different forms and vary from institution to institution. For example, an institution may subject loans with risk ratings to a formal review at least annually to ensure the ratings are accurate and up to date. In that regard, the responsible loan officer may be required to conduct and document a review of each loan that includes, among other things, a formal reassessment of the accuracy of the loan's risk rating. The loan officer's reassessment may then be subjected to a secondary review. The secondary annual review functions as a control over risk ratings assigned to all loans with such ratings, including groups of loans that are not covered by the scope of the independent loan review function in the current year.
- Controls over the loss estimation process. A loss estimation process for individually impaired loans and groups of other loans includes
  - assigning responsibility for identification of impaired loans;
  - assigning responsibility for measurement of impairment;
  - impaired loan tracking and impairment measurement information system:
  - historical loss tracking and loss rates measurement information system;
  - documenting current economic conditions that differ from conditions that prevailed in the prior periods used to determine historical loss rates, and justification for specific adjustments to historical loss rates; and
  - accumulating the component needs of the allowance and provision amounts.

Controls over the loss estimation process include review controls over the judgments within the allowance estimate, as well as controls over the completeness and accuracy of underlying data used in the operation of the review control. These also include controls over models used in the estimation process.

- Adequate review and approval of the allowance estimates by the individuals specified in management's written ALLL policy (or other documents that describe the ALLL methodology). This includes
  - review of sources of relevant information;
  - review of development of assumptions and methodologies (for example, historical loss experience, including the look-back period and loss emergence period, and qualitative adjustment factors);
  - review of reasonableness of assumptions, methodologies, and resulting estimates, including metrics, thresholds, or criteria to identify outliers or exceptions;
  - consideration of the need to use the work of specialists (such as appraisers or construction specialists);
  - consideration of changes in previously established methods to arrive at the allowance; and
  - review of accuracy of data and calculations.
- Back-testing and validation of prior estimates related to the allowance with subsequent results to assess the reliability of the process used to develop the allowance, including controls over the back-testing process.
- Controls over the identification of subsequent events or transactions that provide additional evidence about conditions that existed or did not exist at the balance sheet date. Considerations may include the following:
  - Whether loans that have become delinquent or that have been restructured or charged off subsequent to the balance sheet date were impaired as of the balance sheet date.
  - Updated appraisal information received subsequent to the balance sheet date.
  - Independent loan review results received subsequent to the balance sheet date.
  - Regulatory examination results received subsequent to the balance sheet date.
- **9.71** Because compliance with a well-defined lending policy is essential to an institution's asset quality, failure to follow that policy could have a substantial impact on the reliability of financial statement assertions. For example, authority limits established in management's written underwriting policies are based in large part on (a) the knowledge and skill of the reviewing loan officer or committee and (b) the credit risk the institution is willing to assume on a particular type of loan. A loan made for an amount in excess of an officer's limit, or for an unauthorized loan type, would normally involve greater amounts of credit or other risks. Accordingly, management's financial statement assertions about impairment and valuation of the loan portfolio, for example, may be affected.
- **9.72** AU-C section 330 addresses the auditor's responsibility to design and implement responses to the risks of material misstatement identified and

assessed by the auditor in accordance with AU-C section 315 and to evaluate the audit evidence obtained in an audit of financial statements.

**9.73** In accordance with paragraph .08 of AU-C section 330, the auditor should design and perform tests of controls to obtain sufficient appropriate audit evidence about the operating effectiveness of relevant controls if (a) the auditor's assessment of risks of material misstatement at the relevant assertion level includes an expectation that the controls are operating effectively or (b) substantive procedures alone do not provide sufficient appropriate audit evidence at the relevant assertion level. The auditor may consider performing procedures such as the following to test the effectiveness of internal control over financial reporting in the area of credit losses:

- Review and challenge the design of the institution's accounting policies and procedures describing the process for determining, evaluating, and maintaining the allowance for loan losses and other credit losses in accordance with GAAP. Discuss these policies and procedures with appropriate personnel to determine whether they understand the related financial reporting objective.
- Test the controls over identifying and reporting classified and potential problem loans (for example, a *watch list*) and following up on such loans, as well as delinquent loan reports and procedures for follow-up on delinquencies.
- Examine evidence that senior management and an appropriate board committee review and monitor past due loans, watch list loans, classified loans, and assigned risk ratings.
- Test application controls and related general IT controls, including system interfaces and account mapping, for all relevant systems.
- Test the controls over completeness and accuracy of the underlying data used in the allowance process, including the preparation of the periodic past due, watch list, and classified loans reports.
- Test the controls over how each report is prepared, including which loans are included and excluded.
- Examine evidence that the delinquent loan report interfaces appropriately with the watch list or problem loan report.
- Test the controls over the completeness and accuracy of loans between the trial balance and the applicable report.
- Examine evidence that the institution has an independent loan review function to review and evaluate loan officer analyses of significant loans, including the accuracy of the risk ratings.
- Examine evidence of secondary review of the loan officers' periodic loan risk ratings.
- Review the institution's documentation that analyzes each component of the allowance for loan losses. For example, test the calculation of historical loss experience for one or more periods and one or more pools of loans or test the calculation of discounted expected cash flows for one or more impaired loans.
- Read minutes of meetings of the board or loan committee for evidence of the periodic review and approval of the appropriateness in accordance with GAAP of the allowance for loan losses based on appropriate documentation.

There may be more than one control that addresses the assessed risk of material misstatement to a particular relevant assertion for the ALLL; and therefore the combination of controls would prevent, or detect, a material misstatement.

- 9.74 Management review controls may be higher-level or process-level controls and usually involve a member of management reviewing information contained in documents, reports, or other information produced by the entity, including variance reports, exception reports, or detailed calculations supporting financial statement balances or disclosures. Management review controls also may relate to significant management estimates or judgments incorporated into the allowance process, such as independent loan review and management review of qualitative or environmental factors and adjustments to the historical loss experience. The design of management review controls includes metrics, thresholds, or other criteria to identify outliers or exceptions and should involve the appropriate level of precision to ensure that the controls would detect a material misstatement.
- **9.75** Ordinarily, it is expected that a management review control is performed using information produced by the entity. When testing controls that are dependent on information produced by the entity (for example, system generated or management prepared reports), the auditor should test controls over the completeness and accuracy of that information.

### **Substantive Tests**

- **9.76** Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure, which for a financial institution would include credit losses. In accordance with paragraph .A45 of AU-C section 330, this requirement reflects the facts that (i) the auditor's assessment of risk is judgmental and may not identify all risks of material misstatement and (ii) inherent limitations to internal control exist, including the potential for management override.
- 9.77 Based on the assessed risks of material misstatement, paragraph .12 of AU-C section 540 states that the auditor should determine
  - a. whether management has appropriately applied the requirements
    of the applicable financial reporting framework relevant to the accounting estimate (in this instance, the allowance for loan losses)
    and
  - b. whether the methods for making the accounting estimates are appropriate and have been applied consistently and whether changes from the prior period, if any, in accounting estimates or the method for making them are appropriate in the circumstances.
- **9.78** In responding to the assessed risks of material misstatement, as required by AU-C section 330, paragraph .13 of AU-C section 540 states that the auditor should undertake one or more of the following, taking into account the nature of the accounting estimate:
  - a. Determine whether events occurring up to the date of the auditor's report provide audit evidence regarding the allowance for loan losses.

- b. Test how management made the accounting estimate of the allowance for loan losses and the data on which it is based. In doing so, the auditor should evaluate whether
  - i. the method of measurement used is appropriate in the circumstances,
  - ii. the assumptions used by management are reasonable in light of the measurement objectives of the applicable financial reporting framework (see further discussion in paragraph 9.81), and
  - iii. the data on which the allowance is based are sufficiently reliable for the auditor's purposes.
- c. Test the operating effectiveness of the controls over how management made the accounting estimate of the allowance for loan losses, together with the appropriate substantive procedures.
- d. Develop a point estimate or range to evaluate management's point estimate. For this purpose
  - i. if the auditor uses assumptions or methods that differ from management's, the auditor should obtain an understanding of management's assumptions or methods sufficient to establish that the auditor's point estimate or range takes into account relevant variables and to evaluate any significant differences from management's point estimate.
  - ii. if the auditor concludes that is it appropriate to use a range, the auditor should narrow the range, based on audit evidence available, until all outcomes within the range are considered reasonable.
- **9.79** Generally, the most effective audit strategy for the allowance for loan losses is to test management's methodology and process. This strategy may include
  - testing the extent to which data used in formulating the accounting estimate of the allowance for loan losses is accurate, complete, and relevant and whether the estimate has been properly determined using such data and management assumptions;
  - considering the source, relevance, and reliability of external data or information, including that received from external experts engaged by management to assist in making the accounting estimate;
  - determining how management has taken into account the effect
    of events, transactions, and changes in circumstances occurring
    between the date that the accounting estimate or inputs to the
    estimate were determined and the reporting date (for example, a
    valuation by an independent appraiser made prior to the reporting
    date);
  - recalculating the accounting estimate and reviewing information about the estimate for internal consistency;
  - considering management's review and approval processes for the accounting estimate; and
  - determining consistency of the resulting estimate with applicable GAAP.

- **9.80** In accordance with paragraph .A81 of AU-C section 540, the assumptions on which accounting estimates, such as the allowance for loan losses, are based may reflect what management expects will be the outcome of specific objectives and strategies. In such cases, the auditor may perform audit procedures to evaluate the reasonableness of such assumptions by considering, for example, whether the assumptions are consistent with
  - the general economic environment and the entity's economic circumstances.
  - the plans of the entity.
  - assumptions made in prior periods, if relevant.
  - the experience of, or previous conditions experienced by, the entity to the extent this historical information may be considered representative of future conditions or events. For example, compare current year charge-offs with prior period estimated losses to determine the historical reliability of prior period estimates.
  - other assumptions used by management relating to the financial statements.
- 9.81 The auditor may test key factors and assumptions such as those that are
  - significant to the estimate of the amount of the allowance for loan losses, such as
    - the effectiveness of the institution's internal control related to loans and the allowance for loan losses (the design and operating effectiveness of controls may impact the nature, timing, and extent of substantive procedures);
    - portfolio segmentation based on similar credit risk characteristics;
    - look-back period used to calculate the historical loss component of the allowance;
    - loss emergence period used to calculate the historical loss component of the allowance (this period is typically defined as the period of time from the event that triggers a loss to the charge-off or confirmation of the loss);
    - current local, national, and international economic conditions and trends, particularly as they have affected collateral values;
    - the amount of recoveries of loans previously charged off;
    - composition of the loan portfolio and trends in volume and terms of loans, as well as trends in delinquent and nonaccrual loans that could indicate historical loss averages do not reflect current conditions;
    - identified potential problem loans and large groups of problem loans, including delinquent and nonaccrual loans and loans classified according to regulatory guidelines;

- concentrations of loans to individuals or entities and their related interests, to industries, and in geographic regions;
- size of specific credit exposures (a few large loans versus numerous small loans);
- quality of the internal loan review and internal audit functions;
- the effects of changes in lending policies and procedures, including those for underwriting, credit monitoring, collection, and charge-offs that could indicate historical loss averages do not reflect current conditions;
- results of regulatory examinations; and
- nature and extent of related-party lending.
- sensitive to variations. Assumptions based on historical trends, such as the amount of late or partial payments in a particular period and the amount of charge-offs, can have a significant effect on estimates of the allowance.
- subjective and susceptible to misstatement and bias, such as
  - the risk classification and allowance allocation given to problem loans;
  - estimates of collateral values, and the related assumptions that drive the determination of such values, such as cash flow estimates, discount rates, and projected occupancy rates;
  - current economic or market conditions that in the future may affect a borrower's ability to meet scheduled repayments; and
  - contingencies, such as a commitment for funding from a third party.
- **9.82** For accounting estimates (such as the allowance for loan losses) that give rise to significant risks, in addition to other substantive procedures performed to meet the requirements of AU-C section 330, the auditor should evaluate the following:<sup>7</sup>
  - a. How management has considered alternative assumptions or outcomes and why it has rejected them or how management has otherwise addressed estimation uncertainty in making the accounting estimate
  - b. Whether the significant assumptions used by management are reasonable
  - c. When relevant to the reasonableness of the significant assumptions used by management or the appropriate application of the applicable financial reporting framework, management's intent to carry out specific courses of action and its ability to do so

<sup>&</sup>lt;sup>7</sup> Paragraph .18 of AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards).

If, in the auditor's professional judgment, management has not addressed adequately the effects of estimation uncertainty on the accounting estimates that give rise to significant risks, the auditor should, if considered necessary, develop a range with which to evaluate the reasonableness of the accounting estimate.

- **9.83** Further, the auditor should consider the total credit exposure of particular borrowers, including that related to standby letters of credit, guarantees, commitments to lend, and other off-balance-sheet exposures in addition to the institution's recorded liability for such other credit exposures.
- **9.84** When information included in reports prepared by management is used in the performance of substantive tests, the accuracy and completeness of such information should be evaluated by, for example, testing loan subsidiary ledgers and tracing delinquencies to the past-due reports.
- **9.85** In determining the matters identified in paragraph .12 of AU-C section 540 (see paragraph 9.77) or in responding to the assessed risks of material misstatement in accordance with paragraph .13 of AU-C section 540 (see paragraph 9.78), paragraph .14 of AU-C section 540 states that the auditor should consider whether specialized skills or knowledge with regard to one or more aspects of the allowance for loan losses is required in order to obtain sufficient appropriate audit evidence.
- **9.86** AU-C section 620, *Using the Work of an Auditor's Specialist* (AICPA, *Professional Standards*), addresses the auditor's responsibilities relating to the work of an individual or organization possessing expertise in a field other than accounting or auditing when that work is used to assist the auditor in obtaining sufficient appropriate audit evidence. To properly evaluate the collectibility of certain loans, the auditor may need information outside of his or her usual experience. For example, the auditor might encounter valuation problems that require special knowledge of types of collateral.
- **9.87** The knowledge of a specialist could be useful for loans based on oil and gas reserves. The specialist might review engineering reports on current reserves and production reports if the wells are in production. If fluctuating market conditions exist, a specialist could answer additional inquiries concerning the current status of oil and gas properties. For example, a loan secured by drilling equipment might have only marginal collateral value in a period of declining petroleum prices, even though the loan was highly secured when it was made.
- **9.88** Loans to developing countries are another example of instances in which the auditor may obtain the assistance of a specialist to become familiar with the economic, political, and social factors affecting the country's debt repayment. Other sources of such information include International Monetary Fund publications, international economists, and reports provided to institutions by the ICERC.
- **9.89** For impaired collateral-dependent loans, the specific allowance is generally based on the fair value of collateral less estimated selling costs. In this situation, auditors generally use an appraisal obtained by management as evidence of the fair value of the collateral, which is an example of using a management specialist as contemplated by paragraph .08 of AU-C section 500, *Audit Evidence* (AICPA, *Professional Standards*). Refer to chapter 11, "Real

Estate Investments, Real Estate Owned, and Other Foreclosed Assets," of this guide for audit procedures over appraisals.

- **9.90** In accordance with paragraphs .09–.10 and .12 of AU-C section 620, the auditor should evaluate whether the auditor's specialist has the necessary competence, capabilities, and objectivity for the auditor's purposes. In the case of an auditor's external specialist, the evaluation of objectivity should include inquiry regarding interests and relationships that may create a threat to the objectivity of the auditor's specialist. Furthermore, the auditor should obtain a sufficient understanding of the field of expertise of the auditor's specialist to enable the auditor to
  - a. determine the nature, scope, and objectives of the work of the auditor's specialist for the auditor's purposes and
  - b. evaluate the adequacy of that work for the auditor's purposes.

Further application and explanatory material regarding the competence, capabilities, and objectivity, as well as understanding the field of expertise, of the auditor's specialist can be found in paragraphs .A15—.A24 of AU-C section 620.

- **9.91** The auditor should evaluate the adequacy of the work of the auditor's specialist for the auditor's purposes, including
  - a. the relevance and reasonableness of the findings and conclusions of the auditor's specialist and their consistency with other audit evidence.
  - b. if the work of the auditor's specialist involves the use of significant assumptions and methods,
    - obtaining an understanding of those assumptions and methods and
    - ii. evaluating the relevance and reasonableness of those assumptions and methods in the circumstances, giving consideration to the rationale and support provided by the specialist, and in relation to the auditor's other findings and conclusions.
  - c. if the work of the auditor's specialist involves the use of source data that is significant to the work of the auditor's specialist, the relevance, completeness, and accuracy of that source data.
- **9.92** If the auditor determines that the work of the auditor's specialist is not adequate for the auditor's purposes, paragraph .13 of AU-C section 620 states that the auditor should
  - a. agree with the auditor's specialist on the nature and extent of further work to be performed by the auditor's specialist or
  - b. perform additional audit procedures appropriate to the circumstances (such as selecting and hiring another specialist).
- **9.93** Large groups of loans. For loans that are pooled for purposes of measuring the allowance for loan losses, the focus of testing is not on individual loan files, and the collectibility of individual loans is not tested directly. Rather, the auditor generally reviews and tests for compliance with the institution's charge-off and nonaccrual policy and tests the completeness and accuracy of historical data and reports, such as delinquency reports, that are relied upon in estimating the allowance for such loans.

- **9.94** For example, loan categories represented by large volumes of relatively small loans with similar characteristics, such as residential real estate mortgages, consumer loans, and credit-card loans, are generally evaluated on an aggregate, or pool, basis. The auditor is generally more concerned with the effectiveness of and adherence to procedures related to assessing the collectibility of such loans as a group rather than the collectibility of each individual loan. The testing or procedures and the review of delinquency status reports may permit the auditor to draw a conclusion about the appropriateness in accordance with GAAP of the allowance necessary for those loan categories. In evaluating the appropriateness in accordance with GAAP of the portion of the allowance attributable to those loans, use of unadjusted historical annual charge off experience may not be sufficient in itself but could be considered in light of consistent application of loan policies, and current economic conditions at the balance sheet date.
- **9.95** *Individually evaluated loans*. In contrast to large groups of smaller balance homogeneous loans, an evaluation of larger balance loans (for example, commercial loans) generally requires a more detailed review because of the loan size and the fact that the type of borrower, the purpose of the loan, and the timing of cash flows may be dissimilar from loan to loan. More important, a relatively small number of credit losses on larger balance loans can significantly affect the appropriateness of the allowance in accordance with GAAP. In these circumstances, the auditor will generally select and review in detail a number of those loans. The auditor's review of loans may be performed as a dual purpose test (test of controls and substantive attribute test).
- **9.96** In addition to identified problem loans, the auditor may select other commercial loans to include in the detailed loan file review. The selection of these additional loans generally includes a stratum of large loan balances above specified limits, loans from other sources (such as related parties and industry concentrations), and some loans selected without regard to size or other specific criteria. The auditor generally will be concerned with the total credit exposure of the borrower, including standby letters of credit and other commitments to lend, rather than with individual loan balances. Based on the auditor's evaluations and tests, the number of loans reviewed might be limited when the internal loan review function is deemed effective in identifying and classifying problem credits.
- **9.97** The extent of an individual loan review varies from loan to loan. For example, a loan that has been subjected to a recent management review, an effective internal review, or a recent regulatory review may be reviewed in less detail than a loan that has not had some or all of those reviews.
- **9.98** The exemption of certain loans from examiner review and criticism pursuant to the March 30, 1993, regulatory policy (see paragraph 9.33) does not extend to management's financial reporting responsibilities or to the auditor's responsibility in financial statement audits or other engagements involving management assertions about the exempt loans. An institution's exempt portfolio could be material to its financial statements. An auditor's assessment of management assertions about the allowance for loan losses may depend on the availability of certain documentation, including adequate collateral appraisals or current and complete financial information about borrowers or guarantors. The March 1993 policy may affect the availability of such documentation. Auditors are cautioned against undue reliance on management representations when no supporting evidence exists.

**9.99** For each loan selected for review, the auditor may prepare a loan review worksheet or other memoranda documenting the procedures performed and summarizing the conclusions reached. Exhibit 9-2, "Sample Loan Review Form," is an example of a loan review form that could be used for a commercial loan. It can also be adapted to other types of loans. For loans reviewed previously, the auditor typically updates prior reviews for new information concerning the loan. In addition, the auditor usually reviews correspondence updating classified loans, working papers prepared by the institution's internal loan review personnel, and any regulatory examination reports (including those with information on shared national credits). Such data often provide additional information concerning the loan and how management considered the loan in determining the allowance for loan losses.

303

# Exhibit 9-2 Sample Loan Review Form

Client:					
Audit Date:					
Borrower's Name:					
Nature of Business	s:				
Purpose of Loan: _					
I. Borrower's Note	Effective	Direct	Line of Credit/	Outsta	ndings
Description	Interest Rate	Loan or Participation	Commitment Amount	$\frac{\overline{Principal}}{}$	Interest
Total loans outstanding at preliminary					
Total loans outstanding at year-end			_ / /		
Accrual basis (Y/N	)				
Repayment Sche	dule:				
Indicate probable r	epayment sche	edule if different	from contractual	l schedule.	
Approach used to e	stimate impair	ment (check on	e):		
Present value	of cash flows				
Fair value of o	collateral				
Market value	of loan				
Repayment Statu	ıs:				
		Principal		Interest	
Amount past due					
Last payment:					
Date					
Amount			<del></del>		
II. Contingencies/ with recourse)	Guarantees (1	for example, le	etters of credit,	participatio	ns sold
	Total at prel	iminary			
	Total at year	r-end			
III. Related Loans	<b>;</b>				
		Maturity	Commitment		
Obligor	$\underline{Relationship}$	Date	Amount	Outsta	ndings
	Total Rel	ated Loans			

### IV. Collateral Summary

Description	Gross Value	Prior Liens	Value to Lender	Basis for a Valuation (f appro market val	or example iisal,
Total					
V. Guarantors					
VI. Loan Grade					
	Regulatory		$Institution's\ In\text{-}House$		
classification		Amount	Classification	Amount	
Special mention					
Substandard					
Doubtful					
Loss					
Unclassified					
Total					
Is the loan impaire	ed not defined in	FASB ASC 310	0-10 (Y/N)?		
Has the loan been restructuring?	modified and if	so, does the mod	lification constitu	te a troubled d	lebt
VII. Financial D	ata				
Auditors:					
Type of opinion:		Last audit da	te:		
		Interim		$Fiscal\ Year$	
		Current-Yearmonths ended/_/	Prior-Yearmonths ended/_/	Current Year	Prior Year / /
Current assets					
Current liabiliti	es				
Working capital					
Total assets					
Total liabilities					

©2016, AICPA

Net worth Net sales Net income Cash flow

VIII. Loan Officer	
Comments	

Provide a narrative analysis prepared by [or through inquiry of] the loan officer of collectibility including estimated repayment dates, sources of repayment, appropriateness of collateral in accordance with U.S. generally accepted accounting principles to cover outstanding principal and interest, financial data on guarantors, and rationale for any estimated allowance allocation, charge-off or both

incipal and interest, financial data on guarantors, and rationale for any wance allocation, charge-off, or both.
estimated specific allowance allocation, charge-off, or both and 's supporting rationale:
Summary
n (including the amount and basis for auditor's s exposure)

- **9.100** For many loans, the auditor should discuss the status and background of the loans reviewed with the responsible loan officer and the loan review officer. In addition to providing information about the loans, such discussions may provide the auditor with information about the loan officer's and loan review officer's attitudes and degree of awareness of the status of loans and controls.
- **9.101** In reviewing individual loans, the auditor could review the institution's analysis of the borrower's financial resources, liquidity and future cash flows, and other financial forecasts, particularly for unsecured loans for which repayment is dependent on the borrower's ability to generate funds from profitable operations. The auditor might consider measuring such financial data against the trends and norms, both historical and forecasted, for both the borrower being reviewed and the industry in which the borrower operates. It is preferable that the institution's analysis be supported by current audited financial statements, although financial statements that have been reviewed or compiled by the borrower's accountant or prepared internally by the borrower may be useful.
- **9.102** *Collateral review.* Assessment of collateral is important to the classification or grading of the loan portfolio as well as the measurement of credit losses when the loan, individually evaluated for impairment, is deemed collateral dependent. For loans secured by collateral, a careful evaluation and valuation of that collateral is often necessary. In such circumstances, the auditor may evaluate the security interest in the collateral to determine how the institution knows that it has been perfected by execution and recording of the appropriate legal documents. The auditor could also consider the reasonableness of the institution's collateral valuation by referring to quoted market prices or other pertinent sources, such as a specialist's appraisal.
- **9.103** The auditor may test the existence of the collateral by physical observation, independent confirmation, or other appropriate procedures, especially when the institution is involved in loans secured by marketable securities or in asset-based lending, which may include loans secured by inventories, equipment, or receivables. For collateral in the form of marketable securities, the auditor may evaluate whether such securities are under the institution's

control, either in its own vault or in a safekeeping account in the institution's name maintained with an independent, third-party custodian. In the latter case, the auditor may wish to evaluate the independent custodian's ability to perform under its obligation. For other types of collateral, there should be documentation that the institution has verified the existence of the collateral. In the absence of such documentation, the auditor should perform these or other collateral verification procedures, especially for significant loans for which collectibility is otherwise questionable. The auditor should also consider the potential accounting consequences of collateral arrangements under FASB ASC 860, Transfers and Servicing.

**9.104** *Personal guarantees.* For loans supported by personal guarantees, the auditor may perform a review solely of the borrower's ability to pay. However, if the review indicates the guarantor may be a source of repayment, the auditor might review the financial statements and other pertinent information about the guarantor as if the guarantor were the borrower. It is also important to consider the extent of, as well as the institution's policies and practices for, pursuing guarantees and to evaluate, perhaps in consultation with an attorney, the enforceability and scope of the guarantee.

**9.105** The substance of a personal guarantee depends on (a) the ability and willingness of the guarantor to perform under the guarantee, including a determination of whether the guarantor has other guarantees outstanding that might be pursued, (b) the practicality of enforcing the guarantee in the applicable jurisdiction, (c) the scope of the guarantee (that is, whether it covers all principal and interest or has a limit), and (d) a demonstrated intent by the institution to enforce the guarantee. Even if the guarantee is legally enforceable, the auditor should consider making a determination concerning whether there are business reasons that might preclude the institution from enforcing the guarantee. Those business reasons could include the length of time necessary to enforce a guarantee, whether it is normal business practice to enforce guarantees on similar transactions, or whether it is necessary for the institution to choose between pursuing the guarantee or the underlying collateral, instead of pursuing both. See paragraphs 21–25 of FASB ASC 310-10-25 for additional information.

**9.106** Participation.<sup>8</sup> Management should have the information necessary to authorize, monitor, and review participation loans and to estimate

(continued)

<sup>&</sup>lt;sup>8</sup> As stated in the "Glossary" section of the Federal Financial Institutions Examination Council's Instructions for Preparation of Consolidated Reports of Condition and Income, a loan secured by real estate is a loan that, at origination, is secured wholly or substantially by a lien or liens on real property for which the lien or liens are central to the extension of the credit—that is, the borrower would not have been extended credit in the same amount or on terms as favorable without the lien or liens on real property. To be considered wholly or substantially secured by a lien or liens on real property, the estimated value of the real estate collateral at origination (after deducting any more senior liens held by others) must be greater than 50 percent of the principal amount of the loan at origination. Banks need not reevaluate and, if appropriate, recategorize loans originated before April 1, 2009, that they reported as loans secured by real estate into other loan categories on Schedule RC-C, part I, Loans and Leases.

When a bank acquires either (1) a portion of an entire loan that does not meet the definition of a participating interest (such as, a nonqualifying loan participation) or (2) a qualifying participating interest in a transfer that does not meet all of the conditions for sale accounting, the general instructions for Schedule RC-C, part I, state that the bank should normally report the loan participation or participating interest in Schedule RC, item 4.b, "Loans and leases, net of unearned income." The bank also should report the loan participation or participating interest in Schedule RC-C, part I, in

Credit Losses 307

any related allowance for loan losses. The collectibility of participation loans (whether at the lead institution or at a participating institution) is normally evaluated in light of the entire amount of the loan, not just of the share held by the institution. Accordingly, the participating institution should supplement documentation by the lead institution with its own investigation and credit analysis. The participating institution should not rely solely on the lead institution to monitor the credit. Certain large participation arrangements are reviewed by regulators, who issue a shared national credit report detailing their classification and rationale to the lead and all participating institutions. The auditor's objectives in testing loans for a participating institution are the same as for other loans. For example, the repayment status, borrowers' financial statements, and appraisals should be considered.

- **9.107** The auditor usually confirms the existence and terms of significant participations (both purchased and sold) with the debtor and lead institution. In addition, the auditor normally reviews the related loan file documentation. For participations, the loan files should contain the same information as other loan files.
- **9.108** Charge-offs and recoveries. The auditor should test the propriety of charge-offs and recoveries. Substantive detail testing in this area may be minimized if tests of controls and analytical procedures on charge-offs and recoveries are performed.
- **9.109** Analytical procedures. AU-C section 520, Analytical Procedures (AICPA, Professional Standards), addresses the auditor's use of analytical procedures as substantive procedures (substantive analytical procedures). The auditor should consider analytical procedures as a supplement to the detailed tests of the reasonableness of the allowance. These analytical tests may use statistics relating to the allowance as compared to related income statement accounts, net charge-off rates, nonperforming loan levels and other loan categories, historical experience, and peer results. Various analytical techniques can be utilized to assist the auditor in determining the appropriateness in accordance with GAAP of the allowance. See chapter 5 of this guide for additional guidance regarding analytical procedures.
- **9.110** Conclusions. In accordance with paragraph .18 of AU-C section 540, the auditor should evaluate, based on the audit evidence, whether the allowance for loan losses in the financial statements is either reasonable in the context of the applicable financial reporting framework or is misstated. Based on the audit evidence obtained, paragraph .A122 of AU-C section 540 goes on to state that the auditor may conclude that the evidence points to an accounting estimate of the allowance that differs from management's point estimate. When the audit evidence supports a point estimate, the difference between

 $(footnote\ continued)$ 

the loan category appropriate to the underlying loan (for example, as a "commercial and industrial loan" in item 4 or as a "loan secured by real estate" in item 1).

Considering the actual mechanics of the participation is important for management. Management should consider how legal title changes in the event of foreclosure. For example, depending on the facts and circumstances, each bank may have a perfected interest if the other real estate owned is held in trust or a separate legal entity or the lead bank may have the sole legal interest in the property. In addition, management should consider the following: (a) whether an asset and liability should be recorded, (b) whether the account will reflect the lower of cost or fair value, and (c) if a liability is recorded, when will the liability be extinguished.

the auditor's point estimate and management's point estimate constitutes a misstatement. When the auditor has concluded that using the auditor's range provides sufficient appropriate audit evidence, a management point estimate that lies outside the auditor's range would not be supported by audit evidence. In such cases, the misstatement is no less than the difference between management's point estimate and the nearest point of the auditor's range. (Note that this only applies to situations in which the auditor has chosen to test the reasonableness of the institution's allowance for loan losses by developing a point estimate or range, as opposed to testing how management made the accounting estimate, in accordance with paragraph .13 of AU-C section 540. AU-C section 450, Evaluation of Misstatements Identified During the Audit (AICPA, Professional Standards), provides guidance on distinguishing misstatements for purposes of the auditor's evaluation of the effect of uncorrected misstatements on the financial statements. With regard to accounting estimates, a misstatement, whether caused by fraud or error, may arise as a result of

- misstatements about which no doubt exists (factual misstatements).
- differences arising from management's judgments concerning accounting estimates that the auditor considers unreasonable or the selection or application of accounting policies that the auditor considers inappropriate (judgmental misstatements).
- the auditor's best estimate of misstatements in populations involving the projection of misstatements identified in audit samples to the entire population from which the samples were drawn (projected misstatements).

In some cases involving accounting estimates, a misstatement could arise as a result of a combination of these circumstances, making separate identification difficult or impossible.

**9.111** If the auditor deems the financial information inadequate, the auditor may discuss the situation with an appropriate member of management. In accordance with paragraph .A38 of AU-C section 260, The Auditor's Communication With Those Charged With Governance (AICPA, Professional Standards), this discussion may clarify facts and issues and give management an opportunity to provide further information and explanations. For example, the discussion may include missing information that may be evidence of an internal control deficiency as well. The results of such discussion or the inability of the institution to obtain financial information that is appropriate in accordance with GAAP should be considered in evaluating the collectibility of the loan. In accordance with paragraph .12 of AU-C section 260, the auditor should communicate with those charged with governance9 significant difficulties, if any, encountered during the audit (for example, appropriate financial information in accordance with GAAP that is not available for significant loans). In some circumstances, paragraph .A26 of AU-C section 260 explains that such difficulties may constitute a scope limitation that leads to a modification of the auditor's opinion.

<sup>&</sup>lt;sup>9</sup> For purposes of GAAS, those charged with governance are defined as the person(s) or organization(s) with responsibility for overseeing the strategic direction of the entity and the obligations related to the accountability of the entity. This includes overseeing the financial reporting process. They may include management personnel, for example, executive members of a governance board or an owner-manager.

**9.112** Furthermore, during their examinations of depository institutions, regulators focus a great deal of attention on the allowance for loan losses. Failure to maintain an adequate allowance is considered an unsafe or unsound practice.

©2016, AICPA AAG-DEP 9.112

# Chapter 10

# Transfers and Servicing and Variable Interest Entities

Gray shaded text in this chapter reflects guidance issued but not yet effective as of the date of this guide, July 1, 2016, but becoming effective on or prior to December 31, 2016, exclusive of any option to early adopt ahead of the mandatory effective date. Unless otherwise indicated, all unshaded text reflects guidance that was already effective as of the date of this guide.

### © Update 10-1 Accounting and Reporting: Credit Losses

FASB Accounting Standards Update (ASU) No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, issued in June 2016, is effective for fiscal years of public business entities that are SEC filers beginning after December 15, 2019, including interim periods within those fiscal years.

For all other public business entities, the amendments in FASB ASU No. 2016-13 are effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years.

For all other entities, including not-for-profit entities and employee benefit plans within the scope of FASB Accounting Standards Codification (ASC) 960, Plan Accounting—Defined Benefit Pension Plans, through FASB ASC 965, Plan Accounting—Health and Welfare Benefit Plans, on plan accounting, FASB ASU No. 2016-13 is effective for fiscal years beginning after December 15, 2020, and interim periods within fiscal years beginning after December 15, 2021.

Early application is permitted for all entities as of the fiscal years beginning after December 15, 2018, including interim periods within those fiscal years.

FASB ASU No. 2016-13 creates FASB ASC 326, Financial Instruments— Credit Losses, to amend guidance on reporting credit losses for assets held at amortized cost basis and available-for-sale debt securities.

For assets held at amortized cost basis, FASB ASC 326 eliminates the probable initial recognition threshold in current U.S. generally accepted accounting principles (GAAP) and, instead, requires an entity to reflect its current estimate of all expected credit losses. The allowance for credit losses is a valuation account that is deducted from the amortized cost basis of the financial assets to present the net amount expected to be collected.

For available-for-sale debt securities, credit losses should be measured in a manner similar to current GAAP. However, FASB ASC 326 will require that credit losses be presented as an allowance rather than as a write-down.

FASB ASU No. 2016-13 affects entities holding financial assets and net investment in leases that are not accounted for at fair value through net income. The amendments affect loans, debt securities, trade receivables, net investments in leases, off-balance-sheet credit exposures, reinsurance receivables, and any other financial assets not excluded from the scope that have the contractual right to receive cash.

©2016, AICPA AAG-DEP 10

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this chapter will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-13, see appendix G, "Accounting for Financial Instruments," of this guide.

### Introduction

10.01 This chapter discusses transfers of financial assets and servicing of financial assets (including mortgage banking) and variable interest entities (VIEs) which are often used to facilitate a transfer of financial assets. Common transfers of financial assets for financial institutions include whole loan sales, loan participations, asset securitizations, securities lending transactions, repurchase agreements, and banker's acceptances. Such transfers may result in the seller having continuing involvement with the transferred financial assets, such as servicing rights, guarantees, retained interests in the assets, indemnity clauses, representations and warranties, or beneficial interests in securitized assets. Repurchase agreements are discussed in further detail in chapter 14 of this guide.

10.02 Financial institutions often have mortgage banking activities that consist primarily of the purchase or origination of mortgage loans for sale to secondary market investors and the subsequent servicing of those loans. Mortgage loans can be sold outright through whole loan sales or pooled and securitized. Access to the secondary mortgage market is an important source of liquidity for banks and savings institutions. Many institutions have deposit bases that are indexed to variable rates and, therefore, are particularly sensitive to interestrate risk. Financial institutions with long term, fixed-rate assets funded with variable-rate deposit bases may incur losses in a rising interest-rate environment. Therefore, sales of mortgage loans and, in some cases, servicing rights and retained beneficial interests in the secondary market are an important source of funds to many institutions, supplemented by income streams from servicing and other fees. Access to the secondary market also provides opportunities to restructure existing long term portfolios.

10.03 Participants in the secondary market for residential financing include government-sponsored entities (GSEs) such as Federal Home Loan Mortgage Corporation (Freddie Mac) and Federal National Mortgage Association (Fannie Mae); and federal agencies such as Government National Mortgage Association (Ginnie Mae) and the U.S. Department of Veterans' Affairs (VA). These entities participate in the secondary market as issuers, investors, or guarantors of asset-backed securities (ABSs) such as mortgage-backed securities (MBSs), real estate mortgage investment conduits, and collateralized mortgage obligations. (Chapter 7, "Investments in Debt and Equity Securities," of this guide describes ABS transactions and considerations for investors in ABSs.) Many entities are also active in the secondary market as issuers, investors, servicers, and financial guaranty companies.

10.04 When mortgage loans are originated for sale, the process includes not only finding an investor but also identifying loans that fit the investor's requirements, including underwriting criteria. Mortgage loans originated for sale normally must comply with specific standards governing documentation,

appraisal, mortgage insurance, loan terms, and borrower qualifications. Investors typically review key loan terms and underlying documentation prior to completing their purchase. Individual loans that fail to meet the specified criteria are eliminated from the pool of loans eligible for sale. If exceptions cannot be corrected, the selling institution may have to either find alternative investors or retain the loans in its portfolio. In most cases, the originating institution continues to be subject to recourse by the investor for underwriting exceptions identified subsequent to the sale of the loans and any related defaults by borrowers. These obligations typically arise from the representations and warranties included in the pooling and servicing agreements.

10.05 The extent to which mortgage loans are originated for sale will differ for each institution. Factors such as liquidity, interest-rate exposure, asset/liability management policy, business plan, and capital considerations will influence an institution's mortgage banking activities. One institution may manage its interest-rate risk position by intentionally selling all fixed-rate mortgage loans it originates, whereas another institution may originate a variety of both fixed- and variable-rate loan products for sale in the secondary market.

### **Asset-Backed Securitizations**

10.06 Securitization is often utilized by lending institutions to diversify funding sources. It involves the sale, generally to a trust, of a portfolio of loan receivables. Asset-backed certificates are then sold by the trust to investors through a private placement or public offering. Typically, the seller will retain the servicing rights for the loans sold to the trust and often retains a subordinated interest in the trust , which serves as a credit enhancement to the asset-backed certificates. Such structures provide companies the opportunity to obtain funding at competitive levels given the structural characteristics of securitization vehicles.

# Loan Participations and Loan Syndications

10.07 In certain situations, a customer's borrowing needs may exceed its bank's legal lending limits or its self-imposed counterparty limits. To accommodate such a customer, the bank may participate the loan to other banks. A loan participation, as defined in the FASB ASC glossary, is a transaction in which a single lender makes a large loan to a borrower and subsequently transfers undivided interests in the loan to groups of banks or other entities. Transfers by the originating lender may take the legal form of either assignments or participations. The transfers are usually on a nonrecourse basis, and the transferor (originating lender) continues to service the loan and is the lender of record. The transfere (participating entity) may or may not have the right to sell or transfer its participation during the term of the loan, depending on the terms of the participation agreement. A loan participation is in the scope of FASB ASC 860, Transfers and Servicing. In contrast, a loan

©2016. AICPA AAG-DEP 10.07

<sup>&</sup>lt;sup>1</sup> Under FASB Accounting Standards Codification (ASC) 860, Transfers and Servicing, so-called last in, first out (LIFO) participations, in which all principal cash flows collected on the loan are paid first to the party acquiring the participation, do not meet the definition of a participating interest (see FASB ASC 860-10-40-6A for detailed guidance on characteristics of a participating interest). Similarly, so-called first in, first out (FIFO) participations, in which all principal cash flows collected on the loan are paid first to the lead lender, do not meet the definition of a participating interest. As a result, neither LIFO nor FIFO participations qualify for sale accounting and, instead, must be reported as secured borrowings.

syndication, as defined in the FASB ASC glossary, is a transaction in which several lenders share in lending to a single borrower. In a loan syndication, each lender loans a specific amount to the borrower and has the right to repayment from the borrower. A loan syndication is not a transfer of financial assets because each participant in a syndication is a lender of record, and therefore the transaction is not in the scope of FASB ASC 860. The "Glossary" section of the Federal Financial Institutions Examination Council's *Instructions for Preparation of Consolidated Reports of Condition and Income* provides guidance that may assist in distinguishing between a loan participation and a loan syndication.

### **Loan Servicing**

10.08 If mortgage or other loans are sold, the selling institution sometimes retains the right to service the loans for a servicing fee, normally expressed as a percentage of the unpaid principal balance of the loans that is collected over the life of the loans as payments are received. A typical servicing agreement requires the servicer to carry out the servicing function, including billing and collection of borrowers' payments; remittance of payments to the investor, insurers, and taxing authorities; loss mitigation activities; maintenance of custodial bank accounts; and related activities. The agreement also may involve significant risks being retained by the servicer arising from failure to abide by the terms of the servicing agreement. These loans may have been originated by the servicer institution itself or by other financial institutions. When servicing mortgages for government entities (for example, Ginnie Mae) and GSEs (for example, Fannie Mae, Freddie Mac), institutions must meet certain minimum net-worth requirements and comply with the GSEs' servicing standards. Inability to meet the requirements may result in termination of the servicing contracts.

# **Regulatory Matters**

10.09 The Office of the Comptroller of the Currency (OCC), the FDIC, and the Board of Governors of the Federal Reserve System (Federal Reserve) (collectively, the federal banking agencies) limit the aggregate amount of servicing assets, which includes mortgage servicing assets, that may be included in regulatory capital.

10.10 On December 13, 1999, the federal banking agencies, including the Office of Thrift Supervision (prior to its transfer of powers to the OCC, the FDIC, and the Federal Reserve), jointly issued the *Interagency Guidelines on Asset Securitization* that highlight the risks associated with asset securitization and emphasize the agencies' concerns with certain retained interests generated from the securitization and sale of assets. The guidelines set forth the supervisory expectation that the value of retained interests in securitizations must be supported by objectively verifiable documentation of the assets' fair-market value, utilizing reasonable, conservative valuation assumptions. Retained interests that do not meet such standards or that fail to meet the supervisory standards outlined in the guidance will be disallowed as assets of the bank for

 $<sup>^2</sup>$  See chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide for further discussion on the Office of Thrift Supervision transfer of powers.

regulatory capital purposes. The guidance stresses the need for bank management to implement policies and procedures that include limits on the amount of retained interests that may be carried as a percentage of capital. Institutions that lack effective risk management programs or engage in practices deemed to present other safety and soundness concerns may be subject to more frequent supervisory review, limitations on retained interest holdings, more stringent capital requirements, or other supervisory response. On December 20, 2013, the Federal Reserve issued Risk Transfer Considerations When Assessing Capital Adequacy—Supplemental Guidance on Consolidated Supervision Framework for Large Financial Institutions, an advisory that provides guidance on how certain risk transfer transactions, such as securitizations, affect assessments of capital adequacy at large financial institutions. Readers are encouraged to view this guidance under the "Banking Information and Regulation—Supervision and Regulation Letters" page at www.federalreserve.gov.

- **10.11** On February 25, 2003, the federal banking agencies, including the OTS, issued the *Interagency Advisory on Mortgage Banking*. This guidance focuses on risks associated with valuation and modeling processes, hedging activities, management information systems, and internal audit processes in connection with mortgage banking activities.
- 10.12 On May 3, 2005, the federal banking agencies, including the OTS, and the National Credit Union Administration (NCUA) issued *Interagency Advisory on Accounting and Reporting for Commitments to Originate and Sell Mortgage Loans*. This advisory provides guidance related to the origination of mortgage loans that will be held for resale, and the sale of mortgage loans under mandatory delivery and best efforts contracts.
- 10.13 In September 2010, the FDIC approved a final rule to extend through December 31, 2010, the Safe Harbor Protection for Treatment by the FDIC as Conservator or Receiver of Financial Assets Transferred by an Insured Depository Institution in Connection With a Securitization or Participation. Under this safe harbor, all securitizations or participations in process before the end of 2010 were permanently grandfathered under the existing terms of Title 12 U.S. Code of Federal Regulations (CFR) Part 360.6. The rule defines the conditions for safe harbor protection for securitizations and participations for which transfers of financial assets are made after September 30, 2010, and clarifies the application of the safe harbor to transactions that comply with the accounting standards (that is, FASB Statement Nos. 166, Accounting for Transfers of Financial Assets—an amendment of FASB Statement No. 140 [FASB ASU No. 2009-16, Transfers and Servicing (Topic 860): Accounting for Transfers of Financial Assets, and 167, Amendments to FASB Interpretation No. 46(R) [FASB ASU No. 2009-17, Consolidations (Topic 810): Improvements to Financial Reporting by Enterprises Involved with Variable Interest Entities]) for off-balance-sheet treatment as well as those that do not comply with those accounting standards. The conditions contained in the final rule will serve to protect the Deposit Insurance Fund and the FDIC's interests as deposit insurer and receiver by aligning the conditions for the safe harbor with better and more sustainable securitization practices by insured depository institutions. For more information on this final ruling, readers are encouraged to visit the FDIC website at www.fdic.gov/news/news/press/2010/pr10216.html.
- 10.14 See chapter 9, "Credit Losses," of this guide for regulatory matters related to loan and lease losses and nontraditional mortgage products and

©2016, AICPA AAG-DEP 10.14

chapter 8, "Loans," of this guide for regulatory matters related to commercial and residential mortgages.

10.15 The OCC's Bank Accounting Advisory Series (BAAS) is updated periodically to express the Office of the Chief Accountant's current views on accounting topics of interest to national banks and federal savings associations. See further discussion of the BAAS in paragraph 7.82 of this guide. Topic 9, "Income and Expense Recognition," of the BAAS includes interpretations on accounting for transfers of financial assets and servicing. Readers are encouraged to view this publication under the "Publications—Bank Management" page at www.occ.gov.

# VA No-Bids and Private Mortgage Agencies

- **10.16** Historically, the VA paid lenders 100 percent of the outstanding debt on defaulted loans that the VA guaranteed. In return, the lenders turned the borrowers' residential properties over to the VA, which would dispose of them. The VA had the option of guaranteeing the lesser of 60 percent of a loan's original balance or \$27,500, leaving the property with the lender if that is less costly for the agency. Called a *no-bid option*, this practice was seldom used, especially because inflation pushed up housing prices during the late 1970s and early 1980s. However, as inflation began to slow and the costs of carrying foreclosed houses began to rise, the VA began to invoke the no-bid option.
- 10.17 On October 1, 2008, the VA issued Circular 26-08-19, Implementation of Loan Guarantee Provisions of Public Law 110-389. Section 501 of Public Law 110-389 provides a temporary increase in the maximum guaranty amount for loans closed January 1, 2009, through December 31, 2011. During this period, the "maximum guaranty amount" set forth in this circular should be substituted for the maximum guaranty amount specified at Veterans' Benefits, U.S. Code Title 38, Section 3703(a)(1)(C), Title 38 CFR Parts 36.4302(a)(4) and 36.4802(a)(4), and in the VA Lender's Handbook. The guaranty amount for loans remains unchanged for situations in which the original principal loan amount is \$417,000 or less. If the original principal loan amount is greater than \$417,000, the VA will guarantee 25 percent of the original principal loan amount, up to the maximum guaranty amount. The maximum guaranty amount varies depending upon the location of the property. In February 2009, the VA issued changes to Circular 26-08-19 to clarify entitlement calculations for certain veterans with previously used, unrestored entitlement. Readers are encouraged to refer to the U.S. Department of Veterans Affairs website for details regarding the maximum guarantee calculation.
- **10.18** Institutions may incur losses due to the uncollectibility of receivables from other government programs such as the Federal Housing Administration or Ginnie Mae, from other investors such as Freddie Mac and Fannie Mae, or from insolvent private mortgage insurers.
- 10.19 In an environment with increased risk of foreclosure losses (including unrecoverable servicer advances; foreclosure costs such as attorneys' fees, inspections, and so forth; and the implicit cost to carry the asset until ultimate sale), the evaluation of loss allowances on VA and privately insured mortgage loans becomes increasingly difficult. Chapter 11, "Real Estate Investments, Real Estate Owned, and Other Foreclosed Assets," of this guide provides guidance on the valuation of foreclosed real estate, and chapter 9 of this guide, provides guidance on the evaluation of the collectibility of real estate loans.

# Accounting and Financial Reporting<sup>3</sup>

# Mortgage Loans and MBSs Held for Sale

- **10.20** FASB ASC 948, *Financial Services—Mortgage Banking*, includes accounting and reporting guidance applicable to mortgage banking entities and entities that engage in certain mortgage banking activities.
- 10.21 FASB ASC 948-310-35-1 states that mortgage loans held for sale (HFS) should be reported at the lower of cost or fair value, determined as of the balance sheet date. If a mortgage loan has been the hedged item in a fair value hedge (as addressed in FASB ASC 815, Derivatives and Hedging), the loan's cost basis used in lower-of-cost-or-fair value accounting should reflect the effect of the adjustments of its carrying amount made pursuant to FASB ASC 815-25-35-1. After the securitization of a mortgage loan HFS that meets the conditions for a sale addressed in FASB ASC 860-10-40-5, FASB ASC 948-310-40-1 requires that any MBSs received by the transferor as proceeds should be classified in accordance with the provisions of FASB ASC 320, Investments— Debt and Equity Securities. However, FASB ASC 948-310-35-3A states that a mortgage banking entity should classify as trading any MBSs received as proceeds that it commits to sell before or during the securitization process. An entity is prohibited from reclassifying loans as investment securities unless the transfer of those loans meets the conditions for sale accounting addressed in FASB ASC 860-10-40-5.
- **10.22** FASB ASC 948-310-35-2 requires that the amount by which the cost exceeds fair value of mortgage loans HFS should be accounted for as a valuation allowance. Changes in the valuation allowance should be included in the determination of net income of the period in which the change occurs.
- **10.23** The valuation allowance addressed in paragraph 10.22 is not the same as the allowance for credit losses and thus the change should not be recognized in the provision for credit losses.
- 10.24 Many mortgage banking entities apply the fair value option to loans under FASB ASC 825, *Financial Instruments*. The "Fair Value Option" subsections of FASB ASC 825-10 address circumstances in which entities may choose, at specified election dates, to measure eligible items at fair value (the fair value option). FASB ASC 825-10-15 provides guidance on the scope of the "Fair Value Option" subsections. See chapter 20, "Fair Value," of this guide for a summary of FASB ASC 825.
- 10.25 The fair value of mortgage loans and MBSs HFS should be measured by type of loan, as explained in FASB ASC 948-310-35-3. At a minimum, the fair value of residential (one- to four-family dwellings) and commercial mortgage loans should be measured separately. Either the aggregate or individual loan basis may be used in determining the lower of cost or fair value for each type of mortgage loan. Fair value for loans subject to investor purchase commitments (committed loans) and loans held on a speculative basis (uncommitted loans) should be measured separately as follows:
  - a. Committed loans. Mortgage loans covered by investor commitments should be based on the fair values of the loans.

©2016, AICPA AAG-DEP 10.25

<sup>&</sup>lt;sup>3</sup> Guidance on matters related to fair value measurements and disclosures are provided in chapter 20, "Fair Value," of this guide and should be considered in connection with references to fair value in this chapter.

- b. Uncommitted loans. Fair value for uncommitted loans should be based on fair value in the principal market or, in the absence of a principal market, in the most advantageous market (see paragraphs 5–6C of FASB ASC 820-10-35). That determination relies on the principles in FASB ASC 820, Fair Value Measurement, and would include consideration of the following:
  - i. Market prices and yields sought by market participants in the principal or most advantageous market.
  - ii. Quoted Ginnie Mae security prices or other public market quotations for long-term mortgage loan rates.
  - iii. Freddie Mac and Fannie Mae current delivery prices.
- **10.26** If a loan is held for resale, loan origination fees and the direct loan origination costs as specified in FASB ASC 310, *Receivables*, should be deferred until the related loan is sold as stated in FASB ASC 948-310-25-3. However, for loans subject to the fair value option as discussed in paragraph 10.24, such amounts are recognized in the income statement immediately.
- 10.27 Notwithstanding the characteristics discussed in FASB ASC 815-10-15-83, loan commitments that relate to the origination of mortgage loans that will be HFS, as discussed in FASB ASC 948-310-25-3, should be accounted for as derivative instruments by the issuer of the loan commitment (that is, the potential lender), as stated in FASB ASC 815-10-15-71. The fair value of loan commitments accounted for as derivative instruments should include the inherent value of servicing, in accordance with SEC Staff Accounting Bulletin No. 109, "Written Loan Commitments Recorded at Fair Value Through Earnings" (codified in FASB ASC 815-10-S99-1).
- **10.28** FASB ASC 948-310-30-4 states that mortgage loans transferred from loans HFS to long-term-investment classification should be measured at the lower of cost or fair value on the date of transfer.
- 10.29 The carrying amount of mortgage loans to be sold to an *affiliated entity* (as defined in the FASB ASC glossary) should be adjusted to the lower of cost or fair value of the loans as of the date management decides that a sale to an affiliated entity will occur, as stated in paragraphs 1–2 of FASB ASC 948-310-30. If a particular class of mortgage loans or all loans are originated exclusively for an affiliated entity, the originator is acting as an agent of the affiliated entity, and the loan transfers should be accounted for at the *originator's acquisition cost*, as defined in the FASB ASC glossary.

### Transfers and Servicing of Financial Assets

- 10.30 FASB ASC 860 establishes accounting and reporting standards for transfers and servicing of financial assets. FASB ASC 860-10-05-6 provides examples of the various types of transfers to which FASB ASC 860 applies, which include but are not limited to securitizations, factoring arrangements, transfers of receivables with recourse, securities lending transactions, repurchase agreements, loan participations, and banker's acceptances.
- 10.31 FASB ASC 860 establishes accounting and reporting standards for transfers and servicing of financial assets, as well as establishes the accounting for transfers of servicing rights. Transfers of financial assets often occur in which the transferor has some continuing involvement either with the assets transferred or with the transferee. Transfers of financial assets with continuing

involvement raise issues about the circumstances under which the transfers should be considered as sales of all or part of the assets or as secured borrowings and about how transferors and transferees should account for sales and secured borrowings. *Continuing involvement* is defined in FASB ASC 860-10-20 and includes, but is not limited to, servicing rights, guarantees, retained interests in the loans, or beneficial interests in securitized assets.

- 10.32 The objective of the derecognition guidance in FASB ASC 860-10-40 and related implementation guidance is to determine whether a transferor and its consolidated affiliates included in the financial statements being presented have surrendered control over transferred financial assets or third-party beneficial interests. This determination
  - a. should first consider whether the transferee would be consolidated by the transferor (for implementation guidance, see FASB ASC 860-10-55-17D).
  - b. should consider the transferor's *continuing involvement* (as defined in the FASB ASC glossary) in the transferred financial assets.
  - c. requires the use of judgment that should consider all arrangements or agreements made contemporaneously with, or in contemplation of, the transfer, even if they were not entered into at the time of the transfer.

With respect to item *b*, all continuing involvement by the transferor, its consolidated affiliates included in the financial statements being presented, or its agents should be considered continuing involvement by the transferor. In a transfer between two subsidiaries of a common parent, the transferor subsidiary should not consider its parent's involvements with the transferred financial assets in applying FASB ASC 860-10-40-5.

- 10.33 According to "Pending Content" in FASB ASC 860-10-40-4D, to be eligible for sale accounting, an entire financial asset cannot be divided into components before a transfer unless all of the components meet the definition of a *participating interest*. The legal form of the asset and what the asset conveys to its holders should be considered in determining what constitutes an entire financial asset (for implementation guidance, see FASB ASC 860-10-55-17E). An entity should not account for a transfer of an entire financial asset or a participating interest in an entire financial asset partially as a sale and partially as a secured borrowing.
- **10.34** FASB ASC 860-10-40-6A provides guidance on the characteristics of a participating interest. Critical for qualifying as a participating interest is that the transferee receives from the date of the transfer a proportionate (pro rata) ownership interest in an entire financial asset.
- 10.35 According to "Pending Content" in FASB ASC 860-10-40-4E, if a transfer of a portion of an entire financial asset meets the definition of a participating interest, the transferor should apply the guidance in FASB ASC 860-10-40-5 (see paragraph 10.36). If a transfer of a portion of a financial asset does not meet the definition of a participating interest, the transferor and transferee should account for the transfer as a secured borrowing with a pledge of collateral, in accordance with the guidance in FASB ASC 860-30-25-2. However, if the transferor transfers an entire financial asset in portions that do not individually meet the participating interest definition, FASB ASC 860-10-40-5 (see paragraph 10.36) should be applied to the entire financial asset once all portions have been transferred.

- 10.36 "Pending Content" in FASB ASC 860-10-40-5 states that a transfer of an entire financial asset, a group of entire financial assets, or a participating interest in an entire financial asset in which the transferor surrenders control over those financial assets should be accounted for as a sale if and only if all of the following conditions are met:
  - a. Isolation of transferred financial assets. The transferred assets have been isolated from the transferor—put presumptively beyond the reach of the transferor and its creditors, even in bankruptcy or other receivership. Transferred financial assets are isolated in bankruptcy or other receivership only if the transferred financial assets would be beyond the reach of the powers of a bankruptcy trustee or other receiver for the transferor or any of its consolidated affiliates included in the financial statements being presented. For multiple step transfers, a bankruptcy-remote entity (as defined in the FASB ASC glossary) is not considered a consolidated affiliate for purposes of performing the isolation analysis. Notwithstanding the isolation analysis, each entity involved in the transfer is subject to the applicable guidance on whether it should be consolidated (see paragraphs 7-14 of FASB ASC 860-10-40 and the guidance beginning in FASB ASC 860-10-55-18 [see discussion of paragraphs 18–18C of FASB ASC 860-10-55 in paragraphs 10.39–.42]). A set-off right (as defined in the FASB ASC glossary) is not an impediment to meeting the isolation condition.
  - b. Transferee's rights to pledge or exchange. This condition is met if both the following conditions are met:
    - i. Each transferee (or, if the transferee is an entity whose sole purpose is to engage in securitization or asset-backed financing activities and that entity is constrained from pledging or exchanging the assets it receives, each thirdparty holder of its beneficial interests) has the right to pledge or exchange the assets (or beneficial interests) it received.
    - ii. No condition both constrains the transferee (or third-party holder of its beneficial interests) from taking advantage of its right to pledge or exchange and provides more than a trivial benefit to the transferor (see paragraphs 15–21 of FASB ASC 860-10-40).

If the transferor, its consolidated affiliates included in the financial statements being presented, and its agents have no continuing involvement with the transferred financial assets, the condition under item (b) in FASB ASC 860-10-40-5 is met.

- c. Effective control. The transferor, its consolidated affiliates included in the financial statements being presented, or its agents do not maintain effective control over the transferred assets or third-party beneficial interests related to those transferred assets (see FASB ASC 860-10-40-22A). A transferor's effective control over the transferred financial assets includes, but is not limited to, any of the following:
  - i. An agreement that both entitles and obligates the transferor to repurchase or redeem the transferred financial

- assets before their maturity (see paragraphs 23–25 of FASB ASC 860-10-40).
- ii. An agreement, other than through a cleanup call (see paragraphs 28–39 of FASB ASC 860-10-40), that provides the transferor with both the unilateral ability to cause the holder to return specific financial assets and a more-thantrivial benefit attributable to that ability.
- iii. An agreement that permits the transferee to require the transferor to repurchase the transferred financial assets at a price that is so favorable to the transferee that it is probable that the transferee will require the transferor to repurchase them (see FASB ASC 860-10-55-42D).
- 10.37 For guidance on accounting for a transfer that satisfies the conditions in FASB ASC 860-10-40-5, see FASB ASC 860-20 (sales of financial assets), including derecognition guidance in FASB ASC 860-20-40 and guidance on recognition of any assets obtained and liabilities incurred in FASB ASC 860-20-25. For guidance on accounting for a transfer that does not satisfy the conditions of FASB ASC 860-10-40-5, see FASB ASC 860-30 (secured borrowings and collateral).
- 10.38 Paragraphs 16–18 of FASB ASC 860-10-05 provide background on securities lending transactions and paragraphs 19–21 of FASB ASC 860-10-05 provide background on repurchase agreements. Implementation guidance related to the application of FASB ASC 860 to repurchase agreements and securities lending transactions is discussed in paragraphs 51–56B of FASB ASC 860-10-55. Implementation guidance related to the application of FASB ASC 860 to wash sales is discussed in FASB ASC 860-10-55-57.
- **10.39** Isolation of transferred financial assets. Paragraphs 18–25A of FASB ASC 860-10-55 provide implementation guidance related to the isolation condition in item (a) in FASB ASC 860-10-40-5.
- **10.40** In the context of U.S. bankruptcy laws, a true sale opinion from an attorney is often required to support a conclusion that transferred financial assets are isolated from the transferor, any of its consolidated affiliates included in the financial statements being presented, and its creditors. In addition, a nonconsolidation opinion is often required if the transfer is to an affiliated entity. In the context of U.S. bankruptcy laws,
  - a. a true sale opinion is an attorney's conclusion that the transferred financial assets have been sold and are beyond the reach of the transferor's creditors and that a court would conclude that the transferred financial assets would not be included in the transferor's bankruptcy estate.
  - b. a nonconsolidation opinion is an attorney's conclusion that a court would recognize that an entity holding the transferred financial assets exists separately from the transferor. Additionally, a nonconsolidation opinion is an attorney's conclusion that a court would not order the substantive consolidation of the assets and liabilities of the entity holding the transferred financial assets and the assets and liabilities of the transferor (and its consolidated affiliates included in the financial statements being presented) in the event of the transferor's bankruptcy or receivership.

- **10.41** A legal opinion may not be required if a transferor has a reasonable basis to conclude that the appropriate legal opinion(s) would be given if requested. For example, the transferor might reach a conclusion without consulting an attorney if either of the following conditions exist:
  - a. The transfer is a routine transfer of financial assets that does not result in any continuing involvement by the transferor.
  - b. The transferor had experience with other transfers with similar facts and circumstances under the same applicable laws and regulations.
- 10.42 For entities that are subject to other possible bankruptcy, conservatorship, or other receivership procedures (for example, banks subject to receivership by the FDIC in the United States or other jurisdictions), judgments about whether transferred financial assets have been isolated should be made in relation to the powers of bankruptcy courts or trustees, conservators, or receivers in those jurisdictions.

### Sale of Financial Assets

- **10.43** Sale of a participating interest. Upon completion of a transfer of a participating interest that satisfies the conditions in FASB ASC 860-10-40-5 to be accounted for as a sale, FASB ASC 860-20-40-1A states that the transferor (seller) should
  - a. allocate the previous carrying amount of the entire financial asset between both of the following on the basis of their relative fair values at the date of the transfer:
    - i. The participating interest(s) sold
    - ii. The participating interest that continues to be held by the transferor
  - b. derecognize the participating interest(s) sold.
  - c. apply the guidance in FASB ASC 860-20-25-1 and 860-20-30-1 on recognition and measurement of assets obtained and liabilities incurred in the sale.
  - d. recognize in earnings any gain or loss on the sale.
  - e. report any participating interest(s) that continue to be held by the transferor as the difference between the following amounts measured at the date of the transfer:
    - i. The previous carrying amount of the entire financial asset
    - ii. The amount derecognized
- 10.44 Sale of an entire financial asset or group of entire financial assets. Upon completion of a transfer of an entire financial asset or group of entire financial assets that satisfies the conditions in FASB ASC 860-10-40-5 to be accounted for as a sale, FASB ASC 860-20-40-1B states that the transferor (seller) should
  - *a.* derecognize the transferred financial assets.
  - b. apply the guidance in FASB ASC 860-20-25-1 and 860-20-30-1 on recognition and measurement of assets obtained and liabilities incurred in the sale.
  - c. recognize in earnings any gain or loss on the sale.

If the transferred financial asset was accounted for under FASB ASC 320 as available for sale before the transfer, item a requires that the amount in other comprehensive income be recognized in earnings at the date of the transfer.

- 10.45 Transferor recognition and measurement of assets obtained and liabilities incurred in a sale of financial assets. Upon completion of a transfer of financial assets that satisfies the conditions to be accounted for as a sale in FASB ASC 860-10-50-5, FASB ASC 860-20-25-1 states the transferor (seller) should also recognize any assets obtained or liabilities incurred in the sale, including, but not limited to, any of the following:
  - a. Cash
  - b. Servicing assets
  - c. Servicing liabilities
  - d. In a sale of an entire financial asset or a group of entire financial assets, any of the following:
    - The transferor's beneficial interest in the transferred financial assets
    - ii. Put or call options held or written (for example, guarantee or recourse obligations)
    - Forward commitments (for example, commitments to deliver additional receivables during the revolving periods of some securitizations)
    - iv. Swaps (for example, provisions that convert interest rates from fixed to variable)
- **10.46** According to FASB ASC 860-20-30-1, the transferor should initially measure at fair value any asset obtained (or liability incurred) and recognized under FASB ASC 860-20-25-1.
- 10.47 Transferee recognition and measurement of assets obtained and liabilities incurred in a sale of financial assets. In accordance with FASB ASC 860-20-25-3 and FASB ASC 860-20-30-2, the transferee should recognize and initially measure all assets obtained (including any participating interest[s] obtained) and any liabilities incurred at fair value.
- 10.48 Financial assets subject to prepayment. FASB ASC 860-20-35-2 requires that financial assets, except for instruments that are within the scope of FASB ASC 815-10, that can contractually be prepaid or otherwise settled in such a way that the holder would not recover substantially all of its recorded investment should be subsequently measured like investments in debt securities classified as available-for-sale or trading under FASB ASC 320. Examples of such financial assets include, but are not limited to, interest-only strips, other beneficial interests, loans, or other receivables.

### Transfers of Loans With Recourse

10.49 Institutions may transfer loans with recourse. This may be accomplished to deliver loans into a particular investor's commitment program, to obtain a better price, or both. For example, a seller may be obligated to make full or partial payment to the purchaser if the debtor fails to pay when payment is due. Similarly, a seller may be obligated to make payments to the purchaser as the result of loan prepayments or because of adjustments resulting from defects (such as failure to perfect a security interest in collateral) of

the transferred loans. In some cases (for example, student loans), underwriting exceptions identified subsequent to the transfer of loans may subject the seller to additional recourse risk if the borrower defaults on the loan. Because of the continuing risk of delinquency and foreclosure, the institution's management should carefully evaluate its potential contingent liabilities with respect to such loans.

10.50 FASB ASC 860 provides an overview of transfers of receivables with recourse. A transfer of receivables in their entireties with recourse should be accounted for as a sale, with the proceeds of the sale reduced by the fair value of the recourse obligation, if the conditions in FASB ASC 860-10-40-5 are met, as stated in item (a) in FASB ASC 860-10-55-46. Otherwise, a transfer of receivables with recourse should be accounted for as a secured borrowing. A transfer of a portion of a receivable with recourse, other than that permitted in item (c)(4) in FASB ASC 860-10-40-6A, does not meet the requirements of a participating interest and should be accounted for as a secured borrowing, as stated in item (b) in FASB ASC 860-10-55-46. The form and nature of recourse may have a bearing on legal isolation (see further discussion in paragraph 10.39).

# Servicing Assets and Liabilities

- **10.51** According to FASB ASC 860-50-25-1, an entity should recognize a servicing asset or servicing liability each time it undertakes an obligation to service a financial asset by entering into a servicing contract in any of the following situations:
  - a. A servicer's transfer of any of the following, if that transfer meets the requirements for sale accounting:
    - i. An entire financial asset
    - ii. A group of entire financial assets
    - iii. A participating interest in an entire financial asset, in which circumstance the transferor should recognize a servicing asset or a servicing liability only related to the participating interest sold
  - b. An acquisition or assumption of a servicing obligation that does not relate to financial assets of the servicer or its consolidated affiliates included in the financial statements being presented
- 10.52 Servicing rights are significant assets for some institutions. They have value in addition to the servicing fee value because of the servicer's ability to invest the "float" that results from payments that are received from borrowers but are not immediately passed on to the investors in the loans. Additionally, intrinsic value components of servicing rights may include ancillary income, such as late-payment charges and prepayment charges. Accordingly, servicing rights, either separately or as part of a loan, can be purchased and sold.
- **10.53** A servicer that transfers or securitizes financial assets in a transaction that does not meet the requirements for sale accounting and is accounted for as a secured borrowing with the underlying financial assets remaining on the transferor's balance sheet should not recognize a servicing asset or a servicing liability, as stated in FASB ASC 860-50-25-2.

- 10.54 FASB ASC 860-50-25-4 states that an entity that transfers its financial assets to an unconsolidated entity in a transfer that qualifies as a sale in which the transferor obtains the resulting securities and classifies them as debt securities held-to-maturity in accordance with FASB ASC 320 may either separately recognize its servicing assets or servicing liabilities or report those servicing assets or servicing liabilities together with the asset being serviced.
- **10.55** FASB ASC 860-50-30-1 states that an entity should initially measure at fair value, a servicing asset or servicing liability that qualifies for separate recognition regardless of whether explicit consideration was exchanged.
- **10.56** FASB ASC 860-50-35-1 states that an entity should subsequently measure each class of servicing assets and servicing liabilities using one of the following methods:
  - a. Amortization method. Amortize servicing assets or servicing liabilities in proportion to and over the period of estimated net servicing income (if servicing revenues exceed servicing costs) or net servicing loss (if servicing costs exceed servicing revenues), and assess servicing assets or servicing liabilities for impairment or increased obligation based on fair value at each reporting date.
  - b. Fair value measurement method. Measure servicing assets or servicing liabilities at fair value at each reporting date and report changes in fair value of servicing assets and servicing liabilities in earnings in the period in which the changes occur.
- **10.57** FASB ASC 860-50-35-2 states that the election to subsequently measure servicing assets and servicing liabilities by using either the amortization method or the fair value measurement method should be made separately for each class of servicing assets and servicing liabilities.
- **10.58** Paragraphs 4–5 of FASB ASC 860-50-35 state that an entity should apply the same subsequent measurement method to each servicing asset and servicing liability in a class. Classes of servicing assets and servicing liabilities should be identified based on any of the following:
  - a. The availability of market inputs used in determining the fair value of servicing assets or servicing liabilities.
  - b. An entity's method for managing the risks of its servicing assets or servicing liabilities.

Once an entity elects the fair value measurement method for a class of servicing assets and servicing liabilities, that election should not be reversed, as stated in item (a) in FASB ASC 860-50-35-3. However, at the beginning of each year, for servicing assets and liabilities measured at amortized cost, an entity may make an irrevocable decision to subsequently measure a class of servicing assets or liabilities at fair value. If this election is made the cumulative effect adjustment should be recorded to retained earnings at the beginning of the year.

- 10.59 Transfers of servicing rights. The criteria in paragraphs 2–4 of FASB ASC 860-50-40 apply to transfers of servicing rights relating to loans previously sold and to transfers of servicing rights relating to loans that are retained by the transferor, as stated in FASB ASC 860-50-40-6.
- **10.60** More recently, there has been an increase in bulk transfers of servicing (such as sales of servicing rights from one servicer to another). That activity has been driven by several factors, including changes in bank regulatory

capital requirements for mortgage servicing rights and changes in the tax treatment of certain servicing cash flows.

- **10.61** FASB ASC 860-50-40-2 states that the following criteria should be considered when evaluating whether a transfer of servicing rights qualifies as a sale:
  - Whether the transferor has received written approval from the investor if required.
  - Whether the transferee is a currently approved transferor/servicer and is not at risk of losing approved status.
  - If the transferor finances a portion of the sales price, whether an adequate nonrefundable down payment has been received (necessary to demonstrate the transferee's commitment to pay the remaining sales price) and whether the note receivable from the transferee provides full recourse to the transferee. Nonrecourse notes or notes with limited recourse (such as to the servicing) do not satisfy this criterion.
  - Temporary servicing performed by the transferor for a short period of time should be compensated in accordance with a subservicing contract that provides adequate compensation.
- **10.62** FASB ASC 860-50-40-3 states that the following criteria should also be considered when evaluating whether a transfer of servicing rights qualifies as a sale:
  - a. Title has passed
  - b. Substantially all risks and rewards of ownership have irrevocably passed to the buyer
  - c. Any protection provisions retained by the seller are minor and can be reasonably estimated
- 10.63 Servicing rights may be purchased by brokers or investment bankers that intend to seek buyers for the rights. Although such purchases cannot be canceled, approval of the transfer of the rights is not requested by the seller until the broker enters into a transaction with the third-party purchaser. Thus, such transactions should generally be characterized as financing transactions and a sale has not occurred until an approval of transfer of rights has been requested, even though other contingencies are resolved.
- **10.64** FASB ASC 860-50-40-4 explains that if a sale is recognized and minor protection provisions exist, a liability should be accrued for the estimated obligation associated with those provisions. The seller retains only minor protection provisions if both of the following are met:
  - a. The obligation associated with those provisions is estimated to be no more than 10 percent of the sales price.
  - b. Risk of prepayment is retained for no longer than 120 days.
- **10.65** FASB ASC 860-50-40-5 states that a temporary subservicing contract in which the subservicing will be performed by the transferor for a short period of time would not necessarily preclude recognizing a sale at the closing date.
- 10.66 Paragraphs 7–9 of FASB ASC 860-50-40 provide additional guidance on sales of servicing rights with a subservicing contract.

10.67 Paragraphs 10-11 of FASB ASC 860-50-40 address a situation in which an entity sells the right to service mortgage loans that are owned by other parties. The related mortgage loans have been previously sold, with servicing retained, in a separate transaction. Because of the ability to invest the float that results from payments received from borrowers but not yet passed to the owners of the mortgages, the mortgage servicing rights can be sold for immediate cash or for a participation in the future interest stream of the loans. If a transfer of mortgage servicing rights qualifies as a sale under the criteria stated in FASB ASC 860-50-40-2 and the sale is for a participation in the future interest income stream, gain recognition is appropriate at the sale date. There are difficulties in measuring the amount of the gain if the sales price is based on a participation in future payments and there is no specified upper limit on the computed sales price. The transferor of mortgage servicing rights should consider all available information, including the amount of gain that would be recognized if the servicing rights were to be sold outright for a fixed cash price.

10.68 In general, three to six months elapse between entry into a contract to sell servicing rights and actual delivery of the loan portfolio to be serviced. These delays may result from the purchaser's inability to accept immediate delivery, the seller's inability to immediately transfer the servicing records and loan files, difficulties in obtaining necessary investor approval, requirements to give advance notification to mortgagors, or other planning considerations. Issues relating to the transfer of risks and rewards between buyers and sellers of servicing rights may be complex.

10.69 Paragraphs 1–2 of FASB ASC 860-50-45 state that an entity should report recognized servicing assets and servicing liabilities that are subsequently measured using the fair value measurement method in a manner that separates those carrying amounts on the face of the statement of financial position from the carrying amounts for separately recognized servicing assets and servicing liabilities that are subsequently measured using the amortization method. To accomplish that separate reporting, an entity may either (a) display separate line items for the amounts that are subsequently measured using the fair value measurement method and amounts that are subsequently measured using the amortization method or (b) present the aggregate of those amounts that are subsequently measured at fair value and those amounts that are subsequently measured using the amortization method (see paragraphs 9–11 of FASB ASC 860-50-35) and disclose parenthetically the amount that is subsequently measured at fair value that is included in the aggregate amount.

# Secured Borrowings and Collateral

10.70 Paragraphs 2–3 of FASB ASC 860-30-05 state that a debtor (obligor) may grant a security interest in certain assets to a lender (the secured party) to serve as collateral for its obligation under a borrowing, with or without recourse to other assets of the obligor. An obligor under other kinds of current or potential obligations, for example, interest rate swaps, also may grant a security interest in certain assets to a secured party. If collateral is transferred to the secured party, the custodial arrangement is commonly referred to as a pledge. Secured parties sometimes are permitted to sell or repledge (or otherwise transfer) collateral held under a pledge. The same relationships occur, under different names, in transfers documented as sales that are accounted for as secured borrowings (see FASB ASC 860-30-25-2).

- **10.71** FASB ASC 860-30-25-2 requires that the transferor and transferee account for the transfer as a secured borrowing with pledge of collateral in either of the following circumstances:
  - a. If a transfer of an entire financial asset, a group of entire financial assets, or a participating interest in an entire financial asset does not meet the conditions for a sale in FASB ASC 860-10-40-5
  - b. If a transfer of a portion of an entire financial asset does not meet the definition of a *participating interest*

The transferor should continue to report the transferred financial asset in its statement of financial position with no change in the asset's measurement (that is, basis of accounting).

- 10.72 Cash Collateral. Paragraphs 3–4 of FASB ASC 860-30-25 state that all cash collateral should be recorded as an asset by the party receiving it (the secured party), together with a liability for the obligation to return it to the payer (obligor), whose asset is a receivable. Cash collateral used, for example, in securities lending transactions (see paragraphs 16–18 of FASB ASC 860-10-05) should be derecognized by the obligor and recognized by the secured party, not as collateral but rather as proceeds of either a sale or a borrowing.
- 10.73 Noncash Collateral. FASB ASC 860-30-25-5 states that the accounting for noncash collateral by the debtor (or obligor) and the secured party depends on whether the secured party has the right to sell or repledge the collateral and on whether the debtor has defaulted. Noncash collateral should be accounted for as follows:
  - a. If the secured party (transferee) has the right by contract or custom to sell or repledge the collateral, then FASB ASC 860-30-45-1 requires that the obligor (transferor) should reclassify that asset and report that asset in its statement of financial position separately, (for example, as security pledged to creditors) from other assets not so encumbered.
  - b. If the secured party (transferee) sells collateral pledged to it, it should recognize the proceeds from the sale and its obligation to return the collateral. The sale of the collateral is a transfer subject to the provisions of FASB ASC 860.
  - c. If the obligor (transferor) defaults under the terms of the secured contract and is no longer entitled to redeem the pledged asset, it should derecognize the pledged asset as required by FASB ASC 860-30-40-1 and the secured party (transferee) should recognize the collateral as its asset initially measured at fair value (see FASB ASC 860-30-30-1) or, if it has already sold the collateral, derecognize its obligation to return the collateral (see FASB ASC 860-30-40-1).
  - d. Except as provided in FASB ASC 860-30-40-1, the obligor (transferor) should continue to carry the collateral as its asset, and the secured party (transferee) should not recognize the pledged asset.

# Loans Not Previously HFS

10.74 FASB ASC 310-10-35-49 states that once a decision has been made to sell loans not previously classified as HFS, such loans should be transferred into the HFS classification and carried at the lower of cost or fair value. At the time of the transfer into the HFS classification, any amount by which

cost exceeds fair value should be accounted for as a valuation allowance. This guidance applies to both mortgage and nonmortgage loans.

- **10.75** The following paragraphs discuss whether the adjustments should be recorded by using an allowance method or as a direct write down, depending on the circumstances involved.
- 10.76 On March 26, 2001, the federal banking agencies, including the OTS and the NCUA issued *Interagency Guidance on Certain Loans Held for Sale* to provide instruction to institutions and examiners about the appropriate accounting and reporting treatment for certain loans that are sold directly from the loan portfolio or transferred to a HFS account. This guidance addresses transfers to the HFS account for loans within its scope and states that when a decision is made to sell a loan or portion thereof that was not originated or initially acquired with the intent to sell, the loan should be clearly identified and transferred to the HFS account.
- 10.77 Accordingly, for institutions subject to this guidance, once the decision has been made to sell loans not previously classified as HFS, such loans should be transferred into the HFS classification. For loans with declines in fair value that are attributable to credit quality, any reduction in the loan's value at the time of the transfer into the HFS classification should be reflected as a write-down of the recorded investment resulting in a new cost basis, with a corresponding reduction in the allowance for loan and lease losses. Thus, this guidance adds to the provisions of a requirement to write down the transferred loan and establish a new cost basis for institutions subject to this guidance.
- 10.78 In contrast, for loans with declines in fair value that are attributable to interest or foreign exchange rates and clearly are not attributable, in any respect, to credit or transfer risk, any amount by which the recorded investment exceeds fair value at the time of the transfer into the HFS classification should be accounted for as a valuation allowance.

### **Financial Statement Presentation**

- **10.79** Loans or trade receivables may be presented on the balance sheet as aggregate amounts, as stated in FASB ASC 310-10-45-2. However, such receivables HFS should be a separate balance sheet category.
- 10.80 FASB ASC 310-10-50-3 states that if major categories of loans or trade receivables are not presented separately in the balance sheet (see FASB ASC 310-10-45-2), they should be disclosed in the notes to the financial statements. FASB ASC 860-20-50-5 states that the aggregate amount of gains or losses on sales of loans or trade receivables (including adjustments to record loans HFS at the lower of cost or fair value) should be presented separately in the financial statements or disclosed in the notes to the financial statements.
- 10.81 As explained in FASB ASC 230-10-45-21, cash receipts and cash payments resulting from acquisitions and sales of loans also should be classified as operating cash flows if those loans are acquired specifically for resale and are carried at fair value or at the lower of cost or fair value. For example, mortgage loans HFS are required to be reported at the lower of cost or fair value in accordance with FASB ASC 948. Receipts from collections or sales of loans made by the entity and of other entities' debt instruments (other than cash equivalents and certain debt instruments that are acquired specifically for resale as discussed in FASB ASC 230-10-45-21) that were purchased by the

entity, are cash inflows from investing activities, according to item (a) in FASB ASC 230-10-45-12.

- 10.82 Many financial institutions currently present gains or losses on sales of loans or trade receivables separately on the face of the income statement. The financial statement disclosure requirements do not prohibit this industry practice.
- **10.83** FASB ASC 948-310-50-1 requires an entity to disclose the method used in determining the lower of cost or fair value of mortgage loans HFS (that is, aggregate or individual loan basis).
- **10.84** In addition, see chapter 18, "Derivative Instruments: Futures, Forwards, Options, Swaps, and Other Derivative Instruments," of this guide for accounting and financial reporting of derivative instruments recognized in connection with mortgage banking activities.
- **10.85** FASB ASC 825-10-45 addresses financial statement presentation if the fair value option is elected for mortgage loans HFS.

### Financial Statement Disclosure

- 10.86 FASB ASC 860 provides disclosure requirements for transfers and servicing of financial assets, including sales of financial assets, secured borrowings and collateral, and servicing assets and liabilities. Paragraphs 3–7 of FASB ASC 860-10-50 provide detailed guidance regarding the principal objectives of the disclosure requirements of FASB ASC 860 as well as consideration of disclosure aggregation. Those objectives are to provide financial statement users with an understanding of all of the following:
  - a. A transferor's continuing involvement, if any, with transferred financial assets
  - b. The nature of any restrictions on assets reported by an entity in its statement of financial position that relate to a transferred financial asset, including the carrying amounts of those assets
  - c. How servicing assets and servicing liabilities are reported under FASB ASC 860-50
  - d. For both of the following, how the transfer of financial assets affects an entity's financial position, financial performance, and cash flows:
    - i. Transfers accounted for as sales, if a transferor has continuing involvement with the transferred financial asset
    - Transfers of financial assets accounted for as secured borrowing

The specific disclosures required by FASB ASC 860 are minimum requirements, and an entity may need to supplement the required disclosures depending on (a) the facts and circumstances of a transfer, (b) the nature of an entity's continuing involvement with the transferred financial assets, or (c) the effect of an entity's continuing involvement on the transferor's financial position, financial performance, and cash flows.

**10.87** Secured borrowing and collateral disclosure requirements. An entity should disclose all of the following for collateral, as stated in FASB ASC 860-30-50-1A:

- a. If the entity has entered into repurchase agreements or securities lending transactions, it should disclose its policy for requiring collateral or other security.
- b. As of the date of the latest statement of financial position presented, both of the following:
  - The carrying amount and classification of any assets pledged as collateral that are not reclassified and separately reported in the statement of financial position in accordance with item (a) in FASB ASC 860-30-25-5 and associated liabilities.
  - ii. Qualitative information about the relationship(s) between those assets and associated liabilities; for example, if assets are restricted solely to satisfy a specific obligation, a description of the nature of restrictions placed on those assets.
- c. If the entity has accepted collateral that it is permitted by contract or custom to sell or repledge, it should disclose all of the following:
  - i. The fair value as of the date of each statement of financial position presented of that collateral.
  - The fair value as of the date of each statement of financial position presented of the portion of that collateral that it has sold or repledged.
  - iii. Information about the sources and uses of that collateral.

10.88 To provide an understanding of the nature and risks of short-term collateralized financing obtained through repurchase agreements, securities lending transactions, and repurchase-to-maturity transactions, that are accounted for as secured borrowings at the reporting date, FASB ASC 860-30-50-7 states that an entity should disclose the following information for each interim and annual period about the collateral pledged and the associated risks to which the transferor continues to be exposed after the transfer:

- a. A disaggregation of the gross obligation by the class of collateral pledged. An entity should determine the appropriate level of disaggregation and classes to be presented on the basis of the nature, characteristics, and risks of the collateral pledged. Total borrowings under those agreements should be reconciled to the amount of the gross liability for repurchase agreements and securities lending transactions disclosed in accordance with item (a) in FASB ASC 210-20-50-3 before any adjustments for offsetting. Any difference between the amount of the gross obligation disclosed under this paragraph and the amount disclosed in accordance with item (a) in FASB ASC 210-20-50-3 should be presented as reconciling item(s).
- b. The remaining contractual maturity of the repurchase agreements, securities lending transactions, and repurchase-to-maturity transactions. An entity should use judgment to determine an appropriate range of maturity intervals that would convey an understanding of the overall maturity profile of the entity's financing agreements.
- c. A discussion of the potential risks associated with the agreements and related collateral pledged, including obligations arising from a decline in the fair value of the collateral pledged and how those risks are managed.

- **10.89** Servicing assets and servicing liabilities disclosure requirements. For all servicing assets and servicing liabilities, an entity should disclose all of the following, according to FASB ASC 860-50-2:
  - a. Management's basis for determining its classes of servicing assets and servicing liabilities
  - b. A description of the risks inherent in servicing assets and servicing liabilities and, if applicable, the instruments used to mitigate the income statement effect of changes in fair value of the servicing assets and servicing liabilities
  - c. The amount of contractually specified servicing fees (which is defined in the FASB ASC glossary), late fees, and ancillary fees earned for each period for which results of operations are presented, including a description of where each amount is reported in the statement of income
  - d. Quantitative and qualitative information about the assumptions used to estimate fair value (for example, discount rates, anticipated credit losses, and prepayment speeds)

Disclosure of quantitative information about the instruments used to manage the risks inherent in servicing assets and servicing liabilities, including the fair value of those instruments at the beginning and end of the period, is encouraged but not required. An entity that provides such quantitative information is also encouraged, but not required, to disclose quantitative and qualitative information about the assumptions to estimate the fair value of those instruments. FASB ASC 235-10-50 provides guidance on disclosures of accounting policies.

10.90 For servicing assets and servicing liabilities subsequently measured at fair value, FASB ASC 860-50-50-3 states that for each class of servicing assets and servicing liabilities, the activity in the balance of servicing assets and the activity in the balance of servicing liabilities (including a description of where changes in fair value are reported in the statement of income for each period for which results of operations are presented), including, but not limited to, the following:

- a. The beginning and ending balances
- b. Additions through purchases of servicing assets, assumptions of servicing obligations, and servicing obligations that result from transfers of financial assets
- c. Disposals
- d. Changes in fair value during the period resulting from either changes in valuation inputs or assumptions used in the valuation model or other changes in fair value and a description of those changes
- e. Other changes that affect the balance and a description of those changes
- **10.91** According to FASB ASC 860-50-50-4, all of the following should be disclosed for servicing assets and servicing liabilities measured subsequently under the amortization method in item (a) in FASB ASC 860-50-35-1:
  - a. For each class of servicing assets and servicing liabilities, the activity in the balance of servicing assets and the activity in the balance of servicing liabilities (including a description of where changes in

the carrying amount are reported in the statement of income for each period for which results of operations are presented), including, but not limited to, the following:

- i. The beginning and ending balances.
- ii. Additions through purchases of servicing assets, assumption of servicing obligations, and recognition of servicing obligations that result from transfers of financial assets.
- iii. Disposals.
- iv. Amortization.
- v. Application of valuation allowance to adjust carrying value of servicing assets.
- vi. Other-than-temporary impairments.
- Other changes that affect the balance and a description of those changes.
- b. For each class of servicing assets and servicing liabilities, the fair value of recognized servicing assets and servicing liabilities at the beginning and end of the period.
- c. The risk characteristics of the underlying financial assets used to stratify recognized servicing assets for purposes of measuring impairment in accordance with FASB ASC 860-50-35-9. If the predominant risk characteristics and resulting stratums are changed, that fact and the reasons for those changes should be included in the disclosures about the risk characteristics of the underlying financial assets used to stratify the recognized servicing assets in accordance with this paragraph.
- d. For each period for which results of operations are presented, the activity by class in any valuation allowance for impairment of recognized servicing assets including all of the following:
  - i. Beginning and ending balances.
  - Aggregate additions charged and recoveries credited to operations.
  - iii. Aggregate write-downs charged against the allowance.
- 10.92 Sales of financial assets disclosure requirements. For securitizations, asset-backed financing arrangements, and similar transfers that are accounted for as sales when the transferor has continuing involvement with the transferred financial assets, the entity should disclose all of the following for each income statement presented, as stated in FASB ASC 860-20-50-3:
  - a. The characteristics of the transfer including all of the following:
    - i. A description of the transferor's continuing involvement with the transferred financial assets.
    - ii. The nature and initial fair value of both (a) the asset obtained as proceeds and (b) the liabilities incurred in the transfer.
    - iii. The gain or loss from sale of transferred financial assets.
  - b. For the initial fair value measurements in item a(ii), the level within the fair value hierarchy in FASB ASC 820 in which the fair value measurement fall, segregating fair value measurements using each of the following:

- Quoted prices in active markets for identical assets or liabilities (level 1).
- ii. Significant other observable inputs (level 2).
- iii. Significant unobservable inputs (level 3).
- c. For the initial fair value measurements in item a(ii), the key inputs and assumptions used in measuring the fair value of assets obtained and liabilities incurred as a result of the sale that relate to the transferor's continuing involvement, including quantitative information about all of the following:
  - i. Discount rates.
  - ii. Expected prepayments including the expected weighted-average life of prepayable financial assets. The weighted-average life of prepayable assets in periods (for example, months or years) can be calculated by multiplying the principal collections expected in each future period by the number of periods until that future period, summing those products, and dividing the sum by the initial principal balance
  - iii. Anticipated credit losses, including expected static pool losses.

If an entity has aggregated transfers during a period in accordance with the guidance beginning in FASB ASC 860-10-50-5, it may disclose the range of assumptions.

- d. For the initial fair value measurements in item a(ii), the valuation technique(s) used to measure fair value.
- e. Cash flows between a transferor and transferee, including all of the following:
  - i. Proceeds from new transfers.
  - ii. Proceeds from collections reinvested in revolving-period transfers.
  - iii. Purchases of previously transferred financial assets.
  - iv. Servicing fees.
  - v. Cash flows received from a transferor's interests.
- **10.93** According to FASB ASC 860-20-50-4, for securitizations, asset-backed financing arrangements, and similar transfers accounted for as sales when the transferor has continuing involvement with the transferred financial assets, the entity should disclose all of the following for each statement of financial position presented:
  - a. Qualitative and quantitative information about the transferor's continuing involvement with transferred financial assets that provides financial statement users with sufficient information to assess the reasons for the continuing involvement and the risks related to the transferred financial assets to which the transferor continues to be exposed after the transfer and the extent that the transferor's risk profile has changed as a result of the transfer (including, but not limited to, credit risk, interest rate risk, and other risks), including all of the following:
    - i. The total principal amount outstanding

- ii. The amount that has been derecognized
- The amount that continues to be recognized in the statement of financial position
- iv. Whether the transferor has provided financial or other support during the periods presented that it was not previously contractually required to provide to the transferee or its beneficial interest holders, including—when the transferor assisted the transferee or its beneficial interest holders in obtaining support—both (a) the type and amount of support and (b) the primary reasons for providing the support

An entity also is encouraged to disclose information about any liquidity arrangements, guarantees, or other commitments by third parties related to the transferred financial assets that may affect the fair value or risk of the related transferor's interest.

- b. The entity's accounting policies for subsequently measuring assets or liabilities that relate to the continuing involvement with the transferred financial assets.
- c. The key inputs and assumptions used in measuring the fair value of assets or liabilities that relate to the transferor's continuing involvement including, at a minimum, but not limited to, quantitative information about all of the following:
  - i. Discount rates.
  - ii. Expected prepayments including the expected weightedaverage life of prepayable financial assets (see item (c)(2) in FASB ASC 860-20-50-3).
  - iii. Anticipated credit losses, including expected static pool losses, if applicable. Expected static pool losses can be calculated by summing the actual and projected future credit losses and dividing the sum by the original balance of the pool of assets.

If an entity has aggregated transfers during a period in accordance with the guidance beginning in FASB ASC 860-10-50-5, it may disclose the range of assumptions.

- d. For the transferor's interest in the transferred financial assets, a sensitivity analysis or stress test showing the hypothetical effect on the fair value of those interests (including any servicing assets or servicing liabilities) of two or more unfavorable variations from the expected levels for each key assumption that is reported under item c independently from any change in another key assumption.
- e. A description of the objectives, methodology, and limitations of the sensitivity analysis or stress test.
- f. Information about the asset quality of transferred financial assets and any other financial assets that it manages together with them. This information should be separated between assets that have been derecognized and assets that continue to be recognized in the statement of financial position. This information is intended to provide financial statement users with an understanding of the risks inherent in the transferred financial assets as well as in other

financial assets and liabilities that it manages together with transferred financial assets. For example, information for receivables should include, but is not limited to both of the following:

- i. Delinquencies at the end of the period.
- ii. Credit losses, net of recoveries, during the period.

10.94 To provide an understanding of the nature of the transactions, the transferor's continuing exposure to the transferred financial assets, and the presentation of the components of the transaction in the financial statements, FASB ASC 860-20-50-4D states that an entity should disclose the following for outstanding transactions at the reporting date that meet the scope guidance in paragraphs 4A–4B of FASB ASC 860-20-50 by type of transaction (for example, repurchase agreement, securities lending transaction, and sale and total return swap) (except for those transactions that are excluded from the scope, as described in FASB ASC 860-20-50-4C):<sup>4</sup>

- The carrying amount of assets derecognized as of the date of derecognition:
  - If the amounts that have been derecognized have changed significantly from the amounts that have been derecognized in prior periods or are not representative of the activity throughout the period, a discussion of the reasons for the change should be disclosed.
- b. The amount of gross cash proceeds received by the transferor for the assets derecognized as of the date of derecognition.
- c. Information about the transferor's ongoing exposure to the economic return on the transferred financial assets:
  - As of the reporting date, the fair value of assets derecognized by the transferor.
  - ii. Amounts reported in the statement of financial position arising from the transaction (for example, the carrying

The following items are excluded from the scope of this disclosure requirement:

- a. Transfers of financial assets with an agreement to purchase another financial asset that is not substantially the same as the initial transferred financial asset in accordance with item (a) in FASB ASC 860-10-40-24, for example, a dollar roll transaction accounted for as a sale because the financial asset to be purchased is not substantially the same as the initially transferred financial assets in accordance with item (a) in FASB ASC 860-10-40-24.
- b. Securitizations, asset-backed financing arrangements, and similar transfers, in which both the transfer is accounted for as a sale and the transferor has continuing involvement with the transferred financial assets, that are subject to the disclosures in paragraphs 3–4 of FASB ASC 860-20-50.

<sup>&</sup>lt;sup>4</sup> In accordance with the scope guidance addressed in paragraphs 4A–4C of FASB ASC 860-20-50, this disclosure requirement applies to transactions accounted for as a sale that comprise both a transfer of financial assets to a transferee and an agreement entered into in contemplation of the initial transfer with the transferee that results in the transferor retaining substantially all of the exposure to the economic return on the transferred financial asset throughout the term of the transaction. These transactions include both of the following types:

a. Transfers of financial assets with an agreement to repurchase the transferred financial asset (or a substantially-the-same financial asset) before maturity at a fixed or determinable price that will be settled in a form other than the return of the transferred financial asset (for example, the transaction is cash-settled).

b. Transfers of financial assets with an agreement that requires that the transferor retain substantially all of the exposure to the economic return on the transferred financial asset (for example, a sale with a total return swap).

value or fair value of forward repurchase agreements or swap contracts). To the extent that those amounts are captured in the derivative disclosures presented in accordance with FASB ASC 815-10-50-4B, an entity should provide a cross-reference to the appropriate line item in that disclosure.

- iii. A description of the arrangements that result in the transferor retaining substantially all of the exposure to the economic return on the transferred financial assets and the risks related to those arrangements.
- **10.95** In addition, see chapter 18 of this guide for disclosures required for derivative instruments recognized in connection with mortgage banking activities.
- **10.96** FASB ASC 820-10-50 and FASB ASC 825-10-50 provide additional guidance related to fair value disclosures.

#### VIEs<sup>5</sup>

10.97 Financial institutions use securitization vehicles to pool and repackage mortgage loans, credit card loans and other assets. They also use asset-backed commercial paper conduits and collateralized debt and loan obligation vehicles to issue equity and debt to investor. Other common investment vehicles used by financial institutions include partnerships for low income housing or historic rehabilitation. These vehicles should be evaluated to determine if they are VIEs. A reporting entity that holds a direct or indirect (explicit or implicit) variable interest in a legal entity must determine whether the guidance in the "Variable Interest Entities" subsections of FASB ASC 810-10 applies to that legal entity before considering other consolidation guidance. However, if a reporting entity does not have a direct or indirect (explicit or implicit) variable interest in a legal entity, then the reporting entity is not the primary beneficiary of that legal entity and is not required to provide disclosures for that legal entity under FASB ASC 810-10 "Variable Interest Entities" subsections.<sup>6</sup>

(continued)

<sup>&</sup>lt;sup>5</sup> Many paragraphs in FASB ASC 810-10 are labeled as "Pending Content" and relate to FASB Accounting Standards Update (ASU) No. 2009-17, Consolidations (Topic 810): Improvements to Financial Reporting by Enterprises Involved with Variable Interest Entities. Although labeled as "Pending Content," this guidance is applicable for many entities except those that qualify for the deferral in FASB ASU No. 2010-10, Consolidation (Topic 810): Amendments to Certain Investment Funds. In February 2015, FASB issued ASU No. 2015-02, Consolidation (Topic 810): Amendments to the Consolidation Analysis. Among other provisions, the amendments in this ASU rescind the indefinite deferral in FASB ASU No. 2010-10 and provide a scope exception from FASB ASC 810, Consolidation, for reporting entities with interests in legal entities that are required to comply with or operate in accordance with requirements similar to those in Rule 2a-7 of the Investment Company Act of 1940 for registered money market funds. See footnote 7 for further details on the effective date of ASU No. 2015-02. Readers should carefully review the transition guidance in FASB ASC 810-10-65-2 associated with "Pending Content" to determine the applicability.

<sup>&</sup>lt;sup>6</sup> FASB ASU No. 2010-02, Consolidation (Topic 810): Accounting and Reporting for Decreases in Ownership of a Subsidiary—a Scope Clarification, added a scope exception for deconsolidation/derecognition of sales of in-substance real estate. Emerging Issues Task Force (EITF) Issue No. 10-E, "Accounting for Deconsolidation of a Subsidiary That Is In-Substance Real Estate," highlighted a concern for in-substance real estate in which there is an event or circumstance other than a sale in which an entity may be required to deconsolidate in-substance real estate because the reporting entity has lost a controlling financial interest but the reporting entity would not be able to derecognize the in-substance asset under the guidance in FASB ASC 360-20. FASB ASU No. 2011-10, Property, Plant, and Equipment (Topic 360): Derecognition of in Substance Real Estate—a Scope Clarification

10.98 Per the FASB ASC glossary, a *variable interest* is defined as an investment or other interest that will absorb portions of a VIE's expected losses or receive portions of the entity's expected residual returns. Variable interests in a VIE are contractual, ownership, or other pecuniary interests in a VIE that change with changes in the fair value of the VIE's net assets exclusive of variable interests. Equity interests with or without voting rights are considered variable interests if the legal entity is a VIE and to the extent that the investment is at risk as described in FASB ASC 810-10-15-14 (see paragraph 10.102). FASB ASC 810-10-25-55 explains how to determine whether a variable interest in specified assets of a legal entity is a variable interest in the entity. Paragraphs 16–41 of FASB ASC 810-10-55 describe various types of variable interests and explain in general how they may affect the determination of the primary beneficiary of a VIE.

**10.99** "Pending Content" in paragraphs 37–38 of FASB ASC 810-10-55 provides detailed guidance in determining whether fees paid to a legal entity's decision maker or service provider constitute a variable interest.

**10.100** The VIE guidance in FASB ASC 810-10 explains how to identify a VIE and how to determine when a reporting entity includes the assets, liabilities, noncontrolling interests, and results of activities of a VIE in its consolidated financial statements. VIEs are often created for a single specified purpose, for example, to facilitate securitization, leasing, hedging, research and development, reinsurance, or other transactions or arrangements. The distinction between VIEs and other entities is based on the nature and amount of the equity investment and the rights and obligations of the equity investors.

**10.101** FASB ASC 810-10-15-12 provides exceptions to consolidation that apply to both VIEs and voting interest entities, whereas "Pending Content" in FASB ASC 810-10-15-17 provides a listing of entities exempt from consolidation under VIE guidance but that may be subject to consolidation under other GAAP.

(footnote continued)

(a consensus of the FASB Emerging Issues Task Force), codified the consensus in EITF Issue No. 10-E and resolved diversity in practice by clarifying that when a parent (reporting entity) ceases to have a controlling financial interest (as described in FASB ASC 810-10) in a subsidiary that is in-substance real estate as a result of default on the subsidiary's nonrecourse debt, the reporting entity should apply the guidance in FASB ASC 360-20 to determine whether it should derecognize the in-substance real estate. Further, on the basis of the feedback received, the EITF recommended that the FASB chairman add a project to its agenda to address lenders' accounting in situations in which a borrower ceases to have a controlling financial interest in an in-substance real estate entity because of a default on its nonrecourse debt. After considering the recommendation of the EITF and the comments received on FASB ASU No. 2011-10, the FASB chairman added the research project titled Clarifying the Definition of a Business (formerly Application of Asset- or Entity-Based Guidance to Nonfinancial Assets in an Entity) to FASB's agenda to explore the broader issue of determining when an entity that substantially consists of nonfinancial assets should be accounted for as an in-substance asset or as an entity.

In May 2013, FASB elected to move the project into a standard-setting project. Currently, FASB has two open projects titled Clarifying the Definition of a Business Phase 1 (formerly Application of Asset- or Entity-Based Guidance to Nonfinancial Assets in an Entity) and Clarifying the Scope of Subtopic 610-20 and Accounting for Partial Sales of Nonfinancial Assets (formerly Clarifying the Definition of a Business Phase 2). Phase 1 of the projects is intended to clarify the definition of business with the objective of adding guidance to assist entities when evaluating whether transactions should be accounted for as acquisitions (or disposals) of assets or businesses. Phase 2 of the projects will include clarifying the guidance for partial sales or transfers of nonfinancial assets in the scope of FASB ASC 610-20 and the reference to in-substance nonfinancial assets. Readers are encouraged to visit the "Technical Agenda" page at www.fasb.org for the latest developments regarding these projects, including a summary of decisions reached to date.

# Considerations for Private Companies that Elect to use Standards as Issued by the Private Company Council

FASB ASU No. 2014-07, Consolidation (Topic 810): Applying Variable Interest Entities Guidance to Common Control Leasing Arrangements (a consensus of the Private Company Council), permits a private company lessee (the reporting entity) to elect an alternative not to apply VIE guidance to a lessor entity if the following criteria are met:

- The private company lessee and the lessor legal entity are under common control.
- The private company lessee has a lease arrangement with the lessor legal entity.
- Substantially all activities between the private company lessee and the lessor legal entity are related to leasing activities (including supporting leasing activities) between those two entities.
- If the private company lessee explicitly guarantees or provides collateral for any obligation of the lessor legal entity related to the asset leased by the private company, then the principal amount of the obligation at inception of such guarantee or collateral arrangement does not exceed the value of the asset leased by the private company from the lessor legal entity.

In accordance with FASB ASC 810-10-15-17B, the accounting alternative is an accounting policy election that, when elected, should be applied by a private company lessee to all current and future lessor entities under common control that meet the criteria for applying this approach. If any of the criteria for applying the alternative cease to be met, a private company should apply the VIE guidance at the date of change on a prospective basis.

#### VIE Determination

**10.102** According to "Pending Content" in FASB ASC 810-10-15-14, a legal entity should be subject to consolidation under the VIE guidance subsections of FASB ASC 810 if, by design, any of the following conditions exist:

- a. The total equity investment at risk is not sufficient to permit the legal entity to finance its activities without additional subordinated financial support provided by any parties, including equity holders. For this purpose, the total equity investment at risk has all of the following characteristics:
  - i. Includes only equity investments in the legal entity that participate significantly in profits and losses even if those investments do not carry voting rights.
  - ii. Does not include equity interests that the legal entity issued in exchange for subordinated interests in other VIEs.
  - iii. Does not include amounts provided to the equity investor directly or indirectly by the legal entity or by other parties involved with the legal entity, unless the provider is a parent, subsidiary, or affiliate of the investor that is required to be included in the same set of consolidated financial statements as the investor.

iv. Does not include amounts financed for the equity investor directly by the legal entity or by other parties involved with the legal entity, unless that party is a parent, subsidiary, or affiliate of the investor that is required to be included in the same set of consolidated financial statements as the investor.

Paragraphs 45–47 of FASB ASC 810-10-25 discuss the amount of the total equity investment at risk that is necessary to permit a legal entity to finance its activities without additional subordinated financial support.

- b. As a group the holders of the equity investment at risk lack any of the following three characteristics:
  - i. The power through voting rights or similar rights to direct the activities of a legal entity that most significantly impact the entity's economic performance. The investors do not have the power through voting rights or similar rights if no owners hold voting rights or similar rights (such as those of a common shareholder in a corporation or a general partner in a partnership). Legal entities that are not controlled by the holder of a majority voting interest because of noncontrolling shareholder veto rights, as discussed in paragraphs 2–14 of FASB ASC 810-10-25. are not VIEs if the shareholders as a group have the power to control the entity and the equity investment meets the other requirements of the "Variable Interest Entities" subsections. Kick-out rights or participating rights, as defined in the FASB ASC glossary, held by the holders of the equity investment at risk should not prevent interests other than the equity investment from having this characteristic unless a single equity holder (including its related parties and de facto agents) has the unilateral ability to exercise such rights. Alternatively, interests other than the equity investment at risk that provide the holders of those interests with kick-out rights or participating rights should not prevent the equity holders from having this characteristic unless a single reporting entity (including its related parties and de facto agents) has the unilateral ability to exercise those rights. A decision maker also should not prevent the equity holders from having this characteristic unless the fees paid to the decision maker represent a variable interest based on paragraphs 37–38 of FASB ASC 810-10-55.
  - ii. The obligation to absorb the expected losses of the legal entity. The investor or investors do not have that obligation if they are directly or indirectly protected from the expected losses or are guaranteed a return by the legal entity itself or by other parties involved with the legal entity.
  - iii. The right to receive the expected residual returns of the legal entity. The investors do not have the right if their return is capped by the legal entity's governing documents or arrangements with other variable interest holders or the legal entity.

If interests other than the equity investment at risk provide the holders of that investment with the characteristics of a controlling financial interest or if interests other than the equity investment at risk prevent the equity holders from having the necessary characteristics, the entity is a VIE.

- c. The equity investors as a group also are considered to lack the characteristics in item b(i) if both of the following conditions are present:
  - The voting rights of some investors are not proportional to their obligations to absorb the expected losses of the legal entity, their rights to receive the expected residual returns of the legal entity, or both.
  - ii. Substantially all of the legal entity's activities either involve or are conducted on behalf of an investor that has disproportionately few voting rights. This provision is necessary to prevent a primary beneficiary from avoiding consolidation of a VIE by organizing the legal entity with nonsubstantive voting interests.

For purposes of applying this requirement, reporting entities should consider each party's obligations to absorb expected losses and rights to receive expected residual returns related to all of that party's interests in the legal entity and not only to its equity investment at risk.

According to "Pending Content" in FASB ASC 810-10-15-14, a legal entity should be subject to consolidation under the VIE guidance subsections of FASB ASC 810, *Consolidation*, if, by design, any of the following conditions exist:

- a. The total equity investment at risk is not sufficient to permit the legal entity to finance its activities without additional subordinated financial support provided by any parties, including equity holders. For this purpose, the total equity investment at risk has all of the following characteristics:
  - i. Includes only equity investments in the legal entity that participate significantly in profits and losses even if those investments do not carry voting rights.
  - ii. Does not include equity interests that the legal entity issued in exchange for subordinated interests in other VIEs
  - iii. Does not include amounts provided to the equity investor directly or indirectly by the legal entity or by other parties involved with the legal entity, unless the provider is a parent, subsidiary, or affiliate of the investor that is required to be included in the same set of consolidated financial statements as the investor.
  - iv. Does not include amounts financed for the equity investor directly by the legal entity or by other parties involved with the legal entity, unless that party is a parent, subsidiary, or affiliate of the investor that is required to be included in the same set of consolidated financial statements as the investor.

Paragraphs 45–47 of FASB ASC 810-10-25 discuss the amount of the total equity investment at risk that is necessary to permit a legal entity to finance its activities without additional subordinated financial support.

- b. As a group the holders of the equity investment at risk lack any of the following three characteristics:
  - The power through voting rights or similar rights to direct the activities of a legal entity that most significantly impact the entity's economic performance.
    - For legal entities other than limited partnerships, investors lack the power through voting rights or similar rights if no owners hold voting rights or similar rights (such as those of a common shareholder in a corporation). Legal entities that are not controlled by the holder of a majority voting interest because of noncontrolling shareholder veto rights (participating rights), as discussed in paragraphs 2-14 of FASB ASC 810-10-25, are not VIEs if the holders of the equity investment at risk as a group have the power to control the entity and the equity investment meets the other requirements of the "Variable Interest Entities" subsections. If no owners hold voting rights or similar rights (such as those of a common shareholder in a corporation) over the activities of a legal entity that most significantly impact the entity's economic performance, kickout rights or participating rights, as defined in the FASB ASC glossary, held by the holders of the equity investment at risk should not prevent interests other than the equity investment from having this characteristic unless a single equity holder (including its related parties and de facto agents) has the unilateral ability to exercise such rights. Alternatively, interests other than the equity investment at risk that provide the holders of those interests with kick-out rights or participating rights should not prevent the equity holders from having this characteristic unless a single reporting entity (including its related parties and de facto agents) has the unilateral ability to exercise those rights. A decision maker also should not prevent the equity holders from having this characteristic unless the fees paid to the decision maker represent a variable interest based on paragraphs 37–38 of FASB ASC 810-10-55.
    - (2) For limited partnerships, partners lack that power if neither items (a) nor (b), discussed subsequently, exists. The guidance in this subparagraph does not apply to entities in industries

(see FASB ASC 910-810-45-1 and FASB ASC 932-810-45-1) in which it is appropriate for a general partner to use the pro rata method of consolidation for its investment in a limited partnership (see FASB ASC 810-10-45-14).

- A simple majority or lower threshold of limited partners (including a single limited partner) with equity at risk is able to exercise substantive kick-out rights (according to their voting interest entity definition ) through voting interests over the general partner(s).For purposes of evaluating this threshold, a general partner's kick-out rights held through voting interests should not be included. Kick-out rights through voting interests held by entities under common control with the general partner or other parties acting on behalf of the general partner also should not be included.
- (b) Limited partners with equity at risk are able to exercise substantive participating rights (according to their voting interest entity definition) over the general partner(s).
- (c) For purposes of items (a) and (b), evaluation of the substantiveness of participating rights and kick-out rights should be based on the guidance included in paragraphs 2–14C of FASB ASC 810-10-25-2.7
- ii. The obligation to absorb the expected losses of the legal entity. The investor or investors do not have that obligation if they are directly or indirectly protected from the expected losses or are guaranteed a return by the legal entity itself or by other parties involved with the legal entity.

FASB ASU No. 2015-02 affects reporting entities that are required to evaluate whether they should consolidate certain legal entities. All legal entities are subject to reevaluation under the revised consolidation model.

Readers are encouraged to read the full text of the ASU, available at www.fasb.org. Readers should apply the appropriate guidance based on their facts and circumstances.

<sup>&</sup>lt;sup>7</sup> FASB ASU No. 2015-02 was issued in February 2015. The amendments are effective for public business entities for fiscal years, and for interim periods within those fiscal years, beginning after December 15, 2015. For all other entities, the amendments in this ASU are effective for fiscal years beginning after December 15, 2017. Early adoption is permitted, including adoption in an interim period. If an entity early adopts the amendments in an interim period, any adjustments should be reflected as of the beginning of the fiscal year that includes that interim period. A reporting entity may apply the amendments in this ASU using a modified retrospective approach by recording a cumulative-effect adjustment to equity as of the beginning of the fiscal year of adoption. A reporting entity also may apply the amendments retrospectively.

iii. The right to receive the expected residual returns of the legal entity. The investors do not have the right if their return is capped by the legal entity's governing documents or arrangements with other variable interest holders or the legal entity.

If interests other than the equity investment at risk provide the holders of that investment with the characteristics of a controlling financial interest or if interests other than the equity investment at risk prevent the equity holders from having the necessary characteristics, the entity is a VIE.

- c. The equity investors as a group also are considered to lack the characteristics in item  $b(\mathbf{i})$  if both of the following conditions are present:
  - i. The voting rights of some investors are not proportional to their obligations to absorb the expected losses of the legal entity, their rights to receive the expected residual returns of the legal entity, or both.
  - ii. Substantially all of the legal entity's activities either involve or are conducted on behalf of an investor that has disproportionately few voting rights. This provision is necessary to prevent a primary beneficiary from avoiding consolidation of a VIE by organizing the legal entity with nonsubstantive voting interests.

For purposes of applying this requirement, reporting entities should consider each party's obligations to absorb expected losses and rights to receive expected residual returns related to all of that party's interests in the legal entity and not only to its equity investment at risk.

#### Reconsideration of VIE Status

10.103 A legal entity that previously was not subject to the VIE guidance presented in FASB ASC 810-10 should not become subject to it simply because of losses in excess of its expected losses that reduce the equity investment, according to "Pending Content" in FASB ASC 810-10-35-4. The initial determination of whether a legal entity is a VIE should be reconsidered if any of the following occur:

- a. The legal entity's governing documents or contractual arrangements are changed in a manner that changes the characteristics or adequacy of the legal entity's equity investment at risk
- b. The equity investment or some part thereof is returned to the equity investors, and other interests become exposed to expected losses of the legal entity
- c. The legal entity undertakes additional activities or acquires additional assets, beyond those that were anticipated at the later of the inception of the entity or the latest reconsideration event, that increase the entity's expected losses
- d. The legal entity receives an additional equity investment that is at risk, or the legal entity curtails or modifies its activities in a way that decreases its expected losses

e. Changes in facts and circumstances occur such that the holders of the equity investment at risk, as a group, lose the power from voting rights or similar rights of those investments to direct the activities of the entity that most significantly impact the entity's economic performance

## **Primary Beneficiary Evaluation**

10.104 In accordance with "Pending Content" in paragraphs 38–38A of FASB ASC 810-10-25, a reporting entity should consolidate a VIE when the reporting entity has a variable interest (or combination of variable interests) that provides the reporting entity with a controlling financial interest on the basis of the provisions in paragraphs 38A–38G of FASB ASC 810-10-25. A reporting entity should be deemed to have a controlling financial interest in a VIE, and thus is the VIE's primary beneficiary, if it has both of the following characteristics:

- a. The power to direct the activities of the VIE that most significantly impact the VIE's economic performance
- b. The obligation to absorb losses of the VIE that could potentially be significant to the VIE or the right to receive benefits from the VIE that could potentially be significant to the VIE

In accordance with "Pending Content" in paragraphs 38–38A of FASB ASC 810-10-25, a reporting entity should consolidate a VIE when the reporting entity has a variable interest (or combination of variable interests) that provides the reporting entity with a controlling financial interest on the basis of the provisions in paragraphs 38A–38J of FASB ASC 810-10-25.8 A reporting entity should be deemed to have a controlling financial interest in a VIE, and thus is the VIE's primary beneficiary, if it has both of the following characteristics:

- a. The power to direct the activities of the VIE that most significantly impact the VIE's economic performance
- b. The obligation to absorb losses of the VIE that could potentially be significant to the VIE or the right to receive benefits from the VIE that could potentially be significant to the VIE

10.105 For purposes of determining whether a reporting entity is the primary beneficiary of a VIE held by its related parties, FASB ASC 810-10-25-42 states that the reporting entity with a variable interest should treat variable interests in that same VIE held by its related parties as its own interests. "Pending Content" in FASB ASC 810-10-25-43 provides guidance on related parties and de facto agents. In situations in which a reporting entity concludes that neither it nor one of its related parties has the characteristics in FASB ASC 810-10-25-38A but, as a group, the reporting entity and its related parties (including de facto agents) have those characteristics, then "Pending Content" in FASB ASC 810-10-25-44 states that the party within the related party group that is most closely associated with the VIE is the primary beneficiary. The determination of which party within the related party group is most closely

<sup>&</sup>lt;sup>8</sup> See footnote 7.

associated with the VIE requires judgment and should be based on analysis of all relevant facts and circumstances, including all of the following:

- a. The existence of a principal-agency relationship between parties within the related party group
- b. The relationship and significance of the activities of the VIE to the various parties within the related party group
- c. A party's exposure to the variability associated with anticipated economic performance of the VIE
- d. The design of the VIE

"Pending Content" in paragraphs 42–44B of FASB ASC 810-10-25-42 address the effect of related parties on the primary beneficiary determination.<sup>9</sup>

10.106 Once a reporting entity has determined itself to be the primary beneficiary of a VIE, the reporting entity must consolidate the VIE. In addition, a reporting entity with a significant variable interest in a VIE must also provide disclosures about its involvement with the VIE, regardless of whether it is the primary beneficiary of the VIE.

## Financial Statement Presentation and Disclosure Requirements

**10.107** "Pending Content" in FASB ASC 810-10-45-25 states that a reporting entity should present each of the following separately on the face of the statement of financial position:

- a. Assets of a consolidated VIE that can be used only to settle obligations of the consolidated VIE
- b. Liabilities of a consolidated VIE for which creditors (or beneficial interest holders) do not have recourse to the general credit of the primary beneficiary

**10.108** The principal objectives of the required disclosure in FASB ASC 810-10 are to provide financial statement users with an understanding of all of the following:

- a. The significant judgments and assumptions made by a reporting entity in determining whether it must consolidate a VIE or disclose information about its involvement in a VIE
- b. The nature of restrictions on a consolidated VIE's assets and on the settlement of its liabilities reported by a reporting entity in its statement of financial position, including the carrying amounts of such assets and liabilities
- c. The nature of, and changes in, the risks associated with a reporting entity's involvement with the VIE
- d. How a reporting entity's involvement with the VIE affects the reporting entity's financial position, financial performance, and cash flows

# Considerations for Private Companies that Elect to use Standards as Issued by the Private Company Alternative

Under the private company accounting alternative, a private company lessee would not be required to provide the VIE disclosures about the

<sup>&</sup>lt;sup>9</sup> See footnote 7.

lessor entity. Rather, in accordance with FASB ASC 810-10-50-2AD, the private company lessee would disclose the following information:

- The amount and key terms of liabilities (for example, debt, environmental liabilities, and asset retirement obligations) recognized by the lessor legal entity that expose the private company lessee to providing financial support to the legal entity. For example, a private company lessee exposed to debt of the legal entity should disclose information such as the amount of debt, interest rate, maturity, pledged collateral, and guarantees associated with the debt.
- A qualitative description of circumstances (for example, certain commitments and contingencies) not recognized in the financial statements of the lessor legal entity that expose the private company lessee to providing financial support to the legal entity.

In applying this disclosure guidance, FASB ASC 810-10-50-2AE states that a private company lessee should consider exposures through implicit guarantees. The determination as to whether an implicit guarantee exists is based on facts and circumstances. The facts and circumstances include, but are not limited to, whether there is an economic incentive for the private company lessee to act as a guarantor or to make funds available, such actions have happened in similar situations in the past, and the private company lessee acting as a guarantor or making funds available would be considered a conflict of interest or illegal.

In accordance with FASB ASC 810-10-50-2AF, the disclosures under this alternative are required in combination with the disclosures required by other guidance (for example, FASB ASC 460, *Guarantees*, FASB ASC 850, *Related Party Disclosures*, and FASB ASC 840, *Leases*). Those disclosures could be combined in a single note or by including cross-references within the notes to the financial statements.

#### © Update 10-2 Accounting and Reporting: Leases

FASB ASU No. 2016-02, *Leases (Topic 842)*, issued in February 2016, is effective for fiscal years of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files financial statements with the SEC beginning after December 15, 2018, including interim periods within those fiscal years.

For all other entities, FASB ASU No. 2016-02 is effective for fiscal years beginning after December 15, 2019, and interim periods within fiscal years beginning after December 15, 2019.

Early application is permitted for all entities.

FASB ASU No. 2016-02 supersedes the lease requirements in FASB ASC 840 and creates FASB ASC 842, *Leases*, to establish the principles that lessees and lessors should apply to report useful information to users of financial statements about the amount, timing, and uncertainty of cash flows arising from a lease. FASB ASC 842 affects any entity that enters into a lease (as

that term is defined in FASB ASU No. 2016-02), with some specified scope exceptions.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this discussion will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-02, see appendix F, "The New Leases Standard: FASB ASU No. 2016-02," of this guide.

- **10.109** *Primary beneficiary of a VIE disclosure requirements.* "Pending Content" in FASB ASC 810-10-50-3 states the primary beneficiary of a VIE that is not a business should disclose the amount of gain or loss recognized on the initial consolidation of the VIE. In addition to other disclosures required by FASB ASC 810, the primary beneficiary should disclose all of the following:
  - a. The carrying amounts and classification of the VIE's assets and liabilities in the statement of financial position that are consolidated in accordance with VIE guidance in FASB ASC 810-10, including qualitative information about the relationship(s) between those assets and liabilities
  - Lack of recourse if creditors (or beneficial interest holders) of a consolidated VIE have no recourse to the general credit of the primary beneficiary
  - c. Terms of arrangements, giving consideration to both explicit arrangements and implicit variable interests that could require the reporting entity to provide financial support to the VIE, including events or circumstances that could expose the reporting entity to a loss
- 10.110 Disclosure requirements for a holder of a variable interest in a VIE that is not the primary beneficiary. "Pending Content" in FASB ASC 810-10-50-4 states that in addition to disclosures required by other guidance, a reporting entity that holds a variable interest in a VIE but is not the VIE's primary beneficiary should disclose
  - a. the carrying amounts and classification of the assets and liabilities in the reporting entity's statement of financial position that relate to the reporting entity's variable interest in the VIE.
  - b. the reporting entity's maximum exposure to loss as a result of its involvement with the VIE, including how the maximum exposure is determined and the significant sources of the reporting entity's exposure to the VIE. If the reporting entity's maximum exposure to loss as a result of its involvement with the VIE cannot be quantified, that fact should be disclosed.
  - c. a tabular comparison of the carrying amounts of the assets and liabilities, as required by item a, and the reporting entity's maximum exposure to loss, as required by item b. A reporting entity should provide qualitative and quantitative information to allow financial statement users to understand the differences between the two amounts. That discussion should include, but is not limited to, the terms of arrangements, giving consideration to both explicit arrangements and implicit variable interests, that could require the

- reporting entity to provide financial support (for example, liquidity arrangements and obligations to purchase assets) to the VIE, including events or circumstances that could expose the reporting entity to a loss.
- d. information about any liquidity arrangements, guarantees, and/or other commitments by third parties that may affect the fair value or risk of the reporting entity's variable interest in the VIE is encouraged.
- e. if applicable, significant factors considered and judgments made in determining that the power to direct the activities of a VIE that most significantly impact the VIE's economic performance is shared in accordance with the guidance in FASB ASC 810-10-25-38D.

A VIE may issue voting equity interests, and the entity that holds a majority voting interest also may be the primary beneficiary of the VIE. If so, the disclosures in this paragraph are not required.

- **10.111** *Primary beneficiaries or other holders of interests in VIEs disclosure requirements.* "Pending Content" in FASB ASC 810-10-50-5A states that a reporting entity that is a primary beneficiary of a VIE or a reporting entity that holds a variable interest in a VIE but is not the entity's primary beneficiary should disclose all of the following:
  - a. Its methodology for determining whether the reporting entity is the primary beneficiary of a VIE, including, but not limited to, significant judgments and assumptions made. One way to meet this disclosure requirement would be to provide information about the types of involvements a reporting entity considers significant, supplemented with information about how the significant involvements were considered in determining whether the reporting entity is the primary beneficiary.
  - b. If facts and circumstances change such that the conclusion to consolidate a VIE has changed in the most recent financial statements (for example, the VIE was previously consolidated and is not currently consolidated), the primary factors that caused the change and the effect on the reporting entity's financial statements.
  - c. Whether the reporting entity has provided financial or other support (explicitly or implicitly) during the periods presented to the VIE that it was not previously contractually required to provide or whether the reporting entity intends to provide that support, including both of the following:
    - i. The type and amount of support, including situations in which the reporting entity assisted the VIE in obtaining another type of support.
    - ii. The primary reasons for providing the support.
  - d. Qualitative and quantitative information about the reporting entity's involvement (giving consideration to both explicit arrangements and implicit variable interests) with the VIE, including, but not limited to, the nature, purpose, size, and activities of the VIE, including how the VIE is financed. Paragraphs 49–54 of FASB ASC 810-10-25 provide guidance on how to determine whether a reporting entity has an implicit variable interest in a VIE.

- **10.112** Aggregation of certain disclosures. "Pending Content" in FASB ASC 810-10-50-9 states that disclosures about VIEs may be reported in the aggregate for similar entities if separate reporting would not provide more useful information to financial statement users. A reporting entity should disclose how similar entities are aggregated and should distinguish between
  - a. VIEs that are not consolidated because the reporting entity is not the primary beneficiary but has a variable interest and
  - b. VIEs that are consolidated.

In determining whether to aggregate VIEs, the reporting entity should consider quantitative and qualitative information about the different risk and reward characteristics of each VIE and the significance of each VIE to the entity. The disclosures should be presented in a manner that clearly and fully explains to financial statement users the nature and extent of an entity's involvement with VIEs.

10.113 In accordance with "Pending Content" in FASB ASC 810-10-50-10, a reporting entity should determine, in light of the facts and circumstances, how much detail it should provide to satisfy the requirements of the VIE guidance presented in FASB ASC 810-10. A reporting entity should also determine how it aggregates information to display its overall involvements with VIEs with different risk characteristics. The reporting entity must strike a balance between obscuring important information as a result of too much aggregation and overburdening financial statements with excessive detail that may not assist financial statement users to understand the reporting entity's financial position.

# Auditing<sup>10</sup>

# **Objectives**

10.114 Audit objectives and procedures for loan origination and underwriting are discussed in chapter 8 of this guide. Audit objectives and procedures for securities, including MBSs, are addressed in chapter 7 of this guide. The primary audit objectives in this area are to obtain sufficient appropriate evidence that

- a. loans HFS exist and the institution has the right to them;
- b. loans HFS are carried at the lower of cost or fair value, or at fair value if that election has been made;
- c. loans HFS are properly classified, described, and disclosed in the financial statements;
- d. transfers of financial assets, including through repurchase agreements and securitizations, have been properly accounted for as sales or secured borrowings;

The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board (ASB) and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of issuers and nonissuers as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

- e. gains and losses on the transfer of financial assets or servicing rights are properly measured, recorded, and disclosed;
- f. assets retained or obtained as a result of a transfer of a financial asset accounted for as a sale are properly recognized and measured initially and on an ongoing basis;
- g. representations and warranty liabilities, recourse obligations and other liabilities assumed as a result of a transfer of a financial asset accounted for as a sale are measured appropriately initially and on an ongoing basis;
- transfers of servicing rights have been properly accounted for as sales or secured borrowings;
- i. servicing rights are properly measured upon and subsequent to initial recognition, including key assumptions such as prepayment speeds, discount rates and market cost-to-service;
- j. derivatives are properly identified, valued, recorded and disclosed;
- k. the entity has complied with the disclosure objectives and requirements for transfers of financial assets in FASB ASC 860 for all transfers of financial assets and other applicable disclosures in other FASB ASC topics;
- the entity has identified a complete population of VIEs in which it
  has variable interests and arrived at the appropriate conclusion as
  to whether each should be consolidated; and
- m. the entity has complied with the disclosure objectives and requirements for VIEs in FASB ASC 810.

## **Planning**

10.115 In accordance with AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (as described in chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide). The following procedures related to loan transfers and mortgage banking activities assist the auditor in obtaining the required understanding:

- a. Obtain an understanding of the entity's transfer and servicing activities
- b. Inquire about the nature and frequency of transfers, the types of loans transferred, the extent of continuing involvement with the transferred assets, and the nature of obligations incurred
- c. Inquire about the processes, procedures, and controls employed by the entity at both the time of transfer and thereafter for recognition, measurement, and disclosures related to the transfer and servicing of assets
- 10.116 To obtain an understanding of mortgage banking activities in which the institution is engaged, the auditor may inquire about how the mortgage banking activities relate to management's objectives for managing

interest-rate risk and enhancing liquidity. The auditor may also inquire about the reporting systems used by management to account for mortgage banking activities and may consider whether management has sufficient data to evaluate loan sale transactions, identify loans HFS, and track mortgage loan commitments and applications. Such information is usually needed to manage risks arising from mortgage banking activities.

10.117 Institutions acting as servicers of loans have a fiduciary responsibility to parties under the agreement. Failure to meet these responsibilities may result in contingent liabilities that could have a material effect on an institution's financial statements. Under contracts with third parties such as Ginnie Mae, Freddie Mac, Fannie Mae, and the U.S. Department of Housing and Urban Development (HUD), an institution must meet certain minimum net worth requirements. Failure to meet the requirements could result in termination of the servicing contract or the loss of a seller servicer number. In addition, the auditor should consider and evaluate the risk of material misstatement of the financial statements that could result from the entity's mortgage banking activities including the institution's servicing systems controls over investor and escrow accounts (for example, for taxes and insurance or loan principal and interest) and the potential for contingent liabilities associated with noncompliance with investor-servicing requirements.

10.118 Contractual agreements with Ginnie Mae, Freddie Mac, Fannie Mae, HUD, or other investors may require engagements related to aspects of the contractual agreement or to the SEC Regulation AB or the *Uniform Single Attestation Program for Mortgage Bankers*. These agreements may require confirmation of the loans being serviced under specific contracts.

# Internal Control Over Financial Reporting and Possible Tests of Controls

10.119 AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13—.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit (such as mortgage loans HFS, participations sold, and individually significant securitization transactions) and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

- **10.120** The discussions of internal control activities in chapters 8–9 of this guide are also relevant to loan transfers and mortgage banking activities.
- **10.121** *Policies and procedures.* Examples of typical internal control activities relating to financial reporting of mortgage banking activities include, but are not limited to,
  - use of a quality control function to monitor underwriting and documentation practices;

- executive management review of open and pending commitments to buy or sell and strategies to minimize exposure to changing interest rates;
- loans transferred with continuing involvement are properly identified to determine whether sale or secured borrowing treatment is appropriate;
- review of ongoing valuation of outstanding derivatives;
- periodic reconciliation of cash receipts and payments applied to the servicing (custodial) system;
- periodic reconciliations of custodial accounts (the level of account activity could determine the frequency of reconciliation);
- periodic reconciliation of servicing fees received to servicing fee income recorded in the general ledger;
- periodic evaluation of the recoverability of servicing rights and other capitalized costs;
- procedures are established and reviewed for distinguishing loans HFS from those held for investment;
- procedures and systems are established to ensure that reconsideration of VIEs takes place on an ongoing basis;
- procedures to review investments and interests of the entity for VIE considerations;
- procedures to review the assumptions used in the accounting for servicing rights, including review of data inputs for accuracy, review of reasonableness of assumptions by a qualified person and review of the models used to calculate estimates; and
- review of recourse obligation and evaluation of potential liabilities.
- **10.122** Examples of typical internal control activities relating to financial reporting of loan transfers include, but are not limited to,
  - approval of sales by appropriate officers or committees;
  - periodic reconciliations of detailed trial balances to the general ledger balance of loans HFS:
  - procedures in place to engage counsel to provide a legal opinion when deemed necessary to support the assertion that transferred financial assets have been legally isolated from the transferor;
  - periodic review of the outstanding loans and locked-in borrower commitments for proper valuation;
  - procedures in place to ensure that derivative loan commitments, derivative sales contracts, and loans HFS are properly identified and valued;
  - procedures in place to ensure that interest and fee income and gains or losses on sales are properly recorded, and information required for GAAP disclosures is complete and accurate; and
  - procedures in place to ensure that assets obtained and liabilities incurred as a result of a transfer of a financial asset accounted for as a sale are properly accounted for initially and on an ongoing basis.

- **10.123** AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards), addresses the auditor's responsibility to design and implement responses to the risks of material misstatement identified and assessed by the auditor in accordance with AU-C section 315 and to evaluate the audit evidence obtained in an audit of financial statements.
- **10.124** In accordance with paragraph .08 of AU-C section 330, the auditor should design and perform tests of controls to obtain sufficient appropriate audit evidence about the operating effectiveness of relevant controls if (a) the auditor's assessment of risks of material misstatement at the relevant assertion level includes an expectation that the controls are operating effectively or (b) substantive procedures alone do not provide sufficient appropriate audit evidence at the relevant assertion level. Tests of controls that may be used to obtain evidence to support such an assessment include
  - selecting a sample of borrower remittances and testing reconciliation of payment amounts to income, principal, escrow, and service fee accounts;
  - testing custodial account reconciliations and supporting documentation to assess whether all activity is processed and cleared currently;
  - selecting a sample of delinquent loans serviced and considering whether collection and follow-up procedures are performed on a timely basis and are in accordance with investor requirements and management's policies; and
  - examining controls over loan documentation. (See chapter 8 of this guide.)
- **10.125** The following are examples of tests of controls that may be used by the auditor to ensure that internal control over financial reporting of mortgage banking activities are operating effectively:
  - Review accounting policy manual updates and consultations.
  - Review of valuation of outstanding derivatives.
  - Examine applicable loan sale and servicing agreements for appropriate review and approval.
  - Determine that the appropriate personnel understand the accounting treatment for sales of loans and related financial reporting implications.
  - Examine accounting records to determine that verification procedures are effectively performed to ensure interest and fee income and gains or losses on sales of loans are properly recorded.
  - Examine reconciliations of cash receipts and payments to custodial accounts and servicing fees to the general ledger.
  - Examine and assess management's valuation of servicing rights, including the controls over data inputs, review of assumptions and model utilized.
  - Test management's process for identifying VIEs, evaluating whether VIEs should be consolidations and identifying reconsideration events for VIEs.

- Examine and assess management's impairment analysis for servicing rights, if applicable, including management's stratification of the portfolio for the impairment analysis.
- Examine evidence of management's review of financial statements and footnotes to ensure that all required disclosures are included.
- Review of recourse obligations and potential liabilities to be recognized.

#### Substantive Tests

10.126 Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure, which for a financial institution may include transfers of loans and mortgage banking activities. In accordance with paragraph .A45 of AU-C section 330, this requirement reflects the facts that (a) the auditor's assessment of risk is judgmental and may not identify all risks of material misstatement and (b) inherent limitations to internal control exist, including management override.

10.127 Paragraph .06 of AU-C section 330 states that the auditor should design and perform further audit procedures whose nature, timing, and extent are based on, and responsive to, the assessed risks of material misstatement at the relevant assertion level. Examples of substantive tests that the auditor may perform include the following:

- Selecting a sample of borrower remittances and testing allocation of payment amounts to income, principal, escrow, and service fee accounts.
- Obtaining and testing the documentation supporting escrow and investor account reconciliations. (Custodial accounts may be offbalance-sheet accounts. Accordingly, the auditor may need to select custodial accounts from records independent of the general ledger. In this case, the auditor may need to perform separate tests of the completeness and accuracy of custodial records.)
- Obtaining and testing supporting documentation for derivatives and related fair value determinations.
- Evaluating the propriety of loan classifications to determine that all loans HFS within the loan portfolio are properly identified. (In evaluating whether loans are HFS or held for investment, the auditor should consider management policy and practices, for example, previous loan sale activity, types of loans sold, transactions subsequent to year-end, and pending contracts, and whether management has the ability and intent to hold the loans for the foreseeable future or until maturity.)
- Assessing the documentation and recalculating the amounts supporting the measurement of lower of cost or fair value for loans HFS, or fair value should that election have been made.
- Selecting a sample of financial asset transfers during the period and examining contracts to evaluate whether sale-versus-secured borrowing treatment and servicing assets and liabilities have been recognized properly. Interpretation No. 1, "The Use of Legal Interpretations As Audit Evidence to Support Management's

assertion That A Transfer of Financial Assets Has Met the Isolation Criterion in Paragraphs 7–14 of FASB ASC 860-10-40," of AU-C section 620, *Using the Work of an Auditor's Specialist* (AICPA, *Professional Standards*, AU-C sec. 9620 par. .01–.21), provides guidance to the auditor in obtaining sufficient appropriate audit evidence when an entity has derecognized financial assets in connection with a transfer to another entity.<sup>11</sup>

- Recalculating a sample of financial asset sale transactions to test gains or losses by testing the valuation of assets retained and liabilities assumed and vouching payments received for those transactions.
- Analytically projecting service fees for comparison to such expectations with service fee revenues reported in operating income for the period.
- Analytically assessing gain on sale to the volume of loans sold and comparing this to auditor expectations.
- Assessing the fair values of derivative loan commitments, derivative sales contracts, and loans HFS against the related notional amount and for directional consistency between the accounts.
- Evaluating the adequacy of valuation allowances for servicing and escrow advances. (Some investors require that contractual interest and principal be remitted to them by the servicer regardless of mortgagor performance. Advances of such amounts are frequently made in anticipation of borrower performance and generally must be tracked on an individual basis to limit exposure to uncollectible advances.)
- For servicing rights, assessing the assumptions used in the valuation process, considering their current reasonableness, and evaluating the effect of changes in assumptions on impairment or fair value measurement. Key assumptions include prepayment speeds, discount rates, cost-to-service and ancillary income.
- Analyzing prepayment data used by management to calculate the fair value of servicing rights at sale date and the systems used to update prepayment data over time for actual prepayment experience, selecting a sample of loan pools sold in prior periods,

In light of the issuance of FASB Statement No. 166 and the FDIC's changes to the safe harbor, the AICPA's ASB is currently in the process of revising this interpretation. Auditors should be alert for such revisions; however, the guidance in this interpretation continues to be relevant.

<sup>&</sup>lt;sup>11</sup> Interpretation No. 1, "The Use of Legal Interpretations as Audit Evidence to Support Management's Assertion That a Transfer of Financial Assets Has Met the Isolation Criterion in Paragraphs 7–14 of FASB ASC 860-10-40," of AU-C section 620, Using the Work of an Auditor's Specialist (AICPA, Professional Standards, AU-C sec. 9620 par. .01–.21), has not been updated to reflect the issuance of FASB Statement No. 166, Accounting for Transfers of Financial Assets. FASB Statement No. 166 was incorporated into FASB ASC in FASB ASU No. 2009-16, Transfers and Servicing (Topic 860): Accounting for Transfers of Financial Assets, and is discussed in FASB ASC 860.

In addition, this interpretation has not been updated for changes to the FDIC's safe harbor for financial assets transferred in connection with securitizations and participations. The FDIC's final amendments to the safe harbor, Treatment by the Federal Deposit Insurance Corporation as Conservator or Receiver of Financial Assets Transferred by an Insured Depository Institution in Connection With a Securitization or Participation After September 30, 2010 (www.fdic.gov/news/news/press/2010/pr10216.html), were issued in September 2010. The safe harbor provides important protections for securitizations and participations by confirming that in the event of a bank failure, the FDIC would not try to reclaim loans transferred into such transactions.

and comparing the actual current loan balance with management estimates.

- Evaluating the reasonableness of cost-to-service assumptions used to calculate the fair value of servicing rights, considering that a fair value should consider a market participant's cost-to-service rather than the institution's internal cost-to-service.
- Analytically assessing the ancillary income assumptions used to calculate the fair value of servicing rights.
- Analytically assessing the amount of initial servicing rights and ending servicing rights in relation to the related loan balances at those respective dates to auditor expectations.
- Evaluating the method of amortizing servicing rights, if applicable, including key assumptions like the methodology for stratification of the portfolio for impairment purposes.
- Evaluating the appropriateness of the liability for recourse obligations. Loan sale or servicing agreements generally address recourse provisions and should be reviewed for all substantial investors to ensure that portfolios sold with recourse are included in recourse liability considerations.
- Confirming selected loan balances serviced for others and related information directly with the borrower, investor, or trustee.
- Reviewing legal documents (such as bylaws and servicing agreements) for securitization trusts and other VIEs to determine that consolidation or nonconsolidation is appropriate.
- Obtaining legal opinions for securitizations derecognized from the balance sheet.
- Reviewing financial statements and footnotes to determine that all required disclosures have been made.

©2016. AICPA

# Chapter 11

# Real Estate Investments, Real Estate Owned, and Other Foreclosed Assets

# FASB Accounting Standards Update (ASU) No. 2014-09, Revenue from Contracts with Customers (Topic 606)

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14, Revenue from Contracts with Customers [Topic 606]: Deferral of the Effective Date) for annual reporting periods of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an overthe-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period. Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as of either:

- An annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- An annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB Accounting Standards Codification (ASC) 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605, *Revenue Recognition*, as well as guidance within the 900 series of industry-specific topics, including FASB ASC 942, *Financial Services—Depository and Lending*. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts).

Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

The AICPA has formed 16 industry task forces to assist in developing a new Accounting Guide on revenue recognition that will provide helpful hints and illustrative examples for how to apply the new standard. Revenue recognition implementation issues identified by the Depository and Lending Institutions Revenue Recognition Task Force will be available for informal comment, after review by the AICPA Financial Reporting Executive Committee, at the "Depository and Lending Institutions Revenue Recognition Task Force" page at www.aicpa.org.

Readers are encouraged to submit comments to revreccomments@aicpa.org.

©2016, AICPA AAG-DEP 11

For more information on FASB ASU No. 2014-09, see appendix E, "The New Revenue Recognition Standard: FASB ASU No. 2014-09," of this guide.

## Introduction

11.01 Generally, the largest component of real estate owned (REO), also known as other real estate owned (OREO), by lenders is assets taken in full or partial settlement of troubled loans through foreclosure or receiving physical possession of real estate property collateralizing a loan. Real estate investments, real estate loans that qualify as investments in real estate, and premises that are no longer used in operations may also be included in REO.¹ Once a bank obtains physical possession of the collateral, the real estate is included in REO. Furthermore, institutions may obtain assets other than real estate through foreclosure, and those assets also are addressed in this chapter.

#### Foreclosed Assets

11.02 Foreclosed assets include all assets received in full or partial satisfaction of a receivable and include real and personal property; equity interests in corporations, partnerships, and joint ventures; and beneficial interests in trusts.

#### Real Estate Investments

11.03 Some institutions make direct equity investments in real estate projects.

11.04 Further, some loans have terms that provide virtually the same risks and rewards as those of owners or joint venture participants and are accounted for as real estate investments. In such arrangements, the lender participates in expected residual profits, which may be in the form of an equity kicker or a higher than market effective interest rate. At the outset and during the construction and development of the property, the borrower generally has little or no equity in the property and the institution's only source of repayment is the property. The institution generally (a) agrees to provide substantially all funds to acquire, develop, and construct the property, (b) funds the commitment or origination fees or both, and (c) funds interest during the development and construction of the property.

# **Regulatory Matters**

11.05 As explained in the entry foreclosed assets in the "Glossary" section of the Federal Financial Institutions Examination Council's Instructions for Preparation of Consolidated Reports of Condition and Income, certain provisions of Statement of Position No. 92-3, Accounting for Foreclosed Assets, (rescinded with the issuance of FASB Statement No. 144, Accounting for the Impairment or Disposal of Long Lived Assets, [codified in FASB ASC 360, Property, Plant, and Equipment]) are not present in FASB ASC 360, but the application of these provisions represents prevalent practice in the banking

<sup>&</sup>lt;sup>1</sup> FASB Accounting Standards Update (ASU) No. 2014-04, Receivables—Troubled Debt Restructurings by Creditors (Subtopic 310-40): Reclassification of Residential Real Estate Collateralized Consumer Mortgage Loans upon Foreclosure (a consensus of the FASB Emerging Issues Task Force), clarifies when a creditor is considered to have received physical possession of residential real estate property collateralizing a consumer mortgage loan.

industry and is consistent with safe and sound banking practices and the accounting objectives set forth in Section 37(a) of the Federal Deposit Insurance Act. These provisions have been incorporated into the entry *foreclosed assets* in the "Glossary" section of the Federal Financial Institutions Examination Council's *Instructions for Preparation of Consolidated Reports of Condition and Income*. The instructions explain that

- for purposes of these reports, foreclosed assets (other than residential real estate property collateralizing a consumer mortgage loan) include loans where the bank, as creditor, has received physical possession of a borrower's assets, regardless of whether formal foreclose proceedings take place. The timing of physical possession is also explained.
- when a bank receives a long-lived asset, such as real estate, from a borrower in full satisfaction of a loan, the long-lived asset is rebuttably presumed to be held for sale (HFS) and the bank shall initially measure this asset at its fair value less cost to sell. This fair value (less cost to sell, if applicable) of the asset received in full satisfaction of the loan becomes the "cost" of the asset. Cost to sell includes items like estimated sales commissions on the sale of the property.
- an asset received in partial satisfaction of a loan should be initially
  measured at its fair value less cost to sell and the recorded amount
  of the loan should be reduced by the fair value (less cost to sell, if
  applicable) of the asset at the time of restructuring, foreclosure,
  or repossession.
- after foreclosure, each foreclosed real estate asset (including any real estate for which the bank receives physical possession) must be carried at the lower of (1) the fair value of the asset minus the estimated costs to sell the asset or (2) the cost of the asset. This determination must be made on an asset-by-asset basis.
- 11.06 Voluntary direct investments in real estate are generally limited for national banks, as described in Title 12 U.S. Code of Federal Regulations (CFR) Part 7.100. State member banks may invest in real estate only with the prior approval of the Board of Governors of the Federal Reserve System (Federal Reserve) as described in Regulation H. For REO acquired through foreclosure, the maximum allowable holding period is generally five years from the date of a bank receiving legal title, unencumbered by any redemption period applicable, to the property. A bank may request an extension not to exceed an additional five years, if (1) the institution has made a good faith attempt to dispose of the real estate within the five-year period, or (2) disposal within the five-year period would be detrimental to the institution, and the holding period regulatory guidance applies to all OREO properties the bank holds legal title to, even if the property has been written down to a recorded amount of zero.
- 11.07 Foreclosure management. In July 2012, the Federal Reserve issued guidance to address a lender's decision to discontinue foreclosure proceedings. The objective of the guidance is to emphasize the importance of appropriate risk management practices and controls in connection with a decision not to complete foreclosure proceedings after they have been initiated. Responsibilities of management include the following:

- Communication with borrowers. Supervised banking organizations should use all means possible to provide the previously described notification to affected borrowers, particularly those who prematurely vacated their homes based on the servicers' initial communications regarding foreclosure actions.
- Notification to local authorities. Supervised banking organizations should ensure that their procedures include reasonable efforts to notify appropriate state or local government authorities of the organization's decision not to pursue a foreclosure, including complying with applicable state or local government notification requirements.
- Obtaining and monitoring collateral values. Supervised banking
  organizations should have a process for obtaining the best practicable information on the collateral value of a residential property
  that may be subject to foreclosure and for updating this information on a regular basis.

Readers can access this guidance in Supervision and Regulation (SR) letter 12-11/Consumer Affairs (CA) letter 12-10, *Guidance on a Lender's Decision to Discontinue Foreclosure Proceedings*, from the Federal Reserve website at www.federalreserve.gov.

- 11.08 In April 2013, the Federal Reserve issued guidance on sound business practices for residential mortgage servicing that Federal Reserve supervised financial institutions are expected to address in their collections, loss mitigation, and foreclosure processing functions. The guidance confirms the minimum standards that all regulated institutions are expected to adopt in prioritizing and handling borrowers' files with imminent risk of foreclosure. These minimum review criteria are intended to ensure a level of consistency across servicers and should be used to determine whether a scheduled foreclosure sale should be postponed, suspended, or cancelled because of critical foreclosure defects in the borrower's file. Readers can access this guidance in SR letter 13-9/CA letter 13-6, Minimum Standards for Prioritization and Handling Borrower Files with Imminent Scheduled Foreclosure Sale, from the Federal Reserve website at www.federalreserve.gov.
- 11.09 Disposal of problem assets through exchanges. In December 2011, the Federal Reserve issued SR letter 11-15, Disposal of Problem Assets through Exchanges. The guidance addresses the risks involved in asset exchanges, the associated risk management considerations for institutions, and the supervisory considerations for examiners. The guidance also provides a detailed example of an asset exchange transaction. The example highlights the potential risks associated specifically with transactions that may reduce problem assets in the short term but, because of a lack of appropriate, up-front due diligence, may result in heightened risks over the longer term. Readers can access the guidance in SR letter 11-15 from the Federal Reserve website at www.federalreserve.gov.
- 11.10 Rental of Residential OREO Properties. In April 2012, the Federal Reserve issued SR letter 12-5/CA letter 12-2, Policy Statement on Rental of Residential OREO Properties, (policy statement) to remind banking organizations and examiners that the Federal Reserve's regulations and policies permit the rental of OREO properties as part of an orderly disposition strategy within statutory and regulatory limits. The policy statement applies to state member

banks, bank holding companies, nonbank subsidiaries of bank holding companies, savings and loan holding companies, nonthrift subsidiaries of savings and loan holding companies, and U.S. branches and agencies of foreign banking organizations (collectively, banking organizations).

- 11.11 While the general policy of the Federal Reserve is that banking organizations should make good-faith efforts to dispose of OREO properties at the earliest practicable date. The policy statement indicates that banking organizations may rent one- to four-family residential OREO properties without having to demonstrate continuous active marketing of the properties, provided that suitable policies and procedures are followed.
- 11.12 Key risk management considerations for banking organizations that engage in the rental of residential OREO include compliance with holding-period requirements for OREO, compliance with landlord-tenant and associated requirements, and accounting for OREO assets in accordance with U.S. generally accepted accounting principles (GAAP). Residential OREO is typically treated as a substandard asset as defined by the interagency classification guidelines. However, residential properties with leases in place and demonstrated cash flow from rental operations sufficient to generate a reasonable rate of return should generally not be classified.
- 11.13 The policy statement also establishes specific supervisory expectations for banking organizations that undertake large-scale residential OREO rentals (generally, 50 properties or more available for rent). Such organizations should have formal policies and procedures governing the operation and administration of OREO rental activities, including property-specific rental plans, policies and procedures for compliance with applicable laws and regulations, a risk management framework, and oversight of third-party property managers. For more information, readers can access SR letter 12-5/CA letter 12-3 from the Federal Reserve website at www.federalreserve.gov.
- 11.14 OREO management. In June 2012, the Federal Reserve issued SR letter 12-10/CA letter 12-9, Questions and Answers for Federal Reserve-Regulated Institutions Related to the Management of Other Real Estate Owned (OREO), to clarify existing policies and promote prudent practices for the management of an institution's OREO assets. Topics covered in the question and answer document include transferring an asset to OREO, reporting treatment and classification, appraisal concepts, ongoing property management, operational and legal issues, and sale and transfer of OREO. Readers can access SR letter 12-10/CA letter 12-9 from the Federal Reserve website at www.federalreserve.gov.
- 11.15 Appraisals. All OREO whether actively marketed or not actively marketed, should be supported by appropriate documentation to support appraised value. Readers should refer to the discussion beginning in paragraph 8.63 of this guide for regulatory guidance addressing appraisals. Further discussion on appraisal requirements for OREO can also be found in 12 (CFR) Part 34.85(a)(2).
- 11.16 Bank Accounting Advisory Series (BAAS). The Office of the Comptroller of the Currency's BAAS is updated periodically to express the Office of the Chief Accountant's current views on accounting topics of interest to national banks and federal savings associations. See further discussion of the BAAS in paragraph 7.82 of this guide. Topic 5, "OREO and Other Assets," of the BAAS includes interpretations and responses related to OREO

(for example, when a bank should derecognize a loan collateralized by property other than residential real estate, such as a commercial real estate loan and an auto loan.) In addition, Topic 2 A, "Troubled Debt Restructuring," includes discussion on when discharged debt should not to be reported as OREO and classification and measurement guidance under GAAP for government guaranteed mortgage loans upon a bank's foreclosure of the property that collateralizes the loan. Readers are encouraged to view this publication under the "Publications—Bank Management" page at www.occ.gov.

# **Accounting and Financial Reporting**

### Foreclosed Assets

# © Update 11-1 Accounting and Reporting: Revenue From Contracts With Customers

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14) for annual reporting periods of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period. Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as of either:

- An annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- An annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB ASC 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605 as well as guidance within the 900 series of industry-specific topics. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts). Under FASB ASU No. 2014-09, in addition to other criteria to determine whether a contract exists, the bank must assess collectability to determine if a contract exists. If collection is not deemed probable, then the consideration received from the customer is recognized as a liability.

Readers should be aware that the sale of OREO is within the scope of FASB ASU No. 2014-09.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers

are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2014-09, see appendix E of this guide.

#### © Update 11-2 Accounting and Reporting: Credit Losses

FASB ASU No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, issued in June 2016, is effective for fiscal years of public business entities that are SEC filers beginning after December 15, 2019, including interim periods within those fiscal years.

For all other public business entities, the amendments in FASB ASU No. 2016-13 are effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years.

For all other entities, including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960, *Plan Accounting—Defined Benefit Pension Plans*, through FASB ASC 965, *Plan Accounting—Health and Welfare Benefit Plans*, FASB ASU No. 2016-13 is effective for fiscal years beginning after December 15, 2020, and interim periods within fiscal years beginning after December 15, 2021.

Early application is permitted for all entities as of the fiscal years beginning after December 15, 2018, including interim periods within those fiscal years.

FASB ASU No. 2016-13 creates FASB ASC 326, Financial Instruments—Credit Losses, to amend guidance on reporting credit losses for assets held at amortized cost basis and available-for-sale debt securities.

For assets held at amortized cost basis, FASB ASC 326 eliminates the probable initial recognition threshold in current GAAP and, instead, requires an entity to reflect its current estimate of all expected credit losses. The allowance for credit losses is a valuation account that is deducted from the amortized cost basis of the financial assets to present the net amount expected to be collected.

For available-for-sale debt securities, credit losses should be measured in a manner similar to current GAAP. However, FASB ASC 326 will require that credit losses be presented as an allowance rather than as a write-down.

FASB ASU No. 2016-13 affects entities holding financial assets and net investment in leases that are not accounted for at fair value through net income. The amendments affect loans, debt securities, trade receivables, net investments in leases, off-balance-sheet credit exposures, reinsurance receivables, and any other financial assets not excluded from the scope that have the contractual right to receive cash.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-13, see appendix G, "Accounting for Financial Instruments," of this guide.

©2016, AICPA AAG-DEP 11.16

- 11.17 FASB ASC 310-40-40 applies to the initial measurement of a fore-closed property (including when the creditor receives physical possession of the debtor's assets, regardless of whether formal foreclosure proceedings take place). Paragraphs 2–3 of FASB ASC 310-40-40 provide that a creditor that receives real estate or other assets from a debtor in full satisfaction of a receivable should record the asset received at fair value at the time of the restructuring. If the long-lived asset received will be sold, those assets should be recorded at their fair value less cost to sell, as that term is used in FASB ASC 360-10-35-43. In addition, FASB ASC 310-40-35-7 provides that an asset received in partial satisfaction of a receivable combined with a modification of the debt should account for the asset received as prescribed in paragraphs 2–4 of FASB ASC 310-40-40 (fair value less cost to sell if the asset received will be sold) and reduce the recorded investment in the receivable by a corresponding amount.
- 11.18 Further, "Pending Content" in FASB ASC 310-40-40-6 explains that, except in the circumstances described in FASB ASC 310-40-40-6A (see paragraph 11.19), a troubled debt restructuring that is, in substance, a repossession or foreclosure by the creditor (that is, the creditor receives physical possession of the debtor's assets, regardless of whether formal foreclosure proceedings take place, or in which the creditor otherwise obtains one or more of the debtor's assets in place of all or part of the receivable) should be accounted for according to the provisions of FASB ASC 310-40-35-7 (see paragraph 11.17), paragraphs 2–4 of FASB ASC 310-40-40 (see paragraph 11.17), and if appropriate, FASB ASC 310-40-40-8. See paragraphs 7A–7B of FASB ASC 310-40-40 for the classification and measurement of certain government-guaranteed mortgage loans. For guidance on when a creditor should be considered to have received physical possession (resulting from an in substance repossession or foreclosure) of residential real estate properly collateralizing a consumer mortgage loan, see FASB ASC 310-40-55-10A.
- **11.19** The guidance in FASB ASC 310-40-40-7 (see paragraph 11.20) applies to initial measurement of a foreclosed property in a transaction having all of the following characteristics:
  - *a*. A sale of real estate was financed by the seller.
  - b. The buyer's initial investment was not sufficient for recognition of profit under the full accrual method.
  - c. The seller met the conditions of FASB ASC 970-605 to record a sale and recognized profit on the installment or cost recovery method.
  - d. Subsequently, the buyer defaulted on the mortgage to the seller.
  - e. The seller forecloses on the property.
  - f. At the time of foreclosure, the fair value of the property is less than the seller's gross receivable but greater than the seller's net receivable, that is, the principal and interest receivable less the deferred profit on the sale and related allowances.
- 11.20 At the time of foreclosure, foreclosed property meeting the characteristics of FASB ASC 310-40-6A (see paragraph 11.19) should be recorded at the lower of the net amount of the receivable or fair value of the property, in accordance with FASB ASC 310-40-40-7. The net receivable assumes that the accrual of interest income on the financing, if any, is appropriate under the circumstances. FASB ASC 310, *Receivables*, would be applied to a foreclosure related to a sale accounted for under the full accrual method, and if appropriate, the repossessed property would be recorded at its fair value. The "Impairment

and Disposal of Long-Lived Assets" subsections of FASB ASC 360-10 require a foreclosed asset that is newly acquired and that is classified as HFS to be recognized at the lower of its carrying value or fair value less cost to sell.

- 11.21 In accordance with "Pending Content" in paragraphs 7A–7B of FASB ASC 310-40-40, a guaranteed mortgage loan receivable should be derecognized and a separate other receivable be recognized upon foreclosure (that is, when a creditor receives physical possession of real estate property collateralizing a mortgage loan in accordance with the guidance in FASB ASC 310-40-40-6 (see paragraph 11.18) if the following conditions are met:
  - The loan has a government guarantee that is not separable from the loan before foreclosure.
  - At the time of foreclosure, the creditor has the intent to convey the real estate property to the guarantor and make a claim on the guarantee, and the creditor has the ability to recover under the claim.
  - At the time of foreclosure, any amount of the claim that is determined on the basis of the fair value of the real estate is fixed.

Upon foreclosure, the separate other receivable should be measured based on the amount of the loan balance (principal and interest) expected to be recovered from the guarantor.

11.22 FASB ASC 310-10-35-32 establishes guidance on the accounting for loans when foreclosure is probable and requires that, regardless of the measurement method, a creditor should measure impairment based on the fair value of the collateral if the creditor determines that foreclosure is probable.

#### Financial Statement Presentation and Disclosure

11.23 FASB 310-10-45-3 requires that foreclosed and repossessed assets should be classified as a separate balance sheet amount or included in other assets in the balance sheet, with separate disclosures in the notes to the financial statements. Certain returned or repossessed assets, such as inventory, should not be classified separately if the assets subsequently are to be utilized by the entity in operations. In accordance with FASB ASC 310-10-50-11, an entity should also disclose the carrying amount of foreclosed residential real estate properties held at the reporting date as a result of obtaining physical possession in accordance with FASB ASC 310-40-40-6 (see paragraph 11.18) and FASB ASC 310-40-55-10A.

# Accounting and Reporting for Long-Lived Assets to Be Disposed of by Sale

- 11.24 "Pending Content" in FASB ASC 360-10-05-4 states that the "Impairment or Disposal of Long-Lived Assets" subsections of FASB ASC 360-10 provide guidance for measurement of long-lived assets to be disposed of by sale and disclosures about the impairment or disposal of long-lived assets and disposals of individually significant components of an entity.
- 11.25 FASB ASC 360-10-45-9 states that a long-lived asset (disposal group) to be sold should be classified as HFS in the period in which all of the following criteria are met:
  - a. Management, having the authority to approve the action, commits to a plan to sell the asset (disposal group).

- b. The asset (disposal group) is available for immediate sale in its present condition subject only to terms that are usual and customary for sales of such assets (disposal groups).
- c. An active program to locate a buyer and other actions required to complete the plan to sell the asset (disposal group) have been initiated.
- d. The sale of the asset (disposal group) is probable, and transfer of the asset (disposal group) is expected to qualify for recognition as a completed sale, within one year, except as permitted by FASB ASC 360-10-45-11.
- e. The asset (disposal group) is being actively marketed for sale at a price that is reasonable in relation to its current fair value. The price at which a long-lived asset (disposal group) is being marketed is indicative of whether the entity currently has the intent and ability to sell the asset (disposal group). A market price that is reasonable in relation to fair value indicates that the asset (disposal group) is available for immediate sale, whereas a market price in excess of fair value indicates that the asset (disposal group) is not available for immediate sale.
- f. Actions required to complete the plan indicate that it is unlikely that significant changes to the plan will be made or that the plan will be withdrawn.
- 11.26 FASB ASC 360-10-45-10 states that if, at any time, the preceding criteria are no longer met (except as permitted by FASB ASC 360-10-45-11), a long-lived asset (disposal group) classified as HFS should be reclassified as held and used in accordance with FASB ASC 360-10-35-44.
- 11.27 FASB ASC 360-10-35-43 states that a long-lived asset (disposal group) classified as HFS should be measured at the lower of its carrying amount or fair value less cost to sell. If the asset (disposal group) is newly acquired, the carrying amount of the asset (disposal group) should be established based on its fair value less cost to sell at the acquisition date. A long-lived asset should not be depreciated (amortized) while it is classified as HFS. Interest and other expenses attributable to the liabilities of a disposal group classified as HFS should continue to be accrued.
- **11.28** A valuation allowance for a loan collateralized by a long-lived asset should not be carried over as a separate element of the cost basis for purposes of accounting for the long-lived asset under FASB ASC 360 after foreclosure, as stated in FASB ASC 310-40-40-10.
- 11.29 In accordance with FASB ASC 360-10-35-40, a loss should be recognized for any initial or subsequent write-down to fair value less cost to sell. A gain should be recognized for any subsequent increase in fair value less cost to sell, but not in excess of the cumulative loss previously recognized (for a write-down to fair value less cost to sell). The loss or gain should adjust only the carrying amount of a long-lived asset, whether classified as HFS individually or as part of a disposal group. A gain or loss not previously recognized that results from the sale of a long-lived asset (disposal group) should be recognized at the date of sale according to FASB ASC 360-10-40-5.

#### Financial Statement Presentation and Disclosure

- 11.30 "Pending Content" in FASB ASC 360-10-45-5 states that a gain or loss recognized (see FASB ASC 610-20 on the sale or transfer of a nonfinancial asset) on the sale of a long-lived asset (disposal group) that is not a discontinued operation should be included in income from continuing operations before income taxes in the income statement of a business entity. If a subtotal such as income from operations is presented, it should include the amounts of those gains or losses.
- 11.31 "Pending Content" in FASB ASC 360-10-45-14 states that a long-lived asset classified as HFS (but not qualifying for presentation as a discontinued operation in the statement of financial position in accordance with FASB ASC 205-20-45-10) should be presented separately in the statement of financial position of the current period. The assets and liabilities of a disposal group classified as HFS should be presented separately in the asset and liability sections, respectively, of the statement of financial position. Those assets and liabilities should not be offset and presented as a single amount. The major classes of assets and liabilities classified as HFS should be separately presented on the face of the statement of financial position or disclosed in the notes to the financial statements (see FASB ASC 360-10-50-3 in paragraph 11.32e).
- 11.32 For any period in which a long-lived asset (disposal group) either has been disposed of or is classified as HFS (see FASB ASC 360-10-45-9 in paragraph 11.25), "Pending Content" in FASB ASC 360-10-50-3 states that an entity should disclose the following information in the notes to the financial statements:
  - a. A description of the facts and circumstances leading to the disposal or expected disposal.
  - b. The expected manner and timing of that disposal.
  - c. The gain or loss recognized in accordance with paragraphs 37–45 of FASB ASC 360-10-35 and FASB ASC 360-10-40-5.
  - d. If not separately presented on the face of the statement where net income is reported, the caption in the statement where net income is reported that includes the gain or loss.
  - e. If not separately presented on the face of the statement of financial position, the carrying amounts of the major classes of assets and liabilities included as part of a disposal group classified as HFS. Any loss recognized on the disposal group classified as HFS in accordance with paragraphs 37–45 of FASB ASC 360-10-35 and FASB ASC 360-10-40-5 should not be allocated to the major classes of assets and liabilities of the disposal group.
  - f. If applicable, the segment(s) in which the long-lived asset (disposal group) is reported under FASB ASC 280, Segment Reporting.
- 11.33 In addition to the disclosures in FASB ASC 360-10-50-3 (see paragraph 11.32), "Pending Content" in FASB ASC 360-10-50-3A states that if a long-lived asset (disposal group) includes an individually significant component of an entity that either has been disposed of or is classified as held for sale (see FASB ASC 360-10-45-9 in paragraph 11.25) and does not qualify for presentation and disclosure as a discontinued operation (see FASB ASC 205-20 on discontinued operations), a public business entity and a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market should disclose

the information in item (a). All other entities should disclose the information in item (b).

- a. For a public business entity and a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, both of the following:
  - The pretax profit or loss of the individually significant component of an entity for the period in which it is disposed of or is classified as held for sale and for all prior periods that are presented in the statement where net income is reported calculated in accordance with paragraphs 6–9 of FASB ASC 205-20-45
  - 2. If the individually significant component of an entity includes a noncontrolling interest, the pretax profit or loss attributable to the parent for the period in which it is disposed of or is classified as held for sale and for all prior periods that are presented in the statement where net income is reported
- b. For all other entities, both of the following:
  - 1. The pretax profit or loss of the individually significant component of an entity for the period in which it is disposed of or is classified as held for sale calculated in accordance with paragraphs 6–9 of FASB ASC 205-20-45
  - 2. If the individually significant component of an entity includes a noncontrolling interest, the pretax profit or loss attributable to the parent for the period in which it is disposed of or is classified as held for sale

# Accounting and Reporting for Long-Lived Assets to Be Held and Used

- 11.34 The "Impairment or Disposal of Long-Lived Assets" subsections of FASB ASC 360-10 provide guidance for recognition and measurement of the impairment of long-lived assets to be held and used.
- 11.35 Paragraphs 16–36 of FASB ASC 360-10-35 address how long-lived assets or asset groups that are intended to be held and used in an entity's business should be reviewed for impairment. FASB ASC 360-10-35-21 requires that a long-lived asset (asset group) be tested for recoverability whenever events or changes in circumstances indicate that its carrying amount may not be recoverable. The carrying amount of a long-lived asset (asset group) is not recoverable if it exceeds the sum of the undiscounted cash flows expected to result from the use and eventual disposition of the asset (asset group), as stated in FASB ASC 360-10-35-17. That assessment should be based on the carrying amount of the asset (asset group) at the date it is tested for recoverability, whether in use (see FASB ASC 30-10-35-33) or under development (see FASB ASC 360-10-35-34). In accordance with FASB ASC 360-10-35-29, estimates of future cash flows used to test recoverability of a long-lived asset (asset group) should include only the future cash flows (cash inflows less associated cash outflows) that are directly associated with and that are expected to arise as a direct result of the use and eventual disposition of the asset (asset group). Those estimates should exclude interest charges that will be recognized as an expense when incurred.

- 11.36 FASB ASC 360-10-35-17 goes on to state that an impairment loss should be recognized only if the carrying amount of the long-lived asset (asset group) is not recoverable and exceeds its fair value. An impairment loss should be measured as the amount by which the carrying amount of the long-lived asset (asset group) exceeds its fair value. Paragraphs 23–25 of FASB ASC 360-10-35 provide guidance on grouping long-lived assets classified as held and used.
- 11.37 When a long-lived asset (asset group) is tested for recoverability, FASB ASC 360-10-35-22 states that it also may be necessary to review depreciation estimates and methods as required by FASB ASC 250, Accounting Changes and Error Corrections, or the amortization period as required by FASB ASC 350, Intangibles—Goodwill and Other. Paragraphs 17–20 of FASB ASC 250-10-45 and FASB ASC 250-10-50-4 address the accounting for changes in accounting estimates, including changes in the method of depreciation, amortization, and depletion. Paragraphs 1–5 of FASB 350-30-35 address the determination of the useful life of an intangible asset. Any revision to the remaining useful life of a long-lived asset resulting from that review also should be considered in developing estimates of future cash flows used to test the asset (asset group) for recoverability (see paragraphs 31–32 of FASB ASC 360-10-35). However, any change in the accounting method for the asset resulting from that review should be made only after applying FASB ASC 360-10.
- 11.38 If an impairment loss is recognized, the adjusted carrying amount of a long-lived asset should be its new cost basis according to FASB ASC 360-10-35-20. For a depreciable long-lived asset, the new cost basis should be depreciated (amortized) over the remaining useful life of that asset. Restoration of a previously recognized impairment loss is prohibited.

#### Financial Statement Presentation and Disclosure

- 11.39 In accordance with FASB ASC 360-10-45-4, an impairment loss recognized for a long-lived asset (asset group) to be held and used should be included in income from continuing operations before income taxes in the income statement of a business entity. If a subtotal such as income from operations is presented, it should include the amount of that loss.
- **11.40** As required by FASB ASC 360-10-50-2, the following information should be disclosed in the notes to the financial statements that include the period in which an impairment loss is recognized:
  - a. A description of the impaired long-lived asset (asset group) and the facts and circumstances leading to the impairment
  - b. If not separately presented on the face of the statement, the amount of the impairment loss and the caption in the income statement that includes that loss
  - c. The method or methods for determining fair value (whether based on a quoted market price, prices for similar assets, or another valuation technique)
  - d. If applicable, the segment in which the impaired long-lived asset (asset group) is reported under FASB ASC 280

#### Real Estate Investments

11.41 The "General" subsections of FASB ASC 970, *Real Estate—General*, provide accounting guidance concerning various forms of real estate investments. FASB ASC contains several topics for real estate due to differing accounting treatment for various real estate subindustries.

11.42 The "Acquisition, Development, and Construction" subsections of FASB ASC 310-10 provide guidance for determining whether a lender should account for an acquisition, development, and construction (ADC) arrangement as a loan or as an investment in real estate or a joint venture. The FASB ASC glossary defines an acquisition, development, and construction arrangement as an arrangement in which a lender, usually a financial institution, participates in expected residual profit from the sale or refinancing of property. In an ADC arrangement in which the lender participates in expected residual profit, in addition to the lender's participation in expected residual profit, FASB ASC 310-10-25-19 outlines certain characteristics that would suggest that the risks and rewards of the arrangement are similar to those associated with an investment in real estate or joint venture.

#### Financial Statement Presentation and Disclosure

11.43 FASB ASC 310-10-45-15 states that ADC arrangements accounted for as investments in real estate or joint ventures should be combined and reported in the balance sheet separately from those ADC arrangements accounted for as loans.

#### Sale of Real Estate Assets<sup>2</sup>

# © Update 11-3 Accounting and Reporting: Revenue From Contracts With Customers

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14) for annual reporting periods of a public business

<sup>2</sup> FASB ASU No. 2010-02, Consolidation (Topic 810): Accounting and Reporting for Decreases in Ownership of a Subsidiary—a Scope Clarification, added a scope exception for deconsolidation/derecognition of sales of in-substance real estate. Emerging Issues Task Force (EITF) Issue No. 10-E, "Accounting for Deconsolidation of a Subsidiary That Is In-Substance Real Estate," highlighted a concern for in-substance real estate in which there is an event or circumstance other than a sale in which an entity may be required to deconsolidate in-substance real estate because the reporting entity has lost a controlling financial interest but the reporting entity would not be able to derecognize the in-substance asset under the guidance in FASB Accounting Standards Codification (ASC) 360-20. FASB ASU No. 2011-10, Property, Plant, and Equipment (Topic 360): Derecognition of in Substance Real Estate—a Scope Clarification (a consensus of the FASB Emerging Issues Task Force), codified the consensus in EITF Issue No. 10-E and resolved diversity in practice by clarifying that when a parent (reporting entity) ceases to have a controlling financial interest (as described in FASB ASC 810-10) in a subsidiary that is in-substance real estate as a result of default on the subsidiary's nonrecourse debt, the reporting entity should apply the guidance in FASB ASC 360-20 to determine whether it should derecognize the in-substance real estate. Further, on the basis of the feedback received, the EITF recommended that the FASB chairman add a project to its agenda to address lenders' accounting in situations in which a borrower ceases to have a controlling financial interest in an in-substance real estate entity because of a default on its nonrecourse debt. After considering the recommendation of the EITF and the comments received on FASB ASU No. 2011-10, the FASB chairman added the research project titled Clarifying the Definition of a Business (formerly Application of Asset- or Entity-Based Guidance to Nonfinancial Assets in an Entity) to FASB's agenda to explore the broader issue of determining when an entity that substantially consists of nonfinancial assets should be accounted for as an in-substance asset or as an entity.

In May 2013, FASB elected to move the project into a standard-setting project. Currently, FASB has two open projects titled Clarifying the Definition of a Business Phase 1 (formerly Application of Asset- or Entity-Based Guidance to Nonfinancial Assets in an Entity) and Clarifying the Scope of Subtopic 610-20 and Accounting for Partial Sales of Nonfinancial Assets (formerly Clarifying the Definition of a Business Phase 2). Phase 1 of the projects is intended to clarify the definition of business with the objective of adding guidance to assist entities when evaluating whether transactions should be accounted for as acquisitions (or disposals) of assets or businesses. Phase 2 of the projects will include clarifying the guidance for partial sales or transfers of nonfinancial assets in the scope of FASB ASC 610-20 and the reference to in-substance nonfinancial assets. Readers are encouraged to visit the "Technical Agenda" page at www.fasb.org for the latest developments regarding the project, including a summary of decisions reached to date.

entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an overthe-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period. Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as of either:

- An annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- An annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB ASC 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605 as well as guidance within the 900 series of industry-specific topics. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts). Under FASB ASU No. 2014-09, in addition to other criteria to determine whether a contract exists, the bank must assess collectability to determine if a contract exists. If collection is not deemed probable, then the consideration received from the customer is recognized as a liability.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2014-09, see appendix E of this guide.

#### © Update 11-4 Accounting and Reporting: Leases

FASB ASU No. 2016-02, *Leases (Topic 842)*, issued in February 2016, is effective for fiscal years of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files financial statements with the SEC beginning after December 15, 2018, including interim periods within those fiscal years.

For all other entities, FASB ASU No. 2016-02 is effective for fiscal years beginning after December 15, 2019, and interim periods within fiscal years beginning after December 15, 2019.

Early application is permitted for all entities.

FASB ASU No. 2016-02 supersedes the lease requirements in FASB ASC 840, *Leases*, and creates FASB ASC 842, *Leases*, to establish the principles that lessees and lessors should apply to report useful information to users of financial statements about the amount, timing, and uncertainty of cash flows arising from a lease. FASB ASC 842 affects any entity that enters into a lease (as that term is defined in FASB ASU No. 2016-02), with some specified scope exceptions.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-02, see appendix F, "The New Leases Standard: FASB ASU No. 2016-02," of this guide.

- **11.44** FASB ASC 360-20 establishes standards for recognition of profit on all real estate sales transactions, other than retail land sales, without regard to the nature of the seller's business.
- 11.45 FASB ASC 360-20-40 presents the real estate derecognition guidance primarily from the perspective of the profit recognition upon a sale. This guidance addresses various conditions that may or may not result in derecognition of the real estate asset (and related profit or loss recognition, if any). The six primary methods of accounting for sales transactions are full accrual, installment, percentage-of-completion, reduced profit, cost recovery, and deposit. If an institution is selling discrete real estate projects, it may need to consider whether the projects would be considered a disposal of a component of an entity that represents a strategic shift that has (or will have) a major effect on an entity's operations and financial results and accordingly may need to be reported as a discontinued operation in accordance with paragraphs 1A–5 of FASB ASC 205-20-45. Other guidance on accounting for real estate sales that may apply includes FASB ASC 970-360 and FASB ASC 840-40.

## **Development Costs**

# © Update 11-5 Accounting and Reporting: Revenue From Contracts With Customers

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14) for annual reporting periods of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period. Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as of either:

- An annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- An annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB ASC 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605 as well as guidance within the 900 series of industry-specific topics. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts). Under FASB ASU No. 2014-09, in addition to other criteria to determine whether a contract exists, the bank must assess collectability to determine if a contract exists. If collection is not deemed probable, then the consideration received from the customer is recognized as a liability.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2014-09, see appendix E of this guide.

- 11.46 The "Real Estate Project Costs" subsections in FASB ASC 970-10 establish accounting and reporting standards for ADC, selling, and rental costs associated with real estate projects, as stated in FASB ASC 970-10-05-6. Project costs clearly associated with the ADC of a real estate project should be capitalized as a cost of that project, according to FASB ASC 970-360-25-2.
- 11.47 Payments to obtain an option to acquire real property should be capitalized as incurred, as stated in FASB ASC 970-340-25-3. All other costs related to a property that are incurred before the entity acquires the property, or before the entity obtains an option to acquire it, should be capitalized if all of the following conditions are met and otherwise should be charged to expense as incurred:
  - a. The costs are directly identifiable with the specific property.
  - b. The costs would be capitalized if the property were already acquired.
  - c. Acquisition of the property or of an option to acquire the property is probable (that is, likely to occur). This condition requires that the prospective purchaser is actively seeking to acquire the property and has the ability to finance or obtain financing for the acquisition and that there is no indication that the property is not available for sale.
- 11.48 Paragraphs 8–12 of FASB ASC 970-340-25 address the accounting for costs incurred on real estate for property taxes and insurance, costs of amenities, and incidental operations. Costs incurred on real estate for property taxes and insurance should be capitalized as property cost only during periods in which activities necessary to get the property ready for its intended

use are in progress. Accounting for costs of amenities should be based on management's plans for the amenities in accordance with the criteria stated in FASB ASC 970-340-25-9. Incremental revenues from incidental operations in excess of incremental costs of incidental operations should be accounted for as a reduction of capitalized project costs.

- 11.49 The capitalized costs of real estate projects should be assigned to individual components of the project based on specific identification, as stated in FASB ASC 970-360-30-1. If specific identification is not practical, capitalized costs should be allocated as follows:
  - a. Land costs and all other common costs, including the costs of amenities to be allocated as common costs per paragraphs 9–11 of FASB ASC 970-340-25 (before construction), should be allocated to each land parcel benefited. Allocation should be based on the relative fair value before construction.
  - b. Construction costs should be allocated to individual units in the phase on the basis of relative sales values of each unit.

If allocation based on relative values is also impractical, costs should be allocated based on area methods (for example, square footage) or other value methods as appropriate under the circumstances.

## Allocation of Income and Equity Among Parties to a Joint Venture

- 11.50 If a real estate investment is made through a joint venture arrangement, a formal agreement generally exists that specifies key terms, such as profit or loss allocations, cash distribution, and capital infusion provisions. The terms of these agreements may affect the institution's investment valuation and, accordingly, are considered in the investment evaluation process.
- 11.51 According to FASB ASC 970-323-35-16, joint venture agreements may designate different allocations among the investor for any of the following:
  - a. Profits and losses
  - b. Specified costs and expenses
  - c. Distributions of cash from operations
  - d. Distributions of cash proceeds from liquidation
- 11.52 FASB ASC 970-323-35-17 states that accounting by the investors for their equity in the venture's earnings under such agreements requires careful consideration of substance over form and consideration of the underlying values, as discussed in FASB ASC 970-323-35-10. Specified profit and loss allocation ratios should not be used to determine an investor's equity in venture earnings if the allocation of cash distributions and liquidating distributions are determined on some other basis.
- 11.53 If a specified allocation has no substance (for example, all depreciation is to be allocated to one partner but all cash distributions, including proceeds from the sale of real estate, are shared equally by all partners), it should be ignored. The agreement should be analyzed to determine how changes in net assets of the venture will affect cash payments to investors over the venture's life and at liquidation.
- 11.54 The institution should consider whether it is appropriate to allocate to other partners losses in excess of their capital contributions or whether the institution should record losses in excess of its own investment, including loans

and advances. Items that may affect the institution's decision are (a) the financial strength of the partners, (b) the type of partners (general versus limited) and the partners' legal requirement to fund losses, (c) the fair value of the real estate, and (d) the type of losses being incurred (cash or book). Paragraphs 2–11 of FASB ASC 970-323-35 provide guidance on investor accounting for losses in such circumstances.

11.55 FASB ASC 970-810 provides guidance on consolidation matters incremental to the real estate industry. FASB ASC 970-810-25-2 states that a noncontrolling investor in a general partnership should account for its investment by the equity method and should be guided by the provisions of FASB ASC 323, *Investments—Equity Method and Joint Ventures*. A controlling investor should account for its investment under the principles of accounting applicable to investments in subsidiaries. For guidance on determining control held by an investor, see paragraphs 1–2 of FASB ASC 970-810-25. For guidance on determining whether a general partner should consolidate a limited partnership or apply the equity method of accounting to its interests in the limited partnership, see FASB ASC 970-810-25-3.

# Auditing 4

#### **Objectives**

- 11.56 The primary objectives of audit procedures in the real estate investments, REO, and other foreclosed assets area are to obtain sufficient appropriate evidence that
  - a. the assets exist, are properly recorded, and are owned by the institution;
  - b. the assets are properly classified, described, and disclosed in the financial statements;
  - adequate provisions have been made for impairment, if any, of the assets;

<sup>&</sup>lt;sup>3</sup> FASB ASU No. 2015-02, Consolidation (Topic 810): Amendments to the Consolidation Analysis, was issued in February 2015. The amendments are effective for public business entities for fiscal years, and interim periods within those fiscal years, beginning after December 15, 2015. For all other entities, the amendments are effective for fiscal years beginning after December 15, 2016, and for interim periods within those fiscal years beginning after December 15, 2017. Early adoption is permitted, including adoption in an interim period. If an entity early adopts the amendments in an interim period, any adjustments should be reflected as of the beginning of the fiscal year that includes that interim period. A reporting entity may apply the amendments in this ASU using a modified retrospective approach by recording a cumulative-effect adjustment to equity as of the beginning of the fiscal year of adoption. A reporting entity also may apply the amendments retrospectively.

The amendments in this ASU affect reporting entities that are required to evaluate whether they should consolidate certain legal entities. All legal entities are subject to reevaluation under the revised consolidate model.

Readers are encouraged to read the full text of the ASU, available at www.fasb.org. Readers should apply the appropriate guidance based on their facts and circumstances.

<sup>&</sup>lt;sup>4</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of *issuers* and *nonissuers* as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

- d. depreciation expense, where applicable and other revenues and expenses related to real estate assets are properly allocated and reported;
- e. sales of assets, including the recognition of gains and losses, have been recognized; and
- f. appropriate disclosures have been made.

## **Planning**

11.57 In accordance with AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatements (AICPA, Professional Standards), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (as described in chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide). In obtaining that understanding, the auditor might consider the following factors related to real estate investments, REO, and other foreclosed assets that may indicate higher inherent risk in this area:

- Adverse environmental or economic conditions that may affect real estate markets and the values and liquidity of properties or other assets
- Significant losses on past sales of REO or other foreclosed assets
- Complex real estate assets
- Sales, financed by the institution, of REO or other foreclosed assets
- Lack of experienced real estate staff
- High concentrations of real estate or other assets in a particular geographic area
- Significant fluctuations in the amount and number of foreclosures or in-substance foreclosures
- Inexperienced internal appraisal personnel or the use of lowquality or outdated appraisals
- Appraisals in unstable market conditions

# Internal Control Over Financial Reporting and Possible Tests of Controls

11.58 AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13–.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit (such as, individually significant real estate in-vestments, REO and other foreclosed assets) and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C

section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

- 11.59 Inherent risk is often high for foreclosed assets and ADC arrangements because of the high degree of subjectivity involved in determining real estate values and the classification of ADC arrangements. However, with a high level of inherent risk in the real estate area, the auditor would often conclude that for most of the assertions it is more effective or efficient to assess control risk at the maximum and plan a primarily substantive approach, involving a selection of major real estate assets for detailed review.
- 11.60 AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards), addresses the auditor's responsibility to design and implement responses to the risks of material misstatement identified and assessed by the auditor in accordance with AU-C section 315 and to evaluate the audit evidence obtained in an audit of financial statements.
- 11.61 In accordance with paragraph .08 of AU-C section 330, the auditor should design and perform tests of controls to obtain sufficient appropriate audit evidence about the operating effectiveness of relevant controls if (a) the auditor's assessment of risks of material misstatement at the relevant assertion level includes an expectation that the controls are operating effectively or (b) substantive procedures alone do not provide sufficient appropriate audit evidence at the relevant assertion level. Internal control over financial reporting of real estate investments, REO, and other foreclosed assets may include
  - written policies and procedures, including those that address
    - frequency of appraisals and selection and qualifications of appraisers;
    - disbursement of funds and capitalization of costs;
    - review and monitoring of marketing efforts;
    - nature and amount of facilitating financing;
    - revenue recognition;
    - estimates of cost to sell; and
    - capitalization of interest.
  - proper authorizations for specific transactions.
  - periodic reviews of balances, fair values, realizable values, and policies by persons specified in management's written policy.
  - accounting for dispositions, including whether derecognition (sale) and profit recognition are appropriate.

#### Substantive Tests

11.62 Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure, which for a financial institution would include real estate investments, REO, and other foreclosed assets. In accordance with paragraph .A45 of AU-C section 330, this requirement reflects the facts that (a) the auditor's assessment of risk is judgmental and may

not identify all risks of material misstatement and (b) inherent limitations to internal control exist, including management override.

- about the carrying amount of foreclosed assets (fair values) and real estate investments (including loans that qualify as real estate investments) may involve a review of appraisals, feasibility studies, forecasts, sales contracts or lease commitments, and information concerning the track record of the developer. Being aware of the involvement of related parties may contribute to the design of audit procedures used by the auditor. To obtain appropriate audit evidence of progress to completion under a real estate investment or other real estate project, the auditor may also decide to perform an on-site inspection of certain properties.
- 11.64 Substantive tests of other real estate and real estate investments generally focus on the valuation assertion; however, tests of the other assertions should also be considered. For example, evidence about the completeness assertion may be obtained through the auditor's testing of loans. In addition, the auditor should consider testing the propriety of gains and losses on real estate sales and capitalized interest and other holding costs. The auditor may also evaluate appraisals performed close to a period end that may not be timely processed and included in the entity's records.
- 11.65 Estimates of the fair value of real estate assets are necessary to account for such assets. AU-C section 540, Auditing Accounting Estimates, Including Fair Value Accounting Estimates, and Related Disclosure (AICPA, Professional Standards), addresses the auditor's responsibilities related to accounting estimates, including fair value accounting estimates and related disclosures, in an audit of financial statements. Specifically, it expands on how AU-C section 315, AU-C section 330, and other relevant AU-C sections are to be applied with regard to accounting estimates. It also includes requirements and guidance related to misstatements of individual accounting estimates and indicators of possible management bias.
- 11.66 Many fair values will be based on valuations by independent appraisers. In applying audit procedures to real estate, the auditor often relies on representations of independent experts, particularly appraisers and construction consultants, to assist in the assessment of real estate values. AU-C section 500, Audit Evidence (AICPA, Professional Standards), addresses the auditor's use of the work of an individual or organization possessing expertise in a field other than accounting or auditing whose work in that field is used by the entity to assist the entity in preparing the financial statements (defined as a management's specialist).
- 11.67 If information to be used as audit evidence has been prepared using the work of a management's specialist, paragraph .08 of AU-C section 500 states that the auditor should, to the extent necessary, taking into account the significance of that specialist's work for the auditor's purposes,
  - a. evaluate the competence, capabilities, and objectivity of that specialist;
  - b. obtain an understanding of the work of that specialist; and
  - c. evaluate the appropriateness of that specialist's work as audit evidence for the relevant assertion.

- 11.68 Information regarding the competence, capabilities, and objectivity of a management's specialist may come from a variety of sources, such as knowledge of that specialist's qualifications, membership in a professional body or industry association, license to practice, or other forms of external recognition (a listing of additional sources is addressed in paragraph .A39 of AU-C section 500). Further application and explanatory material regarding the reliability of information produced by a management's specialist is addressed in paragraphs .A35—.A49 of AU-C section 500.
- 11.69 As addressed in the preceding paragraphs, independent appraisals may be considered acceptable audit evidence. The quality of appraisals varies, however, and, in some instances, the auditor may have reason to believe certain assumptions underlying appraisals are unrealistic. Some matters that should be considered by the auditor when evaluating an appraisal are
  - a rise or decline in a particular market area not reflected in an appraisal may warrant that additional procedures, or perhaps a new appraisal, be performed;
  - if the date of appraisal is substantially earlier than the audit date, a rise or decline in a particular market area between the two dates may warrant a new appraisal or the performance of additional procedures;
  - appraised values should be based on current market conditions and must be discounted for costs to complete and sell; and
  - the estimated selling prices should reflect the expectations of a sale in the reasonably near future—not in an indefinite future period.
- 11.70 In periods of declining real estate values, management might consider implementing policies and procedures that address
  - its process for evaluating appraisals that are not as of the balance sheet date to determine whether an adjustment to the appraised value is needed to reflect declines in fair value that are not reflected in the appraisals.
  - circumstances in which an updated appraisal should be obtained and the process for obtaining an updated appraisal.

Depending on materiality and level of inherent risk the auditor may also engage its own experts to review appraisals obtained by management or perform independent appraisals of collateral.

11.71 Because of time and cost considerations, an institution may use various approaches to estimate value without using the services of an independent appraiser. In evaluating internally derived valuation data, the auditor should understand the methods and assumptions used and the qualifications of the individual performing the evaluation and should be aware of inherent subjective determinations in estimating value that may be significant to the valuation process. The auditor should consider the reasonableness of the assumptions and approach used and should test the information underlying the valuation. Further, the auditor may decide to engage an appraiser independent of the institution to test the institution's internally derived valuation. Despite the existence of an appraisal, in certain situations the auditor may wish to physically observe properties for the stage of completion, for deterioration, or for estimating the extent of occupancy.

- 11.72 The auditor should also evaluate whether significant real estate transactions qualify as sales in conformity with criteria set forth in FASB ASC 360-20-40.
- 11.73 Other foreclosed assets. The procedures discussed previously may be applied to other foreclosed assets to the extent that the auditor deems necessary.

# Chapter 12

# Other Assets, Other Liabilities, and Other Investments

#### © Update 12-1 Accounting and Reporting: Leases

FASB Accounting Standards Update (ASU) No. 2016-02, Leases (Topic 842), issued in February 2016, is effective for fiscal years of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files financial statements with the SEC beginning after December 15, 2018, including interim periods within those fiscal years.

For all other entities, FASB ASU No. 2016-02 is effective for fiscal years beginning after December 15, 2019, and interim periods within fiscal years beginning after December 15, 2019.

Early application is permitted for all entities.

FASB ASU No. 2016-02 supersedes the lease requirements in FASB Accounting Standards Codification (ASC) 840, Leases, and creates FASB ASC 842, Leases, to establish the principles that lessees and lessors should apply to report useful information to users of financial statements about the amount, timing, and uncertainty of cash flows arising from a lease. FASB ASC 842 affects any entity that enters into a lease (as that term is defined in FASB ASU No. 2016-02), with some specified scope exceptions.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this chapter will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-02, see appendix F, "The New Leases Standard: FASB ASU No. 2016-02," of this guide.

# Introduction

**12.01** The following assets are among those frequently grouped as *other* assets or *other investments* in institutions' balance sheets; however, any that are individually material should be presented in the balance sheet as a separate line item:

- Accrued interest receivable (see chapter 7, "Investments in Debt and Equity Securities," of this guide for a discussion on securities and chapter 8, "Loans," of this guide for a discussion on loans)
- Premises and equipment, net
- Other real estate, such as foreclosed assets (see chapter 11, "Real Estate Investments, Real Estate Owned, and Other Foreclosed Assets," of this guide for a discussion on real estate investments, real estate owned, and other foreclosed assets)

©2016, AICPA AAG-DEP 12.01

- Servicing assets (SAs) (see chapter 10, "Transfers and Servicing and Variable Interest Entities," of this guide for a discussion of SAs)
- Federal Home Loan Bank (FHLB), Federal Reserve Bank (FRB), or other restricted stocks
- National Credit Union Share Insurance Fund (NCUSIF) deposits or other share insurance deposits
- Mortgage servicing advances
- Identifiable intangible assets, such as core deposit intangibles, and purchased credit-card relationships
- Indemnification assets arising from contractual indemnities provided in business combinations
- Goodwill
- Customers' liabilities on acceptances
- Deferred tax assets (see chapter 16, "Income Taxes," of this guide)
- Investments in equity securities that are not readily marketable (which meet definition of a security in FASB ASC 320, *Investments—Debt and Equity Securities*, but are not subject to its provisions because they are not readily marketable), such as stock in the Federal Agricultural Mortgage Corporation or stock in banker's banks.
- Other equity investments, including investments in joint ventures or venture capital investments, and credit union investments in credit union service organizations (CUSOs) and corporate credit unions
- Investments in nonnegotiable certificates of deposits (see chapter 6, "Cash and Cash Equivalents," of this guide)
- Bank, or credit union, owned life insurance
- Loans to executives for collateralized split-dollar life insurance agreements
- Prepaid pension assets

# © Update 12-2 Accounting and Reporting: Recognition and Measurement of Financial Assets and Financial Liabilities

FASB ASU No. 2016-01, Financial Instruments—Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities, issued in January 2016, is effective for fiscal years, and interim periods within those fiscal years, of a public business entity beginning after December 15, 2017.

For all other entities (including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960, *Plan Accounting—Defined Benefit Pension Plans* through FASB ASC 965, *Plan Accounting—Health and Welfare Benefit Plans* on plan accounting), FASB ASU No. 2016-01 is effective for fiscal years beginning after December 15, 2018, and interim period within fiscal years beginning after December 15, 2019. Early application is permitted for

all other entities as of the fiscal years beginning after December 15, 2017, including interim periods within those fiscal years.

All entities may adopt the presentation guidance in paragraphs 5–7 of FASB ASC 825-10-45 for financial statements of fiscal years or interim periods that have not yet been issued or that have not yet been made available for issuance. Furthermore, entities that are not public business entities may elect not to disclose the information about fair value of financial instruments required by the "General" subsection of FASB ASC 825-10-50 in financial statements of fiscal years or interim periods that have not yet been made available for issuance. Except as indicated previously in this paragraph, early application is not permitted.

FASB ASU No. 2016-01 addresses certain aspects of recognition, measurement, presentation, and disclosure of financial instruments and affects all entities that hold financial assets or owe financial liabilities. Among other provisions of the guidance, the amendments in this ASU supersede the guidance to classify equity securities with readily determinable fair values into different categories (that is, trading or available-for-sale) and require equity securities (including other ownership interests, such as partnerships, unincorporated joint ventures, and limited liability companies) to be measured at fair value with changes in the fair value recognized through net income (other than equity securities accounted for under the equity method of accounting or those that result in consolidation of an investee).

The amendments in this ASU also require an entity to present separately in other comprehensive income the portion of the total change in fair value of a liability resulting from a change in the instrument-specific credit risk when the entity has elected to measure the liability at fair value in accordance with the fair value option for financial instruments.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this discussion will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-01, see appendix G, "Accounting for Financial Instruments," of this guide.

#### **12.02** Other liabilities frequently include the following:

- Accounts payable
- Accrued interest payable (see chapter 13, "Deposits," of this guide for discussion of deposits)
- Accrued expenses
- Borrower's taxes and insurance escrows
- Bankers' acceptance liability
- Guarantee liabilities pursuant to FASB ASC 460, Guarantees
- Recourse liabilities
- Allowance related to unfunded loan commitments
- Servicing liabilities

- Asset retirement obligations
- Liabilities associated with exit or disposal activities
- Estimated litigation and settlement loss allowance
- Capital lease obligations
- Compensation and benefit-related accruals, such as bonuses, pension liabilities, supplemental executive retirement plans, and postretirement health care benefits including split-dollar life insurance arrangements
- Deferred tax liabilities
- Allowance for unrecognized tax benefits

# Premises and Equipment, Net

12.03 Premises and equipment consist primarily of land, buildings, furniture, fixtures, equipment, purchased software, and leasehold improvements used in institution operations. Such assets may be acquired directly or through a special-purpose subsidiary. Institutions may also lease property and equipment to other parties.

#### FHLB or FRB Stock

12.04 FHLB stock. Institutions that are members of the FHLB system are required to maintain a minimum investment in FHLB stock. The minimum is calculated as a percentage of aggregate outstanding mortgages. An additional investment calculated as a percentage of total FHLB advances, letters of credit, and the collateralized portion of interest-rate swaps outstanding may be necessary. FHLB stock is capital stock that is bought from and sold only to the FHLB at \$100 par. Both stock and cash dividends may be received on FHLB stock.

12.05 FRB stock. Members of the Federal Reserve System are required to maintain stock in the district FRB in a specified ratio to its capital. The stock does not provide the owner with control or financial interest in the FRB, is not transferable, and cannot be used as collateral. A member institution's ownership of FRB stock may be canceled in the event of the institution's insolvency or voluntary liquidation, conversion to nonmember status through merger or acquisition, or voluntary or involuntary termination of membership in the Federal Reserve System.

# Identifiable Intangibles

12.06 Identifiable intangible assets may be acquired individually, as part of a group of assets, or in a business combination in accordance with FASB ASC 805, *Business Combinations*. According to the FASB ASC glossary, *intangible assets* are assets (not including financial assets) that lack physical substance. Although goodwill is an intangible asset, the term *intangible assets* is used to refer to intangible assets other than goodwill, which may include, among others, core deposit intangibles (the value of long term deposit relationships) and credit-card customer lists (the value of long term credit-card relationships).

12.07 Paragraphs 9–15A of FASB ASC 805-20-25 provide guidance on recognizing identifiable intangible assets, including operating leases and

reacquired rights, acquired in a business combination. Paragraphs 11–45 of FASB ASC 805-20-55 present examples of identifiable intangible assets acquired in a business combination. See chapter 19, "Business Combinations," of this guide for additional information regarding this topic.

#### Goodwill

**12.08** Goodwill arises in a business combination. *Goodwill* is an asset representing the future economic benefits arising from other assets acquired in a business combination or an acquisition by a not-for-profit entity that are not individually identified and separately recognized, as defined in the FASB ASC glossary.

# **Customers' Liabilities on Acceptances**

**12.09** Customer's liabilities on acceptances represent a customer's outstanding debt to the institution that resulted from a banker's acceptance transaction. A banker's acceptance is a short term negotiable time draft drawn on and accepted by an institution.

#### Other Miscellaneous Items

12.10 Other items that may be classified as other assets include accounts receivable, accruals for miscellaneous fees, other prepaid expenses, payroll deductions receivable, and suspense accounts. Suspense accounts usually contain amounts related to items recorded and held pending classification and transfer to the proper account and may originate from a variety of sources, such as loan remittances, branch clearing transactions, automated teller machine transactions, and payroll transactions. Suspense accounts are generally recorded within "other liabilities" in the balance sheet but may have debit balances.

# Regulatory Matters

12.11 Section 107(4) of the Federal Credit Union Act permits federal credit unions to purchase, hold, and dispose of only that property which is necessary or incidental to their operations. Part 701.36 of the National Credit Union Administration (NCUA) regulations establishes occupancy, planning, and disposal requirements for acquired and abandoned premises, and prohibits certain transactions (such as, acquiring real property from certain related parties.<sup>1</sup>

12.12 As noted in chapter 17, "Equity and Disclosures Regarding Capital Matters," in 2013, the U.S. Banking regulators issued a final rule for U.S. adoption of the Basel III regulatory capital framework. Basel III includes a more stringent definition of capital and all intangible assets, other than mortgage servicing rights, are fully deducted from capital (net of associated deferred tax liabilities) after the current phase-in period which began on January 1, 2015 and will be fully phased-in January 1, 2018. Specific to mortgage servicing rights, under the revised capital rule any amount above 10 percent of a firm's common equity tier 1 capital must be deducted from common equity tier 1 capital. Starting January 1, 2018, any amount of the threshold items that is not

<sup>&</sup>lt;sup>1</sup> In July 2015, the National Credit Union Administration issued a final rule to amend its regulation governing federal credit union ownership of fixed assets. Among other amendments, the regulation eliminated a provision in the prior fixed assets rule that established a five percent aggregate limit on investments in fixed assets for federal credit union's with \$1 million or more in assets.

deducted from common equity tier 1 capital will be risk weighted at 250 percent (an increase from 100 percent). In addition, any amount of mortgage servicing assets, certain deferred tax assets arising from temporary differences, and significant investments in the capital of unconsolidated financial institutions in the form of common stock above 15 percent of a firm's common equity tier 1 capital must also be deducted from common equity tier 1 capital.

- 12.13 Both national banks, federal savings associations, and state member banks are limited as to the amount of their investments in bank premises. National bank and federal savings association limitations are set forth in Title 12 U.S. Code of Federal Regulations (CFR) Part 5.37 Banks and Banking, U.S. Code 12, Section 371d. State member bank limitations are set forth in 12 CFR 208.21.
- **12.14** See paragraph 10.10 of this guide for information on interagency guidelines on asset securitization.
- 12.15 The Office of the Comptroller of the Currency's *Bank Accounting Advisory Series* (BAAS) is updated periodically to express the Office of the Chief Accountant's current views on accounting topics of interest to national banks and federal savings associations. See further discussion of the BAAS in paragraph 7.82 of this guide. Topic 10B, "Intangible Assets," includes interpretations related to intangible assets. Readers are encouraged to view this publication under the "Publications—Bank Management" page at www.occ.gov.

# **Accounting and Financial Reporting**

## Premises and Equipment, Net

- 12.16 Financial institutions account for premises and equipment in the same way that commercial entities account for property and equipment (fixed assets). Institutions carry premises and equipment at cost less accumulated depreciation, and adjust the carrying amount for permanent impairments of value. Capital additions and improvements to premises should be capitalized, including construction period interest in accordance with FASB ASC 835-20. A description of the institution's depreciation and capitalization policies should be included in the notes to the financial statements.
- 12.17 In accordance with paragraphs 1–2 of FASB ASC 942-360-45, premises and equipment are generally shown as a single caption on the balance sheet, net of accumulated depreciation and amortization, the amount of which should be disclosed either on the face of the balance sheet or in the notes to the financial statements. For premises and equipment, net gains or net losses on dispositions should be included in noninterest income or noninterest expense.
- 12.18 Long-lived assets or asset groups should be tested for recoverability whenever events or changes in circumstances indicate that the carrying amount may not be recoverable as stated in FASB ASC 360-10-35-21. Capital leases should be accounted for in accordance with FASB ASC 840-30 and are subject to the requirements of the "Impairment or Disposal of Long-Lived Assets" subsections of FASB ASC 360-10, for purposes of recognizing and measuring impairment.

- **12.19** Premises and equipment held for use, including capital leases, need to be tested for recoverability whenever indicators of impairment are present. Paragraph 12.59 provides further guidance on impairment of long-lived assets.
- **12.20** Consolidation of subsidiaries that own premises and equipment should occur in accordance with FASB ASC 810, *Consolidation*.
- 12.21 In accordance with FASB ASC 810-10-15-3, all reporting entities should apply the guidance in FASB ASC 810 to determine whether and how to consolidate another entity and apply the applicable subsections of FASB ASC 810-10. The "Variable Interest Entities" subsections of FASB ASC 810-10 clarify the application of the "General" subsections to certain legal entities in which equity investors do not have sufficient equity at risk for the legal entity to finance its activities without additional subordinated financial support, or according to "Pending Content" in FASB ASC 810-10-05-8, 2 as a group, the holders of the equity investment at risk lack any of the following three characteristics:
  - a. The power, through voting rights or similar rights, to direct the activities of a legal entity that most significantly impact the entity's economic performance
  - b. The obligation to absorb the expected losses of the legal entity
  - c. The right to receive the expected residual returns of the legal entity.
- 12.22 FASB ASC 840-40 provides guidance on how to account for sale-leaseback transactions and various sale-leaseback matters.
- 12.23 FASB ASC 350-40 provides guidance on accounting for the cost of computer software developed or obtained for internal use and for determining whether the software is for internal use. According to FASB ASC 350-40-05-2, internal use software is acquired, internally developed, or modified solely to meet the entity's internal needs and during the software's development or modification, no substantive plan exists or is being developed to market

©2016, AICPA AAG-DEP 12.23

<sup>&</sup>lt;sup>2</sup> Many paragraphs in FASB Accounting Standards Codification (ASC) 810-10 are labeled as "Pending Content" and relate to FASB Accounting Standards Update (ASU) No. 2009-17, Consolidations (Topic 810): Improvements to Financial Reporting by Enterprises Involved with Variable Interest Entities. Although labeled as "Pending Content," this guidance is applicable for many entities except those that qualify for the deferral in FASB ASU No. 2010-10, Consolidation (Topic 810): Amendments for Certain Investment Funds. In February 2015, FASB issued ASU No. 2015-02, Consolidation (Topic 810): Amendments to the Consolidation Analysis. Among other provisions, the amendments in this ASU rescind the indefinite deferral in FASB ASU No. 2010-10 and provide a scope exception from FASB ASC 810, Consolidation, for reporting entities with interests in legal entities that are required to comply with or operate in accordance with requirements similar to those in Rule 2a-7 of the Investment Company Act of 1940 for registered money market funds. FASB ASU No. 2015-02 is effective for public business entities for fiscal years, and for interim periods within those fiscal years, beginning after December 15, 2015. For all other entities, the amendments in this ASU are effective for fiscal years beginning after December 15, 2016, and for interim periods within fiscal years beginning after December 15, 2017. Early adoption is permitted, including adoption in an interim period. If an entity early adopts the amendments in an interim period, any adjustments should be reflected as of the beginning of the fiscal year that includes that interim period. A reporting entity may apply the amendments in this ASU using a modified retrospective approach by recording a cumulative-effect adjustment to equity as of the beginning of the fiscal year of adoption. A reporting entity also may apply the amendments retrospectively. Readers should carefully review the transition guidance in FASB ASC 810-10-65-2 associated with the "Pending Content" to determine the applicability.

the software externally. Accounting for the cost of either internally developed and produced or purchased computer software to be sold, leased, or otherwise marketed is established in FASB ASC 985-20. 3.4

- **12.24** If the individual categories of assets are material, they should be disclosed on the face of the balance sheet or in the notes to the financial statements. The gross amount of assets recorded under capital leases as of the date of each balance sheet presented by major classes according to nature or function should be disclosed, in accordance with FASB ASC 840-30-50-1.
- **12.25** *Operating leases.* An *operating lease*, from the perspective of a lessor, is a lease that meets the conditions in item (d) in FASB ASC 840-10-25-43 and. from the perspective of a lessee, is any lease other than a capital lease, as defined in the FASB ASC glossary. The capital lease criteria are specified in FASB ASC 840-10-25-1. From the lessor's perspective, the leased assets are recorded on the balance sheet with or near property, plant, and equipment and lease payments are recognized as rental income in the income statement in accordance with FASB ASC 840-20-45-2 and 840-20-25-1, respectively. Long-lived assets of lessors subject to operating leases are also subject to the requirements of "Impairment or Disposal of Long-Lived Assets" subsections of FASB ASC 360-10 for purposes of recognizing and measuring impairment. From the lessee's perspective, rent should be charged to expense by lessees over the lease term as it becomes payable, in accordance with FASB ASC 840-20-25-1. If rental payments are not made on a straight-line basis, rental expense nevertheless should be recognized on a straight-line basis unless another systematic and rational basis is more representative of the time pattern in which use benefit is derived from the leased property, in which case that basis should be used.
- **12.26** State credit unions. Because the NCUA eliminated the 5 percent aggregate limit on investments in fixed assets for federal credit unions with \$1 million or more in assets, state credit unions in some instances may still have limits on the amount they can invest in fixed assets.

<sup>&</sup>lt;sup>3</sup> FASB ASU No. 2015-05, Intangibles—Goodwill and Other—Internal-Use Software (Subtopic 350-40): Customer's Accounting for Fees Paid in a Cloud Computing Arrangement, provides guidance to customers about whether a cloud computing arrangement includes a software license. If a cloud computing arrangement includes a software license, then the customer should account for the software license element of the arrangement consistent with the acquisition of other software licenses. If a cloud computing arrangement does not include a software license, the customer should account for the arrangement as a service contract. The guidance will not change U.S. generally accepted accounting principles for a customer's accounting for service contracts. In addition, the guidance in this ASU supersedes FASB ASC 350-40-25-16. Consequently, all software licenses within the scope of FASB ASC 350-40 will be accounted for consistent with other licenses of intangible assets.

FASB ASU No. 2015-05 is effective for public business entities for annual periods, including interim periods within those annual periods, beginning after December 15, 2015. For all other entities, the amendments in this ASU will be effective for annual periods beginning after December 15, 2015, and interim periods in annual periods beginning after December 15, 2016. Early adoption is permitted for all entities. Readers are encouraged to consult the full text of the ASU on FASB's website at www.fasb.org.

<sup>&</sup>lt;sup>4</sup> FASB currently has an open project titled *Accounting for Goodwill Impairment*. The objective of the project is to reduce the cost and complexity of the subsequent accounting for goodwill by simplifying the impairment test. Readers are encouraged to visit the "Technical Agenda" page at www.fasb.org for the latest developments regarding the project and how it may impact the guidance in this chapter.

#### FHLB or FRB Stock

# © Update 12-3 Accounting and Reporting: Recognition and Measurement of Financial Assets and Financial Liabilities

FASB ASU No. 2016-01, issued in January 2016, is effective for fiscal years, and interim periods within those fiscal years, of a public business entity beginning after December 15, 2017.

For all other entities (including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting), FASB ASU No. 2016-01 is effective for fiscal years beginning after December 15, 2018, and interim period within fiscal years beginning after December 15, 2019. Early application is permitted for all other entities as of the fiscal years beginning after December 15, 2017, including interim periods within those fiscal years.

All entities may adopt the presentation guidance in paragraphs 5–7 of FASB ASC 825-10-45 for financial statements of fiscal years or interim periods that have not yet been issued or that have not yet been made available for issuance. Furthermore, entities that are not public business entities may elect not to disclose the information about fair value of financial instruments required by the "General" subsection of FASB ASC 825-10-50 in financial statements of fiscal years or interim periods that have not yet been made available for issuance. Except as indicated previously in this paragraph, early application is not permitted.

FASB ASU No. 2016-01 addresses certain aspects of recognition, measurement, presentation, and disclosure of financial instruments and affects all entities that hold financial assets or owe financial liabilities. Among other provisions of the guidance, the amendments in this ASU supersede the guidance to classify equity securities with readily determinable fair values into different categories (that is, trading or available-for-sale) and require equity securities (including other ownership interests, such as partnerships, unincorporated joint ventures, and limited liability companies) to be measured at fair value with changes in the fair value recognized through net income (other than equity securities accounted for under the equity method of accounting or those that result in consolidation of an investee).

The amendments in this ASU also require an entity to present separately in other comprehensive income the portion of the total change in fair value of a liability resulting from a change in the instrument-specific credit risk when the entity has elected to measure the liability at fair value in accordance with the fair value option for financial instruments.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-01, see appendix G of this guide.

12.27 On June 23, 2004, the Federal Housing Finance Board voted to require the 12 FHLBs to enhance their financial disclosures by registering

©2016, AICPA AAG-DEP 12.27

with the SEC. Each bank is required to register a class of its equity securities under Section 12(g) of the Securities and Exchange Act of 1934. The banks file quarterly, annual, and supplemental disclosures.

- 12.28 Although FHLB (or FRB) stock is an equity interest in a FHLB (or FRB), it does not have a readily determinable fair value for purposes of FASB ASC 320 because its ownership is restricted and it lacks a market, as stated in FASB ASC 942-325-05-2. FHLB (or FRB) stock can be sold back only at its par value of \$100 per share and only to the FHLBs (or FRBs) or to another member institution. In addition, the equity ownership rights represented by FHLB stock are more limited than would be the case for a public company, because of the oversight role exercised by regulators in the process of budgeting and approving dividends.
- 12.29 FHLB and FRB stock should be classified as a restricted investment security, carried at cost, and evaluated for impairment, as stated in FASB ASC 942-325-25-1 and 942-325-35-1. FASB ASC 942-325-35-2 states that both cash and stock dividends received on FHLB stock should be reported as income. See FASB ASC 505, *Equity*, and FASB ASC 852, *Reorganizations*, for guidance concerning stock splits.
- 12.30 FASB ASC 942-325-35-3 states that FHLB stock is generally viewed as a long term investment. Accordingly, when evaluating FHLB stock for impairment, its value should be determined based on the ultimate recoverability of the par value rather than by recognizing temporary declines in value. The determination of whether the decline affects the ultimate recoverability is influenced by criteria such as the following:
  - The significance of the decline in net assets of the FHLBs as compared to the capital stock amount for the FHLBs and the length of time this situation has persisted.
  - Commitments by the FHLBs to make payments required by law or regulation and the level of such payments in relation to the operating performance of the FHLBs.
  - The impact of legislative and regulatory changes on the institutions and, accordingly, on the customer base of the FHLBs.
  - The liquidity position of the FHLBs.
- 12.31 The evaluation of whether a decline is temporary or whether it affects the ultimate recoverability of the FHLB stock is ultimately made by the member institution based on the facts at the time. This consideration is influenced by (a) the materiality of the carrying amount to the member institution and (b) whether an assessment of the institution's operational needs for the foreseeable future would indicate management intends to dispose of the stock and cause management to dispose of the stock at an amount other than par value.
- **12.32** When addressing other than temporary impairment, readers may also refer to FASB ASC 320-10-35.
- 12.33 Classification of FHLB or FRB stock in the balance sheet varies among financial institutions. Some institutions, with more significant investments in FHLB or FRB stock, present their investment as a separate line item in the balance sheet. Others may combine the investment in FHLB or FRB

stock with other investments or other assets. Either manner of presentation is acceptable.

**12.34** However, investments in FHLB or FRB stock should not be shown with securities accounted for under FASB ASC 320, according to FASB ASC 942-325-45-1.

## Goodwill and Other Intangible Assets

12.35 Goodwill. 5 FASB ASC 350-20 addresses financial accounting and reporting for goodwill subsequent to its acquisition and for the cost of internally developing goodwill. In accordance with FASB ASC 350-20-35-1, goodwill should not be amortized. Instead, goodwill should be tested for impairment at a level of reporting referred to as a reporting unit. (Paragraphs 33-46 of FASB ASC 350-20-35 provide guidance on determining reporting units.) FASB ASC 350-20-35-2 states that impairment is the condition that exists when the carrying amount of goodwill exceeds its implied fair value. The first step of the goodwill impairment test compares the estimated fair value of the identified reporting units with their carrying amount, including goodwill. If the estimated fair value of the reporting unit is less than the carrying value, the second step must be performed to determine the implied fair value of the reporting unit's goodwill and the amount of goodwill impairment, if any. The implied fair value of goodwill is determined as if the reporting unit were being acquired in a business combination. FASB ASC 350-20-35-3 states that an entity may first assess qualitative factors,6 as described in paragraphs 3A-3G of FASB ASC 350-20-35 (see paragraphs 12.36-.38), to determine whether it is necessary to perform the two-step goodwill impairment test discussed in paragraphs 4-19 of FASB ASC 350-20-35. The entity should consider its specific facts and circumstances when determining whether to utilize the qualitative assessment. If determined to be necessary, the two-step impairment test should be used to identify potential goodwill impairment and measure the amount of a goodwill impairment loss to be recognized (if any).

# Considerations for Private Companies That Elect to Use Standards as Issued by the Private Company Council

FASB ASU No. 2014-02, Intangibles—Goodwill and Other (Topic 350): Accounting for Goodwill (a consensus of the Private Company Council), permits a private company to subsequently amortize goodwill on a straight-line basis over a period of 10 years, or less if the company demonstrates that another useful life is more appropriate. It also permits a private company to apply a simplified impairment model to goodwill.

**12.36** In performing the qualitative assessment to evaluate whether it is more likely than not (that is, a likelihood of more than 50 percent) that the fair value of a reporting unit is less than its carrying amount, an entity

©2016, AICPA

<sup>&</sup>lt;sup>5</sup> See footnote 4.

<sup>&</sup>lt;sup>6</sup> In accordance with FASB ASC 350-20-35B, an entity has an unconditional option to bypass the qualitative assessment for any reporting unit in any period and proceed directly to performing the first step of the goodwill impairment test. An entity may resume performing the qualitative assessment in any subsequent period.

should assess relevant events and circumstances. Examples of such events and circumstances include the following:<sup>7,8</sup>

- Macroeconomic conditions such as a deterioration in general economic conditions, limitations on accessing capital, fluctuations in exchange rates, or other developments in equity and credit markets
- b. Industry and market considerations such as deterioration in the environment in which an entity operates, an increased competitive environment, a decline in market-dependent multiples or metrics (consider in both absolute terms and relative to peers), a change in the market for an entity's products or services, or a regulatory or political development
- c. Cost factors such as increases in raw materials, labor, or other costs that have a negative effect on earnings and cash flows
- d. Overall financial performance such as negative or declining cash flows or a decline in actual or planned revenue or earnings compared with actual and projected results of relevant prior periods
- e. Other relevant entity-specific events such as changes in management, key personnel, strategy, or customers; contemplation of bankruptcy; or litigation
- f. Events affecting a reporting unit such as a change in the composition or carrying amount of its net assets, a more-than-likely-than-not expectation of selling or disposing of all, or a portion, of a reporting unit, the testing for recoverability of a significant asset group within a reporting unit, or recognition of a goodwill impairment loss in the financial statements of a subsidiary that is a component of a reporting unit
- g. If applicable, a sustained decrease in share price (consider in both absolute terms and relative to peers)

12.37 An entity should consider the extent to which each of the adverse events and circumstances identified could affect the comparison of a reporting unit's fair value with its carrying amount. An entity should place more weight on the events and circumstances that most affect a reporting unit's fair value or the carrying amount of its net assets. An entity also should consider positive and mitigating events and circumstances that may affect its determination of whether it is more likely than not that the fair value of a reporting unit is less than its carrying amount. If an entity has a recent fair value calculation for a reporting unit, it also should include as a factor in its consideration the difference between the fair value and the carrying amount in reaching its conclusion about whether to perform the first step of the goodwill impairment test. An entity should evaluate, on the basis of the weight of evidence, the significance of all identified events and circumstances in the context of determining whether it is more likely than not that the fair value of a reporting unit is less than its carrying amount. Also, the existence of positive and mitigating events and

<sup>&</sup>lt;sup>7</sup> The examples included are not all-inclusive, and an entity should consider other relevant events and circumstances that affect the fair value or carrying amount of a reporting unit in determining whether to perform the first step of the goodwill impairment test. None of these individual examples of events and circumstances are intended to represent standalone events or circumstances that necessarily require an entity to perform the first step of the goodwill impairment test.

<sup>&</sup>lt;sup>8</sup> See footnote 4.

circumstances is not intended to represent a rebuttable presumption that an entity should not perform the first step of the goodwill impairment test.

- 12.38 If, after assessing the totality of events or circumstances, an entity determines that it is not more likely than not that the fair value of a reporting unit is less than its carrying amount, then the first and second steps of the goodwill impairment test are unnecessary. Conversely, if an entity determines that it is more likely than not that the fair value of a reporting unit is less than its carrying amount, then the entity should perform the first step of the two-step goodwill impairment test.
- **12.39** *Identified intangibles other than goodwill*. In accordance with FASB ASC 350-30-25-1, an intangible asset that is acquired either individually or with a group of other assets should be recognized. FASB ASC 350-30-25-2 states that the cost of a group of assets acquired in a transaction other than a business combination or an acquisition by a not-for-profit entity should be allocated to the individual assets acquired based on their relative fair values and should not give rise to goodwill. Paragraphs 12.06–.07 and chapter 19 of this guide discuss intangible assets acquired in connection with a business combination.
- **12.40** Costs of internally developing, maintaining, or restoring intangible assets that are not specifically identifiable, that have indeterminate lives, or that are inherent in a continuing business or nonprofit activity and related to an entity as a whole, should be recognized as an expense when incurred in accordance with FASB ASC 350-30-25-3.
- 12.41 In accordance with paragraphs 1-4 of FASB 350-30-35, the accounting for a recognized intangible asset is based on its useful life to the reporting entity. An intangible asset with a finite useful life should be amortized; an intangible asset with an indefinite useful life should not be amortized. The useful life of an intangible asset to an entity is the period over which the asset is expected to contribute directly or indirectly to the future cash flows of that entity. The useful life is not the period of time that it would take that entity to internally develop an intangible asset that would provide similar benefits. However, a reacquired right recognized as an intangible asset is amortized over the remaining contractual period of the contract in which the right was granted. If an entity subsequently reissues (sells) a reacquired right to a third party, the entity includes the related unamortized asset, if any, in determining the gain or loss on the reissuance. The estimate of the useful life of an intangible asset to an entity should be based on an analysis of all pertinent factors, including those described in FASB ASC 350-30-35-3. If no legal, regulatory, contractual, competitive, economic, or other factors limit the useful life of an intangible asset to the reporting entity, the useful life of the asset should be considered to be indefinite. The term indefinite does not mean the same as infinite or indeterminate. The useful life of an intangible asset is indefinite if that life extends beyond the foreseeable horizon—that is, there is no foreseeable limit on the period of time over which it is extended to contribute to the cash flows of the reporting entity.
- **12.42** As required by FASB ASC 350-30-35-6, if an intangible asset has a finite useful life, but the precise length of that life is not known, that intangible asset should be amortized over the best estimate of its useful life. The method of amortization should reflect the pattern in which the economic benefits of the intangible asset are consumed or otherwise used up. If that pattern cannot be reliably determined, a straight-line amortization method should be used.

Accordingly, FASB ASC 350-30-35-7 states that an intangible asset should not be written down or off in the period of acquisition unless it becomes impaired during that period.

- 12.43 The amount of an intangible asset to be amortized should be the amount initially assigned to that asset less any residual value, according to FASB ASC 350-30-35-8. The residual value of an intangible asset should be assumed to be zero unless at the end of its useful life to the entity the asset is expected to continue to have a useful life to another entity and either of the following conditions is met:
  - a. The reporting entity has a commitment from a third party to purchase the asset at the end of its useful life.
  - b. The residual value can be determined by reference to an exchange transaction in an existing market for that asset and that market is expected to exist at the end of the asset's useful life.
- 12.44 Paragraphs 9–10 and 12 of FASB ASC 350-30-35 state that an entity should evaluate the remaining useful life of an intangible asset that is being amortized each reporting period to determine whether events and circumstances warrant a revision to the remaining period of amortization. If the estimate of an intangible asset's remaining useful life is changed, the remaining carrying amount of the intangible asset should be amortized prospectively over that revised remaining useful life. If an intangible asset that is being amortized is subsequently determined to have an indefinite useful life, the asset should be tested for impairment in accordance with paragraphs 18–20 of FASB ASC 350-30-35. (FASB ASC 360-10-35-21 includes examples of impairment indicators.) That intangible asset should no longer be amortized and should be accounted for in the same manner as other intangible assets that are not subject to amortization.
- 12.45 FASB ASC 350-30-35-14 states that an intangible asset that is subject to amortization should be reviewed for impairment in accordance with the "Impairment or Disposal of Long-Lived Assets" subsections of FASB ASC 360-10 by applying the recognition and measurement provisions in paragraphs 17–35 of FASB ASC 360-10-35. In accordance with the "Impairment or Disposal of Long-Lived Assets" subsections of FASB ASC 360-10, an impairment loss should be recognized if the carrying amount of an intangible asset is not recoverable and its carrying amount exceeds its fair value. After an impairment loss is recognized, the adjusted carrying amount of the intangible asset should be its new accounting basis. Subsequent reversal of a previously recognized impairment loss is prohibited.
- 12.46 If an intangible asset is determined to have an indefinite useful life, it should not be amortized until its useful life is determined to be no longer indefinite, as required by paragraphs 15–16 of FASB ASC 350-30-35. An entity should evaluate the remaining useful life of an intangible asset that is not being amortized each reporting period to determine whether events and circumstances continue to support an indefinite useful life. If an intangible asset that is not being amortized is subsequently determined to have a finite useful life, FASB ASC 350-30-35-17 states that the asset should be tested for impairment in accordance with paragraphs 18–19 of FASB ASC 350-30-35. That intangible asset should then be amortized prospectively over its estimated remaining useful life and accounted for in the same manner as other intangible assets that are subject to amortization.

- 12.47 Paragraphs 18–19 of FASB ASC 350-30-35 state that an intangible asset that is not subject to amortization should be tested for impairment annually and more frequently if events or changes in circumstances indicate that it is more likely than not that the asset is impaired. An entity may first perform a qualitative assessment, as described in paragraphs 18A-18F of FASB ASC 350-30-35, to determine whether it is necessary to perform the quantitative impairment test. An entity has an unconditional option to bypass the qualitative assessment for any indefinite-lived intangible asset in any period and proceed directly to performing the quantitative impairment test. An entity may resume performing the qualitative assessment in any subsequent period. The quantitative impairment test for an indefinite-lived intangible asset should consist of a comparison of the fair value of the intangible asset with its carrying amount. If the carrying amount of an intangible asset exceeds its fair value, an entity should recognize an impairment loss in an amount equal to the excess. After an impairment loss is recognized, the adjusted carrying amount of the intangible asset should be its new accounting basis. In accordance with FASB ASC 350-30-35-20, subsequent reversal of a previously recognized impairment loss is prohibited.
- 12.48 Indemnification assets. Indemnification assets arise from contractual indemnities provided for in a business combination. According to paragraphs 27-28 of FASB ASC 805-20-25, the acquirer should recognize an indemnification asset at the same time that it recognizes the indemnified item, measured on the same basis as the indemnified item, subject to the need for a valuation allowance for uncollectible amounts. However, the indemnity may relate to an asset or liability that is an exception to the recognition or measurement principal, such as a contingency that is not recognized at the acquisition date because it does not meet the criteria for recognition in paragraphs 18A-19 of FASB ASC 805-20-25 at that date. In such a circumstance, the indemnification asset should be recognized and measured using assumptions consistent with those used to measure the indemnified item, subject to management's assessment of the collectability of the indemnification asset and any contractual limitations on the indemnified amount. At each subsequent reporting date, FASB ASC 805-20-35-4 requires the acquirer to measure an indemnification asset that was recognized in accordance with paragraphs 27-28 of FASB ASC 805-20-25 at the acquisition date on the same basis as the indemnified liability or asset. Further, the measure is subject to any contractual limitations on its amount, except as noted in FASB ASC 805-20-35-4B, and, for an indemnification asset that is not subsequently measured at its fair value, management's assessment of the collectability of the indemnification asset.
- 12.49 Financial statement presentation. Paragraphs 1–2 of FASB ASC 350-30-45 state that at a minimum, all intangible assets should be aggregated and presented as a separate line item in the statement of financial position. However, that requirement does not preclude presentation of individual intangible assets or classes of intangible assets as separate line items. The amortization expense and impairment losses for intangible assets should be presented in income statement line items within continuing operations as deemed appropriate for each entity.
- 12.50 The aggregate amount of goodwill should be presented as a separate line item in the statement of financial position, as provided in paragraphs 1–3

<sup>9</sup> See footnote 4.

of FASB ASC 350-20-45. The aggregate amount of goodwill impairment losses should be presented as a separate line item in the income statement before the subtotal income from continuing operations (or similar caption) unless a goodwill impairment loss is associated with a discontinued operation. A goodwill impairment loss associated with a discontinued operation should be included (on a net-of-tax basis) within the results of discontinued operations. FASB ASC 350-30-50 and 350-20-50 provide additional disclosure requirements for intangible assets and goodwill, respectively.

## **Exit or Disposal Activities**

- **12.51** Exit activities include, but are not limited to, the closure of activities in a particular location, the relocation of activities from one location to another, changes in management structure, sale or termination of a line of business, or a fundamental reorganization that affects the nature and focus of operations.
- 12.52 FASB ASC 420, Exit or Disposal Cost Obligations, discusses recognition of liabilities for the costs associated with exit or disposal activities, including, but not limited to, involuntarily employee termination benefits pursuant to a one-time termination benefit arrangement, costs to terminate a contract that is not a capital lease, and other associated costs, including costs to consolidate or close facilities and relocate employees, according to paragraphs 2–3 of FASB ASC 420-10-05.

# **Asset Retirement Obligations**

- **12.53** FASB ASC 410-20 establishes accounting standards for recognition and measurement of a liability for an asset retirement obligation and the associated asset retirement costs, as stated in FASB ASC 410-20-05-1.
- **12.54** An example of an asset retirement obligation might include an obligation to restore space leased for bank operations to its original condition, including the cost of removing vaults, teller lines and drive-through facilities.
- 12.55 The guidance in FASB ASC 410-20 applies to legal obligations associated with the retirement of a tangible long-lived asset that result from the acquisition, construction, development and (or) the normal operation of a long-lived asset, including any legal obligations that require disposal of a replaced part that is a component of a tangible long-lived asset, according to item (a) in FASB ASC 410-20-15-2. FASB ASC 410-20-25-4 requires that the fair value of a liability for an asset retirement obligation be recognized in the period in which it is incurred if a reasonable estimate of fair value can be made.
- 12.56 The FASB ASC glossary clarifies the term conditional asset retirement obligation, as used in FASB ASC 410-20, as a legal obligation to perform an asset retirement activity in which the timing and (or) method of settlement are conditional on a future event that may or may not be within the control of the entity. In addition, as defined in the FASB ASC glossary, the asset retirement cost is the amount capitalized that increases the carrying amount of the long-lived asset when a liability for an asset retirement obligation is recognized.

# **Customers' Liabilities on Acceptances**

12.57 FASB ASC 942-310-35-8 establishes that provisions for uncollectible amounts for customers' acceptance liabilities should be made, if necessary. Customers' liabilities on acceptances should be reported gross, rather

than net of the related bankers' acceptance liability in accordance with FASB ASC 942-310-45-1.

## **Mortgage Servicing Advances**

**12.58** Mortgage servicers may make payments on a borrower's behalf when escrow amounts, or other similar deposits, are not sufficient. For example, a servicer may pay a borrower's property tax to avoid a tax lien. Such amounts may need to be evaluated as to their significance and collectability by management.

## Impairment of Long-Lived Assets

**12.59** The "Impairment or Disposal of Long-Lived Assets" subsections of FASB ASC 360-10 establish the financial accounting and reporting for the impairment of long-lived assets to be held and used or to be disposed of, including (a) capital leases of lessees, and (b) long-lived assets of lessors subject to operating leases, according to item (a) in FASB ASC 360-10-15-4. See chapter 11 of this guide for the requirements of FASB ASC 360-10.

## **NCUSIF Deposit**

**12.60** Federally insured credit unions are required, under 12 CFR 741.4c, to maintain on deposit with the NCUSIF during each reporting period an amount equal to 1 percent of the credit union's total insured shares at the close of the preceding reporting period. The amount on deposit is adjusted periodically for changes in the amount of a credit union's insured shares. For example, if the insured shares decline, a pro rata portion of the amount on deposit with the NCUSIF is refunded to the credit union. A credit union, according to 12 CFR 741.4d, is also required to pay to the NCUSIF, on dates the NCUA board determines, but not more than twice in any calendar year, an insurance premium in an amount stated as a percentage of insured shares, which will be the same for all insured credit unions. The NCUA board may assess a premium charge only if the NCUSIF's equity ratio is less than 1.3 percent and the premium charge does not exceed the amount necessary to restore the equity ratio to 1.3 percent. If the equity ratio of NCUSIF falls below 1.2 percent, the NCUA board is required to assess a premium in an amount it determines is necessary to restore the equity ratio to, and maintain that ratio at, 1.2 percent.

**12.61** FASB ASC 942-325-25-3 states that amounts deposited with the NCUSIF should be accounted for and reported as assets as long as such amounts are fully refundable.

12.62 The refundability of NCUSIF deposits should be reviewed for impairment, as stated in item (a) in FASB ASC 942-325-35-4. When the refundability of a deposit is evaluated, the financial condition of both the credit union and of the NCUSIF should be considered. Deposits may be returned to solvent credit unions for a number of reasons, including the termination of insurance coverage, conversion to insurance coverage from another source, or the transfer of operations of the insurance fund from the NCUA board. However, insolvent or bankrupt credit unions should not be entitled to a return of their deposits. To the extent that NCUSIF deposits are not refundable, they should be charged to expense in the period in which the deposits are made or the assets become impaired.

©2016, AICPA AAG-DEP 12.62

12.63 Item (b) in FASB ASC 942-325-35-4 states that in years in which the equity of the NCUSIF exceeds normal operating levels, the NCUA is required to make distributions to insured credit unions to reduce the equity of the NCUSIF to normal operating levels. Such distributions may be in the form of a waiver of insurance premiums, premium rebates, or cash payments. Distributions in connection with that reduction in the equity of the NCUSIF should be reported in the income statement in the period in which it is determined that a distribution will be made.

12.64 Item (c) in FASB ASC 942-325-35-4 also states that the system of savings account insurance established by the recapitalization of the NCUSIF, which provided for reserves of 1 percent of insured deposits, is based on the concept that the necessary deposits create a fund with an earning potential sufficient to provide for the risk of losses in the credit union system. In years during which the earnings of the fund have been adequate to provide insurance protection and cover all expenses and losses incurred by the fund, the NCUA has elected to waive the insurance premiums due from insured credit unions. In those years, it has been industry practice to net imputed earnings on the insurance deposits against imputed premium expense rather than present them as gross amounts on the statement of income. In years during which the insurance premiums are not waived by the NCUA, the premiums should be expensed in the period to which they relate. To the extent that the NCUA assesses premiums to cover prior operating losses of the insurance fund or to increase the fund balance to normal operating levels, credit unions should expense those premiums when assessed.

12.65 In a letter to federally insured credit unions (NCUA Letter to Credit Unions No. 09-CU-02, "Corporate Credit Union System Strategy") issued on January 28, 2009, the NCUA announced certain actions it was taking to stabilize the corporate credit union system. The NCUA indicated that the expense of the actions would be passed on proportionately to all federally-insured credit unions through the partial write-off of such credit unions' existing deposits with the NCUSIF, as well as the assessment of an insurance premium sufficient to return the NCUSIF's equity to insured shares ratio to 1.30 percent.

**12.66** On May 20, 2009, the Helping Families Save Their Homes Act of 2009 became law. The legislation amended the Federal Credit Union Act providing several provisions favorable to credit unions including, but not limited to the following:

- The Temporary Corporate Credit Union Stabilization Fund was created to mitigate near term corporate credit union stabilization costs with NCUA authority to assess premiums over 7 years.
- The NCUSIF was provided authority to assess premiums over 8 years to rebuild the equity ratio, if needed.

See NCUA Letter to Credit Unions No. 09-CU-14, "Corporate Stabilization Fund Implementation," dated June 2009.

#### Other Investments

12.67 CUSOs. <sup>10</sup> Credit unions are allowed under the NCUA regulations to own and operate outside entities that conduct business related to the general services of the credit union. The types of businesses are restricted as to

<sup>&</sup>lt;sup>10</sup> See Title 12 U.S. Code of Federal Regulations Part 712.

operations within the regulations. These entities may conduct business with other credit unions, credit union members, and nonmembers. In addition, credit unions can own these entities with other credit unions or outside third parties.

12.68 Credit unions are restricted in the amount of money that can be invested in and loaned to the CUSO. Under current regulations, credit unions can invest up to 1 percent of the credit union's unimpaired capital and surplus. Credit unions can also lend up to an additional 1 percent. This loan authority is independent and separate from the 1 percent investment authority. The CUSO can be structured as a corporation, a limited liability company, or a limited partnership as long as the credit union is not the general partner. All CUSOs must follow U.S. generally accepted accounting principles (GAAP), have an annual opinion audit, and prepare quarterly financial statements. A wholly owned CUSO is not required to obtain a separate annual financial statement audit if it is included in the annual consolidated financial statement audit of the credit union that is its parent. The recording of the investment in or loan to the CUSO must also be accounted for in accordance with GAAP.

 ${f 12.69}$  The following are the general approved categories of CUSOs within the regulations:

- Checking and currency services
- Clerical, professional and management services
- Business loan origination
- Consumer mortgage loan origination
- Electronic transaction services
- Financial counseling services
- Fixed-asset services
- Insurance brokerage services
- Leasing
- Loan support services
- Record retention, security, and disaster recovery services
- Securities brokerage services
- Shared branching
- Student loan origination
- Travel agency
- Trust and trust-related services
- Real estate brokerage services
- Non-CUSO service providers
- Credit card loan origination
- Payroll processing services

**12.70** Under current IRS regulations, federally chartered credit unions do not have to pay taxes on income from flow-through entities. However, state chartered credit unions may be liable for taxes on flow-through income. All CU-SOs must follow all relevant IRS and state reporting and filing requirements.

12.71 In accordance with FASB ASC 323, Investments—Equity Method and Joint Ventures, the equity method of accounting should be used if the institution has the ability to exercise significant influence over the operating and financial policies of the investee or CUSO. The equity method tends to

be most appropriate if an investment enables the investor to influence the operating or financial decisions of the investee, as stated in FASB ASC 323-10-05-5. The investor then has a degree of responsibility for the return on its investment, and it is appropriate to include in the results of operations of the investor its share of the earnings or losses of the investee. Influence tends to be more effective as the investor's percent of ownership in the voting stock of the investee increases. Investments of relatively small percentages of voting stock of an investee tend to be passive in nature and enable the investor to have little or no influence on the operations of the investee.

12.72 A majority-owned subsidiary is an entity separate from the institution and may be a VIE that is subject to consolidation in accordance with the "Variable Interest Entities" subsections of FASB ASC 810-10, according to FASB ASC 810-10-15-9. The "Variable Interest Entities" subsections of FASB ASC 810-10 clarify the application of the "General" subsections to certain legal entities in which equity investors do not have sufficient equity at risk for the legal entity to finance its activities without additional subordinated financial support or, according to "Pending Content" in FASB ASC 810-10-05-8, as a group, the holders of the equity investment at risk lack any one of the following three characteristics:

- a. The power, through voting rights or similar rights, to direct the activities of a legal entity that most significantly impact the entity's economic performance
- b. The obligation to absorb the expected losses of the legal entity
- c. The right to receive the expected residual returns of the legal entity. 11

#### **Contributed Assets**

12.73 Many credit unions receive substantial contributions (for example, use of facilities and utilities, telephone services, data processing, mail services, payroll processing services, pension administration services and pension plan contributions, and other materials and supplies) from their sponsoring organizations. Many credit unions also rely on volunteers to provide various services to their members; some credit unions are staffed exclusively by volunteers. In addition, credit unions occasionally receive contributions of assets of material value.

**12.74** Contributions received, as provided by FASB ASC 958-605-25-2, should be recognized as revenue or gains in the period received and as assets, decreases of liabilities, or expenses depending on the form of the benefits received. FASB ASC 958-605-25-8 clarifies that, pursuant to FASB ASC 958-605-25-2, an unconditional promise to give should be recognized when it is received. However, to be recognized there must be sufficient evidence in the form of verifiable documentation that a promise was made and received. Contributions of services should be recognized only if the services received (a) create or enhance nonfinancial assets or (b) require specialized skills, are provided by individuals possessing those skills, and would typically need to be purchased if not provided by donation, as explained in FASB ASC 958-605-25-16.

12.75 Contributions received should be measured at their fair values, according to FASB ASC 958-605-30-2. FASB ASC 820, Fair Value Measurement,

<sup>&</sup>lt;sup>11</sup> See footnote 2.

defines fair value, establishes a framework for measuring fair value, and expands disclosures about fair value measurements. See chapter 20, "Fair Value," of this guide for a summary of FASB ASC 820.

12.76 A reporting entity should measure the fair value of an asset using the assumptions that market participants would use in pricing the asset, as stated in FASB ASC 820-10-35-9. In accordance with FASB ASC 958-605-30-6, unconditional promises to give that are expected to be collected in less than one year may be measured at net realizable value because that amount results in a reasonable estimate of fair value. Contributions of services that create or enhance nonfinancial assets may be measured by referring to either the fair value of the services received or the fair value of the asset or of the asset enhancement resulting from the services, as stated in FASB ASC 958-605-30-10. A major uncertainty about the existence of value may indicate that an item received or given should not be recognized, according to FASB ASC 958-605-25-4. The future economic benefit or service potential of a tangible item usually can be obtained by exchanging it for cash or by using it to produce goods or services, as stated in FASB ASC 958-605-25-5. FASB ASC 958-605-50-1 sets forth certain disclosure requirements for receipts of contributed services.

## Auditing 12

#### **Objectives**

**12.77** *Other assets.* The primary objectives of audit procedures applied to other assets are to obtain sufficient appropriate evidence that

- *a.* the assets exist and are owned by the institution;
- b. the assets are properly classified, valued, described, and disclosed in the financial statements;
- c. intangible assets that should be amortized are being amortized on a consistent basis over the estimated period of benefit;
- adequate provisions have been made for impairment, if any, of the assets;
- e. sales of assets, including the recognition of gains and losses, have been properly recognized; and
- f. appropriate disclosures, including the existence of liens, have been made.

12.78 Other liabilities. The primary objectives of auditing other liabilities are to obtain reasonable assurance that

- the liabilities represent authorized obligations of the institution;
   and
- all contingencies and estimated current period expenses that will be paid in future periods that should be accrued during the period have been accrued, classified, and described in accordance with GAAP, and the related disclosures are adequate.

©2016, AICPA

<sup>12</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards (GAAS) issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of issuers and nonissuers as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

#### **Planning**

12.79 In accordance with AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (as described in chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide). Presented in the following list are examples of factors related to other assets and other liabilities that may indicate higher risks of material misstatement for these areas:

- A current interest rate environment that may adversely affect the values of intangible assets that derive their value from (a) loan relationships or (b) from the timing and amount of future cash flows
- The use of unobservable inputs and higher degree of estimation uncertainty in the goodwill or intangible impairment test
- Loss of depositor relationships
- Prepayments of loans resulting in loss of servicing revenue and relationships
- Operating losses and uncertainty about future taxable income
- Declining real estate values
- Unreconciled balances in suspense accounts
- Litigation or regulatory enforcement actions or both
- Planned branch dispositions
- Operating losses at FHLBs or NCUSIF

# Internal Control Over Financial Reporting and Possible Tests of Controls

12.80 AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13–.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

12.81 AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards), addresses the auditor's responsibility to design and implement responses to the risks of material misstatement identified and assessed by the

auditor in accordance with AU-C section 315 and to evaluate the audit evidence obtained in an audit of financial statements.

- 12.82 In accordance with paragraph .08 of AU-C section 330, the auditor should design and perform tests of controls to obtain sufficient appropriate audit evidence about the operating effectiveness of relevant controls if (a) the auditor's assessment of risks of material misstatement at the relevant assertion level includes an expectation that the controls are operating effectively or (b) substantive procedures alone do not provide sufficient appropriate audit evidence at the relevant assertion level. Typical financial reporting controls may include the following:
  - Written policies and procedures that, among other things, specify depreciation and amortization methods and periods for property and equipment
  - Proper authorizations for specific transactions, such as approval for property and equipment purchases and sales
  - Periodic reviews of balances, fair values, realizable values, and policies by persons specified in management's written policy
  - Written policies and procedures aimed at assessing possible impairment of assets on a periodic basis

#### Substantive Tests

- 12.83 Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure. In accordance with paragraph .445 of AU-C section 330, this requirement reflects the fact that (a) the auditor's assessment of risk is judgmental and may not identify all risks of material misstatement and (b) inherent limitations to internal control exist, including management override.
- **12.84** *FHLB or FRB stock.* If the institution is a member of the FHLB or Federal Reserve System, the auditor should consider (a) confirming stock ownership with the related FHLB or FRB and (b) reconciling the dollar amount of the shares with the institution's general ledger. For institutions holding FHLB stock, the auditor should consider the status of the FHLB's redemption of its stock at par value before concluding that income recognition is appropriate for any FHLB stock dividends.
- **12.85** *Premises and equipment.* Substantive procedures used to test premises and equipment consist primarily of physical inspection, review of documents of title or other documents supporting the acquisition, tests of disposals and other adjustments, and reasonableness tests of depreciation. Similar procedures are often used to test the classification, recording, and disclosure of leased premises and equipment.
- 12.86 Auditors should be alert for signs that premises and equipment are no longer in use and consider tests to determine whether there are any undisclosed liens on premises and equipment. If those signs are present, the auditor might test whether there is impairment of the premises and equipment that should be recognized. Also, computer hardware and software are particularly vulnerable to obsolescence and their valuation should be reviewed if not in use.

12.87 Intangible assets and goodwill. AU-C section 540, Auditing Accounting Estimates, Including Fair Value Accounting Estimates, and Related Disclosures (AICPA, Professional Standards), addresses the auditor's responsibilities relating to accounting estimates, including fair value accounting estimates and related disclosures, in an audit of financial statements. Specifically, it expands on how AU-C section 315, AU-C section 330, and other relevant AU-C sections are to be applied with regard to accounting estimates. It also includes requirements and guidance related to misstatements of individual accounting estimates and indicators of possible management bias. In accordance with paragraph .06 of AU-C section 540, the objective of the auditor is to obtain sufficient appropriate audit evidence about whether, in the context of the applicable financial reporting framework, (a) accounting estimates, including fair value accounting estimates, in the financial statements, whether recognized or disclosed, are reasonable and (b) related disclosures in the financial statements are adequate.

12.88 Paragraphs .12–.13 of AU-C section 540 states that based on the assessed risks of material misstatement, the auditor should determine (a) whether management has appropriately applied the requirements of the applicable financial reporting framework relevant to the accounting estimate and (b) whether the methods for making the accounting estimates are appropriate and have been applied consistently and whether changes from the prior period, if any, in accounting estimates or the method for making them are appropriate in the circumstances. In responding to the assessed risks of material misstatement, as required by AU-C section 330, the auditor should undertake one or more of the following, taking into account the nature of the accounting estimate:

- a. Determine whether events occurring up to the date of the auditor's report provide audit evidence regarding the accounting estimate.
- b. Test how management made the accounting estimate and the data on which it is based. In doing so, the auditor should evaluate whether
  - i. the method of measurement used is appropriate in the circumstances.
  - ii. the assumptions used by management are reasonable in light of the measurement objectives of the applicable financial reporting framework, and
  - iii. the data on which the estimate is based is sufficiently reliable for the auditor's purposes.
- c. Test the operating effectiveness of the controls over how management made the accounting estimate, together with appropriate substantive procedures.
- d. Develop a point estimate or range to evaluate management's point estimate.  $^{13}$  For this purpose,
  - i. if the auditor uses assumptions or methods that differ from management's, the auditor should obtain an understanding of management's assumptions or methods sufficient

<sup>&</sup>lt;sup>13</sup> For purposes of GAAS, *management's point estimate* is defined as the amount selected by management for recognition or disclosure in the financial statements as an accounting estimate.

- to establish that the auditor's point estimate or range<sup>14</sup> takes into account relevant variables and to evaluate any significant differences from management's point estimate.
- ii. if the auditor concludes that it is appropriate to use a range, the auditor should narrow the range, based on audit evidence available, until all outcomes within the range are considered reasonable.

In relation to intangible assets and goodwill, the auditor might consider such key factors and assumptions as the ability of the assets to generate income in the future, the expected lives of loans, expected losses on assets subject to contractual indemnities, or expected withdrawal rates of deposits. The auditor should consider whether the assumptions continue to be reasonable and evaluate the effect of changes in assumptions on the recoverability of the assets. Additionally, paragraph .14 of AU-C section 540 states that the auditor should consider whether specialized skills or knowledge with regard to one or more aspects of the accounting estimates is required in order to obtain sufficient appropriate audit evidence.

Considerations for Audits Performed in Accordance With PCAOB Standards

The second note in paragraph .A5, of AS 2201, An Audit of Internal Control Over Financial Reporting That Is Integrated with An Audit of Financial Statements (AICPA, PCAOB Standards and Related Rules), discusses the inherent limitations of internal control. Because fair value determinations often involve subjective judgments by management, this may affect the nature of controls that are capable of being implemented, including the possibility of management override of controls. The auditor considers the inherent limitations of internal control in such circumstances in assessing control risk.

Paragraph .47 of AS 2502, Auditing Fair Value Measurements and Disclosures (AICPA, PCAOB Standards and Related Rules), states that the auditor should evaluate the sufficiency and competence of the audit evidence obtained from auditing fair value measurements and disclosures as well as the consistency of that evidence with other audit evidence obtained and evaluated during the audit. The auditor's evaluation of whether the fair value measurements and disclosures in the financial statements are in conformity with GAAP is performed in the context of the financial statements taken as a whole (see paragraphs .12–.18 and .24–.27 of AS 2810, Evaluating Audit Results [AICPA, PCAOB Standards and Related Rules]).

**12.89** *Customers' liabilities on acceptances.* Substantive tests that are performed on loans, such as confirmation and collectibility reviews, are generally used to test customers' liabilities on acceptances. (See chapters 8 and 9, "Credit Losses," of this guide.)

**12.90** *Other liabilities.* Substantive audit procedures relating to interest payable, accrued expenses, and other liability amounts that the auditor might perform include the following:

©2016, AICPA

<sup>&</sup>lt;sup>14</sup> For purposes of GAAS, an auditor's point estimate or auditor's range is defined as the amount or range of amounts, respectively, derived from audit evidence for use in evaluating the recorded or disclosed amount(s).

- Tracing recorded amounts to supporting documentation
- Agreeing rates used in the calculation of recorded amounts of interest payable to board of directors' authorization
- Testing individual calculations of accrued interest (dividends)
- Tracing recorded amounts to subsequent cash disbursements
- Examining evidence supporting the carrying amount of other liabilities, including such items as an actuarial evaluation used to compute accrued pension costs, payroll tax returns, and invoices received from third parties
- Confirming recorded amounts
- Performing a search for unrecorded liabilities
- Circulating attorney letters
- Performing analytical procedures
- Evaluating key assumptions, such as discount rates

**12.91** Suspense accounts. The auditor should consider reviewing the suspense account for material items remaining in the account at year-end for reclassification entries to the appropriate account. The auditor should also consider reviewing for subsequent entries made to clear suspense account items.

Deposits 409

## Chapter 13

# **Deposits**

# FASB Accounting Standards Update (ASU) No. 2014-09, Revenue from Contracts with Customers (Topic 606)

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14, Revenue from Contracts with Customers [Topic 606]—Deferral of the Effective Date) for annual reporting periods of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period.

Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period. For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as of either:

- An annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- An annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB Accounting Standards Codification (ASC) 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605, *Revenue Recognition*, as well as guidance within the 900 series of industry-specific topics, including FASB ASC 942, *Financial Services—Depository and Lending*. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts).

Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

The AICPA has formed 16 industry task forces to assist in developing a new Accounting Guide on revenue recognition that will provide helpful hints and illustrative examples for how to apply the new standard. Revenue recognition implementation issues identified by the Depository and Lending Institutions Revenue Recognition Task Force will be available for informal comment, after review by the AICPA Financial Reporting Executive Committee, at the "Depository and Lending Institutions Revenue Recognition Task Force" page at www.aicpa.org.

Readers are encouraged to submit comments to revreccomments@aicpa.org. For more information on FASB ASU No. 2014-09, see appendix E, "The New Revenue Recognition Standard: FASB ASU No. 2014-09," of this guide.

©2016, AICPA AAG-DEP 13

#### Introduction

- 13.01 Deposits are an important source of funds for banks, credit unions, and savings institutions. Finance and mortgage companies do not take insured deposits. Because a credit union's members are also its owners, credit unions often refer to deposits as *share accounts* and related interest paid as *dividends*. Some credit unions permit nonmembers to deposit funds subject to certain restrictions.
- **13.02** Deposits are often an institution's most significant liability and interest expense on deposits an institution's most significant expense.
- 13.03 Deposits are generally classified by whether they bear interest, by their ownership (for example, public, private, interbank, or foreign), and by their type (for example, demand, time, and savings; or transaction and nontransaction). A description of various deposit products follows. These descriptions may not correspond to regulatory designations under the Board of Governors of the Federal Reserve System (Federal Reserve) Regulation D.

## Demand Deposits 1

13.04 Demand deposits (often called *transaction accounts* or *DDAs*) are accounts that may bear interest and the depositor is entitled to withdraw at any time without prior notice. Checking and negotiable order of withdrawal accounts are the most common form of demand deposits. Withdrawals are typically made through check writing, automated teller machines (ATMs), debit cards at point-of-sale (POS) terminals, electronic funds transfers (EFTs), or preauthorized payment transactions. Deposits are generally made through direct deposit (such as payroll amounts) or EFTs, or at ATMs or teller windows.

#### **Savings Deposits**

13.05 Savings deposits bear interest and have no stated maturity. Savings deposits include passbook and statement savings accounts and money-market deposit accounts (MMDAs). Withdrawals and deposits are typically made at ATMs or teller windows, by EFTs, or by preauthorized payments. Furthermore, MMDAs generally permit the customer to write checks, although the number of checks that may be written is limited by law.

### Time Deposits

- **13.06** Time deposits, which include certificates of deposit (CDs), IRAs, and open accounts, bear interest for a fixed, stated period of time.
- **13.07** CDs bear a stipulated maturity and interest rate, payable either periodically or at maturity. CDs may be issued in bearer form (payable to the

<sup>&</sup>lt;sup>1</sup> Under Regulation D, demand deposit accounts (DDAs) and negotiable order of withdrawal (NOW) accounts are types of transaction accounts. Historically, DDAs could not be interest-bearing under federal regulation, whereas NOW accounts may bear interest.

Section 627 of the Dodd-Frank Wall Street REform and Consumer Protection Act (Dodd-Frank Act) permits insured depository institutions (IDIs) to pay interest on DDAs. In response to this provision of the Dodd-Frank Act, the FDIC released Financial Institution Letter FIL-38-2011, Deposit Insurance Notice Requirement Regarding the Payment of Interest on Demand Deposit Accounts, to remind IDI's that if on or after July 21, 2011, an IDI modifies the terms of a DDA so that the account may pay interest, the IDI must notify the affected customers that the account no longer will be eligible for unlimited deposit insurance coverage as a noninterest-bearing transaction account (issued May 2011).

holder) or registered form (payable only to a specified individual or entity) and may be negotiable or nonnegotiable (always issued in registered form). Negotiable CDs, for which there is an active secondary market, are generally short term and are most commonly sold to corporations, pension funds, and government bodies in large denominations (generally, \$100,000 to \$1 million). Nonnegotiable CDs, including savings certificates, are generally in smaller denominations. Depositors holding nonnegotiable CDs may recover their funds prior to the stated maturity, however, in doing so they normally pay a penalty.

- 13.08 Retirement accounts known as IRAs, Keogh accounts (also known as H.R. 10 plans), and self-employed-person accounts are generally maintained as CDs. However, because of the tax benefits for depositors, they typically have longer terms than most CDs. Many retirement accounts provide for automatic renewal on maturity.
- 13.09 Open accounts are time deposits with specific maturities and fixed interest rates but, unlike savings certificates, amounts may be added to them until maturity. Common types of open accounts are vacation and Christmas club accounts.
- 13.10 According to IRC regulations, employers that withhold federal taxes from employees' pay are required to deposit those funds periodically with a depository institution. Institutions record such deposits, which are noninterest-bearing, as treasury tax and loan accounts and include such accounts with their deposits. (See paragraph 15.18 of this guide.)

#### **Brokered Deposits**

- **13.11** Brokered deposits are time deposits that are third-party deposits placed by or through the assistance of a deposit broker. Deposit brokers sometimes sell interests in placed deposits to third parties. As discussed in subsequent paragraphs, federal law restricts the acceptance and renewal of brokered deposits by an institution based on its capitalization.
- **13.12** Brokered deposits are similar to other deposits in that they represent funds placed at the institution by a third party. However, they are fundamentally different in that the funds were acquired through the use of a broker, not through a customer relationship.
- 13.13 In recent years, a brokered deposit account type known as the Certificate of Deposit Account Registry Service (CDARS) program has become more prevalent. With CDARS, an investor who wishes to buy CDs in excess of the FDIC insurance limit is able to do so through one account. The sponsoring financial institution sells the amount in excess of the FDIC insurance limit to other institutions. The investor ultimately has one CDARS account through which their entire CD balance is FDIC insured. Financial institutions can participate in the process through buying deposits, selling deposits, or a reciprocating relationship.

#### **Dormant Accounts**

13.14 Institutions generally have a policy on classifying accounts as dormant. Before a savings account is classified as dormant, it must be inactive for a standard period of time, which normally exceeds that of checking accounts. After a much longer specific period of inactivity, as determined by the state in which the institution is located, state regulations may require that inactive deposits be turned over to (escheated to) the state.

#### Closed Accounts

**13.15** When an account is closed, the signature card or electronic file is generally removed from the file of active accounts and placed in a closed-account section.

#### Other Deposit Services

**13.16** Institutions often offer other deposit services such as reserve or overdraft protection programs<sup>2</sup> (which combine a checking account and a preauthorized personal loan), check guarantee services, and consolidated account statements (which combine the account information of several services into one monthly statement).

### The Payments Function and Services

- **13.17** The payments function of a depository institution involves facilitating money payments and transferring funds. The payments function is accomplished through checks and EFTs.
- 13.18 Check processing. The check-clearing process, which is highly automated, involves the exchange of checks and the settlement of balances among institutions locally, regionally, and nationally. Check processing traditionally involved the encoding of checks with magnetic ink character recognition (MICR) symbols to facilitate routing, the proof and transit function, and the flow of checks for collection. A correspondent system and the Federal Reserve perform such clearinghouse functions for depository institutions. More recently, check processing occurs through the digital capture of check images and the electronic transfer and settlement between accounts.
- 13.19 An institution receives two types of checks: (a) on-us checks, drawn on a depositor's account and (b) foreign checks, drawn on accounts of other institutions. Such checks may be received from the Federal Reserve, local clearinghouses, other depository institutions, at an ATM or teller window, through the mail, or by other means, such as a loan payment.
- 13.20 Many physical checks that an institution receives have been dollar-amount encoded by the first institution that handles the check. However, checks received through an institution's own operations must go through its proof department or its correspondent bank. A proof department has the responsibility to
  - a. prove the individual transaction against its documentation, such as a deposit slip;
  - b. verify totals for several departments;
  - c. encode the dollar-amount field;
  - d. mechanically endorse the back of the check; and
  - e. sort the items according to destination.
- 13.21 The flow of physical checks for collection depends primarily on the location of the institution on which the check is drawn. Processing an on-us

<sup>&</sup>lt;sup>2</sup> Interagency guidance issued by the Office of the Comptroller of the Currency, the Board of Governors of the Federal Reserve System, the FDIC, and the National Credit Union Administration, addresses the risks and accounting treatment associated with overdraft protection programs.

check for deposit to another account in the same institution is straightforward: The institution debits the check writer's account and credits the check depositor's account. Processing a check drawn on another depository institution, however, can be complex.

- **13.22** Though some direct collections are made in the banking system, most institutions collect foreign checks through a clearing arrangement (clearinghouse), a correspondent bank, or the Federal Reserve.
- 13.23 In a clearing arrangement, a group of depository institutions in a given area that receive large numbers of deposited checks drawn on one another meets to exchange and collect payment for the checks. Checks are physically exchanged among participants, and collection is made by crediting or debiting the net amount presented by each institution against all the others.
- 13.24 When a correspondent institution receives a check drawn on one of its respondent institutions, the check collection process can take several different routes. If the presented check is drawn on an institution that also maintains an account with the correspondent, collection simply involves the correspondent's transfer of deposit credit from one account to another account. If the check is drawn on an institution that does not have an account relationship with the correspondent, the check is credited to the respondent institution's account and then either (a) sent to a second correspondent in which the first correspondent and the institution on which the check is drawn both have an account, (b) sent to a local clearinghouse, or (c) sent to a Federal Reserve bank.
- 13.25 The Federal Reserve collects checks by internally transferring credit balances from one account to another, in much the same way that individual institutions collect on-us checks. For presenting and paying institutions that have accounts at two different Federal Reserve banks, an extra step is involved in the collection process. Each Federal Reserve bank has an interdistrict settlement account that it maintains on the books of the Interdistrict Settlement Fund established in Washington, D.C., to handle settlements. A check presented to one Federal Reserve bank drawn on a depository institution in another Federal Reserve district will result in a transfer of interdistrict settlement account balances from one Federal Reserve bank to another.
- 13.26 *EFT systems*. Institutions have responded to the large volume of checks and the high costs of clearing checks by increasingly using EFT systems. EFT systems are computer-based networks designed to move funds to and from accounts and to and from other institutions electronically, thus eliminating paper-based transactions. Banks and savings institutions transact an enormous volume of daily business between themselves and for customers over regional and national EFT systems. The three principal kinds of EFT systems are direct deposit systems, automated clearinghouse (ACH) systems, and ATMs.
- 13.27 A direct deposit system involves the direct deposit of payments into a customer's account without the use of a definitive check and is widely used for payrolls. The payment information is usually transmitted to the institution from the payer in electronic form and processed through the institution's proof system.
- 13.28 An ACH is used to transfer funds from one institution's account at a Federal Reserve bank to that of another; conduct transactions in the federal

funds market; transfer funds for customers; transfer book entries representing certain securities; and receive, send, and control other specific EFT messages between member banks and other clearinghouses. The largest ACH is Fedwire, operated by the Federal Reserve. The Clearing House Interbank Payments System (CHIPS) is an ACH operated by the New York Clearing House Association and is the focal point for payments in the world's international dollars market. International dollar payments generally do not leave the United States but are held as deposits at money-center and regional banks or the U.S. branches of foreign banks and are transferred between accounts through CHIPS in payment for internationally traded goods and services, international financial transactions, or settlement of debt.

- **13.29** Institutions also provide a variety of retail EFT services, including ATMs, POS terminals, telephone bill payment, and home computer banking.
- 13.30 The Check Clearing for the 21st Century Act (Check 21). Check 21 requires financial institutions to recognize paper checks constructed from digital images as negotiable instruments (Subpart D of Regulation CC, Expedited Funds Availability). These negotiable instruments, known as *substitute checks*, contain certain information to be considered a legal equivalent of an original check. To be a legal equivalent, substitute checks must
  - a. be suitable for automated processing;
  - b. bear a MICR line containing all the information appearing on the original check;
  - c. meet the technical requirements for substitute checks;
  - d. bear a legend that states, "This is a legal copy of your check. You can use it the same way you would use the original check;" and
  - e. be able to be processed in the same manner as the original check with current check processing equipment.
- **13.31** Check 21 includes necessary warranties, indemnities and disclosures. If a financial institution transfers, presents, or returns a substitute check for consideration (payment), it warrants that
  - a. the substitute check has met all the requirements to have legal equivalence to the original check, and
  - b. no party will be asked to pay a check that already has been paid.

Under these warranties, the financial institution will indemnify any person who suffered a loss due to the receipt of a substitute check instead of the original check. If the financial institution transferred a substitute check to a consumer who experienced a loss, it may be responsible for recrediting the consumer. Banks normally have procedures in place for processing recredit amounts for consumer payment. Additionally, upon the implementation of Check 21, certain consumer disclosures explaining Check 21 are required to be sent to consumers.

13.32 With Check 21, financial institutions can convert paper checks into electronic images and deliver Image Replacement Documents in place of the original for payment. Additionally, two banks or a network of multiple banks through mutual agreement can now exchange data taken from the MICR of the original check or an electronic image of the original check, and drastically reduce turnaround time (float).

Deposits 415

## **Regulatory Matters**

## Limitations on Brokered Deposits<sup>3</sup>

13.33 Restrictions on the acceptance of brokered deposits, particularly for institutions that become undercapitalized, could affect an institution's liquidity. The effect of such restrictions on liquidity may be a condition, when considered with other factors that could indicate substantial doubt about an entity's ability to continue as a going concern. (See chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide.)

13.34 Banks and savings institutions. Section 29 of the Federal Deposit Insurance Act (FDI Act) (codified in Title 12 U.S. Code of Federal Regulations [CFR] Part 337) significantly limits the acceptance or use of brokered deposits, funds which the reporting bank obtains directly or indirectly by or through any deposit broker for deposit into one or more accounts, by depository institutions other than those that are well capitalized (as defined for purposes of prompt corrective regulatory action, as discussed in chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide). Adequately capitalized institutions may accept brokered deposits only if they first obtain a waiver from the FDIC. Undercapitalized institutions are prohibited from accepting brokered deposits.

13.35 Also, the federal banking agencies may restrict the use of brokered deposits of an institution about which the agencies have concerns, even though the institution may continue to be deemed well capitalized. Such restriction may be evidenced by a memorandum of understanding, board resolution, cease-and-desist order or other regulatory order.

13.36 Section 29 of the FDI Act also limits the interest rates that may be offered by under- or adequately capitalized institutions. Undercapitalized institutions may not solicit any deposits by offering rates significantly higher (as defined) than prevailing rates in the institution's market area or the prevailing rate in the market area from which the deposit is accepted. Adequately capitalized institutions are prohibited from paying interest on brokered deposits above certain levels. The applicable federal banking agency may also restrict the rate paid by well capitalized institutions.

13.37 Effective January 1, 2010, institutions subject to these interest rate restrictions are required to use the national rate to determine conformance with the restrictions. The *national rate* is defined as a simple average of rates paid by insured depository institutions and branches for which data are available. Institutions subject to the restrictions operating in an area where the rates paid on deposits are higher than the "national rate" can use the local market to determine the prevailing rate if they seek and receive approval from the FDIC. (See FDIC Financial Institution Letter [FIL]-69-2009, *Process for Determining If An Institution Subject to Interest-Rate Restrictions is Operating in a High-Rate Area*, dated December 4, 2009.)

©2016, AICPA AAG-DEP 13.37

<sup>&</sup>lt;sup>3</sup> In January 2015, the FDIC issued *Guidance on Identifying, Accepting, and Reporting Brokered Deposits—Frequently Asked Questions* to address frequently asked questions on such topics as identifying brokered deposits, accepting deposits, listing services, interest rate restrictions, and other brokered deposit-related matters. In November 2015, the FDIC updated the document in response to further inquiries and comments. For further information, readers can access the guidance on the FDIC website at www.fdic.gov.

**13.38** *Credit unions.* Section 107(6) of the Federal Credit Union Act (codified in 12 CFR 701.32(b)) limits the acceptance and use of nonmember and public unit deposits (as defined), including brokered deposits, to the greater of (a) 20 percent of the total deposits of the federal credit union or (b) \$1.5 million.

#### Classification of Deposits of Credit Unions

13.39 The Credit Union Membership Access Act (codified in 12 CFR 715) requires all federally insured credit unions with assets of \$10 million and over to follow U.S. generally accepted accounting principles (GAAP). Accordingly, all federally insured credit unions with over \$10 million in assets are required to file their Call Report on a GAAP basis. However, the Call Report is not structured for GAAP presentation and disclosure and shows deposits in a separate category, not in equity or liabilities. To the National Credit Union Administration (NCUA), the Call Report deals specifically with recognition and measurement for GAAP rather than presentation and disclosure.

**13.40** As discussed further in paragraph 13.64, supervisory committees of federal credit unions are required to perform a verification of member's accounts.

## Accounting and Financial Reporting<sup>4</sup>

- 13.41 FASB ASC 942-405-25-3 states that the institution's liability for deposits originates and should be recognized at the time deposits are received rather than when the institution collects the funds. Checks that are deposited by customers and that are in the process of collection and are currently not available for withdrawal (deposit float) should be recorded as assets and liabilities. Deposits should not be recorded based solely on collections.
- **13.42** Deposit accounts that are overdrawn should be reclassified as loans and should therefore be evaluated for collectibility as part of the evaluation of credit loss allowances.
- 13.43 Credit unions and corporate credit unions should report unequivocally all member deposit accounts, including member shares, as liabilities in the statement of financial condition, as stated in FASB ASC 942-405-25-4. Paragraphs 3-4 of FASB ASC 942-405-45 address presentation guidance related to member deposits and respective interest payable on those deposit accounts. The statement of financial condition should either (a) present deposit accounts as the first item in the liabilities and equity section or (b) include deposit accounts within a captioned subtotal for total liabilities. An unclassified presentation whereby all liabilities and equity are shown together under one subheading and savings accounts are presented as the last item before retained earnings should not be an acceptable presentation. The interest paid or accrued on these accounts, commonly referred to as dividends, should be

<sup>&</sup>lt;sup>4</sup> In June 2010, the AICPA issued Technical Questions and Answers (Q&A) section 2130.38, "Certificates of Deposit and FASB ASC 820, Fair Value Measurements and Disclosures"; Q&A section 2130.39, "Balance Sheet Classification of Certificates of Deposit"; and Q&A section 2130.40, "Certificates of Deposit and FASB ASC 320" (AICPA, Technical Questions and Answers). Q&A section 2130.38 addresses whether certificates of deposit (CDs) are within the scope of the disclosure requirements of FASB Accounting Standards Codification (ASC) 820, Fair Value Measurement. Q&A section 2130.39 addresses where in a classified balance sheet an entity should report a CD. Q&A section 2130.40 addresses whether a CD is within the scope of FASB ASC 320, Investments—Debt and Equity Securities.

reported as an interest expense on the statement of income, and the amount of interest payable to members should be included as a liability in the statement of financial condition.

- **13.44** FASB ASC 942-405-50-1 states that disclosures about deposit liabilities should include all of the following:
  - a. The aggregate amount of time deposit accounts (including CDs) in denominations that meet or exceed the FDIC insurance limit at the balance sheet date
  - b. Securities, mortgage loans, or other financial instruments that serve as collateral for deposits, that are otherwise not disclosed under FASB ASC 860, *Transfers and Servicing*
  - c. The aggregate amount of any demand deposits that have been reclassified as loan balances, such as overdrafts, at the balance sheet date
  - d. Deposits that are received on terms other than those available in the normal course of business
- **13.45** Some additional disclosures about deposits should generally include the following:
  - a. For time deposits having a remaining term of more than one year, the aggregate amount of maturities for each of the five years following the balance sheet date (in accordance with FASB ASC 470-10-50-1)
  - b. The amount of deposits of related parties at the balance sheet date (in accordance with FASB ASC 850, *Related Party Disclosures*)
- 13.46 In accordance with FASB ASC 825, Financial Instruments, the fair value of deposits should also be disclosed. In estimating the fair value of deposit liabilities, FASB ASC 942-470-50-1 states that a financial entity should not take into account the value of its long term relationships with depositors, commonly known as core deposit intangibles, which are separate intangible assets, not financial instruments. For deposits liabilities with no defined maturities, the fair value to be disclosed under FASB ASC 825-10-50-13 and FASB ASC 825-10-60-1 is the amount payable on demand at the reporting date.

# © Update 13-1 Accounting and Reporting: Recognition and Measurement of Financial Assets and Financial Liabilities

FASB ASU No. 2016-01, Financial Instruments—Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities, issued in January 2016, is effective for fiscal years, and interim periods within those fiscal years, of a public business entity beginning after December 15, 2017.

For all other entities (including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting), FASB ASU No. 2016-01 is effective for fiscal years beginning after December 15, 2018, and interim periods within fiscal years beginning after December 15, 2019. Early application is permitted for all other entities as of the fiscal years beginning after December 15, 2017, including interim periods within those fiscal years.

All entities may adopt the presentation guidance in paragraphs 5–7 of FASB ASC 825-10-45 for financial statements of fiscal years or interim periods that have not yet been issued or that have not yet been made available for issuance. Furthermore, entities that are not public business entities may elect not to disclose the information about fair value of financial instruments required by the "General" subsection of FASB ASC 825-10-50 in financial statements of fiscal years or interim periods that have not yet been made available for issuance. Except as indicated previously in this paragraph, early application is not permitted.

FASB ASU No. 2016-01 addresses certain aspects of recognition, measurement, presentation, and disclosure of financial instruments and affects all entities that hold financial assets or owe financial liabilities. Among other provisions of the guidance, the conclusions reached in paragraph BC137 of FASB ASU No. 2016-01 explain that the difficulties of determining the fair value of demand deposit liabilities supported not requiring disclosure of fair value, either parenthetically or in the notes, and as such, upon the effective date of this guidance, demand deposit liabilities are exempt from an entity's requirement to disclose the fair value of financial liabilities.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-01, see appendix G, "Accounting for Financial Instruments," of this guide.

## Auditing<sup>5</sup>

### **Objectives**

**13.47** The primary objectives of auditing procedures for deposit liabilities are to obtain sufficient appropriate evidence that

- a. financial statement amounts for deposit liabilities and related transactions include all deposit obligations of the institution and reflect all related transactions for the period; and
- b. deposit liabilities and related income statement and balance sheet accounts have been properly valued, classified, and disclosed in conformity with GAAP.

### **Planning**

13.48 In accordance with AU-C section 315, *Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement* (AICPA, *Professional Standards*), the objective of the auditor is to identify and assess

<sup>&</sup>lt;sup>5</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards (GAAS) issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of issuers and nonissuers as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

Deposits 419

the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (as described in chapter 5 of this guide). The following factors related to deposits may contribute to higher inherent risk:

- a. Large number of accounts and volumes of transactions
- b. Transactions executed through a variety of means and locations
- c. Accounts typically can be opened in numerous locations or through the Internet
- d. Recurring and significant difficulties in reconciling exception items
- e. A practice of permitting depositors to withdraw funds from their accounts before deposited checks have been collected by the institution
- f. Introduction of new deposit products
- g. The existence and activity of previously inactive or dormant accounts
- h. Significant increases in the number of closed accounts, especially near the end of a reporting period
- Numerous accounts having instructions not to mail account statements to the depositor (no-mail accounts)

13.49 As stated in paragraphs .A36 and .A44 of AU-C section 200, Overall Objectives of the Independent Auditor and the Conduct of an Audit in Accordance With Generally Accepted Auditing Standards (AICPA, Professional Standards), audit risk is a function of the risks of material misstatement (consisting of inherent risk and control risk) and detection risk. (A detailed discussion on audit risk can be found in chapter 5 of this guide.) The assessment of risks is a matter of professional judgment, rather than a matter capable of precise measurement. The assessment of the risks of material misstatement may be expressed in quantitative terms, such as in percentages, or in nonquantitative terms. In any case, the need for the auditor to make appropriate risk assessments is more important than the different approaches by which they may be made. For example, if deposit liabilities have a higher inherent risk, considering factors listed in paragraph 13.48 or other factors, and if there is a basis for concluding that control risk is low, the auditor may conclude that the overall risk of material misstatement is low.

# Internal Control Over Financial Reporting and Possible Tests of Controls

13.50 AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13—.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement

at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

- 13.51 AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards), addresses the auditor's responsibility to design and implement responses to the risks of material misstatement identified and assessed by the auditor in accordance with AU-C section 315 and to evaluate the audit evidence obtained in an audit of financial statements.
- **13.52** Effective internal control (as it relates to financial reporting of deposits) provides reasonable assurance that (a) deposits are accepted in accordance with management's established policies, (b) misstatements caused by error or fraud in the processing of accounting information for deposits are prevented or detected, and (c) deposits are monitored on an ongoing basis to determine whether recorded financial statement amounts necessitate adjustment.
- **13.53** According paragraph .15 of AU-C section 315, the auditor should obtain an understanding of the control environment. As part of obtaining this understanding, the auditor should evaluate whether (a) management, with the oversight of those charged with governance, has created and maintained a culture of honesty and ethical behavior and (b) the strengths in the control environment elements collectively provide an appropriate foundation for the other components of internal control and whether those other components are not undermined by deficiencies in the control environment. Control activities that may contribute to a strong control environment may include
  - policies and procedures approved by the board of directors and that include position limits for each type of deposit (including brokered deposits) and guidelines for setting the interest rates offered on deposits.
  - segregation of duties between persons involved with the proof function, persons having access to cash, persons responsible for opening new accounts and issuing CDs or savings certificates, persons with responsibility for authorizing account adjustments, and persons with responsibility for posting information to the general ledger. (Because many of the potential duty conflicts found in the deposit area also exist for cash, it is usually efficient to coordinate any assessment of segregation of duties in those two areas.)
  - reconciliation of subsidiary ledgers for deposit principal, accrued interest, and related accounts to the general ledger on a periodic basis.
  - daily performance of a proof and transit operation with rejected or exception items segregated and individually reviewed. (Examples of such items include activity in dormant accounts or customer overdrafts.)
  - designation by management of persons such as officers or supervisory employees, to be responsible for reviewing and approving unposted holdover items, overdrafts, return items, and status of inactive or dormant accounts.
  - files, ledger cards, canceled checks, deposit tickets, signature cards, and unissued CDs and savings certificates safeguarded

- from unauthorized access (including dual control over and prenumbering of unissued certificates and official checks).
- periodic depositor account statements mailed regularly. (Returned statements are controlled, with follow-up on a timely basis.)

Deposits

- supervisory personnel designated by management to be responsible for periodically reviewing activity in employee accounts for unusual transactions.
- EFTs subject to control procedures that
  - segregate duties between employees who handle cash, balance EFT transactions, authorize EFTs, and post EFTs to deposit accounts.
  - require authorization for EFTs exceeding a depositor's available balance.
  - establish and maintain current, written agreements with all depositors making EFT requests, particularly for those customers who initiate EFT requests by telephone, cellular phone, Internet, POS terminal and other means not involving signed authorization. These agreements generally should set forth the scope of the institution's liability and the agreed-upon security procedures for authenticating transactions (such as callbacks or passwords).
  - provide for the review of rejected transactions and the correction and reversal of entries by a supervisor.
  - restrict initiation of EFTs and access to computer terminals or other EFT equipment.
  - require that documentation of EFTs is provided to the parties involved on a timely basis.
  - disclose the name of the debit party to the receiver of funds.
  - provide for written instructions to employees and users concerning the EFT function.
  - provide for the use and confidentiality of authorized caller and other access codes or authentication algorithms, including periodic changes in such codes or algorithms.
  - provide for the maintenance of a current list of personnel authorized to initiate EFTs.
  - establish authorization limits for personnel.
  - provide for holds to be placed on customer accounts by EFT personnel when instructions are received directly from the authorized customer to confirm that available funds are in the customer's account or that the EFT funds are within authorized limits before the EFT is made.
  - provide for the maintenance of card files or authorization letters on file for all customers who initiate EFTs.

 controls and verification procedures over requests for EFTs in place at respondent depository institutions.

13.54 In accordance with paragraph .08 of AU-C section 330, the auditor should design and perform tests of controls to obtain sufficient appropriate audit evidence about the operating effectiveness of relevant controls if (a) the auditor's assessment of risks of material misstatement at the relevant assertion level includes an expectation that the controls are operating effectively or (b) substantive procedures alone do not provide sufficient appropriate audit evidence at the relevant assertion level, such as typically would be the case with the audit of deposits. Examples of tests of controls might include

- observing or otherwise obtaining evidence about segregation of duties and supervisory review of activity in employee accounts;
- testing the reconciliations of related accounts, including the disposition of reconciling items and review and approval by a person other than the preparer;
- testing controls over origination of and access to signature cards and mailing address files;
- testing controls over the direct mailing of statements to depositors;
- comparing withdrawal slips with the applicable signature cards;
   and
- testing controls over restrictions on deposits pledged as collateral, inactive or dormant accounts, and mail receipts.

#### 13.55 Tests of control activities related to EFTs might include

- testing compliance with management's established authorization and verification procedures;
- validating sequence numbers on transfers sent and received;
- confirming that acknowledgments are returned for all outgoing messages;
- reviewing management's daily comparison of the total number and dollar amount of EFTs sent and received with summaries received from the Federal Reserve:
- testing the reconciliations of daily reserve or clearing account statements for disposition of reconciling differences and supervisory review and approval;
- testing the procedures for identification and verification of EFTs with respondent institutions; and
- observing control activities that address access.

#### Substantive Tests

**13.56** Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure, which for a financial institution would include deposits. In accordance with paragraph .445 of AU-C section 330, this requirement reflects the facts that (a) the auditor's assessment of risk is judgmental and may not identify all risks of material misstatement and (b) inherent limitations to internal control exist, including management override.

- 13.57 Audit procedures for deposits might include testing the reconciliations of related subsidiary and general ledger accounts, confirmation of account balances, and analytical procedures.
- 13.58 Subsidiary records and reconciliations. In accordance with paragraph .21 of AU-C section 330, the auditor's substantive procedures should include audit procedures relating to the financial statement closing process, such as agreeing or reconciling financial statement balances, including deposit balances and related accounts (accrued interest payable), with the underlying accounting records (trial balance, general ledger, and other subsidiary records). The disposition of reconciling items between general and subsidiary ledgers (such as returned items, adjustment items, holdovers, overdrafts, and service charges) might be investigated to determine whether any adjustments to recorded amounts are necessary.
- **13.59** Confirmations. Paragraph .A8 of AU-C section 500, Audit Evidence (AICPA, Professional Standards), states that corroborating information obtained from a source independent of the entity may increase the assurance that the auditor obtains from audit evidence that is generated internally, such as evidence existing within the accounting records, minutes of meetings, or a management representation. AU-C section 505, External Confirmations (AICPA, Professional Standards), addresses the auditor's use of external confirmation procedures to obtain audit evidence, in accordance with the requirements of AU-C sections 330 and 500.
- 13.60 Confirmation of deposits provides evidence about existence and valuation (accuracy), but it may not provide evidence about other assertions. Paragraph .15 of AU-C section 505 states that negative confirmations provide less persuasive audit evidence than positive confirmations. Accordingly, the auditor should not use negative confirmation requests as the sole substantive audit procedure to address an assessed risk of material misstatement at the assertion level, unless all of the following are present:
  - a. The auditor has assessed the risk of material misstatement as low and has obtained sufficient appropriate audit evidence regarding the operating effectiveness of controls relevant to the assertion.
  - b. The population of items subject to negative confirmation procedures comprises a large number of small, homogeneous account balances, transactions, or conditions.
  - c. A very low exception rate is expected.
  - d. The auditor is not aware of circumstances or conditions that would cause recipients of negative confirmation requests to disregard such requests.
- **13.61** According to paragraph .A5 of AU-C section 505, factors to consider when designing confirmation requests include the following:
  - The assertions being addressed.
  - Specific identified risks of material misstatement, including fraud risk.
  - The layout and presentation of the confirmation request.
  - Prior experience on the audit or similar engagements.
  - The method of communication (for example, in paper form or by electronic or other medium).

- Management's authorization or encouragement to the confirming parties to respond to the auditor. Confirming parties may only be willing to respond to a confirmation containing management's authorization.
- 13.62 Refer to AU-C section 505 for additional guidance on the confirmation process. Refer to AU-C sections 530, Audit Sampling, and 320, Materiality in Planning and Performing an Audit (AICPA, Professional Standards), for guidance on the extent of audit procedures (that is, considerations involved in determining the number of items to confirm). Readers may also refer to the AICPA Audit Guide Audit Sampling that provides guidance to help auditors apply audit sampling in accordance with AU-C section 530. Guidance on the timing of audit procedures is included in AU-C section 330.
- 13.63 In the case of each nonresponse, <sup>6</sup> paragraph .12 of AU-C section 505 states that the auditor should perform alternative procedures to obtain relevant and reliable audit evidence. (See paragraphs .A24–.A27 of AU-C section 505 for additional guidance regarding alternative procedures.) Some depositors may have instructed the institution not to send account statements to the depositor's mailing address. For such no-mail accounts, the auditor might review a written request from the depositor requesting the no-mail status and could use alternative procedures. Alternative procedures for positive confirmation nonreplies and no-mail accounts might include obtaining the respective period deposit account statement and subsequent statements to review for unusual activity and investigating any significant or unusual activity noted.
- **13.64** *Credit union supervisory committee procedures.* Note that one of the explicit duties of a federally insured credit union's supervisory committee is to periodically perform a verification of the members' accounts. In addition to procedures required under generally accepted auditing standards (GAAS), auditors may be asked to extend their procedures to assist in meeting the supervisory committee's requirements. Part 715.8(b) of the NCUA's regulations require that the verification be made using any of the following methods:
  - A controlled verification of 100 percent of members' share and loan accounts.
  - A statistical sampling method that provides for random selection that is expected to be representative of the population from which the sample was selected with an equal chance of selecting each item in the population, which will allow the auditor to test sufficient accounts in both number and scope on which to base conclusions concerning management's financial reporting objectives. The auditor should also perform alternative procedures if evidence provided by confirmations alone is not sufficient. (According to paragraph .08 of AU-C section 530, the auditor should select items for the sample in such a way that the auditor can reasonable expect the sample to be representative of the relevant population and likely to provide the auditor with a reasonable basis for conclusion about the population.)
  - For independent, licensed, CPAs, the additional option of sampling members' accounts using nonstatistical sampling methods

<sup>&</sup>lt;sup>6</sup> For purposes of GAAS, *nonresponse* is defined as a failure of the confirming party to respond, or fully respond, to a positive confirmation request or a confirmation request returned undelivered.

Deposits 425

consistent with applicable GAAS may be chosen as long as the sampling method provides a selection that allows the auditor to test sufficient accounts in both number and scope to provide assurance that the general ledger accounts are fairly stated in relation to the financial statements taken as a whole. (Independent, licensed, CPAs will be responsible for documenting their sampling procedures, and providing evidence to the NCUA, if requested, that the method used is consistent with applicable GAAS.)

**13.65** Check 21. The following are some potential audit concerns that may arise related to Check 21:

- The auditor will see fewer original checks as they are replaced by substitute checks. Audit planning may be adjusted based on the financial institution's processes for disseminating and returning checks.
- Detecting check fraud may become more difficult as substitute checks will no longer contain watermarks, fingerprints, ink or original paper with access to pressure points. Because the original check is no longer used for processing, the security of the electronic systems will reduce human access to the financial information and reduce employee fraud or error. Shorter processing time will allow for a quicker identification of check fraud or forgery.
- The financial institution may have purchased equipment that does not have the proper controls in place to prevent computer hacking.
- 13.66 Accrued interest payable, interest expense, and service charge income. Substantive audit procedures should be performed on accrued interest payable, interest expense, and service charge income in connection with other procedures on deposits. Audit procedures for such amounts might include reviewing and testing reconciliations of subsidiary ledgers with the general ledger, recalculating interest paid, accrued interest payable, and service charge income, and testing of interest expense and service charge income for the period.
- **13.67** Overdraft protection programs. Audit procedures on overdraft protection programs may include verifying that overdraft balances are properly classified along with the related write-offs of uncollectible balances.
- 13.68 Other analytical procedures. AU-C section 520, Analytical Procedures (AICPA, Professional Standards), addresses the auditor's use of analytical procedures as substantive procedures (substantive analytical procedures). Paragraph .05 of AU-C section 520 states that when designing and performing analytical procedures, either alone or in combination with tests of details, as substantive procedures in accordance with AU-C section 330,7 the auditor should
  - a. determine the suitability of particular substantive analytical procedures for given assertions, taking into account the assessed risks of material misstatement and tests of details, if any, for these assertions. (For example, analytical review procedures can provide

<sup>&</sup>lt;sup>7</sup> Paragraph .18 of AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards).

substantive evidence about the completeness of deposit-related financial statement amounts and disclosures; however, such procedures in tests of deposit expense are often less precise than substantive tests such as recalculations. Because institutions generally offer a wide variety of deposit products with rates that change frequently during a financial reporting period, it is normally difficult to develop expectations to be used in analyzing yields on deposits.)

- b. evaluate the reliability of data from which the auditor's expectation of recorded amounts or ratios is developed, taking into account the source, comparability, and nature and relevance of information available and controls over preparation.
- c. develop an expectation of recorded amounts or ratios and evaluate whether the expectation is sufficiently precise (taking into account whether substantive analytical procedures are to be performed alone or in combination with tests of details) to identify a misstatement that, individually or when aggregated with other misstatements, may cause the financial statements to be materially misstated.
- d. determine the amount of any difference of recorded amounts from expected values that is acceptable without further investigation as required by paragraph .07 of AU-C section 520 and compare the recorded amounts, or ratios developed from recorded amounts, with the expectations.

13.69 As discussed in item a in paragraph 13.68, it is normally difficult to develop sufficiently precise expectations to be used in analyzing yields on deposits. Accordingly, analytical procedures in this area might be considered only as a supplement to other substantive procedures, except where an expected yield can be known with some precision (using computer-assisted audit techniques). Analytical review procedures that the auditor may apply in the deposit area include analysis and evaluation of the following:

- Comparison of the percentage of deposit growth during the period with historical percentages
- Comparison of the average deposit account balances during the period with those of prior periods
- Changes in the relative composition of deposits from period to period
- Comparison of the amounts and percentage ratio of dormant accounts to total deposits with those of prior periods
- Comparison of deposit interest rates with those prevailing in the institution's marketing area for the same periods

However, be careful not to view trends entirely from a historical perspective; current environmental and business factors as well as local, regional, and national trends could be considered to determine if the institution's trend appears reasonable.

13.70 Paragraph .A7 of AU-C section 520 states that the auditor's substantive procedures to address the assessed risk of material misstatement for relevant assertions may be tests of details, substantive analytical procedures, or a combination of both. The decision about which audit procedures to

Deposits 427

perform, including whether to use substantive analytical procedures, is based on the auditor's professional judgment about the expected effectiveness and efficiency of the available audit procedures to reduce the assessed risk of material misstatement to an acceptably low level. Further guidance on the auditor's use of analytical procedures as substantive procedures is provided in AU-C section 520. AU-C section 315 addresses the use of analytical procedures as risk assessment procedures, and AU-C section 330 addresses the nature, timing, and extent of audit procedures in response to assessed risks; these audit procedures may include substantive analytical procedures. In addition, see analytical procedures discussion within paragraphs 5.94–.97 of this guide for further guidance on designing and performing substantive analytical procedures.

©2016. AICPA AAG-DEP 13.70

## Chapter 14

# Federal Funds and Repurchase Agreements

#### Introduction

14.01 This chapter addresses two types of transactions—federal funds and repurchase agreements (repos)—that can be either investing or financing transactions, depending on which side of the transaction the financial institution participates. Federal funds transactions can be an important tool for managing liquidity. Repos also can provide a cost-effective source of funds and may provide a means for the institution to leverage its securities portfolio for liquidity and funding needs.

#### Federal Funds Purchased

14.02 Federal funds are funds that commercial banks deposit at Federal Reserve Banks. Banks must meet legal reserve requirements on a daily basis by maintaining a specified total amount of deposits at Federal Reserve Banks and vault cash. A bank with excess reserves on a particular day may lend the excess, at an agreed-rate of interest (the federal funds rate), to another bank needing funds to meet its reserve requirements that day. The federal funds market does not increase or decrease total reserves in the Federal Reserve System, but merely redistributes them to facilitate efficient use of bank reserves and resources. However, by setting reserve requirements, the Federal Reserve System may increase or decrease total reserves in the system. No physical transfer of funds takes place; the Federal Reserve Bank charges the seller's reserve balance and credits the buyer's reserve balance. In addition to buying and selling funds to meet their own needs, banks with correspondent banking relationships absorb or provide funds as a service or accommodation to their correspondents. Accordingly, banks may operate on both sides of the federal funds market on the same day.

**14.03** Two types of transactions involving federal funds are commonly used. In an unsecured transaction, the selling bank sells federal funds on one day and is repaid with interest at maturity (usually the next day). In a collateralized transaction, other than by a repo, a bank purchasing federal funds places U.S. government securities in a custody account for the seller until the funds are repaid. A borrowing bank records a liability (federal funds *purchased*) and a selling bank records an asset (federal funds *sold*).

### **Repos**

**14.04** A repurchase agreement, as stated in "Pending Content" in the FASB Accounting Standards Codification (ASC) glossary, refers to an agreement under which the transferor (repo party) transfers a financial asset to a transferee (repo counterparty or reverse party) in exchange for cash and concurrently agrees to reacquire that financial asset at a future date for an amount equal to the cash exchanged plus a stipulated interest factor. Instead of cash, other securities or letters of credit sometimes are exchanged. Some repos call for repurchase of financial assets that need not be identical to the financial assets transferred. A repurchase agreement may be accounted for either as a

collateralized borrowing or a sale, depending on the terms of the contract. See discussion beginning in paragraph 14.27 for further details.

- **14.05** A repurchase agreement accounted for as a collateralized borrowing, as stated in the FASB ASC glossary, refers to a transaction in which a seller-borrower of securities sells those securities to a buyer-lender with an agreement to repurchase them at a stated price plus interest at a specified date or in specified circumstances. 1 A repo accounted for as a collateralized borrowing is a repo that does not qualify for sale accounting under FASB ASC 860, Transfers and Servicing. The payable under a repo accounted for as a collateralized borrowing refers to the amount of the seller-borrower's obligation recognized for the future repurchase of the securities from the buyer-lender. In certain industries, the terminology is reversed; that is, entities in those industries refer to this type of agreement as a reverse repo. A reverse repurchase agreement accounted for as a collateralized borrowing, per the FASB ASC glossary, refers to a transaction that is accounted for as a collateralized lending in which a buyer-lender buys securities with an agreement to resell them to the seller-borrower at a stated price plus interest at a specified date or in specified circumstances. The receivable under a reverse repo accounted for as a collateralized borrowing refers to the amount due from the seller-borrower for the repurchase of the securities from the buyer-lender. In certain industries, the terminology is reversed; that is, entities in those industries refer to this type of agreement as a repo.<sup>2</sup>
- **14.06** "Pending Content" in the FASB ASC glossary defines a *repurchase-to-maturity transaction* as a repurchase agreement in which the settlement date of the agreement to repurchase a transferred financial asset is at the maturity date of that financial asset and the agreement would not require the transferor to reacquire the financial asset.
- 14.07 Most repos involve obligations of the federal government or its agencies, but other financial instruments, such as commercial paper, banker's acceptances, negotiable certificates of deposit, and non-agency mortgage-backed securities (MBSs) are sometimes used in repos. Repos are similar to the seller-borrower's borrowing funds equal to the sales price of the related securities with the securities as collateral. The difference in the price at which the institution sells its securities and repurchases them represents interest for the use of the funds. Most repo transactions occur with other depository institutions, dealers in securities, state and local governments, insurance companies, investment funds, and customers (retail repo). Maturities of such agreements are flexible and generally vary from 1 day to 270 days.
- 14.08 *Dollar-roll repurchase agreements*, as defined in the FASB ASC glossary, are agreements to sell and repurchase similar but not identical securities. The securities sold and repurchased are usually of the same issuer. Dollar-rolls differ from regular repos in that the securities sold and repurchased have all of the following characteristics:

A seller-borrower represents a seller of securities and borrower of funds, whereas a buyer-lender represents a buyer of securities and a lender of funds.

<sup>&</sup>lt;sup>2</sup> Broker-dealers and depository institutions (and accordingly, this guide) refer to agreements by seller-borrowers to sell and repurchase securities as repurchase agreements (repos) and agreements by buyer-lenders to purchase and resell securities as reverse repurchase agreements (reverse repos). Savings institutions and credit unions have in the past used the opposite terms, calling a seller-borrower's agreement a reverse repo and a buyer-lender's agreement a repo.

- They are represented by different certificates.
- They are collateralized by different but similar mortgage pools (for example, conforming single-family residential mortgages).
- They generally have different principal amounts.

Fixed coupon and yield maintenance dollar-roll agreements comprise the most common agreement variations. In a fixed coupon agreement, the seller and buyer agree that delivery will be made with securities having the same stated interest rate as the interest rate stated on the securities sold. In a yield maintenance agreement, the parties agree that delivery will be made with securities that will provide the seller a yield that is specified in the agreement.

- **14.09** The dollar roll transactions are typically found in the To Be Announced section of the primary issuance markets for agency MBSs.
- **14.10** In a fixed coupon dollar-roll, the securities repurchased are generally priced to result in substantially the same yield. In addition, the seller-borrower retains control over the future economic benefits of the securities sold and assumes no additional market risk.
- 14.11 In a yield-maintenance agreement, the securities repurchased may have a different stated interest rate from that of the securities sold and are generally priced to result in different yields as specified in the agreement.<sup>3</sup> The seller-borrower surrenders control over the future economic benefits of the securities sold and assumes additional market risk. Yield-maintenance agreements may contain  $par\ cap^4$  provisions that could significantly alter the economics of the transactions.
- 14.12 The terms of the agreements often provide criteria to determine whether the securities are similar enough to make the transaction, in substance, a borrowing and lending of funds or whether the securities are so dissimilar that the transaction is a sale and separate forward purchase of securities. For agreements involving existing securities collateralized by dissimilar pools, those transactions are accounted for as sales and forward purchases of securities.
- 14.13 Rollovers and extensions. Occasionally, securities involved in repos are not delivered on the settlement date of the agreement and the contract may be rolled over or extended upon mutual agreement of the buyer-lender and seller-borrower. Overnight repos are often used as a source of short-term financing, and rolled over on a daily basis by agreement between the counterparties.
- 14.14 *Breakage.* Securities repurchased under repos commonly have a principal amount that differs from the principal amount of the security originally sold under the agreement. This is particularly common to dollar rolls, which involve MBSs. That difference is referred to as breakage and occurs because the principal amounts of MBSs generally differ as a result of the monthly amortization of principal balances of mortgages collateralizing the MBSs. The

©2016. AICPA

<sup>&</sup>lt;sup>3</sup> The price-spread relationship between securities with different contractual interest rates does not move in tandem. The existence of yield and price disparities provides opportunities for the buyer-lender to deliver, within the terms of the agreement, certificates providing the greatest benefit to the buyer-lender.

<sup>&</sup>lt;sup>4</sup> A *par cap* provision limits the repurchase price to a stipulated percentage of the face amount of the certificate. Fixed-coupon agreements do not contain par cap provisions.

amount of the breakage is a factor in determining whether substantially the same security is reacquired in the repo transaction, that is, whether good delivery (one in accordance with the agreement terms) has been met on repurchase of the MBSs.

- **14.15** Business risk. Business risks associated with repos arise from the contractual and economic complexities inherent in certain of these transactions and the degree to which the institution's management understands the terms of the agreements and the economics of the transactions. Misunderstandings may result in incorrect pricing of the agreements or an incorrect assessment of the risks that are being assumed, the return that is anticipated to be earned, or the financing costs that are being incurred. Misunderstandings of a contract's terms may also result in improper accounting treatment of the transaction (that is, as a sale and forward purchase or as a secured borrowing).
- **14.16** *Market risk.* The prices of government securities vary inversely with changes in interest rates. Price changes may be small, but they can result in significant changes in the market values of government securities due to the large dollar amounts often involved in government securities transactions. This is generally referred to as *market risk.* Price changes may affect the ability of the seller-borrower under repos to continue the financing without providing additional collateral. Changes in market price of nongovernment securities can also be caused by changes in credit spreads and liquidity in the market. Price changes affect the margin in a transaction and may create a need for the seller-borrower to transfer additional securities or return cash.
- **14.17** The excess of the fair value of the securities transferred by the seller-borrower over the amount of cash transferred by the buyer-lender is called a *haircut*. A haircut represents a margin of safety required by the buyer-lender to guard against a decline in the value of the collateral as a result of rising interest rates during the term of the agreement. Whether an agreement provides for a haircut depends on competition among buyer-lenders and seller-borrowers and their relative bargaining strengths. Haircuts generally range from a fraction of 1 percent to 4 percent or 5 percent but may be higher in certain instances.
- **14.18** All of the following factors are considered in determining the haircut for a particular transaction:
  - a. The term of the agreement
  - b. The creditworthiness of the institution
  - $\it c.$  The type of securities underlying the agreement, the length of time to maturity, and the creditworthiness of the issuer of the securities
  - d. The volatility of the market value of the underlying securities
  - e. The differential between the interest rate specified in the agreement and the interest rate on the securities
- **14.19** *Credit risk.* A repo or reverse repo can be considered a loan of cash by one party and a loan of securities by another. When the agreement is completed, both loans are repaid. Parties to repo and reverse repo transactions are subject to credit risk, that is, the risk that the transaction counterparty will not perform under the terms of the agreement. For example, a seller-borrower is at risk that changes in market prices and resulting economic losses may prevent the buyer-lender from returning the securities at the maturity of the agreement.

- 14.20 Credit risk also exists to the extent that the issuer of the underlying securities may default. However, such risk may be negligible for securities issued or guaranteed by the U.S. government or its agencies. If the issuer of the underlying securities defaults, both participants in the repo are obligated to complete the transaction. This aspect of credit risk is affected by the extent to which the institution's repo position is concentrated in any one type of underlying security or with any one counterparty.
- 14.21 The extent of credit risk faced by a seller-borrower also depends on the buyer-lender's business policies and practices for control and use of collateral (including re-hypothecations), the extent of the haircut on securities serving as collateral, the extent to which the buyer-lender offsets transactions (that is, maintains a matched book), and the buyer-lender's extent of capitalization.
- **14.22** Analyzing credit risk requires an understanding of how securities dealers and other counterparties to repos manage their businesses and of the steps that can be and are taken to reduce their exposure to market risk. Securities dealers are typically highly leveraged, with securities positions that represent large multiples of their net capital and that can quickly be eroded by adverse market changes. Many securities dealers entering into repos frequently employ matched-book transactions. In a matched-book transaction, the securities dealer effects both a repo and a reverse repo with the same underlying securities for the same period of time but usually at slightly different rates. By running a matched book, a dealer can reduce its exposure to market changes, and a seller-borrower may face less credit risk by entering into agreements with a dealer that has a matched book and employs adequate procedures to control credit risk. Even if the dealer runs a matched book, the seller-borrower still faces credit risk associated with the dealer's credit risk, that is, the risk that a customer of the dealer might not be able to complete its agreement with the dealer.
- 14.23 Risk of collateral loss. When an institution transfers the securities sold under an agreement to repurchase, there is a risk that the dealer may not be able to reverse the transaction by selling the securities back at the agreed price. If the institution overcollateralizes the agreement by selling the securities at a relatively large discount from the market price, its rights to the overage may be diminished or lost entirely in the event of the dealer's bankruptcy. In that case, the institution may find that neither the securities nor the funds to replace the securities are available for the dealer to complete the transaction and, as a result, may incur an economic loss. If the institution does not have the legal right of setoff, the potential economic loss extends to the full value of the securities, including accrued interest.
- 14.24 If the institution has the legal right to set off the securities against the borrowed funds, the potential economic loss is limited to the excess of the fair value of the securities, plus accrued interest, at the date of the sale over the amount borrowed, plus or minus any change in that fair value and accrued interest. However, the accounting loss may be greater or less than the economic loss if the book value of the securities is above or below their fair value. (See paragraph 14.39.)
- **14.25** Securities purchased under agreements to resell (reverse repos) pose risk to buyer-lenders to the extent that they do not take possession of the

securities they agreed to resell.<sup>5</sup> If the buyer-lender or securities dealer through whom the transaction is made does not perfect a security interest in securities purchased (by having signed an agreement and by taking possession, either directly or through a custodian acting as its agent), the potential economic loss extends to the full value of the securities and the risk assumed—namely, credit risk—becomes that of an unsecured lender. Institutions reduce such risk by

- a. making sure that definitive collateral is held by the counterparty's custodian as the counterparty's agent with specific identification of the assignee;
- b. settling through the Federal Reserve System, where book-entry collateral is transferred directly or by a notation entry;
- c. evaluating the creditworthiness of the other party to the agreement; and
- d. overcollateralizing the borrowing.

## Regulatory Matters

14.26 On December 18, 2008, the FDIC issued Financial Institution Letter-146-2008, *Recordkeeping Requirements for Qualified Financial Contracts*, which required institutions in troubled condition to produce position level and counterparty level data and other information that is relevant to the resolution and disposition of qualified financial contracts (QFCs). QFCs include repos among other contracts and agreements.

## **Accounting and Financial Reporting**

14.27 FASB ASC 860 establishes accounting and reporting standards for transfers and servicing of financial assets. As addressed in FASB ASC 860-10-05-6, transfers of financial assets take many forms, including but not limited to, repos. Paragraphs 19–21 of FASB ASC 860-10-05 provide an overview of repos. "Pending Content" in paragraphs 51–56B of FASB ASC 860-10-55 provide implementation guidance for accounting for repos.

**14.28** If the conditions in FASB ASC 860-10-40-5 are met, "Pending Content" in FASB ASC 860-10-55-55 states that the transferor should account for the repo as a sale of financial assets and a forward repurchase commitment, and the transferee should account for the agreement as a purchase of financial assets and a forward resale commitment.

**14.29** In accordance with "Pending Content" in FASB ASC 860-10-55-51 repos and securities lending transactions that do not meet all the criteria in FASB ASC 860-10-40-5 should be treated as secured borrowings.

14.30 "Pending Content" in FASB ASC 860-10-55-51A goes on to state that under certain agreements to repurchase transferred financial assets before their maturity, the transferor maintains effective control over the transferred financial assets. If effective control is maintained or the transaction

<sup>&</sup>lt;sup>5</sup> The Gramm-Leach-Bliley Act of 1999 (GLBA) repealed the blanket exemption for banks from the definition of a broker set forth in Section 3(a)(4) of the Securities Exchange Act of 1934 (the 1934 Act). In place of the blanket exemption, GLBA provided 11 specific exceptions for the definition of broker. The statutory exceptions were designed to allow banks to continue to effect securities transactions in connection with certain traditional bank activities. Regulation R, "Exceptions for Banks from the Definition of Broker in the Securities Exchange Act of 1934," defines important terms relating to the broker exceptions in the 1934 Act and provides a number of exemptions.

qualifies for the repurchase-to-maturity transaction exception, the agreement is accounted for as a secured borrowing. If effective control is not maintained or the repurchase-to-maturity transaction exception is not met, the transaction would be assessed under the other derecognition conditions in FASB ASC 860-10-40-5 to determine if the transferred financial asset should be derecognized and accounted for as a sale.

- 14.31 "Pending Content" in FASB ASC 860-10-40-24 states that an agreement that both entitles and obligates the transferor to repurchase or redeem transferred financial assets from the transferee maintains the transferor's effective control over those assets under item (c)(1) in FASB ASC 860-10-40-5, if all of the following conditions are met:
  - a. The financial assets to be repurchased or redeemed are the same or substantially the same as those transferred (see paragraph 14.32).
  - b. The agreement is to repurchase or redeem the financial assets before maturity, at a fixed or determinable price.
  - c. The agreement is entered into contemporaneously with, or in contemplation of, the transfer.
- **14.32** Item (a) in FASB ASC 860-10-40-24 states that, to be substantially the same, the financial asset that was transferred and the financial asset that is to be repurchased or redeemed need to have all of the following characteristics:
  - a. The same primary obligor (except for debt guaranteed by a sovereign government, central bank, government-sponsored enterprise or agency thereof, in which case the guaranter and the terms of the guarantee must be the same)
  - b. Identical form and type to provide the same risks and rights
  - c. The same maturity (or in the case of mortgage-backed pass-through and pay-through securities, similar remaining weighted-average maturities that result in approximately the same market yield)
  - d. Identical contractual interest rates
  - e. Similar assets as collateral
  - f. The same aggregate unpaid principal amount or principal amounts within accepted good delivery standards for the type of security involved

FASB ASC 860-10-55-35 contains implementation guidance related to these conditions.

- 14.33 In accordance with item (a) in FASB ASC 860-10-55-35, the exchange of pools of single-family loans would not meet the criteria in item (a)(1) in FASB ASC 860-10-40-24 (see item (a) in paragraph 14.32 because the mortgages making up the pool do not have the same primary obligor and would therefore not be considered substantially the same.
- 14.34 Item (b) in FASB ASC 860-10-55-35 explains that Government National Mortgage Association (Ginnie Mae) I securities for Ginnie Mae II securities, loans to foreign debtors that are otherwise the same except for different U.S. foreign tax credit benefits (because such differences in the tax receipts associated with the loans result in instruments that vary in form and type), and commercial paper for redeemable preferred stock would not meet the identical form and type criterion (see item [b] in paragraph 14.32).

- 14.35 In accordance with item (c) in FASB ASC 860-10-55-35, an example that would not meet the criterion addressed in item c in paragraph 14.32 would be the exchange of a *fast-pay* Ginnie Mae certificate (that is, a certificate with underlying mortgage loans that have a high prepayment record) for a *slow-pay* Ginnie Mae certificate because differences in the expected remaining lives of the certificates result in different market yields.
- **14.36** Related to item f in paragraph 14.32, participants in the MBSs market have established parameters for what is considered acceptable delivery. As stated in item (a)(6) in FASB ASC 860-10-40-24, these specific standards are defined by the Securities Industry and Financial Markets Association (SIFMA) and can be found in *Uniform Practices for the Clearance and Settlement of Mortgage-Backed Securities and Other Related Securities*, which is published by the SIFMA.
- 14.37 Repurchase financings. A repurchase financing, as defined in the FASB ASC glossary, is a repo that relates to a previously transferred financial asset between the same counterparties (or consolidated affiliates of either counterparty) that is entered into contemporaneously with, or in contemplation of, the initial transfer. In transactions involving a contemporaneous transfer of a financial asset and a repurchase financing of that transferred financial asset with the same counterparty, "Pending Content" in FASB ASC 860-10-40-4C states that a transferor and transferee should separately account for the initial transfer of the financial asset and the related repurchase agreement.
- **14.38** Paragraphs 17A–17C of FASB ASC 860-10-55 provide implementation guidance related to repurchase financings.
- **14.39** Offsetting. Financial institutions may operate on both sides of the federal funds and repo markets on the same day. FASB ASC 210-20-05-1 states that it is a general principle of accounting that the offsetting of assets and liabilities in the balance sheet is improper except where a right of setoff exists. The FASB ASC glossary defines right of setoff as the debtor's legal right, by contract or otherwise, to discharge all or a portion of the debt owed to another party by applying against the debt an amount that the other party owes to the debtor and FASB ASC 210-20-45-1 specifies conditions that must be met to permit offsetting. A debtor having a valid right of setoff may offset the related asset and liability and report the net amount, as stated in FASB ASC 210-20-45-2. According to FASB ASC 210-20-45-11, notwithstanding the condition in item (c) in FASB ASC 210-20-45-1, which requires an intent to net-settle the instruments' cash flows, an entity may, but is not required to, offset amounts recognized as payables under repos accounted for as collateralized borrowings and amounts recognized as receivables under reverse repos accounted for as collateralized borrowings if all the following conditions are met:
  - a. The repos and reverse repos are executed with the same counterparty.
  - b. The repos and reverse repos have the same explicit settlement date specified at the inception of the agreement.
  - c. The repos and reverse repos are executed in accordance with a master netting arrangement.
  - d. The securities underlying the repos and reverse repos exist in book entry form and can be transferred only by means of entries in the records of the transfer system operator or securities custodian.

Book entry securities meeting the criterion in this paragraph exist only as items in accounting records maintained by a transfer system operator. This requirement does not preclude offsetting securities held in book entry form solely because other securities of the same issue exist in other forms.

- e. The repos and reverse repos will be settled on a securities transfer system that operates in the manner described in paragraphs 14–17 of FASB ASC 210-20-45, and the entity must have associated banking arrangement in place as described in those paragraphs. Cash settlements for securities transferred should be made under established banking arrangements that provide that the entity will need available cash on deposit only for any net amounts that are due at the end of the business day. It must be probable that the associated banking arrangements will provide sufficient daylight overdraft or other intraday credit at the settlement date for each of the parties. The term probable is used in FASB ASC 210-20 consistent with its use in FASB ASC 450-20-25-1 to mean that a transaction or event is likely to occur.
- f. The entity intends to use the same account at the clearing bank or other financial institution at the settlement date in transacting both the cash inflows resulting from the settlement of the reverse repo and the cash outflows in settlement of the offsetting repo.

According to FASB ASC 942-210-45-3, FASB ASC 210, *Balance Sheet*, permits offsetting in the statement of financial position of only payables and receivables that represent repos and reverse repos and that meet all of the conditions specified therein and does not apply to securities borrowing or lending transactions.

- **14.40** FASB ASC 210-20-50-1 establishes the scope of the disclosure requirements in FASB ASC 210-20-50. The disclosures apply to both of the following:
  - a. Recognized derivative instruments accounted for in accordance with FASB ASC 815, *Derivatives and Hedging* (including bifurcated embedded derivatives), repos accounted for as collateralized borrowings and reverse repos, and securities borrowing and securities lending transactions that are offset in accordance with either FASB ASC 210-20-45 or FASB ASC 815-10-45
  - b. Recognized derivative instruments accounted for in accordance with FASB ASC 815 (including bifurcated embedded derivatives), repos and reverse repos, and securities borrowing and securities lending transactions that are subject to an enforceable master netting arrangement or similar agreement, irrespective of whether they are offset in accordance with either FASB ASC 210-20-45 or FASB ASC 815-10-45
- 14.41 Paragraphs 2–3 of FASB ASC 210-20-50 state that an entity should disclose information to enable users of its financial statements to evaluate the effect or potential effect of netting arrangements on its financial position for recognized assets and liabilities within the scope of the preceding paragraph. This includes the effect or potential effect of rights of setoff associated with an entity's recognized assets and recognized liabilities that are in the scope of FASB ASC 210-20-50 disclosure requirements. To meet these objectives, an

entity should disclose at the end of the reporting period the following quantitative information separately for assets and liabilities that are in scope:<sup>6</sup>

- a. The gross amounts of those recognized assets and those recognized liabilities
- b. The amounts offset in accordance with the guidance in FASB ASC 210-20-45 and FASB ASC 815-10-45 to determine the net amounts presented in the statement of financial position
- c. The net amounts presented in the statement of financial position
- d. The amounts subject to an enforceable master netting arrangement or similar agreement<sup>7</sup> not otherwise included in item b:
  - The amounts related to recognized financial instruments and other derivative instruments that either management makes an accounting policy election not to offset or do not meet some or all of the guidance in either FASB ASC 210-20-45 or FASB ASC 815-10-45
  - ii. The amounts related to financial collateral (including cash collateral).
- e. The net amount after deducting the amounts in item d from the amounts in item c.
- **14.42** *Disclosures.* FASB ASC 860 requires certain disclosures about transfers and servicing of financial assets. See chapter 10, "Transfers and Servicing and Variable Interest Entities," of this guide for additional information regarding the particular disclosures.
- **14.43** Item (a) in FASB ASC 860-30-50-1A requires an entity that has entered into repos or securities lending transactions to disclose its policy for requiring collateral or other security.
- 14.44 To provide an understanding of the nature and risks of short-term collateralized financing obtained through repurchase agreements, securities lending transactions, and repurchase-to-maturity transactions, that are accounted for as secured borrowings at the reporting date, "Pending Content" in FASB ASC 860-30-50-7 states that an entity should disclose the following information for each interim and annual period about the collateral pledged and the associated risks to which the transferor continues to be exposed after the transfer:
  - a. A disaggregation of the gross obligation by the class of collateral pledged. An entity should determine the appropriate level of disaggregation and classes to be presented on the basis of the nature, characteristics, and risks of the collateral pledged. Total borrowings under those agreements should be reconciled to the amount of the gross liability for repurchase agreements and securities lending transactions disclosed in accordance with item (a) in FASB ASC

<sup>&</sup>lt;sup>6</sup> FASB Accounting Standards Codification (ASC) 210-20-50-4 states that this information should be presented in tabular format, separately for assets and liabilities, unless another format is more appropriate. The total amount disclosed in accordance with FASB ASC 210-20-50-3(d) for an instrument should not exceed the amount disclosed in accordance with FASB ASC 210-20-50-3(c) for that instrument.

 $<sup>^7</sup>$  In accordance with FASB ASC 210-20-50-5, an entity should provide a description of the rights of setoff associated with an entity's recognized assets and recognized liabilities subject to an enforceable master netting arrangement or similar agreement, including the nature of those rights.

- 210-20-50-3 before any adjustments for offsetting. Any difference between the amount of the gross obligation disclosed under this paragraph and the amount disclosed in accordance with item (a) in FASB ASC 210-20-50-3 should be presented as reconciling item(s).
- b. The remaining contractual maturity of the repurchase agreements, securities lending transactions, and repurchase-to-maturity transactions. An entity should use judgment to determine an appropriate range of maturity intervals that would convey an understanding of the overall maturity profile of the entity's financing agreements.
- c. A discussion of the potential risks associated with the agreements and related collateral pledged, including obligations arising from a decline in the fair value of the collateral pledged and how those risks are managed.
- **14.45** In addition, FASB ASC 825-10-50-20 states that, except as indicated in FASB ASC 825-10-50-22, an entity should disclose all significant concentrations of credit risk arising from all financial instruments, whether from an individual counterparty or groups of counterparties. The concentrations-of-credit-risk disclosures apply to debt securities and loans.
- 14.46 The balance sheet classification noncash collateral by the seller-borrower depends on whether the buyer-lender has the right to sell or repledge the collateral under the agreement. If there is a right to sell or repledge the collateral, FASB 860-30-45-1 requires the seller-borrower to classify those assets separately from assets not so encumbered (for example, as security pledged to creditors). See paragraphs 10.70–.73 of this guide for further discussion.
- 14.47 When a smaller financial institution sells securities to another larger financial institution, under agreements to repurchase, the agreements may have default provisions that should be considered for disclosure in the financial statements. For example, a common default provision is if the selling financial institution drops below well capitalized under prompt corrective action provisions. The defaulting institution may be required to pay amounts in excess of the outstanding balance plus accrued interest.

# Auditing<sup>8</sup>

## **Objectives**

**14.48** The primary objectives of audit procedures applied to federal funds and repo transactions are to obtain sufficient appropriate evidence that

- a. the reported amounts include all federal funds purchased or sold and that repos and reverse repos are properly identified, described, and disclosed; include all such agreements; and are stated at appropriate amounts;
- b. interest expense or income and the related balance sheet accounts are properly measured and reported in the proper periods;

<sup>&</sup>lt;sup>8</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of *issuers* and *nonissuers* as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

- c. repos and dollar rolls are appropriately accounted for as sales or secured borrowings based upon whether the assets to be repurchased or redeemed are the same or substantially the same as those transferred;
- d. federal funds and repo transactions have been executed in accordance with management's authorizations and are obligations of the institution;
- e. assets pledged as collateral for federal funds and repo transactions are properly disclosed in the financial statements, including classification of collateral on the balance sheet;
- f. the federal funds sold and securities purchased under reverse repos exist and are either on hand or are held in safekeeping or custody for the bank;
- g. the institution has legal title or similar rights of ownership for all recorded securities:
- recorded amounts include all securities owned by the institution, and the financial statements include all related transactions during the period;
- *i.* the values at which securities are reported are appropriate;
- j. realized and unrealized gains and losses on sales of securities are properly measured, recorded, and disclosed;
- k. securities involved in such agreements are properly described and classified in the financial statements and the related footnote disclosures are adequate; and
- events of default are appropriately disclosed and any related fees are properly accrued; and
- m. amounts that have been offset in the statement of financial position meet the criteria for netting set forth in FASB ASC 210.

## **Planning**

**14.49** In accordance with AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (as described in chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide). Federal funds transactions are fairly routine for most institutions, are generally not complex, and many have matured by the close of the audit; thus, less risk may be associated with this account balance at a specific institution. Normal auditing procedures for borrowed funds could be applied to such obligations. However, certain repo transactions, whether viewed from an accounting, legal, or economic perspective, are extremely complex. Also, the risks involved in repo transactions vary widely, depending on the terms of the agreement, the parties involved, and the legal status of the agreement. The risks faced by an institution entering into a repo are generally reduced if the institution maintains effective controls related to the authorization, processing, and recording of these transactions. The auditing guidance in this chapter focuses on repo transactions.

- 14.50 The auditor inspects the current-year's interim financial statements, board of directors' reports and minutes, supervisory examination reports or related reports, and pertinent financial information and accounting to obtain an understanding of the level of activity in federal funds and repos, types of transactions entered into, accounting treatment (financing versus a sale and repurchase), and compliance with the institution's established investment and asset/liability management policies.
- **14.51** When an institution concentrates its repos with one dealer or a small group of dealers the evaluation of credit risk and counterparty risk gains significant importance to the auditor. The auditor might
  - a. obtain an understanding of the institution's controls over evaluating the reputation and financial strength of the dealer;
  - b. inspect the latest audited financial statements of the dealer;
  - c. obtain an understanding of the specific entity within an affiliated group with which the institution is doing business; and
  - obtain an understanding of transactions between that entity and its affiliates.
- 14.52 The audit procedures applied to federal funds purchased and securities sold under agreements to repurchase are also appropriate for federal funds sold and securities purchased under agreements to resell. It is important for the auditor to be aware that, as a buyer-lender, an institution might not take delivery of the securities that serve as collateral in a repo transaction. If it does take delivery, either directly or indirectly through another party acting as its agent, credit risk is less than may otherwise be the case; the auditor could consider confirming the occurrence and terms of the transaction, and the seller-borrower's obligation to repurchase the securities with the seller-borrower, and might consider counting securities in the institution's possession and confirming securities not in its possession with the custodian.
- 14.53 Whenever a buyer-lender or its agent does not take delivery of the securities, the auditor should consider confirming not only the occurrence and terms of the transaction and the obligation to repurchase the securities but also that they have not been delivered and are being held on the institution's behalf. It is important to note, when delivery is not made, the transaction has many of the attributes of an unsecured loan. Accordingly, the auditor should consider assessing the reputation and financial strength of the seller-borrower and of its custodian. Based on those assessments, the auditor should consider the desirability of obtaining a report from the custodian's auditor on the custodian's internal accounting controls over securities held in safekeeping, about which guidance for user auditors is provided in AU-C section 402, Audit Considerations Relating to an Entity Using a Service Organization (AICPA, Professional Standards), and for service auditors is provided in AT section 801, Reporting on Controls at a Service Organization (AICPA, Professional Standards). If, in accordance with paragraph .16a of AU-C section 402, the

©2016. AICPA

<sup>&</sup>lt;sup>9</sup> The AICPA Guide Service Organizations: Reporting on Controls at a Service Organization Relevant to User Entities' Internal Control Over Financial Reporting, contains information for practitioners reporting on controls at a service organization that affect user entities' internal control over financial reporting. Also, the AICPA Guide Reporting on Controls at a Service Organization Relevant to Security, Availability, Processing Integrity, Confidentiality, or Privacy (SOC 2®) summarizes the three SOC engagements and provides detailed guidance on planning, performing, and reporting on SOC 2® engagements.

user auditor plans to use a type 2 report as audit evidence that controls at the service organization are operating effectively, paragraph .17 of AU-C section 402 states that the user auditor should determine whether the service auditor's report provides sufficient appropriate audit evidence about the effectiveness of the controls to support the user auditor's risk assessment by

- a. evaluating whether the type 2 report is for a period that is appropriate for the user auditor's purposes;
- b. determining whether complementary user entity controls identified by the service organization are relevant in addressing the risks of material misstatement relating to the relevant assertions in the user entity's financial statements and, if so, obtaining an understanding of whether the user entity has designed and implemented such controls and, if so, testing their operating effectiveness;
- evaluating the adequacy of the time period covered by the tests of controls and the time elapsed since the performance of the tests of controls; and
- d. evaluating whether the tests of controls performed by the service auditor and the results thereof, as described in the service auditor's report, are relevant to the assertions in the user entity's financial statements and provide sufficient appropriate audit evidence to support the user auditor's risk assessment.

#### © Update 14-1 Attestation: Clarification and Recodification

Statement on Standards for Attestation Engagements (SSAE) No. 18, Attestation Standards: Clarification and Recodification (AICPA, Professional Standards), issued in April 2016, is effective for practitioners' reports dated on or after May 1, 2017.

To address concerns over the clarity, length, and complexity of its standards, the Auditing Standards Board established clarity drafting conventions and undertook a project to redraft all the standards it issues in clarity format. The redrafting of SSAEs (attestation standards) in SSAE No. 18 represents the culmination of that process. The attestation standards are developed and issued in the form of SSAEs and are codified into sections. SSAE No. 18 recodifies the "AT" section numbers designated by SSAE Nos. 10–17 using the identifier "AT-C" to differentiate the sections of the clarified attestation standards (AT-C sections) from the attestation standards that are superseded by SSAE No. 18 (AT sections). The AT sections in AICPA *Professional Standards* remain effective through April 2017, by which time substantially all engagements for which the AT sections were still effective are expected to be completed.

This edition of the guide has not been updated to reflect changes as a result of this SSAE, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this SSAE on the AICPA's website at www.aicpa.org.

# Internal Control Over Financial Reporting and Possible Tests of Controls

14.54 AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements

through understanding the entity and its environment, including the entity's internal control. Paragraphs .13–.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit (see examples of controls over financial reporting of federal funds and repo transactions in paragraph 14.55) and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

**14.55** Examples of controls over financial reporting of federal funds and repo transactions are as follows:

- The institution has formal, written policies that specify the types of securities that can be sold or repurchased under repos.
- Formal policies and procedures are in place to provide that repo transactions are executed in accordance with written contracts that describe the rights and obligations of the parties and that offsetting is applied only where appropriate based on the underlying contracts.
- Master agreements used by the institution should be entered into by authorized personnel and should specify the terms of the transactions and the intent of the parties.
- Only board-approved securities dealers and other institutions are allowed to enter into transactions with the institution.
- The institution has policies and procedures to provide that only authorized individuals enter into and approve such transactions and that those individuals are aware of the inherent risks and returns of such agreements. The institution's board of directors sets limits on the amount and terms of agreements with particular securities dealers and other institutions.
- The institution has policies and procedures for monitoring the reputation, financial stability, and creditworthiness of securities dealers and other institutions with which the institution may enter into an agreement as a basis for evaluating their ability to fulfill their obligation to return the collateral to the institution.
- The institution has procedures for monitoring communications with securities dealers and other institutions and for reviewing confirmations from securities dealers to detect unrecorded or inappropriately recorded transactions and to determine the reasonableness of interest rates.
- Initial transactions and rollover agreements are reviewed by a responsible official who determines whether the transactions represent sales or financing transactions.
- Written policies mandate frequent evaluation of the market value, including accrued interest, of the agreements and necessary collateral levels.

- The subsidiary ledgers containing information on securities collateralizing agreements are periodically reconciled to the general ledger. The reconciliations are reviewed by supervisory personnel.
- Policies and procedures exist to monitor the use of hedging techniques, if any, to reduce market risk.
- Review of the default provisions to ensure that any required disclosures are included in the financial statements.

Chapter 7, "Investments in Debt and Equity Securities," and chapter 15, "Debt," of this guide discuss related control issues for investments and borrowings, respectively.

- **14.56** AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards), addresses the auditor's responsibility to design and implement responses to the risks of material misstatement identified and assessed by the auditor in accordance with AU-C section 315 and to evaluate the audit evidence obtained in an audit of financial statements.
- 14.57 In accordance with paragraph .08 of AU-C section 330, the auditor should design and perform tests of controls to obtain sufficient appropriate audit evidence about the operating effectiveness of relevant controls if (a) the auditor's assessment of risks of material misstatement at the relevant assertion level includes an expectation that the controls are operating effectively or (b) substantive procedures alone do not provide sufficient appropriate audit evidence at the relevant assertion level. Examples of tests of controls the auditor may perform include the following:
  - Obtain and review the institution's written investment and asset/liability management policies (or other applicable policies relating to the management of federal funds and repo transactions), and consider whether such policies have been reviewed and approved by the institution's board of directors.
  - Obtain the institution's approved list of counterparties to agreements, and compare the list with those dealers with whom borrowing transactions were entered into during the current year.
     Ascertain that counterparty limits set by the board of directors have not been exceeded.
  - Review selected transactions to consider whether all significant terms were specified and documented and whether the amounts and terms were consistent with those established by the institution's formal investment and asset/liability management policies and subject to supervisory review.
  - Review supporting documentation for transactions, and consider whether only authorized individuals entered into or otherwise executed those transactions on behalf of the institution and that these components of the transactions were appropriately reviewed by supervisory personnel.
  - Review management's control over the appropriateness of the accounting for repo transactions, including the determination of sale or secured borrowing treatment, application of netting, recording the difference between the selling price and repurchase price as interest expense and determining whether interest expense is

- properly recorded on other borrowings, such as federal funds purchased. Obtain evidence that supervisory personnel reviewed the recording of the transaction.  $^{10}$
- Review the latest audited financial statements of the counterparties and other available reports, such as reports on internal control or special-purpose reports by the dealer's accountant, to determine whether the dealer has net capital in excess of statutory requirements.

14.58 If there is reason to question the creditworthiness of the counterparty, the auditor might consult with legal counsel regarding whether, in the event of the counterparty's inability to return (sell back) the collateral securities, the institution has the right to set off the loan liability against the collateral.

#### **Substantive Tests**

- 14.59 Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure. In accordance with paragraph .445 of AU-C section 330, this requirement reflects the facts that (a) the auditor's assessment of risk is judgmental and may not identify all risks of material misstatement and (b) inherent limitations to internal control exist, including management override.
- **14.60** Inspection of repo or other documentation of borrowing. Repos and other source documents may be inspected by the auditor and relevant details may be agreed upon concerning the respective recording of the liability in the subsidiary records. The auditor should test that securities collateralizing the borrowing are adequately identified to ensure proper disclosure and that the amounts and description should agree to the respective subsidiary ledger, and that when such securities are sold or re-hypothecated, a liability to return them is recorded by the original buyer-lender. The auditor should also examine the underlying contracts and (a) assess whether the criteria for offsetting assets and liabilities have been met (in cases where netting has been applied) and (b) identify any default provisions and determine whether any events of default have occurred and whether any liabilities require accrual as a result of the event of default.
- 14.61 Confirmation. Paragraph .A8 of AU-C section 500, Audit Evidence (AICPA, Professional Standards), states that corroborating information obtained from a source independent of the entity may increase the assurance that the auditor obtains from audit evidence that is generated internally, such as evidence existing within the accounting records, minutes of meetings, or a management representation. The auditor could consider confirming the amount and all significant terms of federal funds transactions and repos with the respective securities dealers, customers, and institutions. Details of any rollovers or extensions of repos should be agreed to brokers' advices. Confirming the repo transactions provides evidence of the occurrence of the transactions, their terms, and the treatment of the underlying securities, for example, evidence that the securities were delivered to the counterparty. Confirmation does not

<sup>&</sup>lt;sup>10</sup> These procedures also could be performed to provide substantive evidence.

provide evidence about the existence, location, or transferability of the securities or about the counterparty's ability to complete the transactions. It is usually impracticable to confirm the location of the securities delivered to the counterparty as collateral. The counterparty often is not able to determine the exact location of the securities delivered because they are fungible with other securities of the same issue under the dealer's control and are commingled with those securities. In addition, the counterparty may have appropriately used the securities for collateral in another repo or dollar roll in which the counterparty sold the securities to be repurchased at a later date. The seller-borrower and its auditor need not necessarily be concerned, however, about the location of securities transferred to the counterparty as collateral because their location does not necessarily affect the risk that the counterparty may not complete the transactions.

- **14.62** The auditor should consider the need to assess the counterparty's ability to complete the transaction by performing other procedures, such as testing the subsequent completion of the transaction for transactions that are completed after period-end but before the issuance of the financial statements, reviewing audited financial statements of the counterparty, considering the regulatory requirements applicable to the counterparty, and, if necessary, obtaining a special-purpose report from the counterparty's auditor.
- **14.63** Auditing interests in trusts held by a third-party trustee and reported at fair value. In circumstances in which the auditor determines that the nature and extent of auditing procedures should include verifying the existence and testing the measurement of investments held by a trust simply receiving a confirmation from the trustee, either in aggregate or on an investment by investment basis, does not in and of itself constitute adequate audit evidence with respect to the requirements for auditing the fair value of the interest in trust under AU-C section 540, Auditing Accounting Estimates, Including Fair Value Accounting Estimates, and Related Disclosures (AICPA, *Professional Standards*). In addition, receiving conformation from the trustee for investments in aggregate does not constitute adequate audit evidence with respect to the existence assertion. Receiving confirmation from the trustee on an investment-by-investment basis, however, typically would constitute adequate audit evidence with respect to the existence assertion. Also, in discussing obtaining an understanding of how management identifies the need for accounting estimates, paragraph .A15 of AU-C section 540 states that the preparation and fair presentation of the financial statements requires management to determine whether a transaction, an event, or a condition gives rise to the need to make an accounting estimate and that all necessary accounting estimates have been recognized, measured, and disclosed in the financial statements in accordance with the applicable financial reporting framework.
- **14.64** In circumstances in which the auditor is unable to audit the existence or measurement of interests in trusts at the financial statement date, the auditor should consider whether that scope limitation requires the auditor to either qualify his or her opinion or to disclaim an opinion, as discussed in AU-C section 705, *Modifications to the Opinion in the Independent Auditor's Report* (AICPA. *Professional Standards*).
- **14.65** Review of related-party transactions. The auditor might consider reviewing borrowing transactions involving related parties that have been accounted for as sales transactions to determine whether there are potential unrecorded financing transactions. A review of transaction activity may indicate

that an event accounted for as two separate transactions (a sale and subsequent purchase) is in fact a repo that should be accounted for as a financing. The auditor might consider the possibility of related party transactions that are improperly accounted for, possibly to avoid recognizing losses on sales.

14.66 Assess collateral risk. The auditor may assess the collateral risk through consideration of the counterparty's reputation, financial position, and market presence. The auditor may consider reviewing the fair values, including accrued interest, of securities serving as collateral and consider whether the collateral is sufficient or excessive in relation to the requirements of the agreement. The auditor might assess whether those securities repurchased under repos meet the substantially-the-same criteria for financing transactions or whether a gain or loss should have been recorded under a sales transaction. The auditor may test whether collateral held is properly recognized on the balance sheet in accordance with FASB ASC 860. Under FASB ASC 860-30-25-5(c), if the obligor (transferor) defaults under the terms of the secured contract and is no longer entitled to redeem the pledged asset, it should derecognize the pledged asset as required by FASB ASC 860-30-40-1 and the secured party (transferee) should recognize the collateral as its asset.

**14.67** Analytical procedures. Chapters 7 and 15 of this guide discuss analytical procedures that may also be applied in this area.

14.68 Tests of fair value disclosures. AU-C section 540 addresses the auditor's responsibilities relating to accounting estimates, including fair value accounting estimates and related disclosures, in an audit of financial statements. Specifically, it expands on how AU-C section 315, AU-C section 330, and other relevant AU-C sections are to be applied with regard to accounting estimates. It also includes requirements and guidance related to misstatements of individual accounting estimates and indicators of possible management bias. AU-C section 501, Audit Evidence—Specific Considerations for Selected Items (AICPA, Professional Standards), addresses specific considerations by the auditor in obtaining sufficient appropriate audit evidence, in accordance with AU-C sections 330, AU-C section 500, and other relevant AU-C sections, regarding certain aspects of (a) investments in securities and derivative instruments; (b) inventory; (c) litigation, claims, and assessments involving the entity; and (d) segment information in an audit of financial statements.

Considerations for Audits Performed in Accordance With PCAOB Standards

The second note to paragraph .A5 of AS 2201, An Audit of Internal Control Over Financial Reporting That Is Integrated with An Audit of Financial Statements (AICPA, PCAOB Standards and Related Rules), discusses the inherent limitations of internal control. Because fair value determinations often involve subjective judgments by management, this may affect the nature of controls that are capable of being implemented, including the possibility of management override of controls. The auditor considers the inherent limitations of internal control in such circumstances in assessing control risk.

Paragraph .47 of AS 2502, Auditing Fair Value Measurements and Disclosures (AICPA, PCAOB Standards and Related Rules), states that the auditor should evaluate the sufficiency and competence of the audit evidence obtained from auditing fair value measurements and disclosures as well as the consistency of that evidence with other audit evidence obtained and evaluated during the audit. The auditor's

evaluation of whether the fair value measurements and disclosures in the financial statements are in conformity with U.S. generally accepted accounting principles is performed in the context of the financial statements taken as a whole (see paragraphs .12–.18 and .24–.27 of AS 2810, Evaluating Audit Results [AICPA, PCAOB Standards and Related Rules]).

**14.69** *Other procedures.* Other audit procedures related to repos that the auditor may consider performing are as follows:

- Read the board of directors' minutes to determine whether financing transactions have been authorized.
- Test whether approved securities dealers are used, and whether financing arrangements comply with the institution's established policies.
- Recompute gains or losses on reverse repos that are accounted for as sales on a test basis.

Debt 449

# Chapter 15

# Debt

Gray shaded text in this chapter reflects guidance issued but not yet effective as of the date of this guide, July 1, 2016, but becoming effective on or prior to December 31, 2016, exclusive of any option to early adopt ahead of the mandatory effective date. Unless otherwise indicated, all unshaded text reflects guidance that was already effective as of the date of this guide.

### Introduction

- 15.01 Depository institutions use long and short term borrowings to provide funds that supplement deposits and to carry out their overall asset/liability management strategy. Finance and mortgage companies cannot accept deposits, and therefore, rely almost exclusively on borrowings to fund loans and operations.
- 15.02 Debt-to-equity ratios of finance companies generally are higher than those of manufacturing companies because finance company assets consist more of liquid assets, such as receivables, than of inventories and fixed assets. Debt-to-equity ratios of at least four- or five-to-one are not uncommon for finance companies. However, finance companies' leverage has traditionally been much lower than the leverage of depository institutions.
- **15.03** Debt may be classified as senior, senior subordinated, and junior subordinated. The classifications describe priorities of repayment, which become especially significant when solvency becomes questionable.
- 15.04 Internal policy and credit rating goals cause companies to establish diverse target amounts for each priority category of debt. Moreover, debt agreements usually contain restrictions on the amount of debt that may be incurred in each category. For example, a common restriction in debt securities issued to the general public prohibits pledging assets to secure new or existing debt. Other common restrictions may limit dividend payments and the amount of additional senior debt that can be incurred. If an issuer has other restrictions in its current typical public debt issue, lenders commonly demand the same restrictions in a private placement.
- 15.05 The creditworthiness of an institution's debt may be assessed by a rating agency based on analysis of the issuer's financial condition as measured by ratios and other factors. Ratings directly affect the institution's cost of borrowing and, thus, its ability to borrow. Institutional investors, such as other financial institutions, insurance companies, trusts, mutual funds, and pension and profit-sharing plans, rely heavily on credit ratings when making investment decisions. Some institutions are prohibited by law or formal agreement from investing in debt below a specified minimum level. For example, some states prohibit licensed domestic insurance companies from investing in corporate obligations that do not meet specified fixed-charge coverage ratios.

<sup>&</sup>lt;sup>1</sup> For example, many debt agreements consider the debt in default if the issuer fails to pay interest. Accordingly, credit risk may be assessed (in part) through analysis of fixed-charge coverage, which is the ratio of pretax earnings (before interest expense) to interest expense.

Similarly, many government agency pension funds are prohibited by law from investing in securities that do not have an investment grade<sup>2</sup> rating.

## Long Term Debt

- 15.06 The most common long term debt funding sources are debentures and notes. Institutions also may have long term mortgages, obligations and commitments under capital leases, and mandatorily redeemable preferred stock, that have many of the characteristics of debt. Such obligations are similar to those of other kinds of enterprises. Funds are also borrowed through Eurodollar certificates, collateralized mortgage obligations (CMOs) and real estate mortgage investment conduits (REMICs), mortgage-backed bonds (MBBs), mortgage-revenue bonds, and Federal Home Loan Bank (FHLB) advances.
- 15.07 The terms of an institution's long term debt obligations vary widely. They may be secured or unsecured. The debt may be senior or subordinated to other debt. The debt may be convertible into shares of preferred or common stock. Convertible debentures are convertible into stock at a specified price at the option of the holder. In most cases, convertible debt securities are also callable at the option of the issuer, generally beginning a few years after issuance. Interest rates may be fixed or floating.
- 15.08 Credit unions may borrow from individuals (whether or not they are members of the credit union) by issuing promissory notes or *certificates* of indebtedness. Certificates of indebtedness are generally uninsured. Their issuance is governed by Section 701.38 of the National Credit Union Administration (NCUA) regulations. Credit unions can have access to the NCUA maintained Central Liquidity Facility for short term borrowing by either being a member directly, or indirectly through an agent (usually a corporate credit union). Other notes issued by credit unions are generally payable to corporate credit unions or other financial institutions, or a Federal Reserve Bank.
- **15.09** Institutions and their subsidiaries sometimes finance expansion using traditional real estate mortgages.

#### Short Term Debt

- **15.10** Repurchase agreements. Repurchase agreements (repos) are discussed in chapter 14, "Federal Funds and Repurchase Agreements," of this guide.
- **15.11** Federal funds purchased. Federal funds purchased are discussed in chapter 14 of this guide.
- **15.12** *Commercial paper.* Commercial paper is an unsecured promissory note that provides creditworthy institutions, typically, finance companies or holding companies of banks and savings institutions, with short term funds. Commercial paper is generally short term (at most 270 days, but usually much less) and negotiable.
- **15.13** Institutions that rely heavily on commercial paper generally sell and redeem it continuously. They may sell more commercial paper than needed

<sup>&</sup>lt;sup>2</sup> In accordance with Title 12 U.S. Code of Federal Regulations Part 16.2, investment grade means that the issuer of a security has an adequate capacity to meet financial commitments under the security for the projected life of the asset or exposure. An issuer has an adequate capacity to meet financial commitments if the risk of default by the obligor is low and full and timely repayment of principal and interest is expected.

on certain days simply to maintain a market for customers who wish to invest beyond the institution's current needs. Sales of commercial paper may also increase when large amounts of commercial paper or long term debt mature. Proceeds in excess of current needs are often invested by entering into repos or by buying Eurodollar deposits, or commercial paper issued by others.

Debt

- **15.14** *Lines of credit.* Institutions often obtain funds through lines of credit from banks and savings institutions.
- **15.15** Institutions may obtain lines of credit as a source of funds or to provide creditors with assurance that commercial paper and other shorter term debt will be repaid. Further, rating agencies generally will not rate a finance company's commercial paper if it is not supported by a line of credit.
- **15.16** Institutions may pay commitment fees, maintain compensating balances, or do both to have lines of credit available. Interest rates on borrowings under lines of credit are usually based on a spread over the lender's prime rate based on the lender's assessment of credit risk.
- **15.17** Borrowing from Federal Reserve discount windows and FHLB system. Member depository institutions may borrow from their regional Federal Reserve Bank in the form of discounts (often called rediscounts) and advances, which are primarily used to cover shortages in the required reserve account and also in times of liquidity needs. A discounting transaction is technically a note to the Federal Reserve with recourse secured by a member institution's eligible loans. In an advancing transaction, a member institution executes a promissory note, which is collateralized generally by government securities to the Federal Reserve. Most discount-window transactions are in the form of advances. Interest charged in those transactions is referred to as discount. The rates are set biweekly by the individual reserve banks. Such loans usually have short maturities. Members of the FHLB System can obtain callable and non-callable advances of varying maturities from their district FHLBs. FHLB advances often are secured through pledges of loans or securities. Paragraph 15.73 discusses the performance of agreed-upon procedures relating to collateral for FHLB advances.
- 15.18 Treasury tax and loan note accounts. Employers that withhold federal taxes from employees' pay are required to deposit those funds periodically with a bank or savings institution. Institutions record such deposits, that are noninterest-bearing, as treasury tax and loan accounts and include such accounts with their deposits. However, on the day after receipt, such funds must be remitted to the Federal Reserve Bank or converted into an open-ended, interest-bearing note, commonly referred to as a treasury tax and loan note account.
- 15.19 Banker's acceptances. Paragraphs 24–26 of FASB Accounting Standards Codification (ASC) 860-10-05 state that banker's acceptances provide a way for a bank to finance a customer's purchase of goods from a vendor for periods usually not exceeding 6 months. Under an agreement among the bank, the customer, and the vendor, the bank agrees to pay the customer's liability to the vendor upon presentation of specified documents that provide evidence of delivery and acceptance of the purchased goods. The principal document is a draft or bill of exchange drawn by the customer that the bank stamps to signify its acceptance of the liability to make payment on the draft on its due date. Once the bank accepts a draft, the customer is liable to repay the bank at the time the draft matures. The bank recognizes a receivable from the customer

and a liability for the acceptance it has issued to the vendor. The accepted draft becomes a negotiable financial instrument. The vendor typically sells the accepted draft at a discount either to the accepting bank or in the marketplace. A risk participation is a contract between the accepting bank and a participating bank that the participating bank agrees, in exchange for a fee, to reimburse the accepting bank in the event that the accepting bank's customer fails to honor its liability to the accepting bank in connection with the banker's acceptance. The participating bank becomes a guarantor of the credit of the accepting bank's customer.

- 15.20 Banker's acceptances are similar to other short term borrowed funds in that they can be effectively used for short term liquidity needs by avoiding disbursing funds for short term loans to bank customers. Readers may also refer to FASB ASC 460, *Guarantees*, for guidance, because banker's acceptances contain an obligation to stand ready to perform over the term of the guarantee in the event that the specified triggering events or conditions occur, which may require the recognition of an obligation at fair value.
- **15.21** *MBBs*. MBBs are any borrowings (other than those from an FHLB) collateralized in whole or in part by one or more real estate loans. MBBs typically have the following characteristics:
  - a. Fixed maturities or payments of principal and interest
  - b. The use of mortgage loans or mortgage-backed securities (MBSs) owned by the issuer as collateral
  - c. Stated or fixed interest rates with interest payable monthly or semiannually (there may also be call provisions)
  - d. Principal payments made through periodic sinking-fund payments or at final maturity
  - e. Mortgage collateral in which the purchaser does not have an ownership interest
  - f. Collateral values usually ranging from 110 percent to 200 percent of the amount of the debt issue, so that the collateral value exceeds the principal value throughout its term (overcollateralization)
- **15.22** Preferred stock and other securities of finance subsidiaries. Finance subsidiaries, as defined in federal banking regulations, are a means by which institutions can issue preferred stock and other securities at rates lower than those the institutions would otherwise have to pay if they issued the securities directly. Thus, finance subsidiaries afford banking institutions the opportunity to obtain less costly funds.
- **15.23** Finance subsidiaries, as defined in the FASB ASC glossary, are subsidiaries with no assets, operations, revenues, or cash flows other than those related to the issuance, administration, and repayment of the security being issued and any other securities guaranteed by its parent entity.
- 15.24 In a structured financing (the simplest form of a finance subsidiary), the parent entity transfers certain assets to a special-purpose finance subsidiary to collateralize or otherwise support the securities issued by the finance subsidiary. In return for the assets, the subsidiary remits the net proceeds of the offering to the parent for use in operations. Where debt is issued at the subsidiary level, the trustee for the debt perfects a security interest in the transferred collateral. If preferred stock is issued, no security interest is perfected. However, because the finance subsidiary is chartered for the limited

purpose of issuing the securities and can neither incur debt nor engage in any other business (that is, a bankruptcy remote entity), the preferred stock is, in fact, insulated from other encumbrances and is, therefore, backed by the collateral in a manner approximating a security interest. The result is to provide greater protection for preferred stockholders than any of them would have had if the parent entity had been the issuer.

- 15.25 The economic value of this financing technique is made possible by a variety of factors. Because of the requirements established by the rating agencies, preferred stock offerings are significantly overcollateralized by a combination of mortgage securities, short term money-market instruments, treasuries, and other securities. This degree of collateralization, combined with the protection afforded by the structure, enables the rating agencies to issue triple-A ratings. Additionally, because qualified corporate taxpayers holding preferred stock are eligible for a deduction of a specified percentage of dividends received, the dividends paid by the issuer can be low by market standards, making the transaction a low-cost "borrowing" for the parent entity.
- **15.26** CMOs. As introduced in paragraph 7.31 of this guide, CMOs are multiclass, pay-through bonds collateralized by MBSs or mortgage loans and are generally structured so that all, or substantially all, of the collections of principal and interest from the underlying collateral are paid to the holders of the bonds. Typically, the bonds are issued with two or more maturity classes; the actual maturity of each bond class varies depending upon the timing of the cash receipts from the underlying collateral. CMOs are usually issued by a minimally capitalized special-purpose entity (issuer) established by one or more sponsors (frequently the original owners of the mortgages). The assets collateralizing the bonds are acquired by the special-purpose entity and then pledged to an independent trustee until the issuer's obligation under the bond indenture has been fully satisfied. The investor agrees to look solely to those trusteed assets and the issuer's initial capital (collectively referred to as segregated assets) for repayment of the obligations. Therefore, the sponsor and its other affiliates no longer have any financial obligations for the instrument, although one of those entities may retain the responsibility for servicing the underlying mortgage loans.
- 15.27 For the sponsor of the CMO, cash is immediately generated; there is no waiting for the collection of the amounts when the respective mortgage payments come due. Credit enhancement of CMOs is generally achieved by using collateral that carries a third-party guarantee; otherwise, CMOs are overcollateralized to mitigate the risk of default. The excess collateral generally reverts to the sponsor at the maturity of the CMOs. The sponsor of the CMO issuer may retain any residual (see chapter 7, "Investments in Debt and Equity Securities," of this guide), or an unrelated third party may acquire the residual as an investment.
- 15.28 For both the issuer and investor, cash flows may not materialize as scheduled. For example, prepayments of the underlying mortgages at a greater-than-anticipated rate can reduce the yield to maturity expected by the investor.
- **15.29** *REMICs*. REMICs are vehicles for issuing multiclass mortgage-backed obligations that require compliance with a number of technical requirements of the IRC. REMICs refer to the taxable entity (rather than to the security structure like a CMO and other types of mortgage-backed borrowings).

Failure to comply with the requirements could result in imposition of a corporate income tax on the gross income of the REMIC. REMIC certificates of ownership are qualifying real property loans and qualified assets under the IRC.

- 15.30 To qualify for REMIC status as defined by the IRC, all of the assets continuously held by the REMIC must consist of qualified mortgages and permitted investments. In general, the term, *qualified mortgages* refers to mortgages that are principally collateralized by an interest in real property and are transferred to the REMIC at the time of its formation or purchased by the REMIC within three months of its formation. Qualified mortgage also refers to a regular interest in another REMIC. The term *permitted investments* includes cash-flow investments, qualified reserve assets, and foreclosed property.
- 15.31 All of the interests in the REMIC normally consist of either regular interests or residual interests. A regular interest is an interest that unconditionally entitles the holder to receive specified principal and interest payments under terms that are fixed at the time of the REMIC's formation. A residual interest is any interest in a REMIC that is not a regular interest. Only one class of residual interest may exist with respect to a REMIC. In other words, the rights of all of the holders of an interest that does not qualify as a regular interest must be exactly the same.

# Regulatory Matters

- 15.32 Federal savings institutions must notify the Office of the Comptroller of the Currency before borrowing money, unless the institution meets regulatory capital requirements and any applicable minimum capital directive. Regulations may also prohibit growth above a certain level without prior regulatory approval. Further, savings institutions generally must obtain written approval (prior to issuance) for subordinated debt to qualify as regulatory capital.
- **15.33** The Federal Reserve Act limits the availability of certain types of borrowings through the Federal Reserve discount window.

# **Accounting and Financial Reporting**

- 15.34 Significant categories of borrowings should be presented as separate line items in the liability section of the balance sheet, or as a single line item with appropriate note disclosure of components as stated in FASB ASC 942-470-45-1. Institutions may, alternatively, present debt based on the debt's priority (that is, senior or subordinated) if they also provide separate disclosure of significant categories of borrowings. FASB ASC 860-30-45-1 explains that if the secured party (transferee) has the right by contract or custom to sell or repledge the collateral, then the obligor (transferor) should reclassify that asset and report that asset in its statement of financial position separately (for example, as security pledged to creditors) from other assets not so encumbered.
- 15.35 FASB ASC 860-30-45-2 states that liabilities incurred by either the secured party or obligor in securities borrowings or resale transactions should be separately classified. However, FASB ASC 860-30-45 does not specify the classification or the terminology to be used to describe such liabilities, nor does

Debt 455

it specify the classification or the terminology to be used for pledged assets reclassified by the transferror of securities loaned or transferred under a repo accounted for as a collateralized borrowing if the transferee is permitted to sell or repledge those securities.

- 15.36 FASB ASC 942-470-50-3 states that for debt, the notes to the financial statements should describe the principal terms of the respective agreements including, but not limited to, the title or nature of the agreement, or both; the interest rate (and whether it is fixed or floating); the payment terms and maturity date(s); collateral; conversion or redemption features; whether it is senior or subordinated; and restrictive covenants (such as dividend restrictions or earnings requirements), if any.
- 15.37 Accounting and reporting requirements for long term obligations are the same for financial institutions as for other entities, as stated in FASB ASC 942-470-50-2. If the financial institution has an unclassified balance sheet, there is no need to separate balances into current and long term portions. (See FASB ASC 44 for disclosure requirements of future payments on long term borrowings.)
- 15.38 In accordance with FASB ASC 470-10-50-1, institutions should disclose the combined aggregate amount of maturities and sinking-fund requirements for all long term borrowings for each of the five years following the date of the latest balance sheet presented.
- **15.39** According to FASB ASC 835-30-45-3, debt issuance costs should be reported in the balance sheet as deferred charges. FASB ASC 470-10-35-2 states that debt issue costs should be amortized over the same period used in the interest cost determination.

"Pending Content" in FASB ASC 835-30-45-3 states that amortization of discount or premium should be reported as interest expense in the case of liabilities or as interest income in the case of assets. Amortization of debt issuance costs also should be reported as interest expense. FASB ASC 470-10-35-2 states that debt issue costs should be amortized over the same period used in the interest cost determination.

**15.40** Debt instruments with redemption features should be carefully analyzed to determine the appropriate period over which these items should be amortized.

An entity should apply the new guidance on a retrospective basis, wherein the balance sheet of each individual period presented should be adjusted to reflect the period-specific effects of applying the new guidance. Upon transition, an entity is required to comply with the applicable disclosures for a change in an accounting principle. These disclosures include the nature of and reason for the change in accounting principle, the transition method, a description of the prior-period information that has been retrospectively adjusted, and the effect of the change on the financial statement line items (that is, debt issuance cost asset and the debt liability).

Readers are encouraged to read the full text of the ASU, available at www.fasb.org. Readers should apply the appropriate guidance based on their facts and circumstances.

©2016, AICPA

<sup>&</sup>lt;sup>3</sup> FASB Accounting Standards Update (ASU) No. 2015-03, Interest—Imputation of Interest (Subtopic 835-30): Simplifying the Presentation of Debt Issuance Costs, issued in April 2015, is effective, for public business entities, for financial statements issued for fiscal years beginning after December 15, 2015, and interim periods within those fiscal years. For all other entities, the amendments in this ASU is effective for financial statements issued for fiscal years beginning after December 15, 2015, and interim periods within fiscal years beginning after December 15, 2016. Early adoption is permitted for financial statements that have not been previously issued.

- 15.41 FASB ASC 480, *Distinguishing Liabilities from Equity*, <sup>4</sup> establishes standards for how an issuer classifies and measures certain financial instruments with characteristics of both liabilities and equity. "Pending Content" in paragraphs 4–5 of FASB ASC 480-10-25, as well as paragraphs 8 and 14 of FASB ASC 480-10-25, require an issuer to classify the following instruments as liabilities (or assets in some circumstances):
  - a. A mandatorily redeemable financial instrument, unless the redemption is required to occur only upon the liquidation or termination of the reporting entity. A financial instrument that embodies a conditional obligation to redeem the instrument by transferring assets upon an event not certain to occur becomes mandatorily redeemable if that event occurs, the condition is resolved, or the event becomes certain to occur.
  - b. Any financial instrument, other than an outstanding share, that, at inception, embodies an obligation to repurchase the issuer's equity shares, or is indexed to such an obligation, and that requires or may require the issuer to settle the obligation by transferring assets.
  - c. A financial instrument that embodies an unconditional obligation, or a financial instrument other than an outstanding share that embodies a conditional obligation, that the issuer must or may settle by issuing a variable number of its equity shares if, at inception, the monetary value of the obligation is based solely or predominantly on any of the following:
    - A fixed monetary amount known at inception (for example, a payable settleable with a variable number of the issuer's equity shares),
    - ii. Variations in something other than the fair value of the issuer's equity shares (for example, a financial instrument indexed to the S&P 500 and settleable with a variable number of the issuer's equity shares), and
    - iii. Variations inversely related to changes in the fair value of the issuer's equity shares (for example, a written put option that could be net share settled).
- **15.42** As stated in FASB ASC 480-10-25-10, examples of financial instruments that meet the criteria in FASB ASC 480-10-25-8 include forward purchase contracts or written put options on the issuer's equity shares that are to be physically settled or net cash settled.
- **15.43** Convertible instruments. Accounting for debt obligations that are convertible into other instruments such as equity securities can be very complex. Such instruments may require bifurcation of embedded derivatives,

<sup>&</sup>lt;sup>4</sup> For certain mandatorily redeemable financial instruments of certain nonpublic entities, the "Pending Content" in FASB Accounting Standards Codification (ASC) 480-10-25 has been indefinitely deferred. For certain mandatorily redeemable noncontrolling interests that were deferred for both public and nonpublic entities, FASB plans to reconsider implementation issues and, perhaps, classification or measurement guidance for those noncontrolling interests during the deferral period, in conjunction with FASB's ongoing projects. During the deferral period for certain mandatorily redeemable noncontrolling interests, all public entities as well as nonpublic entities that are SEC registrants are required to follow the disclosure requirements in paragraphs 1–3 of FASB ASC 480-10-50 as well as disclosures required by other applicable guidance. For additional effective date information see the transition and open effective date information discussed in FASB ASC 480-10-65-1.

Debt 457

bifurcation between liability and equity, calculation of any beneficial conversion feature, and impact earnings per share calculations.

- 15.44 Redeemable preferred stock. As noted in paragraph 15.41, "Pending Content" in FASB ASC 480-10-25-4 states that a mandatorily redeemable financial instrument should be classified as a liability unless the redemption is required to occur only upon the liquidation or termination of the reporting entity. "Pending Content" in FASB ASC 480-10-45-1 states that items within the scope of FASB ASC 480-10 should be presented as liabilities (or assets in some circumstances). Those items should not be presented between the liabilities section and equity section of the statement of financial position.<sup>5</sup>
- 15.45 Redeemable preferred stock that is conditionally redeemable (for example, stock that is puttable by the holder at a specified date) is not in the scope of FASB ASC 480. For SEC registrants, according to SEC Accounting Series Release No. 268, Presentation in Financial Statements of "Redeemable Preferred Stocks" (codified in the SEC Codification of Financial Reporting Policies), mezzanine presentation applies for conditionally redeemable stock that is not in the scope of FASB ASC 480. (Also see paragraphs 17.17–.26 of this guide for a discussion of preferred stock and regulatory capital).
- 15.46 Banker's acceptances. FASB ASC 860-10-55-65 addresses banker's acceptances and risk participations in them. An accepting bank that obtains a risk participation should not derecognize the liability for the banker's acceptance because the accepting bank is still primarily liable to the holder of the banker's acceptance even though it benefits from a guarantee of reimbursement by a participating bank. The accepting bank should not derecognize the receivable from the customer because it has not transferred the receivable. The accepting bank should, however, record the guarantee purchased, and the participating bank should record a liability for the guarantee issued. FASB ASC 460-10-05-2 explains that certain disclosures are required to be made by a guarantor in its interim and annual financial statements about its obligations under guarantees. The "General" subsections of FASB ASC 460 also address the recognition of a liability by a guarantor at the inception of a guarantee for the obligations the guarantor has undertaken in issuing that guarantee.
- **15.47** *MBBs.* FASB ASC 942-470-45-2 states that transfers of mortgages accounted for under FASB ASC 860, *Transfers and Servicing*, as secured borrowings of the issuing institution should be classified as debt on the institution's balance sheet. Such MBBs should be classified separately from advances, other notes payable, and subordinated debt.
- 15.48 Any discounts or premiums associated with the issuance of MBBs ordinarily should be reported in a contra liability (debit) or liability (credit) account, consistent with FASB ASC 835-30-45-1A and FASB Concept No. 6, *Elements of Financial Statements* (paragraphs 235–239). Bond issuance costs and expenses (legal, accounting, printing, and other expenses) generally should be deferred and amortized to operations using the constant effective yield method over the life of the bonds.

Any discounts or premiums associated with the issuance of MBBs ordinarily should be reported in a contra liability (debit) or liability (credit) account, consistent with "Pending Content" in FASB ASC 835-30-45-1A and FASB Concept No. 6, *Elements of Financial Statements* (paragraphs 235–239).

<sup>&</sup>lt;sup>5</sup> See footnote 4.

Similarly, related bond issuance costs should be reported in the balance sheet as a direct deduction from the face amount of that bond.<sup>6</sup>

- **15.49** Extinguishments of liabilities. FASB ASC 405-20 provides accounting and reporting standards for extinguishments of liabilities. FASB ASC 405-20-40-1 states that a debtor should derecognize a liability if and only if it has been extinguished. A liability has been extinguished if either of the following conditions is met:
  - a. The debtor pays the creditor and is relieved of its obligation for the liability. Paying the creditor includes delivery of cash, other financial assets, goods or services, or reacquisition by the debtor of its outstanding debt securities whether the securities are canceled or held as so-called treasury bonds.
  - b. The debtor is legally released from being the primary obligor under the liability, either judicially or by the creditor. FASB ASC 405-20-40-2 provides related guidance.

15.50 Gain or loss on extinguishments. Most gains or losses on extinguishments are recorded as ordinary items. However, paragraphs 1–2 of FASB ASC 470-50-45 indicate that gains and losses from extinguishment of debt that meet the criteria in FASB ASC 225-20 (unusual nature and infrequency of occurrence) are not precluded from being classified as extraordinary items unless they result from the application of line-of-credit or revolving debt arrangements under item (c) in FASB ASC 470-50-40-21.

Gain or loss on extinguishments. Most gains or losses on extinguishments are recorded as ordinary items. However, in accordance with "Pending Content" in FASB ASC 225-20-45-16, gains or losses from extinguishment of debt that an entity considers to be a material event or transaction of an unusual nature and of a type that indicates infrequency of occurrence should be reported as a separate component of income from continuing operations. The nature and financial effects of each event or transaction should be presented as a separate component of income from continuing operations, or, alternatively disclosed in notes to financial statements. Gains or losses of a similar nature that are not individually material should be aggregated and such items should not be reported on the face of the income statement net of income taxes. Similarly, the earnings per share effects of those items should not be presented on the face of the income statement.

<sup>&</sup>lt;sup>6</sup> See footnote 3.

<sup>&</sup>lt;sup>7</sup> FASB ASU No. 2015-01, Income Statement—Extraordinary and Unusual Items (Subtopic 225-20): Simplifying Income Statement Presentation by Eliminating the Concept of Extraordinary Items, was issued in January 2015. The amendments are effective for fiscal years, and interim periods within those fiscal years, beginning after December 15, 2015. A reporting entity may apply the amendments in this ASU prospectively. A reporting entity also may apply the amendments in this ASU retrospectively to all prior periods presented in the financial statements. Early adoption is permitted provided that the guidance is applied from the beginning of the fiscal year of adoption. The effective date is the same for both public business entities and all other entities.

For an entity that prospectively applies the guidance, the only required transition disclosure will be to disclose, if applicable, both the nature and the amount of an item included in income from continuing operations after adoption that adjusts an extraordinary item previously classified and presented before the date of adoption. An entity retrospectively applying the guidance should provide the disclosures in paragraphs 1–2 of FASB ASC 250-10-50.

Readers are encouraged to read the full text of the ASU, available at www.fasb.org. Readers should apply the appropriate guidance based on their facts and circumstances.

- **15.51** *Modifications or exchanges of debt.* FASB ASC 470-50 provides accounting and reporting standards for assessing modification or exchanges of debt arrangements in determining whether or not the modifications or exchanges will result in extinguishment or modification accounting. FASB ASC 470-60 provides accounting and reporting guidance for determining when a modification is considered a troubled debt restructuring.
- 15.52 Foreign currency debt. Entities with debt payable in a foreign currency may experience fluctuations in the reporting currency value of the debt due to changes in exchange rates. In some instances, entities enter into a currency swap contract to receive a foreign currency and pay the reporting currency. FASB ASC 815-10-45-2 states that none of the provisions in FASB ASC 815-10 support netting a hedging derivative's asset (or liability) position against the hedged liability (or asset) position in the balance sheet. Readers may refer to FASB ASC 815, Derivatives and Hedging, for further guidance.
- 15.53 Dual currency bonds. The guidance in paragraphs 35–36 of FASB ASC 815-20-55 related to foreign-currency-denominated interest payments also applies to dual-currency bonds that provide for repayment of principal in the functional currency and periodic fixed-rate interest payments denominated in a foreign currency, as stated in FASB ASC 815-20-55-37. FASB ASC 830-20 applies to dual-currency bonds and requires the present value of the interest payments denominated in a foreign currency to be remeasured and the transaction gain or loss recognized in earnings.
- 15.54 REMICs. As discussed, REMIC is simply a label that covers various forms of underlying securities. These securities may resemble either CMOs or pass-through certificates that represent a transfer of the underlying receivables. Institutions may enter into REMIC transactions to raise immediate cash utilizing mortgage agreements as collateral. FASB ASC 860 provides accounting and reporting standards for transfers of financial assets, including transfers associated with REMICs and CMOs. FASB ASC 810, Consolidation, provides guidance on the consolidation of these entities which are typically variable interest entities.
- 15.55 Lease financing. Accounting for leases by lessees and lessors is established in FASB ASC 840, Leases.

#### © Update 15-1 Accounting and Reporting: Leases

FASB Accounting Standards Update (ASU) No. 2016-02, Leases (Topic 842), issued in February 2016, is effective for fiscal years of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files financial statements with the SEC beginning after December 15, 2018, including interim periods within those fiscal years.

For all other entities, FASB ASU No. 2016-02 is effective for fiscal years beginning after December 15, 2019, and interim periods within fiscal years beginning after December 15, 2019.

Early application is permitted for all entities

FASB ASU No. 2016-02 supersedes the lease requirements in FASB ASC 840 and creates FASB ASC 842, *Leases*, to establish the principles that lessees

and lessors should apply to report useful information to users of financial statements about the amount, timing, and uncertainty of cash flows arising from a lease. FASB ASC 842 affects any entity that enters into a lease (as that term is defined in FASB ASU No. 2016-02), with some specified scope exceptions.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this discussion will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-02, see appendix F, "The New Leases Standard: FASB ASU No. 2016-02," of this guide.

**15.56** Lending of customers' securities. Banks and savings institutions sometimes lend customers' securities. FASB ASC 860-30 discusses application of the guidance to securities lending transactions. Any contingencies related to the lending of securities normally should be accounted for in conformity with FASB ASC 450, Contingencies.

15.57 Fair value measurements. FASB ASC 820, Fair Value Measurement, defines fair value, establishes a framework for measuring fair value, and expands disclosures about fair value measurements. See chapter 20, "Fair Value," of this guide for a summary of FASB ASC 820. Some institutions have elected to apply fair value accounting to financing agreements under FASB ASC 825, Financial Instruments.

# Auditing<sup>8</sup>

## **Objectives**

15.58 The primary audit objectives in this area are to obtain sufficient appropriate evidence that

- a. short and long term borrowings recorded as of the date of the financial statements, including embedded derivatives that are required to be accounted for in accordance with FASB ASC 815 and debt issuance costs, include all such liabilities of the institution and that they have been properly valued, classified, described and disclosed, and reflect all transactions for the period;
- b. financing subsidiaries are consolidated with the parent institution as required by FASB ASC 810;
- c. interest expense and the related balance sheet accounts (accrued interest payable, unamortized premiums or discounts, and issuance costs) are properly measured and recorded, and amortization has been properly computed;
- d. collateral for borrowings is properly identified and disclosed;

<sup>&</sup>lt;sup>8</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of issuers and nonissuers as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

Debt 461

- e. borrowings have been authorized in accordance with management's written policies and are obligations of the institution; and
- f. the effects on reported amounts and disclosures of any noncompliance with debt covenants are properly identified, described, and disclosed.
- g. the potential impact of failed covenants, debt maturing within 1 year or other related events that are considered in the auditor's going concern analysis.

## **Planning**

**15.59** In accordance with AU-C section 315, *Understanding the Entity and* Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (as described in chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide). Factors related to debt that could influence the risks of material misstatement may include accounting for borrowings, regulatory considerations, the existence of restrictive covenants, the existence of conversion features, and the existence and adequacy of collateral, if applicable. The auditor might also review board of directors' reports, the current year's interim financial statements, and other documents that may include information about whether any significant new debt has been incurred or issued and whether any significant debt has been repaid or refinanced. The auditor may also inquire as a means to obtain audit evidence about the nature of the entity, for example, the existence of financing subsidiaries.

# Internal Control Over Financial Reporting and Possible Tests of Controls

15.60 AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13—.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit, and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

**15.61** AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards), addresses the auditor's responsibility to design and implement responses to the risks of material misstatement identified and assessed by the auditor in accordance with AU-C section 315 and to evaluate the audit evidence obtained in an audit of financial statements.

15.62 In accordance with paragraph .08 of AU-C section 330, the auditor should design and perform tests of controls to obtain sufficient appropriate audit evidence about the operating effectiveness of relevant controls if (a) the auditor's assessment of risks of material misstatement at the relevant assertion level includes an expectation that the controls are operating effectively or (b) substantive procedures alone do not provide sufficient appropriate audit evidence at the relevant assertion level. Controls relating to the financial reporting of debt include, but are not limited to, the following:

- Debt transactions are reviewed and approved by the board of directors or its designated committee and documented in the minutes.
- Debt agreements are reviewed by the appropriate accounting and legal personnel to ensure that borrowings meet GAAP criteria for classification as a liability, debt agreements are evaluated for derivative accounting in accordance with FASB ASC 815 and debt issuance costs are appropriately identified and accounted for.
- Adjustments to liability accounts are reviewed and approved by responsible personnel.
- The subsidiary ledgers for long and short term borrowings and collateral are periodically reconciled with the general ledger. Reconciliations are regularly reviewed and approved by supervisory personnel.
- Reports or statements from outside trustees or transfer agents are periodically reconciled to the institution's records. Reconciliations are regularly reviewed and approved by supervisory personnel.
- Through periodic confirmation with the trustee or transfer agent, the institution ascertains that collateral on borrowings remains sufficient.
- Borrowings (such as CMOs and REMICs) are reviewed to ensure that they meet the GAAP criteria for treatment as financing transactions and consolidation.
- Periodic tests of covenant compliance are performed and reviewed by responsible personnel.

15.63 AU-C section 402, Audit Considerations Relating to an Entity Using a Service Organization (AICPA, Professional Standards), addresses the user auditor's responsibility for obtaining sufficient appropriate audit evidence in an audit of the financial statements of a user entity that uses one or more service organizations (for example, processing CMO or REMIC cash flows by a trustee). Specifically, it expands on how the user auditor applies AU-C sections 315 and 330 in obtaining an understanding of the user entity, including internal control relevant to the audit, sufficient to identify and assess the risks of material misstatement and in designing and performing further audit procedures responsive to those risks. If, in accordance with paragraph .16a of AU-C section 402, the user auditor plans to use a type 2 report as audit evidence that controls at the service organization are operating effectively, paragraph .17 of AU-C section 402 states that the user auditor should determine whether the service auditor's report provides sufficient appropriate audit evidence about the effectiveness of the controls to support the user auditor's risk assessment by

463

a. evaluating whether the type 2 report is for a period that is appropriate for the user auditor's purposes;

Debt

- b. determining whether complementary user entity controls identified by the service organization are relevant in addressing the risks of material misstatement relating to the relevant assertions in the user entity's financial statements and, if so, obtaining an understanding of whether the user entity has designed and implemented such controls and, if so, testing their operating effectiveness;
- c. evaluating the adequacy of the time period covered by the tests of controls and the time elapsed since the performance of the tests of controls; and
- d. evaluating whether the tests of controls performed by the service auditor and the results thereof, as described in the service auditor's report, are relevant to the assertions in the user entity's financial statements and provide sufficient appropriate audit evidence to support the user auditor's risk assessment.

#### **Substantive Tests**

- 15.64 Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure. In accordance with paragraph .445 of AU-C section 330, this requirement reflects the facts that (a) the auditor's assessment of risk is judgmental and may not identify all risks of material misstatement and (b) inherent limitations to internal control exist, including management override.
- **15.65** Review of documentation. The auditor might review documentation such as legal agreements and signed notes supporting long term debt and agree pertinent information to subsidiary ledgers. The auditor might review the following information:
  - a. Type of debt
  - b. Interest rate and dates interest is payable
  - c. Maturity of the debt
  - d. Underlying collateral of the debt, if any
  - e. Subordination of the debt
  - f. Evidence of regulatory approval
  - g. Presence of restrictive covenants
  - h. Unusual features
  - i. Embedded derivatives
- **15.66** Confirmation. Paragraph .A8 of AU-C section 500, Audit Evidence (AICPA, Professional Standards), states that corroborating information obtained from a source independent of the entity may increase the assurance that the auditor obtains from audit evidence that is generated internally, such as evidence existing within the accounting records, minutes of meetings, or a management representation. AU-C section 505, External Confirmations (AICPA, Professional Standards), addresses the auditor's use of external confirmation procedures to obtain audit evidence, in accordance with the requirements of AU-C sections 330 and 500.

15.67 The auditor should consider confirming pertinent information with the trustee or transfer agent, including all terms, unpaid balance, accrued interest payable, principal and interest payments made during the year, collateral description, annual trust accounts activity, and the occurrence of any violations of the terms of the agreement. If collateral is not under the control of the institution and is held by a trustee or transfer agent, the auditor should consider confirming its existence, completeness, and valuation with the trustee or transfer agent. If collateral is deemed deficient with respect to the terms of the debt agreement or is not under the control of the institution, the auditor might consider the need for disclosure.

15.68 Tests of premiums, discounts, and issuance costs. The auditor may consider testing borrowings that were issued at a premium or discount to determine whether amortization has been properly computed and recorded. The auditor may also evaluate the propriety of amortization of costs incurred in connection with a debt issuance. The auditor may also consider the appropriateness of debt issuance costs recorded and test the classification and account of the costs.

**15.69** Analytical procedures. Paragraph .05 of AU-C section 520, Analytical Procedures (AICPA, Professional Standards), states that when designing and performing analytical procedures, either alone or in combination with tests of details, as substantive procedures in accordance with AU-C section 330, the auditor should

- a. determine the suitability of particular substantive analytical procedures for given assertions, taking into account the assessed risks of material misstatement and tests of details, if any, for these assertions. (For example, analytical procedures can provide substantive evidence about the completeness of debt-related financial statement amounts and disclosures; however, such procedures in tests of debt expense are often less precise than substantive tests such as recalculations. Because institutions generally issue a wide variety of debt with rates that vary with each issuance, it is normally difficult to develop expectations to be used in analyzing yields on debt.)
- b. evaluate the reliability of data from which the auditor's expectation of recorded amounts or ratios is developed, taking into account the source, comparability, and nature and relevance of information available and controls over preparation.
- c. develop an expectation of recorded amounts or ratios and evaluate whether the expectation is sufficiently precise (taking into account whether substantive analytical procedures are to be performed alone or in combination with tests of details) to identify a misstatement that, individually or when aggregated with other misstatements, may cause the financial statements to be materially misstated.
- d. determine the amount of any difference of recorded amounts from expected values that is acceptable without further investigation as required by paragraph .07 of AU-C section 520 and compare

<sup>&</sup>lt;sup>9</sup> Paragraph .18 of AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards).

the recorded amounts, or ratios developed from recorded amounts, with the expectations.

- **15.70** As discussed in paragraph 15.69a, it is normally difficult to develop sufficiently precise expectations to be used in analyzing yields on debt. Accordingly, analytical procedures in this area might be considered only as a supplement to other substantive procedures, except where an expected yield can be known with some precision (using computer-assisted audit techniques). Analytical review procedures that the auditor may apply in the debt area include analysis and evaluation of the following:
  - Comparison of interest expense by major category of debt as a percentage of the average amount of the respective debt outstanding during the year with stated rates on the debt instruments (yield test)
  - Reasonableness of balance sheet accruals and other related balance sheet accounts (accrued interest payable, deferred issuance costs, and premiums and discounts) by comparison to prior-year balances
- 15.71 Paragraph .A7 of AU-C section 520 states that the auditor's substantive procedures to address the assessed risk of material misstatement for relevant assertions may be tests of details, substantive analytical procedures, or a combination of both. The decision about which audit procedures to perform, including whether to use substantive analytical procedures, is based on the auditor's judgment about the expected effectiveness and efficiency of the available audit procedures to reduce the assessed risk of material misstatement to an acceptably low level. Further guidance on the auditor's use of analytical procedures as substantive procedures is provided in AU-C section 520. AU-C section 315 addresses the use of analytical procedures as risk assessment procedures, and AU-C section 330 addresses the nature, timing, and extent of audit procedures in response to assessed risks; these audit procedures may include substantive analytical procedures.
- **15.72** *Other procedures.* Other audit procedures the auditor may consider, related to debt and the extinguishment of debt, are as follows:
  - Review debt covenants and test whether the institution has complied with such covenants. Determine whether disclosures are appropriate.
  - Read minutes of meetings of the board of directors to determine whether financing transactions have been authorized in accordance with the institution's written policies.
  - Compare recorded interest expense and accrued interest payable to recorded debt for completeness of debt liabilities.
  - Obtain a detailed supporting schedule of prior-year and current-year account balances. Agree the prior-year balance to prior-year working papers and the current-year balance to the general ledger. Review activity for reasonableness.
  - For CMOs and REMICs, obtain and review compliance and verification letters prepared by the trustee's auditors. (Such letters are prepared on an annual basis and provide for the verification of the principal balance of the collateral and bonds, the cash flows

- associated with the issue, and compliance with the respective terms of the underlying agreements.)
- Examine canceled notes for borrowings that have been paid in full or consider confirmation procedures to validate that borrowings have been paid in full.
- Read lease agreements, identifying those that should be capitalized, and determine whether they were recorded using effective rates of interest.
- For financing transactions resulting from failed sales under FASB ASC 860, revisit agreement terms to validate that financing treatment is still appropriate (that is, a call option may have expired resulting in a reconsideration event under FASB ASC 860 that may change the nature of the transfer from a financing to a true sale).
- Assess the sufficiency of the value of assets collateralizing any borrowings to determine whether the entity has posted and recorded sufficient collateral for its borrowings.
- 15.73 If applicable, the auditor may be engaged to perform agreed-upon procedures relating to collateral for FHLB advances by reference to the security agreement signed by the institution's management that indicates compliance. The procedures depend upon the nature of the agreement (blanket lien, specific lien without delivery, or specific lien with delivery of the collateral). The respective district FHLB provides guidance on procedures to be performed. In light of the professional standards, the auditor considers whether and how to perform all that is requested by the FHLB.
- 15.74 The terms of some debt agreements may mandate companies to have their independent auditors issue compliance reports on various restrictive covenants involving matters such as restrictions on assets, payments of interest, and dividend payments. Such reports, which normally are in the form of negative assurance, are discussed in AU-C section 806, Reporting on Compliance With Aspects of Contractual Agreements or Regulatory Requirements in Connection With Audited Financial Statements (AICPA, Professional Standards).
- 15.75 Finance company credit questionnaires. Finance companies provide creditors with financial and operating information through standard credit questionnaires developed jointly by industry participants and the Risk Management Association (an association of lending officers). Some finance companies include credit questionnaires as information in addition to the finance company's basic financial statements. The auditor's responsibility depends on the services requested by the finance company.
- 15.76 Paragraph .03 of AU-C section 725, Supplementary Information in Relation to the Financial Statements as a Whole (AICPA, Professional Standards), states that the objective of the auditor, when engaged to report on supplementary information in relation to the basic financial statements as a whole, is to evaluate the presentation of the supplementary information in relation to the basic financial statements as a whole and to report on whether the supplementary information (such as a credit questionnaire) is fairly stated, in all material respects, in relation to the financial statements as whole. Supplementary information (for purposes of generally accepted auditing standards)

is defined as information presented outside the basic financial statements, excluding required supplementary information that is not considered necessary for the financial statements to be fairly presented in accordance with the applicable financial reporting framework. Paragraphs .A7–.A8 of AU-C section 725 explain that supplementary information includes additional details or explanations of items in or related to the basic financial statements, consolidating information, historical summaries of items extracted from the basic financial statements, statistical data, and other material, some of which may be from sources outside the accounting system or outside the entity. Supplementary information may be prepared in accordance with an applicable financial reporting framework, by regulatory or contractual requirements, in accordance with management's criteria or in accordance with other requirements. The auditor may report on such information using the guidance in paragraphs .09–.13 of AU-C section 725.

15.77 However, paragraphs .A1–.A2 of AU-C section 725 state that the auditor's responsibility for information that a designated accounting standard setter 10 requires to accompany an entity's basic financial statements is addressed in AU-C section 730, Required Supplementary Information (AICPA, Professional Standards). The auditor's responsibility for financial and nonfinancial information (other than the financial statements and the auditor's report thereon) that is included in a document containing audited financial statements and the auditor's report thereon, excluding required supplementary information, is addressed in AU-C section 720, Other Information in Documents Containing Audited Financial Statements (AICPA, Professional Standards).

 $<sup>^{10}</sup>$  Designated accounting standards setter is defined in paragraph .04 of AU-C section 730, Required Supplementary Information (AICPA, Professional Standards).

Income Taxes 469

# Chapter 16

# **Income Taxes**

#### Introduction 1

16.01 Depository institutions generally are subject to the same tax rules that apply to other corporations, including those that are members of a consolidated group. Generally, credit unions are exempt from federal income taxes. Banks and savings institutions are permitted to elect subchapter S status under IRC Section 1362, provided certain requirements are met. Among these requirements are a 100-shareholder limitation (including family attribution), one class of stock restriction, and prohibition on the use of the reserve method of accounting for bad debts for tax. If elected, S corporation status essentially converts the financial institution to a pass-through entity for tax purposes so that most of the corporate level tax on income is avoided.

**16.02** The IRC contains many provisions specific to taxable depository institutions. Finance and mortgage companies generally are subject to the same tax rules that apply to other corporations. The purpose of this chapter is to highlight certain federal tax matters and related accounting matters specific to the industry and to provide related auditing guidance.

## **Banks and Savings Institutions**

- **16.03** *Definition of a bank for tax purposes.* IRC Section 581 defines a bank for tax purposes and provides special rules governing bank taxation.
- **16.04** Definition of a savings institution for tax purposes. Savings institutions are considered to be mutual savings banks (IRC Section 591), domestic building and loan associations (IRC Section 7701[a][19]), or cooperative banks (IRC Section 7701[a][32]). The failure of an institution to qualify as a savings institution may affect the financial accounting standards that apply.
- 16.05 Securities gains and losses. IRC Section 582 provides banks special treatment for certain asset dispositions. Gains and losses on bonds, debentures, notes, certificates, and other evidences of indebtedness held by banks generally are treated under IRC Section 582 as ordinary (rather than capital) gains and losses. Equity securities and other investments generally are not afforded IRC Section 582 ordinary treatment. IRC Section 582 generally is not applicable to nonbank subsidiaries, including, for example, investment companies or insurance agencies.
- **16.06** Tax bad-debt deductions. IRC Section 585 provides that a bank or savings institution with \$500 million or less in assets is allowed a tax bad-debt deduction for reasonable additions to the bad-debt reserve. This asset test generally is based upon the average adjusted tax basis of all assets. If the institution is a member of a controlled group (as defined), all assets of the group are taken into account. The annual addition to the reserve generally

©2016, AICPA AAG-DEP 16.06

<sup>&</sup>lt;sup>1</sup> This chapter is not intended to provide comprehensive discussion of all possible tax matters an accountant might encounter in the preparation or audit of the financial statements of a financial institution. Further, state tax matters are beyond the scope of the introductory section of this chapter. Consulting this chapter cannot take the place of a careful reading of the related laws, regulations, rulings, and related documents, where appropriate.

cannot exceed the greater of the amount computed using actual experience percentages or the base year fill-up method (as defined).

**16.07** A bank or savings institution with assets exceeding \$500 million generally is allowed to claim a tax bad-debt deduction only under the general rule of IRC Section 166, that generally permits taxpayers to de-duct any debt that becomes worthless, in whole or in part, during the taxable year (that is, the specific charge-off method).2 Treasury Regulation 1.166-2(d) provides circumstances under which a bank may presume a debt to have become worthless, in whole or in part including, if a bank makes a conformity election under Regulation 1.166-2(d)(3), that a debt or portion thereof charged off for regulatory purposes is presumed to be worthless for tax purposes as well. The conformity election must be formally adopted and requires a financial institution to obtain an express determination letter from its primary federal regulator during each examination. Presence of the conformity election can also support the nonrecognition, for tax purposes, of contractually accrued interest that has not been recorded for financial reporting purposes, but only to the extent that no cash has been received that is required to be applied to the debtor's contractual balance of such outstanding interest. ).

16.08 According to IRC Section 593 and the accompanying regulations, a qualifying thrift institution that, prior to 1996, was permitted to establish a reserve for bad debts based on a percentage of taxable income may be required to recapture a portion of such bad-debt reserve if it makes distributions to shareholders that exceed earnings and profits accumulated after 1951. Additionally, if such an institution makes a distribution in redemption of stock or in partial or complete liquidation, notwithstanding the existence of earnings and profits, a portion of the reserve may have to be recaptured.

16.09 Net operating losses. For taxable years beginning after August 5, 1997, net operating losses (NOLs) of banks and savings institutions generally are carried back 2 years and then forward 20 years under the provisions of IRC Section 172. For taxable years prior to 1994, banks and savings institutions also had various special provisions in the IRC that determined the appropriate carryback and carryforward periods. Legislation was passed in 2009 that allowed financial institutions that had not received assistance under the Troubled Asset Relief Program to make a one-time election to carry back NOLs generated in 2008 or 2009 up to 5 years instead of 2.

#### Other

16.10 Alternative minimum tax. Beginning in 1987, corporations must compute their federal tax liability under both the regular tax and alternative minimum tax (AMT) systems and pay the higher amount. The AMT system is a separate but parallel tax system that regular taxable income is increased or decreased by certain AMT adjustments and preference items to arrive at AMT income. A rate of tax generally lower than the regular tax rate is applied to AMT income. The AMT adjustments and preference items most common for banks include tax-exempt interest income on private activity bonds issued after August 7, 1986 (reduced by any related interest expense disallowance), and accelerated depreciation and cost recovery. An adjustment is also required for the adjusted current earnings amount (defined), that frequently includes additional modifications for all tax-exempt interest income, the dividends received

<sup>&</sup>lt;sup>2</sup> Finance companies and mortgage companies also use the specific charge-off method.

471

deduction, and the increase in the cash surrender value of life insurance over the premiums paid. Further, only 90 percent of AMT income may be offset by a NOL. Any excess of tax computed under the AMT system over the regular system generally is eligible to reduce future regular tax (a minimum tax credit).

16.11 Mark to market. IRC Section 475 generally requires any company that is a dealer in securities to mark its securities to market. A dealer in securities is broadly defined as any taxpayer that regularly purchases securities from, or sells securities to, customers in the ordinary course of business or regularly offers to enter into, assume, offset, assign or otherwise terminate positions in securities with customers in the ordinary course of a trade or business. The definition of securities in the IRC differs from and is generally more expansive than the definition of securities in the FASB Accounting Standards Codification (ASC) glossary. For example, a bank regulatory selling loans into the secondary market would be considered a dealer. Further, institutions that are dealers generally may not exempt any security held for investment from mark to market accounting for tax purposes, unless there is a proper identification statement in place and the security is identified as exempt at the date it is acquired. A window (sometimes as much as 30 days) has generally been allowed for identification of certain loans.

16.12 Interest expense relating to tax-exempt income. IRC Section 291 generally provides that 20 percent of the allocable interest expense attributable to tax-exempt obligations acquired by a financial institution after 1982 and before August 8, 1986, is not deductible. For tax-exempt obligations acquired after August 7, 1986, IRC Section 265 generally requires that all of the interest expense attributable to the obligation be nondeductible. An exception exists for certain "qualified small issuer" obligations (as defined), that are subject to IRC Section 291. To be a qualified small issuer, the issuer must either be a state or local government or issue bonds on behalf of a state or local government. The issuer also must reasonably expect to issue not more than \$10 million of tax-exempt obligations (other than certain private activity bonds) within a year. The American Recovery and Reinvestment Act of 2009 that was passed by Congress and signed into law on February 17, 2009, increases the \$10 million limit to \$30 million for tax-exempt obligations issued in 2009 and 2010 to finance new projects and to refund prior obligations.

#### Credit Union

**16.13** Federal credit unions are exempt from federal and state income tax. State chartered credit unions may be subject to state income tax. Credit union service organizations, that are subsidiaries of federal or state credit unions, may be subject to income tax for federal and state purposes.

# Regulatory Matters<sup>3</sup>

**16.14** The Federal Financial Institutions Examination Council (FFIEC) requires, for regulatory reporting purposes, that income taxes be accounted for in conformity with U.S. generally accepted accounting principles (GAAP).

(continued)

<sup>&</sup>lt;sup>3</sup> In July 2013, the Office of the Comptroller of the Currency (OCC), the Board of Governors of the Federal Reserve System (Federal Reserve), and the FDIC issued the final U.S. Basel III rules that overhaul existing capital adequacy rules and implement both the Basel III capital framework

However, income taxes receive special treatment in regulatory capital calculations as the federal banking regulatory agencies limit the amount of deferred tax assets that may be included in regulatory capital.

- **16.15** Under final U.S. Basel III rules, certain deferred tax assets must be excluded from a bank's calculation of regulatory capital:
  - Deferred tax assets that arise from NOL and tax credit carry forwards
  - b. Deferred tax assets arising from temporary differences that could not be realized through carryback of losses if those differences reversed at the report date, but only to the extent the deferred tax assets that could not be realized exceed a 10 percent of common equity tier 1 capital threshold individually and a 15 percent threshold in aggregate with certain other assets (most commonly, mortgage servicing assets)

Deferred tax assets must be analyzed net of any related valuation allowances and on a jurisdiction by jurisdiction basis. A bank may also opt to net deferred tax liabilities, on a pro rata basis, against these two groups of deferred tax assets. Transition rules phase in these exclusions for certain banks through calendar year 2018.

16.16 Certain assets must be excluded from regulatory capital in their entirety (for example, goodwill and other intangible assets) or to the extent they exceed the 10 percent and 15 percent thresholds mentioned previously (for example, mortgage servicing assets). A bank may opt to net associated deferred tax liabilities against these assets in order to reduce the exclusion. The option to net deferred tax liabilities in this paragraph and in paragraph 16.15 is a policy decision that must be applied consistently from each reporting period to the next. In one circumstance, netting is not optional: a bank that opts out of including accumulated other comprehensive income in its regulatory capital must net that exclusion with both associated deferred tax assets and associated deferred tax liabilities.

16.17 In 1998, the federal banking agencies adopted a statement of policy, Interagency Policy Statement on Income Tax Allocation in a Holding Company Structure. The policy statement, which does not materially change any of the guidance previously issued by the agencies, generally, requires that intercorporate tax settlements between an institution and its parent company be no less favorable to the institution than if it had filed its income tax return as a separate entity. Taxes should not be paid to the parent before the payment would

(footnote continued)

issued by the Basel Committee and certain requirements imposed by the Dodd-Frank Wall Street Reform and Consumer Protection Act. The new rules completely replace the agencies' existing Basel I-based capital requirements. In comparison to the Basel I framework, Basel III generally requires U.S. banking organizations to hold higher amounts of capital, especially common equity, against their risk-weighted assets. The new rules consist of a standardized approach to measuring risk-weighted as-sets and advanced approaches to measuring risk-weighted assets. The standardized approach is a nonmodels-based approaches to measuring risk-weighted assets. The standardized approaches are models-based approaches that apply only to large, internationally active banking organizations, or "advanced approaches" banking organizations with \$250 billion or more in total consolidated assets or total consolidated on-balance sheet foreign exposure of \$10 billion or more. Advanced approaches banking organizations are required to calculate their capital ratios under both the standardized approach and the advanced approaches and report the lower of the two ratios. Readers can access the final U.S. Basel III rules from any of the agencies' websites. Further discussion on capital adequacy can also be found in of this guide.

Income Taxes 473

have been due to the taxing authority and if the subsidiary incurs a tax loss, it should receive a refund from the parent. Adjustments for statutory tax considerations that arise in a consolidated return are permitted if they are made on an equitable basis, consistently applied to all affiliates. These rules generally require that deferred taxes of the institution may not be paid or transferred to, or forgiven by, its holding company. The agencies recommend that members of a consolidated group have a written comprehensive tax agreement to address intercorporate tax policies and procedures.<sup>4</sup>

# **Accounting and Financial Reporting**

**16.18** FASB ASC 740, *Income Taxes*, addresses financial accounting and reporting for the effects of income taxes that result from an entity's activities during the current and preceding years, as stated in FASB ASC 740-10-05-1. The objectives of accounting for income taxes, as stated in FASB ASC 740-10-10-1, are to recognize (a) the amount of income taxes payable or refundable for the current year and (b) deferred tax liabilities and assets for future tax consequences of events that have been recognized in an institution's financial statements or tax returns.

**16.19** Two basic principles are related to accounting for income taxes, each of which considers uncertainty through the application of recognition and measurement criteria, according to FASB ASC 740-10-05-5:

- A current income tax liability or asset is recognized for the estimated taxes payable or refundable on tax returns for the current year.
- A deferred tax liability or asset is recognized for the estimated future tax effects attributable to temporary differences and carryforwards.

As stated in FASB ASC 740-10-15-2, the principles and requirements of FASB ASC 740 are applicable to domestic and foreign entities in preparing financial statements in accordance with GAAP.

**16.20** The following basic requirements, as provided in FASB ASC 740-10-30-2, are applied to the measurement of current and deferred income taxes at the date of the financial statements:

- The measurement of current and deferred tax liabilities and assets is based on provisions of the enacted tax law; the effects of future changes in tax laws or rates are not anticipated.
- The measurement of deferred tax assets is reduced, if necessary, by the amount of any tax benefits that, based on available evidence, are not expected to be realized.

#### **Deferred Tax Assets and Liabilities**

**16.21** Subject to certain specific exceptions identified in FASB ASC 740-10-25-3, FASB ASC 740-10-55-7 establishes that a deferred tax liability is

©2016, AICPA

<sup>&</sup>lt;sup>4</sup> In June 2014, the OCC, the Federal Reserve, and the FDIC issued the *Addendum to the Interagency Policy Statement on Income Tax Allocation in a Holding Company*. The addendum instructs insured depository institutions (IDIs) and their holding companies to ensure their tax-allocation agreements expressly acknowledge that the holding companies are acting as agents for the IDI and are consistent with certain requirements of sections 23A and 23B of the Federal Reserve Act.

recognized for all taxable temporary differences. Deferred tax assets are to be recognized for all deductible temporary differences and operating loss and tax credit carryforwards. In accordance with the definition in the FASB ASC glossary, a *deferred tax asset* is reduced by a *valuation allowance* (as defined in the FASB ASC glossary) if, based on the weight of available evidence, it is more likely than not that some portion or all of the deferred tax assets will not be realized. (The term *more likely than not* means a likelihood of more than 50 percent as established in FASB ASC 740-10-25-6.)<sup>5</sup>

**16.22** An example of a deferred tax liability includes book and tax basis differences of assets and liabilities that will result in future taxable amounts. An example of a deferred tax asset includes book and tax basis differences of assets and liabilities that will result in future deductible amounts.

**16.23** FASB ASC 740-10-45-20 requires that the effect of a change in the beginning-of-the-year balance of a valuation allowance that results from a change in circumstances that causes a change in judgment about the realizability of the related deferred tax asset in future years ordinarily should be included in income from continuing operations.

## **Temporary Differences**

16.24 A temporary difference, as defined in the FASB ASC glossary, is a difference between the tax basis of an asset or liability computed pursuant to the requirements in FASB ASC 740-10 for tax positions, and its reported amount in the financial statements that will result in taxable or deductible amounts in future years when the reported amount of the asset or liability is recovered or settled, respectively. FASB ASC 740-10-25-20 cites eight examples of temporary differences.<sup>6</sup>

**16.25** Other examples of temporary differences common to financial institutions may include the following:

- Bad-debt reserves for institutions that deduct bad-debt reserves under IRC Section 585 (that excludes the base year amount discussed at paragraph 16.26) and bad-debt reserves for financial statement purposes in excess of the bad-debt reserve for tax purposes. (For larger institutions that are covered under IRC Section 166, there is no bad-debt reserve for tax purposes and, therefore, the entire allowance for credit losses in the financial statements is a temporary difference.)
- Interest on nonperforming loans is generally not recognized in income for financial and regulatory purposes, but may be required

<sup>&</sup>lt;sup>5</sup> In January 2015, FASB issued proposed Accounting Standards Update (ASU) *Income Taxes (Topic 740): Intra-Entity Asset Transfers* to eliminate the exception in U.S. generally accepted accounting principles that prohibits recognizing current and deferred income tax consequences for an intra-entity asset transfer until the asset or assets have been sold to an outside party. Consequently, this proposed ASU requires that an entity recognize the current and deferred income tax consequences of an intra-entity asset transfer when the transfer occurs. Readers are encouraged to visit the "Technical Agenda" page at www.fasb.org for the latest developments regarding the Accounting for Income Taxes: Intra-Entity Asset Transfers and Balance Sheet Classification of Deferred Taxes project and how it may impact the guidance in this chapter.

<sup>&</sup>lt;sup>6</sup> See footnote 5.

Income Taxes 475

- to be included in taxable income under IRC Section 451. This would result in additional tax basis in the interest receivable.
- Unrealized gains or losses on securities recorded under FASB ASC 320, Investments—Debt and Equity Securities, are generally not recognized for tax purposes.
- Other real estate owned and other assets may reflect postacquisition impairment write-downs in the financial statements; those write-downs are generally not recognized by a bank for tax purposes until the asset is sold or disposed of.<sup>7</sup>
- Accrued deferred compensation is not generally deductible for tax purposes until paid.
- *Accrued loss contingencies* are generally not deductible for tax purposes until the liability is fixed and determinable.
- Depreciation of property, plant, and equipment and the amortization of intangible assets may be different for financial statement and tax purposes.
- Accrual of retirement liabilities is often made in the financial statements in different periods from those in which the expense is recognized for tax purposes.
- Originated mortgage servicing assets recorded in the financial statements are typically not capitalized for tax purposes to the extent not deemed to be excess serving compensation.
- Other basis differences in assets and liabilities may be caused by the following:
  - Gains and losses on sales of loans, foreclosed assets, or property, plant, and equipment recognized in financial reporting periods different from tax periods.
  - Amortization of imputed interest income from transactions involving loans recognized in different periods for financial reporting and tax purposes.
  - Accretion of discount on securities recorded currently for financial reporting purposes, but subject to tax at maturity or sale, or accreted differently for tax purposes.
  - Commitment fees included in taxable income when collected but deferred to a period when earned for financial reporting purposes.
  - Loan fee income recognized on a cash basis for tax purposes while recognized as a yield adjustment for financial reporting purposes.
  - Federal Home Loan Bank stock dividends recognized as current financial reporting income but deferred for tax purposes.

©2016, AICPA

 $<sup>^7</sup>$  In January 2014, Revenue Procedure 2014-16 was issued to allow financial institutions that had been capitalizing the costs to acquire and hold foreclosed property the opportunity to file for an automatic change of accounting method in order to deduct these costs currently under defined circumstances.

- The timing of the recognition of *income or loss for hedges* and swaps that differ for financial reporting and tax purposes.
- 16.26 As described in FASB ASC 740-10-25-3, a deferred tax liability should not be recognized for certain types of temporary differences unless it becomes apparent that those temporary differences will reverse in the foreseeable future. These types of temporary difference include bad debt reserves for tax purposes of U.S. savings and loan associations (and other qualified thrift lenders) that arose in tax years beginning before December 31, 1987 (that is, the base-year amount). See paragraphs 1–3 of FASB ASC 942-740-25 for the specific requirements related to this exception.

#### Financial Statement Presentation and Disclosure

- **16.27** FASB ASC 740-10-50 provides guidance on the financial statement disclosure requirements relating to income taxes applicable to all entities.
- **16.28** FASB ASC 740-10-50-10 requires that institutions disclose the amount of income tax expense or benefit allocated to continuing operations and the amounts separately allocated to other items (in accordance with the intra-period tax allocation provisions of paragraphs 2–14 of FASB ASC 740-20-45) for each year that those items are presented. For example, the amount of income tax expense or benefit allocated to certain items whose tax effects are charged or credited directly to other comprehensive income (such as translation adjustments under FASB ASC 830, *Foreign Currency Matters*, and changes in the unrealized holding gains and losses of securities classified as available-forsale under FASB ASC 320) should be separately allocated and disclosed.
- **16.29** As established in FASB ASC 740-30-50-2, all of the following information should be disclosed whenever a deferred tax liability is not recognized because of the exceptions to comprehensive recognition of deferred taxes related to subsidiaries and corporate joint ventures:
  - A description of the types of temporary differences that a deferred tax liability has not been recognized and the types of events that would cause those temporary differences to become taxable
  - The cumulative amount of each type of temporary difference
  - The amount of unrecognized deferred tax liability for temporary differences related to investments in foreign subsidiaries and foreign corporate joint ventures that are essentially permanent in duration if determination of that liability is practicable or a statement that determination is not practicable
  - The amount of the deferred tax liability for other temporary differences that is not recognized in accordance with the provisions of FASB ASC 740-30-25-18
- **16.30** If a thrift institution does not recognize a deferred tax liability for the base-year amount of a tax bad debt reserve in accordance with FASB ASC 740-10-25-3, it must disclose the amount of the deferred tax liability not recorded and the types of events that would cause the temporary difference (the base-year reserve) to become taxable, per FASB ASC 942-740-50-1.

Income Taxes 477

## Auditing<sup>8</sup>

### **Objectives**

16.31 The objectives of auditing income taxes are to obtain sufficient appropriate evidence that

- a. the provision for income taxes and the reported income tax liability or receivable are properly measured, valued, classified, and described in accordance with GAAP:
- b. deferred income tax liabilities and assets accurately reflect the future tax consequences of events that have been recognized in the institution's financial statements or tax returns (temporary differences and carryovers); and
- c. a valuation allowance has been established for deferred tax assets when it is more likely than not that they will not be realized.

### **Planning**

16.32 In accordance with AU-C section 315, *Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement* (AICPA, *Professional Standards*), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (as described in chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide). Changes in specific tax laws from year to year could affect financial institutions, as well as general corporations. It is necessary for the auditor to be aware of such changes.

## Internal Control Over Financial Reporting and Possible Tests of Controls

16.33 AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13—.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

©2016, AICPA

<sup>&</sup>lt;sup>8</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of *issuers* and *nonissuers* as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

**16.34** AU-C section 330, *Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained* (AICPA, *Professional Standards*), addresses the auditor's responsibility to design and implement responses to the risks of material misstatement identified and assessed by the auditor in accordance with AU-C section 315 and to evaluate the audit evidence obtained in an audit of financial statements.

#### Substantive Tests

**16.35** Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure, which for a financial institution subject to income tax would generally include income taxes. In accordance with paragraph .A45 of AU-C section 330, this requirement reflects the facts that (a) the auditor's assessment of risk is judgmental and may not identify all risks of material misstatement and (b) inherent limitations to internal control exist, including management override.

**16.36** Substantive audit procedures may include the following:

- Review the tax status and consolidated return requirements of subsidiaries.
- Review the status of current-year acquisitions of other companies and their preacquisition tax liabilities and exposures, if acquired in a tax free acquisition.
- Obtain a schedule reconciling net income per books with taxable income for federal, state, and foreign income taxes. Agree amounts to general ledger and supporting documents as appropriate. Consider the reasonableness of the current and deferred tax account balances.
- Test the rollforward of tax balance sheet accounts. Consider vouching significant tax payments and credits.
- Review reconciliation of prior-year tax accrual to the actual filed tax returns and determine the propriety of adjustments made in this regard and consider the impact on the current-year and prioryear tax accruals.
- Consider the deductibility of transactions such as profit-sharing, bonus, contributions, or stock option transactions.
- Ascertain whether changes in income tax laws and rates have been properly reflected in the tax calculations and account balances.
- Review the allocation, apportionment, and sourcing of income and expense applicable to state tax jurisdictions with significant income or franchise taxes.
- Review classification and description of accounts to identify possible tax reporting differences such as reserves for anticipated losses or expenses.
- Review schedule of NOL and other tax credit carryforwards and their utilization.
- Review and determine the need for and appropriateness of any valuation allowance for deferred tax assets. It is important for

auditors to note that institutions often may have a significant deferred tax asset resulting from the loan loss reserve. This asset should be evaluated based upon the likelihood of realization, taking into account the timing of the bad-debt deduction, and the special NOL carryovers and carryback tax rules, if applicable.

- Review tax planning strategies and assumptions utilized in the calculation of deferred income taxes under FASB ASC 740 guidance on accounting for uncertain tax positions.
- Evaluate tax positions and consider the appropriate accounting treatment and disclosure requirements for any uncertain tax positions under FASB ASC 740.
- Evaluate the adequacy of the financial statement disclosures.
- For separate financial statements of affiliates, review terms of all tax-sharing agreements between affiliated entities to determine proper disclosure and accounting treatment. The auditor should be cognizant of and consider whether the institution is in compliance with the regulatory reporting for banks and savings institutions related to intercompany tax allocations, as discussed in the entry income taxes of a bank subsidiary of a holding company in the "Glossary" section of the FFIEC's Instructions for Preparation of Consolidated Reports of Condition and Income, and the Interagency Policy Statement on Income Tax Allocation in a Holding Company Structure.

©2016, AICPA AAG-DEP 16.36

## Chapter 17

# Equity and Disclosures Regarding Capital Matters

Gray shaded text in this chapter reflects guidance issued but not yet effective as of the date of this guide, July 1, 2016, but becoming effective on or prior to December 31, 2016, exclusive of any option to early adopt ahead of the mandatory effective date. Unless otherwise indicated, all unshaded text reflects guidance that was already effective as of the date of this guide.

#### Introduction

17.01 Chapters 1, "Industry Overview—Banks and Savings Institutions," and 2, "Industry Overview—Credit Unions," of this guide discuss the regulatory capital requirements for banks and savings institutions and credit unions, respectively. Chapter 4, "Industry Overview—Mortgage Companies," of this guide discusses similar capital requirements for mortgage companies. This chapter discusses the related financial statement disclosures and auditing guidance. Chapter 3, "Regulatory Considerations," of the AICPA Accounting Guide Brokers and Dealers in Securities, discusses similar capital requirements for broker-dealers. Finance companies unaffiliated with banking organizations are not subject to regulatory capital requirements. Finance companies affiliated with banking organizations are subject to consolidated regulatory capital requirements and prudential supervision by the Board of Governors of the Federal Reserve System (Federal Reserve).

## **Banks and Savings Institutions**

#### Introduction

17.02 Banks are organized with capital stock and shareholders. Savings institutions operate under a capital stock structure, like banks, or a cooperative form of ownership, similar to credit unions. Savings institutions operating under the cooperative form are referred to as "mutual institutions." Although mutual institutions may be incorporated, they issue no capital stock and have no stockholders. The equity section of a mutual institution's statement of financial condition generally consists only of retained earnings and the accumulated other comprehensive income under FASB Accounting Standards Codification (ASC) 220, Comprehensive Income. The equity section for banks and stock savings institutions additionally include common stock and additional paid-in capital.

### Equity

© Update 17-1 Accounting and Reporting: Recognition and Measurement of Financial Assets and Financial Liabilities

FASB Accounting Standards Update (ASU) No. 2016-01, Financial Instruments—Overall (Subtopic 825-10): Recognition and Measurement of

Financial Assets and Financial Liabilities, issued in January 2016, is effective for fiscal years, and interim periods within those fiscal years, of a public business entity beginning after December 15, 2017.

For all other entities (including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960, *Plan Accounting—Defined Benefit Pension Plans* through FASB ASC 965, *Plan Accounting—Health and Welfare Benefit Plans* on plan accounting), FASB ASU No. 2016-01 is effective for fiscal years beginning after December 15, 2018, and interim periods within fiscal years beginning after December 15, 2019. Early application is permitted for all other entities as of the fiscal years beginning after December 15, 2017, including interim periods within those fiscal years.

FASB ASU No. 2016-01 addresses certain aspects of recognition, measurement, presentation, and disclosure of financial instruments and affects all entities that hold financial assets or owe financial liabilities. Among other provisions of the guidance, the amendments in this ASU supersede the guidance to classify equity securities with readily determinable fair values into different categories (that is, trading or available-for-sale) and require equity securities (including other ownership interests, such as partnerships, unincorporated joint ventures, and limited liability companies) to be measured at fair value with changes in the fair value recognized through net income (other than equity securities accounted for under the equity method of accounting or those that result in consolidation of an investee).

The amendments in this ASU also require an entity to present separately in other comprehensive income the portion of the total change in fair value of a liability resulting from a change in the instrument-specific credit risk when the entity has elected to measure the liability at fair value in accordance with the fair value option for financial instruments.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-01, see appendix G, "Accounting for Financial Instruments," of this guide.

#### © Update 17-2 Accounting and Reporting: Credit Losses

FASB ASU No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, issued in June 2016, is effective for fiscal years of public business entities that are SEC filers beginning after December 15, 2019, including interim periods within those fiscal years.

For all other public business entities, the amendments in FASB ASU No. 2016-13 are effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years.

For all other entities, including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting, FASB ASU No. 2016-13 is effective for fiscal years beginning

after December 15, 2020, and interim periods within fiscal years beginning after December 15, 2021.

Early application is permitted for all entities as of the fiscal years beginning after December 15, 2018, *including* interim periods within those fiscal years.

FASB ASU No. 2016-13 creates FASB ASC 326, Financial Instruments—Credit Losses, to amend guidance on reporting credit losses for assets held at amortized cost basis and available-for-sale debt securities.

For assets held at amortized cost basis, FASB ASC 326 eliminates the probable initial recognition threshold in current U.S. generally accepted accounting principles (GAAP) and, instead, requires an entity to reflect its current estimate of all expected credit losses. The allowance for credit losses is a valuation account that is deducted from the amortized cost basis of the financial assets to present the net amount expected to be collected.

For available-for-sale debt securities, credit losses should be measured in a manner similar to current GAAP. However, FASB ASC 326 will require that credit losses be presented as an allowance rather than as a write-down.

FASB ASU No. 2016-13 affects entities holding financial assets and net investment in leases that are not accounted for at fair value through net income. The amendments affect loans, debt securities, trade receivables, net investments in leases, off-balance-sheet credit exposures, reinsurance receivables, and any other financial assets not excluded from the scope that have the contractual right to receive cash.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-13, see appendix G of this guide.

- 17.03 *Common stock*. Common stock consists of stock certificates issued to investors (stockholders) as evidence of their ownership interest. As defined in the FASB ASC glossary, *common stock* is stock that is subordinate to all other stock of the issuer.
- 17.04 Preferred stock. Preferred stock has certain privileges over common stock, such as a first claim on dividends. Typically, preferred stock conveys no voting rights, or only limited voting rights, to the holders. The rights of preferred stockholders are described in the articles of incorporation. As defined in the FASB ASC glossary,  $preferred\ stock$  is a security that has preferential rights compared to common stock.
- 17.05 Preferred stock may have certain characteristics or features that qualify it for different components of regulatory capital consistent with the applicable functional regulations and guidelines, for example, cumulative versus noncumulative dividends and perpetual versus limited life.
- **17.06** Additional paid-in capital. Amounts paid in the excess of par are additional paid-in capital. Absent such a stated par value, the bank or savings institution will assign a nominal par value to capital stock. Adjustments for

treasury stock transactions, stock-based compensation and capital contributions may also be included in additional paid-in capital.

17.07 Retained earnings. Retained earnings include undivided earnings and other appropriations as designated by management or regulatory authorities. Undivided earnings include the transfer of net income, declaration of dividends and transfers to additional paid-in capital.

**17.08** Accumulated other comprehensive income. In accordance with FASB ASC 220-10-45-14, the total of other comprehensive income for a period should be transferred to a component of equity that is presented separately from retained earnings and additional paid-in capital in a statement of financial position at the end of each accounting period.

17.09 In accordance with FASB ASC 220-10-45-10A, other comprehensive income includes, but is not limited to, the following:

- Unrealized holding gains and losses on available-for-sale securities (see FASB ASC 320-10-45-1).
- Amounts recognized in other comprehensive income for debt securities classified as available-for-sale and held-to-maturity related to an other-than-temporary impairment recognized in accordance with FASB ASC 320-10-35 if a portion of the impairment was not recognized in earnings.
- Gains and losses (effective portion) on derivative instruments that
  are designated as, and qualify as, cash flow hedges (see item (c) in
  FASB ASC 815-20-35-1). As stated in FASB ASC 815-30-35-3, the
  ineffective portion of the gain or loss on a derivative instrument
  designated as a cash flow hedge is reported in earnings.

17.10 Minority interest in consolidated subsidiaries. A minority interest is the portion of equity in a bank's subsidiary not attributable, directly or indirectly, to the parent bank. For regulatory capital purposes, generally, banks may include such minority interests in equity capital accounts (both common and noncumulative perpetual preferred stocks) of consolidated subsidiaries if the subsidiary is a depository institution or a foreign bank, unless such accounts would not otherwise qualify for inclusion in common equity tier 1 or tier 1 capital. For example, a bank may not include minority interests representing cumulative preferred stock in consolidated subsidiaries because such preferred stock, if issued directly by the bank, would not be eligible for inclusion in tier 1 capital. The amount of minority interest includable is limited if the subsidiary has equity in excess of the subsidiary's minimum capital requirements and the applicable capital conservation buffer. Minority interests in consolidated asset-backed commercial paper conduits are excluded for regulatory capital purposes if the consolidated program assets are excluded from risk-weighted assets. Although a minority interest in a consolidated subsidiary is generally includable in common equity tier 1 or tier 1 capital, a bank's primary federal regulator may determine that it is not includable if it fails to provide meaningful capital support, fails to contribute to the subsidiary's ability to absorb losses, or presents other safety and soundness concerns.

## Holding Company Equity and Regulatory Capital<sup>1</sup>

17.11 Trust preferred securities. In accordance with FASB ASC 942-810-55-1, trust preferred securities (TPSs) have been issued by banks for a number of years due to favorable regulatory capital treatment. However, under current regulatory capital rules, TPSs no longer receive as favorable a capital treatment. See further discussion beginning in paragraph 17.20. FASB ASC 942-810-55-1 goes on to explain that various trust preferred structures have been developed involving minor differences in terms. Under the typical structure, a bank holding company first organizes a business trust or other special purpose entity. This trust issues two classes of securities: common securities, all of which are purchased and held by the bank holding company, and TPSs, which are sold to investors. The trust's only assets are deeply subordinated debentures of the corporate issuer, which the trust purchases with the proceeds from the sale of its common and preferred securities. The bank holding company makes periodic interest payments on the subordinated debentures to the business trust, which uses these payments to pay periodic dividends on the TPSs to the investors. The subordinated debentures have a stated maturity and may include an embedded call option. Most TPSs are subject to a mandatory redemption upon the repayment of the debentures.

17.12 Under the provisions of FASB ASC 810, *Consolidation*, a bank or a holding company that sponsored a structure described in the preceding paragraph should not consolidate the trust because the trust is a variable interest entity (VIE) and the bank or holding company is not the primary beneficiary of that VIE, as provided by FASB ASC 942-810-55-2.

In June 2011, the Office of the Comptroller of the Currency (OCC), Board of Governors of the Federal Reserve System (Federal Reserve), and the FDIC published a final rule, Risk-Based Capital Standards: Advanced Capital Adequacy Framework—Basel II; Establishment of a Risk-Based Capital Floor. The final rule was effective July 28, 2011, and it amends (a) the advanced risk-based capital adequacy standards (advanced approaches rule) in a manner that is consistent with certain provisions of the Dodd-Frank Wall Street Reform and Consumer Protection Act (Dodd-Frank Act), and (b) the general risk-based capital rules to provide limited flexibility consistent with Section 171(b) of the Dodd-Frank Act for recognizing the relative risk of certain assets generally not held by depository institutions.

In July 2013, the OCC, the Federal Reserve, and the FDIC issued the new regulatory capital rules that implement both the Basel III capital framework issued by the Basel Committee and certain requirements imposed by the Dodd-Frank Act. Readers can access the final U.S. Basel III rules from any of the agencies' websites.

In November 2015, the OCC, the Federal Reserve, and the FDIC issued guidance to clarify the interaction between the agencies' regulatory capital rule and section 13 of the Bank Holding Company Act, also known as the Volcker rule, with respect to the appropriate capital treatment for investments in certain private equity funds and hedge funds (covered funds). Readers can access the guidance from any of the agencies' websites.

In March 2016, the OCC, the Federal Reserve, the FDIC, and the SEC issued a Frequently Asked Questions (FAQ) document to clarify the capital treatment of certain Collateralized Debt Obligations backed by Trust Preferred Securities (TruPS CDO) the Volcker rule. The Volcker rule permits a banking entity to retain its interest in a Qualifying TruPS CDO under Title 12 U.S. Code of Federal Regulations (CFR) Part 351.16 of the interagency interim final rule. Readers can access the FAQ from any of the agencies' websites.

©2016. AICPA

<sup>&</sup>lt;sup>1</sup> In 1988, the Basel Committee on Banking Supervision (Basel Committee) introduced a capital measurement system commonly referred to as the Basel Capital Accord. This system provided for the implementation of a credit risk measurement framework with a minimum capital standard of 8 percent by the end of 1992. Since 1988, this framework has been progressively introduced not only in member countries but also in virtually all other countries with internationally active banks. The committee issued a capital adequacy framework which consists of three pillars: minimum capital requirements; supervisory review of an institution's internal assessment process and capital adequacy; and effective use of disclosure to strengthen market discipline as a complement to supervisory efforts.

- 17.13 The trust's dividend is financed by the trust through the purchase of the debentures.
- 17.14 In accordance with FASB ASC 942-810-45-1, in the typical trust preferred arrangement, the bank holds no variable interest in the trust, and therefore, cannot be the trust's primary beneficiary. If the bank does not consolidate the trust, the bank or holding company should report its debt issued to the trust and an equity-method investment in the common stock of the trust.
- **17.15** FASB ASC 810-10-55-31 states that some assets and liabilities of a VIE have embedded derivatives. For the purpose of identifying variable interests, an embedded derivative that is clearly and closely related economically to its asset or liability host is not to be evaluated separately.
- 17.16 Under this guidance, an embedded call option is not a variable interest in the trust.
- 17.17 Regulatory capital treatment of TPSs. On October 21, 1996, the Federal Reserve approved the use of certain cumulative preferred stock instruments in tier 1 capital for bank holding companies. Similar interpretive guidance and approvals for qualification as tier 1 capital for national banks and state chartered nonmember banks and thrifts also have been provided, on an institution-specific preapproval basis, by the Office of the Comptroller of the Currency (OCC), the FDIC, and the Office of Thrift Supervision (prior to its transfer of powers to the OCC, the FDIC, and the Federal Reserve). The Capital and Dividends Booklet of the Comptroller's Licensing Manual provides additional information regarding the guidance issued by the OCC.
- 17.18 On March 1, 2005, the Federal Reserve issued Risk-Based Capital Standards: Trust Preferred Securities and the Definition of Capital (Title 12) U.S. Code of Federal Regulations [CFR] Parts 208 and 225 [Regulations H and Y). This rule allows the continued limited inclusion of TPSs in the tier 1 capital of bank holding companies. Under this rule, TPSs and other restricted core capital elements are subject to stricter quantitative limits. Prior to the rule, the amount of TPSs, together with other cumulative preferred stock that a bank holding company could include in tier 1 capital was limited to 25 percent of tier 1 capital. This rule limits restricted core capital elements to 25 percent of all core capital elements, net of goodwill less any associated deferred tax liability. Internationally active bank holding companies, defined as those with consolidated assets greater than or equal to \$250 billion or on a consolidated basis reports on-balance sheet foreign exposure greater than or equal to \$10 billion, will be subject to a 15 percent limit (excluding mandatory convertible preferred securities). They may include qualifying mandatory convertible preferred securities up to the generally applicable 25 percent limit. Amounts of restricted core capital elements in excess of these limits generally may be included in tier 2 capital.
- 17.19 The requirement for TPSs to include a call option was eliminated and standards for the junior subordinated debt underlying TPSs eligible for tier 1 capital treatment were clarified. The rule also addressed supervisory concerns, competitive equity considerations, and the accounting for TPSs. The rule also strengthened the definition of regulatory capital by incorporating long

<sup>&</sup>lt;sup>2</sup> See chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide for further discussion on the Office of Thrift Supervision transfer of powers.

standing board policies regarding the acceptable terms of capital instruments included in banking organizations' tier 1 or tier 2 capitals.

- 17.20 Section 171 of the Dodd-Frank Wall Street Reform and Consumer Protection Act (Dodd-Frank Act) addresses deductions from regulatory capital and includes the following provisions:
  - Trust-preferred securities issued by banks and thrift holding companies after May 19, 2010, will no longer count as tier 1 capital.
     Trust-preferred securities may otherwise qualify to be treated as tier 2 capital.
  - Trust-preferred securities issued before May 19, 2010, by bank and thrift holding companies with \$15 billion or more in assets were treated as tier 1 capital (subject to existing limitations, see paragraphs 17.17–.18) until January 2013. Then, the tier 1 capital treatment will be phased out over a 3-year period (see paragraph 17.21).
  - Bank and thrift holding companies with assets of less than \$15 billion as of December 31, 2009, may continue to include trust-preferred securities that were issued before May 19, 2010, as tier 1 capital (subject to existing limitations, see paragraphs 17.17—.18).
  - These provisions of the Dodd-Frank Act do not apply to small bank and savings and loan holding companies (holding companies with less than \$1 billion in assets). See further discussion of small banking and savings and loan holding companies in paragraph 17.26.
- 17.21 In July 2013, the OCC, the Federal Reserve, and the FDIC issued the new regulatory capital rules. Among other provisions, the new rule improves the quality of capital by phasing out of tier 1 capital by 2016 instruments such as trust preferred securities and cumulative preferred securities.<sup>3</sup> However, the new rule grandfathers the inclusion of these instruments in tier 1 capital, subject to limitations, for banking organizations that have consolidated assets of less than \$15 billion as of December 31, 2009. Although new issuances from these institutions will have to meet new stricter criteria, these banking organizations may continue to include instruments issued prior to May 19, 2010, in tier 1 capital subject to current limitations.
- 17.22 Mandatory redeemable preferred stock. Banks may issue mandatorily redeemable preferred stock as part of their capital structure.
- 17.23 "Pending Content" in FASB ASC 480-10-25-4 states that a mandatorily redeemable financial instrument should be classified as a liability unless the redemption is required to occur only upon the liquidation or termination of the reporting entity. "Pending Content" in FASB ASC 480-10-45-1 states that items within the scope of FASB ASC 480-10 should be presented as liabilities (or assets in some circumstances). Those items should not be presented between the liabilities section and the equity section of the statement of financial

<sup>&</sup>lt;sup>3</sup> Advanced approaches depository institution holding companies may include these instruments in tier 2 capital temporarily because the instruments are subject to the phase out schedule. All other depository institution holding companies may include these instruments in tier 2 capital permanently.

position.<sup>4</sup> See further discussion on distinguishing liabilities from equity in paragraph 15.41 of this guide.

17.24 FASB ASC 480, Distinguishing Liabilities from Equity, does not address redeemable preferred stock that is conditionally redeemable (for example, stock that is putable by the holder at a specified date.) Mezzanine presentation would continue to apply for conditionally redeemable stock that is not in the scope of FASB ASC 480. If mandatorily redeemable shares are subject to the deferral under FASB ASC 480-10-65-1, the guidance in the SEC Regulation S-X, Section No. 210.5–02.28, is applicable. This regulation states that mandatory redeemable preferred stock is not to be included in amounts reported as stockholders' equity. Although nonpublic companies are not required to follow Regulation S-X, it would be appropriate for them to do so in most cases.

17.25 Perpetual preferred stock issued to the U.S. Treasury. On June 1, 2009, the Federal Reserve adopted a final rule (Federal Register Vol. 74, No. 103 [1 June 2009], pp. 26081–26084) to allow bank holding companies that have issued senior perpetual preferred stock to the U.S. Department of the Treasury under the capital purchase and other programs established by the Secretary of the Treasury under the Emergency Economic Stabilization Act of 2008, to include such capital instruments in tier 1 capital for purposes of the Federal Reserve's risk-based and leverage capital guidelines for bank holding companies. The final rule became effective on July 1, 2009.

17.26 Bank and savings and loan holding companies under \$1 billion in assets. The Federal Reserve adopted a Small Bank Holding Company Policy that provides flexibility for qualifying bank and savings and loan holding companies to be exempt from the minimum regulatory capital requirements and a surveillance program to assist in the assessment of the capital adequacy of small bank and savings and loan holding companies regulated by the Federal Reserve. The bank and savings and loan holding company capital adequacy guidelines apply on a consolidated basis to bank and savings and loan holding companies with consolidated assets of \$1 billion or more. For bank and savings and loan holding companies with less than \$1 billion in consolidated assets, the guidelines will be applied on a bank only basis unless the parent is engaged in a nonbank activity involving significant leverage, conduct significant off-balance sheet activities, or the parent company has a significant amount of outstanding debt that is held by the general public.

17.27 Foreign banking organizations with U.S. non-branch assets of \$50 billion or more. On February 17, 2014, the Federal Reserve adopted a final rule, which establishes a number of enhanced prudential standards for large U.S. bank holding companies and foreign banking organizations. These standards include liquidity, risk management, and capital. The final rule was required

<sup>&</sup>lt;sup>4</sup> For certain mandatorily redeemable financial instruments of certain nonpublic entities, the "Pending Content" in FASB Accounting Standards Codification (ASC) 480-10-25 has been indefinitely deferred. For certain mandatorily redeemable noncontrolling interests that were deferred for both public and nonpublic entities, FASB plans to reconsider implementation issues and, perhaps, classification or measurement guidance for those noncontrolling interests during the deferral period, in conjunction with FASB's ongoing projects. During the deferral period for certain mandatorily redeemable noncontrolling interests, all public entities as well as nonpublic entities that are SEC registrants are required to follow the disclosure requirements in paragraphs 1–3 of FASB ASC 480-10-50 as well as disclosures required by other applicable guidance. For additional effective date information see the transition and open effective date information discussed in FASB ASC 480-10-65-1.

by section 165 of the Dodd-Frank Act. Foreign banking organizations with U.S. non-branch assets of \$50 billion or more will be required to establish a U.S. intermediate holding company over their U.S. subsidiaries. The foreign-owned U.S. intermediate holding company generally will be subject to the same risk-based and leverage capital standards applicable to U.S. bank holding companies. The intermediate holding companies also will be subject to the Federal Reserve's rules requiring regular capital plans and stress tests. Although the final rule became effective June 1, 2014, the initial compliance date for foreign banking organizations is July 1, 2016 and generally defers application of the leverage ratio to foreign-owned U.S. intermediate holding companies until 2018.

## **Disclosures for Banks and Savings Institutions**

17.28 Noncompliance with regulatory capital requirements could materially affect the economic resources of a bank or savings institution and claims to those resources, as stated in FASB ASC 942-505-50-1. Accordingly, at a minimum, the entity should disclose the following in the footnotes to the financial statements:

- a. A description of regulatory capital requirements for both of the following:
  - i. Those for capital adequacy purposes
  - Those established by the prompt corrective action provisions of Section 38 of the Federal Deposit Insurance Act (FDI Act)
- b. The actual or possible material effects of noncompliance with such requirements
- c. Whether the entity is in compliance with the regulatory capital requirements, including, as of each balance sheet date presented, the following with respect to quantitative measures:
  - The entity's required and actual ratios and amounts of tier 1 leverage, tier 1 risk-based, and total risk-based capital, (for savings institutions) tangible capital, and (for certain banks and bank holding companies) tier 3 capital for market risk
  - Factors that may significantly affect capital adequacy such as potentially volatile components of capital, qualitative factors, and regulatory mandates
- d. As of each balance sheet date presented, the prompt corrective action category in which the entity was classified as of its most recent notification
- e. As of the most recent balance sheet date, whether management believes any conditions or events since notification have changed the institution's category

Noncompliance with regulatory capital requirements may, when considered with other factors, raise substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time.

©2016, AICPA AAG-DEP 17.28

- 17.29 As stated in FASB ASC 942-505-50-1C, a bank or savings institution is (under federal regulations) deemed to be within a given capital category as of the most recent date of any of the following:
  - a. The date the institution filed a regulatory financial report
  - b. The date a final regulatory examination report is delivered to the institution
  - c. The date the institution's primary regulator provides written notice of the entity's capital category or that the institution's capital category has changed
- 17.30 In accordance with FASB ASC 942-505-50-1A, disclosures should also be presented for any state-imposed capital requirements that are more stringent than or significantly different from federal requirements.
- 17.31 For "adequately capitalized" or "undercapitalized" institutions, the disclosure in item c in paragraph 17.28 should present the minimum amounts and ratios the institution must have to be categorized as  $adequately\ capitalized$  under the prompt corrective action framework and should include the effect of any supervisory action that has been imposed, as stated in FASB ASC 942-505-50-1B. The amounts disclosed under that paragraph may be presented in either narrative or tabular form. The percentages disclosed should be those applicable to the entity. Also, if the institution has been advised that it must meet capital adequacy levels that exceed the statutory minimums, those higher levels should be disclosed. Such institution-specific requirements also should be the basis for management's assertion in FASB ASC 942-505-50-1(c) (see item c in paragraph 17.28) about whether the institution is in compliance.
- 17.32 Paragraphs 1D–1E of FASB ASC 942-505-50 state that if, as of the most recent balance sheet date presented, the entity is (a) not in compliance with capital adequacy requirements, (b) considered less than adequately capitalized under the prompt corrective action provisions, or (c) both, the possible material effects of such conditions and events on amounts and disclosures in the financial statements should be disclosed.
- 17.33 The institution should consider also making such disclosures when one or more of the institution's actual ratios is nearing noncompliance or when capital adequacy restrictions are imposed by regulation. Capital ratios higher than those in the prompt corrective action provisions may be required by the federal banking agencies, either informally through a board resolution or memorandum of understanding or formally though a consent order or formal agreement. Elevated capital levels may be considered for disclosure based on the nature of the agreement and the potential impact on operations.
- 17.34 If, after considering management's plans, substantial doubt about an entity's ability to continue as a going concern is alleviated as a result of consideration of management's plans, "Pending Content" in FASB ASC 205-40-50-12 states that an entity should disclose in the footnotes information that enables users of the financial statements to understand all of the following (or refer to similar information disclosed elsewhere in the footnotes):
  - Principal conditions or events that raised substantial doubt about the entity's ability to continue as a going concern (before consideration of management's plans)

- Management's evaluation of the significance of those conditions or events in relation to the entity's ability to meet its obligations
- Management's plans that alleviated substantial doubt about the entity's ability to continue as a going concern<sup>5</sup>

17.35 If, after considering management's plans, substantial doubt about an entity's ability to continue as a going concern is not alleviated, "Pending Content" in FASB ASC 205-40-50-13 states that an entity should include a statement in the footnotes indicating that there is substantial doubt about the entity's ability to continue as a going concern within one year after the date that the financial statements are issued. Additionally, the entity should disclose information that enables users of the financial statements to understand all of the following:

- Principal conditions or events that raised substantial doubt about the entity's ability to continue as a going concern
- Management's evaluation of the significance of those conditions or events in relation to the entity's ability to meet its obligations
- Management's plans that are intended to mitigate the conditions or events that raise substantial doubt about the entity's ability to continue as a going concern<sup>6</sup>

**17.36** Additional information that might be disclosed in situations where there is substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time may include the following:

- Pertinent conditions and events giving rise to the assessment of substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time
- Possible effects of such conditions and events
- Management's evaluation of the significance of those conditions and events and any mitigating factors
- Possible discontinuance of operations
- Management's plans (including relevant prospective financial information)
- Information about the recoverability or classification of recorded asset amounts or the amounts or classification of liabilities.

Additional information that might be disclosed in situations where there is substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time may include the following:

 $Readers\ are\ encouraged\ to\ consult\ the\ full\ text\ of\ the\ ASU\ on\ FASB's\ website\ at\ www.fasb.org.$ 

©2016. AICPA

<sup>&</sup>lt;sup>5</sup> FASB Accounting Standards Update (ASU) No. 2014-15, Presentation of Financial Statements—Going Concern (Subtopic 205-40): Disclosure of Uncertainties about an Entity's Ability to Continue as a Going Concern, was issued in August 2015. The amendments are effective for the annual period ending after December 15, 2016, and for annual periods and interim periods thereafter. Early adoption is permitted.

<sup>&</sup>lt;sup>6</sup> See footnote 5.

- Possible effects of such conditions and events giving rise to the assessment of substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time;
- Possible discontinuance of operations; and
- Information about the recoverability or classification of recorded asset amounts or the amounts or classification of liabilities.<sup>7</sup>

17.37 FASB ASC 942-505-50-1F states that other regulatory limitations may exist despite compliance with minimum regulatory capital requirements. To the extent such limitations could materially affect the economic resources of the institution and claims to those resources, they should similarly be disclosed in the footnotes to the financial statements.

## **Disclosure for Holding Companies**

17.38 FASB ASC 942-505-50-1G states that the disclosures required by paragraphs 1–1F of FASB ASC 942-505-50 should be presented for all significant subsidiaries of a holding company. Bank holding companies should also present the disclosures required by paragraphs 1–1F of FASB ASC 942-505-50 as they apply to the holding company, except for the prompt corrective action disclosure required by item (d) in FASB ASC 942-505-50-1. Savings institution holding companies are not subject to regulatory capital requirements separate from those of their subsidiaries. Bank holding companies are not subject to the prompt corrective action provisions of the FDI Act.

17.39 A bank holding company that is a financial holding company (FHC) should disclose the applicable regulatory requirements for maintaining its status as a FHC, including that each of its insured deposit taking subsidiaries must be well capitalized, well managed, and have at least a satisfactory Community Reinvestment Act rating. If the Federal Reserve were to find that any depository institution subsidiary owned or controlled by the bank holding company ceases to be well capitalized or well managed and such noncompliance is not subsequently corrected, the Federal Reserve could require the banking organization to cease its FHC related activities or divest its banking subsidiaries. Paragraph 1.19 of this guide and Section 3901.0.2, "Holding Company Fails to Continue Meeting Financial Holding Company Capital and Management Requirements," of the Federal Reserve's Bank Holding Company Supervision Manual provide additional guidance.

**17.40** If a bank holding company is an FHC, the significant subsidiaries include all U.S. insured deposit taking subsidiaries.

## Illustrative Disclosures for Banks and Savings Institutions (The example disclosures that follow are for illustrative purposes only)

17.41 Well capitalized. Following is an illustrative disclosure for an institution that is in compliance with capital adequacy requirements and considers itself well capitalized under the prompt corrective action framework. Comparative disclosures should be included for each balance sheet presented.

<sup>&</sup>lt;sup>7</sup> See footnote 5.

The Bank is subject to various regulatory capital requirements administered by the federal banking agencies. Failure to meet minimum capital requirements can initiate certain mandatory—and possibly additional discretionary—actions by regulators that, if undertaken, could have a direct material effect on the Bank's financial statements. Under capital adequacy guidelines and the regulatory framework for prompt corrective action, the Bank must meet specific capital guidelines that involve quantitative measures of the Bank's assets, liabilities, and certain off-balance-sheet items as calculated under U.S. GAAP, regulatory reporting requirements, and regulatory capital standards. The Bank's capital amounts and classification are also subject to qualitative judgments by the regulators about components, risk weightings, and other factors.

Quantitative measures established by regulatory capital standards to ensure capital adequacy require the Bank to maintain minimum amounts and ratios (set forth in the following table) of total, Tier 1 and common equity Tier 1 capital (as defined) to risk-weighted assets (as defined) and of Tier 1 capital (as defined) to average assets (as defined). Management believes, as of December 31, 200X, that the Bank meets all capital adequacy requirements to which it is subject.

As of December 31, 200X, and December 31, 200W, the most recent notification from [institution's primary regulator] categorized the Bank as [well capitalized] under the regulatory framework for prompt corrective action. To be categorized as [well capitalized] the Bank must maintain minimum total risk-based, Tier 1 risk-based, common equity Tier 1 risk-based capital, and Tier 1 leverage ratios as set forth in the table. There are no conditions or events since that notification that management believes have changed the institution's category.

**17.42** Adequately capitalized. Following is an illustrative paragraph to be added to the disclosures illustrated in paragraph 17.41 when an institution considers itself adequately capitalized:

Under the framework, the Bank's capital levels do not allow the Bank to accept brokered deposits without prior approval from regulators [describe the possible effects of this restriction].

17.43 *Undercapitalized.* Following are illustrative paragraphs to be added to the disclosures illustrated in paragraphs 17.41–.42 when an institution considers itself *undercapitalized*, *significantly undercapitalized*, or *critically undercapitalized*. For a discussion about the auditor's consideration of noncompliance, see discussion beginning in paragraph 5.212 of this guide.

The Bank may not issue dividends or make other capital distributions, and may not accept brokered or high rate deposits, as defined, due to the level of its risk-based capital. [Describe the possible effects of these restrictions.]

©2016, AICPA AAG-DEP 17.43

<sup>&</sup>lt;sup>8</sup> See paragraph 17.31.

<sup>&</sup>lt;sup>9</sup> Paragraphs 1.81–.91 of this guide describe the prompt corrective action ratios. For some institutions, the calculation of required amounts and ratios under the prompt corrective action framework may differ from calculations under the capital adequacy requirements. The disclosure should provide the relevant amounts and ratios accordingly.

Under the regulatory framework for prompt corrective action, the Bank's capital status may preclude the Bank from access to borrowings from the Federal Reserve System through the discount window. [Describe the possible effects of these restrictions.] Also, as required by the framework, the Bank has a capital plan that has been filed with and accepted by the FDIC. The plan outlines the Bank's steps for attaining the required levels of regulatory capital. Management believes, at this time, that the Bank will meet all the provisions of the capital plan and all the regulatory capital requirements by December 31, 200Y (or earlier if stated in the capital plan). [The disclosure should continue with discussion of management plans such as reducing the size of the institution by converting noncash assets and reducing liabilities, issuing additional equity securities, or other plans for financial restructuring.]

17.44 Advanced approaches banks and savings institutions. Following is an illustrative table for presentation in financial statements for an advanced approaches bank or savings institution's actual capital amounts and ratios as of the balance sheet date. All disclosures required by paragraphs 1–1F of FASB ASC 942-505-50 should be presented.

	Actual		For Capital Adequacy Purposes <sup>1, 2</sup>		To Be Well Capitalized Under Prompt Corrective Action Provisions	
	Amount	Ratio	Amount	Ratio	Amount	Ratio
Total Capital (to Risk Weighted Assets)	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	8.0%	\$X,XXX,XXX	10.0%
Tier 1 Capital (to Risk Weighted Assets)	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	6.0%	\$X,XXX,XXX	8.0%
Common Equity Tier 1 Capital (to Risk Weighted Assets)	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	4.5%	\$X,XXX,XXX	6.5%
Tier 1 Capital (to Average Assets)	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	4.0%	\$X,XXX,XXX	5.0%
Tier 1 Capital (to Total Leverage Exposure)	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	3	3	3

See paragraph 17.31.

17.45 Banks and savings institutions not subject to advanced approaches regulations. Following is an illustrative table for presentation in financial statements for a bank or saving institution's actual capital amounts and ratios as of the balance sheet date. All disclosures required by paragraphs 1–1F of FASB ASC 942-505-50 should be presented.

T- D- W-11

For adequately capitalized or undercapitalized institutions, this column should present the minimum amounts and ratios the institution must have to be categorized as adequately capitalized under the prompt corrective action framework and should include the effect of any prompt corrective action capital directive.

<sup>&</sup>lt;sup>3</sup> Advanced approaches banking organizations are required to calculate and report their minimum supplementary leverage ratio as of January 1, 2015, but they do no need to comply with the minimum capital adequacy requirements until January 1, 2018.

To Re Well

	Actual		For Capital Adequacy Purposes <sup>1, 2</sup>		Capitalized Under Prompt Corrective Action Provisions	
	Amount	Ratio	Amount	Ratio	Amount	Ratio
Total Capital (to Risk Weighted Assets)	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	8.0%	\$X,XXX,XXX	10.0%
Tier 1 Capital (to Risk Weighted Assets)	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	4.0%	\$X,XXX,XXX	6.0%
Common Equity Tier 1 Capital (to Risk Weighted Assets)	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	4.5%	\$X,XXX,XXX	6.5%
Tier 1 Capital (to Average Assets)	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	4.0%	\$X,XXX,XXX	5.0%

See paragraph 17.31.

17.46 Holding companies. Following is an illustrative table for presentation in consolidated financial statements for a bank (or savings and loan association) holding company and each significant subsidiary as of the balance sheet date. Tier 3 capital market risk requirements are required to be disclosed only for certain banks and bank holding companies. All disclosures required by paragraphs 1–1F of FASB ASC 942-505-50 should be presented except item (d) disclosures in FASB ASC 942-505-50-1 related to prompt corrective action.

	Actual		For Capital Adequacy Purposes 1, 2		To Be Well Capitalized Under Prompt Corrective Action Provisions	
	Amount	Ratio	Amount	Ratio	Amount	Ratio
Total Capital (to Risk Weigh	hted Assets):					
Consolidated	\$X,XXX,XXX	<b>X.X</b> %	\$X,XXX,XXX	<b>X.X</b> %	N/A	
Subsidiary Bank A	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%
Subsidiary Bank B	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%
Subsidiary Bank C	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%
Tier 1 Capital (to Risk Weig	hted Assets):					
Consolidated	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	N/A	
Subsidiary Bank A	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%
Subsidiary Bank B	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%
Subsidiary Bank C	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%
					(conti	inued)

©2016, AICPA AAG-DEP 17.46

For adequately capitalized or undercapitalized institutions, this column should present the minimum amounts and ratios the institution must have to be categorized as adequately capitalized under the prompt corrective action framework and should include the effect of any prompt corrective action capital directive.

<sup>&</sup>lt;sup>10</sup> Bank holding companies subject to the advanced approaches rule are required to calculate and report their minimum supplementary leverage ratio, but they do not need to comply with the requirement until January 1, 2018 (see Update 17-3).

	Actual		For Capital Ad		Capitalized V Prompt Corr Action Provi	Inder ective
	Amount	Ratio	Amount	Ratio	Amount	Ratio
Common Equity Tier 1 Capit	al (to Risk Weig	hted Ass	sets):			
Consolidated	\$X,XXX,XXX	<b>X.X</b> %	\$X,XXX,XXX	<b>X.X</b> %	N/A	
Subsidiary Bank A	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%
Subsidiary Bank B	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%
Subsidiary Bank C	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%
Tier 1 Capital (to Average As	sets):					
Consolidated	\$X,XXX,XXX	<b>X.X</b> %	\$X,XXX,XXX	<b>X.X</b> %	N/A	
Subsidiary Bank A	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%
Subsidiary Bank B	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%
Subsidiary Bank C	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	X.X%

<sup>&</sup>lt;sup>1</sup> See paragraph 17.31.

#### © Update 17-3 Regulatory: Regulatory Capital Rules

The regulatory final rule Regulatory Capital Rules: Regulatory Capital, Enhanced Supplementary Leverage Ratio Standards for Certain Bank Holding Companies and Their Subsidiary Insured Depository Institutions, issued by the Federal Reserve, the OCC, and the FDIC in April 2014, will become effective on January 1, 2018. Banking organizations subject to the requirements were required to calculate and publicly disclose the ratio beginning January 1, 2015.

The final regulatory rule strengthens the agencies' supplementary leverage ratio standards for large, interconnected U.S. banking organizations and is applicable to any U.S. top-tier bank holding company with more than \$700 billion in total consolidated assets or more than \$10 trillion in assets under custody and any insured depository institution subsidiary of these bank holding companies. The final rule establishes enhanced supplementary leverage ratio standards for covered bank holding companies and their subsidiary insured depository institutions. Among other provisions of the final rule, an insured depository institution that is a subsidiary of a covered Bank Holding Company (BHC) must maintain a supplementary leverage ratio of at least 6 percent to be well capitalized under the agencies' prompt corrective action framework.

Readers are encouraged to consult the full text of this final rule at any of the respective agencies' websites.

To Do Wall

For adequately capitalized or undercapitalized institutions, this column should present the minimum amounts and ratios the institution must have to be categorized as adequately capitalized under the prompt corrective action framework and should include the effect of any prompt corrective action capital directive.

#### Credit Unions

#### © Update 17-4 Regulatory: Risk-Based Capital

The National Credit Union Administration (NCUA) regulatory final rule *Risk-Based Capital*, issued in October 2015, is effective on January 1, 2019 for federally insured, natural person credit unions with assets over \$100 million.

The final regulation restructures the NCUA's prompt corrective action regulations and make various revisions, including amending the agency's current risk-based net worth requirement by replacing it with a new risk-based capital ratio for federally insured natural person credit unions. The risk-based capital requirement set forth in the final rule is more consistent with the NCUA's risk-based capital measure for corporate credit unions and, as the law requires, more comparable to the regulatory risk-based capital measures used by the FDIC, the Federal Reserve, and the OCC.

The final rule also eliminates several provisions in the NCUA's current prompt corrective action regulations, including provisions relating to regular reserve accounts, risk-mitigation credits, and alternative risk-weights.

This edition of the guide has not been updated to reflect changes as a result of this final ruling, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this regulation on NCUA's website at www.ncua.gov.

#### Introduction

17.47 Credit unions operate under a cooperative form of ownership. Members, in effect, "own" the credit union, although their interests in the credit union (that is, their shares) have the characteristics of deposits. Although the equity section of a credit union's statement of financial condition generally consists of retained earnings and accumulated other comprehensive income, GAAP requires other items to be classified as equity. As GAAP evolves, other items may also be included in members' equity. Retained earnings includes statutory reserves, undivided earnings, and other appropriations as designated by management or regulatory authorities. Although credit unions may be incorporated, no stock is issued. Retained earnings is generally shown as a single line item in the statement of financial condition. The components of retained earnings may be presented in the body of the statement of financial condition, the notes to the financial statements, or the statement of retained earnings. All appropriations and other restrictions of retained earnings should be disclosed.

## Members' Equity

17.48 Regular reserve (statutory reserve). The Federal Credit Union Act and certain states require that a regular (or statutory) reserve be established and maintained to provide a restriction on undivided earnings. At one time, the regular (or statutory) reserve was used to record charged-off loans and its funding was based on a periodic (typically quarterly) charge to undivided earnings and a credit to the reserve account. For federal credit unions, the amount currently required to be transferred, if any, is defined in Sections 702.201 and 702.303 of the NCUA regulations. The prompt corrective action rules describe the mandatory and discretionary prompt corrective actions that a credit union is subject to in cases where the credit union's capital level is

below the *well capitalized* level, or the credit union fails to meet its required risk-based net worth requirement (RBNWR), or the credit union is subject to regulatory restrictions because of activities that are judged by the NCUA to be unsafe and unsound. In cases relating to inadequate capital with respect to the net worth requirement or the RBNWR, a credit union is generally required to increase its net worth by the equivalent of at least 0.1 percent of assets each quarter until the credit union is classified as *well capitalized*.

- **17.49** Certain states may have adopted similar regulations that apply to state chartered credit unions. The statutes for each state should be consulted for applicable requirements.
- 17.50 *Undivided earnings*. Undivided earnings represent unappropriated accumulated earnings or losses of the credit union since its inception. The undivided earnings may also be increased or decreased as a result of transfers to or from appropriated accounts such as the regular reserve or appropriated undivided earnings.
- 17.51 Appropriated undivided earnings. The board of directors of a credit union may restrict or appropriate portions of undivided earnings for specific purposes in accordance with paragraphs 3–4 of FASB ASC 505-10-45. Examples may include appropriations for loss contingencies and for major expenditures. The amount of such appropriations is normally transferred from undivided earnings, pending resolution of its purpose. Amounts appropriated may be returned to undivided earnings when they are no longer deemed necessary.
- 17.52 Federally insured state chartered credit unions are required under terms of the insurance agreement to establish an investment valuation reserve, displayed as an equity classification, for held-to-maturity nonconforming investments. Nonconforming investments are those investments permissible under state law for a state chartered credit union, but which are impermissible for federally chartered credit unions.
- **17.53** Other components of equity. GAAP provides guidance for other items that should be classified as equity (for example, equity acquired in a combination arising from a mutual to mutual acquisition in which no purchase price is paid).
- 17.54 In accordance with FASB ASC 220-10-45-14, the total of other comprehensive income for a period should be transferred to a component of equity that is presented separately from retained earnings and additional paid-in capital in a statement of financial position at the end of an accounting period. See FASB ASC 220-10-45-10A and paragraph 17.09 for examples of items required to be reported as other comprehensive income.

## New Credit Unions and Low Income Designated Credit Unions

17.55 The prompt corrective action regulations for credit unions designated as new are different than for other natural person credit unions. According to regulations, to be designated as new a credit union must have been in existence for less than 10 years and have \$10 million or less in total assets. For credit unions designated as  $low\ income$  by the NCUA, the net worth calculation includes certain uninsured, secondary capital accounts (as defined in the regulations). low

<sup>&</sup>lt;sup>11</sup> Prompt corrective action regulations can be found at 12 CFR 702.301-.307. Additional regulations for "low-income" credit unions can be found at 12 CFR 701.34.

#### **Disclosures for Natural Person Credit Unions**

17.56 FASB ASC 942-505-50-1H states noncompliance with regulatory capital requirements could materially affect the economic resources of a credit union and claims to those resources. Accordingly, at a minimum, a credit union within the scope of FASB ASC 942-10-15-2 should disclose all of the following in the footnotes to the financial statements:

- a. A description of regulatory capital requirements (a) for capital adequacy purposes and (b) for prompt corrective action
- b. The actual or possible material effects of noncompliance with such requirements
- c. Whether the entity is in compliance with the regulatory capital requirements, including, as of each balance sheet date presented, the following with respect to quantitative measures:
  - i. Whether the institution meets the definition of a complex credit union as defined by the NCUA
  - ii. The institution's required and actual capital ratios and required and actual capital amounts
  - iii. Factors that may significantly affect capital adequacy such as potentially volatile components of capital, qualitative factors, and regulatory mandates
- d. As of each balance sheet date presented, the prompt corrective action category in which the institution was classified
- e. If, as of the most recent balance sheet date or date financial statements are issued or are available to be issued (as discussed in FASB ASC 855-10-25), the institution is not in compliance with capital adequacy requirements, the possible material effects of such conditions on amounts and disclosures in the financial statements
- f. Whether subsequent to the balance sheet date and before the financial statements are issued or are available to be issued (as discussed in FASB ASC 855-10-25), management believes any events or changes have occurred to change the institution's prompt corrective action category

Noncompliance with regulatory capital requirements may, when considered with other factors, raise substantial doubt about a credit union's ability to continue as a going concern for a reasonable period of time.

- 17.57 The institution should consider also making such disclosures, as discussed in the preceding paragraph, when the institution's actual ratio is nearing noncompliance.
- 17.58 Additional information that might be disclosed in situations where there is substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time may include the following:
  - Pertinent conditions and events giving rise to the assessment of substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time
  - Possible effects of such conditions and events
  - Management's evaluation of the significance of those conditions and events and any mitigating factors

©2016. AICPA AAG-DEP 17.58

- Possible discontinuance of operations
- Management's plans (including relevant prospective financial information)
- Information about the recoverability or classification of recorded asset amounts or the amounts or classification of liabilities

See paragraphs 17.34—.35 for required going concern disclosure guidance. Additional information that might be disclosed in situations where there is substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time may include the following:

- Possible effects of such conditions and events giving rise to the assessment of substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time
- Possible discontinuance of operations
- Information about the recoverability or classification of recorded asset amounts or the amounts or classification of liabilities.<sup>12</sup>

17.59 Disclosures should also be presented for any state-imposed capital requirements that are more stringent than or significantly different from federal requirements, in accordance with FASB ASC 942-505-50-2.

17.60 The NCUA board adopted prompt corrective action rules in response to the Credit Union Membership Access Act requirement that the NCUA adopt a system to restore the net worth of inadequately capitalized federally insured credit unions. In conjunction with the adopted Prompt Corrective Action Rule, the NCUA board also issued a rule, which defines a *complex* credit union and establishes RBNWRs. Readers should refer to the NCUA regulations for the risk-based net worth and prompt corrective action requirements.

#### Illustrative Disclosures for Natural Person Credit Unions

17.61 Well capitalized. The example disclosures that follow are for illustrative purposes only. Following is an illustrative disclosure for an institution that is in compliance with capital adequacy requirements and considers itself well capitalized under the prompt corrective action framework. Comparative disclosures should be included for each balance sheet presented.

The Credit Union is subject to various regulatory capital requirements administered by the NCUA. Failure to meet minimum capital requirements can initiate certain mandatory—and possibly additional discretionary—actions by regulators that, if undertaken, could have a direct material effect on the Credit Union's financial statements. Under capital adequacy guidelines and the regulatory framework for prompt corrective action, the Credit Union must meet specific capital guidelines that involve quantitative measures of the Credit Union's assets, liabilities, and certain off-balance-sheet items as calculated under U.S. GAAP, regulatory reporting requirements, and regulatory capital standards. The Credit Union's capital amounts and net worth classification are also subject to qualitative judgments by the regulators about components, risk weightings, and other factors.

Quantitative measures established by regulatory capital standards to ensure capital adequacy require the Credit Union to maintain minimum amounts and ratios (set forth in the following table) of net worth (as defined) to total assets (as defined). Credit unions are also required to calculate a RBNWR which establishes whether or not the Credit Union will be considered "complex" under the regulatory framework. The Credit Union's RBNW ratio as of December 31, 200X was—percent. The minimum ratio to be considered complex under the regulatory framework is 6 percent. Management believes, as of December 31, 200X, that the Credit Union meets all capital adequacy requirements to which it is subject.

As of December 31, 200X, the most recent call reporting period, the NCUA categorized the Credit Union as *well capitalized* under the regulatory framework for prompt corrective action. To be categorized as *well capitalized* the Credit Union must maintain a minimum net worth ratio of 7 percent of assets.<sup>13</sup> There are no conditions or events since that notification that management believes have changed the institution's category.

The Credit Union's actual capital amounts and ratios are also presented in the table.

To Po Adoquately

To Do Wall

	Actual		Capitalized Under Prompt Corrective Action Provisions <sup>1</sup>		Capitalized Under Prompt Corrective Action Provisions	
	Amount	Ratio	Amount	Ratio	Amount	Ratio
Net worth	\$2,000,000	7.5%	\$1,600,000	6.0%	\$1,800,000	7.0%
Risk-Based Net Worth Requirement	\$1,700,000	6.5%	N/A	N/A	N/A	N/A

For adequately capitalized or undercapitalized institutions, this column should present the minimum amounts and ratios the institution must have to be categorized as adequately capitalized under the prompt corrective action framework and should include the effect of any mandatory or discretionary supervisory actions.

Because the RBNWR, 6.5 percent, is less than the net worth ratio, 7.5 percent, the Credit Union retains its original category. Further, in performing its calculation of total assets, the Credit Union used the [select one: average of the quarter-end balances of the four most recent quarters, monthly average over the quarter, daily average over the quarter, or quarter-end balance] option, as permitted by regulation.

17.62 Adequately capitalized. Following is an illustrative paragraph to be added in place of the third illustrative paragraph in paragraph 17.61 for an institution that is in compliance with capital adequacy requirements and considers itself adequately capitalized under the prompt corrective action framework:

As of December 31, 200X, and December 31, 200W, the most recent call reporting period, the NCUA categorized the Credit Union

<sup>&</sup>lt;sup>13</sup> Paragraphs 2.28-.42 of this guide describe the prompt corrective action net worth ratios for natural person credit unions. For some institutions, the calculation of required amounts and net worth ratios under the prompt corrective action framework may differ from calculations under the capital adequacy requirements. The disclosure should provide the relevant amounts and ratios accordingly.

as adequately capitalized under the regulatory framework for prompt corrective action. To be categorized as adequately capitalized the Credit Union must maintain a minimum net worth ratio of 6 percent of assets and, if applicable, must maintain adequate net worth to meet the Credit Union's RBNWR of X percent as set forth in the table. <sup>14</sup> As an adequately capitalized credit union, the NCUA's prompt corrective action regulations require that the Credit Union increase its net worth quarterly by an amount equivalent to at least 0.1 percent of its total assets for the current quarter, and must transfer that amount (or more by choice) from undivided earnings to its regular reserve account until it is well capitalized, while continuing to meet its RBNWR. There are no conditions or events since that filing date that management believes have changed the institution's category.

17.63 Undercapitalized. Following are illustrative paragraphs to be added to the disclosures illustrated in paragraph 17.61 when a credit union considers itself undercapitalized, significantly undercapitalized, or critically undercapitalized [for existing credit unions].

The Credit Union may not increase assets and must restrict member business loans due to its net worth. [Describe the possible effects of these restrictions.] Under the regulatory framework for prompt corrective action, the Credit Union has a net worth restoration plan that has been filed with and accepted by the NCUA. The plan outlines the Credit Union's steps for attaining the required levels of net worth. Management believes, at this time, that the Credit Union will implement the steps and meet all the provisions of the plan and all the regulatory net worth requirements by December 31, 200Y (or earlier if stated in the restoration plan). [The disclosure should continue with discussion of any discretionary actions required by the NCUA.]

17.64 New credit unions. Following are illustrative paragraphs to be added to the disclosures illustrated in paragraph 17.61 when a new credit union considers itself moderately capitalized, marginally capitalized, minimally capitalized, or uncapitalized.

The Credit Union must restrict member business loans due to its net worth. [Describe the possible effects of these restrictions.] Under the regulatory framework for prompt corrective action, a revised business plan has been filed, as required, with and accepted by the NCUA. The plan outlines the Credit Union's steps for attaining the required levels of net worth. Management believes, at this time, that the Credit Union will implement the steps and meet all the provisions of the plan and all the regulatory net worth requirements by December 31, 200Y (or earlier if stated in the revised business plan). [The disclosure should continue with discussion of any discretionary actions required by the NCUA.]

<sup>&</sup>lt;sup>14</sup> For some institutions, the calculation of required amounts and net worth ratios under the prompt corrective action framework may differ from calculations under the capital adequacy requirements. The disclosure should provide the relevant amounts and ratios accordingly.

## **Corporate Credit Unions**

#### Introduction

17.65 Corporate credit unions operate under a cooperative form of ownership similar to natural person credit unions. Corporate credit unions are established to serve the financial needs of natural person credit unions that join the corporate. Although, the equity section of a corporate credit union's statement of financial condition generally consists of paid-in capital, retained earnings and accumulated other comprehensive income, GAAP requires other items to be classified as equity. As GAAP evolves, other items may also be included in members' equity. Retained earnings include all forms of retained earnings such as regular or statutory reserves and undivided earnings. Although some corporate credit unions may be incorporated under state laws, no stock is issued.

## Equity

17.66 Nonperpetual capital. <sup>16</sup> Corporate credit unions are different from natural person credit unions in that they have specific nonperpetual capital accounts (NCAs). NCAs, as defined in 12 CFR 704.2, are funds contributed by members or nonmembers that (a) are term certificates with an original minimum term of five years or that have an indefinite term (that is, no maturity) with a minimum withdrawal notice of five years, (b) are available to cover losses that exceed retained earnings and perpetual contributed capital (PCC), (c) are not insured by the National Credit Union Share Insurance Fund (NCUSIF) or other share or deposit insurers, and (d) cannot be pledged against borrowings. In the event the corporate credit union is liquidated, the holders of NCAs will claim equally. These claims will be subordinate to all other claims (including NCUSIF claims), except that any claims by the holders of PCC will be subordinate to the claims of holders of NCAs.

17.67 *PCC.* PCC, as defined in 12 CFR 704.2, means accounts or other interests of a corporate credit union that are perpetual, noncumulative dividend accounts; are available to cover losses that exceed retained earnings; are not insured by the NCUSIF or other share or deposit insurers; and cannot be pledged against borrowings. In the event the corporate is liquidated, any claims made by the holders of PCC will be subordinate to all other claims (including NCUSIF claims). These funds are callable only if the corporate credit union meets its minimum required capital and net economic value ratios after the funds are called and only with the prior approval of the NCUA and, for state chartered corporate credit unions, the applicable state regulator. They are callable on a pro-rata basis across an issuance class. <sup>17</sup> A corporate credit union may issue PCC to both members and nonmembers. <sup>18,19</sup>

©2016, AICPA

<sup>&</sup>lt;sup>15</sup> See paragraph 17.52.

<sup>&</sup>lt;sup>16</sup> 12 CFR 704.3(f) addresses membership capital accounts that qualified as corporate capital prior to October 21, 2011, but which no longer satisfy the definitions of capital because the accounts were not converted by the member to nonperpetual capital accounts or perpetual contributed capital.

<sup>&</sup>lt;sup>17</sup> See 12 CFR 704.3(c)(3).

<sup>&</sup>lt;sup>18</sup> See 12 CFR 704.3(c)(4).

 $<sup>^{19}</sup>$  Auditors may need to be familiar with Part 704, "Corporate Credit Unions" of the National Credit Union Administration (NCUA) regulations to help assess the propriety and adequacy of financial statement disclosures.

17.68 Reserves and undivided earnings. Reserves and undivided earnings represent unappropriated accumulated earnings or losses of the corporate credit union since its inception. The accounting treatment of transactions in undivided earnings of a credit union is similar to that of transactions in retained earnings of corporate entities. Corporate credit unions must maintain at all times (a) a leverage ratio<sup>20</sup> of 4.0 percent or greater; (b) a tier 1 risk-based capital ratio<sup>21</sup> of 4.0 percent or greater; and (c) a total risk-based capital ratio<sup>22</sup> of 8.0 percent or greater. To ensure it meets its capital requirements, a corporate credit union must develop and ensure implementation of written short and long term capital goals, objectives, and strategies that provide for the building of capital consistent with regulatory requirements, the maintenance of sufficient capital to support the risk exposures that may arise from current and projected activities, and the periodic review and reassessment of the capital position of the corporate credit union.<sup>23</sup>

17.69 When significant circumstances or events warrant, 12 CFR 704.3(d) notes that the NCUA may establish increased minimum capital requirements, including modification of the minimum capital requirements related to being either significantly and critically undercapitalized for purposes of 12 CFR 704.4, upon a determination that the corporate credit union's capital is or may become inadequate in view of the credit union's circumstances. Examples of when higher capital levels may be appropriate can be found in 12 CFR 704.3(d)(2).

17.70 Other components of equity. GAAP provides guidance for other items that should be classified as equity.

17.71 In accordance with FASB ASC 220 10-45-14, the total of other comprehensive income for a period should be transferred to a component of equity that is presented separately from retained earnings and additional paidin capital in a statement of financial position at the end of an accounting period. See FASB ASC 220-10-45-10A and paragraph 17.09 for examples of items required to be reported as other comprehensive income.

## **Disclosures for Corporate Credit Unions**

17.72 FASB ASC 942-505-50-1H states that noncompliance with regulatory capital requirements could materially affect the economic resources of a credit union and claims to those resources. Accordingly, at a minimum, a corporate credit union within the scope of FASB ASC 942-10-15-2 should disclose all of the following in the notes to the financial statements:

- a. A description of regulatory capital requirements (a) for capital adequacy purposes and (b) for prompt corrective action
- b. The actual or possible material effects of noncompliance with such requirements

 $<sup>^{20}</sup>$  Leverage ratio, as defined in 12 CFR 704.2, means the ratio of tier 1 capital to moving daily average net assets. The moving daily average net assets, as defined in 12 CFR 704.2, means the average of daily average net assets for the month being measured and the previous 11 months.

<sup>&</sup>lt;sup>21</sup> Tier 1 risk-based capital ratio, as defined in 12 CFR 704.2, means the ratio of tier 1 capital to the moving monthly average net risk-weighted assets. The moving monthly average net risk-weighted assets, as defined in 12 CFR 704.2, means the average of the net risk-weighted assets for the month being measured and the previous 11 months. Measurements must be taken on the last day of each month.

 $<sup>^{22}</sup>$  Total risk-based capital ratio, as defined in 12 CFR 704.2, means the ratio of total capital to moving monthly average net risk-weighted assets.

<sup>&</sup>lt;sup>23</sup> See 12 CFR 704.3(a).

- c. Whether the entity is in compliance with the regulatory capital requirements, including, as of each balance sheet date presented, all of the following with respect to quantitative measures:
  - i. Whether the institutions meets the definition of a complex credit union as defined by the NCUA
  - ii. The institution's required and actual capital ratios and required and actual capital amounts
  - iii. Factors that may significantly affect capital adequacy such as potentially volatile components of capital, qualitative factors, and regulatory mandates
- d. As of each balance sheet date presented, the prompt corrective action category in which the institution was classified
- e. If, as of the most recent balance sheet date or date financial statements are issued or are available to be issued (as discussed in FASB ASC 855-10-25), the institution is not in compliance with capital adequacy requirements, the possible material effects of such conditions on amounts and disclosures in the financial statements
- f. Whether subsequent to the balance sheet date and prior financial statements are issued or are available to be issued (as discussed in FASB ASC 855-10-25), management believes any events or changes have occurred to change the institution's prompt corrective action category
- 17.73 FASB ASC 942-505-50-1H also states that noncompliance with regulatory capital requirements may, when considered with other factors, raise substantial doubt about the credit union's ability to continue as a going concern for a reasonable period of time.
- FASB ASC 942-505-50-1H also states that noncompliance with regulatory capital requirements may, when considered with other factors, raise substantial doubt about the credit union's ability to continue as a going concern for a reasonable period of time. See paragraphs 17.34–.35 for required going concern disclosure guidance.<sup>24</sup>
- 17.74 Disclosures should also be presented for any state-imposed capital requirements that are more stringent than or significantly differ from federal requirements, in accordance with FASB ASC 942-505-50-2.
- 17.75 Additional information that might be disclosed in situations where there is substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time may include the following:
  - Pertinent conditions and events giving rise to the assessment of substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time
  - Possible effects of such conditions and events
  - Management's evaluation of the significance of those conditions and events and any mitigating factors
  - Possible discontinuance of operations

<sup>&</sup>lt;sup>24</sup> See footnote 5.

- Management's plans (including relevant prospective financial information)
- Information about the recoverability or classification of recorded asset amounts or the amounts or classification of liabilities

Additional information that might be disclosed in situations where there is substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time may include the following:

- Possible effects of such conditions and events giving rise to the assessment of substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time
- Possible discontinuance of operations
- Information about the recoverability or classification of recorded asset amounts or the amounts or classification of liabilities.<sup>25</sup>

17.76 Effective October 20, 2011, the NCUA adopted prompt corrective action requirements for corporate credit unions. The purpose of 12 CFR 704.4 is to define, for corporate credit unions that are not adequately capitalized, the capital measures and capital levels that are used for determining appropriate supervisory actions. It also establishes procedures for the submission and review of capital restoration plans and the issuance and review of capital directives, orders, and other supervisory directives.

## **Illustrative Disclosures for Corporate Credit Unions**

17.77 Well capitalized. The example disclosures that follow are for illustrative purposes only. Following is an illustrative disclosure for a corporate credit union that is in compliance with capital adequacy requirements and is considered well capitalized under the prompt corrective action framework. Comparative disclosures should be included for each balance sheet presented.

The Corporate Credit Union is subject to various regulatory capital requirements administered by the NCUA. Failure to meet minimum capital requirements can initiate certain mandatory—and possibly additional discretionary—actions by regulators that, if undertaken, could have a direct material effect on the Corporate Credit Union's financial statements. Under capital adequacy guidelines and the regulatory framework for prompt corrective action, the Corporate Credit Union must meet specific capital guidelines that involve quantitative measures of the Corporate Credit Union's assets, liabilities, and certain off-balance-sheet items as calculated under regulatory reporting requirements. The Credit Union's capital amounts and classification are also subject to qualitative judgments by the regulators about components, risk weightings, and other factors.

Quantitative measures established by regulatory capital standards to ensure capital adequacy require the Corporate Credit Union to maintain minimum amounts and ratios (set forth in the following table) of total and Tier 1 capital (as defined) to moving monthly average net risk-weighted assets (as defined) and total capital to moving daily average net assets (as defined). Management believes, as of December

31, 200X, that the Corporate Credit Union meets all capital adequacy requirements to which it is subject.

As of December 31, 200X, the most recent call reporting period, the NCUA categorized the Corporate Credit Union as [well capitalized] under the regulatory framework for prompt corrective action. To be categorized as [well capitalized] the Corporate Credit Union must maintain a minimum total risk-based, Tier 1 risk-based, and leverage ratios as set forth in the table. <sup>26</sup> There are no conditions or events since that notification that management believes have changed the Corporate Credit Union's category.

The Corporate Credit Union's actual capital amounts and ratios are also presented in the table.

	Actual		To Be Adequately Capitalized Under Prompt Corrective Action Provisions <sup>1</sup>		To Be Well Capitalized Under Prompt Corrective Action Provisions	
	Amount	Ratio	Amount	Ratio	Amount	Ratio
Total Capital (to Moving Monthly Average Net Risk-Weighted Assets) Tier 1 Risk-Based Capital (to Moving Monthly	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	8.0%	\$X,XXX,XXX	10.0%
Average Net Risk-Weighted Assets) Total Capital (to Moving	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	4.0%	\$X,XXX,XXX	6.0%
Daily Average Net Assets)	\$X,XXX,XXX	X.X%	\$X,XXX,XXX	4.0%	\$X,XXX,XXX	5.0%

For adequately capitalized or undercapitalized institutions, this column should present the minimum amounts and ratios the institution must have to be categorized as adequately capitalized under the prompt corrective action framework and should include the effect of any mandatory or discretionary supervisory actions.

17.78 Adequately capitalized. The following is an illustrative paragraph to be added in place of the third illustrative paragraph in paragraph 17.77 for a corporate credit union that is in compliance with capital adequacy requirements and is considered adequately capitalized under the prompt corrective action framework.

As of December 31, 200X, the most recent call reporting period, the NCUA categorized the Corporate Credit Union as [adequately capitalized] under the regulatory framework for prompt corrective action. To be categorized as [adequately capitalized] the Corporate Credit Union must maintain a minimum total risk-based, Tier 1 risk-based, and leverage ratios as set forth in the table.<sup>27</sup> There are no conditions or events since that notification that management believes have changed the Corporate Credit Union's category.

17.79 Undercapitalized. The following is an illustrative paragraph to be added to the disclosures illustrated in paragraph 17.77 when a corporate credit union is considered undercapitalized, significantly undercapitalized, or critically undercapitalized.

 $<sup>^{26}</sup>$  Paragraphs 2.49–.50 of this guide describe the prompt corrective action ratios for corporate credit unions.

<sup>&</sup>lt;sup>27</sup> See footnote 26.

The Corporate Credit Union may not increase its daily average net assets during any calendar month to exceed its moving daily average net assets. In addition, the Corporate Credit Union is prohibited from making any capital distribution, including payment of dividends on perpetual and nonperpetual capital accounts while the Corporate Credit Union is undercapitalized. [Describe the possible effects of these restrictions.] Also, as required by the framework, the Corporate Credit Union has a capital restoration plan that has been filed and accepted by the NCUA. The plan outlines the Corporate Credit Union's steps for attaining the required levels of regulatory capital. Management believes, at this time, that the Corporate Credit Union will meet all the provisions of the capital plan and all the regulatory capital requirements by December 31, 200Y (or earlier if stated in the capital plan) [The disclosure should continue with discussion of any discretionary actions required by the NCUA.]

## Mortgage Companies and Mortgage Banking Activities

#### Introduction

17.80 Mortgage companies are organized with capital stock and shareholders. Mortgage banking activities primarily consist of two separate but interrelated activities: (1) the origination or acquisition of mortgage loans for the purpose of selling those loans to permanent investors in the secondary market, and (2) the subsequent long term servicing of those loans. Mortgage loans are acquired for sale to permanent investors from a variety of sources, including in-house origination and purchases from third party correspondents. Certain common requirements are discussed in chapter 4 of this guide. For example, to participate in the Federal Housing Administration mortgage insurance program, a mortgage lender must obtain, U.S. Department of Housing and Urban Development (HUD) approval by meeting various requirements prescribed by HUD, including maintaining minimum net worth requirements. Net worth requirements vary depending on the program. To obtain approval to sell and service mortgage loans for Federal National Mortgage Association (Fannie Mae) or Federal Home Loan Mortgage Corporation (Freddie Mac), a mortgage lender must meet various federal requirements including maintaining an acceptable net worth.

## Disclosure for Mortgage Companies and Mortgage Banking Activities

17.81 FASB ASC 948-10-50-3 states that noncompliance with minimum net worth (capital) requirements imposed by secondary market investors or state imposed regulatory mandates could materially affect the economic resources of a mortgage banking entity and claims to those resources. To the extent an entity is subject to such requirements, the entity should disclose all of the following in the notes to the financial statements:

a. A description of the minimum net worth requirements related to the following:

<sup>&</sup>lt;sup>28</sup> The asset growth restriction may be waived if the NCUA has accepted the corporate credit union's capital restoration plan and an increase in total assets is consistent with that plan.

- i. Secondary market investors
- ii. State imposed regulatory mandates
- b. The actual or possible material effects of noncompliance with those requirements
- c. Whether the entity is in compliance with the regulatory capital requirements, including, as of each balance sheet date presented, the following with respect to quantitative measures:
  - i. The entity's required and actual net worth amounts
  - ii. Factors that may significantly affect adequacy of net worth such as potentially volatile components of capital, qualitative factors, or regulatory mandates
- d. If, as of the most recent balance sheet date, the entity is not in compliance with capital adequacy requirements, the possible material effects of such conditions on amounts and disclosures in the notes to the financial statements

17.82 Further, FASB ASC 948-10-50-4 states that noncompliance with minimum net worth requirements may, when considered with other factors, raise substantial doubt about an entity's ability to continue as a going concern for a reasonable period of time. Additional information that might be disclosed in these situations may include the following:

- Pertinent conditions and events giving rise to the assessment of substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time
- Possible effects of such conditions and events
- Management's evaluation of the significance of those conditions and events and any mitigating factors
- Possible discontinuance of operations
- Management's plans (including any relevant financial information)
- Information about the recoverability or classification of recorded asset amounts or the amounts or classifications of liabilities

Further, FASB ASC 948-10-50-4 states that noncompliance with minimum net worth requirements may, when considered with other factors, raise substantial doubt about an entity's ability to continue as a going concern for a reasonable period of time. See paragraphs 17.34–.35 for required going concern disclosure guidance. Additional information that might be disclosed in situations where there is substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time may include

- Possible effects of such conditions and events giving rise to the assessment of substantial doubt about the entity's ability to continue as a going concern for a reasonable period of time
- Possible discontinuance of operations
- Information about the recoverability or classification of recorded asset amounts or the amounts or classification of liabilities.<sup>29</sup>

<sup>&</sup>lt;sup>29</sup> See footnote 5.

17.83 FASB ASC 948-10-50-5 states that servicers with net worth requirements from multiple sources should disclose, in the notes to the financial statements, the net worth requirement of the following:

- a. Significant servicing covenants with secondary market investors with commonly defined servicing requirements (common secondary market investors include the U.S. HUD, Fannie Mae, Government National Mortgage Association, and Freddie Mac
- b. Any other secondary market investor where violation of the requirement would have a significant adverse effect on the business
- c. The most restrictive third-party agreement if not previously included

## Illustrative Disclosures for Mortgage Companies and Mortgage Banking Activities

17.84 The disclosures that follow are for illustrative purposes only, and represent a mortgage company that is in compliance with capital adequacy requirements. Comparative disclosures should be included for each balance sheet presented.

The Company is subject to various capital requirements in connection with seller-servicer agreements that the Company has entered into with secondary market investors. Failure to maintain minimum capital requirements could result in the Company's inability to originate and service loans for the respective investor and, therefore, could have a direct material effect on the Company's financial statements. Management believes, as of December 31, 200X and 200W, that the Company met all capital requirements to which it is subject.

The Company's actual capital amounts and the minimum amounts required for capital adequacy purposes, by investor, are as follows:

	Actual Capital	Minimum Capital Requirement
As of December 31, 200X:		
HUD	\$X,XXX,XXX	\$XXX,XXX
FHLMC	\$X,XXX,XXX	\$ XX,XXX
FNMA	\$X,XXX,XXX	\$ XX,XXX

## **Regulatory Capital Matters for All Entities**

## Regulatory Capital Disclosures for Branches of Foreign Institutions

17.85 The disclosure requirements related to capital adequacy and prompt corrective action do not apply to branches of foreign organizations because such branches do not have capital. See discussion of enhanced prudential standards for foreign banking organizations discussed in paragraph 17.27.

17.86 Paragraphs 3–4 of FASB ASC 942-505-50 state that branches of foreign financial institutions, although they do not have regulatory capital requirements, may be required to maintain capital-equivalent deposits and,

depending on facts and circumstances, supervisory mandated reserves. These requirements carry regulatory uncertainty of a nature similar to that posed by the regulatory capital rules in that failure to meet such mandates can result in supervisory action and ultimately going-concern questions. Accordingly, branches should disclose such requirements. Quantitative disclosure should be made, highlighting mandated deposit or reserve requirements and actual balances in those reserve or deposit accounts at the balance sheet date(s) reported. Further, if an uncertainty exists related to a parent that creates a higher than normal risk as to the viability of a branch or subsidiary, then that matter should be adequately disclosed in the notes to the financial statements of the branch or subsidiary. If factors do not exist that indicate a higher than normal amount of risk or uncertainty regarding parent capital and other regulatory matters, then disclosures of capital and supervisory issues of the parent would not be required.

## **Regulatory Capital Disclosures for Trust Operations**

17.87 Trust banks are required by certain federal regulators to hold capital as a percentage of discretionary and nondiscretionary assets under management. The percentages vary for each category. The percentages are not standardized as with other capital requirements and are communicated on an entity by entity basis in the application to obtain a trust charter or by other supervisory processes. Depending on the type of charter, these entities may be subject to risk-based standards as well. Because these are not published requirements, these guidelines are applied on a discretionary basis by the agencies and may not be uniformly applied to all entities.

17.88 FASB ASC 942-505-50-5 states that if an institution is subject to capital requirements based on trust assets under management, a discussion of the existence of these requirements, ramifications of failure to meet them, and a measurement of the entity's position relative to imposed requirements should be disclosed in the notes to the financial statements.

# Auditing 30

# Banks, Savings Institutions, and Credit Unions

## **Objectives**

17.89 In addition to testing of disclosures, as discussed subsequently, the auditor should consider the implications of capital noncompliance, as discussed in chapter 5, "Audit Considerations and Certain Financial Reporting Matters," and chapter 23, "Reporting Considerations," of this guide.

**17.90** In addition to the normal objectives sought in auditing equity (for example balances are presented in accordance with GAAP), the auditor's objective in this area is to obtain reasonable assurance that the financial statements

<sup>&</sup>lt;sup>30</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of *issuers* and *nonissuers* as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

include proper and understandable description and disclosure of regulatory matters (as discussed earlier in this chapter) in the context of the financial statements taken as a whole. Similarly, the audit objective for regulatory capital matters relates primarily to disclosure. <sup>31</sup> Capital amounts are determined under GAAP, regulatory reporting requirements, and regulatory capital standards.

17.91 An auditor's report on financial statements containing the required regulatory capital disclosures does not constitute an opinion on the fair presentation of the institution's regulatory reports (in part or taken as a whole) in accordance with underlying instructions for such reports. Nor does the opinion indicate that the auditor has confirmed with any regulatory agency that the agency has examined or otherwise evaluated or opined on the fair presentation of such reports.

#### Planning

17.92 In accordance with AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (see chapter 5 of this guide for additional information). In complying with the requirements of AU-C section 315, auditors should obtain an understanding of capital regulations sufficient to understand application and classification decisions made by management. In addition, the auditor should obtain audit evidence about the changes in regulatory reporting instructions and related capital requirements since the preceding audit. AU-C section 500, Audit Evidence (AICPA, Professional Standards), explains what constitutes audit evidence in an audit of financial statements and addresses the auditor's responsibility to design and perform audit procedures to obtain sufficient appropriate audit evidence to be able to draw reasonable conclusions on which to base the auditor's opinion (further discussion on AU-C section 500 can be found in chapter 5 of this guide).

**17.93** Paragraphs 5.223–.228 of this guide discuss the auditor's responsibility relative to review of supervisory reports and coordination with examiners.

17.94 While planning and carrying out procedures in other audit areas, the auditor should consider the regulatory reporting requirements and regulatory capital standards that affect the risk weighting of assets resulting from the institution's transactions. This information will help the auditor assess (a) the regulatory interpretations and reporting requirements that provide guidance consistent with GAAP in greater detail, (b) GAAP equity amounts adjusted in calculating regulatory capital amounts (for example, tier 1 capital and tier 2 capital), and (c) GAAP asset amounts and off-balance-sheet amounts risk weighted for regulatory capital purposes. The information will also be useful for performing any procedures applied to such adjustments (including consideration of the relative risk weightings assigned to certain amounts or transactions).

<sup>&</sup>lt;sup>31</sup> Notwithstanding the disclosure objective, regulatory matters may also affect preparation of the auditor's report, as discussed in chapter 23, "Reporting Considerations," of this guide.

- 17.95 As a part of understanding the entity and its environment, paragraph .12 of AU-C section 315 states that the auditor should obtain an understanding of relevant industry, regulatory, and other external factors, including the applicable financial reporting framework. In this regard, some components of regulatory capital ratios, including related amounts, asset measures, and risk weightings, may be difficult to determine due to (a) the complexity and subjectivity of capital standards and related regulatory reporting requirements or (b) the complexity of the institution's transactions. The number and variety of adjustments between GAAP and regulatory capital amounts affecting the institution also will affect inherent risk in this area.
- 17.96 Management's regulatory financial reporting classification and risk weighting decisions involve a high degree of subjective analysis by management and might be challenged by examiners. Accordingly, such decisions that could have a material impact on regulatory disclosures should be carefully considered by the auditor.
- 17.97 The following are examples of factors related to regulatory matters that may indicate higher risks of material misstatement:
  - A high volume or high degree of complexity of off-balance-sheet transactions
  - Actual or borderline noncompliance with minimum capital requirements
  - A poor regulatory rating
  - Past disagreements between management and regulators about classifications, risk weightings, other regulatory reporting requirements, or application of regulatory capital standards in general
  - Frequent corrections to filed regulatory reports
  - Regulatory restrictions or other regulatory actions taken related to capital compliance (for example, the federal banking agencies may issue informal enforcement actions such as a memorandum of understanding or commitment letter)
  - Unusual, material, or frequent related party transactions
  - Capital calculations, including management's classification or risk weighting decisions, that are not well documented

# Internal Control Over Financial Reporting

17.98 AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13–.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

©2016, AICPA AAG-DEP 17.98

17.99 Effective internal control over financial reporting in this area should provide reasonable assurance that errors or fraud in financial statement disclosures about regulatory matters are prevented or detected. In part, these controls may overlap with controls the institution has established for compliance with capital requirements. Institutions' systems for gathering the necessary information and preparing regulatory financial reports vary in sophistication. Examples of factors that may contribute to effective internal control in this area include the following:

- Responsibilities for capital planning, monitoring compliance with capital laws and regulations, and preparation of Call Reports have been assigned to competent individuals in the institution.
- Regulatory financial reporting is subject to risk assessment and supervisory control procedures and is overseen by officers of the institution who review the details supporting classifications and risk weightings.
- Capital amounts reported to regulators are reconciled to underlying detailed schedules and subsidiary ledgers with reconciling items supported by appropriate computations and documentation and with appropriate supervisory review and oversight.
- Procedures are in place for collection and reporting by branches, divisions, and subsidiaries of amounts necessary for regulatory capital calculations.
- Management obtains competent outside advice, as warranted, on significant classification or risk weighting questions before and after major transactions are executed.
- Regulatory capital analyses, calculations, and supporting documentation are well prepared and readily accessible.
- The regulatory financial reporting process (including classifications and risk weightings) is reviewed periodically by the internal audit function.

#### Substantive Procedures

17.100 Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards), states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure. In accordance with paragraph .445 of AU-C section 330, this requirement reflects the facts that (a) the auditor's assessment of risk is judgmental and may not identify all risks of material misstatement and (b) inherent limitations to internal control exist, including management override.

17.101 The extent to which the auditor applies tests to specific transactions or amounts will depend on the auditor's assessment of the risks of material misstatement and the materiality of the accounts. Where the risks of material misstatement are assessed at lower levels, the auditor may consider testing a reconciliation of differences between GAAP as applied in the financial statements versus GAAP as applied in regulatory reports (if management chose to use an option available under GAAP for financial statement purposes that is different than the regulatory interpretation of GAAP for regulatory reporting purposes) before year-end, reviewing classifications made for

risk weighting purposes, reviewing examination findings, and testing material differences, risk weighting classifications, and ratio calculations in preparation for any substantive tests to be applied to disclosures of year-end amounts and ratios.

- 17.102 Satisfaction that regular reserve transfers are in compliance with regulatory requirements is necessary when conducting a credit union engagement. To gain such satisfaction, the auditor should obtain an understanding about of the applicable federal and state laws and regulations. Other entries, including direct charges and credits in accordance with regulatory requirements, should be tested for propriety. Other appropriations of net "retained earnings" should be traced to authorization by the board of directors. Certain changes to the regular reserve are subject to regulatory approval and the auditor should be familiar with these requirements.
- **17.103** Paragraphs 5.223–.228 of this guide discuss the auditor's responsibility relative to review of supervisory reports and coordination with examiners. Such review and coordination should involve consideration of the adequacy of the financial statement disclosures in this area.
- 17.104 Paragraphs .A46—.A50 of AU-C section 330 provide additional guidance on the nature and extent of substantive procedures. Depending on the circumstances, the auditor may determine that performing only substantive analytical procedures will be sufficient to reduce audit risk to an acceptably low level (for example, when the auditor's assessment of risk is supported by audit evidence from tests of controls), only tests of details are appropriate, or a combination of substantive analytical procedures and tests of details are most responsible to the assessed risks. Substantive analytical procedures are generally more applicable to large volumes of transactions that tend to be predictable over time. The nature of the risk and assertion is relevant to the design of tests of details.

**17.105** Substantive audit procedures might include the following:

- Obtain and test management's schedules supporting calculation
  of the institution's actual and required regulatory capital ratios,
  including regulatory capital amounts (ratio numerators) and related asset bases (ratio denominators).
- Review and evaluate management's analyses of significant nonrecurring transactions and their impact on regulatory capital.
- Inquire about, and discuss with officers having responsibility for regulatory financial reporting, the existence and nature of any differences between GAAP as applied in the financial statements versus GAAP as applied in regulatory reports (if management chose to use an option available under GAAP for financial statement purposes that is different than the regulatory interpretation of GAAP for regulatory reporting purposes). Review copies of prior year regulatory reports (and, as necessary, client's supporting working papers), and obtain management's analysis of classification issues concerning preparation of Call Reports, including risk weighting classifications assigned. In assessing the completeness of any reconciliation, consider the potential for GAAP to have been applied differently for any other of the institution's transactions in the financial statements versus regulatory reports.

- Obtain any reconciliation of amounts supporting the institution's regulatory capital ratio calculations to amounts in the institution's financial statements prepared in conformity with GAAP:
  - Test management's supporting schedules and reconciliations for completeness and mathematical accuracy.
  - Agree GAAP amounts to general or subsidiary ledgers, or both, and obtain supporting schedules for non-GAAP amounts.
- Review the nature and amount of material non-GAAP amounts for propriety and consistency with prior years.
- Consider current treatment of items that resulted in past corrections or changes to regulatory financial reports.
- Consider whether significant changes in instructions for preparation of Call Reports have been applied to material transactions.
- Inquire about, and discuss with officers having responsibility for call reporting, any significant reclassification of transactions since the last filed regulatory report.

# Mortgage Companies and Activities

#### **Objectives**

17.106 The auditor's objective in this area includes the normal objectives sought in auditing equity (for example, balances are presented in accordance with GAAP), including obtaining reasonable assurance that the financial statements include proper description and disclosure of capital matters in the context of the financial statements taken as a whole. Capital noncompliance is an important consideration for auditors when conducting a mortgage company engagement.

## **Planning**

**17.107** In accordance with paragraph .A1 of AU-C section 300, *Planning* an Audit (AICPA, Professional Standards), the nature and extent of planning activities will vary according to the size and complexity of the entity, the key engagement team members' previous experience with the entity, and changes in circumstances that occur during the audit engagement. Paragraphs .09-.10 of AU-C section 210, Terms of Engagement (AICPA, Professional Standards), state that the auditor should agree upon the terms of the audit engagement with management or those charged with governance, as appropriate. The agreed-upon terms of the audit engagement should be documented in an audit engagement letter or other suitable form of written agreement (see paragraph .10 of AU-C section 210 for a listing of agreed-upon terms that should be included). Both management and the auditor have an interest in documenting the agreed-upon terms of the audit engagement before the commencement of the audit to help avoid misunderstandings with respect to the audit as stated in paragraph .A22 of AU-C section 210. For example, it might be necessary for an auditor to obtain an understanding about the capital requirements that the entity is subject to as a result of seller-servicer agreements entered into with investors, as well as capital requirements that may be imposed as a result of other business transactions such as borrowing arrangements. In connection with these requirements, it is important that the auditors understand the elements that constitute capital, as defined in the various agreements.

- 17.108 In accordance with paragraph .12 of AU-C section 315, the auditor should obtain an understanding of the entity's objectives and strategies and those related business risks that may result in risks of material misstatement. For purposes of generally accepted auditing standards, business risks are defined as risks resulting from significant conditions, events, circumstances, actions, or inactions that could adversely affect the entity's ability to achieve its objectives and execute its strategies or from the setting of inappropriate objectives and strategies.
- **17.109** The following are examples of factors related to capital matters that may indicate higher risks of material misstatement:
  - Actual or borderline noncompliance with minimum capital requirements
  - Communications or restrictions from investors regarding capital compliance issues
  - Capital requirements and calculations that are not well documented

#### Internal Control Over Financial Reporting

- 17.110 In accordance with paragraph .08 of AU-C section 330, the auditor should design and perform tests of controls to obtain sufficient appropriate audit evidence about the operating effectiveness of relevant controls if (a) the auditor's assessment of risks of material misstatement at the relevant assertion level includes an expectation that the controls are operating effectively or (b) substantive procedures alone cannot provide sufficient appropriate audit evidence at the relevant assertion level.
- 17.111 Examples of factors that may contribute to effective internal control in this area follow:
  - Responsibilities for capital planning and monitoring compliance with capital requirements have been assigned to competent officials in the company.
  - Capital analyses, calculations, and supporting documentation are well prepared and readily accessible.

#### Substantive Procedures

- 17.112 Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure.
- 17.113 The extent to which the auditor applies tests to specific transactions or amounts will depend on the auditor's assessment of inherent and control risks and the materiality of the accounts.
- 17.114 Paragraphs .A46—.A50 of AU-C section 330 provide additional guidance on the nature and extent of substantive procedures. Depending on the circumstances, the auditor may determine that performing only substantive analytical procedures will be sufficient to reduce audit risk to an acceptably low level (for example, when the auditor's assessment of risk is supported by audit evidence from tests of controls), only tests of details are appropriate, or a combination of substantive analytical procedures and tests of details are most responsible to the assessed risks. Substantive analytical procedures are

generally more applicable to large volumes of transactions that tend to be predictable over time. The nature of the risk and assertion is relevant to the design of tests of details.

#### 17.115 Such procedures might include the following:

- Obtain and read new seller-servicer agreements entered into during the period, or amendments to existing agreements, for capital requirements in effect
- Obtain and test management's schedules supporting calculation of the entity's actual and required capital amounts
- Review and evaluate management's analyses of significant nonrecurring transactions and their impact on capital
- Obtain any reconciliation of amounts supporting the entity's capital calculations to amounts in the entity's financial statements prepared in accordance with GAAP
- Test management's supporting schedules and reconciliations for completeness and mathematical accuracy
- Agree GAAP amounts to general or subsidiary ledgers, or both

# Chapter 18

# Derivative Instruments: Futures, Forwards, Options, Swaps, and Other Derivative Instruments

## Introduction

**18.01** The following section provides a discussion about the economic uses of derivative instruments and hedging activities. Refer to FASB *Accounting Standards Codification* (ASC) 815, *Derivatives and Hedging*, for accounting guidance on these topics.

**18.02** The derivative instruments addressed in this chapter, which include futures, forwards, swaps, and option contracts, as well as other financial contracts with similar characteristics, have become important financial management tools for banks and savings institutions. These instruments collectively are referred to in this chapter as derivative instruments, which are defined for accounting purposes in paragraphs 83–139 of FASB ASC 815-10-15. This chapter provides background information on basic contracts, risks, and other general considerations to provide a context for related accounting and auditing guidance.

18.03 This chapter focuses on end uses of derivatives, rather than on the broader range of activities that includes the marketing of derivatives to others. Some banks and savings institutions, primarily large commercial banks, act as market makers or dealers in derivatives that are not traded under uniform rules through an organized exchange. The primary goals of those activities are to make a market and earn income on the difference between the bid and offer prices. Although the nature of the transactions often causes individual exposures to offset each other, such activities may be subject to different permutations of risks and different accounting, auditing, and regulatory considerations.

**18.04** The AICPA Audit Guide *Special Considerations in Auditing Financial Instruments* provides background information about financial instruments and discussion of audit considerations relating to financial instruments.

#### Risks Inherent in Derivatives

18.05 Risks inherent in derivatives, such as credit risk, market risk, legal risk, and control risk, are the same as risks inherent in other types of financial instruments. However, derivatives have unique risks because derivative transactions are designed to create price exposure, and thereby transfer risk, by having their value determined—or derived—from the value of an underlying commodity, security, index, rate, or event. Unlike other financial instruments, derivatives generally do not involve the transfer of a title or principle, require little or no considerations to be exchanged at inception, and as such can be thought of as creating pure price exposure by linking their value to a notional amount (that is, principle of the underlying item). These characteristics give rise to potential risks and rewards substantially greater than the amounts recognized in the statement of financial position. Also, many derivatives are

less liquid and have values that are more volatile than those of other financial instruments, potentially alternating between positive and negative values in a short period of time.

- 18.06 Given these features, a derivative's risks can be difficult to segregate because the interaction of such risks may be complex. This complexity is increased (a) when two or more basic derivatives are used in combination, (b) by the difficulty of valuing complex derivatives, and (c) by the volatile nature of markets for some derivatives. The economic interaction between an institution's position in derivatives and that institution's other on- or off-balance-sheet positions (whether assets or liabilities) is an important determinant of the total risk associated with an institution's derivatives use. Risk assessment, therefore, involves consideration of the specific instrument and its interaction with other on- and off-balance-sheet portfolios and activities. No list of risk characteristics exists that can cover all those complex interactions, but a discussion of the basic risk characteristics associated with derivatives follows.
- 18.07 Counterparty credit risk of derivative instruments is the risk that the counterparty to a transaction could default or deteriorate in creditworthiness before the final settlement of a transaction's cash flows. Counterparty credit risk creates a bilateral risk of loss because the market value of a transaction can be positive or negative to either counterparty. For example, if a derivative instrument is in an unrealized gain position to the holder, and the counterparty enters bankruptcy, it is unlikely that the holder would be able to collect the unrealized value of the derivative from the bankrupt counterparty. Conversely, if the derivative instrument is in an unrealized loss position to the holder and the holder experienced financial difficulties, it is possible that the counterparty would not recover the unrealized value of the contract.
- 18.08 Entities often quantify this risk of loss based upon the derivative's replacement cost—that is, the fair value of an identical contract. Master netting agreements and collateral maintenance provisions are some of the ways that financial institutions and other derivative counterparties seek to reduce counterparty credit risk. Requirements that participants settle or collateralize changes in the value of their positions can mitigate the credit risk associated with many derivatives. Such settlement or collateralization requirements, which are typically based on the change in the value of a position, or group of positions, may occur daily or when the change exceeds a threshold, as governed by the rules of the exchange (for exchange traded derivatives) or through credit support annex agreements of each party's International Swaps and Derivatives Association master contracts.
- **18.09** Settlement risk is the related exposure that a counterparty may fail to perform under a contract after the institution has delivered funds or assets according to its obligations under the contract. Settlement risk relates almost solely to over-the-counter (OTC) derivative contracts (that is, nonexchange-traded derivatives). Institutions can reduce settlement risk through master netting agreements, which allows the parties to set off all their related payable and receivable positions at settlement.
- **18.10** Concentration risk broadly defines the sensitivity of the entity to an excessive concentration of exposure to specific counterparties, industry sectors, geographical locations, or individual transactions; the concept also extends to the degree of anticipated positive correlation in the behavior of such variables under stressed economic conditions.

- 18.11 Market risk relates broadly to economic losses due to adverse changes in the fair value of the derivative. Related risks include price risk, basis risk, and liquidity risk. Price risk relates to changes in the level of prices due to changes in (a) interest rates, (b) foreign exchange rates, or (c) other factors that relate to market volatilities of the rate, index, or price underlying the derivative. Basis risk relates to the differing effect market forces have on the performance or value of two or more distinct underlyings, possibly in instruments used in combination (see the discussion of hedging that follows). Liquidity risk relates to changes in the ability to sell, dispose of, or close out the derivative, thus affecting its value. This may be due to a lack of sufficient contracts or willing counterparties. Valuation or model risk is the risk associated with the imperfection and subjectivity of models and the related assumptions used to value derivatives.
- **18.12** Legal risk relates to losses due to a legal or regulatory action that invalidates or otherwise precludes performance by the institution or its counterparty under the terms of the contract or related netting arrangements. Such risk could arise, for example, from insufficient documentation for the contract, an inability to enforce a netting arrangement in bankruptcy, adverse changes in tax laws, or statutes that prohibit entities (such as certain state and local governmental entities) from investing in certain types of financial instruments.
- **18.13** Control risk relates to losses that result from the failure (or absence) of controls to prevent or detect problems (such as human error, fraud, or system failure) that hinder an institution from achieving its operational, financial reporting, or compliance objectives. Such failure could result, for instance, from unauthorized trading in derivative instruments or in an institution's failing to understand a contract's economic characteristics, all of which could impact published financial information or compliance with applicable contracts, laws, or regulations. Failure to understand the derivatives used may lead to inadequate design of controls over their use.

# Types of Derivatives

- $\bf 18.14~$  Paragraphs 83–139 of FASB ASC 815-10-15 define the term derivative~instrument.
- **18.15** A key feature of derivatives, as defined in this chapter, is that resulting cash flows are decided by reference to an underlying, such as the following:
  - Rates, indexes (which measure changes in specified markets), or other independently observable factors
  - b. The value of underlying positions in the following:
    - Financial instruments such as government securities, equity instruments (such as common stock), or foreign currencies
    - ii. Commodities such as corn, gold bullion, or oil
    - iii. Other derivatives
  - c. The occurrence or nonoccurrence of a specified event
- **18.16** Derivatives can generally be described as either *forward-based* or *option-based* or combinations of the two. A traditional forward contract obligates one party to buy and another counterparty to sell an underlying financial instrument, foreign currency, or commodity at a future date at an agreed-upon

price. Thus, a forward-based derivative (examples are futures, forward, and swap contracts) is a two-sided contract in that each party potentially has a favorable or unfavorable outcome resulting from changes in the value of the underlying position or the amount of the underlying reference factor. A traditional option contract provides one party that pays a premium (the option holder) with a right, but not an obligation, to buy (call options) or sell (put options) an underlying financial instrument, foreign currency, or commodity at an agreed-upon price on or before a predetermined date. The counterparty (the option writer) is obligated to sell (buy) the underlying position if the option holder exercises the right. Thus, an option-based derivative (examples are option contracts, interest rate caps, interest-rate floors, and swaptions) is one-sided in the sense that, in the event the right is exercised, only the holder can have a favorable outcome (or the loss is limited to any premium paid) and the writer can have only an unfavorable outcome reduced by any premium received. If market conditions would result in an unfavorable outcome for the holder, the holder will allow the right to expire unexercised. The expiration of the option contract results in a neutral outcome for both parties (except for any premium paid to the writer by the holder). Although there are a variety of derivatives, they generally are variants or combinations of these two types of contracts.

18.17 Derivatives also are either exchange-traded or traded OTC. Institutions and dealers trade futures, certain option, and other standardized contracts under uniform rules through an organized exchange. Most of the risk inherent in such exchange-traded derivatives relates to market risk rather than to credit risk. OTC derivatives are privately traded instruments (primarily swap, option, and forward contracts) customized to meet specific needs and for which the counterparty is not an organized exchange. As a result, although OTC derivatives are more flexible, they potentially involve higher credit and liquidity risk.

18.18 A description of the basic contracts and variations follows.

**18.19** Forward contracts are contracts negotiated between two parties to purchase and sell a specified quantity of a financial instrument, foreign currency, or commodity at a price specified at origination of the contract, with delivery and settlement at a specified future date. Forward contracts are not traded on exchanges and, accordingly, may be less liquid and generally involve more credit and liquidity risk than futures contracts.

18.20 Forward-rate agreements, which are widely used to manage interest rate risk (IRR), are forward contracts that specify a reference interest rate and an agreed-upon interest rate (one to be paid and one to be received) on an assumed deposit until a specified future date (the settlement date). The assumed deposit is also known as a notional amount. The term of the assumed deposit may begin at a future date; for example, the contract period may be for 6 months, commencing in 3 months. At the settlement date, the seller of the forward-rate agreement pays the buyer if interest calculated at the reference rate is higher than that calculated at the agreed-upon rate; conversely, the buyer pays the seller if interest calculated at the agreed-upon rate is higher than that calculated at the reference rate. Treasury rate locks are a type of forward-rate agreement where the settlement is based upon the price movements of a particular reference U.S. Treasury issuance rather than to a deposit. Examples of such reference instruments would be the current on-the-run 10-year U.S. Treasury security or possibly the Committee on Uniform Security Identification Procedures identifier of a particular U.S. Treasury issuance.

- 18.21 Futures contracts are forward-based contracts to make or take delivery of a specified financial instrument, foreign currency, or commodity at a specified future date or during a specified period at a specified price or yield. Futures are standardized contracts traded on an organized exchange. The deliverable financial instruments underlying interest-rate futures contracts are specified investment-grade financial instruments, such as U.S. Treasury securities or mortgage-backed securities (MBSs). Foreign-currency futures contracts involve specified deliverable amounts of a particular foreign currency. The deliverable products under commodities futures contracts are specified amounts and grades of commodities, such as oil, gold bullion, or coffee.
- 18.22 Active markets exist for many financial and commodities futures contracts. Active markets provide a mechanism by which entities may transfer their exposures to price risk to other parties. Those parties may, in turn, be trying to manage their own financial risks or achieve gains through speculation. Recognized exchanges, such as the International Monetary Market (a division of the Chicago Mercantile Exchange) or the Chicago Board of Trade, establish conditions governing transactions in futures contracts. U.S. Treasury bond (interest-rate) futures contracts are the most widely traded financial futures contracts. To ensure an orderly market, the exchanges specify maximum daily price fluctuations for each type of contract. If the change in price from the previous day's close reaches a specified limit, no trades at a higher or lower price are allowed. Consequently, trading in the contract is stopped until buy orders and sell orders can be matched either within the daily price limits or on the next business day. Such limits may affect liquidity and thereby hinder the effectiveness of futures contracts used as hedges.
- 18.23 Brokers require both buyers and sellers of futures contracts to deposit assets (such as cash, government securities, or letters of credit) with a broker. Such assets represent the initial margin (which is a good-faith deposit) at the time the contract is initiated. The brokers mark open positions to market daily and either call for additional assets to be maintained on deposit when losses are experienced (a margin call) or credit customers' accounts when gains are experienced. This daily margin adjustment is called *variation margin*. Variation margin payments generally must be settled daily in cash or acceptable collateral, thus reducing credit risk. The broker returns the initial margin when the futures contract is closed out or the counterparty delivers the underlying financial instrument according to the terms of the contract.
- **18.24** Delivery of the commodity or financial instrument underlying futures contracts occurs infrequently, as contracts usually are closed out before maturity. This close-out process involves the participants entering a futures contract that is equal and opposite to a currently held futures contract. This provides the participant with equal and opposite positions and obligations and eliminates any subsequent market movements from affecting the net asset or obligation during the remaining lives of the futures contracts.
- **18.25** Swap contracts are forward-based contracts in which two parties agree to swap streams of payments over a specified period. The payment streams are based on an agreed-upon (or notional) principal amount. The term notional is used because swap contracts generally involve no exchange of principal at either inception or maturity. Rather, the notional amount serves as a basis for calculation of the payment streams to be exchanged. Many swap contracts are not exchange-traded; however, certain types of interest rate, currency, and credit default swaps are required to be executed on an exchange or

swap execution facility and settled through a clearinghouse. These execution and clearing requirements are the result of regulations and are applicable depending on the nature and size of the parties in the transaction. OTC swaps are generally tailored to the needs of the parties and thus often have unique terms; accordingly, they are not as liquid as futures contracts and may lack the credit protection provided by regulated exchanges. The failure by a counterparty to make payments under a swap contract usually results in an economic loss to an institution only if the underlying prices (for example, interest rates or foreign exchange rates) have moved in an adverse direction (that is, in the direction that the swap contract was intended to protect against). The economic loss corresponds to the cost to replace the swap contract. That cost would be the present value of any discounted net cash inflows that the swap contract would have generated over its term.

- **18.26** Interest-rate swaps are the most prevalent type of swap contract. One party generally agrees to make periodic payments, which are fixed at the outset of the swap contract. The counterparty agrees to make variable payments based on a market interest rate (index rate). Swap contracts allow institutions to achieve net payments similar to those that would be achieved if the institution actually changed the interest rate of designated assets or liabilities (the underlying cash position) from floating to fixed rate or vice versa.
- 18.27 Interest-rate swap contracts are considered a flexible means of managing IRR. Because swap contracts are customized for institutions, terms may be longer than futures contracts, which generally have delivery dates from three months to three years. Swap contract documentation usually is standardized and transactions can be concluded quickly, making it possible to rapidly take action against anticipated interest-rate movements.
- **18.28** Interest-rate swap contracts normally run to maturity. However, there may be circumstances that eliminate an institution's need for the swap contract before maturity. In such instances, an institution may cancel contracts, sell its position, or enter an offsetting swap contract.
- **18.29** In some swap contracts, the timing of payments varies. For example, in an interest-rate swap contract, one party might pay interest quarterly while the counterparty pays interest semiannually. An added element of credit risk exists for the quarterly payer because of the risk that the semiannual payer may default. Here, the economic loss equals the lost quarterly payment and the cost of replacing the swap contract.
- **18.30** Many entities enter legally enforceable master netting agreements that may reduce total credit risk. Upon default by an applicable counterparty, the agreements provide that entities may set off (for settlement purposes) all their related payable and receivable derivative contract positions.
- **18.31** Foreign-currency swaps (sometimes called cross-currency exchange agreements) are used to fix (for example, in U.S. dollar terms) the value of foreign exchange transactions that will occur in the future. Foreign-currency swap contracts are also used to transfer a stream of cash flows denominated in a particular currency or currencies into another currency or currencies. Basic features of foreign-currency swap contracts include the following:
  - The principal amount is usually exchanged at the initiation of the swap contract.

- Periodic interest payments are made based on the outstanding principal amounts at the respective interest rates agreed to at inception.
- The principal amount is usually re-exchanged at the maturity date of the swap contract.
- **18.32** In *fixed-rate-currency swaps*, two counterparties exchange fixed-rate interest in one currency for fixed-rate interest in another currency. Currency coupon or cross-currency interest-rate swap contracts combine the features of an interest-rate swap contract and a fixed-rate-currency swap contract. That is, the counterparties exchange fixed-rate interest in one currency for floating-rate interest in another currency.
- **18.33** Basis swaps are a variation on interest-rate swap contracts where both rates are variable but are tied to different index rates. For example, one party's rate may be indexed to three-month London Interbank Offered Rate (LIBOR) while the other party's rate is indexed to six-month LIBOR.
- **18.34** Equity swaps, also known as total return swaps, are contracts in which the counterparties exchange a series of cash payments based on (a) an equity index and (b) a fixed or floating interest rate on a notional principal amount. Equity swap contracts typically are tied to a stock index, but sometimes they relate to a particular stock or a defined basket of stocks. One party (the equity payer) pays the counterparty (the equity receiver) an amount equal to the increase in the stock index at regular intervals specified in the contract. Conversely, the equity receiver must pay the equity payer if the stock index declines. The counterparties generally make quarterly payments. Whatever the index performance, the party designated as the equity receiver may also receive an amount representing dividends paid by the companies making up the index during the period.
- 18.35 The equity payer, on a floating-rate equity swap contract, typically receives LIBOR (plus or minus a spread) on the notional principal amount defined in the equity swap contract. This notional principal amount is based on the underlying equity index value at the contract's inception. The notional principal amount is adjusted at each payment date to reflect the settlement of the equity gain or loss. The floating rate is also reset on the periodic payment dates. A fixed-rate equity swap contract is essentially the same, except that the interest rate is fixed for the term of the contract.
- **18.36** Commodity swaps are contracts in which the counterparties agree to exchange cash flows based on the difference between an agreed-upon, fixed price and a price that varies with changes in a specified commodity index, as applied to an agreed-upon quantity of the underlying commodity.
- 18.37 In *mortgage swaps*, two counterparties exchange contractual payments designed to replicate the net cash flows of a portfolio of MBSs financed by short term floating-rate funds. For example, mortgage swaps enable an institution to finance mortgage securities at a rate tied to a floating-rate index below LIBOR on a guaranteed, multiyear basis. Mortgage swaps have been described as being similar to an amortizing interest-rate swap (rather than one with a fixed notional principal amount) with a long term forward commitment to purchase MBSs. In a typical mortgage swap transaction, an investor contracts with a third party to receive cash flows based on a generic class of MBSs over a specified period in exchange for the payment of interest at a rate

typically based on LIBOR. The payments are made as if there were an underlying notional pool of mortgage securities. Payments are exchanged on a monthly basis. The cash flows received by the investor are derived not only from the fixed coupon on the generic class of securities but also, to the extent that the coupon is above or below par, from the benefit or loss implicit to the discount or premium. The notional amount of the mortgage swap is adjusted monthly, based on the amortization and prepayment experience of the generic class of MBSs.

- 18.38 The contract may require the investor either to take physical delivery of mortgages at a predetermined price (for example, a percentage of the par amount of mortgages remaining in the pool) when the contract expires or to settle in cash for the difference between the predetermined price of the mortgages and their fair value as determined by the dealer.
- **18.39** Overnight index swaps are swap contracts where the floating rate is based upon the geometric average of the overnight rates for the period. The overnight index swap rate is an interest rate representative of the amount of interest paid on funds deposited for one day, such as posted collateral. It is becoming more common to see overnight index swap rates used in the pricing of swaps where collateral is required to be posted to reduce counterparty nonperformance risks.
- **18.40** Option contracts are traded on an exchange or OTC (that is, they are negotiated between two parties). Option contracts allow, but do not require, the holder (or purchaser) to buy (call) or sell (put) a specific or standard commodity, or financial or equity instrument, at a specified price during a specified period (an American option), at a specified date (a European option), or dates (a Bermudian option). Furthermore, certain option contracts may involve cash settlements based on changes in specified indexes, such as stock indexes. Again, the principal difference between option contracts and either futures or forward contracts are that an option contract does not require the holder to exercise the option, whereas performance under a futures or forward contract is mandatory.
- **18.41** At the inception of an option contract, the holder typically pays a fee, which is called a *premium*, to the writer (or seller) of the option. The premium includes 2 components of value: the intrinsic value and the time value. The intrinsic value of a call option is the excess, if any, of the market price of the item underlying the option contract over the price specified in the option contract (the strike price or the exercise price). The intrinsic value of a put is the excess, if any, of the option contract's strike price over the market price of the item underlying the option contract. The intrinsic value of an option cannot be less than zero. The other component of the premium's value is the time value. The time value reflects the probability that the price of the underlying item will move above the strike price (for a call) or below the strike price (for a put) during the exercise period. For example, suppose an entity owned a call option that granted it the right to purchase a given stock at \$50 per share. If the price of the underlying stock is \$50, then the intrinsic value of the option is \$0. If the price of the stock rises to \$55 per share, then the intrinsic value is \$5 because the entity can purchase for \$50 an asset that has a market value of \$55. If the market value of the shares drops to \$45 per share, then the option will not be exercised: it has an intrinsic value of \$0.
- **18.42** The advantage of option contracts held is that they can be used to either mitigate downside price risk or to permit upside profit potential. This

is because the loss on a purchased option contract is limited to the amount paid for the option contract. Profit on written option contracts is limited to the premium received but the loss potential is unlimited because the writer is obligated to settle at the strike price if the option is exercised.

- 18.43 Option contracts are frequently processed through a clearinghouse that guarantees the writer's performance under the contract. This reduces credit risk, much like organized exchanges reduce credit risk for futures contracts. Thus, such option contracts are primarily subject to market risk. However, for option contracts that are not processed through the clearinghouse, the holder may have significant credit and liquidity risks.
- **18.44** Different option contracts can be combined to transfer risks from one entity to another. Examples of such option-based derivatives are caps, floors, collars, and swaptions (an option to enter a swap).
- **18.45** *Interest-rate caps* are contracts in which the cap writer, in return for a premium, agrees to limit, or cap, the cap holder's risk associated with an increase in interest rates for a certain period of time. If rates go above a specified interest-rate level (the strike price or the cap rate), the cap holder is entitled to receive cash payments equal to the excess of the market rate over the strike rate multiplied by the notional amount. Issuers of floating-rate liabilities often purchase caps to protect against rising interest rates while retaining the ability to benefit from a decline in rates.
- **18.46** Because a cap is an option-based contract, the cap holder has the right but not the obligation to exercise the option. If rates move down, the cap holder has lost only the premium paid. Because caps are not exchange-traded, however, they expose the cap holder to credit risk because the cap writer could fail to fulfill its obligations to the cap holder.
- **18.47** A cap writer has virtually unlimited risk resulting from increases in interest rates above the cap rate. However, the cap writer's premium may potentially provide an attractive return.
- **18.48** *Interest-rate floors* are similar to interest-rate caps. Interest-rate floors are contracts in which the floor writer, in return for a premium, agrees to limit the risk associated with a decline in interest rates based on a notional amount for a certain period of time. If rates fall below an agreed rate, the floor holder will receive cash payments from the floor writer equal to the difference between the market rate and an agreed rate multiplied by the notional amount. Floor contracts allow floating-rate lenders to limit the risk associated with a decline in interest rates while benefiting from an increase in rates. As with interest-rate caps, the floor holder is exposed to credit risk because the floor writer could fail to fulfill its obligations.
- **18.49** *Interest-rate collars* combine a cap and a floor (one held and one written). Interest-rate collars enable an institution with a floating-rate contract to lock into a predetermined interest-rate range.
- **18.50** *Swaptions* are option contracts to enter into an interest-rate swap contract at some future date or to cancel an existing swap contract in the future. As such, a swaption contract may act as a floor or a cap for an existing swap contract or be used as an option to enter, close out, or extend a swap contract in the future.
- 18.51 Various contracts may need to be evaluated as possible derivatives including warrants, the embedded conversion feature in convertible debt or

convertible preferred stock, and other embedded derivatives that may or may not require bifurcation. Warrants generally meet the definition of a derivative and should be accounted for as such unless they meet the exception criteria in paragraphs 74-75 of FASB ASC 815-10-15 (see also FASB ASC 815-40 for additional information). Features embedded in contracts or agreements may require separate accounting as a derivative, and complex pricing structures may increase the complexity of the assumptions used in estimating the fair value of a derivative. Warrants and embedded derivatives are less likely to be identified by management as requiring derivative treatment, which increases the inherent risk for certain assertions. For example, an option to convert the principal outstanding under a loan agreement into equity securities is less likely to be identified for valuation and disclosure considerations if it is a clause in a loan agreement than if it is a freestanding agreement. Similarly, a structured note may include a provision for payments related to changes in a stock index or commodities prices that requires separate accounting. Certain embedded interest structures also require separate accounting, FASB ASC 815 requires bifurcation (separation) of embedded derivatives from host contracts if (a) the economic characteristics and risks of the embedded derivative are not clearly and closely related to those of the host, (b) a separate instrument with the same terms as the embedded derivative would be a derivative instrument subject to the requirements of FASB ASC 815-15, and (c) the hybrid instrument is not currently measured at fair value through profit or loss. For additional information on the scope of embedded derivatives guidance and evaluating embedded derivatives for bifurcation, see FASB ASC 815-15-15 and FASB ASC 815-15-25, respectively.<sup>1</sup>

In addition, the amendments in this ASU clarify that, in evaluating the nature of a host contract, an entity should assess the substance of the relevant terms and features (that is, the relative strength of the debt-like or equity-like terms and features given the facts and circumstances) when considering how to weight those terms and features. Specifically, the assessment of the substance of the relevant terms and features should incorporate a consideration of (1) the characteristics of the terms and features themselves (for example, contingent versus noncontingent, in-the-money versus out-of-the-money), (2) the circumstances under which the hybrid financial instrument was issued or acquired (for example, issuer-specific characteristics, such as whether the issuer is thinly capitalized or profitable and well-capitalized), and (3) the potential outcomes of the hybrid financial instrument (for example, the instrument may be settled by the issuer issuing a fixed number of shares, the instrument may be settled by the issuer specified amount of cash, or the instrument may remain legal-form equity), as well as the likelihood of those potential outcomes.

FASB ASU No. 2014-16 is effective, for public business entities, for fiscal years, and interim periods within those fiscal years, beginning after December 15, 2015. For all other entities, FASB ASU No. 2014-16 is effective for fiscal years beginning after December 15, 2015, and interim periods within fiscal years beginning after December 15, 2016. Early adoption, including adoption in an interim period, is permitted.

Readers are encouraged to consult the full text of the ASU on FASB's website at www.fasb.org.

<sup>&</sup>lt;sup>1</sup> FASB Accounting Standards Update (ASU) No. 2014-16, Derivatives and Hedging (Topic 815): Determining Whether the Host Contract in a Hybrid Financial Instrument Issued in the Form of a Share Is More Akin to Debt or to Equity (a consensus of the FASB Emerging Issues Task Force), eliminates the use of different methods in practice and thereby reduces existing diversity under U.S. generally accepted accounting principles (GAAP) in the accounting for hybrid financial instruments issued in the form of a share. The amendments in this ASU do not change the current criteria in GAAP for determining when separation of certain embedded derivative features in a hybrid financial instrument is required. That is, an entity will continue to evaluate whether the economic characteristics and risks of the embedded derivative feature are clearly and closely related to those of the host contract, among other relevant criteria. The amendments clarify how current GAAP should be interpreted in evaluating the economic characteristics and risks of a host contract in a hybrid financial instrument that is issued in the form of a share. Specifically, the amendments clarify that an entity should consider all relevant terms and features—including the embedded derivative feature being evaluated for bifurcation—in evaluating the nature of the host contract. Furthermore, the amendments clarify that no single term or feature would necessarily determine the economic characteristics and risks of the host contract. Rather, the nature of the host contract depends upon the economic characteristics and risks of the entire hybrid financial instrument.

**18.52** Credit derivatives are derivatives based on an index, basket of reference entities, or single name reference entity where settlement is determined by the credit performance of the reference entity. The writer of a credit derivative receives a premium (either paid up front or over the life of the contract) as compensation for an obligation to make a payment should a defined credit event occur on the underlying referenced obligation(s). Credit events may include bankruptcy, failure to pay, decline in credit rating, or repudiation. These instruments may be used by financial institutions to reduce or modify the credit exposure of their loan or investment portfolios. Credit derivatives should not be confused with financial guarantee contracts as described in FASB ASC 815-10-15-58, which are not derivatives. Credit derivatives differ from financial guarantees in that credit derivatives require settlement regardless of whether the holder has suffered a loss based on the underlying referenced obligation(s), whereas a settlement under a financial guarantee contract is based on loss suffered by the holder. A letter of credit is an example of a financial guarantee contract.

#### Uses of Derivatives to Alter Risk

- 18.53 Financial market participants have created a large variety of derivatives. Not only are there basic contracts, but there are variants tailored to add, subtract, multiply, or divide the related risk and reward characteristics and thereby satisfy specific risk objectives of the parties to the transactions. Such innovation has been driven by the users' desire to cope with (or attempt to take advantage of) market volatility in foreign exchange rates, interest rates, and other market prices; deregulation; tax law changes; and other broad economic or business factors. An institution may attempt to alter such risks (a) at a general level (that is, the overall risk exposures faced by the institution), (b) at the level of specific portfolios of assets or liabilities, or (c) narrowly to a specific asset, liability, or anticipated transaction. Uses of derivatives to alter risks range from uses that help mitigate or control volatile risk exposures (activities that include the idea of taking defensive action against risk through hedging) to uses that increase exposures to risk and, by that, the potential rewards (the idea of offensive action, often considered as trading or speculation).
- **18.54** Speculation. Speculation involves the objective of profiting by entering into an exposed position. That is, assuming risk in exchange for the opportunity to profit from anticipated market movements. A speculator believes that the cash market price of an underlying commodity, financial instrument, or index will change so that the derivative produces net cash inflows or can be closed out in the future at a profit. By simultaneously buying the relatively cheaper item and selling the relatively more expensive item, speculation serves a key function in the financial markets by eliminating price differences and works to keep markets and their prices efficient. See discussion of the Volcker Rule prohibiting banks from propriety trading for their own account in paragraphs 1.54 and 18.77–.78 of this guide.
- **18.55** *Risk management.* Most institutions that use derivatives do so to increase or decrease risks associated with existing or anticipated on- or off-balance-sheet transactions, not for speculation. Institutions often manage financial risks both generally (through management of the overall mix of financial assets and liabilities) and specifically (through hedges of specific risks or transactions). Risk managers typically look at the mix of the institution's interest-based assets and liabilities and attempt to economically hedge those instruments where right of setoff exists. Any excess or shortfalls in the

economic hedges may then be offset by looking to derivatives and other financial instruments to achieve the degree of offset necessary to insulate the institution from movements in market interest rates.

- 18.56 Some entities continually analyze and manage financial assets and liabilities based on their payment streams and interest rates, the timing of their maturities, and their sensitivity to actual or potential changes in market prices or interest rates. Such activities fall under the broad definition of asset/liability management. Some institutions enter into derivatives to help manage and select their total exposure to IRR. Institutions also enter into derivatives to convert the cash flow pattern and market risk profile of assets and liabilities. Those instruments can be used in the institution's asset/liability management activities to synthetically alter the interest income and expense flows of certain assets or liabilities. For example, an institution can convert the cash flow pattern and market risk profile of floating-rate debt to those of fixed-rate debt by entering an interest-rate swap contract. (Note that synthetic instrument accounting is prohibited by FASB ASC 815-10-25-4).
- **18.57** *Hedging* connotes a risk alteration activity to protect against the risk of adverse price movements on certain of an institution's assets, liabilities, or anticipated transactions. A hedge is a defensive strategy. It is used to avoid or reduce risk by creating a relationship by which losses on certain positions (assets, liabilities, or anticipated transactions) are expected to be counterbalanced in whole or in part by gains on separate positions in another market. For example, an institution may want to attempt to fix the value of an asset, the sales price of some portion of its future production, the rate of exchange for payments to its suppliers, or the interest rates of existing or an anticipated issuance of debt. Derivatives may be designated as accounting hedges, as discussed further in the "Accounting and Financial Reporting" section of this chapter. Derivatives not designated as accounting hedges either because they do not qualify for specialized hedge accounting, or by management choice, but held to offset a risk are commonly referred to as economic hedges. If not held for an accounting or economic hedge, and if unrelated to transactions with customers and others (for instance, a bank may underwrite floating rate debt and also enter into a fixed to floating rate swap with the same customer or compensate employees with derivative instruments), the use of a derivative is generally considered speculative.
- **18.58** The use of various financial instruments to reduce certain risks results in the hedger assuming a different set of risks. Effective control and management of risks through hedging, usually depends on a thorough understanding of the market risks associated with the financial instrument that is part of the hedging program.
- **18.59** Basis risk is an important risk encountered with most hedging contracts. As introduced previously, basis is the difference between the cash market price of the instrument or other position being hedged and the price of the related hedging contract. The institution is subject to the risk that the basis will change while the hedging contract is open (that is, the price correlation will not be perfect). Changes in basis can occur continually and may be significant. Changes in basis can occur even if the position underlying the hedging contract is the same as the position being hedged. However, entities often enter a hedging contract, such as a futures contract, on a position that is different from the position being hedged. Such *cross-hedging* increases the basis risk.

- **18.60** As cash market prices change, the prices of related hedging contracts change, but not necessarily to the same degree. *Correlation* is the degree to which hedging contract prices reflect the price movement in the cash market. The higher the correlation between changes in the cash market price and the hedging contract's price, the higher the precision with which the hedging contract will offset the price changes of the position being hedged (that is, less basis risk).
- **18.61** Gains or losses on the hedge position will not exactly offset the exposed cash market positions when the basis changes. The institution might enter a hedge when (a) it is perceived that the risk of a change in basis is lower than the risk associated with the cash market price exposure or (b) there is the ability to monitor the basis and to adjust the hedge position in response to basis changes.
- **18.62** Basis changes occur in response to many factors. Among them are (a) economic conditions, (b) supply and demand for the position being hedged, (c) liquidity of the cash market and the futures market for the instrument, (d) the credit rating of the cash instrument, and (e) the maturity of the instrument being hedged as compared with the instrument represented in the hedging contract. A discussion of how these factors affect basis is beyond the scope of this chapter. However, convergence—a significant contributor to a change in the basis over time—warrants mention.
- **18.63** Convergence is the shrinking of the basis between the hedging contract's price and the cash market price as the contract delivery date approaches. The hedging contract's price includes an element related to the time value up to the expiration of the contract. Convergence results from the delivery feature of hedging contracts that encourages the price of an expiring contract to equal the price of the deliverable cash market instrument on the day that the contract expires. As the delivery day approaches, prices generally fluctuate less and less from the cash market prices because the effect of expectations related to time is diminishing.
- 18.64 The correlation factor represents the potential effectiveness of hedging a cash market instrument with a contract where the deliverable financial instrument differs from the cash market instrument. The correlation factor generally is determined by regression analysis or another method of technical analysis of market behavior. When a high degree of positive correlation has historically existed between the hedging instrument price and the cash market price of the instrument being hedged, the risk of price variance associated with a cross-hedge is expected to be lower than the risk of not being hedged. Institutions usually employ the correlation factor to analyze cross-hedging risk at the inception of the hedge, while actual changes in the relative values of the hedge instrument and the hedged item usually are employed throughout the hedge period to measure correlation.

#### Variations on Basic Derivatives

**18.65** Some derivatives combine two or more basic contracts and thereby the risk and reward characteristics of several different products. Written options and other variations embedded in certain derivative and non-derivative contracts can magnify interest-rate and other risks assumed by the institution as end user. Included may be variations affecting the term, notional amount, interest rate, or specified payments. These variations have the potential to produce higher cash inflows or outflows than similar instruments that do not

contain the option feature. This follows the general rule that the greater the potential return, the higher the risk.

**18.66** Some swap contracts involve the institution's writing of options that the counterparty issuer may exercise if certain changes occur in the index rate or under other specified circumstances. As with most option contracts (and allowing for the effect of the premium paid for the contract) the holder of the option (here, the counterparty) has a potentially favorable (or neutral) outcome, while the writer of the option (here, the institution) has a potentially unfavorable (or neutral) outcome if the option is exercised. For example, the counterparty will exercise an option to sell securities to the institution at a specified price only when that price exceeds the current market prices. Accordingly, the institution must analyze such contracts carefully to understand the nature of the derivative and how it will work under various interest-rate and other conditions.

**18.67** Other variations. Other variations built into derivatives may necessitate certain actions be taken by the institution or may result in changes in terms if specified events or conditions occur. For example, such variations might involve

- increases or decreases in the notional amount based on certain changes in interest rates;
- increases or decreases in interest rates based on a multiplier (that is, leverage);
- additional payments as a result of specified conditions; and
- a settlement payment based upon the expiration of a contract.

18.68 Some swap contracts magnify changes in the specified index rate by tying floating payments to an exponent of the index rate over a specified denominator. The risks of this variation, a contract with embedded leverage terms, are similar to the risks posed by written options. Consider a contract that specifies the floating rate as 3-month LIBOR squared and divided by 5 percent. Assume that 3-month LIBOR is 5 percent at inception. If 3-month LIBOR were to climb 5 basis points to 5.05 percent, the increase would be magnified. The floating rate would increase 10 basis points to approximately 5.10 percent (5.05 percent squared and divided by 5 percent). Thus, at this level of interest rates, an increase of 1 basis point in the index rate for the contract would result in an increase of 2 basis points in the contractual rate—in other words, 1 basis point on twice the stated notional amount.

**18.69** Finally, the notional principal amount of certain swap contracts changes with changes in the rate to which the floating payments are indexed. These are called index amortizing swaps. For example, the notional principal amount may decrease when interest rates decline. Thus, the floating-rate payer would lose some of the benefit of declining interest rates but would not get a corresponding benefit if interest rates increase.

# Regulatory Matters<sup>2</sup>

**18.70** Banking Circular 277, *Risk Management of Financial Derivatives*, issued by the Office of the Comptroller of the Currency (OCC), addresses banks'

<sup>&</sup>lt;sup>2</sup> Chapter 7, "Investments in Debt and Equity Securities," of this guide discusses the regulatory matters affecting the permissibility of certain investments.

risk management of derivatives and sets forth best practices and procedures for managing risk. Bulletin OCC 94-31, Risk Management of Financial Derivatives Q & A's, answers commonly asked questions about Banking Circular 277. Through Bulletin OCC 2014-8, End-User Derivatives and Trading Activities: Comptroller's Handbook Supplemental Examination Procedures, the OCC issued supplemental exam procedures applicable to banks that are active end users of derivatives or that have significant trading activity. The Board of Governors of the Federal Reserve System (Federal Reserve) issued detailed guidance to its examiners for evaluating derivatives with respect to management oversight, measurements and monitoring procedures, and internal control in Supervision and Regulation (SR) letters 96-17, Supervisory Guidance for Credit Derivatives, 97-18, Application of Market Risk Capital Requirements to Credit Derivatives, and 02-10, Derivative Contracts Hedging Trust Preferred Stock. The FDIC issued guidance for its examiners in Financial Institution Letter (FIL)-62-96, Credit Derivatives, and FIL-45-98, Investment Activities. Derivatives: Practice and Principles and related Appendices, published by the Global Derivatives Study Group in 1993, also recommends a set of sound risk management practices for dealers and end-users of derivatives. Readers can access the guidance from any of the respective agencies' websites.

**18.71** Intercompany transactions and performance of services between two or more U.S. Bank legal vehicles and all their affiliates must be conducted in compliance with the applicable statutory and regulatory requirements of Sections 23A-23B of the Federal Reserve Act (Banks and Banking, U.S. Code 12. Section 371c and 371c-1) and Regulation W (Title 12 U.S. Code of Federal Regulations [CFR] Part 223) of the Federal Reserve Act. Derivative transactions are subject to the market terms requirement of Section 23B. Accordingly. each institution would price, and require collateral in, derivative transactions with affiliates in a way that is at least as favorable to the institution as the way the institution would price, or require collateral in, a derivative transaction with comparable unaffiliated counterparties. Credit derivatives that are the functional equivalent of a guarantee are also subjected to Section 23A, and the Federal Reserve Board suggests that other derivative transactions that are the functional equivalent of a loan might be subject to the requirements of Section 23A, depending on facts and circumstances. Regulation W also requires the bank to have policies and procedures in place to manage the credit exposure arising from derivatives with affiliates in a safe and sound manner (see 12 CFR 223.33[b]).

**18.72** The Federal Financial Institutions Examination Council's (FFIEC's) *Advisory on Interest Rate Risk Management*, issued in January 2010,<sup>3</sup> reminds institutions of supervisory expectations of sound practices for

(continued)

<sup>&</sup>lt;sup>3</sup> In January 2012, the Office of the Comptroller of the Currency (OCC), the Board of Governors of the Federal Reserve System, the FDIC, the National Credit Union Administration, and State Liaison Committee issued the Interagency Advisory on Interest Rate Risk Management Frequently Asked Questions. This document was created to clarify points in the 2010 interagency Advisory on Interest Rate Risk Management by providing responses to the most common questions. The responses address interest rate risk (IRR) exposure measurement and reporting, model risk management, stress testing, assumption development, and model and systems validation. Financial institution management should consider the responses in the context of the institution's complexity, risk profile, business model, and scope of operations. Readers can access the guidance from any of the respective agencies' websites.

In conjunction with the release of this interagency advisory, Bulletin OCC 2012-5, Interest Rate Risk Management: FAQs on 2010 Interagency Advisory on Interest Rate Risk Management, also

managing IRR. This advisory reiterates the importance of effective corporate governance, policies and procedures, risk measuring and monitoring systems, stress testing, and internal control related to the IRR exposures of depository institutions. It also clarifies elements of existing guidance and describes some IRR management techniques used by effective risk managers. For the complete text of the advisory, see the FFIEC's website at www.ffiec.gov.

18.73 On March 22, 2010, the OCC, the Federal Reserve, and the FDIC (collectively, the federal banking agencies), along with the Office of Thrift Supervision (prior to its transfer of powers to the OCC, the Federal Reserve, and the FDIC),<sup>4</sup> and the National Credit Union Administration issued *Interagency Policy Statement on Funding and Liquidity Risk Management*. The policy statement summarizes the principles of sound liquidity risk management that the agencies have issued in the past and, when appropriate, supplements them with the *Principles for Sound Liquidity Risk Management and Supervision* issued by the Basel Committee on Banking Supervision in September 2008. This policy statement emphasizes supervisory expectations for all depository institutions including banks, thrifts, and credit unions.

**18.74** Policies on funding and liquidity risk management should clearly articulate a liquidity risk tolerance that is appropriate for the business strategy of the institution considering its complexity, business mix, liquidity risk profile, and its role in the financial system. Policies should also contain provisions for documenting and periodically reviewing assumptions used in liquidity projections. Policy guidelines should employ both quantitative targets and qualitative guidelines. For example, these measurements, limits, and guidelines may be specified in terms of the measures and conditions mentioned in the policy statement, as applicable to derivatives (for example, contingent liability exposures such as unfunded loan commitments, lines of credit supporting asset sales or securitizations, and collateral requirements for derivative transactions and various types of secured lending; or exposures of material activities. such as securitization, derivatives, trading, transaction processing, and international activities, to broad systemic and adverse financial market events. This is most applicable to institutions with complex and sophisticated liquidity risk profiles).

18.75 In response to the increased risk for model management, the OCC and the Federal Reserve jointly developed and issued Supervisory Guidance on Model Risk Management, which was released as Bulletin OCC 2011-12, Sound Practices for Model Risk Management: Supervisory Guidance on Model Risk Management, and SR letter 11-7, Guidance on Model Risk Management, in April 2011. The guidance replaces Bulletin OCC 2000-16, Risk Modeling: Model Validation. Although model validation remains at the core of the guidance, the broader scope of model risk management encompasses model development, implementation, and use, as well as governance and controls related to models. All banks and savings and loan holding companies should ensure that internal policies and procedures are consistent with the risk management principles and

<sup>(</sup>footnote continued)

required that savings associations have an independent IRR measurement process that measures both earnings and capital at risk, as described in the 2010 interagency advisory and the 1996 IRR policy statement.

<sup>&</sup>lt;sup>4</sup> See chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide for further discussion on the Office of Thrift Supervision transfer of powers.

supervisory expectations contained in this guidance. For further information, readers can access the supervisory guidance from either the OCC website at www.occ.gov or the Federal Reserve website at www.federalreserve.gov.

**18.76** In June 2011, the federal banking agencies issued *Interagency Su*pervisory Guidance on Counterparty Credit Risk. The guidance clarifies supervisory expectations and sound practices for an effective counterparty credit risk management framework. The guidance emphasizes that banks should use appropriate reporting metrics and limits systems, have well-developed and comprehensive stress testing, and maintain systems that facilitate measurement and aggregation of counterparty credit risk throughout the organization. The guidance is intended for banks with significant derivatives portfolios. Banks with limited derivatives exposure, particularly noncomplex exposures that are typical for community banks (such as embedded caps and floors on assets or liabilities, forward agreements to sell mortgages, or simple interest rate swaps) should apply this guidance as appropriate. Banks using derivatives that are more complex or those with significant noncomplex derivatives exposure should refer to the guidance for applicable risk management principles and practices. Readers can access the guidance from any of the federal banking agencies' websites.

18.77 In January 2014, the OCC, the Federal Reserve, the FDIC, and the SEC published a final rule, *Prohibitions and Restrictions on Proprietary Trading and Certain Interests in, and Relationships With, Hedge Funds and Private Equity Funds*, implementing Section 13 of the Bank Holding Company Act, which was added by Section 619 (commonly referred to as the *Volcker Rule*) of the Dodd-Frank Wall Street Reform and Consumer Protection Act (see additional discussion in chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide). National banks (other than certain limited-purpose trust banks), federal savings associations, and federal branches and agencies of foreign banks are required to fully conform their activities and investments to the requirements of the final regulations by the end of the conformance period, which the Federal Reserve extended to July 21, 2015.

#### 18.78 The final regulations

- prohibit banks from engaging in short-term proprietary trading of certain securities, derivatives commodity futures, and options on these instruments for their own accounts.
- impose limits on banks' investments in, and other relationships with, hedge funds and private equity funds.
- provide exemptions for certain activities, including marketmaking-related activities, underwriting, risk-mitigating hedging, trading in government obligations, insurance company activities, and organizing and offering hedge funds and private equity funds.
- clarify that certain activities are not prohibited, including acting as agent, broker, or custodian.
- scale compliance requirements based on the size of the bank and
  the scope of the activities. Larger banks are required to establish
  detailed compliance programs and their CEOs must attest to the
  OCC that the bank's programs are reasonably designed to achieve
  compliance with the final regulations. Smaller banks engaged in
  modest activities are subject to a simplified compliance program.

# Accounting and Financial Reporting<sup>5</sup>

18.79 FASB ASC 815-10-05-4 requires that an entity recognize derivative instruments, including certain derivative instruments embedded in other contracts as assets or liabilities, in the statement of financial position and measure them at fair value. If certain conditions are met, an entity may elect, under FASB ASC 815, to designate a derivative instrument in any one of the following ways:

- a. A hedge of the exposure to changes in the fair value of a recognized asset or liability, or an unrecognized firm commitment that is attributable to a particular risk (referred to as a fair value hedge)
- b. A hedge of the exposure to variability in the cash flows of a recognized asset or liability, or of a forecasted transaction, that is attributable to a particular risk (referred to as a cash flow hedge)
- c. A hedge of the foreign currency exposure of a net investment in a foreign operation, an unrecognized firm commitment (a foreign currency fair value hedge), an available-for-sale security (a foreign currency fair value hedge), or a forecasted transaction (a foreign currency cash flow hedge)

# © Update 18-1 Accounting and Reporting: Recognition and Measurement of Financial Assets and Financial Liabilities

FASB Accounting Standards Update (ASU) No. 2016-01, Financial Instruments—Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities, issued in January 2016, is effective for fiscal years, and interim periods within those fiscal years, of a public business entity beginning after December 15, 2017.

For all other entities (including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960, *Plan Accounting—Defined Benefit Pension Plans*, through FASB ASC 965, *Plan Accounting—Health and Welfare Benefit Plans*, on plan accounting), FASB ASU No. 2016-01 is effective for fiscal years beginning after December 15, 2018, and interim periods within fiscal years beginning after December 15, 2019. Early application is

<sup>&</sup>lt;sup>5</sup> In May 2010, FASB issued proposed ASU Accounting for Financial Instruments and Revisions to the Accounting for Derivative Instruments and Hedging Activities—Financial Instruments (Topic 825) and Derivatives and Hedging (Topic 815). The main objective of this proposal is to provide financial statement users with a more timely and representative depiction of an entity's involvement in financial instruments while reducing the complexity in accounting for those instruments. It develops a consistent framework for classifying financial instruments, removes the threshold for recognizing credit impairments creating a single credit impairment model for both loans and debt securities, and makes changes to the requirements to qualify for hedge accounting.

On February 9, 2011, FASB issued a discussion paper, Invitation to Comment—Selected Issues about Hedge Accounting, to solicit input on the International Accounting Standards Board's (IASB's) exposure draft Hedge Accounting in order to improve, simplify, and converge the financial reporting requirements for hedging activities. Specifically, it requests stakeholders to comment on whether the IASB's exposure draft is a better starting point for any changes to GAAP as it relates to derivatives and hedging activities.

With the other major financial instruments related projects completed, FASB has re-engaged in the hedging project. As of the release of this guide, the FASB staff are drafting an exposure draft for public comment.

Readers are encouraged to visit the "Technical Agenda" page at www.fasb.org for the latest developments regarding FASB's research project titled Accounting for Financial Instruments—Hedging and how it may impact the guidance in this chapter.

permitted for all other entities as of the fiscal years beginning after December 15, 2017, including interim periods within those fiscal years.

All entities may adopt the presentation guidance in paragraphs 5–7 of FASB ASC 825-10-45 for financial statements of fiscal years or interim periods that have not yet been issued or that have not yet been made available for issuance. Furthermore, entities that are not public business entities may elect not to disclose the information about fair value of financial instruments required by the "General" subsection of FASB ASC 825-10-50 in financial statements of fiscal years or interim periods that have not yet been made available for issuance. Except as indicated previously in this paragraph, early application is not permitted.

FASB ASU No. 2016-01 addresses certain aspects of recognition, measurement, presentation, and disclosure of financial instruments and affects all entities that hold financial assets or owe financial liabilities. Among other provisions of the guidance, the amendments in this ASU supersede the guidance to classify equity securities with readily determinable fair values into different categories (that is, trading or available-for-sale) and require equity securities (including other ownership interests, such as partnerships, unincorporated joint ventures, and limited liability companies) to be measured at fair value with changes in the fair value recognized through net income (other than equity securities accounted for under the equity method of accounting or those that result in consolidation of an investee).

The amendments in this ASU also require an entity to present separately in other comprehensive income the portion of the total change in fair value of a liability resulting from a change in the instrument-specific credit risk when the entity has elected to measure the liability at fair value in accordance with the fair value option for financial instruments.

Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-01, see appendix G, "Accounting for Financial Instruments," of this guide.

**18.80** FASB ASC 820-10-20 defines fair value as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. Paragraphs 5 and 5A of FASB ASC 820-10-35 state that a fair value measurement assumes that the transaction to sell the asset or transfer the liability takes place either in the principal market for the asset or liability or, in the absence of a principal market, in the most advantageous market for the asset or liability. The FASB ASC glossary defines the principal market as the market with the greatest volume and level of activity for the asset or liability. A reporting entity need not undertake an exhaustive search of all possible markets to identify the principal market or, in the absence of a principal market, the most advantageous market, but it should take into account all information that is reasonably available. In the absence of evidence to the contrary, the market in which the reporting entity normally would enter into a transaction to sell the asset or to transfer the liability is presumed to be the principal market or, in the absence of a principal market, the most advantageous market. If the market in which the transaction takes place is different from the principal market (or most advantageous market), the transaction price might not represent the fair value of an asset or liability at initial recognition as stated in FASB ASC 820-10-30-3A(d). Under FASB ASC 820-10-35-53, when relevant observable inputs are not otherwise available, the use of unobservable data to measure fair value is permitted. Unobservable inputs should reflect assumptions that market participants would use when pricing the asset or liability including assumptions about risk. See chapter 20, "Fair Value," of this guide and FASB ASC 820, Fair Value Measurement, for additional information.

18.81 The specific criteria for qualifying for hedge accounting vary depending on the type of hedge. (FASB ASC 815-20-25 sets forth criteria that must be met for designated hedging instruments and hedged items or transactions to qualify for fair value hedge accounting, cash flow hedge accounting, and accounting for a hedge of a net investment in a foreign operation.) FASB ASC 815-20-25-3 prescribes requirements for designation and documentation of the hedge at inception for cash flow and fair value hedges. One aspect of qualification should include an assessment of the expectation of effective off-setting changes in fair values or cash flows during the term of the hedge for the risk being hedged, as stated in FASB ASC 815-10-10-1(d). To meet the documentation requirement applicable to fair value, cash flow, and net investment hedges, at the inception of the hedge, management must designate the derivative as a hedge and contemporaneously formally document the hedging relationship, including identification of all of the following as stated in FASB ASC 815-20-25-3(b):

- The hedging relationship.
- The entity's risk management objective and strategy for undertaking the hedge, including identification of all of the following:
  - The hedging instrument.
  - The hedged item or transaction.
  - The nature of the risk being hedged.
  - The method that will be used on an ongoing basis to retrospectively and prospectively assess the hedging instrument's effectiveness in offsetting the exposure to changes in the hedged item's fair value (if a fair value hedge) or hedged transaction's variability in cash flows (if a cash flow hedge) attributable to the hedged risk. There should be a reasonable basis for how the entity plans to assess the hedging instrument's effectiveness.
  - The method that will be used to measure hedge ineffectiveness (including those situations in which the change in fair value method as described in paragraphs 31–32 of FASB ASC 815-30-35 will be used).
  - If the entity is hedging foreign currency risk on an aftertax basis, that the assessment of the effectiveness, including the calculation of ineffectiveness, will be on an after-tax basis (rather than on a pretax basis).

# Considerations for Private Companies That Elect to Use Standards as Issued by the Private Company Council

FASB ASC 815 also allows the use of the simplified hedge accounting approach to account for swaps that are entered into for the purpose of economically converting a variable-rate borrowing into a fixed-rate

borrowing. The approach may be applied to all entities, except for public business entities and not-for-profit entities as defined in the FASB ASC glossary, employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting, and financial institutions. As described in FASB ASC 942-320-50-1, the term financial institutions includes banks, savings and loan associations, savings banks, credit unions, finance companies, and insurance entities. Because mortgage companies do not fall under the financial institutions exclusion, mortgage companies that are not public business entities would be permitted to follow the simplified hedge accounting approach. Under this approach, the income statement charge for interest expense will be similar to the amount that would result if the entity had directly entered into a fixed-rate borrowing instead of a variable-rate borrowing and a receive-variable, pay-fixed interest rate swap. Alternatively, that entity may continue to follow the current guidance in FASB ASC 815.

The simplified hedge accounting approach provides entities within its scope with a practical expedient to qualify for cash flow hedge accounting under FASB ASC 815. Under this approach, an entity may assume no ineffectiveness for qualifying swaps designated in a hedging relationship under FASB ASC 815. This approach can be applied to a cash flow hedge of a variable-rate borrowing with a receive-variable, pay-fixed interest rate swap provided all of the following criteria are met:

- Both the variable rate on the swap and the borrowing are based on the same index and reset period (for example, both the swap and borrowing are based on one-month LIBOR or both the swap and borrowing are based on three-month LIBOR). In complying with this condition, an entity is not limited to benchmark interest rates described in FASB ASC 815-20-25-6A.
- The terms of the swap are typical (in other words, the swap is what is generally considered to be a "plainvanilla" swap), and there is no floor or cap on the variable interest rate of the swap unless the borrowing has a comparable floor or cap.
- The repricing and settlement dates for the swap and the borrowing match or differ by no more than a few days.
- The swap's fair value at inception (that is, at the time the derivative was executed to hedge the interest rate risk of the borrowing) is at or near zero.
- The notional amount of the swap matches the principal amount of the borrowing being hedged. In complying with this condition, the amount of the borrowing being hedged may be less than the total principal amount of the borrowing.
- All interest payments occurring on the borrowing during the term of the swap (or the effective term of the swap underlying the forward starting swap) are designated as hedged whether in total or in proportion to the principal amount of the borrowing being hedged.

Under the simplified hedge accounting approach, an entity that qualifies for the accounting alternative has the option to measure the designated swap at settlement value instead of fair value. The primary difference between settlement value and fair value is that nonperformance risk is not considered in determining settlement value. One approach for estimating the receive-variable, pay-fixed interest rate swap's settlement value is to perform a present value calculation of the swap's remaining estimated cash flows using a valuation technique that is not adjusted for nonperformance risk.

Under the simplified hedge accounting approach, documentation required by FASB ASC 815-20-25-3 to qualify for hedge accounting must be completed by the date on which the first annual financial statements are available to be issued after hedge inception rather than concurrently at hedge inception. Because FASB ASC 815 permits election of hedge accounting on a swap-by-swap basis, an entity that qualifies for the accounting alternative can elect to apply this approach to any qualifying swap, whether existing at the date of adoption of this approach or entered into after that date. In determining whether an existing swap otherwise meets all of the requirements for applying this approach at adoption, the criterion that the swap's fair value at the time of the application of this approach is at or near zero does not need to be considered as long as the swap's fair value was at or near zero at the time the swap was entered into.

All other requirements in FASB ASC 815 for cash flow hedge accounting apply for the simplified hedge accounting approach. In addition, the current disclosure requirements in FASB ASC 815 and 820 on fair value measurement continue to apply for a swap accounted for under the simplified hedge accounting approach. In providing those disclosures, amounts recorded at settlement value may be used in place of fair value wherever applicable with amounts disclosed at settlement value subject to all of the same disclosure requirements as amounts disclosed at fair value. Any amounts disclosed at settlement value should be clearly stated as such and disclosed separately from amounts disclosed at fair value.

For the purpose of evaluating whether FASB ASC 825, *Financial Instruments*, disclosures about the fair value of financial instruments are required, a swap recorded under the simplified hedge accounting approach is not considered a derivative instrument under FASB ASC 815.

- **18.82** Items (c) and (d) in FASB ASC 815-20-25-3 also include additional documentation requirements applicable specifically to either fair value hedges or cash flows hedges. Paragraphs 4–72 of FASB ASC 815-20-25 address eligibility criteria for hedged items and transactions.
- **18.83** FASB ASC 815-10-35-2 states that the accounting for changes in the fair value (that is, gains and losses) of a derivative instrument depends on whether it has been designated and qualifies as part of a hedging relationship and, if so, the reason for holding it.
- 18.84 Companies typically receive periodic fair value reporting from the counterparty. These fair values may or may not conform to the requirements of FASB ASC 820. As such, additional consideration of these values may be warranted. See chapter 20 of this guide for additional information regarding fair value measurements and disclosures.

18.85 *Guarantees.* According to FASB ASC 460-10-25-1, a guarantee that is accounted for as a derivative instrument at fair value under FASB ASC 815 is not subject to the recognition provisions of FASB ASC 460, *Guarantees.* The contingent aspect of the guarantee should be accounted for in accordance with FASB ASC 450-20 unless the guarantee is accounted for as a derivative under FASB ASC 815, according to FASB ASC 460-10-35-4.

#### © Update 18-2 Accounting and Reporting: Credit Losses

FASB ASU No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, issued in June 2016, is effective for fiscal years of public business entities that are SEC filers beginning after December 15, 2019, including interim periods within those fiscal years.

For all other public business entities, the amendments in FASB ASU No. 2016-13 are effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years.

For all other entities, including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting, FASB ASU No. 2016-13 is effective for fiscal years beginning after December 15, 2020, and interim periods within fiscal years beginning after December 15, 2021.

Early application is permitted for all entities as of the fiscal years beginning after December 15, 2018, *including* interim periods within those fiscal years.

FASB ASU No. 2016-13 creates FASB ASC 326, Financial Instruments—Credit Losses, to amend guidance on reporting credit losses for assets held at amortized cost basis and available-for-sale debt securities.

For assets held at amortized cost basis, FASB ASC 326 eliminates the probable initial recognition threshold in current U.S. generally accepted accounting principles (GAAP) and, instead, requires an entity to reflect its current estimate of all expected credit losses. The allowance for credit losses is a valuation account that is deducted from the amortized cost basis of the financial assets to present the net amount expected to be collected.

For available-for-sale debt securities, credit losses should be measured in a manner similar to current GAAP. However, FASB ASC 326 will require that credit losses be presented as an allowance rather than as a write-down.

FASB ASU No. 2016-13 affects entities holding financial assets and net investment in leases that are not accounted for at fair value through net income. The amendments affect loans, debt securities, trade receivables, net investments in leases, off-balance-sheet credit exposures, reinsurance receivables, and any other financial assets not excluded from the scope that have the contractual right to receive cash.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-13, see appendix G of this guide.

©2016, AICPA AAG-DEP 18.85

18.86 In accordance with FASB ASC 460-10-50-1, the loss contingency disclosures required by FASB ASC 460-10-50 apply to all guarantees, including guarantees that are outside the scope of FASB ASC 460-10-15-4 (such as a guarantee accounted for as derivative instruments at fair value under FASB ASC 815); however, this guidance does not apply to guarantees or indemnifications excluded from the scope of FASB ASC 450, *Contingencies*, as described in FASB ASC 460-10-15-7(a). According to FASB ASC 460-10-50-5(c), the disclosure requirements of FASB ASC 460-10-50 do not eliminate or affect the disclosure requirements in the disclosure sections of FASB ASC 815, which apply to guarantees that are accounted for as derivatives.

18.87 Offsetting. Paragraphs 1–2 of FASB ASC 210-20-05 state that it is a general principle of accounting that the offsetting of assets and liabilities in the balance sheet is improper except where a right of setoff exists. The general principle that the offsetting of assets and liabilities is improper except where a right of setoff exists is usually thought of in the context of unconditional receivables from and payables to another party. The general principle also applies to conditional amounts recognized for contracts under which the amounts to be received or paid or items to be exchanged in the future depend on future interest rates, future exchange rates, future commodity prices, or other factors. The FASB ASC glossary defines right of setoff as a debtor's legal right, by contract or otherwise, to discharge all or a portion of the debt owed to another party by applying against the debt an amount that the other party owes to the debtor. FASB ASC 210-20-45-1 specifies what conditions must be met to have that right. FASB ASC 210-20-45-9 states that the phrase enforceable at law encompasses the idea that the right of setoff should be upheld in bankruptcy.

18.88 Paragraphs 1-7 of FASB ASC 815-10-45 address offsetting certain amounts related to derivative instruments. For purposes of this guidance, derivative instruments include those that meet the definition of a derivative instrument but are not included in the scope of FASB ASC 815-10. None of the provisions in FASB ASC 815-10 support netting a hedging derivative's asset (or liability) position against the hedged liability (or asset) position in the balance sheet. Unless the conditions in FASB ASC 210-20-45-1 are met, the fair value of derivative instruments in a loss position should not be offset against the fair value of derivative instruments in a gain position. Similarly, amounts recognized as accrued receivables should not be offset against amounts recognized as accrued payables unless a right of setoff exists. Without regard to the condition in FASB ASC 210-20-45-1(c), a reporting entity may offset fair value amounts recognized for derivative instruments and fair value amounts recognized for the right to reclaim cash collateral (a receivable) or the obligation to return cash collateral (a payable) arising from derivative instrument(s) recognized at fair value executed with the same counterparty under the same master netting arrangement. Regardless of whether or not instruments have been netted for presentation in the statement of financial position, FASB ASC 815-10-50-4B requires derivative instruments to be reported gross within the footnote disclosures.

**18.89** FASB ASC 230-10-45-27 states that cash flows from a derivative instrument that is accounted for as a fair value hedge or cash flow hedge may be classified in the same category as the cash flows from the items being hedged provided that the derivative instrument does not include an other-than-insignificant financing element at inception, other than a financing element inherently included in an at-the-market derivative instrument with no

prepayments (that is, the forward points in an at-the-money forward contract) and that the accounting policy is disclosed. If the derivative instrument includes an other-than insignificant financing element at inception, all cash inflows and outflows of the derivative instrument should be considered cash flows from financing activities by the borrower. If for any reason hedge accounting for an instrument that hedged an identifiable transaction or event is discontinued, then any cash flows after the date of discontinuance should be classified consistent with the nature of the instrument.

- **18.90** SEC Market Risk Disclosure Rules.<sup>6</sup> Item 305 of Regulation S-K requires entities filing with the SEC to disclose certain information about market risk. In general, the rule
  - a. requires quantitative and qualitative disclosures about market risk inherent in derivatives and other financial instruments outside the financial statements; and
  - b. provides a reminder to registrants to supplement existing disclosures about financial instruments, commodity positions, firm commitments, and other forecasted transactions with related disclosures about derivatives.
- **18.91** SEC Management's Discussion and Analysis (MD&A) Requirements. Item 303 of Regulation S-K requires institutions to discuss, in their MD&A, any known trends or any known demands, commitments, events or uncertainties that the institution reasonably expects to have a material favorable or unfavorable impact on their results of operations, liquidity, and capital resources.
- **18.92** Written loan commitments. In accordance with FASB ASC 815-10-15-71, loan commitments that relate to the origination of mortgage loans that will be held for sale, as discussed in FASB ASC 948-310-25-3, should be accounted for, by the issuer of the loan commitment, as derivative instruments at fair value. Recording a written loan commitment at its initial fair value may result in the recognition in an initial asset and gain, or an initial liability and loss.
- 18.93 Consistent with the guidance in FASB ASC 860-50 and FASB ASC 825-10, the SEC's *Codification of Staff Accounting Bulletins* topic 5DD, "Written Loan Commitments Recorded at Fair Value Through Earnings," states the expected net future cash flows related to the associated servicing of a loan should be included in the measurement of all written loan commitments that are accounted for at fair value through earnings. The expected net future cash flows should be determined in the same manner that the fair value of a recognized servicing asset or liability is measured under FASB ASC 860-50. However, as discussed in FASB ASC 860-50-25-1, a separate and distinct servicing asset or liability is not recognized for accounting purposes until the servicing rights have been contractually separated from the underlying loan by sale or securitization of the loan with servicing retained.
- **18.94** The preceding guidance should be applied irrespective of whether the resulting loan is intended to be sold servicing released or servicing retained.

©2016. AICPA

 $<sup>^6</sup>$  Refer to the "Division of Corporation Finance: Rules, Regulations and Schedule" page at  ${\it www.sec.gov.}$ 

<sup>&</sup>lt;sup>7</sup> See footnote 6.

#### **Financial Statement Disclosures**

- **18.95** According to FASB ASC 815-10-50-1, an entity with derivative instruments (or non-derivative instruments that are designated and qualify as hedging instruments pursuant to paragraphs 58 and 66 of FASB ASC 815-20-25) should disclose information to enable users of the financial statements to understand all of the following:
  - How and why an entity uses derivative instruments (or such nonderivative instruments)
  - How derivative instruments (or such non-derivative instruments) and related hedged items are accounted for under FASB ASC 815
  - How derivative instruments (or such non-derivative instruments) and related hedged items affect an entity's financial position, financial performance, and cash flows
- **18.96** Paragraphs 4A–4F of FASB ASC 815-10-50 require certain quantitative disclosures about fair value amounts of and gains and losses on derivative instruments.
- 18.97 FASB ASC 815-10-50-4K requires certain disclosures about credit derivatives and hybrid instruments (for example, a credit-linked note) that have embedded credit derivatives to enable users of financial statements to assess their potential effect on the entity's financial position, financial performance, and cash flows. These disclosures, however, do not apply to an embedded derivative feature related to the transfer of credit risk that is only in the form of subordination of one financial instrument to another as described in FASB ASC 815-15-15-9.
- 18.98 FASB ASC 815-10-50-5 provides additional information to consider related to qualitative disclosures. Qualitative disclosures about an entity's objectives and strategies for using derivative instruments (and non-derivative instruments that are designated and qualify as hedging instruments pursuant to FASB ASC 815-20-25-58 and FASB ASC 815-20-25-66) may be more meaningful if such objectives and strategies are described in the context of an entity's overall risk exposures relating to IRR, foreign exchange risk, commodity price risk, credit risk, and equity price risk. Those additional qualitative disclosures, if made, should include a discussion of those exposures even though the entity does not manage some of those exposures by using derivative instruments. An entity is encouraged, but not required, to provide such additional qualitative disclosures about those risks and how they are managed.
- **18.99** As stated in FASB ASC 825-10-50-20, except as indicated in FASB ASC 825-10-50-22, an entity should disclose all significant concentrations of credit risk arising from all financial instruments, whether from an individual counterparty or groups of counterparties. Throughout paragraphs 20–21 of FASB ASC 825-10-50, the term *financial instruments* includes derivative instruments accounted for under FASB ASC 815. Group concentrations of credit risk exist if a number of counterparties are engaged in similar activities and have similar economic characteristics that would cause their ability to meet contractual obligations to be similarly affected by changes in economic or other conditions.
- **18.100** FASB ASC 210-20-50 requires enhanced disclosures about financial instruments and derivative instruments that are either offset in accordance with FASB ASC 210-20-45 or FASB ASC 815-10-45 or subject to an enforceable

master netting arrangement or similar agreement (irrespective of whether they are offset in accordance with the aforementioned FASB ASC sections). Further discussion on the scope of and specific disclosures required by FASB ASC 210-20-50 can be found beginning in paragraph 14.40 of this guide.

# Auditing Considerations<sup>8</sup>

18.101 AU-C section 501, Audit Evidence—Specific Considerations for Selected Items (AICPA, Professional Standards), addresses specific considerations by the auditor in obtaining sufficient appropriate audit evidence, in accordance with AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards); AU-C section 500, Audit Evidence (AICPA, Professional Standards); and other relevant AU-C sections regarding, among other considerations, certain aspects of investments in securities and derivative instruments in an audit of financial statements. In addition, the companion AICPA Audit Guide Special Considerations in Auditing Financial Instruments provides background information about financial instruments and discussion of audit considerations relating to financial instruments. The suggested auditing procedures contained in the guide do not increase or otherwise modify the auditor's responsibilities described in AU-C section 501. Rather, the suggested procedures in the guide are intended to clarify and illustrate the application of the requirements of AU-C section 501. In addition, AU-C section 540, Auditing Accounting Estimates, Including Fair Value Accounting Estimates, and Related Disclosures (AICPA. Professional Standards), addresses the auditor's responsibilities relating to accounting estimates, including fair value accounting estimates and related disclosures, in an audit of financial statements. Specifically, it expands on how AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards); AU-C section 330; and other relevant AU-C sections are to be applied with regard to accounting estimates. It also includes requirements and guidance related to misstatements of individual accounting estimates and indicators of possible management bias.

Considerations for Audits Performed in Accordance With PCAOB Standards<sup>9</sup>

PCAOB Staff Audit Practice Alert No. 2, Matters Related to Auditing Fair Value Measurements of Financial Instruments and the Use of Specialists (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance sec. 400.02), provides guidance on auditors' responsibilities for auditing fair value measurements of financial instruments and when using the work of specialists under the existing standards of the PCAOB. This alert is focused on specific matters that are likely to increase audit risk related to the fair value of financial instruments

<sup>&</sup>lt;sup>8</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of issuers and nonissuers as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

<sup>&</sup>lt;sup>9</sup> PCAOB Staff Audit Practice Alerts are not rules of the board and do not reflect any board determination or judgment about the conduct of any particular firm, auditor, or any other person.

in a rapidly changing economic environment. This practice alert highlights certain requirements in the auditing standards related to fair value measurements and disclosures in the financial statements and certain aspects of GAAP that are particularly relevant to the current economic environment.

PCAOB Staff Audit Practice Alert No. 4, Auditor Considerations Regarding Fair Value Measurements, Disclosures, and Other-Than-Temporary Impairments (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance sec. 400.04), informs auditors about potential implications of the recent guidance on reviews of interim financial information and annual audits. This alert addresses the following topics: (a) reviews of interim financial information (reviews); (b) audits of financial statements, including integrated audits; (c) disclosures; and (d) auditor reporting considerations.

### Chapter 19

### **Business Combinations**

#### Introduction

19.01 Business combinations, as defined in the FASB Accounting Standards Codification (ASC) glossary, are transactions or other events in which an acquirer obtains control of one or more businesses. A business, as defined in the FASB ASC glossary, is an integrated set of activities and assets that is capable of being conducted and managed for the purpose of providing a return in the form of dividends, lower costs, or other economic benefits directly to investors or other owners, members, or participants. In accordance with FASB ASC 805-10-55-3, a business combination may be structured in a variety of ways for legal, taxation, or other reasons, which include but are not limited to, the following:

- One or more businesses become subsidiaries of an acquirer or the net assets of one or more businesses are legally merged into the acquirer
- One combining entity transfers its nets assets or its owners transfer their equity interests to another combining entity or its owners
- All of the combining entities transfer their net assets or the owners of those entities transfer their equity interests to a newly formed entity (sometimes referred to as a roll-up or put-together transaction)
- A group of former owners of one of the combining entities obtains control of the combined entity

Business combinations may involve one entity acquiring the equity interests or net assets of another entity or both entities transferring their equity interests or net assets to a newly formed entity. Business combinations involving depository institutions are common and result from voluntary decisions as well as regulatory mandates. Most business combination issues are the same for depository institutions as for other business entities. This chapter addresses only significant issues that are unique to depository institutions.<sup>1</sup>

#### **Regulatory Matters**

**19.02** Staff Accounting Bulletin (SAB) No. 112, which was issued by the SEC in June 2009, amends or rescinds portions of the SEC's interpretative guidance regarding business combinations (codified in the SEC *Codification of* 

<sup>&</sup>lt;sup>1</sup> FASB has two open projects titled Clarifying the Definition of a Business Phase 1 (formerly Application of Asset- or Entity-Based Guidance to Nonfinancial Assets in an Entity) and Clarifying the Scope of Subtopic 610-20 and Accounting for Partial Sales of Nonfinancial Assets (formerly Clarifying the Definition of a Business Phase 2). Phase 1 of the projects is intended to clarify the definition of a business with the objective of adding guidance to assist entities with evaluating whether transactions should be accounted for as acquisitions (or disposals) of assets or businesses. Phase 2 of the projects will include clarifying the guidance for partial sales or transfers of nonfinancial assets in the scope of FASB Accounting Standards Codification (ASC) 610-20 and the reference to in-substance nonfinancial assets. Readers are encouraged to visit the "Technical Agenda" page at www.fasb.org for the latest developments regarding these projects, including a summary of decisions reached to date.

Staff Accounting Bulletins, topic 2, "Business Combinations"). This guidance was amended to make the SEC's interpretative guidance consistent with accounting guidance in FASB ASC 805, Business Combinations, and FASB ASC 810, Consolidations. Among other changes, SEC Codification of Staff Accounting Bulletins topic 2(A)(5), which provided guidance on assigning acquisition costs to loans receivable acquired in a business combination, was removed. FASB ASC 805 provides new guidance that requires acquired receivables, including loans, to be measured at their acquisition date fair value and precludes the acquirer from recognizing a separate valuation allowance at the acquisition date for assets acquired in a business combination. In addition, SAB No. 112 amended SEC Codification of Staff Accounting Bulletins topic 2(A)(6), "Debt Issue Costs," to conform to the requirement in FASB ASC 805-10-25-23 that acquisition-related costs be accounted for as expenses in the period in which the costs are incurred and services are received, except for costs incurred to issue debt or equity securities which are recognized in accordance with other applicable U.S. generally accepted accounting principles (GAAP). The full text of SAB No. 112 is located on the SEC website.

- 19.03 In certain circumstances, an acquired bank or savings institution uses the acquiring institution's basis of accounting in preparing the acquired institution's financial statements. GAAP provides that an acquired entity may elect to apply pushdown accounting in its separate financial statements upon a change-in-control event in which an acquirer obtains control of the acquired entity. FASB ASC 805-50 addresses the specific guidance on pushdown accounting for all entities.
- 19.04 As addressed in the entry business combinations in the "Glossary" section of the Federal Financial Institutions Examination Council's Instructions for Preparation of Consolidated Reports of Condition and Income, the federal bank supervisory agencies note that the pushdown accounting election available under GAAP can be used to produce a particular result in the Consolidated Reports of Condition and Income (Call Report) that may not be reflective of the economic substance of the underlying business combination. Therefore, an institution's primary federal regulator reserves the right to require or prohibit the institution's use of pushdown accounting for Call Report purposes based on the regulator's evaluation of whether the election best reflects the facts and circumstances of the business combination.
- 19.05 The Office of the Comptroller of the Currency's (OCC's) Bank Accounting Advisory Series (BAAS) is updated periodically to express the Office of the Chief Accountant's current views on accounting topics of interest to national banks and federal savings associations. See further discussion of the BAAS in paragraph 7.82 of this guide. Topic 10, "Acquisitions, Corporate Reorganizations, and Consolidations," of the BAAS includes interpretations and responses on (a) acquisitions, (b) intangible assets, (c) push-down accounting, (d) corporate reorganizations, and (e) related party transactions. Readers are encouraged to view this publication under the "Publications—Bank Management" page at www.occ.gov.
- 19.06 SEC Codification of Staff Accounting Bulletins topic 11(N), "Disclosures of The Impact of Assistance From Federal Financial Institution Regulatory Agencies," discusses accounting for transfers of nonperforming assets by financial institutions and disclosure of the impact of financial assistance from regulators. SEC Codification of Staff Accounting Bulletins topic 11(N) states the SEC staff's belief that users of financial statements must be able to

assess the impact of credit and other risks on a company following a regulatory assisted acquisition, transfer, or other reorganization on a basis comparable with that disclosed by other institutions, that is, as if the assistance did not exist. In that regard, the SEC staff believes that the amount of regulatory assistance should be separately disclosed and should be separately identified in the statistical information furnished pursuant to Industry Guide 3, *Statistical Disclosures by Bank Holding Companies*, to the extent that it affects such information. Further, the nature, extent, and impact of such assistance should be fully disclosed in management's discussion and analysis.

19.07 On June 7, 2010, the OCC, the Board of Governors of the Federal Reserve System (Federal Reserve), the FDIC, the National Credit Union Administration (NCUA), and the Office of Thrift Supervision (prior to its transfer of powers to the OCC, the Federal Reserve, and the FDIC)<sup>2</sup> issued *Interagency Supervisory Guidance on Bargain Purchases and FDIC- and NCUA-Assisted Acquisitions* to address supervisory considerations related to bargain purchase gains and the impact such gains have on the application (licensing) approval process. This guidance also highlights the accounting and reporting requirements unique to business combinations resulting in bargain purchase gains and FDIC- and NCUA-assisted acquisitions of failed institutions. Although this guidance principally focuses on bargain purchase gains, it is also relevant to business combinations in general. Readers can access the full text of this guidance from the OCC, the Federal Reserve, the FDIC, or the NCUA websites.

### **Accounting and Financial Reporting**

#### © Update 19-1 Accounting and Reporting: Credit Losses

FASB Accounting Standards Update (ASU) No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, issued in June 2016, is effective for fiscal years of public business entities that are SEC filers beginning after December 15, 2019, including interim periods within those fiscal years.

For all other public business entities, the amendments in FASB ASU No. 2016-13 are effective for fiscal years beginning after December 15, 2020, including interim periods within those fiscal years.

For all other entities, including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting, FASB ASU No. 2016-13 is effective for fiscal years beginning after December 15, 2020, and interim periods within fiscal years beginning after December 15, 2021.

Early application is permitted for all entities as of the fiscal years beginning after December 15, 2018, *including* interim periods within those fiscal years.

FASB ASU No. 2016-13 creates FASB ASC 326, Financial Instruments—Credit Losses, to amend guidance on reporting credit losses for assets held at amortized cost basis and available-for-sale debt securities.

<sup>&</sup>lt;sup>2</sup> See chapter 1, "Industry Overview—Banks and Savings Institutions," of this guide for further discussion on the Office of Thrift Supervision transfer of powers.

For assets held at amortized cost basis, FASB ASC 326 eliminates the probable initial recognition threshold in current GAAP and, instead, requires an entity to reflect its current estimate of all expected credit losses. The allowance for credit losses is a valuation account that is deducted from the amortized cost basis of the financial assets to present the net amount expected to be collected.

For available-for-sale debt securities, credit losses should be measured in a manner similar to current GAAP. However, FASB ASC 326 will require that credit losses be presented as an allowance rather than as a write-down.

FASB ASU No. 2016-13 affects entities holding financial assets and net investment in leases that are not accounted for at fair value through net income. The amendments affect loans, debt securities, trade receivables, net investments in leases, off-balance-sheet credit exposures, reinsurance receivables, and any other financial assets not excluded from the scope that have the contractual right to receive cash.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-13, see appendix G, "Accounting for Financial Instruments," of this guide.

- 19.08 Accounting for business combinations involving financial institutions is similar to that for entities in other industries. According to paragraphs 2–3 of FASB ASC 805-10-15, the guidance in FASB ASC 805 applies to all entities and to all transactions that meet the definition of a business combination. Certain specific qualifications and exceptions are listed in FASB ASC 805-10-15-4.
- 19.09 Per FASB ASC 805-10-25-1, an entity should determine whether a transaction or other event is a business combination which requires that the assets acquired and liabilities assumed constitute a business, and each business combination should be accounted for by applying the acquisition method. If the assets acquired are not a business, the reporting entity should account for the transaction or other event as an asset acquisition. An entity should account for each business combination by applying the acquisition method.
- **19.10** FASB ASC 805-10-05-4 explains that the acquisition method requires all of the following steps:
  - a. Identifying the acquirer (see FASB ASC 805-10)
  - b. Determining the acquisition date (see FASB ASC 805-10)
  - c. Recognizing and measuring the identifiable assets acquired, the liabilities assumed, and any noncontrolling interest in the acquiree (see FASB ASC 805-20)
  - d. Recognizing and measuring goodwill or a gain from a bargain purchase (see FASB ASC 805-30)
- **19.11** Regarding paragraph 19.10a–b, paragraphs 4 and 6 of FASB ASC 805-10-25 state that for each business combination, one of the combining entities should be identified as the acquirer. The acquirer should identify the acquisition date, which is the date on which it obtains control of the acquiree.

19.12 Regarding item in paragraph 19.10c, recognition of identifiable assets acquired and liabilities assumed is subject to the conditions specified in paragraphs 2–3 of FASB ASC 805-20-25. However, an entity (the acquirer) within the scope of FASB ASC 805-20-15-2 may elect to apply the accounting alternative for the recognition of identifiable assets acquired in a business combination as described in paragraphs 29–32 of FASB ASC 805-20-25. Under the acquisition method, the acquirer should measure the identifiable assets acquired, the liabilities assumed, and any noncontrolling interest in the acquiree at their acquisition-date fair values, as explained in paragraphs 1–2 of FASB ASC 805-20-30, with exceptions to the measurement principle identified and their accounting treatment addressed in paragraphs 10–23 of FASB ASC 805-20-30.

### Considerations for Private Companies that Elect to use Standards as Issued by the Private Company Council

FASB ASU No. 2014-18. Business Combinations (Topic 805): Accounting for Identifiable Intangible Assets in a Business Combination (a consensus of the Private Company Council), denotes that a private company that elects the accounting alternative to recognize or otherwise consider the fair value of intangible assets as a result of any in-scope transactions should no longer recognize separately from goodwill (1) customer-related intangible assets unless they are capable of being sold or licensed independently from the other assets of the business and (2) noncompetition agreements. An entity that elects the accounting alternative in FASB ASU No. 2014-18 must adopt the private company alternative to amortize goodwill as described in FASB ASU No. 2014-02, Intangibles—Goodwill and Other (Topic 350): Accounting for Goodwill. However, an entity that elects the accounting alternative in FASB ASU No. 2014-02 is not required to adopt the amendments in FASB ASU No. 2014-18. Readers of this guide are encouraged to consult the full text of FASB ASU No. 2014-18, available at www.fasb.org.

- 19.13 FASB ASC 805-20-30-4 states that the acquirer should not recognize a separate valuation allowance as of the acquisition date for assets acquired in a business combination that are measured at their acquisition-date fair values because the effects of uncertainty about future cash flows are included in the fair value measure. For example, because FASB ASC 805-20 requires the acquirer to measure acquired receivables, including loans, at their acquisition date fair values, the acquirer does not recognize a separate valuation allowance for the contractual cash flows that are deemed to be uncollectible at that date.
- **19.14** Regarding paragraph 19.10d and as stated in FASB ASC 805-30-30-1, the acquirer should recognize goodwill as of the acquisition date, measured as the excess of items (a) over (b):
  - a. The aggregate of the following:
    - The consideration transferred measured in accordance with FASB ASC 805-30-30, which generally requires acquisition-date fair value (see FASB ASC 805-30-30-7)
    - The fair value of any noncontrolling interest in the acquiree

- iii. In a business combination achieved in stages, the acquisition-date fair value of the acquirer's previously held equity interest in the acquiree
- b. The net of the acquisition-date amounts of the identifiable assets acquired and the liabilities assumed measured in accordance with FASB ASC 805
- 19.15 FASB ASC 350, Intangibles—Goodwill and Other, provides guidance on financial accounting and reporting related to goodwill and other intangibles, other than the accounting at acquisition for goodwill and other intangibles acquired in a business combination. Acquisition guidance for intangible assets acquired in a business combination is provided in FASB ASC 805-20 and guidance on recognition and initial measurement of goodwill acquired in a business combination is provided in FASB ASC 805-30. See chapter 12, "Other Assets, Other Liabilities, and Other Investments," of this guide for a discussion of the requirements of FASB ASC 350.
- **19.16** The consideration transferred in a business combination should be measured at fair value, which should be calculated as the sum of the acquisition-date fair values of the assets transferred by the acquirer, the liabilities incurred by the acquirer to former owners of the acquiree, and the equity interests issued by the acquirer, as stated in FASB ASC 805-30-30-7.
- 19.17 For certain assets and liabilities acquired, such as loans and deposits, for which there is not an active market, determining fair values usually involves estimating cash flows and discounting those cash flows at prevailing risk adjusted market rates of interest. However, other valuation techniques may be used to determine *fair value* as defined in FASB ASC 820, *Fair Value Measurement*. Core deposit intangibles represent a separate asset, and, therefore, demand deposits are generally valued at their face amount plus any accrued interest.

#### Purchase of a Loan or Group of Loans

- 19.18 FASB ASC 310-20-30-5 explains that the initial investment in a loan or group of loans acquired in other than a business combination should include the amount paid to the seller plus any fees paid or less any fees received. FASB ASC 310-20-25-22 explains that the initial investment frequently differs from the related loan's principal amount at the date of purchase. This difference should be recognized as an adjustment of yield over the life of the loan, in accordance with FASB ASC 310-20-35-15 except for loans acquired with deteriorated credit quality, which are discussed in more detail beginning in paragraph 19.21.
- 19.19 FASB ASC 310-20-30-5 further explains that, in applying the provisions of FASB ASC 310-20 to loans purchased as a group, the purchaser may allocate the initial investment to the individual loans or may account for the initial investment in the aggregate. In accordance with FASB ASC 310-20-35-16, the cash flows provided by the underlying loan contracts should be used to apply the interest method, except as set forth in FASB ASC 310-20-35-26. If prepayments are not anticipated pursuant to that paragraph and prepayments occur or a portion of the purchased loans is sold, a proportionate amount of the related deferred fees and purchase premium or discount should be recognized in income so that the effective interest rate on the remaining portion of loans continues unchanged.

19.20 Loan receivables may be acquired with a fair value (if acquired through a business combination) or relative fair value (if acquired through an asset purchase) lower than the contractual amounts due (principal amount) that are not required to be accounted for in accordance with the guidance in FASB ASC 310-30, which is addressed in the subsequent paragraphs. The discount relating to such acquired loan receivables must be accounted for subsequently through accretion. In a letter to the SEC on December 18, 2009, the AICPA addressed accounting in subsequent periods for discount accretion associated with such loan receivables. The letter confirmed the understanding that the SEC Staff would not object to an accounting policy based on contractual cash flows or an accounting policy based on expected cash flows. Readers may access this letter on the AICPA website.

## Loans and Debt Securities Acquired With Deteriorated Credit Quality

19.21 FASB ASC 310-30 provides recognition, measurement, and disclosure guidance regarding loans acquired with evidence of deterioration of credit quality since origination acquired by completion of a transfer for which it is probable, at acquisition, that the investor will be unable to collect all contractually required payments receivable. The application of FASB ASC 310-30 requires that each loan should be evaluated individually to determine whether the loan meets the scope criteria of FASB ASC 310-30-15-2. A loan may be acquired at a discount because of a change in credit quality or rate or both, according to FASB ASC 310-30-30-2. When a loan is acquired at a discount that relates, at least in part, to the loan's credit quality, the effective interest rate is the discount rate that equates the present value of the investor's estimate of the loan's future cash flows to the purchase price of the loan. There is further discussion on accounting for loans and debt securities acquired with deteriorated credit quality in chapter 8, "Loans," of this guide beginning with paragraph 8.110.

## Special Considerations in Applying the Acquisition Method to Combinations of Mutual Entities

19.22 Although similar in many ways to other businesses, mutual entities have distinct characteristics that arise primarily because their members are both customers and owners, according to FASB ASC 805-30-55-4. Members of mutual entities generally expect to receive benefits for their membership, often in the form of reduced fees charged for goods and services or patronage dividends. The portion of patronage dividends allocated to each member is often based on the amount of business the member did with the mutual entity during the year.

19.23 As stated in FASB ASC 805-30-55-3, when two mutual entities combine, the fair value of the equity or member interests in the acquiree (or the fair value of the acquiree) may be more reliably measurable than the fair value of the member interests transferred by the acquirer. In that situation, paragraphs 2–3 of FASB ASC 805-30-30 require the acquirer to determine the amount of goodwill by using the acquisition-date fair value of the acquiree's equity interests instead of the acquisition-date fair value of the acquirer's equity interests transferred as consideration.

**19.24** FASB ASC 805-30-55-5 states that a fair value measurement of a mutual entity should include the assumptions that market participants would

make about future member benefits as well as any other relevant assumptions market participants would make about the mutual entity. For example, an estimated cash flow model may be used to determine the fair value of a mutual entity. The cash flows used as inputs to the model should be based on the expected cash flows of the mutual entity, which are likely to reflect reductions for member benefits, such as reduced fees charged for goods and services.

**19.25** The acquirer in a combination of mutual entities should recognize the acquiree's net assets as a direct addition to capital or equity in its statement of financial position, not as an addition to retained earnings, which is consistent with the way in which other types of entities apply the acquisition method.

#### Impairment and Disposal Accounting for Certain Acquired Long Term Customer Relationship Intangible Assets

19.26 The impairment and disposal provisions of FASB ASC 360-10 apply to long term customer relationship intangible assets, except for servicing assets, recognized in the acquisition of a financial institution. Examples of long term customer relationship intangible assets may include depositor and borrower relationship intangible assets, and credit cardholder intangible assets. See chapter 11, "Real Estate Investments, Real Estate Owned, and Other Foreclosed Assets," of this guide for impairment guidance on intangibles addressed in FASB ASC 360-10.

19.27 Servicing assets are tested for impairment under paragraphs 9–14 of FASB ASC 860-50-35. See chapter 10, "Transfers and Servicing and Variable Interest Entities," of this guide for further discussion over servicing assets and the related requirements set forth in FASB ASC 860, *Transfers and Servicing*.

#### **Branch Acquisitions**

19.28 Depository institutions may acquire branch office locations. Such transactions typically involve the assumption of deposit liabilities by the acquiring institution in exchange for the receipt of a lesser amount of cash, or other assets, such as loans.

19.29 In accordance with FASB ASC 805, an entity should determine whether a transaction or other event is a business combination by applying the definition of a business combination. If the assets acquired are not a business, the reporting entity should account for the transaction or the other event as an asset acquisition, as stated in FASB ASC 805-10-25-1. As discussed in FASB ASC 350-30-25-2 and consistent with FASB ASC 805-50-30-3, the cost of a group of assets acquired in a transaction other than a business combination should be allocated to the individual assets acquired based on their relative fair values and should not give rise to goodwill.

**19.30** The "Acquisition of Assets Rather than a Business" subsections of FASB ASC 805-50 address a transaction in which the assets acquired and liabilities assumed do not constitute a business, such a transaction is accounted for as an asset acquisition.

<sup>&</sup>lt;sup>3</sup> The referenced discussion originated from paragraph A133 of FASB Statement No. 141(R), *Business Combinations*; however, upon restructuring of the accounting and reporting standards into the codified standards, the discussion was not carried forward. Although the discussion was not carried forward into the codified standards, the impairment and disposal provisions of FASB ASC 360-10 are still considered applicable to long term customer relationship intangible assets.

### Auditing Considerations<sup>4</sup>

19.31 Business combinations may take many forms and generally involve complex transactions that have a pervasive effect on the financial statements impacting numerous financial statement line items and assertions. The primary objective of audit procedures for business combinations is to obtain reasonable assurance that the transaction is properly recorded in accordance with GAAP. Supervisory management personnel need to review and adequately support accounting entries made to record the transaction initially including the recognition and measurement of the assets and liabilities of the acquired entity and those required in subsequent years. Supervisory management personnel also need to evaluate the disclosures made with respect to the business combinations. Moreover, subsequent to the acquisition date management should review assumptions used in assigning values to assets and liabilities for continuing validity.

**19.32** AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards), addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13-.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. See chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide for further discussion of the components of internal control. To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and the relevant assertion level for classes of transactions, account balances, and disclosures.

19.33 Paragraph .06 of AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards), states that the auditor should design and perform further audit procedures whose nature, timing, and extent are based on, and responsive to, the assessed risks of material misstatement at the relevant assertion level. Irrespective of the assessed risks of material misstatement, paragraph .18 of AU-C section 330 states that the auditor should design and perform substantive procedures for all relevant assertions related to each material class of transactions, account balance, and disclosure. Paragraph .445 of AU-C section 330 further states that this requirement reflects the facts that (a) the auditor's assessment of risk is judgmental and may not be sufficiently precise to identify all risks of material misstatement, and (b) inherent limitations to internal control exist, including management override.

<sup>&</sup>lt;sup>4</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of *issuers* and *nonissuers* as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

19.34 AU-C section 501, Audit Evidence—Specific Considerations for Selected Items (AICPA, Professional Standards), addresses specific considerations by the auditor in obtaining sufficient appropriate audit evidence, in accordance with AU-C section 330; AU-C section 500, Audit Evidence (AICPA, Professional Standards); and other relevant AU-C sections, regarding certain aspects of (a) investments in securities and derivative instruments; (b) inventory; (c) litigation, claims, and assessments involving the entity; and (d) segment information in an audit of financial statements. In addition, AU-C section 540, Auditing Accounting Estimates, Including Fair Value Accounting Estimates, and Related Disclosures (AICPA, Professional Standards), addresses the auditor's responsibilities relating to accounting estimates, including fair value accounting estimates and related disclosures, in an audit of financial statements.<sup>5</sup> Specifically, it expands on how AU-C section 315, AU-C section 330, and other relevant AU-C sections are to be applied with regard to accounting estimates. It also includes requirements and guidance related to misstatements of individual accounting estimates and indicators of possible management bias. The companion AICPA Audit Guide Special Considerations in Auditing Financial Instruments provides background information about financial instruments and discussion of audit considerations relating to financial instruments.

19.35 Many fair values assigned to an acquired depository institution's assets and liabilities will be based on valuations by independent third-party appraisals. In applying audit procedures over these fair values, the auditor often relies on representations of independent experts. AU-C section 500 addresses the auditor's use of the work of an individual or organization possessing expertise in a field other than accounting or auditing, whose work in that field is used by the entity to assist the entity in preparing the financial statements (defined in the auditing standards as a management's specialist). AU-C section 620, Using the Work of an Auditor's Specialist (AICPA, Professional Standards), addresses the auditor's responsibilities relating to the work of an individual or organization possessing expertise in a field other than accounting or auditing when that work is used to assist the auditor in obtaining sufficient appropriate audit evidence (defined in the auditing standards as an auditor's specialist). See paragraph .A1 of AU-C section 620 for further examples of expertise in a field other than accounting or auditing.

**19.36** Using the work of a management's specialist. If information to be used as audit evidence has been prepared using the work of a management's specialist, paragraph .08 of AU-C section 500 states that the auditor should, to the extent necessary, taking into account the significance of that specialist's work for the auditor's purposes,

- a. evaluate the competence, capabilities, and objectivity of that specialist;
- b. obtain an understanding of the work of that specialist; and
- evaluate the appropriateness of that specialist's work as audit evidence for the relevant assertion.

<sup>&</sup>lt;sup>5</sup> In response to issues identified during the inspection of accounting firms and comments received from members of the Standing Advisory Group, the PCAOB staff is drafting a proposed standard on auditing accounting estimates, including fair value measurements and related disclosures. Readers should be alert to developments and are encouraged to refer to the Standard-Setting Agenda at www.pcaobus.org.

- 19.37 Information regarding the competence, capabilities, and objectivity of a management's specialist may come from a variety of sources, such as knowledge of that specialist's qualifications, membership in a professional body or industry association, license to practice, or other forms of external recognition (a listing of additional sources is addressed in paragraph .A39 of AU-C section 500). Further application and explanatory material regarding the reliability of information produced by a management's specialist is addressed in paragraphs .A35—.A49 of AU-C section 500.
- **19.38** Using the work of an auditor's specialist. In accordance with paragraph .09 of AU-C section 620, the auditor should evaluate whether the auditor's specialist has the necessary competence, capabilities, and objectivity for the auditor's purposes. In the case of an auditor's external specialist, the evaluation of objectivity should include inquiry regarding interests and relationships that may create a threat to the objectivity of the auditor's specialist.
- 19.39 Paragraphs .12—.13 of AU-C section 620 state that the auditor should evaluate the adequacy of the work of the auditor's specialist for the auditor's purposes, including
  - a. the relevance and reasonableness of the findings and conclusions
    of the auditor's specialist and their consistency with other audit
    evidence.
  - b. if the work of the auditor's specialist involves the use of significant assumptions and methods,  $^6$ 
    - obtaining an understanding of those assumptions and methods and
    - ii. evaluating the relevance and reasonableness of those assumptions and methods in the circumstances, giving consideration to the rationale and support provided by the specialist, and in relation to the auditor's other findings and conclusions.
  - c. if the work of the auditor's specialist involves the use of source data that is significant to the work of the auditor's specialist, the relevance, completeness, and accuracy of that source data.
- **19.40** If the auditor determines that the work of the auditor's specialist is not adequate for the auditor's purposes, the auditor should
  - a. agree with the auditor's specialist on the nature and extent of further work to be performed by the auditor's specialist or
  - perform additional audit procedures appropriate to the circumstances.
- **19.41** The audit objectives and procedures applied to an acquisition of assets and assumption of liabilities (for example a branch acquisition) will generally be substantially the same as the objectives and procedures applied to a business combination, as described more fully in the preceding paragraphs.

<sup>&</sup>lt;sup>6</sup> When the work of an auditor's specialist involves the use of significant assumptions and methods, the appropriateness and reasonableness of those assumptions and methods used and their application are the responsibility of the auditor's specialist. See paragraph .A40 of AU-C section 620, Using the Work of an Auditor's Specialist (AICPA, Professional Standards), for further discussion of factors relevant to the auditor's evaluation of those assumptions and methods.

Fair Value 559

### Chapter 20

### Fair Value

Gray shaded text in this chapter reflects guidance issued but not yet effective as of the date of this guide, July 1, 2016, but becoming effective on or prior to December 31, 2016, exclusive of any option to early adopt ahead of the mandatory effective date. Unless otherwise indicated, all unshaded text reflects guidance that was already effective as of the date of this guide.

#### Introduction

**20.01** The following section provides a discussion about fair value measurements and disclosures and the fair value option. FASB *Accounting Standards Codification* (ASC) 820, *Fair Value Measurement*, defines *fair value*, establishes a framework for measuring fair value, and requires certain disclosures about fair value measurements. FASB ASC 825, *Financial Instruments*, allows entities to choose, at specified election dates, to measure eligible items at fair value (the fair value option). The following paragraphs summarize FASB ASC 820 and FASB ASC 825 but are not intended as a substitute for reviewing this guidance in its entirety.

### Accounting and Financial Reporting<sup>1</sup>

#### Definition of Fair Value

**20.02** FASB ASC 820-10-20 defines *fair value* as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. The FASB ASC glossary defines an *orderly transaction* as a transaction that assumes exposure to the market for a period before the measurement date to allow for marketing activities that are usual and customary for transactions involving such assets and liabilities; it is not a forced transaction (for example, a forced liquidation or distress sale).

#### Unit of Account

**20.03** For recognition and disclosure purposes, the reporting entity must determine the appropriate unit of account to apply the provision of FASB ASC 820. FASB ASC 820-10-35-2E states that whether the asset or liability is a standalone asset or liability, a group of assets, a group of liabilities, or a group of assets and liabilities depends on its unit of account. The unit of account is generally based on the applicable guidance that requires or permits fair value measurement, except for instances explicitly discussed in FASB ASC 820.

©2016, AICPA AAG-DEP 20.03

<sup>&</sup>lt;sup>1</sup> The Office of the Comptroller of the Currency's Bank Accounting Advisory Series (BAAS) is updated periodically to express the Office of the Chief Accountant's current views on accounting topics of interest to national banks and federal savings associations. See further discussion of the BAAS in paragraph 7.82 of this guide. Topic 11D, "Fair Value Accounting," includes interpretations related to fair value measurement. Readers are encouraged to view this publication under the "Publications—Bank Management" page at www.occ.gov.

#### The Principal (or Most Advantageous) Market

20.04 FASB ASC 820-10-35-5 states that a fair value measurement assumes that the transaction to sell the asset or transfer the liability takes place either in the principal market for the asset or liability or, in the absence of a principal market, in the most advantageous market for the asset or liability. The FASB ASC glossary defines the *principal market* as the market with the greatest volume and level of activity for the asset or liability. The most advantageous market, as defined in the FASB ASC glossary, is the market that maximizes the amount that would be received to sell the asset or minimizes the amount that would be paid to transfer the liability, after taking into account transaction costs and transportation costs. Paragraphs 5A-6 of FASB ASC 820-10-35 state that a reporting entity need not undertake an exhaustive search of all possible markets to identify the principal market or, in the absence of a principal market, the most advantageous market, but it should take into account all information that is reasonably available. If there is a principal market for the asset or liability, the fair value measurement should represent the price in that market (whether that price is directly observable or estimated using another valuation technique), even if the price in a different market is potentially more advantageous at the measurement date.

20.05 The reporting entity must have access to the principal (or most advantageous) market at the measurement date. Because different entities (and businesses within those entities) with different activities may have access to different markets, the principal (or most advantageous) market for the same asset or liability might be different for different entities (and businesses within those entities). Therefore, paragraphs 6A–6B of FASB ASC 820-10-35 state that the principal (or most advantageous) market (and thus, market participants), should be considered from the perspective of the reporting entity, thereby allowing for differences between and among entities with different activities. Although a reporting entity must be able to access the market, the reporting entity does not need to be able to sell the particular asset or transfer the particular liability on the measurement date to be able to measure fair value on the basis of the price in that market.

20.06 Paragraphs 3–3A of FASB ASC 820-10-30 explain that, in many cases, the transaction price will equal the fair value (for example, that might be the case when, on the transaction date, the transaction to buy an asset takes place in the market in which the asset would be sold). When determining whether fair value at initial recognition equals the transaction price, a reporting entity should take into account factors specific to the transaction and to the asset or liability. Even when there is no observable market to provide pricing information about the sale of an asset or the transfer of a liability at the measurement date, FASB ASC 820-10-35-6C states that a fair value measurement should assume that a transaction takes place at that date, considered from the perspective of a market participant that holds the asset or owes the liability. That assumed transaction establishes a basis for estimating the price to sell the asset or to transfer the liability.

**20.07** The transaction price might not represent the fair value of an asset or a liability at initial recognition under certain conditions. For example, if the transaction is conducted between related parties, the seller is under duress, the unit of account differs, the market in which the transaction was conducted differs from the principal (or most advantageous) market that would be used by the entity. Paragraphs 46–49 of FASB ASC 820-10-55 (example 5) illustrate

Fair Value 561

when the price in a transaction involving a derivative instrument might (and might not) equal the fair value of the instrument at initial recognition.

#### **Market Participants**

**20.08** Fair value measurement should be based on the assumptions of market participants, as stated in FASB ASC 820-10-35-9, where a reporting entity should measure the fair value of an asset or a liability using the assumptions that a market participant would use in pricing the asset or liability. The reporting entity is not required to identify specific market participants but rather identify characteristics that distinguish market participants generally considering factors specific to the asset or liability, the principal (or most advantageous) market for the asset or liability, and market participants with which the reporting entity would enter into a transaction in the market.

#### The Price

20.09 Paragraphs 9A–9C of FASB ASC 820-10-35 provide additional guidance on the price. Fair value is the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction in the principal (or most advantageous) market at the measurement date under current market conditions (that is, an exit price) regardless of whether that price is directly observable or estimated using another valuation technique. The price should not be adjusted for transaction costs because they are not a characteristic of an asset or a liability; rather, they are specific to the transaction and will differ depending on the transaction. Further, transaction costs do not include transportation costs. The price in the principal (or most advantageous) market should be adjusted for the costs, if any, that would be incurred to transport the asset from its current location to that market.

#### **Application of Fair Value Measurement to Nonfinancial Assets**

**20.10** Highest and best use for nonfinancial assets. FASB ASC 820-10-35-10A states that a fair value measurement of a nonfinancial asset takes into account a market participant's ability to generate economic benefits by using the asset in its highest and best use or by selling it to another market participant that would use the asset in its highest and best use. (Application of the highest and best use concept is limited to the fair value measurement of nonfinancial assets and cannot be used to measure the fair value of financial assets.)

- **20.11** FASB ASC 820-10-35-10B states that the highest and best use of a nonfinancial asset takes into account the use of the asset that is physically possible, legally permissible, and financial feasible, as follows:
  - a. A use that is physically possible takes into account the physical characteristics of the asset that market participants would take into account when pricing the asset (for example, the location or size of a property).
  - b. A use that is legally feasible takes into account any legal restrictions on the use of the asset that market participants would take into account when pricing the asset (for example, the zoning regulation application to a property).
  - c. A use that is financially feasible takes into account whether a use of the asset that is physically possible and legally permissible generates adequate income or cash flows (taking into account the costs

of converting the asset to that use) to produce an investment return that market participants would require from an investment in that asset put to that use.

- **20.12** Even if the reporting entity intends a different use, paragraphs 10C–10D of FASB ASC 820-10-35 state that the highest and best use is determined from the perspective of market participants. However, a reporting entity's current use of a nonfinancial asset is presumed to be its highest and best use unless market or other factors suggest that a different use by market participants would maximize the value of the asset. To protect its competitive position, or for other reasons, a reporting entity may intend not to use an acquired nonfinancial asset actively, or it may intend not to use the asset according to its highest and best use. For example, that might be the case for an acquired intangible asset that the reporting entity plans to use defensively by preventing others from using it. Nevertheless, the reporting entity should measure the fair value of a nonfinancial asset assuming its highest and best use by market participants.
- **20.13** Valuation premise for nonfinancial assets. FASB ASC 820-10-35-10E states that the highest and best use of a nonfinancial asset establishes the valuation premise used to measure the fair value of the asset, as follows:
  - a. The highest and best use of a nonfinancial asset might provide maximum value to market participants through its use in combination with other assets as a group (as installed or otherwise configured for use) or in combination with other assets and liabilities (for example, a business).
    - i. If the highest and best use of the asset is to use the asset in combination with other assets or with other assets and liabilities, the fair value of the asset is the price that would be received in a current transaction to sell the asset assuming that the asset would be used with other assets or with other assets and liabilities and that those assets and liabilities (that is, its complementary assets and the associated liabilities) would be available to market participants.
    - ii. Liabilities associated with the asset and with the complementary assets include liabilities that fund working capital, but do not include liabilities used to fund assets other than those within the group of assets.
    - iii. Assumptions about the highest and best use of a nonfinancial asset should be consistent for all of the assets (for which highest and best use is relevant) of the group of assets or the group of assets and liabilities within which the asset would be used.
  - b. The highest and best use of a nonfinancial asset might provide maximum value to market participants on a standalone basis. If the highest and best use of the asset is to use it on a standalone basis, the fair value of the asset is the price that would be received in a current transaction to sell the asset to market participants that would use the asset on a standalone basis.

FASB ASC 820-10-55-25 (example 1) illustrates the application of the highest and best use and valuation premise concepts for nonfinancial assets.

Fair Value 563

## Application of Fair Value Measurement to Liabilities and Instruments Classified in a Reporting Entity's Shareholders' Equity<sup>2</sup>

**20.14** According to paragraphs 16–16AA of FASB ASC 820-10-35, a fair value measurement assumes that a financial or nonfinancial liability or an instrument classified in a reporting entity's shareholders' equity (for example, equity interests issued as consideration in a business combination) is transferred to a market participant at the measurement date. The transfer of such a liability or instrument assumes the following:

- a. A liability would remain outstanding and the market participant transferee would be required to fulfill the obligation. The liability would not be settled with the counterparty or otherwise extinguished on the measurement date.
- b. An instrument classified in a reporting entity's shareholders' equity would remain outstanding and the market participant transferee would take on the rights and responsibilities associated with the instrument. The instrument would not be canceled or otherwise extinguished on the measurement date.

Even when there is no observable market to provide pricing information about the transfer of such a liability or instrument (for example, because contractual or other legal restrictions prevent the transfer of such items), there might be an observable market for such items if they are held by other parties as assets (for example, a corporate bond or a call option on a reporting entity's shares). In all cases, a reporting entity should maximize the use of relevant observable inputs and minimize the use of unobservable inputs to meet the objective of a fair value measurement, which is to estimate the price at which an orderly transaction to transfer the liability or instrument classified in shareholders' equity would take place between market participants at the measurement date under current market conditions.

**20.15** Liabilities and instruments classified in a reporting entity's shareholders' equity held by other parties as assets. Paragraphs 16B–16BB of FASB ASC 820-10-35 state that when a quoted price for the transfer of an identical liability or a similar liability or instrument classified in a reporting entity's shareholders' equity is not available and the identical item is held by another party as an asset, a reporting entity should measure the fair value of the liability or equity instrument from the perspective of a market participant that holds the identical items as an asset at the measurement date. In such cases, a reporting entity should measure the fair value of the liability or equity instrument as follows:

- a. Using the quoted price in an active market (defined in the FASB ASC glossary as a market in which transactions for the asset or liability take place with sufficient frequency and volume to provide pricing information on an ongoing basis) for the identical item held by another party as an asset, if that price is available.
- b. If that price is not available, using other observable inputs, such as the quoted price in a market that is not active for the identical item held by another party as an asset.
- c. If the observable prices discussed in item (a) or item (b) are not available, using another valuation technique, such as

<sup>&</sup>lt;sup>2</sup> The guidance discussed in this section does not apply to share-based payment transactions.

- an income approach (for example, a present value technique that takes into account the future cash flows that a
  market participant would expect to receive from holding
  the liability or equity instrument as an asset; see FASB
  ASC 820-10-55-3F).
- a market approach (for example, using quoted prices for similar liabilities or instruments classified as shareholders' equity held by other parties as assets; see FASB ASC 820-10-55-3A).
- **20.16** According to FASB ASC 820-10-35-16D, a reporting entity should adjust the quoted price only if there are factors specific to the asset that are not applicable to the fair value measurement of the liability or equity instrument. A reporting entity should ensure that the price of the asset does not reflect the effect of a restriction preventing the sale of that asset. Some factors that may indicate that the quoted price of the asset should be adjusted include the following:
  - The quoted price for the asset relates to a similar (but not identical) liability or equity instrument held by another party as an asset. For example, the liability or equity instrument may have a particular characteristic (for example, the credit quality of the issuer) that is different from that reflected in the fair value of the similar liability or equity instrument held as an asset.
  - The unit of account for the asset is not the same as for the liability or equity instrument. For example, for liabilities, in some cases the price for an asset reflects a combined price for a package comprising both the amounts due from the issuer and a third-party credit enhancement. If the unit of account for the liability is not for the combined package, the objective is to measure the fair value of the issuer's liability, not the fair value of the combined package. Thus, in such cases, the reporting entity would adjust the observed price for the asset to exclude the effect of the third-party credit enhancement. See FASB ASC 820-10-35-18A for further guidance.
- **20.17** Liabilities and instruments classified in a reporting entity's shareholders' equity not held by other parties as assets. Paragraphs 16H–16I of FASB ASC 820-10-35 explain that when a quoted price for the transfer of an identical or a similar liability or instrument classified in a reporting entity's shareholders' equity is not available and the identical item is not held by another party as an asset, a reporting entity should measure the fair value of the liability or equity instrument using a valuation technique from the perspective of a market participant that owes the liability or has issued the claim on equity. For example, when applying a present value technique, a reporting entity might take into account either of the following:
  - The future cash outflows that a market participant would expect to incur in fulfilling the obligation, including the compensation that a market participant would require for taking on the obligation (see paragraphs 16J–16K of FASB ASC 820-10-35).
  - The amount that a market participant would receive to enter into or issue an identical liability or equity instrument, using the assumptions that market participants would use when pricing, the

identical item (for example, having the same credit characteristics) in the principal (or most advantageous) market for issuing a liability or an equity instrument with the same contractual terms.

**20.18** *Nonperformance risk*. Paragraphs 17–18 of FASB ASC 820-10-35 state that the fair value of a liability reflects the effect of nonperformance risk. Nonperformance risk includes, but may not be limited to, a reporting entity's own credit risk and is assumed to be the same before and after the transfer of the liability. When measuring the fair value of a liability, a reporting entity should take into account the effect of its credit risk (credit standing) and any other factors that might influence the likelihood that the obligation will or will not be fulfilled. That effect may differ depending on the liability. For example, whether the liability is an obligation to deliver cash (a financial liability) or an obligation to deliver goods or services (a nonfinancial liability), or the terms of credit enhancements related to the liability, if any.

FASB ASC 820-10-55-56 illustrates the effect of credit risk on the fair value measurement of a liability.

**20.19** Third-party credit enhancements. As stated in paragraph 18A of FASB ASC 820-10-35, the issuer of a liability issued with an inseparable third-party credit enhancement that is accounted for separately from the liability should not include the effect of the credit enhancement in the fair value measurement of the liability. Instead, the issuer would take into account its own credit standing and not that of the third-party guarantor when measuring the fair value of the liability.

**20.20** Restriction preventing the transfer of a liability or an instrument classified in a reporting entity's shareholders' equity. When measuring the fair value, FASB ASC 820-10-35-18B states that a reporting entity should not include a separate input or an adjustment to other inputs relating to the existence of a restriction that prevents the transfer of the item. The effect of that restriction is either implicitly or explicitly included in the other inputs to the fair value measurement.

# Application of Fair Value Measurement to Financial Assets and Financial Liabilities with Offsetting Positions in Market Risks or Counterparty Credit Risk

20.21 Paragraphs 18D–18L of FASB ASC 820-10-35 provide fair value guidance to reporting entities that hold a group of financial assets and financial liabilities with offsetting positions in market risks or counterparty credit risks. The guidance permits a reporting entity that manages a group of financial assets and financial liabilities on the basis of its net exposure to either market risks or credit risk to apply an exception to FASB ASC 820 for measuring fair value. That exception permits a reporting entity to measure the fair value of a group of financial assets and financial liabilities on the basis of the price that would be received to sell a net long position (that is, an asset) for a particular risk exposure or paid to transfer a net short position (that is, a liability) for a particular risk exposure in an orderly transaction between market participants at the measurement date under current market conditions. Accordingly, a reporting entity should measure the fair value of the group of financial assets and financial liabilities consistently with how market participants would price the net risk exposure at the measurement date.

- **20.22** A reporting entity is permitted to use the exception in the preceding paragraph only if the reporting entity does all of the following:
  - a. Manages the group of financial assets and financial liabilities on the basis of the reporting entity's net exposure to a particular market risk (or risks) or to the credit risk of a particular counterparty in accordance with the reporting entity's documented risk management or investment strategy
  - b. Provides information on that basis about the group of financial assets and financial liabilities to the reporting entity's management
  - c. Is required or has elected to measure those financial assets and financial liabilities at fair value in the statement of financial position at the end of each reporting period

The exception does not pertain to financial statement presentation. In some cases, the basis for the presentation of financial instruments in the statement of financial position differs from the basis for the measurement of financial instruments (for example, if a FASB ASC topic does not require or permit financial instruments to be presented on a net basis). In such cases, a reporting entity may need to allocate the portfolio-level adjustments (see subsequent discussion of exposure to market risks and credit risk of a particular counterparty) to the individual assets or liabilities that make up the group of financial assets and financial liabilities managed on the basis of the reporting entity's net risk exposure. A reporting entity should perform such allocations on a reasonable and consistent basis using a methodology appropriate in the circumstances.

- **20.23** A reporting entity should make an accounting policy decision to use the portfolio measurement exception described in paragraph 20.21. A reporting entity that uses the exception should apply that accounting policy, including its policy for allocating bid-ask adjustments (see subsequent discussion of exposure to market risks in paragraph 20.24) and credit adjustments (see subsequent discussion of exposure to the credit risk of a particular counterparty in paragraph 20.27), if applicable, consistently from period to period for a particular portfolio. The exception applies only to financial assets and financial liabilities within the scope of FASB ASC 815, *Derivatives and Hedging*, or FASB ASC 825.
- **20.24** Exposure to market risks. When using the exception to measure the fair value of a group of financial assets and financial liabilities managed on the basis of the reporting entity's net exposure to a particular market risk (or risks), the reporting entity should apply the price within the bid-ask spread that is most representative of fair value in the circumstances to the reporting entity's net exposure to those market risks (see paragraphs 36C–36D of FASB ASC 820-10-35).
- **20.25** In regard to applying the portfolio measurement exception, a reporting entity should also ensure that the market risk (or risks) to which the reporting entity is exposed within that group of financial assets and financial liabilities is substantially the same. For example, a reporting entity would not combine the interest rate risk associated with a financial asset with the commodity price risk associated with a financial liability, because doing so would not mitigate the reporting entity's exposure to interest rate risk or commodity price risk. When using the exception, any basis risk resulting from the market risk parameters not being identical should be taken into account in the fair

value measurement of the financial assets and financial liabilities within the group.

**20.26** Similarly, the duration of the reporting entity's exposure to a particular market risk (or risks) arising from the financial assets and financial liabilities should be substantially the same. For example, a reporting entity that uses a 12-month futures contract against the cash flows associated with 12 months' worth of interest rate risk exposure on a 5-year financial instrument within a group made up of only those financial assets and financial liabilities measures the fair value of the exposure to 12-month interest rate risk on a net basis and the remaining interest rate risk exposure (that is, years 2–5) on a gross basis.

20.27 Exposure to the credit risk of a particular counterparty. When using the exception to measure the fair value of a group of financial assets and financial liabilities entered into with a particular counterparty, the reporting entity should include the effect of the reporting entity's net exposure to the credit risk of that counterparty or the counterparty's net exposure to the credit risk of the reporting entity in the fair value measurement when market participants would take into account any existing arrangements that mitigate credit risk exposure in the event of default (for example, a master netting agreement with the counterparty or an agreement that requires the exchange of collateral on the basis of each party's net exposure to the credit risk of the other party). The fair value measurement should reflect market participants' expectations about the likelihood that such an arrangement would be legally enforceable in the event of default.

#### **Valuation Techniques**

20.28 Paragraphs 24–24A of FASB ASC 820-10-35 explain that a reporting entity should use valuation techniques that are appropriate in the circumstances and for which sufficient data are available to measure fair value. Three widely used valuation techniques are the market approach, cost approach, and income approach. The main aspects of those approaches are summarized in paragraphs 3A–3G of FASB ASC 820-10-55. An entity should use valuation techniques consistent with one or more of these approaches to measure fair value:

- a. Market approach. The market approach uses prices and other relevant information generated by market transactions involving identical or comparable (that is, similar) assets, liabilities, or a group of assets and liabilities (for example, a business). Valuation techniques consistent with the market approach include matrix pricing and those that use market multiples derived from a set of comparables.
- b. Cost approach. The cost approach reflects the amount that would be required currently to replace the service capacity of an asset (often referred to as current replacement cost). Fair value is determined based on the cost to a market participant buyer to acquire or construct a substitute asset of comparable utility, adjusted for obsolescence. That is because a market participant buyer would not pay more for an asset than the amount for which it could replace the service capacity of that asset.

- c. Income approach. The income approach uses valuation techniques to convert future amounts (for example, cash flows or earnings) to a single present amount (discounted). The measurement is based on the value indicated by current market expectations about those future amounts. Valuation techniques consistent with the income approach include present value techniques, option-pricing models, and the multiperiod excess earnings method. See discussion beginning in paragraph 20.31 regarding guidance for using the present value technique.
- **20.29** FASB ASC 820-10-35-24B states that, in some cases, a single valuation technique will be appropriate (for example, when valuing an asset or liability using quoted prices in an active market for identical assets or liabilities). In other cases, multiple valuation techniques will be appropriate (for example, that might be the case when valuing a reporting unit). If multiple valuation techniques are used to measure fair value, the results (that is, respective indications of fair value) should be evaluated considering the reasonableness of the range of values indicated by those results. A fair value measurement is the point within that range that is most representative of fair value in the circumstances. Example 3 beginning in FASB ASC 820-10-55-35 illustrates the use of multiple valuation techniques.

**20.30** As explained in paragraphs 25–26 of FASB ASC 820-10-35, valuation techniques used to measure fair value should be applied consistently. However, a change in a valuation technique or its application is appropriate if the change results in a measurement that is equally or more representative of fair value in the circumstances. Such a change would be accounted for as a change in accounting estimate in accordance with the provisions of FASB ASC 250, Accounting Changes and Error Corrections. However, FASB ASC 250-10-50-5 explains that the disclosures in FASB ASC 250 for a change in accounting estimate are not required for revisions resulting from a change in a valuation technique or its application. See discussion of FASB ASC 820-10-50-2bbb for disclosure requirements when there has been a change in valuation techniques.

#### Present Value Techniques

**20.31** The components of a present value measurement. Paragraphs 4–19 of FASB ASC 820-10-55 and 820-10-55-20 describe the use of present value techniques to measure fair value. Those paragraphs neither prescribe the use of a single specific present value technique nor limit the use of present value techniques to measure fair value to the techniques discussed. Present value (that is, the application of the income approach) is a tool used to link future amounts (for example, cash flows or values) to a present amount using a discount rate. A fair value measurement of an asset or a liability using a present value technique captures all of the following elements from the perspective of market participants at the measurement date: (a) an estimate of future cash flows for the asset or liability being measured; (b) expectations about possible variations in the amount and timing of the cash flows representing the uncertainty inherent in the cash flows; (c) the time value of money, represented by the rate on risk-free monetary assets that have maturity dates or durations that coincide with the period covered by the cash flows and pose neither uncertainty in timing nor risk of default to the holder (that is, a risk-free interest rate); (d) the price for bearing the uncertainty inherent in the cash flows (that is, a risk premium); (e) other factors that market participants would take into account in the circumstances; and (f) for a liability, the nonperformance risk

relating to that liability, including the reporting entity's (that is, the obligor's) own credit risk.

**20.32** *General principles.* Present value techniques differ in how they capture the elements discussed in the preceding paragraph. FASB ASC 820-10-55-6 provides the general principles that govern the application of any present value technique used to measure fair value, as follows:

- Cash flows and discount rates should reflect assumptions that market participants would use when pricing the asset or liability.
- Cash flows and discount rates should take into account only factors attributable to the asset (or liability) being measured.
- To avoid double counting or omitting the effects of risk factors, discount rates should reflect assumptions that are consistent with those inherent in the cash flows. For example, a discount rate that reflects the uncertainty in expectations about future defaults is appropriate if using the contractual cash flows of a loan (that is, a discount rate adjustment technique), but is not appropriate if using expected (that is, probability-weighted) cash flows (that is, an expected present value technique) because the expected cash flows already reflect assumptions about the uncertainty in future defaults; instead, a discount rate that is commensurate with the risk inherent in the expected cash flows should be used.
- Assumptions about cash flows and discount rates should be internally consistent. For example, nominal cash flows, which include the effects of inflation, should be discounted at a rate that includes the effects of inflation.
- Discount rates should be consistent with the underlying economic factors of the currency in which the cash flows are denominated.

20.33 Risks and uncertainty. FASB ASC 820-10-55-9 describes how present value techniques differ in how they adjust for risk and in the type of cash flows they use. For example, the discount rate adjustment technique uses a risk-adjusted discount rate and contractual, promised, or most likely cash flows. In contrast, method 1 of the expected present value technique uses risk-adjusted expected cash flows and a risk-free rate. Expected cash flow, as defined in the FASB ASC glossary, is the probability-weighted average (that is, the mean of the distribution) of possible future cash flows. Method 2 of the expected present value technique uses expected cash flows that are not risk adjusted and a discount rate (this rate is different from the rate used in the discount rate adjustment technique) adjusted to include the risk premium that market participants require. The discount rate adjustment technique and two methods of expected present value techniques are discussed more fully in paragraphs 10–16 of FASB ASC 820-10-55.

#### The Fair Value Hierarchy

**20.34** Because fair value is a market-based measurement, FASB ASC 820-10-35-9 states that fair value should be measured using the assumptions that market participants would use in pricing the asset or liability (referred to as *inputs*), assuming that market participants act in their economic best interest. The FASB ASC glossary defines *inputs* as assumptions that market participants would use when pricing the asset or liability, including assumptions about risk, such as the risk inherent in a particular valuation technique

used to measure fair value (such as a pricing model) or the risk inherent in the inputs to the valuation technique. Inputs may be observable or unobservable (both as defined in the FASB ASC glossary):

- Observable inputs are developed using market data, such as publicly available information about actual events or transactions, and that reflect the assumptions that market participants would use when pricing the asset or liability.
- *Unobservable inputs* are inputs for which market data are not available and that are developed using the best information available about the assumptions that market participants would use when pricing the asset or liability.

Paragraphs 37–54B of FASB ASC 820-10-35 establish a fair value hierarchy that distinguishes between observable and unobservable inputs. FASB ASC 820-10-35-36 states that valuation techniques used to measure fair value should maximize the use of relevant observable inputs and minimize the use of unobservable inputs.

**20.35** The fair value hierarchy in FASB ASC 820 categorizes the inputs to valuation techniques used to measure fair value into three broad levels. The three levels are discussed as follows:

Level 1. FASB ASC 820-10-35-40 defines Level 1 inputs as quoted prices (unadjusted) in active markets for identical assets or liabilities that the reporting entity can access at the measurement date. The FASB ASC glossary defines active market as a market in which transactions for the asset or liability take place with sufficient frequency and volume to provide pricing information on an ongoing basis. Paragraphs 41-41C and 44 of FASB ASC 820-10-35 go on to explain that a quoted price in an active market provides the most reliable evidence of fair value and should be used without adjustment to measure fair value whenever available, except as specified in FASB ASC 820-10-35-41C. A Level 1 input will be available for many financial assets and financial liabilities, some of which might be exchanged in multiple active markets (for example, on different exchanges). Therefore, the emphasis within Level 1 is on determining both the principal market for the asset or liability or, in the absence of a principal market, the most advantageous market for the asset or liability and whether the reporting entity can enter into a transaction for the asset or liability at the price in the market for the asset or liability at the measurement date. If a reporting entity holds a position in a single asset or liability (including a position comprising a large number of identical assets or liabilities, such as a holding of financial instruments) and the asset or liability is traded in an active market, the fair value of the asset or liability should be measured within Level 1 as the product of the quoted price for the individual asset or liability and the quantity held by the reporting entity. That is the case, even if a market's normal daily trading volume is not sufficient to absorb the quantity held and placing orders to sell the position in a single transaction might affect the quoted price. FASB ASC 820-10-55-42 (example 4) illustrates the use of Level 1 inputs to measure the fair value of a financial asset that trades in multiple active markets with different prices.

- Level 2. FASB ASC 820-10-35-47 defines Level 2 inputs as inputs other than quoted prices included within Level 1 that are observable for the asset or liability, either directly or indirectly. Paragraphs 48-51 of FASB ASC 820-10-35 go on to explain that if the asset or liability has a specified (contractual) term, a Level 2 input must be observable for substantially the full term of the asset or liability. Adjustments to Level 2 inputs will vary depending on factors specific to the asset or liability. Those factors include the condition and location of the asset, the extent to which inputs relate to items that are comparable to the asset or liability (including those factors described in FASB ASC 820-10-35-16D [see paragraph 20.16]), and the volume or level of activity in the markets within which the inputs are observed. An adjustment to a Level 2 input that is significant to the entire measurement might result in a fair value measurement categorized within Level 3 of the fair value hierarchy if the adjustment uses significant unobservable inputs. Level 2 inputs include the following:
  - Quoted prices for similar assets or liabilities in active markets
  - Quoted prices for identical or similar assets or liabilities in markets that are not active
  - Inputs other than quoted prices that are observable for the asset or liability (for example, interest rates and yield curves observable at commonly quoted intervals, implied volatilities, or credit spreads)
  - Market-corroborated inputs (defined in the FASB ASC glossary as inputs that are derived principally from or corroborated by observable market data by correlation or other means).
- Level 3. Level 3 inputs, as defined in the FASB ASC glossary, are unobservable inputs for the asset or liability. Paragraphs 53-54A of FASB ASC 820-10-35 state that unobservable inputs should be used to measure fair value to the extent that relevant observable inputs are not available, thereby allowing for situations in which there is little, if any, market activity for the asset or liability at the measurement date. Unobservable inputs should reflect the assumptions that market participants would use when pricing the asset or liability (including assumptions about risk). Assumptions about risk include the risk inherent in a particular valuation technique used to measure fair value (such as a pricing model) and the risk inherent in the inputs to the valuation technique. A measurement that does not include an adjustment for risk would not represent a fair value measurement if market participants would include one when pricing the asset or liability. A reporting entity should develop unobservable inputs using the best information available in the circumstances, which might include the reporting entity's own data. In developing unobservable inputs, a reporting entity may begin with its own data, but it should adjust those data if reasonably available information indicates that other market participants would use different data or there is something particular to the reporting entity that is not available to other market

participants (for example, an energy-specific synergy). A reporting entity need not undertake exhaustive efforts to obtain information about market participant assumptions. However, a reporting entity should take into account all information about market participant assumptions that is reasonably available. Unobservable inputs developed in the manner described previously are considered market participant assumptions and meet the objectives of a fair value measurement.

As explained in FASB ASC 820-10-35-37A, in some cases, the inputs used to measure fair value of an asset or liability might be categorized within different levels of the fair value hierarchy. In those cases, the fair value measurement is categorized in its entirety in the same level of the fair value hierarchy as the lowest level input that is significant to the entire measurement. Assessing the significance of a particular input to the entire measurement requires judgment, taking into account factors specific to the asset or liability. Adjustments to arrive at measurements based on fair value, such as costs to sell when measuring fair value less costs to sell, should not be taken into account when determining the level of the fair value hierarchy within which a fair value measurement is categorized.

20.36 As discussed in paragraphs 38 and 38A of FASB ASC 820-10-35, the availability of relevant inputs and their relative subjectivity might affect the selection of appropriate valuation techniques (see FASB ASC 820-10-35-24 discussed in paragraph 20.28). However, the fair value hierarchy prioritizes the inputs to valuation techniques, not the valuation techniques used to measure fair value. For example, a fair value measurement developed using a present value technique might be categorized within Level 2 or Level 3, depending on the inputs that are significant to the entire measurement and the level of the fair value hierarchy within which those inputs are categorized. If an observable input requires an adjustment using an unobservable input and that adjustment results in a significantly higher or lower fair value measurement. the resulting measurement would be categorized within Level 3 of the fair value hierarchy. For example, if a market participant would take into account the effect of a restriction on the sale of an asset when estimating the price for the asset, a reporting entity would adjust the quoted price to reflect the effect of that restriction. If that quoted price is a Level 2 input and the adjustment is an unobservable input that is significant to the entire measurement, the measurement would be categorized within Level 3 of the fair value hierarchy.

## Categorizing Investments in Certain Entities That Calculate Net Asset Value per Share (or Its Equivalent) Within the Fair Value Hierarchy

**20.37** According to FASB ASC 820-10-35-54B, categorization within the fair value hierarchy of a fair value measurement of an investment within the scope of paragraphs 4–5 of FASB ASC 820-10-15 that is measured at net asset value per share (or its equivalent, for example member units or an ownership interest in partners' capital to which a proportionate share of net assets is attributed) requires judgment, considering the following:

• If a reporting entity has the ability to redeem its investment with the investee at net asset value per share (or its equivalent) at the measurement date, the fair value measurement of the investment should be categorized within Level 2 of the fair value hierarchy.

Fair Value 573

- If a reporting entity would never have the ability to redeem its investment with the investee at net asset value per share (or its equivalent), the fair value measurement of the investment should be categorized within Level 3 of the fair value hierarchy.
- If a reporting entity cannot redeem its investment with the investee at net asset value per share (or its equivalent) at the measurement date but the investment may be redeemable with the investee at a future date (for example, investments subject to a lockup or gate or investments whose redemption period does not coincide with the measurement date), the reporting entity should take into account the length of time until the investment will become redeemable in determining whether the fair value measurement of the investment should be categorized within Level 2 or Level 3 of the fair value hierarchy. For example, if the reporting entity does not know when it will have the ability to redeem the investment or it does not have the ability to redeem the investment in the near term at net asset value per share (or its equivalent), the fair value measurement of the investment should be categorized within Level 3 of the fair value hierarchy.

### Investments in Certain Entities That Calculate Net Asset Value per Share (or Its Equivalent) $^3$

FASB ASC 820-10-35-54B states that an investment within the scope of paragraphs 14-15 of FASB ASC 820-10-15 for which fair value is measured using net asset value per share (or its equivalent, for example member units or an ownership interest in partners' capital to which a proportionate share of net assets is attributed) as a practical expedient, as described in FASB ASC 820-10-35-59, should not be categorized within the fair value hierarchy. In addition, the disclosure requirements in FASB ASC 820-10-50-2 do not apply to that investment. Disclosures required for an investment for which fair value is measured using net asset value per share (or its equivalent) as a practical expedient are described in FASB ASC 820-10-50-6A. Although the investment is not categorized within the fair value hierarchy, a reporting entity should provide the amount measured using the net asset value per share (or its equivalent) practical expedient to permit reconciliation of the fair value of investments included in the fair value hierarchy to the line items presented in the statement of financial position in accordance with FASB ASC 820-10-50-2B.

Readers are encouraged to read the full text of the ASU, available at www.fasb.org. Readers should apply the appropriate guidance based on their facts and circumstances.

©2016. AICPA

<sup>&</sup>lt;sup>3</sup> FASB Accounting Standards Update (ASU) No. 2015-07, Fair Value Measurement (Topic 820): Disclosures for Investments in Certain Entities That Calculate Net asset Value per Share (or Its Equivalent) (a consensus of the FASB Emerging Issues Task Force), issued in May 2015, applies to reporting entities that elect to measure the fair value of an investment within the scope of paragraphs 4–5 of FASB Accounting Standards Codification (ASC) 820-10-15 using the net asset value per share (or its equivalent) practical expedient in FASB ASC 820-10-35-59. FASB ASU No. 2015-07 is effective for public business entities for fiscal years beginning after December 15, 2015, and interim periods within those fiscal years. For all other entities, the amendments in this ASU are effective for fiscal years beginning after December 15, 2016, and interim periods within those fiscal years. A reporting entity should apply the amendments retrospectively to all periods presented. The retrospective approach requires that an investment for which fair value is measured using the net asset value per share practical expedient be removed from the fair value hierarchy in all periods presented in an entity's financial statements. Earlier application is permitted.

## Measuring the Fair Value of Investments in Certain Entities That Calculate Net Asset Value per Share (or Its Equivalent)<sup>4</sup>

20.38 Paragraphs 59–62 of FASB ASC 820-10-35 contain guidance intended to improve financial reporting by permitting the use of a practical expedient, with appropriate disclosures, when measuring the fair value of an alternative investment that does not have a readily determinable fair value. This practical expedient permits a reporting entity to estimate the fair value of an investment within the scope of paragraphs 4–5 of FASB ASC 820-10-15 using the net asset value per share (or its equivalent) of the investment, if the net asset value per share of the investment is calculated in a manner consistent with the measurement principles of FASB ASC 946, Financial Services—Investment Companies, as of the reporting entity's measurement date.

**20.39** According to FASB ASC 820-10-15-4, the guidance contained in paragraphs 59–62 of FASB ASC 820-10-35 and FASB ASC 820-10-50-6A should apply only to an investment that meets both of the following criteria as of the entity's measurement date:

- a. The investment does not have a readily determinable fair value.<sup>5</sup>
- b. The investment is an investment company within the scope of FASB ASC 946 or is an investment in a real estate fund for which it is industry practice to measure investment assets at fair value on a recurring basis and to issue financial statements that are consistent with the measurement principles in FASB ASC 946.

**20.40** FASB ASC 820-10-35-60 explains that if the net asset value per share of the investment obtained from the investee is not as of the reporting entity's measurement date or is not calculated in a manner consistent with the measurement principles of FASB ASC 946, the reporting entity should consider whether an adjustment to the most recent net asset value per share is necessary.

**20.41** FASB ASC 820-10-35-61 states that a reporting entity should decide on an investment-by-investment basis whether to apply the practical expedient in FASB ASC 820-10-35-59 and should apply the practical expedient consistently to the fair value measurement of the reporting entity's entire position in a particular investment, unless it is probable at the measurement date that the reporting entity will sell a portion of an investment at an amount different from net asset value per share (or its equivalent) as described in FASB ASC

<sup>&</sup>lt;sup>4</sup> Technical Questions and Answers (Q&A) sections 2220.18–.27 (AICPA, *Technical Questions and Answers*) are intended to assist reporting entities when implementing the provisions of FASB ASC 820, *Fair Value Measurement*, to estimate the fair value of their investments in certain entities that calculate net asset value. Q&A sections 2220.18–.27 apply to investments that are required to be measured and reported at fair value and are within the scope of paragraphs 4–5 of FASB ASC 820-10-15.

<sup>&</sup>lt;sup>5</sup> FASB ASC 820-10-15-5 notes that the FASB ASC glossary definition of *readily determinable* fair value indicates that an equity security would have a readily determinable fair value if any one of the three conditions is met. One of those conditions is that sales prices or bid-and-asked quotations are currently available on a securities exchange registered with the SEC or in the over-the-counter market, provided that those prices or quotations for the over-the-counter market are publicly reported by the National Association of the Securities Dealers Automated Quotations systems or by OTC Markets Group, Inc. The definition notes that restricted stock meets that definition if the restriction expires within one year. If an investment otherwise would have a readily determinable fair value, except that the investment has a restriction expiring in more than one year, the reporting entity should not apply paragraphs 59–62 of FASB ASC 820-10-35 and in FASB ASC 820-10-50-6A to the investment.

820-10-35-62. In those situations, the reporting entity should account for the portion of the investment that is being sold in accordance with FASB ASC 820 (that is, the reporting entity should not apply the guidance discussed in FASB ASC 820-10-35-59).

## Measuring Fair Value When the Volume or Level of Activity for an Asset or a Liability Has Significantly Decreased

- **20.42** Paragraphs 54C–54J of FASB ASC 820-10-35 clarify the application of FASB ASC 820 in measuring fair value when the volume and level of activity for an asset or liability has significantly decreased. Guidance is also included in identifying transactions that are not orderly. In addition, paragraphs 90–98 of FASB ASC 820-10-55 include illustrations on the application of this guidance.
- 20.43 The fair value of an asset or liability might be affected when there has been a significant decrease in the volume or level of activity for the asset or liability in relation to normal market activity for the asset or liability (or similar assets or liabilities). To determine whether there has been a significant decrease, FASB ASC 820-10-35-54C provides a number of factors a reporting entity should evaluate for significance and relevance. According to FASB ASC 820-10-35-54D, if, after evaluating the factors, the reporting entity concludes that there has been a significant decrease, further analysis of the transactions or quoted prices is needed. A decrease in the volume or level of activity on its own may not indicate that a transaction price or quoted price does not represent fair value or that a transaction in that market is not orderly. However, if a reporting entity determines that a transaction or quoted price does not represent fair value (for example, there may be transactions that are not orderly), an adjustment to the transaction or quoted price will be necessary if the reporting entity uses that price as a basis for measuring fair value and that adjustment may be significant to the fair value measurement in its entirety.
- 20.44 If determined there has been a significant decrease, FASB ASC 820-10-35-54F states that a change in valuation technique or the use of multiple valuation techniques may be appropriate (for example, the use of a market approach and a present value technique). When weighing indications of fair value resulting from the use of multiple valuation techniques, a reporting entity should consider the reasonableness of the range of fair value measurements. The objective is to determine the point within the range that is most representative of fair value under current market conditions. A wide range of fair value estimates may be an indication that further analysis is needed.
- **20.45** Even when there has been such a significant decrease, paragraphs 54G–54H of FASB ASC 820-10-35 state that the objective of a fair value measurement remains the same. Estimating the price at which market participants would be willing to enter into a transaction at the measurement date under current market conditions if there has been a significant decrease in the volume and level of activity for the asset or liability depends on the facts and circumstances at the measurement date and requires judgment. A reporting entity's intention to hold the asset or to settle or otherwise fulfill the liability is not relevant when measuring fair value because fair value is a market-based measurement, not an entity-specific measurement.
- **20.46** *Identifying transactions that are not orderly.* According to FASB ASC 820-10-35-54I, the determination of whether a transaction is orderly (or is not orderly) is more difficult if there has been a significant decrease in

the volume or level of activity for the asset or liability in relation to normal market activity for the asset or liability (or similar assets or liabilities). In such circumstances, it is not appropriate to conclude that all transactions in that market are not orderly (that is, forced liquidations or distressed sales). Circumstances that may indicate that a transaction is not orderly include the following:

- There was not adequate exposure to the market for a period before the measurement date to allow for marketing activities that are usual and customary for transactions involving such assets or liabilities under current market conditions.
- There was a usual and customary marketing period, but the seller marketed the asset or liability to a single market participant.
- The seller is in or near bankruptcy or receivership (that is, the seller is distressed).
- The seller was required to sell to meet regulatory or legal requirements (that is, the seller was forced).
- The transaction price is an outlier when compared with other recent transactions for the same or a similar asset or liability.

A reporting entity should evaluate the circumstances to determine whether the transaction is orderly based on the weight of the evidence available.

**20.47** FASB ASC 820-10-35-54J states that a reporting entity should consider all the following when measuring fair value or estimating market risk premiums:

- If the evidence indicates the transaction is not orderly, a reporting entity should place little, if any, weight (compared with other indications of fair value) on that transaction price.
- If the evidence indicates that a transaction is orderly, a reporting entity should take into account that transaction price. The amount of weight placed on the transaction price when compared with other indications of fair value will depend on the facts and circumstances, such as the following:
  - The volume of the transaction.
  - The comparability of the transaction to the asset or liability being measured.
  - The proximity of the transaction to the measurement date.
- If a reporting entity does not have sufficient information to conclude whether a transaction is orderly, it should take into account the transaction price. However, the transaction price may not represent fair value (that is, the transaction price is not necessarily the sole or primary basis for measuring fair value or estimating market risk premiums). When a reporting entity does not have sufficient information to conclude whether particular transactions are orderly, the reporting entity should place less weight on those transactions when compared with other transactions that are known to be orderly.

Fair Value 577

A reporting entity need not undertake exhaustive efforts to determine whether a transaction is orderly, but it should not ignore information that is reasonably available. When a reporting entity is a party to a transaction, it is presumed to have sufficient information to conclude whether a transaction is orderly.

#### **Using Quoted Prices Provided by Third Parties**

20.48 Paragraphs 54K-54M of FASB ASC 820-10-35 address using quoted prices provided by third parties. Specifically, FASB ASC 820 does not preclude the use of quoted priced provided by third parties, such as pricing services or brokers, if a reporting entity has determined that the quoted prices provided by those parties are developed in accordance with FASB ASC 820. If there has been a significant decrease in the volume or level of activity for the asset or liability, a reporting entity should evaluate whether the quoted prices provided by third parties are developed using current information that reflects orderly transactions or a valuation technique that reflects market participation assumptions (including assumptions about risk). In weighing a quoted price as an input to a fair value measurement, a reporting entity places less weight (when compared with other indications of fair value that reflect the results of transactions) on quotes that do not reflect the result of transactions. Furthermore, the nature of a quote (for example, whether the quote is an indicative price or a binding offer) should be taken into account when weighing the available evidence, with more weight given to quotes provided by third parties that represent binding offers. Management is ultimately responsible for the estimates recorded on its financial statements and accordingly should establish processes and procedures for persons with an appropriate level of expertise to review prices provided by third parties for reasonableness.

#### Disclosures<sup>6</sup>

**20.49** FASB ASC 820-10-50 discusses the disclosures required for assets and liabilities measured at fair value. FASB ASC 820-10-50-1 explains that the reporting entity should disclose information that helps users of its financial statements assess both (a) for assets and liabilities that are measured at fair value on a recurring or nonrecurring basis in the statement of financial position after initial recognition, the valuation techniques and the inputs used to develop those measurements and (b) for recurring fair value measurements using significant unobservable inputs (Level 3), the effect of the measurements on earnings (or changes in net assets) or other comprehensive income for the period.

**20.50** To meet the objectives described in FASB ASC 820-10-50-1 (see paragraph 20.49), FASB ASC 820-10-50-2 requires a reporting entity to disclose, at a minimum, the following information for each of assets and liabilities measured at fair value in the statement of financial position after initial recognition:

<sup>&</sup>lt;sup>6</sup> FASB has an open project titled Disclosure Framework—Disclosure Review: Fair Value Measurement. This project is a subset of the Disclosure Framework project intended to improve the effectiveness of disclosures in notes to financial statements by clearing communicating the information that is most important to users of each entity's financial statements. In December 2015, FASB issued proposed ASU Fair Value Measurement (Topic 820): Disclosure Framework—Changes to the Disclosure Requirements for Fair Value Measurement to propose modification to the disclosure requirements on fair value measurements.

Readers are encouraged to visit the "Technical Agenda" page at www.fasb.org for the latest developments regarding the project and how it may impact the guidance in this chapter.

- a. For recurring fair value measurements, the fair value measurement at the end of the reporting period, and for nonrecurring fair value measurements, the fair value measurement at the relevant measurement date and the reasons for the measurement.
- b. For recurring and nonrecurring fair value measurements, the level of the fair value hierarchy within which the fair value measurements are categorized in their entirety (Level 1, 2, or 3).
- c. For assets and liabilities held at the end of the reporting period that are measured at fair value on a recurring basis, the amounts of any transfers between Level 1 and Level 2 of the fair value hierarchy, the reasons for the transfers, and the reporting entity's policy for determining when transfers between levels are deemed to have occurred (see FASB ASC 820-10-50-2C discussed in paragraph 20.52). Transfers into each level should be disclosed and discussed separately from transfers out of each level.
- d. For recurring and nonrecurring fair value measurements categorized within Level 2 and Level 3 of the fair value hierarchy, a description of the valuation technique(s) and the inputs used in the fair value measurement. If there has been a change in valuation technique, the reporting entity should disclose that change and the reason(s) for making it. For fair value measurements categorized within Level 3 of the fair value hierarchy, a reporting entity should provide quantitative information about the significant unobservable inputs used in the fair value measurement. A reporting entity is not required to create quantitative information to comply with this disclosure requirement if quantitative unobservable inputs are not developed by the reporting entity when measuring fair value (for example, when a reporting entity uses prices from prior transactions or third-party pricing information without adjustment). However, when providing this disclosure, a reporting entity cannot ignore quantitative unobservable inputs that are significant to the fair value measurement and are reasonably available to the reporting entity.
- e. For recurring fair value measurements categorized within Level 3 of the fair value hierarchy, a reconciliation from the opening balances to the closing balances, disclosing separately changes during the period attributable to the following:
  - Total gains or losses for the period recognized in earnings (or changes in net assets), and the line item(s) in the statement of income (or activities) in which those gains or losses are recognized.
  - Total gains or losses for the period recognized in other comprehensive income, and the line item(s) in other comprehensive income in which those gains or losses are recognized.
  - iii. Purchases, sales, issuances, and settlements (each of those types of changes disclosed separately).
  - iv. The amounts of any transfers into or out of Level 3 of the fair value hierarchy, the reasons for those transfers, and the reporting entity's policy for determining when transfers between levels are deemed to have occurred (see FASB)

ASC 820-10-50-2C discussed in paragraph 20.52). Transfers into Level 3 should be disclosed and discussed separately from transfers out of Level 3.

- f. For recurring fair value measurements categorized within Level 3 of the fair value hierarchy, the amount of the total gains or losses for the period in item (e)(i) included in earnings (or changes in net assets) that is attributable to the change in unrealized gains or losses relating to those assets and liabilities held at the end of the reporting period, and the line item(s) in the statement of income (or activities) in which those unrealized gains or losses are recognized.
- g. For recurring and nonrecurring fair value measurements categorized within Level 3 of the fair value hierarchy, a description of the valuation processes used by the reporting entity (including, for example, how an entity decides its valuation policies and procedures and analyzes changes in fair value measurements from period to period).
- h. For recurring fair value measurements categorized within Level 3 of the fair value hierarchy, a narrative description of the sensitivity of the fair value measurement to changes in unobservable inputs if a change in those inputs to a different amount might result in a significantly higher or lower fair value measurement. If there are interrelationships between those inputs and other unobservable inputs used in the fair value measurement, a reporting entity should also provide a description of those interrelationships and how they magnify or mitigate the effect of changes in the unobservable inputs on the fair value measurement. To comply with that disclosure requirement, the narrative description of the sensitivity to changes in unobservable inputs should include, at a minimum, the unobservable inputs disclosed when complying with item (d).
- i. For recurring and nonrecurring fair value measurements, if the highest and best use of a nonfinancial asset differs from its current use, a reporting entity should disclose that fact and why the nonfinancial asset is being used in a manner that differs from its highest and best use.

FASB ASC 820-10-50-2F states that a nonpublic entity is not required to disclose the information discussed in items (c) and (h) unless required by another FASB ASC topic.

**20.51** FASB ASC 820-10-50-2B explains that a reporting entity should determine appropriate classes of assets and liabilities on the basis of (a) the nature, characteristics, and risks of the asset or liability; and (b) the level of the fair value hierarchy within which the fair value measurement is categorized. Further, the number of classes may need to be greater for fair value measurements within Level 3 of the fair value hierarchy because those measurements have a greater degree of uncertainty and subjectivity. A class of assets and liabilities will often require greater disaggregation than the line items presented in the statement of financial position.

**20.52** FASB ASC 820-10-50-2C explains that a reporting entity should disclose and consistently follow its policy for determining when transfers between levels of the fair value hierarchy are deemed to have occurred in accordance with items (bb) and (c)(3) in FASB ASC 820-10-50-2 (see items (c) and (e)(iv) in paragraph 20.50). The policy about the timing of recognizing transfers should

be the same for transfers into the levels as for transfers out of the levels. Examples of policies for determining the timing of transfers include the following:

- The date of the event or change in circumstances that caused the transfer
- The beginning of the reporting period
- The end of the reporting period
- **20.53** If a reporting entity makes an accounting policy decision to use the exception in FASB ASC 820-10-35-18D (see paragraph 20.21), FASB ASC 820-10-50-2D requires disclosure of that fact.
- **20.54** For each class of assets and liabilities not measured at fair value in the statement of financial position but for which the fair value is disclosed, FASB ASC 820-10-50-2E requires a reporting entity to disclose the information required by items (b), (bbb)(1), and (h) in FASB ASC 820-10-50-2 (see items (b), (d), and (i) in paragraph 20.50). However, a reporting entity is not required to provide the quantitative disclosures about significant unobservable inputs used in fair value measurements categorized within Level 3 of the fair value hierarchy required by item (bbb)(2) in FASB ASC 820-10-50-2 (see item (d) in paragraph 20.50). For such assets and liabilities, a reporting entity does not need to provide the other disclosures required by FASB ASC 820.
- **20.55** FASB ASC 820-10-50-3 states that, for derivative assets and liabilities, the reporting entity should present both the fair value disclosures required by items (a)–(bb) in FASB ASC 820-10-50-2 (see items (a)–(c) in paragraph 20.50) on a gross basis and the reconciliation disclosures required by items (c)–(d) in FASB ASC 820-10-50-2 (see items (e)–(f) in paragraph 20.50) on either a gross or net basis.
- **20.56** For a liability measured at fair value and issued with an inseparable third-party credit enhancement, FASB ASC 820-10-50-4A requires an issuer to disclose the existence of that credit enhancement.

#### Additional Disclosures for Financial Instruments

**20.57** Paragraphs 2A–7 of FASB ASC 825-10-50 provide guidance about which entities are required to apply the disclosure requirements for financial instruments. For annual reporting periods, the disclosure guidance related to fair value of financial instruments in paragraphs 10–19 of FASB ASC 825-10-50 applies to all entities but is optional for an entity that is a nonpublic entity, the entity's total assets are less than \$100 million on the date of the financial statements, and the entity has no instrument that, in whole or in part, is accounted for as a derivative instrument under FASB ASC 815 other than commitments related to origination of mortgage loans to be held for sale during the reporting period. For all interim periods, the disclosure guidance applies to all entities but is optional for those entities that do not meet the definition of a *publicly traded company*, as defined in generally accepted accounting principles.

### © Update 20-1 Accounting and Reporting: Recognition and Measurement of Financial Assets and Financial Liabilities

FASB Accounting Standards Update (ASU) No. 2016-01, Financial Instruments—Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities, issued in January 2016, is effective

for fiscal years, and interim periods within those fiscal years, of a public business entity beginning after December 15, 2017.

For all other entities (including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting), FASB ASU No. 2016-01 is effective for fiscal years beginning after December 15, 2018, and interim periods within fiscal years beginning after December 15, 2019. Early application is permitted for all other entities as of the fiscal years beginning after December 15, 2017, including interim periods within those fiscal years.

All entities may adopt the presentation guidance in paragraphs 5–7 of FASB ASC 825-10-45 for financial statements of fiscal years or interim periods that have not yet been issued or that have not yet been made available for issuance. Furthermore, entities that are not public business entities may elect not to disclose the information about fair value of financial instruments required by the "General" subsection of FASB ASC 825-10-50 in financial statements of fiscal years or interim periods that have not yet been made available for issuance. Except as indicated previously in this paragraph, early application is not permitted.

FASB ASU No. 2016-01 addresses certain aspects of recognition, measurement, presentation, and disclosure of financial instruments and affects all entities that hold financial assets or owe financial liabilities. Among other provisions of the guidance, the amendments in this ASU exempt all entities that are not public business entities from disclosing fair value information for financial instruments measured at amortized cost. In addition, for public business entities, the amendments supersede the requirement to disclose the methods and significant assumptions used in calculating the fair value of financial instruments required to be disclosed for financial instruments measured at amortized cost on the balance sheet.

Furthermore, the amendments in this ASU require public business entities that are required to disclose the fair value of financial instruments measured at amortized cost on the balance sheet to measure that fair value using the exit price notion consistent with FASB ASC 820. This change to GAAP eliminates the entry price method previously used by some entities for disclosure purposes for some financial assets.

In addition, the amendments in this ASU require separate presentation of financial assets and financial liabilities by measurement category and form of financial asset (that is, securities or loans and receivables) on the balance sheet or in the accompanying notes to the financial statements.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-01, see appendix G, "Accounting for Financial Instruments," of this guide.

**20.58** Concentrations of credit risk of all financial instruments. Except as indicated in FASB ASC 825-10-50-22, an entity should disclose all significant concentrations of credit risk arising from all financial instruments, whether from an individual counterparty or groups of counterparties. When applying

this disclosure requirement, the term *financial instrument* includes derivative instruments accounted for under FASB ASC 815. Refer to FASB ASC 825-10-50-21 for specific information to be disclosed about each significant concentration. Entities are not required to disclose information about concentrations of credit risk for the financial instruments included in FASB ASC 825-10-50-22.

**20.59** Market risk of all financial instruments. FASB ASC 825-10-50-23 states that an entity is encouraged, but not required, to disclose quantitative information about the market risks of financial instruments that is consistent with the way it manages or adjusts those risks. Appropriate ways of reporting that quantitative information will differ for different entities and will likely evolve over time as management approaches and measurement techniques evolve.

## Fair Value Measurements of Investments in Certain Entities That Calculate Net Asset Value per Share (or its Equivalent)

**20.60** For investments that are within the scope of paragraphs 4–5 of FASB ASC 820-10-15 and measured at fair value on a recurring or nonrecurring basis during the period, FASB ASC 820-10-50-6A requires disclosure of information that will help users of the financial statements to understand the nature and risks of the investments and whether the investments are probable of being sold at amounts different from net asset value per share (or its equivalent). These disclosures, to the extent applicable, are required for each class of investment. The required disclosures, at a minimum, are as follows:

- a. The fair value measurement (as determined by applying paragraphs 59–62 of FASB ASC 820-10-35) of the investments in the class at the reporting date and a description of the significant investment strategies of the investee(s) in the class.
- b. For each class of investment that includes investments that can never be redeemed with the investees, but the reporting entity receives distributions through the liquidation of the underlying assets of the investees, the reporting entity's estimate of the period of time over which the underlying assets are expected to be liquidated by the investees.
- c. The amount of the reporting entity's unfunded commitments related to investments in the class.
- d. A general description of the terms and conditions upon which the investor may redeem investments in the class (for example, quarterly redemption with 60 days' notice).
- e. The circumstances in which an otherwise redeemable investment in the class (or a portion thereof) might not be redeemable (for example, investments subject to a lockup or gate). Also, for those otherwise redeemable investments that are restricted from redemption as of the reporting entity's measurement date, the reporting entity should disclose its estimate of when the restriction from redemption might lapse. If an estimate cannot be made, the reporting entity should disclose that fact and how long the restriction has been in effect.
- f. Any other significant restriction on the ability to sell investments in the class at the measurement date.

Fair Value 583

- g. If a reporting entity determines that it is probable that it will sell an investment(s) for an amount different from net asset value per share (or its equivalent), as described in FASB ASC 820-10-35-62, the reporting entity should disclose the total fair value of all investments that meet the criteria in FASB ASC 820-10-35-62 and any remaining actions required to complete the sale.
- h. If a group of investments would otherwise meet the criteria in FASB ASC 820-10-35-62 but the individual investments to be sold have not been identified (for example, if a reporting entity decides to sell 20 percent of its investments in private equity funds, but the individual investments to be sold have not been identified), so the investments continue to qualify for the practical expedient in FASB ASC 820-10-35-59, the reporting entity should disclose its plans to sell and any remaining actions required to complete the sale(s).

For investments that are within the scope of paragraphs 4–5 of FASB ASC 820-10-15 and that are measured using the practical expedient in FASB ASC 820-10-35-59 on a recurring or nonrecurring basis during the period, "Pending Content" in FASB ASC 820-10-50-6A requires disclosure of information that helps users of the financial statements to understand the nature and risks of the investments and whether the investments, if sold, are probable of being sold at amounts different from net asset value per share (or its equivalent). To meet that objective, to the extent applicable, a reporting entity should disclose, at a minimum, the following information for each class of investment:

- a. The fair value measurement (as determined by applying paragraphs 59–62 of FASB ASC 820-10-35) of the investments in the class at the reporting date and a description of the significant investment strategies of the investee(s) in the class.
- b. For each class of investment that includes investments that can never be redeemed with the investees, but the reporting entity receives distributions through the liquidation of the underlying assets of the investees, the reporting entity's estimate of the period of time over which the underlying assets are expected to be liquidated by the investees.
- c. The amount of the reporting entity's unfunded commitments related to investments in the class.
- d. A general description of the terms and conditions upon which the investor may redeem investments in the class (for example, quarterly redemption with 60 days' notice).
- e. The circumstances in which an otherwise redeemable investment in the class (or a portion thereof) might not be redeemable (for example, investments subject to a lockup or gate). Also, for those otherwise redeemable investments that are restricted from redemption as of the reporting entity's measurement date, the reporting entity should disclose its estimate of when the restriction from redemption might lapse. If an estimate cannot be made, the reporting entity should disclose that fact and how long the restriction has been in effect.

<sup>&</sup>lt;sup>7</sup> See footnote 3.

- f. Any other significant restriction on the ability to sell investments in the class at the measurement date.
- g. If a group of investments would otherwise meet the criteria in FASB ASC 820-10-35-62 but the individual investments to be sold have not been identified (for example, if a reporting entity decides to sell 20 percent of its investments in private equity funds, but the individual investments to be sold have not been identified), so the investments continue to qualify for the practical expedient in FASB ASC 820-10-35-59, the reporting entity should disclose its plans to sell and any remaining actions required to complete the sale(s).

## **Fair Value Option**

**20.61** FASB ASC 825 creates a fair value option under which an entity may irrevocably elect fair value as the initial and subsequent measure for many financial instruments and certain other items, with changes in fair value recognized in earnings as those changes occur. FASB ASC 825-10-35-4 explains that a business entity should report unrealized gains and losses on items for which the fair value option has been elected in earnings at each subsequent reporting date. An election is made on an instrument-by-instrument basis (with certain exceptions as discussed in FASB ASC 825-10-25-7), generally when an instrument is initially recognized in the financial statements and certain other election dates as outlined in FASB ASC 825-10-25-4. The fair value option need not be applied to all identical items, except as required by FASB ASC 825-10-25-7. Most financial assets and financial liabilities are eligible to be recognized using the fair value option, as are firm commitments for financial instruments and certain nonfinancial contracts (described in FASB ASC 825-10-15-4).

**20.62** As explained by FASB ASC 825-10-15-5, specifically excluded from eligibility is an investment in a subsidiary that the entity is required to consolidate, an interest in a variable interest entity that the entity is required to consolidate, employer's and plan's obligations for pension benefits, other postretirement benefits (including health care and life insurance benefits), postemployment benefits, employee stock option and stock purchase plans, and other forms of deferred compensation arrangements (or assets representing net overfunded positions in those plans), financial assets and liabilities recognized under leases (this does not apply to a guarantee of a third-party lease obligation or a contingent obligation arising from a cancelled lease), deposit liabilities of depository institutions, and financial instruments that are, in whole or in part, classified by the issuer as a component of shareholder's equity (including temporary equity).

**20.63** FASB ASC 825-10-45 and 825-10-50<sup>8</sup> also include presentation and disclosure requirements designed to facilitate comparisons between entities

(continued)

<sup>&</sup>lt;sup>8</sup> Q&A section 1800.05, "Applicability of Fair Value Disclosure Requirements and Measurement Principles in Financial Accounting Standards Board (FASB) Accounting Standards Codification (ASC) 820, Fair Value Measurements and Disclosures, to Certain Financial Instruments" (AICPA, Technical Questions and Answers), addresses the applicability of FASB ASC 820 to financial instruments that are not recognized at fair value in the statement of financial position, but that fair value is required to be disclosed in the notes to financial statements in accordance with paragraphs 10–19 of FASB ASC 825-10-50.

Fair Value 585

that choose different measurement attributes for similar types of assets and liabilities. Paragraphs 1-2 of FASB ASC 825-10-45 state that entities should report assets and liabilities that are measured using the fair value option in a manner that separates those reported fair values from the carrying amounts of similar assets and liabilities measured using another measurement attribute. To accomplish that, an entity should either (a) report the aggregate of both fair value and non-fair-value items on a single line, with the fair value amount parenthetically disclosed or (b) present separate lines for the fair value carrying amounts and the non-fair-value carrying amounts. As discussed in FASB ASC 825-10-25-3, upfront costs and fees, such as debt issue costs, may not be deferred for items that the fair value option has been elected.

# Auditing Considerations<sup>9</sup>

**20.64** AU-C section 540, Auditing Accounting Estimates, Including Fair Value Accounting Estimates, and Related Disclosures (AICPA, Professional Standards), addresses the auditor's responsibilities relating to accounting estimates, including fair value accounting estimates and related disclosures, in an audit of financial statements. Specifically, it expands on how AU-C sections 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement; 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards); and other relevant AU-C sections are to be applied with regard to accounting estimates. It also includes requirements and guidance related to misstatements of individual accounting estimates and indicators of possible management bias.

**20.65** If the preparation of the financial statements involves the use of expertise in a field other than accounting, paragraph .A7 of AU-C section 620, Using the Work of an Auditor's Specialist (AICPA, Professional Standards), states that the auditor, who is skilled in accounting and auditing, may not possess the necessary expertise to audit those financial statements. The engagement partner is required by AU-C section 220, Quality Control for an Engagement Conducted in Accordance With Generally Accepted Auditing Standards (AICPA, Professional Standards), to be satisfied that the engagement team and any external auditor's specialists (defined in the auditing standards as an individual or organization possessing expertise in a field other than accounting or auditing, whose work in that field is used by the auditor to assist the auditor in obtaining sufficient appropriate audit evidence) who are not part of the engagement team, collectively, have the appropriate competence and capabilities to perform the audit engagement. <sup>10</sup> Further, the auditor is required

(footnote continued)

As a result of FASB ASU No. 2013-03, *Financial Instruments (Topic 825): Clarifying the Scope and Applicability of a Particular Disclosure to Nonpublic Entities*, FASB ASC 825-10-50-3A was added to exempt all nonpublic entities from providing the disclosure in item (d) in FASB ASC 825-10-50-10 for items disclosed at fair value but not measured at fair value in the statement of financial position.

<sup>&</sup>lt;sup>9</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the Auditing Standards Board and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of *issuers* and *nonissuers* as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

<sup>&</sup>lt;sup>10</sup> Paragraph .16 of AU-C section 220, Quality Control for an Engagement Conducted in Accordance With Generally Accepted Auditing Standards (AICPA, Professional Standards).

by AU-C section 300, *Planning an Audit* (AICPA, *Professional Standards*), to ascertain the nature, timing, and extent of resources necessary to perform the engagement. The auditor's determination of whether to use the work of an auditor's specialist, and, if so, when and to what extent, assists the auditor in meeting these requirements. As the audit progresses or as circumstances change, the auditor may need to revise earlier decisions about using the work of an auditor's specialist.

20.66 Auditing interests in trusts held by a third-party trustee and reported at fair value. In circumstances in which the auditor determines that the nature and extent of auditing procedures should include verifying the existence and testing the measurement of investments held by a trust simply receiving a confirmation from the trustee, either in aggregate or on an investment by investment basis, does not in and of itself constitute adequate audit evidence with respect to the requirements for auditing the fair value of the interest in trust under AU-C section 540. In addition, receiving conformation from the trustee for investments in aggregate does not constitute adequate audit evidence with respect to the existence assertion. Receiving confirmation from the trustee on an investment-by-investment basis, however, typically would constitute adequate audit evidence with respect to the existence assertion. Also, in discussing obtaining an understanding of how management identifies the need for accounting estimates, paragraph .A15 of AU-C section 540 states that the preparation and fair presentation of the financial statements requires management to determine whether a transaction, an event, or a condition gives rise to the need to make an accounting estimate and that all necessary accounting estimates have been recognized, measured, and disclosed in the financial statements in accordance with the applicable financial reporting framework.

**20.67** In circumstances in which the auditor is unable to audit the existence or measurement of interests in trusts at the financial statement date, the auditor should consider whether that scope limitation requires the auditor to either qualify his or her opinion or to disclaim an opinion, as discussed in AU-C section 705, *Modifications to the Opinion in the Independent Auditor's Report* (AICPA, *Professional Standards*).

#### 20.68

Considerations for Audits Performed in Accordance With PCAOB Standards<sup>12</sup>

PCAOB Staff Audit Practice Alert No. 2, Matters Related to Auditing Fair Value Measurements of Financial Instruments and the Use of Specialists (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance sec. 400.02), provides guidance on auditors' responsibilities for auditing fair value measurements of financial instruments and when using the work of specialists under the existing standards of the PCAOB. This alert is focused on specific matters that are likely to increase audit risk related to the fair value of financial instruments in a rapidly changing economic environment. This practice alert highlights certain requirements in the auditing standards related to fair value measurements and disclosures in the financial statements and

<sup>&</sup>lt;sup>11</sup> Paragraph .08e of AU-C section 300, Planning an Audit (AICPA, Professional Standards).

<sup>&</sup>lt;sup>12</sup> PCAOB Staff Audit Practice Alerts are not rules of the board and do not reflect any board determination or judgment about the conduct of any particular firm, auditor, or any other person.

Fair Value 587

certain aspects of U.S. generally accepted accounting principles that are particularly relevant to the current economic environment.

PCAOB Staff Audit Practice Alert No. 4, Auditor Considerations Regarding Fair Value Measurements, Disclosures, and Other-Than-Temporary Impairments (AICPA, PCAOB Standards and Related Rules, PCAOB Staff Guidance sec. 400.04), informs auditors about potential implications of recently issued FASB guidance on reviews of interim financial information and annual audits. This alert addresses the following topics: (a) reviews of interim financial information; (b) audits of financial statements, including integrated audits; (c) disclosures; and (d) auditor reporting considerations.

©2016, AICPA AAG-DEP 20.68

## Chapter 21

# **Trust and Asset Management Activities**

Gray shaded text in this chapter reflects guidance issued but not yet effective as of the date of this guide, July 1, 2016, but becoming effective on or prior to December 31, 2016, exclusive of any option to early adopt ahead of the mandatory effective date. Unless otherwise indicated, all unshaded text reflects guidance that was already effective as of the date of this guide.

# FASB Accounting Standards Update (ASU) No. 2014-09, Revenue from Contracts with Customers (Topic 606)

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14, *Revenue from Contracts with Customers (Topic 606): Deferral of the Effective Date)* for annual reporting periods of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an overthe-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period. Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as of either

- an annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- an annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB Accounting Standards Codification (ASC) 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605, *Revenue Recognition*, as well as guidance within the 900 series of industry-specific topics, including FASB ASC 942, *Financial Services—Depository and Lending*. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts).

Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

The AICPA has formed 16 industry task forces to assist in developing a new Accounting Guide on revenue recognition that will provide helpful hints and illustrative examples for how to apply the new standard. Revenue recognition implementation issues identified by the Depository and Lending Institutions Revenue Recognition Task Force will be available for informal comment,

©2016, AICPA AAG-DEP 21

after review by the AICPA Financial Reporting Executive Committee, at the "Depository and Lending Institutions Revenue Recognition Task Force" page at www.aicpa.org.

Readers are encouraged to submit comments to revreccomments@aicpa.org.

For more information on FASB ASU No. 2014-09, see appendix E, "The New Revenue Recognition Standard: FASB ASU No. 2014-09," of this guide.

### Introduction

**21.01** Among other engagements, auditors may be engaged to (a) report on trust company financial statements, particularly of common or collective funds, (b) report on internal control over financial reporting in the institution's trust department, or (c) perform procedures agreed to by management or regulators or extended audit services to supplement the institution's internal audit efforts.  $^{1,2,3}$ 

<sup>1</sup> Usually, such an engagement is the result of the need of auditors of the financial statements of pension plans, mutual funds, and other entities to obtain audit evidence regarding internal control in the departments of a bank or savings institution controlling assets of other entities. Because an institution may administer many plans, it may not be economically feasible for each plan's auditor to carry out audit procedures at the trustee institution. Accordingly, an auditor may perform procedures in the area or department administering all plans at the institution and issue a report to the user institution on internal accounting controls related to administration of the plans. Guidance is provided in AT section 801, Reporting on Controls at a Service Organization (AICPA, Professional Standards). The related AICPA Guide Service Organizations: Reporting on Controls at a Service Organization Relevant to User Entities' Internal Control Over Financial Reporting provides further guidance. AT section 501, An Examination of an Entity's Internal Control Over Financial Reporting That Is Integrated With an Audit of Its Financial Statements (AICPA, Professional Standards), states that an auditor may apply the relevant concepts described in AU-C section 402, Audit Considerations Relating to an Entity Using a Service Organization (AICPA, Professional Standards), to the examination of internal control.

Usually, such an engagement is the result of the need of auditors of the financial statements of pension plans, mutual funds, and other entities to obtain audit evidence regarding internal control in the departments of a bank or savings institution controlling assets of other entities. Because an institution may administer many plans, it may not be economically feasible for each plan's auditor to carry out audit procedures at the trustee institution. Accordingly, an auditor may perform procedures in the area or department administering all plans at the institution and issue a report to the user institution on internal accounting controls related to administration of the plans. Guidance is provided in AT section 801, Reporting on Controls at a Service Organization (AICPA, Professional Standards). The related AICPA Guide Service Organizations: Reporting on Controls at a Service Organization Relevant to User Entities' Internal Control Over Financial Reporting provides further guidance. Paragraph .89 of AU-C section 940, An Audit of Internal Control Over Financial Reporting That Is Integrated With an Audit of Its Financial Statements (AICPA, Professional Standards), states that the auditor in required to perform the procedures in AU-C section 402, Audit Considerations Relating to an Entity Using a Service Organization (AICPA, Professional Standards), with respect to the activities performed by the service organization. In an audit of internal control over financial reporting, the auditor should also obtain evidence that controls at the service organization that are relevant to the auditor's opinion on internal control over financial reporting are operating effectively.

Statement on Auditing Standards (SAS) No. 130, An Audit of Internal Control Over Financial Reporting That Is Integrated With an Audit of Financial Statements (AICPA, Professional Standards, AU-C sec. 940), was issued in November 2015. The SAS is effective for integrated audits for periods ending on or after December 15, 2016.

The ASB concluded that, because engagements performed under AT section 501, as well as related attestation Interpretation No. 1, "Reporting Under Section 112 of the Federal Deposit Insurance Corporation Improvement Act" (AICPA, *Professional Standards*, AT sec. 9501 par. .01–.07), are required to be integrated with an audit of financial statements, it is appropriate to move the content of AT section 501 from the attestation standards into GAAS.

Statement on Standards for Attestation Engagements (SSAE) No. 18, Attestation Standards: Clarification and Recodification (AICPA, Professional Standards), issued in April 2016, is effective for practitioners' reports dated on or after May 1, 2017.

(continued)

- **21.02** Trust and asset management activities include fiduciary services provided to customers. One type of fiduciary service involves acting as a trustee. Trust activities of an institution may be an integral part of the institution's services; however, because of strict laws<sup>4</sup> governing fiduciary responsibilities, institutions conduct trust activities independently through
  - a. a separate department or division of the institution;
  - b. a separately chartered trust company; and
  - c. a contractual arrangement with the trust department or a trust company of another depository institution.
- **21.03** The organizational structures of institutions' trust departments or of trust companies vary greatly depending upon factors such as the scope of trust activities, the complexity of trust services offered, management's preference, and the historical development of the entity. Trust organizations vary from small operations with one person devoted to trust activities on a part-time basis to large organizations with a variety of specialized staff such as tax attorneys, employee benefit specialists, and investment specialists.
- **21.04** Trusts can be broadly categorized as personal, corporate, or employee benefit.
- **21.05** This chapter deals primarily with how trust services and activities affect audits of the financial statements of financial institutions. However, it is important that auditors be fully aware of any regulatory expectations that may exist in the area of trust departments and design any engagements arising from those expectations in an appropriate manner.

(footnote continued)

To address concerns over the clarity, length, and complexity of its standards, the Auditing Standards Board (ASB) established clarity drafting conventions and undertook a project to redraft all the standards it issues in clarity format. The redrafting of SSAEs (attestation standards) in SSAE No. 18 represents the culmination of that process. The attestation standards are developed and issued in the form of SSAEs and are codified into sections. SSAE No. 18 recodifies the "AT" section numbers designated by SSAE Nos. 10–17 using the identifier "AT-C" to differentiate the sections of the clarified attestation standards (AT-C sections) from the attestation standards that are superseded by SSAE No. 18 (AT sections). The AT sections in AICPA *Professional Standards* remain effective through April 2017, by which time substantially all engagements for which the AT sections were still effective are expected to be completed.

This edition of the guide has not been updated to reflect changes as a result of this SSAE; however, this footnote will be updated in a future edition.

Readers are encouraged to consult the full text of SAS No. 130 and SSAE No. 18 on the AICPA's website at www.aicpa.org.

- <sup>2</sup> Auditors should consider all applicable professional independence rules when providing extended audit services, such as supplementing internal audit efforts. In particular, the auditor should comply with the AICPA's Code of Professional Conduct and, if applicable, Section 201 of the Sarbanes-Oxley Act of 2002 and related SEC and PCAOB rulings.
- <sup>3</sup> The Gramm-Leach-Bliley Act of 1999 (GLBA) repealed the blanket exemption for banks from the definition of a *broker* set forth in Section 3(a)(4) of the Securities Exchange Act of 1934 (the 1934 Act). In place of the blanket exemption, GLBA provided 11 specific exceptions to the definition of *broker*. The statutory exceptions were designed to allow banks to continue to effect securities transactions in connection with certain traditional bank activities. Regulation R, "Definition of Terms and Exemptions Relating to the 'Broker' Exceptions for Banks," defines important terms relating to the broker exceptions in the 1934 Act and provides a number of exemptions.
- $^4$  Most notably, Title 12 U.S. Code of Federal Regulations (CFR) Parts 9 and 150; state fiduciary laws often provide additional requirements.

21.06 Regulatory focus on the adequacy of auditing of trust operations of financial institutions has increased in recent years. The proliferation of trust charters in bank holding companies has led the bank and savings institution regulators to more closely assess the adequacy of secondary monitoring provided by audit functions. In cases where internal audit departments do not exist or lack the expertise necessary to audit the complexities of financial institutions or trust operations, or both, the regulators are looking often to independent auditors to supplement the existing resources. In their respective rules on audits of fiduciary activities for both federal savings associations (Title 12 U.S. Code of Federal Regulations [CFR] Part 150.440) and national banks (12 CFR 9.9), the Office of the Comptroller of the Currency (OCC) requires that management arrange for a suitable audit of trust operations through the efforts of external or internal auditors, or both, on an annual basis or as part of a continuous audit process. Trusts maintaining state charters would be required to follow rules issued by the individual states.

21.07 The audit requirements of the OCC are largely related to operating and compliance controls that may not be tested in the audit of financial statements of a financial institution. In addition, the depth of testing of financial reporting controls will likely be greater than in a financial statement audit. Accordingly, the testing in these areas should be the subject of separate engagements under standards for attestation engagements or consulting standards as they relate to extended audit services.

#### Personal Trusts

**21.08** Personal trust accounts may be established for individuals or other entities such as foundations, college endowments, and not-for-profit organizations. A brief description of the primary kinds of personal trusts follows:

- a. Testamentary trusts are created under a will. Administrative responsibility begins when assets are transferred from the estate to the trust. Almost all testamentary trusts are irrevocable.
- b. Voluntary trusts (inter vivos), also referred to as living trusts, are established by individuals during their lifetimes. This type of trust is often established with powers of revocation or amendment. Furthermore, it has been increasingly common for the grantor of the trust to retain the power to control or participate in deciding on investments resulting in a self-directed trust.
- c. Court trusts are trusts in which the trustee is accountable to a court. Court trusts generally include decedents' estates (under which the courts appoint administrator institutions to settle the estates of persons who either died without leaving wills or who nominated the institutions as executors in their wills), guardianships, and some testamentary trusts.
- d. Agency agreements provide for the care of other parties' securities and properties. Safekeeping and custodianship agreements are two of the more common types.
- e. Property management agreements provide for the management of property, for example, real estate or securities investments, by the trustee institution. The institution, as agent, has managerial duties and responsibilities appropriate to the kind of property being managed. (Such agreements also may exist for employee benefit trusts.)

**21.09** Closely held business management responsibilities may arise through the normal course of events when an institution serves as trustee of a personal trust (or employee benefit trust) that holds ownership of the enterprise, through involvement in winding down the affairs of an estate, or through a specialized property management agreement.

## **Corporate Trusts**

- **21.10** A brief description of the primary kinds of corporate trust activities follows:
  - a. As *transfer agent*, the trust department or trust company transfers registered (in contrast to bearer) securities from one owner to another and maintains the records of ownership.
  - b. As registrar, the trust department or trust company maintains for corporations control over the number of shares issued and outstanding.
  - c. As joint registrar transfer agent, the trust department or trust company acts jointly as registrar and transfer agent for the same issue.
  - d. As paying agent, the trust department or trust company distributes interest or dividend payments or redeems bonds and bond coupons of corporations and political subdivisions within the terms of an agency agreement.
  - e. When an institution is a trustee under indenture, the trust department or trust company acts as an agent designated by a municipality, corporation, or other entity to administer specified cash receipt or payment functions. The trust department or trust company performs the duties specified in the agreement, which might include holding collateral; issuing bond instruments; maintaining required records, accounts and documentation; monitoring for default; ensuring legal compliance; and effecting the payment of principal and interest.

## **Employee Benefit Trusts**

- 21.11 In most cases, the assets of an employee benefit plan must be held in trust with the trustee named in the trust instrument or benefit plan. The trustee manages and controls the assets of an employee benefit plan, although actual investment decisions might be turned over to an investment committee or investment manager. Trustees of employee benefit plans are, in most cases, fiduciaries. The fiduciary's responsibility is to manage the assets of the employee benefit plan so that those assets are used exclusively to provide benefits for employees and beneficiaries and to defray administrative costs of the plan. The fiduciary is charged with using the care and skill in managing the plan that a prudent man would in similar circumstances. The fiduciary must diversify the investments of the plan to minimize the risk of large losses. Finally, the fiduciary must discharge its duties in accordance with the documents and instruments governing the employee benefit plan and the Employee Retirement Income Security Act of 1974 (ERISA), the federal law dealing with employee benefit plans. A brief description of the primary kinds of employee benefit trusts follows:
  - a. Pension or profit sharing trusts provide for a trustee institution to manage trust funds established for the benefit of eligible company officers or employees or for members of a union, professional

- organization, or association. Such trusts are established by comprehensive written plans in which the trustees' powers are limited and their duties are well defined. These trusts may exist in connection with a variety of types of benefit plans, including defined benefit plans, defined contribution plans, individual retirement accounts, and health and welfare plans.
- b. Master trusts are special trust devices used to bring together various employee benefit trusts of a plan sponsor for ease of administration. For instance, an employer may have similar benefit plans for different subsidiaries, divisions, or classes of employees. Rather than maintain separate employee benefit trusts for each plan, all of the plans, subject to restrictions of ERISA, may pool the trust assets in a single master trust and maintain separate subaccounts for each plan to preserve accountability. A master trust may also be structured to establish separate pools of trust assets managed by different investment advisers selected by the plan sponsor.

#### Common or Collective Trust Funds

21.12 A common or collective trust fund is a bank-administered trust that holds commingled individual trust accounts (that is, personal or employee benefit trusts). The account assets are pooled to achieve greater diversification of investment, stability of income, or other investment objectives. They are similar to a mutual fund but are not regulated by the SEC and are not required to be registered with the SEC under an exclusion from the Investment Company Act of 1940. They are instead required to follow OCC regulations at 12 CFR 9. Under OCC regulations, there are two types of funds: (a) common trust funds, which are maintained exclusively for the collective investment of accounts for which the institution serves as trustee, executor, administrator, conservator, and guardian, and (b) collective trust funds or commingled pension trust funds, which consist solely of assets of retirement, pension, profit sharing, stock bonus, or other trusts that are exempt from federal income taxes.

## **Regulatory Matters**

- **21.13** Some institutions are also involved with mutual funds. Their involvement may range from corporate trust activities, which are generally administrative in nature, to investment advisory activities, or may simply involve custodial activities. Some institutions sell funds sponsored by an independent fund group. Others may use their name on a fund sponsored by an affiliate. Readers may refer to the FDIC's nondeposit investment products exam procedures as a resource for evaluating these transactions.
- **21.14** The federal banking agencies use the Uniform Interagency Trust Rating System (UITRS) as a tool to evaluate the soundness of fiduciary activities of financial institutions on a uniform basis and to identify those institutions requiring special supervisory attention. The UITRS was revised in 1998 to place more emphasis on risk management and more closely align the ratings

<sup>&</sup>lt;sup>5</sup> Common trust funds are exempt from federal income taxes under Section 584 of the IRC. Collective investment funds are exempt from federal income taxes under IRS Revenue Ruling 81-100.

definitions language and tone with those of the capital adequacy, asset quality, management, earnings, and liquidity ratings definitions.

# **Accounting and Financial Reporting**

# © Update 21-1 Accounting and Reporting: Revenue From Contracts With Customers

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14) for annual reporting periods of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an over-the-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period. Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as either

- an annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- an annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB ASC 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605 as well as guidance within the 900 series of industry-specific topics. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts).

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2014-09, see appendix E of this guide.

**21.15** Although a trust department or trust entity may have responsibility for the custody of trust assets, they are not assets of the institution and, therefore, should not be included in the institution's financial statements according to FASB ASC 942-605-25-3. However, cash accounts of individual trusts are often deposited with the institution in demand and time deposit accounts, and revenues and expenses related to fees for trust activities are recognized in

©2016, AICPA AAG-DEP 21.15

the institution's income. Trust department income should be presented on the accrual basis.  $^6$ 

**21.16** Financial institutions often make financial statement disclosures describing the nature of the trust activities and are required to apply the provisions of FASB ASC 450, *Contingencies*, to any contingencies that may exist related to trust activities.

# Auditing 7,8

## **Objectives**

**21.17** The primary objectives of financial statement audit procedures applied in the trust operations area are to obtain sufficient appropriate evidence that

- a. the institution has properly described and disclosed contingent liabilities associated with trust activities in the financial statements, and
- b. fee income resulting from trust activities is recognized properly in the institution's financial statements.

## **Planning**

21.18 In accordance with AU-C section 315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement (AICPA, Professional Standards), the objective of the auditor is to identify and assess the risks of material misstatement, whether due to fraud or error, at the financial statement and relevant assertion levels through understanding the entity and its environment, including the entity's internal control, thereby providing a basis for designing and implementing responses to the assessed risks of material misstatement (see chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide for additional information). The following is

 $<sup>^6</sup>$  In accordance with 12 CFR 9.10(b), a national bank may deposit funds of a fiduciary account that are awaiting investment or distribution in the commercial, savings, or another department of the bank, unless prohibited by applicable law. To the extent that the funds are not insured by the FDIC, the bank should set aside collateral as security, under the control of the appropriate fiduciary officers and employees, in accordance with 12 CFR 9.10(b)(2). The market value of the collateral set aside must at all times equal or exceed the amount of the uninsured fiduciary funds.

<sup>&</sup>lt;sup>7</sup> In December 2009, the SEC adopted rules designed to substantially increase the protections for investor funds and securities of which an investment adviser registered with the SEC has custody. Depending on the investment adviser's custody arrangement, the rules require the adviser to be subject to a surprise exam and custody controls review generally not required previously. Readers are encouraged to review the full text of final rule Release No. IA-2968, Custody of Funds or Securities of Clients by Investment Advisers and the related interpretive Release No. IA-2969, Commission Guidance Regarding Independent Public Accountant Engagements Performed Pursuant to Rule 206(4)-2 Under the Investment Advisers Act of 1940. Additionally, readers can access the SEC's Staff Responses to Questions about the Custody Rule from the SEC website. See the section "Compliance With Laws and Regulations" in chapter 5, "Audit Considerations and Certain Financial Reporting Matters," of this guide for further discussion on controls associated with laws and regulations.

<sup>&</sup>lt;sup>8</sup> The auditing content in this guide focuses primarily on generally accepted auditing standards issued by the ASB and is applicable to audits of nonissuers. See the section "Applicability of GAAS and PCAOB Standards" of the preface to this guide for a discussion of the definitions of *issuers* and *nonissuers* as used throughout this guide. Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: *Considerations for Audits Performed in Accordance With PCAOB Standards*.

a list of factors related to trust services and activities that could influence the risks of material misstatement:

- a. The organization of the trust department and the degree of separation from the commercial banking departments (for example, the role of legal counsel in trust account administration and the vulnerability to disclosure of insider information)
- b. The nature of comments on trust operations indicated in supervisory agency or internal audit reports
- c. The extent and nature of insurance coverage
- d. The type and frequency of lawsuits, if any, brought against the institution and arising from trust operations
- e. The nature, complexity, and reliability of data processing systems
- f. The nature and extent of lending of securities from trust accounts

The significance of an institution's exposure to liability (including liability related to the reporting of tax information) is a function of (a) the relative significance of the trust assets administered, (b) whether the institution has discretionary investment authority, (c) the complexity of transactions entered into by the trust, (d) the number of trusts administered, and (e) the effectiveness of administration of the trust. It is important for auditors not to underestimate the importance of an institution's trust department. Auditors might also consider the complexity of the assets owned and the investment mandate.

# Internal Control Over Financial Reporting and Possible Tests of Controls

- 21.19 AU-C section 315 addresses the auditor's responsibility to identify and assess the risks of material misstatement in the financial statements through understanding the entity and its environment, including the entity's internal control. Paragraphs .13—.14 of AU-C section 315 state that the auditor should obtain an understanding of internal control relevant to the audit and, in doing so, should evaluate the design of those controls and determine whether they have been implemented by performing procedures in addition to inquiry of the entity's personnel. (See chapter 5 of this guide for further discussion of the components of internal control.) To provide a basis for designing and performing further audit procedures, paragraph .26 of AU-C section 315 states that the auditor should identify and assess the risks of material misstatement at the financial statement level and at the relevant assertion level related to classes of transactions, account balances, and disclosures.
- **21.20** Accounting systems for trust departments generally use sophisticated electronic data processing systems. The accounting records of a trust department generally reflects the department's asset holdings and liabilities to trust customers, the status of each trust account, and all transactions relating to each trust account. Records providing detailed information for each trust account generally should include the following:
  - Principal (corpus) control account
  - Principal cash account
  - Income cash account
  - Investment records for each asset owned, such as stocks, bonds, notes and mortgages, savings and time accounts, real property, and sundry assets

©2016. AICPA AAG-DEP 21.20

- Liability record for each principal trust liability
- Investment income
- **21.21** The auditor may need to evaluate trust departments' and trust companies' overall internal control over financial reporting, including, but not limited to, the following controls:
  - Individual account and departmental transactions (activity control) and suspense items are reconciled and recorded in a complete, accurate, and timely manner. Appropriate supervisory personnel routinely review and approve reconciliations.
  - Written policies, procedures, and controls exist for securities lending activities, including review of the borrower's creditworthiness, a formal lending agreement, and minimum collateral requirements.
  - Written policies and procedures exist for collateral maintenance and reinvestment.
  - Periodic reconciliations of the trust funds on deposit with the institution or its custodian are performed by an employee having no check signing authority or access to unissued checks and related records. Appropriate supervisory personnel routinely review and approve reconciliations.
  - Measures have been taken to safeguard trust assets by dual control.
  - Accurate files of documents creating trusts and authorizing transactions are maintained. A review of document files for completeness is performed by an independent individual.
  - Vault deposits and withdrawals are reconciled with accounting records to promptly reflect the purchase and sale of trust assets. Appropriate supervisory personnel routinely review and approve reconciliations.
  - Reconciliation of agency accounts (for example, dividends, coupons, bond redemptions, and holdings or principal balance) is performed regularly by an employee having no access to unissued checks or participation in the disbursement function.
  - Periodic physical inspection of assets or confirmation of trust assets is conducted by an independent person.
  - There is frequent reporting and written approval of uninvested cash balances and overdrafts.
  - Procedures exist to ensure compliance with income and other tax filing and remittance requirements.
  - Reviews are conducted to make sure all duties required by the governing trust instruments or agency contracts (legal compliance) are performed.
  - Trust fees are posted to the general ledger by persons who do not have conflicting duties from detailed records produced by the trust system.
  - Trust fee income is periodically reviewed for accuracy by appropriate personnel with no conflicting duties.

Management or independent employees perform an analytical review of trust fee income. Differences are investigated timely.

## **Financial Reporting Controls of the Trust**

#### © Update 21-2 Attestation: Clarification and Recodification

Statement on Standards for Attestation Engagements (SSAE) No. 18, Attestation Standards: Clarification and Recodification (AICPA, Professional Standards), issued in April 2016, is effective for practitioners' reports dated on or after May 1, 2017.

To address concerns over the clarity, length, and complexity of its standards, the Auditing Standards Board (ASB) established clarity drafting conventions and undertook a project to redraft all the standards it issues in clarity format. The redrafting of SSAEs (attestation standards) in SSAE No. 18 represents the culmination of that process. The attestation standards are developed and issued in the form of SSAEs and are codified into sections. SSAE No. 18 recodifies the "AT" section numbers designated by SSAE Nos. 10–17 using the identifier "AT-C" to differentiate the sections of the clarified attestation standards (AT-C sections) from the attestation standards that are superseded by SSAE No. 18 (AT sections). The AT sections in AICPA *Professional Standards* remain effective through April 2017, by which time substantially all engagements for which the AT sections were still effective are expected to be completed.

This edition of the guide has not been updated to reflect changes as a result of this SSAE; however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this SSAE on the AICPA's website at www.aicpa.org.

**21.22** Additional controls that the auditor may wish to consider for engagements not limited to the audit of financial statements (for example, directors' exams, engagements under AT section 801, *Reporting on Controls at a Service Organization* [AICPA, *Professional Standards*], and agreed upon procedures or other extended audit services) include the following:

- Authorization and review procedures are in place to ensure that assets accepted into a trust conform to provisions of the trust and applicable laws and regulations.
- The physical and administrative security (physical control) of assets for which the trust department has responsibility is segregated from transaction authorization and recordkeeping.
- Trust assets are segregated from the institution's assets and are periodically inspected by people outside the trust department or trust company.
- Trust assets are registered in the name of the institution as fiduciary or in the name of the nominee.
- Proper approval is obtained from cofiduciaries (or investment power holders in self-directed trusts) for investment changes, disbursements, and so forth.
- Approval of the individual purchase and sale of all trust investments is performed by the trust or investment committee or its

designees. It is important that for assets where the trustee has discretionary (investment powers) authority, investment restrictions imposed by the client are being adhered to. Paragraph .19 of AU-C section 315 states that the auditor should obtain an understanding of the information system, including the related business processes relevant to financial reporting.

- Procedures exist to ensure proper classification of trust assets, both by trust title and by nature of asset, daily posting of journals containing detailed descriptions of principal and income transactions, and establishment of control accounts for various asset classifications, including principal and income cash.
- Procedures exist to safeguard unissued supplies of stocks and bonds by dual control.
- Periodic mailings, at least quarterly, are made of account statements of activity to an external party designated by the client.
- Policies and procedures exist related to identification and resolution of failed trades and the contractual settlement of trades posted to trust accounts.
- Complete legal files are maintained.
- Based on the nature of trust contracts, accurate tax reporting is performed.

#### Substantive Tests Related to Financial Statement Audits

**21.23** Testing of trust department revenues and expenses. Although a substantial amount of activity may be conducted and reported on within the trust department, items typically reflected in the institution's financial statements are income from trust or agency services and trust operation expenses. Those areas may be tested independently or may be integrated, as appropriate, with other tests of trust operations.

**21.24** Contingent liabilities. The auditor designs audit procedures to determine whether any contingent liabilities should be recognized or disclosed in the institution's financial statements. Acceptance of certain assets, such as real estate with environmental contamination that subjects the trustee to environmental liabilities and ineligible investments in employee benefit trusts subject to ERISA, may result in substantial liabilities for both the trust and trustee. If the institution is providing guarantees to beneficiaries or others associated with the trusts, procedures may need to be performed to determine if the institution has complied with the requirements of FASB ASC 460, Guarantees or FASB ASC 815, Derivatives and Hedging, if the guarantee meets the definition of a derivative. Further, the auditor considers determining the extent to which an institution has engaged in off-balance-sheet activities that create commitments or contingencies, including innovative transactions involving securities and loans (such as transfers with recourse or put options), that could affect the financial statements, including disclosures in the notes. Inquiries of management relating to such activities might be formalized in the representation letter normally obtained at year-end. The auditor also considers reviewing the institution's documentation to determine whether particular transactions are sales or financing arrangements.

#### Substantive Procedures Related to the Trust

**21.25** Additional substantive procedures that the auditor may wish to consider for engagements not limited to the audit of financial statements (for example, directors' exams, engagements under AT section 801 and agreed upon procedures or other extended audit services) are included in the following paragraphs.

#### © Update 21-3 Attestation: Clarification and Recodification

SSAE No. 18, issued in April 2016, is effective for practitioners' reports dated on or after May 1, 2017.

To address concerns over the clarity, length, and complexity of its standards, the ASB established clarity drafting conventions and undertook a project to redraft all the standards it issues in clarity format. The redrafting of attestation standards in SSAE No. 18 represents the culmination of that process. The attestation standards are developed and issued in the form of SSAEs and are codified into sections. SSAE No. 18 recodifies the "AT" section numbers designated by SSAE Nos. 10–17 using the identifier "AT-C" to differentiate the sections of the clarified attestation standards from the attestation standards that are superseded by SSAE No. 18. The AT sections in AICPA *Professional Standards* remain effective through April 2017, by which time substantially all engagements for which the AT sections were still effective are expected to be completed.

This edition of the guide has not been updated to reflect changes as a result of this SSAE; however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this SSAE on the AICPA's website at www.aicpa.org.

- **21.26** Examination of a trust department's activity includes tests of systems and procedures that are common to the management of all or most individual trusts or agency accounts and tests of the activity in selected representative individual trust accounts in each area of trust department service (for example, personal, corporate, and employee benefit).
- **21.27** Testing of trust activities' common procedures. The procedures followed for the numerous types of trusts and agency activities involve many common or similar functions. Tests of the department's conduct of those activities may be on the department as a whole rather than on individual trusts. Functions that may be tested by the department include, but are not limited to, the following:
  - Opening of new accounts
  - Receipt and processing of the initial assets that constitute an account
  - Processing of purchases, sales, and exchanges of principal assets
  - Receipt and payment of cash or other assets
  - Collateralization of trust assets held in deposit accounts at the institution, affiliate, or outside custodian, where contractually required
  - Execution of specified trust or agency activities

- Determination of fees and charging of fees to accounts
- Processing of trust assets in and out of the trust vault
- Closing of accounts

**21.28** Testing of account activity. Depending on the circumstances, paragraph .A46 of AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards), states that the auditor may determine the following:

- Performing only substantive analytical procedures will be sufficient to reduce audit risk to an acceptably low level, such as, for example, when the auditor's assessment of risk is supported by audit evidence from tests of controls.
- Only tests of details are appropriate.
- A combination of substantive analytical procedures and tests of details are most responsive to the assessed risks.

The auditor's determination as to the substantive procedures that are most responsive to the planned level of detection risk is affected by whether the auditor has obtained audit evidence about the operating effectiveness of controls.

**21.29** The tests may cover asset validation, asset valuation, and account administration. For asset validation, a sample of accounts may be selected, trial balances of assets obtained, and the physical existence of assets for which the trust is responsible determined on a test basis. For account administration, a sample of trust accounts may be selected for testing of individual transactions. If appropriate, certain of those transactions may be incorporated in testing of common procedures in the trust department. The auditor may coordinate the selection of accounts for testing asset validation and account administration. The auditor might perform the following procedures for the selected accounts:

- a. Read the governing instrument and note the significant provisions.
- b. Review activity during the period being audited for compliance with the governing trust instrument and applicable laws and regulations.
- c. Review the assets held for compliance with the provisions of the governing trust instrument.
- d. Examine brokers' advices or other documentary evidence supporting the purchase and sale of investments.
- e. For real estate accepted or acquired, determine that appropriate measures are taken to identify potential environmental liability and to properly document the evaluation.
- f. Ascertain that real estate holdings are insured and are inspected on a periodic basis and that appraisals are performed or otherwise obtained as required by the governing trust instrument and applicable laws and regulations.
- g. Obtain reasonable assurance that income from trust assets has been received and credited to the account.
- h. Obtain reasonable assurance that necessary payments have been made.
- *i*. Test computation and collection of fees.

- *j*. Determine whether the account has been reviewed by the investment committee as required by the supervisory authorities or by local regulations.
- k. Test the amounts of uninvested cash to determine whether amounts maintained and time held are not unreasonable.
- Review any overdrafts and obtain reasonable assurance that they
  have a valid business purpose and are covered by appropriate borrowings to avoid violations of laws and regulations.
- m. Independently test market values used in valuing investments.
- Review the "soft dollar" charges allocated to funds for appropriateness.
- Determine whether required tax returns have been filed in accordance with IRC regulations.
- Review the adequacy of trust reporting of co-trustees and beneficiaries.
- q. Confirm individual trust account assets, liabilities, and activity with co-trustees and beneficiaries.
- r. Test valuation procedures.

#### Audits of Unit Investment Trusts

**21.30** The AICPA Audit and Accounting Guide *Investment Companies* provides guidance on the auditing of financial statements of investment companies and unit investment trusts.

©2016, AICPA AAG-DEP 21.30

# Chapter 22

# **Insurance Activities**

Gray shaded text in this chapter reflects guidance issued but not yet effective as of the date of this guide, July 1, 2016, but becoming effective on or prior to December 31, 2016, exclusive of any option to early adopt ahead of the mandatory effective date. Unless otherwise indicated, all unshaded text reflects guidance that was already effective as of the date of this guide.

# FASB Accounting Standards Update (ASU) No. 2014-09, Revenue from Contracts with Customers (Topic 606)

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14, *Revenue from Contracts with Customers (Topic 606): Deferral of the Effective Date)* for annual reporting periods of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an overthe-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period. Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as of either

- an annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- an annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB Accounting Standards Codification (ASC) 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605, *Revenue Recognition*, as well as guidance within the 900 series of industry-specific topics, including FASB ASC 944, *Financial Services—Insurance*. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts).

Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

The AICPA has formed 16 industry task forces to assist in developing a new Accounting Guide on revenue recognition that will provide helpful hints and illustrative examples for how to apply the new standard. Revenue recognition implementation issues identified by the Depository and Lending Institutions Revenue Recognition Task Force will be available for informal comment,

©2016, AICPA AAG-DEP 22

after review by the AICPA Financial Reporting Executive Committee, at the "Depository and Lending Institutions Revenue Recognition Task Force" page at www.aicpa.org.

Readers are encouraged to submit comments to revreccomments@aicpa.org.

For more information on FASB ASU No. 2014-09, see appendix E, "The New Revenue Recognition Standard: FASB ASU No. 2014-09," of this guide.

### Introduction

**22.01** Insurance operations ordinarily are an integral part of consumer finance activities. This chapter deals primarily with insurance business generated from finance customers, though it also addresses insurance coverage provided to others who are not also finance customers.

## **Types of Insurance Coverage**

**22.02** Insurance activities of finance companies often involve insuring risks related to loan transactions. Following are the three general types of insurance coverage associated with those transactions:

- a. Credit life coverage for loan repayment in the event of the debtor's death
- b. Credit accident and health coverage for installment loan payments in the event of the debtor's illness or disability for an extended period
- c. Property and liability coverage on collateral or other property associated with the loan transaction

### **Credit Life**

**22.03** Credit life insurance is a form of term insurance that provides for loan repayment if the debtor dies before the loan is fully paid. It ordinarily is written on a single-premium basis, with the amount of the premium added to the loan balance and paid as part of the scheduled installments on the loan.

**22.04** Credit life insurance includes level term insurance and decreasing term insurance. *Level term insurance* provides a fixed amount of coverage, generally the original amount of the loan. *Decreasing term insurance*, the more common type, insures the debtor's life to the extent of the unpaid balance of the loan, sometimes less any delinquent payments, at the date of death. However, decreasing term insurance usually is based on the contractual loan period. Therefore, the insurer may not pay off the entire uncollected balance on the loan if it is in delinquency status at the time of the debtor's death. The extent to which delinquent installments are covered generally depends on the insurance contract and on applicable state insurance rules and regulations.

**22.05** The insurer's risk exposure on a policy at a given point in time under level term insurance differs from that under decreasing term credit life insurance. Because level term insurance provides coverage equal to the original amount of the loan, the insurer's risk exposure is constant throughout the term of the loan. In contrast, the insurer's exposure under decreasing term insurance decreases as scheduled loan repayments become due, usually in direct proportion to the regular monthly reductions of the loan balance.

### Credit Accident and Health

**22.06** Credit accident and health insurance requires the insurer to make the debtor's monthly loan payments during extended periods of illness or disability. Ordinarily it is written on a single-premium basis, with the premium added to the loan amount and, hence, paid as part of the periodic installments. Under an accident and health policy, the insurer's total risk exposure decreases—as in a decreasing term credit life insurance policy—as loan repayments are made. However, the size of potential claims and the related risk exposure do not decrease in direct proportion to the reduction in the unpaid loan balance, because most credit accident and health insurance claims are for short-duration disabilities that are cured in a period shorter than the remaining loan term.

## **Property and Liability**

22.07 Ordinarily, a finance company requires that the collateral pledged as security to a loan be protected by property insurance. Such coverage may be obtained from the lender's insurance subsidiary or from an unaffiliated insurer. The amount of coverage is usually based on the value of the collateral and does not necessarily bear a relationship to the unpaid balance of the loan. Property insurance policies issued in connection with finance transactions can be written either on a single-premium basis for the loan term or for an annual or other period of less than the remaining loan term, and the policy renewed as desired. Premiums charged by lenders' insurance affiliates for property insurance coverage related to finance transactions frequently are added to the loan amount and paid as part of the regular installment payments on the loan.

## **Writing Policies**

**22.08** An insurance subsidiary of a finance company may be a direct writing or a reinsurance company. A direct writing company writes the insurance policies in its name. A reinsurance company insures policies written by direct writing companies.

**22.09** The insurance can be issued on either a group or an individual policy basis. For group coverage, the insurer issues the policy to the finance company, which in turn issues individual certificates to its debtor-customers. Group policies may be subject to experience-rated premium adjustments based on experience and profitability of the group being covered.

#### Commissions

**22.10** Insurers, both insurance subsidiaries and independent companies, may pay commissions to companies. Those payments may be in the form of advance commissions computed as a percentage of premiums, retrospective or experience-rated commissions, or combinations of advance commissions and retrospective commissions.

## **Regulatory Matters**

**22.11** Credit unions may offer, through a credit union service organization (CUSO), the following insurance brokerage or agency services:

- a. Agency for sale of insurances
- b. Provision of vehicle warranty programs

- c. Provision of group purchasing programs
- d. Real estate settlement services

Other activities or services that CUSOs may provide are outlined in Part 712.5 of the National Credit Union Administration regulations.

# **Accounting**

**22.12** The primary source of accounting guidance for entities that issue insurance contracts is FASB ASC 944. Additionally, the AICPA Audit and Accounting Guides *Life and Health Insurance Entities* and *Property and Liability Insurance Entities* provide further information.

#### Premium Income

- **22.13** FASB ASC 944-605 provides guidance to insurance entities on accounting for and financial reporting of revenue from insurance contracts. As discussed in FASB ASC 944-20-15-2, insurance contracts should be classified as short or long-duration contracts, depending on whether the contracts are expected to remain in force for an extended period.
- **22.14** FASB ASC 944-20-15 provides the following guidance on factors to consider in determining whether a contract is of short or long duration:
  - a. Short-duration contracts provide insurance protection for a fixed period of short duration and enable insurers to cancel the contracts or to adjust provisions of the contracts at the end of any contract period, such as adjusting the amount of premiums charged or coverage provided, according to FASB ASC 944-20-15-7.
  - Long-duration contracts generally are not subject to unilateral changes in their provisions, such as noncancelable or guaranteed

- For short-duration contracts, the targeted improvements were limited to enhancing disclosures (not including measurement and recognition). FASB ASU No. 2015-09, Financial Services—Insurance (Topic 944): Disclosures about Short-Duration Contracts, issued in May 2015, requires additional disclosures about the liability for unpaid claims and claim adjustment expenses for all insurance entities that issue short-duration contracts as defined in FASB Accounting Standards Codification (ASC) 944, Financial Services—Insurance. FASB ASU No. 2015-09 is effective for public business entities for annual periods beginning after December 15, 2015, and interim periods within annual periods beginning after December 15, 2016, For all other entities, FASB ASU No. 2015-09 is effective for annual periods beginning after December 15, 2017. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.
- For long-duration contracts, the project should focus on making targeted improvements to
  existing GAAP addressing recognition, measurement, presentation and disclosure. Readers are encouraged to visit the "Technical Agenda" page at www.fasb.org for the latest
  developments regarding the Insurance—Targeted Improvements to the Accounting for
  Loan-Duration Contracts project and how it may impact the guidance in this chapter.

<sup>&</sup>lt;sup>1</sup> In August 2007, FASB issued an Invitation to Comment, An FASB Proposal: Accounting for Insurance Contracts by Insurers and Policyholders. That invitation to comment included a discussion paper issued by the International Accounting Standards Board (IASB), Preliminary Views on Insurance Contracts, setting forth its preliminary views on the main components of an accounting model for an issuer's rights and obligations (assets and liabilities) under an insurance contract. FASB's objective in undertaking the proposed project was to develop a common, high-quality standard that addresses recognition, measurement, presentation, and disclosure requirements for insurance contracts. Although IASB and FASB reached common decisions in many areas, they reached different conclusions in others. In June 2013, FASB issued a proposed Accounting Standards Update (ASU), Insurance Contracts (Topic 834). In light of feedback received on the proposed ASU, FASB elected to limit the scope to insurance entities as described in existing U.S. generally accepted accounting principles (GAAP) (instead of continuing to include all entities that issue insurance contracts or purchase reinsurance contracts as proposed in the June 2013 exposure draft). FASB also decided the following:

renewable contracts, and require performance of various functions and services (including insurance protection) for extended periods, according to FASB ASC 944-20-15-10.

- **22.15** Paragraphs 1 and 5 of FASB ASC 944-20-55 state that examples of short duration contracts include most property and liability insurance contracts and certain term life insurance contracts, such as credit life insurance. Accident and health insurance contracts may be of short duration or long duration, depending on whether the contracts are expected to remain in force for an extended period. For example, individual and group insurance contracts that are noncancelable or guaranteed renewable (renewable at the option of the insured), or collectively renewable (individual contracts within a group are not cancelable), ordinarily are long-duration contracts.
- **22.16** Insurance policies issued in connection with consumer lending generally are considered to represent short-duration contracts.
- **22.17** FASB ASC 944-605-25-1 states that premiums from short duration contracts should be recognized as revenue over the period of the contract in proportion to the amount of insurance protection provided. For those few types of contracts for which the period of risk differs significantly from the contract period, premiums should be recognized as revenue over the period of risk in proportion to the amount of insurance protection provided. That generally results in premiums being recognized as revenue evenly over the contract period (or the period of risk, if different), except for those few cases in which the amount of insurance protection declines according to a predetermined schedule.

### **Acquisition Costs**

**22.18** Under U.S. generally accepted accounting principles (GAAP), FASB ASU No. 2010-26, Financial Services—Insurance (Topic 944): Accounting for Costs Associated with Acquiring or Renewing Insurance Contracts (a consensus of the FASB Emerging Issues Task Force), that was codified into FASB ASC 944-30 states that only acquisition costs that are directly related to the successful acquisition of a contract can be capitalized as deferred acquisition costs (DAC). As noted in FASB ASU No. 2010-26, if the initial application of the amendments in FASB ASU No. 2010-26 results in the capitalization of acquisition costs that had not been capitalized previously by an entity, the entity may elect not to capitalize those types of costs. These deferred amounts are recorded as an asset on the balance sheet and amortized to income in a systematic manner based on related contract revenues or gross profits (or gross margins), as appropriate. Previously, the guidance of FASB ASC 944-30 did not address successful versus unsuccessful efforts.<sup>2</sup>

(continued)

<sup>&</sup>lt;sup>2</sup> The AICPA issued two technical practice aids to assist with the retrospective adoption of FASB ASU No. 2010-26, Financial Services—Insurance (Topic 944): Accounting for Costs Associated with Acquiring or Renewing Insurance Contracts (a consensus of the FASB Emerging Issues Task Force):

Technical Questions and Answers (Q&A) section 6300.38, "Retrospective Application of ASU No. 2010-26" (AICPA, Technical Questions and Answers), addresses these questions:

<sup>—</sup> If different levels of historical information are available for various products, how should this information be included when retrospectively applying FASB ASU No. 2010-26 if it is impracticable to determine the cumulative effect of applying a change in accounting principle to all prior periods?

- **22.19** Determination of deferrable costs. The FASB ASC glossary defines acquisition costs as costs that are related directly to the successful acquisition of new or renewal insurance contracts.
- **22.20** As stated in FASB ASC 944-30-25-1A, an insurance entity should capitalize only the following as acquisition costs related directly to the successful acquisition of new or renewal insurance contracts:
  - a. Incremental direct costs of contract acquisition
  - b. The portion of the employee's total compensation (excluding any compensation that is capitalized as incremental direct costs of contract acquisition) and payroll-related fringe benefits related directly to time spent performing any of the following acquisition activities for a contract that actually has been acquired:
    - i. Underwriting
    - ii. Policy issuance and processing
    - iii. Medical and inspection
    - iv. Sales force contract selling
  - c. Other costs related directly to the insurer's acquisition activities in item (b) that would not have been incurred by the insurance entity had the acquisition contract transaction(s) not occurred
  - d. Advertising costs that meet the capitalization criteria in FASB ASC 340-20-25-4
- **22.21** Incremental direct costs of contract acquisition. The FASB ASC glossary defines incremental direct costs of contract acquisition as a cost to acquire an insurance contract that has both of the following characteristics:
  - a. It results directly from, and is essential to, the contract transaction(s).
  - b. It would not have been incurred by the insurance entity had the contract transaction(s) not occurred.
- **22.22** FASB ASC 944-30-55-1 discusses the types of incremental direct cost of contract acquisition to be capitalized under item (a) in FASB ASC 944-30-25-1A. Such costs include the following:
  - a. An agent or broker commission or bonus for successful contract acquisition(s)
  - b. Medical and inspection fees for successful contract acquisition(s)
- **22.23** Total compensation, benefits, and other costs directly related to acquisition activities. Items (a)–(c) in FASB ASC 944-30-25-1A require that only the portion of costs related directly to time spent performing specified acquisition activities for a contract that actually has been acquired (that is, successful efforts) may be deferred.

(footnote continued)

- Can FASB ASU No. 2010-26 be applied retrospectively to different points in time for various products?
- Q&A section 6300.39, "Cumulative Effect of Change in Accounting Principle—ASU No. 2010-26" (AICPA, Technical Questions and Answers), provides some items to consider when evaluating the direct effects of retrospective application of FASB ASU No. 2010-26.

- **22.24** FASB ASC 944-30-55-1C discusses that payroll-related fringe benefits include any costs incurred for employees as part of the total compensation and benefits program. Examples of such benefits include all of the following:
  - a. Payroll taxes
  - b. Dental and medical insurance
  - c. Group life insurance
  - d. Retirement plans
  - e. 401(k) plans
  - f. Stock compensation plans, such as stock options and stock appreciation rights
  - g. Overtime meal allowances
- **22.25** FASB ASC 944-30-55-1G discusses that the portion of total compensation of executive employees that relates directly to the time spent approving successful contracts may be deferred as acquisition costs. For example, the amount of compensation allocable to time spent by members of a contract approval committee is a component of acquisition costs.
- **22.26** FASB ASC 944-30-55-1A discusses that examples of other costs related directly to the insurer's acquisition activities in item (b) in FASB ASC 944-30-25-1A that would not have been incurred by the insurance entity had the acquisition contract transaction(s) not occurred include all of the following:
  - a. Reimbursement of costs for air travel, hotel accommodations, automobile mileage, and similar costs incurred by personnel relating to the specified activities
  - b. Costs of itemized long-distance telephone calls related to contract underwriting
  - c. Reimbursement for mileage and tolls to personnel involved in onsite reviews of individuals before the contract is executed
- **22.27** *Direct-response advertising*. FASB ASC 340-20-25-4 notes that the costs of direct-response advertising should be capitalized if both of the following conditions are met:
  - a. The primary purpose of the advertising is to elicit sales to customers who could be shown to have responded specifically to the advertising.
  - b. The direct-response advertising results in probable future benefits.
- **22.28** FASB ASC 340-20-25-6 specifies that in order to conclude that advertising elicits sales to customers who could be shown to have responded specifically to the advertising, there must be a means of documenting that response, including a record that can identify the name of the customer and the advertising that elicited the direct response. Examples of such documentation include the following:
  - a. Files indicating the customer names and related direct-response advertisement
  - b. A coded order form, coupon, or response card included with an advertisement indicating the customer name
  - c. A log of customers who have made phone calls to a number appearing in an advertisement that links those calls to the advertisement

**22.29** Additional guidance for issuers to consider can be found in section 1(B), "Accounting for Advertising Costs," of the SEC's *Division of Corporation Finance: Frequently Requested Accounting and Financial Reporting Interpretations and Guidance*:

Certain direct response advertising costs may be deferred under [FASB ASC 340-20-25]. Qualifying costs relating to a specific advertising activity must meet all of the following criteria:

- a. A direct relationship between a sale and the specific advertising activity for which cost is deferred must be demonstrated clearly. More than trivial marketing effort after customer response to the advertising and before the sale is consummated (such as customer contact with a sales person or furnishing of additional product or financing information) will disqualify the sale as being deemed a direct result of the advertising. A significant lapse of time between the advertising activity and the ultimate sale in an environment of broad general advertising may disqualify the sale as being deemed a direct result of the advertising.
- b. The advertisement's purpose must be one of eliciting a direct response in the form of a sale. For example, if the primary purpose (based on either intent or most frequent actual outcome) is identification of customers to which additional marketing efforts will be targeted, the advertising costs do not qualify.
- c. Deferrable costs do not include administrative costs, occupancy costs, or depreciation of assets other than those used directly in advertising activities. Payroll related costs that are deferrable include only that portion of employees' total compensation and payroll-related fringe benefits that can be shown to relate directly relate to time spent performing the qualifying activities. Costs of prizes, gifts, membership kits and similar items are not deferrable under [FASB ASC 340-20-25], but are accounted for as inventory in most circumstances.
- d. The costs must be probable of recovery from future benefits. Objective historical evidence directly relevant to the particular advertising activity is necessary to demonstrate probability of recoverability. Ancillary income from sources other than the responding customer may not be included in the calculation of future benefits for the test. Future benefits to be included in the calculation are limited to revenues derived from the customer which are the direct result of the advertising activity alone, without significant additional marketing effort. Revenues from subsequent sales and renewals may be included only if insignificant market effort is required to obtain those revenues.

**22.30** If the capitalization criteria in FASB ASC 340-20-25-4 are met, the direct-response advertising costs should be included as DAC for classification,

subsequent measurement, and premium deficiency test purposes, in accordance with FASB ASC 944 applicable to insurance industry DAC.

- **22.31** As noted in FASB ASC 340-20-25-12, the cost of the direct-response advertising directed to all prospective customers, not only the cost related to the portion of the potential customers that is expected to respond to the advertising, should be used to measure the amounts of such reported assets.
- **22.32** As noted in FASB ASC 340-20-25-8, the probable future benefits of direct-response advertising activities are probable future revenues arising from that advertising in excess of future costs to be incurred in realizing those revenues. FASB ASC 340-20-25-9 discusses that demonstrating that direct-response advertising will result in future benefits requires persuasive evidence that its effects will be similar to the effects of responses to past direct-response advertising activities of the entity that resulted in future benefits. Such evidence should include verifiable historical patterns of results for the entity. Attributes to consider in determining whether the responses will be similar include the following:
  - a. The demographics of the audience
  - b. The method of advertising
  - c. The product
  - d. The economic conditions

# © Update 22-1 Accounting and Reporting: Revenue From Contracts With Customers

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14) for annual reporting periods of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an overthe-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period. Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as either

- an annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- an annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB ASC 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605 as well as guidance within the 900 series of industry-specific topics. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the

transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts).<sup>3</sup>

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this discussion will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2014-09, see appendix E of this guide.

- **22.33** *Nondeferrable expenses.* As stated in FASB ASC 944-720-25-2, an insurance entity should charge to expense as incurred any of the following costs:
  - a. An acquisition-related cost that cannot be capitalized in accordance with FASB ASC 944-30-25-1A (for implementation guidance, see FASB ASC 944-720-55-1)
  - b. An indirect cost (for implementation guidance, see FASB ASC 944-720-55-2)
- **22.34** FASB ASC 944-720-55-1 includes examples of acquisition-related costs that cannot be capitalized in accordance with FASB ASC 944-30-25-1A:
  - a. Soliciting potential customers (except direct-response advertising capitalized in accordance with item [d] in FASB ASC 944-30-25-1A)
  - b. Market research
  - c. Training
  - d. Administration
  - e. Unsuccessful acquisition or renewal efforts (except direct-response advertising capitalized in accordance with item [d] in FASB ASC 944-30-25-1A)
  - f. Product development
- **22.35** As discussed in FASB ASC 944-30-55-1F, employees' compensation and fringe benefits related to the activities described in chapter 9, "Commissions, General Expenses, and Deferred Acquisition Costs," of the AICPA Audit and Accounting Guide *Life and Health Insurance Entities*, unsuccessful contract acquisition efforts, and idle time should be charged to expense as incurred.
- **22.36** FASB ASC 944-720-55-2 includes examples of indirect costs, per item (b) in FASB ASC 944-720-25-2, that should be expensed as incurred:
  - a. Administrative costs
  - b. Rent

<sup>&</sup>lt;sup>3</sup> FASB ASU No. 2014-09, *Revenue from Contracts with Customers (Topic 606)*, has codified the guidance on direct response advertising from FASB ASC 340-20 into FASB ASC 944-30 and clarified that the entire cost of a qualifying advertising campaign can be capitalized as direct-response advertising costs consistent with the methodology in FASB ASC 340-20. This will be a change from current guidance. FASB ASC 944-30 will require the entire cost of direct-response advertising campaign to be capitalized; FASB ASC 944-30-25-1A limits the capitalization to successful efforts only.

It should be noted that in the 2013 exposure draft *Insurance Contracts*, FASB proposed that direct response advertising costs would not qualify for inclusion in deferred acquisition costs and would be expensed as incurred. Final conclusions on direct response advertising being included in deferred acquisition costs for insurance entities will be determined after FASB redeliberates and codifies any final conclusions in its insurance project.

- c. Depreciation
- d. Occupancy costs
- e. Equipment costs, including data processing equipment dedicated to acquiring insurance contracts
- f. Other general overhead
- 22.37 FASB ASC 944-30-55-1B discusses that costs for software dedicated to contract acquisition are not eligible for deferral as DAC under the definition of that term. Such costs are not other costs directly related to the insurer's acquisition activities that would not have been incurred but for that contract under the definition of that term. Notwithstanding that the guidance, as described in chapter 9 of the AICPA Audit and Accounting Guide *Life and Health Insurance Entities*, indicates that equipment costs are expensed as incurred, insurance entities should consider the criteria in FASB ASC 350-40 to determine whether the costs qualify for capitalization as internal-use software.
- **22.38** As is the case with other business entities, expenses that do not qualify to be capitalized are charged to expense or operating expenses in the period incurred.
- **22.39** Cost determination. The identification of acquisition costs requires considerable judgment, such as how to determine successful versus unsuccessful efforts and how to determine what types of activities performed by employees are considered to be directly related to sales. The determination of the costs to be deferred can often be determined separately or via a standard costing technique or through a combination of both. Paragraphs 1D–1E of FASB ASC 944-30-55 provide additional discussion.
  - **22.40** Allocation of DAC. FASB ASC 944-30-25-1B requires the following:

To associate acquisition costs with related premium revenue, capitalized acquisition costs should be allocated by groupings of insurance contracts consistent with the entity's manner of acquiring, servicing, and measuring the profitability of its insurance contracts.

- **22.41** Determining deferrable acquisition costs. FASB ASC 944-30-25-1A discusses what types of costs an insurance entity should capitalize as acquisition costs. An insurance entity should evaluate whether employee compensation and payroll-related fringe benefits are related directly to time spent performing acquisition activities for contracts that actually have been acquired. This determination may be accomplished by a two-step process:
  - a. Determine the portion of the employee's time spent performing acquisition activities.
  - b. Determine the portion of the employee's time spent in acquisition activities directly related to contracts that have been acquired (that is, successful efforts).
- **22.42** Both of the following examples are meant to be illustrative, and the actual determination of deferrable acquisition costs under FASB ASC 944 should be based on the facts and circumstances of an entity's specific situation.
  - Example 1: In 201X, an employee of an insurance entity whose responsibility is sales force contract selling is compensated solely on a commission basis, based on the volume of business sold directly by the employee. The employee's current year commission of \$125,000 is calculated as a percentage of premiums relating to business sold directly

by the employee in the current year. In this fact pattern, the \$125,000 commission is an incremental direct cost of contract acquisition, and the entire \$125,000 would be deferrable by the insurance entity.

Example 2: In 201X, an employee of an insurance entity earned a salary and payroll-related fringe benefits of \$120,000 and spent approximately 80 percent of his or her time on qualifying acquisition activities, as described in chapter 9 of the AICPA Audit and Accounting Guide Life and Health Insurance Entities. Approximately 50 percent of this time resulted in successful contract acquisitions. The amount of costs that would be deferrable as acquisition costs would be \$48,000 (\$120,000 x 80% qualifying acquisition activities x 50% successful efforts).

- **22.43** In situations when an employee is compensated by both commission and salary, judgment will be needed to determine what costs can be capitalized as acquisition costs, based on the facts and circumstances of each specific situation.
- **22.44** Estimated gross profit. FASB ASC 944-30-35-5 notes that estimated gross profits should include estimates of amounts expected to be assessed for contract administration less costs incurred for contract administration, including acquisition costs not included in capitalized acquisition costs. The guidance in FASB ASC 944-30-35-5 also specifies that those acquisition costs to be included in estimated gross profits that are not included in capitalized acquisition costs consist of policy-related acquisition costs that are not capitalized under paragraphs 3–4 of FASB ASC 944-30-25, such as ultimate renewal commission and recurring premium taxes. Also, as stated in FASB ASC 944-30-35-5, nonpolicy-related expenses, such as certain overhead costs, and costs that are related to the acquisition of business that are not capitalized under FASB ASC 944-30-25, such as certain advertising costs, should not be included in estimated gross profit.
- **22.45** It should be noted that FASB ASU No. 2010-26 does not change the guidance related to the components or calculation of estimated gross profits. Costs that no longer qualify for deferral under FASB ASU No. 2010-26 do not meet the criteria for inclusion in estimated gross profits.
- **22.46** Commissions paid to affiliated companies and premium taxes normally are the most significant elements of acquisition costs for captive insurance companies. Deferred costs associated with payment of such commissions and other intercompany items should be eliminated in consolidation.
- **22.47** Technical Questions and Answers (Q&A) section 6300.40, "Deferrable Commissions and Bonuses Under ASU No. 2010-26" (AICPA, *Technical Questions and Answers*), states that commissions and bonuses are not deferrable solely due to an insurance entity having a sales transaction. To be deferrable as an incremental direct acquisition cost, the costs must result directly from, and be essential to, the sales transaction(s) and would not have been incurred by the insurance entity had the sales transaction(s) not occurred. Entities will need to use judgment to determine whether acquisition costs related to commissions and bonuses for employees or nonemployees meet the criterion to be deferrable under FASB ASU No. 2010-26 of resulting directly from and being essential to, the sale transaction. Paragraphs 1F–1G of FASB ASC 944-30-55 provide examples of some of the types of activities for which related costs are deferrable and those that are not. Chapter 9 of the AICPA Audit

and Accounting Guide *Life and Health Insurance Entities* contains discussion of the guidance in FASB ASU No. 2010-26.

- **22.48** The "Internal Replacement Transactions" subsections in FASB ASC 944-30 provide guidance to insurance entities on accounting for and financial reporting of unamortized acquisition costs in the event of an internal replacement transaction. This guidance is applicable to modifications and replacements made to short-duration and long-duration contracts (including those contracts defined as *investment contracts* per the FASB ASC glossary), according to FASB ASC 944-30-15-8.
- **22.49** The FASB ASC glossary defines an *internal replacement* as a modification in product benefits, features, rights, or coverages that occurs by a contract exchange, by amendment, endorsement, or rider to a contract; or by the election of a benefit, feature, right, or coverage within a contract. The FASB ASC glossary defines a *contract exchange* as the legal extinguishment of one contract and the issuance of another contract.
- **22.50** The AICPA also issued a series of Q&A sections on accounting and financial reporting issues related to the "Internal Replacement Transactions" subsections in FASB ASC 944-30 (see Q&A sections 6300.25–.34 [AICPA, *Technical Questions and Answers*]).

### **Investment Portfolios**

- **22.51** Insurance subsidiaries maintain investment portfolios usually composed of the same types of securities found in the portfolios of independent insurance companies. State regulations restrict the types of investments that insurance companies may make.
- **22.52** FASB ASC 320-10-15-2 states that the guidance in FASB ASC 320, *Investments—Debt and Equity Securities*, applies to all entities including cooperatives and mutual entities (such as credit unions and mutual insurance entities) and trusts that do not report substantially all of their securities at fair value. FASB ASC 320 establishes standards of financial accounting and reporting for both investments in equity securities that have readily determinable fair values and all investments in debt securities, including debt instruments that have been securitized.

# © Update 22-2 Accounting and Reporting: Recognition and Measurement of Financial Assets and Financial Liabilities

FASB ASU No. 2016-01, Financial Instruments—Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities, issued in January 2016, is effective for fiscal years, and interim periods within those fiscal years, of a public business entity beginning after December 15, 2017.

For all other entities (including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960 through FASB ASC 965 on plan accounting), FASB ASU No. 2016-01 is effective for fiscal years beginning after December 15, 2018, and interim periods within fiscal years beginning after December 15, 2019. Early application is permitted for all other entities as of the fiscal years beginning after December 15, 2017, including interim periods within those fiscal years.

All entities may adopt the presentation guidance in paragraphs 5–7 of FASB ASC 825-10-45 for financial statements of fiscal years or interim periods that have not yet been issued or that have not yet been made available for issuance. Furthermore, entities that are not public business entities may elect not to disclose the information about fair value of financial instruments required by the "General" subsection of FASB ASC 825-10-50 in financial statements of fiscal years or interim periods that have not yet been made available for issuance. Except as indicated previously in this paragraph, early application is not permitted.

FASB ASU No. 2016-01 addresses certain aspects of recognition, measurement, presentation, and disclosure of financial instruments and affects all entities that hold financial assets or owe financial liabilities. Among other provisions of the guidance, the amendments in this ASU supersede the guidance to classify equity securities with readily determinable fair values into different categories (that is, trading or available-for-sale) and require equity securities (including other ownership interests, such as partnerships, unincorporated joint ventures, and limited liability companies) to be measured at fair value with changes in the fair value recognized through net income (other than equity securities accounted for under the equity method of accounting or those that result in consolidation of an investee).

The amendments in this ASU also require an entity to present separately in other comprehensive income the portion of the total change in fair value of a liability resulting from a change in the instrument-specific credit risk when the entity has elected to measure the liability at fair value in accordance with the fair value option for financial instruments.

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2016-01, see appendix G, "Accounting for Financial Instruments," of this guide.

**22.53** FASB ASC 825, *Financial Instruments*, allows entities to choose, at specified election dates, to measure eligible items at fair value (the fair value option) with gains and losses recorded in earnings at each subsequent reporting date.

**22.54** FASB ASC 944-320-50-1 states that an entity should disclose the carrying amount of securities deposited by insurance subsidiaries with state regulatory authorities.

#### State Laws

**22.55** Insurance companies are regulated by state insurance laws, which require maintenance of accounting records and adoption of accounting practices. The insurance laws and regulations of the states require insurance companies domiciled in those states to comply with the guidance provided in the National Association of Insurance Commissioners *Accounting Practices and Procedures Manual* except as otherwise prescribed or permitted by state law. Some prescribed or permitted statutory accounting practices differ from GAAP. Accordingly, the financial statements of insurance subsidiaries prepared for submission to regulatory authorities must be adjusted to conform to GAAP

before they can be consolidated with the financial statements of the parent companies.

## Commissions

## © Update 22-3 Accounting and Reporting: Revenue From Contracts With Customers

FASB ASU No. 2014-09, issued in May 2014, is effective (as amended by FASB ASU No. 2015-14) for annual reporting periods of a public business entity, a not-for-profit entity that has issued, or is a conduit bond obligor for, securities that are traded, listed, or quoted on an exchange or an overthe-counter market, and an employee benefit plan that files or furnishes financial statements with or to the SEC beginning after December 15, 2017, including interim periods within that reporting period. Early application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

For all other entities, FASB ASU No. 2014-09 is effective for annual reporting periods beginning after December 15, 2018, and interim periods within annual periods beginning after December 15, 2019. Other entities may elect to adopt the standard earlier, however, only as either

- an annual reporting period beginning after December 15, 2016, including interim periods within that reporting period, or
- an annual reporting period beginning after December 15, 2016, and interim periods within annual periods beginning one year after the annual reporting period in which an entity first applies the "Pending Content" that links to FASB ASC 606-10-65-1.

FASB ASU No. 2014-09 provides a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605 as well as guidance within the 900 series of industry-specific topics. The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets unless those contracts are within the scope of other standards (for example, insurance or lease contracts).<sup>4</sup>

This edition of the guide has not been updated to reflect changes as a result of this ASU, however, this section will be updated in a future edition. Readers are encouraged to consult the full text of this ASU on FASB's website at www.fasb.org.

For more information on FASB ASU No. 2014-09, see appendix E of this guide.

**22.56** A finance company may receive commissions from an independent insurer for policies issued to finance customers, according to FASB ASC 942-605-05-2.

**22.57** According to FASB ASC 942-605-25-1, insurance commissions received from an independent insurer should be deferred and systematically amortized to income over the life of the related insurance contracts because

<sup>&</sup>lt;sup>4</sup> See footnote 3.

the insurance and lending activities are integral parts of the same transactions. The method of commission amortization should be consistent with the method of premium income recognition for that type of policy as set forth in FASB ASC 944.

**22.58** FASB ASC 605-20-25-7 states that income from experience-rated or retrospective commission arrangements should be recognized over the applicable insurance risk period.

**22.59** Commissions paid to the parent company by an insurance subsidiary are eliminated in consolidation.

## **Consolidation Policy**

**22.60** Consolidation is appropriate if a reporting entity has a controlling financial interest in another entity and a specific scope exception does not apply, as stated in FASB ASC 810-10-25-1. The usual condition for a controlling financial interest is ownership of a majority voting interest, but in some circumstances control does not rest with the majority owner. See FASB ASC 810-10-15 for entities that are subject to consolidation.

For legal entities other than limited partnerships, consolidation is appropriate if a reporting entity has a controlling financial interest in another entity and a specific scope exception does not apply, as stated in "Pending Content" in FASB ASC 810-10-25-1.<sup>5</sup> The usual condition for a controlling financial interest is ownership of a majority voting interest, but in some circumstances control does not rest with the majority owner. See FASB ASC 810-10-15 for entities that are subject to consolidation.

- **22.61** The "Variable Interest Entities" subsections of FASB ASC 810-10 clarify the application of the "General" subsections to certain legal entities in which equity investors do not have sufficient equity at risk for the legal entity to finance its activities without additional subordinated financial support, or, as a group, according to "Pending Content" in FASB ASC 810-10-05-8, the holders of the equity investment at risk lack any one of the following three characteristics:
  - a. The power, through voting rights or similar rights, to direct the activities of a legal entity that most significantly impact the entity's economic performance

FASB ASU No. 2015-02 affects reporting entities that are required to evaluate whether they should consolidate certain legal entities. All legal entities are subject to reevaluation under the revised consolidation model.

Readers are encouraged to read the full text of the ASU, available at www.fasb.org. Readers should apply the appropriate guidance based on their facts and circumstances.

<sup>&</sup>lt;sup>5</sup> FASB ASU No. 2015-02, Consolidation (Topic 810): Amendments to the Consolidation Analysis, was issued in February 2015. The amendments are effective for public business entities for fiscal years, and for interim periods within those fiscal years, beginning after December 15, 2015. For all other entities, the amendments in this ASU are effective for fiscal years beginning after December 15, 2016, and for interim periods within fiscal years beginning after December 15, 2017. Early adoption is permitted, including adoption in an interim period. If an entity early adopts the amendments in an interim period, any adjustments should be reflected as of the beginning of the fiscal year that includes that interim period. A reporting entity may apply the amendments in this ASU using a modified retrospective approach by recording a cumulative-effect adjustment to equity as of the beginning of the fiscal year of adoption. A reporting entity also may apply the amendments retrospectively.

- b. The obligation to absorb the expected losses of the legal entity
- c. The right to receive the expected residual returns of the legal entity

If the activities of the legal entity, according to "Pending Content" in item (d) in FASB ASC 810-10-15-17, are primarily related to securitizations or other forms of asset-backed financings or single-lessee leasing arrangements, then the legal entity should be evaluated by a reporting entity to determine if the legal entity is a variable interest entity (VIE) under the requirements of the "Variable Interest Entities" subsections of FASB ASC 810-10.6

## Considerations for Private Companies that Elect to use Standards as Issued by the Private Company Council

FASB ASU No. 2014-07 permits a private company lessee (the reporting entity) to elect an alternative not to apply VIE guidance to a lessor entity if the following criteria are met:

- The private company lessee and the lessor legal entity are under common control.
- The private company lessee has a lease arrangement with the lessor legal entity.
- Substantially all activities between the private company lessee and the lessor legal entity are related to leasing activities (including supporting leasing activities) between those two entities.
- If the private company lessee explicitly guarantees or provides collateral for any obligation of the lessor legal entity related to the asset leased by the private company, then the principal amount of the obligation at inception of such guarantee or collateral arrangement does not exceed the value of the asset leased by the private company from the lessor legal entity.

The accounting alternative is an accounting policy election that, when elected, should be applied by a private company lessee to all current and future lessor entities under common control that meet the criteria for applying this approach. If any of the criteria for applying the alternative cease to be met, a private company should apply the VIE guidance at the date of change on a prospective basis.

**22.62** Insurance subsidiaries of financial institutions may also participate in VIEs through investing in other structured investments, such as synthetic asset-backed securities and catastrophe bonds, certain structured reinsurance

<sup>&</sup>lt;sup>6</sup> Many paragraphs in FASB ASC 810-10 are labeled as "Pending Content" and relate to FASB ASU No. 2009-17, Consolidations (Topic 810): Improvements to Financial Reporting by Enterprises Involved with Variable Interest Entities. Although labeled as "Pending Content," this guidance is applicable for many entities except those that qualify for the deferral in FASB ASU No. 2010-10, Consolidation (Topic 810): Amendments for Certain Investment Funds. In February 2015, FASB issued ASU No. 2015-02. Among other provisions, the amendments in this ASU rescind the indefinite deferral in FASB ASU No. 2010-10 and provide a scope exception from FASB ASC 810, Consolidation, for reporting entities with interests in legal entities that are required to comply with or operate in accordance with requirements similar to those in Rule 2a-7 of the Investment Company Act of 1940 for registered money market funds. See footnote 5 for further details on the effective date of FASB ASU No. 2015-02. Readers should carefully review the transition guidance in FASB ASC 810-10-65-2 associated with "Pending Content" to determine the applicability.

transactions, joint ventures without substantive operations, financial guarantees, debt issuance vehicles, synthetic leases, collateralized bond obligation issuances, or limited partnerships.

**22.63** Per "Pending Content" in FASB ASC 810-10-15-17, separate accounts of life insurance entities as described in FASB ASC 944 are not subject to consolidation according to the requirements of the "Variable Interest Entities" subsections. FASB ASC 944-80 provides insurance entities guidance on accounting for and financial reporting of separate accounts, including an insurance entity's accounting for separate account assets and liabilities related to contracts for which all or a portion of the investment risk is borne by the insurer.

**22.64** Separate accounts of life insurance entities are described in more detail in the AICPA Audit and Accounting Guide *Life and Health Insurance Entities*.

## **Financial Statement Presentation**

**22.65** FASB ASC 942-210-45-1 states that, unearned premiums and unpaid claims on certain insurance coverage issued to finance customers by a subsidiary may represent intra-entity items because premiums are added to the consumer loan account, which is in turn classified as a receivable until paid, and most or all of the payments on claims are applied to reduce the related finance receivables. Therefore, unearned premiums and unpaid claims on certain credit life and credit accident and health insurance policies issued to finance customers should be deducted from finance receivables in the consolidated balance sheet.

**22.66** The following illustrates that type of presentation:

Finance receivables	XXX	
Less:		
Allowance for losses	(XXX)	
Unearned premiums and unpaid claim		
liabilities related to finance receivables	(XXX)	
Finance receivables, net	XXX	

**22.67** Alternatively, the balance sheet may present only the net finance receivables if the notes to the financial statements contain sufficient disclosure of unearned premiums and unpaid claims and the allowance for losses. Unearned premiums and unpaid claims for credit life and accident and health coverage should not be applied in consolidation against related finance receivables for which the related receivables are assets of unrelated entities as stated in FASB ASC 942-210-45-1.

**22.68** FASB ASC 942-210-45-2 states that, in the consolidated financial statements, unpaid claims for property insurance and level term life insurance should not be offset against related finance receivables because finance companies generally do not receive substantially all proceeds of such claims. That

<sup>&</sup>lt;sup>7</sup> See footnote 6.

prohibition also applies to credit life and accident and health coverage written on policies for which the related receivables are assets of unrelated entities. In those circumstances, such amounts should be presented as liabilities.

## **Auditing**

**22.69** The AICPA Audit and Accounting Guides *Life and Health Insurance Entities* and *Property and Liability Insurance Entities* provide guidance on auditing concepts and procedures for insurance companies. In addition, the auditor should consider whether accounts between the finance company and the insurance subsidiaries are reconciled regularly.

**22.70** Based on the significance of the premiums and commissions associated with insurance provided to finance customers the auditor might consider audit procedures that include insurance activities. Similarly, branch office controls over loans usually apply to insurance products. The auditor should be satisfied that the income recognition methods for insurance premiums and commission income conform to the principles discussed in this chapter.

©2016, AICPA AAG-DEP 22.70

## Chapter 23

## Reporting Considerations

Gray shaded text in this chapter reflects guidance issued but not yet effective as of the date of this guide, July 1, 2016, but becoming effective on or prior to December 31, 2016, exclusive of any option to early adopt ahead of the mandatory effective date. Unless otherwise indicated, all unshaded text reflects guidance that was already effective as of the date of this guide.

The auditing content in this guide focuses primarily on generally accepted auditing standards (GAAS) issued by the Auditing Standards Board (ASB) and is applicable to audits of nonissuers. These standards are codified in AICPA *Professional Standards* and referenced by AU-C section numbers within the codification. Issuers are defined by Section 3 of the Securities Exchange Act of 1934. Audits of issuers are required to be performed under PCAOB standards. Readers of this guide should evaluate their audit engagements to determine which auditing standards are applicable.

In April 2003, the PCAOB adopted as interim standards certain preexisting auditing and related standards via PCAOB Rules 3200T, Interim Auditing Standards; 3300T, Interim Attestation Standards; 3400T, Interim Quality Control Standards; 3500T, Interim Ethics Standards; and 3600T, Interim Independence Standards (AICPA, PCAOB Standards and Related Rules). Although the PCAOB has subsequently issued standards that supersede or amend those interim standards, the reporting guidance that was adopted by the PCAOB remains largely consistent with the reporting guidance applicable to audits conducted in accordance with GAAS as they existed prior to adoption of the clarified Statements on Auditing Standards. As a consequence, significant differences will exist between audit reports for audits conducted in accordance with GAAS as compared to those conducted in accordance with PCAOB standards. In 2010, the PCAOB initiated its own project to consider changes to the auditor's reporting model. This chapter addresses only reporting guidance applicable to audits conducted in accordance with GAAS and does not include reporting guidance applicable to audits conducted in accordance with PCAOB standards. Appendix A, "Illustrative Unqualified PCAOB Reports," of this chapter provides illustrations of unqualified audit reports prepared in accordance with PCAOB standards.

Considerations for audits of issuers in accordance with PCAOB standards may be discussed within this guide's chapter text. When such discussion is provided, the related paragraphs are designated with the following title: Considerations for Audits Performed in Accordance With PCAOB Standards.

(continued)

©2016, AICPA AAG-DEP 23

<sup>&</sup>lt;sup>1</sup> Differences in reporting guidance prior to the Clarity Project were the result of the PCAOB's issuance of Auditing Standard No. 1, References in Auditors' Reports to the Standards of the Public Company Accounting Oversight Board, and Auditing Standard No. 5, An Audit of Internal Control Over Financial Reporting That Is Integrated with An Audit of Financial Statements (AICPA, PCAOB Standards and Related Rules).

## Introduction

**23.01** This chapter applies the reporting guidance found in AU-C sections 700, Forming an Opinion and Reporting on Financial Statements; 705, Modifications to the Opinion in the Independent Auditor's Report; and 706, Emphasis-of-Matter Paragraphs and Other-Matter Paragraphs in the Independent Auditor's Report (AICPA, Professional Standards), to audit reports on the financial statements of depository and lending institutions. Such reports may contain an unmodified opinion, an unmodified opinion with emphasis-of-matter (EOM) or other-matter paragraphs, a qualified opinion, an adverse opinion, or a disclaimer of opinion. This chapter contains a brief discussion of each of those reports, with an emphasis on illustrating issues that an auditor may encounter in the industry. The reports are illustrative; the facts and circumstances of each particular audit will govern the appropriate form of report. Paragraphs 23.22–24 and 23.28–34 apply only to credit unions.

## Forming an Opinion on the Financial Statements

**23.02** Paragraphs .13–.18 of AU-C section 700 address the auditor's responsibility to form an opinion on the financial statements. The auditor should form an opinion on whether the financial statements are presented fairly, in all material respects, in accordance with the applicable financial reporting framework. In order to form that opinion, the auditor should conclude whether the auditor has obtained reasonable assurance about whether the financial statements as a whole are free from material misstatement, whether due to fraud or error. That conclusion should take into account the following:

- a. The auditor's conclusion, in accordance with AU-C section 330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained (AICPA, Professional Standards), about whether sufficient appropriate audit evidence has been obtained
- b. The auditor's conclusion, in accordance with AU-C section 450, Evaluation of Misstatements Identified During the Audit (AICPA, Professional Standards), about whether uncorrected misstatements are material, individually or in aggregate
- c. The evaluations required by paragraphs .15–.18 of AU-C section 700 (as discussed in paragraphs 23.03–.04)

**23.03** The auditor should evaluate whether the financial statements are prepared, in all material respects, in accordance with the requirements of the applicable financial reporting framework. This evaluation should include consideration of the qualitative aspects of the entity's accounting practices, including indicators of possible bias in management's judgments. In particular, the

(footnote continued)

The PCAOB adopted, and SEC approved, amendments that reorganize its auditing standards using a topical structure and a single, integrated numbering system. The reorganization applies a standardized organizational structure to the PCAOB's auditing standards intended to help users navigate the standards more easily. Among other amendments, Auditing Standard No. 1 will be superseded as a result of amendments made to the other standards and Auditing Standard No. 5 will become AS 2201. The reorganization and related amendments will be effective as of December 31, 2016. Nothing precludes auditors and others from using and referencing the reorganized standards prior to the effective date. With the issuance of the reorganized standards, any references to PCAOB standards within this guide will be to the reorganized PCAOB standards.

auditor should evaluate whether, in view of the requirements of the applicable financial reporting framework

- a. the financial statements adequately disclose the significant accounting policies selected and applied;
- b. the accounting policies selected and applied are consistent with the applicable financial reporting framework and are appropriate;
- c. the accounting estimates made by management are reasonable;
- d. the information presented in the financial statements is relevant, reliable, comparable, and understandable;
- e. the financial statements provide adequate disclosures to enable
  the intended users to understand the effect of material transactions
  and events on the information conveyed in the financial statements;
  and
- f. the terminology used in the financial statements, including the title of each financial statement, is appropriate.

**23.04** The auditor's evaluation about whether the financial statements achieve fair presentation should also include consideration of the overall presentation, structure, and content of the financial statements and whether the financial statements, including the related notes, represent the underlying transactions and events in a manner that achieves fair presentation. Finally, the auditor should evaluate whether the financial statements adequately refer to or describe the applicable financial reporting framework.

## Reports

## **Unmodified Opinion**

**23.05** The auditor should express an unmodified opinion when the auditor concludes that the financial statements are presented fairly, in all material respects, in accordance with the applicable financial reporting framework. The following is an illustration of an auditor's report (unmodified opinion) on the comparative financial statements of a bank or savings institution prepared in accordance with U.S. generally accepted accounting principles (GAAP).

## **Independent Auditor's Report**

To the [Institution, Board of Directors, or Stockholders]:

#### Report on the Financial Statements<sup>2</sup>

We have audited the accompanying financial statements of ABC Institution, which comprise the balance sheets as of December 31, 20X1 and 20X0, and the related statements of income and comprehensive income, changes in stockholders' equity, and cash flows for the years then ended, and the related notes to the financial statements.<sup>3</sup>

©2016. AICPA

 $<sup>^2</sup>$  The subtitle "Report on the Financial Statements" is unnecessary in circumstances when the second subtitle, "Report on Other Legal and Regulatory Requirements," is not applicable.

<sup>&</sup>lt;sup>3</sup> These references to the financial statements should conform to the titles and periods of the statements being reported upon. For example, pursuant to FASB Accounting Standards Update (ASU) No. 2011-05, Comprehensive Income (Topic 220): Presentation of Comprehensive Income, institutions may present the components of net income and other comprehensive income either in a single continuous statement or in two separate but consecutive statements. The statements identified in the first paragraph of the audit would depend on an institution's selected presentation. In addition, there may be circumstances (such as for Emerging Growth or Smaller Reporting Companies, a change in auditor, or a newly organized registrant, for example) where less than two years of a balance sheet and three years of the other financial statements are being reported upon.

#### Management's Responsibility for the Financial Statements

Management is responsible for the preparation and fair presentation of these financial statements in accordance with accounting principles generally accepted in the United States of America; this includes the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

## Auditor's Responsibility

Our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement.

An audit involves performing procedures to obtain audit evidence about the amounts and disclosures in the financial statements. The procedures selected depend on the auditor's judgment, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error. In making those risk assessments, the auditor considers internal control relevant to the entity's preparation and fair presentation of the financial statements in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the entity's internal control.<sup>4</sup> Accordingly, we express no such opinion. An audit also includes evaluating the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluating the overall presentation of the financial statements.

We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

#### **Opinion**

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of ABC Institution as of December 31, 20X1 and 20X0, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

## Report on Other Legal and Regulatory Requirements

[Form and content of this section of the auditor's report will vary depending on the nature of the auditor's other reporting responsibilities.]

[Auditor's signature]
[Auditor's city and state]
[Date of the auditor's report]

<sup>&</sup>lt;sup>4</sup> In circumstances when the auditor also has responsibility to express an opinion on the effectiveness of internal control in conjunction with the audit of the financial statements, this sentence would be worded as follows: "In making those risk assessments, the auditor considers internal control relevant to the entity's preparation and fair presentation of the financial statements in order to design audit procedures that are appropriate in the circumstances." In addition, the next sentence, "Accordingly, we express no such opinion," would not be included.

23.06 In accordance with paragraphs .37-.38 of AU-C section 700, if the auditor addresses other reporting responsibilities in the auditor's report on the financial statements that are in addition to the auditor's responsibility under GAAS to report on the financial statements, these other reporting responsibilities should be addressed in a separate section in the auditor's report that follows the section titled "Report on the Financial Statements" and that should be subtitled "Report on Other Legal and Regulatory Requirements" or otherwise, as appropriate to the content of the section. Paragraphs .A32-.A34 of AU-C section 700 indicate that, for example, for audits conducted under Government Auditing Standards, the auditor may be required to report on internal control over financial reporting and compliance with laws, regulations, and provisions of contracts or grant agreements. However, reporting requirements in AU-C section 935, Compliance Audits (AICPA, Professional Standards), apply when the auditor is engaged or required by law or regulation to perform a compliance audit in accordance with GAAS, Government Auditing Standards, and a governmental audit requirement. In some cases, the relevant law or regulation may require or permit the auditor to report on these other responsibilities within the auditor's report on the financial statements. In other cases, the auditor may be required or permitted to report on them in a separate report. As previously discussed in paragraph 4.33 of this guide, financial institutions may participate in the U.S. Department of Housing and Urban Development (HUD) programs and must comply with the requirements of the Consolidated Audit Guide for Audits of HUD Programs (HUD Audit Guide), which includes reporting on internal control over financial reporting and compliance with laws, regulations, and provisions of contracts or grant agreements. Chapter 2 of the HUD Audit Guide provides illustrative reporting examples for audits conducted under Government Auditing Standards for HUD programs.

# EOM and Other-Matter Paragraphs Added to the Independent Auditor's Report

## **EOM Paragraphs**

**23.07** If the auditor considers it necessary to draw users' attention to a matter appropriately presented or disclosed in the financial statements that, in the auditor's professional judgment, is of such importance that it is fundamental to users' understanding of the financial statements, paragraphs .06–.07 of AU-C section 706 state that the auditor should include an EOM paragraph in the auditor's report, provided that the auditor has obtained sufficient appropriate audit evidence that the matter is not materially misstated in the financial statements. Such a paragraph should refer only to information presented or disclosed in the financial statements. When the auditor includes an EOM paragraph in the auditor's report, the auditor should

- a. include it immediately after the opinion paragraph in the auditor's report,
- use the heading "Emphasis of Matter" or other appropriate heading,
- c. include in the paragraph a clear reference to the matter being emphasized and to where relevant disclosures that fully describe the matter can be found in the financial statements, and
- d. indicate that the auditor's opinion is not modified with respect to the matter emphasized.

23.08 Exhibit B, "List of AU-C Sections Containing Requirements for Emphasis-of-Matter Paragraphs," of AU-C section 706 identifies other AU-C sections that require the auditor to include an EOM paragraph in the auditor's report in certain circumstances. One such circumstance is when the auditor concludes that there is substantial doubt about an institution's ability to continue as a going concern for a reasonable period of time.<sup>5</sup> AU-C section 570, The Auditor's Consideration of an Entity's Ability to Continue as a Going Concern (AICPA, Professional Standards), addresses the auditor's responsibilities in an audit of financial statements with respect to evaluating whether there is substantial doubt about the ability of the entity to continue as a going concern for a reasonable period of time, which is defined as a period of time not to exceed one year beyond the date of the financial statements being audited.<sup>6</sup> Chapter 17, "Equity and Disclosures Regarding Capital Matters," of this guide describes going-concern considerations as they relate to banks and savings institutions and discusses how an institution's regulatory capital position should be considered in the auditor's assessment of whether there is substantial doubt about the institution's ability to continue as a going concern. If, after considering management's plans that are intended to mitigate the adverse effects of conditions or events that indicate substantial doubt could exist, the auditor concludes that substantial doubt about an entity's ability to continue as a going concern for a reasonable period of time remains, paragraph .15 of AU-C section 570 states that the auditor should include an EOM paragraph in the auditor's report to reflect that conclusion. Paragraph .16 of AU-C section 570 further states that the auditor's conclusion about the entity's ability to continue as a going concern should be expressed through the use of the phrase "substantial doubt about its (the entity's) ability to continue as a going concern" or similar wording that includes the terms substantial doubt and going concern. In a going concern EOM paragraph, the auditor should not use conditional language in expressing a conclusion concerning the existence of substantial doubt about the entity's ability to continue as a going concern. If the auditor concludes that

<sup>&</sup>lt;sup>5</sup> FASB Accounting Standards Codification 205-40 provides guidance for evaluating whether there is substantial doubt about an entity's ability to continue as a going concern and about related footnote disclosures.

FASB ASU No. 2014-15, Presentation of Financial Statements—Going Concern (Subtopic 205-40): Disclosure of Uncertainties about an Entity's Ability to Continue as a Going Concern, issued in August 2014, requires management to assess an entity's ability to continue as a going concern by incorporating and expanding upon certain principles that are currently in U.S. auditing standards. Specifically, the amendments (1) provide a definition of the term substantial doubt, (2) require an evaluation every reporting period including interim periods, (3) provide principles for considering the mitigating effect of management's plans, (4) require certain disclosures when substantial doubt is alleviated as a result of consideration of management's plans, (5) require an express statement and other disclosures when substantial doubt is not alleviated, and (6) require an assessment for a period of one year after the date that the financial statements are issued (or available to be issued).

FASB ASU No. 2014-15 is effective for the annual period ending after December 15, 2016, and for annual periods and interim periods thereafter. Early application is permitted.

<sup>&</sup>lt;sup>6</sup> When the applicable financial reporting framework includes a definition of substantial doubt about an entity's ability to continue as a going concern, Interpretation No. 1, "Definition of Substantial Doubt About an Entity's Ability to Continue as a Going Concern," of AU-C section 570, The Auditor's Consideration of an Entity's Ability to Continue as a Going Concern (AICPA, Professional Standards, AU-C sec. 9570 par. .01–.02), requires that definition to be used by the auditor when applying the requirements of AU-C section 570. Interpretation No. 2, "Definition of Reasonable Period of Time," of AU-C section 570 (AICPA, Professional Standards, AU-C sec. 9570 par. .03–.05), provides guidance on how an auditor should apply the term reasonable period of time when the applicable financial reporting framework requires management to evaluate whether there are conditions and events that raise substantial doubt for a period of time greater than one year from the date of the financial statements. Specifically, Interpretation No. 2 states that the auditor's assessment of management's going concern evaluation would be for the same period of time as required by the applicable financial reporting framework.

the entity's disclosures with respect to the entity's ability to continue as a going concern for a reasonable period of time are inadequate, paragraph .17 of AU-C section 570 states that the auditor should modify the opinion in accordance with AU-C section 705. The following is an illustration of an auditor's report on the financial statements of a bank or savings institution that includes an EOM paragraph because of the existence of substantial doubt about the institution's ability to continue as a going concern for a reasonable period of time.

## **Independent Auditor's Report**

To the [Institution, Board of Directors, or Stockholders]:

## Report on the Financial Statements<sup>7</sup>

[Same first, second, third, fourth, fifth, and sixth paragraphs as the unmodified report. See illustrative example at paragraph 23.05.]

## **Emphasis of Matter Going Concern**

The accompanying financial statements have been prepared assuming that ABC Institution will continue as a going concern. As discussed in Note XX to the financial statements, at December 31, 20X1, the Institution did not meet its minimum capital requirements established by the Office of the Comptroller of the Currency (OCC). The Institution also has suffered recurring losses from operations. The Institution has filed a capital restoration plan with the OCC outlining its plans for attaining the required levels of regulatory capital by December 31, 20X2. To date, the Institution has not received notification from the OCC regarding acceptance or rejection of its capital restoration plan. Failure to meet the capital requirements and interim capital targets included in the capital restoration plan would expose the Institution to regulatory sanctions that may include restrictions on operations and growth, mandatory asset dispositions, and seizure. These matters raise substantial doubt about the ability of the Institution to continue as a going concern. The ability of the Institution to continue as a going concern is dependent on many factors, one of which is regulatory action, including ultimate acceptance of its capital restoration plan. Management's plans in regard to these matters are described in Note XX. The financial statements do not include any adjustments that might result from the outcome of this uncertainty. Our opinion is not modified with respect to this matter.

#### Report on Other Legal and Regulatory Requirements

[Form and content of this section of the auditor's report will vary depending on the nature of the auditor's other reporting responsibilities.]

[Auditor's signature]

[Auditor's city and state]

[Date of the auditor's report]

**23.09** Paragraph .A6 of AU-C section 570 states that the inclusion of a going-concern EOM paragraph in the auditor's report (as described previously) is sufficient to inform users of the financial statements that substantial doubt exists about the entity's ability to continue as a going concern for a reasonable period of time. Nevertheless, AU-C section 570 does not preclude an auditor from disclaiming an opinion in cases involving uncertainties. When an auditor

<sup>&</sup>lt;sup>7</sup> See footnote 2.

disclaims an opinion, paragraph .18 of AU-C section 570 states that the report should not include the going-concern EOM paragraph but, rather describe the substantive reasons for the auditor's disclaimer of opinion as required by paragraph .17 of AU-C section 705. The auditor should consider the adequacy of disclosure of the uncertainties and their possible effects on the financial statements as described in paragraph .12 of AU-C section 570, even when disclaiming an opinion. The following is an illustration of an auditor's disclaimer of opinion because of the existence of substantial doubt about an institution's ability to continue as a going concern for a reasonable period of time.

## **Independent Auditor's Report**

To the [Institution, Board of Directors, or Stockholders]:

## Report on Financial Statements<sup>8</sup>

We have audited the accompanying financial statements of ABC Institution, which comprise the balance sheet as of December 31, 20X0, and the related statements of income and comprehensive income, changes in stockholders' equity, and cash flows for the years then ended, and the related notes to the financial statements. Further, we were engaged to audit the accompanying financial statements of ABC Institution, which comprise the balance sheet as of December 31, 20X1, and the related statements of income, comprehensive income, changes in stockholders' equity, and cash flows for the year then ended, and the related notes to the financial statements.

## Management's Responsibility for the Financial Statements

Management is responsible for the preparation and fair presentation of these financial statements in accordance with accounting principles generally accepted in the United States of America; this includes the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

#### Auditor's Responsibility

Our responsibility is to express an opinion on these financial statements based on our audits. Except as explained in the Basis for Disclaimer of Opinion paragraph, we conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement. <sup>10</sup>

An audit involves performing procedures to obtain audit evidence about the amounts and disclosures in the financial statements. The procedures selected depend on the auditor's judgment, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error. In making those risk assessments, the auditor considers internal control relevant to the entity's

<sup>&</sup>lt;sup>8</sup> See footnote 2.

<sup>9</sup> See footnote 3.

<sup>&</sup>lt;sup>10</sup> If the auditor was disclaiming an opinion due to a scope limitation, this paragraph would state the following: "Our responsibility is to express an opinion on these financial statements based on conducting the audit in accordance with auditing standards generally accepted in the United States of America. Because of the matters described in the Basis for Disclaimer of Opinion paragraph, however, we were not able to obtain sufficient appropriate audit evidence to provide a basis for an audit opinion." (See paragraph 23.21 on "Disclaimer of Opinion.")

preparation and fair presentation of the financial statements in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the entity's internal control. Accordingly, we express no such opinion. An audit also includes evaluating the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluating the overall presentation of the financial statements. 12

We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinions on the balance sheet as of December 31, 20X0, and the statements of income, comprehensive income, changes in stockholders equity, and cash flows for the year then ended.<sup>13</sup>

## Basis for Disclaimer of Opinion on 20X1 Financial Statements

The accompanying financial statements have been prepared assuming that the Institution will continue as a going concern. As discussed in Note X to the financial statements, at December 31, 20X1, the Institution did not meet its minimum capital requirements established by the Office of the Comptroller of the Currency (OCC). The Institution also has suffered recurring losses from operations. The Institution has filed a capital restoration plan with the OCC outlining its plans for attaining the required levels of regulatory capital by December 31, 20X2. To date, the Institution has not received notification from the OCC regarding acceptance or rejection of its capital restoration plan. Failure to meet the capital requirements and interim capital targets included in the capital restoration plan would expose the Institution to regulatory sanctions that may include restrictions on operations and growth, mandatory asset dispositions, and seizure. These matters raise substantial doubt about the ability of the Institution to continue as a going concern. The ability of the Institution to continue as a going concern is dependent on many factors, one of which is regulatory action, including ultimate acceptance of its capital restoration plan. Management's plans in regard to these matters are described in Note XX. The financial statements do not include any adjustments that might result from the outcome of this uncertainty.

## Disclaimer of Opinion on 20X1 Financial Statements

Because of the significance of the matter described in the Basis for Disclaimer of Opinion paragraph, we have not been able to obtain sufficient appropriate audit evidence to provide a basis for an audit opinion on the financial statements for the year ended December 31, 20X1.

## Opinion on 20X0 Financial Statements

In our opinion, the 20X0 financial statements referred to above present fairly, in all material respects, the financial position of ABC Institution as of December 31, 20X0, and the results of its operations and its cash flows for the year then ended in accordance with accounting principles generally accepted in the United States of America.

<sup>&</sup>lt;sup>11</sup> See footnote 4.

<sup>&</sup>lt;sup>12</sup> If the auditor was disclaiming an opinion due to a scope limitation, this paragraph would be omitted. (See paragraph 23.21 on "Disclaimer of Opinion.")

<sup>&</sup>lt;sup>13</sup> See footnote 12.

## Report on Other Legal and Regulatory Requirements

[Form and content of this section of the auditor's report will vary depending on the nature of the auditor's other reporting responsibilities.]

[Auditor's signature]

 $[Auditor's\ city\ and\ state]$ 

[Date of the auditor's report]

23.10 AICPA Technical Questions and Answers (Q&A) section 9160.28, "Combining a Going Concern Emphasis-of-Matter Paragraph With Another Emphasis-of-Matter Paragraph" (AICPA, Technical Questions and Answers), addresses whether an auditor may combine a going concern EOM paragraph with another EOM paragraph. Q&A section 9160.28 states that AU-C section 706 does not preclude the auditor from combining matters in an EOM paragraph that are either required to be included in the auditor's report or included at the auditor's discretion, recognizing that the requirements in paragraph .07 of AU-C section 706 should be followed for each matter identified. Whether the auditor decides to combine more than one matter in the same paragraph is a matter of professional judgment. However, it may be prudent to emphasize each matter separately in separate paragraphs to make it clear that you have more than one matter of emphasis in your report.

**23.11** In addition to the required EOM paragraphs listed in exhibit B of AU-C section 706 (referenced in paragraph 23.08), paragraph .A2 of AU-C section 706 provides the following examples of circumstances when the auditor may consider it necessary to include an EOM paragraph:

- An uncertainty relating to the future outcome of unusually important litigation or regulatory action
- A major catastrophe that has had, or continues to have, a significant effect on the entity's financial position
- Significant transactions with related parties
- Unusually important subsequent event

The following is an illustration of an unmodified opinion with an EOM paragraph regarding an institution's failure to meet minimum regulatory capital standards on the institution's financial statements (note that in this illustration, this EOM is not a going concern matter).

## **Independent Auditor's Report**

To the [Institution, Board of Directors, or Stockholders]:

## Report on the Financial Statements<sup>14</sup>

[Same first, second, third, fourth, fifth, and sixth paragraphs as the unmodified report. See illustrative example at paragraph 23.05.]

## **Emphasis of Matter**

As discussed in Note XX to the financial statements, at December 31, 20X1, the Institution failed to meet the risk-based capital requirement established by the FDIC. The Institution has filed, and the FDIC has accepted, a capital restoration plan for attaining the required level of regulatory risk-based capital by December 31, 20X2. Our opinion is not modified with respect to this matter.

<sup>&</sup>lt;sup>14</sup> See footnote 2.

## Report on Other Legal and Regulatory Requirements

[Form and content of this section of the auditor's report will vary depending on the nature of the auditor's other reporting responsibilities.]

[Auditor's signature]

[Auditor's city and state]

[Date of the auditor's report]

- 23.12 Q&A section 9160.29, "Modification to the Auditor's Report When a Client Adopts a PCC Accounting Alternative" (AICPA, Technical Questions and Answers), addresses the need for an EOM paragraph in the auditor's report for a change in accounting principle as a result of an entity adopting a Private Company Council (PCC) accounting alternative. Q&A section 9160.29 states that according to paragraphs .07-.12 of AU-C section 708, Consistency of Financial Statements (AICPA, Professional Standards), if an entity has adopted a change in accounting principle and that change has a material effect on the financial statements, the auditor should include in the auditor's report an EOM paragraph in the period of the change and in subsequent periods until the new accounting principle is applied in all periods presented. If the change in accounting principle is accounted for by retrospective application to the financial statements of all prior periods presented, the EOM paragraph is needed only in the period of such change. The auditor's EOM paragraph should describe the change in accounting principle and provide a reference to the entity's note disclosure.
- 23.13 Q&A section 9160.30, "Modification to the Auditor's Report When a Client Adopts a PCC Accounting Alternative that Results in a Change to a Previously Issued Report" (AICPA, *Technical Questions and Answers*), addresses the need to include an EOM paragraph in the auditor's report when an entity adopts a PCC alternative that results in a change to a previously issued report, specifically if the prior year report included a departure from GAAP and now that the PCC alternative had been adopted, the prior year no longer contains the departure from GAAP. According to paragraph .54 of AU-C section 700, when reporting on prior period financial statements in connection with an audit of the current period, the auditor should disclose the following matters in an EOM paragraph in accordance with AU-C section 706 if the auditor's opinion on such prior period financial statements differs from the opinion the auditor previously expressed:
  - a. The date of the auditor's previous report.
  - b. The type of opinion previously expressed.
  - c. The substantive reason for the different opinion.
  - d. The auditor's opinion on the amended financial statements is different from the auditor's previous opinion.

## Other-Matter Paragraphs

23.14 If the auditor considers it necessary to communicate a matter other than those that are presented or disclosed in the financial statements that, in the auditor's professional judgment, is relevant to users' understanding of the audit, the auditor's responsibilities, or the auditor's report, paragraph .08 of AU-C section 706 states that the auditor should do so in a paragraph in the auditor's report with the heading "Other Matter" or other appropriate heading. The auditor should include this paragraph immediately after the opinion paragraph and any EOM paragraph or elsewhere in the auditor's report if

the content of the other-matter paragraph is relevant to the "Other Reporting Responsibilities" section. Exhibit C, "Lists of AU-C Sections Containing Requirements for Other-Matter Paragraphs," of AU-C section 706 identifies other AU-C sections that require the auditor to include an other-matter paragraph in the auditor's report in certain circumstances. The following is an illustration of an unmodified opinion that contains an other-matter paragraph to draw user attention to an updated auditor's report on the financial statements of a prior period that contains an opinion different from the opinion previously expressed.

#### **Independent Auditor's Report**

To the [Institution, Board of Directors, or Stockholders]:

## Report on the Financial Statements<sup>15</sup>

[Same first, second, third, fourth, fifth, and sixth paragraphs as the unmodified report. See illustrative example at paragraph 23.05.]

#### Other Matter

In our report dated March 1, 20X1, we expressed an opinion that the 20X0 financial statements did not fairly present the financial position, results of operations, and cash flows of ABC Institution in accordance with accounting principles generally accepted in the United States of America because of two departures from such principles: (1) ABC Institution carried its property, plant, and equipment at appraisal values, and provided for depreciation on the basis of such values, and (2) ABC Institution did not provide for deferred income taxes with respect to differences between income for financial reporting purposes and taxable income. As described in Note X, the Company has changed its method of accounting for these items and restated its 20X0 financial statements to conform with accounting principles generally accepted in the United States of America. Accordingly, our present opinion on the restated 20X0 financial statements, as presented herein, is different from that expressed in our previous report.

#### Report on Other Legal and Regulatory Requirements

[Form and content of this section of the auditor's report will vary depending on the nature of the auditor's other reporting responsibilities.]

[Auditor's signature]

[Auditor's city and state]

[Date of the auditor's report]

## **Modified Opinions**

**23.15** AU-C section 705 addresses the auditor's responsibility to issue an appropriate report in circumstances when, in forming an opinion in accordance with AU-C section 700, the auditor concludes that a modification to the auditor's opinion on the financial statements is necessary. Paragraph .07 of AU-C section 705 states that the auditor should modify the opinion in the auditor's report when

a. the auditor concludes that, based on the audit evidence obtained, the financial statements as a whole are materially misstated or

<sup>15</sup> See footnote 2.

- b. the auditor is unable to obtain sufficient appropriate audit evidence to conclude that the financial statements as a whole are free from material misstatement.
- **23.16** AU-C section 705 establishes three types of modified opinions: a qualified opinion, an adverse opinion, and a disclaimer of opinion. The decision regarding which type of modified opinion is appropriate depends upon the following:
  - a. The nature of the matter giving rise to the modification (that is, whether the financial statements are materially misstated or, in the case of an inability to obtain sufficient appropriate audit evidence, may be materially misstated)
  - b. The auditor's professional judgment about the pervasiveness of the effects or possible effects of the matter on the financial statements

AU-C section 705 also provides guidance on the circumstances when a modification to the auditor's opinion is required as well as guidance on determining the type of modification to the auditor's opinion (for example, guidance illustrating how the auditor's professional judgment about the nature of the matter giving rise to the modification and the pervasiveness of its effects or possible effects on the financial statement opinion).

23.17 When the auditor modifies the opinion on the financial statements, paragraph .17 of AU-C section 705 states that the auditor should, in addition to the specific elements required by AU-C section 700, include a paragraph in the auditor's report that provides a description of the matter giving rise to the modification. The auditor should place this paragraph immediately before the opinion paragraph in the auditor's report and use a heading that includes "Basis for Qualified Opinion," "Basis for Adverse Opinion," or "Basis for Disclaimer of Opinion," as appropriate. Paragraphs .18–.22 of AU-C section 705 provide further discussion on information that should be included within the basis for modification paragraph. In addition, paragraph .23 of AU-C section 705 states that the auditor should use a heading that includes "Qualified Opinion," "Adverse Opinion," or "Disclaimer of Opinion," as appropriate, for the opinion paragraph.

## **Qualified Opinion**

- **23.18** In accordance with paragraph .08 of AU-C section 705, the auditor should express a qualified opinion when
  - a. the auditor, having obtained sufficient appropriate audit evidence, concludes that misstatements, individually or in the aggregate, are material but not pervasive to the financial statements, or
  - b. the auditor is unable to obtain sufficient appropriate audit evidence on which to base the opinion, but the auditor concludes that the possible effects on the financial statements of undetected misstatements, if any, could be material but not pervasive.
- **23.19** For purposes of GAAS, AU-C section 705 defines *pervasive* as a term used in the context of misstatements to describe the effects on the financial statements of misstatements or the possible effects on the financial statements of misstatements, if any, that are undetected due to an inability to obtain sufficient appropriate audit evidence. Pervasive effects on the financial statements are those that, in the auditor's professional judgment

- are not confined to specific elements, accounts, or items of the financial statements;
- if so confined, represent or could represent a substantial proportion of the financial statements; or
- with regard to disclosures, are fundamental to users' understanding of the financial statements.

Paragraph 23.24 describes a circumstance in which a qualified opinion is appropriate and provides an illustrative qualified opinion.

## **Adverse Opinion**

23.20 In accordance with paragraph .09 of AU-C section 705, the auditor should express an adverse opinion when the auditor, having obtained sufficient appropriate audit evidence, concludes that misstatements, individually or in the aggregate, are both material and pervasive to the financial statements. When the auditor expresses an adverse opinion, paragraph .25 of AU-C section 705 states that the auditor should state in the opinion paragraph that, in the auditor's opinion, because of the significance of the matter(s) described in the basis for adverse opinion paragraph, the financial statements are not presented fairly in accordance with the applicable financial reporting framework. In accordance with paragraph .27 of AU-C section 705, the auditor should also amend the description of the auditor's responsibility to state that the auditor believes that the audit evidence the auditor has obtained is sufficient and appropriate to provide a basis for the auditor's modified audit opinion.

## **Disclaimer of Opinion**

**23.21** In accordance with paragraph .10 of AU-C section 705, the auditor should disclaim an opinion when the auditor is unable to obtain sufficient appropriate audit evidence on which to base the opinion, and the auditor concludes that the possible effects on the financial statements of undetected misstatements, if any, could be both material and pervasive. When the auditor disclaims an opinion due to an inability to obtain sufficient appropriate audit evidence (also referred to as a limitation on the scope of the audit), paragraph .26 of AU-C section 705 states that the auditor should state in the opinion paragraph that

- a. because of the significance of the matter(s) described in the basis for disclaimer of opinion paragraph, the auditor has not been able to obtain sufficient appropriate audit evidence to provide a basis for an audit opinion and
- b. accordingly, the auditor does not express an opinion on the financial statements.

When the auditor disclaims an opinion due to an inability to obtain sufficient appropriate audit evidence, paragraph .28 of AU-C section 705 states that the auditor should amend the introductory paragraph of the auditor's report to state that the auditor was engaged to audit the financial statements. The auditor should also amend the description of the auditor's responsibility and the description of the scope of the audit to state only the following:

Our responsibility is to express an opinion on the financial statements based on conducting the audit in accordance with auditing standards generally accepted in the United States of America. Because of the matter(s) described in the Basis for Disclaimer of Opinion paragraph,

however, we have not been able to obtain sufficient appropriate audit evidence to provide a basis for an audit opinion.

# Financial Statements Prepared in Accordance With a Special Purpose Framework

23.22 Title II of the Credit Union Membership Access Act of 1998, requires all federally insured credit unions with assets of \$10 million or more to follow GAAP. Credit unions with assets under \$10 million may use a basis of accounting other than GAAP; the National Credit Union Administration (NCUA) provides the Accounting Manual for Federal Credit Unions as a guide in accounting for financial transactions and reporting in accordance with the regulatory-basis of accounting described therein that may be adopted by federally insured, state-chartered credit unions with under \$10 million in assets. AU-C section 800, Special Considerations—Audits of Financial Statements Prepared in Accordance With Special Purpose Frameworks (AICPA, Professional Standards), addresses special considerations in an audit of financial statements prepared in accordance with a special purpose framework, which includes a regulatory basis of accounting. For purposes of GAAS, a regulatory basis is defined in AU-C section 800 as a basis of accounting that the entity uses to comply with the requirements or financial reporting provisions of a regulatory agency to whose jurisdiction the entity is subject. The objective of the auditor, when applying GAAS in an audit of financial statements prepared in accordance with a special purpose framework, is to address appropriately the special considerations that are relevant to (a) the acceptance of the engagement, (b) the planning and performance of that engagement, and (c) forming an opinion and reporting on the financial statements. The illustration included in paragraph 23.23 specifically focuses on reporting considerations in an audit of financial statements prepared in accordance with a special purpose framework.

**23.23** The following is an illustration of an auditor's report on financial statements prepared in accordance with the financial reporting provisions prescribed by the NCUA (a special purpose framework). In this illustration, the financial statements together with the auditor's report are not intended for general use.

## **Independent Auditor's Report**

To the [Institution, Board of Directors, or Stockholders]:

## Report on the Financial Statements<sup>16</sup>

We have audited the accompanying financial statements of XYZ Credit Union, which comprise the statements of financial condition—regulatory basis as of December 31, 20X1 and 20X0, and the related statements of income and comprehensive income—regulatory basis, members' equity—regulatory basis, and cash flows—regulatory basis for the years then ended, and the related notes to the financial statements.

#### Management's Responsibility for the Financial Statements

Management is responsible for the preparation and fair presentation of these financial statements in accordance with the financial reporting provisions prescribed or permitted by the National Credit Union

<sup>16</sup> See footnote 2.

Administration (NCUA) as more fully described in Note X; this includes determining that the regulatory basis of accounting described in Note X is an acceptable basis for the preparation of the financial statements in the circumstances. <sup>17</sup> Management is also responsible for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of financial statements that are free from material misstatement, whether due to fraud or error.

## Auditor's Responsibility

Our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement.

An audit involves performing procedures to obtain audit evidence about the amounts and disclosures in the financial statements. The procedures selected depend on the auditor's judgment, including the assessment of the risks of material misstatement of the financial statements, whether due to fraud or error. In making those risk assessments, the auditor considers internal control relevant to the entity's preparation and fair presentation of the financial statements in order to design audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the entity's internal control. Accordingly, we express no such opinion. An audit also includes evaluating the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluating the overall presentation of the financial statements.

We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

#### **Opinion**

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of XYZ Credit Union as of December 31, 20X1 and 20X0, and the results of its operations and its cash flows for the years then ended in accordance with the financial reporting provisions prescribed or permitted by the NCUA as described in Note X.

## Basis of Accounting

We draw attention to Note X to the financial statements, which describes the basis of accounting. As described in Note X to the financial statements, the financial statements are prepared by XYZ Credit Union on the basis of the financial reporting provisions prescribed or permitted by the NCUA, which is a basis of accounting other than accounting principles generally accepted in the United States of

<sup>&</sup>lt;sup>17</sup> If management does not have a choice of financial reporting frameworks, the auditor is not required to make reference to management's responsibility for determining that the applicable financial reporting framework is acceptable in the circumstances.

<sup>&</sup>lt;sup>18</sup> See footnote 4.

America, to meet the requirements of the NCUA. Our opinion is not modified with respect to this matter.

#### Restriction on Use

Our report is intended solely for the information and use of the board of directors and management of XYZ Credit Union and the NCUA, and is not intended to be and should not be used by anyone other than these specified parties.

## Report on Other Legal and Regulatory Requirements

[Form and content of this section of the auditor's report will vary depending on the nature of the auditor's other reporting responsibilities.]

[Auditor's signature]

[Auditor's city and state]

[Date of the auditor's report]

## Members' Shares Reported as Equity

**23.24** As discussed in paragraph 13.43 of this guide, GAAP require that members' shares of a credit union be reported as liabilities in the statement of financial condition. If members' shares are not reported as such, or in any other manner in which it is not unequivocal that members' shares are liabilities, and the shares are material to the financial statements, the auditor should express a qualified opinion or, in certain cases, an adverse opinion on the financial statements unless the financial statements are prepared using a special purpose framework (see paragraphs 23.22–.23). An illustration of a report modified for a qualified opinion in those circumstances follows.

## Independent Auditor's Report

To the [Institution, Board of Directors, or Stockholders]:

#### Report on the Financial Statements<sup>19</sup>

[Same first and second paragraphs as the standard report]

## Auditor's Responsibility

[Same third and fourth paragraphs as the standard report]

We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our qualified audit opinion.

## **Basis for Qualified Opinion**

XYZ Credit Union has reported members' shares as equity in the accompanying balance sheets. Accounting principles generally accepted in the United States of America require members' shares to be reported as liabilities in the balance sheet. If the Credit Union had properly reported these shares as liabilities, liabilities would increase and equity would decrease by \$\_\_\_\_\_ and \$\_\_\_ as of December 31, 20X1 and 20X0, respectively.

## **Qualified Opinion**

In our opinion, except for the effects of the matter described in the Basis for Qualified Opinion paragraph, the financial statements referred to above present fairly, in all material respects, the financial position of XYZ Credit Union as of December 31, 20X1 and 20X0, and

©2016, AICPA

<sup>19</sup> See footnote 2.

the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

## Report on Other Legal and Regulatory Requirements

[Form and content of this section of the auditor's report will vary depending on the nature of the auditor's other reporting responsibilities.]

[Auditor's signature]

[Auditor's city and state]

[Date of the auditor's report]

## **Communication of Internal Control Related Matters**

23.25 AU-C section 265, Communicating Internal Control Related Matters Identified in an Audit (AICPA, Professional Standards), addresses the auditor's responsibility to appropriately communicate to those charged with governance and management deficiencies in internal control that the auditor has identified in an audit of financial statements. Paragraphs .11–.12 of AU-C section 265 state that the auditor should communicate, in writing to those charged with governance, on a timely basis, significant deficiencies and material weaknesses identified during the audit, including those that were remediated during the audit. The auditor also should communicate to management at an appropriate level of responsibility, on a timely basis

- a. in writing, significant deficiencies and material weaknesses that the auditor has communicated or intends to communicate to those charged with governance, unless it would be inappropriate to communicate directly to management in the circumstances.
- b. in writing or orally, other deficiencies in internal control identified during the audit that have not been communicated to management by other parties and that, in the auditor's professional judgment, are of sufficient importance to merit management's attention. If other deficiencies in internal control are communicated orally, the auditor should document the communication.

These communications, as stated in paragraph .13 of AU-C section 265, should be made no later than 60 days following the report release date. However, as further explained in paragraph .A16 of AU-C section 265, the communication is best made by the report release date because receipt of such communication may be an important factor in enabling those charged with governance to discharge their oversight responsibilities. Nothing in AU-C section 265 precludes the auditor from communicating to those charged with governance or management other internal control matters that the auditor has identified during the audit. Paragraph .16 of AU-C section 265 states that the auditor should not issue a written communication stating that no significant deficiencies were identified during the audit.

**23.26** AU-C section 265 is not applicable if the auditor is engaged to report on the effectiveness of an entity's internal control over financial reporting under AT section 501, An Examination of an Entity's Internal Control Over Financial Reporting That Is Integrated With an Audit of Its Financial Statements (AICPA, Professional Standards). AT section 501 establishes requirements and provides guidance that applies when an auditor is engaged to perform an examination of the design and operating effectiveness of an entity's internal control over financial reporting that is integrated with an audit of financial statements.

Among other reporting considerations paragraph .106 of AT section 501 states the auditor should not issue a report indicating that no material weaknesses were identified during the integrated audit.

AU-C section 265 is not applicable if the auditor is engaged to perform an audit of internal control over financial reporting that is integrated with an audit of financial statements. In such circumstances, AU-C section 940, An Audit of Internal control Over Financial Reporting That Is Integrated With an Audit of Financial Statements (AICPA, Professional Standards), applies. AU-C section 940 establishes requirements and provides guidance that applies only when an auditor is engaged to perform an audit of internal control over financial reporting that is integrated with an audit of financial statements. Among other reporting considerations, because the auditor issues a report that expresses an opinion on the effectiveness of the entity's internal control over financial reporting, paragraph .65 of AU-C section 940 states that the auditor should not issue a report indicating that no material weaknesses were identified during the integrated audit.<sup>20</sup>

**23.27** Part 363 of the FDIC's Rules and Regulations requires the independent public accountant's report on internal control over financial reporting to include a statement that the evaluation included controls over the preparation of regulatory financial statements in accordance with regulatory reporting instructions including identification of the regulatory reporting instructions. Exhibit C, "Reporting Under Section 112 of the Federal Deposit Insurance Corporation Improvement Act (FDICIA)," of AT section 501 provides guidance and an illustrative definition paragraph to assist auditors in complying with FDICIA and Part 363.

Part 363 of the FDIC's Rules and Regulations requires the independent public accountant's report on internal control over financial reporting to include a statement that the evaluation included controls over the preparation of regulatory financial statements in accordance with regulatory reporting instructions including identification of the regulatory reporting instructions. Exhibit D, "Reporting Under Section 112 of the Federal Deposit Insurance Corporation Improvement Act," of AU-C section 940 provides guidance and an illustrative definition paragraph to assist auditors in complying with the Federal Deposit Insurance Corporation Improvement Act and Part 363. <sup>21</sup>

## **Reports on Supervisory Committee Audits**

**23.28** Part 715 of the NCUA's regulations requires a credit union's Supervisory Committee to obtain an annual audit of the credit union according

<sup>&</sup>lt;sup>20</sup> Statement on Auditing Standards (SAS) No. 130, An Audit of Internal Control Over Financial Reporting That Is Integrated With an Audit of Financial Statements (AICPA, Professional Standards, AU-C section 940), was issued in November 2015. The SAS is effective for integrated audits for periods ending on or after December 15, 2016.

The Auditing Standards Board concluded that, because engagements performed under AT section 501, An Examination of an Entity's Internal Control Over Financial Reporting That Is Integrated With an Audit of Its Financial Statements (AICPA, Professional Standards), as well as related attestation Interpretation No. 1, "Reporting Under Section 112 of the Federal Deposit Insurance Corporation Improvement Act" (AICPA, Professional Standards, AT sec. 9501), are required to be integrated with an audit of financial statements, it is appropriate to move the content of AT section 501 from the attestation standards into generally accepted auditing standards.

Readers are encouraged to consult the full text of this SAS on the AICPA's website at www.aicpa.org.

<sup>&</sup>lt;sup>21</sup> See footnote 20.

to its charter type and asset size. Any federally insured credit union can meet this requirement by obtaining an independent audit of its financial statements performed in accordance with GAAS. Depending on the credit union's charter type and asset size, this requirement may also be met through other audit options, one of which may be an audit conducted in accordance with the procedures prescribed in NCUA's Supervisory Committee Guide for Federal Credit Unions. 22 The form and content of reports that are currently prepared by independent auditors in connection with supervisory committee audits reflect a diversity of practice. As a result, supervisory committee members may not understand the fundamental differences between an engagement for the application of agreed-upon procedures to specified elements, accounts, or items of a financial statement in connection with a supervisory committee audit and an audit of a credit union's financial statements in accordance with GAAS. This is of particular concern when the limitations of the supervisory committee audit relate to areas of higher risk in the credit union industry. Further, supervisory committee members may incorrectly assume that the application of agreed-upon procedures included obtaining an understanding of the credit union's internal control similar to that obtained in an audit of the credit union's financial statements in accordance with GAAS.

**23.29** Independent auditors' reports on audits of financial statements should comply with the reporting provisions contained in applicable AICPA professional standards. AU-C sections 700, 705, 706, and 805, Special Considerations—Audits of Single Financial Statements and Specific Elements, Accounts, or Items of a Financial Statement (AICPA, Professional Standards), provide guidance on reports on audited financial statements.

**23.30** Independent accountants' reports on the performance of agreed-upon procedures in connection with a supervisory committee audit should be prepared in accordance with Statements on Standards for Attestation Engagements. Paragraph .31 of AT section 201, Agreed-Upon Procedures Engagements (AICPA, Professional Standards), states that such reports should contain the following elements:

- A title that includes the word *independent*.
- Identification of specified parties (see paragraph .36 of AT section 201).
- Identification of the subject matter (or the written assertion related thereto) and the character of the engagement
- Identification of the responsible party.
- A statement that the subject matter is the responsibility of the responsible party.
- A statement that the procedures performed were those agreed to by the specified parties identified in the report.
- A statement that the agreed-upon procedures engagement was conducted in accordance with attestation standards established by the AICPA

<sup>&</sup>lt;sup>22</sup> An audit in accordance with the Supervisory Committee Guide for Federal Credit Unions may be performed by the supervisory committee, the credit union's internal auditor, or any other qualified person (including an independent public accountant); qualified persons that are not state-licensed cannot provide assurance services.

- A statement that the sufficiency of the procedures is solely the responsibility of the specified parties and a disclaimer of responsibility for the sufficiency of those procedures
- A list of the procedures performed (or reference thereto) and related findings. The practitioner should not provide negative assurance (see paragraph .24 of AT section 201).
- Where applicable, a description of any agreed-upon materiality limits (see paragraph .25 of AT section 201).
- A statement that the practitioner was not engaged to and did not conduct an examination of the subject matter, the objective of which would be the expression of an opinion, a disclaimer of opinion on the subject matter, and a statement that if the practitioner had performed additional procedures, other matters might have come to his or her attention that would have been reported.
- A statement of restrictions on the use of the report because it is intended to be used solely by the specified parties.
- Where applicable, reservations or restrictions concerning procedures or findings as discussed in paragraphs .33, .35, and .39-.40 of AT section 201.
- For an agreed-upon procedures engagement on prospective financial information, all items included in paragraph .55 of AT section 301, Financial Forecasts and Projections (AICPA, Professional Standards).
- Where applicable, a description of the nature of the assistance provided by a specialist as discussed in paragraphs .19–.21 of AT section 201.
- The manual or printed signature of the practitioner's firm.
- The date of the report.

## © Update 23-1 Attestation: Clarification and Recodification

Statement on Standards for Attestation Engagements (SSAE) No. 18, Attestation Standards: Clarification and Recodification (AICPA, Professional Standards), issued in April 2016, is effective for practitioners' reports dated on or after May 1, 2017.

To address concerns over the clarity, length, and complexity of its standards, the ASB established clarity drafting conventions and undertook a project to redraft all the standards it issues in clarity format. The redrafting of SSAEs (attestation standards) in SSAE No. 18 represents the culmination of that process. The attestation standards are developed and issued in the form of SSAEs and are codified into sections. SSAE No. 18 recodifies the "AT" section numbers designated by SSAE Nos. 10–17 using the identifier "AT-C" to differentiate the sections of the clarified attestation standards (AT-C sections) from the attestation standards that are superseded by SSAE No. 18 (AT sections). The AT sections in AICPA *Professional Standards* remain effective through April 2017, by which time substantially all engagements for which the AT sections were still effective are expected to be completed.

©2016, AICPA

This edition of the guide has not been updated to reflect changes as a result of this SSAE, however, this paragraph will be updated in a future edition. Readers are encouraged to consult the full text of this SSAE on the AICPA's website at www.aicpa.org.

- **23.31** As mentioned previously, some regulatory agencies require that supervisory committee audit reports include financial statements or other data. In such instances, the supervisory committee usually includes the auditor's report on the application of agreed-upon procedures and the unaudited financial statements or data in its report to the regulatory agency.
- **23.32** An auditor may be requested to perform specific procedures in conjunction with a compilation or review of financial statements. The procedures employed in compilation and review engagements, and reports thereon, should comply with, among other professional standards, the provisions of AR-C section 60, General Principles for Engagements Performed in Accordance With Statements on Standards for Accounting and Review Services; AR-C section 80, Compilation Engagements; and AR-C section 90, Review of Financial Statements (AICPA, Professional Standards). Any procedures that the accountant might have performed before or during the review engagement, including those performed in connection with a compilation of the financial statements, should not be described in his or her report. However, this would not preclude the auditor from issuing a separate, special-purpose report on the nature and extent of procedures performed.
- 23.33 Q&A section 9150.34, "Modifications to the Accountant's Compilation or Review Report When a Client Adopts a Private Company Council Accounting Alternative That Results in a Changes to a Previously Issued Report" (AICPA, *Technical Questions and Answers*), addresses the need to include an explanatory paragraph in the auditor's report when an entity adopts a PCC alternative that results in a change to a previously issued report, specifically if the prior year report included a departure from GAAP and now that the PCC alternative had been adopted, the prior year no longer contains the departure from GAAP. Paragraph .49 of AR-C section 90, states that when the accountant's report on the financial statements of the prior period contains a changed reference to a departure from the applicable financial reporting framework, the accountant's review report should include an other-matter paragraph indicating
  - *a.* the date of the accountant's previous review report.
  - b. the circumstances or events that caused the reference to be changed.
  - c. when applicable, that the financial statements of the prior period have been changed.

According to paragraph .86 of AR-C section 90, a changed reference to a departure from the applicable financial reporting framework includes the removal of a prior reference or the inclusion of a new reference.

## Example Report on the Application of Agreed-Upon Procedures Performed in Connection With a Supervisory Committee Audit

**23.34** The following is an example of a report on the application of agreed-upon procedures performed in connection with a supervisory committee audit.

## Independent Accountant's Report on Applying Agreed-Upon Procedures

Supervisory Committee XYZ Credit Union

We have performed the procedures enumerated in the attached supplement, which were agreed to by [list specified parties, 23] ordinarily the Supervisory Committee of XYZ Credit Union], solely to assist you in connection with your supervisory audit of XYZ Credit Union conducted pursuant to section 715 of the National Credit Union Administration regulations for the period ended June 30, 20X0. The procedures performed by us and enumerated in the attached supplement are in accordance with the minimum procedures described in appendix A of the National Credit Union Administration's Supervisory Committee Guide for Federal Credit Unions. Because the committee is responsible to ensure that a complete set of procedures is performed and because appendix A procedures are designed for smaller, less complex credit unions, we performed supplemental procedures at the committee's request. This engagement to apply agreed-upon procedures was conducted in accordance with attestation standards established by the American Institute of Certified Public Accountants. The sufficiency of the procedures is solely the responsibility of those parties specified in this report. Consequently, we make no representation regarding the sufficiency of the procedures described in the supplement either for the purpose for which this report has been requested or for any other purpose.

We were not engaged to and did not perform an audit, the objective of which would be the expression of an opinion on XYZ Credit Union's financial statements or specified elements, accounts, or items thereof. Accordingly, we do not express such an opinion. Had we performed additional procedures, other matters might have come to our attention that would have been reported to you.

This report is intended solely for the information and use of [the specified parties] and is not intended to be and should not be used by anyone other than these specified parties.

[Signature] [Date]

## Supplement to Illustrative Report<sup>24</sup>

#### Loans

We obtained trial balances or subsidiary ledgers of the notes or both from the service center and reconciled the totals to the general ledger in the following amounts:

<sup>&</sup>lt;sup>23</sup> The National Credit Union Administration (NCUA) should not be named as a specified party.

<sup>&</sup>lt;sup>24</sup> This supplement is for illustrative purposes only and, therefore, is not considered to be an all-inclusive list of either the accounts on which procedures may be performed or the procedures that may be performed. The illustrative procedures listed may or may not be relevant to a particular engagement. The independent accountant should describe those accounts on which procedures were performed and the procedures relevant to the specific engagement. The accounts and procedures described in the report should generally conform to those described in the engagement letter. Procedures for other accounts should be specified in detail, and differences and subsequent disposition should be reported.

Appendix A to the NCUA's Supervisory Committee Guide for Federal Credit Unions presents the minimum procedures that a supervisory committee or its auditor or other accountant must complete when a Supervisory Committee chooses the Supervisory Committee Guide for Federal Credit option for completing its annual audit requirement under Part 715 of the NCUA regulations.

Account	Amount Outstanding at June 30, 20X0
Business loans	\$
Consumer loans	
Real estate loans	
Participations purchased	
	\$

Certain [specify number] loans, including lines of credit that had not been fully funded, were selected for confirmation directly with borrowers. The results of our confirmation procedures are summarized in schedule A. Borrowers with lines of credit of \$\_\_\_\_\_\_ or more as of June 30, 20X0, who did not respond to confirmation requests by July 31, 20X0, are listed in schedule B.

We obtained and read selected [specify number] loan agreements on hand and inspected readily marketable securities and other collateral recorded as held in respect of certain selected secured loans.

We obtained the Credit Union's listing of business loans, real estate loans, and participations purchased five days or more past due as of June 30, 20X0, and compared it with a similar listing as of July 31, 20X0. The following loans were listed in both reports:

		Amount	Amount
		Outstanding at	Outstanding at
Name	$Due\ Date$	June 30, $20X0$	July 31, 20X0

Similarly, we obtained the Credit Union's listing of consumer loans ten days or more past due as of June 30, 20X0, and compared it to a like listing as of July 31, 20X0. The following loans were listed in both reports:

		$Amount\ at$	Amount at July
Name	$Due\ Date$	June 30, $20X0$	31, 20X0

Loan participations [Specify "all" or number] "sold" and serviced by the credit union were confirmed with the purchasers, without exception.

We obtained the Credit Union's listing of overdrafts as of June 30, 20X0, and compared it to a similar listing as of July 31, 20X0. The following overdrafts were listed in both reports:

Name	Date of Overdraft	Amount at June 30, 20X0	Amount at July 31, 20X0

The interest rates and repayment terms of five judgmentally selected loans granted to directors, officers, and other related parties during May 20X0 were compared to the interest rate and repayment terms of similar loans granted to outsiders during the same month. We found no instances where favorable interest rates or repayment terms were granted to directors, officers, and other related parties.

The maturity date and amount of loan commitments in excess of \$50,000 were confirmed as of May 20X0 by the customers for whose benefit they were issued, without exception. We judgmentally selected five loan commitments and tested the computation of deferred fee income. [Specify results of computations.]

Requests for confirmation of loan balances could not be mailed to the following borrowers due to lack of sufficient addresses:

Name Account Balance as of Number June 30, 20X0

Lack of Evaluation of Collectibility and Adequacy of Collateral

As noted in our engagement letter and report, we did not evaluate the collectibility of loans or the adequacy of collateral thereon.

Lack of Evaluation of the Allowance for Loan Losses

As noted in our engagement letter and report, we did not evaluate the reasonableness of the allowance for loan losses determined by management.

## **Confirmation Statistics**

[Confirmation Date]

Share
Draft Savings Certificates
Loans Accounts Accounts of Deposit

Dollar amounts Total

Circularized

Percent circularized to total

Replies received to total circularized

Selected but not circularized

Not delivered by post office

Number of accounts

Total

Circularized

Percent circularized to total

Replies received

Percent replies received to total

circularized

Selected but not circularized

Not delivered by post office

## **Confirmation Requests Not Mailed**

Name		Balance as of
and	$Reason\ for$	June~30,
Address	Not Mailing	20X0

Loans Share draft accounts Savings accounts Certificates of deposit

*Note:* An indication of how the samples were selected (that is, on a random, statistical, or judgmental basis), as well as an indication of the type of confirmation (that is, positive or negative requests), should be included. If the loans are categorized by type in the report, similar categories would normally be used in this schedule.

# Example Reports on the FDIC Loss Sharing Purchase and Assumption Transactions

23.35 The FDIC's Resolutions Handbook (handbook) states that a loss sharing transaction is a purchase and assumption (P&A) transaction that the FDIC commonly uses as a resolution tool for handling failed institutions with more than \$500 million in assets. A P&A is a resolution transaction in which a healthy institution purchases some or all of the assets of a failed bank or thrift and assumes some or all of the liabilities, including all insured deposits. The handbook also states that a loss sharing P&A uses the basic P&A structure, except for the provision regarding transferred assets. Instead of selling some or all of the assets to the acquirer at a discounted price, the FDIC agrees to share in future losses experienced by the acquirer on a fixed pool of assets. The handbook for P&A agreements requires that within 90 days after each calendar year end, the acquiring bank must furnish the FDIC a report signed by its independent public accountants containing specified statements<sup>25</sup> relative to the accuracy of any computations made regarding shared loss assets. It must also perform a semi-annual internal audit of the shared loss compliance and provide the FDIC with copies of the internal audit reports and access to the internal audit work papers.

**23.36** Q&A section 9110.16, "Example Reports on Federal Deposit Insurance Corporation Loss Sharing Purchase and Assumption Transactions" (AICPA, *Technical Questions and Answers*), provides examples of how the auditor might respond to FDIC reporting requirements for an engagement covering an FDIC loss sharing P&A transaction. Q&A section 9110.16 suggests that the auditor may respond by issuing a report in accordance with the requirements of AU-C section 806, *Reporting on Compliance With Aspects of Contractual Agreements or Regulatory Requirements in Connection With Audited Financial Statements* (AICPA, *Professional Standards*), and also provides illustrative auditor reports for three possible outcomes for which the independent auditor might report.

<sup>&</sup>lt;sup>25</sup> The term *specified statements* is not defined in the FDIC's Resolutions Handbook. The practitioner is advised to read the terms of the loss share agreement and confirm that the audit requirement in that agreement provides for the receipt of a report expressing negative assurance.

## **Small Business Lending Fund Auditor Certification Guidance**

23.37 Under the terms of the Small Business Lending Fund (SBLF), a community bank is required to calculate and report to the Treasury Department the amount of its qualified small business lending in a supplemental report. The bank's management is required to certify that the information provided in the report is accurate and also to submit a certification from its external auditors that the processes and controls used to generate the supplemental reports are satisfactory. Q&A section 9110.18, "Small Business Lending Fund Auditor Certification Guidance" (AICPA, Technical Questions and Answers), indicates that an independent auditor may satisfy this requirement by issuing a report in accordance with the requirements of AU-C section 806 and provides an illustration of a report that an auditor may use when, as a result of the auditor's audit procedures, nothing has come to the auditor's attention to indicate that the bank failed to comply with the terms of the SBLF.

23.38

## Appendix A—Illustrative Unqualified PCAOB Reports

This appendix is nonauthoritative and is included for informational purposes only.

## Introduction

This appendix provides illustrative unqualified reports applying the requirements of the PCAOB. The reports are illustrative; the facts and circumstances of each particular audit will govern the appropriate form of report. A discussion of the types of modifications of the reports that may be appropriate (for example, adverse opinion, disclaimer of opinion, and explanatory paragraphs) is not presented herein.

## Reports

## General Form of Unqualified Opinion on Financial Statements

The following is an illustration of an auditor's report (unqualified opinion) on the financial statements of a bank or savings institution when a report on internal controls over financial reporting is not being issued.

## Report of Independent Registered Public Accounting Firm

To the [Institution, Board of Directors, or Stockholders]:

We have audited the accompanying balance sheets of ABC Institution as of December 31, 20X1 and 20X0, and the related statements of income and comprehensive income, stockholders' equity, and cash flows for each of the years in the three-year period ended December 31, 20X1.<sup>2</sup> These financial statements are the responsibility of ABC

Readers are encouraged to visit the PCAOB website at www.pcaobus.org for the full text of the proposed standards and should remain alert for further developments.

<sup>&</sup>lt;sup>1</sup> In August 2013, the PCAOB issued the proposed standard *The Auditor's Responsibilities Regarding Other Information in Certain Documents Containing Audited Financial Statements and the Related Auditor's Report.* The proposed standard would respond to investor's interests in obtaining information regarding the auditor's responsibilities for other information outside the financial statements that is contained in documents that include audited financial statements and the related auditor's report.

In May 2016, the PCAOB reproposed the auditing reporting standard, *The Auditor's Report on an Audit of Financial Statements When the Auditor Expresses and Unqualified Opinion*. The reproposed auditor reporting standard would retain the pass/fail model and would provide more information to investors and other financial statement users regarding the audit and the auditor. While the reproposed standard would still require the auditor to communicate in the auditor's report "critical audit matters" that would be specific to each audit, the requirements would not apply to audits of brokers and dealers reporting under the Securities Exchange Act of 1934 Rule 17a-5, investment companies other than business development companies, and employee stock purchase, savings and similar plans. The auditor's required communication would focus on those matters the auditor addresses during the audit of the financial statements that involved the most difficult, subjective, or complex auditor judgments or posed the most difficulty to the auditor in obtaining sufficient appropriate audit evidence or forming an opinion on the financial statements.

<sup>&</sup>lt;sup>2</sup> These references to the financial statements should conform to the titles and periods of the statements being reported upon. For example, pursuant to FASB Accounting Standards Update No. 2011-05, Comprehensive Income (Topic 220): Presentation of Comprehensive Income, institutions may present the components of net income and other comprehensive income either in a single continuous statement or in two separate but consecutive statements. The financial statements identified in the first paragraph would depend on an institution's selected presentation. In addition, there may be circumstances (such as for Emerging Growth or Smaller Reporting Companies, a change in auditor, or a newly organized registrant, for example) where less than two years of a balance sheet and three years of the other financial statements are being reported upon.

Institution's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements. An audit also includes assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of ABC Institution as of December 31, 20X1 and 20X0, and the results of its operations and its cash flows for each of the years in the three-year period ended December 31, 20X1 in conformity with U.S. generally accepted accounting principles.

[Signature]

[City and State or Country]

[Date]

## General Form of Unqualified Opinion on the Effectiveness of Internal Control over Financial Reporting

When performing an integrated audit of financial statements and internal control over financial reporting in accordance with PCAOB standards, the auditor may choose to issue a combined report or separate reports on the company's financial statements and on internal control over financial reporting. Refer to paragraphs .85–.98 of AS 2201, An Audit of Internal Control Over Financial Reporting That Is Integrated with An Audit of Financial Statements (AICPA, PCAOB Standards and Related Rules), for direction on reporting on internal control over financial reporting.

The following is an illustration of an auditor's *combined* report expressing unqualified opinions on each of the financial statements and the effectiveness of internal control over financial reporting of a bank or savings institution.

## Report of Independent Registered Public Accounting Firm

To the [Institution, Board of Directors, or Stockholders]:

We have audited the accompanying balance sheets of ABC Institution as of December 31, 20X1 and 20X0, and the related statements of income and comprehensive income, stockholders' equity, and cash flows for each of the years in the three-year period ended December 31, 20X1.<sup>3</sup> We also have audited ABC Institution's internal control over financial reporting as of December 31, 20X1, based on [identify criteria].<sup>4</sup> ABC Institution's management is responsible for these

<sup>&</sup>lt;sup>3</sup> See footnote 2.

<sup>&</sup>lt;sup>4</sup> For example, the following may be used to identify the criteria: "Criteria established in the 2013 (or 1992) *Internal Control—Integrated Framework* issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO)." On May 14, 2013, COSO published an updated *Internal Control—Integrated Framework*.

financial statements, for maintaining effective internal control over financial reporting, and for its assessment of the effectiveness of internal control over financial reporting, included in the accompanying [title of management's report]. Our responsibility is to express an opinion on these financial statements and an opinion on ABC Institution's internal control over financial reporting based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement and whether effective internal control over financial reporting was maintained in all material respects. Our audits of the financial statements included examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. Our audit of internal control over financial reporting included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, and testing and evaluating the design and operating effectiveness of internal control based on the assessed risk. Our audits also included performing such other procedures as we considered necessary in the circumstances. We believe that our audits provide a reasonable basis for our opinion.

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection and correction of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of ABC Institution as of December 31, 20X1 and 20X0, and the results of its operations and its cash flows for each of the years in the three-year period ended December 31, 20X1, in conformity with U.S. generally accepted ac-

counting principles. Also in our opinion, ABC Institution maintained, in all material respects, effective internal control over financial reporting as of December 31, 20X1, based on [identify criteria].<sup>5</sup>

[Signature]

[City and State or Country]

[Date]

The following is an illustration of an auditor's report expressing an unqualified opinion on the effectiveness of internal control over financial reporting of a bank or savings institution when a separate report expressing an unqualified opinion on the financial statements is issued.

# Report of Independent Registered Public Accounting Firm

To the [Institution, Board of Directors, or Stockholders]:

We have audited ABC Institution's internal control over financial reporting as of December 31, 20X1, based on [identify criteria]. ABC Institution's management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting, included in the accompanying [title of management's report]. Our responsibility is to express an opinion on ABC Institution's internal control over financial reporting based on our audit.

We conducted our audit in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. Our audit included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, and testing and evaluating the design and operating effectiveness of internal control based on the assessed risk. Our audit also included performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

<sup>&</sup>lt;sup>5</sup> See footnote 4.

<sup>&</sup>lt;sup>6</sup> See footnote 4.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

In our opinion, ABC Institution maintained, in all material respects, effective internal control over financial reporting as of December 31, 20X1, based on [identify criteria].<sup>7</sup>

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the balance sheets of ABC Institution as of December 31, 20X1 and 20X0, and the related statements of income and comprehensive income, stockholders' equity, and cash flows for each of the years in the three-year period ended December 31, 20X1, and our report dated [date of report, which should be the same as the date of the report on the effectiveness of internal control over financial reporting] expressed [include nature of opinion].<sup>8</sup>

If the auditor issues separate reports on the company's financial statements and on the effectiveness of internal control over financial reporting, the following paragraph, as presented in paragraph 88 of AS 2201, should be added to the auditor's report on the company's financial statements:

We also have audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), ABC Institution's internal control over financial reporting as of December 31, 20X1, based on [identify control criteria]<sup>9</sup> and our report dated [date of report, which should be the same as the date of the report on the financial statements] expressed [include nature of opinion].

When performing an integrated audit of financial statements and internal control over financial reporting in accordance with PCAOB standards, the auditor's report on the company's financial statements and on internal control over financial reporting should be dated the same date. Refer to paragraph 89 of AS 2201.

<sup>&</sup>lt;sup>7</sup> See footnote 4.

<sup>&</sup>lt;sup>8</sup> See footnote 4.

<sup>&</sup>lt;sup>9</sup> See footnote 4.

# Appendix A

# **FDI Act Reporting Requirements**

This appendix is nonauthoritative and is included for informational purposes only.

**A.01** Section 36 of the Federal Deposit Insurance Act (FDI Act) and its implementing regulation Title 12 U.S. *Code of Federal Regulations* Part 363 require reports by each institution's management and its auditors over financial statements and internal control over financial reporting. Section 36 and Part 363 also establish minimum qualifications for auditors that provide audit and attest services to insured depository institutions. Section 36 and Part 363 apply to each FDIC-insured depository institution having total assets of \$500 million or greater at the beginning of its fiscal year. The requirements specified in Section 36 and Part 363 are in addition to any other statutory and regulatory requirements otherwise applicable to an insured depository institution.

**A.02** Part 363 was initially adopted by the FDIC's Board of Directors in 1993 and was most recently amended in 2013. The general requirements are summarized in the following paragraphs; the side-by-side analysis of the detailed regulation and guidelines is presented in the exhibit that follows. Each institution's management, board of directors, and audit committee, as well as independent public accountants that provide audit and attestation services to institutions subject to Part 363 are encouraged to read and become familiar with the Part 363 regulatory text, the guidelines and interpretations in appendix A to Part 363, "Guidelines and Interpretations," and the illustrative management reports in appendix B to Part 363, "Illustrative Management Reports," to obtain a complete understanding of the compliance requirements of Part 363.

# Part 363 Annual Reports for Institutions With \$500 Million or More but Less Than \$1 Billion in Total Assets

**A.03** Insured depository institutions with at least \$500 million but less than \$1 billion in total assets are required to file a Part 363 Annual Report that must include the following:

- a. Audited comparative annual financial statements.
- b. The independent public accountant's report on the audited financial statements.
- c. A management report that contains
  - i. a statement of management's responsibilities for
    - (1) preparing the annual financial statements;
    - (2) establishing and maintaining an adequate internal control structure over financial reporting;<sup>1</sup> and

<sup>&</sup>lt;sup>1</sup> For purposes of Title 12 U.S. Code of Federal Regulations Part 363, financial reporting encompasses both financial statements prepared in accordance with generally accepted accounting principles and those prepared for regulatory reporting purposes.

- (3) complying with the designated safety and soundness laws and regulations pertaining to insider loans and dividend restrictions.
- ii. an assessment by management of the institution's compliance with the designated laws and regulations pertaining to insider loans and dividend restrictions during the year, which must state management's conclusion regarding compliance and disclose any noncompliance with these laws and regulations. The assessment must clearly state whether the institution has or has not complied with these regulations. Disclosure is not dependent on the degree or materiality of any noncompliance. Statements such as "management believes that the institution complied, in all material respects with the designated safety and soundness laws and regulations" do not present a definitive and unconditional conclusion regarding compliance as envisioned under Part 363.

**A.04** In general, an institution that is required to file, or whose parent holding company is required to file, management's assessment of the effectiveness of internal control over financial reporting with the SEC or the appropriate federal banking agency in accordance with Section 404 of the Sarbanes-Oxley Act of 2002 must submit a copy of such assessment with its Part 363 Annual Report as additional information. However, this assessment will not be considered part of the institution's Part 363 Annual Report.

# Part 363 Annual Reports for Institutions With \$1 Billion or More in Total Assets

**A.05** Insured depository institutions with \$1 billion or more in total assets are required to file a Part 363 Annual Report that must include the following:

- a. Audited comparative annual financial statements.
- b. The independent public accountant's report on the audited financial statements.
- c. A management report that contains
  - i. a statement of management's responsibilities for
    - (1) preparing the annual financial statements;
    - (2) establishing and maintaining an adequate internal control structure over financial reporting;<sup>2</sup> and
    - (3) complying with the designated safety and soundness laws and regulations pertaining to insider loans and dividend restrictions.
  - ii. an assessment by management on the effectiveness of the institution's internal control structure over financial reporting as of the end of the fiscal year that must

<sup>&</sup>lt;sup>2</sup> See footnote 1.

- identify the internal control framework<sup>3</sup> used by management to evaluate the effectiveness of internal control over financial reporting;
- (2) state that the assessment included controls over the preparation of regulatory financial statements in accordance with regulatory reporting instructions and identify the regulatory reporting instructions;
- (3) state management's conclusion regarding whether internal control over financial reporting is effective as of the institution's fiscal year-end;<sup>4</sup> and
- (4) disclose all material weaknesses in internal control over financial reporting, if any, that management has identified that have not been remediated prior to the institution's fiscal year-end.
- iii. an assessment by management of the institution's compliance with the designated laws and regulations pertaining to insider loans and dividend restrictions during the year, which must state management's conclusion regarding compliance and disclose any noncompliance with these laws and regulations. The assessment must clearly state whether the institution has or has not complied with these regulations. Disclosure is not dependent on the degree or materiality of any noncompliance. Statements such as "management believes that the institution complied, in all material respects with the designated safety and soundness laws and regulations" do not present a definitive and unconditional conclusion regarding compliance as envisioned under Part 363.
- d. The independent public accountant's attestation report concerning the effectiveness of the institution's internal control structure over financial reporting. The accountant's report must not be dated prior to the date of the management report and management's assessment of the effectiveness of internal control over financial reporting and must.
  - i. identify the internal control framework used by the independent public accountant, which must be the same as the internal control framework used by management, to evaluate the effectiveness of the institution's internal control over financial reporting;

<sup>&</sup>lt;sup>3</sup> For example, in the United States, the Committee of Sponsoring Organizations of the Treadway Commission (COSO) published on May 14, 2013, its updated *Internal Control—Integrated Framework*, including an addendum on safeguarding assets. Known as the COSO report, this publication provides a suitable and available framework for purposes of management's assessment. In addition to the updated Framework, COSO also issued the following companion documents: *Illustrative Tools for Assessing Effectiveness of a System of Internal Control and Internal Control over External Financial Reporting (ICEFR): A Compendium of Approaches and Examples.* 

<sup>&</sup>lt;sup>4</sup> If there are one or more material weaknesses that have not been remediated prior to the institution's fiscal year-end, management must conclude that internal control over financial reporting is ineffective.

- ii. state that the independent public accountant's evaluation included controls over the preparation of regulatory financial statements in accordance with regulatory reporting instructions and identify the regulatory reporting instructions:
- iii. state the independent public accountant's conclusion regarding whether internal control over financial reporting is effective as of the institution's fiscal year-end;<sup>5</sup> and
- iv. disclose all material weaknesses in internal control over financial reporting, if any, that the independent public accountant has identified that have not been remediated prior to the institution's fiscal year-end.

# Filing Deadlines for Part 363 Annual Reports

**A.06** An institution shall file its Part 363 Annual Report within 120 days after the end of its fiscal year if (1) it is neither a public company nor a subsidiary of a public company or (2) it is a subsidiary of a public holding company and its consolidated total assets (or the consolidated total assets of all of its parent holding company's insured depository institution subsidiaries) compose less than 75 percent of the consolidated total assets of the public holding company as of the beginning of its fiscal year.

**A.07** An institution shall file its Part 363 Annual Report within 90 days after the end of its fiscal year if (1) it is a public company or (2) it is a subsidiary of a public holding company and its consolidated total assets (or the consolidated total assets of all of its parent holding company's insured depository institution subsidiaries) compose 75 percent or more of the consolidated total assets of the public holding company as of the beginning of its fiscal year.

**A.08** If an institution will be unable to file its Part 363 Annual Report by the specified deadline, it must submit a notification of late filing.

# Other Requirements—All Institutions With \$500 Million or More in Total Assets

# Other Reports and Letters Issued by the Independent Public Accountant

**A.09** Except for the independent public accountant's reports that are included in its Part 363 Annual Report, each insured depository institution must file with the FDIC, the appropriate federal banking agency, and any appropriate state bank supervisor a copy of any management letter or other report issued by its independent public accountant with respect to the institution and the audit and attestation services provided by the accountant within 15 days after receipt. Such reports include, but are not limited to

any written communication regarding matters that the accountant is required to communicate to the audit committee (for example, critical accounting policies, alternative accounting treatments

<sup>&</sup>lt;sup>5</sup> If there are one or more material weaknesses that have not been remediated prior to the institution's fiscal year-end, the independent public accountant must conclude that internal control over financial reporting is ineffective.

- discussed with management, and any schedule of unadjusted differences).
- any written communication of significant deficiencies and material weaknesses in internal control required by the auditing or attestation standards of the AICPA or the PCAOB, as appropriate.
- for an institution with consolidated total assets of \$500 million or more but less than \$1 billion as of the beginning of its fiscal year that is (1) a public company or (2) a subsidiary of a public holding company and its consolidated total assets (or the consolidated total assets of all of its parent holding company's insured depository institution subsidiaries) compose 75 percent or more of the consolidated total assets of the public holding company as of the beginning of its fiscal year, any report by the independent public accountant on the audit of internal control over financial reporting required by Section 404 of the Sarbanes-Oxley Act of 2002 and the PCAOB's auditing standards.
- for an institution that is (1) a public company or (2) a subsidiary of a public holding company and its consolidated total assets (or the consolidated total assets of all of its parent holding company's insured depository institution subsidiaries) comprise 75 percent or more of the consolidated total assets of the public holding company as of the beginning of its fiscal year, any written communication by the independent public accountant of all deficiencies in internal control over financial reporting that are of a lesser magnitude than significant deficiencies, which is required by the PCAOB's auditing standards.
- for an institution that is (1) a nonpublic company or (2) a subsidiary of a nonpublic holding company and its consolidated total assets (or the consolidated total assets of all of its parent holding company's insured depository institution subsidiaries) compose 75 percent or more of the consolidated total assets of the nonpublic holding company as of the beginning of its fiscal year, any written communication by the independent public accountant of all deficiencies in internal control over financial reporting that are of a lesser magnitude than significant deficiencies, which is required by the AICPA's auditing and attestation standards.

# Notice of Engagement, Change, Dismissal, or Resignation of Accountants

**A.10** Within 15 days after a change in or the dismissal or resignation of the institution's independent public accountant or the engagement of a new independent public accountant, the institution must file written notice with the FDIC, the appropriate federal banking agency, and any appropriate state bank supervisor. Also, within 15 days after the institution's independent public accountant resigns or is dismissed, the independent public accountant must file written notice with the FDIC, the appropriate federal banking agency, and any appropriate state bank supervisor. These written notices should set forth in reasonable detail the reasons for the resignation or dismissal of the institution's independent public accountant.

**A.11** In this regard, before engaging an independent public accountant, the institution's audit committee should satisfy itself that the independent public accountant is in compliance with the qualifications and other requirements applicable to independent public accountants set forth in Part 363, including the independence standards of the AICPA, the SEC, and the PCAOB. Also, the audit committee should ensure that engagement letters and any related agreements with the independent public accountant for audit and attestation services to be performed under Part 363 do not contain any limitation of liability provisions that (1) indemnify the independent public accountant against claims made by third parties; (2) hold harmless or release the independent public accountant from liability for claims or potential claims that might be asserted by the client institution, other than claims for punitive damages; or (3) limit the remedies available to the client institution.

# Peer Reviews and Inspection Reports

**A.12** Within 15 days of receiving notification that a peer review has been accepted or a PCAOB inspection report has been issued, or before commencing any audit or attestation service under Part 363, whichever is earlier, the independent public accountant must file two copies of its most recent peer review report and the public portion of its most recent PCAOB inspection report, if any, accompanied by any letters of comments, response, and acceptance, with the FDIC, Accounting and Securities Disclosure Section, 550 17th Street, NW, Washington, DC 20429, if the report has not already been filed. Also, within 15 days of the PCAOB making public a previously nonpublic portion of an inspection report, the independent public accountant must file 2 copies of the previously nonpublic portion of the inspection report with the FDIC.

# Notification of Late Filing

**A.13** An institution that is unable to timely file all or any portion of its Part 363 Annual Report or any other report or notice required to be filed by Part 363 must submit a written notice of late filing to the FDIC, the appropriate federal banking agency, and any appropriate state bank supervisor. The notice shall disclose the institution's inability to timely file the report or notice and the reasons for the late filing in reasonable detail and state the date by which the report or notice will be filed. The written notice should be filed on or before the deadline for filing the Part 363 Annual Report or any other required report or notice, as appropriate.

# Standards for Audits of Financial Statements and Internal Control Over Financial Reporting

**A.14** The financial statement audit is to be performed in accordance with generally accepted auditing standards or the PCAOB's auditing standards, if applicable and Section 37 of the FDI Act. The examination of management's assertion about the institution's internal controls over financial reporting is to be performed in accordance with generally accepted standards for attestation engagements or the PCAOB's auditing standards, if applicable.

# General Qualifications of Auditors

**A.15** To provide audit and attest services to insured depository institutions, an independent public accountant should be registered or licensed to

practice as a public accountant and be in good standing under the laws of the state or other political subdivision of the United States in which the home office of the institution (or the insured branch of a foreign bank) is located. The accountant must also agree to provide regulators with copies of any working papers, policies, and procedures related to services performed under Part 363. Independent accountants should be familiar with Interpretation No. 1, "Providing Access to or Copies of Audit Documentation to a Regulator," of AU-C section 230, *Audit Documentation* (AICPA, *Professional Standards*, AU-C sec. 9230 par. .01–.15).

**A.16** The independent public accountant must comply with the independence standards and interpretations of the AICPA, the SEC, and the PCAOB. To the extent that any of the rules within any one of these independence standards (AICPA, SEC, and PCAOB) is more or less restrictive than the corresponding rule in the other independence standards, the independent public accountant must comply with the more restrictive rule.

# **Enforcement Actions Against Accountants**

**A.17** In August 2003, the FDIC, the Office of the Comptroller of the Currency, the Board of Governors of the Federal Reserve System, and the Office of Thrift Supervision, prior to its transfer of powers to the federal banking agencies, (the agencies) jointly issued final rules that establish procedures under which the agencies can remove, suspend, or bar an accountant or firm from performing audit and attestation services for insured depository institutions subject to the annual audit and reporting requirements of Section 36 of the FDI Act. The final rule can be accessed at www.fdic.gov/news/news/financial/2003/fil0366.html.

**A.18** Under the final rules, certain violations of law, negligent conduct, reckless violations of professional standards, or lack of qualifications to perform auditing services may be considered good cause to remove, suspend, or bar an accountant or firm from providing audit and attestation services for institutions subject to Section 36 of the FDI Act and Part 363. In addition, the rules prohibit an accountant or accounting firm from performing these services if the accountant or firm has been removed, suspended, or debarred by one of the agencies, or if the SEC or the PCAOB takes certain disciplinary actions against the accountant or firm. The rules also permit immediate suspensions of accountants and firms in limited circumstances.

# **Communication With Auditors**

**A.19** Section 36(h) of the FDI Act and Guideline 17 to Part 363 require an institution to provide its auditor with certain information including copies of the institution's most recent reports of condition and examination; any supervisory memorandum of understanding or written agreement with any federal or state regulatory agency; and a report of any action initiated or taken by federal or state banking regulators.

### **Audit Committees**

**A.20** Each insured depository institution is required to establish an audit committee of its board of directors, the composition of which complies with paragraphs (a)(1), (2), and (3) and (b) of Section 363.5. The duties of the audit committees shall include the appointment, compensation, and oversight of the independent public accountant who performs services required under Part 363,

and review with management and the independent public accountant the basis for the reports issued under Part 363. Each insured depository institution with total assets of \$1 billion or more as of the beginning of its fiscal year shall establish an independent audit committee of its board of directors, the members of which shall be outside directors who are independent of management of the institution. Each insured depository institution with total assets of \$500 million or more but less than \$1 billion as of the beginning of its fiscal year shall establish an audit committee of its board of directors, the members of which shall be outside directors, the majority of whom shall be independent of management of the institution. Each insured depository institution with total assets of \$3 billion or more as of the beginning of its fiscal year shall include in its audit committee members with banking or financial management expertise, have access to outside counsel, and not include any large customers of the institution. Guideline 35 to Part 363 provides transition guidance for forming and restructuring audit committees.

# **Audit and Reporting Requirements**

**A.21** Reprinted here is Part 363 of the FDIC's rules and regulations, *Part 363—Annual Independent Audits and Reporting Requirements*, (left column) and appendix A to Part 363 (right column). Part 363 and appendix A were initially published in 1993. The most recent amendments to this regulation were published in June 2009 and include a new appendix B to Part 363. Appendix B provides guidance regarding reporting scenarios that satisfy the annual reporting requirements of Part 363, illustrative management reports, and an illustrative cover letter for use when an institution complies with the annual reporting requirements at the holding company level. <sup>6</sup>

#### Regulation

### §363.1 Scope and definitions

(a) Applicability. This part applies to any insured depository institution with respect to any fiscal year in which its consolidated total assets as of the beginning of such fiscal year are \$500 million or more. The requirements specified in this part are in addition to any other statutory and regulatory requirements otherwise applicable to an insured depository institution.

### Guidelines and Interpretations

### Scope of Rule and Definitions (§363.1)

- 1. Measuring Total Assets. To determine whether this part applies, an institution should use total assets as reported on its most recent Report of Condition (Call Report) or Thrift Financial Report (TFR), the date of which coincides with the end of its preceding fiscal year. If its fiscal year ends on a date other than the end of a calendar quarter, it should use its Call Report or TFR for the quarter end immediately preceding the end of its fiscal year.
- 2. Insured Branches of Foreign Banks. Unlike other institutions, insured branches of foreign banks are not separately incorporated or capitalized. To determine whether this part applies, an insured branch should measure claims on non-related parties reported on its Report of Assets and Liabilities of U.S. Branches and Agencies of Foreign Banks (form FFIEC 002).

<sup>&</sup>lt;sup>6</sup> Readers should be aware that the FDIC has not updated Part 363 of its rules and regulations to reflect that the Office of the Comptroller of the Currency (OCC) eliminated the use of the Thrift Financial Report and now requires all federal savings associations to file using the Call Report.

- (b) Compliance by subsidiaries of holding companies. (1) For an insured depository institution that is a subsidiary of a holding company, the audited financial statements requirement of § 363.2(a) may be satisfied:
- (i) For fiscal years ending on or before June 14, 2010, by audited consolidated financial statements of the top-tier or any mid-tier holding company.
- (ii) For fiscal years ending on or after June 15, 2010, by audited consolidated financial statements of the top-tier or any mid-tier holding company provided that the consolidated total assets of the insured depository institution (or the consolidated total assets of all of the holding company's insured depository institution subsidiaries, regardless of size, if the holding company owns or controls more than one insured depository institution) comprise 75 percent or more of the consolidated total assets of this top-tier or mid-tier holding company as of the beginning of its fiscal year.
- (2) The other requirements of this part for an insured depository institution that is a subsidiary of a holding company may be satisfied by the top-tier or any mid-tier holding company if the insured depository institution meets the criterion specified in § 363.1(b)(1) and if:
- (i) The services and functions comparable to those required of the insured depository institution by this part are provided at this top-tier or mid-tier holding company level; and
- (ii) The insured depository institution has as of the beginning of its fiscal year:
- (A) Total assets of less than \$5 billion; or
- (B) Total assets of \$5 billion or more and a composite CAMELS rating of 1 or 2.
- (3) The appropriate Federal banking agency may revoke the exception in paragraph (b)(2) of this section for any institution with total assets in excess of \$9 billion for any period of time during which the appropriate Federal banking agency determines that the institution's exemption would create a significant risk to the Deposit Insurance Fund.

## Guidelines and Interpretations

- 3. Compliance by Holding Company Subsidiaries. Audited consolidated financial statements and other reports or notices required by this part which are submitted by a holding company for any subsidiary institution, should be accompanied by a cover letter identifying all subsidiary institutions subject to part 363 that are included in the holding company's submission. When submitting a Part 363 Annual Report, the cover letter should identify all subsidiary institutions subject to part 363 included in the consolidated financial statements and state whether the other annual report requirements (i.e., management's statement of responsibilities, management's assessment of compliance with designated safety and soundness laws and regulations, and, if applicable, management's assessment of the effectiveness of internal control over financial reporting and the independent public accountant's attestation report on management's internal control assessment) are being satisfied for these institutions at the holding company level or at the institution level. An institution filing holding company consolidated financial statements as permitted by §363.1(b)(1) also may report on changes in its independent public accountant on a holding company basis. An institution that does not meet the criteria in §363.1(b)(2) must satisfy the remaining provisions of this part on an individual institution basis and maintain its own audit committee. Subject to the criteria in §§363.1(b)(1) and (2), a multi-tiered holding company may satisfy all of the requirements of this part at the top-tier or any mid-tier holding company level.
- 4. *Comparable Services and Functions.* Services and functions will be considered "comparable" to those required by this part if the holding company:
- (a) Prepares reports used by the subsidiary institution to meet the requirements of this part;
- (b) Has an audit committee that meets the requirements of this part appropriate to its largest subsidiary institution; and
- (c) Prepares and submits management's assessment of compliance with the Designated Laws and Regulations defined in guideline 7A and, if applicable, management's assessment of the effectiveness of internal control over financial reporting based on information concerning the relevant activities and operations of those subsidiary institutions within the scope of the Rule.

(c) Financial reporting. For purposes of the management report requirement of § 363.2(b) and the internal control reporting requirement of § 363.3(b), "financial reporting," at a minimum, includes both financial statements prepared in accordance with generally accepted accounting principles for the insured depository institution or its holding company and financial statements prepared for regulatory reporting purposes. For recognition and measurement purposes, financial statements prepared for regulatory reporting purposes shall conform to generally accepted accounting principles and section 37 of the Federal Deposit Insurance Act.

- (d) *Definitions*. For purposes of this part, the following definitions apply:
- (1) AICPA means the American Institute of Certified Public Accountants.
- (2) *GAAP* means generally accepted accounting principles.
- (3) *PCAOB* means the Public Company Accounting Oversight Board.
- (4) Public company means an insured depository institution or other company that has a class of securities registered with the U.S. Securities and Exchange Commission or the appropriate Federal banking agency under Section 12 of the Securities Exchange Act of 1934 and nonpublic company means an insured depository institution or

# Guidelines and Interpretations

4A. Financial Statements Prepared for Regulatory Reporting Purposes. (a) As set forth in § 363.3(c)<sup>7</sup> of this part, "financial reporting," at a minimum, includes both financial statements prepared in accordance with generally accepted accounting principles for the insured depository institution or its holding company and financial statements prepared for regulatory reporting purposes. More specifically, financial statements prepared for regulatory reporting purposes include the schedules equivalent to the basic financial statements that are included in an insured depository institution's or its holding company's appropriate regulatory report (for example, Schedules RC, RI, and RI-A in the Consolidated Reports of Condition and Income (Call Report) for an insured bank; and Schedules SC and SO, and the Summary of Changes in Equity Capital section in Schedule SI in the Thrift Financial Report (TFR) for an insured thrift institution). For recognition and measurement purposes, financial statements prepared for regulatory reporting purposes shall conform to generally accepted accounting principles and section 37 of the Federal Deposit Insurance Act.

(b) Financial statements prepared for regulatory reporting purposes do not include regulatory reports prepared by a non-bank subsidiary of a holding company or an institution. For example, if a bank holding company or an insured depository institution owns an insured subsidiary, financial statements prepared for regulatory reporting purposes would not include any regulatory reports that the insurance subsidiary is required to submit to its appropriate insurance regulatory agency.

 $<sup>^7</sup>$  The reference to the definition of financial reporting in Guideline 4A is not correct. It should be  $\S$  363.1(c), not  $\S$  363.3(c).

# Guidelines and Interpretations

other company that does not meet the definition of a *public company*.

- (5) SEC means the U.S. Securities and Exchange Commission.
- (6) SOX means the Sarbanes—Oxley Act of 2002.

### §363.2 Annual reporting requirements

(a) Audited financial statements. Each insured depository institution shall prepare annual financial statements in accordance with GAAP, which shall be audited by an independent public accountant. The annual financial statements must reflect all material correcting adjustments necessary to conform with GAAP that were identified by the independent public accountant.

# Annual Reporting Requirements (§363.2)

5. Annual Financial Statements. Each institution (other than an insured branch of a foreign bank) should prepare comparative annual consolidated financial statements (balance sheets and statements of income, changes in equity capital, and cash flows, with accompanying footnote disclosures) in accordance with GAAP for each of its two most recent fiscal years. Statements for the earlier year may be presented on an unaudited basis if the institution was not subject to this part for that year and audited statements were not prepared.

5A. Institutions Merged Out of Existence. An institution that is merged out of existence after the end of its fiscal year, but before the deadline for filing its Part 363 Annual Report (120 days after the end of its fiscal year for an institution that is neither a public company nor a subsidiary of a public company that meets the criterion specified in § 363.1(b)(1), and 90 days after the end of its fiscal year for an institution that is a public company or a subsidiary of a public company that meets the criterion specified in § 363.1(b)(1)), is not required to file a Part 363 Annual Report for the last fiscal year of its existence.

- 6. Holding Company Statements. Subject to the criterion specified in § 363.1(b)(1), subsidiary institutions may file copies of their holding company's audited financial statements filed with the SEC or prepared for their FR Y-6 Annual Report under the Bank Holding Company Act of 1956 to satisfy the audited financial statements requirement of § 363.2(a).
- 7. Insured Branches of Foreign Banks. An insured branch of a foreign bank should satisfy the financial statements requirement by filing one of the following for each of its two most recent fiscal years:
- (a) Audited balance sheets, disclosing information about financial instruments with off-balance-sheet risk;
- (b) Schedules RAL and L of form FFIEC 002, prepared and audited on the basis of the instructions for its preparation; or

# Guidelines and Interpretations

- (c) With written approval of the appropriate Federal banking agency, consolidated financial statements of the parent bank.
- 7A. Compliance with Designated Laws and Regulations. The designated laws and regulations are the Federal laws and regulations concerning loans to insiders and the Federal and, if applicable, State laws and regulations concerning dividend restrictions (the Designated Laws and Regulations). Table 1 to this Appendix A lists the designated Federal laws and regulations pertaining to insider loans and dividend restrictions (but not the State laws and regulations pertaining to dividend restrictions) that are applicable to each type of institution.
- 8. Management Report. Management should perform its own investigation and review of compliance with the Designated Laws and Regulations and, if required, the effectiveness of internal control over financial reporting. Management should maintain records of its determinations and assessments until the next Federal safety and soundness examination, or such later date as specified by the FDIC or appropriate Federal banking agency. Management should provide in its assessment of the effectiveness of internal control over financial reporting, or supplementally, sufficient information to enable the accountant to report on its assertions. The management report of an insured branch of a foreign bank should be signed by the branch's managing official if the branch does not have a chief executive officer or a chief accounting or financial officer.
- 8A. Management's Reports on Internal Control over Financial Reporting under Part 363 and Section 404 of SOX. An institution with \$1 billion or more in total assets as of the beginning of its fiscal year that is subject to both part 363 and the SEC's rules implementing section 404 of SOX (as well as a public holding company permitted under the holding company exception in § 363.1(b)(2) to file an internal control report on behalf of one or more subsidiary institutions with \$1 billion or more in total assets) can choose either of the following two options for filing management's report on internal control over financial reporting.
- (i) Management can prepare two separate reports on the institution's or the holding company's internal control over financial reporting to satisfy the FDIC's part 363 requirements and the SEC's section 404 requirements; or

(b) Management report. Each insured depository institution annually shall prepare, as of the end of the institution's most recent fiscal year, a management report that must contain the following:

(1) A statement of management's responsibilities for preparing the institution's annual financial statements, for establishing and maintaining an adequate internal control structure and procedures for financial reporting, and for complying with laws and regulations relating to safety and soundness that are designated by the FDIC and the appropriate Federal banking agency;

- (ii) Management can prepare a single report on internal control over financial reporting provided that it satisfies all of the FDIC's part 363 requirements and all of the SEC's section 404 requirements.
- 8B. Internal Control Reports and Part 363 Annual Reports for Acquired Businesses. Generally, the FDIC expects management's and the related independent public accountant's report on an institution's internal control over financial reporting to include controls at an institution in its entirety, including all of its consolidated entities. However, it may not always be possible for management to conduct an assessment of the internal control over financial reporting of an acquired business in the period between the consummation date of the acquisition and the due date of management's internal control assessment.
- (a) In such instances, the acquired business's internal control structure and procedures for financial reporting may be excluded from management's assessment report and the accountant's attestation report on internal control over financial reporting. However, the FDIC expects management's assessment report to identify the acquired business, state that the acquired business is excluded, and indicate the significance of this business to the institution's consolidated financial statements. Notwithstanding management's exclusion of the acquired business's internal control from its assessment, management should disclose any material change to the institution's internal control over financial reporting due to the acquisition of this business. Also, management may not omit the assessment of the acquired business's internal control from more than one annual part 363 assessment report on internal control over financial reporting. When the acquired business's internal control over financial reporting is excluded from management's assessment, the independent public accountant may likewise exclude this acquired business's internal control over financial reporting from the accountant's evaluation of internal control over financial reporting.
- (b) If the acquired business is or has a consolidated subsidiary that is an insured depository institution subject to part 363 and the institution is not merged out of existence before the deadline for filing its Part 363 Annual Report (120 days after the end of its fiscal year for an institution that is neither a public company nor a subsidiary of a public company that meets the criterion specified

(2) An assessment by management of the insured depository institution's compliance with such laws and regulations during such fiscal year. The assessment must state management's conclusion as to whether the insured depository institution has complied with the designated safety and soundness laws and regulations during the fiscal year and disclose any noncompliance with these laws and regulations; and

- (3) For an insured depository institution with consolidated total assets of \$1 billion or more as of the beginning of such fiscal year, an assessment by management of the effectiveness of such internal control structure and procedures as of the end of such fiscal year that must include the following:
- (i) A statement identifying the internal control framework<sup>1</sup> used by management to evaluate the effectiveness of the insured depository institution's internal control over financial reporting;
- (ii) A statement that the assessment included controls over the preparation of regulatory financial statements in accordance with regulatory reporting instructions including identification of such regulatory reporting instructions; and

# Guidelines and Interpretations

in § 363.1(b)(1), and 90 days after the end of its fiscal year for an institution that is a public company or a subsidiary of public company that meets the criterion specified in § 363.1(b)(1)), the acquired institution must continue to comply with all of the applicable requirements of part 363, including filing its Part 363 Annual Report.

8C. Management's Disclosure of Noncompliance with the Designated Laws and Regulations. Management's disclosure of noncompliance, if any, with the Designated Laws and Regulations should separately indicate the number of instances or frequency of noncompliance with the Federal laws and regulations pertaining to insider loans and the Federal (and, if applicable, State) laws and regulations pertaining to dividend restrictions. The disclosure is not required to specifically identify by name the individuals (e.g., officers or directors) who were responsible for or were the subject of any such noncompliance. However, the disclosure should include appropriate qualitative and quantitative information to describe the nature, type, and severity of the noncompliance and the dollar amount of the insider loan(s) or dividend(s) involved. Similar instances of noncompliance may be aggregated as to number of instances and quantified as to the dollar amounts or the range of dollar amounts of insider loans and/or dividends for which noncompliance occurred. Management may also wish to describe any corrective actions taken in responses to the instances of noncompliance as well any controls or procedures that are being developed or that have been developed and implemented to prevent or detect and correct future instances of noncompliance on a timely basis.

- (iii) A statement expressing management's conclusion as to whether the insured depository institution's internal control over financial reporting is effective as of the end of its fiscal year. Management must disclose all material weaknesses in internal control over financial reporting, if any, that it has identified that have not been remediated prior to the insured depository institution's fiscal year-end. Management is precluded from concluding that the institution's internal control over financial reporting is effective if there are one or more material weaknesses.
- (c) Management report signatures. Subject to the criteria specified in § 363.1(b):
- (1) If the audited financial statements requirement specified in § 363.2(a) is satisfied at the insured depository institution level and the management report requirement specified in § 363.2(b) is satisfied in its entirety at the insured depository institution level, the management report must be signed by the chief executive officer and the chief accounting officer or chief financial officer of the insured depository institution:
- (2) If the audited financial statements requirement specified in § 363.2(a) is satisfied at the holding company level and the management report requirement specified in § 363.2(b) is satisfied in its entirety at the holding company level, the management report must be signed by the chief executive officer and the chief accounting officer or chief financial officer of the holding company; and
- (3) If the audited financial statements requirement specified in \$363.2(a) is satisfied at the holding company level and (i) the management report requirement specified in § 363.2(b) is satisfied in its entirety at the insured depository institution level or (ii) one or more of the components of the management report specified in § 363.2(b) is satisfied at the holding company level and the remaining components of the management report are satisfied at the insured depository institution level, the management report must be signed by the chief executive officers and the chief accounting officers or chief financial officers of both the holding company and the insured depository institution and the management report must clearly indicate the level (institution or holding company) at which each of its components is being satisfied.

# Guidelines and Interpretations

9. Safeguarding of Assets. "Safeguarding of assets," as the term relates to internal control policies and procedures regarding financial reporting, and which has precedent in accounting and auditing literature, should be encompassed in the management report and the independent public accountant's attestation discussed in guideline 18. Testing the existence of and compliance with internal controls on the management of assets, including loan underwriting and documentation, presents a reasonable implementation of section 36. The FDIC expects such internal controls to be encompassed by the assertion in the management report, but the term "safeguarding of assets" need not be specifically stated. The FDIC does not require the accountant to attest to the adequacy of safeguards, but does require the accountant to determine whether safeguarding policies exist.

10. Standards for Internal Control. The management of each insured depository institution with \$1 billion or more in total assets as of the beginning of its fiscal year should base its assessment of the effectiveness of the institution's internal control over financial reporting on a suitable, recognized control framework established by a body of experts that followed due-process procedures, including the broad distribution of the framework for public comment. In addition to being available to users of management's reports, a framework is suitable only when it:

- Is free from bias:
- Permits reasonably consistent qualitative and quantitative measurements of an institution's internal control over financial reporting;
- Is sufficiently complete so that those relevant factors that would alter a conclusion about the effectiveness of an institution's internal control over financial reporting are not omitted; and
- Is relevant to an evaluation of internal control over financial reporting.

In the United States, Internal Control—Integrated Framework, including its addendum on safeguarding assets, which was published by the Committee of Sponsoring Organizations of the Treadway Commission, and is known as the COSO report, provides a suitable and recognized framework for purposes of management's assessment. Other suitable frameworks have been published in other countries or may be developed in the future. Such other suitable frameworks may

be used by management and the institution's independent public accountant in assessments, attestations, and audits of internal control over financial reporting.<sup>2</sup>

11. Service Organizations. Although service organizations should be considered in determining if internal control over financial reporting is effective, an institution's independent public accountant, its management, and its audit committee should exercise independent judgment concerning that determination. Onsite reviews of service organizations may not be necessary to prepare the report required by the Rule, and the FDIC does not intend that the Rule establish any such requirement.

#### 12. [Reserved]

# Role of Independent Public Accountant (§363.3)

13. General Qualifications. To provide audit and attest services to insured depository institutions, an independent public accountant should be registered or licensed to practice as a public accountant, and be in good standing, under the laws of the State or other political subdivision of the United States in which the home office of the institution (or the insured branch of a foreign bank) is located. As required by Section 36(g) (3)(A)(i), the accountant must agree to provide copies of any working papers, policies, and procedures relating to services performed under this part.

#### 14. [Reserved]

- 15. Peer Review Guidelines. The following peer review guidelines are acceptable:
- (a) The external peer review should be conducted by an organization independent of the accountant or firm being reviewed, as frequently as is consistent with professional accounting practices;
- (b) The peer review (other than a PCAOB inspection) should be generally consistent with AICPA Peer Review Standards; and
- (c) The review should include, if available, at least one audit of an insured depository institution or consolidated depository institution holding company.
- 16. [Reserved]

(continued)

### §363.3 Independent public accountant

(a) Annual audit of financial statements. Each insured depository institution shall engage an independent public accountant to audit and report on its annual financial statements in accordance with generally accepted auditing standards or the PCAOB's auditing standards, if applicable, and Section 37 of the Federal Deposit Insurance Act (12 U.S.C. 1831n). The scope of the audit engagement shall be sufficient to permit such accountant to determine and report whether the financial statements are presented fairly and in accordance with GAAP.

- (b) Internal control over financial reporting. For each insured depository institution with total assets of \$1 billion or more at the beginning of the institution's fiscal year, the independent public accountant who audits the institution's financial statements shall examine, attest to, and report separately on, the assertion of management concerning the effectiveness of the institution's internal control structure and procedures for financial reporting. The attestation and report shall be made in accordance with generally accepted standards for attestation engagements or the PCAOB's auditing standards, if applicable. The accountant's report must not be dated prior to the date of the management report and management's assessment of the effectiveness of internal control over financial reporting. Notwithstanding the requirements set forth in applicable professional standards, the accountant's report must include the following:
- (1) A statement identifying the internal control framework used by the independent public accountant, which must be the same as the internal control framework used by management, to evaluate the effectiveness of the insured depository institution's internal control over financial reporting;
- (2) A statement that the independent public accountant's evaluation included controls over the preparation of regulatory financial statements in accordance with regulatory reporting instructions including identification of such regulatory reporting instructions; and
- (3) A statement expressing the independent public accountant's conclusion as to whether the insured depository institution's internal control over financial reporting is effective as of the end of its fiscal year. The report must disclose all material weaknesses in internal control over financial reporting that the independent public accountant has identified that have not been remediated prior to the insured

# Guidelines and Interpretations

- 17. Information to be Provided to the Independent Public Accountant. Attention is directed to section 36(h) which requires institutions to provide specified information to their accountants. An institution also should provide its accountant with copies of any notice that the institution's capital category is being changed or reclassified under Section 38 of the FDI Act, and any correspondence from the appropriate Federal banking agency concerning compliance with this part.
- 18. Attestation Report and Management Letters. The independent public accountant should provide the institution with any management letter and, if applicable, an internal control attestation report, (as required by section 36(c)(1)) at the conclusion of the audit. The independent public accountant's attestation report on internal control over financial reporting must specifically include a statement as to regulatory reporting. If a holding company subsidiary relies on its holding company management report to satisfy the Part 363 Annual Report requirements, the accountant may attest to and report on the management's assertions in one report, without reporting separately on each subsidiary covered by the Rule. The FDIC has determined that management letters are exempt from public disclosure.

depository institution's fiscal year-end. The independent public account is precluded from concluding that the insured depository institution's internal control over financial reporting is effective if there are one or more material weaknesses.

# Guidelines and Interpretations

- 18A. Internal Control Attestation Standards for Independent Auditors. (a) § 363.3(b) provides that the independent public accountant's attestation and report on management's assertion concerning the effectiveness of an institution's internal control structure and procedures for financial reporting shall be made in accordance with generally accepted standards for attestation engagements or the PCAOB's auditing standards, if applicable. The standards that should be followed by the institution's independent public accountant concerning internal control over financial reporting for institutions with \$1 billion or more in total assets can be summarized as follows:
- (1) For an insured institution that is neither a public company nor a subsidiary of a public company, its independent public accountant need only follow the AICPA's attestation standards.
- (2) For an insured institution that is a public company that is required to comply with the auditor attestation requirement of section 404 of SOX, its independent public accountant should follow the PCAOB's auditing standards.
- (3) For an insured institution that is a public company but is not required to comply with the auditor attestation requirement of section 404 of SOX, its independent public accountant is not required to follow the PCAOB's auditing standards. In this case, the accountant need only follow the AICPA's attestation standards.
- (4) For an insured institution that is a subsidiary of a public company that is required to comply with the auditor attestation requirement of section 404 of SOX, but is not itself a public company, the institution and its independent public accountant have flexibility in complying with the internal control requirements of Part 363. If the conditions specified in § 363.1(b) (2) are met, management and the independent public accountant may choose to report on internal control over financial reporting at the consolidated holding company level. In this situation, the independent public accountant's work would be performed for the public company in accordance with the PCAOB's auditing standards.

# Guidelines and Interpretations

Alternatively, the institution may choose to comply with the internal control reporting requirements of Part 363 at the institution level and its independent public accountant could follow the AICPA's attestation standards

- (b) If an independent public accountant need only follow the AICPA's attestation standards, the accountant and the insured institution may instead agree to have the internal control attestation performed under the PCAOB's auditing standards.
- 19. Reviews with Audit Committee and Management. The independent public accountant should meet with the institution's audit committee to review the accountant's reports required by this part before they are filed. It also may be appropriate for the accountant to review its findings with the institution's board of directors and management.
- 20. Notice of Termination. The notice required by §363.3(c) should state whether the independent public accountant agrees with the assertions contained in any notice filed by the institution under §363.4(d), and whether the institution's notice discloses all relevant reasons for the accountant's termination. Subject to the criterion specified in § 363.1(b)(1) regarding compliance with the audited financial statements requirement at the holding company level, the independent public accountant for an insured depository institution that is a public company and files reports with its appropriate Federal banking agency, or is a subsidiary of a public company that files reports with the SEC, may submit the letter it furnished to management to be filed with the institution's or the holding company's current report (e.g., SEC Form 8-K) concerning a change in accountant to satisfy the notice requirements of § 363.3(c). Alternatively, if the independent public accountant confirms that management has filed a current report (e.g., SEC Form 8-K) concerning a change in accountant that satisfies the notice requirements of § 363.4(d) and includes an independent public accountant's letter that satisfies the requirements of § 363.3(c), the independent public accountant may rely on the current report (e.g., SEC Form 8-K) filed with the FDIC by management concerning a change in accountant to satisfy the notice requirements of § 363.3(c).
- 21. Reliance on Internal Auditors. Nothing in this part or this Appendix is intended to preclude the ability of the independent public accountant to rely on the work of an institution's internal auditor.

(c) Notice by accountant of termination of services. An independent public accountant performing an audit under this part who ceases to be the accountant for an insured depository institution shall notify the FDIC, the appropriate Federal banking agency, and any appropriate State bank supervisor in writing of such termination within 15 days after the occurrence of such event, and set forth in reasonable detail the reasons for such termination. The written notice shall be filed at the place identified in § 363.4(f).

- (d) Communications with audit committee. In addition to the requirements for communications with audit committees set forth in applicable professional standards, the independent public accountant must report the following on a timely basis to the audit committee:
- (1) All critical accounting policies and practices to be used by the insured depository institution,
- (2) All alternative accounting treatments within GAAP for policies and practices related to material items that the independent public accountant has discussed with management, including the ramifications of the use of such alternative disclosures and treatments, and the treatment preferred by the independent public accountant, and
- (3) Other written communications the independent public accountant has provided to management, such as a management letter or schedule of unadjusted differences.
- (e) Retention of working papers. The independent public accountant must retain the working papers related to the audit of the insured depository institution's financial statements and, if applicable, the evaluation of the institution's internal control over financial reporting for seven years from the report release date, unless a longer period of time is required by law.
- (f) Independence. The independent public accountant must comply with the independence standards and interpretations of the AICPA, the SEC, and the PCAOB. To the extent that any of the rules within any one of these independence standards (AICPA, SEC, and PCAOB) is more or less restrictive than the corresponding rule in the other independence standards, the independent public accountant must comply with the more restrictive rule.
- (g) Peer reviews and inspection reports.
- (1) Prior to commencing any services for an insured depository institution under this part, the independent public accountant must have received a peer review, or be enrolled in a peer review program, that meets acceptable guidelines. Acceptable peer reviews include peer reviews performed in accordance with the AICPA's Peer Review Standards and inspections conducted by the PCAOB.

(2) Within 15 days of receiving notification that a peer review has been accepted or a PCAOB inspection report has been issued, or before commencing any audit under this part, whichever is earlier, the independent public accountant must file two copies of the most recent peer review report and the public portion of the most recent PCAOB inspection report, if any, accompanied by any letters of comments, response, and acceptance, with the FDIC, Accounting and Securities Disclosure Section, 550 17th Street, NW, Washington, DC 20429, if the report has not already been filed. The peer review reports and the public portions of the PCAOB inspection reports will be made available for public inspection by the FDIC.

(3) Within 15 days of the PCAOB making public a previously nonpublic portion of an inspection report, the independent public accountant must file two copies of the previously nonpublic portion of the inspection report with the FDIC, Accounting and Securities Disclosure Section, 550 17th Street, NW, Washington, DC 20429. Such previously nonpublic portion of the PCAOB inspection report will be made available for public inspection by the FDIC.

### §363.4 Filing and notice requirements

- (a) Part 363 Annual Report.
- (1) Each insured depository institution shall file with each of the FDIC, the appropriate Federal banking agency, and any appropriate State bank supervisor, two copies of its Part 363 Annual Report. A Part 363 Annual Report must contain audited comparative annual financial statements, the independent public accountant's report thereon, a management report, and, if applicable, the independent public accountant's attestation report on management's assessment concerning the institution's internal control structure and procedures for financial reporting as required by §§ 363.2(a),363.3(a), 363.2(b), and 363.3(b) respectively.
- (2) Subject to the criteria specified in § 363.1(b), each insured depository institution with consolidated total assets of less than \$1 billion as of the beginning of its fiscal year that is required to file, or whose parent holding company is required to file, management's assessment of the effectiveness of internal control over financial reporting with the SEC or the appropriate Federal banking agency in accordance

Filing and Notice Requirements (§363.4)

22. [Reserved]

with section 404 of SOX must submit a copy of such assessment to the FDIC, the appropriate Federal banking agency, and any appropriate State bank supervisor with its Part 363 Annual Report as additional information. This assessment will not be considered part of the institution's Part 363 Annual Report.

- (3)(i) Each insured depository institution that is neither a public company nor a subsidiary of a public company that meets the criterion specified in § 363.1(b)(1) shall file its Part 363 Annual Report within 120 days after the end of its fiscal year. (ii) Each insured depository institution that is a public company or a subsidiary of a public company that meets the criterion specified in § 363.1(b)(1) shall file its Part 363 Annual Report within 90 days after the end of its fiscal year.
- (b) Public availability. Except for the annual report in paragraph (a)(1) of this section and the peer reviews and inspection reports in § 363.3(g), which shall be available for public inspection, the FDIC has determined that all other reports and notifications required by this part are exempt from public disclosure by the FDIC.

# Guidelines and Interpretations

- 23. Notification of Late Filing. (a) An institution's submission of a written notice of late filing does not cure the requirement to timely file the Part 363 Annual Report or other reports or notices required by § 363.4. An institution's failure to timely file is considered an apparent violation of Part 363.
- (b) If the late filing notice submitted pursuant to § 363.4(e) relates only to a portion of a Part 363 Annual Report or any other report or notice, the insured depository institution should file the other components of the report or notice within the prescribed filing period together with a cover letter that indicates which components of its Part 363 Annual Report or other report or notice are omitted. An institution may combine the written late filing notice and the cover letter into a single notice that is submitted together with the other components of the report or notice that are being timely filed.

# Guidelines and Interpretations

- 24. Public Availability. Each institution's Part 363 Annual Report should be available for public inspection at its main and branch offices no later than 15 days after it is filed with the FDIC. Alternatively, an institution may elect to mail one copy of its Part 363 Annual Report to any person who requests it. The Part 363 Annual Report should remain available to the public until the Part 363 Annual Report for the next year is available. An institution may use its Part 363 Annual Report under this part to meet the annual disclosure statement required by 12 CFR 350.3, if the institution satisfies all other requirements of 12 CFR Part 350.
- 25. [Reserved]
- (c) Independent public accountant's letters and reports. Except for the independent public accountant's reports that are included in its Part 363 Annual Report, each insured depository institution shall file with the FDIC, the appropriate Federal banking agency, and any appropriate State bank supervisor, a copy of any management letter or other report issued by its independent public accountant with respect to such institution and the services provided by such accountant pursuant to this part within 15 days after receipt. Such reports include, but are not limited to:
- (1) Any written communication regarding matters that are required to be communicated to the audit committee (for example, critical accounting policies, alternative accounting treatments discussed with management, and any schedule of unadjusted differences),
- (2) Any written communication of significant deficiencies and material weaknesses in internal control required by the AICPA's or the PCAOB's auditing standards;
- (3) For institutions with total assets of less than \$1 billion as of the beginning of their fiscal year that are public companies or subsidiaries of public companies that meet the criterion specified in \$ 363.1(b)(1), any independent public accountant's report on the audit of internal control over financial reporting required by section 404 of SOX and the PCAOB's auditing standards; and
- (4) For all institutions that are public companies or subsidiaries of public companies that meet the criterion specified in § 363.1(b)(1), any independent public accountant's written communication of all deficiencies in internal control over financial reporting that are of a lesser magnitude than significant deficiencies required by the PCAOB's auditing standards.

(d) Notice of engagement or change of accountants. Each insured depository institution shall provide, within 15 days, after the occurrence of any such event, written notice to the FDIC, the appropriate Federal banking agency, and any appropriate State bank supervisor of the engagement of an independent public accountant, or the resignation or dismissal of the independent public accountant previously engaged. The notice shall include a statement of the reasons for any such resignation or dismissal in reasonable detail.

- (e) Notification of late filing. No extensions of time for filing reports required by § 363.4 shall be granted. An insured depository institution that is unable to timely file all or any portion of its Part 363 Annual Report or any other report or notice required by § 363.4 shall submit a written notice of late filing to the FDIC, the appropriate Federal banking agency, and any appropriate State bank supervisor. The notice shall disclose the institution's inability to timely file all or specified portions of its Part 363 Annual Report or any other report or notice and the reasons therefore in reasonable detail. The late filing notice shall also state the date by which the report or notice will be filed. The written notice shall be filed on or before the deadline for filing the Part 363 Annual Report or any other report or notice, as appropriate.
- (f) *Place for filing*. The Part 363 Annual Report, any written notification of late filing, and any other report or notice required by § 363.4 should be filed as follows:
- (1) FDIC: Appropriate FDIC Regional or Area Office (Division of Supervision and Consumer Protection), i.e., the FDIC regional or area office in the FDIC region or area that is responsible for monitoring the institution or, in the case of a subsidiary institution of a holding company, the consolidated company. A filing made on behalf of several covered institutions owned by the same parent holding company should be accompanied by a transmittal letter identifying all of the institutions covered.

# Guidelines and Interpretations

26. Notices Concerning Accountants. With respect to any selection, change, or termination of an independent public accountant, an institution's management and audit committee should be familiar with the notice requirements in § 363.4(d) and guideline 20, and management should send a copy of any notice required under § 363.4(d) to the independent public accountant when it is filed with the FDIC. An insured depository institution that is a public company and files reports required under the Federal securities laws with its appropriate Federal banking agency, or is a subsidiary of a public company that files such reports with the SEC, may use its current report (e.g., SEC Form 8-K) concerning a change in accountant to satisfy the notice requirements of § 363.4(d) subject to the criterion of § 363.1(b)(1) regarding compliance with the audited financial statements requirement at the holding company level.

# Guidelines and Interpretations

- (2) Office of the Comptroller of the Currency (OCC): Appropriate OCC Supervisory Office.
- (3) Federal Reserve: Appropriate Federal Reserve Bank.
- (4) Office of Thrift Supervision (OTS): Appropriate OTS District Office.
- (5) State bank supervisor: The filing office of the appropriate State bank supervisor.

#### §363.5 Audit committees

(a) Composition and duties. Each insured depository institution shall establish an audit committee of its board of directors, the composition of which complies with paragraphs (a)(1), (2), and (3) of this section. The duties of the audit committee shall include the appointment, compensation, and oversight of the independent public accountant who performs services required under this part, and reviewing with management and the independent public accountant the basis for the reports issued under this part.

(1) Each insured depository institution with total assets of \$1 billion or more as of the beginning of its fiscal year shall establish an independent audit committee of its board of directors, the members of which shall be outside directors who are independent of management of the institution.

# Audit Committees (§363.5)

27. Composition. The board of directors of each institution should determine whether each existing or potential audit committee member meets the requirements of section 36 and this part. To do so, the board of directors should maintain an approved set of written criteria for determining whether a director who is to serve on the audit committee is an outside director (as defined in § 363.5(a)(3)) and is independent of management. At least annually, the board of each institution should determine whether each existing or potential audit committee member is an outside director. In addition, at least annually, the board of an institution with \$1 billion or more in total assets as of the beginning of its fiscal year should determine whether all existing and potential audit committee members are "independent of management of the institution" and the board of an institution with total assets of \$500 million or more but less than \$1 billion as of the beginning of its fiscal year should determine whether the majority of all existing and potential audit committee members are "independent of management of the institution." The minutes of the board of directors should contain the results of and the basis for its determinations with respect to each existing and potential audit committee member. Because an insured branch of a foreign bank does not have a separate board of directors, the FDIC will not apply the audit committee requirements to such branch. However, any such branch is encouraged to make a reasonable good faith effort to see that similar duties are performed by persons whose experience is generally consistent with the Rule's requirements for an institution the size of the insured branch.

- (2) Each insured depository institution with total assets of \$500 million or more but less than \$1 billion as of the beginning of its fiscal year shall establish an audit committee of its board of directors, the members of which shall be outside directors, the majority of whom shall be independent of management of the institution. The appropriate Federal banking agency may, by order or regulation, permit the audit committee of such an insured depository institution to be made up of less than a majority of outside directors who are independent of management, if the agency determines that the institution has encountered hardships in retaining and recruiting a sufficient number of competent outside directors to serve on the audit committee of the institution.
- (3) An outside director is a director who is not, and within the preceding fiscal year has not been, an officer or employee of the institution or any affiliate of the institution

28. "Independent of Management" Considerations. It is not possible to anticipate, or explicitly provide for, all circumstances that might signal potential conflicts of interest in, or that might bear on, an outside director's relationship to an insured depository institution and whether the outside director should be deemed "independent of management." When assessing an outside director's relationship with an institution, the board of directors should consider the issue not merely from the standpoint of the director himself or herself, but also from the standpoint of persons or organizations with which the director has an affiliation. These relationships can include, but are not limited to, commercial, banking, consulting, charitable. and family relationships. To assist boards of directors in fulfilling their responsibility to determine whether existing and potential members of the audit committee are "independent of management," paragraphs (a) through (d) of this guideline provide guidance for making this determination.

# Guidelines and Interpretations

- (a) If an outside director, either directly or indirectly, owns or controls, or has owned or controlled within the preceding fiscal year, 10 percent or more of any outstanding class of voting securities of the institution, the institution's board of directors should determine, and document its basis and rationale for such determination, whether such ownership of voting securities would interfere with the outside director's exercise of independent judgment in carrying out the responsibilities of an audit committee member, including the ability to evaluate objectively the propriety of management's accounting, internal control, and reporting policies and practices. Notwithstanding the criteria set forth in paragraphs (b), (c), and (d) of this guideline, if the board of directors determines that such ownership of voting securities would interfere with the outside director's exercise of independent judgment, the outside director will not be considered "independent of management."
- (b) The following list sets forth additional criteria that, at a minimum, a board of directors should consider when determining whether an outside director is "independent of management." The board of directors may conclude that additional criteria are also relevant to this determination in light of the particular circumstances of its institution. Accordingly, an outside director will not be considered "independent of management" if:
- (1) The director serves, or has served within the last three years, as a consultant, advisor, promoter, underwriter, legal counsel, or trustee of or to the institution or its affiliates.
- (2) The director has been, within the last three years, an employee of the institution or any of its affiliates or an immediate family member is, or has been within the last three years, an executive officer of the institution or any of its affiliates.
- (3) The director has participated in the preparation of the financial statements of the institution or any of its affiliates at any time during the last three years.
- (4) The director has received, or has an immediate family member who has received, during any twelve-month period within the last three years, more than \$100,000 in direct and indirect compensation from the institution, its subsidiaries, and its affiliates for consulting, advisory, or other services other than director and committee fees and pension or other forms of deferred compensation for prior service (provided such compensation is not contingent in any way on

continued service). Direct compensation also would not include compensation received by the director for former service as an interim chairman or interim chief executive officer.

- (5) The director or an immediate family member is a current partner of a firm that performs internal or external auditing services for the institution or any of its affiliates; the director is a current employee of such a firm; the director has an immediate family member who is a current employee of such a firm and who participates in the firm's audit, assurance, or tax compliance practice; or the director or an immediate family member was within the last three years (but no longer is) a partner or employee of such a firm and personally worked on the audit of the insured depository institution or any of its affiliates within that time.
- (6) The director or an immediate family member is, or has been within the last three years, employed as an executive officer of another entity where any of the present executive officers of the institution or any of its affiliates at the same time serves or served on that entity's compensation committee.
- (7) The director is a current employee, or an immediate family member is a current executive officer, of an entity that has made payments to, or received payments from, the institution or any of its affiliates for property or services in an amount which, in any of the last three fiscal years, exceeds the greater of \$200 thousand, or 5 percent of such entity's consolidated gross revenues. This would include payments made by the institution or any of its affiliates to not-for-profit entities where the director is an executive officer or where an immediate family member of the director is an executive officer.
- (8) For purposes of paragraph (b) of this guideline:
- (i) An "immediate family member" includes a person's spouse, parents, children, siblings, mothers- and fathers-in-law, sons- and daughters-in-laws, brothers- and sisters-inlaw, and anyone (other than domestic employees) who shares such person's home.
- (ii) The term affiliate of, or a person affiliated with, a specified person, means a person or entity that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, the person specified.

- (iii) The term indirect compensation for consulting, advisory, or other services includes the acceptance of a fee for such services by a director's immediate family member or by an organization in which the director is a partner or principal that provides accounting, consulting, legal, investment banking, or financial advisory services to the institution, any of its subsidiaries, or any of its affiliates.
- (iv) The terms direct and indirect compensation and payments do not include payments such as dividends arising solely from investments in the institution's equity securities, provided the same per share amounts are paid to all shareholders of that class; interest income from investments in the institution's deposit accounts and debt securities; loans from the institution that conform to all regulatory requirements applicable to such loans except that interest payments or other fees paid in association with such loans would be considered payments; and payments under non-discretionary charitable contribution matching programs.
- (c) An insured depository institution that is a public company and a listed issuer (as defined in Rule 10A-3 of the Securities Exchange Act of 1934 (Exchange Act)), or is a subsidiary of a public company that meets the criterion specified in § 363.1(b)(1) and is a listed issuer, may choose to use the definition of audit committee member independence set forth in the listing standards applicable to the public institution or its public company parent for purposes of determining whether an outside director is "independent of management."
- (d) All other insured depository institutions may choose to use the definition of audit committee member independence set forth in the listing standards of a national securities exchange that is registered with the SEC pursuant to section 6 of the Exchange Act or a national securities association that is registered with the SEC pursuant to section 15A(a) of the Exchange Act for purposes of determining whether an outside director is "independent of management."

29. [Reserved]

- 30. Holding Company Audit Committees. (a) When an insured depository institution satisfies the requirements for the holding company exception specified in §§ 363.1(b) (1) and (2), the audit committee requirement of this part may be satisfied by the audit committee of the top-tier or any mid-tier holding company. Members of the audit committee of the holding company should meet all the membership requirements applicable to the largest subsidiary depository institution subject to Part 363 and should perform all the duties of the audit committee of a subsidiary institution subject to Part 363, even if the holding company directors are not directors of the institution.
- (b) When an insured depository institution subsidiary with total assets of \$1 billion or more as of the beginning of its fiscal year does not meet the requirements for the holding company exception specified in §§ 363.1(b)(1) and (2) or maintains its own separate audit committee to satisfy the requirements of this part, the members of the audit committee of the top-tier or any mid-tier holding company may serve on the audit committee of the subsidiary institution if they are otherwise independent of management of the subsidiary institution, and, if applicable, meet any other requirements for a large subsidiary institution covered by this part.
- (c) When an insured depository institution with total assets of \$500 million or more but less than \$1 billion as of the beginning of its fiscal year does not meet the requirements for the holding company exception specified in §\$ 363.1(b)(1) and (2) or maintains its own separate audit committee to satisfy the requirements of this part, the members of the audit committee of the top-tier or any midtier holding company may serve on the audit committee of the subsidiary institution provided a majority of the institution's audit committee members are independent of management of the subsidiary institution.
- (d) Officers and employees of a top-tier or any mid-tier holding company may not serve on the audit committee of a subsidiary institution subject to Part 363.

- 31. Duties. The audit committee should perform all duties determined by the institution's board of directors and it should maintain minutes and other relevant records of its meetings and decisions. The duties of the audit committee should be appropriate to the size of the institution and the complexity of its operations, and at a minimum, should include the appointment, compensation, and oversight of the independent public accountant; reviewing with management and the independent public accountant the basis for their respective reports issued under §§363.2(a) and (b) and §§363.3(a) and (b); reviewing and satisfying itself as to the independent public accountant's compliance with the required qualifications for independent public accountants set forth in §§ 363.3(f) and (g) and guidelines 13 through 16; ensuring that audit engagement letters comply with the provisions of § 363.5(c) before engaging an independent public accountant; being familiar with the notice requirements in § 363.4(d) and guideline 20 regarding the selection, change, or termination of an independent public accountant; and ensuring that management sends a copy of any notice required under § 363.4(d) to the independent public accountant when it is filed with the FDIC. Appropriate additional duties could include:
- (a) Reviewing with management and the independent public accountant the scope of services required by the audit, significant accounting policies, and audit conclusions regarding significant accounting estimates;
- (b) Reviewing with management and the accountant their assessments of the effectiveness of internal control over financial reporting, and the resolution of identified material weaknesses and significant deficiencies in internal control over financial reporting, including the prevention or detection of management override or compromise of the internal control system;
- (c) Reviewing with management the institution's compliance with the Designated Laws and Regulations identified in guideline 7A;
- (d) Discussing with management and the independent public accountant any significant disagreements between management and the independent public accountant; and
- (e) Overseeing the internal audit function.

(b) Committees of large institutions. The audit committee of any insured depository institution with total assets of more than \$3 billion, as of the beginning of its fiscal year, shall include members with banking or related financial management expertise, have access to its own outside counsel, and not include any large customers of the institution. If a large institution is a subsidiary of a holding company and relies on the audit committee of the holding company to comply with this rule, the holding company's audit committee shall not include any members who are large customers of the subsidiary institution.

# Guidelines and Interpretations

32. Banking or Related Financial Management Expertise. At least two members of the audit committee of a large institution shall have "banking or related financial management expertise" as required by Section 36(g)(1)(C)(i). This determination is to be made by the board of directors of the insured depository institution. A person will be considered to have such required expertise if the person has significant executive, professional, educational, or regulatory experience in financial, auditing, accounting, or banking matters as determined by the board of directors. Significant experience as an officer or member of the board of directors or audit committee of a financial services company would satisfy these criteria. A person who has the attributes of an "audit committee financial expert" as set forth in the SEC's rules would also satisfy these criteria.

33. Large Customers. Any individual or entity (including a controlling person of any such entity) which, in the determination of the board of directors, has such significant direct or indirect credit or other relationships with the institution, the termination of which likely would materially and adversely affect the institution's financial condition or results of operations, should be considered a "large customer" for purposes of §363.5(b).

34. Access to Counsel. The audit committee should be able to retain counsel at its discretion without prior permission of the institution's board of directors or its management. Section 36 does not preclude advice from the institution's internal counsel or regular outside counsel. It also does not require retaining or consulting counsel, but if the committee elects to do either, it also may elect to consider issues affecting the counsel's independence. Such issues would include whether to retain or consult only counsel not concurrently representing the institution or any affiliate, and whether to place limitations on any counsel representing the institution concerning matters in which such counsel previously participated personally and substantially as outside counsel to the committee.

# Guidelines and Interpretations

- 35. Transition Period for Forming and Restructuring Audit Committees.
- (a) When an insured depository institution's total assets as of the beginning of its fiscal year are \$500 million or more for the first time and it thereby becomes subject to Part 363, no regulatory action will be taken if the institution (1) develops and approves a set of written criteria for determining whether a director who is to serve on the audit committee is an outside director and is independent of management and (2) forms or restructures its audit committee to comply with § 363.5(a)(2) by the end of the fiscal year.
- (b) When an insured depository institution's total assets as of the beginning of its fiscal year are \$1 billion or more for the first time, no regulatory action will be taken if the institution forms or restructures its audit committee to comply with  $\S$  363.5(a)(1) by the end of that fiscal year, provided that the composition of its audit committee meets the requirements specified in  $\S$  363.5(a)(2) at the beginning of that fiscal year, if such requirements were applicable.
- (c) When an insured depository institution's total assets as of the beginning of its fiscal year are \$3 billion or more for the first time, no regulatory action will be taken if the institution forms or restructures its audit committee to comply with \$363.5(b) by the end of that fiscal year, provided that the composition of its audit committee meets the requirements specified in \$363.5(a)(1) at the beginning of that fiscal year, if such requirements were applicable.
- (c) Independent public accountant engagement letters. (1) In performing its duties with respect to the appointment of the institution's independent public accountant, the audit committee shall ensure that engagement letters and any related agreements with the independent public accountant for services to be performed under this part do not contain any limitations of liability provisions that:
- (i) Indemnify the independent public accountant against claims made by third parties:
- (ii) Hold harmless or release the independent public accountant from liability for claims or potential claims that might be asserted by the client insured depository institution, other than claims for punitive damages; or

#### Regulation

- Guidelines and Interpretations
- (iii) Limit the remedies available to the client insured depository institution.
- (2) Alternative dispute resolution agreements and jury trial waiver provisions are not precluded from engagement letters provided that they do not incorporate any limitations of liability provisions set forth in paragraph (c)(1) of this section.

#### Other

36. Modifications of Guidelines. The FDIC's Board of Directors has delegated to the Director of the FDIC's Division of Supervision and Consumer Protection authority to make and publish in the Federal Register minor technical amendments to the Guidelines in this Appendix and the guidance and illustrative reports in Appendix B, in consultation with the other appropriate federal banking agencies, to reflect the practical experience gained from implementation of this part. It is not anticipated any such modification would be effective until affected institutions have been given reasonable advance notice of the modification. Any material modification or amendment will be subject to review and approval of the FDIC Board of Directors.

<sup>&</sup>lt;sup>1</sup> For example, in the United States, the Committee of Sponsoring Organizations (COSO) of the Treadway Commission has published *Internal Control—Integrated Framework*, including an addendum on safeguarding assets. Known as the COSO report, this publication provides a suitable and available framework for purposes of management's assessment.

<sup>&</sup>lt;sup>2</sup> It is management's responsibility to establish policies concerning underwriting and asset management and to make credit decisions. The auditor's role is to test compliance with management's policies relating to financial reporting.

#### TABLE 1 TO APPENDIX A

#### ${\bf Designated\ Federal\ Laws\ and\ Regulations\ Applicable\ to:}$

		National Banks	State Member Banks	State Non- member Banks	Savings Associations
Insid	ler Loans—Parts and/or Sec	tions of Title	e 12 of the U	Inited State	es Code
375a	Loans to Executive Officers of Banks	$\checkmark$	<b>√</b>	( <b>A</b> )	( <b>A</b> )
375b	Extensions of Credit to Executive Officers, Directors, and Principal Shareholders of Banks	√	√	( <b>A</b> )	( <b>A</b> )
1468(b)	Extensions of Credit to Executive Officers, Directors, and Principal Shareholders				$\checkmark$
1828(j)(2)	Extensions of Credit to Officers, Directors, and Principal Shareholders			$\checkmark$	
1828(j)(3)(B)	Extensions of Credit to Officers, Directors, and Principal Shareholders	<b>(B</b> )		( <b>C</b> )	
P	arts and/or Sections of Title	e 12 of the Co	ode of Feder	al Regulat	ions
31	Extensions of Credit to Insiders	$\checkmark$			
32	Lending Limits	√			
215	Loans to Executive Officers, Directors, and Principal Shareholders of Member Banks	√	√	<b>(D</b> )	<b>(E</b> )
337.3	Limits on Extensions of Credit to Executive Officers, Directors, and Principal Shareholders of Insured Nonmember Banks			$\checkmark$	
563.43	Loans by Savings Associations to Their Executive Officers, Directors, and Principal Shareholders				$\checkmark$
Dividend	l Restrictions—Parts and/or	Sections of	Title 12 of t	he United	States Code
56	Prohibition on Withdrawal of Capital and Unearned Dividends	√	√		
60	Dividends and Surplus Fund	√	√		
1467a(f)	Declaration of Dividends				$\checkmark$
1831o(d)(1)	Prompt Corrective Action—Capital Distributions Restricted	√	$\checkmark$	√	$\checkmark$

State

#### TABLE 1 TO APPENDIX A—continued

#### Designated Federal Laws and Regulations Applicable to:

		National Banks	State Member Banks	Non- member Banks	Savings Associations
Pa	rts and/or Sections of Title	$\frac{12 \text{ of the } C_0}{2}$	ode of Feder	ral Regulat	ions
5 Subpart E	Payment of Dividends	$\checkmark$			
6.6	Prompt Corrective Action—Restrictions on Undercapitalized Institutions	$\checkmark$			
208.5	Dividends and Other Distributions		<b>√</b>		
208.45	Prompt Corrective Action—Restrictions on Undercapitalized Institutions		√		
325.105 or 324.403, as applicable	Prompt Corrective Action—Restrictions on Undercapitalized Institutions			√	
563 Subpart E	Capital Distributions				√
565.6	Prompt Corrective Action—Restrictions on Undercapitalized Institutions				$\checkmark$

 $<sup>^{(</sup>A)}$  Subsections (g) and (h) of section 22 of the Federal Reserve Act [12 U.S.C. 375a, 375b].

 $<sup>^{(\</sup>mbox{\footnotesize B})}$  Applies only to insured Federal branches of foreign banks.

 $<sup>^{(</sup>C)}$  Applies only to insured State branches of foreign banks.

<sup>(</sup>D) See 12 CFR 337.3. (E) See 12 CFR 563.43.

#### Appendix B

#### Regulatory Reporting Matters—Interpretation and Reporting Related to U.S. GAAP<sup>1</sup>

This appendix is nonauthoritative and is included for informational purposes only.

[See table on following page.]

©2016, AICPA AAG-DEP APP B

<sup>&</sup>lt;sup>1</sup> Every national bank and saving and loan association, state member bank, state chartered saving and loan association, and insured state nonmember bank is required to file Federal Financial Institutions Examination Council (FFIEC) Consolidated Reports of Condition and Income, commonly referred to as the Call Report. Every federally insured credit union is required to file the National Credit Union Administration (NCUA) 5300 Call Report. Call Reports (for example, FFIEC and NCUA) present an institution's financial condition and results of operations on a consolidated basis in accordance with U.S. generally accepted accounting principles (GAAP). Regulatory reporting topics noted herein are consistent with acceptable practices under GAAP. The Call Report explains certain specific reporting guidance in greater detail. Information may often be found in the appropriate entries in the glossary section of the regulatory report instructions or, in more detail, in the GAAP standards. Financial institutions are encouraged to discuss specific events and transactions not covered by GAAP or the guidance in the regulatory report instructions with their primary supervisory agency for more technical detail on the application of the GAAP accounting standards. This appendix serves as an aid in specific selected areas and is not intended to be a comprehensive discussion of the principles of bank accounting or reporting.

# U.S. Generally Accepted Accounting Principles (GAAP)

Description

Description

References

Regulatory Reporting Matters

# I. RECOGNITION AND MEASUREMENT

## Interest Income Practices for Loans and Other Assets: Accrual and Nonaccrual Matters

Interest income on loans that are not impaired should be accrued and credited to interest income as it is earned. However, U.S. GAAP does not provide specific criteria for when a loan should be placed on nonaccrual.

Regulatory guidance provides a general rule to ensure an institution's net income is not overstated by defining that loans (and other assets) should be placed on nonaccrual (1) when the loan (or asset) is maintained on a cash basis because of deterioration in the financial condition of the or interest is not expected, or (3) when principal or interest has been in default for a period of 90 borrower, (2) when payment in full of principal days or more unless the asset is both well secured FASB Accounting cation (ASC) 310-Standards Codifi-

crual status, such as loans and other assets that have been acquired at a discount and consumer loans. Nevertheless, consistent with the objective to not overstate income, financial institutions Certain other loans need not be placed on nonacshould use alternative methods of evaluation to ensure that this objective is met and in the process of collection.

solidated Reports of

Condition and Income

(Call Report).

(FFIEC's) Instructions Financial Institutions Examination Council's solidated Reports of and Schedule RC-N of the FFIEC Con-Refer to the entry nonaccrual status in tion of the Federal for Preparation of Con-Condition and Income, "Glossarv" References

## Regulatory Reporting Matters

# U.S. Generally Accepted Accounting Principles (GAAP)

Description

### References Income Recognition and Restoration Matters Description

FASB ASC 310-10-FASB ASC 310-10-35-40 provides two alternative income recognition methods to account for changes in the net carrying amount of an impaired loan subsequent to the initial measure of impairment:

interest on the net carrying amount of changes in the net carrying amount of the loan as an adjustment to bad-debt the impaired loan and report other Under the first income recognition method, a creditor should accrue expense. ä.

for a disclosure requirement related to method, a creditor should recognize all changes in the net carrying amount of the loan as an adjustment to bad-debt expense. See FASB ASC 310-10-50-19 Under the second income recognition this method. *b*.

Those income recognition methods are not required, and a creditor is not precluded from using either of those methods.

nonaccrual status in the "Glossary" section of the FFIEC's Instructions for Preparation of Consolidated Reports of Condition Refer to the entry and Income. References As a general rule, when a loan (or other asset) has been placed on nonaccrual status, interest payments received may be recognized as interest income on a cash basis as long as the remaining recorded investment in the loan (or other asset), after charge-off of any identified losses, is deemed

If doubt exists as to the collectibility of the reon nonaccrual status, any payments received must be applied to reduce the recorded investment in the asset to the extent necessary to eliminate such doubt (for example, cost recovery bamaining recorded investment in an asset placed

to be fully collectible.

paid and the institution expects repayment of the status (using the effective interest rate) when (1) none of its principal and interest is due and unremaining contractual principal and interest<sup>2</sup> or Loans (or other assets) may be restored to accrual (2) when the loan (or other asset) becomes well secured and in the process of collection.

the borrower's historical payment performance over a reasonable period of time. The regulators consider a reasonable period of payment perfor-The determination to return a loan (or other asset) to an accrual status may also consider mance to generally be a minimum of 6 months for an amortizing loan. (continued)

There are additional requirements for applying the first part of this test, please see the "Nonaccrual Status" section of the Call Report instructions.

Regulator	
s (GAAP)	
ng Principle	
ted Accounti	
nerally Accep	
. Ge	

ory Reporting Matters References Description References Description

## Impaired Collateral Dependent Loans Matters Measuring Loan Impairment

U.S. GAAP requires the measurement of impairment on a collateral dependent impaired loan at fair value of the collateral when the creditor determines foreclosure is probable.

FASB ASC 310-10-

The FASB ASC glossary defines probable as the future event or events are likely to The creditor may choose from alternative pendent impaired loan unless foreclosure is probable, including the, (1) present value of expected future cash flows at the loan's effective interest rate, (2) observable market price of the loan or (3) the measurement methods for a collateral deair value of the collateral.

### Charge-Offs

No specific guidance exists on the timing of charge-offs

No specific ance exists.

In general, any portion of the recorded investment ing recorded accrued interest, net deferred loan fees in an impaired collateral-dependent loan (includor costs, and unamortized premium or discount) in excess of the fair value of the collateral that can be identified as uncollectible should be promptly charged off against the allowance for loan and lease osses (ALLL) guid-

2006.

ment on the Allowance

Interagency Policy Statefor Loan and Lease Losses issued December 16,

Preparation of Consolidated Reports of Condition and Income.

The agencies require the measurement of impairment on a collateral dependent loan to be measured using the fair value of collateral method.

sary" section of the Refer to the entry loan impairment in the "Glos-FFIEC's Instructions for

# Regulatory Reporting Matters

References

# U.S. Generally Accepted Accounting Principles (GAAP)

References Description

## Restructured Lease Matters

No specific ance exists. No specific guidance exists on disclosure of impaired modified finance lease receivables in FASB ASC 840-10-50.

Restructured loans and leases are reported in bank Call Report schedules RC-C/RC-N, and in National Credit Union Administration Loan related schedules (which equally applies to finance and leveraged ease restructures that would be akin to a troubled debt restructuring). Description guid-

In certain situations a modified finance lease with a debtor in financial difficulty will result in removal of the finance lease from these loan and lease reated schedules. If the restructured finance lease is eclassified as a new operating lease, the operating ease property is reported as an "Other asset" and depreciated.

## Foreclosed Assets

FASB ASC 360-10-U.S. GAAP allows for the grouping of assets when subsequently measuring a long-lived asset (disposal group) classiment should be measured at the lower of fied as held for sale. Such a measureits carrying amount or fair value less cost to sell

After foreclosure, each foreclosed real estate asset must be carried at the lower of (a) the fair value of the asset minus the estimated costs to sell the asset or (b) the cost of the asset (as defined in the glossary definition of foreclosed assets). This determination nust be made on an asset-by-asset basis.

closed assets in the "Glossary" section of the Refer to the entry fore-FFIEC's Instructions for Preparation of Consolidated Reports of Condi-

tion and Income.

### Other Matters:

# Credit Losses on International Loans

FASB ASC 310-10-

Valuation allowances (for example, an ALLL) are deducted from the assets or ate to. Additions and deductions from a group of assets that the allowances revaluation allowance should not be made for impairments or credit losses of unreated assets.

Allocated transfer risk reserves are intended to recognize and measure impairment and credit losses of sured separately from loans. However, institutions are allowed to recognize and measure in the ALLL credit losses on loan impairments or credit losses on international assets unrelated to those loans in the nternational assets, which are recognized and mea-

28.52(c) (2) and (4), Section 905(a) of the Inderal Regulations Part ternational Lending Su-Title 12 U.S. Code of Fepervision Act of 1983; reported in Schedule RI-B, of the Call Report

### (continued)

# U.S. Generally Accepted Accounting Principles (GAAP)

## Acquirer's Accounting Pushed Down to Acquiree's Financial Statements ReferencesDescription

FASB ASC 805-50-25 SEC SAB No. 115

FASB Accounting Standards Update (ASU) No. 2014-17, Pushnonpublic, the option to apply occurrence of an event in which down Accounting, provides an acquired entity, either public or arate financial statements upon an acquirer obtains control of the pushdown accounting in its sepacquired entity.

under Staff Accounting Bulletins (SAB), Topic 5.J., "New Basis tain Circumstances", pushdown accounting generally was permitted when 80 percent or more quired, required when 95 per-Previously, for SEC registrants, of Accounting Required in Cerof an entity's ownership was accent or more was acquired, and prohibited when less than 80 percent was acquired.

Prior to the issuance of FASB ASU No. 2014-17, GAAP offered imited guidance on pushdown accounting for non-SEC regis-

Pushdown accounting is an acquiree's establishment of a new accounting bawhen an acquirer obtains control of the sis in its separate financial statements acquired entity. Under Call Report instructions, an acquiree that retains its Description

separate corporate existence may apply

pushdown accounting upon a change-in-

of the FFIEC's Instructions for Refer to entry business combinations in the "Glossary" section Preparation of Consolidated Reports of Condition and Income.

> A change-in-control event occurs when an acquirer obtains a controlling financial interest, as defined under U.S. GAAP, in the acquiree. A controlling financial interest typically requires ownership of more than 50 percent of the voting rights in an acquired entity. control event.

In all cases, the institution's primary federal regulators reserve the right to require or prohibit the institution's use of pushdown accounting for Call Report ourposes based on the regulator's evaluation of whether the election best reflects the facts and circumstances of the business combination

## Regulatory Reporting Matters

References

AAG-DEP APP B

# Regulatory Reporting Matters U.S. Generally Accepted Accounting Principles (GAAP)

### Description Income Taxes of Subsidiaries Within a Consolidated Group References Description

#### No specific guidance exists. In practice, income taxes genwithin a consolidated group in accordance with a tax sharing erally are allocated to members agreement

The agencies treat each financial instituion subsidiary as if it were filing a separate return for intercompany tax allocation purposes. This treatment is also apolied in determining net deferred tax asset limitations for regulatory capital pur-

Refer to entry income taxes of ing company in the "Glossary"

References

a bank subsidiary of a holdsection of the FFIEC's Instrucidated Reports of Condition and Statement on Income Tax Allocation in a Holding Company Structure, issued December 23,

Income; and Interagency Policy

tions for Preparation of Consol-

If the holding company forgives payment by the subsidiary of the current portion of the income taxes required on a separate entity basis, the forgiveness is treated as quired on a separate return basis or if it ure of the holding company to pay the the bank's net operating loss should also a capital contribution. If the subsidiary pays an amount greater than that repays its deferred tax liability in addition to its current tax liability, the institution must report these differences as a cash dividend to the holding company. Failsubsidiary bank an equitable refund of be considered a cash dividend paid by the bank to the parent holding company.

(continued)

and the FDIC issued the Addendum to the Interagency Policy Statement on Income Tax Allocation in a Holding Company. The addendum instructs insured depository institutions (IDIs) and their holding companies to ensure their tax-allocation agreements expressly acknowledge that the holding companies are acting as agents for the IDI and are consistent with certain requirements of sections 23A and 23B of In June 2014, the Office of the Comptroller of the Currency, the Board of Governors of the Federal Reserve System (Federal Reserve), the Federal Reserve Act.

U.S. Generally Accepted Accounting Principles (GAAP)	ounting Principles (GAAP)		Regulatory Reporting Matters
Description	References	Description	References
Related Party Transactions $^4$			

lated party disclosures is established in FASB ASC General guidance on re-FASB ASC 845-10 FASB ASC 805-50-30 350-10-50 Transfers of nonmonetary assets to a stockholder (such as divicounting does not apply to transfers of nonmonetary assets solely dends in-kind) must be recorded at fair value; however, that acder common control. Rather, such transfers are accounted for between entities or persons unat historical cost.

Institutions must disclose the aggregate amount of loans to directors, officers, employees, and stockholders, as well as to entities that directors, officers, employees, and stockholders are affiliated with.

Refer to entry property dividends within the entry dividends in the "Glossary" section of the FFIEC's Instructions for Preparation of tion and Income and Schedule Consolidated Reports of Condi-RC-M of the Call Report. Transfers of assets other than cash (for example, securities of another company value, as if the transfer were completed at arm's length. However, this accounting is governed by the facts and circumor real estate) to a stockholder or reated party must be recorded at their fair stances of the transaction and there are and

often exceptions to this policy.

Institutions must disclose extensions of credit to executive officers, directors, principal shareholders, and their related interests.

legal vehicles must also be conducted in compliance with the applicable statutory and regulatory requirements of sections 23A-23B of Intercompany transactions, product offerings, performance of services and other business relationships between, or involving two or more, the Federal Reserve Act (Banks and Banking, U.S. Code 12, Section 371c and 371c-1) and Regulation W (Title 12 U.S. Code of Federal Regulations Part 223) of the Federal Reserve (the "Guidelines").

U.S. Generally Accepted Accounting Principles (GAAP)	unting Principles (GAAP)		Regulatory Reporting Matters
Description	References	Description II. DISPLAY	References
Accounting Changes			
U.S. GAAP provides guidance on the accounting for and reporting of accounting changes and error corrections. FASB ASC 250-10 establishes, unless impracticable, retrospective application as the required method for reporting a change in accounting principle in the absence of explicit transition requirements specific to a newly adopted accounting principle.	FASB ASC 250-10	Regulatory reports cover a single discrete period. As a result, the agencies require that the effect (on undivided profits) of required retroactive application of a new pronouncement be excluded from net income and reported as a direct adjustment to equity capital in the related regulatory reports in the period of adoption.	Refer to the entry accounting changes in the "Glossary" section of the FFIEC's Instructions for Preparation of Consolidated Reports of Condition and Income. Cumulative catch-up adjustment due to a change in accounting principle is reported on line 2 of Schedule RI-A and on line 4 of Schedule RI-E of the Call Report.
Gain/Loss on Certain Sales			
SEC registrants are required to include in noninterest expense the net cost of foreclosed real estate, including gains, losses, and rental income. In practice, classification by nonregistrants of gains and losses on sales of loans, real estate owned, fixed assets (including branch office assets), and other assets varies.	Article 9.04.14(d) of SEC Regulation S-X	The agencies require that institutions always include rental income in noninterest income. In addition, there are specific Call Report line items for gains (losses) on sales of loans and leases, real estate owned, and other assets.  Banks should consistently report net gains (losses) on the sale of certain others assets, such as foreign currency, as either other noninterest income or as other noninterest expense.	Line 5(i)–5(l) and 7(d) of Schedule RI of the Call Report and the related FFIEC's Instructions for Preparation of Consolidated Reports of Condition and Income.

#### **Appendix C**

#### Information Sources

This appendix is nonauthoritative and is included for informational purposes only.

Further information on matters addressed in this guide is available through various publications and services listed in the table that follows. Many nongovernment and some government publications and services involve a charge or membership requirement.

Fax services allow users to follow voice cues and request that selected documents be sent by fax machine. Some fax services require the user to call from the handset of the fax machine, others allow the user to call from any phone. Most fax services offer an index document, which lists titles and other information describing available documents.

Recorded announcements allow users to listen to announcements about a variety of recent or scheduled actions or meetings.

©2016, AICPA AAG-DEP APP C

Organization	General Information	Fax Services	Website	Recorded Announcements
American Bankers Association	1120 Connecticut Ave., NW Washington, DC 20036 800.BANKERS Custserv@aba.com		www.aba.com	
American Institute of Certified Public Accountants	Member Service Center: 220 Leigh Farm Road Durham, NC 27707 888.777.7077 Information about AICPA continuing professional education programs is available through the AICPA CPE Division 888.777.7077	800.362.5066	www.aicpa.org www.cpa2biz.com	
Board of Governors of the Federal Reserve System	20th Street and Constitution Avenue, N.W. Washington, DC 20551 General Comments for the Chair and Board of Governors: 202.974.7008 Publications Services: Publications Fulfillment Mail Stop N-127 Washington, DC 20551 202.452.3245 Consumer Assistance: 888.851.1920 Main Telephone: 202.452.3000	Publication Services: 202.728.5886	www.federalreserve .gov	Federal Reserve Board Highlights: 202.452.3206

Organization	General Information	Fax Services	Website	Recorded Announcements
Conference of State Bank Supervisors	1129 20th St., NW 9th Floor Washington, DC 20036 202.296.2840	202.296.1928	www.csbs.org	
Consumer Financial Protection Bureau	P.O. Box 4503 Iowa City, IA 52244 851.411.2372	855.237.2392	www.consumerfinance	
Credit Union National Association	Mailing Address: P.O. Box 431 Madison, WI 53701 Shipping Address: 5710 Mineral Point Road Madison, WI 53705 CUNA, Inc.: 800.356.9655 CUNA Member Services: 800.356.8010 balderson@cuna.com	CUNA, Inc.: 608.231.4333 CUNA Member Services: 608.231.1869	www.cuna.org	
Federal Deposit Insurance Corporation	Public Information Center: 3501 North Fairfax Drive Room E-1021 Arlington, VA 22226 877.275.3342 703.562.2200 publicinfo@fdic.gov	703.562.2296	www.fdic.gov	

Organization	General Information	Fax Services	Website	Recorded Announcements
Federal Home Loan Mortgage Corporation (Freddie Mac)	Corporate Office Headquarter 1: 8200 Jones Branch Drive McLean, VA 22102-3110 800.424.5401 703.903.2000		www.freddiemac.com	
Federal Housing Finance Agency	400 7th Street, SW Washington, DC 20219 202.649.3800	202.649.1071	www.fhfa.gov	
Federal National Mortgage Association (Fannie Mae)	3900 Wisconsin Ave., NW Washington, DC 20016-2892 800.232.6643		www.fanniemae.com	
Financial Accounting Standards Board	401 Merritt 7 PO Box 5116 Norwalk, CT 06856-5116 800.748.0659 203.847.0700 fasbpubs@fasb.org	203.849.9714	www.fasb.org	
Financial Managers Society	1 North LaSalle Street Suite 3100 Chicago, IL 60602-4003 800.275.4367 info@fmsinc.org		www.fmsinc.org	
Government National Mortgage Association (Ginnie Mae)	451 7th Street, SW Room B-133 Washington, DC 20410 Ginnie Mae Hotline: 888.446.6434 GinnieNET Hotline: 800.234.4662		www.ginniemae.gov	

Organization	General Information	Fax Services	Website	Recorded Announcements
Independent Community Bankers of America	1615 L Street, NW Suite 900 Washington, DC 20036 800.422.8439 202.659.8111 info@icba.org	202.659.3604	www.icba.org	
Mortgage Bankers Association	1919 M Street, NW, 5th Floor Washington, DC 20036 202.557.2700		www.mbaa.org	
NACHA—The Electronic Payments Association	2550 Wasser Terrace Suite 400 Herndon, VA 20171 703.561.1100 info@nacha.org	703.787.0996	www.nacha.org	
National Association of Federal Credit Union	3138 10th Street North Arlington, VA 22201-2149 800.344.5580 msc@nafcu.org		www.nafcu.org	
National Association of Insurance Commissioners	Central Office: 1100 Walnut Street Suite 1500 Kansas City, MO 64106-2197 816.842.3600 Customer Service: 816.783.8300 prodserv@naic.org Help Desk: 816.783.8500	Central Office: 816.783.8175 Customer Service: 816.460.7593 Help Desk: 816.460.7456	www.naic.org	

Organization	General Information	Fax Services	Website	Recorded Announcements
National Association of Real Estate Investment Trusts	1875 I Street, NW Suite 600 Washington, DC 20006 800.362.7348 202.739.9400	202.739.9401	www.reit.com	
National Credit Union Administration	1775 Duke Street Alexandria, VA 22314-3428 703.518.6300 Consumer Assistance: 800.755.1030 consumerassistance@ncua.gov		www.ncua.gov	
Public Company Accounting Oversight Board	Headquarters: 1666 K Street, NW Washington, DC 20006 202.207.9100 New York Office: 1251 Avenue of the Americas New York, NY 10020 646.437.5100 General Information: info@pcaobus.org	Headquarters: 202.862.8430 New York Office: 646.792.0001	www.pcaobus.org	

(continued)

Organization	General Information	Fax Services	Website	Recorded Announcements
Securities Industry and Financial Markets Association	New York: 120 Broadway, 35th Floor New York, NY 10271-0080 212.313.1200 Washington, DC: 1101 New York Avenue, NW 8th Floor Washington, DC 20005 202.962.7300 Membership & Membership Inquiries: 212.313.1000 conferences@sifma.org Website Inquiries: 212.313.1000 webmaster@sifma.org	New York: 212.313.1301 Washington, DC: 202.962.7305	www.sifma.org	
U.S. Department of the Treasury—Office of the Comptroller of the Currency	OCC Headquarters: 400 7th Street, SW Washington, DC 20219 Customer Assistance: 800,613.6743 713.658.0340		www.occ.gov	

Organization	General Information	Fax Services	Website	Recorded Announcements
U.S. Department of Education	400 Maryland Avenue, SW Washington, DC 20202 General Inquiries: 800.USA.LEARN Organizational Directory: 202.401.2000 800.872.5327		www.ed.gov	
U.S. Department of Housing and Urban Development	451 7th Street, SW Washington, DC 20410 202.708.1112		www.hud.gov	
U.S. General Accountability Office	441 G Street, NW Washington, DC 20548 202.512.3000 contact@gao.gov		www.gao.gov	
U.S. Government Printing Office	732 North Capitol Street, NW Washington, DC 20401-0001 866.512.1800 202.512.1800 contactenter@gpo.gov	202.512.2104	www.gpo.gov	
U.S. Securities and Exchange Commission	SEC Headquarters: 100 F Street, NE Washington, DC 20549 202.942.8088 SEC Investor Information Service: 800.SEC.0330	202.772.9295	www.sec.gov	

#### Appendix D

#### Overview of Statements on Quality Control Standards

This appendix is nonauthoritative and is included for informational purposes only.

This appendix is a partial reproduction of chapter 1 of the AICPA practice aid *Establishing and Maintaining a System of Quality Control for a CPA Firm's Accounting and Auditing Practice*, available at www.aicpa.org/interestareas/frc/pages/enhancingauditqualitypracticeaid.aspx.

This appendix highlights certain aspects of the quality control standards issued by the AICPA. If appropriate, readers should also refer to the quality control standards issued by the PCAOB, available at www.pcaobus.org/Standards/QC/Pages/default.aspx.

**1.01** The objectives of a system of quality control are to provide a CPA firm with reasonable assurance¹ that the firm and its personnel comply with professional standards and applicable regulatory and legal requirements, and that the firm or engagement partners issue reports that are appropriate in the circumstances. QC section 10, A Firm's System of Quality Control (AICPA, Professional Standards), addresses a CPA firm's responsibilities for its system of quality control for its accounting and auditing practice. That section is to be read in conjunction with the AICPA Code of Professional Conduct and other relevant ethical requirements.

**1.02** A system of quality control consists of policies designed to achieve the objectives of the system and the procedures necessary to implement and monitor compliance with those policies. The nature, extent, and formality of a firm's quality control policies and procedures will depend on various factors such as the firm's size; the number and operating characteristics of its offices; the degree of authority allowed to, and the knowledge and experience possessed by, firm personnel; and the nature and complexity of the firm's practice.

#### **Communication of Quality Control Policies and Procedures**

**1.03** The firm should communicate its quality control policies and procedures to its personnel. Most firms will find it appropriate to communicate their policies and procedures in writing and distribute them, or make them available electronically, to all professional personnel. Effective communication includes the following:

- A description of quality control policies and procedures and the objectives they are designed to achieve
- The message that each individual has a personal responsibility for quality
- A requirement for each individual to be familiar with and to comply with these policies and procedures

©2016, AICPA AAG-DEP APP D

<sup>&</sup>lt;sup>1</sup> The term reasonable assurance, which is defined as a high, but not absolute, level of assurance, used because absolute assurance cannot be attained. Paragraph .53 of QC section 10, A Firm's System of Quality Control (AICPA, Professional Standards), states, "Any system of quality control has inherent limitations that can reduce its effectiveness."

Effective communication also includes procedures for personnel to communicate their views or concerns on quality control matters to the firm's management.

#### Elements of a System of Quality Control

- **1.04** A firm must establish and maintain a system of quality control. The firm's system of quality control should include policies and procedures that address each of the following elements of quality control identified in paragraph .17 of QC section 10:
  - Leadership responsibilities for quality within the firm (the "tone at the top")
  - Relevant ethical requirements
  - Acceptance and continuance of client relationships and specific engagements
  - Human resources
  - Engagement performance
  - Monitoring
- 1.05 The elements of quality control are interrelated. For example, a firm continually assesses client relationships to comply with relevant ethical requirements, including independence, integrity, and objectivity, and policies and procedures related to the acceptance and continuance of client relationships and specific engagements. Similarly, the human resources element of quality control encompasses criteria related to professional development, hiring, advancement, and assignment of firm personnel to engagements, all of which affect policies and procedures related to engagement performance. In addition, policies and procedures related to the monitoring element of quality control enable a firm to evaluate whether its policies and procedures for each of the other five elements of quality control are suitably designed and effectively applied.
- 1.06 Policies and procedures established by the firm related to each element are designed to achieve reasonable assurance with respect to the purpose of that element. Deficiencies in policies and procedures for an element may result in not achieving reasonable assurance with respect to the purpose of that element; however, the system of quality control, as a whole, may still be effective in providing the firm with reasonable assurance that the firm and its personnel comply with professional standards and applicable regulatory and legal requirements and that the firm or engagement partners issue reports that are appropriate in the circumstances.
- **1.07** If a firm merges, acquires, sells, or otherwise changes a portion of its practice, the surviving firm evaluates and, as necessary, revises, implements, and maintains firm-wide quality control policies and procedures that are appropriate for the changed circumstances.

#### Leadership Responsibilities for Quality Within the Firm (the "Tone at the Top")

**1.08** The purpose of the leadership responsibilities element of a system of quality control is to promote an internal culture based on the recognition that

quality is essential in performing engagements. The firm should establish and maintain the following policies and procedures to achieve this purpose:

- Require the firm's leadership (managing partner, board of managing partners, CEO, or equivalent) to assume ultimate responsibility for the firm's system of quality control.
- Provide the firm with reasonable assurance that personnel assigned operational responsibility for the firm's quality control system have sufficient and appropriate experience and ability to identify and understand quality control issues and develop appropriate policies and procedures, as well as the necessary authority to implement those policies and procedures.
- **1.09** Establishing and maintaining the following policies and procedures assists firms in recognizing that the firm's business strategy is subject to the overarching requirement for the firm to achieve the objectives of the system of quality control in all the engagements that the firm performs:
  - Assign management responsibilities so that commercial considerations do not override the quality of the work performed.
  - Design policies and procedures addressing performance evaluation, compensation, and advancement (including incentive systems) with regard to personnel to demonstrate the firm's overarching commitment to the objectives of the system of quality control.
  - Devote sufficient and appropriate resources for the development, communication, and support of its quality control policies and procedures.

#### **Relevant Ethical Requirements**

- 1.10 The purpose of the relevant ethical requirements element of a system of quality control is to provide the firm with reasonable assurance that the firm and its personnel comply with relevant ethical requirements when discharging professional responsibilities. Relevant ethical requirements include independence, integrity, and objectivity. Establishing and maintaining policies such as the following assist the firm in obtaining this assurance:
  - Require that personnel adhere to relevant ethical requirements such as those in regulations, interpretations, and rules of the AICPA, state CPA societies, state boards of accountancy, state statutes, the U.S. Government Accountability Office, and any other applicable regulators.
  - Establish procedures to communicate independence requirements to firm personnel and, where applicable, others subject to them.
  - Establish procedures to identify and evaluate possible threats
    to independence and objectivity, including the familiarity threat
    that may be created by using the same senior personnel on an
    audit or attest engagement over a long period of time, and to take
    appropriate action to eliminate those threats or reduce them to
    an acceptable level by applying safeguards.

- Require that the firm withdraw from the engagement if effective safeguards to reduce threats to independence to an acceptable level cannot be applied.
- Require written confirmation, at least annually, of compliance with the firm's policies and procedures on independence from all firm personnel required to be independent by relevant requirements.
- Establish procedures for confirming the independence of another firm or firm personnel in associated member firms who perform part of the engagement. This would apply to national firm personnel, foreign firm personnel, and foreign-associated firms.<sup>2</sup>
- Require the rotation of personnel for audit or attest engagements where regulatory or other authorities require such rotation after a specified period.

#### Acceptance and Continuance of Client Relationships and Specific Engagements

1.11 The purpose of the quality control element that addresses acceptance and continuance of client relationships and specific engagements is to establish criteria for deciding whether to accept or continue a client relationship and whether to perform a specific engagement for a client. A firm's client acceptance and continuance policies represent a key element in mitigating litigation and business risk. Accordingly, it is important that a firm be aware that the integrity and reputation of a client's management could reflect the reliability of the client's accounting records and financial representations and, therefore, affect the firm's reputation or involvement in litigation. A firm's policies and procedures related to the acceptance and continuance of client relationships and specific engagements should provide the firm with reasonable assurance that it will undertake or continue relationships and engagements only where it

- is competent to perform the engagement and has the capabilities, including the time and resources, to do so;
- can comply with legal and relevant ethical requirements;
- has considered the client's integrity and does not have information that would lead it to conclude that the client lacks integrity; and
- has reached an understanding with the client regarding the services to be performed.

**1.12** This assurance should be obtained before accepting an engagement with a new client, when deciding whether to continue an existing engagement, and when considering acceptance of a new engagement with an existing client. Establishing and maintaining policies such as the following assist the firm in obtaining this assurance:

 $<sup>^2</sup>$  A foreign-associated firm is a firm domiciled outside of the United States and its territories that is a member of, correspondent with, or similarly associated with an international firm or international association of firms.

- Evaluate factors that have a bearing on management's integrity and consider the risk associated with providing professional services in particular circumstances.<sup>3</sup>
- Evaluate whether the engagement can be completed with professional competence; undertake only those engagements for which the firm has the capabilities, resources, and professional competence to complete; and evaluate, at the end of specific periods or upon occurrence of certain events, whether the relationship should be continued.
- Obtain an understanding, preferably in writing, with the client regarding the services to be performed.
- Establish procedures on continuing an engagement and the client relationship, including procedures for dealing with information that would have caused the firm to decline an engagement if the information had been available earlier.
- Require documentation of how issues relating to acceptance or continuance of client relationships and specific engagements were resolved.

#### **Human Resources**

**1.13** The purpose of the human resources element of a system of quality control is to provide the firm with reasonable assurance that it has sufficient personnel with the capabilities, competence, and commitment to ethical principles necessary (a) to perform its engagements in accordance with professional standards and regulatory and legal requirements, and (b) to enable the firm to issue reports that are appropriate in the circumstances. Establishing and maintaining policies such as the following assist the firm in obtaining this assurance:

- Recruit and hire personnel of integrity who possess the characteristics that enable them to perform competently.
- Determine capabilities and competencies required for an engagement, especially for the engagement partner, based on the characteristics of the particular client, industry, and kind of service being performed. Specific competencies necessary for an engagement partner are discussed in paragraph .A27 of QC section 10.
- Determine the capabilities and competencies possessed by personnel.
- Assign the responsibility for each engagement to an engagement partner.

©2016. AICPA AAG-DEP APP D

 $<sup>^3</sup>$  Such considerations would include the risk of providing professional services to significant clients or to other clients for which the practitioner's objectivity or the appearance of independence may be impaired. In broad terms, the significance of a client to a member or a firm refers to relationships that could diminish a practitioner's objectivity and independence in performing attest services. Examples of factors to consider in determining the significance of a client to an engagement partner, office, or practice unit include (a) the amount of time the partner, office, or practice unit devotes to the engagement, (b) the effect on the partner's stature within the firm as a result of his or her service to the client, (c) the manner in which the partner, office, or practice unit is compensated, or (d) the effect that losing the client would have on the partner, office, or practice unit.

- Assign personnel based on the knowledge, skills, and abilities required in the circumstances and the nature and extent of supervision needed.
- Have personnel participate in general and industry-specific continuing professional education and professional development activities that enable them to accomplish assigned responsibilities and satisfy applicable continuing professional education requirements of the AICPA, state boards of accountancy, and other regulators.
- Select for advancement only those individuals who have the qualifications necessary to fulfill the responsibilities they will be called on to assume.

#### **Engagement Performance**

- 1.14 The purpose of the engagement performance element of quality control is to provide the firm with reasonable assurance (a) that engagements are consistently performed in accordance with applicable professional standards and regulatory and legal requirements, and (b) that the firm or the engagement partner issues reports that are appropriate in the circumstances. Policies and procedures for engagement performance should address all phases of the design and execution of the engagement, including engagement performance, supervision responsibilities, and review responsibilities. Policies and procedures also should require that consultation takes place when appropriate. In addition, a policy should establish criteria against which all engagements are to be evaluated to determine whether an engagement quality control review should be performed.
- **1.15** Establishing and maintaining policies such as the following assist the firm in obtaining the assurance required relating to the engagement performance element of quality control:
  - Plan all engagements to meet professional, regulatory, and the firm's requirements.
  - Perform work and issue reports and other communications that meet professional, regulatory, and the firm's requirements.
  - Require that work performed by other team members be reviewed by qualified engagement team members, which may include the engagement partner, on a timely basis.
  - Require the engagement team to complete the assembly of final engagement files on a timely basis.
  - Establish procedures to maintain the confidentiality, safe custody, integrity, accessibility, and retrievability of engagement documentation.
  - Require the retention of engagement documentation for a period of time sufficient to meet the needs of the firm, professional standards, laws, and regulations.
  - Require that
    - consultation take place when appropriate (for example, when dealing with complex, unusual, unfamiliar, difficult, or contentious issues);

- sufficient and appropriate resources be available to enable appropriate consultation to take place;
- all the relevant facts known to the engagement team be provided to those consulted;
- the nature, scope, and conclusions of such consultations be documented; and
- the conclusions resulting from such consultations be implemented.

#### • Require that

- differences of opinion be dealt with and resolved;
- conclusions reached are documented and implemented;
   and
- the report not be released until the matter is resolved.

#### Require that

- all engagements be evaluated against the criteria for determining whether an engagement quality control review should be performed;
- an engagement quality control review be performed for all engagements that meet the criteria; and
- the review be completed before the report is released.
- Establish procedures addressing the nature, timing, extent, and documentation of the engagement quality control review.
- Establish criteria for the eligibility of engagement quality control reviewers.

#### **Monitoring**

- 1.16 The purpose of the monitoring element of a system of quality control is to provide the firm and its engagement partners with reasonable assurance that the policies and procedures related to the system of quality control are relevant, adequate, operating effectively, and complied with in practice. Monitoring involves an ongoing consideration and evaluation of the appropriateness of the design, the effectiveness of the operation of a firm's quality control system, and a firm's compliance with its quality control policies and procedures. The purpose of monitoring compliance with quality control policies and procedures is to provide an evaluation of the following:
  - Adherence to professional standards and regulatory and legal requirements
  - Whether the quality control system has been appropriately designed and effectively implemented
  - Whether the firm's quality control policies and procedures have been operating effectively so that reports issued by the firm are appropriate in the circumstances
- **1.17** Establishing and maintaining policies such as the following assist the firm in obtaining the assurance required relating to the monitoring element of quality control:

- Assign responsibility for the monitoring process to a partner or partners or other persons with sufficient and appropriate experience and authority in the firm to assume that responsibility.
- Assign performance of the monitoring process to competent individuals.
- Require the performance of monitoring procedures that are sufficiently comprehensive to enable the firm to assess compliance with all applicable professional standards and the firm's quality control policies and procedures. Monitoring procedures consist of the following:
  - Review of selected administrative and personnel records pertaining to the quality control elements.
  - Review of engagement documentation, reports, and clients' financial statements.
  - Summarization of the findings from the monitoring procedures, at least annually, and consideration of the systemic causes of findings that indicate that improvements are needed.
  - Determination of any corrective actions to be taken or improvements to be made with respect to the specific engagements reviewed or the firm's quality control policies and procedures.
  - Communication of the identified findings to appropriate firm management personnel.
  - Consideration of findings by appropriate firm management personnel who should also determine that any actions necessary, including necessary modifications to the quality control system, are taken on a timely basis.
  - Assessment of
    - the appropriateness of the firm's guidance materials and any practice aids;
    - new developments in professional standards and regulatory and legal requirements and how they are reflected in the firm's policies and procedures where appropriate;
    - compliance with policies and procedures on independence;
    - the effectiveness of continuing professional development, including training;
    - decisions related to acceptance and continuance of client relationships and specific engagements;
    - firm personnel's understanding of the firm's quality control policies and procedures and implementation thereof.
- Communicate at least annually, to relevant engagement partners and other appropriate personnel, deficiencies noted as a result

of the monitoring process and recommendations for appropriate remedial action.

- Communicate the results of the monitoring of its quality control system process to relevant firm personnel at least annually.
- Establish procedures designed to provide the firm with reasonable assurance that it deals appropriately with the following:
  - Complaints and allegations that the work performed by the firm fails to comply with professional standards and regulatory and legal requirements.
  - Allegations of noncompliance with the firm's system of quality control.
  - Deficiencies in the design or operation of the firm's quality control policies and procedures, or noncompliance with the firm's system of quality control by an individual or individuals, as identified during the investigations into complaints and allegations.

This includes establishing clearly defined channels for firm personnel to raise any concerns in a manner that enables them to come forward without fear of reprisal and documenting complaints and allegations and the responses to them.

- Require appropriate documentation to provide evidence of the operation of each element of its system of quality control. The form and content of documentation evidencing the operation of each of the elements of the system of quality control is a matter of judgment and depends on a number of factors, including the following, for example:
  - The size of the firm and the number of offices.
  - The nature and complexity of the firm's practice and organization.
- Require retention of documentation providing evidence of the operation of the system of quality control for a period of time sufficient to permit those performing monitoring procedures and peer review to evaluate the firm's compliance with its system of quality control, or for a longer period if required by law or regulation.
- **1.18** Some of the monitoring procedures discussed in the previous list may be accomplished through the performance of the following:
  - Engagement quality control review
  - Review of engagement documentation, reports, and clients' financial statements for selected engagements after the report release date
  - Inspection<sup>4</sup> procedures

©2016. AICPA

<sup>&</sup>lt;sup>4</sup> Inspection is a retrospective evaluation of the adequacy of the firm's quality control policies and procedures, its personnel's understanding of those policies and procedures, and the extent of the firm's compliance with them. Although monitoring procedures are meant to be ongoing, they may include inspection procedures performed at a fixed point in time. Monitoring is a broad concept; inspection is one specific type of monitoring procedure.

#### **Documentation of Quality Control Policies and Procedures**

**1.19** The firm should document each element of its system of quality control. The extent of the documentation will depend on the size, structure, and nature of the firm's practice. Documentation may be as simple as a checklist of the firm's policies and procedures or as extensive as practice manuals.

AAG-DEP APP D

#### Appendix E

#### The New Revenue Recognition Standard: FASB ASU No. 2014-09

This appendix is nonauthoritative and is included for informational purposes only.

#### **Overview**

On May 28, 2014, the International Accounting Standards Board (IASB) and FASB issued a joint accounting standard on revenue recognition to address a number of concerns regarding the complexity and lack of consistency surrounding the accounting for revenue transactions. Consistent with each board's policy, FASB issued Accounting Standards Update (ASU) No. 2014-09, Revenue from Contracts with Customers (Topic 606), and the IASB issued International Financial Reporting Standard (IFRS) 15, Revenue from Contracts with Customers. FASB ASU No. 2014-09 will amend the FASB Accounting Standards Codification® (ASC) by creating a new Topic 606, Revenue from Contracts with Customers, and a new Subtopic 340-40, Other Assets and Deferred Costs—Contracts with Customers. The guidance in ASU No. 2014-09 provides what FASB describes as a framework for revenue recognition and supersedes or amends several of the revenue recognition requirements in FASB ASC 605, Revenue Recognition, as well as guidance within the 900 series of industry-specific topics.

As part of the boards' efforts to converge U.S. generally accepted accounting principles (GAAP) and IFRSs, the standard eliminates the transaction- and industry-specific revenue recognition guidance under current GAAP and replaces it with a principles-based approach for revenue recognition. The intent is to avoid inconsistencies of accounting treatment across different geographies and industries. In addition to improving comparability of revenue recognition practices, the new guidance provides more useful information to financial statement users through enhanced disclosure requirements. FASB and the IASB have essentially achieved convergence with these standards, with some minor differences related to the collectibility threshold, interim disclosure requirements, early application and effective date, impairment loss reversal, and nonpublic entity requirements.

The standard applies to any entity that either enters into contracts with customers to transfer goods or services or enters into contracts for the transfer of nonfinancial assets, unless those contracts are within the scope of other standards (for example, insurance or lease contracts).

#### **Effective or Applicability Date**

The guidance in ASU No. 2014-09 was originally effective for annual reporting periods of public entities beginning after December 15, 2016, including interim periods within that reporting period. Early application was not permitted for public entities, including not-for-profit entities (NFPs) that have issued, or are conduit bond obligors for, securities that are traded, listed, or quoted on an

exchange or an over-the-counter market and for employee benefit plans that file or furnish financial statements to the SEC.

For nonpublic entities, the amendments in the new guidance were originally effective for annual reporting periods beginning after December 15, 2017, and interim periods within annual periods beginning after December 15, 2018.

On August 12, 2015, FASB issued ASU No. 2015-14, Revenue from Contracts with Customers (Topic 606): Deferral of the Effective Date, to allow entities additional time to implement systems, gather data, and resolve implementation questions. This update allows for public business entities, certain NFPs, and certain employee benefit plans to apply the new requirements to annual reporting periods beginning after December 15, 2017, including interim reporting periods within that reporting period. Earlier application is permitted only as of annual reporting periods beginning after December 15, 2016, including interim reporting periods within that reporting period.

All other entities will now apply the guidance in ASU No. 2014-09 to annual reporting periods beginning after December 15, 2018, and interim reporting periods within annual reporting periods beginning after December 15, 2019. Application is permitted earlier only as of an annual reporting period beginning after December 15, 2016, including interim reporting periods within that reporting period, or an annual reporting period beginning after December 15, 2016, and interim reporting periods within annual reporting periods beginning one year after the annual reporting period in which an entity first applies the guidance in ASU No. 2014-09.

#### Overview of the New Guidance

The core principle of the revised revenue recognition standard is that an entity should recognize revenue to depict the transfer of goods or services to customers in an amount that reflects the consideration to which the entity expects to be entitled in exchange for those good or services.

To apply the proposed revenue recognition standard, ASU No. 2014-09 states that an entity should follow these five steps:

- 1. Identify the contract(s) with a customer.
- 2. Identify the performance obligations in the contract.
- 3. Determine the transaction price.
- 4. Allocate the transaction price to the performance obligations in the contract.
- 5. Recognize revenue when (or as) the entity satisfies a performance obligation.

Under the new standard, revenue is recognized when a company satisfies a performance obligation by transferring a promised good or service to a customer (which is when the customer obtains control of that good or service). See the following discussion of the five steps involved when recognizing revenue under the new guidance.

#### **Understanding the Five-Step Process**

#### Step 1: Identify the Contract(s) With a Customer

ASU No. 2014-09 defines a contract as "an agreement between two or more parties that creates enforceable rights and obligations." The new standard affects contracts with a customer that meet the following criteria:

- Approval (in writing, orally, or in accordance with other customary business practices) and commitment of the parties
- Identification of the rights of the parties
- Identification of the payment terms
- Contract has commercial substance
- Probable that the entity will collect substantially all the consideration to which it will be entitled in exchange for the goods or services that will be transferred to the customer

A contract does not exist if each party to the contract has the unilateral enforceable right to terminate a wholly unperformed contract without compensating the other party (parties).

#### Step 2: Identify the Performance Obligations in the Contract

A *performance obligation* is a promise in a contract with a customer to transfer a good or service to the customer.

At contract inception, an entity should assess the goods or services promised in a contract with a customer and identify as a performance obligation (possibly multiple performance obligations) each promise to transfer to the customer either

- a good or service (or bundle of goods or services) that is distinct, or
- a series of distinct goods or services that are substantially the same and that have the same pattern of transfer to the customer.

A good or service that is not distinct should be combined with other promised goods or services until the entity identifies a bundle of goods or services that is distinct. In some cases, that would result in the entity accounting for all the goods or services promised in a contract as a single performance obligation.

#### Step 3: Determine the Transaction Price

The transaction price is the amount of consideration (fixed or variable) the entity expects to receive in exchange for transferring promised goods or services to a customer, excluding amounts collected on behalf of third parties. To determine the transaction price, an entity should consider the effects of

- variable consideration,
- constraining estimates of variable consideration,
- the existence of a significant financing component,
- noncash considerations, and
- consideration payable to the customer.

©2016. AICPA AAG-DEP APP E

If the consideration promised in a contract includes a variable amount, then an entity should estimate the amount of consideration to which the entity will be entitled in exchange for transferring the promised goods or services to a customer. An entity would then include in the transaction price some or all of an amount of variable consideration only to the extent that it is probable that a significant reversal in the amount of cumulative revenue recognized will not occur when the uncertainty associated with the variable consideration is subsequently resolved.

An entity should consider the terms of the contract and its customary business practices to determine the transaction price.

#### Step 4: Allocate the Transaction Price to the Performance Obligations in the Contract

The transaction price is allocated to separate performance obligations in proportion to the standalone selling price of the promised goods or services. If a standalone selling price is not directly observable, then an entity should estimate it. Reallocation of the transaction price for changes in the standalone selling price is not permitted. When estimating the standalone selling price, entities can use various methods, including the adjusted market assessment approach, expected cost plus a margin approach, and residual approach (only if the selling price is highly variable and uncertain).

Sometimes, the transaction price includes a discount or a variable amount of consideration that relates entirely to one of the performance obligations in a contract. Guidance under the new standard specifies when an entity should allocate the discount or variable consideration to one (or some) performance obligation(s), rather than to all the performance obligations in the contract.

#### Step 5: Recognize Revenue When (or as) the Entity Satisfies a Performance Obligation

The amount of revenue recognized when transferring the promised good or service to a customer is equal to the amount allocated to the satisfied performance obligation, which may be satisfied at a point in time or over time. Control of an asset refers to the ability to direct the use of, and obtain substantially all the remaining benefits from, the asset. Control also includes the ability to prevent *other entities* from directing the use of, and obtaining the benefits from, an asset.

When performance obligations are satisfied over time, the entity should select an appropriate method for measuring its progress toward complete satisfaction of that performance obligation. The standard discusses methods of measuring progress, including input and output methods, and how to determine which method is appropriate.

#### Additional Guidance Under the New Standard

In addition to the five-step process for recognizing revenue, ASU No. 2014-09 also addresses the following areas:

- Accounting for incremental costs of obtaining a contract, as well as costs incurred to fulfill a contract
- Licenses
- Warranties

Lastly, the new guidance enhances disclosure requirements to include more information about specific revenue contracts entered into by the entity, including performance obligations and the transaction price.

## **Transition Resource Group**

Due to the potential for significant changes that may result from the issuance of the new standard, FASB and the IASB have received an abundance of implementation questions from interested parties. To address these questions, the boards have formed a joint Transition Resource Group (TRG) for revenue recognition to promote effective implementation and transition to the converged standard.

Since the issuance of the standard, the TRG has met several times to discuss implementation issues raised by concerned parties and actions to take to address these issues. Refer to FASB's TRG website for more information on this group and the status of their efforts, including meeting materials and meeting summaries.

## **Latest Developments**

Based on discussions held thus far on individual areas affected by the new standard, the TRG informed the boards that technical corrections are needed to further articulate the guidance in the standard. As a result, FASB has issued updates to clarify guidance on performance obligations, licensing, principal versus agent considerations, and other narrow-scope improvements and practical expedients.

ASU No. 2016-08, Revenue from Contracts with Customers (Topic 606): Principle versus Agent Considerations (Reporting Revenue Gross versus Net), was issued in March 2016 to clarify the guidance in FASB ASC 606 with respect to principal versus agent. There is little disagreement that an entity who is a principal recognizes revenue in the gross amount of consideration when a performance obligation is satisfied. An entity who is an agent (collecting revenue on behalf of the principal) recognizes revenue only to the extent of the commission or fee that the agent collects. This ASU hopes to eliminate the potential diversity in practice when determining whether an entity is a principal or an agent by clarifying the following:

- An entity determines whether it is a principal or an agent for each distinct good or service.
- An entity determines the nature of each specified good or service (including whether it is a right to a good or service)
- When an entity is a principal, it obtains control of
  - a good or another asset from the other party that it then transfers to the customer;
  - a right to a service that will be performed by another party, which gives the entity the ability to direct that party to provide the service to the customer on the entity's behalf; or

- a good or service from the other party that it combines with other goods or services to provide the specified good or service to the customer.
- Indicators in the assessment of control may be more or less relevant or persuasive, or both, to the control assessment, depending on the facts and circumstances.

Additional illustrative examples are also provided in ASU No. 2016-08 to further assist practitioners in applying this guidance. The effective date of this update is in line with the guidance in ASU No. 2014-09.

ASU No. 2016-10, Revenue from Contracts with Customers (Topic 606): Identifying Performance Obligations and Licensing, was issued in April 2016 to reduce potential for diversity in practice at initial application of FASB ASC 606, as well as the cost and complexity of applying FASB ASC 606 at transition and on an ongoing basis. When identifying promised goods and services in a contract, this ASU states that entities

- are not required to assess whether promised goods or services are performance obligations if they are immaterial to the contract.
- can elect to account for shipping and handling activities as an activity to fulfill promises within the contract, rather than as an additional promised service.

When assessing whether promised goods or services are distinct, this ASU emphasizes the need to determine whether the nature of the promise is to transfer

- each of the goods or services, or
- a combined item (or items) to which the promised goods or services are inputs.

With regards to licensing, ASU No. 2016-10 clarifies whether revenue should be recognized at a point in time or over time, based on whether the license provides a right to use an entity's intellectual property or a right to access the entity's intellectual property. Specifically,

- if the intellectual property has significant standalone functionality, the license does not include supporting or maintaining that intellectual property during the license period. Therefore, the performance obligation would be considered satisfied at a point in time. Examples of this type of intellectual property include software, biological compounds or drug formulas, and media.
- licenses for symbolic intellectual property include supporting or maintaining that intellectual property during the license period and, therefore, are considered to be satisfied over time. Examples of symbolic intellectual property include brands, team or trade names, logos, and franchise rights.

Lastly, ASU No. 2016-10 provides clarification on implementation guidance on recognizing revenue for sales-based or usage-based royalty promised in exchange for a license of intellectual property. The effective date of this ASU is in line with the guidance in ASU No. 2014-09.

In addition to ASU Nos. 2016-08 and 2016-10, ASU No. 2016-12, Revenue from Contracts with Customers (Topic 606): Narrow-Scope Improvements and

Practical Expedients, was issued in May 2016. Topics covered in this ASU include

- clarification on contract modifications. This amendment permits an entity to determine and allocate the transaction price on the basis of all satisfied and unsatisfied performance obligations in a modified contract as of the beginning of the earliest period presented in accordance with the guidance in FASB ASC 606. An entity would not be required to separately evaluate the effects of each contract modification. An entity that chooses to apply this practical expedient would apply the expedient consistently to similar types of contracts.
- how to assess the collectibility criterion. The amendment introduces new criteria to meet the collectibility requirement. An entity should assess the collectibility of the consideration promised in a contract for the goods or services that will be transferred to the customer, rather than assessing the collectibility of the consideration promised in the contract for all the promised goods or services.
- how to report sales taxes and similar taxes. This amendment states that an entity may make an accounting policy election to exclude from the measurement of the transaction price all taxes assessed by a governmental authority that are both imposed on and concurrent with a specific revenue-producing transaction and collected by the entity from a customer (for example, sales, use, value added, and some excise taxes). Taxes assessed on an entity's total gross receipts or imposed during the inventory procurement process should be excluded from the scope of the election. An entity that makes this election should exclude from the transaction price all taxes in the scope of the election and should comply with the applicable accounting policy guidance, including disclosure requirements.
- when to measure noncash consideration. This amendment clarifies that the measurement date for noncash consideration is contract inception. If the fair value of the noncash consideration varies because of the form of the consideration and for reasons other than the form of the consideration, an entity should apply the guidance on variable consideration only to the variability resulting from reasons other than the form of the consideration.
- how to apply transition guidance. This amendment clarifies that a completed contract for purposes of transition is a contract for which all (or substantially all) the revenue was recognized under legacy GAAP before the date of initial application. Accounting for elements of a contract that do not affect revenue under legacy GAAP are irrelevant to the assessment of whether a contract is complete. In addition, the amendment permits an entity to apply the modified retrospective transition method either to all contracts or only to contracts that are not completed contracts.

The effective date of this ASU is in line with the guidance in ASU No. 2014-09.

FASB also issued a proposed ASU, Technical Corrections and Improvements to Update 2014-09, Revenue from Contracts with Customers (Topic 606), in May

2016. These proposed amendments affect narrow aspects of guidance issued in ASU No. 2014-09, including but not limited to, guidance on

- preproduction costs related to long-term supply arrangements,
- impairment testing,
- provisions for losses on construction-type and production-type contracts, and,
- disclosure of remaining performance obligations.

Comments on this proposed ASU are due July 2, 2016. Refer to FASB's website for more information on these updates.

#### Conclusion

Upon implementation of the new standard, consistency of revenue recognition principles across geography and industry will be enhanced and financial statement users will be provided better insight through improved disclosure requirements. To provide CPAs with guidance during this time of transition, the AICPA's Financial Reporting Center (FRC) offers invaluable resources on the topic, including a roadmap to ensure that companies take the necessary steps to prepare themselves for the new standard. In addition, the FRC includes a list of conferences, webcasts, and other products to keep you informed on upcoming changes in revenue recognition. Refer to www.aicpa.org/INTERESTAREAS/FRC/ACCOUNTINGFINANCIALREPORTING/REVENUERECOGNITION/Pages/RevenueRecognition.aspx to stay updated on the latest information available on revenue recognition.

# Appendix F

# The New Leases Standard: FASB ASU No. 2016-02

This appendix is nonauthoritative and is included for informational purposes only.

#### Overview

## Issuance and Objective

On February 25, 2016, FASB issued Accounting Standards Update (ASU) No. 2016-02, *Leases (Topic 842)*. The objective of the ASU is to increase transparency and comparability in financial reporting by requiring balance sheet recognition of leases and note disclosure of certain information about lease arrangements. This ASU codifies the new FASB ASC topic 842, *Leases*, and makes conforming amendments to other FASB ASC topics.

The new FASB ASC topic on leases consists of these subtopics:

- a. Overall
- b. Lessee
- c. Lessor
- d. Sale and leaseback transactions
- e. Leveraged lease arrangements

## **Applicability and Effective Date**

ASU No. 2016-02 is applicable to any entity that enters into a lease and is effective as follows:

	Fiscal Years Beginning After	Interim Periods Within Fiscal Years Beginning After
Public business entities, certain not-for-profit entities with conduit financing arrangements, and employee benefit plans	December 15, 2018	December 15, 2018
All other entities	December 15, 2019	December 15, 2020

FASB ASC 842 applies to all leases and subleases of property, plant, and equipment; it specifically does not apply to the following nondepreciable assets accounted for under other FASB ASC topics:

- a. Leases of intangible assets
- b. Leases to explore for or use nonregenerative resources such as minerals, oil, and natural gas

- c. Leases of biological assets, such as timber
- d. Leases of inventory
- e. Leases of assets under construction

#### Main Provisions

#### Overall

#### Identifying a Lease

Key changes in the guidance are illustrated by comparing the definition of a lease in FASB ASC 840 (extant GAAP) and FASB ASC 842.

FASB ASC 840	FASB ASC 842
An agreement conveying the right to use property, plant, or equipment (land and/or depreciable assets) usually for a stated period of time.	A contract, or part of a contract, that conveys the right to control the use of identified property, plant, or equipment (an identified asset) for a period of time in exchange for consideration.

The identification of a lease under FASB ASC 842 should be based on the presence of key elements in the definition.

#### Separating Components of a Lease Contract

Under FASB ASC 842, a contract that contains a lease should be separated into lease and nonlease components. Separation should be based on the right to use; each underlying asset should be considered to be separate from other lease components when both of the following criteria are met:

- $\it a.$  The lessee can benefit from the right-of-use of the asset (either alone or with other readily available resources)
- b. The right-of-use is neither highly dependent on or highly interrelated with other underlying assets in the contract

The consideration in the contract should be allocated to the separate lease and nonlease components in accordance with provisions of FASB ASC 842.

Lessees can make an accounting policy election to treat both lease and nonlease elements as a single lease component.

#### Lease Classification

When a lease meets any of the following specified criteria at commencement, the lease should be classified by the lessee and lessor as a finance lease and a sales-type lease, respectively. These criteria can be summarized as follows:

- a. Transfers ownership to lessee
- b. Purchase option reasonably certain to be exercised
- c. Lease term for major portion of asset's remaining economic life
- d. Present value of lease payments and residual value exceeds substantially all of the fair value of the underlying asset
- e. Specialized nature of underlying asset results in no expectation of alternative use after the lease term

If none of the above criteria are met, the lease should be classified as follows:

Lessee—classify as an operating lease

Lessor—classify as an operating lease unless (1) the present value of the lease payments and any residual value guarantee that equals or exceeds substantially all of the fair value of the underlying asset and (2) it is probable that the lessor will collect the lease payments plus any residual value guarantee. If both of these summarized criteria from FASB ASC 842-10-25-3 are met, the lessor should classify the lease as a direct financing lease.

#### Lease Term and Measurement

The lease term is the noncancellable period of the lease together with all of the following:

- a. Period covered by the option for the lessee to extend the lease if the option is reasonably certain to be exercised
- b. Period covered by option for lessee to terminate the lease if reasonably certain not to be exercised
- c. Period covered by option for lessor to extend or not terminate the lease if option is controlled by lessor.

#### **Lease Payments**

Lease payments relating to use of the underlying asset during the lease term include the following at the commencement date:

- a. Fixed payments less incentives payable to lessee
- b. Variable lease payments based on an index or other rate
- c. Exercise price of an option to purchase the underlying asset if it is reasonably certain to be exercised
- d. Payments for penalties for terminating a lease if the lease term reflects exercise of lessee option
- e. Fees paid by the lessee to the owners of a special purpose entity for structuring the lease
- f. For lessee only, amounts probable of being owed under residual value guarantees

Lease payments specifically exclude the following:

- a. Certain other variable lease payments
- b. Any guarantee by the lessee of the lessor's debt
- c. Certain amounts allocated to nonlease components

Reassessment of the lease term and purchase options, and subsequent remeasurement by either the lessee or lessor are limited to certain specified circumstances.

#### Lessee

## Recognition and Measurement

#### Commencement Date

At the commencement date of the lease, a lessee should recognize a right-ofuse asset and a lease liability; for short term leases, an alternative accounting policy election is available. The lease liability should be measured at the present value of the unpaid lease payments. The right-of-use asset should consist of the following: the amount of the initial lease liability; any lease payments made to lessor at or before the commencement date minus any incentives received; and initial direct costs.

A short term lease is defined by the FASB ASC master glossary as a lease that, at the commencement date has a lease term of 12 months or less and does not include an option to purchase the underlying asset that the lessee is reasonably certain to exercise. The accounting policy election for short term leases should be made by class of underlying asset. The election provides for recognition of the lease payments in profit or loss on a straight-line basis over the lease term and variable lease payments in the period in which the obligation for those payments is incurred.

#### After the Commencement Date

After the commencement date, the lessee should recognize in profit or loss (unless costs are included in the carrying amount of another asset) the following:

- Finance leases:
  - a. Amortization of the right-of-use asset and interest on the lease liability
  - b. Variable lease payments not included in the lease liability in the period obligation incurred
  - c. Any impairment
- Operating leases:
  - a. A single lease cost calculated such that the remaining cost is allocated on a straight line basis over the remaining lease term (unless another allocation is more representative of the benefit from use of the asset)
  - b. Variable lease payments not included in the lease liability in the period in which the obligation is incurred
  - c. Any impairment

#### Subsequent Measurement

FASB ASC 842-20-35 provides guidance for subsequent measurement.

#### Presentation and Disclosure

Key presentation matters include the following:

- Statement of financial position.
  - Separate presentation of right-of-use assets and lease liabilities from finance leases and operating leases.
- Statement of comprehensive income.
  - Finance leases—interest expense on the lease liability and amortization of right-of-use asset in a manner consistent with how the entity presents other interest expense and depreciation or amortization of similar assets.
  - Operating leases—expense to be included in the lessee's income from continuing operations.

- Statement of cash flows.
  - Presentation within financing activities—the repayment of the principal portion of the lease liability arising from finance leases.
  - Presentation within operating activities—payments arising from operating leases; interest payments on the lease liability; variable lease payments and short term lease payments not included in lease liability.

Disclosure requirements include qualitative and quantitative information for leases, significant judgements, and amounts recognized in the financial statements, including certain specified information and amounts.

#### Lessor

#### Recognition and Measurement

FASB ASC 842 provides recognition guidance for sales-type leases, direct financing leases, and operating leases. The following table summarizes the guidance:

Sales-Type Leases		
At the Commencement Date	After the Commencement Date	
Lessor should derecognize the underlying asset and recognize the following:  a. Net investment in the lease (lease receivable and unguaranteed residual asset)  b. Selling profit or loss arising from the lease  c. Initial direct costs as an expense	Lessor should recognize all of the following:  a. Interest income on the net investment in the lease b. Certain variable lease payments c. Impairment	
Direct Financi	ing Leases	
At the Commencement Date	After the Commencement Date	
Lessor should derecognize the underlying asset and recognize the following:  a. Net investment in the lease (lease receivable and unguaranteed residual asset reduced by selling profit)  b. Selling loss arising from the lease, if applicable	Lessor should recognize all of the following:  a. Interest income on the net investment in the lease b. Certain variable lease payments c. Impairment	

Operating Leases		
At the Commencement Date	After the Commencement Date	
Lessor should defer initial direct costs.	Lessor should recognize all of the following:  a. The lease payments as income in profit or loss over the lease term on a straight line basis (unless another method in more representative of the benefit received)  b. Certain variable lease payments as income in profit or loss  c. Initial direct costs as an expense over the lease term on the same basis as lease income	

FASB ASC 842-30-35 provides guidance for subsequent measurement.

#### Presentation and Disclosure

Key presentation matters include the following:

For sales-type and direct financing leases:

- Statement of financial position
  - Separate presentation of lease assets (that is, aggregate of lessor's net investment in sales-type leases and direct financing leases) from other assets.
  - Classified as current or noncurrent based on same considerations as other assets.
- Statement of comprehensive income
  - Presentation of income from leases in the statement of comprehensive income or disclosure of income from leases in the notes with a reference to the corresponding line in the statement of comprehensive income.
  - Presentation of profit or loss recognized at commencement date in a manner appropriate to lessor's business model.
- Statement of cash flows
  - Presentation within operating activities—cash receipts from leases.

For operating leases:

- Statement of financial position
  - Presentation of an underlying asset subject to an operating leases in accordance with other FASB ASC topics.
- Statement of cash flows
  - Presentation within operating activities—cash receipts from leases.

Disclosure requirements include qualitative and quantitative information for leases, significant judgements, and amounts recognized in the financial statements, including certain specified information and amounts.

#### Sale and Leaseback Transactions

FASB ASC 842 provides guidance for both the transfer contract and the lease in a sale and leaseback transaction (a transaction in which a seller-lessee transfers an asset to a buyer-lessor and leases that asset back). Determination of whether the transfer is a sale should be based on provisions of FASB ASC 606, *Revenue from Contracts with Customers*. FASB ASC 842-40-25 provides measurement guidance for a transfer that is either determined to be a sale or determined not to be a sale.

FASB ASC 842-40 provides guidance for subsequent measurement, financial statement presentation, and disclosures.

## **Leveraged Lease Arrangements**

The legacy accounting model for leveraged leases continues to apply to those leveraged leases that commenced before the effective date of FASB ASC 842. There is no separate accounting model for leveraged leases that commence after the effective date of FASB ASC 842.

©2016, AICPA AAG-DEP APP F

# Appendix G

# **Accounting for Financial Instruments**

This appendix is nonauthoritative and is included for informational purposes only.

# **Overall Project Objective**

The objective of FASB's Accounting for Financial Instruments project is to significantly improve the decision usefulness of financial instrument reporting for users of financial statements. The project was initiated to reconsider recognition and measurement of financial instruments, address issues related to impairment of financial instruments and hedge accounting, and increase convergence in accounting for financial instruments. In replacing the existing financial instruments standards, an expected outcome is the simplification of the accounting requirements for financial instruments. The project was split into three phases including classification and measurement, impairment, and hedge accounting. This appendix focusses on the latest developments in each of these phases.

#### Classification and Measurement

#### Overview

On January 5, 2016, FASB issued Accounting Standards Update (ASU) No. 2016-01, Financial Instruments—Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities, to enhance the reporting model for financial instruments and to provide users of financial statements with more decision-useful information. The amendments in the ASU are intended to improve certain aspects of recognition, measurement, presentation, and disclosure of financial instruments.

The new guidance will accomplish the following:

- Require equity investments (except those accounted for under the
  equity method of accounting or those that result in consolidation
  of the investee) to be measured at fair value with changes in fair
  value recognized in net income
- Replace the impairment model for equity investments without readily determinable fair values with a qualitative impairment assessment
- Eliminate the requirement to disclose the fair values of financial assets and financial liabilities measured at amortized cost for entities that are not public business entities
- Eliminate the requirement for public business entities to disclose
  the methods and significant assumptions used to estimate fair
  value that is required to be disclosed for financial assets and financial liabilities measured at amortized cost on the balance sheet
- Require public business entities to use the exit price notion when measuring the fair value of financial instruments for disclosure purposes

- Require an entity to present separately in other comprehensive income the portion of the total change in the fair value of a liability resulting from a change in instrument-specific credit risk when the entity has elected to measure the liability at fair value in accordance with the fair value option for financial instruments
- Require separate presentation of financial assets and financial liabilities by measurement category and form of financial asset (that is, securities or loans and receivables) on the balance sheet or the accompanying notes to the financial statements
- Clarify that an entity should evaluate the need for a valuation allowance on a deferred tax asset related to available-for-sale debt securities in combination with an entity's other deferred tax assets
- Eliminate an entity's ability to estimate the disclosed fair values
  of financial assets and financial liabilities on the basis of entry
  prices

## **Applicability and Effective Date**

ASU No. 2016-01 affects all entities that hold financial assets or have financial liabilities and is effective as follows:

	Fiscal Years Beginning After	Interim Periods Within Fiscal Years Beginning After
Public business entities	December 15, 2017	December 15, 2017
All other entities, including not-for-profit entities and employee benefit plans within the scope of FASB Accounting Standards Codification (ASC) 960–965 on plan accounting	December 15, 2018	December 15, 2019

All entities that are not public business entities may adopt the amendments in this ASU earlier as of the fiscal years beginning after December 15, 2017, including interim periods within those fiscal years.

Early application by public business entities to financial statements of fiscal years or interim periods that have not yet been issued or, by all other entities, that have not yet been made available for issuance of the following amendments in this ASU are permitted as of the beginning of the fiscal year of adoption:

- An entity should present separately in other comprehensive income the portion of the total change in the fair value of a liability resulting from a change in the instrument-specific credit risk if the entity has elected to measure the liability at fair value in accordance with the fair value option for financial instruments.
- Entities that are not public business entities are not required to apply the fair value of financial instruments disclosure guidance in the "General" subsection of FASB ASC 825-10-50.

With the exception of this early application guidance, early adoption of the amendments in this ASU is not permitted.

## **Impairment**

#### Overview

On June 16, 2016, FASB issued ASU No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments, to provide financial statement users with more decision-useful information about the expected credit losses on financial instruments and other commitments to extend credit held by a reporting entity at each reporting date. Upon the effective date of this ASU, the incurred loss impairment methodology in current general accepted accounting principles (GAAP) is replaced with a methodology that reflects expected credit losses and requires consideration of a broader range of reasonable and supportable information to inform credit loss estimates.

#### Assets Measured at Amortized Cost

ASU No. 2016-13 eliminates the probable initial recognition threshold under current GAAP and requires entities that measure financial assets (or a group of financial assets) at amortized cost basis to present such assets at the net amount expected to be collected. The amendments in this ASU broaden the information that an entity must consider in developing its expected credit loss estimate for assets measured either collectively or individually. In addition to past events and current conditions, entities should also consider reasonable and supportable forecasts that affect the collectibility of the reported amount. However, an entity may revert to historical loss information that is reflective of the contractual term (considering the effect of prepayments) for periods that are beyond the time frame for which the entity is able to develop reasonable and supportable forecasts.

An entity may apply any method for measuring expected credit losses as long as their method reasonably reflects its expectations of the credit loss estimate.

#### Purchased Financial Assets With Credit Deterioration

ASU No. 2016-13 defines purchased financial assets with credit deterioration (PCD assets) as acquired individual financial assets (or acquired groups of financial assets with similar risk characteristics) that as of the date of acquisition have experienced a more-than-insignificant deterioration in credit quality since origination, as determined by the acquirer's assessment. The allowance for credit losses for PCD assets that are measured at amortized cost basis is determined in a similar manner to other financial assets measured at amortized cost basis. The initial allowance for credit losses is added to the purchase price, rather than being reported as a credit loss expense. Entities record only subsequent changes in the allowance for credit losses as a credit loss expense for PCD assets. Furthermore, an entity should recognize interest income for PCD assets based on the effective interest rate, excluding the discount embedded in the purchase price that is attributable to the acquirer's assessment of credit losses at acquisition.

#### Disclosures

In an effort to increase users' understanding of underwriting standards and credit quality trends, ASU No. 2016-13 requires the current disclosure on credit

quality indicators in relation to the amortized cost of financing receivables to be further disaggregated by year of origination (or vintage). Entities that are not public business entities are not required to disclose the disaggregation by year of origination.

#### Available for Sale Debt Securities

Entities will now be required to present credit losses on available-for-sale debt securities as an allowance, rather than as a permanent write-down.

An entity will now be able to record reversals of credit losses on debt securities (in situations in which the estimate of credit declines) in current period net income. Thus, aligning the income statement recognition of credit losses with the reporting period in which changes occur. However, an entity may not record an allowance for credit losses exceeding the amount by which fair value is below amortized cost.

#### Purchased Debt Securities With Credit Deterioration

The allowance for credit losses for purchased available-for-sale debt securities with a more-than-insignificant amount of credit deterioration since origination is also determined in a similar manner to other available-for-sale debt securities. However, ASU No. 2016-13 requires an entity to add the initial allowance for credit losses to the purchase price, rather than reporting it as a credit loss expense. Entities record only subsequent changes in the allowance for credit losses as a credit loss expense. Furthermore, an entity should recognize interest income based on the effective interest rate, excluding the discount embedded in the purchase price that is attributable to the acquirer's assessment of credit losses at acquisition.

## **Troubled Debt Restructurings**

The ASU does not change the definition or derecognition guidelines for troubled debt restructurings (TDRs), but, rather, changes the impairment recognized on restructuring. Credit losses for TDRs now will be measured using the current expected credit loss model. The ASU eliminates the current GAAP requirement to use a discounted cash flow technique. Credit losses, including concessions given to a borrower under a TDR, will be recognized through an allowance account.

## **Applicability and Effective Date**

ASU No. 2016-13 affects entities holding financial assets and net investment in leases that are not accounted for at fair value through net income. It also affects loans, debt securities, trade receivables, net investments in leases, off-balance sheet credit exposures, reinsurance receivables, and any other financial assets not excluded from the scope that have the contractual right to receive cash.

Because there is diversity in practice in applying the incurred loss methodology, ASU No. 2016-13 will affect entities to varying degrees depending on the credit quality of the assets held by the entities, their duration, and how the entity applies current GAAP.

ASU No. 2016-13 is effective as follows:

	Fiscal Years Beginning After	Interim Periods Within Fiscal Years Beginning After
Public business entities that are SEC filers	December 15, 2019	December 15, 2019
All other public entities	December 15, 2020	December 15, 2020
All other entities, including not-for-profit entities and employee benefit plans within the scope of FASB ASC 960–965 on plan accounting	December 15, 2020	December 15, 2021

All entities may adopt the amendments in this ASU earlier as of the fiscal years beginning after December 15, 2018, including interim periods within those fiscal years.

## **Transition Resource Group**

Due to the potential for significant changes that may result from the issuance of the new standard, FASB has formed the Transition Resource Group (TRG) for Credit Losses to

- solicit, analyze, and discuss stakeholder issues arising from implementation of the new guidance.
- inform FASB about those implementation issues, which will help FASB determine what, if any, action will be needed to address those issues.
- provide a forum for stakeholders to learn about the new guidance from others involved with implementation.

The TRG will meet to discuss and share their views on potential implementation issues raised by concerned parties and, subsequent to each meeting, FASB will determine what actions, if any, will be taken on each issue. Refer to the page "Transition Resource Group for Credit Losses" on FASB's website for more information on this group and the status of their efforts, including meeting materials and meeting summaries.

## **Hedge Accounting**

#### Overview

Hedge accounting is the third phase in FASB's overall project on accounting for financial instruments. The objective of this project is to make targeted improvements to the hedge accounting model based on the feedback received from preparers, auditors, users, and other stakeholders. FASB has also noted it will consider opportunities to align with IFRS 9, *Financial Instruments*.

### Latest Developments

FASB staff are drafting a proposed ASU based on the tentative decisions reached by the board. Readers are encouraged to visit the "Technical Agenda" page under "Projects" at www.fasb.org for the latest developments regarding the hedge accounting phase.

### Conclusion

The extent of the effect of the new financial instruments standards will depend upon the relative significance of financial instruments to an entity's operations and financial positon as well as the entity's business strategy. To provide CPAs with guidance during this time of transition, the AICPA's Financial Reporting Center (FRC) offers invaluable resources on the topic. In addition, the FRC includes a list of conferences, webcasts, and products to keep you informed on the latest developments in accounting for financial instruments. Refer to www.aicpa.org/InterestAreas/FRC/AccountingFinancialReporting/Financial-Instruments/Pages/financial-instruments.aspx to stay updated on the latest information available on accounting for financial instruments.

# Appendix H

# Schedule of Changes Made to the Text From the Previous Edition

This appendix is nonauthoritative and is included for informational purposes only.

#### As of July 1, 2016

This schedule of changes identifies areas in the text and footnotes of this guide that have that have changed since the previous edition. Entries in the table of this appendix reflect current numbering, lettering (including that in appendix names), and character designations that resulted from the renumbering or reordering that occurred in the updating of this guide.

Reference	Change
General	Guide content included in shaded areas and "Guidance Update" boxes within the chapters have been updated to appropriately reflect guidance not yet effective as of the date of the guide. See the preface of this guide for more explanation to this "dual guidance" treatment.
General	Editorial changes, including rephrasing, have been made in this guide to improve readability where necessary.
Preface	Updated.
Paragraph 1.25	Revised to reflect the issuance of FDIC's final regulation <i>Assessments</i> .
Footnote 4 in paragraph 1.40	Revised to reflect the issuance of final regulation Margin and Capital Requirements for Covered Swap Entities.
Paragraph 1.62	Revised for the passage of time.
Paragraphs 1.98 and 1.114	Revised to reflect the issuance of the PCAOB's Reorganization of PCAOB Auditing Standards and Related Amendments to PCAOB Auditing Standards and Rules.
Footnote 25 in paragraph 1.118	Added to reflect the issuance of Regulation AB II.
Former paragraph 2.18	Deleted for the passage of time.

Reference	Change
Former footnote 1 to heading before paragraph 2.24	Deleted to reflect the issuance of the National Credit Union Administration's (NCUA) final regulation Risk-Based Capital.
Paragraph 2.45 and footnotes 3–5 in paragraphs 2.45 and 2.47	Revised to reflect the issuance of NCUA's final regulation <i>Corporate Credit Unions</i> .
Former footnote 1 in paragraph 4.16	Deleted to reflect the issuance of FASB Accounting Standards Update (ASU) No. 2016-01, Financial Instruments—Overall (Subtopic 825–10): Recognition and Measurement of Financial Assets and Financial Liabilities.
Former footnote 3 in paragraph 4.33	Deleted for the passage of time.
Footnote 3 in paragraph 4.33	Added to reflect the issuance of a revised HUD Handbook.
Former footnote 8 in paragraph 5.36	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Paragraph 5.193	Revised to reflect the issuance of FASB ASU No. 2014-10, Development Stage Entities (Topic 915): Elimination of Certain Financial Reporting Requirements, Including an Amendment to Variable Interest Entities Guidance in Topic 810, Consolidation.
Former footnotes 1 and 3 in paragraphs 6.01 and 6.11	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Footnote 2 in paragraph 6.12	Added to reflect the issuance of FASB proposed ASU Statement of Cash Flows (Topic 230): Restricted Cash (a consensus of the FASB Emerging Issues Task Force).
Footnote 3 in paragraph 6.17	Added to reflect the issuance of FASB proposed ASU Statement of Cash Flows (Topic 230): Classification of Certain Cash Receipts and Cash Payments (a consensus of the FASB Emerging Issues Task Force).
Former footnote 4 in paragraph 6.19	Deleted to reflect the issuance of FASB ASU No. 2016-01.

Reference	Change
Footnote 4 in paragraph 6.21	Added to reflect the issuance of FASB proposed ASU Statement of Cash Flows (Topic 230): Classification of Certain Cash Receipts and Cash Payments (a consensus of the FASB Emerging Issues Task Force).
Former footnote 5 in paragraph 6.23 and former footnote 8 to heading before paragraph	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Footnote 12 in paragraph 7.134	Revised to reflect the issuance of FASB ASU No. 2015-01, Income Statement—Extraordinary and Unusual Items (Subtopic 225-20): Simplifying Income Statement Presentation by Eliminating the Concept of Extraordinary Items.
Footnote 5 in paragraph 8.64	Revised for the passage of time.
Former paragraph 8.70	Deleted to reflect the issuance of the Office of the Comptroller of the Currency's (OCC) revised "Residential Real Estate Lending" booklet to the Comptroller's Handbook.
Former footnotes 9–10 in paragraph 8.89	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Footnote 9 in paragraph 8.90	Revised for the passage of time.
Former footnotes 12–15 and 19 in paragraphs 8.90–.92, 8.96, and 8.117	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Paragraph 8.137	Revised to reflect the issuance of FASB ASU No. 2014-04, Receivables—Troubled Debt Restructurings by Creditors (Subtopic 310-40): Reclassification of Residential Real Estate Collateralized Consumer Mortgage Loans upon Foreclosure (a consensus of the FASB Emerging Issues Task Force) and FASB ASU No. 2014-14, Receivables—Troubled Debt Restructurings by Creditors (Subtopic 310-40): Classification of Certain Government-Guaranteed Mortgage Loans upon Foreclosure (a consensus of the FASB Emerging Issues Task Force).

Reference	Change
Former footnote 21 in former paragraph 8.137	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Former footnote 22 to heading before paragraph 8.140	Deleted to reflect the issuance of FASB ASU No. 2016-02, <i>Leases (Topic 842)</i> .
Former footnote 23 to heading before paragraph 8.141	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Footnote 15 in paragraph 8.153	Revised to reflect the issuance of FASB ASU No. 2016-01.
Former footnotes 26–27 in paragraphs 8.153–.154	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Former footnote 28 in paragraph 8.169	Deleted to reflect the issuance of FASB ASU No. 2016-02.
Former footnote 29 in paragraph 8.170	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Footnote 19 to heading before paragraph 8.185	Revised to reflect the issuance of Statement on Auditing Standards (SAS) No. 130, An Audit of Internal Control Over Financial Reporting That Is Integrated With an Audit of Financial Statements (AICPA, Professional Standards, AU-C sec. 940).
Former footnote 34 in paragraph 8.208	Deleted for the passage of time.
Paragraph 9.29	Revised to reflect the issuance of FDIC Financial Institution Letter [FIL]-49-2015.
Former footnote 5 to heading before paragraph 9.39	Deleted to reflect the issuance of FASB ASU No. 2016-13, Financial Instruments—Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments.
Former footnotes 6–7 in paragraphs 9.44–.45	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Former footnote 4 to heading before paragraph 10.20	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Paragraphs 10.33, 10.35–.36, and 10.38	Revised to reflect the issuance of FASB ASU No. 2014-11, Transfers and Servicing (Topic 860): Repurchase-to-Maturity Transactions, Repurchase Financings, and Disclosures.

Reference	Change
Former footnotes 5–7 in paragraphs 10.44, 10.48, and 10.54; former footnotes 8–9 to headings before paragraphs 10.74 and 10.79	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Paragraph 10.80	Revised to reflect the issuance of FASB Maintenance Update 2015-11, Inventory (Topic 330): Simplifying the Measurement of Inventory.
Paragraphs 10.88 and 10.94 and footnote 4 in paragraph 10.94	Added to reflect the issuance of FASB ASU No. 2014-11.
Footnote 5 to heading before paragraph 10.97	Revised to reflect the issuance of FASB ASU No. 2015-02, Consolidation (Topic 810): Amendments to the Consolidation Analysis.
Footnote 6 in paragraph 10.97	Revised to reflect updates to the FASB Technical Agenda.
Paragraphs 10.101, 10.108, and 10.111	Revised to reflect the issuance of FASB ASU No. 2014-07, Consolidation (Topic 810): Applying Variable Interest Entities Guidance to Common Control Leasing Arrangements (a consensus of the Private Company Council).
Footnote 1 in paragraph 11.01	Revised for the passage of time.
Paragraph 11.16	Revised to reflect updates to the Bank Accounting Advisory Series.
Paragraph 11.18	Revised to reflect the issuance of FASB ASU Nos. 2014-04 and 2014-14.
Paragraph 11.21	Added to reflect the issuance of FASB ASU No. 2014-14.
Paragraph 11.23	Revised to reflect the issuance of FASB ASU No. 2014-04.
Paragraphs 11.24 and 11.30	Revised to reflect the issuance of FASB ASU No. 2014-08, Presentation of Financial Statements (Topic 205) and Property, Plant, and Equipment (Topic 360): Reporting Discontinued Operations and Disclosures of Disposals of Components of an Entity.
Former paragraphs 11.30–.31	Deleted to reflect the issuance of FASB ASU No. 2014-08.

Reference	Change
Paragraphs 11.31–11.33	Added to reflect the issuance of FASB ASU No. 2014-08.
Footnote 2 to heading before paragraph 11.44	Revised for the passage of time.
Paragraph 11.45	Revised to reflect the issuance of FASB ASU No. 2014-08.
Former footnote 1 in paragraph 12.01	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Paragraph 12.11 and footnote 1	Revised to reflect the issuance of NCUA final regulation Federal Credit Union Ownership of Fixed Assets.
Paragraph 12.12	Revised to reflect the issuance of Regulatory Capital Rules.
Former paragraph 12.13 and former footnote 3	Deleted to reflect the issuance of Bulletin OCC 2015-38.
Former footnote 4 to heading before paragraph 12.17	Deleted to reflect the issuance of FASB ASU No. 2016-02.
Footnote 2 in paragraph 12.21	Revised to reflect the issuance of FASB ASU No. 2015-02.
Footnote 3 in paragraph 12.23	Added to reflect the issuance of FASB ASU No. 2015-05.
Footnote 4 in paragraph 12.23	Added to reflect the issuance of FASB proposed ASU Intangibles—Goodwill and Other (Topic 350): Simplifying the Accounting for Goodwill Impairment.
Paragraph 12.26 and former footnote 6	Revised to reflect the issuance of NCUA final regulation Federal Credit Union Ownership of Fixed Assets; Deleted.
Former footnotes 7–11 in paragraphs 12.28–.30, 12.32, and 12.34.	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Footnote 5 in paragraph 12.35	Added to reflect the issuance of FASB proposed ASU Intangibles—Goodwill and Other (Topic 350): Simplifying the Accounting for Goodwill Impairment.
Paragraph 12.35	Revised to reflect the issuance of FASB ASU No. 2014-02.
Footnotes 8–9 in paragraphs 12.36 and 12.47	Added to reflect the issuance of FASB proposed ASU Intangibles—Goodwill and Other (Topic 350): Simplifying the Accounting for Goodwill Impairment.

Reference	Change
Former footnote 14 in paragraph 12.57	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Paragraph 12.66	Revised for the passage of time.
Former footnote 16 in paragraph 12.71	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Footnote 1 to header before paragraph 13.04	Revised for the passage of time.
Footnote 3 to header before paragraph 13.33	Revised to reflect the issuance of updates incorporated into the FDIC's Guidance on Identifying, Accepting, and Reporting Brokered Deposits—Frequently Asked Questions.
Former footnote 5 in paragraph 13.46	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Paragraph 14.04	Revised to reflect the issuance of FASB ASU No. 2014-11.
Paragraph 14.06	Added to reflect the issuance of FASB ASU No. 2014-11.
Paragraph 14.09	Revised for changes in industry practice or developments that are not related to an issuance of authoritative guidance.
Paragraphs 14.27–.28	Revised to reflect the issuance of FASB ASU No. 2014-11.
Former paragraph 14.28	Deleted to reflect the issuance of FASB ASU No. 2014-11.
Paragraph 14.29	Revised to reflect the issuance of FASB ASU No. 2014-11.
Paragraph 14.30	Added to reflect the issuance of FASB ASU No. 2014-11.
Paragraph 14.31	Revised to reflect the issuance of FASB ASU No. 2014-11.
Paragraph 14.36	Revised to reflect the issuance of FASB Maintenance Update 2014-20.
Paragraph 14.37	Revised to reflect the issuance of FASB ASU No. 2014-11.
Former paragraph 14.37	Deleted to reflect the issuance of FASB ASU No. 2014-11.

Reference	Change
Paragraph 14.44	Added to reflect the issuance of FASB ASU No. 2014-11.
Former footnote 5 in paragraph 15.55	Deleted to reflect the issuance of FASB ASU No. 2016-02.
Former footnotes 7 and 9 in paragraphs 16.25 and 16.28	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Former footnote 1 in paragraph 17.09	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Paragraphs 17.10–.11	Revised to reflect the issuance of Regulatory Capital Rules.
Footnote 1 to header before paragraph 17.11	Revised to reflect the issuance of interagency guidance.
Former footnote 9 to header before paragraph 17.47	Deleted to reflect the issuance of the NCUA's final regulation <i>Risk-Based Capital</i> .
Footnote 16 in paragraph 17.66, paragraph 17.67, and footnotes 20–21 in paragraph 17.68	Revised to reflect the issuance of NCUA's <i>Corporate Credit Union</i> regulation amendments.
Paragraph 18.70	Revised for the passage of time.
Footnote 5 to header before paragraph 18.79	Revised for the issuance of FASB ASU No. 2016-13.
Paragraph 18.81	Revised for the issuance of FASB ASU No. 2014-03, Derivatives and Hedging (Topic 815): Accounting for Certain Receive-Variable, Pay-Fixed Interest Rate Swaps—Simplified Hedge Accounting Approach (a consensus of the Private Company Council).
Former footnote 8 in paragraph 18.92	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Former footnote 9 in paragraph 18.96	Deleted for the passage of time.
Former footnote 10 in paragraph 18.97	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Footnote 1 in paragraph 19.01	Revised for the passage of time.

Reference	Change
Paragraph 19.12	Revised to reflect the issuance of FASB ASU No. 2014-18, Business Combinations (Topic 805): Accounting for Identifiable Intangible Assets in a Business Combination (a consensus of the Private Company Council) and FASB ASU No. 2016-03, Intangibles—Goodwill and Other (Topic 350), Business Combinations (Topic 805), Consolidation (Topic 810), Derivatives and Hedging (Topic 815): Effective Date and Transition Guidance (a consensus of the Private Company Council).
Footnote 5 in paragraph 19.34	Revised for the passage of time.
Former footnote 5 to heading before paragraph 20.49	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Footnote 6 to heading before paragraph 20.49	Added to reflect the issuance of FASB's open project Disclosure Framework—Disclosure Review: Fair Value Measurement.
Former footnote 6 to heading before paragraph 20.61	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Footnote 1 to paragraph 21.01	Revised to reflect the issuance of SAS No. 130 and Statement on Standards for Attestation Engagements No. 18, Attestation Standards: Clarification and Recodification.
Former footnotes 4 to heading before paragraph 22.51 and former footnote 5 in paragraph 22.52	Deleted to reflect the issuance of FASB ASU No. 2016-01.
Paragraph 22.61	Revised to reflect the issuance of FASB ASU No. 2014-07.
Footnote 1 to General Note Box	Revised to reflect the issuance of the PCAOB's Reorganization of PCAOB Auditing Standards and Related Amendments to PCAOB Auditing Standards and Rules.

Reference	Change
Paragraph 23.13	Revised as result of the issuance of SAS No. 131, Amendment to Statement on Auditing Standards No. 122 Section 700, Forming an Opinion and Reporting on Financial Statements (AICPA, Professional Standards AU-C sec. 700).
Paragraphs 23.32–.33	Revised to reflect the issuance of SSARS No. 21, Statement on Standards for Accounting and Review Services: Clarification and Recodification (AICPA, Professional Standards).
Footnote 1 to appendix A to chapter 23	Revised to reflect the issuance of reproposed auditing reporting standard, <i>The Auditor's Report on an Audit of Financial Statements When the Auditor Expresses and Unqualified Opinion</i>
Appendix A to chapter 23	Revised to reflect the issuance of the PCAOB's Reorganization of PCAOB Auditing Standards and Related Amendments to PCAOB Auditing Standards and Rules.
Former appendixes A and B	Deleted.
Appendixes A, B, C, D and E	Updated.
Appendixes F and G	Added.
Index of Pronouncements and Other Technical Guidance	Updated.
Subject Index	Updated.

# **Index of Pronouncements and Other Technical Guidance**

### Α

Title	Paragraphs
AR Section	
60, Framework for Performing and Reporting on Compilation and Review Engagements	23.32
80, Compilation of Financial Statements	23.32
90, Review of Financial Statements	23.32
AT Section	
201, Agreed-Upon Procedures Engagement	1.118, 23.30
301, Financial Forecasts and Projections	23.30
501, An Examination of an Entity's Internal Control Over Financial Reporting That Is Integrated With an Audit of Its Financial Statements	1.98, 1.115, 5.75, 23.26–.27
601, Compliance Attestation	1.118, 4.36
801, Reporting on Controls at a Service Organization	14.53, 21.22, 21.25
AU-C Section	
200, Overall Objectives of the Independent Auditor and the Conduct of an Audit in Accordance With Generally Accepted Auditing Standards	5.01–.02, 5.04–.09, 5.126, 13.49
210, Terms of Engagement	17.107
220, Quality Control for an Engagement Conducted in Accordance With Generally Accepted Auditing Standards	5.111, 7.142, 20.65
230, Audit Documentation	1.101, 5.102–.109
Interpretation No. 1, "Providing Access to or Copies of Audit Documentation to a Regulator"	1.101, 5.256
240, Consideration of Fraud in a Financial Statement Audit	1.98, 5.32, 5.123–.151, 5.238, 5.256, 7.150
250, Consideration of Laws and Regulations in an Audit of Financial Statements	5.152–.158, 5.160–.162, 5.239

©2016, AICPA AAG-DEP AU-C

Title	Paragraphs
260, The Auditor's Communication With Those Charged With Governance	1.98, 5.70, 9.111
265, Communicating Internal Control Related Matters Identified in an Audit	1.98, 5.71–.75, 9.13, 23.25–.26
300, Planning an Audit	5.1415, 5.111, 5.141, 7.142, 17.107, 20.65
315, Understanding the Entity and Its Environment and Assessing the Risks of Material Misstatement	$\begin{array}{c} 5.15, 5.2128, \\ 5.3134, 5.5865, \\ 5.7782, 5.120, \\ 5.123, 5.129, 5.133, \\ 5.137, 5.168, 5.215, \\ 5.237238, \\ 6.3536, 6.38, 7.139, \\ 7.144, 7.148149, \\ 7.153, 8.183, 8.185, \\ 8.190, 9.6465, \\ 9.6869, 10.115, \\ 10.119, 10.123, \\ 11.5758, 11.60, \\ 11.65, 12.7980, \\ 12.8788, 13.48, \\ 13.5051, 13.53, \\ 13.70, 14.49, 14.54, \\ 14.56, 14.68, \\ 15.5960, 15.63, \\ 15.71, 16.3234, \\ 17.92, 17.95, 17.98, \\ 17.108, 18.101, \\ 19.32, 19.34, 20.64, \\ 21.1819, 21.22, \end{array}$
320, Materiality in Planning and Performance of an Audit	5.1619, 5.24, 8.208209, 13.62

AAG-DEP AU-C ©2016, AICPA

Title	Paragraphs
330, Performing Audit Procedures in Response to Assessed Risks and Evaluating the Audit Evidence Obtained	$\begin{array}{c} 5.15, 5.8293, 5.98, \\ 5.120, 5.123, 5.140, \\ 5.168, 5.238, \\ 6.3839, 6.41, \\ 7.148, 7.151153, \\ 7.155156, 7.159, \\ 8.199200, \\ 8.203204, \\ 8.207208, 9.65, \\ 9.7273, 9.76, 9.82, \\ 10.123124, 10.126, \\ 11.6062, 11.65, \\ 12.8183, 13.51, \\ 13.54, 13.56, \\ 13.5859, 13.62, \\ 13.68, 13.70, \\ 14.5658, 14.68, \\ 15.6164, 15.66, \\ 15.69, 15.71, \\ 16.3435, 17.100, \\ 17.104, 17.110, \\ 17.112, 17.114, \\ 18.101, 19.3334, \\ 20.64, 21.28, 23.02 \end{array}$
402, Audit Considerations Relating to an Entity Using a Service Organization	5.67, 5.120–.122, 7.153, 8.198, 8.218–.220, 14.53, 15.63
450, Evaluation of Misstatements Identified During the Audit	5.99–.101, 5.147, 9.110, 23.02
500, Audit Evidence	5.111, 5.116118, 7.156, 8.204, 8.209, 8.229, 9.89, 11.6668, 13.59, 14.61, 14.68, 15.66, 17.92, 18.101, 19.3437
501, Audit Evidence—Specific Considerations for Selected Items	7.103, 7.156, 14.68, 18.101, 19.34
505, External Confirmations	8.209211, 13.6063, 15.66
520, Analytical Procedures	5.9497, 8.201, 8.203204, 9.109, 13.68, 13.70, 15.69, 15.71

©2016, AICPA AAG-DEP AU-C

Title	Paragraphs
530, Audit Sampling	8.208, 13.62, 13.64
540, Auditing Accounting Estimates, Including Fair Value Accounting Estimates and Related Disclosures	7.148149, 7.157161, 8.229, 9.7778, 9.80, 9.85, 9.110, 11.65, 12.8788, 14.63, 14.68, 18.101, 19.34, 20.64, 20.66
$550, Related\ Parties$	5.36–.37
570, The Auditor's Consideration of an Entity's Ability to Continue as a Going Concern	5.163, 5.165–.176, 23.08–.09
Interpretation No. 1, "Definition of Substantial Doubt About an Entity's Ability to Continue as a Going Concern"	5.166
Interpretation No. 2, "Definition of Reasonable Period of Time"	5.166
Interpretation No. 4, "Consideration of Financial Statement Effects"	5.174
580, Written Representations	5.180189
600, Special Considerations—Audits of Group Financial Statements	5.14
620, Using the Work of an Auditor's Specialist	5.20, 5.110115, 7.142, 8.229, 9.86, 9.90, 9.92, 19.35, 19.3839, 20.65
Interpretation No. 1, "The Use of Legal Interpretations As Audit Evidence to Support Management's Assertion That a Transfer of Financial Assets Has Met the Isolation Criterion in Paragraphs 7–14 of Financial Accounting Standards Board Accounting Standards Codification 860-10-40"	5.118119, 10.127
700, Forming an Opinion and Reporting on Financial Statements	5.37, 5.101, 5.147, 5.254, 23.01–.02, 23.06, 23.13, 23.15, 23.17, 23.29
705, Modifications to the Opinion in the Independent Auditor's Report	5.37, 5.189, 5.254, 7.162, 14.64, 20.67, 23.01, 23.0809, 23.1521, 23.29

Title	Paragraphs
706, Emphasis-of-Matter Paragraphs and Other-Matter Paragraphs in the Independent Auditor's Report	5.254, 23.01, 23.0708, 23.1011, 23.1314, 23.29
720, Other Information in Documents Containing Audited Financial Statements	5.190, 15.77
725, Supplementary Information in Relation to the Financial Statements as a Whole	5.191, 15.76–.77
730, Required Supplementary Information	5.192, 15.77
800, Special Considerations—Audits of Financial Statements Prepared in Accordance With Special Purpose Frameworks	23.22
805, Special Considerations—Audits of Single Financial Statements and Specific Elements, Accounts, or Items of a Financial Statement	23.29
806, Reporting on Compliance With Aspects of Contractual Agreements or Regulatory Requirements in Connections With Audited Financial Statements	1.118, 15.74, 23.36–.37
935, Compliance Audits	23.06
940, An Audit of Internal Control Over Financial Reporting that is Integrated With an Audit of Financial Statements	5.75
Audit and Accounting Guides (AAG)	
Brokers and Dealers in Securities	17.01
Code of Professional Conduct	1.94
Government Auditing Standards	4.33
Investment Companies	21.30
Life and Health Insurance Entities	22.12, 22.35, 22.37, 22.42, 22.47, 22.64, 22.69
Property and Liability Insurance Entities	22.12, 22.69
Audit Guides	
Audit Sampling	13.62
Special Considerations in Auditing Financial Instruments	7.156, 18.04, 18.101, 19.34

©2016, AICPA AAG-DEP AUD

## В

Title	Paragraphs
Basel Committee on Banking Supervision	
Principles for Sound Liquidity Risk Management and Supervision	18.73

# C

Title	Paragraphs
CFR Title	
12 Banks and Banking	1.72, 5.221
12 CFR 3, subpart F	1.76
12 CFR 3.10	1.73
12 CFR 5.37	12.13
12 CFR 7.100	11.06
12 CFR 9	21.12
12 CFR 9.9	21.06
12 CFR 38.85(a)(2)	11.15
12 CFR 150.440	21.06
12 CFR 162	1.107
12 CFR 162.4	1.94
12 CFR 208	17.18
12 CFR 208.21	12.13
12 CFR 215	5.38
12 CFR 215.4	5.38
12 CFR 217, subpart F	1.76
12 CFR 217.10	1.73
12 CFR 223	18.72
12 CFR 223.33(b)	18.72
12 CFR 225	17.18
12 CFR 225.5	1.9394, 17.18
12 CFR 238.5	1.94, 1.96, 1.109
12 CFR 324, subpart F	1.76
12 CFR 324.10	1.73

**AAG-DEP BAS** 

Title	Paragraphs
12 CFR 337	13.34
12 CFR 360.6	10.13
12 CFR 363	1.9293, 1.9596,
	1.99–.101, 1.103–.104,
	1.107109,
0.111.40	1.112113
Guideline 13	1.101
Guideline 17	1.104
Guideline 18A	1.116
Guidelines 27 to 35	1.100
12 CFR 363.2	1.96
12 CFR 363.3(b)	1.112
12 CFR 363.3(d)	1.9798
12 CFR 363.4	1.96, 1.98–.99
12 CFR 363.5	1.100
12 CFR 390, Subpart R	1.107
12 CFR 390.322	1.94
12 CFR 700	2.24
12 CFR 701.32(b)	13.38
12 CFR 702	2.24, 2.28
12 CFR 702.2	2.24
12 CFR 704	17.67
12 CFR 704.2	2.45, 17.66–.67
12 CFR 704.3(d)	17.69
12 CFR 704.3(d)(2)	17.69
12 CFR 704.4	2.49, 17.69, 17.76
12 CFR 704.4(h)(1)	2.51
12 CFR 704.15	2.61
12 CFR 712.3	2.60, 2.62
12 CFR 715	2.08, 2.59, 13.39
12 CFR 741	2.24
12 CFR 741.4c	12.60
12 CFR 741.4d	12.60
12 CFR 747	2.24

©2016, AICPA AAG-DEP CFR

## **Depository and Lending Institutions**

Title	Paragraphs
12 CFR 1761(d)	2.08
31 CFR Chapter X	1.20
38 CFR 36.4302(a)(4)	10.17
Code of Federal Regulations	8.67

## D

Title	Paragraphs
Dodd-Frank Wall Street Reform and Consumer Protection Act	
Section 165	17.27
Section 165(d)	1.60
Section 165(i)	1.66
Section 171	17.20
Section 301-326	1.54
Section 610	1.4243
Section 619	1.41
Section 941	1.52
Section 942(a)	1.51
Section 945	1.50
Section 998G	1.113
Title I, "Financial Stability"	1.35
Title VI, "Improvements to Regulation"	1.35

# F

Title	Paragraphs
FASB ASC	
205, Presentation of Financial Statements	
205-40	17.34–.35
210, Balance Sheet	14.39, 14.48
210-20	10.88, 14.39–.41, 14.44, 18.87–.88, 18.100

Title	Paragraphs
220, Comprehensive Income	17.02
220-10	7.104–.106, 17.08–.09, 17.54, 17.71
230, Statement of Cash Flows	6.12
230-10	6.12–.13, 6.15, 6.17–.19, 6.21, 6.23–.26, 6.29, 10.81, 18.89
235, Notes to Financial Statements	
235-10	8.175, 10.89
250, Accounting Changes and Error Corrections	8.119–.120, 11.37, 20.30
250-10	9.62, 11.37, 20.30, 22.45
275, Risks and Uncertainties	5.205
275-10	5.193, 5.196–.201, 5.204, 5.206
280, Segment Reporting	11.40
280-10	5.34, 5.210
310, Receivables	8.118, 8.132, 8.137, 9.19, 10.26, 11.20
310-10	7.98, 8.73, 8.81, 8.89, 8.91, 8.9394, 8.99100, 8.136, 8.139, 8.149166, 8.153, 8.227, 9.19, 9.3942, 9.4548, 9.53, 9.55, 9.5861, 9.105, 10.74, 10.7980, 11.2223, 11.4243
310-20	7.109111, 7.113114, 8.56, 8.93, 8.99109, 8.146, 8.151, 8.224, 19.1819
310-30	7.114, 7.125, 8.110–.121, 8.136, 9.05, 9.56, 9.60, 19.20–.21

©2016, AICPA AAG-DEP FAS

Title	Paragraphs
310-40	$7.117, 7.123124,\\ 8.81, 8.123,\\ 8.125138,\\ 8.163164, 8.227,\\ 9.05, 9.55, 11.1721,\\ 11.23, 11.28$
320, Investments—Debt and Equity Securities	6.01, 6.11, 6.23, 7.20, 7.53, 7.68, 7.83, 7.88–.89, 7.117, 7.122, 8.89–.90, 8.133, 9.45, 10.21, 10.44, 10.48, 10.54, 12.01, 12.28, 12.34, 16.25, 16.28, 22.52
320-10	7.83, 7.85, 7.87, 7.90, 7.92–.95, 7.98–.99, 7.101–.102, 7.108–.109, 7.114, 7.126–.129, 7.132–.135, 7.137, 8.117, 8.133, 12.32, 17.09, 22.52
323, Investments—Equity Method and Joint Ventures	11.55, 12.71
323-10	5.36, 12.71
323-20	7.94
325, Investments—Other	
325-40	7.94, 7.114
340, Other Assets and Deferred Costs	
340-20	22.20, 22.2732
350, Intangibles—Goodwill and Other	11.37, 19.15
350-20	12.35, 12.50
350-30	11.37, 12.39–.47, 12.49–.50, 19.29
350-40	12.23, 22.37
360, Property, Plant and Equipment	11.05, 11.28
360-10	8.135, 11.17, 11.20, 11.24–.40, 12.18, 12.25, 12.44–.45, 12.59, 19.26
360-20	11.4445, 11.70

AAG-DEP FAS ©2016, AICPA

Title	Paragraphs
405, Liabilities	
405-20	15.49
410, Asset Retirement and Environmental Obligations	
410-20	12.5456
420, Exit or Disposal Cost Obligations	12.52
420-10	12.52
440, Commitments	
440-10	15.37
450, Contingencies	8.103, 8.118, 8.148, 15.56, 18.86, 21.16
450-20	5.197200, 8.75, 8.180181, 9.19, 9.27, 9.39, 9.4142, 9.4952, 9.58, 9.60, 18.85
460, Guarantees	8.178, 10.108, 12.02, 15.20, 15.46, 18.85, 21.24
460-10	8.178–.181, 15.46, 18.85–.86
470, Debt	8.108
470-10	13.45, 15.38–.39
470-50	15.50–.51
470-60	15.51
480, Distinguishing Liabilities From Equity	15.41, 15.45, 17.24
480-10	15.4142, 15.44, 17.2324
505, Equity	
505-10	17.51
605, Revenue Recognition	8.01, 11.01, 13.01, 21.01
605-20	22.58
740, Income Taxes	16.1819, 16.36
740-10	16.1821, 16.2324, 16.2628, 16.30
740-20	7.89, 16.28

Title	Paragraphs
740-30	16.29
805, Business Combinations	12.06, 19.02, 19.08, 19.14, 19.29
805-10	19.01–.02, 19.08–.11, 19.29
805-20	12.07, 12.48, 19.10, 19.12–.13, 19.15
805-30	19.10, 19.14–.16, 19.22–.24
805-50	19.03, 19.29–.30
810, Consolidation	1.53, 10.102, 12.20–.21, 15.54, 15.58, 17.12, 19.02
810-10	7.115, 10.97103, 10.105, 10.107113, 12.21, 12.72, 17.15, 22.6061, 22.63
810-40	12.22
815, Derivatives and Hedging	7.20, 7.84, 8.90, 8.150, 8.175, 10.21, 14.40, 15.52, 15.58, 15.62, 18.01, 18.51, 18.79, 18.81, 18.85–.86, 18.95, 18.99, 20.23, 20.57–.58, 21.24
815-10	7.118, 8.89, 8.104, 8.145147, 8.150, 9.57, 10.27, 10.48, 10.94, 14.4041, 15.52, 18.02, 18.14, 18.5152, 18.56, 18.79, 18.81, 18.83, 18.88, 18.92, 18.9598, 18.100
815-15	18.51, 18.97
815-20	15.53, 17.09, 18.81–.82, 18.95, 18.98
815-25	7.83, 8.90, 10.21
815-30	17.09
815-40	18.51

Title	Paragraphs
820, Fair Value Measurement	7.91, 7.154, 8.135, 10.25, 10.92, 12.75, 15.57, 18.80, 18.84, 19.17, 20.01, 20.21, 20.35, 20.41, 20.46–.47, 20.54
820-10	8.174, 10.25, 10.96, 12.76, 18.80, 20.02, 20.04–.18, 20.20–.21, 20.24, 20.28–.39, 20.50, 20.60
825, Financial Instruments	5.205, 6.01, 7.84, 8.92, 8.174, 10.24, 13.46, 15.57, 20.01, 20.23, 20.61, 22.53
825-10	5.36, 7.99, 7.136, 8.92, 8.170–.172, 9.57, 10.24, 10.85, 10.96, 13.46, 14.45, 18.93, 18.99, 20.57–.58, 20.61–.63
830, Foreign Currency Matters	16.28
830-20	15.53
835, Interest	8.138
835-20	10.16
835-30	8.96, 8.138, 15.39, 15.48
840, Leases	8.140, 8.217, 9.45, 10.108, 15.55
840-10	8.169, 12.25
840-20	12.25
840-30	12.18, 12.24
840-40	8.140, 11.45
850, Related Party Disclosures	8.177, 10.108, 13.45
852, Reorganizations	12.29
855, Subsequent Events	
855-10	5.196, 5.204, 17.56, 17.72

©2016, AICPA AAG-DEP FAS

Title	Paragraphs
860, Transfers and Servicing	1.53, 5.119, 7.131, 8.56, 8.182, 8.190, 8.196, 9.103, 10.07, 10.30, 10.38, 10.50, 10.73, 10.86, 10.114, 13.44, 14.05, 14.27, 14.42, 14.66, 15.47, 15.72, 19.27
860-10	7.83, 7.87–.88, 7.119, 8.90, 8.93–.94, 10.21, 10.30–.39, 10.43–.45, 10.50, 10.71–.72, 10.86, 10.92–.93, 14.27–.38, 15.19, 15.46
860-20	7.90, 7.118, 8.89, 10.37, 10.43–.48, 10.80, 10.92–.94
860-30	8.94, 8.167, 10.35, 10.37, 10.70–.73, 10.87–.88, 14.43–.44, 14.46, 14.66, 15.34–.56
860-50	4.16, 10.52–.63, 10.66–.67, 10.69, 10.86, 10.89, 18.93, 19.27
942, Financial Services—Depository and Lending	6.28, 8.01, 11.01, 13.01, 21.01
942-10	17.56, 17.72
942-210	6.33, 14.39, 22.65, 22.67–.68
942-230	6.15, 6.28
942-305	6.3031
942-310	8.142–.144, 9.44, 12.57
942-320	7.111, 7.129–.131, 7.137
942-325	7.120, 12.28–.30, 12.34, 12.61, 12.63–.64
942-360	12.17

Title	Paragraphs
942-405	7.121, 13.41, 13.43–.44
942-470	13.46, 15.34, 15.36–.37, 15.47, 16.26
942-505	17.28–.32, 17.37–.38, 17.45–.46, 17.56, 17.59, 17.72–.74, 17.86, 17.88
942-605	21.15, 22.57
942-740	16.30
942-810	17.1112, 17.14
942-825	8.175176
944, Financial Services—Insurance	22.01, 22.12, 22.30, 22.42, 22.57, 22.63
944-20	22.13-22.15
944-30	22.18, 22.20, 22.2226, 22.3335, 22.37, 22.3941, 22.44, 22.4748, 22.50
944-80	22.63
944-320	22.54
944-605	22.13, 22.17
944-720	22.3334, 22.36
948, Financial Services—Mortgage Banking	6.23, 8.146, 10.20, 10.81
948-10	17.81–17.83
948-310	7.88, 8.90, 8.145, 10.21–.22, 10.25–.29, 10.83, 18.92
948-320	18.81
958, Not-For-Profit Entities	
958-605	12.7476
970, Real Estate—General	8.139, 11.41
970-10	11.46
970-323	11.5152, 11.54

©2016, AICPA AAG-DEP FAS

Title	Paragraphs
970-340	11.4749
970-360	11.4546, 11.49
970-605	11.19
970-810	11.55
985, Software	
985-20	12.23
FASB ASU	
No. 2009-16, Transfers and Servicing (Topic 860): Accounting For Transfers of Financial Assets	10.13
No. 2009-17, Consolidations (Topic 810): Improvements to Financial Reporting by Enterprises Involved With Variable Interest Entities	10.13
No. 2010-26, Financial Services—Insurance (Topic 944): Accounting for Costs Associated With Acquiring or Renewing Insurance Contracts	22.18, 22.45, 22.47
No. 2013-01, Balance Sheet (Topic 210): Clarifying the Scope of Disclosures About Offsetting Assets and Liabilities	18.98
No. 2014-02 Intangibles—Goodwill and Other (Topic 350): Accounting for Goodwill (a consensus of the Private Company Council)	12.35, 19.12
No. 2014-07, Consolidation (Topic 810): Applying Variable Interest Entities Guidance to Common Control Leasing Arrangements	10.101, 22.61
No. 2014-09, Revenue from Contracts With Customers (Topic 606): Summary and Amendments That Create Revenue from Contracts With Customers	8.01, 11.01, 13.01, 21.01, 22.01, Appendix E
No. 2014-18, Business Combinations (Topic 805): Accounting for Identifiable Intangible Assets in a Business Combination	19.12
No. 2016-01, Financial Instruments-Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities	Appendix G
No. 2016-02, Leases (Topic 842)	Appendix F
No. 2016-13, Financial Instruments - Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments	Appendix G

AAG-DEP FAS ©2016, AICPA

Title	Paragraphs
FASB SFAC No. 6, Elements of Financial Statements	15.48
FASB SFAS	
No. 118, Accounting by Creditors for Impairment of a Loan-Income Recognition and Disclosures- an Amendment of FASB Statement No. 114	8.81
No. 144, Accounting for the Impairment or Disposal of Long Lived Assets	11.05
No. 166, Accounting for Transfers of Financial Assets—an Amendment of FASB Statement No. 140	1.53, 10.13
No. 167, Amendments to FASB Interpretation No. 46(R)	1.53, 10.13
Federal Credit Union Act (FCUA)	
Section 107(4)	12.11
Section 107(6)	13.38
Section 115	2.08
Federal Deposit Insurance Act (FDI ACT)	
Holding companies	17.38
Section 29	13.34, 13.36
Section 36	1.92–.93, 1.95, 1.102–.103, 1.107–.109, 1.118
Section 36(g)(3)(A)	1.101
Section 36(h)	1.104, 5.250
Section 37	1.98
Section 37(a)	11.05
Section 37(a)(2)	5.244
Section 38	1.8182, 1.85, 17.28
Section 38(c)(3)(A)	1.85
Federal Deposit Insurance Corporation (FDIC)	
Final Rule, Safe Harbor Protection for Treatment by the FDIC as Conservator or Receiver of Financial Assets Transferred by an Insured Depository Institution in Connection With a Securitization or Participation	10.13
Regional Director Memorandum No. 2000-019, Reviews of External Auditors' Workpapers	5.256

©2016, AICPA AAG-DEP FED

Title	Paragraphs
Risk Management Manual of Examination Policies	7.74
Federal Deposit Insurance Corporation Improvement Act of 1991 (FDICIA)	
Section 112	1.92
Federal Financial Institutions Examination Council (FFIEC)	
Advisory on Interest Rate Risk Management	18.72
Instructions for Preparation of Consolidated Reports of Condition and Income	9.31, 11.05, 16.36
Supervisory Policy Statement on Investment Securities and End-User Derivatives Activities	7.67–.69
Supplemental Instructions for September 2014 Call Reports	8.75, 8.123
Uniform Bank Performance Reports	5.29
Federal Interagency Regulations	
Account Management and Loss Allowance Guidance for Credit Card Lending	8.76
Advisory on Accounting and Reporting for Commitments to Originate and Sell Mortgage Loans	10.12
Advisory on Mortgage Banking	10.11
Advisory on the Unsafe and Unsound Use of Limitation of Liability Provisions in External Audit Engagement Letters	5.12
Appraisal and Evaluation Guidelines	8.63
Appraisals for Higher Priced Mortgage Loans	8.6466
Concentrations in Commercial Real Estate Lending, Sound Risk Management Practices	8.61, 9.24
Correspondent Concentration Risks Interagency Guidance	8.79
Credit Risk Retention	1.52
Guidance on Certain Loans Held for Sale	8.66, 8.69, 10.76
Guidance on High Loan-to-Value Residential Real Estate Lending	8.60
Guidance on Leveraged Lending	8.80
Guidelines for Real Estate Lending Policies	8.60
Guidelines on Asset Securitization	10.10

Title	Paragraphs
Illustrations of Consumer Information for Nontraditional Mortgage Products	8.78
Interagency Guidance on Nontraditional Mortgage Product Risks	8.78
Interagency Guidelines on Asset Securitization	7.75
Interagency Policy Statement on Allowance for Loan and Lease Losses Methodologies and Documentation for Banks and Savings Institutions	5.246–.247, 9.23, 9.25
Interagency Policy Statement on Documentation for Loans to Small-and-Medium-sized Businesses and Farms	9.33
Interagency Policy Statement on External Auditing Programs for Banks and Savings Associations	5.256
Interagency Policy Statement on Income Tax Allocation in a Holding Company Structure	16.17, 16.36
Interagency Supervisory Guidance on Allowance for Loan and Lease Losses Estimation Practices for Loans and Lines of Credit Secured by Junior Liens on 1-4 Family Residential Properties	9.28
Interagency Supervisory Guidance on Bargain Purchases and FDIC- and NCUA-Assisted Acquisitions	19.07
Joint Agency Policy Statement on Interest Rate Risk	1.75
Policy Statement on Funding and Liquidity Risk Management	18.73
Policy Statement on Prudent Commercial Real Estate Loan Workouts	8.62, 8.73–.74
Policy Statement on the Internal Audit Function and Its Outsourcing	1.107
Policy Statement on the Review and Classification of Commercial Real Estate Loans	8.62
Prohibitions and Restrictions on Proprietary Trading and Certain Interests in, and Relationships With, Hedge Funds and Private Equity Funds	18.77
Supervisory Guidance Addressing Certain Issues Related to Troubled Debt Restructurings	8.74
Supervisory Guidance on Counterparty Credit Risk	18.76

©2016, AICPA AAG-DEP FED

Title	Paragraphs
Uniform Retail Credit Classification and Account Management Policy	8.68
Update on Accounting for Loan and Lease Losses	9.38
Federal Register	5.221, 9.23
Vol. 74, No. 103, pp. 26081–26084	17.25
Vol. 74, No. 137, pp. 35726-35761	1.95
Federal Reserve Act	
Section 23A	18.71
Section 23B	18.71
Sections 23A-23B	18.72
Federal Reserve Regulations	
Regulation CC	13.30
Regulation H	11.06, 17.18
Regulation LL	1.57
Regulation MM	1.57
Regulation O	5.3839
Regulation W	18.71
Regulation Y	1.9394, 17.18
Regulation Z	8.84
Federal Reserve System Publications	
Federal Reserve Bank Holding Company Supervision Manual Section 3901.0.2, "Holding Company Fails to Continue Meeting Financial Holding Company Capital and Management Requirements"	17.39
Risk-Based Capital Standards: Trust Preferred Securities and the Definition of Capital	17.18
Risk Transfer Considerations When Assessing Capital Adequacy – Supplemental Guidance on Consolidated Supervision Framework for Large Financial Institutions	10.10
Supervisory Guidance on Model Risk Management	18.75
FIL-8-2011, Deposit Insurance Assessment Base, Assessment Rate Adjustments, Dividends, Assessment Rates, and Large Bank Pricing Methodology	1.26
FIL-16-99, Trust Preferred Securities	7.76

Title	Paragraphs
FIL-17-2003, Corporate Governance, Audits, and Reporting Requirements	1.107
FIL-19-2009, Classification Treatment for High Loan-to-Value (LTV) Residential Refinance Loans	8.68
FIL-20-2009, Risk Management of Investments in Structured Credit Products	7.51, 7.72–.73
FIL-22-2008, Managing Commercial Real Estate Concentrations in a Challenging Environment	9.26
FIL-33-2009, Annual Audit and Reporting Requirements: Final Amendments to Part 363	1.96
FIL-34-2012, Investments in Corporate Debt Securities by Savings Associations	7.77
FIL-38-2012, FDIC Advisory on Effective Credit Risk Management Practices for Purchased Loan Participations	9.29
FIL-43-2009, Allowance for Loan and Lease Losses: Residential Mortgages Secured by Junior Liens	9.27
FIL-44-2012, Assessments: Final Rule on Assessments, Large Bank Pricing	1.29
FIL-45-1998, Investment Activities	18.71
FIL-49-2015, Advisory on Effective Risk Management Practices for Purchased Loans and Purchased Loan Participations	9.29
FIL-57-2014, Assessments: Final Rule	1.30
FIL-62-1996, Credit Derivatives	18.71
FIL-64-2011, Assessments: Assessment Rate Adjustment Guidelines	1.28
FIL-69-2009, Process for Determining if an Institution Subject to Interest-Rate Restrictions Is Operating in a High-Rate Area	13.37
FIL-146-2008, Recordkeeping Requirements for Qualified Financial Contracts	14.26

## G

Title	Paragraphs
Global Derivatives Study Group	
Derivatives: Practice and Principles and Related Appendices	18.70

Title	Paragraphs
Internal Revenue Code	
Section 166	16.07
Section 172	16.09
Section 265	16.12
Section 291	16.12
Section 451	16.25
Section 475	16.11
Section 581	16.03
Section 582	16.05
Section 585	16.06, 16.25
Section 591	16.04
Section 593	16.08
Section 1362	16.01
Section 7701(A)(19)	16.04
Section 7701(A)(32)	16.04

#### L

Title	Paragraphs
Letter to Credit Unions	
No. 02-CU-09, Allowance for Loan and Lease Losses	9.37
No. 03-CU-01, Loan Charge-Off Guidance	9.37
No. 03-CU-07, Guidance on Selected Provisions of the Sarbanes-Oxley Act of 2002 for Federal Credit Unions (FCUS)	2.20
No. 09-CU-02, Corporate Credit Union System Strategy	12.65
No. 09-CU-14, Corporate Stabilization Fund Implementation	12.66

AAG-DEP INT ©2016, AICPA

#### Ν

Title	Paragraphs
National Association of Insurance Commissioners	
Accounting Practices and Procedures Manual	22.55
NCUA Accounting Bulletins	2.21
No. 06-01	9.38
NCUA IRPS 02-3, Allowance for Loan and Lease Losses Methodologies and Documentation for Federally-Insured Credit Unions	5.247, 9.37–.38
NCUA Publications	
Accounting Manual for Federal Credit Unions	2.22
Chartering and Field of Membership Manual	2.21
Examiner's Guide	2.21
Federal Credit Union Bylaws	2.21
The Federal Credit Union Handbook	2.21
Letter to Credit Unions No. 09-CU-19, Evaluating Residential Real Estate Mortgage Loan Modification Program	8.71
Statement on Application of Recent Corporate Governance Initiatives to Non-Public Banking Organizations	1.107
Supervisory Committee Guide for Federal Credit Unions	2.21, 23.28
NCUA Regulations	
Part 701.36	12.11
Part 701.38	15.08
Part 702	9.34
Part 702.201	17.48
Part 702.303	17.48
Part 715	23.28
Part 715.18(b)	13.64

©2016, AICPA AAG-DEP NCU

0

Title	Paragraphs
OCC Banking Circular 277, Risk Management of Financial Derivatives	18.69
OCC Bulletin	
No. 94-31, Risk Management of Financial Derivatives Q&A's	18.71
No. 2000-16, Model Validation	18.75
No. 2011-12, Sound Practices for Model Risk Management: Supervisory Guidance on Model Risk Management	18.75
No. 2011-47, OTS Integration: Supervisory Policy Integration Process	1.59
No. 2012-10, Troubled Debt Restructurings: Supervisory Guidance on Accounting and Reporting Requirements	8.73–.74
No. 2013-17, Lending Limits: Final Rule	1.4347
No. 2014-8, End-User Derivatives and Trading Activities: Comptroller's Handbook Supplemental Examination	18.70
OCC Interpretation Letter No. 777	7.76
OCC Publications	
Bank Accounting Advisory Series (BAAS)	7.82, 8.73, 8.88, 9.30–.31, 10.15, 11.16, 12.15, 19.05
Topic 10, "Acquisitions, Corporate Reorganizations, and Consolidations"	19.05
Supervisory Guidance on Model Risk Management	18.75
Office of Inspector General's (OIG)	
Consolidated Audit Guide for Audits of HUD Programs	1.118, 4.33, 23.06
OTS CEO Memo 321, No Interest, No Payment Credit Card Programs	8.76
OTS Publications	
Statement on Application of Recent Corporate Governance Initiatives to Non-Public Banking Organizations	1.107
OTS TB 73A, Investing in Complex Securities	7.76

AAG-DEP OCC

#### P

Title	Paragraphs
PCAOB AS	
1301, Communications With Audit Committees	1.98
2201, An Audit of Internal Control Over Financial Reporting that is Integrated With an Audit of Financial Statements	14.68
2502, Auditing Fair Value Measurements and Disclosures	14.68
PCAOB Interim Standard AU Section 332, Auditing Derivative Instruments, Hedging Activities, and Investments in Securities	12.88
PCAOB Staff Audit Practice Alert	
No. 2, Matters Related to Auditing Fair Value Measurements of Financial Instruments and the Use of Specialists	7.163, 8.230, 18.101, 20.68
No. 4, Auditor Considerations Regarding Fair Value Measurements, Disclosures, and Other-Than-Temporary Impairments	7.163, 8.230, 18.101, 20.68
No. 7, Matters Related to Auditing Fair Value Measurements of Financial Instruments and the Use of Specialists	8.230
No. 10, Maintaining and Applying Professional Skepticism in Audits	5.05
No. 11, Considerations for Audits of Internal Control Over Financial Reporting	5.64, 7.142, 9.65
No. 12, Matters Related to Auditing of Financial Statements	5.138
No. 13, Matters Related to Auditor's Consideration of a Company's Ability to Continue as a Going Concern	5.163
Professional Ethics Executive Committee Interpretation No. 501-8, "Failure to Follow Requirements of Governmental Bodies, Commissions, or Other Regulatory Agencies on Indemnification of Liability Provisions in Connection With Audit and Other Attest Services"	5.12

©2016, AICPA AAG-DEP PRO

#### Q

Title	Paragraphs
Q&A Section	
6300.25, "Integrated/Nonintegrated Contract Features in Applying FASB ASC 944-30"	22.50
6300.26, "Evaluation of Significance Modification in Applying FASB ASC 944-30"	22.50
6300.27, "Changes in Investment Management Fees and Other Administration Changes in Applying FASB ASC 944-30"	22.50
6300.28, "Definition of Reunderwriting for Purposes of Applying FASB ASC 944-30"	22.50
6300.29, "Contract Reinstatements in Applying FASB ASC 944-30"	22.50
6300.30, "Commissions Paid on an Increase in Insurance Coverage or Incremental Deposits in Applying FASB ASC 944-30"	22.50
6300.31, "Participating Dividends and the Interaction of Guidance in FASB ASC 944"	22.50
6300.32, "Premium Changes to Long Duration Contracts in Applying FASB ASC 944-30"	22.50
6300.33, "Evaluation of Changes Under FASB ASC 944-30-35-37(A)"	22.50
6300.34, "Nature of Investment Return Rights in FASB ASC 944-30-35-7(B)"	22.50
6300.39, "Cumulative Effect of Change in Accounting Principle – ASU No. 2010-26"	22.45
6300.40, "Deferrable Commissions and Bonuses Under ASU No. 2010-26"	22.47
9110.16, "Example Reports on Federal Deposit Insurance Corporation Loss Sharing Purchase and Assumption Transactions"	1.118, 23.36
9110.18, "Small Business Lending Fund Auditor Certification Guidance"	23.37
9150.34, "Modifications to the Accountant's Compilation or Review Report When a Client Adopts a Private Company Council Accounting Alternative That Results in a Change to a Previously Issued Report"	23.33
9160.28, "Combining a Going Concern Emphasis-of-Matter Paragraph With Another Emphasis-of-Matter Paragraph"	23.10

Title	Paragraphs
9160.29, "Modification to the Auditor's Report When a Client Adopts a PCC Accounting Alternative"	23.12
9160.30, "Modification to the Auditor's Report When a Client Adopts a PCC Accounting Alternative That Results in a Change to a Previously Issued Report"	23.13

### S

Title	Paragraphs
Sarbanes-Oxley Act	
Section 201	1.107
Section 404	1.110, 1.114, 1.117
Section 404(a)	1.108
Section 404(b)	1.112–.113
Section 404(c)	1.113
SEC Accounting Series Release No. 268, Presentation in Financial Statements of "Redeemable Preferred Stocks"	15.45
SEC Division of Corporation Finance: Frequently Requested Accounting and Financial Reporting Interpretations and Guidance	
Section 1(B), "Accounting for Advertising Costs"	22.29
SEC Final Reporting Release No. 28	9.23
SEC Final Rule Release	
No. 33-8238, Management's Report on Internal Control Over Financial Reporting and Certification of Disclosure in Exchange Act Periodic Reports	1.108, 1.117
No. 33-9142, Internal Control Over Financial Reporting in Exchange Act Periodic Reports of Non-Accelerated Filers	1.113
No. 33-9175, Disclosure for Asset-Backed Securities Required By Section 943 of the Dodd-Frank Wall Street Reform and Consumer Protection Act	1.49
No. 33-9176, Issuer Review of Assets in Offerings of Asset-Backed Securities	1.50

©2016, AICPA AAG-DEP SEC

Title	Paragraphs
No. 34-65148, Suspension of the Duty to File Reports for Classes of Asset-Backed Securities Under Section 15(D) of the Securities Exchange Act of 1934	1.51
SEC Industry Guide No. 3, Statistical Disclosures By Bank Holding Companies	5.29, 19.06
SEC Regulation S-K	
Item 303	18.91
Item 305	18.90
SEC Regulation S-X	17.24
SEC Rules and Regulations	
AB	4.37, 10.118
3	23.01
12	1.107
12b-2	1.111
12(g)	12.27
12(i)	1.13, 1.107
13a-15	1.111
15(d)	1.51, 1.107
15d-15	1.111
15G	1.52
SEC SAB	
No. 102, Selected Loan Loss Allowance Methodology and Documentation Issues	9.20, 9.23
No. 109, Written Loan Commitments Recorded at Fair Value Through Earnings	10.27
No. 112	19.02
Topic 2, Codification of Staff Accounting Bulletins "Business Combinations"	19.02
Topic 5, Codification of Staff Accounting Bulletins "Miscellaneous Accounting"	18.93
Topic 5(M), Other Than Temporary Impairment of Certain Investments in Equity Securities	7.103
Topic 11, Codification of Staff Accounting Bulletins "Disclosures of the Impact of Assistance From Federal Financial Institution Regulatory Agencies"	19.06

Title	Paragraphs
Securities Industry and Financial Markets Association (SIFMA)	
Uniform Practices for the Clearance and Settlement of Mortgage-Backed Securities and Other Related Securities	14.36
SOP 92-3, Accounting for Foreclosed Assets	11.05
SR Letter	
02-10, Derivative Contracts Hedging Trust Preferred Stock	18.71
11-7, Guidance on Model Risk Management	18.75
11-15, Disposal of Problem Assets Through Exchanges	11.09
12-10/CA Letter 12-9, Questions and Answers for Federal Reserve-Regulated Institutions Related to the Management of Other Real Estate Owned (OREO)	11.14
12-11/CA Letter 12-10, Guidance on Lender's Decision to Discontinue Foreclosure Proceedings	11.07
12-15/CA Letter 12-2, Policy Statement on Rental of Residential OREO Properties	11.10
12-15/CA Letter 12-3, Policy Statement on Rental of Residential OREO Properties	11.13
13-9/CA Letter 13-6, Minimum Standards for Prioritization and Handling Borrower Files With Imminent Scheduled Foreclosure Sale	11.08
96-17, Supervisory Guidance for Credit Derivatives	18.71
97-18, Application of Market Risk Capital Requirements to Credit Derivatives	18.71

T

Title	Paragraphs
Treasury Regulation	
1.166-2(D)	16.07
1.166-2(D)(3)	16.07

#### U

Title	Paragraphs
Uniform Commercial Code (UCC)	
Article 9	8.8587
Uniform Single Attestation Program for Mortgage Bankers	1.118, 5.11, 10.118
U.S. Code	
12 USC 371c	18.72
12 USC 371c-1	18.72
12 USC 371d	12.13
38 USC 3703(a)(1)(C)	10.17
U.S. Department of Education Audit Guides	
Compliance Audits (Attestation Engagements) for Lenders and Lender Servicers Participating in the Federal Family Education Loan Program	1.118
U.S. Department of Housing and Urban Development (HUD)	
Consolidated Audit Guide for Audits of Department of Housing and Urban Development programs	5.11
Office of Inspector General's (OIG) Consolidated Audit Guide for Audits of HUD Programs	1.118, 4.33, 23.06

#### ۷

Title	Paragraphs
VA Circular 26-08-19, Implementation of Loan Guarantee Provisions of Public Law 110-389	10.17

AAG-DEP UNI ©2016, AICPA

# **Subject Index**

Α	ACCOUNTS PAYABLE. See also other assets,
ADO 0	other liabilities, and other
ABS. See asset-backed securities	investments
ACCEPTANCES, CUSTOMERS' LIABILITIES	ACCOUNTS RECEIVABLE LOANS8.26
ON12.57	ACCRETABLE YIELD 8.113115, 8.119
ACCOUNT, UNIT OF20.03	ACCRETION OF DISCOUNT ON
ACCOUNTING ESTIMATES 5.193203,	SECURITIES
7.149, 7.157159, 9.65, 9.82	ACCRUED COMPENSATION12.02, 16.25
ACCOUNTING HEDGES, DERIVATIVES	-
AS18.57	ACCRUED DEFERRED
ACCOUNTING STANDARDS	COMPENSATION
· Acquisition arrangements 8.139	ACCRUED EXPENSES12.02
· Business combinations	ACCRUED INTEREST PAYABLE. See also
· Cash and cash equivalents6.12–.29	other assets, other liabilities, and other
· Collateral10.70–.73	investments12.02, 13.66
· Credit losses 9.39–.52, 9.64–.75	ACCRUED INTEREST RECEIVABLE
Debt (institution's borrowings) 15.34–.57	(AIR) 8.221, 12.01
Deposits	ACCRUED LOSS CONTINGENCIES 16.25
Derivative instruments	ACH SYSTEMS. See automated
Fair value measurements and	clearinghouse (ACH) systems
disclosures	ACQUISITION, DEVELOPMENT, AND
Foreclosed assets 8.137, 11.06–.07,	CONSTRUCTION (ADC) ARRANGEMENTS.
11.17–.23	See also real estate investments, real
· Foreign loans	estate owned, and other foreclosed
· Goodwill12.35–.38	assets
· Hedging18.79–.82	· Accounting standards 8.139
· Held for sale8.69, 8.148, 10.20–.29,	Definitions
	Development costs
Income taxes	Financial statement presentation 11.43
Insurance activities	· Inherent risk
Lease financing	ACQUISITION COSTS, INSURANCE. See also
Loan commitments	insurance activities22.1850
· Loan transfers and servicing 10.20–.96,	· Compensation, benefits, and other costs of
	acquisition activities
· Loans8.90–.152	Deferred
· Mortgage-backed securities 7.88,	· Direct-response advertising 22.27–.32
10.20–.29	· Estimated gross profit 22.44–.50
Mortgage companies and banking	· Incremental direct costs of contract
activities 4.16–.17	acquisition
Other assets, other liabilities, and other	Nondeferrable expenses 22.33–.38
investments	ACQUISITION METHOD19.0917,
Premises and equipment	
and other foreclosed assets8.139,	ACQUISITIONS. See business combinations
	ACTIVE MARKETS20.15, 20.35
· Real estate mortgage investment	
conduits	ACTIVITIES, INTERNAL CONTROL OVER FINANCIAL REPORTING9.70
· Repurchase agreements 14.27–.39	
· Residential mortgage loans 8.90, 8.92,	ADC ARRANGEMENTS. See acquisition,
	development, and construction (ADC) arrangements
Securities	•
• Troubled debt restructurings 7.122–.124,	ADDITIONAL PAID-IN CAPITAL (PIC) 17.06
	ADEQUATELY CAPITALIZED BANKS AND
Trust and asset management activities	SAVINGS INSTITUTIONS 1.85–.86,
acaviaco	17.31, 17.42

ADEQUATELY CAPITALIZED CREDIT UNION CLASSIFICATION	ALLOWANCE FOR LOAN LOSSES5.30 9.59
Complex credit unions	ALLOWANCE FOR UNFUNDED LOAN COMMITMENTS12.02
<ul> <li>Defined</li></ul>	ALTERNATIVE COMPONENTS
New credit unions	APPROACH
Prompt corrective action Exhibit 2-1 at 2.30 Supervisory actions	AMERICAN RECOVERY AND REINVESTMENT
ADJUSTABLE-RATE MORTGAGE	ACT OF 2009 (ARRA)16.12
(ARM)	AMORTIZATION
ADJUSTED CORE CAPITAL2.45	· Debt securities
ADVANCE FACTORING 8.24–.25	· Fair value measurements and
ADVERSE OPINIONS23.20	disclosures
ADVERTISING,	· Index amortizing swaps18.69
DIRECT-RESPONSE22.2732	· Loans
AFFILIATED ENTITIES. See also related-party	<ul> <li>Negative amortization mortgages 8.41</li> <li>Other-than-temporary impairment cost</li> </ul>
transactions5.36, 10.29	basis7.98
AGENCY AGREEMENTS AS PERSONAL TRUSTS21.08	Planned amortization class
AGGREGATION OF DISCLOSURES, OF	· Servicing assets or liabilities
VARIABLE INTEREST	AMT. See alternative minimum tax
ENTITIES10.112113	ANALYTICAL PROCEDURES
AGRICULTURAL LOANS. See commercial,	· Debt15.69–.71
industrial, and agricultural loans	• Deposits
AIR. See accrued interest receivable	Federal funds and repurchase agreements14.67
ALLL. See allowance for loan and lease losses	Internal control used in5.63
ALLOCATED TRANSFER RISK RESERVES	· Loans
(ATRR)9.14	Risk assessment process
ALLOWANCE FOR CREDIT LOSSES 8.155,	ANNUAL INDEPENDENT AUDITS
9.0607, 9.4248, 9.59, 9.110	AND REPORTING
ALLOWANCE FOR DOUBTFUL	REQUIREMENTS
ACCOUNTS8.155	Additional requirements for services outsourced to external
ALLOWANCE FOR LOAN AND LEASE LOSSES	auditors
(ALLL) • Accounting standards9.43	• Annual reporting requirements
· Audit considerations9.03, 9.66, 9.70,	Audit committees
9.110112	· Credit unions, CUMAA provisions2.05
<ul> <li>Credit unions</li></ul>	2.55–.62
Failure to maintain adequate	Enforcement actions against     auditors
allowance 9.112	· FDIC loss sharing purchase and assumption
Federal banking regulation	transactions
(GAAP)9.42–.48	· Federal Home Loan Mortgage Corporation borrowing
· Internal control9.73	· Filing and notice requirements1.99
<ul> <li>Loans not previously held for sale 10.77</li> <li>Mathematical calculation basis for 9.06</li> </ul>	Financial statements. See financial statements
Qualitative adjustment based on historical loss	Holding companies
experience	· Independent public accountant
Regulatory matters	standards
Related to financing receivables9.59 Residential mortgage loans8.78	Loan servicing
Substantive tests	Report1.108

ANNUAL INDEPENDENT AUDITS AND REPORTING REQUIREMENTS—continued  Management's Discussion and Analysis	ASSETS  Audit objectives for other
Audit considerations	Intangiole assets. See Intangiole assets  Loans. See loans  Long-lived assets. See long-lived assets  Misappropriation ofExhibit 5-1 at 5.257  Other assets. See other assets, other liabilities, and other investments  Ratios
Fair value	ATM. See automated teller machine ATM NETWORK
Regulatory guidance       10.10         Residuals       7.42–47         Risk analysis       7.33–34, 7.75         Senior/subordinated debt       7.48–49         Versatility       7.13         ASSET-BASED FINANCIAL SERVICES       3.05         ASSET/LIABILITY       MANAGEMENT         5.43–44, 18.56	Allowance for loan and lease losses9.03,
ASSET MANAGEMENT. See trust and asset management activities  ASSET-QUALITY RISK	Compliance with laws and regulations

AUDIT CONSIDERATIONS—continued	AUDIT ENGAGEMENT
· Documentation 1.98, 5.102–.109, 5.256	· Letters1.100, 5.10–.12, 5.256
• Engagement terms	· Teams
· Equity and disclosure regarding capital	· Unable to continue 5.148
matters 17.89–.115	AUDIT EVIDENCE. See also risk assessment
• Examinations	Appraisals as
· Fair value measurements and	· Audit documentation
disclosures	· Electronic data
· Federal funds and repurchase	· Evaluation of sufficiency and
agreements	appropriateness
Financial reporting matters 5.193–.203	· Fraud consideration in financial statement
Fraud in financial statement audit. See fraud,	audit 5.144–.147
consideration of in financial statement audit Going concerns	· Obtainment of5.76–.98
Income taxes	· Written representations as 5.180
Information other than financial	AUDIT OBJECTIVES
statements	· Auditor's reports 5.02
· Insurance activities	· Debt15.58
· Internal control. See internal control over	· Deposits13.47
financial reporting	· Equity and disclosure regarding capital
<ul> <li>Limitations on scope of audit5.254</li> </ul>	matters
<ul> <li>Loan transfers and servicing 10.114–.127</li> </ul>	Federal funds and repurchase
· Loans	agreements
• Materiality	· Income taxes
· Misstatements, evaluating5.99–.101	· Loans 9.182 9.214
Other assets, other liabilities, and other	<ul> <li>Loans</li></ul>
investments	activities17.106
Performance materiality	Other assets, other liabilities, and other
Planning. See audit planning	investments
· Purpose	· Real estate investments, real estate owned,
· Real estate investments, real estate owned,	and other foreclosed assets11.56
and other foreclosed assets 11.56–.73	· Securities
· Regulation and supervision. See also regulation	· Trust and asset management
and oversight5.212–.257	activities
Reports	AUDIT PLANNING
· Risk assessment. See also risk	· Allowance for loan and lease losses 9.66
assessment	· Benefits5.14
• Securities	· Cash and cash equivalents6.35
<ul> <li>Service organizations 5.11, 5.120–.122</li> <li>Specialists. See specialists</li> </ul>	· Credit losses 9.63–.68
Supplementary information5.190–.192	Debt
Trust and asset management	Depository institution regulation and
activities	supervision
· Understanding the entity and its environment.	Equity and disclosure regarding capital
See understanding the entity and its	matters17.92–.97, 17.107–.109
environment	Federal funds and repurchase
· Written representations 5.180–.189	agreements
AUDIT DOCUMENTATION. See also auditor's	· Generally accepted accounting principles
reports 5.102109	(GAAP)17.94
<ul> <li>Assembly and retention of final audit</li> </ul>	· Income taxes
file5.109	· Loan transfers and servicing 10.115–.118
· Audit evidence obtained5.106–.108	· Loans
• Audit procedures 5.106–.108	Materiality
• Defined	Mortgage companies and banking     17 107 100
• Examiner's access	activities
Going concerns	Other assets, other liabilities, and other investments
Overview	• Overview
· Retention	Real estate investments, real estate owned,
· Timely preparation 5.105	and other foreclosed assets11.57
- y p - p	

AUDIT PLANNING—continued	AUTOMATED TELLER MACHINE
· Securities	(ATM)13.0405, 13.26
· Security investments7.139–.143	AVAILABLE-FOR-SALE SECURITIES. See also
· Trust and asset management	other-than-temporary impairment
activities	(OTTI)
AUDIT RISK. See also material	7.128, 7.134, 17.09
misstatements, risk of5.0304,	_
5.06–.09, 13.49	В
AUDIT TEAM, DISCUSSIONS	BACK-TESTING PROCESS FOR LOAN LOSS
AMONG5.129132	ESTIMATES9.70
AUDITORS	BAD-DEBT DEDUCTIONS ON INCOME
· Access to institution's documents 1.104,	TAXES
5.250, 5.254	BAD-DEBT RESERVES 16.25
· Communication with	
· Compliance with laws and regulations	BALANCE SHEETS. See also other assets,
responsibility	other liabilities, and other investments
Enforcement actions against1.102–.103,	Borrowings
5.233	Deposits
· Engagement5.10–.13, 5.148	Federal Home Loan Banks or Federal Reserve
External auditors	Bank stock12.33–.34
· Independence standards	· Foreclosed assets 11.17, 11.23
· Liability limitations	Insurance activities22.18, 22.65, 22.67
· Materiality determinations5.16–.20	· Loan classifications 8.223
Objectives	· Loans
Professional judgment 5.17–.18, 5.26	<ul><li>Members' shares reported as equity23.24</li><li>Operating leases</li></ul>
• Qualifications	Premises and equipment
<ul> <li>Relationship with examiners 5.248–.257</li> <li>Reports. See auditor's reports</li> </ul>	BALLOON PAYMENTS8.17
· Sarbanes-Oxley requirements 1.107	
· Specialists5.110–.115, 19.35–.40,	BANK CREDIT FACILITIES
20.65, 20.68	BANK FAILURES1.06
· Standards1.98	BANK HOLDING COMPANY ACT18.77
AUDITOR'S REPORTS 23.0138	BANK INSURANCE FUND (BIF) 1.1517
· Audit objectives 5.02	BANK SECRECY ACT/ANTI-MONEY
· Emphasis-of-matter paragraphs23.01,	LAUNDERING (BSA/AML) EXAMINATION
	MANUAL
FDIC loss sharing purchase and assumption transactions	BANK SECRECY ACT OF 1970
Financial statements prepared with special	(BSA) 1.2021, 2.62, 5.226, 9.29
purpose framework23.22–.23	BANKERS' ACCEPTANCES. See also other
· Going concerns	assets, other liabilities, and other
· Internal control 23.25–.27	investments 12.02, 15.1920, 15.46
Meetings to review	BANKRUPTCY10.4042
· Members' shares reported as equity23.24	BANKS AND SAVINGS
Modified opinions	INSTITUTIONS 1.01118
Not opinion on fair presentation of institution's	· Advanced approaches17.44
regulatory reports	· Annual independent audits and reporting
<ul> <li>Opinion formation 5.04, 5.37, 23.02–.04</li> <li>Other-matter paragraphs</li></ul>	requirements1.92–.118
• Overview	Assurance services
· PCAOB unqualified	Brokered deposits
opinions	Business description
· Small Business Lending Fund 23.37	Capital requirement. See capital requirement     Competition
· Supervisory committee audits 23.28–.34	Consolidation
· Unmodified opinion23.05–.06	· Credit exposure
AUTOMATED CLEARINGHOUSE (ACH)	· Definitions for tax purposes 16.03–.04
SYSTEMS2.62, 13.2628	Deposit insurance fund

BANKS AND SAVINGS INSTITUTIONS—continued	BROKERED DEPOSITS 5.242, 13.1113, 13.3338
Disclosures of capital adequacy 17.41–.46 Equity and disclosure regarding capital matters17.02–.46, 17.89–.105	BSA/AML EXAMINATION MANUAL. See Bank Secrecy Act/Anti-Money Laundering (BSA/AML) Examination Manual
Federal funds. See federal funds	BUSINESS, DEFINED 19.01, 19.09
Financial intermediation role	BUSINESS COMBINATIONS 19.01–.41  Accounting standards 19.08–.30  Acquisition method 19.09–.17, 19.22–.25  Audit considerations 19.31–.41  Branch acquisitions 19.28–.30  Corporate credit unions 2.45  Impairment and disposal for certain acquired long term customer relationship intangible assets 19.26–.27  Indemnification assets 12.01, 12.48  Intangible assets 12.06–.07  Loans and debt securities acquired with deteriorated credit quality 19.21  Mutual entities 19.22–.25  Overview 19.01  Purchased loans 19.18–.20  Regulatory matters 19.02–.07  Structures for 19.01  Substantive tests 19.33
COMMITTEE)1.76, 18.73	BUSINESS LOANS. See commercial, industrial, and agricultural loans
BASEL III	BUSINESS PLANS
BASIS POINT	BUSINESS RISK14.15, 17.108
BASIS RISK18.11, 18.5962	·
	BYLAWS
BASIS SWAPS18.33	BYLAWS2.11, 2.21, 5.35
BASIS SWAPS	С
BENEFICIARIES, OF VARIABLE INTEREST	C CALL OPTIONS17.16, 17.19
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C CALL OPTIONS17.16, 17.19 CALL REPORTS
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C CALL OPTIONS
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C CALL OPTIONS
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C         CALL OPTIONS       17.16, 17.19         CALL REPORTS       . Analytical procedures       5.29         Business combinations       . 19.04         Capital adequacy       1.72, 1.79         Copies to auditor       5.249–.250         Credit unions       2.45, 2.57
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C         CALL OPTIONS       17.16, 17.19         CALL REPORTS       5.29         Business combinations       19.04         Capital adequacy       1.72, 1.79         Copies to auditor       5.249–250         Credit unions       2.45, 2.57         Deposits       13.39
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C         CALL OPTIONS       17.16, 17.19         CALL REPORTS       . Analytical procedures       5.29         Business combinations       . 19.04         Capital adequacy       1.72, 1.79         Copies to auditor       5.249–.250         Credit unions       2.45, 2.57
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C  CALL OPTIONS
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C  CALL OPTIONS
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C  CALL OPTIONS
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C  CALL OPTIONS
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C  CALL OPTIONS
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C CALL OPTIONS
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C CALL OPTIONS
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C CALL OPTIONS
BENEFICIARIES, OF VARIABLE INTEREST ENTITIES	C CALL OPTIONS

CAPITAL REQUIREMENT—continued	CASH FLOW HEDGE ACCOUNTING
· Credit unions2.39–.42, 17.56–.64	APPROACH18.81
• Equity regarding17.03–.10, 17.47–.54,	CASH FLOW HEDGES6.29, 17.09
	CASH FLOW STATEMENTS
disclosure 1.79, 17.28–.46, 17.56–.64,	· Acquisition and sales of certain securities and
17.72–.79	loans
· Foreign institution branches 17.85–.86	· Cash receipts and payments related to
· Holding companies 1.34–.35, 17.11–.27,	hedging
	Derivatives
Income tax treatment	· Gross and net cash flows6.25–.26
• Minimal net worth	· Transfers and servicing10.81
· Minority interest in consolidated	CASH FLOWS6.1222, 20.3133
subsidiaries	CASH ITEMS IN THE PROCESS OF
Mortgage companies and banking	COLLECTION (CIPC)
activities	CASH ON HAND
17.56–.64	CASH RESERVES. See reserves
New federal interagency rules17.21, 17.27	CCPA. See Consumer Credit Protection Act
Noncompliance1.71, 1.91, 17.28, 17.56,	CD. See certificates of deposit
	CDARS. See Certificate of Deposit Account
PCA. See prompt corrective action	Registry System
· Tier 1 capital. See Tier 1 capital	CDO. See collateralized debt obligations
Tier 2 capital	CENTRAL LIQUIDITY FACILITY (CLF)2.23
Trust banks	CERTIFICATE OF DEPOSIT ACCOUNT
	REGISTRY SYSTEM (CDARS) 13.13
CAPITAL RESTORATION PLANS2.53	CERTIFICATES OF DEPOSIT (CD)
CAPTIVE FINANCE ACTIVITY3.02, 8.96	· Characteristics
· ·	· Defined as other assets or other
CARRYING AMOUNT, FORECLOSED RESIDENTIAL REAL ESTATE	investments
PROPERTY11.23	Defined as security
CASH, DEFINED	Negotiable compared to
CASH AND CASH EQUIVALENTS6.0142	nonnegotiable
· Accounting standards 6.09–.29	Retirement accounts
Acquisition and sales of certain securities and	· As time deposit
loans	CERTIFICATES OF INDEBTEDNESS 15.08
Audit considerations6.34–.42	CFPB. See Consumer Financial Protection
Cash items in the process of collection6.02     Cash on hand	Bureau
· Classification of cash flows 6.12–.22	CFR. See Code of Federal Regulations
· Definitions	CFTC. See Commodity Futures Trading
· Deposits with other financial	Commission
institutions	CHANGES IN ACCOUNTING POLICY, CREDIT
Federal funds. See federal funds     Financial statement presentation and	LOSS ALLOWANCE AFFECTED
disclosure	BY
· Financing activities6.20	CHARGE-OFFS AND RECOVERIES, LOANS9.53, 9.59, 9.108
· Gross and net cash flows 6.25–.28	
Hedging activities 6.29, 17.09	CHARGING OFF UNCOLLECTABLE RECEIVABLES
· Internal control	
· Investing activities	CHARTERS, CREDIT UNIONS. See also state-chartered credit
Reserves with Federal Reserve Bank or Federal	unions2.1117, 5.35
Home Loan Banks6.06, 6.35	CHECK CLEARING FOR THE 21ST CENTURY
· Substantive tests6.41–.42	ACT OF 2003 (CHECK 21) 13.3032,
CASH COLLATERAL10.73	

CHECK CLEARING PROCESS 13.2226	COLLATERAL—continued
CHECKING ACCOUNTS AS DEMAND DEPOSITS13.04	Cash collateral
CHECKS (CHECKING ACCOUNTS)13.1832	Credit loss estimation issues
CHIEF ACCOUNTING OFFICERS1.97	· Discontinued foreclosures
CHIEF EXECUTIVE OFFICERS (CEO). See also management1.97, 5.183	Financial statement disclosure8.167–.169,
CHIEF FINANCIAL OFFICERS (CFO). See also management1.97, 5.183	Loan review process
CHIPS. See Clearing House Interbank Payments System	Noncash collateral
C&I LOANS. See commercial, industrial, and agricultural loans	Property and liability insurance
CIPC. See cash items in the process of collection	Risk 8.07, 14.23–.25, 14.44, 14.66 Secured parties
CIVIL MONEY PENALTIES5.232	· Structured credit products
CLASSIFIED LOANS TO TOTAL LOANS RATIO	Uniform Commercial Code financing statements
CLEARED SWAPS	COLLATERAL-DEPENDENT LOANS9.47,
CLEARING HOUSE INTERBANK PAYMENTS SYSTEM (CHIPS)13.28	9.89, 9.102104 COLLATERAL MAINTENANCE
CLEARING PROCESS, CHECK13.2226	AGREEMENTS 18.08
CLEARINGHOUSE FOR SWAP CONTRACTS18.25	COLLATERAL MYOPIA, MANAGEMENT LOAN EVALUATION9.15
	COLLATERAL RISK8.07, 14.2325,
CLF. See Central Liquidity Facility	· · · · · · · · · · · · · · · · · · ·
CLF. See Central Liquidity Facility CLIENT REPRESENTATIONS, CREDIT LOSS	
CLIENT REPRESENTATIONS, CREDIT LOSS ESTIMATION	14.44, 14.66  COLLATERALIZED BORROWING, REPURCHASE AGREEMENTS
CLIENT REPRESENTATIONS, CREDIT LOSS ESTIMATION	
CLIENT REPRESENTATIONS, CREDIT LOSS ESTIMATION	14.44, 14.66   COLLATERALIZED BORROWING, REPURCHASE AGREEMENTS
CLIENT REPRESENTATIONS, CREDIT LOSS ESTIMATION	14.44, 14.66   COLLATERALIZED BORROWING, REPURCHASE AGREEMENTS
CLIENT REPRESENTATIONS, CREDIT LOSS ESTIMATION	14.44, 14.66   COLLATERALIZED BORROWING, REPURCHASE AGREEMENTS
CLIENT REPRESENTATIONS, CREDIT LOSS ESTIMATION	14.44, 14.66   COLLATERALIZED BORROWING, REPURCHASE AGREEMENTS

COMMERCIAL, INDUSTRIAL, AND AGRICULTURAL LOANS—continued  Receivables portfolio purchase agreements 8.28–.29 Review 9.97 Revolving loans 8.26–.27 COMMERCIAL FINANCING 3.01, 3.05–.07	COMPLEX CREDIT UNIONS—continued  Undercapitalized credit union classification
COMMERCIAL LETTERS OF CREDIT8.50 COMMERCIAL-MORTGAGE BACKED	COMPLIANCE WITH LAWS AND REGULATIONS5.152162
SECURITIES (CMBS)1.118	COMPREHENSIVE INCOME
COMMERCIAL PAPER	Accounting and reporting update7.103     Accounting standards
15.1213 COMMERCIAL REAL ESTATE (CRE)	Classification
LOANS	Other comprehensive income17.08–.09
COMMISSIONS, INSURANCE22.10,22.42, 22.46, 22.56–.59	Other-than-temporary impairment recognized as7.101–.102 Security investments7.89, 7.104
COMMITMENTS. See loan commitments	COMPUTER ASSISTED AUDIT
COMMITTED LOANS10.25	TECHNIQUES 8.222
COMMODITIES FUTURES CONTRACTS18.21	COMPUTER SYSTEMS AND SOFTWARE4.20, 12.23
COMMODITY FUTURES TRADING COMMISSION (CFTC)1.39	CONCENTRATION RISK 5.204209, 7.136, 8.07, 8.79, 8.172174,
COMMODITY SWAPS	
COMMON BOND AFFILIATION2.0304	CONCENTRATIONS-OF-CREDIT-RISK
COMMON EQUITY17.03, 17.10	DISCLOSURES
COMMON RISK CHARACTERISTICS 8.111	DOCUMENTATION9.06, 9.110
COMMON STOCK17.03	CONDITIONAL ASSET RETIREMENT
COMMON-TRUST FUNDS7.61, 21.12	OBLIGATIONS 12.56
COMMUNICATION  · With auditors	CONDUIT DEBT SECURITIES
· Discontinued foreclosures	CONFERENCE OF STATE BANK SUPERVISORS1.09
Examiners	CONFIDENTIALITY, OF REGULATORY
Internal loan review and audit staff9.67	REPORTS 5.256
· Management	CONFIRMATIONS
Regulatory and enforcement authorities	Debt
· With those charged with governance 1.98,	Loan documents8.208–.211
5.70–.75, 5.149, 9.111, 23.25–.27  COMMUNITY BANKS1.47, 23.37	• Substantive tests
COMMUNITY CREDIT UNIONS 2.03	CONSOLIDATED REPORTS OF CONDITION AND INCOME. See Call Reports
COMMUNITY REINVESTMENT ACT1.19,	CONSOLIDATED SUBSIDIARIES, MINORITY
	INTEREST IN17.10
COMPENSATION	<b>CONSOLIDATION</b> • Insurance
· Accrued	Other investments
· Executive compensation, credit unions2.52	Premises and equipment 12.20–.21
· Insurance activities	Securities
COMPLEX CREDIT UNIONS2.28–.38  · Adequately capitalized credit union	· Trust preferred securities17.12, 17.15
classification2.33	CONSTRUCTION ARRANGEMENTS. See
Defined       2.29–.32         Net worth ratio       2.32–.33	acquisition, development, and construction (ADC) arrangements

CONSTRUCTION LOANS 8.17, 8.52,	CORPORATE GOVERNANCE. See also those
8.216	charged with governance1.98,
CONSUMER CREDIT PROTECTION	
ACT8.84	CORPORATE TRUSTS21.10
CONSUMER FINANCIAL PROTECTION	CORRELATION
BUREAU (CFPB)1.3638, 2.02, 2.20, 3.14, 4.27, 8.84	CORRELATION FACTOR
CONSUMER INFORMATION, SUBPRIME	CORRESPONDENT CONCENTRATION
MORTGAGES4.25	RISKS8.79
CONSUMER LOANS8.3139	CORRESPONDENTS6.03
· Credit cards 8.36–.38, 8.76–.77,	COST APPROACH VALUATION20.28
8.197–.198	COST BASIS
· Defined	· Amortized, other-than-temporary
<ul> <li>Finance companies</li></ul>	impairment
· Laws governing	restructuring
· Overdraft protection	· Held to maturity security
· Retail sales contracts 8.33–.35	COST DETERMINATION FOR INSURANCE
CONTACT INFORMATION Appendix C	ACTIVITIES22.39
CONTINGENT ASSETS AND	COST OF LOAN8.224
LIABILITIES21.24	COUNTERPARTIES, LOAN PARTICIPATION
CONTINGENT LIABILITIES AND RELATED	AND SYNDICATION10.07
DISCLOSURES5.240	COUNTERPARTY RISK 7.22, 18.0708,
CONTINUATION STATEMENTS 8.87	20.21–.27
CONTINUING GENERAL LIEN	COURT TRUSTS21.08
ARRANGEMENTS8.87	COVERED FUNDS, OWNERSHIP IN7.78
CONTINUING INVOLVEMENT 10.3132	CRE. See commercial real estate (CRE) loans
CONTRACTUALLY SPECIFIED SERVICING	CREDIT ACCIDENT AND HEALTH
FEES	INSURANCE22.02, 22.06
CONTRIBUTED ASSETS12.7376	CREDIT CARDS 8.3638, 8.7677,
CONTROL ACTIVITIES. See also internal	8.197–.198
control over financial reporting 5.61	CREDIT COMMITTEES, CREDIT
CONTROL ENVIRONMENT. See also internal control over financial reporting5.61,	UNIONS2.10
9.70	CREDIT DEFAULT SWAPS1.3941
CONTROL RISK5.09, 18.13	CREDIT DERIVATIVES18.52
CONVENTIONAL LOANS8.42	CREDIT ENHANCEMENT 7.2122, 20.19
CONVERGENCE	CREDIT EXPOSURE1.4246
CONVERSION FACTOR MATRIX	CREDIT EXPOSURE MEASUREMENT 1.46
METHOD1.44, 1.47	CREDIT FILES8.193
CONVERTIBLE INSTRUMENTS 15.43,	CREDIT INITIATION8.10
18.51	CREDIT INVESTIGATIONS8.10, 8.192
CORE CAPITAL2.45	CREDIT LIFE INSURANCE 22.0205
CORPORATE BONDS	CREDIT LOSSES9.01112
CORPORATE CREDIT UNIONS. See also	· Accounting standards 9.39–.52, 9.64–.75
credit unions	· Allowance for 8.155, 9.06–.07,
· Capital requirement2.45–.54, 7.81,	9.42–.48, 9.59, 9.110
	Audit considerations 9.32, 9.63–.112     Credit unions
· Cash and cash equivalents	Estimation of overall credit
· Defined2.17	losses
<ul> <li>Equity and disclosure regarding capital</li> </ul>	9.70, Exhibit 9-1 at 9.22
matters17.65–.79, 17.89–.105	• Financial statement disclosure7.133, 9.53–.62
· Role6.05	9.55–.62

CREDIT LOSSES—continued  Institution estimates of	CREDIT UNION NATIONAL ASSOCIATION, INC. (CUNA)
Overview	5.234, 17.56–.64 Charters, bylaws, and minutes 2.11–.17, 5.35 Complex credit unions 2.28–.38 Contributed assets 12.73–.76
CREDIT ORIGINATION AND	<ul> <li>Corporate credit unions. See corporate credit unions</li> </ul>
DISBURSEMENT	• Credit committees       2.10         • Credit losses       9.34–38, 9.62         • Debt, long-term debt source       15.08
CREDIT QUESTIONNAIRES 15.7577	· Defined
CREDIT RATING AGENCIES 3.08	Deposits
CREDIT RATINGS3.08	matters
CREDIT RISK	Financial statements
Concentration-of-credit-risk disclosures	<ul> <li>Financial structure</li></ul>
· Concentration risk 5.204–.209, 7.136,	· History2.01
8.07, 8.79, 8.172–.174, 18.10	· Income tax exemption16.01, 16.13
· Counterparty credit risk	Insurance activities
Derivatives	<ul> <li>Internal control requirements</li></ul>
· Loans	Lending restrictions
Mortgage-backed securities7.21–.22	· Low-income designation
<ul> <li>Off-balance-sheet instruments8.175–.176,</li> <li> 9.02, 9.21</li> </ul>	Member shares
• Offsetting positions	Members' equity
Repurchase agreements 14.19–.22	<ul><li>Membership</li></ul>
Securities	
Swaps	· New
CREDIT RISK RETENTION	New credit unions
CREDIT STRATEGY8.0304	Other assets, other liabilities, and other investments
CREDIT SUPERVISION8.11	· Real estate investment limitations 12.11
CREDIT UNION EXECUTIVES	Regulation and oversight. See also National
SOCIETY2.16	Credit Union Administration (NCUA) 2.02,
CREDIT UNION MEMBERSHIP ACCESS ACT	
OF 1998 (CUMAA)	• Reporting requirements
· Annual audits 2.05, 2.55, 2.62	· Reserves 9.35–.36
• Enactment	Security investments
<ul> <li>Financial statements</li></ul>	State-chartered credit unions 2.19, 2.21, 2.27, 2.59
Prompt correction action	State credit unions
regulations	· Supervisory committees 2.08–.09,
· Provisions	2.26–.27, 2.58–.59, 5.11, 13.64
· Title III	· System
0.5. GMAF use13.39	· Types2.03

CREDIT WEAKNESS, ESTIMATES OF	DEBT (INSTITUTION'S BORROWINGS)—continued  Modifications or exchanges
CRITICALLY UNDERCAPITALIZED CREDIT UNION CLASSIFICATION2.50, 2.53,	Redeemable preferred stock
CROSS-CURRENCY EXCHANGE AGREEMENTS	<b>DEBT SECURITIES.</b> See also securities  · Accounting and reporting standards 7.125,
CROSS-HEDGING	Acquired with deteriorated credit quality7.125, 8.110–.122, 19.21  Amortization or accretion7.111
Association, Inc. CURRENT REPLACEMENT COST20.28	Asset-backed. See asset-backed securities Certificates of deposit
CURTAILMENTS	Corporate bonds
CUSTOMERS' LIABILITIES ON ACCEPTANCES	Financial statement disclosure
D DDA. See demand deposit accounts	(HTM) securities International organization issues7.60 Municipal obligations7.08–.11
DEALER RESERVES AND HOLDBACKS	Other-than-temporary impairment
DEBT-EQUITY SWAPS8.142143 DEBT FINANCING3.0811	credit products  Troubled debt restructurings8.133–.134
DEBT (INSTITUTION'S	DEBT SERVICE COVERAGE RATIO8.14
<b>BORROWINGS)</b>	DEBT-TO-EQUITY RATIO
Audit considerations	DEBTOR'S FINANCIAL DIFFICULTIES DETERMINATION8.129
· Creditworthiness ratings15.05	DECREASING TERM INSURANCE 22.04
<ul> <li>Distinguishing liabilities from equity</li></ul>	DEFERRED ACQUISITION COSTS (DAC)22.1819, 22.4043
· Dual currency bonds	DEFERRED COMPENSATION 16.25, 20.62
· Extinguishments of	DEFERRED COSTS8.102
<ul> <li>Finance companies</li></ul>	DEFERRED TAX ASSETS12.01, 16.2123
disclosure	DEFERRED TAX LIABILITIES. See also other assets, other liabilities, and other investments
reporting	Capital requirement       1.80         Income taxes       16.15–16, 16.21–23,         16.26, 16.29–30         Tier 1 capital       1.80

DEFICIENCIES, IN INTERNAL CONTROL	DEPOSITS—continued
MATTERS 5.71–.72, 5.74, 23.25	Internal control over financial
DELINQUENCY FEES8.97, 8.99	reporting
DEMAND DEPOSIT ACCOUNTS	· Liquidity risk
(DDA)	· Loans to deposits ratio
	· With other financial institutions 6.03–.05
DEMAND DEPOSITS	• Overview
DEMAND LOANS 8.17, 8.106	Payments function 13.17–.32, 13.53,
DEPOSIT INSURANCE FUND (DIF)1.17,	Product types
1.24–.30	Regulatory matters
DEPOSITORY AND LENDING INSTITUTIONS	· Savings deposits
· Audit considerations 5.01–.257	Substantive tests
Audit considerations	• Time deposits
Banks and savings institutions. See banks and	
savings institutions	DEPRECIATION
Business combinations	DERECOGNITION, TRANSFER10.32,
· Cash and cash equivalents6.01–.42	10.37
· Credit losses	DEREGULATION
· Credit unions. See credit unions	DERIVATIVE INSTRUMENTS18.01101
Debt (institution's borrowings) 15.01–.77	· Accounting standards18.79–.94
Deposits	• Analysis
· Equity and disclosure regarding capital	• Audit considerations 7.156, 18.101
matters	· Auditing considerations
· Fair value measurements and	· Banks' risk management of
disclosures	· Community banks, information for 1.47
· Federal funds and repurchase	· Credit derivatives
agreements14.0169	· Credit exposure measurement1.44, 1.46
· Finance companies. See finance companies	Defined
· Income taxes	· Designation18.79
· Insurance activities22.01–.70	· Disclosures
· Loan transfers and servicing 10.01–.127	· Embedded derivatives
· Loans	· Exchange-traded compared to
Mortgage companies. See mortgage	over-the-counter
companies and banking activities	• Fair value
Other assets, other liabilities, and other     investments	
investments	Financial statement presentation and
and other foreclosed assets 11.01–.73	disclosure
Securities investments	Forward-based derivatives. See also
Trust and asset management	swaps
activities	· Gains and losses on cash flow
DEPOSITORY INSTITUTION DEREGULATION	hedges17.09
AND MONETARY CONTROL ACT	· Guarantees
(1980)	· Hedging. See hedging
DEPOSITS	Interest rate risk (IRR)
· Accounting standards and financial	Intracompany transactions
reporting	Lending limits
· Audit considerations	· Loan commitments as8.104, 8.145–.150
Balance sheet presentation 6.32–.33	Mortgage companies and banking     activities
· Brokered deposits 5.242, 13.11–.13,	• Offsetting
	Option-based
· Classification	· Overview
· Closed accounts	Private company election to use standards
· Credit unions	issued by the private company
Demand deposits	council18.81
· Dormant accounts	· Regulatory matters 1.39–.41, 18.71–.78
· Financial statement presentation and	· Risk
disclosure	
· As funding source	· Risk management 18.55–.56, 18.71

DERIVATIVE INSTRUMENTS—continued	DISCLOSURES—continued
· Trust and asset management	· Foreign institution branches, regulatory capital
activities	disclosures for
Types of	• Guarantees
· Uses	<ul> <li>Held for sale</li></ul>
· Variations	• Income taxes
Warrants	Loan transfers and servicing 10.27=.30
DETECTION RISK5.09	Loans 7.133, 8.153–.181, 9.53–.62
DETERIORATED CREDIT QUALITY, LOANS	<ul> <li>Management's evaluation of substantial</li> </ul>
ACQUIRED WITH 7.125, 8.110122,	doubt
9.56, 19.21	Mortgage banking disclosure regarding capital
DEVELOPMENT ARRANGEMENTS. See	matters
acquisition, development, and	activities
construction (ADC) arrangements	Other assets, other liabilities, and other
DEVELOPMENT COSTS 11.4649	investments
DIF. See Deposit Insurance Fund	Other-than-temporary
DIRECT CONSUMER LOANS3.03	impairment
DIRECT DEPOSIT13.04, 13.27	and other foreclosed assets11.17,
DIRECT FINANCING LEASES 8.46	11.23, 11.30–.33, 11.39–.40, 11.43
DIRECT INVESTMENTS IN REAL ESTATE. See	· Regulatory report information,
also real estate investments, real estate	restrictions5.256
owned, and other foreclosed	Repurchase agreements 14.40–.47 Sales or transfers of securities 7.134–.135
assets	Securities
DIRECT LOAN ORIGINATION	Short-term debt
COSTS8.101102	· Significant estimates5.194–.201
DIRECT METHOD OF REPORTING CASH	· Significant risks and
FLOWS 6.15, 6.17, 6.24	uncertainties
DIRECT-RESPONSE	Troubled debt restructurings 8.166–.167  Trust banks, regulatory conital displayures.
ADVERTISING22.2732	Trust banks, regulatory capital disclosures for17.87–.88
DIRECT WRITING COMPANIES22.08	· Variable interest entities
DISCLAIMERS OF OPINION 23.09, 23.21	
DISCLOSURES	DISCOUNT RATES IN PRESENT VALUE
· Banks and savings institutions 17.28–.37,	TECHNIQUES 20.3133
17.41–.46	DISCOUNTS 7.107113, 8.224, 15.17
Business combinations	DISINTERMEDIATION1.0405
Capital matters. See equity and disclosure	DISPOSAL ACTIVITIES. See exit or disposal
regarding capital matters  Capital requirement1.79, 17.28–.46,	activities
	DIVIDEND INCOME
· Cash and cash equivalents 6.14, 6.3033	DIVIDENDS 2.12, 2.52, 13.01, 13.43
· Charging off uncollectable	DOCUMENTATION AND RECORD KEEPING
receivables8.156	· Audits. See audit documentation
Concentration-of-credit risk 7.136, 14.45	· Communication with those charged with
· Concentration risk 5.204–.209, 8.172–.174	governance5.74
· Corporate credit unions	· Confirmation of loan
Debt (institution's borrowings) 15.34–.39	documents
· Deferred tax liabilities	· Credit
· Deposits	· Credit loss allowance documentation 9.06,
Derivative instruments 18.95–.100	9.110
Fair value measurements18.96, 20.49–.60	· Exemption guidelines 9.33
Financial instruments 20.57–.60,	• Going concerns
· Financial statements. See financial statements	Inspection of borrowing documentation, federal
· Foreclosed assets	funds and repurchase agreements14.60

DOCUMENTATION AND DECORD	5004 0 F 10 100 1 11 A 1
DOCUMENTATION AND RECORD	ECOA. See Equal Credit Opportunity Act
KEEPING—continued Inspection of loans documents 8.212–.215	ECONOMIC CLIMATE, RISKS RESULTING
Loan files	FROM 5.03, 20.68
· Loan subsidiary records	ECONOMIC HEDGES18.57
· Long-term debt	EDUCATION DEPARTMENT1.118
· Trust and asset management	
activities	EFFECTIVE CONTROL FOR TRANSFERRED
DODD-FRANK WALL STREET REFORM AND	FINANCIAL ASSETS10.36
CONSUMER PROTECTION ACT	EFT SYSTEMS13.0405, 13.2629,
(2010)1.3169	
Collateralized loan obligations 7.78	ELECTRONIC AUDIT EVIDENCE 5.66
· Consumer Financial Protection	<b>EMBEDDED DERIVATIVES17.15, 18.51</b>
Bureau	EMERGENCY ECONOMIC STABILIZATION
Creditworthiness standards	ACT (2008)
Derivatives trading	
• Enactment	EMPHASIS-OF-MATTER (EOM) PARAGRAPHS23.01, 23.0713
· Financial Stability Oversight Council 1.33	·
<ul> <li>Foreign banking organizations with U.S.</li> </ul>	EMPLOYEE BENEFIT TRUSTS. See also
non-branch capitalization 17.27	related-party transactions 5.36,
· Full copy	21.11
Goals	EMPLOYEE RETIREMENT INCOME SECURITY
reporting1.113	ACT OF 1974 (ERISA) 21.11, 21.24
· Lending limits	EMPLOYEE STOCK OPTION PLANS20.62
<ul> <li>Leverage and risk-based capital</li> </ul>	EMPLOYEE STOCK PURCHASE
requirements	PLANS20.62
Loan appraisals	EMPLOYER WITHHOLDING13.10, 15.18
abolishment	ENFORCEABLE AT LAW18.87
· Ownership in "covered funds"	ENFORCEMENT,
Proprietary trading1.41, 1.54	REGULATORY 5.229–.236
· Regulatory capital deductions 17.20	ENGAGEMENT, UNABLE TO
Reserve rate ratio	CONTINUE5.148
Resolution plans	ENGAGEMENT LETTERS 1.100, 5.10–.12,
Securitization	5.256
· Stress testing	ENGAGEMENT TEAMS5.31
Thrift regulations	
· Volcker Rule	ENTIRE FINANCIAL ASSET, SALE OF10.44
7.86, 18.54, 18.77	ENTITY AND ITS ENVIRONMENT,
DOLLAR-ROLL REPURCHASE	UNDERSTANDING THE. See understanding the entity and its
AGREEMENTS14.0812, 14.14	environment
DORMANT ACCOUNTS13.14	
DOUBTFUL, LOANS CLASSIFIED AS 9.08,	EOM PARAGRAPH. See emphasis-of-matter (EOM) paragraphs
9.11	
DUAL CURRENCY BONDS 15.53	EQUAL CREDIT OPPORTUNITY ACT OF 1974
"DUE FROM BANKS"6.03	(ECOA)
DURATION ANALYSIS 5.45	EQUIPMENT. See also premises and equipment 12.85, 16.25
	• •
E	<b>EQUITY</b> • Common
FADRUNICO	Other components of, corporate credit
EARNINGS	union
Credit unions17.50–.52, 17.68–.69 From dividends or interest7.108	EQUITY AND DISCLOSURE REGARDING
Internal control over financial	CAPITAL MATTERS17.01115
reporting7.100	· Audit considerations
Retained 2.45, 17.07, 17.102	Banks and savings institutions 17.02–.46,
· Undivided17.50–.52, 17.68–.69	

EQUITY AND DISCLOSURE REGARDING CAPITAL MATTERS—continued  Banks/savings institutions	EXAMINATIONS—continued           Purpose         5.223           Ratings         5.228           Reports         5.249–.250           Requirements         5.224           Risk assessment         5.230           Risk Management Manual of Examination Policies (FDIC)         7.74           Scope         5.225           Specific targets         5.226           Targeted to specific area of operations         5.226
Overview         17.01           Substantive tests         17.100105           Trust operations         17.8788           Understanding the entity and its environment         17.95           EQUITY MODELS, ECONOMIC VALUE	EXAMINERS       5.223, 5.248257         EXCHANGE       Cross-currency exchange agreements       18.31         Disposal of problem assets through       11.09,
OF5.46 EQUITY PARTICIPANTS8.47	· Right to, transferred financial assets10.36
EQUITY SECURITIES. See also securities	EXCHANGE ACT. See Securities Exchange Act (1934 Act)
Common stock	EXECUTIVE COMPENSATION, CREDIT
investments	UNIONS2.52 EXIT OR DISPOSAL ACTIVITIES11.09,
Bank stock12.2734	
Financial statement disclosure 7.130 Other-than-temporary impairment 7.94	EXIT PRICE
	EXPECTED CASH FLOW 20.33
· OTTI. See other-than-temporary impairment	EXPECTED CASHTEOW20.33
Preferred stock15.44–.45, 17.04–.05,	EXPENSES
Preferred stock15.44–.45, 17.04–.05,	EXPENSES
Preferred stock15.44–.45, 17.04–.05,	EXPENSES  · Accrued
Preferred stock15.44–.45, 17.04–.05,	EXPENSES         Accrued       12.02         Interest expense       13.01, 13.66, 16.12         Nondeferrable       22.33–38         Total deposit interest expense to average total deposits ratio       5.30         Trust department, substantive tests of       21.23
Preferred stock	EXPENSES       . Accrued       . 12.02         . Interest expense       . 13.01, 13.66, 16.12         . Nondeferrable       . 22.33–.38         . Total deposit interest expense to average total deposits ratio       . 5.30         . Trust department, substantive tests of       . 21.23         EXPORT-IMPORT BANK       . 8.20, 8.50
Preferred stock	EXPENSES  Accrued
Preferred stock	EXPENSES
Preferred stock	EXPENSES  Accrued
Preferred stock	EXPENSES
Preferred stock	EXPENSES  Accrued
Preferred stock	EXPENSES

• Accounting standards 20.01–.68	FAST-PAY GINNIE MAE CERTIFICATES14.35
· Asset-backed securities 7.39, 7.75	FCRA. See Fair Credit Reporting Act
· Audit considerations	
· Business combinations19.16–.17, 19.35	FCUA. See Federal Credit Union Act
Cash flow reporting6.23	FDI ACT. See Federal Deposit Insurance Act
· Contributed assets12.73–.76	FDIC. See Federal Deposit Insurance
· Counterparty credit risk 20.21–.27	Corporation .
· Credit risk20.21–.27, 20.58	FDIC IMPROVEMENT ACT OF 1991. See
· Debt15.57	Federal Deposit Insurance Corporation
· Definitions	Improvement Act
Deposits	FEDERAL CREDIT UNION ACT (FCUA)
Derivatives18.8084, 18.8889, 18.96,	· Credit union
	• Enactment
Disclosures	· Field of membership 2.03
Fair value hierarchy	· Prompt corrective action 5.231
· Fair value option. See fair value option	Publication
Federal funds and repurchase	· Real property limitations 12.11
agreements	Supervisory committee
Foresland assets, not receivable	FEDERAL DEPOSIT INSURANCE ACT (FDI
Foreclosed assets, net receivable amount	ACT)
· Intangibles	Annual independent audits and reporting
Investments in certain entities that calculate net	requirements1.92
asset value per share20.60	Auditor qualifications
· Liabilities	Bank Insurance Fund1.16
· Loan transfers and servicing10.85	<ul> <li>Brokered deposit limitations 13.34, 13.36</li> </ul>
· Loans 8.170–.174, 8.230	· Communication with independent
Market participants20.08	auditors1.104, 5.250
Market risk	• Enforcement actions against auditors5.233
Mortgage loans and mortgage-backed	Going concerns
securities4.1617, 10.2125	Regulatory enforcement and
Mutual entities19.2324	reporting
· Nonfinancial assets 20.10–.13	<ul> <li>Reporting requirements</li></ul>
<ul> <li>Offsetting positions in market</li> </ul>	receivership or conservatorship 5.119
risks20.21–.27	·
· Overview	FEDERAL DEPOSIT INSURANCE
Present value techniques20.31–.33	CORPORATION (FDIC)
Price, guidance on	<ul><li>Auditor liability advisory</li></ul>
Real estate assets	Business combinations
Residuals	Capital requirement. See capital requirement
Sales of financial assets	· Credit losses
Securities 7.83, 7.87, 7.99, 7.105, 7.153	· Credit risk classifications9.12
<ul><li>Servicing assets or liabilities 10.56, 10.69</li><li>Shareholders' equity 20.14–.20</li></ul>	· Creditworthiness standards 7.77
Swap contracts	<ul> <li>Enforcement actions against</li> </ul>
Trusts	auditors
· Unit of account	· Establishment
· Using quoted prices provided by third	· Examinations. See examinations
parties	FFIEC membership1.11
· Valuation	· As independent agency of federal
· Valuation techniques	government
· When volume or level of activity has	Large Bank Pricing Rule
significantly decreased 20.42–.47	Laws protecting funds
FAIR VALUE OPTION	<ul> <li>Loan transfers and servicing10.09–.12</li> <li>Loss sharing purchase and</li> </ul>
· Loans	assumption
· For loans, compared to hedge	· OTS's authority transferred to1.55–.59
accounting8.92	· Quarterly Banking Profile
· Overview	Real estate lending standards8.57–.67

FEDERAL DEPOSIT INSURANCE	FEDERAL FUNDS—continued
CORPORATION (FDIC)—continued	· Purpose14.01
Resolution plans	Substantive tests
Resolutions Handbook 1.118, 23.35	Transaction types
Restoration Plan	FEDERAL HOME LOAN DANK BOARD 1.00
· Revision of insurance assessment	FEDERAL HOME LOAN BANK BOARD1.08
system	FEDERAL HOME LOAN BANKS (FHLB)
· Risk assessment procedures5.29	· Advances
· Risk Management Manual of Examination	· Cash and cash equivalents 6.35
Policies	Reserves6.06
· Role	SEC registration
Rules and regulations	· Stock 12.01, 12.04, 12.27–.34, 12.84,
Safe harbor for financial assets 10.13	
	· Structured notes issued by7.52
· Separate insurance funds under	•
FIRREA	FEDERAL HOME LOAN MORTGAGE
· Stress tests	CORPORATION (FHLMC, FREDDIE MAC).
FEDERAL DEPOSIT INSURANCE	See also mortgage-backed securities
CORPORATION IMPROVEMENT ACT	- Bonds 7.07
(FDICIA)	· Government bailout4.06
· Annual independent audits and	· Pass through security sales4.04
reporting	· Reporting requirements
· Credit union compliance 2.61	· Sale and service of mortgage loans for4.30
· History	<ul> <li>Secondary market participation 4.02–.03,</li> </ul>
Prompt corrective action	
• Purpose	4.05, 10.03
Sarbanes-Oxley Act compared to 1.109	Structured notes issued by
	· Uncollectability of receivables10.18
FEDERAL DEPOSIT INSURANCE REFORM ACT	FEDERAL HOUSING ADMINISTRATION
OF 2005 (REFORM ACT)1.17	(FHA)2.62, 4.02, 4.26, 4.29, 8.42,
FEDERAL FAMILY EDUCATION LOAN	10.18
PROGRAM	
	EEDEDAL HOUSING EINANGE AGENGV
	FEDERAL HOUSING FINANCE AGENCY
FEDERAL FINANCIAL INSTITUTIONS	(FHFA)4.26
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)	(FHFA) 4.26 FEDERAL HOUSING FINANCE
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC) Advisory on Interest Rate Risk	(FHFA)4.26
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC) · Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  · Advisory on Interest Rate Risk Management	(FHFA)       4.26         FEDERAL HOUSING FINANCE       12.27         FEDERAL LAWS. See specific federal
FEDERAL FINANCIAL INSTITUTIONS	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA) 4.26  FEDERAL HOUSING FINANCE BOARD 12.27  FEDERAL LAWS. See specific federal legislation, departments and regulations  FEDERAL NATIONAL MORTGAGE ASSOCIATION (FNMA, FANNIE MAE). See also mortgage-backed securities Bonds 7.07 Government bailout 4.06
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management 18.72 BSA/AML Examination Manual 1.21 Business combinations 19.04 Consolidated Reports of Condition and Income. See Call Reports Income taxes 16.14 Instructions for Preparation of Consolidated Reports of Condition and Income10.07,	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA) 4.26  FEDERAL HOUSING FINANCE BOARD 12.27  FEDERAL LAWS. See specific federal legislation, departments and regulations  FEDERAL NATIONAL MORTGAGE ASSOCIATION (FNMA, FANNIE MAE). See also mortgage-backed securities  Bonds 7.07  Government bailout 4.06  Sale and service of mortgage loans for4.30  Secondary market participation4.02–.03,4.05, 10.03  Structured notes issued by 7.52  Uncollectability of receivables 10.18
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management 18.72 BSA/AML Examination Manual 1.21 Business combinations 19.04 Consolidated Reports of Condition and Income. See Call Reports Income taxes 16.14 Instructions for Preparation of Consolidated Reports of Condition and Income 10.07, 11.05, 19.04 Investment security policy statement 7.67–.75 Loan loss allowance 9.20 Members 1.11 Role 1.11 Troubled debt restructurings 8.75 Uniform Financial Institutions Rating	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management 18.72  BSA/AML Examination Manual 1.21 Business combinations 19.04  Consolidated Reports of Condition and Income. See Call Reports Income taxes 16.14 Instructions for Preparation of Consolidated Reports of Condition and Income 10.07, 11.05, 19.04 Investment security policy statement 7.67–.75 Loan loss allowance 9.20 Members 1.11 Role 1.11 Troubled debt restructurings 8.75 Uniform Financial Institutions Rating System 5.228	(FHFA)       4.26         FEDERAL HOUSING FINANCE       BOARD       12.27         FEDERAL LAWS. See specific federal legislation, departments and regulations         FEDERAL NATIONAL MORTGAGE       ASSOCIATION (FNMA, FANNIE MAE). See also mortgage-backed securities         Bonds       7.07         Government bailout       4.06         Sale and service of mortgage loans for       4.30         Secondary market participation       4.05, 10.03         Structured notes issued by       7.52         Uncollectability of receivables       10.18         FEDERAL RESERVE ACT       15.33, 18.72         FEDERAL RESERVE BANK (FRB)       6.06,
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management 18.72 BSA/AML Examination Manual 1.21 Business combinations 19.04 Consolidated Reports of Condition and Income. See Call Reports Income taxes 16.14 Instructions for Preparation of Consolidated Reports of Condition and Income 10.07, 11.05, 19.04 Investment security policy statement 7.67–.75 Loan loss allowance 9.20 Members 1.11 Role 1.11 Troubled debt restructurings 8.75 Uniform Financial Institutions Rating System 5.228 FEDERAL FUNDS. See also cash and cash	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)       4.26         FEDERAL HOUSING FINANCE BOARD       12.27         FEDERAL LAWS. See specific federal legislation, departments and regulations       12.27         FEDERAL NATIONAL MORTGAGE ASSOCIATION (FNMA, FANNIE MAE). See also mortgage-backed securities       8004         Bonds       7.07         Government bailout       4.06         Sale and service of mortgage loans for       4.30         Secondary market participation       4.02-03,         Most of the company of the com
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management 18.72 BSA/AML Examination Manual 1.21 Business combinations 19.04 Consolidated Reports of Condition and Income. See Call Reports Income taxes 16.14 Instructions for Preparation of Consolidated Reports of Condition and Income 10.07, 11.05, 19.04 Investment security policy statement 7.67–.75 Loan loss allowance 9.20 Members 1.11 Role 1.11 Troubled debt restructurings 8.75 Uniform Financial Institutions Rating System 5.228 FEDERAL FUNDS. See also cash and cash equivalents Accounting standards 14.03, 14.27–.39 Audit considerations 14.48–.69 Definitions 6.07, 14.02 Federal funds sold 6.01, 6.07	(FHFA)       4.26         FEDERAL HOUSING FINANCE       BOARD       12.27         FEDERAL LAWS. See specific federal legislation, departments and regulations       legislation, departments and regulations         FEDERAL NATIONAL MORTGAGE       ASSOCIATION (FNMA, FANNIE MAE). See also mortgage-backed securities         Bonds       7.07         Government bailout       4.06         Sale and service of mortgage loans for       4.30         Secondary market participation       4.02-03,         Mostructured notes issued by       7.52         Uncollectability of receivables       10.18         FEDERAL RESERVE ACT       15.33, 18.72         FEDERAL RESERVE BANK (FRB)       6.06,         Mostructured interagency Regulations in Pronouncements index       4.04         ACH transfers       13.28         Board of Governors       5.12         BOPEC ratings       5.228         Business combinations       19.07
FEDERAL FINANCIAL INSTITUTIONS EXAMINATION COUNCIL (FFIEC)  Advisory on Interest Rate Risk Management	(FHFA)

FEDERAL RESERVE SYSTEM—continued	FIDUCIARY RISK5.55
· Consolidated Financial Statements for Bank	FIELDS OF MEMBERSHIP2.03
Holding Companies (Reporting Form	FINANCE COMPANIES
Y-9C)	· Asset-based financial services3.05
· Credit risk classifications9.12	· Business description 3.01–.11
· Credit union regulation	· Captive finance activity
· Debt	· Commercial financing 3.01, 3.05–.07
· Discount windows	· Consumer financing
· Disposal of problem assets through	Debt financing
exchanges11.09	services
· Enforcement actions against	Regulation and oversight3.12–.14
auditors	FINANCE SUBSIDIARIES15.2225
Examinations. See examinations     Federal funds. See federal funds	FINANCIAL ASSETS
FFIEC membership	Impairment of5.52–.54, 11.20–.23,
Holding company regulation	12.35–.38, 19.26
As independent agency of federal	· Isolation of transferred 10.39–.42
government	· Prepayment and prepayment risk10.48
· Loan transfers and servicing10.09–.12	· Sale of entire financial asset 10.44
<ul> <li>Market Risk Rule 1.76–.78</li> </ul>	· Sales
· Monetary policy 6.06	<ul><li>Securities. See securities</li><li>Transfers and servicing10.30–.42,</li></ul>
Mortgage servicing	
OREO management	FINANCIAL CRIMES ENFORCEMENT
OTS's authority transferred to1.55–.59	NETWORK (FINCEN)
Real estate investment regulations 11.06 Real estate lending standards 8.57–.67	FINANCIAL GUARANTEE
Regulation 0	CONTRACTS18.52
· Reserves	
• Resolution plans 1.60–.65	FINANCIAL HOLDING COMPANIES (FHC)
· Role	•
· Rules and regulations	FINANCIAL INFORMATION9.10, 9.15
overview	FINANCIAL INSTITUTIONS REFORM,
Small Bank Holding Company Policy 17.26	RECOVERY, AND ENFORCEMENT ACT
Stock	OF 1989 (FIRREA) 1.14–.15, 2.02
· Stress tests	FINANCIAL INSTRUMENTS. See also specific
FEDERAL SAVINGS	instruments 18.99, 20.57–.60
ASSOCIATIONS7.76–.78	FINANCIAL INTERMEDIATION PROCESS
FEDWIRE	
FEES	FINANCIAL POSITION STATEMENT 7.132
· Contractually specified servicing fees10.89	FINANCIAL REFORM1.06
Delinquency fees	FINANCIAL REPORTING. See financial
<ul> <li>Loan</li></ul>	statements
• Prepayment fees	FINANCIAL SERVICE MODERNIZATION ACT
FFIEC. See Federal Financial Institutions	(1999)
Examination Council	FINANCIAL SERVICES REGULATORY RELIEF ACT (2006)2.02
FHA. See Federal Housing Administration	FINANCIAL STABILITY OVERSIGHT COUNCIL
FHC. See financial holding companies	(FSOC)1.33, 1.60
FHFA. See Federal Housing Finance Agency	FINANCIAL STANDBY LETTERS OF
FHLB. See Federal Home Loan Banks	CREDIT
FHLMC. See Federal Home Loan Mortgage	FINANCIAL STATEMENTS  · Annual reporting requirements
Corporation	Additing of. See audit considerations
FIDUCIARIES, EMPLOYEE BENEFIT PLAN	· Auditor's report. See auditor's reports
TRUSTEES21.11	Balance sheets. See balance sheets

FINANCIAL STATEMENTS—continued	FINANCING ACTIVITIES, CASH FLOWS
· Capital requirement 1.79, 17.28–.46, 17.56–.64, 17.72–.79	FROM6.18, 6.20
• Cash and cash equivalents6.30–.33	FINCEN. See Financial Crimes Enforcement Network
· Cash flow statements. See cash flow	
statements	FIRREA. See Financial Institutions Reform, Recovery, and Enforcement Act
· Collateralized debt obligations 7.130	FIXED COUPON AGREEMENTS14.08
Concentrations, current vulnerability     to	14.06
· Credit loss	FIXED-RATE-CURRENCY SWAPS18.32
· Debt (institution's borrowings) 15.34–.57	
· Deposits	FIXED-RATE RESIDUALS
Derivatives	FLOATING-RATE RESIDUALS7.47
Disclosures. See disclosures     Fatimates.	FLOOR PLAN LOANS8.30
Estimates	FNMA. See Federal National Mortgage
Financial asset sales	Association
Foreclosed assets	FORECLOSED ASSETS
• Forming opinion on	· Accounting standards 8.137, 11.06–.07
Going concern effects 5.173–.174	
· Guarantees 8.178–.182	· Audit considerations
· Held for sale10.79–.83, 10.85, 11.20,	· Characteristics of initial
	measurement
Impaired loans	Defined
Income statement	Discontinued proceedings
· Income taxes	Financial statement presentation and
· Insurance activities	disclosure
Intangible assets	• Inherent risk
	· Initial measurement of foreclosed
· Loans 7.133, 8.153–.181, 9.53–.62	property
Mortgage companies and banking	Loan modifications
activities4.33–.37, 17.80–.84	· Net receivable amount of receivable or fair
<ul> <li>Other assets, other liabilities, and other</li> </ul>	value11.20
investments	FORECLOSURES
· Other-than-temporary	FOREIGN BANKING ORGANIZATIONS, U.S.
impairment	NON-BRANCH CAPITALIZATION 17.27
Real estate investments, real estate owned,	FOREIGN CHECKS
and other foreclosed assets11.17,	FOREIGN CURRENCY DEBT15.52
11.23, 11.30–.33, 11.39–.40, 11.43	FOREIGN-CURRENCY FUTURES
· Sales or transfers of securities 7.134–.135	CONTRACTS18.21
Secured borrowing and collateral10.87 Securities	FOREIGN-CURRENCY SWAPS18.31
Segment reporting	
Servicing assets and liabilities 10.89–.91	FOREIGN EXCHANGE RISK8.07
· Short-term debt	FOREIGN GOVERNMENT DEBT
· Significant risks and	SECURITIES7.60
uncertainties5.193–.203	FOREIGN INSTITUTION BRANCH
Special purpose framework23.22–.23	REGULATORY CAPITAL
· Substantive tests related to	DISCLOSURES 17.85–.86
audits	FOREIGN LOANS 8.07, 8.54, 8.141142
activities	9.14, 9.88
· Uniform Commercial Code8.86–.87	FORWARD-BASED DERIVATIVES. See also
· Variable interest entities7.115,	swaps 18.16, 18.1924
	FORWARD-RATE AGREEMENTS18.20
FINANCIAL STRUCTURE OF CREDIT	FORWARD STANDBY
UNIONS2.1213	COMMITMENTS

FRAUD, CONSIDERATION OF IN FINANCIAL STATEMENT AUDIT	GENERALLY ACCEPTED ACCOUNTING PRINCIPLES (GAAP)—continued  Disclosures for financial instruments 20.57  Equity and disclosure regarding capital matters 17.53, 17.90, 17.94, 17.101, 17.105  Going concerns 5.166 Income taxes 16.14, 16.19 Individually evaluated loans 9.95 Insurance activities 22.55 Interpretation and reporting related to 5.243–247, Appendix B Investments in securities 7.143 Loan documentation 9.33 Management estimates of credit losses 9.04 Perpetual contributed capital alternative adoption 23.13, 23.33 Real estate lending standards 8.62 Regulatory reporting matters Appendix B Related-party transactions 5.37 Substantive procedures 17.101, 17.105 GENERALLY ACCEPTED AUDITING STANDARDS (GAAS)
Council	Audit considerations
FUNCTIONAL REGULATION1.1819 FUTURES CONTRACTS18.2124	Audit documentation
_	Auditor specialist definition
G	Mortgage activity planning17.108
GAAP. See generally accepted accounting principles	GLOBAL ISSUES7.60, 8.49–.50,
GAAS. See generally accepted auditing standards	GLOBALLY ACTIVE BANKS 8.4950, 17.18
GAINS OR LOSSES         0n cash flow hedges         17.09           Extinguishments         15.50           Real estate investment joint ventures         11.54	GNMA. See Government National Mortgage Association GOING CONCERNS
Sale or transfer of assets	Auditor's report effects5.176, 23.08     Combining with other emphasis-of-matters
· Unrealized. See unrealized gains and losses  GAP ANALYSIS	Corporate credit unions
GENERAL PARTNERSHIPS11.55	· Overview
GENERALLY ACCEPTED ACCOUNTING PRINCIPLES (GAAP)  · Allowance for loan and lease losses9.23,9.42–.48, 9.54	Regulatory issues
Auditor's communication with those charged with governance	Accounting standards

GOVERNMENT NATIONAL MORTGAGE ASSOCIATION (GNMA, GINNIE MAE)           Bonds         7.07           Fast-pay Ginnie Mae certificates         14.35           MBS. See mortgage-backed securities         Reporting requirements         2.62           Role         4.31           Secondary market participation         4.02–.04,           10.03         Slow-pay Ginnie Mae certificates         14.35           Uncollectability of receivables         10.18           GOVERNMENT-SPONSORED ENTERPRISES	HELD FOR SALE (HFS)  - Accounting standards
(GSE)	Mortgage loans
GROSS PROFIT, ESTIMATED, FOR INSURANCE ACTIVITIES 22.44–.50 GROUPS OF LOANS, CREDIT LOSSES	Substantive tests
GSE. See government-sponsored enterprises GUARANTEE LIABILITIES	Defined       7.83         Designation       7.90         Disclosure       7.128
GUARANTEED MORTGAGE LOANS11.16,	Other-than-temporary impairment. See also other-than-temporary impairment 7.93  Sales or transfers 7.85, 7.135
GUARANTEES. See also other assets, other liabilities, and other investments  Bankers' acceptances	HELPING FAMILIES SAVE THEIR HOMES ACT (2009)
As derivative instrument	HIGH VOLATILITY COMMERCIAL REAL ESTATE (HVCRE) LOANS8.139
	ESTATE (HVCRE) LOANS8.139 HIGHEST AND BEST USE FOR NONFINANCIAL ASSETS20.1013 HMDA. See Home Mortgage Disclosure Act
Financial statement presentation and disclosure	ESTATE (HVCRE) LOANS8.139 HIGHEST AND BEST USE FOR NONFINANCIAL ASSETS20.1013 HMDA. See Home Mortgage Disclosure Act HOLDING COMPANIES  · Annual independent audits and reporting
Financial statement presentation and disclosure	ESTATE (HVCRE) LOANS
Financial statement presentation and disclosure	HIGHEST AND BEST USE FOR NONFINANCIAL ASSETS

HOME EQUITY LINES OF CREDIT8.41	INCOME TAXES—continued
HOME MORTGAGE DISCLOSURE ACT OF 1975 (HMDA)8.84	· Deferred-tax assets and liabilities
HTM SECURITIES. See held-to-maturity (HTM) securities	Definition of banks and savings institutions
HUD. See U.S. Department of Housing and Urban Development	Financial statement presentation and disclosure
HVCRE LOANS. See high volatility commercial real estate (HVCRE) loans	income
HYBRID ADJUSTABLE RATE MORTGAGE PRODUCTS4.25	reporting
1	Net operating losses         16.09           Overview         16.0102
ICERC. See Interagency Country Exposure Risk Committee	Regulatory matters
IDENTIFIABLE INTANGIBLE ASSETS	Substantive tests
IDI. See insured depository institutions	· Temporary differences 16.24–.26,
IMAGE REPLACEMENT DOCUMENTS (IRD)	
IMPAIRED FINANCIAL ASSETS	INCREMENTAL DIRECT COSTS OF
· Audit considerations5.54	CONTRACT ACQUISITION22.2122 INDEMNIFICATION ASSETS 12.01, 12.48
• Examples	INDEPENDENT AUDITORS. See auditors
Long-lived assets	INDEX AMORTIZING SWAPS 18.69
· NCUSIF deposits	
IMPAIRED LOANS	INDIRECT CONSUMER LOANS 3.03
Collective evaluation of	INDIRECT METHOD OF REPORTING CASH FLOWS6.15, 6.17
	INDIVIDUAL RETIREMENT ACCOUNTS (IRA)13.06, 13.08
9.95–97 Substantive tests 9.89 Troubled debt restructurings 8.136, 8.164	INDIVIDUALLY EVALUATED LOANS, CREDIT LOSSES9.16, 9.19, 9.45–.48, 9.95–.97
IMPAIRMENT, OTHER-THAN-TEMPORARY.  See other-than-temporary impairment	INDUSTRIAL LOANS. See commercial, industrial, and agricultural loans
(OTTI)	INDUSTRY RISK FACTORS5.4057
INCOME  Comprehensive. See comprehensive income	INFORMATION, FINANCIAL. See financial information
<ul> <li>Dividend</li></ul>	INFORMATION AND COMMUNICATION SYSTEMS
· Premium	INFORMATION SOURCES Appendix C
Recognition on problem loans8.81–.82 Service charge	INFORMATION TECHNOLOGY (IT)
INCOME APPROACH VALUATION 20.28	INHERENT RISK5.09
INCOME STATEMENT10.82	INPUTS, IN FAIR VALUE
INCOME TAXES16.0136	HIERARCHY20.34–.37, 20.39
· Accounting standards	INSIDER RISK8.07
Alternative minimum tax	INSPECTION 8.212215, 14.60
· Capital requirement calculations 16.14 · Credit unions 12.70, 16.01, 16.13	INSTALLMENT LOANS8.17, 8.32, 8.100, 8.194

INSURANCE ACTIVITIES22.0170	INTEREST-RATE CAPS 18.4547
· Accounting standards 22.12–.70	INTEREST-RATE COLLARS18.49
Acquisition costs	
· Audit considerations 22.69–.70	INTEREST-RATE FLOORS 18.48
· Commissions22.10, 22.42, 22.56–.59	INTEREST RATE RISK (IRR)
· Consolidation	· Asset/liability management 5.43–.45
· Credit accident and health insurance22.02,	Banks and savings institutions
	· Loans
· Finance companies 3.04	· Mortgage-backed securities
Financial statement presentation22.54,	Mortgage banking
22.65–.68	· Overview 5.41–.49
Investment portfolios	Prepayment risk5.49
• Overview	Security investments
Policy writing	-
Property and liability insurance 22.13–17	INTEREST-RATE SWAPS18.2628
	INTEREST RECEIVABLE, ACCRUED. See
· Real estate	accrued interest receivable
· Regulatory matters22.11	INTERNAL AUDIT DEPARTMENTS9.67
• State laws	INTERNAL CONTROL OVER FINANCIAL
· Types of coverage22.02	REPORTING 5.58–.75
INSURANCE AND OTHER ANCILLARY	· Annual report requirements 1.106,
PRODUCTS AND SERVICES3.04	
INSURED DEPOSITORY INSTITUTIONS	· Auditor's reports
(IDI) 1.24–.29, 1.34–.35, 1.60–.65	· Cash and cash equivalents6.36–.40
INTANGIBLE ASSETS	· Communication with those charged with
· Accounting standards 12.08, 12.35–.50	governance5.70–.75, 23.25–.27
<ul> <li>Acquisitions in business combination 19.15</li> <li>Banks and savings institutions,</li> </ul>	· Components
limitations12.12	· Credit losses
· Defined	Debt
<ul> <li>Defined as other assets or other</li> </ul>	· Deposits
investments	· Derivatives
Fair value measurements 12.36–.48	· Equity and disclosure regarding capital
Financial statement presentation12.49–.50	matters17.98–.99, 17.110–.111
Goodwill. See goodwill Identifiable	<ul> <li>Federal funds</li></ul>
· Identified, other than goodwill12.39–.47	· Income taxes
Impairment12.35–.38, 12.45, 19.26	· Information technology
Substantive tests	considerations5.64–.69
INTERAGENCY COUNTRY EXPOSURE RISK	<ul> <li>Lending policies and procedures 8.08–.16</li> </ul>
COMMITTEE (ICERC)9.14	Loan transfers and servicing 10.119–.125
INTEREST	Loans
· Participating10.43, 10.71	activities
· Unearned 8.17	Other assets, other liabilities, and other
INTEREST-BEARING DEPOSITS6.32	investments
INTEREST CLASSES, MORTGAGE-BACKED	· Overview 5.58–.59
SECURITIES	Purpose
INTEREST EXPENSE13.01, 13.66, 16.12	Real estate investments, real estate owned,
INTEREST INCOME7.108, 7.114, 8.95,	and other foreclosed assets 11.58–.61 Repurchase agreements 14.54–.58
8.144, 8.221, 16.25	· Risk assessment procedures 5.58–.75
INTEREST METHOD7.110113	· SEC reports
INTEREST-ONLY (IO) SECURITIES7.29,	Security investments
	Substantive tests. See substantive tests     Tests of controls. See tests of controls.
INTEREST PAYABLE, ACCRUED. See accrued	<ul> <li>Tests of controls. See tests of controls</li> <li>Trust and asset management</li> </ul>
interest payable	activities
	200.000

INTERNAL REPLACEMENT TRANSACTIONS	J
INTERNAL REVENUE CODE	JOINT REGISTRAR TRANSFER
(IRC)	AGENTS
INTERNAL REVENUE SERVICE (IRS). See also	JOINT VENTURES
income taxes	JUNIOR LIENS
INTERNATIONAL ISSUES	JUNIOR SUBORDINATED DEBT3.09
INTERNATIONAL SUPERVISION ACT	K
(1983)9.14	KEOGH ACCOUNTS13.05
INTERNATIONAL SWAPS AND DERIVATIVES ASSOCIATION MASTER	KICK-OUT RIGHTS10.102
CONTRACTS	L
INTERNATIONALLY ACTIVE BANKS8.4950, 17.18	- LABOE DANK PRIOING BUILE 1 27 20
INTRINSIC VALUE	LARGE BANK PRICING RULE 1.2729
INVENTORY LOANS8.27	LAWS AND LEGISLATION. See specific laws or legislative acts
INVESTMENT COMPANIES	LEASE BROKERS8.48
INVESTMENT COMPANY ACT OF 1940 (1940	LEASE FINANCING
ACT)21.12	Accounting standards 8.140, 15.55
INVESTMENT CONTRACTS, INSURANCE ACTIVITIES22.48	<ul> <li>Allowance for. See allowance for loan and lease losses (ALLL)</li> </ul>
INVESTMENT INCOME TO AVERAGE TOTAL SECURITIES RATIO	· Audit considerations
INVESTMENT VALUATION RESERVE, CREDIT UNIONS	
INVESTMENTS	· Disclosure
• Cash flows from	Financial statement disclosure
· Fair value measurements and	· Loans
disclosures	· Operating leases 8.45, 12.25
disclosures	<ul> <li>Operating leases</li></ul>
disclosures	· Operating leases 8.45, 12.25
disclosures	Operating leases
disclosures	Operating leases       8.45, 12.25         Private company lessee and lessor, variable interest entities       22.61         Sale-leaseback transactions       12.22         Substantive tests       8.217         Transactional lease financing       8.48
disclosures	Operating leases
disclosures	Operating leases       8.45, 12.25         Private company lessee and lessor, variable interest entities       22.61         Sale-leaseback transactions       12.22         Substantive tests       8.217         Transactional lease financing       8.48
disclosures	Operating leases

LEVERAGED LENDING8.80	LOAN LOSS RECOVERIES TO PRIOR-YEAR
LIABILITIES. See also other assets, other liabilities, and other investments  · Asset/liability management	WRITE-OFFS RATIO
· Audit objectives for other	LOAN MODIFICATION PROGRAMS 4.26
· Borrowings. See debt (institution's borrowings)	LOAN OFFICERS
• Contingent	LOAN REVIEW
Contingent liabilities	Auditor's communication with       9.67         Collateral       9.102–.104         Credit losses       9.13–.15         Exemptions       9.98         Foreign loans       9.14         Groups of loans       9.93–.94
Insurance	· Independence of
<ul> <li>Limitations on</li></ul>	Management controls
Right of setoff	· Sample form Exhibit 9-2 at 9.99
· Sale of financial assets	· Substantial tests9.95–.105
• Servicing	LOAN SYNDICATIONS10.07
<ul> <li>Shareholders' equity 5.30, 20.14–.20</li> <li>LIABILITIES TO SHAREHOLDERS' EQUITY</li> </ul>	LOAN-TO-VALUE (LTV) RATIO 8.57, 8.60,
RATIO5.30	8.68 LOAN TRANSFERS AND
LIBOR (LONDON INTERBANK OFFERED	SERVICING
RATE) 18.33, 18.35, 18.37, 18.81	· Accounting standards and financial
LIFE INSURANCE. See also other assets,	reporting
other liabilities, and other	Advances
investments	Asset backed securitization
LINES OF CREDIT8.17, 15.1416	Audit considerations 5.11, 10.114–.127
LIP. See loans in process	Authoritative guidance8.198
LIQUIDITY2.23, 7.02	· Consolidation
LIQUIDITY IN MARKET7.25	Financial assets
LIQUIDITY LOANS8.17	
LIQUIDITY RISK 5.50–.51, 7.71, 7.140,	Held for sale10.20–.29, 10.74–.78
	Internal control over financial reporting
LITIGATION ALLOWANCE. See also other assets, other liabilities, and other investments	Loan participation 10.01, 10.03, 10.07,
LIVING TRUSTS	· Overview
LOAN APPROVAL PROCESS8.10	• Recourse loans
	Regulatory matters
LOAN COLLECTION RISK. See also credit losses	· Secured borrowings and
LOAN COMMITMENTS	collateral10.70–.73
· Accounting standards8.145–.152	Securitization. See securitization
· Allowance for unfunded12.02	<ul> <li>Servicing agreements</li></ul>
· As derivative instrument8.104,	Substantive tests10.126–.127
8.145–.150, 18.92	· Tests of controls
• Fair value option	· Variable interest entities 10.97–.113
Forward standby commitments8.149–.152	LOANS 8.01230
· Written	Accounting standards
LOAN FILES 8.15, 8.191196	<ul> <li>Acquired with deteriorated credit quality 7.125, 8.110–.122, 9.56, 19.21</li> </ul>
LOAN INCOME TO AVERAGE NET LOANS RATIO	Allowance for loan losses. See allowance for loan and lease losses (ALLL)

LOANS—continued	LOANS—continued
· Analytical procedures9.109	Prepayment fees
· Audit considerations 8.182–.230	Prepayment penalties
Bank Accounting Advisory Series	Regulation O loans         5.3839           Regulatory matters         8.5784
(OCC)	Related-party transactions
acquisition and sales of 6.23–.24	· Residential mortgage. See residential
· Categories	mortgage loans
· Charge-offs and recoveries 9.53, 9.108	Restructuring. See also troubled debt
· Collateral. See collateral	restructurings (TDR)
· Collectively evaluated for	Retail credit loans
impairment	• Secondary market 4.01–.06, 4.08,
<ul> <li>Commercial, industrial, and agricultural. See commercial, industrial, and agricultural loans</li> </ul>	4.11–.12, 10.02–.03
Commercial real estate (CRE)	· Servicing. See loan transfers and servicing
loans	<ul><li>Substantive tests8.190, 8.199–.230</li><li>Syndication lending arrangements8.55–.56</li></ul>
Commitments8.103104, 8.145152,	Trade financing
16.25, 18.92–.93	· Transfers. See loan transfers and servicing
· Construction loans8.17, 8.52, 8.216	· Troubled debt restructurings 8.71–.75,
· Consumer. See consumer loans	8.123–.138
Consumer loans	· Understanding the entity and its
Credit cards 8.76–.77	environment
· Credit risk 8.05–.07	LOANS BY TYPE TO TOTAL LOANS
· Credit strategy8.03–.04	RATIO5.30
· Credit union lending restrictions 8.83	
Debt securities acquired with deteriorated	LOANS IN PROCESS (LIP) 8.225–.226
credit quality	LOANS TO DEPOSITS RATIO5.30
Delinquency fees	LOANS TO TOTAL ASSETS RATIO5.30
Documents8.15, 8.191–.196, 8.212–.215	LOANS WITH RECOURSE, TRANSFERS AND
Fees, costs, discounts, and	SERVICING10.4950
premiums 8.101–.109, 8.224, 16.25	LONG-DURATION CONTRACTS22.1415
· Financial statement presentation and	LONG-LIVED ASSETS
disclosure7.133, 8.153–.181, 9.53–.62 • Foreclosed assets	• To be disposed by sale
· Foreign. See foreign loans	To be held and used
· Guarantees. See guarantees	Initial measurement of foreclosed
· Impaired loans 8.136, 8.164, 9.55,	property 11.17
9.59–.60, 9.89	Of lessors subject to operating
· Income recognition on problem loans8.1–.82	leases
Interest income	<ul> <li>Premises and equipment. See premises and equipment</li> </ul>
· Internal control over financial	Presentation when classified as
reporting	held-for-sale11.36
· Involving more than one lender8.55–.56	· Recoverability
Lease financing. See lease financing	LONG-TERM DEBT15.0608, 15.3439
Lending arrangements	LOOK-BACK PERIOD9.17
	LOSS
- Lending statutes8.84	· Accrued loss contingencies16.25
· Leveraged lending8.80	Contingencies for
· Loan files	· Credit losses contingencies9.53–.54
Losses. See credit losses	· Loans classified as 9.08, 9.11
• Modifications	Swap contracts
Nontraditional residential mortgage	LOSS SHARING TRANSACTIONS
products 8.78	LOW-INCOME DESIGNATED CREDIT
Overview	UNIONS17.55
Participation arrangements	LTV. See loan-to-value (LTV) ratio
9.100–.107	LIV. See Ivali-to-value (LIV) fatto

M	MARKETING OREO RESIDENTIAL
MAGNETIC INK CHARACTER RECOGNITION	PROPERTIES11.11, 11.15
(MICR)13.18, 13.32	MASTER TRUSTS
MAJORITY-OWNED SUBSIDIARY12.72	MATERIAL MISSTATEMENTS, RISK OF
MANAGEMENT. See also those charged with governance	At assertion level
Allowance for loans. See allowance for loan	<ul> <li>Audit considerations5.124, 5.137–.143</li> <li>Auditor's detection of5.238–.239</li> </ul>
and lease losses	Business combinations
· Assertions 5.21–.22, 5.37	· Cash and cash equivalents6.35
· Auditor engagement terms5.13	· Credit losses9.64, 9.69–.70
Communication with	· Debt15.59
· Compliance with laws and regulations5.11,	Depository institution regulation and supervision
· Credit loss estimation methodology9.04–.12	Deposits
· Credit strategy8.03	<ul> <li>Due to fraud. See fraud, consideration of in financial statement audit</li> </ul>
· Defined as related party. See also related-party	Equity and disclosure regarding capital
transactions5.36	matters
Discontinued foreclosures 11.07	· Evaluation of misstatements identified during
Evaluation of substantial doubt5.174	audit
Going concerns, plans to continue as5.170–.171	• Expressing
Investment oversight 7.70–.71, 7.150	Federal funds and repurchase agreements
· Loan evaluation	Further audit procedures 5.76–.98
· Loan review controls 9.74–.75	· Identification of5.79–.81
· Override of controls 5.142–.143	· Lending activities indicative of8.186
Reports from 1.97, 1.108, 1.117	· Loans8.183–.185
• Representations by	Other assets, other liabilities, and other investments
	At overall financial statement level 5.07,
· Subjective analysis by	
· Substantial doubt about entity's ability to	· Overall responses to 5.83–.84
continue as a going concern 17.34–.35	· Overview
• Written representations from5.175, 5.180–.189	Real estate investments, real estate owned,
MANAGEMENT RISK 8.07	and other foreclosed assets 11.57–.61 Regulatory matters indicative of 17.97
	Risk assessment procedures. See risk
MANAGEMENT'S DISCUSSION AND ANALYSIS (MD&A)18.91	assessment
MANDATORY REDEEMABLE PREFERRED	• Security investments 7.139, 7.144, 7.149,
STOCK	
MARGINALLY CAPITALIZED CREDIT UNION	Tolerable misstatement distinguished
CLASSIFICATION2.39, 2.41, 17.64	from5.19
MARK TO MARKET SECURITIES16.11	Trust and asset management     activities
MARKET APPROACH VALUATION20.28	MATERIAL WEAKNESS 5.63, 5.7172,
MARKET LIQUIDITY, PREPAYMENT 7.25	5.74
MARKET PARTICIPANTS, FAIR VALUE	MATERIALITY5.1620, 5.186
MEASUREMENTS AND DISCLOSURES20.08	MATURITY FACTORING8.24
MARKET PREMIUM, ESTIMATING20.47	MBB. See mortgage-backed bonds
MARKET RISK	MBS. See mortgage-backed securities
· Capital requirement 1.76–.78	MD&A. See Management's Discussion and
Derivative instruments	Analysis
Fair value measurements20.21–.27, 20.59	MEETING MINUTES5.35
Repurchase agreements (repos) 14.16–.18	MEETINGS, OF EXAMINERS WITH
MARKET RISK RULE (MRR) 1.7678	INSTITUTIONS 5.252–.253. 5.256

MEMBERS' EQUITY17.4854	MORTGAGE-BACKED SECURITIES
MEMBERSHIP FIELDS2.03	(MBS)—continued
MERGERS. See business combinations	Fair value
MEZZANINE DEBT17.24	HFS. See held for sale
MICR. See magnetic ink character recognition	History
MINIMAL NET WORTH CAPITAL REQUIREMENTS 17.80–.81	securities
MINIMALLY CAPITALIZED CREDIT UNION CLASSIFICATION2.39, 2.41, 17.64	As mortgage swap
MINIMUM CAPITAL REQUIREMENTS. See also net worth standard2.06, 2.48, 17.69,	Plain vanilla
MINIMUM RATIO OF CAPITAL1.71	Repurchase agreements14.07, 14.09,
	14.14
MINUTES AND BYLAWS, CREDIT UNIONS	Risk analysis
MISAPPROPRIATION OF	Risk characteristics
ASSETSExhibit 5-1 at 5.257	Secondary market
MISSTATEMENTS. See material	MORTGAGE COMPANIES AND BANKING
misstatements, risk of	ACTIVITIES4.0137
MMDA. See money-market deposit accounts	· Accounting and reporting
MODEL METHOD1.43	standards
	• Activities
MODEL RISK	· Asset servicing for investors4.36–.37 · Audit objectives
MODERATELY CAPITALIZED CREDIT UNION CLASSIFICATION 2.39–.41, 17.64	· Audit planning17.107–.109
MODIFICATIONS	Business description
Debt (institution's borrowings)	Computer systems support
· Loans4.26, 8.71, 8.108, 8.163–.166	matters17.80–.84, 17.106–.115
· Troubled debt restructurings 8.71–.75,	• Funding
8.132–.133, 8.163–.166	· Hedge accounting approach 18.81
MODIFIED OPINIONS	HUD programs. See also U.S. Department of
MONETARY POLICY6.06	Housing and Urban Development4.33–.35
MONEY LAUNDERING	· Interest rate risk4.18
PREVENTION	· Internal control over financial
MONEY-MARKET DEPOSIT ACCOUNTS	reporting
(MMDA)13.05	Loans. See residential mortgage loans
MONITORING, INTERNAL CONTROL 5.61	Net worth standard
MORE LIKELY THAN NOT, DEFINED 16.21	Regulation and oversight4.19
MORTGAGE-BACKED BONDS	Reporting considerations 4.33–.37
(MBB) 15.06, 15.21, 15.47–.48	Substantive tests
MORTGAGE-BACKED SECURITIES	· Understanding activities10.116
(MBS)	MORTGAGE LOANS. See residential
· Accounting standards 7.88, 10.20–.29 · Adjustable-rate mortgage 7.27	mortgage loans
· As asset-backed security	MORTGAGE PARTNERSHIP FINANCE (MPF)
· Collateral	PROGRAM 4.09
Collateralized mortgage obligations 7.31,	MORTGAGE PURCHASE PROGRAM
	(MPP)4.09
· Complex structures	MORTGAGE-REVENUE BONDS15.06
· Defined	MORTGAGE SERVICING ADVANCES 12.01
$\cdot$ Defined as structured credit product $\dots\dots7.51$	MORTGAGE SERVICING ASSETS 12.12,
· Dollar-roll repurchase agreements 14.14	

MORTGAGE SWAPS18.3738	NET WORTH STANDARD
MOST ADVANTAGEOUS MARKET20.0407	CUMAA provisions2.06, 2.24, 5.234 Mortgage companies and banking
MRR. See Market Risk Rule MULTIPLE COMMON BOND CREDIT UNIONS2.03	activities
MUNICIPAL OBLIGATIONS7.0811	
MUTUAL ENTITIES19.2225	· New credit unions2.39–.42
MUTUAL FUNDS. See also securities	<ul> <li>Noncompliance</li></ul>
N	Risk-based net worth requirement 2.28–.38 Service organizations10.117, 17.83
NATIONAL ACCOMPTION OF FEDERAL	NETTING AGREEMENTS18.08, 18.30
NATIONAL ASSOCIATION OF FEDERAL CREDIT UNIONS	NEW CREDIT UNIONS 2.24, 2.3942, 17.55, 17.64
NATIONAL CREDIT UNION ADMINISTRATION (NCUA)  • Call Reports filed with	1934 ACT. See Securities Exchange Act (1934 Act)
Corporate credit unions	NO-BID OPTION10.1617
· Credit losses from credit unions 9.34, 9.37	NOL. See net operating losses
<ul> <li>Minimum capital requirements</li></ul>	NONACCRETABLE DIFFERENCE 8.114, 8.119
17.76	NONACCRUAL POLICIES8.8182
Regular reserve (statutory reserve)17.48	NONCASH COLLATERAL10.73
State credit unions	NONCASH INVESTING AND FINANCING ACTIVITIES6.2122
INSURANCE FUND (NCUSIF)2.19, 	NONCOMPLIANCE. See also material misstatements, risk of 2.44, 5.154, 5.158–162, 5.239
DEPOSITS	NONCONSOLIDATION OPINION10.40
NATURAL PERSON CREDIT UNIONS2.03,	NONDEFERRABLE EXPENSES22.33–.38
2.17, 2.24–.44, 17.56–.64	NONDEPOSIT INVESTMENT PRODUCTS. See
NCA. See nonperpetual capital accounts	also trust and asset management
NCUA. See National Credit Union Administration	activities
NCUSIF. See National Credit Union Share Insurance Fund	NONINTEREST INCOME TO TOTAL REVENUE RATIO5.30
NEGATIVE AMORTIZATION	NONPERFORMANCE RISK20.18
MORTGAGES	NONPERPETUAL CAPITAL, CORPORATE
NET ASSET VALUE PER SHARE20.3741, 20.60	CREDIT UNION2.46
NET CASH FLOWS6.25–.28	NONPERPETUAL CAPITAL ACCOUNTS (NCA)17.66
NET INCOME, UNREALIZED GAINS AND	NONSUPERVISED MORTGAGEES4.33
LOSSES7.106	NOTIFICATION TO LOCAL AUTHORITIES,
NET INCOME TO AVERAGE CAPITAL RATIO5.30	DISCONTINUED FORECLOSURES
NET INCOME TO AVERAGE TOTAL ASSETS RATIO5.30	NOTIONAL AMOUNT18.05, 18.20, 18.45,
NET INTEREST INCOME MODELS 5.46	
NET OPERATING LOSSES	0
(NOL)16.09–.10, 16.36	OCC. See Office of the Comptroller of the
NET WORTH, OF CREDIT UNION 2.25	Currency

OFF-BALANCE-SHEET INSTRUMENTS	OPERATIONAL (TRANSACTION)
· Credit losses 9.02, 9.21, 9.54, 9.57, 9.83	RISK
· Credit risk8.175–.176	OPINIONS
· Risk management	· Disclaimers of
OFFICE OF THE COMPTROLLER OF THE	• Modified opinions
CURRENCY (OCC). See also Federal	Not opinion on fair presentation of institution's
Interagency Regulations in Pronouncements index	regulatory reports
Auditor liability advisory	· Qualified opinions
· Borrowings by savings institutions15.32	· Reporting considerations 23.05–.21
· Business combinations19.05, 19.07	· Unmodified opinion23.05–.06
· Capital requirement. See capital requirement	· Unqualified opinions in PCAOB
Credit losses	reports
Enforcement actions against	OPTION-BASED DERIVATIVES18.16,18.40–.52
auditors	
· Establishment	OPTION CONTRACTS
· Examinations. See examinations	ORDERLY TRANSACTIONS 20.02, 20.42,
· FFIEC membership	
· As independent bureau of Treasury  Department	OREO. See other real estate owned
· Loan transfers and servicing10.09–.12	ORIGINATED MORTGAGE SERVICING
· OTS's authority transferred to1.55–.59	ASSETS
Retail sales contracts         8.34           Role         1.08	OTC. See over-the-counter (OTC) derivatives
Rules and regulations 1.107, 21.12	OTHER ASSETS, OTHER LIABILITIES, AND
· Savings associations audit rules1.94	OTHER INVESTMENTS 12.0191
· Stress tests	· Accounting standards12.16–.76
· Trust and asset management 21.06–.07,	· Asset retirement obligations12.53–.56
	Audit considerations
OFFICE OF THRIFT SUPERVISION (OTS)	Contributed assets
· Abolishment and integration	organizations
process	· Customers' liabilities on
Business combinations	acceptances12.09, 12.57, 12.89
Credit losses	Exit or disposal activities 12.02, 12.51–.52
· Enforcement actions against	<ul> <li>Federal Home Loan Banks or Federal Reserve Bank stock12.04, 12.27–.34, 12.84,</li> </ul>
auditors	
Real estate lending standards8.57–.67 Rules and regulations1.107	Financial statement presentation and
_	disclosure
<b>OFFSETTING</b> • Derivatives	· Impairment of long-lived assets12.45,
Fair value presentation and	
disclosure	Internal control over financial
· Federal funds and repurchase	reporting
agreements	· Mortgage servicing advances12.58
ON-US CHECKS	• NCUSIF deposits
125 LOANS 8.41	Other investments
OPEN ACCOUNTS13.09	
OPERATING ACTIVITIES, CASH FLOW	Regulatory matters
FROM 6.1518	• Substantive tests
OPERATING CASH FLOW	OTHER COMPREHENSIVE
OPERATING INCOME9.54	INCOME17.0809
OPERATING LEASES8.45, 12.25	OTHER-MATTER PARAGRAPHS23.14
OPERATING POLICIES5.35	OTHER REAL ESTATE OWNED
OPERATING STATISTICS	(OREO) 11.01, 11.10–.13, 11.14–.16

C	THER-THAN-TEMPORARY IMPAIRMENT	PCA. See prompt corrective action
	(OTTI)	PCC. See perpetual contributed capital
	<ul> <li>Authoritative guidance</li></ul>	PEER REVIEW1.98
	• Debt securities	PENSION TRUSTS21.11
	· Decline in value of security7.103	PENSIONS. See also related-party
	Equity securities	transactions5.36, 12.01, 20.62
	Fair value measurement and disclosure	PERFORMANCE MATERIALITY5.1920
	Financial statement presentation and	PERMITTED INVESTMENTS15.30
	disclosure	PERPETUAL CONTRIBUTED CAPITAL
	<ul> <li>Held-to-maturity securities</li></ul>	(PCC) 2.28, 2.45–.46, 17.66–.67
	Risk assessment procedures7.144	23.1213, 23.33
	OTS. See Office of Thrift Supervision	PERPETUAL PREFERRED STOCK ISSUED TO U.S. TREASURY17.25
	OTTI. See other-than-temporary impairment	PERSONAL BIAS, CREDIT LOSS
	OUTSIDE PROFESSIONALS. See specialists	ESTIMATION9.15
c	OVER-THE-COUNTER (OTC)	PERSONAL GUARANTEES FOR
	DERIVATIVES1.39, 16.17, 18.17	LOANS9.104105
C	OVERADVANCES, IN FACTORING	PERSONAL TRUSTS21.0809
	ARRANGEMENTS8.25	PERVASIVE, DEFINED23.19
C	OVERALL RESPONSES TO ASSESSED RISK OF MATERIAL	PIC. See paid-in capital
	MISSTATEMENTS5.8384, 5.140	PLAIN VANILLA MORTGAGE-BACKED
C	OVERDRAFT PROTECTION 8.39, 13.67	SECURITIES (MBS)
c	OVERDRAWN DEPOSIT ACCOUNTS13.42	PLANNED AMORTIZATION CLASS7.29
c	VERHEAD TO TOTAL REVENUE	PLANNING ACTIVITIES. See also audit planning
	RATIO5.30	PLEDGES OF COLLATERAL 10.70, 10.88
C	OVERNIGHT INDEX SWAPS18.39	PO SECURITIES. See principal-only (PO)
C	OVERRIDE OF CONTROLS, BY	securities
_	MANAGEMENT 5.142143	POINT ESTIMATE, MANAGEMENT 7.159
C	OWNERS, DEFINED AS RELATED PARTY. See also related-party transactions 5.36	POINTS7.26
	also related party transactions	POLICY WRITING22.0809
	Р	POOLED TRUST PREFERRED
P	%A. See purchase and assumption (P&A)	SECURITIES
•	transactions	PORTFOLIO MEASUREMENT
P	PAID-IN CAPITAL (PIC)17.06	EXCEPTION 20.21–.25, 20.27
	AR CAP PROVISIONS14.11	PORTFOLIO PURCHASE AGREEMENTS8.2829
P	PARTICIPATING INTEREST 10.43, 10.71	PORTFOLIO SEGMENTATION. ESTIMATED
P	PARTICIPATING RIGHTS10.102	ALLOWANCE FOR LOSSES 9.07, 9.16
P	ARTICIPATION LENDING	POSTRETIREMENT BENEFITS20.62
	ARRANGEMENTS	PPE. See property, plant, and equipment
_	9.106–.107	PREFERRED STOCK
	PARTICIPATIONS PURCHASED 8.218	17.04–.05, 17.22–.24, 17.25
	PASS RATINGS9.11	PREMISES AND EQUIPMENT
P	ASS THROUGH SECURITIES. See	<ul> <li>Accounting standards</li></ul>
	mortgage-backed securities PATRIOT ACT. See USA-Patriot Act	investments
		· Examples12.03
	PAYING AGENTS	Financial statement presentation 12.17
۲	PAYMENTS FUNCTION, DEPOSITS13.1732	Substantive procedures to test

PREMIUM INCOME22.1317	PROOF DEPARTMENT, PHYSICAL CHECK
PREMIUMS 7.107113, 8.224, 18.41	PROCESSING13.20
PREPAID PENSION ASSETS. See also other assets, other liabilities, and other	PROPERTY, PLANT, AND EQUIPMENT (PPE).  See also premises and
investments 12.01	equipment16.25
PREPAYMENT AND PREPAYMENT RISK 4.17, 5.49, 7.23–.28, 8.98–.99,	PROPERTY AND LIABILITY INSURANCE22.02, 22.07
8.100, 10.48	PROPERTY MANAGEMENT AGREEMENTS AS
PREPAYMENT PENALTIES8.9899	PERSONAL TRUSTS21.08
PRESENT VALUE TECHNIQUES 20.3133	PROPERTY TAXES11.48
PRICE 20.0407, 20.09, 20.4647	PROPRIETARY TRADING1.41, 1.54,
PRICE DISCOVERY, SERVICING	
RIGHTS 4.15	PUBLIC COMPANY ACCOUNTING OVERSIGHT
PRICE RISK18.11	BOARD (PCAOB)
PRICING SERVICE,	<ul><li> Establishment</li></ul>
SECURITIES 7.153154	· Independent public accountant
PRIMARY BENEFICIARIES OF VARIABLE	standards 1.98
INTEREST ENTITIES 10.104–.106,	· Inspections
10.109, 10.111, 10.112–.113	Security investments
PRIMARY MORTGAGE MARKET4.01	<ul> <li>Unqualified opinions in PCAOB reports</li></ul>
PRINCIPAL MARKET18.80, 20.04–.07	Unqualified reports Appendix A at 23.38
PRINCIPAL-ONLY (PO) SECURITIES 7.29,	PUBLIC ENTITY5.211, 7.11
7.3841	PUBLICLY HELD COMPANIES, SECURITIES
PRIVATE COMPANIES  Derivative considerations	LAW 1.13, 3.12
· Electing to use standards as issued by the	PURCHASE AND ASSUMPTION (P&A)
private company alternative 10.108,	TRANSACTIONS1.118, 23.3536
	PURCHASED LOANS8.218, 19.1820
Electing to use standards as issued by the private company council 10.101, 12.35	PUSH DOWN ACCOUNTING19.04
Variable interest entity lessee and	PUTS
lessor22.61	
PRIVATE MORTGAGE	Q
AGENCIES	_
PROCESSING RISK 5.56–.57	QUALIFIED FINANCIAL CONTRACTS (QFC).  See also repurchase agreements
PROFESSIONAL ETHICS EXECUTIVE	(repos)14.26
COMMITTEE5.12	QUALIFIED MORTGAGES15.30
PROFESSIONAL JUDGMENT5.17–.18,	QUALIFIED OPINIONS23.1819
	QUALITATIVE/QUANTITATIVE LOAN
PROFESSIONAL SKEPTICISM 5.126128	IMPAIRMENT FACTORS 9.17–.18
PROFIT  • Estimated, on insurance	QUALITY CONTROL STANDARDS,
activities22.44–.50	GENERALLY OF
· Residual, from real estate	STATEMENTSAppendix D
investments8.140	QUOTED PRICES, PROVIDED BY THIRD
PROFIT SHARING TRUSTS21.11	PARTIES, FAIR VALUE20.48
PROMPT CORRECTIVE ACTION (PCA)	D
Banks and savings institutions 1.81–.91	R
<ul> <li>Complex credit unions 2.32–.33, 2.37</li> <li>Corporate credit unions 2.49–.54, 17.76</li> </ul>	RATINGS
Credit union net worth standard 2.44	$\cdot$ Creditworthiness of institution's debt $\ldots15.05$
· Credit unions 2.06, 2.24–.44, 17.60	· Loans 9.08–.12, 9.70
· Levels of1.86	· Residuals 7.44
· Regulation enforcement 5 231	· Systems

RATIOS	REAL ESTATE INVESTMENTS, REAL ESTATE
<ul> <li>Allowance for loan losses to classified loans</li> </ul>	OWNED, AND OTHER FORECLOSED
ratio5.30	ASSETS—continued
· Allowance for loan losses to total loans ratio5.30	Long-lived assets to be disposed by sale
· For analytical procedures5.29–.30	· Long-lived assets to be held and
· Capital ratios	used11.34–.40
· Classified loans to total loans ratio5.30	· Overview
Debt service coverage ratio8.14	· Property taxes and insurance costs 11.48
Debt-to-equity ratio	· Real estate owned
· Investment income to average total securities	Regulatory matters 11.05–.16
ratio	· Sales transactions
Investments by type divided by total investments ratio	• Substantive tests
Investments to total assets ratio5.30	· Temporary differences
Key ratios for financial statement	• Types
audits	
· Liabilities to shareholders' equity ratio 5.30	REAL ESTATE LENDING
<ul> <li>Loan income to average net loans ratio5.30</li> </ul>	STANDARDS 8.57–.67
· Loan loss recoveries to prior-year write-offs	REAL ESTATE MORTGAGE INVESTMENT
ratio	CONDUITS (REMIC)7.31, 15.06,
Loans by type to total loans ratio	15.29–.31, 15.54
Loans by type to total loans ratio5.30     Loans to deposits ratio5.30	REAL ESTATE OWNED (REO)11.01, 11.06
· Loans to total assets ratio 5.30	REAL ESTATE SETTLEMENT PROCEDURES
Minimum ratio of capital	ACT OF 1974 (RESPA)8.84
Net income to average capital ratio5.30	REALIZED AND UNREALIZED GAINS AND
Net income to average total assets	LOSSES7.104106, 7.132134,
ratio5.30	7.138
Net worth ratio	REASONABLE ASSURANCE 5.02, 5.05,
	5.155
Noninterest income to total revenue ratio5.30	REASONABLE PERIOD OF
Overhead to total revenue ratio5.30	TIME5.165166
• Reserve ratio	REBATES8.100
· Tangible capital ratio	REBORROWING, DEMAND LOANS8.106
· Tier 1 risk-based capital ratio 1.54, 1.71,	·
2.47, 17.41	RECEIPTS OF ASSETS, TROUBLED DEBT RESTRUCTURINGS
· Total deposit interest expense to average total	
deposits ratio5.30	RECEIVABLES
RBC. See risk based capital	· Charging off uncollectable
RBNWR. See risk-based net worth	• Foreclosed assets
requirement	Nonaccrual and past due financing
REAL ESTATE INVESTMENTS, REAL ESTATE	receivables8.157–.160
OWNED, AND OTHER FORECLOSED	Partial satisfaction of8.137
ASSETS11.0173	RECEIVABLES PORTFOLIO PURCHASE
• Accounting standards8.139, 11.17–.29	AGREEMENTS8.2829
· Audit considerations	RECIPROCAL ACCOUNT BALANCES 6.31
Development costs	
· Financial statement presentation and disclosure 11.17, 11.23, 11.30–.33,	RECONCILIATION REVIEW, AS INTERNAL
	CONTROL 15.62
· Foreclosed assets. See foreclosed assets	RECORD KEEPING. See documentation and
· Inherent risk	record keeping
· Internal control over financial	RECOURSE LIABILITIES12.02
reporting	RECOURSE LOANS10.4950
· Investments in real estate11.03–.04,	RECOURSE RISK
	RECOVERABILITY TESTING11.35–.37,
- Lending standards	
5	· · · · · · · · · · · · · · · · · · ·

RECOVERIES AND CHARGE-OFFS,	REGULATION AND OVERSIGHT—continued
LOANS9.108	<ul> <li>Other assets, other liabilities, and other</li> </ul>
REDEEMABLE PREFERRED	investments
STOCK	Overview
REFINANCING8.107	Real estate investments, real estate owned,
	and other foreclosed assets 11.05–.16
REGISTRARS21.10	Real estate lending standards8.57–.67
REGULAR RESERVE (STATUTORY	Regulatory authority
RESERVE)	Repurchase agreements
•	Retail credit loans
REGULATION AND OVERSIGHT. See also	• Risks resulting from
specific legislative acts and regulatory	Securities
institutions1.08–.91, 5.212–.257  · Anti-money laundering	• Troubled debt restructurings 8.71–.75,
,	
Audit considerations	Trust and asset management
, , , , , , , , , , , , , , , , , , , ,	activities
(OCC)	Trust operation capital
Bank affiliates and holding companies1.19	disclosures
• Brokered deposits5.242, 13.33–.38	· Uniform Commercial Code (UCC)8.8587
Business combinations	REGULATORY CAPITAL. See also capital
Capital requirement. See capital requirement	requirement
· Communication about possible fraud 5.150	
• Compliance with5.152–.162, 5.238	REGULATORY RISK8.07
· Correspondence with regulators 5.35	REINSURANCE COMPANIES22.08
· Correspondent concentration risks 8.79	RELATED-PARTY TRANSACTIONS
· Credit cards	• Defined
· Credit losses	· Financial statement disclosure 8.177
· Credit unions 2.02, 2.18–.23, 2.51,	Review, federal funds and repurchase
5.234–.236, 8.83, 13.38–.40	agreements
Debt	Risk assessment procedures 5.36–.39
Deposit insurance fund	
Derivative instruments18.71–.78	REMIC. See real estate mortgage investment
• Economic issues	conduits
· Enforcement 5.229–.236	RENEWALS, EXCESSIVE, CREDIT LOSS
· Engagement impact5.214	ESTIMATION
<ul> <li>Equity and disclosure regarding capital</li> </ul>	RENTAL OF RESIDENTIAL OREO
matters17.94	PROPERTIES11.1013
· Error detection 5.238–.239	
• Examinations 5.35, 5.223–.228,	REO. See real estate owned
5.248–.257	REPORTS AND REPORTING. See also
· Exclusion of deferred tax assets 16.15–.16	disclosures; financial statements;
· Finance companies	specific topics23.0138
Foreign institution branch capital	<ul> <li>Annual independent reporting1.92–.117</li> </ul>
disclosures	<ul> <li>Auditor's reports. See auditor's reports</li> </ul>
· Fraud detection 5.238–.239	Under FDI Act Appendix A
· Functional regulation 1.18–.19	Financial statements. See financial statements
· Going concerns 5.241–.242	· Internal control over financial
· Gramm-Leach-Bliley Act1.18	reporting
· History and background 1.15–.23	Loss sharing transactions 23.35–.36
· Income recognition on problem	Members' shares reported as equity23.24
loans 8.81–.82	• Opinions
· Income taxes	Small Business Lending Fund
· Insurance activities22.11	reporting
· Lending statutes	Special purpose framework, financial statements proposed in accordance.
· Leveraged lending8.80	statements prepared in accordance
· Loan transfers and servicing10.09–.19	of
· Loans	· Supervisory committee audits 23.28–.34
· Mortgage companies4.21–.32	Trust and asset management
Nontraditional products	activities
· Objectives	reportsAppendix A at 23.38
,	reportsAppendix A at 23.30

REPURCHASE AGREEMENTS (REPOS). See	RESIDENTIAL MORTGAGE LOANS—continued
also cash and cash equivalents	· Loan files
· Accounting standards14.27–.39	
• Accounting standards	Loans in process
	· Modification program 4.26, 8.71
Breakage	Mortgage banking. See mortgage companies
	and banking activities
• Characteristics	Nontraditional products 4.24, 8.78
· Collateral risk 14.23–.25, 14.44, 14.66	· Origination for sale
· Credit risk	Points
Debt, reclassified pledged assets 15.35	Primary market4.01
Definitions	Regulation
Disclosures	Reporting requirements
	· Secondary market
agreements	4.11–.12, 10.02–.03
· Internal control over financial	Securitization
reporting	· Servicing. See loan transfers and servicing
• Market risk	· Sound business practices
· Offsetting	• Structures 8.40
Provide background on securities lending	· Transfers. See loan transfers and servicing
transactions10.38	· Types8.41
· Purpose	· Veteran's Administration8.42, 10.16–.17
Regulatory matters14.26	RESIDENTIAL MORTGAGE
· Reverse repurchase agreements,	SERVICING11.08
securities 7.65, 14.05, 14.25, 14.39	RESIDUAL PROFIT, FROM REAL ESTATE
Rollovers and extensions	INVESTMENTS8.140
· Securities	
· Short-term collateralized financing10.88	RESIDUALS7.29, 7.42–.47
Substantive tests	RESOLUTION PLANS, DODD-FRANK
REPURCHASE FINANCING 14.3738	ACT1.6065
REPURCHASE RISK4.19	RESOLUTION TRUST CORPORATION
	(RTC)1.15
REPURCHASE-TO-MATURITY	RESPA. See Real Estate Settlement
TRANSACTIONS14.06	Procedures Act
REPUTATION RISK	RESTORATION PLAN 1.2427
RESERVES	
· Corporate credit unions 17.68–.69	RESTRUCTURING. See also troubled debt
· Credit unions 9.35–.36, 17.48–.49	restructurings (TDR)7.117, 8.107
· Defined as cash and cash equivalents6.06	RETAIL CREDIT LOANS8.6870
· Financial statement disclosure6.30	RETAIL SALES CONTRACTS 8.3335,
· Ratio 1.17, 1.24–.25	8.84–.87
RESIDENTIAL-MORTGAGE BACKED	RETAINED EARNINGS2.45, 17.07,
SECURITIES (RMBS)1.118, 4.07	17.102
RESIDENTIAL MORTGAGE	
LOANS 8.4043	RETIREMENT ACCOUNTS13.06, 13.08
· Accounting standards 8.90, 8.92,	RETIREMENT LIABILITIES16.25
	RETURN, COMPARED TO RISK7.02
· Adjustable-rate mortgages 7.26–.28,	REVENUE, TRUST DEPARTMENT 21.23
8.40–.41	
· Allowance for loan losses 8.78	REVENUE RECOGNITION STANDARD, NEW
· Audit considerations 8.225–.226	(FASB ASU NO. 2014-09)Appendix E
· Commitments8.147–.148	REVERSE MORTGAGES8.41
· Defined 8.40	REVERSE REPURCHASE AGREEMENTS
· FHA loans8.42	(REVERSE REPOS)7.65, 14.05,
<ul> <li>Foreclosed guaranteed mortgage</li> </ul>	
loans 11.18, 11.21	•
· Foreclosures10.19	REVOLVING CREDIT
· HFS. See held for sale	ARRANGEMENTS8.17, 8.3639, 8.106
· Junior liens9.27	
· Laws governing 2.01, 4.22–.23, 8.84	REVOLVING LOANS8.2627

RIGHTS  • To exchange, transferred financial	RISK ASSESSMENT—continued  Design of further audit
assets	procedures5.76–.98
· Kick-out rights	· Examination focus5.230
Participating rights	· Fraud consideration in financial statement
Right of setoff 14.39, 14.41, 18.87	audit 5.133–.136, Exhibit 5-1 at 5.25
· To servicing. See servicing rights	<ul> <li>Identifying and assessing risk of material</li> </ul>
RISK	misstatement5.77–.8
· Assessment of. See risk assessment	· Industry risk factors5.40–.5
Asset-quality risk	<ul> <li>Internal control. See internal control over</li> </ul>
· Basis risk18.11	financial reporting
· Business risk	· Loan internal controls8.219–.22
· Collateral risk 8.07, 14.23–.25, 14.44,	<ul> <li>Material misstatements. See material</li> </ul>
14.66	misstatements, risk of
· Compared to return7.02	Overview 5.23–.2
· Concentration risk 7.136, 8.07, 8.79,	Portfolio segmentation estimated allowance fo
8.172–.174, 18.10	losses9.07, 9.1
• Control risk	• Procedures
• Counterparty risk 7.22, 20.21–.27	Repurchase agreements14.44, 14.6
· Credit risk. See credit risk	Security investments
Derivatives	
	Understanding the entity and its environment.  See understanding the entity and its
Fiduciary risk	See understanding the entity and its environment
· Insider risk8.07	
Interest rate. See interest rate risk (IRR)	· Use of assertions in
Interest rate. See interest rate risk (intr) Internal controls. See internal control over	RISK-BASED CAPITAL (RBC)1.3435
financial reporting	1.74, 1.107, 2.2
Legal risk	RISK-BASED NET WORTH REQUIREMENT
Liquidity risk 5.50–.51, 7.71, 7.140,	(RBNWR) 2.2738, 2.4344, 17.4
	RISK MANAGEMENT 18.555
Loan collection risk	RISK MANAGEMENT ASSOCIATION3.1
· Market risk. See market risk	RISK MITIGATION CREDIT2.343
Of material misstatements. See material misstatements, risk of	RISK RATINGS, CONTROLS OVER9.7
• Model risk	RMBS. See residential-mortgage backed
Nonperformance risk	securities
· Operational (transaction) risk 7.71, 8.07	ROLLFORWARD1.13
Prepayment risk 5.49, 7.23–.28, 8.98–.99	ROLLOVER8.17, 14.1
· Present value techniques 20.33	
Price risk	RTC. See Resolution Trust Corporation
Processing risk	RULE MAKING, OVERVIEW5.221–.22
• Recourse risk	•
	S
• Reputation risk	SA. See servicing assets
Settlement risk	SAFE HARBOR FOR FINANCIAL
· Significant risks 5.62, 5.193–.203	
· Sovereign country risk 8.07	ASSETS 1.53, 10.1
· Transfer risk	SAIF. See Savings Associations Insurance
· Valuation risk	Fund
RISK ANALYSIS, ASSET-BACKED	SALE
SECURITIES7.3334, 7.75	· Financial assets 10.43–.48, 10.92–.9
RISK ASSESSMENT5.2198	· Participating interest
· Additional guidance	Repurchase agreements as14.0
· Analytical procedures5.28–.30	Of securities. See sales or transfers of
· Audit engagement team discussion5.31	securities
	T   10 A
· Credit losses 9.64–.68	· True sale opinion10.4

SALES OR TRANSFERS OF SECURITIES	SECURITIES—continued
· Accounting and reporting 7.118–.125	· Borrowing and lending7.63–.64
<ul> <li>Acquired with deteriorated credit</li> </ul>	· Cash flows from purchases, sales, and
quality 7.125	maturities
· Audit considerations7.138	<ul> <li>Cash receipts and cash payments from</li> </ul>
· Authoritative guidance7.117	acquisition and sales of 6.23–.24
Borrowing and lending securities 7.63–.64	· Classification 7.83–.91, 7.143
Between categories 7.87	Consolidation
· Financial statement presentation and	<ul> <li>Corporate credit union shares and</li> </ul>
disclosure 7.130–.131, 7.134–.135	certificates
Held-to-maturity securities7.85, 7.135	<ul> <li>Cost basis of debt security received in</li> </ul>
Marketable securities with put     arrangements	restructuring7.117
Repurchase agreements	<ul> <li>Debt securities. See debt securities</li> </ul>
· Short sales	· Defined16.11
· Trade date accounting7.120	· Equity securities. See equity securities
· Troubled debt restructurings 7.122–.124,	Financial statement presentation and
8.138	disclosure
SAMPLING BY CREDIT UNIONS	Financing transactions
SUPERVISORY COMMITTEES13.64	Income tax treatment 7.89, 16.05, 16.11,
SARBANES-OXLEY ACT (2002)	16.28
Banking regulatory agency	· Internal control
regulations1.107	Laws governing
· Credit union compliance 2.61	<ul><li>Liquidity compared to yield</li></ul>
· Credit unions	Municipal obligations
· Finance companies	Mutual-investment funds
· Mortgage companies	Negotiable certificates of deposit 7.61
· Scope	Other-than-temporary
SEC regulations	impairment
Section 404: Assessment of internal control	· Premiums and discounts7.107–.113
	Public entity
SATISFACTORY BANKS AND SAVINGS	· Realized and unrealized gains and
INSTITUTIONS17.39	losses 7.89, 7.104106, 7.132134,
SAVINGS AND LOANS HOLDING COMPANIES	7.138, 16.25
(SLHC). See also holding	Regulatory matters7.66–.82
companies 1.54, 1.57, 1.94	Repurchase agreements. See repurchase
SAVINGS ASSOCIATIONS INSURANCE FUND	agreements (repos)
(SAIF)1.15, 1.17	Risk compared to return
SAVINGS DEPOSITS13.05	<ul> <li>Sales or transfers. See sales or transfers of securities</li> </ul>
SAVINGS INSTITUTIONS. See also banks and	Structured credit. See structured credit
savings institutions 1.15, 17.2837,	products
17.41–.46	Substantive tests
SBLF. See Small Business Lending Fund	Troubled debt restructurings 7.122–.124
SECOND-LIEN FIXED TERM MORTGAGE	SECURITIES ACT (1933 ACT) 1.13, 3.12,
LOANS	4.21
SECONDARY MARKET4.0106, 4.08,	SECURITIES AND EXCHANGE COMMISSION
4.1112, 10.0203	(SEC)
SECURED BORROWINGS 10.7073,	Derivatives trading oversight
10.87	Federal Home Loan Bank registration12.27
SECURED PARTIES8.85	· Independent public accountant
	standards
SECURITIES	Market Risk Disclosure Rules18.90
· Accounting standards	· Reporting requirements 1.13, 1.108–.117
Asset-backed. See asset-backed securities     Asset-backed securities 7.12–.49	· Sarbanes-Oxley implementation1.105,
· Asset-backed securities	
Banks and savings institutions 1.13, 1.19,	SECURITIES EXCHANGE ACT (1934
	ACT)1.13, 1.106–.107, 3.12, 4.21

SECURITIES INDUSTRY AND FINANCIAL MARKETS ASSOCIATION (SIFMA)14.36	SETTLEMENT LOSS ALLOWANCE. See also other assets, other liabilities, and other investments
SECURITIZATION	SETTLEMENT RISK
· Asset-backed. See asset-backed securities	SHARE ACCOUNT DEPOSITS 13.01
Authoritative guidelines	SHAREHOLDERS' EQUITY 20.1420,
Finance companies	
· Mortgage loans4.08, 7.88	SHORT-DURATION
· Purpose10.06	CONTRACTS22.1417
Safe harbor protection	SHORT SALES
SEGMENT REPORTING5.210	SHORT-TERM DEBT15.1031
SEGREGATED ASSETS15.26	· Bankers' acceptances15.19–.20
SELF-DEALING, CREDIT LOSS	· Collateralized financing
ESTIMATION	· Collateralized mortgage obligations15.26–.28
SELF-EMPLOYED PERSON (SEP) ACCOUNTS13.08	Commercial paper 3.09, 7.61, 15.12–.13
	· Federal funds. See federal funds
SENIOR DEBT	· Federal Home Loan Banks advances 15.17
SENIOR/SUB SECURITIES7.48–.49	· Federal Reserve discount windows 15.17
SENIOR SUBORDINATED DEBT3.09	Finance subsidiary securities 15.22–.25
SEP ACCOUNTS. See self-employed person	Financial statement presentation and disclosure
(SEP) accounts SERVICE CHARGE INCOME	Institution's borrowings15.10–.31
	· Lines of credit
SERVICE ORGANIZATIONS · Audit considerations 5.11, 5.120–.122	<ul> <li>Mortgage-backed bonds 15.06, 15.21,</li> </ul>
· Credit unions	
· Fair value measurement of investment	conduits
securities	· Repos. See repurchase agreements
Fiduciary responsibility	$\cdot$ Treasury tax and loan note accounts $\dots15.18$
Loan transfer. See loan transfers and servicing	SIFMA. See Securities Industry and Financial
· Net worth requirements 10.117, 17.83	Markets Association
· Reporting requirements1.118, 2.62	SIGNIFICANT RISKS5.62, 5.8081, 5.90,
SERVICING AGREEMENTS FOR	5.93, 5.193–.203
LOANS10.08	SIGNIFICANTLY UNDERCAPITALIZED BANKS
SERVICING ASSETS (SA) 10.5169,	AND SAVINGS INSTITUTIONS1.85–.86, 17.43
12.01, 19.27	
SERVICING LIABILITIES12.02	SIGNIFICANTLY UNDERCAPITALIZED CREDIT UNION CLASSIFICATION2.38, 2.50,
SERVICING MORTGAGES, INTANGIBLE BENEFITS4.13	2.53, 17.63, Exhibit 2-1 at 2.30
SERVICING OF LOANS. See loan transfers	SIMULATION MODELS5.46
and servicing	SINGLE COMMON BOND CREDIT
SERVICING RIGHTS	UNIONS2.03
· As an asset	SLOW-PAY GINNIE MAE
· Bulk transfers10.60	CERTIFICATES14.35
Delay in actual delivery of the loan portfolio to	SMALL BUSINESS LENDING FUND
be serviced	(SBLF)23.37
· Sale of	SOFTWARE DEVELOPED FOR INTERNAL
· Subservicing contracts10.65–.66	USE12.23
<ul> <li>Temporary subservicing contracts 10.65</li> </ul>	SOVEREIGN COUNTRY RISK8.07
• Transfer of	SPE. See special-purpose entities
SETTLEMENT DATES, REPURCHASE-TO-MATURITY	SPECIAL-MENTION LOANS9.0911
TRANSACTIONS14.06	SPECIAL-PURPOSE ENTITIES (SPE)7.31

SPECIFIC CHARGEOFF METHOD	SPECIAL PURPOSE FRAMEWORK, FINANCIAL STATEMENTS PREPARED IN ACCORDANCE OF 23.2223	STRUCTURED FINANCING
STABILIZATION FUND  12.66 STANDBY COMMITMENTS  8.149–152 STANDBY LETTERS OF CREDIT  8.50  8.178 STATE AND FEDERAL SAVINGS ASSOCIATIONS  7.76–78 STATE-CHARTERED CREDIT UNIONS  Audit requirements  2.59 Capital adequacy  2.27 Net worth classification  2.45 Regulation and oversight  2.02, 2.19, 2.21, 2.21 Insurance activities  1.19, 22.55 Loans  8.84–87 Mortgage loans  4.22–23 Trusts  STATEMENT OF FINANCIAL POSITION. See balance sheets  STATEMENT OF SINANCIAL POSITION. See balance sheets  STATEMENT OF FINANCIAL POSITION. See balance sheets  STATEMENT OF FINANCIAL POSITION. See balance sheets  STATEMENT OF ORDIT STANDARDS, OVERVIEW OF  OF  STATUTORY RESERVE. See regular reserve  STOCK. See equity securities  STOCK OPTIONS  2.062 STRESS TESTING  1.66–69, 8.78 STRUCTURED CREDIT  PRODUCTS  7.50–59 Authoritative guidance  7.751–59 Collateral CREDIT STANDARDS. See mortgage-backed securities  Pooled trust preferred securities  Pooled trust preferred securities  Pooled trust preferred securities  Pooled trust preferred securities  7.55–59 Structured notes  7.55  5UBSTANTIAL DOUBT, ABOUT ENTITY'S ABILITY TO CONTINUE AS A GOING CONCERN  5.165–174, 5.176  CONCERN  5.165–174, 5.176  CORCERN  5.165–174, 5.176  CORCERN  5.165–174, 5.176  Allowance for loan and lease losses  9.79–82  All		CLASSIFICATION8.228, 9.08, 9.11
STANDBY COMMITMENTS	SPECULATION 18.54	SUBSTANDARD LOAN RENEWAL8.73
STATE AND FEDERAL SAVINGS ASSOCIATIONS	STANDBY COMMITMENTS8.149152 STANDBY LETTERS OF CREDIT8.50,	ABILITY TO CONTINUE AS A GOING CONCERN 5.165174, 5.176, 17.3435, 23.0809
ASSOCIATIONS 7.76–.78  STATE-CHARTERED CREDIT UNIONS Audit requirements 2.59 Capital adequacy 2.27 Net worth classification 2.45 Regulation and oversight 2.02, 2.19, 2.21, 7.80 STATE CREDIT UNIONS 12.26 STATE LAWS AND REGULATIONS Conference of State Bank Supervisors 1.09 Credit unions 2.18–20, 2.21 Insurance activities 1.19, 22.55 Loans 8.84–87 Mortgage loans 4.22–23 Trusts 21.06 STATEMENT OF CASH FLOWS. See cash flow statements STATEMENT OF FINANCIAL POSITION. See balance sheets STATEMENT OF FINANCIAL POSITION. See balance sheets STATEMENTS ON QUALITY CONTROL STANDARDS, OVERVIEW OF Appendix D STATUTORY RESERVE. See regular reserve STOCK. See equity securities STOCK OPTIONS 20.62 STRESS TESTING 1.66–.69, 8.78 STRUCTURED CREDIT PRODUCTS 7.50–.59 Collateralized mortgage obligations 7.31, 0cllateral 7.59 Collateralized mortgage obligations 7.31, MBS. See mortgage-backed securities 7-00et drust preferred securities 7-055 Pooled trust preferred securities 7-055 Structured notes 7.55–59 STRUCTURED CREDIT Proportion 7.55–759 Structured notes 7.55–59 STATE ARRESTED CREDIT Procedures See analytical procedures Sea analytical procedures Sea analytical procedures 5.247 Standatequacy 2.25 Business combinations 1.9.3 Scash and cash equivalents .6.41–42 Credit losses .9.76–112 Business combinations .19.33 Cash and		
STATE CREDIT UNIONS         12.26           STATE LAWS AND REGULATIONS	ASSOCIATIONS         7.76–.78           STATE-CHARTERED CREDIT UNIONS         . Audit requirements         2.59           . Capital adequacy         2.27           . Net worth classification         2.45           . Regulation and oversight         . 2.02, 2.19, 2.21,	Allowance for loan and lease     losses
STATE LAWS AND REGULATIONS Conference of State Bank Supervisors 1.09 Credit unions 2.18–20, 2.21 Insurance activities 1.19, 22.55 Loans 8.84–87 Mortgage loans 4.22–23 Trusts 21.06 STATEMENT OF CASH FLOWS. See cash flow statements STATEMENT OF FINANCIAL POSITION. See balance sheets STATEMENTS ON QUALITY CONTROL STANDARDS, OVERVIEW OF Appendix D STATUTORY RESERVE. See regular reserve STOCK. See equity securities STOCK OPTIONS 20.62 STRESS TESTING 1.66–.69, 8.78 STRUCTURED CREDIT PRODUCTS 7.50–.59 Authoritative guidance 7.72–.73 Collateral 7.59 Collateral 7.51, 15.06, 15.26–.28 Defined 7.51 MBS. See mortgage-backed securities 7.09 Structured notes 7.55–.59		· Credit losses 9.76–.112
Conference of State Bank Supervisors 1.09 Credit unions 2.18–20, 2.21 Insurance activities 1.19, 22.55 Loans 8.84–87 Mortgage loans 4.22–23 Trusts 21.06  STATEMENT OF CASH FLOWS. See cash flow statements  STATEMENT OF FINANCIAL POSITION. See balance sheets  STATEMENTS ON QUALITY CONTROL STANDARDS, OVERVIEW OF Appendix D  STATUTORY RESERVE. See regular reserve STOCK. See equity securities  STOCK OPTIONS 20.62  STRUCTURED CREDIT PRODUCTS 7.50–59 Authoritative guidance 7.72–73 Collateral 7.51, 15.06, 15.26–28 Defined 7.51, 15.06, 15.26–28 Defined 7.55–59 Structured notes 7.50–59 Structured notes 7.55–59		Debt
STATEMENT OF CASH FLOWS. See cash flow statements         to	Conference of State Bank Supervisors       1.09         Credit unions       2.18–20, 2.21         Insurance activities       1.19, 22.55         Loans       8.84–87         Mortgage loans       4.22–23         Trusts       21.06	Design and performance
STATEMENT OF FINANCIAL POSITION. See balance sheets  STATEMENTS ON QUALITY CONTROL STANDARDS, OVERVIEW OF Appendix D  STATUTORY RESERVE. See regular reserve  STOCK. See equity securities  STOCK OPTIONS 20.62  STRESS TESTING 1.66–.69, 8.78  STRUCTURED CREDIT PRODUCTS 7.50–.59  Authoritative guidance 7.72–.73  Collateral 7.59  Collateral 7.51, 15.06, 15.26–.28  Defined 7.51, 15.06, 15.26–.28  Defined 7.51  MBS. See mortgage-backed securities Pooled trust preferred securities 7.55–.59  Structured notes 7.52  Structured notes 7.52  Loan transfers and servicing 10.126–.127  Loans 8.190, 8.199–.230  Mortgage companies and banking activities 17.112–.115  Other assets, other liabilities, and other investments 12.82–.91  Overview 5.85–.86  Real estate investments, real estate owned, and other foreclosed assets 11.62–.73  Securities 7.161–.162, 21.21–.30  SUBSTITUTE CHECKS 13.30–.32  SUPERVISION 5.237, 8.11, 11.13  SUPERVISORY COMMITTEES  Audits and reporting 23.28–.34  Credit unions 2.08–.09, 2.26–.27  STRUCTURED CREDIT Trust and asset management activities 7.161–.162, 21.21–.30  SUBSTITUTE CHECKS 13.30–.32  SUPERVISION 5.237, 8.11, 11.13  SUPERVISORY COMMITTEES  Audits and reporting 23.28–.34  Credit unions 2.08–.09, 2.26–.27  STRUCTURED CREDIT Trust and asset management activities 7.161–.162, 21.21–.30  SUBSTITUTE CHECKS 13.30–.32  SUPERVISORY COMMITTEES  Audits and reporting 23.28–.34  Credit unions 2.08–.09, 2.26–.27  STRUCTURED CREDIT Trust and asset management activities 7.161–.162, 21.21–30  SUPERVISORY COMMITTEES  Audits and reporting 23.28–.34  Credit unions 2.08–.09, 2.26–.27		to
STATEMENTS ON QUALITY CONTROL STANDARDS, OVERVIEW OF Appendix D  STATUTORY RESERVE. See regular reserve  STOCK. See equity securities  STOCK OPTIONS 20.62  STRESS TESTING 1.66–.69, 8.78  STRUCTURED CREDIT PRODUCTS 7.50–.59 Collateral 7.59 Collateral 7.51 MBS. See mortgage obligations 7.31, MBS. See mortgage-backed securities 7.55–.59 Structured notes 7.55 Structured notes 7.55 STATEMENTS ON QUALITY CONTROL Mortgage companies and banking activities 17.112–.115 Other assets, other liabilities, and other investments 12.82–.91 Overview 5.85–.86 Real estate investments, real estate owned, and other foreclosed assets 11.62–.73 Securities 7.155–.163 Trust and asset management activities 7.161–.162, 21.21–.30 SUBSTITUTE CHECKS 13.30–.32 SUPERVISION 5.237, 8.11, 11.13 SUPERVISORY COMMITTEES Audits and reporting 23.28–.34 Credit unions 2.08–.09, 2.26–.27 Structured notes 7.55 Structured notes 7.55		· Loan transfers and servicing $\dots10.126$ –.127
STOCK. See equity securities           STOCK OPTIONS         20.62           STRESS TESTING         1.66–.69, 8.78           STRUCTURED CREDIT         Securities           PRODUCTS         7.50–.59           Authoritative guidance         7.72–.73           Collateral         7.59           Collateralized mortgage obligations         7.31,           MBS. See mortgage-backed securities         7.51, 15.06, 15.26–.28           Pooled trust preferred securities         7.55–.59           Structured notes         7.55	STANDARDS, OVERVIEW	activities17.112–.115
Real estate investments, real estate owned, and other foreclosed assets   11.6273	STATUTORY RESERVE. See regular reserve	investments
STOCK OPTIONS         20.62           STRESS TESTING         1.66–.69, 8.78           STRUCTURED CREDIT PRODUCTS         5.50–.59           Authoritative guidance         7.72–.73           Collateral         7.59           Collateralized mortgage obligations         7.31,           5.237, 8.11, 11.13           SUPERVISION           SUPERVISION           5.237, 8.11, 11.13           5.237, 8.11, 11.13           5.237, 8.11, 11.13           5.237, 8.11, 11.13           5.237, 8.11, 11.13           5.237, 8.11, 11.13           6.23           6.24           7.55           8.25           8.26           8.27           8.27           8.29           8.20           8.20           8.20           8.20           8.20           8.20           8.20           8.20		
STRESS TESTING         1.6669, 8.78         Securities         7.155163           STRUCTURED CREDIT PRODUCTS         7.5059           • Authoritative guidance         7.7273         Collateral         7.59           • Collateral collisteral c	STOCK OPTIONS20.62	
PRODUCTS	STRESS TESTING 1.6669, 8.78	· Securities
Collateral	PRODUCTS	activities7.161162, 21.2130
Collateralized mortgage obligations       7.31,		SUBSTITUTE CHECKS13.3032
Defined       .7.51         MBS. See mortgage-backed securities       .7.55–.59         Structured notes       .7.52    Audits and reporting Credit unions 2.08–.09, 2.26–.27 2.58–.59, 5.11, 13.64	· Collateralized mortgage obligations 7.31,	SUPERVISION5.237, 8.11, 11.13
	Defined	SUPERVISORY COMMITTEES         Audits and reporting       23.28–.34         Credit unions       2.08–.09, 2.26–.27,

SUPPLEMENTARY	TESTS OF CONTROLS—continued
INFORMATION5.190192,	· Income taxes
15.76–.77	Loans 8.196, 9.70
SUSPENSE ACCOUNTS12.91	Other assets, other liabilities, and other
SWAPATIONS 18.44, 18.50	investments
SWAPS	Possible tests of controls. See internal control
· Basis swaps	over financial reporting
· Cleared swaps	· Real estate investments, real estate owned,
· Clearinghouse for swap contracts18.25	and other foreclosed assets 11.62–.73
Commodity swaps	• Securities
· Credit default swaps	· Substantive. See substantive tests
Credit risk	· Transfers and servicing10.123–.124
Equity swaps	TESTS OF VALUATION15.68
· Fair value measurements 18.38, 18.81	THIRD PARTIES 20.19, 20.48, 20.6667
· Fixed-rate-currency swaps 18.32	THOSE CHARGED WITH GOVERNANCE. See
Foreign-currency swaps	also management
Hedge accounting approaches	Communication with
Index amortizing swaps	
• Mortgage swaps	Written representations from 5.175,
SYNDICATION LENDING	5.180–.189
ARRANGEMENTS8.55–.56	THREE-PARTY PAPER
_	THRIFT INDUSTRY. See also banks and
T	savings institutions 1.15, 1.55–.59
T-BILLS. See Treasury bills	TIER 1 CAPITAL
T-BONDS. See Treasury bonds	· Basel III regulations
T-NOTES. See Treasury notes	· Capital requirement 1.71, 1.77, 1.80,
TANGIBLE CAPITAL RATIO1.83	· Corporate credit union
TANGIBLE EQUITY	· Deductions of certain assets from1.80
TARGETED AMORTIZATION CLASS7.29	· Minority interest in consolidated
TARP. See Trouble Asset Relief Program	subsidiaries
TAX BAD-DEBT DEDUCTIONS16.0608	Prompt corrective action 1.82–.83, 1.86 Trust preferred securities 17.17–.19
TAXATION. See income taxes; property taxes	TIER 1 RISK-BASED CAPITAL RATIO1.54,
TDR. See troubled debt restructurings	1.71, 2.47, 17.41
TEASER RATES7.27	TIER 2 CAPITAL
TEMPORARY CORPORATE CREDIT UNION	TIME DEPOSITS
STABILIZATION FUND (STABILIZATION	TIME LOANS 8.17
FUND)12.66	TIME VALUE
TEMPORARY DIFFERENCES, INCOME TAXES16.2426, 16.2930	TOLERABLE MISSTATEMENT 5.19
TERM INSURANCE	TOTAL ASSETS, CREDIT UNIONS 2.26
	TOTAL DEPOSIT INTEREST EXPENSE TO
TERM LOANS8.17 TERMS OF ENGAGEMENT5.1013	AVERAGE TOTAL DEPOSITS
TESTAMENTARY TRUSTS	RATIO5.30
	TOTAL RETURN SWAPS. See equity swaps
• Cash and cash equivalents6.39–.40	TOTAL RISK-BASED CAPITAL RATIO 2.47
Credit losses	TPS. See trust preferred securities
· Debt	TRADE DATE ACCOUNTING7.120
· Deposits	TRADE FINANCING 8.49–.50
Design and performance of 5.87–.90	TRADING SECURITIES, DEFINED7.83
Federal funds and repurchase	•
agreements14.57	TRANCHES

TRANSACTION ACCOUNTS 13.04	TRUST AND ASSET MANAGEMENT
TRANSACTION PRICE IN PRINCIPAL	ACTIVITIES—continued
MARKET20.04–.07	· Common or collective trust funds 21.12 · Corporate trusts
TRANSACTION TESTING. See also orderly	Derivatives
transactions5.230	· Employee benefit trusts
TRANSACTIONAL LEASE FINANCING 8.48	Federal funds and repurchase
TRANSFER AGENTS21.10	agreements
TRANSFER RISK, DERIVATIVES 18.05	· Financial reporting controls of the
•	trust
TRANSFERS OF FINANCIAL ASSETS8.168	Financial statement disclosure21.15–.16
TRANSFERS OF LOANS. See loan transfers	· Internal control over financial reporting
and servicing	· Overview
TREASURY BILLS (T-BILLS) 6.09, 7.0506	• Personal trusts
TREASURY BONDS (T-BONDS) 7.0506	· Regulatory matters 21.06, 21.13–.14
TREASURY DEPARTMENT. See U.S.	Reporting
Department of the Treasury	· Substantive tests 7.161–.162, 21.21–.30
TREASURY NOTES (T-NOTES)6.09,	· Trusts held by third-party trustee20.66–.67
7.05–.06	TRUST BANKS, REGULATORY CAPITAL
TREASURY RATE LOCKS18.20	DISCLOSURES17.87–.88
TREASURY TAX AND LOAN (TT&L)	TRUST DEPARTMENT, SUBSTANTIVE
ACCOUNTS	TESTS21.23
TROUBLE ASSET RELIEF PROGRAM	TRUST PREFERRED POOLS7.5559
(TARP)16.09	TRUST PREFERRED SECURITIES
TROUBLED DEBT RESTRUCTURINGS (TDR)	(TPS)
· Accounting standards 7.122–.124,	17.17–.21
8.71–.75, 8.123–.138, 11.18	TRUSTEES UNDER INDENTURE21.10
· Audit considerations 8.227–.228	TRUTH IN LENDING ACT8.84
Credit losses	TRUTH-IN-LENDING LAWS 2.20
Creditor's concession     determination	TRUTH-IN-SAVINGS REGULATIONS2.20
Debt security received in8.133–.134	
Debtor's financial difficulties	TT&L. See treasury tax and loan (TT&L) accounts
determination8.129	
· Defined	TYPE 1 REPORTS8.219
• Evaluating	TYPE 1 SECURITIES1.47
• Foreclosed assets in	TYPE 2 REPORTS8.219220
Loans	
• Modifications 8.71–.75	U
· Partial satisfaction of a receivable8.137	UCC. See Uniform Commercial Code
· Receipts of assets 8.135	UIT. See unit investment trusts
Regulatory issues8.71–.75	
Renewals, extensions or modifications8.71–.75	UITRS. See Uniform Interagency Trust Rating
Reporting requirements8.71	System
Restructuring characteristics 8.130–.131	UNCAPITALIZED CREDIT UNION CLASSIFICATION17.6364
· Sales of assets from 8.138	
Substantive tests	UNCERTAINTIES 5.193203, 9.61, 20.33
TRUE SALE OPINION10.40	UNCOMMITTED LOANS10.25
TRUST AND ASSET MANAGEMENT	UNDERCAPITALIZED BANKS AND SAVINGS
ACTIVITIES21.0130	INSTITUTIONS 1.8586, 1.90, 17.31,
· Accounting and financial	17.43
reporting	UNDERCAPITALIZED CREDIT UNION
· Accounting standards	CLASSIFICATION
· Audit considerations 5.11, 21.17–.30	· Corporate credit unions 2.50–.51, 2.53,
Classifications of trusts	17.79

Credit unions 2.27, 2.33, 2.36, 2.38, 17.63 Financial statement disclosure 17.64 Net worth ratio Exhibit 2.1 at 2.30 New credit unions 2.39, 2.41, 17.64 Prompt corrective action Exhibit 2.1 at 2.30 UNDERSTANDING CLIENT'S BUSINESS 5.34–39 UNDERSTANDING THE ENTITY AND ITS ENVIRONMENT 5.33–75 Asset-quality risk 5.52–54 Audit planning. See audit planning Banks, savings institutions, and credit unions 17.95 Cash and cash equivalents 6.36 Client's business 5.34–39 Credit losses 9.64, 9.69–70 Debt 15.59–60 Depository institution regulation and supervision 17.95 Federal funds and repurchase agreements 14.54 Industry risk factors 5.40–57 Interest rate risk. See interest rate risk (IRR) Internal control. See internal control over financial reporting 2.30 Loans 8.185, 8.188–189 Other assets, other liabilities, and other investments 12.79 Overview 5.33 Purpose of 5.24 Real estate investments, real estate owned, and other foreclosed assets 11.58 Requirements 5.33 Security investments 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNDISBURSED PORTION OF MORTGAGE LOANS 8.225–226 UNDIVIDED LOAN COMMITTMENT 8.17 UNFORM COMMERCIAL CODE (UCC) 2.20, 8.85–87 UNIFORM CONSUMER CODE 2.20 UNIFORM CONSUMER CODE 2.20 UNIFORM CONSUMER CODE 2.20 UNIFORM CONSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING	UNDERCAPITALIZED CREDIT UNION CLASSIFICATION—continued	UNIFORM INTERAGENCY TRUST RATING SYSTEM (UITRS)21.14
Financial statement disclosure 1.7.64 Net worth ratio Exhibit 2-1 at 2.30 New credit unions 2.39, 2.41, 17.64 Prompt corrective action Exhibit 2-1 at 2.30 UNDERSTANDING CLIENT'S BUSINESS 5.34-39 UNDERSTANDING THE ENTITY AND ITS ENVIRONMENT 5.33-75 Asset-quality risk 5.52-54 Audit planning. See audit planning Banks, savings institutions, and credit unions 1.7.95 Cash and cash equivalents 6.36 Client's business 5.34-39 Credit losses 9.64, 9.69-70 Debt 15.59-60 Depository institution regulation and supervision 15.59-70 Debt 15.59-60 Depository institution regulation and supervision 5.237 Deposits 1.10 Equity and disclosure regarding capital matters 17.95 Federal funds and repurchase agreements 1.7.95 Interest rate risk. See internal control over financial reporting Loans 8.185, 8.188-189 Other assets, other liabilities, and other investments 1.12.79 Overview 5.33 Security investments 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS REQUirements 5.33 Security investments 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS UNISUAL EVENTS 1.50 UNISUAL EVENTS 1.50 U.S. DEPARTMENT OF HOUSING AND URBAN DEVELOPMENT (HUD) Credit union reporting 2.662 Loans, reporting requirements 1.10 S. DEPARTMENT OF THE TREASURY. See also Office of the Comptroller of the Currency (OCC); Office of Thrift Supervision (OTS) Anti-money laundering program 1.20 Credit union regulation 2.20 UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.85-87 UNDEANNET OF THE TREASURY See also Office of The Comptroller of the Currency (OCC); Office of Thrift Supervision (OTS)  UNISUAL EVENTS 1.50 UN		• •
Net worth ratio		
New credit unions 2.39, 2.41, 17.64 Prompt corrective action . Exhibit 21 at 2.30 UNDERSTANDING CLIENT'S BUSINESS 5.3439 UNDERSTANDING THE ENTITY AND ITS ENVIRONMENT 5.3375 Asset-quality risk 5.3375 Asset-quality risk 5.3254 Audit planning. See audit planning Banks, savings institutions, and credit unions 17.95 Cash and cash equivalents 6.36 Client's business 5.3439 Credit losses 9.64, 9.6970 Debt 15.5960 Depository institution regulation and supervision 5.237 Deposits 13.50 Equity and disclosure regarding capital matters 17.95 Federal funds and repurchase agreements 14.54 Industry risk factors 5.4057 Interest rate risk. See internal control over financial reporting 1.0ans 8.185, 8.188189 Other assets, other liabilities, and other investments 12.79 Overview 5.33 Purpose of 4.29 Real estate investments, real estate owned, and other foreclosed assets 5.13.3 Purpose of 5.24 Real estate investments, real estate owned, and other foreclosed assets 5.13.3 Security investments 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNDISBURSED PORTION OF MORTGAGE LOANS 8.225-226 UNDIVIDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 17.5052, 17.6869 UNEARNED INTEREST 8.17 UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other seems of the company regulation 1.19 Home Affordable Mortgage Program (VALUATION) Los GOVERNMENT OF THE TREASURY. See also other assets, other liabilities, and other liabilit		
Prompt corrective action Exhibit 2-1 at 2.30 UNDERSTANDING CLIENT'S BUSINESS		
## SUSINESS		-
## SUNDERSTANDING THE ENTITY AND ITS ENVIRONMENT	UNDERSTANDING CLIENT'S	UNREALIZED GAINS AND LOSSES7.89.
ENVIRONMENT 5.3375 Asset-quality risk 5.52-54 Audit planning. See audit planning Banks, savings institutions, and credit unions 17.95 Cash and cash equivalents 6.36 Client's business 5.3439 Credit losses 9.64, 9.6970 Debt 15.5960 Depository institution regulation and supervision 5.237 Deposits 13.50 Equity and disclosure regarding capital matters 14.54 Industry risk factors 5.4057 Interest rate risk. See interest rate risk (IRR) Internal control. See internal control over financial reporting Loans 8.185, 8.188189 Other assets, other liabilities, and other investments 12.79 Overview 5.33 Security investments, real estate owned, and other foreclosed assets 11.58 Requirements 5.33 Security investments 7.139, 7.144 UNDEBRYRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNDISBURSED PORTION OF MORTGAGE LOANS 8.225226 UNDIVIDED EARNINGS 17.5052, 17.6869 UNEARNED INTEREST 8.17 UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02 UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.8587 UNIFORM COMSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING  PRACTICE 2.51, 5.232 UNISCURED FEDERAL FUND TRANSACTIONS 14.03 UNUSUAL EVENTS 15.50 UNSCURED FEDERAL FUND TRANSACTIONS 14.03 UNISUAL EVENTS 15.50 UNSCURED FEDERAL FUND TRANSACTIONS 14.03 UNISUAL EVENTS 15.50 U.S. DEPARTMENT OF HOUSING AND URBAN DEVELOPMENT (HUD) Credit union reporting 2.62 Loans, reporting requirements 1.103 Net worth requirements 1.118 U.S. DEPARTMENT OF HOUSING AND URBAN DEVELOPMENT (HUD) Credit union reporting 2.62 U.S. DEPARTMENT OF HOUSING AND URBAN DEVELOPMENT (HUD) Credit union reporting 2.62 U.S. DEPARTMENT OF HOUSING AND URBAN DEVELOPMENT (HUD) Credit union reporting 2.62 U.S. DEPARTMENT OF HOUSING AND URBAN DEVELOPMENT (HUD) Credit union reporting 2.62 U.S. DEPARTMENT OF THE TREASURY. See also Office of the Comptrolled of the Co	BUSINESS5.3439	7.104–.106, 7.132–.134, 7.138
Asset-quality risk 5.52–.54 Audit planning. See audit planning Banks, savings institutions, and credit unions 17.95 Cash and cash equivalents 6.36 Client's business 5.34–.39 Credit losses 9.64, 9.69–.70 Debt 15.59–.60 Depository institution regulation and supervision 5.237 Deposits 13.50 Equity and disclosure regarding capital matters 17.95 Federal funds and repurchase agreements 14.54 Industry risk factors 5.40–.57 Interest rate risk. See interenal control over financial reporting Loans 8.185, 8.188–189 Other assets, other liabilities, and other investments 12.79 Overview 5.33 Purpose of 5.24 Real estate investments, real estate owned, and other foreclosed assets 11.58 Requirements 5.33 Security investments 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNIDISBURSED PORTION OF MORTGAGE LOANS 8.225–.226 UNDIVIDED EARNINGS 17.50–.52, 17.68–69 UNEARNED INTEREST 8.17 UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02 UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.85–87 UNIFORM CONSUMER CODE 2.20 UNIFORM CONSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING  Audit planning. See aid and the red investments 12.02 UNIFORM FINANCIAL INSTITUTIONS RATING  UNSCURED FEDERAL FUND TRANSACTIONS 14.03 UNISUAL EVENTS 15.50 U.S. DEPARTMENT OF FUUSING AND URBAN DEVELOPMENT (HUD)  Loans reporting equirements 1.103 Net worth requirements 1.103 Net worth requirements 1.118, 4.33–.35 Reporting considerations 1.118, 4.33–.35 Reporting considerations 1.118, 4.33–.35 Reporting considerations 1.118, 4.33–.35 Reporting considerations 1.118, 4.33–.35 Reporting requirements 1.103 Net worth requirements 1.106 Loans reporting capital maters 1.17.80 Loans capital maters 1.17.80 Loans capital maters 1.18 Requirements 1.20 UNIFORM COMMERCIAL CODE (10 months of the Currency (OCC): Office of Thrift Supervision (TS)	UNDERSTANDING THE ENTITY AND ITS ENVIRONMENT5.3375	
Audit planning. See audit planning Banks, savings institutions, and credit unions 17.95 Cash and cash equivalents 6.36 Client's business 5.34–39 Credit losses 9.64, 9.69–70 Debt 15.59–60 Depository institution regulation and supervision 5.237 Deposits 13.50 Equity and disclosure regarding capital matters 17.95 Federal funds and repurchase agreements 14.54 Industry risk factors 5.40–57 Interest rate risk. See internal control over financial reporting Loans 8.185, 8.188–189 Other assets, other liabilities, and other investments 12.79 Overview 5.33 Purpose of 5.24 Real estate investments, real estate owned, and other foreclosed assets 11.58 Requirements 5.33 Security investments, real estate owned, and other foreclosed assets 11.58 Requirements 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNIDISBURSED PORTION OF MORTGAGE LOANS 8.225–226 UNDIVIDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02 UNIFORM CONSUMER CODE (UCC) 2.20, 8.85–87 UNIFORM CONSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING 17.95 UNIFORM FINANCIAL INSTITUTIONS RATING 17.95 Cash and cash equivalents 5.34 UNISUAL EVENTS 15.50 U.S. DEPARTMENT OF FOUSING AND URBAN DEVELOPMENT (HUD) Credit union reputing 9.2.62 Loans, reporting equivements 1.103 Net worth requirements 1.118, 4.33–35 Reporting considerations 1.118, 4.33–35 Unmodified opinions 2.3.06 U.S. DEPARTMENT OF THE TREASURY, See also Office of the Comptroller of the Currency (OCC); Office of Thrift Supervision (OTS) Anti-money laundering programs 1.20 Credit union reputing equivements 1.20 Credit union reputing equivements 1.103 Net worth requirements 1.118, 4.33–35 Reporting considerations 1.118 U.S. DEPARTMENT OF THE TREASURY, See also Office of the Courrency (OCC); Office of Thrift Super	· Asset-quality risk 5.52–.54	
Banks, savings institutions, and credit unions		
Cash and cash equivalents 6.36 Client's business 5.34–39 Credit losses 9.64, 9.69–70 Debt 15.59–60 Depository institution regulation and supervision 5.237 Deposits 13.50 Equity and disclosure regarding capital matters 17.95 Federal funds and repurchase agreements 14.54 Industry risk factors 5.40–57 Interest rate risk. See interest rate risk (IRR) Internal control. See internal control over financial reporting Loans 8.185, 8.188–189 Coher investments 12.79 Overview 5.33 Purpose of 5.24 Real estate investments, real estate owned, and other foreclosed assets 11.58 Requirements 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNDISBURSED PORTION OF MORTGAGE LOANS 8.225–226 UNDIVIDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02 UNIFORM CONSUMER CODE (UCC) 2.20, 8.85–87 UNIFORM CONSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING  US. DEPARTMENT OF EDUCATION 1.106, 1118 U.S. DEPARTMENT OF HOUSING AND URBAN DEVELOPMENT (HUD) Credit union reporting 2.62 Loans, reporting requirements 1.108 Net worth requirements 1.108 Net worth requirements 1.118 U.S. DEPARTMENT OF HOUSING AND URBAN DEVELOPMENT (HUD) Credit union reporting 2.62 Loans, reporting requirements 1.108 Net worth requirements 1.108 Net worth requirements 1.108 Net worth requirements 1.108 Net worth requirements 1.118 U.S. DEPARTMENT OF HOUSING AND URBAN DEVELOPMENT (HUD) Credit union reporting 2.62 Loans, reporting equirements 1.108 Net worth requirements 1.108 Net worth requirement	Banks, savings institutions, and credit	
Client's business 5.34–39 Credit losses 9.64, 9.69–70 Debt 15.59–60 Depository institution regulation and supervision 5.237 Deposits 1.350 Equity and disclosure regarding capital matters 17.95 Federal funds and repurchase agreements 14.54 Industry risk factors 5.40–57 Interest rate risk. See interest rate risk (IRR) Internal control. See internal control over financial reporting 8.185, 8.188–189 Other assets, other liabilities, and other investments 1.533 Security investments 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNDISBURSED PORTION OF MORTGAGE LOANS 8.225–226 UNDIVIDED LAAN COMMERCIAL CODE (UCC) 2.20, 8.85–87 UNIFORM CONSUMER CODE 2.20 UNIFORM CONSUMER CODE 2.20 UNIFORM CONSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING  Lequity and disclosure regarding capital matters 1.350 DEVELOPMENT (HUD) Credit union reporting 2.62 Loans, reporting requirements 1.780 Net worth requirem		
Credit losses 9.64, 9.69–70 Debt 15.59–60 Depository institution regulation and supervision 5.237 Deposits 13.50 Equity and disclosure regarding capital matters 17.95 Federal funds and repurchase agreements 14.54 Industry risk factors 5.40–57 Interest rate risk. See internal control over financial reporting Loans 8.185, 8.188–189 Other assets, other liabilities, and other investments 12.79 Overview 5.33 Purpose of 15.24 Real estate investments, real estate owned, and other foreclosed assets 11.58 Requirements 5.33 Security investments 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNDISBURSED PORTION OF MORTCAGE LOANS 8.225–226 UNDIVIDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02 UNFORM CONSUMER CODE (UCC) 2.20, 8.85–87 UNIFORM CONSUMER CODE 2.20 UNIFORM CONSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING  Deposits 15.59–60 Credit union reporting 1.103 Net worth requirements 1.103 Net wor	·	
Debt	· Credit losses9.64, 9.69–.70	II S DEPARTMENT OF HOUSING AND LIRRAN
Depository institution regulation and supervision		
Supervision 5.237 Deposits 13.50 Equity and disclosure regarding capital matters 17.95 Federal funds and repurchase agreements 14.54 Industry risk factors 5.40–57 Interest rate risk. See internal control over financial reporting 2.00 Cher assets, other liabilities, and other investments 12.79 Overview 5.33 Purpose of 5.24 Real estate investments, real estate owned, and other foreclosed assets 11.58 Requirements 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNDISBURSED PORTION OF MORTGAGE LOANS 8.225–226 UNDIVIDED EARNINGS 17.50–52, 17.68–69 UNEARNED INTEREST 8.17 UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02 UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.85–87 UNIFORM CONSUMER CODE 2.20 UNIFORM CONSUMER CODE 2.20 UNIFORM CONSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING		• • •
Equity and disclosure regarding capital matters		
Reporting considerations 1.118, 4.33–35 Federal funds and repurchase agreements 14.54 Industry risk factors 5.40–57 Interest rate risk. See interest rate risk (IRR) Internal control. See internal control over financial reporting 8.185, 8.188–189 Other assets, other liabilities, and other investments 12.79 Overview 5.33 Purpose of 5.24 Requirements 5.33 Security investments, real estate owned, and other foreclosed assets 11.58 Requirements 5.33 Security investments 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNDISBURSED PORTION OF MORTGAGE LOANS 8.225–226 UNDIVIDED LARNINGS 17.50–52, 17.68–69 UNEARNED INTEREST 8.17 UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02 UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.85–87 UNIFORM CONSUMER CODE 2.20 UNIFORM CONSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING Reporting considerations 1.118, 4.33–35 Unmodified opinions 23.06 US. DEPARTMENT OF THE TREASURY. See also Office of the Comptroller of the Currency (OCC); Office of Thrift Supervision (OTS) Anti-money laundering programs 1.20 Credit union regulation 2.20 Fannie Mae/Freddie Mac bailout 4.06 Holding company regulation 1.19 Home Affordable Mortgage Program ((HAMP) 4.26 (HAMP) 4.26 UNS. DEPARTMENT OF THE TREASURY. See also Office of the Comptroller of the Currency (OCC); Office of Thrift Supervision (OTS) Anti-money laundering programs 1.20 Credit union regulation 1.20 US. DEPARTMENT OF THE TREASURY. See also Office of the Comptroller of the Currency (OCC); Office of Thrift Supervision (OTS) Anti-money laundering programs 1.20 US. DEPARTMENT OF THE TREASURY. See also Office of the Comptroller of the Currency (OCC); OFFice of Tarift Supervision (OTS) Anti-money laundering program 1.		· Net worth requirements17.80
Federal funds and repurchase agreements		
agreements		
Industry risk factors	•	· Unmodified opinions23.06
Interest rate risk. See interest rate risk (IRR) Internal control. See internal control over financial reporting Loans		
Internal control. See internal control over financial reporting Loans 8.185, 8.188–189 Other assets, other liabilities, and other investments 12.79 Overview 5.33 Purpose of 5.24 Real estate investments, real estate owned, and other foreclosed assets 11.58 Requirements 5.33 Security investments 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNDISBURSED PORTION OF MORTGAGE LOANS 8.225–226 UNDIVIDED EARNINGS 17.50–52, 17.68–69 UNEARNED INTEREST 8.17 UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02 UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.85–87 UNIFORM CONSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING  Currency (OCC); Unice of Inritt Supervision (OTS) Anti-money laundering programs 1.20 Credit union regulation 2.20 Fannie Mae/Freddie Mac bailout 4.06 Holding company regulation 1.19 Home Affordable Mortgage Program (HAMP) 4.26 Ferpetual preferred stock issued to 17.25 Small Business Lending Fund reporting 23.37 Tax bad-debt deductions 16.07 U.S. DEPARTMENT OF VETERANS' AFFAIRS (VA) 4.26, 8.42, 10.03, 10.16–17 U.S. GOVERNMENT SECURITIES 7.05–07 USA PATRIOT ACT (2001) 1.20, 5.226  VA. See U.S. Department of Veterans' Affairs VALUATION Deferred tax assets 16.21 Fair value techniques 8.230, 20.28–30 Loan substantive testing 8.229–230 Loans acquired with deteriorated credit quality 8.121–122 Premise for nonfinancial assets 20.13 Real estate investments, real estate owned,		
Loans	Internal control. See internal control over	
Other assets, other liabilities, and other investments 12.79 Overview 5.33 Purpose of 5.24 Real estate investments, real estate owned, and other foreclosed assets 11.58 Requirements 5.33 Security investments 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNDISBURSED PORTION OF MORTGAGE LOANS 8.225226 UNDIVIDED EARNINGS 17.5052, 17.6869 UNEARNED INTEREST 8.17 UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02 UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.8587 UNIFORM CONSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING  Fannie Mae/Freddie Mac bailout 4.06 Holding company regulation 1.19 Home Affordable Mortgage Program (HAMP) 4.26 Holding company regulation 1.19 Home Affordable Mortgage Program (HAMP) 4.26 Small Business Lending Fund reporting 23.37 Tax bad-debt deductions 16.07 U.S. DEPARTMENT OF VETERANS' AFFAIRS (VA) 4.26, 8.42, 10.03, 10.1617 USA PATRIOT ACT (2001) 1.20, 5.226  VA. See U.S. Department of Veterans' Affairs VALUATION Deferred tax assets 16.21 Fannie Mae/Freddie Mac bailout 4.06 Holding company regulation 1.19 Home Affordable Mortgage Program (HAMP) 4.26 Perpetual preferred stock issued to 17.25 Small Business Lending Fund reporting 23.37 Tax bad-debt deductions 16.07 U.S. GOVERNMENT SECURITIES 7.0507 USA PATRIOT ACT (2001) 1.20, 5.226  VA. See U.S. Department of Veterans' Affairs VALUATION Deferred tax assets 16.21 Fair value techniques 8.230, 20.2830 Loans sacquired with deteriorated credit quality 8.121122 Premise for nonfinancial assets 20.13 Real estate investments, real estate owned,		Anti-money laundering programs1.20
investments 12.79 Overview 5.33 Purpose of 5.24 Real estate investments, real estate owned, and other foreclosed assets 11.58 Requirements 5.33 Security investments 7.139, 7.144 UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10 UNDISBURSED PORTION OF MORTGAGE LOANS 8.225226 UNDIVIDED EARNINGS 17.5052, 17.6869 UNEARNED INTEREST 8.17 UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02 UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.8587 UNIFORM CONSUMER CODE 2.20 UNIFORM FINANCIAL INSTITUTIONS RATING Holding company regulation 1.19 Home Affordable Mortgage Program (HAMP) 4.26 Holding company regulation 1.19 Home Affordable Mortgage Program (HAMP) 4.26 Small Business Lending Fund reporting 23.37 Tax bad-debt deductions 16.07 U.S. DEPARTMENT OF VETERANS' AFFAIRS (VA) 4.26, 8.42, 10.03, 10.1617 U.S. GOVERNMENT SECURITIES 7.0507 USA PATRIOT ACT (2001) 1.20, 5.226  V VA. See U.S. Department of Veterans' Affairs VALUATION Deferred tax assets 16.21 Fair value techniques 8.230, 20.2830 Loans sacquired with deteriorated credit quality 8.121122 Premise for nonfinancial assets 20.13 Real estate investments, real estate owned,		
Overview		
Purpose of		
Real estate investments, real estate owned, and other foreclosed assets		
and other foreclosed assets 11.58 Requirements 5.33 Security investments 7.139, 7.144  UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10  UNDISBURSED PORTION OF MORTGAGE LOANS 8.225226  UNDIVIDED EARNINGS 17.5052, 17.6869  UNEARNED INTEREST 8.17  UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02  UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.8587  UNIFORM CONSUMER CODE 2.20  UNIFORM FINANCIAL INSTITUTIONS RATING  Acquirements 5.33  Small Business Lending Fund reporting 23.37  Tax bad-debt deductions 16.07  U.S. DEPARTMENT OF VETERANS' AFFAIRS (VA) 4.26, 8.42, 10.03, 10.1617  U.S. GOVERNMENT SECURITIES 7.0507  USA PATRIOT ACT (2001) 1.20, 5.226  VALUATION  Deferred tax assets 16.21  Fair value techniques 8.230, 20.2830  Loan substantive testing 8.229230  Loans acquired with deteriorated credit quality 8.121122  Premise for nonfinancial assets 20.13  Real estate investments, real estate owned,		
Security investments	and other foreclosed assets11.58	
UNDERWRITING PROCEDURES, CREDIT LOSS RISK 9.10  UNDISBURSED PORTION OF MORTGAGE LOANS 8.225226  UNDIVIDED EARNINGS 17.5052, 17.6869  UNEARNED INTEREST 8.17  UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02  UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.8587  UNIFORM CONSUMER CODE 2.20  UNIFORM FINANCIAL INSTITUTIONS RATING 1.03, 7.144  U.S. DEPARTMENT OF VETERANS' AFFAIRS (VA)4.26, 8.42, 10.03, 10.1617  U.S. GOVERNMENT SECURITIES7.0507  USA PATRIOT ACT (2001)1.20, 5.226  VALUATION	Requirements	
UNDERWRITING PROCEDURES, CREDIT LOSS RISK	· Security investments 7.139, 7.144	
RISK 9.10  UNDISBURSED PORTION OF MORTGAGE LOANS 8.225-226  UNDIVIDED EARNINGS 17.5052, 17.6869  UNEARNED INTEREST 8.17  UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02  UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.8587  UNIFORM CONSUMER CODE 2.20  UNIFORM FINANCIAL INSTITUTIONS RATING  (VA)4.26, 8.42, 10.03, 10.1617  U.S. GOVERNMENT SECURITIES7.0507 USA PATRIOT ACT (2001)1.20, 5.226  VALUATION  VALUATION  Deferred tax assets	UNDERWRITING PROCEDURES, CREDIT LOSS	
LOANS 8.225–226  UNDIVIDED EARNINGS 17.50–52, 17.68–69  UNEARNED INTEREST 8.17  UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments 12.02  UNIFORM COMMERCIAL CODE (UCC) 2.20, 8.85–87  UNIFORM CONSUMER CODE 2.20  UNIFORM FINANCIAL INSTITUTIONS RATING  U.S. GOVERNMENT SECORTIES 7.7.05–.07  USA PATRIOT ACT (2001) 1.20, 5.226  USA PATRIOT ACT (2001)		
UNDIVIDED EARNINGS		U.S. GOVERNMENT SECURITIES7.0507
UNEARNED INTEREST	UNDIVIDED EARNINGS 17.50–.52,	USA PATRIOT ACT (2001)1.20, 5.226
UNFUNDED LOAN COMMITMENT ALLOWANCE. See also other assets, other liabilities, and other investments	17.68–.69	V
ALLOWANCE. See also other assets, other liabilities, and other investments	UNEARNED INTEREST8.17	•
ALLOWANCE. See also other assets, other liabilities, and other investments	UNFUNDED LOAN COMMITMENT	VA. See U.S. Department of Veterans' Affairs
other liabilities, and other investmentsDeferred tax assets16.21UNIFORM COMMERCIAL CODE (UCC)Loan substantive testing8.229230UNIFORM CONSUMER CODELoans acquired with deteriorated credit quality8.121122UNIFORM FINANCIAL INSTITUTIONS RATINGPremise for nonfinancial assets20.13Real estate investments, real estate owned,		VALUATION
UNIFORM COMMERCIAL CODE (UCC)		
(UCC)		
UNIFORM CONSUMER CODE	UNIFORM COMMERCIAL CODE	
UNIFORM CONSUMER CODE2.20 UNIFORM FINANCIAL INSTITUTIONS RATING quality8.121–.122 Premise for nonfinancial assets20.13 Real estate investments, real estate owned,		
UNIFORM FINANCIAL INSTITUTIONS RATING  Real estate investments, real estate owned,		
	SYSTEM 5.228	Real estate investments, real estate owned,     and other foreclosed assets

VALUATION ALLOWANCE 8.121122, 16.21	WELL CAPITALIZED BANKS AND SAVINGS INSTITUTIONS 1.8586, 1.88, 17.39,
VALUATION RISK18.11	17.41
VARIABLE INTEREST, DEFINED10.98	WELL CAPITALIZED CREDIT UNION
VARIABLE INTEREST ENTITIES (VIE). See also consolidation 10.97113  Beneficiaries of 10.104106  Determination 10.97, 10.102  Fair value option exclusion 20.62  Financial statement presentation and disclosure 7.115, 10.107113  Insurance activities 22.6163  Overview 10.9795  Primary beneficiary evaluation .10.104106  Private company lessee and lessors 22.61  Reconsideration of status 10.103	CLASSIFICATION         Capital adequacy disclosures       17.61, 17.77         Complex credit unions       2.33, 2.36         Corporate credit unions       2.50         Net worth ratio       Exhibit 2-1 at 2.30         New credit unions       2.39, 2.40         Prompt corrective action       Exhibit 2-1 at 2.30         Reserves       17.48         WELL MANAGED BANKS AND SAVINGS INSTITUTIONS         INSTITUTIONS       17.39         WHOLE LOANS PURCHASED       8.218
Trust preferred securities 17.12–.16  VARIATION MARGIN	WHOLESALE LOANS 8.30
VENTURE CAPITAL INVESTMENTS	WITHDRAWALS, DEMAND DEPOSITS
VETERANS' AFFAIRS (VA). See U.S. Department of Veterans' Affairs	WORKING PAPERS OR WORKPAPERS. See audit documentation
VOLCKER RULE1.41, 1.54, 7.78, 7.86, 18.54, 18.77	WORLD COUNCIL OF CREDIT UNIONS2.14
VOLUME OF ACTIVITY, FAIR VALUE MEASUREMENTS WHEN SIGNIFICANTLY	WRITTEN LOAN COMMITMENTS18.92
DECREASED	WRITTEN REPRESENTATIONS 5.175, 5.180189
W	Υ
WAREHOUSING       8.17         WARRANTS       18.51         WATCH LIST, TEST OF CONTROLS       9.73         WATCH-LISTED LOANS       8.228	YIELD, COMPARED TO LIQUIDITY7.02 YIELD ANALYSIS

## WILEY END USER LICENSE AGREEMENT

Go to www.wiley.com/go/eula to access Wiley's ebook EULA.